AS FILED WITH THE SECURITIES AND EXCHANGE COMMISSION ON NOVEMBER 20, 1998 REGISTRATION NO. 333-64793 SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549 AMENDMENT NO. 3 TO FORM S-4 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933 HMC MERGER CORPORATION (EXACT NAME OF REGISTRANT AS SPECIFIED IN ITS GOVERNING INSTRUMENT) MARYLAND 7011 53-0085950 (STATE OR OTHER (PRIMARY STANDARD (I.R.S. EMPLOYER JURISDICTION INDUSTRIAL IDENTIFICATION NO.) OF INCORPORATION OR CLASSIFICATION CODE ORGANIZATION) NUMBER) 10400 FERNWOOD ROAD BETHESDA, MARYLAND 20817 (301) 380-9000 (ADDRESS, INCLUDING ZIP CODE, AND TELEPHONE NUMBER, INCLUDING AREA CODE, OF REGISTRANT'S PRINCIPAL EXECUTIVE OFFICES) CHRISTOPHER G. TOWNSEND GENERAL COUNSEL HMC MERGER CORPORATION 10400 FERNWOOD ROAD BETHESDA, MARYLAND 20817 (301) 380-9000 (NAME, ADDRESS, INCLUDING ZIP CODE, AND TELEPHONE NUMBER, INCLUDING AREA CODE, OF AGENT FOR SERVICE) COPIES TO: J. WARREN GORRELL, JR., ESQ. HOWARD I. FLACK, ESQ. HOGAN & HARTSON L.L.P. 555 THIRTEENTH STREET, N.W. WASHINGTON, D.C. 20004-1109 (202) 637-5600 APPROXIMATE DATE OF COMMENCEMENT OF PROPOSED SALE OF THE SECURITIES TO THE PUBLIC: As soon as practicable after the effective date of this Registration Statement. If the securities being registered on this form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, check the following box.  $[\_]$ If this form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. [\_]

If this form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. [\_] -----CALCULATION OF REGISTRATION FEE **PROPOSED** MAXIMUM PR0P0SED TITLE OF EACH CLASS OF AMOUNT AGGREGATE AMOUNT OF MAXIMUM SECURITIES TO BE OFFERING PRICE TO BE OFFERING REGISTRATION REGISTERED PER SHARE(1) REGISTERED PRICE(1) Shares of Common Stock, par value \$.01 per \_\_\_\_\_ (1) Estimated solely for the purpose of calculating the registration fee pursuant to Rule 457(f) promulgated under the Securities Act of 1933, as

(2) Includes the maximum number of shares of common stock, par value \$.01 per share ("Host REIT Common Stock"), of HMC Merger Corporation, a Maryland corporation (referred to herein as "Host REIT", which will be renamed "Host Marriott Corporation" following the merger described below), to be registered hereunder that may be issuable pursuant to the merger (the "Merger") of Host Marriott Corporation, a Delaware corporation ("Host"), with and into Host REIT pursuant to the Agreement and Plan of Merger (the "Agreement") by and among Host. Host REIT and Host Marriott L.P., as

described in the Proxy Statement/Prospectus that forms a part of this Registration Statement. Based upon the number of shares of common stock, par value \$1.00 per share, of Host ("Host Common Stock") outstanding at the close of business on November 18, 1998. See "The Restructuring Transactions--Terms of the Merger."

(3) Host REIT previously registered 204,586,035 shares of Common Stock, for which a registration fee of \$844,940 was paid.

THE REGISTRANT HEREBY AMENDS THIS REGISTRATION STATEMENT ON SUCH DATE OR DATES AS MAY BE NECESSARY TO DELAY ITS EFFECTIVE DATE UNTIL THE REGISTRANT SHALL FILE A FURTHER AMENDMENT WHICH SPECIFICALLY STATES THAT THIS REGISTRATION STATEMENT SHALL THEREAFTER BECOME EFFECTIVE IN ACCORDANCE WITH SECTION 8(A) OF THE SECURITIES ACT OF 1933 OR UNTIL THE REGISTRATION STATEMENT SHALL BECOME EFFECTIVE ON SUCH DATE AS THE COMMISSION, ACTING PURSUANT TO SAID SECTION 8(A) MAY DETERMINE.

- ------

#### HOST MARRIOTT CORPORATION 10400 FERNWOOD ROAD BETHESDA, MARYLAND 20817-1109 (301) 380-9000

NOVEMBER , 1998

Dear Fellow Stockholder:

You are cordially invited to attend a special meeting of stockholders of Host Marriott Corporation, a Delaware corporation ("Host"), which will be held at the Ritz-Carlton Hotel, 1700 Tysons Boulevard, McLean, Virginia, 22102, on December 15, 1998 at 10:00 a.m., local time (the "Special Meeting").

At the Special Meeting, you will be asked to approve (a) the Agreement and Plan of Merger (the "Agreement"), entered into by and among Host, HMC Merger Corporation, a wholly-owned, newly formed Maryland subsidiary of Host (referred to herein as "Host REIT", which will be renamed "Host Marriott Corporation" following the merger described below), and Host Marriott, L.P., a recently formed Delaware limited partnership organized and currently whollyowned by Host and of which Host will be the sole general partner (the "Operating Partnership") and (b) such other matters as may properly come before the meeting or any adjournments or postponements thereof. The Agreement contemplates certain restructuring transactions (the "Restructuring Transactions") consisting of (i) the contribution by Host of its wholly-owned full-service hotels, its interests in certain hotel partnerships and certain other assets to the Operating Partnership in exchange for units of limited and general partnership interest in the Operating Partnership and the assumption of liabilities (the "OP Contribution") and (ii) the reincorporation of Host from the State of Delaware to the State of Maryland (the "Reincorporation") by means of a merger (the "Merger") of Host with and into Host REIT, which was formed for the sole purpose of effecting the Reincorporation. As a result of the Reincorporation by means of the Merger, each outstanding share of common stock of Host, together with the associated right issued under Host's existing stockholder rights plan, will be converted into one share of common stock of Host REIT, together with the associated right under a stockholder rights plan to be adopted by Host REIT prior to the Merger.

As described in the attached Proxy Statement/Prospectus, the Restructuring Transactions are part of an overall plan (the "REIT Conversion") adopted by Host to restructure its business operations so that it will qualify as a real estate investment trust ("REIT") for federal income tax purposes. If Host REIT qualifies as a REIT, it generally will not be subject to federal corporate income taxes on that portion of its ordinary income or capital gain that is distributed to its stockholders. Such treatment would substantially eliminate the federal "double taxation" on earnings (at the corporate and stockholder levels) that generally results from investment in a corporation. If the Agreement is approved by the stockholders of Host at the Special Meeting and the Restructuring Transactions are consummated, Host REIT expects to qualify as a REIT beginning with its first full taxable year commencing after the REIT Conversion is completed, which currently is expected to be the year commencing January 1, 1999 (but which might not be until the year beginning January 1, 2000).

The Host Board of Directors is proposing the Restructuring Transactions primarily for the following reasons:

- . The Restructuring Transactions are essential components of Host's conversion to REIT status for federal income tax purposes. In particular, the OP Contribution will enable Host REIT, following the Merger, to operate, together with the Operating Partnership, in an umbrella partnership REIT ("UPREIT") structure, through which Host REIT would continue the full-service hotel ownership business currently conducted by Host. Host believes that the UPREIT structure will improve its ability to acquire additional properties in the future on favorable terms.
- . The provisions of Maryland law have generally been viewed as favorable to REITs organized in corporate or trust form, as evidenced by the large number of publicly traded REITs that have chosen to operate as a regular Maryland corporation or as a special statutory Maryland real estate investment trust.

- . In particular, in order to satisfy certain requirements that are applicable to REITs in general, many REITs impose through their charters ownership limits and transfer restrictions similar to the ownership limit proposed by Host REIT in its charter, as described in the accompanying Proxy Statement/Prospectus. Under Delaware law, such restrictions would not be binding with respect to securities issued prior to adoption of the restriction unless holders of such securities agree to, or vote in favor of, such restriction. However, under Maryland law and by reason of the Merger, all shares of common stock of Host REIT issued in the Merger and thereafter would be subject to the ownership limit under Host REIT's charter, for which authority exists under Maryland law.
- . Host's principal executive offices and a substantial number of Host's employees are employed in Maryland.

The Board of Directors of Host believes that the Agreement, which contemplates the Restructuring Transactions, and the other transactions comprising the REIT Conversion described in the accompanying Proxy Statement/Prospectus are advisable for Host and its stockholders, based on the belief that: (i) the REIT structure, as a more efficient tax structure, will provide improved operating results through changing economic conditions and all phases of the hotel economic cycle; (ii) the REIT Conversion, which will reduce corporate-level taxes and the need to incur debt to reduce corporate taxes through interest deductions, will improve its financial flexibility and allow it to continue to strengthen its balance sheet by reducing its overall debt to equity ratio over time; (iii) as a REIT, Host will be able to compete more effectively with other public lodging real estate companies that already are organized as REITs and to make performance comparisons with its peers more meaningful; (iv) by becoming a dividend paying company, Host's stockholder base will expand to include investors attracted by yield as well as asset quality, which is expected to facilitate Host REIT's capital-raising efforts and provide a less volatile stockholder base; and (v) the adoption of an UPREIT structure will facilitate tax-deferred acquisition of additional hotels. Host believes that these benefits justify the REIT Conversion even if the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999 (in which event the effectiveness of Host REIT's election could be delayed until January 1, 2000).

The Merger will not be consummated unless Host's Board of Directors shall have determined prior to consummating the Merger that the conditions to the Merger (including approval of the Agreement by the stockholders of Host) have been satisfied or waived, and in particular, that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur, and based on advice of counsel, that Host REIT can elect to be treated as a REIT for federal income tax purposes effective no later than the first full taxable year commencing after the REIT Conversion is completed (which might not be until the year commencing January 1, 2000 if the REIT Conversion is not completed prior to January 1, 1999). In the event the Agreement is not approved by Host stockholders at the Special Meeting, Host will continue to operate as a Delaware corporation, and the REIT Conversion will not be completed at this time.

Details of the Agreement, including the proposed Restructuring Transactions, as well as the other transactions comprising the REIT Conversion, are contained in the attached Proxy Statement/Prospectus, which you are encouraged to read carefully.

AFTER CAREFUL CONSIDERATION, YOUR BOARD OF DIRECTORS HAS DETERMINED THE AGREEMENT, WHICH CONTEMPLATES THE RESTRUCTURING TRANSACTIONS, AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION, TO BE ADVISABLE FOR HOST AND ITS STOCKHOLDERS. ACCORDINGLY, YOUR BOARD OF DIRECTORS HAS APPROVED THE AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY AND RECOMMENDS THAT ALL STOCKHOLDERS VOTE "FOR" APPROVAL OF THE AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY.

The affirmative vote of holders of two-thirds (66 2/3%) of the outstanding shares of Host common stock is necessary to approve the Agreement. Therefore, whether or not you plan to attend the Special Meeting, please complete, sign and date the enclosed proxy card and return it promptly in the enclosed postage prepaid envelope, or vote your proxy by telephone or the internet in accordance with the instructions on the admission ticket attached to the enclosed proxy card. You may revoke your proxy in the manner discussed in the

accompanying Proxy Statement/Prospectus at any time before it has been voted at the Special Meeting. If you attend the Special Meeting, you may vote in person if you wish, even if you have previously submitted your proxy. Your prompt cooperation will be greatly appreciated. This solicitation is made on behalf of the Board of Directors of Host.

Sincerely,

RICHARD E. MARRIOTT Chairman of the Board

YOUR VOTE IS IMPORTANT
PLEASE PROMPTLY COMPLETE, SIGN AND DATE AND RETURN
THE ENCLOSED PROXY CARD OR PROMPTLY VOTEYOUR PROXY BY TELEPHONE OR THE
INTERNET

IF THE RESTRUCTURING TRANSACTIONS AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION DO NOT OCCUR IN TIME FOR HOST REIT TO ELECT REIT STATUS EFFECTIVE JANUARY 1, 1999, THE EFFECTIVENESS OF HOST REIT'S ELECTION COULD BE DELAYED TO JANUARY 1, 2000, WHICH WOULD RESULT IN HOST OR HOST REIT CONTINUING TO PAY SUBSTANTIAL CORPORATE-LEVEL INCOME TAXES IN 1999 AND COULD CAUSE OTHER RELATED TRANSACTIONS NOT TO BE CONSUMMATED. THEREFORE, IT IS EXTREMELY IMPORTANT THAT STOCKHOLDERS SUBMIT THEIR PROXIES AS SOON AS POSSIBLE. ANY DELAY IN RETURNING PROXIES COULD CAUSE THE SPECIAL MEETING TO BE DELAYED, WHICH COULD PREVENT THE RESTRUCTURING TRANSACTIONS AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION FROM BEING EFFECTIVE JANUARY 1, 1999.

#### HOST MARRIOTT CORPORATION 10400 FERNWOOD ROAD BETHESDA, MARYLAND 20817-1109 (301) 380-9000

-----

NOTICE OF SPECIAL MEETING OF STOCKHOLDERS TO BE HELD ON DECEMBER 15, 1998

-----

To the Stockholders of Host Marriott Corporation:

NOTICE IS HEREBY GIVEN that a special meeting of stockholders of Host Marriott Corporation, a Delaware corporation ("Host"), will be held at the Ritz-Carlton Hotel, 1700 Tysons Boulevard, McLean, Virginia, 22102, on December 15, 1998 at 10:00 a.m., local time (the "Special Meeting"), for the following purposes:

1. To consider and vote upon a proposal to approve the Agreement and Plan of Merger dated November , 1998 (the "Agreement") by and among Host, HMC Merger Corporation, a wholly-owned, newly formed Maryland subsidiary of Host (referred to as "Host REIT", which will be renamed "Host Marriott Corporation" following the merger described below), and Host Marriott, L.P., a recently formed Delaware limited partnership organized and currently wholly-owned by Host and of which Host will be the sole general partner (the "Operating Partnership"). The Agreement contemplates certain restructuring transactions (the "Restructuring Transactions") consisting of (i) the contribution by Host of its wholly-owned full-service hotels, its interests in certain hotel partnerships and certain other assets to the Operating Partnership in exchange for units of limited and general partnership interest in the Operating Partnership and the assumption of liabilities (the "OP Contribution") and (ii) the reincorporation of Host from the State of Delaware to the State of Maryland (the "Reincorporation") by means of a merger (the "Merger") of Host with and into Host REIT, which was formed for the sole purpose of effecting the Reincorporation. If the Agreement is approved by Host stockholders and the Merger is consummated, Host REIT will be the surviving entity in the Merger, the separate existence of Host will terminate and each issued and outstanding share of common stock, par value \$1.00 per share, of Host (the "Host Common Stock"), together with the associated right issued under the Rights Agreement dated as of February 3, 1989 between Host and the Bank of New York, will be converted into one share of common stock, par value \$.01 per share, of Host REIT (the "Host REIT Common Stock"), together with the associated right issued under a stockholder rights plan to be adopted by Host REIT prior to the Merger. In addition, the Agreement provides that, in the Merger, outstanding options and other rights to acquire Host Common Stock will be converted into rights to acquire Host REIT Common Stock, subject to certain adjustments and except as otherwise set forth in the Agreement, and any outstanding shares of preferred stock of Host would be converted into an equal number of shares of preferred stock of Host REIT having substantially the same rights and preferences. Upon consummation of the Merger, holders of Host REIT Common Stock will become subject to the Ownership Limit (described below) under the Articles of Incorporation of Host REIT that will be in effect upon consummation of the Merger (the "Host REIT Charter"). Primarily to satisfy certain requirements under the Internal Revenue Code of 1986, as amended (the "Code"), that are applicable to REITs in general, the Host REIT Charter will provide that no person or group of persons may own, or be deemed to own by virtue of the attribution provisions of the Code, more than 9.8% of the lesser of the number or value of shares of Host REIT Common Stock (or any class or series of Host REIT preferred stock) outstanding (the "Ownership Limit"), subject to waiver or modification by Host REIT in certain limited circumstances and to certain limited exceptions for a holder of shares of Host REIT Common Stock solely by reason of the Merger in excess of the Ownership Limit so long as such holder would not own, directly or by attribution under the Code, more than 9.9% by value of the outstanding capital stock of Host REIT as of 12:01 a.m. (Eastern Time) on the first day after the effective time of the  $% \left( 1\right) =\left( 1\right) \left( 1\right$ Merger. THE AGREEMENT, THE RESTRUCTURING TRANSACTIONS, INCLUDING THE TERMS OF THE MERGER, AND THE OWNERSHIP LIMIT ARE MORE COMPLETELY DESCRIBED IN THE ACCOMPANYING PROXY STATEMENT/PROSPECTUS AND THE APPENDICES THERETO, WHICH FORM A PART OF THIS NOTICE. A COPY OF THE AGREEMENT IS ATTACHED AS APPENDIX A TO THE PROXY STATEMENT/PROSPECTUS.

2. To transact such other business as may properly come before the Special Meeting or any adjournments or postponements thereof.

Host reserves the right to cancel or defer the Restructuring Transactions even if stockholders of Host approve the Agreement and the other conditions to the consummation of the Restructuring Transactions are satisfied or waived.

Only stockholders of record of Host Common Stock at the close of business on November 13, 1998 (the "Record Date") are entitled to notice of, and to vote at, the Special Meeting and any adjournments or postponements thereof.

Approval of the Agreement requires the affirmative vote of the holders of two-thirds (66 2/3%) of the shares of Host Common Stock outstanding on the Record Date for the Special Meeting.

ALL STOCKHOLDERS ARE CORDIALLY INVITED TO ATTEND THE SPECIAL MEETING IN PERSON. WHETHER OR NOT YOU PLAN TO ATTEND THE SPECIAL MEETING, PLEASE COMPLETE, DATE AND SIGN THE ENCLOSED PROXY AND RETURN IT IN THE ENCLOSED ENVELOPE AS PROMPTLY AS POSSIBLE OR VOTE YOUR PROXY BY TELEPHONE OR THE INTERNET. IF A PROXY IS SIGNED BUT NO VOTING INSTRUCTIONS ARE INDICATED THEREON, SUCH PROXY WILL BE VOTED "FOR" APPROVAL OF THE AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY. IF YOU ATTEND THE SPECIAL MEETING, YOU MAY WITHDRAW YOUR PROXY AND VOTE IN PERSON.

 $\ensuremath{\mathsf{HOST'S}}$  STOCKHOLDERS SHOULD NOT SEND ANY STOCK CERTIFICATES WITH THEIR PROXY CARDS.

By Order of the Board of Directors,

CHRISTOPHER G. TOWNSEND Secretary

Bethesda, Maryland November , 1998

## SUBJECT TO COMPLETION DATED NOVEMBER 20, 1998

PROXY STATEMENT
OF
HOST MARRIOTT CORPORATION
AND
PROSPECTUS
OF
HMC MERGER CORPORATION

This Proxy Statement/Prospectus is being furnished to the holders of the common stock, par value \$1.00 per share (the "Host Common Stock"), of Host Marriott Corporation, a Delaware corporation ("Host"), in connection with the solicitation of proxies on behalf of the Board of Directors of Host for use at a special meeting of stockholders of Host to be held at the Ritz-Carlton Hotel, 1700 Tysons Boulevard, McLean, Virginia, 22102, on December 15, 1998 at 10:00 a.m., local time, and at any adjournments or postponements thereof (the "Special Meeting"). At the Special Meeting, stockholders of Host will be asked to approve (a) the Agreement and Plan of Merger dated November , 1998 (the "Agreement"), by and among Host, HMC Merger Corporation, a wholly-owned, newly formed Maryland subsidiary of Host (referred to herein as "Host REIT", which will be renamed "Host Marriott Corporation" following the merger described below), and Host Marriott, L.P., a recently formed Delaware limited partnership organized and currently wholly-owned by Host and of which Host will be the sole general partner (the "Operating Partnership"); and (b) such other matters as may properly come before the meeting or any adjournments or postponements thereof. The Agreement contemplates certain restructuring transactions (the "Restructuring Transactions") consisting of (i) the contribution by Host of its wholly-owned full-service hotels, its interests in certain hotel partnerships and certain other assets to the Operating Partnership in exchange for units of limited and general partnership interest in the Operating Partnership and the assumption of liabilities (the "OP Contribution") and (ii) the reincorporation of Host from the State of Delaware to the State of Maryland (the "Reincorporation") by means of a merger (the "Merger") of Host with and into Host REIT, which was formed for the sole purpose of effecting the Reincorporation. A copy of the Agreement is attached hereto as Appendix A. The Restructuring Transactions are part of an overall plan (the "REIT Conversion") adopted by Host to restructure its business operations so that it will qualify as a real estate investment trust ("REIT") for federal income tax purposes.

SEE "RISK FACTORS" BEGINNING ON PAGE 32 FOR MATERIAL RISKS THAT SHOULD BE CONSIDERED WHEN VOTING ON THE AGREEMENT, INCLUDING THE FOLLOWING:

- . Since Host REIT will lease virtually all of its hotels to lessees, Host REIT will be dependent for its revenue upon the ability of the lessees of Host REIT's hotels, Marriott International, Inc. ("Marriott International"), which currently manages all but 16 of Host's hotels, and other companies that manage the hotels and upon various non-controlled subsidiaries of Host REIT, and Host REIT will have limited control over the operation of the hotels and no control over the non-controlled subsidiaries.
- . Rental payments from the lessees of Host REIT's hotels will be the primary source of Host REIT's revenues.
- . If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the effectiveness of Host REIT's election could be delayed until January 1, 2000, which would result in Host or Host REIT continuing to pay substantial corporate-level income taxes in 1999 (which would reduce Host REIT's cash distributions) and could cause the Blackstone Acquisition (as defined herein) not to be consummated.
- . The current stock price of Host reflects the current market valuation of Host's current business and assets and not the business and assets of Host REIT following the REIT Conversion.
- After the REIT Conversion, Host REIT, as the sole general partner of the Operating Partnership, will have fiduciary obligations to the limited partners in the Operating Partnership, and the discharge of such obligations could result in decisions that may fail to reflect fully the interests of all holders of Host REIT Common Stock (as defined below) and limited partners of the Operating Partnership. Conflicts of interest also may be involved in Host REIT's relationships with Marriott International and the lessees of Host REIT's hotels.
- . The preliminary estimated initial annual cash distributions of the Operating Partnership during the twelve months ending December 31, 1999 (\$231 million) will exceed its estimated cash available for distribution and cash from contingent rents during the twelve months ending December 31, 1999, which would require borrowings by the Operating Partnership to make such distributions.
- . There are several uncertainties relating to the REIT Conversion that will exist at the time Host stockholders vote on the Agreement. There is no

assurance as to the outcome of various matters, and if certain of these events or transactions are not consummated as expected, it is possible that the REIT Conversion may not be completed, the value of the Host REIT Common Stock and the cash available for distribution to stockholders of Host REIT could be materially adversely affected or Host REIT could fail to qualify as a REIT for federal income tax purposes.

The inability of Host, the Operating Partnership and Host REIT to obtain one or more third-party consents prior to consummation of the REIT Conversion could have a material adverse effect on the Operating Partnership and Host REIT, and thus could reduce the value of the Host REIT Common Stock.

- . The Ownership Limit under Host REIT's charter will adversely affect the value of any Host REIT Common Stock held in excess of such Ownership Limit and may have the effect of delaying, deferring or preventing a change in control of Host REIT.
- change in control of Host REIT.

  Host REIT's charter and bylaws, the Maryland General Corporation Law, as amended (the "MGCL"), Host REIT's stockholder rights plan and certain existing rights of Marriott International contain or will contain a number of provisions that may limit the ability of outside parties to acquire, or discourage them from acquiring, control of Host REIT.

  Taxation of Host REIT as a regular corporation if it fails to qualify as
- . Taxation of Host REIT as a regular corporation if it fails to qualify as a REIT, or taxation of the Operating Partnership as a corporation if it fails to qualify as a partnership for federal income tax purposes, would, among other things, result in a material decrease in cash available for distribution and a material reduction in the value of the Host REIT Common Stock.
- . No assurance can be provided that new legislation, Treasury Regulations, administrative interpretations or court decisions will not significantly change the tax laws with respect to Host REIT's qualification as a REIT or the federal income tax consequences of such qualification.

If the Agreement is approved by the stockholders of Host at the Special Meeting and the Restructuring Transactions are consummated, Host REIT expects to qualify as a REIT beginning with its first full taxable year commencing after the REIT Conversion is completed, which currently is expected to be the year commencing January 1, 1999 (but which might not be until the year beginning January 1, 2000). See "Federal Income Tax Consequences."

This Proxy Statement/Prospectus also constitutes the prospectus of Host REIT relating to the common stock, par value \$.01 per share, of Host REIT (the "Host REIT Common Stock") into which the outstanding shares of Host Common Stock will be converted as part of the Reincorporation by means of the Merger. If the Agreement is approved by the Host stockholders and the Merger is consummated, Host REIT will be the surviving entity in the Merger, the separate existence of Host will terminate and each issued and outstanding share of Host Common Stock, together with the associated right (the "Host Right") issued under the Rights Agreement dated as of February 3, 1989 between Host and the Bank of New York (the "Host Rights Agreement"), will be converted into one share of Host REIT Common Stock, together with the associated right ("Host REIT Right") issued under a stockholder rights plan to be adopted by Host REIT prior to the Merger (the "Host REIT Rights Agreement"). In addition, the Agreement provides that, in the Merger, outstanding options and other rights to acquire Host Common Stock will be converted into rights to acquire Host REIT Common Stock, subject to certain adjustments to take into account aspects of the REIT Conversion, and except as set forth in the Agreement and any outstanding shares of preferred stock of Host would be converted into an equal number of shares of preferred stock of Host REIT having substantially the same rights and preferences. See "The Restructuring Transactions."

ANY SHARES OF HOST REIT COMMON STOCK HELD AT THE EFFECTIVE TIME OF THE MERGER AND NOT SUBSEQUENTLY TRANSFERRED PRIOR TO THE SPECIAL MERGER OWNERSHIP LIMIT EFFECTIVE TIME (AS DEFINED HEREIN), OR ACQUIRED OR HELD AT ANY TIME AFTER THE EFFECTIVE TIME OF THE MERGER IN VIOLATION OF THE OWNERSHIP LIMIT (AS DEFINED HEREIN), WILL BE TRANSFERRED AUTOMATICALLY TO A TRUST FOR THE BENEFIT OF A DESIGNATED CHARITABLE BENEFICIARY, AND THE PERSON WHO HOLDS SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL NOT BE ENTITLED TO ANY DISTRIBUTIONS THEREON OR TO VOTE SUCH EXCESS SHARES OF HOST REIT COMMON STOCK. TO AVOID THE ADVERSE EFFECTS OF THE OWNERSHIP LIMIT, ANY HOLDER OF HOST COMMON STOCK WHO WOULD OWN SHARES IN EXCESS OF THE OWNERSHIP LIMIT AT THE EFFECTIVE TIME OF THE MERGER SHOULD DISPOSE OF ANY SUCH EXCESS SHARES PRIOR THERETO. SEE "DESCRIPTION OF HOST REIT CAPITAL STOCK--RESTRICTIONS ON OWNERSHIP AND TRANSFER."

The Board of Directors has fixed the close of business on November 13, 1998 as the record date for the determination of stockholders entitled to receive notice of and vote at the Special Meeting (the "Record Date"). As of the Record Date, there were 204,954,447 outstanding shares of Host Common Stock and 47,845 holders of record. See "Voting and Proxies."

This Proxy Statement/Prospectus and the accompanying proxy card are first being mailed to stockholders of Host on or about November  $\,$  , 1998.

This Proxy Statement/Prospectus does not cover any resales of Host REIT Common Stock to be received by stockholders of Host upon consummation of the Merger, and no person is authorized to make use of this Proxy Statement/Prospectus in connection with any such resale.

THE SECURITIES DESCRIBED HEREIN HAVE NOT BEEN APPROVED OR DISAPPROVED BY THE SECURITIES AND EXCHANGE COMMISSION OR ANY STATE SECURITIES COMMISSION NOR HAS THE SECURITIES AND EXCHANGE COMMISSION OR ANY STATE SECURITIES COMMISSION PASSED UPON THE ACCURACY OR ADEQUACY OF THIS PROXY STATEMENT/PROSPECTUS. ANY REPRESENTATION TO THE CONTRARY IS A CRIMINAL OFFENSE.

## TABLE OF CONTENTS

	PAGE
SUMMARY	1
Forward-Looking Statements	1
Certain Key Definitions	2
0verview	4
Risk Factors	10
Date, Time, Place and Purpose of Special Meeting	15
Stockholders Entitled to Vote	15
Vote Required; No Dissenters' Rights	15
The Restructuring Transactions	16 21
Federal Income Tax Consequences	27
Recommendation of the Board of Directors	28
Distributions and Market Prices of Host Common Stock	28
Distribution and Dividend Policy After the Merger	29
Unaudited Per Share Data	30
Summary Financial Information	30
RISK FACTORS	32
Risks and Effects of the Merger and the REIT Conversion	32
Lack of Control over Hotel Operations	32
Lack of Control over Non-Controlled Subsidiaries	32
Dependence upon Crestline  Current Host Common Stock Price Is Not Necessarily Indicative of the	32
Price of Host REIT Common Stock Following the REIT Conversion	33
Cash Distributions May Exceed Cash Available for Distribution	33
Timing of the Restructuring Transactions and the REIT Conversion	33
Conflicts of Interest	34
Uncertainties at the Time of Voting on the Agreement	34
Inability to Obtain Third-Party Consents May Have a Material Adverse	
Effect	35
Expiration of the Leases and Possible Inability to Find Other	
Lessees	35
Leases Could Impair the Sale or Other Disposition of Host REIT's Hotels	35
Limitations on Sale or Refinancing of Certain Hotels	35 35
Risks of Ownership of Host REIT Common Stock	36
Possible Adverse Consequences of Limits on Ownership of Host REIT	00
Common Stock	36
Limitations on Acquisition of Host REIT Common Stock and Change in	
Control	37
Effect on Stock Price of Shares Available for Future Sale	40
Effect on Stock Price of Market Conditions	40
Effect on Stock Price of Earnings and Cash Distributions	40
Effect on Stock Price of Market Interest Rates Effect on Stock Price of Unrelated Events	41 41
Dependence on External Sources of Capital	41
Federal Income Tax Risks Relating to REIT Qualification	41
GeneralGeneral	41
Required Distributions and Payments	41
Consequences of Failure to Qualify as a REIT	42
"Earnings and Profits" Attributable to "C" Corporation Taxable Years	42
Treatment of Leases	42
Treatment of Leases of Personal Property	43
Other Tax Liabilities; Host REIT's Substantial Deferred and Contingent	40
Tax Liabilities	43
Failure of the Operating Partnership to Qualify as a Partnership	43

	PAGE
Other Tay Biglia	40
Other Tax Risks  Tax Consequences to Host of the OP Contribution and Other Transactions	43
Comprising the REIT Conversion	43
Effects of Subsequent Events upon Recognition of Gain	44
Risks of Operation	44
Competition in the Lodging Industry	44
Substantial Indebtedness of Host REIT	44
No Limitation on Debt  The Board May Change Investment and Other Policies Without Stockholder	45
Approval Management Agreements Could Impair the Sale or Other Disposition of	45
Host REIT's Hotels	45 45
Rental Revenues from Hotels Subject to Prior Rights of Lenders  Hotels Subject to Ground Leases May Affect Host REIT's Revenues	45 45
General Real Estate Investment Risks	46
Possible Underperformance of New Acquisitions	46
Seasonality	46
Illiquidity of Real Estate	46
Miscellaneous Risks	46
Changes in Laws	46
Dependence upon Key Personnel	47
Potential Litigation Related to the REIT Conversion	47
Risk Involved in Investments through Partnerships or Joint Ventures	47
Year 2000 Problem	47
Uninsured Loss	48
Americans with Disabilities Act	48 48
Possible Environmental Liabilities	48
CONFLICTS OF INTEREST	50
Responsibilities of Host REIT to Other Limited Partners in the Operating	
Partnership	50
Potential Conflicts Involving Marriott International and Crestline	50
Policies with Respect to Conflicts of Interest	50
/OTING AND PROXIES	51
Matters to be Considered at the Special Meeting	51
Record Date and Outstanding Shares	51
Voting of Proxies	51
Vote RequiredQuorum; Abstentions and Treatment of Broker Non-Votes; Adjournment; Rev-	51
ocation	52
Solicitation of Proxies and Expenses	53
THE RESTRUCTURING TRANSACTIONS	54
Background and Reasons for the Restructuring Transactions and the REIT	•
Conversion	54
The OP Contribution	56
Terms of the Merger	57
Absence of Dissenters' Rights	59
Accounting Treatment	59
Conditions to the Merger	59
Comparison of Rights of Stockholders of Host and Host REIT	60
Limitation of Liability and Indemnification of Directors and Officers	75 77
THE REIT CONVERSIONThe Initial E&P Distribution	77 78
Other Transactions Commission the DETT Conversion	70

	PAGE
007117011 05 57114110711 4017000	
OPINION OF FINANCIAL ADVISOROpinion of BT Wolfensohn, Financial Advisor to Host	85 85
BUSINESS AND PROPERTIES	89
Business of the Company	89
General	90
Business Objectives	90
Business Strategy	90
Hotel Lodging Industry	93
Hotel Lodging Properties	94
Hotel Properties	99
1998 Acquisitions	101
Blackstone Acquisition	101
Investments in Affiliated Partnerships	102
Marketing	102
Competition	103
Relationship with HM Services	103
Relationship with Marriott International; Marriott International	102
Distribution	103 104
Employees Environmental and Regulatory Matters	104
Legal Proceedings	104
The Leases	105
The Management Agreements	111
Noncompetition Agreement	114
Indebtedness	115
DISTRIBUTION AND OTHER POLICIES	117
Distribution Policy	117
Investment Policies	120
Financing Policies	121
Lending Policies	121
Conflicts of Interest Policies	121
Policies with Respect to Other Activities	123
SELECTED FINANCIAL DATA	124
MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF	100
OPERATIONSLack of Comparability Following the Merger and the REIT Conversion	126 126
Historical Results of Operations	126
First Three Quarters 1998 Compared to First Three Quarters 1997	120
(Historical)	126
1997 Compared to 1996 (Historical)	128
1996 Compared to 1995 (Historical)	130
Pro Forma Results of Operations	131
100% Participation with No Notes IssuedFirst Three Quarters 1998	
Compared to First Three Quarters 1997 (Pro Forma)	132
100% Participation with Notes IssuedFirst Three Quarters 1998	
Compared to First Three Quarters 1997 (Pro Forma)	133
100% Participation with No Notes Issued1997 Compared to 1996 (Pro	
Forma)	134
100% Participation with Notes Issued1997 Compared to 1996 (Pro	405
Forma)	135
Pro Forma Results if No Partnerships Participate in the REIT	126
Conversion	136
Pro Forma Results if Host REIT's Election is Delayed until January 1, 2000	136
Liquidity and Capital Resources	137
Year 2000 Problem	144
MANAGEMENT	148
Directors and Executive Officers of Host REIT	148

		PAGE
	Committees of the Board of Directors	150
	Compensation of Directors	
	Executive Compensation	151
	Aggregated Stock Option Exercises and Year-End Value	
	Long-Term Incentive Plan	154
	Employment Agreements	154
	1998 Employee Benefits Allocation Agreement	154
	Comprehensive Stock Incentive Plan	155
	Stock Purchase Plan	
	401(k) Plan	156
	Non-Employee Director Plan	156
	Deferred Compensation Plan	
	Limitation of Liability and Indemnification	
	Indemnification Agreements	
CE	RTAIN RELATIONSHIPS AND RELATED TRANSACTIONS	
	Relationship Between Host and Marriott International	
	Relationship Between Host and Host Marriott Services Corporation	161
	Relationship Between Host and Crestline After the Initial E&P	
	Distribution	
	RINCIPAL STOCKHOLDERS	
DE	SCRIPTION OF HOST REIT CAPITAL STOCK	
	General	
	Host REIT Common Stock	
	Preferred Stock	
	Power to Issue Additional Host REIT Common Stock and Preferred Stock	
	Restrictions on Ownership and Transfer	
۰.	Transfer Agent and Registrar	
UE	Number of Directors; Classification and Removal of Board of Directors;	1/2
	Other Provisions	172
	Changes in Control Pursuant to Maryland Law	
	Advance Notice of Director Nominations and New Business	174
	Meetings of Stockholders; Call of Special Meetings; Stockholder Action	1/4
	in Lieu of Meeting by Unanimous Consent	174
	Merger, Consolidation, Share Exchange and Transfer of Assets of Host	1,7
	REIT	175
	Amendments to the Host REIT Charter and Bylaws	
	Anti-Takeover Effect of Certain Provisions of Maryland Law and the Host	1.0
	REIT Charter and Bylaws	175
	Marriott International Purchase Right	
	Stockholder Rights Plan	
DE	SCRIPTION OF THE PARTNERSHIP AGREEMENT AND OP UNITS	
	General	
	Formation	
	Purposes, Business and Management	178
	Host REIT May Not Engage in Other Businesses; Conflicts of Interest	
	Distributions; Allocations of Income and Loss	
	Borrowing by the Operating Partnership	
	Reimbursement of Host REIT; Transactions with Host REIT and its	
	Affiliates	
	Liability of Host REIT and Limited Partners	
	Exculpation and Indemnification of Host REIT	180

	PAGE
Sales of Assets	101
Removal or Withdrawal of Host REIT; Transfer of Host REIT's Interests	
Certain Voting Rights of Holders of OP Units During the First Year	
Following the Effective Date of the Partnership Mergers	
Restrictions on Transfers of Interests by Limited Partners	
Unit Redemption Right	
No Withdrawal by Limited Partners	
Issuance of Limited Partnership Interests	
Meetings; Voting	184
Amendment of the Partnership Agreement	
Books and Reports	
Power of Attorney	
Dissolution, Winding Up and Termination	
Ownership Limitation	
ERISA CONSIDERATIONS	
Status of Host REIT and the Operating Partnership Under ERISA	
FEDERAL INCOME TAX CONSEQUENCES	
Introduction	
Federal Income Tax Consequences of the Merger	
Federal Income Tax Consequences of the Initial E&P Distribution	
Federal Income Tax Consequences of the OP Contribution	
Federal Income Taxation of Host REIT Following the Merger	
Taxation of Taxable U.S. Stockholders Generally	
Backup Withholding for Host REIT Distributions	
Taxation of Tax-Exempt Stockholders of Host REIT	
Taxation of Non-U.S. Stockholders	
Tax Aspects of Host REIT's Ownership of OP Units	
Other Tax Consequences for Host REIT and Its Stockholders	
LEGAL MATTERS	
EXPERTS	
OTHER MATTERS	
STOCKHOLDER PROPOSALS	
AVAILABLE INFORMATION; INCORPORATION OF CERTAIN DOCUMENTS BY REFERENCE	
GLOSSARY	
INDEX TO FINANCIAL STATEMENTS	F-1

# APPENDICES

Appendix A--Agreement and Plan of Merger

Appendix B--Opinion of BT Wolfensohn

Appendix C--Tax opinion of Hogan & Hartson L.L.P. with respect to the tax consequences of the Merger  $\,$ 

Appendix D--Tax opinion of Hogan & Hartson L.L.P. with respect to the qualification of Host REIT as a REIT

This Summary does not purport to be complete and is qualified in its entirety by the more detailed information appearing elsewhere in this Proxy Statement/Prospectus, including the appendices attached hereto (this "Proxy Statement/Prospectus"), and is presented solely to provide an overview of the transactions described in detail in the remainder of this Proxy Statement/Prospectus and of the business and investment considerations and risks that should be considered carefully when voting on the Agreement. Stockholders are advised not to rely on this Summary, but to carefully review this entire Proxy Statement/Prospectus.

The information contained herein, unless otherwise indicated, assumes the Restructuring Transactions and the other transactions comprising the REIT Conversion (including the Blackstone Acquisition, as defined below) occur, all Partnerships (as defined below) participate in the Partnership Mergers (as defined below) and no shares of Host REIT Common Stock or Notes (as defined below) are issued in the Partnership Mergers (the "Full Participation Scenario").

#### FORWARD-LOOKING STATEMENTS

Certain matters discussed herein or delivered in connection with this Proxy Statement/Prospectus are forward-looking statements. Certain, but not necessarily all, of such forward-looking statements can be identified by the use of forward-looking terminology, such as "believes," "expects," "may," "will," "should," "estimates" or "anticipates" or the negative thereof or other variations thereof or comparable terminology. All forward-looking statements involve known and unknown risks, uncertainties and other factors which may cause the actual transactions, results, performance or achievements of Host or Host REIT to be materially different from any future transactions, results, performance or achievements expressed or implied by such forward-looking statements. The cautionary statements set forth under the caption "Risk Factors" and elsewhere in this Proxy Statement/Prospectus identify important factors with respect to such forward-looking statements, including the following factors that could affect such forward-looking statements: (i) national and local economic and business conditions that will, among other things, affect demand for hotels and other properties, the level of rates and occupancy that can be achieved by such properties and the availability and terms of financing; (ii) the ability to maintain the properties in a firstclass manner (including meeting capital expenditure requirements); (iii) Host REIT's ability to compete effectively in areas such as access, location, quality of accommodations and room rate structures; (iv) Host REIT's ability to acquire or develop additional properties and the risk that potential acquisitions or developments may not perform in accordance with expectations; (v) Host's or Host REIT's ability to obtain required consents of stockholders, lenders, debt holders, partners and ground lessors of Host and its affiliates and of other third parties in connection with the REIT Conversion and to consummate all of the transactions constituting part of the REIT Conversion (including the Blackstone Acquisition); (vi) changes in travel patterns, taxes and government regulations which influence or determine wages, prices, construction procedures and costs; (vii) governmental approvals, actions and initiatives, including the need for compliance with environmental and safety requirements, and changes in laws and regulations or the interpretation thereof; (viii) the effects of tax legislative action; (ix) the effect on Host and Host REIT of the Year 2000 issue; and (x) the timing of Host REIT's election to be taxed as a REIT and the ability of Host REIT to satisfy complex rules in order to qualify for taxation as a REIT for federal income tax purposes and to operate effectively within the limitations imposed by these rules. Although Host and Host REIT believe the expectations reflected in such forward-looking statements are based upon reasonable assumptions, they can give no assurance that their expectations will be attained or that any deviations will not be material. Neither Host nor Host REIT undertakes any obligation to publicly release the result of any revisions to these forward-looking statements that may be made to reflect any future events or circumstances.

1

#### CERTAIN KEY DEFINITIONS

The following terms have the meanings set forth below. See the "Glossary" on page 218 for the definitions of other capitalized terms used in this Proxy Statement/Prospectus.

"Host"..... Host Marriott Corporation, a Delaware

corporation, and either the general partner or an affiliate of the general partner of each Partnership (as defined below), or, as the context may require, Host Marriott Corporation together with its subsidiaries or any of such subsidiaries.

"Host Common Stock"..... Shares of common stock, par value \$1.00 per share, of Host.

"Host REIT"..... HMC Merger Corporation, a wholly-owned, newly formed Maryland subsidiary of Host (to be renamed "Host Marriott Corporation" connection with the Merger), which was formed for the sole purpose of effecting the Reincorporation by means of the Merger, and which will be the sole general partner of the Operating Partnership, and the successor to Host, or, as the context may require, HMC Merger Corporation and the Operating Partnership collectively together with their subsidiaries or any of such subsidiaries.

"Host REIT Common Stock".....

Shares of common stock, par value \$.01 per share, of Host REIT.

"Operating Partnership".....

Host Marriott L.P., a Delaware limited partnership, the sole general partner of which will be Host REIT, or as the context may require, such entity together with its subsidiaries, including the Non-Controlled Subsidiaries (as defined below), or any of them.

"Company".....

Host (to the extent of its business and assets to be contributed to the Operating Partnership) with respect to the period prior to the REIT Conversion, and Host REIT and the Operating Partnership collectively with respect to the period after the REIT Conversion.

"Non-Controlled

Subsidiaries".....

The taxable corporations in which the Operating Partnership will own 95% of the economic interest but no voting stock and which will hold various assets contributed by Host and its subsidiaries to the Operating Partnership, which assets, if owned directly by the Operating Partnership, could jeopardize Host REIT's status as a REIT.

"Crestline"..... Crestline Capital Corporation (formerly HMC Senior Communities, Inc.), a Maryland corporation, or, as the context may require, such entity together with the Lessees (as defined below) and its other subsidiaries or any of them, which currently is a wholly-owned subsidiary of Host but will become a separate public company as part of the REIT Conversion when Host or Host REIT distributes the common stock of Crestline to its existing stockholders as part of the Initial E&P Distribution.

"REIT Conversion"	(i) The OP Contribution, (ii) the Reincorporation by means of the Merger, (iii) the Initial E&P Distribution, (iv) the recently completed refinancing and amendment of the debt securities and certain credit facilities of Host substantially in the manner described herein, (v) the Partnership Mergers (if and to the extent consummated), (vi) the Private Partnership Transactions (if and to the extent consummated), (vii) the Blackstone Acquisition (if and to the extent consummated), (viii) the creation and capitalization of the Non-Controlled Subsidiaries, (ix) the leasing of virtually all of the full-service Hotels (as defined below) owned or controlled by the Operating Partnership for initial terms ranging generally from seven to ten years (the "Leases") to lessees, which will be indirect wholly-owned subsidiaries of Crestline (the "Lessees"), whereby the Lessees will operate the hotels under their existing brand names and pursuant to their existing management agreements, and (x) such other related transactions described in "The REIT Conversion" and such other steps occurring prior to, substantially concurrent with or within a reasonable time after the Effective Date, as Host may determine in its sole discretion to be necessary or desirable to complete or facilitate the transactions contemplated herein or otherwise to permit Host REIT to elect to be treated as a REIT for federal income tax purposes for the first full taxable year commencing after the REIT Conversion.
"OP Contribution"	The contribution by Host of its wholly-owned full-service hotels, its interests in certain hotel partnerships and certain other assets to the Operating Partnership in exchange for units of limited and general partnership interest in the Operating Partnership and the assumption of liabilities.
"Reincorporation"	The reincorporation of Host from the State of Delaware to the State of Maryland by means of the Merger.
"Merger"	The proposed merger of Host with and into Host REIT, as described in this Proxy Statement/Prospectus.
"Effective Date"	The date upon which the Reincorporation by means of the Merger is consummated.
"Restructuring Transactions"	The OP Contribution and the Reincorporation by means of the Merger.
"Initial E&P Distribution"	One or more taxable distributions by Host or Host REIT to its stockholders in connection with the REIT Conversion consisting of shares of common stock of Crestline and cash or other consideration in an amount to be determined.
"OP Units"	The limited partnership interests in the Operating Partnership.
"Partnership Mergers"	The acquisition by merger, if and to the extent consummated, of up to eight limited partnerships (the "Partnerships") that own

full-service hotels in which Host or its subsidiaries are general partners in exchange for OP Units (which may, in turn, be exchanged for shares of Host REIT Common Stock or unsecured notes of the Operating Partnership (the "Notes")).

"Private Partnership Transactions".....

The acquisition of the partnership interests from unaffiliated partners of four partnerships ("Private Partnerships") that own one or more full-service hotels and that, prior to the REIT Conversion, are partially but not wholly-owned by Host or one of its subsidiaries in exchange for OP Units, if and to the extent such acquisitions are consummated.

"Blackstone Acquisition".....

The expected acquisition from The Blackstone Group and a series of funds controlled by Blackstone Real Estate Partners (collectively, the "Blackstone Entities") of ownership of, or controlling interests in, twelve hotels and a mortgage loan secured by a thirteenth hotel in exchange for OP Units, the assumption of certain liabilities and other consideration, including up to 18% of the shares of Crestline common stock, to the extent such acquisition is consummated.

"Ownership Limit".....

The prohibition in the Articles of Incorporation of Host REIT that will be in effect upon consummation of the Merger (the "Host REIT Charter") against ownership, directly or indirectly by virtue of the attribution provisions of the Internal Revenue Code of 1986, as amended (the "Code"), by any person or persons acting as a group of more than (i) 9.8% of the lesser of the number or value of shares of Host REIT Common Stock outstanding or (ii) 9.8% of the lesser of the number or value of the issued and outstanding shares of any class or series of  $\ensuremath{\mathsf{Host}}$   $\ensuremath{\mathsf{REIT}}$ preferred stock, subject to waiver or modification by Host REIT in certain limited circumstances and to certain limited exceptions for a holder of shares of Host REIT Common Stock solely by reason of the Merger in excess of the Ownership Limit so long as such holder would not own, directly or by attribution under the Code, more than 9.9% by value of the outstanding capital stock of Host REIT as of 12:01 a.m. (Eastern Time) on the first day after the effective time of the Merger (the "Special Merger Ownership Limit Effective Time").

## OVERVIEW

General. Host has adopted a plan to restructure its business operations so that it will qualify as a REIT for federal income tax purposes. Host REIT was organized as a Maryland subsidiary of Host on September 28, 1998 for the sole purpose of effecting the Reincorporation by means of the Merger. Upon consummation of the Restructuring Transactions, Host REIT will succeed to and continue the business of Host. The Restructuring Transactions are part of a series of transactions comprising the REIT Conversion. If Host REIT qualifies as a REIT, it generally will not be subject to federal corporate income taxes on that portion of its ordinary income or capital gain that is distributed to its stockholders. Such treatment would substantially eliminate the federal "double taxation" on earnings (at the corporate and stockholder levels) that generally results from investment in a corporation. See "Federal Income Tax Consequences." Host REIT has conducted no business to date other

than that incident to the REIT Conversion. This Proxy Statement/Prospectus is being furnished to the holders of Host Common Stock to solicit their approval of the Agreement, which contemplates the OP Contribution and the Reincorporation by means of the Merger.

The principal executive offices of both Host and Host REIT are located at 10400 Fernwood Road, Bethesda, Maryland, 20817-1109, telephone number (301) 380-9000.

The OP Contribution. Subject to approval of the Agreement at the Special Meeting, after the Special Meeting Host will contribute, as the OP Contribution, its wholly-owned full-service hotels, its interests in certain of the Partnerships and the Private Partnerships and certain other assets to the Operating Partnership. As a preliminary step in the REIT Conversion, during 1998, various subsidiaries of Host also have contributed or will contribute the wholly-owned full-service hotels held by such entities, their interests in certain of the Partnerships and Private Partnerships and certain other assets to the Operating Partnership (including its subsidiaries). In exchange for these contributions, the Operating Partnership will issue to Host in the aggregate a number of OP Units equal to the number of shares of Host REIT Common Stock that will be outstanding upon completion of the REIT Conversion (reduced by any OP Units to be owned by subsidiaries of Host REIT) and will assume Host's liabilities. Following these contributions, the Operating Partnership and its subsidiaries will directly or indirectly own all of Host's and its subsidiaries' wholly-owned hotels, substantially all of Host's direct and indirect interests in both the Partnerships and the Private Partnerships and all of Host's other assets (excluding its senior living assets and the cash or other consideration to be distributed as part of the Initial E&P Distribution and certain other de minimis assets that cannot be contributed to the Operating Partnership). These contributions, which include the OP Contribution, will enable Host REIT, following the Reincorporation by means of the Merger, to operate together with the Operating Partnership in an umbrella partnership REIT ("UPREIT") structure, through which Host REIT would continue the full-service hotel ownership business currently conducted by Host, and which is expected to improve Host REIT's ability to acquire additional hotel and other properties in the future on favorable terms. See "The Restructuring Transactions -- The OP Contribution."

Federal Income Tax Consequences of the OP Contribution. Section 721 of the Code provides that no gain or loss is recognized in the case of a contribution of property to a partnership in exchange for an interest in the partnership. However, there are a number of potential exceptions to the availability of such treatment. The application of these exceptions is highly complex and depends upon a number of factual determinations and other outside events which may or may not occur, but Host believes that it will not recognize a material amount of gain in connection with the OP Contribution. See "Federal Income Tax Consequences--Federal Income Tax Consequences of the OP Contribution." Host is not obtaining an opinion of outside counsel as to the tax consequences to it of the OP Contribution.

The Reincorporation by Means of the Merger. If the requisite stockholder approval of the Agreement is obtained and certain other conditions to consummation of the Merger are satisfied (including, among others, determination by the Host Board of Directors that Host REIT can elect to be treated as a REIT for federal income tax purposes effective no later than the first full taxable year commencing after the REIT Conversion is completed (which might not be until the year commencing January 1, 2000 if the REIT Conversion is not completed prior to January 1, 1999), approval of the Host REIT Common Stock for listing on the New York Stock Exchange (the "NYSE"), the receipt of governmental and material third-party consents and certain other conditions relating to the REIT Conversion), then pursuant to the Merger, Host would be merged with and into Host REIT, Host REIT would be the surviving entity in the Merger, the separate existence of Host would terminate and each issued and outstanding share of Host Common Stock, together with the associated Host Right under the Host Rights Agreement, would be converted into one share of Host REIT Common Stock, together with the associated Host REIT Right under the Host REIT Rights Agreement, in accordance with the terms of the Agreement. In addition, in the Merger, outstanding options and other rights to acquire Host Common Stock would be converted into rights to acquire Host REIT Common Stock, subject to adjustment for the Initial E&P Distribution and except as otherwise set forth in the Agreement, and any outstanding shares of preferred stock of Host would be converted into an equal number of shares of preferred stock of Host REIT having substantially the same rights and preferences. It is expected that, upon consummation of the Merger, the Host REIT Common

Stock will be listed and trade on the NYSE under the "HMT" symbol in the same manner as shares of Host Common Stock currently trade on the NYSE. See "The Restructuring Transactions--Terms of the Merger" and "--Conditions to the Merger."

Federal Income Tax Consequences of the Merger. Host has received an opinion of counsel to the effect that the Merger will qualify as a tax-free reorganization within the meaning of Section 368(a) of the Code, and accordingly: (i) no gain or loss will be recognized by Host or Host REIT as a result of the Merger; (ii) no gain or loss will be recognized by holders of shares of Host Common Stock upon the conversion of such shares into Host REIT Common Stock (except for certain stockholders who are not considered "U.S. persons" for purposes of the Code and who own (or have owned) in excess of 5% of Host); (iii) the tax basis of shares of Host REIT Common Stock received by a former holder of Host Common Stock pursuant to the Merger in the aggregate will be the same as the holder's adjusted tax basis in the shares of Host Common Stock being converted in the Merger (subject to any adjustment resulting from the Initial E&P Distribution); and (iv) the holding period of shares of Host REIT Common Stock received by a former holder of Host Common Stock pursuant to the Merger will include the holder's holding period with respect to the shares of Host Common Stock being converted in the Merger, assuming that the Host Common Stock was held as a capital asset at the effective time of the Merger. See "Federal Income Tax Consequences -- Federal Income Tax Consequences of the Meraer.'

The Initial E&P Distribution. In order to qualify as a REIT for federal income tax purposes, Host REIT and/or Host, as its predecessor, must have distributed all of the accumulated "earnings and profits" ("E&P") of Host to its stockholders in one or more taxable dividends prior to the end of the first full taxable year for which the REIT election of Host is effective, which currently is expected to be the taxable year commencing January 1, 1999 (but which might not be until the year beginning January 1, 2000). In order to help accomplish the requisite distributions of the accumulated E&P of Host, Host or Host REIT will make the Initial E&P Distribution, consisting of one or more taxable distributions to its stockholders consisting of shares of common stock of Crestline and cash or other consideration in an amount to be determined. Although there is no assurance as to what form such other consideration comprising the Initial E&P Distribution will take, it is currently contemplated that it will include a special dividend (the "Special Dividend") payable, at each stockholder's election, in cash or Host Common Stock (or Host REIT Common Stock if the Merger has occurred). The actual amount of the Initial E&P Distribution will be based in part upon the estimated amount of accumulated E&P of Host as of the last day of its taxable year ending on or immediately following completion of the REIT Conversion. To the extent that the Initial E&P Distribution is not sufficient to eliminate Host's estimated accumulated E&P, Host REIT will make one or more additional taxable distributions to its stockholders (in the form of cash or securities) prior to the last day of Host REIT's first full taxable year as a REIT (currently expected to be December 31, 1999 but which instead might be December 31, 2000) in an amount intended to be sufficient to eliminate such E&P, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions. See "The REIT Conversion -- The Initial E&P Distribution."

Federal Income Tax Consequences of the Initial E&P Distribution. Generally, the Initial E&P Distribution will be a taxable dividend to a Host stockholder to the extent that the Initial E&P Distribution is made out of the Host stockholder's share of the portion of the current and accumulated E&P of Host and Host REIT allocable to the Initial E&P Distribution. Host and Host REIT currently believe that the substantial majority, if not all, of the Initial E&P Distribution (the fair market value of which Host currently estimates will range from approximately \$2.10 to \$2.50 per share of Host Common Stock) will be considered made out of such E&P and, therefore, will be taxable as a dividend. See "Federal Income Tax Consequences of the Initial E&P Distribution."

The Ownership Limit. Primarily to satisfy certain additional requirements under the Code that are applicable to REITs in general, the Ownership Limit under the Host REIT Charter will provide that no person or persons acting as a group may own, or be deemed to own by virtue of the attribution provisions of the Code,

6

more than 9.8% of the lesser of the number or value of shares of Host REIT Common Stock (or any other class or series of Host REIT preferred stock) outstanding, subject to waiver or modification by Host REIT in certain limited circumstances and to certain limited exceptions for a holder of shares of Host REIT Common Stock solely by reason of the Merger in excess of the Ownership Limit so long as such holder would not own, directly or by attribution under the Code, more than 9.9% by value of the outstanding capital stock of Host REIT after the Special Merger Ownership Limit Effective Time. See "Description of Host REIT Capital Stock--Restrictions on Ownership and Transfer."

Other Transactions Comprising the REIT Conversion. In addition to the Restructuring Transactions and the Initial E&P Distribution, the REIT Conversion includes the following transactions, among others: (i) the recently completed refinancing and amendment of the debt securities and certain credit facilities of Host substantially in the manner described herein; (ii) the Partnership Mergers (if and to the extent consummated); (iii) the Private Partnership Transactions (if and to the extent consummated); (iv) the Blackstone Acquisition (if consummated); (v) the creation and capitalization of the Non-Controlled Subsidiaries; (vi) the leasing of virtually all of the full-service Hotels (as defined herein) to the Lessees; and (vii) such other related transactions and such other steps occurring prior to, concurrent with or within a reasonable time after the Effective Date as may be necessary or desirable to complete the transactions contemplated herein or otherwise to permit Host REIT to elect to be treated as a REIT for federal income tax purposes for the first full taxable year commencing after the REIT Conversion. See "The REIT Conversion--Other Transactions Comprising the REIT Conversion."

Because REITs are not permitted under current federal income tax law to derive revenues directly from the operation of hotels, Host REIT will lease the Hotels to Lessees that will operate the Hotels under the existing long-term management agreements (the "Management Agreements") and pay rent to Host REIT as more fully described under "Business and Properties--The Leases." The Lessees will be indirect, wholly-owned subsidiaries of Crestline. Crestline, which currently is a wholly-owned subsidiary of Host, will become a separate public company when Host distributes common stock of Crestline to its existing stockholders as part of the Initial E&P Distribution. Shares of Host REIT Common Stock and Crestline common stock will be separately traded securities, and the companies will operate independently. There will be no overlap between the boards of Host REIT and Crestline. There will be a substantial overlap of stockholders of the two companies initially, but this overlap will diverge over time.

As the first step in a strategy to acquire non-Marriott as well as Marriott branded hotels (which strategy will be subject to certain limitations through October 8, 2000), Host has entered into an agreement with various affiliates of the Blackstone Entities to acquire from the Blackstone Entities ownership of, or controlling interests in, twelve upscale and luxury full-service hotel properties (the "Blackstone Hotels") and certain other related assets (including a mortgage loan secured by an additional hotel) in exchange for a combination of cash and the assumption of debt totalling \$862 million, 43.7 million OP Units (based upon a negotiated value of \$20.00 per OP Unit) and up to 18% of the shares of Crestline common stock and other consideration. If the Blackstone Acquisition is consummated, the interests in the Blackstone Hotels will be contributed by the Blackstone Entities to the Operating Partnership as part of the REIT Conversion. The Blackstone Hotels will be leased to Lessees that are subsidiaries of Crestline and will continue to be managed under their existing management agreements. See "Business and Properties--Blackstone Acquisition."

Reasons for the Restructuring Transactions and the REIT Conversion. The Host Board of Directors is proposing the Restructuring Transactions primarily for the following reasons:

. The Restructuring Transactions are essential components of Host's conversion to REIT status for federal income tax purposes. In particular, the OP Contribution will enable Host REIT, following the Merger, to operate, together with the Operating Partnership, in an UPREIT structure, through which Host REIT would continue the full-service hotel ownership business currently conducted by Host. Host believes that the UPREIT structure will improve its ability to acquire additional properties in the future on favorable terms. Specifically, under certain circumstances, OP Units could be issued to acquire properties in

7

transactions that would not trigger immediate tax obligations for certain sellers. Accordingly, converting to an UPREIT structure could enable Host REIT to acquire hotel and other properties in the future at lower prices because of the tax advantages to some sellers of receiving OP Units as consideration. OP Units would subsequently be redeemable for cash or common stock of Host REIT (at the option of Host REIT) at such time as the recipient desires liquidity.

- . The provisions of Maryland law have generally been viewed as favorable to REITs organized in corporate or trust form. As discussed below, Maryland law facilitates qualification as a REIT by authorizing the charter of a Maryland corporation to provide for restrictions on ownership and transferability designed to permit a corporation to qualify as a REIT under the Code or for any other purpose. In addition, unlike Delaware, Maryland does not impose a franchise tax on corporations, which will result in cost savings to Host in annual franchise tax payments and related fees of approximately \$150,000. Maryland's status as a jurisdiction favorable to REITs is evidenced by the large number of publicly-traded REITs that have chosen to operate as a regular Maryland corporation or as a special statutory Maryland real estate investment trust. According to the National Association of Real Estate Investment Trusts, Inc. ("NAREIT"), as of September 1998, there were over 100 publicly-traded REITs organized under Maryland law.
- . In order to satisfy certain requirements that are applicable to REITs in general, many REITs impose ownership limits and transfer restrictions, similar to the Ownership Limit under the Host REIT Charter, by inclusion of such provisions in their charters. Under Delaware law, such restrictions would not be binding with respect to securities issued prior to adoption of the restriction unless holders of such securities agree to or vote in favor of such restriction. However, under Maryland law and by reason of the Merger, all shares of Host REIT Common Stock issued in the Merger and thereafter would be subject to the Ownership Limit, for which authority exists under Maryland law.
- . Host's principal executive offices and a substantial number of Host's employees are employed in Maryland.

The Board of Directors of Host believes that the Restructuring Transactions and the other transactions comprising the REIT Conversion are advisable for Host and its stockholders based on the belief that:

- . The REIT structure, as a more tax efficient structure, will provide improved operating results through changing economic conditions and all phases of the hotel economic cycle.
- . The REIT Conversion, which will reduce corporate-level taxes and the need to incur debt to reduce corporate taxes through interest deductions, will improve its financial flexibility and allow it to continue to strengthen its balance sheet by reducing its overall debt to equity ratio over time.
- . As a REIT, Host will be able to compete more effectively with other public lodging real estate companies that already are organized as REITs and to make performance comparisons with its peers more meaningful.
- . By becoming a dividend paying company, Host's stockholder base will expand to include investors attracted by yield as well as asset quality, which is expected to facilitate Host REIT's capital-raising efforts and provide a less volatile stockholder base.
- . The adoption of an UPREIT structure will facilitate tax-deferred acquisition of additional hotels (such as in the case of the Blackstone Acquisition and the Partnership Mergers).

Host believes that these benefits justify the REIT Conversion even if the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999 (in which event the effectiveness of Host REIT's election could be delayed until January 1, 2000). See "The Restructuring Transactions--Background and Reasons for the Restructuring Transactions and the REIT Conversion."

Opinion of Financial Advisor. Host's Board of Directors has received an opinion dated November , 1998 from BT Wolfensohn, which acted as financial advisor to Host in connection with the REIT Conversion, to the effect that the REIT Transactions (as defined in such opinion), taken together, are fair from a financial point of view to the holders of Host Common Stock. See "Opinion of Financial Advisor." The Host Board of Directors considered such opinion as well as other matters it deemed relevant in determining the advisability of the Agreement and the other transactions comprising the REIT Conversion.

Certain Effects of the REIT Conversion. The Host Board believes that Host's conversion to a REIT and the distribution of Crestline shares as part of the Initial E&P Distribution will benefit Host's stockholders by providing them with a tax advantaged REIT security that is expected to provide both the opportunity for regular cash dividends and capital appreciation as Host REIT acquires additional properties, as well as a continuing interest in Crestline, Host's senior living company and the initial lessee of substantially all of Host REIT's hotels, if a Host stockholder continues to hold the Crestline common stock. As a REIT, Host REIT would be able to benefit from the tax advantages that apply to REITs, and stockholders would receive quarterly distributions that are expected to be at least sufficient to satisfy the annual distribution requirements applicable to REITs under the Code. The Host Board believes that this will highlight the value of Host REIT's hotel properties and permit stockholders to realize a regular cash return on that value. Upon completion of the REIT Conversion, Crestline is expected to own Host's 31 senior living communities, which will continue to be managed by Marriott International, and a 25% interest in the Swissotel management company expected to be acquired in the Blackstone Acquisition, and will lease substantially all of the Hotels (as defined below) owned by Host REIT and its affiliates, as further described in this Proxy Statement/Prospectus. At that time, Crestline will operate independently of Host, will be publicly traded and separately listed on the NYSE, and will pursue its own growth opportunities.

Following the Restructuring Transactions and the other transactions comprising the REIT Conversion (including the Blackstone Acquisition), Host REIT and its subsidiaries are expected initially to own outright, or have controlling interests in, approximately 125 full-service hotels operated primarily under the Marriott, Ritz-Carlton, Four Seasons, Swissotel and Hyatt brand names (the "Hotels"). Upon completion of the REIT Conversion, Host REIT will be the sole general partner of the Operating Partnership and is expected to own approximately 74% of the OP Units in the Operating Partnership, and the remaining OP Units, which are intended to be substantially equivalent on an economic basis to shares of Host REIT Common Stock, will be owned by the limited partners (that are unaffiliated with Host) of the Partnerships participating in the Partnership Mergers (the "Limited Partners"), unaffiliated partners of the four Private Partnerships participating in the Private Partnership Transactions and the Blackstone Entities. Host REIT and its subsidiaries will own a number of OP Units equal to the number of shares of Host REIT Common Stock outstanding. Host REIT will be managed by its Board of Directors and will have no employees who are not also employees of the Operating Partnership.

Although a number of the transactions comprising the REIT Conversion are expected to be consummated immediately prior to, or in certain instances immediately following, the Merger, the Merger will not be consummated unless the conditions to the Merger have been satisfied or waived. In particular, Host's Board of Directors will have determined, among other things, that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur, and based on advice of counsel, that Host REIT can elect to be treated as a REIT for federal income tax purposes effective no later than the first full taxable year commencing after the REIT Conversion is completed (which might not be until the year commencing January 1, 2000 if the REIT Conversion is not completed prior to January 1, 1999). Consistent with the foregoing, Host intends to pursue the transactions constituting the REIT Conversion at least through the date of the Special Meeting of Host stockholders. If the Agreement is approved by Host stockholders at the Special Meeting, Host intends to continue pursuing those transactions constituting the REIT Conversion which have not yet been completed, including the Blackstone

Acquisition (which is not expected to be consummated any earlier than December 29, 1998). In addition, if the Board of Directors has determined that the conditions to the Merger have been or likely will be satisfied or waived (and, in particular, that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur), the Board intends to declare the Initial E&P Distribution and enter into the Leases with the Lessees (which will be indirect, wholly-owned subsidiaries of Crestline). Under such circumstances, however, the Merger and other transactions comprising the REIT Conversion still might be delayed or possibly might never be consummated. Assuming the Agreement is approved by Host stockholders at the Special Meeting and the Host Board of Directors makes the determination described above, it is currently contemplated that the Merger would be consummated on or about December 29, 1998, subject to satisfaction or waiver of the remaining conditions. If, however, the Agreement is not approved by the Host stockholders at the Special Meeting or the Host Board of Directors does not make the requisite determinations, Host will continue to operate as a Delaware corporation, the REIT Conversion will not be completed at this time and the Initial E&P Distribution will not be made.

If the REIT Conversion otherwise does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the effectiveness of Host REIT's election could be delayed until January 1, 2000, which would result in Host REIT continuing to pay substantial corporate-level income taxes in 1999 (which would reduce Host REIT's estimated cash distributions per share of Host REIT Common Stock) and could cause the Blackstone Acquisition (which is conditioned, among other things, on consummation of the REIT Conversion by March 31, 1999 and Host REIT qualifying as a REIT for 1999) not to be consummated. If Host REIT's election to be taxed as a REIT is not effective on January 1, 1999, Host REIT intends to operate following the REIT Conversion in a manner that would permit it to qualify as a REIT at the earliest time practicable, and it might pursue a merger with another entity or other transaction that would permit it to commence a new taxable year and elect REIT status prior to January 1, 2000, although no assurance can be given that the Company will enter into or consummate such other transaction or otherwise qualify as a REIT prior to January 1, 2000. Host REIT in any event would elect to be treated as a REIT for federal income tax purposes no later than its taxable year commencing January 1, 2000. See "The REIT Conversion."

## RISK FACTORS

The following is a summary of the material risks associated with the Restructuring Transactions and the other transactions comprising the REIT Conversion. This summary is qualified in its entirety by the detailed discussion in the section entitled "Risk Factors" contained in this Proxy Statement/Prospectus. Some of the significant matters that stockholders of Host should consider carefully when voting on the Agreement, and the transactions contemplated thereby, include:

- . Lack of Control over Hotel Operations and Non-Controlled Subsidiaries. Due to current federal income tax law restrictions on a REIT's ability to derive revenues directly from the operation of a hotel, Host REIT will lease virtually all of its consolidated Hotels to the Lessees, which will operate the Hotels by continuing to retain the existing managers of the Hotels (the "Managers") pursuant to the existing long-term Management Agreements. Host REIT will not operate the Hotels or participate in the decisions affecting the daily operations of the Hotels. Host REIT will have only a limited ability to require the Lessees or the Managers to operate or manage the Hotels in any particular manner, and no ability to govern any particular aspect of their day-to-day operation or management. Host REIT also will not own any of the voting stock of the Non-Controlled Subsidiaries, which may own, in the aggregate, up to 20% by value of Host REIT's assets. Therefore, Host REIT will be dependent for its revenue upon the ability of the Lessees and the Managers to operate and manage the Hotels and the Non-Controlled Subsidiaries to operate and manage their businesses.
- Dependence upon Crestline. Subsidiaries of Crestline will be the Lessees of virtually all of the Hotels and their rent payments will be the primary source of Host REIT's revenues. Upon the REIT Conversion, all fees payable under the Management Agreements for subsequent periods will become the primary obligations of the Lessees, to be paid by the Lessees for so long as the Leases remain in effect. The obligations of the Lessees will be guaranteed by Crestline, subject to specified liability limitations. Crestline's and each Lessee's financial condition and ability to meet its obligations under the Leases and the Management Agreements will determine the Operating Partnership's ability to make distributions to

holders of OP Units, including Host REIT, and Host REIT's ability, in turn, to make distributions to its stockholders. As of September 11, 1998, on a pro forma basis, after giving effect to the REIT Conversion, Crestline would have had approximately \$300 million of indebtedness (including approximately \$85 million due to Host REIT to pay for hotel working capital purchased from Host REIT but not including guarantees of obligations of Crestline's subsidiaries under the Leases and the Management Agreements), and Crestline can incur additional indebtedness in the future. There can be no assurance that Crestline or any Lessee will have sufficient assets, income and access to financing to enable it to satisfy its obligations under the Leases or to make payments of fees under the Management Agreements. Host REIT remains obligated to the Managers in case the Lessee fails to pay these fees (but it would be entitled to reimbursement from the Lessee under the terms of the Leases). The credit rating of the Operating Partnership and Host REIT will be affected by the general creditworthiness of Crestline.

- . Current Host Common Stock Price Is Not Necessarily Indicative of the Price of Host REIT Common Stock Following the REIT Conversion. Host's current stock price is not necessarily indicative of how the market will value Host REIT Common Stock following the REIT Conversion. The current stock price of Host reflects the current market valuation of Host's current business and assets (including the Crestline common stock and cash or other consideration to be distributed in connection with the Initial E&P Distribution) and not solely the business and assets of Host REIT following the REIT Conversion. Host's current stock price also is affected by general market conditions.
- . Cash Distributions May Exceed Cash Available for Distribution. The preliminary estimated initial annual cash distributions of the Operating Partnership during the twelve months ending December 31, 1999 (\$231 million) will exceed the estimated cash available for distribution (\$144 million) and cash from contingent rents (\$64 million) during the twelve months ending December 31, 1999, which would require borrowings by the Operating Partnership of approximately \$23 million to make such distributions in accordance with the Operating Partnership's distribution policy. Moreover, if estimated cash from contingent rents were less than \$64 million, then the Operating Partnership also would be required to borrow any such shortfall in order to make such distributions.
- . Timing of the Restructuring Transactions and the REIT Conversion. Host intends to cause the Restructuring Transactions and the other transactions comprising the REIT Conversion to be completed as soon as possible, but there is no assurance that the REIT Conversion will be completed during 1998 in time for Host REIT to elect REIT status effective January 1, 1999. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the effectiveness of Host REIT's election could be delayed until January 1, 2000, which would result in Host REIT continuing to pay substantial corporate-level income taxes in 1999 (which would be expected to reduce Host REIT's estimated cash distributions during 1999 to \$0.53 per share from \$0.84 per share if its REIT election were effective on January 1, 1999) and could cause the Blackstone Acquisition (which is conditioned, among other things, on consummation of the REIT Conversion by March 31, 1999 and Host REIT qualifying as a REIT for 1999) not to be consummated.
- Responsibilities of Host REIT to Other Limited Partners in the Operating Partnership. Maryland law imposes certain duties on the Board of Directors of Host REIT to its stockholders. In addition, after the REIT Conversion, Host REIT, as the sole general partner of the Operating Partnership, will have fiduciary obligations under Delaware law with respect to the other limited partners in the Operating Partnership (to the extent such duties have not been modified or eliminated pursuant to the terms of the amended and restated agreement of limited partnership of the Operating Partnership (the "Partnership Agreement"). Although Delaware law provides that Host REIT, as general partner, is subject to the duties of care and loyalty with respect to the limited partners of the Operating Partnership, the Partnership Agreement imposes certain limitations on Host REIT's fiduciary obligations with respect to such limited partners. See "Distribution and Other Policies--Conflicts of Interest Policies." Although some courts

have interpreted the fiduciary duties of the Board of Directors in the same way as the duties of a general partner in a limited partnership, it is unclear whether, or to what extent, there are differences in such fiduciary duties. It is possible that, notwithstanding the contractual limitations in the Partnership Agreement, the fiduciary duties of the directors of Host REIT to the stockholders may be less than those of Host REIT, as general partner of the Operating Partnership, to the limited partners of the Operating Partnership. Thus, the discharge of Host REIT's obligations to its stockholders and to the limited partners in the Operating Partnership could result in decisions that may fail to reflect fully the interest of all holders of Host REIT Common Stock and limited partners of the Operating Partnership.

- . Relationships with Marriott International and Crestline. Marriott International currently serves as manager for all but 16 of Host's Hotels, and will continue to manage those Hotels pursuant to the Management Agreements that will be assigned to the Lessees. In addition, Marriott International acts as manager of hotels that compete with Host REIT's Hotels. As a result, Marriott International may make decisions regarding competing lodging facilities which it manages that would not necessarily be in the best interests of Host REIT or the Lessees. Further, J.W. Marriott, Jr. and Richard E. Marriott, who are brothers, currently serve as directors of Host and directors (and, in the case of J.W. Marriott, Jr., also an officer) of Marriott International. After the REIT Conversion, J.W. Marriott, Jr. will serve as a director of Host REIT and will continue to serve as a director and officer of Marriott International, and Richard E. Marriott will serve as Chairman of the Board of Host REIT and continue to serve as a director of Marriott International. J.W. Marriott, Jr. and Richard E. Marriott also beneficially own (as determined for securities law purposes) approximately 10.6% and 10.2%, respectively, of the outstanding shares of common stock of Marriott International, and will beneficially own approximately 5.33% and 5.31%, respectively, of the outstanding shares of common stock of Crestline (but neither will serve as an officer or director thereof). As a result, J.W. Marriott, Jr. and Richard E. Marriott may have a potential conflict of interest with respect to their obligations as directors of Host REIT in connection with any decisions regarding Marriott International itself (including decisions relating to the Management Agreements involving the Hotels), Marriott International's management of competing lodging properties and Crestline's leasing and other businesses that would not necessarily be in the best interests of Host RETT.
- . Uncertainties at the Time of Voting on the Agreement. There are several uncertainties relating to the REIT Conversion that will exist at the time Host stockholders vote on the Agreement, including (i) the occurrence of the Blackstone Acquisition, (ii) the results of the Partnership Mergers and the number and value of the OP Units to be issued to the Limited Partners therein, (iii) the assets and liabilities of the Non-Controlled Subsidiaries and (iv) the value and components of the Initial E&P Distribution. There is no assurance as to the outcome of such matters, and if certain of these transactions or events are not consummated as expected, either the REIT Conversion may not be consummated or the value of Host REIT Common Stock and the amount of cash available for distribution to stockholders could be adversely affected, and Host REIT could fail to qualify as a REIT for federal income tax purposes.
- . Inability to Obtain Third-Party Consents May Have a Material Adverse Effect. There are numerous third-party consents which are required to be obtained in order to consummate the Restructuring Transactions and the other transactions comprising the REIT Conversion. The inability of Host, the Operating Partnership or Host REIT to obtain one or more such consents could cause a default under cross-default provisions of the Company's principal credit facilities or otherwise have a material adverse effect on the Operating Partnership and Host REIT and thus could reduce the value of Host REIT Common Stock.
- Expiration of the Leases and Possible Inability to Find Other Lessees. The Leases generally will expire seven to ten years after the Effective Date, and there can be no assurance that the affected Hotels will be relet (or if relet, will be relet on terms as favorable to Host REIT). If the Hotels are not relet to the Lessees, Host REIT will be required to find other lessees, which lessees must meet certain requirements set forth in the Management Agreements and the Code. There can be no assurance that satisfactory lessees could be found or as to the terms and conditions on which Host REIT would be able to relet the Hotels or enter into new leases with such lessees, which could result in a failure of Host REIT to qualify as a REIT or in reduced cash available for distribution.
- . Leases Could Impair the Sale or Other Disposition of Host REIT's Hotels. Each Lease generally provides for a termination payment if the Lease is terminated by Host REIT prior to the expiration of the term of such Lease, except following a default by a Lessee and in certain other circumstances (including in connection with the sale of up to 12 Hotels without a termination payment) or unless Host REIT leases

to the Lessee a comparable substitute hotel. The payment of such termination fee under the Leases could have the effect of impairing the ability of Host REIT to sell its Hotels if market conditions otherwise warrant such a sale and would reduce the net proceeds of any such sale.

- . Limitations on Sale or Refinancing of Certain Hotels. For reasons relating to federal income tax considerations, the agreements by which the Operating Partnership will acquire certain Hotels (or obtain consent to lease certain Hotels to the Lessees) will also restrict the ability of the Operating Partnership to dispose of or refinance the debt secured by such Hotels for varying periods from the Effective Date, depending on the Hotel. Similarly, the Operating Partnership will agree generally not to dispose of the Blackstone Hotels for five or ten years, depending on the circumstances. Thus, it may be difficult or impossible for Host REIT to sell such Hotels or refinance such debt during their respective lock-out periods.
- Possible Adverse Consequences of the Ownership Limit. Any shares of Host REIT Common Stock held at the effective time of the Merger and not subsequently transferred prior to the Special Merger Ownership Limit Effective Time, or acquired or held at any time after the effective time of the Merger, in violation of the Ownership Limit, will be transferred automatically to a trust for the benefit of a designated charitable beneficiary, and the person holding such excess shares of Host REIT Common Stock will not be entitled to any distributions thereon or to vote such excess shares of Host REIT Common Stock. The holder of any such excess shares of Host REIT Common Stock will receive the lesser of the value of such excess shares as of the effective time of the Merger or the cash proceeds from the sale of such excess shares of Host REIT Common Stock by the designated charitable beneficiary. After the effective time of the Merger, any person who acquires Host REIT Common Stock in excess of the Ownership Limit will not receive any proceeds from the subsequent sale thereof in excess of the lesser of the price paid therefor or the amount realized from such sale. As a result, the Ownership Limit would adversely effect the value of any Host REIT Common Stock held in excess of the Ownership Limit. In addition, the Ownership Limit may have the effect of delaying, deferring or preventing a change in control of Host REIT and, as a result, could adversely effect the value of Host REIT
- . Anti-Takeover Effect of Certain Provisions of the Host REIT Charter and Bylaws, Maryland Law and the Host REIT Rights Agreement. The Host REIT Charter and the bylaws of Host REIT (the "Host REIT Bylaws") to be effective upon completion of the Merger, as well as provisions of Maryland law, will contain certain provisions that could have the effect of delaying, deferring or preventing a change in control of Host REIT. These provisions could limit the price that certain investors might be willing to pay in the future for Host REIT Common Stock. Certain of these provisions provide for a staggered board and allow Host REIT to issue, without stockholder approval, preferred stock or other stock having rights senior to those of the Host REIT Common Stock. The Board of Directors also will be authorized, without a vote of stockholders, to classify or reclassify unissued common stock or preferred stock into another class or series of stock. Other provisions impose various procedural and other requirements that could make it difficult for stockholders to effect certain corporate actions. The Host REIT Charter also will provide that no person or persons acting as a group may own more that 9.8% (in number or value) of the outstanding shares of any class or series of Host REIT stock, subject to certain exceptions. Host REIT also intends to adopt the Host REIT Rights Agreement to replace the existing stockholder rights plan of Host. Host REIT also will become subject to the business combination and control share provisions under Maryland law. Marriott International has the right to purchase 20% of each class of Host's outstanding voting stock at the then fair market value upon the occurrence of certain change of control (or potential change of control) events involving Host, which right will continue in effect after the Merger until June 2017, subject to certain limitations intended to protect the REIT status of Host REIT. See "Certain Provisions of Maryland Law and the Host REIT Charter and Bylaws--Marriott International Purchase Right."
- . Failure of Host REIT to Qualify as a REIT for Tax Purposes. Taxation of Host REIT as a corporation if it fails to qualify as a REIT, and Host REIT's subsequent liability for federal, state and local taxes on

its income and property, would, among other things, have the effect of reducing cash available for distribution to Host REIT's stockholders and materially reducing the value of the Host REIT Common Stock.

- . Change in Tax Laws. No assurance can be provided that new legislation, Treasury Regulations, administrative interpretations or court decisions will not significantly change the tax laws with respect to Host REIT's qualification as a REIT or the federal income tax consequences of such qualification.
- . Failure of the Operating Partnership to Qualify as a Partnership for Tax Purposes. Taxation of the Operating Partnership as a corporation if it fails to qualify as a partnership and the Operating Partnership's subsequent liability for federal, state and local income taxes would, among other things, have the effect of reducing cash available for distribution to Host REIT stockholders and causing Host REIT to fail to qualify as a REIT for tax purposes, thereby materially reducing the value of the Host REIT Common Stock.
- . Failure of the Leases to Qualify as Leases for REIT Purposes. If one or more of the Leases were to be disregarded for tax purposes (for example, because a Lease was determined to lack economic substance) or if the rent payable under a Lease were determined to be based upon the income or profits of the applicable Lessees, Host REIT would fail to qualify as a REIT and the Operating Partnership might be treated as a corporation for federal income tax purposes, all of which would have a material adverse impact on Host REIT and the value of the Host REIT Common Stock.
- . Host REIT's Substantial Deferred and Contingent Tax Liabilities. Host REIT will have substantial deferred tax liabilities attributable to Host's assets and operations that are likely to be recognized in the next ten years (notwithstanding Host REIT's status as a REIT), and the IRS could assert substantial additional liabilities for taxes against Host for taxable years prior to the time Host REIT qualifies as a REIT.
- . Competition in the Lodging Industry. The profitability of the Hotels is subject to general economic conditions, the management abilities of the Managers (including primarily Marriott International), competition, the desirability of particular locations and other factors relating to the operation of the Hotels. The full-service segment of the lodging industry, in which virtually all of the Hotels operate, is highly competitive, and the Hotels generally operate in geographical markets that contain numerous competitors. The Hotels' success will be dependent, in large part, upon their ability to compete in such areas as access, location, quality of accommodations, room rate structure, the quality and scope of food and beverage facilities and other services and amenities. The lodging industry, including the Hotels (and thus Host REIT), may be adversely affected in the future by (i) national and regional economic conditions, (ii) changes in travel patterns, (iii) taxes and government regulations which influence or determine wages, prices, interest rates, construction procedures and costs, (iv) the availability of credit and (v) other factors beyond the control of Host REIT.
- . Substantial Indebtedness of Host REIT. Host REIT will have substantial indebtedness. As of September 11, 1998, on a pro forma basis assuming the Full Participation Scenario, Host REIT had outstanding indebtedness totaling approximately \$5 billion, which represents an approximately 59% debt-to-total market capitalization ratio on a pro forma basis at such date (based upon an assumed price per share of Host REIT Common Stock of \$12.50 after giving effect to the REIT Conversion). Host REIT's business is capital intensive and it will have significant capital requirements in the future. Host REIT's leverage level and other factors beyond its control (including market conditions) could affect its ability to (i) obtain financing in the future, (ii) undertake refinancings on terms and subject to conditions deemed acceptable by Host REIT, (iii) make distributions, (iv) pursue its acquisition strategy or (v) compete effectively or operate successfully under adverse economic conditions.
- . No Limitation on Debt. There are no limitations in the Operating Partnership's or Host REIT's organizational documents limiting the amount of indebtedness that either may incur, although certain of Host REIT's debt instruments will contain a number of restrictions on the amount of indebtedness that Host REIT may incur. In addition, Host REIT will have a policy of incurring debt only if upon such

incurrence the debt-to-total market capitalization of Host REIT and the Operating Partnership would be 60% or less. Host REIT's Board of Directors, however, may unilaterally change this policy without stockholder approval. If Host REIT's policy of incurring debt were changed, Host REIT could become more highly leveraged which could adversely affect Host REIT's cash flow and the cash available for distribution to stockholders of Host REIT, and could increase the risk of default on Host REIT's indebtedness.

- . The Board May Change Investment and Other Policies Without Stockholder Approval. Host REIT's Board of Directors may change the investment, financing and other policies of Host REIT without stockholder approval. Such policy changes may have adverse consequences to Host REIT.
- . Management Agreements Could Impair the Sale or Other Disposition of Host REIT's Hotels. The Hotels generally may not be sold, leased or otherwise transferred unless the transferee assumes the Management Agreements relating thereto and meets certain other conditions. The possible desire of Host REIT, from time to time, to finance, refinance or effect a sale of any of the properties managed by Marriott International (which currently serves as manager for all but 16 of Host REIT's Hotels) may, depending upon the structure of such transactions, result in a need to obtain Marriott International's consent, which could include modification of the Management Agreement with Marriott International with respect to such property. The lack of any required consent from Marriott International would prohibit Host REIT from consummating such financing or sale without breaching such Management Agreement.
- .Year 2000 Problem. Year 2000 issues have arisen because many existing computer programs and chip-based embedded technology systems use only the last two digits to refer to a year, and therefore do not properly recognize a year that begins with "20" instead of the familiar "19". Although Host has adopted a compliance program in recognition of the possible interruptions that may occur as a result of Year 2000 issues and taken certain steps toward Year 2000 remediation, there can be no assurances that the steps taken toward Year 2000 remediation by Host or third parties will be properly and timely completed, and failure to do so could have a material adverse effect on Host, its business and its financial condition. Following the REIT Conversion, Host REIT will be dependent upon Crestline to interface with third parties in addressing Year 2000 issues at the Hotels leased to Crestline.
- .Tax Consequences to Host of the OP Contribution and the Other Transactions Comprising the REIT Conversion. Because of the numerous transactions that comprise the OP Contribution and the REIT Conversion and the inherently factual nature of a number of the critical determinations relevant to the tax consequences to Host of the OP Contribution and the REIT Conversion, Host has not sought (and likely could not obtain) an opinion of outside counsel as to the tax consequences to it (in terms of the actual amount of income and gain that would be recognized) of the OP Contribution and the other transactions comprising the REIT Conversion (other than the Merger). Host believes, based upon its knowledge of the relevant facts and advice that it has received from various professional advisers (including both attorneys and accountants on its staff and outside attorneys and accountants) as to the applicable general principles of law, that it will not recognize a material amount of income and gain as a result of the OP Contribution and the other transactions comprising the REIT Conversion. Host, however, will recognize some income and gain in connection with these various transactions, and there can be no assurance that under the relevant facts and circumstances such income and gain would not be material. If Host were to recognize a material amount of income and gain as a result of these various transactions, the resulting income tax liability could be material to Host, and the increase in E&P resulting from such income and gain could materially increase the amount that Host must distribute to its stockholders in order to qualify as a REIT, all of which could have adverse consequences to Host REIT and its stockholders.

15

## DATE, TIME, PLACE AND PURPOSE OF SPECIAL MEETING

The Special Meeting will be held at the Ritz-Carlton Hotel, 1700 Tysons Boulevard, McLean, Virginia, 22102, on December 15, 1998 at 10:00 a.m., local time, to approve the Agreement.

## STOCKHOLDERS ENTITLED TO VOTE

The Board of Directors has fixed the close of business on November 13, 1998 as the Record Date for the determination of stockholders entitled to receive notice of and vote at the Special Meeting. As of the Record Date, Host had outstanding and entitled to vote 204,954,447 shares of Host Common Stock and 47,845 holders of record. See "Voting and Proxies."

## VOTE REQUIRED; NO DISSENTERS' RIGHTS

Under the Delaware General Corporation Law, as amended (the "DGCL"), the affirmative vote of a majority of the outstanding capital stock of the corporation entitled to vote thereon is needed to adopt a merger agreement. Similarly, if the OP Contribution were deemed to constitute a sale, lease or exchange of all or substantially all of the assets of Host under the DGCL, the affirmative vote of a majority of the outstanding

capital stock of the corporation entitled to vote thereon would be needed to approve the OP Contribution. Pursuant to Host's Restated Certificate of Incorporation (the "Host Certificate"), the affirmative vote of the holders of two-thirds (66 2/3%) of the outstanding shares of Host Common Stock is necessary for approval of the Merger, as well as the OP Contribution if it were deemed to constitute a sale, lease or exchange of all or substantially all of the assets of Host. On the Record Date, directors and executive officers of Host, together with their affiliates, as a group, beneficially owned approximately 13.7% of the issued and outstanding shares of Host Common Stock. It is currently expected that each director and executive officer of Host will vote the shares of Host Common Stock beneficially owned by such director or executive officer for approval of the Agreement and the transactions contemplated thereby. See "Principal Stockholders."

Under the DGCL, Host's stockholders will not be entitled to dissenters' rights of appraisal as a result of the Merger or other Restructuring Transactions. See "Voting and Proxies" and "The Restructuring Transactions--Absence of Dissenters' Rights."

IF THE RESTRUCTURING TRANSACTIONS AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION DO NOT OCCUR IN TIME FOR HOST REIT TO ELECT REIT STATUS EFFECTIVE JANUARY 1, 1999, THE EFFECTIVENESS OF HOST REIT'S ELECTION COULD BE DELAYED TO JANUARY 1, 2000, WHICH WOULD RESULT IN HOST OR HOST REIT CONTINUING TO PAY SUBSTANTIAL CORPORATE-LEVEL INCOME TAXES IN 1999 AND COULD CAUSE THE BLACKSTONE ACQUISITION (WHICH IS CONDITIONED, AMONG OTHER THINGS, ON CONSUMMATION OF THE REIT CONVERSION BY MARCH 31, 1999 AND HOST REIT QUALIFYING AS A REIT FOR 1999) NOT TO BE CONSUMMATED. THEREFORE, IT IS EXTREMELY IMPORTANT THAT STOCKHOLDERS RETURN THE PROXY CARDS AS SOON AS POSSIBLE OR PROPERLY SUBMIT THEIR PROXIES BY TELEPHONE OR THE INTERNET IN ACCORDANCE WITH THE INSTRUCTIONS ON THE ADMISSION TICKET ATTACHED TO THE ENCLOSED PROXY CARD. ANY DELAY IN RETURNING PROXIES COULD CAUSE THE SPECIAL MEETING TO BE DELAYED, WHICH COULD PREVENT THE RESTRUCTURING TRANSACTIONS AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION FROM BEING EFFECTIVE JANUARY 1, 1999.

#### THE RESTRUCTURING TRANSACTIONS

Background and Reasons for the Restructuring Transactions and the REIT Conversion. Host is proposing the Restructuring Transactions in connection with a plan adopted by Host to restructure its business operations so that it will qualify as a REIT under the Code. Host REIT expects to qualify as a REIT beginning with its first full taxable year commencing after the REIT Conversion is completed, which currently is expected to be the year commencing January 1, 1999 (but which might not be until the year beginning January 1, 2000). The Host Board of Directors is proposing the Restructuring Transactions primarily for the following reasons:

- . The Restructuring Transactions are essential components of Host's conversion to REIT status for federal income tax purposes. In particular, the OP Contribution will enable Host REIT, following the Merger, to operate, together with the Operating Partnership, in an UPREIT structure, through which Host REIT would continue the full-service hotel ownership business currently conducted by Host. Host believes that the UPREIT structure will improve its ability to acquire additional properties in the future on favorable terms. Specifically, under certain circumstances, OP Units could be issued to acquire properties in transactions that would not trigger immediate tax obligations for certain sellers. Accordingly, converting to an UPREIT structure could enable Host REIT to acquire hotels and other properties in the future at lower prices because of the tax advantages to some sellers of receiving OP Units as consideration. OP Units would subsequently be redeemable for cash or common stock of Host REIT (at the option of Host REIT) at such time as the recipient desires liquidity.
- . The provisions of Maryland law have generally been viewed as favorable to REITs organized in corporate or trust form. As discussed below, Maryland law facilitates qualification as a REIT by authorizing the charter of a Maryland corporation to provide for restrictions on ownership and transferability designed to permit a corporation to qualify as a REIT under the Code or for any other purpose. In addition, unlike Delaware, Maryland does not impose a franchise tax on corporations, which will result in cost savings to

16

Host in annual franchise tax payments and related fees of approximately \$150,000. Maryland's status as a jurisdiction favorable to REITs is evidenced by the large number of publicly-traded REITs that have chosen to operate as a regular Maryland corporation or as a special statutory Maryland real estate investment trust. According to NAREIT, as of September 1998, there were over 100 publicly-traded REITs organized under Maryland law.

- . In particular, in order to satisfy certain requirements that are applicable to REITs in general, many REITs impose ownership limits and transfer restrictions, similar to the Ownership Limit under the Host REIT Charter, by inclusion of such provisions in their charters. Under Delaware law, such restrictions would not be binding with respect to securities issued prior to adoption of the restriction unless holders of such securities agree to or vote in favor of such restriction. However, under Maryland law and by reason of the Merger, all shares of Host REIT Common Stock issued in the Merger and thereafter would be subject to the Ownership Limit, for which authority exists under Maryland law.
- . Host's principal executive offices and a substantial number of Host's employees are employed in Maryland.

The Board of Directors of Host believes that the Restructuring Transactions and the other transactions comprising the REIT Conversion are advisable for Host and its stockholders based on the belief that:

- . The REIT structure, as a more tax efficient structure, will provide improved operating results through changing economic conditions and all phases of the hotel economic cycle.
- . The REIT Conversion, which will reduce corporate-level taxes and the need to incur debt to reduce corporate taxes through interest deductions, will improve its financial flexibility and allow it to continue to strengthen its balance sheet by reducing its overall debt to equity ratio over time.
- . As a REIT, Host will be able to compete more effectively with other public lodging real estate companies that already are organized as REITs and to make performance comparisons with its peers more meaningful.
- . By becoming a dividend paying company, Host's stockholder base will expand to include investors attracted by yield as well as asset quality, which is expected to facilitate Host REIT's capital-raising efforts and provide a less volatile stockholder base.
- . The adoption of an UPREIT structure will facilitate tax-deferred acquisition of additional hotels (such as in the case of the Blackstone Acquisition and the Partnership Mergers).

Host believes that these benefits justify the REIT Conversion even if the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999 (in which event the effectiveness of Host's REIT election could be delayed until January 1, 2000).

Host explored the possibility of engaging in a business combination with a so-called "paired share" REIT, Santa Anita, in December 1996 and January 1997. Based upon an analysis of potential costs, the pricing of the transaction, the time requirement to complete such a transaction and the possible legislative risks associated with the "paired share" structure, Host decided not to pursue such a transaction. During the fourth quarter of 1997, Host began to explore internally the possibility of reorganizing as a REIT on a stand-alone basis. In April 1998, Host and the Operating Partnership entered into agreements with the Blackstone Entities for the Blackstone Acquisition and concurrently reached a decision and publicly announced that it would be advantageous if Host were to convert to a REIT. See "Restructuring Transactions--Background and Reasons for the Restructuring Transactions and the REIT Conversion."

The Host Board believes that Host's conversion to a REIT and the distribution of Crestline shares as part of the Initial E&P Distribution will benefit Host's stockholders by providing them with a tax advantaged REIT security that is expected to provide both the opportunity for regular cash dividends and capital appreciation as

Host REIT acquires additional properties, as well as a continuing interest in Crestline, Host's senior living company and the initial lessee of substantially all of Host REIT's Hotels, if a Host stockholder continues to hold the Crestline common stock. If Host REIT qualifies for taxation as a REIT, it generally will not be subject to federal corporate income taxes on that portion of its ordinary income or capital gain that is distributed to stockholders. As a REIT, Host REIT would be able to benefit from the tax advantages that apply to REITs, and stockholders will receive quarterly distributions that are at least sufficient to satisfy the annual distribution requirements applicable to REITs under the Code. The Host Board believes that this will highlight the value of Host REIT's hotel properties and permit stockholders to realize a regular cash return on that value. Upon completion of the REIT Conversion, Crestline is expected to own Host's 31 senior living communities, which will continue to be managed by Marriott International, and a 25% interest in the Swissotel management company expected to be acquired in the Blackstone Acquisition, and will lease substantially all of the Hotels owned by Host REIT and its affiliates. At that time, Crestline will operate independently of Host, will be publicly traded and separately listed on the NYSE, and will pursue its own growth opportunities.

Host's Board of Directors has received an opinion dated November , 1998 from BT Wolfensohn, which acted as financial advisor to Host in connection with the REIT Conversion, to the effect that the REIT Transactions (as defined in the opinion), taken together, are fair from a financial point of view to the holders of Host Common Stock. See "Opinion of Financial Advisor." The Host Board of Directors considered such opinion as well as other matters it deemed relevant in determining the advisability of the Agreement and the other transactions comprising the REIT Conversion.

Following the Restructuring Transactions and the other transactions comprising the REIT Conversion (including the Blackstone Acquisition), Host REIT and its subsidiaries are expected initially to own outright, or have controlling interests in, approximately 125 full-service Hotels operated primarily under the Marriott, Ritz-Carlton, Four Seasons, Swissotel and Hyatt brand names. Upon completion of the REIT Conversion, Host REIT will be the sole general partner of the Operating Partnership and is expected to own approximately 74% of the OP Units in the Operating Partnership, and the remaining OP Units, which are intended to be substantially equivalent on an economic basis to shares of Host REIT Common Stock, will be owned by the Limited Partners, unaffiliated partners of the Private Partnerships and the Blackstone Entities. Host REIT and its subsidiaries will own a number of OP Units equal to the number of shares of Host REIT Common Stock outstanding. Host REIT will be managed by its Board of Directors and will have no employees who are not also employees of the Operating Partnership.

THE BOARD OF DIRECTORS OF HOST BELIEVES THAT THE AGREEMENT, WHICH CONTEMPLATES THE RESTRUCTURING TRANSACTIONS AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION DESCRIBED IN THIS PROXY STATEMENT/PROSPECTUS, ARE ADVISABLE FOR HOST AND ITS STOCKHOLDERS AND RECOMMENDS THAT STOCKHOLDERS VOTE "FOR" APPROVAL OF THE AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY. In the event the Agreement is not approved by Host stockholders at the Special Meeting, Host will continue to operate as a Delaware corporation and the REIT Conversion (including the Initial E&P Distribution) will not be completed at this time. See "The Restructuring Transactions--Background and Reasons for the Restructuring Transactions and the REIT Conversion" and "--Comparison of Rights of Stockholders of Host and Host REIT."

The OP Contribution. Subject to approval of the Agreement at the Special Meeting, after the Special Meeting Host will contribute, as the OP Contribution, its wholly-owned full-service hotels, interests in certain of the Partnerships and the Private Partnerships (other than their interests in the general partners of the Partnerships, each of which is Host or a wholly-owned, direct or indirect subsidiary of Host, and certain other subsidiaries of Host who will remain in existence as subsidiaries of Host REIT and will receive OP Units in the Partnership Mergers) and certain other assets (excluding Host's senior living assets and the cash or other consideration to be distributed in connection with the Initial E&P Distribution and certain other de minimis assets that cannot be contributed to the Operating Partnership) to the Operating Partnership. As a preliminary step in the REIT Conversion, during 1998, various subsidiaries of Host also have contributed or will contribute the wholly-owned full-service hotels held by such entities, their interests in certain of the Partnerships and Private Partnerships and certain other assets to the Operating Partnership (including its subsidiaries). In exchange for these contributions,

18

the Operating Partnership (i) will issue to Host in the aggregate a number of OP Units equal to the number of outstanding shares of Host Common Stock at the completion of the REIT Conversion, reduced by the number of OP Units to be received by the general partners of the Partnerships and other subsidiaries of Host in the Partnership Mergers (together with preferred partnership interests in the Operating Partnership corresponding to any shares of Host preferred stock outstanding at the time of the REIT Conversion) and (ii) will assume all liabilities of Host, including past and future contingent liabilities, other than liabilities of Crestline. Following these contributions, the Operating Partnership and its subsidiaries will directly or indirectly own all of Host's and its subsidiaries' wholly-owned hotels, substantially all of Host's direct and indirect interests in both the Partnerships and the Private Partnerships and all of Host's other assets (excluding its senior living assets, the cash or other consideration to be distributed in connection with the Initial E&P Distribution, and certain other de minimis assets that cannot be contributed to the Operating Partnership). See "The Restructuring Transactions--The OP Contribution."

The above-described contributions, which include the OP Contribution, will enable Host REIT, following the Reincorporation by means of the Merger, to operate together with the Operating Partnership in an UPREIT structure, which is expected to improve Host REIT's ability to acquire additional hotels and other properties in the future on favorable terms.

Terms of the Merger. The Reincorporation will be effected through the merger of Host with and into Host REIT. As a result of the Merger, Host REIT will be the surviving entity, the separate existence of Host will terminate and each issued and outstanding share of Host Common Stock, together with the associated Host Right, will be converted into one share of Host REIT Common Stock, together with the associated Host REIT Right under the Host REIT Rights Agreement. In addition, in the Merger, outstanding options and certain other rights to acquire Host Common Stock will be converted into rights to acquire  $\hbox{Host REIT Common Stock, subject to adjustment for the Initial $\tt E\&P Distribution}$ and except as otherwise set forth in the Agreement, and any outstanding shares of preferred stock of Host would be converted into an equal number of shares of preferred stock of Host REIT having substantially the same rights and preferences. Upon consummation of the Merger, the holders of Host REIT Common Stock will become subject to the Ownership Limit under the Host REIT Charter. THE OWNERSHIP LIMIT WILL PROVIDE THAT NO PERSON OR PERSONS ACTING AS A GROUP MAY OWN, OR BE DEEMED TO OWN BY VIRTUE OF THE ATTRIBUTION PROVISIONS OF THE CODE, MORE THAN (I) 9.8% OF THE LESSER OF THE NUMBER OR VALUE OF SHARES OF HOST REIT COMMON STOCK OUTSTANDING OR (II) 9.8% OF THE LESSER OF THE NUMBER OR VALUE OF THE ISSUED AND OUTSTANDING SHARES OF ANY CLASS OR SERIES OF HOST REIT PREFERRED STOCK, SUBJECT TO WAIVER OR MODIFICATION BY HOST REIT IN CERTAIN LIMITED CIRCUMSTANCES AND SUBJECT TO (A) AN EXCEPTION FOR A HOLDER OF SHARES OF HOST REIT COMMON STOCK SOLELY BY REASON OF THE MERGER IN EXCESS OF THE OWNERSHIP LIMIT SO LONG AS SUCH HOLDER WOULD NOT OWN, DIRECTLY OR BY ATTRIBUTION UNDER THE CODE, MORE THAN 9.9% BY VALUE OF THE OUTSTANDING CAPITAL STOCK OF HOST REIT AS OF THE SPECIAL MERGER OWNERSHIP LIMIT EFFECTIVE TIME, AND (B) A LIMITATION ON THE APPLICATION OF THE "GROUP" LIMITATION (BUT NO OTHER ELEMENT OF THE OWNERSHIP LIMIT) TO ANY "GROUP" THAT OTHERWISE WOULD EXCEED THE OWNERSHIP LIMIT AT THE EFFECTIVE TIME OF THE MERGER SOLELY BY REASON OF ITS STATUS AS A "GROUP." THE OWNERSHIP LIMIT WILL BECOME APPLICABLE TO ALL HOST REIT STOCKHOLDERS AS OF THE EFFECTIVE TIME OF THE MERGER. ANY SHARES OF HOST REIT COMMON STOCK HELD AT THE EFFECTIVE TIME OF THE MERGER AND NOT SUBSEQUENTLY TRANSFERRED PRIOR TO THE SPECIAL MERGER OWNERSHIP LIMIT EFFECTIVE TIME, OR OTHERWISE HELD AT ANY TIME AFTER THE EFFECTIVE TIME OF THE MERGER IN VIOLATION OF THE OWNERSHIP LIMIT, WILL BE TRANSFERRED AUTOMATICALLY TO A TRUST FOR THE BENEFIT OF A DESIGNATED CHARITABLE BENEFICIARY, AND THE PERSON WHO ACQUIRED (OR HELD) SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL NOT BE ENTITLED TO ANY DISTRIBUTIONS THEREON OR TO VOTE SUCH EXCESS SHARES OF HOST REIT COMMON STOCK. THE HOLDER OF ANY SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL RECEIVE THE LESSER OF THE VALUE OF SUCH EXCESS SHARES AS OF THE EFFECTIVE TIME OF THE MERGER OR THE CASH PROCEEDS FROM THE SALE OF SUCH EXCESS SHARES OF HOST REIT COMMON STOCK BY THE TRUSTEE OF THE TRUST. AFTER THE EFFECTIVE TIME OF THE MERGER, ANY PERSON WHO ACQUIRES HOST REIT COMMON STOCK IN EXCESS OF THE OWNERSHIP LIMIT WILL NOT RECEIVE ANY PROCEEDS FROM THE SUBSEQUENT SALE THEREOF IN EXCESS OF THE LESSER OF THE PRICE PAID THEREFOR OR THE AMOUNT REALIZED FROM SUCH SALE. A TRANSFER OF HOST REIT COMMON STOCK TO

A PERSON WHO, AS A RESULT OF THE TRANSFER, VIOLATES THE OWNERSHIP LIMIT MAY BE VOID UNDER CERTAIN CIRCUMSTANCES, AND, IN ANY EVENT, WOULD DENY THE TRANSFEREE ANY OF THE ECONOMIC BENEFITS OF OWNING SHARES OF HOST REIT COMMON STOCK IN EXCESS OF THE OWNERSHIP LIMIT. TO AVOID THE ADVERSE EFFECTS OF THE OWNERSHIP LIMIT, ANY HOLDER OF HOST COMMON STOCK WHO WOULD OWN SHARES IN EXCESS OF THE OWNERSHIP LIMIT AT THE EFFECTIVE TIME OF THE MERGER SHOULD DISPOSE OF ANY SUCH EXCESS SHARES PRIOR THERETO. See "Description of Host REIT Capital Stock--Restrictions on Ownership and Transfer."

As a consequence of the Merger, among other things, the assets of Host will become assets of Host REIT without further action, and Host REIT will become liable for all the debts and obligations of Host. As soon as practicable following the Merger, Host REIT will complete any other transactions comprising the REIT Conversion that have not already been completed. See "The REIT Conversion."

Following the Merger, each person who was a director of Host will continue as a director of Host REIT, certain of the officers of Host will remain officers of Host REIT (others will have become officers of Crestline in connection the distribution of Crestline shares as part of the Initial E&P Distribution) and the rights of stockholders of Host REIT will be governed by the Host REIT Charter attached to the Agreement as Exhibit B and the Host REIT Bylaws attached to the Agreement as Exhibit C. See "The Restructuring Transactions--Comparison of Rights of Stockholders of Host and Host REIT."

The Agreement has been adopted by the Board of Directors of Host, subject to stockholder approval. The Merger will become effective at the time the Articles of Merger are accepted for filing by the State Department of Assessments and Taxation of Maryland (the "SDAT") in accordance with the MGCL and the time the Certificate of Merger is accepted for filing by the Secretary of State of Delaware in accordance with the DGCL, or later if so specified in such Articles and Certificate (the "Effective Time"). It is expected that the listing of Host REIT Common Stock on the NYSE will occur at or prior to the Effective Time and that the listing of Host Common Stock on the NYSE will be terminated at the Effective Time.

Host anticipates that the Merger will become effective as promptly as practicable following stockholder approval of the Agreement at the Special Meeting and the satisfaction or waiver of the other conditions to the Merger. The Agreement provides that the Merger may be abandoned by Host or Host REIT at any time prior to its effectiveness. However, Host has no current intention of abandoning or causing Host REIT to abandon the Merger subsequent to the Special Meeting if stockholder approval is obtained and the other conditions to the Merger are satisfied or waived. See "The Restructuring Transactions--Conditions to the Merger."

In connection with the REIT Conversion and the Initial E&P Distribution, Host expects to enter into the Employee Benefits and Other Employment Matters Allocation Agreement (the "1998 Employee Benefits Allocation Agreement") among Host, the Operating Partnership and Crestline. The 1998 Employee Benefits Allocation Agreement is expected to provide that the Operating Partnership will assume from Host, to the extent provided therein, the rights and obligations of Host under the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan (formerly called the Host Marriott Corporation 1993 Comprehensive Stock Inventive Plan), the Host Marriott Corporation Employee Stock Purchase Plan, the Host Marriott Corporation (HMC) Retirement and Savings Plan and the Host Marriott Corporation Executive Deferred Compensation Plan (the "Plans"). See "The Restructuring Transactions--Terms of the Merger," "Description of Host REIT Capital Stock" and "Management--1998 Employee Benefits Allocation Adreement."

Neither Host nor Host REIT is aware of any federal, state or local regulatory requirements that must be complied with or approvals that must be obtained prior to consummation of the Merger pursuant to the Agreement, other than compliance with applicable federal and state securities laws, the filing of the Articles of Merger as required under the MGCL and the filing of a Certificate of Merger as required under the DGCL and various state and local governmental authorizations. See "The Restructuring Transactions--Terms of the Merger."

Conditions to the Merger. Consummation of the Merger is subject to the satisfaction or waiver of various conditions, including satisfaction of the following conditions: (i) approval of the Agreement by the stockholders of Host; (ii) determination by the Host Board of Directors that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur, and that Host REIT can elect to be treated as a REIT for federal income tax purposes effective no later than the first full taxable year commencing after the REIT Conversion is completed (which might not be until the year commencing January 1, 2000 if the REIT Conversion is not completed prior to January 1, 1999); (iii) approval of the Host REIT Common Stock for listing on the NYSE; (iv) the receipt of all governmental and third-party consents to the Restructuring Transactions except for consents as would not reasonably be expected to have a material adverse effect on the business, financial condition or results of operations of Host REIT, the Operating Partnership and their subsidiaries taken as a whole; and (v) the United States Congress shall not have proposed or enacted adverse tax legislation. See "The Restructuring Transactions -- Conditions to the Merger.

In addition, Host will amend the Host Rights Agreement to provide that each Host Right issuable pursuant to the Host Rights Agreement will be converted into a Host REIT Right issuable under the Host REIT Rights Agreement. Host REIT intends to adopt the Host REIT Rights Agreement prior to the completion of the Merger and each share of Host REIT Common Stock issued in the Merger will have a Host REIT Right attached to it. See "Certain Provisions of Maryland Law and the Host REIT Charter and Bylaws--Stockholder Rights Plan."

Comparison of Rights of Stockholders of Host and Stockholders of Host REIT. The rights of stockholders of Host are currently governed by the DGCL, the Host Certificate, the Amended Bylaws of Host (the "Host Bylaws") and the Host Rights Agreement. If the Agreement is approved by Host's stockholders and the Merger is consummated, Host REIT will be the surviving entity in the Merger and the rights of the stockholders of Host REIT will be governed by the MGCL, the Host REIT Charter, the Host REIT Bylaws and the Host REIT Rights Agreement. Certain important differences exist between the rights of stockholders of Host and the rights of stockholders of Host REIT. See "The Restructuring Transactions--Comparison of Rights of Stockholders of Host and Host REIT." Copies of the Host REIT Charter and Host REIT Bylaws are attached to the Agreement (which is attached to this Proxy Statement/Prospectus) as Exhibits B and C, respectively.

Limitation of Liability and Indemnification of Directors and Officers. The Host REIT Charter and Host REIT Bylaws contain provisions limiting the liability of Host REIT's present and former directors and officers to the corporation and its stockholders and obligating Host REIT to indemnify present and former directors and officers all in accordance with Maryland law. See "The Restructuring Transactions--Limitation of Liability and Indemnification of Directors and Officers."

## THE REIT CONVERSION

The transactions summarized below, together with the Restructuring Transactions, constitute the transactions pursuant to which Host will restructure its business so that it will qualify as a REIT. If the required approvals for the various transactions are obtained and other conditions to the different steps in the REIT Conversion are satisfied or waived, these transactions are expected to occur at various times prior to the end of 1998 (or as soon thereafter as practicable). Although the Restructuring Transactions are expected to occur at the final stages of the REIT Conversion, certain of the REIT Conversion transactions (such as the Partnership Mergers, the Private Partnership Transactions and the Blackstone Acquisition) may occur following the Restructuring Transactions to the extent they are consummated.

Although a number of the transactions comprising the REIT Conversion are expected to be consummated immediately prior to, or in certain instances immediately following, the Merger, the Merger will not be consummated unless the conditions to the Merger have been satisfied or waived. In particular, Host's Board of Directors will have determined, among other things, that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur, and based on advice of counsel, that Host REIT can elect to be treated as a REIT for federal

income tax purposes effective no later than the first full taxable year commencing after the REIT Conversion is completed (which might not be until the year commencing January 1, 2000 if the REIT Conversion is not completed prior to January 1, 1999). Consistent with the foregoing, Host intends to pursue the transactions constituting the REIT Conversion at least through the date of the Special Meeting of Host stockholders. If the Agreement is approved by Host stockholders at the Special Meeting, Host intends to continue pursuing those transactions constituting the REIT Conversion which have not yet been completed, including the Blackstone Acquisition (which is not expected to be consummated any earlier than December 29, 1998). In addition, if the Board of Directors has determined that the conditions to the Merger have been or likely will be satisfied or waived (and, in particular, that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur), the Board intends to declare the Initial E&P Distribution and enter into the Leases with the Lessees (which will be indirect, wholly-owned subsidiaries of Crestline). Under such circumstances, however, the Merger and other transactions comprising the REIT Conversion still might be delayed or possibly might never be consummated. Assuming the Agreement is approved by Host stockholders at the Special Meeting and the Host Board of Directors makes the determination described above, it is currently contemplated that the Merger would be consummated on or about December 29, 1998, subject to satisfaction or waiver of the remaining conditions. If, however, the Agreement is not approved by the Host stockholders at the Special Meeting or the Host Board of Directors does not make the requisite determinations, Host will continue to operate as a Delaware corporation, the REIT Conversion will not be completed at this time and the Initial E&P Distribution will not be made.

If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the effectiveness of Host REIT's election could be delayed until January 1, 2000, which would result in Host REIT continuing to pay substantial corporate-level income taxes in 1999 (which would reduce Host REIT's estimated cash distributions per share of Host REIT Common Stock) and could cause the Blackstone Acquisition (which is conditioned, among other things, on consummation of the REIT Conversion by March 31, 1999 and Host REIT qualifying as a REIT for 1999) not to be consummated. In view of the complexity of the REIT Conversion, the number of transactions that must occur to complete the REIT Conversion and the time and expense involved, Host believes that it is advisable to complete the REIT Conversion, or as many of the transactions comprising the REIT Conversion as possible, as soon as practicable even if such completion is after January 1, 1999 (and, thus, a REIT election would not be effective until January 1, 2000). Moreover, completion of the REIT Conversion (or a significant number of the transactions comprising the REIT Conversion) could possibly facilitate another transaction (such as a merger into another entity) that would enable Host (or its successor) to elect to be treated as a REIT prior to January 1, 2000. If Host REIT's election to be taxed as a REIT is not effective on January 1, 1999, Host REIT intends to operate following the REIT Conversion in a manner that would permit it to qualify as a REIT at the earliest time practicable, and it might pursue a merger with another entity or other transaction that would permit it to commence a new taxable year and elect REIT status prior to January 1, 2000, although no assurance can be given that the Company will enter into or consummate such other transaction or otherwise qualify as a REIT prior to January 1, 2000. Host REIT in any event would elect to be treated as a REIT for federal income tax purposes no later than its taxable year commencing January

The Initial E&P Distribution. In order to qualify as a REIT for federal income tax purposes, among other things, Host REIT and/or Host, as its predecessor, must have distributed all of the accumulated E&P of Host to its stockholders in one or more taxable dividends prior to the end of the first full taxable year for which the REIT election of Host is effective, which currently is expected to be the taxable year commencing January 1, 1999 (but which might not be until the year beginning January 1, 2000).

In an effort to help accomplish the requisite distributions of the accumulated E&P of Host, Host or Host REIT will make the Initial E&P Distribution, consisting of one or more taxable distributions, to its stockholders in connection with the REIT Conversion. Such taxable distributions would consist of shares of common stock of

Crestline and cash or other consideration in an amount to be determined. Although there is no assurance as to what form such other consideration comprising the Initial E&P Distribution will take, it is currently contemplated that it will include the Special Dividend payable, at each stockholder's election, in cash or Host Common Stock (or Host REIT Common Stock if the Merger has occurred). If the Special Dividend is declared, stockholders of record on the Special Dividend record date are expected to be given approximately 20 days to decide whether to take cash, Host Common Stock (or Host REIT Common Stock if the Merger has occurred), or a combination of cash and stock in payment of the Special Dividend. It is anticipated that the cash/stock election will be available on a per-share basis and that, once made, stockholders' elections will be irrevocable. Stockholders entitled to the Special Dividend who fail to make a timely election will receive shares of Host Common Stock (or Host REIT Common Stock if the Merger has occurred), subject to the Ownership Limit, in payment of the Special Dividend. In any event, cash would be paid in lieu of fractional shares, and the Special Dividend payment would be made promptly following expiration of the election period.

The aggregate value of the Crestline common stock and the cash or other consideration to be distributed to Host stockholders (and the Blackstone Entities) in connection with the Initial E&P Distribution is currently estimated to be approximately \$525 million to \$625 million (approximately \$2.10 to \$2.50 per share to the Host stockholders), of which approximately \$200 to \$300 million (or approximately \$.80 to \$1.20 per share) is expected to be represented by the Special Dividend. The actual amount of the Initial E&P Distribution will be based in part upon the estimated amount of accumulated E&P of Host as of the last day of its taxable year ending on or immediately following completion of the REIT Conversion. To the extent that the Initial E&P Distribution is not sufficient to eliminate Host's accumulated E&P, Host REIT will make one or more additional taxable distributions to its stockholders (in the form of cash or securities) prior to the last day of Host REIT's first full taxable year as a REIT (currently expected to be December 31, 1999 but which instead might be December 31, 2000) in an amount intended to be sufficient to eliminate such E&P, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

Limited Partners who elect to receive Host REIT Common Stock in exchange for OP Units in connection with the Partnership Mergers will not receive the Crestline common stock or any other portion of the Initial E&P Distribution, which will have been distributed before they become stockholders of Host REIT (approximately 25 trading days after the effective date of the Partnership Mergers). However, following receipt of such shares of Host REIT Common Stock, such Limited Partners would participate as stockholders of Host REIT in other dividends or distributions by Host REIT to holders of Host REIT Common Stock, including any additional taxable distributions necessary to eliminate Host's accumulated E&P to the extent the Initial E&P Distribution is not sufficient to eliminate such accumulated E&P. The Limited Partners who elect to retain OP Units also will participate in such additional distributions because distributions by the Operating Partnership to its partners (including Host REIT) would be the source of such additional distributions.

No holder of Host Common Stock will be required to pay any cash or other consideration to Host or Host REIT for shares of Crestline common stock received in the Initial E&P Distribution or to surrender or exchange their Host Common Stock or Host REIT Common Stock in order to receive shares of Crestline common stock or other cash or securities as part of the Initial E&P Distribution. See "The REIT Conversion--The Initial E&P Distribution."

In addition, following the Restructuring Transactions, the Blackstone Entities are entitled to receive a pro rata portion of the same consideration received by Host REIT's stockholders in connection with the Initial E&P Distribution, except to the extent the Blackstone Entities elected to receive additional OP Units in lieu thereof pursuant to the terms of the Blackstone Acquisition. The payment to the Blackstone Entities of Crestline common stock and other consideration is expected to be approximately \$90 to \$110 million of the aggregate Initial E&P Distribution of approximately \$525 to \$625 million if the REIT Conversion and the Blackstone Acquisition are consummated. See "Business and Properties--Blackstone Acquisition." Following the Initial E&P Distribution, Crestline's principal assets will include the senior living assets of Host, which are expected to consist of 31 senior living communities, a 25% interest in the Swissotel management company acquired from the Blackstone Entities

and the Lessees. The shares of Crestline common stock distributed to Host stockholders as part of the Initial E&P Distribution and the Blackstone Entities will become a separately traded security from the Host REIT Common Stock, and Crestline will operate independently from Host REIT. There will be substantial overlap of stockholders of Crestline and Host REIT initially, but this overlap likely will diverge over time.

The Host Board believes that the distribution of Crestline common stock to Host stockholders as part of the Initial E&P Distribution will provide those Host stockholders who continue to hold Crestline common stock with a separate identifiable interest in a diversified company that generates revenue from both its senior living business and its leasing business. Even under circumstances where the Crestline common stock is distributed to Host stockholders as part of the Initial E&P Distribution but the Merger or other transactions comprising the REIT Conversion are delayed or possibly never consummated, the Host Board believes that having existing leasing arrangements in place with Crestline could facilitate any subsequent efforts by Host to qualify as a REIT for federal income tax purposes (including efforts to pursue a merger with another entity or another transaction that would permit it to commence a new taxable year and elect REIT status prior to January 1, 2000).

Other Transactions Comprising the REIT Conversion. In addition to the Restructuring Transactions and the Initial E&P Distribution, the REIT Conversion includes the following additional transactions:

- . Debt Refinancing. In August 1998, Host refinanced \$1.55 billion of outstanding senior notes (the "Senior Note Refinancing") through offers to purchase such debt securities for cash and a concurrent solicitation of consents to amend the terms of the debt securities to facilitate the transactions constituting the REIT Conversion. Host obtained the funds for the Senior Note Refinancing primarily from the issuance of new debt securities and a new \$1.25 billion credit facility (the "New Credit Facility"). See "Business and Properties--Indebtedness."
- . Treatment of Convertible Preferred Securities. In the REIT Conversion, the Operating Partnership will assume primary liability for repayment of the \$567 million of convertible subordinated debentures of Host underlying the \$550 million of outstanding Convertible Quarterly Income Preferred Securities of Host (the "Convertible Preferred Securities"). As the successor to Host, Host REIT also will be liable on the debentures and the debentures will become convertible into Host REIT Common Stock, but the Operating Partnership will have primary responsibility for payment of the debentures, including all costs of conversion. Upon conversion by a Convertible Preferred Securities holder, Host REIT will issue shares of Host REIT Common Stock, which will be delivered to such holder. Upon the issuance of such shares by Host REIT, the Operating Partnership will issue to Host REIT a number of OP Units equal to the number of shares of Host REIT Common Stock issued in exchange for the debentures. As a result of the distribution of Crestline common stock and any cash and other consideration to Host or Host REIT stockholders in connection with the Initial E&P Distribution, the conversion ratio of the Convertible Preferred Securities will be adjusted to take into account certain effects of the REIT Conversion. See "Management's Discussion and Analysis of Financial Condition and Results of Operations--Liquidity and Capital Resources."
- . The Partnership Mergers. Immediately following the Effective Date, each Partnership that participates in the Partnership Mergers will merge with a subsidiary of the Operating Partnership. Such participating Partnerships will be the surviving entities of the Partnership Mergers and will continue in existence as indirect subsidiaries of the Operating Partnership. In the Partnership Mergers, each Limited Partner will receive a number of OP Units with a deemed value equal to the stated exchange value of his respective partnership interest. If a Limited Partner elects to receive Host REIT Common Stock or a Note in exchange for OP Units in connection with the Partnership Mergers, such Limited Partner will, upon receipt of his OP Units, tender (or be deemed to tender) all of such OP Units to Host REIT in exchange for an equal number of shares of Host REIT Common Stock or to the Operating Partnership in exchange for a Note with a specified principal amount. The general partners of the Partnerships and other subsidiaries of Host also will receive OP Units in exchange for their interests in the Partnerships, and

such general partners will continue as wholly-owned, direct or indirect subsidiaries of Host REIT. Partnerships that do not participate in a Partnership Merger will continue as separate partnerships, but the Operating Partnership would contribute some or all of the interests in certain of these Partnerships that it receives from Host and its subsidiaries to a Non-Controlled Subsidiary.

- Restructuring of the Private Partnerships. The Operating Partnership will acquire in the Private Partnership Transactions the partnership interests from unaffiliated partners of four Private Partnerships in exchange for OP Units and, accordingly, will own all of the interests in those Private Partnerships. For the remaining Private Partnerships, (i) the Operating Partnership will be a partner in the partnership if the unaffiliated partners consent to a lease of the partnership's Hotel(s) to a Lessee or (ii) if the requisite consents to enter into a lease are not obtained, the Operating Partnership may transfer its interest in such partnership to a Non-Controlled Subsidiary.
- .The Blackstone Acquisition. Subject to various terms and conditions, the Operating Partnership expects to acquire from the Blackstone Entities ownership of, or controlling interests in, the Blackstone Hotels. In addition, Host REIT will acquire a 25% interest in the Swissotel management company from the Blackstone Entities, which Host REIT will transfer to Crestline. If the Blackstone Acquisition is consummated, the Operating Partnership expects to issue approximately 43.7 million OP Units (based upon a negotiated value of \$20.00 per OP Unit), assume debt and make cash payments totaling approximately \$862 million and distribute up to 18% of the shares of Crestline common stock and other consideration to the Blackstone Entities. See "The REIT Conversion--Other Transactions Comprising the REIT Conversion--The Blackstone Acquisition."
- . Contribution of Assets to Non-Controlled Subsidiaries. The Operating Partnership will organize the Non-Controlled Subsidiaries to hold various assets (not exceeding, in the aggregate, 20% by value of the assets of the Operating Partnership) contributed by Host and its subsidiaries to the Operating Partnership. The direct ownership of most of these assets by the Operating Partnership could jeopardize Host REIT's status as a REIT. These assets primarily will consist of partnership or other interests in hotels which are not leased, certain furniture, fixtures and equipment ("FF&E") used in the Hotels and certain international hotels in which Host owns interests. In exchange for the contribution of these assets to the Non-Controlled Subsidiaries, the Operating Partnership will receive nonvoting common stock representing 95% of the total economic interests of the Non-Controlled Subsidiaries. In addition, the Operating Partnership and, prior to the Partnership Mergers, certain of the Partnerships (assuming they participate in the Partnership Mergers) will sell to a Non-Controlled Subsidiary an estimated \$180 million in value of personal property associated with certain Hotels for notes or cash that has been contributed or loaned to the Non-Controlled Subsidiary by the Operating Partnership, or a combination thereof. The Operating Partnership could not lease this personal property to the Lessees without potentially jeopardizing Host REIT's qualification as a REIT. The Non-Controlled Subsidiary will lease such personal property to the applicable Lessees. The Host Marriott Statutory Employee/Charitable Trust, the beneficiaries of which will be certain employees of Host REIT and a designated charity (the "Host Employee/Charitable Trust"), and possibly certain other investors will acquire all of the voting common stock representing the remaining 5% of the total economic interests, and 100% of the control, of each Non-Controlled Subsidiary. See "The REIT Conversion--Other Transactions Comprising the REIT Conversion.
- Leases of Hotels. The Operating Partnership, its subsidiaries and its controlled partnerships, including the Partnerships participating in the Partnership Mergers, will lease virtually all of their Hotels to the Lessees pursuant to the Leases. See "Business and Properties--The Leases." The leased Hotels will be operated by the Lessees under their existing brand names pursuant to their existing long-term Management Agreements, which will be assigned to the Lessees for the terms of the applicable Leases but under which the Operating Partnership will remain obligated. See "Business and Properties--The Management Agreements."

Following the REIT Conversion, assuming the Full Participation Scenario, the organizational structure of Host REIT is expected to be as follows:

## [FLOW CHART APPEARS HERE]

- (1) Represents Limited Partners and others who retain OP Units and do not elect to receive shares of Host REIT Common Stock or Notes; excludes Host and its subsidiaries.
- (2) Also will include Limited Partners in the Partnership Mergers who elect to receive shares of Host REIT Common Stock in exchange for the OP Units received in the Partnership Mergers. Immediately following the Restructuring Transactions and the distribution by Host of Crestline common stock to its stockholders and receipt of Crestline common stock by the Blackstone Entities, the stockholders of Crestline will consist of the stockholders of Host REIT (other than Limited Partners who elect to receive Host REIT Common Stock in connection with the Partnership Mergers) and the Blackstone Entities. The common ownership of the two public companies, however, will diverge over time.
- (3) Percentage ownership in the Operating Partnership assumes no Limited Partners elect to receive either Host REIT Common Stock or Notes in connection with the Partnership Mergers and that the price per share of Host REIT Common Stock is \$ 12.50 for purposes of the Partnership Mergers.
  (4) The Operating Partnership will own all or substantially all of the equity
- (4) The Operating Partnership will own all or substantially all of the equity interests in the Partnerships participating in the Partnership Mergers, certain Private Partnerships and other Host subsidiaries that own Hotels, both directly and through other direct or indirect, wholly-owned subsidiaries of the Operating Partnership or Host REIT. Host will contribute its partial equity interests in the Partnerships that do not participate in the Partnership Mergers and those Private Partnerships whose partners have not elected to exchange their interests for OP Units to the Operating Partnership, and the Operating Partnership will either hold such partial interests or contribute them to the Non-Controlled Subsidiaries.

Ownership Interests in the Operating Partnership Following the Restructuring Transactions and the Other Transactions Comprising the REIT Conversion. Following the Restructuring Transactions and the other transactions comprising the REIT Conversion, the Operating Partnership is expected to be owned as set forth below:

## OWNERSHIP OF THE OPERATING PARTNERSHIP

ENTITY	PERCENTAGE INTEREST(1)
Host REITLimited Partners of the PartnershipsPrivate PartnershipsBlackstone Entities	8.4 1.2
TOTAL	100.0% =====

(1) Assumes that all Partnerships participate in the Partnership Mergers, that the Blackstone Acquisition is consummated, that all Limited Partners in the Partnership Mergers elect to retain OP Units and that the price of an OP Unit is \$12.50 per share for purposes of the Partnership Mergers. The percentage interest of Host REIT will increase, and the percentage interest of the Limited Partners will decrease, if Limited Partners elect to receive Host REIT Common Stock or Notes in exchange for their OP Units in connection with the Partnership Mergers or if the price per OP Unit in the Partnership Mergers is greater than \$12.50. The percentage interest of Host REIT will decrease, and the percentage interest of the Limited Partners will increase, if the price per OP Unit in the Partnership Mergers is less than \$12.50.

## FEDERAL INCOME TAX CONSEQUENCES

Federal Income Tax Consequences of the OP Contribution. Section 721 of the Code provides that no gain or loss is recognized in the case of a contribution of property to a partnership in exchange for an interest in the partnership. However, there are a number of potential exceptions to the availability of such treatment. The application of these exceptions is highly complex and depends upon a number of factual determinations and other outside events which may or may not occur, but Host believes that it will not recognize a material amount of gain in connection with the OP Contribution. See "Federal Income Tax Consequences--Federal Income Tax Consequences of the OP Contribution." Host is not obtaining an opinion of outside counsel as to the tax consequences to it of the OP Contribution.

Federal Income Tax Consequences of the Merger. Host has received an opinion of counsel to the effect that the Merger will qualify as a tax-free reorganization within the meaning of Section 368(a) of the Code, and accordingly, (i) no gain or loss will be recognized by Host or Host REIT as a result of the Merger; (ii) no gain or loss will be recognized by holders of shares of Host Common Stock upon the conversion of such shares into Host REIT Common Stock (except for certain stockholders who are not considered "U.S. persons" for purposes of the Code and who own (or have owned) in excess of 5% of Host); (iii) the tax basis of shares of Host REIT Common Stock received by a former holder of Host Common Stock pursuant to the Merger in the aggregate will be the same as the holder's adjusted tax basis in the shares of Host Common Stock being converted in the Merger (subject to any adjustment resulting from the Initial E&P Distribution); and (iv) the holding period of shares of Host REIT Common Stock received by a former holder of Host Common Stock pursuant to the Merger will include the holder's holding period with respect to the shares of Host Common Stock being converted in the Merger, assuming that the Host Common Stock was held as a capital asset at the Effective Time. See "Federal Income Tax Consequences--Federal Income Tax Consequences of the Merger.

Federal Income Tax Consequences of the Initial E&P Distribution. Generally, the Initial E&P Distribution will be a taxable dividend to a Host stockholder to the extent that the Initial E&P Distribution is made out of the Host stockholder's share of the portion of the current and accumulated E&P of Host and Host REIT allocable to

the Initial E&P Distribution. Host and Host REIT currently believe that the substantial majority, if not all, of the Initial E&P Distribution (the fair market value of which Host currently estimates will range from approximately \$2.10 to \$2.50 per share of Host Common Stock) will be considered made out of such E&P and, therefore, will be taxable as a dividend. See "Federal Income Tax Consequences --Federal Income Tax Consequences of the Initial E&P Distribution."

Qualification of Host REIT as a REIT. Host REIT expects to qualify as a REIT for federal income tax purposes effective for its first full taxable year commencing after the REIT Conversion is completed, which Host currently expects to be the year beginning January 1, 1999 (but which might not be until the year beginning January 1, 2000). If it so qualifies, Host REIT will be permitted to (i) deduct dividends paid to its stockholders, allowing the income represented by such dividends to avoid taxation at the entity level and to be taxed only at the stockholder level and (ii) treat retained net capital gains in a manner so that such gains are taxed at the Host REIT level but effectively avoid taxation at the stockholder level. Host REIT, however, will be subject to a separate corporate income tax on any gains recognized during the ten years following the REIT Conversion that are attributable to "built-in" gain with respect to the assets that Host owned at the time of the REIT Conversion (which tax would be paid by the Operating Partnership). Host REIT has substantial deferred tax liabilities that are likely to be recognized as "built-in" gain (or by a Non-Controlled Subsidiary) during such period without any corresponding receipt of cash, and the Operating Partnership will be responsible for paying such taxes. Host REIT's ability to qualify as a REIT will depend upon its continuing satisfaction following the REIT Conversion of various requirements related to the nature of its assets, the sources of its income and the distributions to its stockholders, including a requirement that Host REIT distribute to its stockholders at least 95% of its taxable income each year.

## RECOMMENDATION OF THE BOARD OF DIRECTORS

FOR THE REASONS STATED HEREIN, THE BOARD OF DIRECTORS OF HOST RECOMMENDS THAT HOST'S STOCKHOLDERS VOTE "FOR" APPROVAL OF THE AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY.

## DISTRIBUTIONS AND MARKET PRICES OF HOST COMMON STOCK

Host Common Stock has been listed on the NYSE under the symbol "HMT" since October 8, 1993. Prior thereto, Host and Marriott International were operated as a single consolidated company. See "Certain Relationships and Related Transactions--Relationship Between Host and Marriott International."

The table below sets forth, for the periods indicated, the reported high and low sale prices of Host Common Stock on the NYSE. On April 15, 1998, the last full trading day prior to the public announcement of the proposed REIT Conversion, the closing sale price of Host Common Stock on the NYSE was \$20.00 per share. On November 19, 1998, the latest practicable date before the printing of this Proxy Statement/Prospectus, the closing sale price of Host Common Stock on the NYSE was \$14 1/4 per share. Host has not declared any cash dividends on the Host Common Stock during the two fiscal years ended January 2, 1998 and through the date hereof.

	HIGH	-	LOW	
1997				
First Quarter	\$18	3/4	\$15	3/4
Second Quarter	18	1/8	15	1/4
Third Quarter	20	13/16	17	1/2
Fourth Quarter	23	3/4	18	1/16
1998				
First Quarter	\$20	7/16	\$17	1/2
Second Quarter	21	3/8	17	1/8
Third Quarter	18	15/16	12	13/16
Fourth Quarter (through November 19, 1998)	15	5/16	9	7/8

It is expected that, upon consummation of the Merger, the Host REIT Common Stock will be listed and trade on the NYSE in the same manner as shares of Host Common Stock currently trade on the NYSE. The historical trading prices of Host's Common Stock are not necessarily indicative of the future trading prices of Host REIT Common Stock because, among other things, the current stock price of Host reflects the current market valuation of Host's current business and assets (including the Crestline common stock and the cash or other consideration to be distributed in connection with the Initial E&P Distribution) and does not necessarily take into account the changes in Host's business and operations (including the leasing of its Hotels to Crestline) that will occur in connection with the REIT Conversion. See "Risk Factors--Current Host Common Stock Price Is Not Necessarily Indicative of the Price of Host REIT Common Stock Following the REIT Conversion."

## DISTRIBUTION AND DIVIDEND POLICY AFTER THE MERGER

The Operating Partnership and Host REIT intend to pay regular quarterly distributions to holders of OP Units and Host REIT Common Stock, respectively. Host REIT and the Operating Partnership anticipate that distributions will be paid during January, April, July and October of each year, except that the first distribution in 1999 is expected to be paid at the end of February if the REIT Conversion is completed in 1998.

The Operating Partnership intends to distribute an amount that will enable Host REIT to distribute to its stockholders an amount equal to 100% of Host REIT's taxable income (other than capital gains, which will be addressed on a case-by-case basis) for each year no later than the end of January of the following year. Host REIT anticipates that distributions generally will be paid from cash available for distribution (generally equal to cash from operations less capital expenditures and principal amortization on indebtedness); however, to the extent that cash available for distribution is insufficient to make such distributions, the Operating Partnership intends to borrow funds in order to make distributions consistent with this policy.

Based upon Host's preliminary estimates of Host REIT's taxable income for the twelve months ending December 31, 1999, Host and the Operating Partnership currently estimate that this policy will result in an initial annual distribution by the Operating Partnership of approximately \$0.84 per OP Unit (\$0.21 per quarter) during the twelve months ending December 31, 1999. If Host's preliminary estimate of \$231 million of cash distributions by the Operating Partnership during the twelve months ending December 31, 1999 proves accurate but the Operating Partnership's cash available for distribution were only equal to its estimated cash available for distribution (\$144 million) and cash from contingent rents (\$64 million) during 1999, then the Operating Partnership would be required to borrow approximately \$23 million (or \$0.08 per OP Unit) to make such distributions. While Host does not believe this will be necessary, it believes that the Operating Partnership would be able to borrow the necessary amounts under the New Credit Facility or from other sources and that any such borrowing would not have a material adverse effect on its financial condition or results of operations. The distributions to stockholders per share of Host REIT Common Stock are expected to be in an amount equal to the amount distributed by the Operating Partnership per OP Unit. However, if the REIT Conversion is not completed prior to January 1, 1999, then Host REIT's distributions to stockholders in 1999 would be lower than the Operating Partnership's distributions per OP Unit (by the amount of Host REIT's 1999 corporate income tax payments) until its REIT election becomes effective, which would be no later than January 1, 2000. The Operating Partnership intends to make distributions during 1999 at the estimated level of \$0.84 per OP Unit even if the REIT election of Host REIT were not effective until January 1, 2000, which would result in estimated distributions by Host REIT (after estimated federal and state income tax payments) of \$0.53 per share of Host REIT Common Stock for the full year 1999.

Distributions will be made in the discretion of the Board of Directors of Host REIT and will be affected by a number of factors, many of which are beyond the control of Host REIT and the Operating Partnership. In order to maintain its qualification as a REIT under the Code, Host REIT is required to distribute (within a certain period after the end of each year) at least 95% of its REIT taxable income for such year. See "Distribution and Other Policies--Distribution Policy." Host REIT intends to establish a dividend reinvestment plan.

## UNAUDITED PER SHARE DATA

The following table sets forth selected historical per share data for Host. The per share data presented below are based on and derived from, and should be read in conjunction with the historical consolidated statements and the related notes thereto of Host, which are included in this Proxy Statement/Prospectus. Interim data for the thirty-six weeks ended September 11, 1998 ("First Three Quarters 1998") and September 12, 1997 ("First Three Quarters 1997") and as of September 11, 1998 and September 12, 1997 are unaudited, but include, in the opinion of management of Host, all adjustments (which are normal and recurring) necessary for a fair presentation of such data. Results for the First Three Quarters 1998 are not necessarily indicative of the results that may be expected for any other interim periods or for the year as a whole.

AS OF AND FOR THE FISCAL YEAR(1)			AS OF AND FOR THE FIRST THREE QUARTERS			
1997	1996	1995	199	8 	199	7
\$0.23	\$(0.07)	\$(0.39)	\$	0.49	\$	0.19
0	` o´	` o´		0		0.19 0 5.89
	\$0.23 \$0.23	FISCAL YEAR	FISCAL YEAR(1)  1997 1996 1995  \$0.23 \$(0.07) \$(0.39)  \$0.23 \$(0.07) \$(0.39)  0 0 0	FISCAL YEAR(1) FIRST  1997 1996 1995 199  \$0.23 \$(0.07) \$(0.39) \$  \$0.23 \$(0.07) \$(0.39) \$  0 0 0	FISCAL YEAR(1) FIRST THREE  1997 1996 1995 1998  \$0.23 \$(0.07) \$(0.39) \$ 0.49  \$0.23 \$(0.07) \$(0.39) \$ 0.48 0 0 0 0	FISCAL YEAR(1) FIRST THREE QUART  1997 1996 1995 1998 199  \$0.23 \$(0.07) \$(0.39) \$ 0.49 \$  \$0.23 \$(0.07) \$(0.39) \$ 0.48 \$ 0 0 0 0 0

- (1) Host's fiscal year ends on the Friday nearest to December 31. Fiscal years 1997 and 1995 included 52 weeks compared to 53 weeks for fiscal year 1996.
- (2) See "--Distributions and Market Prices of Host Common Stock."

#### SUMMARY FINANCIAL INFORMATION

The following table sets forth unaudited pro forma financial and other information for the Operating Partnership and Host REIT and combined consolidated historical financial information for Host. The following summary financial information should be read in conjunction with the financial statements and notes thereto and Management's Discussion and Analysis of Results of Operations and Financial Condition included elsewhere in this Consent Solicitation.

The unaudited pro forma financial statements as of September 11, 1998 and for the fiscal year ended January 2, 1998 and the First Three Quarters 1998 for the 100% Participation Presentation are presented as if the REIT Conversion occurred as of September 11, 1998 for the pro forma balance sheets and at the beginning of the fiscal year for the pro forma statements of operations. The unaudited REIT 2000 pro forma information assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur and Host does not become a REIT until January 1, 2000. The pro forma information incorporates certain assumptions that are described in the Notes to the Unaudited Pro Forma Financial Statements included elsewhere in this Proxy Statement/Prospectus.

The pro forma information does not purport to represent what the Operating Partnership's or Host REIT's financial position or results of operations would actually have been if these transactions had, in fact, occurred on such date or at the beginning of the period indicated, or to project the Operating Partnership's or Host REIT's financial position or results of operations at any future date or for any future period.

In addition, the historical information contained in the following table is not comparable to the operations of the Operating Partnership or Host REIT on a going-forward basis because the historical information relates to an operating entity which owns and operates hotels and senior living communities, while the Operating Partnership will own the Hotels but will lease them to the Lessees and receive rental payments in connection therewith.

# SUMMARY FINANCIAL INFORMATION (IN MILLIONS)

	OPERATING PARTNERSHIP PRO FORMA FISCAL YEAR 1997		HOST REIT PRO FORMA FISCAL YEAR 1997 HIS			OPERATING PARTNERSHIP PRO FORMA FIRST THREE QUARTERS 1998		HOST REIT PRO FORMA FIRST THREE QUARTERS 1998	
					HOST HISTORICAL				
	100% PARTICIPATION WITH NO NOTES ISSUED	NO NOTES	100% PARTICIPATION WITH NO NOTES ISSUED	NO NOTES	FISCAL	100% PARTICIPATION WITH NO NOTES ISSUED	NO NOTES	100% PARTICIPATION WITH NO NOTES ISSUED	REIT 2000 WITH NO NOTES ISSUED(1)
REVENUES: Hotel revenues Rental revenues Other revenues	\$ 1,135 (4)	\$ 998 (4)	\$ 1,135 (4)	\$ 998 (4)	\$1,093  54	\$ 540 2	\$ 468 2	\$ 540 2	\$ 468 2
Total revenues	1,131	994	1,131	994	1,147	542	470	542	470
OPERATING COSTS AND EXPENSES:									
Hotel	600	513	600	513	649	394	336	394	336
Other	11 	11 	11 	11 	49	15 	15 	15 	15 
Total operating costs and									
expenses	611	524 	611	524 	698	409	351	409 	351 
Operating profit Minority interest Corporate	520 (10)	470 (10)	520 (17)	470 (12)	449 (32)	133 (14)	119 (14)	133 38	119 (1)
expenses REIT Conversion	(44)	(44)	(44)	(44)	(47)	(30)	(30)	(30)	(30)
expenses Interest expense Dividends on Convertible Preferred	(468)	(420)	(430)	(382)	(302)	(328)	(292)	(302)	(266)
Securities Interest income	 29	 36	(37) 29	(37) 36	(37) 52	 24	 30	(26) 24	(26) 30
Income (loss) before income									
taxes	27	32	21	31	83	(215)	(187)	(163)	(174)
Benefit (provision) for income taxes	(1)	(2)	(1)	(13)	(36)	11	9	8	71
Income (loss) before extraordinary items									
	\$ 26 =====	\$ 30 =====	\$ 20 =====	\$ 18 =====	\$ 47 =====	\$(204) =====	\$(178) =====	\$(155) =====	\$(103) =====
	HOST								

	HOST HISTORICAL
	FIRST THREE QUARTERS 1998
REVENUES:	
Hotel revenues	\$ 922
Rental revenues	
Other revenues	118
Total revenues	1,040
OPERATING COSTS AND EXPENSES:	
Hotel	502
Other	45
Total operating costs and	
expenses	547
Operating profit	493
Minority interest	(36)
Corporate	(30)
expenses	(33)
REIT Conversion	
expenses	
Interest expense	(245)
Dividends on	
Convertible	

Preferred	
Securities	(26)
Interest income	36
Income (loss)	
before income	
taxes	175
Benefit (provision)	
for income taxes	(75)
Income (loss)	
before	
extraordinary items	
	\$ 100
	========

AS OF SEPTEMBER 11, 1998

# OPERATING PARTNERSHIP

	PRO FORMA		HOST REIT PRO F			
		REIT 2000 WITH NO NOTES ISSUED(1)	100% PARTICIPATION WITH NO NOTES ISSUED	` ,		
BALANCE SHEET DATA: Property and equipment,						
net	\$7,069	\$5,619	\$7,069	\$5,619	\$5,937	
Total assets Debt, excluding	7,952	6,696	7,952	6,696	6,969	
convertible debt	4,970	4,370	4,970	4,370	4,224	
Convertible debt	567	567				
Total liabilities Convertible Preferred	6,560	6,112	6,000	5,546	5,267	
Securities Limited Partner interests of third parties at redemption			550	550	550	
value	989	333				
Equity	403	251	1,048	467	1,152	

<sup>(1)</sup> Assumes all Partnerships participate in the Partnership Mergers, the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur and Host does not become a REIT until January 1, 2000.

## RISK FACTORS

In considering whether to approve the Agreement and the transactions contemplated thereby, stockholders of Host should consider carefully, among other factors, the material risks described below.

## RISKS AND EFFECTS OF THE MERGER AND THE REIT CONVERSION

LACK OF CONTROL OVER HOTEL OPERATIONS. Due to current federal income tax law restrictions on a REIT's ability to derive revenues directly from the operation of a hotel, Host REIT will lease virtually all of its consolidated Hotels to the Lessees, which will operate the Hotels by continuing to retain the Managers pursuant to the Management Agreements. Host REIT will not operate the Hotels or participate in the decisions affecting the daily operations of the Hotels. Host REIT will have only limited ability to require the Lessees or the Managers to operate or manage the Hotels in any particular manner and no ability to govern any particular aspect of their day-to-day operation or management. Even if Host REIT's management believes the Lessees or the Managers are operating or managing the Hotels inefficiently or in a manner that does not result in the maximization of rental payments to Host REIT under the Leases, Host REIT has only a limited ability to require the Lessees or the Managers to change their method of operation or management. Therefore, Host REIT will be dependent for its revenue upon the ability of the Lessees and the Managers to operate and manage the Hotels. Host REIT is limited to seeking redress only if the Lessees violate the terms of the Leases and then only to the extent of the remedies set forth therein. Remedies under the Leases include Host REIT's ability to terminate a Lease upon certain events of default, such as the Lessee's failure to pay rent or failure to maintain certain net worth requirements and breaches of other specified obligations under the Leases. See "Business and Properties--The Leases." Termination of a Lease, however, could impair Host REIT's ability to qualify as a REIT for federal income tax purposes unless another suitable lessee could be found.

LACK OF CONTROL OVER NON-CONTROLLED SUBSIDIARIES. The Non-Controlled Subsidiaries will hold various assets (not exceeding in the aggregate 20% by value of the assets of Host REIT), consisting primarily of interests in hotels which are not leased, certain FF&E used in the Hotels and certain international hotels. The direct ownership or control of most of these assets by Host REIT could jeopardize Host REIT's status as a REIT. Although Host REIT will own 95% of the total economic interests of the Non-Controlled Subsidiaries, the Host Employee/Charitable Trust and possibly certain other investors will own all of the voting common stock of the Non-Controlled Subsidiaries (which will represent the remaining 5% of the total economic interest thereof). As the owner of the voting stock of the Non-Controlled Subsidiaries, the Host Employee/Charitable Trust and possibly certain other investors will select the directors of the Non-Controlled Subsidiaries, who will be responsible for overseeing the operations of those entities. As a result, Host REIT will have no control over the operation or management of the hotels or other assets owned by the Non-Controlled Subsidiaries even though it will depend upon the Non-Controlled Subsidiaries for a significant portion of its revenues (and the activities of the Non-Controlled Subsidiaries could cause Host REIT to be in default under its principal debt facilities).

DEPENDENCE UPON CRESTLINE. Subsidiaries of Crestline will be the Lessees of virtually all of the Hotels and their rent payments will be the primary source of Host REIT's revenues. Upon the REIT Conversion, all fees payable under the Management Agreements for subsequent periods will become the primary obligations of the Lessees, to be paid by the Lessees for so long as the Leases remain in effect. The obligations of the Lessees will be guaranteed by Crestline, subject to specified liability limitations. Crestline's and each Lessee's financial condition and ability to meet its obligations under the Leases and the Management Agreements will determine the Operating Partnership's ability to make distributions to holders of OP Units, including Host REIT, and Host REIT's ability, in turn, to make distributions to its stockholders. As of September 11, 1998, on a pro forma basis, after giving effect to the REIT Conversion, Crestline would have had approximately \$300 million of indebtedness (including approximately \$85 million due to Host REIT to pay for hotel working capital purchased from Host REIT but not including guarantees of obligations of Crestline's subsidiaries under the Leases and the Management Agreements) and Crestline can incur additional indebtedness in the future. There can be no assurance that Crestline or any Lessee will have sufficient assets, income and access to financing to enable it to

satisfy its obligations under the Leases or to make payments of fees under the Management Agreements. Host REIT remains obligated to the Managers in case the Lessee fails to pay these fees (but it would be entitled to reimbursement from the Lessee under the terms of the Leases). In addition, the credit rating of Host REIT will be affected by the general creditworthiness of Crestline.

CURRENT HOST COMMON STOCK PRICE IS NOT NECESSARILY INDICATIVE OF THE PRICE OF HOST REIT COMMON STOCK FOLLOWING THE REIT CONVERSION. Host's current stock price is not necessarily indicative of how the market will value Host REIT Common Stock following the REIT Conversion, because of the effect of the distribution of Crestline common stock and cash or other consideration in connection with the Initial E&P Distribution, the acquisition of additional assets in connection with the REIT Conversion, including the Blackstone Acquisition, and the change in Host's organization from a taxable corporation to a REIT. The current stock price of Host reflects the current market valuation of Host's current business and assets (including the Crestline common stock and the cash or other consideration that may be distributed in connection with the Initial E&P Distribution) and does not necessarily take into account the changes in Host's business and operations (including the leasing of its Hotels to Crestline) that will occur in connection with the REIT Conversion. Host's current stock price also is affected by general market conditions.

CASH DISTRIBUTIONS MAY EXCEED CASH AVAILABLE FOR DISTRIBUTION. Distributions will be made at the discretion of Host REIT's Board of Directors and will be affected by a number of factors, including the rental payments received by the Operating Partnership from the Lessees with respect to the Leases of the Hotels, the operating expenses of the Operating Partnership, the level of borrowings and interest expense incurred in borrowing, the Operating Partnership's financial condition and cash available for distribution, the taxable income of Host REIT and the Operating Partnership, the effects of acquisitions and dispositions of assets, unanticipated capital expenditures and distributions required to be made on any preferred units issued by the Operating Partnership. To the extent that cash available for distribution (generally cash from operations less capital expenditures and principal amortization of indebtedness) is insufficient to pay distributions in accordance with the Operating Partnership's distribution policy or to maintain the REIT qualification of Host REIT, the Operating Partnership intends to borrow to make such distributions. The preliminary estimated initial annual cash distributions of the Operating Partnership during the twelve months ending December 31, 1999 (\$231 million) will exceed its estimated cash available for distribution (\$144 million) and cash from contingent rents (\$64 million) during the twelve months ending December 31, 1999, which would require borrowings by the Operating Partnership of approximately \$23 million (or \$0.08 per OP Unit) to make such distributions in accordance with the Operating Partnership's distribution policy. Moreover, if estimated cash from contingent rents were less than \$64 million, then the Operating Partnership also would be required to borrow any such shortfall in order to make such distributions. Actual results may vary substantially from the estimates and no assurance can be given that the Operating Partnership's estimates will prove accurate or that any level of distributions will be made or sustained.

TIMING OF THE RESTRUCTURING TRANSACTIONS AND THE REIT CONVERSION. Host currently expects to complete the Restructuring Transactions and the other transactions comprising the REIT Conversion during 1998, which would permit Host REIT to qualify as a REIT for its 1999 taxable year. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the effectiveness of Host REIT's election could be delayed until January 1, 2000, which would result in Host REIT continuing to pay substantial corporate-level income taxes in 1999 (which would be expected to reduce Host REIT's estimated cash distributions during 1999 to \$0.52 per share from \$0.84 if its REIT election were not effective on January 1, 1999) and could cause the Blackstone Acquisition (which is conditioned, among other things, on consummation of the REIT Conversion by March 31, 1999 and Host REIT qualifying as a REIT for 1999) not to be consummated. In view of the complexity of the REIT Conversion and the number of transactions that must occur to complete the REIT Conversion, Host believes that it is beneficial to the stockholders of Host to complete the REIT Conversion as soon as practicable, even if the REIT Conversion cannot be completed prior to January 1, 1999. If Host REIT's

election to be taxed as a REIT is not effective on January 1, 1999, Host REIT intends to operate following the REIT Conversion in a manner that would permit it to qualify as a REIT at the earliest time practicable, and it might pursue a merger with another entity or other transaction that would permit it to commence a new taxable year and elect REIT status prior to January 1, 2000. Host REIT in any event would elect to be treated as a REIT for federal income tax purposes no later than its taxable year commencing January 1, 2000 assuming it so qualified.

#### CONFLICTS OF INTEREST

RESPONSIBILITIES OF HOST REIT TO OTHER LIMITED PARTNERS IN THE OPERATING PARTNERSHIP. Maryland law imposes certain duties on the Board of Directors of Host REIT to its stockholders. In addition, after the REIT Conversion, Host REIT, as the sole general partner of the Operating Partnership, will have fiduciary obligations under Delaware law with respect to the other limited partners in the Operating Partnership (to the extent such duties have not been modified or eliminated pursuant to the terms of the Partnership Agreement). Although Delaware law provides that Host REIT, as general partner, is subject to the duties of care and loyalty with respect to the limited partners of the Operating Partnership, the Partnership Agreement imposes certain limitations on Host REIT's fiduciary obligations with respect to such limited partners. See "Distribution and Other Policies--Conflicts of Interest Policies." Although some courts have interpreted the fiduciary duties of the Board of Directors in the same way as the duties of a general partner in a limited partnership, it is unclear whether, or to what extent, there are differences in such fiduciary duties. It is possible that, notwithstanding the contractual limitations in the Partnership Agreement, the fiduciary duties of the directors of Host REIT to the stockholders may be less than those of Host REIT, as general partner of the Operating Partnership, to the limited partners of the Operating Partnership. Thus, the discharge of Host REIT's obligations to its stockholders and to the limited partners in the Operating Partnership could result in decisions that may fail to reflect fully the interests of all holders of Host REIT Common Stock and limited partners of the Operating Partnership.

RELATIONSHIPS WITH MARRIOTT INTERNATIONAL AND CRESTLINE. Marriott International currently serves as manager for all but 16 of Host's Hotels, and will continue to manage those Hotels pursuant to the Management Agreements that will be assigned to the Lessees. In addition, Marriott International acts as manager of hotels that will compete with Host REIT's Hotels. As a result, Marriott International may make decisions regarding competing lodging facilities which it manages that would not necessarily be in the best interests of Host REIT or the Lessees. Further, J.W. Marriott, Jr. and Richard E. Marriott, who are brothers, currently serve as directors of Host and directors (and, in the case of J.W. Marriott, Jr., also an officer) of Marriott International. After the REIT Conversion, J.W. Marriott, Jr. will serve as a director of Host REIT and will continue to serve as a director and officer of Marriott International, and Richard E. Marriott will serve as Chairman of the Board of Host REIT and continue to serve as a director of Marriott International. J.W. Marriott, Jr. and Richard E. Marriott also beneficially own (as determined for securities law purposes) approximately 10.6% and 10.2%, respectively, of the outstanding shares of common stock of Marriott International, and will beneficially own approximately 5.33% and 5.31%, respectively, of the outstanding shares of common stock of Crestline (but neither will serve as an officer or director thereof). As a result, J.W. Marriott, Jr. and Richard E. Marriott may have a potential conflict of interest with respect to their obligations as directors of Host REIT in connection with any decisions regarding Marriott International itself (including decisions relating to the Management Agreements involving the Hotels), Marriott International's management of competing lodging properties and Crestline's leasing and other businesses that would not necessarily be in the best interests of Host REIT.

UNCERTAINTIES AT THE TIME OF VOTING ON THE AGREEMENT. There are several uncertainties relating to the REIT Conversion, including the uncertainties described below, that will exist at the time Host stockholders vote on the Agreement. The results of the Partnership Mergers, including whether all or less than all of the Partnerships voted to participate in the Partnership Mergers, will not be known at the time Host stockholders vote on the Agreement. The benefits to Host of the REIT Conversion will be markedly reduced if one or more of the Partnerships do not participate in the Partnership Mergers. In addition, the price of an OP Unit for purposes

of the Partnership Mergers will be equal to the average closing price on the NYSE of a share of Host REIT Common Stock for the first 20 trading days after the effective date of the Partnership Mergers (but, subject to adjustment will not be less than \$9.50 or greater than \$15.50 per OP Unit). The maximum and minimum prices per OP Unit will be reduced if the Blackstone Acquisition is not consummated and, as a result thereof, the Initial E&P Distribution exceeds \$2.50 per share of Host Common Stock or Host REIT Common Stock. Because the value of the OP Units issued to the Limited Partners in the Partnership Mergers will not be determined until after the Partnership Mergers, stockholders cannot know at the time they vote on the Agreement (a) the value of the OP Units to be issued to the Limited Partners, which value could be higher or lower than the value of Host REIT Common Stock at the time of the Merger, and (b) the percentage interest in the Operating Partnership that

such OP Units represent, which affects the allocation of distributions from the Operating Partnership to Host REIT and the Limited Partners. Moreover, because the Limited Partners can tender the OP Units they receive in the Partnership Mergers to Host REIT in exchange for an equal number of shares of Host REIT Common Stock, which Host REIT Common Stock will be issued promptly following the 20th trading day after the effective date of the Partnership Mergers, or to the Operating Partnership for Notes, Host stockholders cannot know the exact number of Notes or shares of Host REIT Common Stock that may be issued in connection with the Partnership Mergers or the extent of the dilutive effect on Host REIT stockholders from the issuance of additional shares of Host REIT Common Stock to the Limited Partners. In addition, the assets and liabilities of the Non-Controlled Subsidiaries will not be known at the time stockholders vote on the Agreement. Although the Restructuring Transactions may be consummated with the expectation that certain other transactions comprising the REIT Conversion, such as the Private Partnership Transactions and the Blackstone Acquisition, will occur, there is no assurance that such will be the case. There is no assurance as to the outcome of such matters, and if certain of these transactions or events are not consummated as  $\frac{1}{2}$ expected, either the REIT Conversion may not be consummated or the value of Host REIT Common Stock and the amount of cash available for distribution to stockholders could be adversely affected, and Host REIT could fail to qualify as a REIT for federal income tax purposes.

INABILITY TO OBTAIN THIRD-PARTY CONSENTS MAY HAVE A MATERIAL ADVERSE EFFECT. There are numerous third-party consents which are required to be obtained in order to consummate the Restructuring Transactions and the other transactions comprising the REIT Conversion. These include consents of many hotel project lenders, ground lessors, joint venture partners, Marriott International and others. The inability of Host, the Operating Partnership or Host REIT to obtain one or more such consents could cause a default under cross-default provisions of the Company's principal credit facilities. Although Host will not consummate the Restructuring Transactions or the REIT Conversion unless it believes that the inability of Host, the Operating Partnership or Host REIT to obtain one or more consents would not reasonably be expected to have a material adverse effect on the Company's business, financial condition or results of operations, there can be no assurance that such a material adverse effect will not occur, which could reduce the value of Host REIT Common Stock.

EXPIRATION OF THE LEASES AND POSSIBLE INABILITY TO FIND OTHER LESSEES. The Leases generally will expire seven to ten years after the Effective Date, and there can be no assurance that the affected Hotels will be relet to the Lessees (or if relet, will be relet on terms as favorable to Host REIT). If the Hotels are not relet, Host REIT will be required to find other lessees, which lessees must meet certain requirements set forth in the Management Agreements and the Code. There can be no assurance that satisfactory lessees could be found or as to the terms and conditions on which Host REIT would be able to relet the Hotels or enter into new leases with such lessees, which could result in a failure of Host REIT to qualify as a REIT or in reduced cash available for distribution.

LEASES COULD IMPAIR THE SALE OR OTHER DISPOSITION OF HOST REIT'S HOTELS. Each Lease generally provides for a termination payment if the Lease is terminated by Host REIT prior to the expiration of the term of such Lease (including due to a change in the federal income tax laws that allows Host REIT to operate the Hotels without jeopardizing Host REIT's status as a REIT), except following a default by a Lessee and in certain other circumstances (including in connection with the sale of up to 12 Hotels without a termination payment) or unless Host REIT leases to the Lessee a comparable substitute hotel. The termination fee is equal to the fair market value of the Lessee's leasehold interest in the remaining term of the Lease. The payment of such termination fee under the Leases could have the effect of impairing the ability of Host REIT to sell its Hotels if market conditions otherwise warrant such a sale and would reduce the net proceeds of any such sale. See "Business and Properties--The Leases--Termination of Leases upon Disposition of Full-Service Hotels."

LIMITATIONS ON SALE OR REFINANCING OF CERTAIN HOTELS. For reasons relating to federal income tax considerations, the agreements by which the Operating Partnership will acquire certain Hotels (or obtain consent to lease certain Hotels to the Lessees) will also restrict the ability of the Operating Partnership to dispose of or refinance the debt secured by such Hotels for varying periods from the Effective Date, depending on the Hotel. Similarly, upon acquiring the Blackstone Hotels, the Operating Partnership will agree not to dispose of the

Blackstone Hotels for ten years (although the Operating Partnership may dispose of up to 50% of the value of the assets contributed to the Operating Partnership by the Blackstone Entities commencing after five years). Thus, even if it were in the best interests of Host REIT to sell such hotels or refinance the debt secured by any of these Hotels, it may be difficult or impossible for Host REIT to do so during their respective lock-out periods.

# RISKS OF OWNERSHIP OF HOST REIT COMMON STOCK

POSSIBLE ADVERSE CONSEQUENCES OF LIMITS ON OWNERSHIP OF HOST REIT COMMON STOCK. To maintain its qualification as a REIT for federal income tax purposes, not more than 50% in value of the outstanding shares of capital stock of Host REIT may be owned, directly or indirectly, by five or fewer individuals (as defined in the Code to include certain entities). See "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Merger--Requirements for Qualification." In addition, a person who owns, directly or by attribution, 10% or more of an interest in a tenant of Host REIT (or a tenant of any partnership in which Host REIT is a partner) cannot own, directly or by attribution, 10% or more of the shares of Host REIT without jeopardizing Host REIT's qualification as a REIT. Primarily to facilitate maintenance of its qualification as a REIT for federal income tax purposes, the Ownership Limit under the Host REIT Charter will prohibit ownership, directly or by virtue of the attribution provisions of the Code, by any person or persons acting as a group of more than 9.8% of the lesser of the number or value of the issued and outstanding shares of Host REIT Common Stock and will prohibit ownership, directly or by virtue of the attribution provisions of the Code, by any person or persons acting as a group of more than 9.8% of the lesser of the number or value of the issued and outstanding shares of any class or series of Host REIT's preferred stock. The Ownership Limit will become applicable to all Host REIT stockholders at the Effective Time (subject to a limited exception for a holder of shares of Host REIT Common Stock solely by reason of the Merger in excess of the Ownership Limit so long as the holder thereof would not own, directly or by attribution under the Code, more than 9.9% in value of the outstanding shares of capital stock of Host REIT as of the Special Merger Ownership Limit Effective Time, and to a limitation on the application of the "group" limitation (but not any other element of the Ownership Limit) to any "group" that otherwise would exceed the Ownership Limit at the Effective Time solely by reason of its status as a "group"). The Board of Directors, in its sole and absolute discretion, may waive or modify the Ownership Limit with respect to one or more persons who would not be treated as "individuals" for purposes of the Code if it is satisfied, based upon information required to be provided by the party seeking the waiver and upon an opinion of counsel satisfactory to the Board of Directors, that ownership in excess of this limit will not cause a person who is an individual to be treated as owning shares in excess of the Ownership Limit, applying the applicable constructive ownership rules, and will not otherwise jeopardize Host REIT's status as a REIT for federal income tax purposes (for example, by causing any tenant of Host REIT or any of the Partnerships or Private Partnerships (including but not limited to Crestline and the Lessees) to be considered a "related party tenant" for purposes of the REIT qualification rules). The constructive ownership rules are complex and may cause Host REIT Common Stock owned directly or constructively by a group of related individuals or entities to be constructively owned by one individual or entity. As a result, the acquisition of less than 9.8% of the outstanding shares of Host REIT Common Stock or 9.8% of the outstanding shares of any class of Host REIT preferred stock (or the acquisition of an interest in an entity which owns shares of Host REIT's capital stock) by an individual or entity could cause that individual or entity (or another individual or entity) to own constructively in excess of 9.8% of the outstanding shares of Host REIT Common Stock or 9.8% of the outstanding shares of any class of Host REIT's preferred stock and thus subject such shares to the remedy provisions under the Ownership Limit.

ANY SHARES OF HOST REIT COMMON STOCK HELD AT THE EFFECTIVE TIME AND NOT SUBSEQUENTLY TRANSFERRED PRIOR TO THE SPECIAL MERGER OWNERSHIP LIMIT EFFECTIVE TIME, OR ACQUIRED OR HELD AT ANY TIME AFTER THE EFFECTIVE TIME IN VIOLATION OF THE OWNERSHIP LIMIT, WILL BE TRANSFERRED AUTOMATICALLY TO A TRUST FOR THE BENEFIT OF A DESIGNATED CHARITABLE BENEFICIARY, AND THE PERSON WHO ACQUIRED SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL NOT BE ENTITLED TO ANY DISTRIBUTIONS THEREON OR TO VOTE SUCH EXCESS SHARES OF HOST REIT COMMON STOCK. THE HOLDER OF ANY SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL RECEIVE THE LESSER OF THE VALUE OF SUCH EXCESS SHARES AS OF THE EFFECTIVE TIME OR THE CASH PROCEEDS FROM THE SALE OF SUCH EXCESS SHARES OF HOST REIT COMMON STOCK BY THE TRUSTEE OF THE TRUST. AFTER THE EFFECTIVE TIME, ANY PERSON WHO ACQUIRES HOST REIT COMMON STOCK IN EXCESS OF THE OWNERSHIP LIMIT WILL NOT RECEIVE ANY PROCEEDS

FROM THE SUBSEQUENT SALE THEREOF IN EXCESS OF THE LESSER OF THE PRICE PAID THEREFOR OR THE AMOUNT REALIZED FROM SUCH SALE. A TRANSFER OF HOST REIT COMMON STOCK TO A PERSON WHO, AS A RESULT OF THE TRANSFER, VIOLATES THE OWNERSHIP LIMIT MAY BE VOID UNDER CERTAIN CIRCUMSTANCES, AND, IN ANY EVENT, WOULD DENY THE TRANSFEREE ANY OF THE ECONOMIC BENEFITS OF OWNING HOST REIT COMMON STOCK IN EXCESS OF THE OWNERSHIP LIMIT. TO AVOID THE ADVERSE EFFECTS OF THE OWNERSHIP LIMIT, ANY HOLDER OF HOST COMMON STOCK WHO WOULD OWN SHARES IN EXCESS OF THE OWNERSHIP LIMIT AT THE EFFECTIVE TIME SHOULD DISPOSE OF ANY SUCH EXCESS SHARES PRIOR THERETO. See "Description of Host REIT Capital Stock--Restrictions on Ownership and Transfer." The Ownership Limit may have the effect of delaying, deferring or preventing a change in control and, therefore, could adversely affect the stockholders' ability to realize a premium over the then-prevailing market price for Host REIT Common Stock in connection with such transaction.

LIMITATIONS ON ACQUISITION OF HOST REIT COMMON STOCK AND CHANGE IN CONTROL. The Host REIT Charter and Host REIT Bylaws, the Partnership Agreement, the Host REIT Rights Agreement (to be adopted by Host REIT to replace the existing stockholder rights plan under the Host Rights Agreement) and Maryland law will contain a number of provisions (as further described under "Certain Provisions of Maryland Law and the Host REIT Charter and Bylaws" and "The Restructuring Transactions--Comparison of Rights of Stockholders of Host and Host REIT") that could delay, defer or prevent a transaction or a change of control of Host REIT that might involve a premium price for holders of Host REIT Common Stock or otherwise be in their best interests, including the following:

OWNERSHIP LIMIT. The 9.8% Ownership Limit described under "--Possible Adverse Consequences of Limits on Ownership of Host REIT Common Stock" above may have the effect of precluding a change in control of Host REIT by a third party without the consent of the Board of Directors, even if such change in control would be in the interest of the stockholders of Host REIT (and even if such change in control would not reasonably jeopardize the REIT status of Host REIT).

STAGGERED BOARD. The Host REIT Charter will provide that the Board of Directors initially shall consist of eight members and may be thereafter increased or decreased in accordance with the Host REIT Bylaws, provided that the total number of directors may not be fewer than three nor more than thirteen. Pursuant to the Host REIT Bylaws, the number of directors shall be fixed by the Board of Directors within the limit set forth in the Host REIT Charter. The Board of Directors of Host REIT will be divided into three classes of directors. The terms of the first, second and third classes will expire in 1999, 2000 and 2001, respectively. Directors for each class will be chosen for a three-year term upon the expiration of the then current class' term, beginning in 1999. The staggered terms for directors may affect the stockholders' ability to effect a change in control of Host REIT even if a change in control would be in the interest of the stockholders of Host REIT.

REMOVAL OF BOARD OF DIRECTORS. The Host REIT Charter will provide that, except for any directors who may be elected by holders of a class or series of shares of capital stock other than Host REIT Common Stock, directors may be removed only for cause and only by the affirmative vote of stockholders holding at least two-thirds of the shares then outstanding and entitled to be cast for the election of directors. Vacancies on the Board of Directors may be filled by the concurring vote of a majority of the remaining directors and, in the case of a vacancy resulting from the removal of a director by the stockholders by at least two-thirds of all the votes entitled to be cast in the election of directors.

PREFERRED STOCK; CLASSIFICATION OR RECLASSIFICATION OF UNISSUED SHARES OF CAPITAL STOCK WITHOUT STOCKHOLDER APPROVAL. The Host REIT Charter provides that the total number of shares of stock of all classes which Host REIT has authority to issue is 800,000,000 shares of stock, initially consisting of 750,000,000 shares of Host REIT Common Stock, par value \$.01 per share, and 50,000,000 shares of preferred stock, par value \$.01 per share ("Host REIT Preferred Stock"). The Board of Directors is authorized, without a vote of stockholders, to classify or reclassify any unissued shares of stock, including Host REIT Common Stock into Host REIT Preferred Stock or vice versa, and to establish the preferences and rights of any preferred or other class or series of stock to be issued. The issuance of preferred stock or other stock having special preferences or rights could have the effect of delaying or preventing a change in control of Host REIT even if a change in control would be in the interest of the stockholders of Host REIT. Because the Board of Directors will have the power to establish the preferences and rights of additional classes or series of stock without a stockholder vote, the Board of Directors may afford the holders of any

such class or series preferences, powers and rights, including voting rights, senior to the rights of holders of Host REIT Common Stock.

CONSENT RIGHTS OF THE LIMITED PARTNERS. Under the Partnership Agreement, Host REIT generally will be able to merge or consolidate with another entity with the consent of partners holding percentage interests in the Operating Partnership ("Percentage Interests") that are more than 50% of the aggregate Percentage Interests of the outstanding partnership interests entitled to vote thereon (including any such partnership interests held by Host REIT) as long as the holders of OP Units either will receive or will have the right to receive the same consideration as the holders of Host REIT Common Stock. Host REIT, as holder of a majority of the OP Units, would be able to control the outcome of such vote. Under the Host REIT Charter, the approval of the holders of at least two-thirds of the outstanding shares of Host REIT Common Stock generally will be necessary to effectuate such merger or consolidation.

MARYLAND BUSINESS COMBINATION LAW. Under the MGCL, unless an exemption is available, certain "business combinations" (including certain issuances of equity securities) between a Maryland corporation and any person who owns 10% or more of the voting power of the corporation's then outstanding shares of capital stock (an "Interested Stockholder") or an affiliate of the Interested Stockholder are prohibited for five years after the most recent date in which the Interested Stockholder becomes an Interested Stockholder. Thereafter, any such business combination must be approved by (i) a supermajority (80%) of outstanding voting shares, and (ii) two-thirds of voting shares (other than voting shares held by an Interested Stockholder) unless, among other conditions, the corporation's common stockholders receive a minimum price (as defined in the MGCL) for their shares and the consideration is received in cash or in the same form as previously paid by the Interested Stockholder. A business combination that is approved by the board of directors of a Maryland corporation at any time before an Interested Stockholder first becomes an Interested Stockholder is not subject to the special voting requirements. Host REIT has not "optedout" of the business combination provisions of the MGCL and, accordingly, will be subject to such provisions although Host REIT may elect to opt-out of these provisions in the future. The Board of Directors of Host REIT has adopted a resolution exempting from the operation of the "business combination" statute the acquisition of shares by Marriott International (together with its successors and permitted assignees, if any) pursuant to the terms of the Marriott International Purchase Right as well as any other transactions involving Host REIT, Marriott International, J.W. Marriott, Jr. and Richard E. Marriott; provided that any such other transaction with Marriott International that is not in the ordinary course of business or with J.W. Marriott, Jr. or Richard E. Marriott must be approved by a majority of the directors of Host REIT present at a meeting at which a quorum is present, including a majority of the disinterested directors, in addition to any vote of stockholders required by other provisions of the MGCL. See "Certain Provisions of Maryland Law and the Host REIT Charter and Bylaws--Changes in Control Pursuant to Maryland Law.

MARYLAND CONTROL SHARE ACQUISITION LAW. Under the MGCL, unless a corporation elects not to be subject thereto, "control shares" acquired in a "control share acquisition" have no voting rights except to the extent approved by stockholders by a vote of two-thirds of the votes entitled to be cast on the matter, excluding shares owned by the acquiror and by officers or directors who are employees of the corporation. "Control shares" are voting shares which, if aggregated with all other such shares previously acquired by the acquiror or in respect of which the acquiror is able to exercise or direct the exercise of voting power (except solely by virtue of a revocable proxy), would entitle the acquiror to exercise voting power in electing directors within one of the following ranges of voting power: (i) one-fifth or more but less than one-third, (ii) one-third or more but less than a majority or (iii) a majority or more of the voting power. Control shares do not include shares the acquiring person is then entitled to vote as a result of having previously obtained stockholder approval. A "control share acquisition" means the acquisition of control shares, subject to certain exceptions. Host REIT will be subject to these control share provisions of Maryland law. The Bylaws of Host REIT will contain an exemption from the control share acquisition provisions for any shares acquired by Marriott International (together with its successors and permitted assignees, if any) pursuant to the Marriott International Purchase Right, to the extent it is exercised.

ADVANCE NOTICE OF DIRECTOR NOMINATIONS AND NEW BUSINESS. The Host REIT Bylaws impose certain advance notice requirements that must be met for nominations of persons for election to the Board of

Directors and the proposal of business to be considered by stockholders. The advance notice provisions contained in the Host REIT Bylaws generally require nominations and new business proposals by stockholders to be delivered to the Secretary of Host REIT not later than the close of business on the 60th day nor earlier than the close of business on the 90th day before the date on which Host REIT mailed its proxy materials for the prior year's annual meeting of stockholders.

MEETINGS OF STOCKHOLDERS; CALL OF SPECIAL MEETINGS; STOCKHOLDER ACTION IN LIEU OF MEETING BY UNANIMOUS CONSENT. The Host REIT Bylaws provide that annual meetings of stockholders shall be held on a date and at the time set by the Board of Directors during the month of May each year (commencing in May 1999). Special meetings of the stockholders may be called by the President or the Board of Directors or on the written request of stockholders entitled to cast a majority of all the votes entitled to be cast at the meeting. Any action required or permitted to be taken by the stockholders must be effected at a duly called annual or special meeting of stockholders or by unanimous written consent.

MERGER, CONSOLIDATION, SHARE EXCHANGE AND TRANSFER OF ASSETS OF HOST REIT. Pursuant to provisions to be contained in the Host REIT Charter, subject to the terms of any class or series of capital stock at the time outstanding, Host REIT may merge with or into another entity, may consolidate with one or more other entities, may participate in a share exchange or may transfer its assets within the meaning of the MGCL if approved (i) by the Board of Directors in the manner provided in the MGCL and (ii) by stockholders by the affirmative vote of two-thirds of all the votes entitled to be cast on the matter, except that any merger of Host REIT with or into a trust organized for the purpose of changing Host REIT's form of organization from a corporation to a trust will require the approval of stockholders of Host REIT by the affirmative vote only of a majority of all the votes entitled to be cast on the matter. Under the MGCL, certain mergers may be accomplished without a vote of stockholders and a share exchange need be approved by a Maryland successor only by its Board of Directors. A voluntary dissolution of Host REIT also would require the affirmative vote of two-thirds of all the votes entitled to be cast on the matter.

AMENDMENTS TO THE HOST REIT CHARTER AND BYLAWS. The provisions to be contained in the Host REIT Charter relating to restrictions on transferability of Host REIT Common Stock, the classified Board and fixing the size of the Board within the range set forth in the Host REIT Charter, as well as the provisions relating to removal of directors and the filling of Board vacancies may be amended only by a resolution adopted by the Board of Directors and approved by stockholders by the affirmative vote of the holders of not less than two-thirds of the votes entitled to be cast on the matter. As permitted under the MGCL, the Host REIT Charter and the Host REIT Bylaws will provide that directors have the exclusive right to amend the Host REIT Bylaws. Amendments to this provision of the Host REIT Charter also would require Board action and approval by two-thirds of all votes entitled to be cast on the matter.

HOST REIT RIGHTS AGREEMENT. Host REIT intends to adopt the Host REIT Rights Agreement to replace the existing stockholder rights plan under the Host Rights Agreement. The new Host REIT Rights Agreement is expected to provide, among other things, that upon the occurrence of certain events, stockholders will be entitled to purchase from Host REIT a newly created series of junior preferred stock, subject to Host REIT's Ownership Limit. The preferred stock purchase rights will be triggered by the earlier to occur of (i) ten days following the date of a public announcement that a person or group acting in concert has acquired, or obtained the right to acquire, beneficial ownership of 20% or more of the outstanding shares of Host REIT Common Stock or (ii) ten business days following the commencement of or announcement of an intention to make a tender or exchange offer, the consummation of which would result in the acquiring person becoming the beneficial owner of 20% or more of such outstanding shares of Host REIT Common Stock. The preferred stock purchase rights would cause substantial dilution to a person or group that attempts to acquire Host REIT on terms not approved by the Board of Directors. See "Description of Host REIT Capital Stock" and "Certain Provisions of Maryland Law and the Host REIT Charter and Bylaws."

MARRIOTT INTERNATIONAL PURCHASE RIGHT. In connection with Host's spin-off of Marriott International in 1993, Marriott International obtained the right to purchase up to 20% of each class of Host's outstanding voting shares at the then fair market value upon the occurrence of certain change of control events involving

Host (the "Marriott International Purchase Right"). The Marriott International Purchase Right will continue in effect after the Merger (until June 2017), subject to certain limitations intended to protect the REIT status of Host REIT. The Marriott International Purchase Right may have the effect of discouraging a takeover of Host REIT, because any person considering acquiring a substantial or controlling block of Host REIT Common Stock will face the possibility that its ability to obtain or exercise control would be impaired or made more expensive by the exercise of the Marriott International Purchase Right.

EFFECT ON STOCK PRICE OF SHARES AVAILABLE FOR FUTURE SALE. Sales of a substantial number of shares of Host REIT Common Stock, or the perception that such sales could occur, could adversely affect prevailing market prices for Host REIT Common Stock. Limited Partners who elect to receive shares of Host REIT Common Stock in connection with the Partnership Mergers will be able to sell such shares at any time after they are received (unless held by an affiliate of Host REIT). Beginning July 1, 1999, half of the approximately 43.7 million OP Units expected to be issued in the Blackstone Acquisition will become redeemable pursuant to the holder's right to redeem them for shares of Host REIT Common Stock or the cash equivalent thereof (as elected by Host REIT) ("Unit Redemption Right"), an additional 25% will be redeemable on October 1, 1999, and the balance will be redeemable on January 1, 2000. In addition, beginning at least one year after the effective date of the Partnership Mergers (or after a lesser period in certain circumstances), other holders of OP Units, including Limited Partners who retain OP Units received in the Partnership Mergers, may be able to sell shares of Host REIT Common Stock received upon exercise of their Unit Redemption Right in the public market pursuant to registration or exemptions from registration. Further, a substantial number of shares of Host REIT Common Stock would, pursuant to employee benefit plans, be issued or reserved for issuance from time to time, including Host REIT Common Stock reserved for issuance pursuant to options granted prior to the consummation of the REIT Conversion, and these shares of Host REIT Common Stock would be available for sale in the public markets from time to time pursuant to exemptions from registration or upon registration. Moreover, the issuance of additional Host REIT Common Stock by Host REIT in the future (including any Host REIT Common Stock that may be issued in connection with the Initial E&P Distribution) would be available for sale in the public markets. Although not yet certain, it is currently contemplated that the Initial E&P Distribution will include the Special Dividend entitling Host stockholders who receive the Initial E&P Distribution and the Blackstone Entities to elect to receive either a specified dollar amount of cash or a specified fraction of a share of Host Common Stock (or a share of Host REIT Common Stock if the Merger has occurred). No prediction can be made about the effect that future sales of shares of Host REIT Common Stock would have on the market price of Host REIT Common Stock.

EFFECT ON STOCK PRICE OF MARKET CONDITIONS. As with other publicly traded equity securities, the value of Host REIT Common Stock will depend upon various market conditions, which may change from time to time. Among the market conditions that may affect the value of the Host REIT Common Stock are the following: (i) the extent of institutional investor interest in Host REIT, (ii) the general market perception of REITs in general and hotel REITs in particular and the attractiveness of their equity securities in comparison to other equity securities (including securities issued by other real estate-based companies), (iii) Host REIT's financial performance, (iv) changes in the tax laws affecting REITs (particularly REITs that primarily own hotels) and (v) general stock and bond market conditions. There can be no assurance that these market conditions would not have a material adverse effect on the market price of Host REIT Common Stock or that the Host REIT Common Stock will not trade at prices below the net asset value of Host REIT's business and assets.

EFFECT ON STOCK PRICE OF EARNINGS AND CASH DISTRIBUTIONS. It is generally believed that the market value of the equity securities of a REIT is primarily based upon the market's perception of the REIT's growth potential for its core portfolio, the value of its real estate portfolio and its prospects for accretive acquisitions and development. The combination of these factors creates a market perception of a REIT's current and potential future cash distributions, whether from operations, sales, acquisitions, development or refinancings, and is secondarily based upon the value of the underlying assets. For that reason, Host REIT Common Stock may trade at prices that are higher or lower than the net asset value per share. To the extent Host REIT retains operating cash flow for investment purposes, working capital reserves or other purposes rather than distributing such cash

flow to stockholders, these retained funds, while increasing the value of Host REIT's underlying assets, may not correspondingly increase the market price of Host REIT Common Stock. The failure of Host REIT to meet the market's expectation with regard to future earnings and cash distributions would likely adversely affect the market price of Host REIT Common Stock.

EFFECT ON STOCK PRICE OF MARKET INTEREST RATES. One of the factors that will influence the price of Host REIT Common Stock will be the dividend yield on Host REIT Common Stock (as a percentage of the price of Host REIT Common Stock) relative to market interest rates. Thus, an increase in market interest rates may lead prospective purchasers of Host REIT Common Stock to expect a higher dividend yield, which would adversely affect the market price of Host REIT Common Stock.

EFFECT ON STOCK PRICE OF UNRELATED EVENTS. As with other publicly traded equity securities, the value of Host REIT Common Stock will depend upon various market conditions, including conditions unrelated to real estate investments generally. Thus, events which depress equity market prices may not have any effect on real estate market values, with the result that Host REIT Common Stock may trade at prices below Host REIT's net asset value.

DEPENDENCE ON EXTERNAL SOURCES OF CAPITAL. As with other REITs, but unlike corporations generally, Host REIT's ability to reduce its debt and finance its growth largely must be funded by external sources of capital because Host REIT generally will have to distribute to its stockholders 95% of its taxable income in order to qualify as a REIT (including taxable income where Host REIT does not receive corresponding cash). Host REIT's access to external capital will depend upon a number of factors, including general market conditions, the market's perception of Host REIT's growth potential, its current and potential future earnings, cash distributions and the market price of Host REIT Common Stock.

## FEDERAL INCOME TAX RISKS RELATING TO REIT QUALIFICATION

GENERAL. Host REIT intends to operate so as to qualify as a REIT under the Code effective for Host REIT's first full taxable year commencing following the REIT Conversion. A REIT generally is not taxed at the corporate level on income it currently distributes to its stockholders as long as it distributes currently at least 95% of its taxable income (excluding net capital gain). No assurance can be provided, however, that Host REIT will so qualify or be able to remain so qualified or that new legislation, Treasury Regulations, administrative interpretations or court decisions will not significantly change the tax laws with respect to Host REIT's qualification as a REIT or the federal income tax consequences of such qualification. In this regard, Host REIT has received an opinion of Hogan & Hartson L.L.P. to the effect that Host REIT, effective for its first full taxable year commencing after the REIT Conversion is completed, will be organized in conformity with the requirements for qualification as a REIT under the Code, and that Host REIT's proposed method of operation will enable it to satisfy the requirements for qualification and taxation as a REIT. This opinion is conditioned upon completion of the REIT Conversion and upon certain factual representations made by Host REIT and the Operating Partnership as to matters relating to the organization and operation of Host REIT, the Operating Partnership, the Partnerships, the Private Partnerships, the Subsidiary Partnerships, the Non-Controlled Subsidiaries, the Host Employee/Charitable Trust, and Crestline and the Lessees, and the economic and other terms of the Leases and the expectations of Host REIT and the Lessees with respect thereto. In addition, this opinion is based upon the factual representations of Host REIT concerning its business and properties as set forth in this Proxy Statement/Prospectus and assumes that the actions described in this Proxy Statement/Prospectus are completed in a timely fashion (generally, prior to the first day of the first taxable year for which Host REIT's election to be treated as a REIT would be effective). Moreover, an opinion of counsel does not bind the IRS or the courts, and no assurance can be provided that such opinion will not be challenged by the IRS or will be sustained by a court if so challenged.

REQUIRED DISTRIBUTIONS AND PAYMENTS. In order to qualify as a REIT, Host REIT will be required each year to distribute to its stockholders at least 95% of its net taxable income (excluding any net capital gain). Due to certain transactions entered into in prior years, Host REIT is expected to recognize substantial amounts of "phantom" taxable income in future years that is not matched by cash flow or EBITDA to the Operating Partnership or Host REIT. As discussed below in "--"Earnings and Profits" Attributable to "C" Corporation

Taxable Years," to qualify as a REIT, Host REIT also will have to distribute to its stockholders not later than the end of its first full taxable year as a REIT an amount equal to the E&P accumulated by Host and its subsidiaries and not distributed before or at the time of the REIT Conversion (including any increases thereto resulting from subsequent IRS audits of years prior to Host REIT's first taxable year as a REIT). In addition, Host REIT will be subject to a 4% nondeductible excise tax on the amount, if any, by which certain distributions made by it with respect to the calendar year are less than the sum of (i) 85% of its ordinary income, (ii) 95% of its capital gain net income for that year, and (iii) any undistributed taxable income from prior periods. Host REIT intends to make distributions to its stockholders to comply with the 95% distribution requirement and to avoid the nondeductible excise tax and will rely for this purpose on distributions from the Operating Partnership. However, differences in timing between taxable income and cash available for distribution due to, among other things, the seasonality of the hospitality industry and the fact that some taxable income will be "phantom" income (i.e., taxable income that is not matched by cash flow or EBITDA to the Operating Partnership) could require the Operating Partnership to borrow funds or to issue additional equity to enable Host REIT to meet the 95% distribution requirement (and therefore to maintain its REIT status) and to avoid the nondeductible excise tax. Host REIT also could be required to pay taxes and liabilities attributable to periods and events prior to the REIT Conversion and taxes in the event it were to fail to qualify as a REIT. In addition, the Operating Partnership's inability to retain earnings (resulting from Host REIT's 95% and other distribution requirements) will generally require the Operating Partnership to refinance debt that matures with additional debt or equity. There can be no assurance that any of these sources of funds, if available at all, would be available to meet the Operating Partnership's distribution and tax obligations.

CONSEQUENCES OF FAILURE TO QUALIFY AS A REIT. If Host REIT fails to qualify as a REIT, it will be subject to federal income tax (including any applicable alternative minimum tax) on its taxable income at regular corporate rates. In addition, unless entitled to relief under certain statutory provisions, Host REIT will be disqualified from treatment as a REIT for the four taxable years following the year during which REIT qualification is lost. The additional tax would significantly reduce the cash available for distribution by Host REIT to its stockholders. Failure of Host REIT to qualify as a REIT could reduce materially the value of the Host REIT Common Stock and would cause all distributions to stockholders to be taxable as ordinary income to the extent of Host REIT's current and accumulated E&P (although, subject to certain limitations under the Code, corporate distributees may be eligible for the dividends received deduction with respect to these distributions). See "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Merger--Failure of Host REIT to Qualify as a REIT." Failure of Host REIT to qualify as a REIT also would result in a default under the New Senior Notes and the New Credit Facility.

"EARNINGS AND PROFITS" ATTRIBUTABLE TO "C" CORPORATION TAXABLE YEARS. In order to qualify as a REIT, Host REIT cannot have at the end of any taxable year any undistributed E&P that is attributable to a "C" corporation taxable year. A REIT has until the close of its first full taxable year as a REIT in which it has non-REIT E&P to distribute such accumulated E&P. Host REIT will be required to distribute this E&P prior to the end of 1999 (the first full taxable year for which the REIT election of Host REIT currently is expected to be effective). Failure to do so would result in disqualification of Host REIT as a REIT at least for taxable year 1999. Host REIT believes that the Initial E&P Distribution, together with any additional distributions of non-REIT E&P made after the REIT Conversion but prior to December 31, 1999, will be sufficient to distribute all of the non-REIT E&P as of December 31, 1999, but there are substantial uncertainties relating to the estimate of Host REIT's non-REIT E&P and the value of noncash consideration to be distributed as part of the Initial E&P Distribution and, thus, there can be no assurance that this requirement will be met. Hogan & Hartson L.L.P. will not provide any opinion as to the amount of Host's undistributed E&P and has relied, for purposes of its opinion as to the qualification of Host REIT as a REIT, upon a representation from Host and Host REIT that Host REIT will not have any undistributed non-REIT E&P as of the end of 1999. See "Federal Income Tax Consequences -- Federal Income Taxation of Host REIT Following the Merger--Requirements for Qualification.'

TREATMENT OF LEASES. To qualify as a REIT, a REIT must satisfy two gross income tests. Rent paid pursuant to the Leases will constitute substantially all of the gross income of Host REIT. In order for the rent

paid pursuant to the Leases to constitute qualifying income for purposes of the gross income tests (a) the Leases must be respected as true leases for federal income tax purposes and not be treated as service contracts, joint ventures or some other type of arrangement, (b) the Lessees must not be regarded as "related party tenants" (as defined in the Code) and (c) no portion of the rent can be based upon the income or profits of the tenant (although rent based upon the gross revenue of a tenant is permissible unless the arrangement does not conform with normal business practice but in reality is used as a means of basing rent on the income or profits of the tenant). The Host REIT Charter will contain the Ownership Limit, which is intended in part to help ensure that neither the Lessees nor future lessees will be treated as "related party tenants," but there can be no assurance that the Ownership Limit will be effective for achieving that result in all circumstances. With regard to the treatment of the Leases for federal income tax purposes, Hogan & Hartson L.L.P. has provided to Host REIT an opinion to the effect that, based upon certain representations of Host REIT regarding the economic and other terms of the Leases and the expectations of Host REIT and the Lessees with respect thereto, the Leases will be respected as leases for federal income tax purposes. An opinion of counsel, however, does not bind the IRS or the courts and this determination ultimately will depend upon the accuracy of the factual representations of Host REIT regarding the Leases. Finally, with regard to whether the rent under the Leases is based upon the income or profits of the Lessees, such rent is based only upon the gross revenues of the Lessees, and Host REIT has represented that the rental provisions and other terms of the Leases conform with normal business practice and were not intended to be used as a means of basing rent on the income or profits of the Lessees. If the Leases were not respected as true leases for federal income tax purposes, if the Lessees were regarded as "related party tenants" or if the rent payable under the Leases were considered based, in whole or in part, upon the income or profits of the Lessees, Host REIT would not be able to satisfy either of the two gross income tests applicable to REITs and, as a result, would lose its REIT status. Accordingly, Host REIT would be subject to corporate level income taxation, which would significantly reduce the cash available for distribution to its stockholders. See "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Merger--Income Tests Applicable to REITs."

TREATMENT OF LEASES OF PERSONAL PROPERTY. Rent that is considered attributable to personal property under the applicable provisions of the Code is not "qualifying" income for purposes of the gross income tests that a REIT must satisfy. Significant portions of hotels (e.g., FF&E) constitute personal property for purposes of these rules. In order to protect Host REIT's ability to qualify as a REIT, the Operating Partnership will sell to a Non-Controlled Subsidiary an estimated \$180 million in value of personal property associated with certain Hotels. The Non-Controlled Subsidiary will lease the personal property it acquires associated with each Hotel directly to the Lessee that is leasing the Hotel. Under each personal property lease, the Non-Controlled Subsidiary will receive rent payments directly from the applicable Lessee, which Host REIT believes represent the fair rental value of the personal property. If for any reason these lease arrangements are not respected for federal income tax purposes, Host REIT likely would not qualify as a REIT.

OTHER TAX LIABILITIES; HOST REIT'S SUBSTANTIAL DEFERRED AND CONTINGENT TAX LIABILITIES. Even if Host REIT qualifies as a REIT, it will be subject, through the Operating Partnership, the Partnerships and the Private Partnerships, to certain federal, state and local taxes on its income and property. See "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Merger--General." In addition, Host REIT will be subject to tax at the regular corporate rate (currently 35%) upon its share of any gain recognized as a result of any sale by the Operating Partnership (within the 10-year period beginning on the effective date of the OP Contribution) of assets, including the full-service hotels, contributed to the Operating Partnership by Host in connection with the OP Contribution, to the extent that such gain existed as of the first day of Host REIT's first taxable year as a REIT. Host has substantial deferred tax liabilities that likely will be recognized by Host REIT in the next ten years as "built-in gain" under these rules (or by a Non-Controlled Subsidiary), without any corresponding receipt of cash by Host REIT or the Operating Partnership. In addition, the IRS may assert liabilities against Host REIT for corporate income taxes for taxable years of Host prior to the time Host REIT qualifies as a REIT. Finally, the Non-Controlled Subsidiaries will be taxable "C" corporations and will pay federal and state income tax on their net income at the full applicable corporate rates.

FAILURE OF THE OPERATING PARTNERSHIP TO QUALIFY AS A PARTNERSHIP. The Operating Partnership and Host REIT have received an opinion of Hogan & Hartson L.L.P. to the effect that the Operating Partnership will be treated as a partnership for federal income tax purposes. An opinion of counsel, however, does not bind the IRS or the courts, and no assurance can be provided that such opinion will not be challenged by the IRS or will be sustained by a court if so challenged. If the IRS were to treat successfully the Operating Partnership as an entity that is taxable as a corporation, Host REIT would cease to qualify as a REIT because the value of Host REIT's ownership interest in the Operating Partnership would exceed 5% of Host REIT's assets and because Host REIT would be considered to hold more than 10% of the voting securities of another corporation. See "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Merger--Asset Tests Applicable to REITs." Moreover, the imposition of a corporate tax on the Operating Partnership would reduce significantly the amount of cash available for distribution to Host REIT (and thus its stockholders). See "Federal Income Tax Consequences--Tax Aspects of Host REIT's Ownership of OP Units."

#### OTHER TAX RISKS

TAX CONSEQUENCES TO HOST OF THE OP CONTRIBUTION AND OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION. Because of the numerous transactions that comprise the OP Contribution and the REIT Conversion and the inherently factual nature of a number of the critical determinations relevant to the tax consequences to Host of the OP Contribution and the REIT Conversion, Host has not sought (and likely could not obtain) an opinion of outside counsel as to the tax consequences to it (in terms of the actual amount of income and gains that would be recognized) of the OP Contribution and the other transactions comprising the REIT Conversion (other than the Merger). Host believes, based upon its knowledge of the relevant facts and advice that it has received from various professional advisers (including both attorneys and accountants on its staff and outside attorneys and accountants) as to the applicable general principles of law, that it will not recognize a material amount of income and gain as a result of the OP Contribution and the other transactions comprising the REIT Conversion. Host, however, will recognize some income and gain in connection with these various transactions, and there can be no assurance that under the relevant facts and circumstances such income and gain would not be material. If Host were to recognize a material amount of income and gain as a result of these various transactions, the resulting income tax liability could be material to Host, and the increase in E&P resulting from such income and gain could materially increase the amount that Host must distribute to its stockholders in order to qualify as a REIT, all of which could have adverse consequences to Host REIT and its stockholders.

43--1

EFFECTS OF SUBSEQUENT EVENTS UPON RECOGNITION OF GAIN. In addition to any gain that might be recognized by Host at the time of the OP Contribution, there are a variety of subsequent events and transactions including (i) the sale or other taxable disposition of appreciated assets contributed by Host to the Operating Partnership in the OP Contribution (including one or more of the Hotels contributed by Host), (ii) the refinancing or repayment of certain liabilities secured by one or more of the Hotels contributed to the Operating Partnership by Host in the OP Contribution, (iii) the issuance of additional OP Units, including in connection with the acquisition of additional properties by the Operating Partnership in exchange for OP Units or other equity interests in the Operating Partnership, (iv) an increase to the basis of one or more of the Hotels contributed to the Operating Partnership by Host in the OP Contribution resulting from capital expenditures and (v) the elimination over time of the disparity between the current tax basis of one or more of the Hotels contributed to the Operating Partnership by Host in the OP Contribution and the "book basis" of such assets (based upon their fair market value at the time of the OP Contribution) that could cause Host REIT to recognize part or all of the taxable gain that otherwise has been deferred pursuant to the Merger and the OP Contribution.

## RISKS OF OPERATION

COMPETITION IN THE LODGING INDUSTRY. The profitability of the Hotels is subject to general economic conditions, the management abilities of the Managers (including primarily Marriott International), competition, the desirability of particular locations and other factors relating to the operation of the Hotels. The full-service segment of the lodging industry in which the Hotels primarily operate is highly competitive and the Hotels generally operate in geographical markets that contain numerous competitors. The Hotels' success will be dependent, in large part, upon their ability to compete in such areas as access, location, quality of accommodations, room rate structure, the quality and scope of food and beverage facilities and other services and amenities. Although the competitive position of each of Host REIT's hotel properties differs from market to market, Host REIT believes that its properties generally compare favorably to their competitive set in the markets in which they operate on the basis of these factors. Furthermore, Host REIT's strategy is to affiliate its properties with managers operating under the highest quality brand names in the industry which Host REIT believes will enhance their competitive position. Nonetheless, there can be no assurance that these managers will maintain the quality of their brand names. Furthermore, competing properties may be built or existing products enhanced such that they offer characteristics more favorable than those offered by Host REIT's properties. See "Business and Properties--Competition." The lodging industry, including the Hotels (and thus Host REIT), may be adversely affected in the future by (i) national and regional economic conditions, (ii) changes in travel patterns, (iii) taxes and government regulations which influence or determine wages, prices, interest rates, construction procedures and costs, (iv) the availability of credit and (v) other factors beyond the control of Host REIT.

SUBSTANTIAL INDEBTEDNESS OF HOST REIT. Host REIT will have substantial indebtedness. As of September 11, 1998, on a pro forma basis assuming the Full Participation Scenario, Host REIT had outstanding indebtedness totaling approximately \$5 billion, which represents an approximately 59% debt-to-total market capitalization ratio on a pro forma basis at such date (based upon a price per share of Host REIT Common Stock of \$12.50 after giving effect to the REIT Conversion). Host REIT's business is capital intensive and it will have significant capital requirements in the future. Host REIT's leverage level and other factors beyond its control (including market conditions) could affect its ability to (i) obtain financing in the future, (ii) undertake refinancings on terms and subject to conditions deemed acceptable by Host REIT, (iii) make distributions, (iv) pursue its acquisition strategy or (v) compete effectively or operate successfully under adverse economic conditions. In the event that Host REIT's cash flow and working capital are not sufficient to fund Host REIT's expenditures or to service its indebtedness, Host REIT would be required to raise additional funds through capital contributions, the refinancing of all or part of its indebtedness, the incurrence of additional permitted indebtedness or the sale of assets. There can be no assurance that any of these sources of funds would be available, if at all, in amounts sufficient for Host REIT to meet its obligations. Moreover, even if Host REIT were able to meet its obligations, its leveraged capital structure could significantly limit its ability to finance its acquisition program and other capital expenditures, to compete effectively or to operate successfully, especially under adverse economic conditions.

NO LIMITATION ON DEBT. Host REIT will have a policy of incurring debt only if, immediately following such incurrence, its debt-to-total market capitalization ratio on a pro forma basis would be 60% or less. However, there are no limitations in Host REIT's or the Operating Partnership's organizational documents that limit the amount of indebtedness that either entity may incur, although the Operating Partnership's and Host REIT's debt instruments will contain certain restrictions on the amount of indebtedness that Host REIT may incur. Accordingly, the Board of Directors could alter or eliminate this policy unilaterally without stockholder approval from time to time to the extent permitted by its debt agreements. If this policy were changed, Host REIT could become more highly leveraged, resulting in an increase in debt service payments that could adversely affect Host REIT's cash flow and consequently, the cash available for distributions to holders of OP Units, including Host REIT and, in turn, to stockholders of Host REIT and could increase the risk of default on Host REIT's indebtedness.

THE BOARD MAY CHANGE INVESTMENT AND OTHER POLICIES WITHOUT STOCKHOLDER APPROVAL. Host REIT's Board of Directors may change the investment, financing and other policies of Host REIT without stockholder approval. Such policy changes may have adverse consequences to Host REIT.

MANAGEMENT AGREEMENTS COULD IMPAIR THE SALE OR OTHER DISPOSITION OF HOST REIT'S HOTELS. Marriott International serves as the manager for all but 16 of Host REIT's Hotels and provides various other services to Host and its subsidiaries. Although the Lessees will have primary liability under the Management Agreements as long as the Leases are in effect, Host REIT will remain liable thereunder. The Hotels generally may not be sold, leased or otherwise transferred unless the transferee assumes the Management Agreements relating thereto and meets certain other conditions. The possible desire of Host REIT, from time to time, to finance, refinance or effect a sale of any of the properties managed by Marriott International or another manager may, depending upon the structure of such transactions, result in a need to obtain Marriott International's consent, which could include modification of the Management Agreements with Marriott International or such other manager with respect to such property. Any such modification proposed by Host REIT may not be acceptable to Marriott International or such other manager, and the lack of any required consent from Marriott International or such other manager would prohibit Host REIT from consummating such financing, refinancing or sale without breaching such Management Agreement. In addition, certain situations could arise where actions taken by Marriott International or another manager in its capacity as manager of competing lodging properties would not necessarily be in the best interests of Host REIT. Nevertheless, Host REIT believes that there is sufficient mutuality of interest between Host REIT and Marriott International or another manager to result in a mutually productive relationship.

RENTAL REVENUES FROM HOTELS SUBJECT TO PRIOR RIGHTS OF LENDERS. In accordance with the mortgage loan agreements with respect to outstanding indebtedness of certain of the Partnerships or the Private Partnerships, the rental revenues received by such Partnerships under certain Leases first will be used to satisfy the debt service on such outstanding indebtedness with only the cash flow remaining after debt service being available to satisfy other obligations of such Partnership or Private Partnership (including paying property taxes and insurance, funding the required FF&E reserves for the Hotels and capital improvements and paying debt service with respect to unsecured debt) and to make distributions to holders of OP Units (including Host REIT), which affects Host REIT's ability, in turn, to make distributions to stockholders of Host REIT.

HOTELS SUBJECT TO GROUND LEASES MAY AFFECT HOST REIT'S REVENUES. Of the approximately 125 Hotels in which Host REIT initially is expected to hold an interest, approximately 60 are subject to ground leases. Such ground leases generally require increases in ground rent payments every five years. To the extent that the rents payable under the Leases do not increase at the same rate as the increases under the ground leases, it could affect the Operating Partnership's cash available for distributions to holders of OP Units, including Host REIT, and Host REIT's ability, in turn, to make cash available for distributions to its stockholders. In addition, any sale of a Hotel encumbered by a ground lease would be made subject to such ground lease and the value realized by

Host REIT in such sale might not be as high if such Hotel were not sold subject to such ground lease or were sold subject thereto.

GENERAL REAL ESTATE INVESTMENT RISKS. Host REIT stockholders will continue to bear risks associated with real estate investments. The yields available from equity investments in real estate and Host REIT's ability to service debt depend, in large part, upon the amount of rental revenues generated, expenses incurred and capital expenditures required in the operation of its business. Host REIT's income and ability to make distributions will be dependent upon the rent payable by the Lessees exceeding the amounts required for debt service, property taxes and other expenses payable by Host REIT (including required FF&E reserves and capital expenditures). The rental payments payable by the Lessees will be affected in part by the sales generated by the Managers from operation of the Hotels. The Lessees' ability to pay rent accrued under the Leases will depend in significant part upon the ability of the Managers to generate gross sales in excess of its requirements to meet operating expenses. Host REIT's rental income from the Hotels may, therefore, directly or indirectly, be adversely affected by a number of factors, including the general economic climate, local real estate conditions, such as an oversupply of, or a reduction in demand for, hotel space, the attractiveness of the Hotels to consumers, the quality, philosophy and performance of management, the ability of the Lessees to maximize rental payments to the Operating Partnership, the ability of the Manager to effectively operate the Hotels, competition from comparable hotels, changes in room rates and increases in operating costs due to inflation and other factors, which increases may not necessarily be passed through fully to guests. In addition, Host REIT's rental income from the Hotels and real estate values also are affected by such factors as the cost of compliance with government regulation, including zoning and tax laws, the potential for liability under applicable laws, interest rate levels and the availability of financing. Certain significant expenditures associated with each equity investment in a Hotel (such as mortgage payments, if any, real estate taxes and maintenance costs) also may not decrease even though circumstances cause a reduction in Host REIT's rental income from the Hotel. If any of the above occurs, the Operating Partnership's ability to make distributions to holders of OP Units, including Host REIT, and Host REIT's ability, in turn, to make distributions to its stockholders, could be adversely affected.

POSSIBLE UNDERPERFORMANCE OF NEW ACQUISITIONS. In the future, Host REIT expects to pursue acquisitions of additional full-service hotels and other types of real estate. Acquisitions entail the risk that such investments will fail to perform in accordance with expectations. Host REIT anticipates that, in certain circumstances, it may use OP Units in the Operating Partnership as consideration to acquire hotels from tax-sensitive sellers and, in connection with such acquisitions, it may agree to certain restrictions on the Operating Partnership's ability to sell, or reduce the amount of mortgage indebtedness on, such acquired hotels, which may increase Host REIT's leverage and which may impair Host REIT's ability to take actions that would otherwise be in the best interests of Host REIT.

SEASONALITY. The hotel industry is seasonal in nature. The seasonality of the industry may, from time to time, affect either the amount of rent that accrues under the Leases or the ability of the Lessees to make timely rent payments under the Leases. An inability of the Lessees to make timely rent payments to Host REIT could adversely affect the ability of the Operating Partnership to make distributions to holders of OP Units, including Host REIT, and Host REIT's ability, in turn, to make distributions to its stockholders.

ILLIQUIDITY OF REAL ESTATE. Real estate investments are relatively illiquid and, therefore, will tend to limit the ability of Host REIT to sell and purchase hotels promptly in response to changes in economic or other conditions. This could make it difficult for Host REIT to sell any of its Hotels, even if a sale were in the interest of Host REIT.

# MISCELLANEOUS RISKS

CHANGES IN LAWS. Increases in real estate or business improvement district taxes will not result in increased rental payments to Host REIT under the Leases, with the result that they may adversely affect the Operating Partnership's cash flow from operations and its ability to maintain the expected level of distributions

to holders of OP Units, including Host REIT, and, in turn, to Host REIT's stockholders. Similarly, changes in laws increasing the potential liability for environmental conditions existing at Hotels or increasing the restrictions on discharges or other conditions, as well as changes in laws affecting construction and safety requirements, may result in significant unanticipated capital expenditures, which, to the extent such expenditures must be borne by Host REIT as the lessor of the Hotels, would adversely affect the Operating Partnership's cash flow from operations and its ability to make distributions to holders of OP Units, including Host REIT and Host REIT's ability, in turn, to make distributions to its stockholders.

DEPENDENCE UPON KEY PERSONNEL. Host REIT is dependent upon the efforts of its executive officers. While Host REIT believes that it could find replacements for these key personnel, the loss of their services could have a significant adverse effect on the operations of Host REIT. Host REIT does not intend to obtain key-man life insurance with respect to any of its executive officers.

POTENTIAL LITIGATION RELATED TO THE REIT CONVERSION. Over the last several years, business reorganizations involving the combination of several partnerships into a single entity occasionally have given rise to investor lawsuits. These lawsuits have involved claims against the general partners of the participating partnerships, the partnerships themselves and related persons involved in the structuring of, or benefiting from, the conversion or reorganization, as well as claims against the surviving entity and its directors and officers. Certain lawsuits are pending against Host and its affiliates by limited partners in certain limited partnerships. If any lawsuits are filed in connection with the Partnership Mergers or other transactions comprising the REIT Conversion, such lawsuits could delay the closing of the REIT Conversion or result in substantial damage claims against Host REIT or the general partners of the Partnerships, each of which is a wholly-owned, direct or indirect subsidiary of Host. The Partnerships are each obligated to indemnify their general partner for claims against them arising from their role as general partner other than to the extent they are guilty of negligence, fraud, misconduct or breach of fiduciary duty. Because Host REIT will be acquiring the Partnerships or Private Partnerships through the Partnership Mergers, the Operating Partnership and Host REIT indirectly will be subject to the indemnification obligations of the Partnerships and the Private Partnerships to their general partners and any obligations of the Partnerships and the Private Partnerships to pay damages to the extent not covered by any available insurance. See "Business and Properties--Legal Proceedings.'

RISK INVOLVED IN INVESTMENTS THROUGH PARTNERSHIPS OR JOINT VENTURES. Instead of purchasing hotel properties directly, Host REIT may invest as a coventurer. Joint venturers often have shared control over the operation of the joint venture assets. Therefore, such investments may, under certain circumstances, involve risks such as the possibility that the co-venturer in an investment might become bankrupt, or have economic or business interests or goals that are inconsistent with the business interests or goals of Host REIT, or be in a position to take action contrary to the instructions or the requests of Host REIT or contrary to Host REIT's policies or objectives. Consequently, actions by a co-venturer might result in subjecting hotel properties owned by the joint venture to additional risk. Although Host REIT generally will seek to maintain sufficient control of any joint venture to permit Host REIT's objectives to be achieved, it may be unable to take action without the approval of its joint venture partners or its joint venture partners could take actions binding on the joint venture without Host REIT's consent. Additionally, should a joint venture partner become bankrupt, Host REIT could become liable for such partner's share of joint venture liabilities.

YEAR 2000 PROBLEM. Year 2000 issues have arisen because many existing computer programs and chip-based embedded technology systems use only the last two digits to refer to a year, and therefore do not properly recognize a year that begins with "20" instead of the familiar "19". If not corrected, many computer applications could fail or create erroneous results. Although Host has adopted a compliance program in recognition of the possible interruptions that may occur as a result of Year 2000 issues and taken certain steps toward Year 2000 remediation, there can be no assurances that the steps taken toward Year 2000 remediation by

Host or third parties will be properly and timely completed, and failure to do so could have a material adverse effect on Host, its business and its financial condition. Host cannot predict the actual effects on its business of Year 2000 issues, which depend on uncertainties such as whether significant third parties properly and timely address Year 2000 issues and whether broadbased or systematic economic failures may occur. Host is also unable to predict the severity and duration of any such failures, which could include disruptions in passenger transportation or transportation systems generally, loss of utility and/or telecommunications services, the loss or disruption of hotel reservations made on centralized reservation systems and errors or failures in financial transactions or payment processing systems such as credit cards. Moreover, following the REIT Conversion, Host REIT will be dependent upon Crestline to interface with third parties in addressing Year 2000 issues at the Hotels leased to Crestline. Due to the general uncertainty inherent with respect to Year 2000 issues and Host's dependence on third parties (including Crestline following the REIT Conversion), Host is unable to determine at this time whether the consequences of Year 2000 failures will have a material impact on Host or Host REIT. Although Host's Year 2000 compliance program and Crestline's adoption thereof following the REIT Conversion are expected to significantly reduce the level of uncertainty concerning Year 2000 issues and management believes that the possibility of significant interruptions of normal operations should be reduced, there is no assurance that this will be the case.

UNINSURED LOSS. Host REIT will carry comprehensive liability, fire, flood, extended coverage and rental loss (for rental losses extending up to 12 months) with respect to its Hotels with policy specification and insured limits customarily carried for similar hotels. Certain types of losses (such as from earthquakes and environmental hazards), however, may be either uninsurable or not economically insurable. Should an uninsured loss occur, Host REIT could lose both its capital invested in, and anticipated profits from, one or more of its Hotels.

AMERICANS WITH DISABILITIES ACT. The Hotels must comply with Title III of the Americans with Disabilities Act (the "ADA") to the extent that such Hotels are "public accommodations" or "commercial facilities" as defined by the ADA. The ADA may require removal of structural barriers to access by persons with disabilities in certain public areas of Host REIT's Hotels where such removal is readily achievable. Host REIT believes that the Hotels will not be required to make substantial non-budgeted capital expenditures to address the requirements of the ADA. However, noncompliance with the ADA could result in substantial capital expenditures to remove structural barriers, as well as the imposition of fines or an award of damages to private litigants which might adversely affect the Operating Partnership's ability to make expected distributions to holders of OP Units, including Host REIT and Host REIT's ability, in turn, to make distributions to its stockholders. Under the Leases, Host REIT would be required to fund all such expenditures.

OTHER REGULATORY ISSUES. Host REIT's Hotels will be subject to various forms of regulation in addition to the ADA, including building codes, regulations pertaining to fire safety and other regulations which may from time to time be enacted. Host REIT may be required to incur significant costs to comply with any future changes in such regulations.

POSSIBLE ENVIRONMENTAL LIABILITIES. Under various federal, state and local laws, ordinances and regulations, owners or operators of real estate may be required to investigate and clean up certain hazardous substances released at a property, and may be held liable to a governmental entity or to third parties for property damage or personal injuries and for investigation and clean-up costs incurred by the parties in connection with any contamination. In addition, some environmental laws create a lien on a contaminated site in favor of the government for damages and costs it incurs in connection with the contamination. The presence of contamination or the failure to remediate contamination may adversely affect the owner's ability to sell or lease real estate or to borrow using the real estate as collateral. No assurances can be given that (i) a prior owner, operator or occupant, such as a tenant, did not create a material environmental condition with respect to any Hotel does not exist or (iii) future uses or conditions (including, without limitation, changes in applicable environmental laws and regulations) will not result in the imposition of environmental liability.

In addition, no assurances can be given that all potential environmental liabilities have been identified or properly quantified or that no prior owner, operator or past or current guest has created an environmental condition not known to Host REIT. Moreover, no assurances can be given that (i) future laws, ordinances, or regulations will not impose any material environmental liability or (ii) the current environmental condition of the Hotels will not be affected by the condition of land or operations in the vicinity of the Hotels (such as the presence of underground storage tanks) or by third parties unrelated to Host REIT.

## CONFLICTS OF INTEREST

As discussed below, the operation of Host REIT involves various potential conflicts of interest.

RESPONSIBILITIES OF HOST REIT TO OTHER LIMITED PARTNERS IN THE OPERATING PARTNERSHIP

Maryland law imposes certain duties on the Board of Directors of Host REIT to its stockholders. In addition, after the REIT Conversion, Host REIT, as the sole general partner of the Operating Partnership, will have fiduciary obligations under Delaware law with respect to the other limited partners in the Operating Partnership (to the extent such duties have not been modified or eliminated pursuant to the terms of the Partnership Agreement). Although Delaware law provides that Host REIT, as general partner, is subject to the duties of care and loyalty with respect to the limited partners of the Operating Partnership, the Partnership Agreement imposes certain limitations on Host REIT's fiduciary obligations with respect to such limited partners. See "Distribution and Other Policies -- Conflicts of Interest Policies." Although some courts have interpreted the fiduciary duties of the Board of Directors in the same way as the duties of a general partner in a limited partnership, it is unclear whether, or to what extent, there are differences in such fiduciary duties. It is possible that, notwithstanding the contractual limitations in the Partnership Agreement, the fiduciary duties of the directors of Host REIT to the stockholders may be less than those of Host REIT, as general partner of the Operating Partnership, to the limited partners of the Operating Partnership. Thus, the discharge of Host REIT's obligations to its stockholders and to the limited partners in the Operating Partnership could result in decisions that may fail to reflect fully the interests of all holders of Host REIT Common Stock and limited partners of the Operating Partnership.

#### POTENTIAL CONFLICTS INVOLVING MARRIOTT INTERNATIONAL AND CRESTLINE

Marriott International currently serves as manager for all but 16 of Host's Hotels, and will continue to manage those Hotels pursuant to the Management Agreements that will be assigned to the Lessees. In addition, Marriott International acts as manager of hotels that will compete with Host REIT's Hotels. As a result, Marriott International may make decisions regarding competing lodging facilities which it manages that would not necessarily be in the best interests of Host REIT or the Lessees. Further, J.W. Marriott, Jr. and Richard E. Marriott, who are brothers, currently serve as directors of Host and directors (and, in the case of J.W. Marriott, Jr., also an officer) of Marriott International. After the REIT Conversion, J.W. Marriott, Jr. will serve as a director of Host REIT and will continue to serve as a director and officer of Marriott International, and Richard E. Marriott will serve as Chairman of the Board of Host REIT and continue to serve as a director of Marriott International. J.W. Marriott, Jr. and Richard E. Marriott also beneficially own (as determined for securities law purposes) approximately 10.6% and 10.2%, respectively, of the outstanding shares of common stock of Marriott International, and will beneficially own approximately 5.33% and 5.31%, respectively, of the outstanding shares of common stock of Crestline (but neither will serve as an officer or director thereof). As a result, J.W. Marriott, Jr. and Richard E. Marriott may have a potential conflict of interest with respect to their obligations as directors of Host REIT in connection with any decisions regarding Marriott International itself (including decisions relating to the Management Agreements involving the Hotels), Marriott International's management of competing lodging properties and Crestline's leasing and other businesses that would not necessarily be in the best interests of Host REIT.

# POLICIES WITH RESPECT TO CONFLICTS OF INTEREST

Host REIT has adopted certain policies and will enter into agreements with the Operating Partnership and others designed to minimize the adverse effects of these potential conflicts of interest. See "Distribution and Other Policies--Conflicts of Interest Policies" and "Business and Properties--Noncompetition Agreement." There can be no assurance, however, that the policies and agreements will be successful in eliminating the influence of such conflicts, and if they are not successful, decisions could be made at the Host REIT level that might not fully reflect fully the interests of the stockholders of Host REIT.

## **VOTING AND PROXIES**

This Proxy Statement/Prospectus is being furnished to holders of Host Common Stock on the Record Date in connection with the solicitation of proxies by Host's Board of Directors for use at the Special Meeting to be held at the Ritz-Carlton Hotel, 1700 Tysons Boulevard, McLean, Virginia, 22102, on December 15, 1998 at 10:00 a.m., local time, or at any adjournments or postponements thereof, for the purposes set forth herein and in the accompanying Notice of Special Meeting of Stockholders of Host.

This Proxy Statement/Prospectus and the accompanying proxy cards are being first mailed to Host's stockholders on or about November , 1998.

## MATTERS TO BE CONSIDERED AT THE SPECIAL MEETING

At the Special Meeting, stockholders of record as of the close of business on the Record Date will be asked to (i) consider and vote upon the Agreement which contemplates (a) the OP Contribution and (b) the Merger, and (ii) transact such other business as may properly come before the Special Meeting or any adjournments or postponements thereof.

## RECORD DATE AND OUTSTANDING SHARES

Only holders of record of Host Common Stock at the close of business on the Record Date are entitled to notice of and to vote at the Special Meeting. As of the Record Date, there were 204,954,447 shares of Host Common Stock and no shares of preferred stock of Host issued and outstanding and 47,845 holders of record. Each stockholder of Host is entitled to one vote for each share of Host Common Stock held as of the close of business on the Record Date.

## **VOTING OF PROXIES**

General Information. The proxy accompanying this Proxy Statement/Prospectus is solicited on behalf of Host's Board of Directors for use at the Special Meeting. All properly executed written proxy cards, and all properly completed proxies voted by telephone or the internet, which are delivered pursuant to this solicitation (and not later revoked) will be voted at the Special Meeting in accordance with the instructions given in the proxy. If no instructions are indicated, such proxies will be voted FOR approval of the Agreement at the Special Meeting. Host's Board of Directors currently is not aware of any matters other than the Agreement referred to herein which will come before the Special Meeting. If any other matter should be properly presented at the Special Meeting for action, the persons named in the accompanying proxy card will vote the proxy in their own discretion.

Voting by Written Proxy Card. Stockholders intending to vote by written proxy card are requested to complete, date and sign the accompanying proxy and promptly return it in the accompanying envelope or otherwise mail it to Host.

Voting by Telephone or the Internet. Instructions for voting by telephone or the internet are set forth on the enclosed admission ticket attached to the proxy card. The telephone and internet voting procedures are designed to authenticate votes cast by use of a personal identification number. The procedures, which comply with Delaware law, allow stockholders to appoint a proxy to vote their shares and to confirm that their instructions have been properly recorded.

## VOTE REQUIRED

Under the DGCL, the affirmative vote of a majority of the outstanding capital stock of the corporation entitled to vote thereon is needed to adopt a merger agreement. Similarly, if the OP Contribution were deemed to constitute a sale, lease or exchange of all or substantially all of the assets of Host under the DGCL, the affirmative vote of a majority of the outstanding capital stock of the corporation entitled to vote thereon would be needed to approve the OP Contribution. Pursuant to the Host Certificate, the affirmative vote of the holders of

two-thirds (66 2/3%) of the outstanding shares of Host Common Stock is necessary for approval of the Merger, as well as the OP Contribution if it were deemed to constitute a sale, lease or exchange of all or substantially all of the assets of Host.

On the Record Date, directors and executive officers of Host, together with their affiliates, as a group, beneficially owned approximately 13.7% of the issued and outstanding shares of Host Common Stock. It is currently expected that each director and executive officer of Host will vote the shares of Host Common Stock beneficially owned by such director or executive officer "FOR" approval of the Agreement and the transactions contemplated thereby. See "Principal Stockholders."

Under the DGCL, the stockholders of Host will not be entitled to dissenters' rights of appraisal as a result of the Merger or other Restructuring Transactions. See "The Restructuring Transactions--Absence of Dissenters' Rights."

QUORUM; ABSTENTIONS AND TREATMENT OF BROKER NON-VOTES; ADJOURNMENT; REVOCATION

The holders of a majority of the shares of Host Common Stock issued and outstanding and entitled to vote at the Special Meeting, present in person or by proxy, will constitute a quorum at the Special Meeting. Abstentions and broker non-votes will be counted for purposes of determining the presence of a quorum at the Special Meeting.

Votes cast in person or by proxy at the Special Meeting will be tabulated by the inspectors of election appointed for the Special Meeting, who will determine whether or not a quorum is present. Votes may be cast for, against or as abstentions. Broker/dealers who hold their customers' shares in street name may, under the applicable rules of the exchange and other self-regulatory organizations of which the broker/dealers are members, submit proxies for such shares and may vote such shares on routine matters, which, under such rules, typically include the election of directors. However, broker/dealers may not vote such shares on certain other matters, which typically include transactions related to mergers, without specific instructions from the customer who owns such shares. Properly executed proxies submitted by broker/dealers which have not been voted on certain matters as described in the previous sentence are referred to as broker non-votes. Because approval of the Agreement requires the affirmative vote of not less than two-thirds percent (66 2/3%) of the shares of Host Common Stock outstanding, abstentions and broker non-votes will have the same effect as votes AGAINST the Agreement.

In the event that a quorum is not present at the time the Special Meeting is convened, or if for any other reason Host believes that additional time should be allowed for the solicitation of proxies, Host may adjourn the Special Meeting with or without a vote of the stockholders. If Host proposes to adjourn the Special Meeting by a vote of the stockholders, the persons named in the enclosed proxy card will vote all shares of Host Common Stock for which they have voting authority in favor of such adjournment.

Each stockholder who signs and returns a proxy in the form enclosed with this Proxy Statement/Prospectus or properly submits their proxy by telephone or the internet may revoke it at any time prior to its exercise by giving notice of such revocation in writing to the Secretary of Host, by a later dated proxy either signed and returned by mail or by using the telephone or internet voting procedures or by voting in person at the Special Meeting. Unless so revoked, the shares of Host Common Stock represented by each such proxy will be voted at the meeting and any adjournment thereof. Presence at the meeting of a stockholder who has properly submitted a proxy but does not duly revoke it or request to vote in person does not revoke that proxy.

IF THE RESTRUCTURING TRANSACTIONS AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION DO NOT OCCUR IN TIME FOR HOST REIT TO ELECT REIT STATUS EFFECTIVE JANUARY 1, 1999, THE EFFECTIVENESS OF HOST REIT'S ELECTION COULD BE DELAYED TO JANUARY 1, 2000, WHICH WOULD RESULT IN HOST OR HOST REIT CONTINUING TO PAY SUBSTANTIAL CORPORATE-LEVEL INCOME TAXES IN 1999 AND COULD CAUSE THE BLACKSTONE ACQUISITION (WHICH IS CONDITIONED, AMONG OTHER THINGS, ON CONSUMMATION OF THE REIT CONVERSION BY MARCH 31, 1999 AND HOST REIT QUALIFYING AS A REIT FOR 1999) NOT TO BE CONSUMMATED. THEREFORE, IT IS EXTREMELY IMPORTANT THAT

STOCKHOLDERS RETURN THE PROXY CARDS AS SOON AS POSSIBLE OR PROPERLY SUBMIT THEIR PROXIES BY TELEPHONE OR THE INTERNET IN ACCORDANCE WITH THE INSTRUCTIONS ON THE ADMISSION TICKET ATTACHED TO THE ENCLOSED PROXY CARD. ANY DELAY IN RETURNING PROXIES COULD CAUSE THE SPECIAL MEETING TO BE DELAYED, WHICH COULD PREVENT THE RESTRUCTURING TRANSACTIONS AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION FROM BEING EFFECTIVE JANUARY 1, 1999.

## SOLICITATION OF PROXIES AND EXPENSES

The solicitation of proxies for the Special Meeting is being made by the Board of Directors of Host and will be done principally by mail. Host will bear the entire cost of solicitation of proxies from Host's stockholders and of preparing, assembling, printing and mailing this Proxy Statement/Prospectus, the proxy and any additional information furnished to Host's stockholders. Copies of solicitation materials will be furnished to brokerage houses, banks, fiduciaries and other custodians or nominees holding in their names shares of Host Common Stock beneficially owned by others to forward to such beneficial owners. Host may reimburse persons representing owners of shares of Host Common Stock for their expenses in forwarding solicitation materials to such beneficial owners. Original solicitation of proxies by mail may be supplemented by telephone, telegram or personal solicitation by directors, officers or other regular employees of Host. In addition, Host has engaged MacKenzie Partners, Inc. to furnish solicitation services on its behalf. No additional compensation will be paid to directors, officers or other regular employees for such services, but MacKenzie Partners, Inc. will be paid a fee for its services, estimated to be approximately \$5,500, plus reimbursement of its expenses.

HOST'S STOCKHOLDERS SHOULD NOT SEND ANY STOCK CERTIFICATES WITH THEIR PROXY CARDS.

THE BOARD OF DIRECTORS OF HOST HAS DETERMINED THE AGREEMENT, WHICH CONTEMPLATES THE RESTRUCTURING TRANSACTIONS, AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION TO BE ADVISABLE FOR HOST AND ITS STOCKHOLDERS. ACCORDINGLY, THE BOARD OF DIRECTORS HAS APPROVED THE AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY AND RECOMMENDS THAT ALL STOCKHOLDERS VOTE "FOR" APPROVAL OF THE AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY.

### THE RESTRUCTURING TRANSACTIONS

The following summary of the terms of the Restructuring Transactions is qualified in its entirety by reference to the Agreement, a copy of which is attached as Appendix A to this Proxy Statement/Prospectus and is incorporated by reference into this Proxy Statement/Prospectus. Stockholders of Host are urged to review the Agreement in its entirety.

BACKGROUND AND REASONS FOR THE RESTRUCTURING TRANSACTIONS AND THE REIT CONVERSION

Host is proposing the Restructuring Transactions in connection with a plan adopted by Host to restructure its business operations so that it will qualify as a REIT under the Code. If the Agreement is approved by Host stockholders at the Special Meeting and the Restructuring Transactions are consummated, Host REIT expects to qualify as a REIT beginning with its first full taxable year commencing after the REIT Conversion is completed, which currently is expected to be the year commencing January 1, 1999 (but which might not be until the year beginning January 1, 2000). The Host Board of Directors is proposing the Restructuring Transactions primarily for the following reasons:

- . The Restructuring Transactions are essential components of Host's conversion to REIT status for federal income tax purposes. In particular, the OP Contribution will enable Host REIT, following the Merger, to operate, together with the Operating Partnership, in an UPREIT structure, through which Host REIT would continue the full-service hotel ownership business currently conducted by Host. Host believes that the UPREIT structure will improve its ability to acquire additional properties in the future on favorable terms. Specifically, under certain circumstances, OP Units could be issued to acquire properties in transactions that would not trigger immediate tax obligations for certain sellers. Accordingly, converting to an UPREIT structure could enable Host REIT to acquire hotels and other properties in the future at lower prices because of the tax advantages to some sellers of receiving OP Units as consideration. OP Units would subsequently be redeemable for cash or common stock of Host REIT (at the option of Host REIT) at such time as the recipient desires liquidity.
- . The provisions of Maryland law have generally been viewed as favorable to REITs organized in corporate or trust form. As discussed below, Maryland law facilitates qualification as a REIT by authorizing the charter of a Maryland corporation to provide for restrictions on ownership and transferability designed to permit a corporation to qualify as a REIT under the Code or for any other purpose. In addition, unlike Delaware, Maryland does not impose a franchise tax on corporations, which will result in cost savings to Host in annual franchise tax payments and related fees of approximately \$150,000. Maryland's status as a jurisdiction favorable to REITs is evidenced by the large number of publicly-traded REITs that have chosen to operate as a regular Maryland corporation or as a special statutory Maryland real estate investment trust. According to NAREIT, as of September 1998, there were over 100 publicly-traded REITs organized under Maryland law.
- . In order to satisfy certain requirements that are applicable to REITs in general, many REITs impose ownership limits and transfer restrictions, similar to the Ownership Limit under the Host REIT Charter, by inclusion of such provisions in their charters. Under Delaware law, such restrictions would not be binding with respect to securities issued prior to adoption of the restriction unless holders of such securities agree to or vote in favor of such restriction. However, under Maryland law and by reason of the Merger, all shares of Host REIT Common Stock issued in the Merger and thereafter would be subject to the Ownership Limit, for which authority exists under Maryland law.
- . Host's principal executive offices and a substantial number of Host's employees are employed in Maryland.

The Board of Directors of Host believes that the Restructuring Transactions and the other transactions comprising the REIT Conversion are advisable for Host and its stockholders based on the belief that:

. The REIT structure, as a more tax efficient structure, will provide improved operating results through changing economic conditions and all phases of the hotel economic cycle.

- . The REIT Conversion, which will reduce corporate-level taxes and the need to incur debt to reduce corporate-level taxes through interest deductions, will improve its financial flexibility and allow it to continue to strengthen its balance sheet by reducing its overall debt to equity ratio over time.
- . As a REIT, Host will be able to compete more effectively with other public lodging real estate companies that already are organized as REITs and to make performance comparisons with its peers more meaningful.
- . By becoming a dividend paying company, Host's stockholder base will expand to include investors attracted by yield as well as asset quality, which is expected to facilitate Host REIT's capital-raising efforts and provide a less volatile stockholder base.
- . The adoption of the UPREIT structure will facilitate tax-deferred acquisition of additional hotels (such as in the case of the Blackstone Acquisition and the Partnership Mergers).

Host believes that these benefits justify the REIT Conversion even if the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999 (in which event the effectiveness of Host's REIT election could be delayed until January 1, 2000).

Host explored the possibility of engaging in a business combination with a so-called "paired share" REIT, Santa Anita, in December 1996 and January 1997. Based upon an analysis of potential costs, the pricing of the transaction, the time requirement to complete such a transaction and the possible legislative risks associated with the "paired share" structure, Host decided not to pursue such a transaction.

During the fourth quarter of 1997, Host began to explore internally the possibility of reorganizing as a REIT on a stand-alone basis. Host analyzed the various consents that would need to be obtained and other requirements that would need to be met in order to restructure its assets and operations (including its indirect interests in the Partnerships and the Private Partnerships) in order to qualify as a REIT. In light of the importance of its relationship with Marriott International as the manager of substantially all of Host's Hotels, Host began preliminary discussions with Marriott International during January 1998 in order to ascertain whether or not Marriott International would cooperate in Host's potential conversion to a REIT. In February 1998, Host tentatively concluded that it would be desirable for the Operating Partnership to use OP Units to acquire the Partnerships and the Private Partnerships, subject to determination of satisfactory terms and conditions for such acquisitions. In order to determine the feasibility of this approach, Host commenced preliminary discussions in February and March 1998 with the outside partners of certain Private Partnerships to determine whether or not they would have an interest in such a transaction. In March 1998, Host also entered into discussions with the Blackstone Group regarding the potential acquisition of the twelve full-service hotels and certain other assets owned by the Blackstone Entities because Host believed that these hotels represented the premier hotel portfolio on the market and an acquisition would be consistent with its desire to pursue a strategy of owning both Marriott and other upscale and luxury hotel brands. In April 1998, Host and the Operating Partnership entered into agreements with the Blackstone Entities for the Blackstone Acquisition and concurrently reached a decision and publicly announced that it would be advantageous if Host were to convert to a REIT. In May and June 1998, Host and the Operating Partnership entered into agreements to acquire the interests of certain outside partners in four Private Partnerships. Each of these transactions is contingent upon the REIT Conversion. On June 2, 1998, the Operating Partnership filed a registration statement for the OP Units and Notes to be issued to the Limited Partners of the Partnerships in the Partnership Mergers. On October 9, 1998, the Operating Partnership commenced the offering of such OP Units and Notes if the Partnership Mergers are approved and Host REIT commenced its offer to exchange Host REIT Common Stock for OP Units to be issued in the Partnership Mergers.

The Host Board believes that Host's conversion to a REIT and the distribution of Crestline shares as part of the Initial E&P Distribution will benefit Host's stockholders by providing them with a tax advantaged REIT security that is expected to provide both the opportunity for regular cash dividends and capital appreciation as Host REIT acquires additional properties, as well as a continuing interest in Crestline, Host's senior living company and the initial lessee of substantially all of Host REIT's Hotels, if a Host stockholder continues to hold the Crestline common stock. If Host REIT qualifies for taxation as a REIT, it generally will not be subject to

federal corporate income taxes on that portion of its ordinary income or capital gain that is distributed to stockholders. As a REIT, Host REIT would be able to benefit from the tax advantages that apply to REIT's, and stockholders will receive quarterly distributions that are at least sufficient to satisfy the annual distribution requirements applicable to REITs under the Code. The Host Board believes that this will highlight the value of Host REIT's hotel properties and permit stockholders to realize a regular cash return on that value. Upon completion of the REIT Conversion, Crestline is expected to own Host's 31 senior living communities, which will continue to be managed by Marriott International, and a 25% interest in the Swissotel management company expected to be acquired in the Blackstone Acquisition, and will lease substantially all of the Hotels owned by Host REIT and its affiliates. At such time, Crestline will operate independently of Host, will be publicly traded and separately listed on the NYSE, and will pursue its own growth opportunities.

Host's Board of Directors has received an opinion dated November , 1998 from BT Wolfensohn, which acted as financial advisor to Host in connection with the REIT Conversion, to the effect that the REIT Transactions (as defined in such opinion), taken together, are fair from a financial point of view to the holders of Host Common Stock. See "Opinion of Financial Advisor." The Host Board of Directors considered such opinion as well as other matters it deemed relevant in determining the advisability of the Agreement and the other transactions comprising the REIT Conversion.

Following the Restructuring Transactions and the other transactions comprising the REIT Conversion (including the Blackstone Acquisition), Host REIT and its subsidiaries are expected initially to own outright, or have controlling interests in, approximately 125 full-service hotels operated primarily under the Marriott, Ritz-Carlton, Four Seasons, Swissotel and Hyatt brand names. Upon completion of the REIT Conversion, Host REIT will be the sole general partner of the Operating Partnership and is expected to own approximately 74% of the OP Units in the Operating Partnership, and the remaining OP Units, which are intended to be substantially equivalent on an economic basis to shares of Host REIT Common Stock, will be owned by the Limited Partners, unaffiliated partners of the four Private Partnerships participating in the Private Partnership Transactions and the Blackstone Entities. Host REIT and its subsidiaries will own a number of OP Units equal to the number of shares of Host REIT Common Stock outstanding. Host REIT will be managed by its Board of Directors and will have no employees who are not also employees of the Operating Partnership.

THE BOARD OF DIRECTORS OF HOST BELIEVES THAT THE AGREEMENT, WHICH CONTEMPLATES THE RESTRUCTURING TRANSACTIONS AND THE OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION DESCRIBED IN THIS PROXY STATEMENT/PROSPECTUS, ARE ADVISABLE FOR HOST AND ITS STOCKHOLDERS AND RECOMMENDS THAT STOCKHOLDERS VOTE "FOR" APPROVAL OF THE AGREEMENT AND THE TRANSACTIONS CONTEMPLATED THEREBY. In the event the Agreement is not approved by Host stockholders at the Special Meeting, Host will continue to operate as a Delaware corporation, and the other transactions comprising the REIT Conversion (including the Initial E&P Distribution) will not be completed at this time.

## THE OP CONTRIBUTION

Subject to approval of the Agreement at the Special Meeting, after the Special Meeting Host will contribute, as the OP Contribution, its wholly-owned full-service hotels, interests in certain of the Partnerships and the Private Partnerships (other than their interests in the general partners of the Partnerships, each of which is Host or a wholly-owned direct or indirect subsidiary of Host, and certain other subsidiaries of Host who will remain in existence as subsidiaries of Host REIT and will receive OP Units in the Partnership Mergers) and certain other assets (excluding Host's senior living assets and the cash or other consideration to be distributed in connection with the Initial E&P Distribution and certain de minimis assets that cannot be contributed to the Operating Partnership) to the Operating Partnership. As a preliminary step in the REIT Conversion, during 1998, various subsidiaries of Host also have contributed or will contribute the wholly-owned full-service hotels held by such entities, their interests in certain of the Partnerships and Private Partnerships and certain other assets to the Operating Partnership (including its subsidiaries). In exchange for these contributions, the Operating Partnership (i) will issue to Host in the aggregate a number of OP Units equal to the number of outstanding shares of Host Common Stock at the time of the REIT Conversion, reduced by the number of OP Units to be received by the general partners of the Partnerships and other subsidiaries of Host in the Partnership Mergers (together with

56

preferred partnership interests in the Operating Partnership corresponding to any shares of Host preferred stock outstanding at the time of the REIT Conversion) and (ii) will assume all liabilities of Host (including past and future contingent liabilities and liabilities for the Plans in accordance with the 1998 Employee Benefits Allocation Agreement), other than liabilities of Crestline. A portion of Host REIT's OP Units represents a 0.1% general partnership interest in the Operating Partnership. Following these contributions, the Operating Partnership and its subsidiaries will directly or indirectly own all of Host's and its subsidiaries' wholly-owned hotels, substantially all of Host's direct and indirect interests in the Partnerships and the Private Partnerships and all of Host's other assets (excluding its senior living assets and the cash or other consideration to be distributed in connection with the Initial E&P Distribution, and certain de minimis assets that cannot be contributed to the Operating Partnership).

The above-described contributions, which include the OP Contribution, will enable Host REIT, following the Reincorporation by means of the Merger, to operate together with the Operating Partnership in an UPREIT structure, which is expected to improve Host REIT's ability to acquire additional hotels and other properties in the future on favorable terms.

### TERMS OF THE MERGER

The Reincorporation will be effected through the merger of Host with and into Host REIT. As a result of the Merger, Host REIT will be the surviving entity and the separate existence of Host will terminate. At the Effective Time, each outstanding share of Host Common Stock, together with the associated Host Right, will be converted into one share of Host REIT Common Stock, together with the associated Host REIT Right, and holders thereof will become subject to the Ownership Limit under the Host REIT Charter. In addition, rights and obligations of Host under the Plans and related and other agreements will be assumed by Host REIT, and all rights of the participants therein to acquire shares of Host Common Stock on the terms and conditions of the Plans and such agreements will be converted into rights to acquire shares of Host REIT Common Stock in accordance with the 1998 Employee Benefits Allocation Agreement (other than with respect to individuals who will be employed by Crestline following the Initial E&P Distribution). Any outstanding shares of preferred stock of Host would be converted into an equal number of shares of preferred stock of Host REIT having substantially the same rights and preferences. Certain rights and obligations of Host under the Plans and related agreements to acquire shares of Host Common Stock, to the extent held by participants who will be employed by Crestline following the Initial E&P Distribution, will be assumed by Crestline and converted into rights to acquire shares of Crestline common stock in accordance with the 1998 Employee Benefits Allocation Agreement. See "Description of Host REIT Capital Stock" and "Management--1998 Employee Benefits Allocation Agreement."

As a consequence of the Merger, among other things, the assets of Host will become assets of Host REIT without further action, and Host REIT will become liable for all the debts and obligations of Host. As soon as practicable following the Merger, Host REIT will complete any other transactions comprising the REIT Conversion that have not already been completed. See "The REIT Conversion."

Following the Merger, each person who was a director of Host will continue as a director of Host REIT and certain of the officers of Host will remain officers of Host REIT (others will have become officers of Crestline in connection with the distribution of Crestline shares as part of the Initial E&P Distribution). Certain information about each person who currently is a member of the Board of Directors and an executive officer of Host is set forth herein under the caption "Management--Directors and Executive Officers of Host REIT." In addition, upon consummation of the Merger, Host REIT and its stockholders will be governed by the MGCL and by the Host REIT Charter and Host REIT Bylaws, copies of which are attached as Exhibits B and C, respectively, to the Agreement, which is attached to this Proxy Statement/Prospectus as Appendix A. See "--Comparison of Rights of Stockholders of Host and Host REIT."

PRIMARILY TO SATISFY CERTAIN REQUIREMENTS UNDER THE CODE THAT ARE APPLICABLE TO REITS IN GENERAL, THE OWNERSHIP LIMIT WILL PROVIDE THAT NO PERSON OR PERSONS ACTING AS A GROUP MAY OWN, OR BE DEEMED TO OWN BY VIRTUE OF THE ATTRIBUTION PROVISIONS OF THE CODE, MORE THAN (I) 9.8% OF THE LESSER OF THE NUMBER OR VALUE OF SHARES OF HOST REIT COMMON STOCK OUTSTANDING OR (II) 9.8% OF THE LESSER OF THE NUMBER OR VALUE OF

THE ISSUED AND OUTSTANDING SHARES OF ANY CLASS OR SERIES OF HOST REIT PREFERRED STOCK, SUBJECT TO WAIVER OR MODIFICATION BY HOST REIT IN CERTAIN LIMITED CIRCUMSTANCES AND SUBJECT TO (A) AN EXCEPTION FOR A HOLDER OF SHARES OF HOST REIT COMMON STOCK SOLELY BY REASON OF THE MERGER IN EXCESS OF THE OWNERSHIP LIMIT SO LONG AS SUCH HOLDER WOULD NOT OWN, DIRECTLY OR BY ATTRIBUTION UNDER THE CODE, MORE THAN 9.9% BY VALUE OF THE OUTSTANDING CAPITAL STOCK OF HOST REIT AS OF THE SPECIAL MERGER OWNERSHIP LIMIT EFFECTIVE TIME, AND (B) A LIMITATION ON THE APPLICATION OF THE "GROUP" LIMITATION (BUT NO OTHER ELEMENT OF THE OWNERSHIP LIMIT) TO ANY "GROUP" THAT OTHERWISE WOULD EXCEED THE OWNERSHIP LIMIT AT THE EFFECTIVE TIME SOLELY BY REASON OF ITS STATUS AS A "GROUP." THE OWNERSHIP LIMIT WILL BECOME APPLICABLE TO ALL HOST REIT STOCKHOLDERS AS OF THE EFFECTIVE TIME. ANY SHARES OF HOST REIT COMMON STOCK HELD AT THE EFFECTIVE TIME AND NOT SUBSEQUENTLY TRANSFERRED PRIOR TO THE SPECIAL MERGER OWNERSHIP LIMIT EFFECTIVE TIME, OR ACQUIRED OR OTHERWISE HELD AT ANY TIME AFTER THE EFFECTIVE TIME IN VIOLATION OF THE OWNERSHIP LIMIT WILL BE TRANSFERRED AUTOMATICALLY TO A TRUST FOR THE BENEFIT OF A DESIGNATED CHARITABLE BENEFICIARY, AND THE PERSON WHO ACQUIRED (OR HELD) SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL NOT BE ENTITLED TO ANY DISTRIBUTIONS THEREON OR TO VOTE SUCH EXCESS SHARES OF HOST REIT COMMON STOCK. THE HOLDER OF ANY SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL RECEIVE THE LESSER OF THE VALUE OF SUCH EXCESS SHARES AS OF THE EFFECTIVE TIME OR THE CASH PROCEEDS FROM THE SALE OF SUCH EXCESS SHARES OF HOST REIT COMMON STOCK BY THE TRUSTEE OF THE TRUST. AFTER THE EFFECTIVE TIME, ANY PERSON WHO ACQUIRES HOST REIT COMMON STOCK IN EXCESS OF THE OWNERSHIP LIMIT WILL NOT RECEIVE ANY PROCEEDS FROM THE SUBSEQUENT SALE THEREOF IN EXCESS OF THE LESSER OF THE PRICE PAID THEREFOR OR THE AMOUNT REALIZED FROM SUCH SALE. A TRANSFER OF HOST REIT COMMON STOCK TO A PERSON WHO, AS A RESULT OF THE TRANSFER, VIOLATES THE OWNERSHIP LIMIT MAY BE VOID UNDER CERTAIN CIRCUMSTANCES, AND, IN ANY EVENT, WOULD DENY THE TRANSFEREE ANY OF THE ECONOMIC BENEFITS OF OWNING SHARES OF HOST REIT COMMON STOCK IN EXCESS OF THE OWNERSHIP LIMIT. TO AVOID THE ADVERSE EFFECTS OF THE OWNERSHIP LIMIT, ANY HOLDER OF HOST COMMON STOCK WHO WOULD OWN SHARES IN EXCESS OF THE OWNERSHIP LIMIT AT THE EFFECTIVE TIME SHOULD DISPOSE OF ANY SUCH EXCESS SHARES PRIOR THERETO.

Host anticipates that the Merger will become effective as promptly as practicable following stockholder approval of the Agreement at the Special Meeting and satisfaction or waiver of the other conditions to the Merger. The Agreement provides that the Merger may be abandoned by Host or Host REIT at any time prior to its effectiveness. Host has no current intention of abandoning or causing Host REIT to abandon the Merger subsequent to the Special Meeting if stockholder approval is obtained and the other conditions to the Merger are satisfied or waived. See "--Conditions to the Merger."

The Board of Directors of Host has approved the Agreement, and the transactions contemplated thereby, subject to stockholder approval. Assuming the stockholders of Host approve the Agreement at the Special Meeting and the other conditions to the Merger are satisfied or waived, the Merger will become effective at the time the Articles of Merger are accepted for filing by the SDAT in accordance with the MGCL and the time the Certificate of Merger is accepted for filing by the Secretary of State of Delaware in accordance with the DGCL, or later if so specified in such Articles and Certificate. It is expected that the listing of Host REIT Common Stock on the NYSE will occur at or prior to the Effective Time and that the listing of Host Common Stock on the NYSE will be terminated at the Effective Time.

At the Effective Time, each certificate representing shares of Host Common Stock will be deemed for all purposes to evidence the same number of shares of Host REIT Common Stock. As soon as reasonably practicable after the Effective Time, Host REIT will instruct First Chicago Trust Company of New York, the transfer agent and registrar for Host REIT Common Stock and the exchange agent for purposes of the Merger (the "Transfer and Exchange Agent"), to mail a letter of transmittal and instructions to each holder of a certificate or certificates which immediately prior to the Effective Time represented outstanding shares of Host Common Stock ("Certificates"), advising the securityholder of the procedure for surrendering the Certificates in exchange for certificates representing shares of Host REIT Common Stock. Upon surrender of the Certificates for cancellation to the Transfer and Exchange Agent, together with a duly executed letter of transmittal and such other documents as may be reasonably required by the Transfer and Exchange Agent or the Agreement, the holder of the Certificates will be entitled to receive in exchange therefor a certificate evidencing that number of shares of Host

58

REIT Common Stock which such holder has the right to receive in respect of the rights formerly evidenced by such Certificates. CERTIFICATES SHOULD NOT BE SURRENDERED UNTIL THE LETTER OF TRANSMITTAL IS RECEIVED.

Neither Host nor Host REIT is aware of any federal, state or local regulatory requirements that must be complied with or approvals that must be obtained prior to consummation of the Merger pursuant to the Agreement, other than compliance with applicable federal and state securities laws, the filing and acceptance for record of the Articles of Merger by the SDAT as required under the MGCL and the filing of a Certificate of Merger as required under the DGCL and the receipt of various state and local governmental authorizations.

### ABSENCE OF DISSENTERS' RIGHTS

Pursuant to Section 262(b)(1) of the DGCL, the stockholders of Host will not be entitled to appraisal rights as a result of the Merger or other Restructuring Transactions.

#### ACCOUNTING TREATMENT

The OP Contribution and the Merger will be accounted for as a transfer or exchange between enterprises under common control. Therefore, the assets and liabilities so transferred will be accounted for at historical cost in a manner similar to that in pooling-of-interests accounting.

#### CONDITIONS TO THE MERGER

Consummation of the Merger is subject to the satisfaction or waiver of various conditions, including satisfaction of the following:

- . Host Stockholder Approval. The affirmative vote of the holders of Host Common Stock representing not less than two-thirds (66 2/3%) of the outstanding shares of Host Common Stock shall have approved the Agreement, which shall be deemed to be approval of the Merger for purposes of Section 251 of the DGCL (applicable to mergers), as well as approval of the OP Contribution for purposes of Section 271 of the DGCL (applicable to the sale, lease or exchange of all or substantially all of the assets of a corporation) if such OP Contribution were deemed to constitute a sale, lease or exchange of all or substantially all of the assets of Host.
- . REIT Qualification. Host's Board of Directors shall have determined, (i) that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur, and (ii) based upon the advice of counsel, that Host REIT can elect to be treated as a REIT for federal income tax purposes effective no later than the first full taxable year commencing after the REIT Conversion is completed (which might not be until the year commencing January 1, 2000 if the REIT Conversion is not completed prior to January 1, 1999).
- . NYSE Listing. The Host REIT Common Stock shall have been approved for listing on the NYSE.
- . Governmental and Third-Party Consents. Host shall have received all governmental and third-party consents to the Restructuring Transactions, including consents of lenders and Marriott International, except for consents as would not reasonably be expected to have a material adverse effect on the business, financial condition or results of operations of Host REIT, the Operating Partnership and their subsidiaries taken as a whole.
- . No Adverse Tax Legislation. The United States Congress shall not have enacted legislation, or proposed legislation with a reasonable possibility of being enacted, that would have the effect of (i) substantially impairing the ability of Host REIT to qualify as a REIT or the Operating Partnership to qualify as a partnership, (ii) substantially increasing the federal tax liabilities of Host REIT resulting from the REIT Conversion or (iii) substantially reducing the expected benefits to Host REIT resulting from the REIT Conversion. The determination that this condition has been satisfied will be made by Host, in its discretion.

In addition, Host will amend the Host Rights Agreement to provide that each Host Right issuable pursuant to the Host Rights Agreement will be converted into a Host REIT Right issuable under the Host REIT Rights Agreement. Host REIT intends to adopt the Host REIT Rights Agreement prior to the completion of the Merger and each share of Host REIT Common Stock issued in the Merger will have a Host REIT Right attached to it. See "Certain Provisions of Maryland Law and the Host REIT Charter and Bylaws--Stockholder Rights Plan."

### COMPARISON OF RIGHTS OF STOCKHOLDERS OF HOST AND HOST REIT

The rights of stockholders of Host are currently governed by the DGCL, the Host Certificate, the Host Bylaws and the Host Rights Agreement. If the Agreement is approved by Host's stockholders and the Merger is consummated, Host REIT will be the surviving entity in the Merger and the rights of the stockholders of Host REIT will be governed by the MGCL, the Host REIT Charter, the Host REIT Bylaws and Host REIT's Stockholder Rights Plan. The following discussion compares certain of the existing rights of stockholders of Host with those of stockholders of Host REIT.

## FORM OF ORGANIZATION AND PURPOSE

Host. Host is a Delaware corporation. Pursuant to the Host Certificate, Host is authorized to engage in any lawful acts or activities for which corporations may be organized under the DGCL, including conducting a general hotel business.

Host REIT. Host REIT is a Maryland corporation and will be the sole general partner of the Operating Partnership. Host REIT will make an election to be taxed as a REIT under the Code and intends to maintain its qualification as a REIT. Host REIT's only significant asset will be its interest in the Operating Partnership and consequently an indirect investment in the hotels owned by the Operating Partnership and its subsidiaries.

#### CAPTTAL TZATTON

Host. The Host Certificate authorizes a total of 601,000,000 shares of stock consisting of 600,000,000 shares of Host Common Stock, par value \$1.00 per share, and 1,000,000 shares of Host preferred stock without par value ("Host Preferred Stock"). Of the Host Preferred Stock, 300,000 shares are designated as Series A Junior Participating Preferred Stock without par value, and 4,000 shares are designated as Series A Cumulative Convertible Preferred Stock without par value. At November 13, 1998, Host had 204,954,447 shares of Host Common Stock issued and outstanding and no shares of Host Preferred Stock issued and outstanding.

Host REIT. The Host REIT Charter provides that the total number of shares of stock of all classes which Host REIT has authority to issue is 800,000,000 shares of stock, initially consisting of 750,000,000 shares of Host REIT Common Stock, par value \$.01 per share, and 50,000,000 shares of Host REIT Preferred Stock, par value \$.01 per share. The Board of Directors is authorized, without a vote of stockholders, to classify or reclassify any unissued shares of stock and to establish the preferences and rights of any preferred or other class or series of stock to be issued. At November 13, 1998, 100 shares of Host REIT Common Stock were issued and outstanding.

## PREEMPTIVE RIGHTS

DGCL. Under the DGCL, stockholders of a corporation have only such preemptive rights as may be provided in the corporation's certificate of incorporation.

Host. The Host Certificate does not provide for preemptive rights.

MGCL. Under the MGCL, stockholders of a Maryland corporation have only such preemptive rights as may be provided in the Maryland corporation's charter.

Host REIT. The Host REIT Charter will not provide for preemptive rights.

DGCL. Under the DGCL, a written restriction on the transfer of a security, if permitted by the DGCL and noted conspicuously on the certificate representing the security or, in the case of uncertificated shares, contained in the notice required to be sent to the security holder pursuant to the DGCL, may be enforced against the holder or any successor or transferee of the holder. A restriction on the transfer of securities of a corporation may be imposed either by the certificate of incorporation or by the by-laws or by an agreement among any number of security holders or among such holders and the corporation. No restriction so imposed is binding with respect to securities issued prior to adoption of the restriction unless the holders of the securities are parties to an agreement or voted in favor of the restriction. A restriction on the transfer of securities of a corporation is permitted under the DGCL if, among other things, it prohibits the transfer of the restricted securities to designated persons or classes of persons, and such designation is not manifestly unreasonable. Any other lawful restriction on the transfer of securities also is permitted under the DGCL. The DGCL expressly provides that any restriction on the transfer of shares imposed for the purpose of maintaining a tax advantage to the corporation is conclusively presumed to be for a reasonable purpose.

Host. Neither the Host Certificate nor the Host Bylaws will provide for restrictions on the transfer of Host securities.

MGCL. Under the MGCL, a Maryland corporation may impose a restriction on the transfer of its stock in the corporation's charter, articles of incorporation or bylaws. If a Maryland corporation which issues stock imposes a restriction on its transferability, the stock certificate must (i) contain a full statement of the restriction, or (ii) state that the corporation will furnish information about the restriction to the stockholder on request and without charge. The MGCL expressly authorizes the charter of a Maryland corporation to provide for restrictions on transferability designed to permit a corporation to qualify as a REIT under the Code or for any other purpose.

Host REIT. The Ownership Limit under the Host REIT Charter will provide that, subject to certain exceptions, no person or persons acting as a group may own, or be deemed to own by virtue of the attribution provisions of the Code, more than (i) 9.8% of the lesser of the number or value of shares of Host REIT Common Stock outstanding or (ii) 9.8% of the lesser of the number or value of the issued and outstanding preferred or other stock of any class or series of Host REIT (subject to a limited exception for a holder of shares of Host REIT Common Stock solely by reason of the Merger in excess of the Ownership Limit so long as the holder thereof would not own, directly or by attribution under the Code, more than 9.9% in value of the outstanding shares of capital stock of Host REIT as of the Special Merger Ownership Limit Effective Time, and to a limitation on the application of the "group" limitation). The Host REIT Charter will further prohibit (a) any person from actually or constructively owning shares of the capital stock of Host REIT that would result in Host REIT being "closely held" under Section 856(h) of the Code or otherwise cause Host REIT to fail to qualify as a REIT and (b) any person from transferring shares of Host REIT Common Stock or Host REIT Preferred Stock if such transfer would result in all classes and series of the capital stock of Host REIT being owned by fewer than 100 persons. See "Description of Host REIT Capital Stock--Restrictions on Ownership and Transfer.'

## AMENDMENT OF HOST CERTIFICATE/HOST REIT CHARTER

DGCL. Under the DGCL, an amendment to a corporation's certificate of incorporation generally requires the approval of the board of directors and the approval of a majority of the outstanding stock entitled to vote thereon and a majority of the outstanding stock of each class entitled to vote thereon. Under the DGCL, the holders of the outstanding shares of a class are entitled to vote as a separate class on a proposed amendment that would increase or decrease the aggregate number of authorized shares of such class, increase or decrease the par value of the shares of such class or alter or change the powers, preferences or special rights of the shares of such class so as to affect them adversely. If any proposed amendment would alter or change the powers, preferences or special rights of one or more series of any class so as to affect them adversely, but would not so affect the entire class, then only the shares of the series so affected by the amendment will be considered a separate class for purposes of voting by classes.

Host. The Host Certificate provides that, unless otherwise specified by the Host Certificate, Host reserves the right to amend, alter, change or repeal any provision contained in the Host Certificate in the manner prescribed by the DGCL.

The Host Certificate provides further that the affirmative vote of at least 66 2/3% of the voting power of all the shares of the corporation entitled to vote generally in the election of directors ("Voting Stock") is required to alter, amend or repeal, or to adopt any provisions inconsistent with certain provisions in the Host Certificate relating to, among other things: the number and election of Host Directors, the removal of Host Directors, vacancies on the Host Board, the voting requirements for specified business combinations, stockholder action by written consent, special stockholder meetings, the amendment of the Host Bylaws, and the voting requirements for certain "interested stockholder" transactions. See "--Number and Election of Directors," "--Removal of Directors," "--Vacancies on the Board of Directors," "--Changes in Control Pursuant to Delaware/Maryland Law," "--Transactions Involving Directors or Officers," "--Stockholder Action by Written Consent," "--Special Stockholder Meetings," and "--Amendment of Bylaws."

MGCL. Under the MGCL, in order to amend the charter, the board of directors first must adopt a resolution setting forth the proposed amendment and declaring its advisability and direct that the proposed amendment be submitted to stockholders for their consideration either at an annual or special meeting of stockholders. Thereafter, the proposed amendment must be approved by stockholders by the affirmative vote of two-thirds of all the votes entitled to be cast on the matter, unless a greater or lesser proportion of votes (but not less than a majority of all votes entitled to be cast) is specified in the charter.

Host REIT. The provisions to be contained in the Host REIT Charter relating to restrictions on transferability of Host REIT Common Stock, preemptive rights, the classified Board and fixing the size of the Board within the range set forth in the Host REIT Charter, as well as the provisions relating to removal of directors, the filling of Board vacancies, liability and indemnification of directors and officers, the calling of special stockholder meetings, the authority of the Board of Directors to issue stock, amendment of the Host REIT Charter and the exclusive authority of the Board to amend the Bylaws may be amended only by a resolution adopted by the Board of Directors and approved at an annual or special meeting of the stockholders by the affirmative vote of the holders of not less than two-thirds of the votes entitled to be cast on the matter. Other amendments to the Host REIT Charter generally may be effected by requisite action of the Board of Directors and approval by stockholders by the affirmative vote of not less than a majority of the votes entitled to be cast on the matter.

## AMENDMENT OF BYLAWS

DGCL. Under the DGCL, the adoption of, amendment to, or repeal of, a corporation's bylaws requires the approval of the stockholders, unless the certificate of incorporation confers the power to amend the bylaws upon the board of directors. Even if the power to adopt, amend or repeal the bylaws is conferred upon the board of directors, the stockholders may not be divested of the power to adopt, amend or repeal the bylaws.

Host. Subject to the exceptions described below, the Host Certificate provides that the Host Board of Directors is authorized to make, alter, amend and repeal the Host Bylaws (except insofar as the Host Bylaws adopted by the shareholders shall otherwise provide). Any bylaws made by the Board of Directors may be altered, amended or repealed by the directors or by the stockholders. The Host Bylaws provide that, subject to the exceptions described below, the Host Bylaws may be altered, amended or repealed (a) by a majority vote of the shares represented and entitled to vote at any regular meeting of the stockholders (or at any special meeting called for that purpose), or (b) subject to Delaware law, by a majority vote of those directors present at any meeting at which a quorum of the Board of Directors is present.

The Host Certificate and/or the Host Bylaws provides that certain provisions in the Host Bylaws may not be altered, amended or repealed, and no provision inconsistent therewith adopted, without the affirmative vote of the holders of at least 66 2/3% of the Voting Stock, including provisions relating to, among other things: the number and election of Host directors; vacancies on the Board of Directors and the removal of directors from

the Host Board; director nominations; notice to directors and stockholders; bylaw amendments; and the prohibition on stockholder action by written consent. See "--Number and Election of Directors," "--Vacancies on the Board of Directors," "--Removal of Directors" and "--Stockholder Action by Written Consent."

MGCL. Under the MGCL, the adoption of, an amendment to, or the repeal of, a Maryland corporation's bylaws requires the approval of the stockholders, unless the charter or bylaws confer the power to amend the bylaws upon the board of directors.

Host REIT. As permitted under the MGCL, the Host REIT Charter and Bylaws will provide that directors have the exclusive right to adopt, amend or repeal the Host REIT Bylaws.

### STOCKHOLDER VOTING RIGHTS GENERALLY

DGCL. Under the DGCL, unless otherwise provided in the certificate of incorporation and subject to certain provisions of the DGCL, each stockholder is entitled to one vote for each share of capital stock held by such stockholder. Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to corporate action in writing without a meeting may authorize another person or persons to act for him by proxy, but no such proxy shall be voted or acted upon after three years from its date, unless the proxy provides for a longer period. The DGCL further provides that in all matters other than the election of directors, the affirmative vote of the majority of shares present in person or represented by proxy at a duly held meeting at which a quorum is present and entitled to vote on the subject matter is deemed to be the act of the stockholders, unless the DGCL, the certificate of incorporation or the bylaws specify a different voting requirement. Where a separate vote by a class or classes is required, a majority of the outstanding shares of such class or classes, present in person or represented by proxy, shall constitute a quorum entitled to take action with respect to that vote on that matter and the affirmative vote of the majority of shares of such class or classes present in person or represented by proxy at the meeting shall be the act of such class.

Host. The Host Bylaws provide that in all matters, when a quorum is present at any meeting, the vote of the holders of a majority of the stock having voting power present in person or represented by proxy shall decide any question brought before such meeting, unless the question is one upon which by express provision of the statutes or of the Host Certificate, a different vote is required in which case such express provision shall govern and control the decision of such question. The Host Bylaws also provide that each stockholder shall at every meeting of the stockholders be entitled to one vote in person or by proxy for each share of the capital stock having voting power held by such stockholder, but no proxy shall be voted on after three years from its date, unless the proxy provides for a longer period. Each share of Host Common Stock has one vote and Host's Certificate of Incorporation permits the Board of Directors to classify and issue shares of capital stock in one or more series having voting power which may differ from that of the Common Stock.

MGCL. Under the MGCL, unless the charter provides for a greater or lesser number of votes per share or limits or denies voting rights, each outstanding share of stock, regardless of class, is entitled to one vote on each matter submitted to a vote at a meeting of stockholders. A stockholder may vote the stock the stockholder owns either in person or by proxy, which proxy is not valid more than eleven months after its date, unless such proxy provides otherwise. Unless the MGCL or charter specify a different voting requirement, a majority of all the votes cast at a duly held meeting at which a quorum is present and entitled to vote on the subject matter is deemed to be the act of the stockholders. Additionally, unless the MGCL or charter provide otherwise, if two or more classes of stock are entitled to vote separately on any matter for which the MGCL requires approval by two-thirds of all the votes entitled to be cast, the matter must be approved by two-thirds of all the votes of each class.

Host REIT. Each share of Host REIT Common Stock will have one vote and Host REIT's Charter will permit the Board of Directors to classify and issue shares of capital stock in one or more classes or series having voting power which may differ from that of the shares of Host REIT Common Stock. The Host REIT Bylaws provide that, except with respect to the election of directors, and unless otherwise provided in the Host REIT

Charter, a majority of the votes cast at a meeting of stockholders duly called and at which a quorum is present is sufficient to approve any matter properly before the meeting, unless more than a majority is required by statute or the Host REIT Charter. Each stockholder is entitled to cast the votes owned of record by him either in person or by proxy, but no proxy shall be valid after 11 months from the date of its execution, unless otherwise provided in the proxy.

### STOCKHOLDER ACTION BY WRITTEN CONSENT

DGCL. Under the DGCL, unless otherwise provided in a corporation's certificate of incorporation, any action that may be taken at any annual or special meeting of stockholders may be taken without a meeting, without prior notice and without a vote, if a consent (or consents) in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered to the corporation.

Host. The Host Certificate and Host Bylaws provide that any action required or permitted to be taken by the stockholders of the corporation must be effected at a duly called annual or special meeting of such holders and may not be effected by any consent in writing by such holders.

MGCL. Under the MGCL, any action required or permitted to be taken at a meeting of stockholders may be taken without a meeting if the following are filed with the records of stockholder meetings: (i) an unanimous written consent which sets forth the action and is signed by each stockholder entitled to vote on the matter; and (ii) a written waiver of any right to dissent signed by each stockholder entitled to notice of the meeting but not entitled to vote at it.

Host REIT. Pursuant to the MGCL and Host REIT Bylaws, subject to the rights of the holders of any class or series of stock (other than Host REIT Common Stock) to elect additional directors under specific circumstances, any action required or permitted to be taken by the stockholders must be effected at a duly called annual or special meeting of stockholders and may not be effected by any consent in writing by stockholders, unless such consent is unanimous.

## SPECIAL STOCKHOLDER MEETINGS

DGCL. Under the DGCL, special meetings of the stockholders may be called by the board of directors or by such person or persons as may be authorized by the certificate of incorporation or by the bylaws.

Host. The Host Certificate and Host Bylaws provide that, subject to the rights of holders of any series of preferred stock, special meetings of stockholders of the corporation may be called only by the Board of Directors pursuant to a resolution approved by a majority of the entire Board of Directors. Stockholders are not permitted to call a special meeting or to require that the Host Board call a special meeting of stockholders. Moreover, the business permitted to be conducted at any special meeting of stockholders is limited to the business brought before the meeting by or at the direction of the Host Board.

MGCL. Under the MGCL, a special meeting of a Maryland corporation's stockholders may be called by (i) the president, (ii) the board of directors, or (iii) any other person specified in the charter or the by-laws. The MGCL further provides that the secretary of a Maryland corporation shall call a special meeting of the stockholders on the written request of stockholders entitled to cast at least 25% of all the votes entitled to be cast at the meeting, unless a Maryland corporation includes in its charter or bylaws a provision that requires the written request of stockholders entitled to cast a greater or lesser percentage of all votes entitled to be cast at the meeting, except that in no event shall the percentage provided for in the charter or bylaws be greater than a majority of all the votes entitled to be cast at the meeting. Notwithstanding the foregoing, the MGCL provides that unless requested by stockholders entitled to cast a majority of all the votes entitled to be cast at the meeting, a special meeting need not be called to consider any matter which is substantially the same as a matter voted on at any special meeting of the stockholders held during the preceding 12 months.

Host REIT. The Host REIT Charter will provide that special meetings of the stockholders may be called by the President, the Board of Directors of Host REIT or any other person specified in the Host REIT Bylaws. The Host REIT Charter will provide further that the Secretary of Host REIT also is required to call a special meeting of the stockholders on the written request of stockholders entitled to cast a majority of all the votes entitled to be cast at the meeting. The Host REIT Bylaws contain similar provisions to the Host REIT Charter regarding the calling of special stockholder meetings. In addition, the Host REIT Bylaws provide that special stockholder meetings may be called by the holders of any class or series of stock having a preference over the Host REIT Common Stock as to dividends or upon liquidation in the manner (if any) specified in articles supplementary filed as part of the Host REIT Charter.

## NUMBER AND ELECTION OF DIRECTORS

DGCL. Under the DGCL, the minimum number of directors is one. The DGCL provides that the number of directors shall be fixed by, or in the manner provided in, the bylaws, unless the certificate of incorporation fixes the number of directors, in which case a change in the number of directors may be made only upon amendment of the certificate of incorporation. In addition, the DGCL permits, but does not require, a classified board of directors, with staggered terms under which one-half or one-third of the directors are elected for terms of two or three years, respectively. The DGCL provides that directors shall be elected by a plurality of the votes of the shares present in person or represented by proxy at a stockholder meeting and entitled to vote on the election of directors. The DGCL further provides that the vote of the majority of the directors present at a meeting at which a quorum is present shall be the act of the board of directors unless the certificate of incorporation or the bylaws shall require a vote of greater number.

Host. The Host Certificate provides that the number of directors shall be fixed from time to time by or pursuant to the Host Bylaws. The Host Bylaws provide that the Host Board shall be fixed from time to time by the Board of Directors but shall not be less than three. The Host Certificate and Host Bylaws provide that the Board of Directors is divided into three classes, each consisting of approximately one-third of the total number of directors. The term of office for each director is three years and such terms expire in successive years at the time of the annual meeting of stockholders. The current number of directors is eight. There are no cumulative voting rights in the election of directors under the Host Certificate.

MGCL. Under the MGCL, the minimum number of directors is three. The MGCL provides that the number of directors shall be provided by the charter until changed by the by-laws. The bylaws may both (i) alter the number of directors set by the charter, and (ii) authorize a majority of the entire board of directors to alter within specified limits the number of directors set by the charter or the bylaws, but the action may not affect the tenure of office of any director. In addition, the MGCL permits, but does not require, a classified board of directors. If the directors are divided into classes, the term of office may be provided in the by-laws, except that (i) the term of office of a director may not be longer than five years or, except in the case of an initial or substitute director, shorter than the period between annual meetings, and (ii) the term of office of at least one class must expire each year. Each share of stock may be voted for as many individuals as there are directors to be elected and for whose election the share is entitled to be voted. Unless the charter or bylaws provide otherwise, a plurality of all the votes cast at a meeting at which a quorum is present is sufficient to elect a director

Host REIT. The Host REIT Charter will provide that, effective upon completion of the Merger, the Board of Directors will consist of eight members and may thereafter be increased or decreased in accordance with the Host REIT Bylaws, provided that the total number of directors may not be fewer than three nor more than thirteen. Pursuant to the Host REIT Bylaws, the number of directors shall be fixed by the Board of Directors within the limits set forth in the Host REIT Charter. Further, the Host REIT Charter will provide that the Board of Directors, other than those who may be elected by the holders of any class or series of stock having a preference over the Host REIT Common Stock as to dividends or upon liquidation, will be divided into three classes of directors, with each class to consist as nearly as possible of an equal number of directors. The term of office of the first class of directors will expire at the 1999 annual meeting of stockholders; the term of the second class of directors will expire at the 2000 annual meeting of stockholders; and the term of the third class of

directors will expire at the 2001 annual meeting of stockholders. The Host REIT Charter will provide further that in the event of any increase or decrease in the number of directors, other than resulting from the rights of the holders of any class or series of stock having a preference over the Host REIT Common Stock as to dividends or upon liquidation to elect additional directors pursuant to the specified provisions of the Host REIT Charter, the newly created or eliminated directorship resulting from such increase or decrease shall be apportioned by the Board of Directors among the three classes of directors so as to maintain such classes as nearly equal in number as possible. At each annual meeting of stockholders, the class of directors to be elected at such meeting will be elected for a three-year term, and the directors in the other two classes will continue in office. Because stockholders of Host REIT will have no right to cumulative voting for the election of directors, at each annual meeting of stockholders the holders of a majority of the outstanding shares of Host REIT Common Stock will be able to elect all of the successors to the class of directors whose term expires at that meeting.

## REMOVAL OF DIRECTORS

DGCL. The DGCL provides that a director of a corporation may be removed with or without cause by the holders of a majority of shares then entitled to vote at an election of directors, provided, that, when a corporation has a classified board of directors, a director may be removed only for cause, unless the certificate of incorporation provides otherwise.

Host. The Host Certificate and Host Bylaws provide that directors may be removed from office, with or without cause, only by the affirmative vote of the holders of at least 66 2/3% of the Voting Stock.

MGCL. Under the MGCL, unless the charter provides otherwise, the stockholders of a Maryland corporation may remove any director, with or without cause, by the affirmative vote of a majority of all the votes entitled to be cast for the election of directors, except that, unless the charter provides otherwise, (i) if the stockholders of any class or series are entitled separately to elect one or more directors, a director elected by a class or series may not be removed without cause except by the affirmative vote of a majority of all the votes of that class or series, (ii) if a corporation has cumulative voting for the election of directors and less than the entire board is to be removed, a director may not be removed without cause if the votes cast against his removal would be sufficient to elect him if then cumulatively voted at an election of the entire board of directors, or, if there is more than one class of directors, at an election of the class of directors of which he is a member, and (iii) if the directors have been divided into classes, a director may not be removed without cause.

Host REIT. The Host REIT Charter and Host REIT Bylaws will provide that, except for any directors who may be elected by holders of a class or series of shares other than Host REIT Common Stock, directors may be removed only for cause and only by the affirmative vote of stockholders holding at least two-thirds of all the votes entitled to be cast for the election of directors.

# VACANCIES ON THE BOARD OF DIRECTORS

DGCL. Under the DGCL, vacancies and newly created directorships resulting from any increase in the authorized number of directors elected by all of the stockholders having the right to vote as a single class may be filled by a majority of the directors then in office, although less than a quorum, unless otherwise provided in the certificate of incorporation or the bylaws. However, if the certificate of incorporation directs that a particular class is to elect such director, such vacancy may be filled only by the other directors elected by such class. If, at the time of filling any vacancy or newly created directorship, the directors then in office constitute less than a majority of the whole board as constituted immediately prior to such increase, the Delaware Court of Chancery may, upon application of stockholders holding at least ten percent of the total number of shares outstanding having the right to vote for such directors, order an election to be held to fill any such vacancies or newly created directorships or to replace the directors chosen by the directors then in office. Under the DGCL, unless otherwise provided in the certificate of incorporation or bylaws, when one or more directors resigns from the board, effective at a future date, a majority of the directors then in office, including those who have so resigned, shall have the power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or

resignations shall become effective, and each director so chosen shall hold office as provided in the DGCL in the filling of other vacancies.

Host. The Host Certificate provides that, subject to any rights of the holders of preferred stock, newly created directorships resulting from any increase in the number of directors and any vacancies on the board of directors resulting from death, resignation, disqualification, removal or other cause shall be filled by the affirmative vote of a majority of the remaining directors then in office, even though less than a quorum of the board of directors. Any director elected in this manner shall hold office for the remainder of the full term of the class of directors in which the new directorship was created or the vacancy occurred and until such director's successor shall have been elected and qualified. No decrease in the number of directors may shorten the term of any incumbent director.

MGCL. Under the MGCL, the stockholders may fill vacancies on the board of directors caused by the removal of a director, except that if the stockholders of any class or series are entitled separately to elect one or more directors, the stockholders of that class or series may elect a successor to fill a vacancy which results from the removal of a director elected by that class or series. The MGCL further provides that unless the charter or bylaws provide otherwise, a majority of the remaining directors, whether or not sufficient to constitute a quorum, may fill a vacancy on the board of directors which results from any cause except that (i) a vacancy caused by an increase in the number of directors may be filled by a majority of the entire board, and (ii) if the stockholders of any class or series are entitled separately to elect one or more directors, a majority of the remaining directors elected by that class or series or the sole remaining director elected by that class or series may fill any vacancy among the number of directors elected by that class or series.

Under the MGCL, a director elected by the board of directors to fill a vacancy serves until the next annual meeting of stockholders and until his successor is elected and qualifies. A director elected by the stockholders to fill a vacancy which results from the removal of a director serves for the balance of the term of the removed director.

Host REIT. The Host REIT Charter and Host REIT Bylaws provide that except in the case of a vacancy on the Board of Directors among the directors elected by a class or series of stock other than Host REIT Common Stock, any vacancy on the Board of Directors may be filled by the affirmative vote of a majority of the remaining directors (except that a vacancy which results from an increase in the number of directors may be filled by a majority of the entire Board of Directors). In addition, the Host REIT Charter will provide that in the case of a vacancy resulting from the removal of a director, by the stockholders by the affirmative vote of two-thirds of the votes entitled to be cast for the election of directors. The Host REIT Charter will provide further that any vacancy on the Board of Directors among the directors elected by a class or series of stock other than Host REIT Common Stock may be filled by a majority of the remaining directors elected by that class or series, or by the stockholders of that class or series unless otherwise provided in the articles supplementary for that class or series.

# ADVANCE NOTICE OF DIRECTOR NOMINATIONS AND OF NEW BUSINESS PROPOSALS

Host. The Host Bylaws provide that (i) nominations for persons for election to the Board of Directors may be made (A) with respect to an annual meeting of the stockholders, by the Board of Directors or a proxy committee appointed by the Board of Directors, or by any stockholder entitled to vote in the election of directors if written notice of such stockholder's intent to make such nomination or nominations is given to the Secretary of Host not later than ninety days in advance of the date established by the Host Bylaws for the holding of such meeting, or (B) with respect to special meetings of the stockholders, by the Board of Directors or a proxy committee appointed by the Board of Directors, or by any stockholder entitled to vote in the election of directors if written notice of such stockholder's intent to make such nomination or nominations is given to the Secretary of Host not later than the close of business on the seventh day following the date on which notice of such meeting is first given to stockholders, and (ii) stockholder proposals may be made by any stockholder entitled to vote in the election of directors and who satisfies the requirements of the proxy rules under the Exchange Act by notice in writing, such notice to be received by the Secretary of Host not less than (X) with respect to an annual meeting

of the stockholders, one hundred and twenty days in advance of the date established for the holding of such meeting, or (Y) with respect to special meeting of the stockholders, the close of business on the seventh day following the date on which notice of such meeting is first given to stockholders.

Host REIT. The Host REIT Bylaws provide that (i) with respect to an annual meeting of stockholders, subject to the rights of holders of any class or series of stock having a preference over the Host REIT Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, nominations of persons for election to the Board of Directors and the proposal of business to be considered by stockholders may be made only (A) pursuant to Host REIT's notice of meeting, (B) by or at the direction of the Board of Directors or (C) by a stockholder who was a stockholder of record both at the time of giving notice provided for in the Host REIT Bylaws and at the time of the annual meeting, and who is entitled to vote at the meeting and has complied with the advance notice procedures set forth in the Host REIT Bylaws and (ii) with respect to special meetings of the stockholders, only the business specified in Host REIT's notice of meeting may be brought before the meeting of stockholders and except as otherwise provided for or fixed by or pursuant to the provisions of the Host REIT Charter relating to the rights of the holders of any class or series of stock having a preference over the Host REIT Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, nominations of persons for election to the Board of Directors may be made only (X) pursuant to Host REIT's notice of the meeting, (Y) by or at the direction of the Board of Directors or (Z) provided that the Board of Directors has determined that directors shall be elected at such meeting, by a stockholder who was a stockholder of record both at the time of giving notice provided for in the Host REIT Bylaws and at the time of the special meeting, and who is entitled to vote at the meeting and has complied with the advance notice provisions set forth in the Host REIT Bylaws. The advance notice provisions contained in the Host REIT Bylaws generally require nominations and new business proposals by stockholders to be delivered to the Secretary of Host REIT not later than the close of business on the 60th day nor earlier than the close of business on the 90th day before the date on which Host REIT first mailed its proxy materials for the prior year's annual meeting of stockholders.

## TRANSACTIONS INVOLVING DIRECTORS OR OFFICERS

DGCL. Under the DGCL, no contract or transaction between a corporation and one or more of its directors or officers, or between a corporation and any other corporation, partnership, association, or other organization in which one or more of its directors or officers, are directors or officers, or have a financial interest, shall be void or voidable solely for this reason, or solely because the director or officer is present at or participates in the meeting of the board or committee which authorizes the contract or transaction, or solely because such director's votes are counted for such purpose, if (i) the material facts as to such director's relationship or interest and as to the contract or transaction are disclosed or are known to the board of directors or the committee, and the board or committee in good faith authorizes the contract or transaction by the affirmative votes of a majority of the disinterested directors, even though the disinterested directors be less than a quorum; (ii) the material facts as to such director's relationship or interest and as to the contract or transaction are disclosed or are known to the stockholders entitled to vote thereon, and the contract or transaction is specifically approved in good faith by vote of the stockholders; or (iii) the contract or transaction is fair to the corporation as of the time it is authorized, approved or ratified by the board of directors, a committee or the stockholders.

A corporation may lend money to, or guarantee any obligation of, or otherwise assist any officer or other employee of the corporation or of its subsidiary, including any officer who is a director of the corporation or its subsidiary, whenever, in the judgment of the directors, such loan, guaranty or assistance may reasonably be expected to benefit the corporation.

Host. The Host Certificate provides that the corporation may enter into contracts or transact business with one or more of its officers or directors, or with any firms of which one or more of its officers or directors is a member, or may invest its funds in the securities of and may enter into contracts, or transact business with any corporation or association in which any one or more of its officers or directors is a stockholder, officer or director, and in the absence of bad faith, or unfair dealing, such contract or transaction or investment shall not be

invalidated or to any extent affected by the fact that any such officer or officers or any such director or directors has or may have interests therein which are or might be adverse to the interests of the corporation, provided that the remaining directors are sufficient in number to ratify and approve the transaction.

 $\operatorname{MGCL}$ . Under the  $\operatorname{MGCL}$ , a contract or transaction between a Maryland corporation and any of its directors or between a Maryland corporation and any other corporation, firm, or other entity in which any of its directors is a director, or has a material financial interest, is not void or voidable solely for this reason, or solely because the director is present at the meeting of the board or committee of the board which authorizes, approves, or ratifies the contract or transaction, or solely because such director's or directors' votes are counted for such purpose, if (i) the fact of common directorship or interest is disclosed or known to the board of directors or the committee, and the board or committee authorizes, approves, or ratifies the contract or transaction by the affirmative vote of a majority of disinterested directors, even if the disinterested directors constitute less than a quorum, (ii) the fact of common directorship or interest is disclosed or known to the stockholders entitled to vote, and the contract or transaction is authorized, approved, or ratified by a majority of the votes cast by the stockholders entitled to vote other than the votes of shares owned of record or beneficially by the interested corporation, firm or other entity, or (iii) the contract or transaction is fair and reasonable to the corporation. Common or interested directors or the stock owned by them or by an interested corporation, firm, or other entity may be counted in determining the presence of a quorum at a meeting of the board of directors or a committee of the board or at a meeting of the stockholders, as the case may be, at which the contract or transaction is authorized, approved or ratified.

Host REIT. As a Maryland corporation, Host REIT will be subject to the foregoing provisions with respect to transactions between Host REIT and its directors. Host REIT's Board of Directors also intends to adopt a policy which would require that all material contracts and transactions between Host REIT, the Operating Partnership or any of its subsidiaries, on the one hand, and a director or executive officer of Host REIT or any entity in which such director or executive officer is a director or has a material financial interest, on the other hand, must be approved by the affirmative vote of a majority of the disinterested directors. Where appropriate in the judgment of the disinterested directors, the Board of Directors may obtain a fairness opinion or engage independent counsel to represent the interests of nonaffiliated security holders, although the Board of Directors will have no obligation to do so. See "Distribution and Other Policies--Conflicts of Interest Policies."

## LIMITATION OF LIABILITY AND INDEMNIFICATION OF DIRECTORS AND OFFICERS

DGCL. Under the DGCL, directors may be indemnified for liabilities incurred in connection with specified actions (other than any action brought by or in the right of the corporation), if they acted in good faith and in a manner they reasonably believed to be in and not opposed to the best interests of the corporation and, with respect to any criminal action or proceeding, had no reasonable cause to believe their conduct was unlawful. The same standard of conduct is applicable for indemnification in the case of derivative actions brought by or in the right of the corporation, except that in such cases the DGCL authorizes indemnification only for expenses (including attorneys' fees) incurred in connection with the defense or settlement of such cases. Moreover, the DGCL requires court approval before there can be any such indemnification where the person seeking indemnification has been found liable to the corporation in a derivative action. To the extent that a present or former director or officer has been successful in defense of any action, suit or proceeding, the DGCL provides for indemnification of such person for expenses (including attorneys' fees). The DGCL states expressly that the indemnification provided by or granted pursuant to the DGCL is not deemed exclusive of any non-statutory indemnification rights existing under any bylaw, agreement, vote of stockholders or disinterested directors or otherwise.

Host. Host's Certificate of Incorporation and Bylaws provide that every director, officer and employee of Host shall be indemnified against all expenses and liabilities, including counsel fees, reasonably incurred by or imposed upon him by reason of his being or having been a director, officer or employee of Host. Under Host's Certificate, no director shall be liable to Host or its shareholders for monetary damages, for breach of fiduciary duty as a director, except for liability (a) for any breach of the director's duty of loyalty to the corporation or its

shareholders, (b) for acts or omissions not in good faith or which involve intentional misconduct or knowing violation of law, (c) under section 174 of the DGCL (concerning unlawful payment of dividend or unlawful stock purchase or redemption), or (d) for any transaction from which the directors derived an improper personal benefit.

MGCL. The MGCL permits a Maryland corporation to include in its charter a provision limiting the liability of its directors and officers to the corporation and its shareholders for money damages except for liability resulting from (i) actual receipt of an improper benefit or profit in money, property or services or (ii) acts committed in bad faith or active and deliberate dishonesty established by a final judgment as being material to the cause of action; or (iii) in the case of any criminal proceeding, the director having reasonable cause to believe that an act or omission was unlawful. The MGCL further provides that a Maryland corporation may not indemnify for an adverse judgment in a suit by or in the right of the corporation. The MGCL expressly states that the indemnification or advancement of expenses authorized by the MGCL shall not be deemed exclusive of any other rights, by indemnification or otherwise, to which a director or officer may be entitled under the charter, the bylaws, a resolution of the stockholders or directors, an agreement or otherwise.

Host REIT. The Host REIT Charter will provide that to the fullest extent permitted by Maryland statutory or decisional law, as amended or interpreted, no director or officer of Host REIT shall be personally liable to Host REIT or its stockholders for money damages. No amendment of the Host REIT Charter or repeal of its provisions will limit the benefits provided to directors and officers under this provision with respect to any act or omission which occurred prior to such amendment or repeal.

The Host REIT Charter and Host REIT Bylaws will provide broad indemnification to directors and officers, whether serving Host REIT, or at its request, any other entity, to the fullest extent permitted under Maryland law. Host REIT will indemnify any present or former director and officer, (A) who has been successful, on the merits or otherwise, in defense of a proceeding to which he or she was made a party by reason of service in such capacity against reasonable expenses incurred by such officer or director in connection with the proceeding, and (B) against judgments, penalties, fines, settlements and reasonable expenses actually incurred by them in connection with any proceeding to which they may be made a party by reason of their service in those or other capacities unless it is established that (i) the act or omission of the director or officer was material to the matter giving rise to the proceeding and (a) was committed in bad faith or (b) was the result of active and deliberate dishonesty, (ii) the director or officer actually received an improper personal benefit in money, property or services or (iii) in the case of any criminal proceeding, the director or officer had reasonable cause to believe that the act or omission was unlawful.

The Bylaws of Host REIT require Host REIT, as a condition to advancing expenses, to obtain (i) a written affirmation by the director or officer of his good faith belief that he has met the standard of conduct necessary for indemnification by Host REIT as authorized by the Bylaws and (ii) a written statement by or on his behalf to repay the amount paid or reimbursed by Host REIT if it shall ultimately be determined that the standard of conduct was not met.

Host REIT also intends to enter into indemnification agreements indemnifying each of its directors and officers to the fullest extent permitted by Maryland law and to advance to its directors and officers all related expenses subject to reimbursement if it is subsequently determined that indemnification is not permitted. See "--Limitation of Liability and Indemnification of Directors and officers" for a description of the limitations on liability of directors and officers of Host REIT and the provisions for indemnification of directors and officers provided for under applicable Maryland law and the Host REIT Charter.

# DECLARATION OF DIVIDENDS

DGCL. Under the DGCL, a corporation is permitted to declare and pay dividends out of surplus (as defined in the DGCL) or, if there is no surplus, out of net profits for the fiscal year in which the dividend is declared and/or for the preceding fiscal year as long as the amount of capital of the corporation following the declaration and payment of the dividend is not less than the aggregate amount of the capital represented by the issued and outstanding stock of all classes having a preference upon the distribution of assets. Dividends may be

paid in cash, property or shares of a corporation's capital stock. In addition, the DGCL generally provides that a corporation may redeem or repurchase its shares only if such redemption or repurchase would not impair the capital of the corporation.

Host. The Host Bylaws provide that dividends upon the capital stock of Host, subject to the provisions of the Host Certificate, if any, may be declared by the Host Board at any regular or special meeting, pursuant to law. Dividends may be paid in cash, in property, or in shares of the capital stock, subject to the provisions of the Host Certificate.

MGCL. Under the MGCL, if authorized by its board of directors, a Maryland corporation may declare and pay dividends subject to any restriction in its charter unless, after giving effect to the dividend, (i) the corporation would not be able to pay indebtedness of the corporation as the indebtedness becomes due in the

usual course of business, or (ii) the corporation's total assets would be less than the sum of the corporation's total liabilities plus, unless the charter permits otherwise, the amount that would be needed, if the corporation were to be dissolved at the time of the distribution, to satisfy the preferential rights upon dissolution of the stockholders whose preferential rights on dissolution are superior to those receiving the dividend.

Host REIT. Under the Host REIT Bylaws, dividends upon the shares of stock of Host REIT may be authorized and declared by the directors, subject to the provisions of law and the Host REIT Charter. Under the Host REIT Charter, subject to the provisions of law and any preferences of any class of the capital stock of Host REIT, classified or reclassified, dividends (including dividends payable in shares of another class of Host REIT stock) will be able to be paid on Host REIT Common Stock at such time and in such amounts as the Board of Directors may deem advisable and the holders of Host REIT Common Stock shall share ratably in any such dividends, in proportion to the number of shares of Host REIT Common Stock held by them respectively, on a share for share basis. The Host REIT Bylaws provide that dividends and other distributions may be paid in cash, property, or shares of the Corporation, subject to Maryland law and the Charter. Host REIT and the Operating Partnership currently intend to pay regular quarterly distributions to holders of shares of Host REIT Common Stock and OP Units. For a discussion of Host REIT's distribution policy, see "Distribution and Other Policies--Distribution Policy."

#### APPRAISAL RIGHTS

DGCL. Under the DGCL, the right to receive the fair value of dissenting shares is made available to stockholders of a constituent corporation in a merger or consolidation effected under the DGCL. Dissenters' rights of appraisal are not available for the shares of any class or series of stock, which stock, or depository receipts in respect thereof, at the record date fixed to determine stockholders entitled to receive notice and vote on such transaction, were either (i) listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Security Dealers, Inc. or (ii) held of record by more than 2,000 holders. Further, no appraisal rights are available for any shares of stock of the constituent corporation surviving a merger if the merger did not require for its approval the vote of the stockholders of the surviving corporation as provided by the DGCL.

Notwithstanding the foregoing, appraisal rights under the DGCL are available for the shares of any class or series of stock of a constituent corporation if the holders thereof are required by the terms of an agreement of merger or consolidation pursuant to the DGCL to accept for such stock anything except (i) shares of stock of the corporation surviving or resulting from such merger or consolidation, or depository receipts in respect thereof, (ii) shares of stock of any other corporation, or depository receipts in respect thereof, which shares of stock (or depository receipts in respect thereof) will be either listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc., or held of record by more than 2,000 holders, (iii) cash in lieu of fractional shares, or (iv) any combination of the shares of stock, depository receipts and cash in lieu of such fractional shares.

MGCL. Under the MGCL, a stockholder of a Maryland corporation has the right to demand and receive payment of the fair value of the stockholder's stock from the corporation if, among other things, the corporation consolidates or merges with another corporation. However, unless the transaction is governed by certain provisions specified by the MGCL, a stockholder may not demand the fair value of his stock and is bound by the terms of the transaction if (i) the stock is listed on a national securities exchange or is designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc.; (ii) the stock is that of the successor in a merger, unless (a) the merger alters the contract rights of the stock as expressly set forth in the charter, and the charter does not reserve the right to do so, or (b) the stock is to be changed or converted in whole or in part in the merger into something other than either stock in the successor or cash, scrip, or other rights or interests arising out of the provisions for the treatment of fractional shares of stock in the successor; or (iii) the stock is that of an open-end investment company registered with the SEC under the Investment Company Act of 1940 and the value placed on the stock in the transaction is its net asset value.

MERGER, CONSOLIDATION, SHARE EXCHANGE AND TRANSFER OF ALL OR SUBSTANTIALLY ALL ASSETS

DGCL. Under the DGCL, the principal terms of a merger or consolidation generally require the approval of the stockholders of each of the constituent corporations. Unless otherwise required in a corporation's certificate of incorporation, the DGCL does not require a stockholder vote of the surviving corporation in a merger if (i) the agreement of merger does not amend in any respect the certificate of incorporation of such constituent corporation, (ii) each share of stock of such constituent corporation outstanding immediately prior to the effective date of the merger is to be an identical outstanding or treasury share of the surviving corporation after the effective date of the merger, and (iii) either no shares of common stock of the surviving corporation and no shares, securities or obligations convertible into such stock are to be issued or delivered under the plan of merger, or the authorized unissued shares or the treasury shares of common stock of the surviving corporation to be issued or delivered under the plan of merger plus those initially issuable upon conversion of any other shares, securities or obligations to be issued or delivered under such plan do not exceed 20% of the shares of common stock of such constituent corporation outstanding immediately prior to the effective date of the merger.

When a stockholder vote is required under the DGCL to approve a merger or consolidation, unless the certificate of incorporation provides otherwise, the affirmative vote of a majority of shares present in person or represented by proxy for each class of shares entitled to vote on the merger or consolidation shall be required to approve the merger or consolidation. If multiple classes of stock are entitled to vote on the merger or consolidation as separate classes, then a majority of each class entitled to vote to approve the merger or consolidation, voting separately as a class, shall be required to approve the merger or consolidation.

Under the DGCL, a corporation may at any meeting of its board of directors or governing body, sell, lease or exchange all or substantially all of its property and assets, including its goodwill and its corporate franchises, upon such terms and conditions and for such consideration, which may consist in whole or in part of money or other property, including shares of stock in, and/or other securities of, any other corporation or corporations, as its board of directors or governing body deems expedient and for the best interests of the corporation, when and as authorized by a resolution adopted by the holders of a majority of the outstanding stock of the corporation entitled to vote thereon.

Host. The Host Certificate provides that the affirmative vote of the holders of shares representing not less than 66 2/3% of the voting power of the corporation shall be required for the approval of any proposal for the corporation to reorganize, merge, or consolidate with any other corporation, or to sell, lease, or exchange substantially all of its assets or business.

MGCL. The MGCL generally provides that mergers, consolidations, share exchanges or transfers of assets must first be approved by a majority of the board of directors and thereafter approved by stockholders by the affirmative vote of two-thirds of all the votes entitled to be cast on the matter (unless the charter provides for a greater or lesser stockholder vote but not less than a majority of the number of votes entitled to be cast on the matter). However, under the MGCL, certain mergers may be accomplished without a vote of stockholders. For

example, no stockholder vote is required for a merger of a subsidiary of a Maryland corporation into its parent, provided the parent owns at least 90 percent of the subsidiary. In addition, a merger need not be approved by stockholders of a Maryland successor corporation if the merger does not reclassify or change the outstanding shares or otherwise amend the charter, and the number of shares to be issued or delivered in the merger is not more than 20 percent of the number of its shares of the same class or series outstanding immediately before the merger becomes effective. A share exchange need be approved by a Maryland successor only by its board of directors and by any other action required by its charter. Under the MGCL, a "transfer of assets" is defined to mean any sale, lease, exchange or other transfer of all or substantially all of the assets of the corporation but does not include (i) a transfer of assets by a corporation in the ordinary course of business actually conducted by it, (ii) a mortgage, pledge or creation of any other security interest in any or all of the assets of the corporation, whether or not in the ordinary course of its business, (iii) an exchange of shares of stock through voluntary action under any agreement with the stockholders, or (iv) a transfer of assets to one or more persons if all the equity interests of the person or persons are owned, directly or indirectly, by the corporation.

Host REIT. Pursuant to the Host REIT Charter, subject to the terms of any class or series of capital stock at the time outstanding, Host REIT will be able to merge with or into another entity, consolidate with one or more other entities, participate in a share exchange or transfer its assets within the meaning of the MGCL, but any such merger consolidation, share exchange or transfer of assets must be approved by the Board of Directors in the manner provided in the MGCL. The Host REIT Charter generally will provide for stockholder approval of such transactions by a two-thirds vote of all votes entitled to be cast (subject to the above enumerated statutory exceptions to a stockholder vote), except that any merger of Host REIT with or into a trust organized for the purpose of changing Host REIT's form of organization from a corporation to a trust will require the approval of stockholders of Host REIT by the affirmative vote only of a majority of all the votes entitled to be cast on the matter, provided that (i) the stockholders of the trust immediately following the merger are the same as the stockholders of Host REIT immediately prior to the merger and (ii) the trust's declaration of trust contains amendment provisions substantially equivalent to those contained in specified provisions of the Host REIT Bylaws.

### CHANGES IN CONTROL PURSUANT TO DELAWARE/MARYLAND LAW

DGCL. Section 203 of the DGCL provides that, subject to certain exceptions specified therein, a corporation will not engage in any business combination with any "interested stockholder" for a three-year period following the time that such stockholder becomes an interested stockholder unless (i) prior to such time the board of directors of the corporation approved either the business combination or the transaction which resulted in the stockholder becoming an interested stockholder, (ii) upon consummation of the transaction which resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the corporation outstanding at the time the transaction commenced (excluding for purposes of determining the number of shares outstanding those shares owned by persons who are directors and also officers, and employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer), or (iii) at or subsequent to such time the business combination is approved by the board of directors and authorized at an annual or special meeting of stockholders, and not by written consent, by the affirmative vote of at least 66 2/3% of the outstanding voting stock which is not owned by the interested stockholder. Except as specified in Section 203 of the DGCL, an interested stockholder is defined to include any person that (a) is the owner of 15% or more of the outstanding voting stock of the corporation, or (b) is an affiliate or associate of the corporation and was the owner of 15% or more of the outstanding voting stock of the corporation at any time within the 3-year period immediately prior to the date on which it is sought to be determined whether such person is an interested stockholder; and the affiliates and associates of such person. Section 203(b)(4) exempts from the restrictions in Section 203 a corporation that does not have a class of voting stock that is (i) listed on a national securities exchange, (ii) authorized for quotation on The NASDAQ Stock Market, or (iii) held of record by more than 2,000 stockholders, unless any of the foregoing results from action taken, directly or indirectly, by an interested stockholder or from a transaction in which a person becomes an interested stockholder.

73

Host. The Host Certificate provides that in addition to any affirmative vote required by law or by the Host Certificate, certain Business Combinations (as defined in the Host Certificate) involving "interested shareholders" (as defined below) require the affirmative vote of the holders of at least 66 2/3% of the Voting Stock, voting together as a single class unless the Business Combination is approved by a majority of the Disinterested Directors (as defined in the Host Certificate) or certain other conditions are satisfied. An "interested shareholder" is defined by the Host Certificate as any person (other than Host or any subsidiary of Host) who or which (i) is the beneficial owner, directly or indirectly, of more than 25% of the voting power of the outstanding Voting Stock, (ii) is an affiliate of Host (as defined in the Host Certificate) and at any time within the two-year period immediately prior to the date in question was the beneficial owner, directly or indirectly, of 25% or more of the voting power of the then outstanding Voting Stock, or (iii) is an assignee of or has otherwise succeeded to any shares of Voting Stock which were at any time within the two-year period immediately prior to the date in question beneficially owned by any "interested shareholder," if such assignment or succession shall have occurred in the course of a transaction or series of transactions not involving a public offering within the meaning of the Securities Act.

MGCL. Under the MGCL, certain "business combinations" (including certain issuances of equity securities) between a Maryland corporation and any Interested Stockholder or an affiliate of the Interested Stockholder are prohibited for five years after the most recent date on which the Interested Stockholder becomes an Interested Stockholder. Thereafter, any such business combination must be approved by a supermajority (80%) of outstanding voting shares, and by two-thirds of voting shares other than those held by an Interested Stockholder unless, among other conditions, the corporation's common stockholders receive a minimum price (as defined in the MGCL) for their shares and the consideration is received in cash or in the same form as previously paid by the Interested Stockholder. A business combination that is approved by the board of directors of a Maryland corporation at any time before an Interested Stockholder first becomes an Interested Stockholder is not subject to the special voting requirements. Host REIT has not "opted-out" of the business combination provisions of the MGCL and, accordingly, will be subject to such provisions although Host REIT may elect to opt-out of these provisions in the future. The Board of Directors of Host REIT has adopted a resolution exempting from the operation of the "business combination" statute the acquisition of shares by Marriott International (together with its successors and permitted assignees, if any) pursuant to the terms of the Marriott International Purchase Right as well as any other transactions involving Host REIT, Marriott International, J.W. Marriott, Jr. and Richard E. Marriott; provided that any such transaction with Marriott International that is not in the ordinary course of business or with J.W. Marriott, Jr. or Richard E. Marriott must be approved by a majority of the directors of the Company present at a meeting at which a quorum is present, including a majority of the disinterested directors, in addition to any vote of stockholders required by other provisions of the MGCL.

The MGCL further provides that "control shares" acquired in a "control share acquisition" have no voting rights except to the extent approved by a vote of two-thirds of the votes entitled to be cast on the matter, excluding shares owned by the acquiror, by officers or by directors who are employees of the corporation. "Control shares" are voting shares which, if aggregated with all other such shares previously acquired by the acquiror or in respect of which the acquiror is able to exercise or direct the exercise of voting power (except solely by virtue of a revocable proxy), would entitle the acquiror to exercise voting power in electing directors within one of the following ranges of voting power: (i) one-fifth or more but less than one-third, (ii) one-third or more but less than a majority, (iii) a majority or more of the voting power. Control shares do not include shares the acquiring person is then entitled to vote as a result of having previously obtained stockholder approval. A "control share acquisition" means the acquisition of control shares, subject to certain exceptions. The Bylaws of Host REIT will contain an exemption from the control share acquisition provisions for any shares acquired by Marriott International (together with its successors and permitted assignees, if any) pursuant to the Marriott International Purchase Right, to the extent it is exercised.

Host REIT. The Host REIT Charter will not contain a Fair Price Provision similar to the provision in the Host Certificate. Host REIT has not "optedout" of the business combination or "control share" acquisition provisions of the MGCL, and, accordingly will be subject to such provisions. However, as permitted by the MGCL, Host REIT's Board of Directors may elect to opt out of these provisions in the future.

### STOCKHOLDER RIGHTS PLAN

Host. Host currently has in effect a stockholder rights plan pursuant to the Host Rights Agreement and it has preferred stock purchase rights attached to the Host Common Stock pursuant to such rights plan.

Host REIT. Host REIT intends to adopt the Stockholder Rights Plan to replace the existing stockholder rights plan of Host. All shares of Host REIT Common Stock issued by Host REIT between the date of adoption of the Stockholder Rights Plan and the rights distribution date, or the date, if any, on which the preferred stock purchase rights are redeemed would have preferred stock purchase rights attached to them. See "Description of Host REIT Capital Stock--Certain Provisions of Maryland Law and the Host REIT Charter and Host REIT Bylaws--Stockholder Rights Plan." The Stockholder Rights Plan is expected to provide, among other things, that upon the occurrence of certain events, stockholders will be entitled to purchase from Host REIT a newly created series of junior preferred stock, subject to Host REIT's Ownership Limit. See "Description of Host REIT Capital Stock" and "Certain Provisions of Maryland Law and the Host REIT Charter and Bylaws."

The discussion of the comparative rights of stockholders of Host and shareholders of Host REIT set forth above is not complete and is subject to and qualified in its entirety by reference to the DGCL and the MGCL and to the Host Certificate, Host Bylaws and the Host Rights Agreement and the Host REIT Charter, Host REIT Bylaws and the Shareholder Rights Plan. Copies of the Host REIT Charter and Host REIT Bylaws are attached as Exhibits B and C, respectively, to the Agreement, which is attached to this Proxy Statement/Prospectus as Appendix A.

# LIMITATION OF LIABILITY AND INDEMNIFICATION OF DIRECTORS AND OFFICERS

The MGCL permits a Maryland corporation to include in its charter a provision limiting the liability of its directors and officers to the corporation and its stockholders for money damages except for liability resulting from (a) actual receipt of an improper benefit or profit in money, property or services or (b) acts committed in bad faith or active and deliberate dishonesty established by a final judgment as being material to the cause of action. The Host REIT Charter will contain such a provision which limits such liability to the maximum extent permitted by Maryland law.

The Host REIT Charter will authorize it, to the maximum extent permitted by Maryland law, to obligate itself to indemnify and to pay or reimburse reasonable expenses in advance of final disposition of a proceeding to (i) any present or former director or officer or (ii) any individual who, while a director of Host REIT and at the request of Host REIT, serves or has served another corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or any other enterprise from and against any claim or liability to which such person may become subject or which such person may incur by reason of his or her status as a present or former Director or officer of Host REIT. The Host REIT Bylaws obligate it, to the maximum extent permitted by Maryland law, to indemnify and to pay or reimburse reasonable expenses in advance of final disposition of a proceeding to (a) any present or former director or officer who is made party to the proceeding by reason of his service in that capacity or (b) any individual who, while a director or officer of Host REIT and at the request of Host REIT, serves or has served another corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or any other enterprise as a director, trustee, officer or partner of such corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or other enterprise and who is made a party to the proceeding by reason of his service in that capacity, against any claim or liability to which he may become subject by reason of such status. The Host REIT Charter and Host REIT Bylaws also permit Host REIT to indemnify and advance expenses to any person who served as a predecessor of Host REIT in any of the capacities described above and to any employee or agent of Host REIT or a predecessor of Host REIT. The Host REIT Bylaws require Host REIT to indemnify a director or officer who has been successful on the merits or otherwise in the defense of any proceeding to which he is made a party by reason of his service in that capacity.

The MGCL permits a Maryland corporation to indemnify and advance expenses to its directors, officers, employees and agents, and permits a corporation to indemnify its present and former directors and officers,

among others, against judgments, penalties, fines, settlements and reasonable expenses actually incurred by them in connection with any proceeding to which they may be made a party by reason of their service in those or other capacities unless it is established that (a) the act or omission of the director or officer was material to the matter giving rise to the proceeding and (i) was committed in bad faith or (ii) was the result of active and deliberate dishonesty, (b) the director or officer actually received an improper personal benefit in money, property or services or (c) in the case of any criminal proceeding, the director or officer had reasonable cause to believe that the act or omission was unlawful. However, under the MGCL, a Maryland corporation may not indemnify a director or officer in a suit by or in the right of the corporation if such director or officer has been adjudged to be liable to the corporation. In accordance with the MGCL, the Host REIT Bylaws require it, as a condition to advancing expenses, to obtain (1) a written affirmation by the director or officer of his good faith belief that he has met the standard of conduct necessary for indemnification by Host REIT as authorized by the Host REIT Bylaws and (2) a written statement by or on his behalf to repay the amount paid or reimbursed by Host REIT if it shall ultimately be determined that the standard of conduct was not met.

Host REIT intends to enter into indemnification agreements with each of its directors and officers. The indemnification agreements will require, among other things, that Host REIT indemnify its directors and officers to the fullest extent permitted by law and advance to its directors or officers all related expenses, subject to reimbursement if it is subsequently determined that indemnification is not permitted.

The Partnership Agreement also provides for indemnification of Host REIT and its officers and directors to the same extent that indemnification is provided to officers and directors of Host REIT in the Host REIT Charter, and limits the liability of Host REIT and its officers and directors to the Operating Partnership and its respective partners to the same extent that the liability of the officers and directors of Host REIT to Host REIT and its stockholders is limited under the Host REIT Charter. See "Description of the Partnership Agreement and OP Units--Exculpation and Indemnification of Host REIT."

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers or persons controlling the registrant pursuant to the foregoing provisions, Host REIT has been informed that in the opinion of the Commission such indemnification is against public policy as expressed in the Securities Act and is therefore unenforceable.

### THE REIT CONVERSION

The transactions summarized below, together with the Restructuring Transactions, collectively constitute the transactions pursuant to which Host will restructure its business so that it will qualify as a REIT. If the required corporate (Board and stockholder) and partnership approvals for the various transactions are obtained and other conditions to the different steps in the REIT Conversion are satisfied or waived, these transactions are expected to occur at various times prior to the end of 1998 (or as soon thereafter as practicable). The Restructuring Transactions are expected to occur at the final stage of the REIT Conversion, although certain of the transactions comprising the REIT Conversion (such as the Partnership Mergers, the Private Partnership Transactions and the Blackstone Acquisition) may occur after the Restructuring Transactions to the extent they are consummated.

Although a number of the transactions comprising the REIT Conversion are expected to be consummated immediately prior to, or in certain instances immediately following, the Merger, the Merger will not be consummated unless the other conditions to the Merger have been satisfied or waived. In particular, Host's Board of Directors will have determined, among other things, that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur, and based on advice of counsel, that Host REIT can elect to be treated as a REIT for federal income tax purposes effective no later than the first full taxable year commencing after the REIT Conversion is completed (which might not be until the year commencing January 1, 2000 if the REIT Conversion is not completed prior to January 1, 1999). Consistent with the foregoing, Host intends to pursue the transactions constituting the REIT Conversion at least through the date of the Special Meeting of Host stockholders. If the Agreement is approved by Host stockholders at the Special Meeting, Host intends to continue pursuing those transactions constituting the REIT Conversion which have not yet been completed, including the Blackstone Acquisition (which is not expected to be consummated any earlier than December 29, 1998). In addition, if the Board of Directors has determined that the conditions to the Merger have been or likely will be satisfied or waived (and, in particular, that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur), the Board intends to declare the Initial E&P Distribution and enter into the Leases with the Lessees (which will be indirect, wholly-owned subsidiaries of Crestline). Under such circumstances, however, the Merger and other transactions comprising the REIT Conversion still might be delayed or possibly might never be consummated. Assuming the Agreement is approved by Host stockholders at the Special Meeting and the Host Board of Directors makes the determination described above, it is currently contemplated that the Merger would be consummated on or about December 29, 1998, subject to satisfaction or waiver of the remaining conditions. If, however, the Agreement is not approved by the Host stockholders at the Special Meeting or the Host Board of Directors does not make the requisite determinations, Host will continue to operate as a Delaware corporation, the REIT Conversion will not be completed at this time and the Initial E&P Distribution will not be made.

If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the effectiveness of Host REIT's election could be delayed until January 1, 2000, which would result in Host REIT continuing to pay substantial corporate-level income taxes in 1999 (which would reduce Host REIT's estimated cash distributions) and could cause the Blackstone Acquisition (which is conditioned, among other things, on consummation of the REIT Conversion by March 31, 1999 and Host REIT qualifying as a REIT for 1999) not to be consummated. In view of the complexity of the REIT Conversion, the number of transactions that must occur to complete the REIT Conversion and the time and expense involved, Host believes that it is advisable to complete the REIT Conversion, or as many of the transactions comprising the REIT Conversion as possible, as soon as practicable even if such completion is after January 1, 1999 (and, thus, a REIT election would not be effective until January 1, 2000). Moreover, completion of the REIT Conversion (or a significant number of the transactions comprising the REIT Conversion) could possibly facilitate another transaction (such as a merger into another entity) that would enable Host (or its successor) to elect to be treated as a REIT prior to January 1, 2000. If Host REIT's election to be taxed as a REIT is not effective on January 1, 1999, Host REIT intends to operate following the REIT Conversion in a manner that would permit it to qualify as a REIT at the earliest time practicable, and it might pursue a merger with another entity or another transaction that would

permit it to commence a new taxable year and elect REIT status prior to January 1, 2000, although no assurance can be given that the Company will enter into or consummate such other transaction or otherwise qualify as a REIT prior to January 1, 2000. Host REIT in any event would elect to be treated as a REIT for federal income tax purposes no later than its taxable year commencing January 1, 2000, assuming it so qualified.

### THE INITIAL E&P DISTRIBUTION

In order to qualify as a REIT for federal income tax purposes, among other things, Host REIT and/or Host, as its predecessor, must have distributed all of the accumulated E&P of Host to its stockholders in one or more taxable dividends prior to the end of the first full taxable year for which the REIT election of Host is effective, which currently is expected to be the taxable year commencing January 1, 1999 (but which might not be until the year beginning January 1, 2000).

In an effort to help accomplish the requisite distributions of the accumulated E&P of Host, Host or Host REIT will make the Initial E&P Distribution, consisting of one or more taxable distributions, to its stockholders in connection with the REIT Conversion. Such taxable distributions would consist of shares of common stock of Crestline and cash or other consideration in an amount to be determined. Although there is no assurance as to what form such other consideration comprising the Initial E&P Distribution will take, it is currently contemplated that it will include the Special Dividend, payable, at each stockholder's election, in cash or Host Common Stock (or Host REIT Common Stock if the Merger has occurred). If the Special Dividend is declared, stockholders of record on the Special Dividend record date will be given approximately 20 days to decide whether to take cash, Host Common Stock (or Host REIT Common Stock if the Merger has occurred), or a combination of cash and stock in payment of the Special Dividend. It is anticipated that the cash/stock election will be available on a per-share basis and that, once made, stockholders' elections will be irrevocable. Stockholders entitled to the Special Dividend who fail to make a timely election will receive shares of Host Common Stock (or Host REIT Common Stock if the Merger has occurred), subject to the Ownership Limit, in payment of the Special Dividend. In any event, cash would be paid in lieu of fractional shares, and the Special Dividend payment would be made promptly following expiration of the election period.

The aggregate value of the Crestline common stock and the cash or other consideration to be distributed to Host stockholders (and the Blackstone Entities) in connection with the Initial E&P Distribution is currently estimated to be approximately \$525 million to \$625 million (approximately \$2.10 to \$2.50 per share to the Host stockholders), of which approximately \$200 to \$300 million (or approximately \$.80 to \$1.20 per share) is expected to be represented by the Special Dividend. The actual amount of the Initial E&P Distribution will be based in part upon the estimated amount of accumulated E&P of Host as of the last day of its taxable year ending on or immediately following completion of the REIT Conversion. To the extent that the Initial E&P Distribution is not sufficient to eliminate Host's accumulated E&P, Host REIT will make one or more additional taxable distributions to its stockholders (in the form of cash or securities) prior to the last day of Host REIT's first full taxable year as a REIT (currently expected to be December 31, 1999 but which instead might be December 31, 2000) in an amount intended to be sufficient to eliminate such E&P, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

Limited Partners who elect to receive Host REIT Common Stock in exchange for OP Units in connection with the Partnership Mergers will not receive the Crestline common stock or any other portion of the Initial E&P Distribution, which will have been distributed before they become stockholders of Host REIT (approximately 25 trading days after the effective date of the Partnership Mergers). However, following receipt of such shares of Host REIT Common Stock, such Limited Partners would participate as stockholders of Host REIT in other dividends or distributions by Host REIT to holders of Host REIT Common Stock, including any additional taxable distributions necessary to eliminate Host's accumulated E&P to the extent the Initial E&P Distribution is not sufficient to eliminate such accumulated E&P. The Limited Partners who elect to retain OP Units also will participate in such additional distributions because distributions by the Operating Partnership to its partners (including Host REIT) would be the source of such additional distributions.

No holder of Host Common Stock will be required to pay any cash or other consideration to Host or Host REIT for shares of Crestline common stock received in the Initial E&P Distribution or to surrender or exchange their Host Common Stock or Host REIT Common Stock in order to receive shares of Crestline common stock or other cash or securities as part of the Initial E&P Distribution. See "The REIT Conversion--The Initial E&P Distribution."

In addition, following the Restructuring Transactions, the Blackstone Entities are entitled to receive a pro rata portion of the same consideration received by Host REIT's stockholders in connection with the Initial E&P Distribution, except to the extent the Blackstone Entities elected to receive additional OP Units in lieu thereof pursuant to the terms of the Blackstone Acquisition. The payment to the Blackstone Entities of Crestline common stock shares and other consideration is expected to be approximately \$90 to \$110 million of the aggregate Initial E&P Distribution of approximately \$525 to \$625 million if the REIT Conversion and the Blackstone Acquisition are consummated. See "Business and Properties--Blackstone Acquisition."

Following the Initial E&P Distribution, Crestline's principal assets will include the senior living assets of Host, which are expected to consist of 31 senior living communities, a 25% interest in the Swissotel management company acquired from the Blackstone Entities and the Lessees. Certain REITs have spun-off public operating companies to conduct certain activities which REITs are prohibited from conducting and have described such structure as a "paperclip" structure. There is no established definition of a "paper-clip" structure. While Host REIT and Crestline clearly expect to have a mutually beneficial, long-term relationship, they do not believe that their relationship should be characterized as a "paper-clip" structure because they will operate as separate public companies with independent business plans, there will be no overlap between officers and directors of the two companies (other than one officer of Host who will be a director but not an officer of Crestline), there are no rights of first refusal or other similar arrangements (other than the non-competition arrangements) with respect to future acquisitions between Host REIT and Crestline and they expect stockholders of the two companies to diverge over time. Crestline also will be engaged in the businesses of owning senior living communities and asset management of hotels, neither of which will be conducted by Host REIT. Crestline further intends to pursue leasing opportunities for both full-service and limited-service hotels with majority owners other than Host REIT.

The Host Board believes that the distribution of Crestline common stock to Host stockholders as part of the Initial E&P Distribution will provide those Host stockholders who continue to hold Crestline common stock with a separate identifiable interest in a diversified company that generates revenue from both its senior living business and its leasing business. Even under circumstances where the Crestline common stock is distributed to Host stockholders as part of the Initial E&P Distribution but the Merger or other transactions comprising the REIT Conversion are delayed or possibly never consummated, the Host Board believes that having existing leasing arrangements in place with Crestline could facilitate any subsequent efforts by Host to qualify as a REIT for federal income tax purposes (including efforts to pursue a merger with another entity or another transaction that would permit it to commence a new taxable year and elect REIT status prior to January 1, 2000).

## OTHER TRANSACTIONS COMPRISING THE REIT CONVERSION

In addition to the Restructuring Transactions and the Initial E&P Distribution, the REIT Conversion includes the following additional transactions:

- . Debt Refinancing. In August 1998, Host refinanced \$1.55 billion of outstanding senior notes through offers to purchase such debt securities for cash and a concurrent solicitation of consents to amend the terms of the debt securities to facilitate the transactions constituting the REIT Conversion. Host obtained the funds for this Senior Note Refinancing primarily from the issuance of new debt securities and the New Credit Facility. See "Business and Properties--Indebtedness."
- . Treatment of Convertible Preferred Securities. In the REIT Conversion, the Operating Partnership will assume primary liability for repayment of the \$567 million of convertible subordinated debentures of Host underlying the \$550 million of Convertible Preferred Securities. As the successor to Host, Host REIT also will be liable on the debentures and the debentures will become convertible into Host REIT

Common Stock, but the Operating Partnership will have primary responsibility for payment of the debentures, including all costs of conversion. Upon conversion by a Convertible Preferred Securities holder, Host REIT will issue shares of Host REIT Common Stock, which will be delivered to such holder. Upon the issuance of such shares by Host REIT, the Operating Partnership will issue to Host REIT a number of OP Units equal to the number of shares of Host REIT common Stock issued in exchange for the debentures. As a result of the distribution of Crestline common stock and any cash and other consideration to Host or Host REIT stockholders in connection with the Initial E&P Distribution, the conversion ratio of the Convertible Preferred Securities will be adjusted to take into account certain effects of the REIT Conversion. See "Management's Discussion and Analysis of Financial Condition and Results of Operations--Liquidity and Capital Resources."

- . The Partnership Mergers. Immediately following the Effective Date, each Partnership participating in the Partnership Mergers will merge with a subsidiary of the Operating Partnership. Such participating Partnerships will be the surviving entities of the Partnership Mergers and will continue in existence as indirect subsidiaries of the Operating Partnership. In the Partnership Mergers, each Limited Partner will receive a number of OP Units with a deemed value equal to the stated exchange value of his respective partnership interest. If a Limited Partner elects to receive Host REIT Common Stock or a Note in exchange for OP Units in connection with the Partnership Mergers, such Limited Partner will, upon receipt of his OP Units, tender (or be deemed to tender) all of such OP Units to Host REIT in exchange for an equal number of shares of Host REIT Common Stock or to the Operating Partnership in exchange for a Note with a specified principal amount. The general partners of the Partnerships, each of which is a wholly-owned, direct or indirect subsidiary of Host, and other subsidiaries of Host also will receive OP Units in exchange for their interests in the Partnerships, and the general partners will continue as wholly-owned direct or indirect subsidiaries of Host REIT. Any Partnership that does not participate in a Partnership Merger will continue as a separate partnership with its own assets and liabilities and with its current Limited Partners. There will be no change in its investment objectives, policies or restrictions or the fees or distributions payable to the applicable general partner or Manager. Each Partnership that does not participate in a Partnership Merger will remain subject to the terms of its current partnership agreement. The Operating Partnership would contribute some or all of the interests in certain of these Partnerships that it receives from Host and its subsidiaries to a Non-Controlled Subsidiary. In addition, the Operating Partnership has reserved the right to exclude any Partnership from participation in the REIT Conversion (even if the requisite percentage of Limited Partners has voted to approve the Partnership Merger and each of the other conditions to such Partnership Merger has been satisfied or waived) if the Operating Partnership determines, in its sole discretion, that such exclusion is in the best interests of the Operating Partnership. Any such Partnership that is so excluded shall be treated as a Partnership that does not participate in a Partnership Merger, as described above, and its Limited Partners will continue to hold their respective interests in the Partnerships.
- Restructuring of the Private Partnerships. The Operating Partnership will acquire the partnership interests from unaffiliated partners of four Private Partnerships in exchange for OP Units and, accordingly, will own all of the interests in those Private Partnerships. For the remaining Private Partnerships, (i) the Operating Partnership will be a partner in the partnership if the unaffiliated partners consent to a lease of the partnership's Hotel(s) to a Lessee or (ii) if the requisite consents to enter into a lease are not obtained, the Operating Partnership may transfer its interest in such partnership to a Non-Controlled Subsidiary. The determination of the action to be taken with respect to the Operating Partnership's interest in these Private Partnerships will be based primarily upon the character of the income therefrom under the REIT tax rules.

80

The partners in the following Private Partnerships will receive the estimated number of OP Units set forth below in connection with the REIT Conversion (assuming a price of \$12.50 per OP Unit):

	NEGOTIATED VALUE OF	NUMBER OF
ARTNERSHIP	OP UNITS	OP UNITS
HMC BN Limited Partnership	\$20,600,000	1,648,000
Ivy Street Hotel Limited Partnership	4,050,000	324,000
Times Square Marquis Hotel, L.P	7,499,000	599,920
HMC/RGI Hartford Limited Partnership	10.500.000	840.000

. The Blackstone Acquisition. Subject to various terms and conditions, the Operating Partnership expects to acquire from the Blackstone Entities ownership of, or controlling interests in, the Blackstone Hotels. In addition, Host REIT will acquire a 25% interest in the Swissotel management company from the Blackstone Entities, which Host REIT will transfer to Crestline. If the Blackstone Acquisition is consummated, the Operating Partnership expects to issue approximately 43.7 million OP Units (based upon a negotiated value of \$20.00 per OP Unit), assume debt and make cash payments totaling approximately \$862 million and distribute up to 18% of the shares of Crestline common stock and other consideration to the Blackstone Entities. Fifty percent of the OP Units issued in the Blackstone Acquisition will become redeemable pursuant to a Unit Redemption Right on July 1, 1999, an additional 25% will become redeemable on October 1, 1999 and the balance will become redeemable on January 1, 2000. Holders of OP Units issued in the Blackstone Acquisition will have registration rights under a shelf registration statement with respect to Host REIT Common Stock received in connection with the exercise of their redemption rights.

In connection with the Blackstone Acquisition, Host agreed to cause a person designated by Blackstone Real Estate Acquisitions L.L.C. ("Blackstone") to be appointed to serve as a director of Host (or a director of Host REIT following the Merger) and to continue to include a person designated by Blackstone in the slate of directors nominated by the Board of Directors for so long as Blackstone and its affiliates own at least 5% of the outstanding OP Units. Mr. Schreiber has been appointed to be the initial Blackstone designee. If the Blackstone Acquisition does not close, the Blackstone designee will resign.

Host also agreed to certain limitations on sales of the properties acquired in the Blackstone Acquisition lasting for five years after the REIT Conversion for 50% of the properties and for an additional five years for the remaining properties.

Each Blackstone Entity has agreed that, until the earlier of the fifth anniversary of the closing of the Blackstone Acquisition and the date on which the Blackstone Entities do not own, in the aggregate, more than 5% of the outstanding OP Units and Host REIT Common Stock, such Blackstone Entity will not, and will use its best efforts to cause its affiliates to not, directly or indirectly (i) subject to certain exceptions, acquire or agree to acquire beneficial ownership of any securities or partnership interests of Host REIT, the Operating Partnership or Crestline, if after giving effect thereto, such Blackstone Entity and its affiliates (together with the other members of any group (as defined in Section 13d-1 of the Exchange Act) of which any of them is part) would (A) directly or indirectly own more than 9.8% of any class of voting securities of such entity or more than 19.9% of the aggregate value of all outstanding voting securities of Host REIT and OP Units or (B) violate the ownership limitations or transfer restrictions set forth in the Host REIT Charter, the Partnership Agreement of the Operating Partnership or the Articles of Incorporation of Crestline, (ii) sell, transfer, pledge or otherwise dispose of any OP Units or any voting securities of Host REIT or Crestline in violation of such ownership limitations or transfer restrictions, (iii) participate in any proxy contest in opposition to the position taken by the directors or general partner, as applicable, of Host REIT, the Operating Partnership or Crestline, (iv) seek to cause a disposition (by way of merger, business combination, sale or otherwise) of a material portion of the assets or securities or partnership interests, or a change in the composition of the directors or management, of Host REIT, the Operating Partnership or Crestline or (v) initiate or propose to the holders of securities or partnership interests, as applicable, of Host REIT, the Operating Partnership or Crestline, or otherwise solicit their approval of, any proposal to be voted by such holders.

- . Contribution of Assets to Non-Controlled Subsidiaries. The Operating Partnership will organize the Non-Controlled Subsidiaries to hold various assets (not exceeding, in the aggregate, 20% by value of the assets of the Operating Partnership) contributed by Host and its subsidiaries to the Operating Partnership. The direct ownership of most of these assets by the Operating Partnership could jeopardize Host REIT's status as a REIT. These assets primarily will consist of partnership or other interests in hotels which are not leased, certain FF&E used in the Hotels and certain international hotels in which Host owns interests. In exchange for the contribution of these assets to the Non-Controlled Subsidiaries, the Operating Partnership will receive nonvoting common stock representing 95% of the total economic interests of the Non-Controlled Subsidiaries. In addition, the Operating Partnership and, prior to the Partnership Mergers, certain of the Partnerships (assuming they participate in the Partnership Mergers) will sell to a Non-Controlled Subsidiary an estimated \$180 million in value of personal property associated with certain Hotels for notes or cash that has been contributed or loaned to the Non-Controlled Subsidiary by the Operating Partnership, or a combination thereof. The Operating Partnership could not lease this personal property to the Lessees without potentially jeopardizing Host REIT's qualification as a REIT. The Non-Controlled Subsidiary will lease such personal property to the applicable Lessees. The Host Employee/Charitable Trust, a Delaware statutory business trust, and possibly certain other investors, will acquire all of the voting common stock representing the remaining 5% of the total economic interests, and 100% of the control, of each Non-Controlled Subsidiary. The indirect income beneficiaries of the Host Employee/Charitable Trust will be employees of Host REIT eligible to participate in Host's Comprehensive Stock Incentive Plan (excluding Directors of Host REIT and certain other highly compensated employees). Upon termination of the Host Employee/Charitable Trust, the residual assets, if any, are to be distributed to a charitable organization designated in its charter.
- Leases of Hotels. The Operating Partnership, its subsidiaries and its controlled partnerships, including the Partnerships participating in the Partnership Mergers, will lease virtually all of their Hotels to the Lessees pursuant to the Leases. See "Business and Properties--The Leases." The leased Hotels will be operated by the Lessees under their existing brand names pursuant to their existing long-term Management Agreements, which will be assigned to the Lessees for the terms of the applicable Leases but under which the Operating Partnership will remain obligated. See "Business and Properties--The Management Agreements."

Following the REIT Conversion, assuming the Full Participation Scenario, the organizational structure of Host REIT is expected to be as follows:

- (1) Represents Limited Partners and others who retain OP Units and do not elect to receive shares of Host REIT Common Stock or Notes; excludes Host and its subsidiaries.
- (2) Also will include Limited Partners in the Partnership Mergers who elect to receive shares of Host REIT Common Stock in exchange for the OP Units received in the Partnership Mergers. Immediately following the Restructuring Transactions and the distribution by Host or Host REIT of Crestline common stock to its stockholders and receipt of Crestline common stock by the Blackstone Entities, the stockholders of Crestline will consist of the stockholders (other than Limited Partners who elect to receive Host REIT Common Stock in connection with the Partnership Mergers) and the Blackstone Entities. The common ownership of the two public companies, however, will diverge over time.
- (3) Percentage ownership in the Operating Partnership assumes no Limited Partners elect to receive either Host REIT Common Stock or Notes in connection with the Partnership Mergers and that the price per share of Host REIT Common Stock is \$12.50 for purposes of the Partnership Mergers.
- (4) The Operating Partnership will own all or substantially all of the equity interests in the Partnerships participating in the Partnership Mergers, certain Private Partnerships and other Host subsidiaries that own Hotels, both directly and through other direct or indirect, wholly-owned subsidiaries of the Operating Partnership or Host REIT. Host will contribute its partial equity interests in the Partnerships that do not participate in the Partnership Mergers and those Private Partnerships whose partners have not elected to exchange their interests for OP Units to the Operating Partnership, and the Operating Partnership will either hold such partial interests or contribute them to the Non-Controlled Subsidiaries.

Ownership Interests in the Operating Partnership Following the Restructuring Transactions and the other transactions comprising the REIT Conversion. Following the Restructuring Transactions and the other transactions comprising the REIT Conversion, the Operating Partnership is expected to be owned as set forth below:

### OWNERSHIP OF THE OPERATING PARTNERSHIP

ENTITY	PERCENTAGE INTEREST(1)
Host REIT  Limited Partners of the Partnerships  Private Partnerships  Blackstone Entities	1.2
TOTAL	100.0% =====

(1) Assumes that all Partnerships participate in the Partnership Mergers, that the Blackstone Acquisition is consummated, that all Limited Partners in the Partnership Mergers elect to retain OP Units and that the price of an OP Unit is \$12.50 per share for purposes of the Partnership Mergers. The percentage interest of Host REIT will increase, and the percentage interest of the Limited Partners will decrease, if Limited Partners elect to receive Host REIT Common Stock or Notes in exchange for their OP Units in connection with the Partnership Mergers or if the price per OP Unit in the Partnership Mergers is greater than \$12.50. The percentage interest of Host REIT will decrease, and the percentage interest of the Limited Partners will increase, if the price per OP Unit in the Partnership Mergers is less than \$12.50.

### OPINION OF BT WOLFENSOHN, FINANCIAL ADVISOR TO HOST

BT Wolfensohn has acted as financial advisor to Host in connection with the REIT Conversion. On November , 1998, BT Wolfensohn delivered its written opinion to the Host Board of Directors to the effect that, as of the date of such opinion, based upon and subject to the assumptions made, matters considered and limits of the review undertaken by BT Wolfensohn, certain transactions to be effected by Host in connection with the conversion to a REIT described below, taken together, (the "REIT Transactions") were fair, from a financial point of view, to Host stockholders.

For the purposes of BT Wolfensohn's opinion, the REIT Transactions mean (i) the contribution by Host of its wholly-owned full-service hotels, certain interests in hotel partnerships and certain other assets to the Operating Partnership in exchange for units of limited and general partnership interest in the Operating Partnership and the assumption of liabilities, (ii) the reincorporation of Host from the State of Delaware to the State of Maryland by merging Host into Host REIT, (iii) the taxable Initial E&P Distribution by Host to its stockholders which Host has advised BT Wolfensohn will consist of shares of common stock of Crestline and a special dividend of approximately \$250 million in cash or, at the election of Host common stockholders, in Host REIT Common Stock, (iv) the assumption by the Operating Partnership of the debentures underlying the Convertible Preferred Securities and other indebtedness of Host and the related conversion price adjustment of such debentures, (v) the acquisition by merger of the Partnerships pursuant to the Partnership Mergers, (vi) the acquisition of partnership interests from unaffiliated partners of four limited partnerships that own one or more fullservice hotels and that are partially but not wholly-owned by Host or one of its subsidiaries in exchange for OP Units, (vii) the Blackstone Acquisition, (viii) the creation and capitalization of the taxable Non-Controlled Subsidiaries, and (ix) the leasing of virtually all of the full-service hotels owned or controlled by the Operating Partnership for initial terms ranging generally from seven to ten years to the Lessees, whereby the Lessees will operate the hotels under the existing brand names and pursuant to their existing management agreements.

The terms and conditions of the transactions comprising the REIT Transactions are more fully described in this Proxy Statement/Prospectus.

THE FULL TEXT OF BT WOLFENSOHN'S WRITTEN OPINION, DATED NOVEMBER , 1998 (THE "BT WOLFENSOHN OPINION"), WHICH SETS FORTH, AMONG OTHER THINGS, THE ASSUMPTIONS MADE, MATTERS CONSIDERED AND LIMITS ON THE REVIEW UNDERTAKEN BY BT WOLFENSOHN IN CONNECTION WITH THE OPINION, IS ATTACHED AS APPENDIX B TO THIS PROXY STATEMENT/PROSPECTUS AND IS INCORPORATED HEREIN BY REFERENCE. HOST STOCKHOLDERS ARE URGED TO READ THE BT WOLFENSOHN OPINION IN ITS ENTIRETY. THE SUMMARY OF THE BT WOLFENSOHN OPINION SET FORTH IN THIS PROXY STATEMENT/PROSPECTUS IS QUALIFIED IN ITS ENTIRETY BY REFERENCE TO THE FULL TEXT OF THE BT WOLFENSOHN OPINION.

In connection with BT Wolfensohn's role as financial advisor to Host, and in arriving at its opinion, BT Wolfensohn has, among other things, reviewed certain publicly available financial information and other information concerning Host and certain internal analyses and other information furnished to it by Host and Crestline. BT Wolfensohn also held discussions with the members of the senior management of Host and Crestline regarding the businesses and prospects of Host and Crestline. In addition, BT Wolfensohn (i) reviewed the reported prices and trading activity for the common stock of Host, (ii) compared certain financial and stock market information for Host with similar information for certain companies whose securities are publicly traded, (iii) reviewed certain public information of certain companies it deemed appropriate in analyzing Host and Crestline, and (iv) performed such other studies and analyses and considered such other factors as it deemed appropriate. BT Wolfensohn did not receive any instructions from Host or any affiliate of Host.

In preparing its opinion, BT Wolfensohn did not assume responsibility for the independent verification of, and did not independently verify, any information, whether publicly available or furnished to it, concerning Host and Crestline, including, without limitation, any financial information, forecasts or projections, considered in connection with the rendering of its opinion. Accordingly, for purposes of its opinion, BT Wolfensohn assumed

and relied upon the accuracy and completeness in all material respects of all such information. BT Wolfensohn did not conduct a physical inspection of any of the properties or assets, and did not prepare or obtain any independent evaluation or appraisal of any of the assets or liabilities of Host and Crestline. With respect to the financial forecasts and projections made available to BT Wolfensohn and used in its analysis, BT Wolfensohn has assumed that they have been reasonably prepared on bases reflecting the best currently available estimates and judgments of the management of Host and Crestline as to the matters covered thereby. In rendering its opinion, BT Wolfensohn expressed no view as to the reasonableness of such forecasts and projections, or the assumptions on which they are based. The BT Wolfensohn Opinion was necessarily based upon economic, market and other conditions as in effect on, and the information made available to BT Wolfensohn as of, the date of such opinion.

For purposes of rendering its opinion, BT Wolfensohn has assumed that the REIT Transactions will take place prior to or in connection with Host's election of REIT status in accordance with Sections 856 through 860 of the Code (the "REIT Rules"). For purposes of rendering its opinion, BT Wolfensohn has analyzed the REIT Transactions assuming alternatively that the Blackstone Acquisition is not consummated or is consummated in accordance with the terms of such transaction. BT Wolfensohn has also assumed that all material federal, state, local and other approvals and consents required in connection with the REIT Transactions will be obtained and that in connection with obtaining any necessary federal, state, local and other approvals and consents, or any amendments, modifications or waivers to any agreements, instruments or orders to which Host is a party or subject or by which it is bound, no limitations, restrictions or conditions will be imposed or amendments, modifications or waivers made that would have a material adverse effect on Host or Crestline or materially reduce the contemplated benefits of the REIT Transactions to Host. In addition, BT Wolfensohn has been advised by Host, and accordingly has assumed for purposes of its opinion, that the Initial E&P Distribution will be taxable to the Host stockholders.

BT Wolfensohn has also assumed that upon consummation of the REIT Transactions, Host will qualify for treatment as a REIT under the REIT Rules no later than the full year beginning January 1, 2000 (and BT Wolfensohn understands that Host will obtain an opinion, which will be based on certain assumptions and representations of Host and subject to certain qualifications, to that effect from its outside counsel prior to consummation of the REIT Transactions). At the request of the Host Board of Directors, BT Wolfensohn has not solicited any proposals from any third parties for the acquisition of any of the assets of Host or Crestline nor made any determination as to whether any such proposals could be obtained if solicited.

Set forth below is a brief summary of certain financial analyses performed by BT Wolfensohn in connection with its opinion.

Comparable Company Trading Analysis. BT Wolfensohn performed a comparable company trading analysis to derive a range of implied future equity values per share for Host, assuming it retains its existing corporate structure, and for Host and Crestline after giving effect to the proposed REIT Transactions. BT Wolfensohn reviewed market statistics and financial and operating information with respect to selected publicly traded companies considered by BT Wolfensohn to be comparable in certain respects to Host and Crestline. BT Wolfensohn selected publicly traded companies organized as C-Corporations engaged in the lodging industry (Choice Hotels International, Inc., Four Seasons Hotels Inc., Hilton Hotel Corporation, John Q. Hammons Hotels, L.P., Marriott International, Inc., Meristar Hotels & Resorts, Inc., Prime Hospitality Corporation, Promus Hotel Corporation, Red Roof Inns, Inc., Servico, Inc. and ShoLodge, Inc.; the "Lodging C-Corp Peer Group"), publicly traded lodging REITS (Boykin Lodging Company, Equity Inns, Inc., FelCor Lodging Trust Incorporated, Hospitality Properties Trust, Innkeepers USA Trust, Jameson Inns, Inc., RFS Hotel Investors, Inc. and Winston Hotels, Inc.; the "Lodging REIT Peer Group") and publicly traded healthcare REITs (American Health Properties, Inc., Health Care Property Investors, Inc., Health Care REIT, Inc., Health & Retirement Properties Trust, LTC Properties, Inc., Nationwide Health Properties, Inc. and Omega Healthcare Investors, Inc.; the "Healthcare REIT Peer Group" and, together with the Lodging C-Corp Peer Group and the Lodging REIT Peer Group, the "BT Wolfensohn Peer Groups"). The business composition, market positions and capital

86

closely comparable to Host and Crestline. In BT Wolfensohn's judgment, the Health Care REIT Peer Group is the most comparable set of companies for the purposes of valuing Crestline due to its ownership of senior living real estate assets. In using the Healthcare REIT Peer Group as a comparison, BT Wolfensohn took several differences into consideration, including that Crestline would be a taxable C-Corporation, that a significant portion of Crestline's earnings would be derived from leases on properties owned by Host, and that Crestline could potentially further diversify its revenue sources in the future. Based upon the market price of their common shares as of November 2, 1998 and the number of common shares outstanding, BT Wolfensohn calculated for each company in the BT Wolfensohn Peer Groups the market value of its common equity. For each C-Corporation engaged in the lodging industry, BT Wolfensohn added the market value of outstanding net debt, preferred stock instruments and capitalized leases (if applicable) in order to arrive at a total enterprise value ("TEV") for each company. BT Wolfensohn calculated the multiple

structures of these companies made them, in BT Wolfensohn's judgment, most

of each company's TEV to its historical earnings before interest, taxes, depreciation and amortization ("EBITDA") using publicly available information regarding each company's last twelve months of reported earnings. BT Wolfensohn also calculated the multiple of each company's stock price to its earnings per share ("EPS") for the latest twelve months and to its estimated earnings for calendar years 1998 and 1999 as reported by the Institutional Brokers Estimate System ("I/B/E/S"). I/B/E/S is a data service that monitors and publishes compilations of earning estimates by selected research analysts regarding companies of interest to institutional investors. BT Wolfensohn also calculated the current market value to current and projected Funds from Operations ratio for each company in the Lodging REIT Peer Group and Healthcare REIT Peer Group, using publicly available information regarding each company's latest twelve months Funds from Operations and Funds from Operations estimates for calendar years 1998 and 1999 as reported by I/B/E/S. Such multiples were then reviewed to determine the median, mean, high and low multiples for the BT Wolfensohn Peer Groups. BT Wolfensohn's analysis was not predicated upon a comparison of Host's stock price performance multiples to the BT Wolfensohn Peer Group multiples. In its professional opinion, BT Wolfensohn believed that Host's stock has been trading with the expectation that Host would convert to a REIT. BT Wolfensohn applied the resulting Lodging C-Corp Peer Group and Lodging REIT Peer Group multiple ranges (6.0x to 8.0x for EBITDA, 10x to 15x for earnings and 5.0x to 8.0x for Funds from Operations) to forecasted EBITDA, EPS and Funds from Operations of Host to determine a range of expected future equity values per share of Host under its existing corporate structure each year over the five-year period beginning January 1, 1999 and ending December 31, 2003. BT Wolfensohn also applied the resulting Lodging REIT Peer Group Funds from Operations multiple range (5.0x to 8.0x) to forecasted Funds from Operations of Host and the resulting Health Care REIT Peer Group Funds from Operations multiple range (5.0x to 8.0x) to forecasted Funds from Operations per share (less applicable corporate taxes) of Crestline to determine a range of expected future equity values per share of Host and Crestline each year over the five-year period beginning January 1, 1999 and ending December 31, 2003 after giving effect to the proposed REIT Transactions. In arriving at its conclusion as to the expected future trading values of Host and Crestline's stock, BT Wolfensohn also considered various factors affecting the trading values of specific members of the BT Wolfensohn Peer Groups as well as possible deviations from the indicated valuation range based upon specific factors related to Host and Crestline.

IRR Analysis. Utilizing trading values calculated by BT Wolfensohn from the above Comparable Company Trading Analysis, BT Wolfensohn analyzed the internal rate of return ("IRR") to a Host stockholder buying one share of Host stock on December 31, 1998 ("Beginning Stock Price") at an assumed price of \$15 per share and holding such share for a period of five years ending December 31, 2003 ("Ending Stock Price"), both under Host's existing corporate structure and after giving effect to the proposed REIT Transactions. The stock price on December 31, 2003 of Host under its existing corporate structure was computed using the methodology described above under Comparable Company Trading Analysis. The stock price on December 31, 2003 of Host based on the proposed REIT Transactions was computed using the methodology described above under Comparable Company Trading Analysis by adding the value of Host stock to the value of Crestline stock at such date. The IRR was then computed by taking into account the Beginning Stock Price, the Ending Stock Price, the assumed cash component of the Initial E&P Distribution and the annual dividends assumed to be received by a stockholder over the period. The analysis did not take into account taxes outside of corporate level taxes which would depend upon each individual shareholder's tax status. In each comparable case across a wide range of sensitivity analyses performed by BT Wolfensohn, the IRR giving effect to the REIT Transactions exceeded the IRR for Host under its existing corporate structure.

The foregoing summary describes all analyses and factors that BT Wolfensohn deemed material in rendering its fairness opinion and each of such analyses supported BT Wolfensohn's opinion. In addition, BT Wolfensohn performed other analyses and considered other factors in connection with preparing its opinion. The preparation of a fairness opinion is a complex process involving the application of subjective business judgment in determining the most appropriate and relevant methods of financial analysis and the application of those methods to the particular circumstances and, therefore, is not readily susceptible to summary description. BT Wolfensohn believes that its analyses must be considered as a whole and that considering any portion of such analyses and of the factors considered without considering all analyses and factors could create a misleading

87

view of the process underlying the opinion. In arriving at its fairness determination, BT Wolfensohn did not assign specific weights to any particular analyses.

In conducting its analyses and arriving at its opinion, BT Wolfensohn utilized a variety of generally accepted valuation methods. The analyses were prepared solely for the purpose of enabling BT Wolfensohn to provide its opinion to the Host Board of Directors as to the fairness to Host stockholders of the REIT Transactions and does not purport to be appraisals or necessarily reflect the prices at which businesses or securities actually may be sold, which are inherently subject to uncertainty. In connection with its analyses, BT Wolfensohn made, and was provided by Host and Crestline management with, numerous assumptions regarding

general market and economic trends and each company's estimated performance, much of which is beyond Host's and Crestline's control. The principal assumptions regarding Host's expected performance relate to occupancy rates, average daily rates ("ADRs") and the resulting revenue per available room ("REVPAR") for the period covered by the estimates. BT Wolfensohn assumed that demand would generally fall across the industry, resulting in a decline in occupancy rates of up to 100 basis points and a gradual decline of REVPAR growth from approximately 6% to 3% between 1999 and 2001. BT Wolfensohn also conducted a series of sensitivity analyses regarding the impact of market and economic performance variances on Host's and Crestline's businesses with occupancy rates, ADRs and REVPAR ranging between 75.1% to 78.6%, \$138 to \$172 and \$108 to \$133, respectively. Analyses based on estimates or forecasts of future results are not necessarily indicative of actual past or future values or results, which may be significantly more or less favorable than suggested by such analyses. Because such analyses are inherently subject to uncertainty, being based upon numerous factors or events beyond the control of Host and Crestline, or their respective advisors, neither Host nor Crestline nor BT Wolfensohn nor any other person assumes responsibility if future results or actual values are materially different from these forecasts or assumptions.

The terms of the REIT Transactions were approved by the Host Board of Directors. Although BT Wolfensohn provided advice to Host during the course of these discussions, the decision to enter into the REIT Transactions was solely that of the Host Board of Directors. As described above, the opinion and presentation of BT Wolfensohn to the Host Board of Directors were only one of a number of factors taken into consideration by the Host Board of Directors in making its determination to approve the REIT Transactions. BT Wolfensohn's written opinion was provided to the Host Board of Directors to assist it in connection with its consideration of the REIT Transactions and does not constitute a recommendation to any holder of Host Common Stock as to how to vote with respect to the Merger.

Host selected BT Wolfensohn as financial advisor in connection with the REIT Conversion based on BT Wolfensohn's qualifications, expertise, reputation and experience in mergers and acquisitions. BT Wolfensohn is engaged in the merger and acquisition and client advisory business of Bankers Trust (together with its affiliates, the "BT Group") and, for legal and regulatory purposes, is a division of BT Alex. Brown Incorporated, a registered broker dealer and member of the New York Stock Exchange. Since 1993, Host has paid BT Wolfensohn an annual retainer fee for providing ongoing merger, acquisition and corporate advisory services. This retainer fee totaled \$500,000 in 1998. Host separately retained BT Wolfensohn pursuant to a letter agreement dated April 16, 1998 (the "Engagement Letter"). As compensation for BT Wolfensohn's services in connection with the REIT Conversion, Host has paid BT Wolfensohn a cash fee of \$1 million and has agreed to pay an additional cash fee of \$4 million if the REIT Conversion is consummated. This amount will be reduced by BT Wolfensohn's annual retainer fee received for 1998. Regardless of whether the REIT Conversion is consummated, Host has agreed to reimburse BT Wolfensohn for reasonable fees and disbursements of BT Wolfensohn's counsel and all of BT Wolfensohn's reasonable travel and other out-of-pocket expenses incurred in connection with the REIT Conversion or otherwise arising out of the retention of BT Wolfensohn under the Engagement Letter. Host has also agreed to indemnify BT Wolfensohn and certain related persons to the full extent lawful against certain liabilities, including certain liabilities under the federal securities laws arising out of its engagement or the REIT Conversion. The fee to be paid to BT Wolfensohn as Host's financial advisor was negotiated on an arms-length basis with Host's management.

BT Wolfensohn is an internationally recognized investment banking firm experienced in providing advice in connection with mergers and acquisitions and related transactions. One or more members of the BT Group have, from time to time, provided investment banking, commercial banking (including extension of credit) and other financial services to Host or its affiliates for which it has received compensation. A member of the BT Group is the lead administrative agent for Host's \$1.25 billion bank facility and was the co-administrative agent for Host's recent bond tender and subsequent \$1.7 billion re-offering. The BT Group may actively trade securities of Host for its own account or the account of its customers and, accordingly, may from time to time hold a long or short position in such securities.

#### BUSINESS OF THE COMPANY

Host REIT was organized as a Maryland corporation on September 28, 1998 by Host to succeed to and continue the business of Host upon consummation of the Merger of Host with and into Host REIT pursuant to the Agreement. Host REIT has conducted no business to date other than that incident to the Merger and the other transactions comprising the REIT Conversion and has no material assets or liabilities.

Host REIT and the Operating Partnership have been formed primarily to continue, in an UPREIT structure, the full-service hotel ownership business currently conducted by Host. The primary business objectives of the Company will be to (i) achieve long-term sustainable growth in Funds From Operations (as defined below) and cash flow per share of Host REIT Common Stock, (ii) increase asset values by improving and expanding the initial Hotels, as appropriate, (iii) acquire additional existing and newly developed upscale and luxury full-service hotels in targeted markets (primarily focusing on downtown hotels in core business districts in major metropolitan markets and select airport and resort/convention locations), (iv) develop and construct upscale and luxury full-service hotels and (v) potentially pursue other real estate investments. Host REIT will operate as a self-managed and self-administered REIT and its operations will be conducted solely through the Operating Partnership and its subsidiaries. Following the REIT Conversion, the Hotels are expected to consist of approximately 125 hotels, representing approximately 59,000 rooms, located throughout the United States and Canada.

The Hotels will be generally operated under the Marriott, Ritz-Carlton, Four Seasons, Swissotel and Hyatt brand names and managed by subsidiaries of Marriott International and other companies. These brand names are among the most respected and widely recognized brand names in the lodging industry. Subsequent to the REIT Conversion, the Hotels will be leased by the Company to the Lessees and will be managed on behalf of the Lessees by subsidiaries of Marriott International and other Managers.

Host REIT will be the sole general partner of the Operating Partnership and will manage all aspects of the business of the Operating Partnership. This will include decisions with respect to (i) sales and purchases of hotels, (ii) the financing of the hotels, (iii) the leasing of the hotels and (iv) capital expenditures for the hotels (subject to the terms of the leases and the Management Agreements). Host REIT will be managed by its Board of Directors and will have no employees who are not also employees of the Operating Partnership.

Under current federal income tax law, REITs are not permitted to derive revenues directly from the operations of hotels. Therefore, the Company will lease the Hotels, through its subsidiaries, to the Lessees under the Leases. See "--The Leases" below. The Lessees will pay rent to the Company generally equal to a specified Minimum Rent plus, to the extent it would exceed Minimum Rent, Percentage Rent. The Lessees will operate the Hotels pursuant to the Management Agreements with the Managers. Each of the Management Agreements provides for certain base and incentive management fees, plus reimbursement of certain costs, as further described below. See "--The Management Agreements." Such fees and cost reimbursements will be the obligation of the Lessees and not the Company (although the obligation to pay such fees could adversely affect the ability of the Lessees to pay the required rent to the Company).

The Leases, through the Percentage Rent provisions, are designed to allow the Company to participate in any growth in room sales at the Hotels above specified levels, which management expects can be achieved through increases in room rates and occupancies. Although the economic trends affecting the hotel industry will be the major factor in generating growth in lease revenues, the abilities of the Lessees and the Managers will also have a material impact on future sales growth.

In addition to external growth generated by new acquisitions, the Company intends to carefully and periodically review its portfolio to identify opportunities to selectively enhance existing assets to improve operating performance through major capital improvements. The Company's Leases will provide the Company with the right to approve and finance major capital improvements.

#### **GENERAL**

The Company's primary focus is on the acquisition of upscale and luxury full-service hotel lodging properties. Since the beginning of 1994 through the date hereof, the Company has acquired 79 full-service hotels representing more than 36,000 rooms for an aggregate purchase price of approximately \$3.9 billion. Based upon data provided by Smith Travel Research, the Company believes that its full-service hotels outperform the industry's average occupancy rate by a significant margin and averaged 78.4% occupancy for 1997 compared to a 71.1% average occupancy for competing hotels in the upscale and luxury full-service segment of the lodging industry, the segment which is most representative of the Company's full-service hotels.

The upscale and luxury full-service segments of the lodging industry are benefiting from a favorable supply and demand relationship in the United States, especially in the principal sub-markets in which the Company operates, considering hotels of similar size and quality. Management believes that demand increases have primarily resulted from a strong domestic economic environment and a corresponding increase in business travel. In spite of increased demand for rooms, the room supply growth rate in the full-service segment has not similarly increased. Management believes that this slower increase in the supply growth rate in the full-service segment is attributable to many factors, including (i) the limited availability of attractive building sites for full-service hotels, (ii) the lack of available financing for new full-service hotel construction and (iii) the availability of existing fullservice properties for sale at a discount to their replacement cost. The relatively high occupancy rates of the Company's hotels, along with increased demand for full-service hotel rooms, have allowed the Managers of the Company's hotels to increase average daily room rates by selectively raising room rates and by replacing certain discounted group business with higher-rate group and transient business. As a result, on a comparable basis, room revenue per available room ("REVPAR") for the Company's full-service properties increased approximately 12.6% in 1997. The Company expects this supply/demand imbalance in the upscale and luxury full-service segments to continue, which should result in improved REVPAR at its hotel properties in the near term; however, there can be no assurance that such supply/demand imbalance will continue or that REVPAR will continue to improve.

#### **BUSINESS OBJECTIVES**

The Company's primary business objective is to increase its "Funds From Operations" or "FFO" (defined as net income (or loss) computed in accordance with generally accepted accounting principles ("GAAP"), excluding gains (or losses) from debt restructuring and sales of properties, plus real estate related depreciation and amortization, and after adjustments for unconsolidated partnerships and joint ventures) per share of Host REIT Common Stock and cash flow and enhance its value by:

- . Acquiring additional existing upscale and luxury full-service hotels, including Marriott and Ritz-Carlton hotels and other hotels operated by leading management companies such as Four Seasons, Hyatt and Swissotel, which satisfy the Company's investment criteria, including entering into joint ventures when the Company believes its return on investment will be maximized by doing so.
- . Developing new upscale and luxury full-service hotels, including Marriott and Ritz-Carlton hotels and other hotels operated by leading management companies such as Four Seasons, Hyatt and Swissotel, which satisfy the Company's investment criteria, employing transaction structures which mitigate risk to the Company.
- . Participating in the growth in sales for each of the hotels through leases which provide for the payment of rent based upon the lessees' gross hotel sales in excess of specified thresholds.
- . Enhancing existing hotel operations by funding selective capital improvements which are designed to increase gross hotel sales.

#### BUSINESS STRATEGY

The Company's primary business strategy is to continue to focus on maximizing the profitability of its existing full-service hotel portfolio and acquiring and, in limited cases, constructing, additional high quality, full-

service hotel properties, including controlling interests in joint ventures, partnerships or other entities holding such hotel properties. Although competition for acquisitions has increased, the Company believes that the upscale and luxury full-service segments of the market offer opportunities to acquire assets at attractive multiples of cash flow and at discounts to replacement value, including underperforming hotels which can be improved by conversion to the Marriott or Ritz-Carlton brands. The Company believes that the upscale and luxury full-service segments are very promising because:

- . There is a limited supply of new upscale and luxury full-service hotel rooms currently under construction in the sub-markets in which the Company operates. According to Smith Travel Research, from 1988 to 1991, upscale and luxury full-service room supply for the Company's competitive set increased an average of approximately 4% annually which resulted in an oversupply of rooms in the industry. However, this growth slowed to an average of approximately 1% from 1992 through 1997. Furthermore, the lead time from conception to completion of construction of a full-service hotel is generally three to five years or more in the markets in which the Company is principally pursuing acquisitions, which management believes will contribute to the continued low growth of room supply relative to the growth of room demand in the upscale and luxury full-service segments through 2000.
- . Many desirable hotel properties continue to be held by inadvertent owners such as banks, insurance companies and other financial institutions, both domestic and international, which are motivated and willing sellers. In recent years, the Company has acquired a number of properties from inadvertent owners at significant discounts to replacement cost, including luxury hotels operating under the Ritz-Carlton brand. While in the Company's experience to date, these sellers have been primarily U.S. financial organizations, the Company believes that numerous international financial institutions are also inadvertent owners of U.S. lodging properties and have only recently begun to dispose of such properties. The Company expects that there will be increased opportunities to acquire lodging properties from international financial institutions and expects to dedicate significant resources to aggressively pursue these opportunities.
- The Company believes that there are numerous opportunities to improve the performance of acquired hotels by replacing the existing hotel manager with Marriott International and converting the hotel to the Marriott brand. Based upon data provided by Smith Travel Research, the Company believes that Marriott-flagged properties have consistently outperformed the industry. Demonstrating the strength of the Marriott brand name, the average occupancy rate for the Company's comparable full-service properties was 79.4%, compared to the average occupancy rate of 71.1% for competing upscale and luxury full-service hotels for 1997. In addition, the Company's comparable properties generated a 29% REVPAR premium over its competitive set. Accordingly, management anticipates that any additional full-service properties acquired by the Company in the future and converted from other brands to the Marriott brand should achieve higher occupancy rates and average room rates than has previously been the case for those properties as the properties begin to benefit from Marriott's brand name recognition, reservation system and group sales organization. The Company intends to pursue additional full-service hotel acquisitions, some of which may be conversion opportunities. Sixteen of the Company's 79 acquired full-service hotels from the beginning of 1994 through the date hereof were converted to the Marriott brand following their acquisition.

91

. The Company intends to increase its pool of potential acquisition candidates by considering acquisitions of select non-Marriott and non-Ritz-Carlton hotels, subject to certain contractual limitations through October 8, 2000 imposed under the noncompetition agreement between Host and Marriott International (see "Certain Relationships and Related Transactions--Relationship Between Host and Marriott International-- Noncompetition Agreement"), that offer long-term growth potential and are consistent with the overall quality of its current portfolio. The Company will focus on upscale and luxury full-service properties in difficult to duplicate locations with high barriers to entry, such as hotels located in downtown, airport and resort/convention locations, which are operated by quality managers. In April 1998, the Company reached a definitive agreement with the Blackstone Entities to acquire interests in twelve upscale and luxury full-service hotels and a mortgage loan secured by a thirteenth hotel in the U.S. and certain other assets in a transaction valued at the time of the agreement, including the assumption of debt. The Company expects to pay approximately \$862 million in cash and assumed debt, issue approximately 43.7 million OP Units (based upon a negotiated value of \$20.00 per OP Unit) and

distribute up to 18% of the shares of Crestline common stock to the Blackstone Entities in exchange for the assets received from the Blackstone Entities. The Blackstone portfolio consists of two Ritz-Carltons, three Four Seasons (including one in which the Company's only interest will be a mortgage loan), one Grand Hyatt, three Hyatt Regencies and four Swissotel properties. See "--Blackstone Acquisition."

The Company believes it is well qualified to pursue its acquisition and development strategy. Management has extensive experience in acquiring and financing lodging properties and believes its industry knowledge, relationships and access to market information provide a competitive advantage with respect to identifying, evaluating and acquiring hotel assets.

During 1997, the Company acquired, or purchased controlling interests in, 17 full-service hotels, containing 8,624 rooms, for an aggregate purchase price of approximately \$765 million (including the assumption of approximately \$418 million of debt). The Company also completed the acquisition of the 504-room New York Marriott Financial Center, following the acquisition of the mortgage on the hotel for \$101 million in late 1996.

The Company holds minority interests and serves as a general partner or limited partner in various partnerships that own, as of the date hereof, an aggregate of 240 hotel properties, 20 of which are full-service properties, managed or franchised by Marriott International. In 1997, the Company acquired, or obtained controlling interests in, five affiliated partnerships, adding 10 hotels to its portfolio. In January, the Company acquired a controlling interest in Marriott Hotel Properties Limited Partnership, a Delaware limited partnership ("MHP"). MHP owns the 1,503-room Marriott Orlando World Center and a 50.5% interest in the 624-room Marriott Harbor Beach Resort. In April, the Company acquired a controlling interest in the 353-room Hanover Marriott. In the fourth quarter, the Company acquired the Chesapeake Hotel Limited Partnership ("CHLP"). CHLP owns the 430-room Boston Marriott Newton; the 681-room Chicago Marriott O'Hare; the 595-room Denver Marriott Southeast; the 588-room Key Bridge Marriott in Virginia; the 479-room Minnesota Airport Marriott; and the 221-room Saddle Brook Marriott in New Jersey. In December 1997, the Company obtained a controlling interest in the partnership that owns the 884-room Marriott's Desert Springs Resort and Spa in California.

In 1998, the Company acquired a controlling interest in the partnership that owns the Atlanta Marriott Marquis, containing 1,671 rooms, for approximately \$239 million, including the assumption of approximately \$164 million of mortgage debt. The Company also acquired a controlling interest in a partnership that owns three full-service hotels, containing a total of 1,029 rooms, for approximately \$50 million and the outstanding interest in the 289-room Park Ridge Marriott in New Jersey for \$24 million. More recently, the Company acquired the 281-room Ritz-Carlton, Phoenix for \$75 million, the 397-room Ritz-Carlton in Tysons Corner, Virginia for \$96 million and the 487-room Torrance Marriott for \$52 million. In the third quarter of 1998, the Company acquired the 308-room Ritz-Carlton, Dearborn for approximately \$65 million, the 336-room Ritz-Carlton, San Francisco for approximately \$161 million and the 404-room Memphis Marriott (which was converted to the Marriott brand upon acquisition) for approximately \$16 million. The Company is continually engaged in discussions with respect to other potential acquisition properties.

In addition to investments in partnerships in which it already held minority interests, the Company has been successful in adding properties to its portfolio through partnership arrangements with either the seller of the property or the incoming managers (typically Marriott International or a Marriott franchisee). During 1997, the Company acquired interests in five such partnerships which owned five full-service hotels, including the 197-room Waterford Hotel in Oklahoma City, Oklahoma; the 404-room Norfolk Waterside Marriott in Norfolk, Virginia; the 380-room Hartford/Farmington Marriott near Farmington, Connecticut; the 380-room former Manhattan Beach Radisson Plaza in Manhattan Beach, California; and the 299-room Ontario Airport Marriott in Ontario, California. The Waterford Hotel and the Manhattan Beach Radisson Plaza have been converted to the Marriott brand. As discussed above, in 1998, the Company acquired a controlling interest in a partnership that owns three hotels: the 359-room Albany Marriott in New York; the 350-room San Diego Marriott Mission Valley in California; and the 320-room Minneapolis Marriott Southwest in Minnesota. The Company has the financial flexibility and, due to its existing partnership investment portfolio, the administrative infrastructure in place to

accommodate such arrangements. The Company views this ability as a competitive advantage and expects to enter into similar arrangements to acquire additional properties in the future.

The Company believes there is a significant opportunity to acquire additional Ritz-Carlton hotels due to the Company's relationship with Marriott International and due to the number of Ritz-Carlton brand hotels currently owned by inadvertent owners. The Company also intends to purchase upscale and luxury full-service hotels with the intention of converting them to the Ritz-Carlton brand.

The Company currently owns six international properties, with 2,550 rooms, located in Canada and Mexico. The overbuilding and economic stress currently being experienced in some European and Pacific Rim countries may eventually lead to additional international acquisition opportunities. The Company will acquire international properties only when such acquisitions achieve satisfactory returns after adjustments for currency and country risks.

In addition to acquisitions, the Company plans to selectively develop new upscale and luxury full-service hotels in major urban markets and convention/resort locations with strong growth prospects, unique or difficult to duplicate sites, high barriers to entry for other new hotels and limited new supply. The Company intends to target only development projects that show promise of providing financial returns that represent a premium to acquisitions. In 1997, the Company announced that it will develop the 717-room Tampa Convention Center Marriott for \$104 million, including a \$16 million subsidy provided by the City of Tampa.

The Company may also expand certain existing hotel properties where strong performance and market demand exists. Expansions to existing properties creates a lower risk to the Company as the success of the market is generally known and development time is significantly shorter than new construction. The Company recently committed to add approximately 500 rooms and an additional 15,000 square feet of meeting space to the 1,503-room Marriott Orlando World Center.

#### HOTEL LODGING INDUSTRY

The upscale and luxury full-service segments of the lodging industry continue to benefit from a favorable cyclical imbalance in the supply/demand relationship in which room demand growth has exceeded supply growth, which has remained fairly limited. The lodging industry posted strong gains in revenues and profits in 1997, as demand growth continued to outpace additions to supply. The Company believes that upscale and luxury full-service hotel room supply growth will remain limited through at least 1998. Accordingly, the Company believes this supply/demand imbalance will result in improving occupancy and room rates which should result in improved REVPAR and operating profit.

Following a period of significant overbuilding in the mid-to-late 1980s, the lodging industry experienced a severe downturn. Since 1991, new hotel construction, excluding casino-related construction, has been modest and largely offset by the number of rooms taken out of service each year. Due to an increase in travel and an improving economy, hotel occupancy has grown steadily over the past several years and room rates have improved. The Company believes that room demand for upscale and luxury full-service properties will continue to grow at approximately the rate of the Gross Domestic Product ("GDP"), which recently has been growing at a rate of approximately 2% to 3%. Increased room demand should result in increased hotel occupancy and room rates. According to Smith Travel Research, upscale and luxury full-service occupancy for the Company and its competitive set grew in 1997 to 72.5% from 72.2% in 1996, while room rate growth continued to exceed inflation. While room demand has been rising, new hotel supply growth has been minimal. Smith Travel Research data shows that upscale and luxury full-service room supply increased an average of only 1% annually from 1991 through 1997. The increase in room demand and minimal growth in new hotel supply has also led to increased room rates. The Company believes that these recent trends will continue, with overall occupancy changing slightly and room rates increasing at more than one and one-half times the rate of inflation in 1998.

While the supply/demand relationship has generally remained favorable in the upper upscale and luxury markets in which the Company's properties operate, a number of new construction projects have been announced or commenced in 1998, particularly in suburban and smaller metropolitan markets where the Company does not have a significant presence. This growth in new supply, together with slowing demand growth, have served to reduce the rate of REVPAR growth at the Company's properties from year earlier growth rates. In part, as a reaction to a concern regarding the potential for lodging industry overbuilding as well as general economic concerns which have been heightened by the Asian crisis and other factors, the availability of capital to the lodging industry has diminished greatly in the second half of 1998. While this condition is having a favorable effect by curtailing construction of a number of potentially competing hotel projects, it is also limiting the Company's ability to grow through acquisitions.

The lodging industry is cyclical with operating results correlated highly to GDP. During recent months, a number of investment banks and economists have substantially reduced their estimates of growth in GDP through 1999. Should such estimates of diminished GDP growth prove accurate, the Company believes that REVPAR growth would be substantially reduced at its properties.

As a result of the overbuilding in the mid-to-late 1980s, many full-service hotels have not performed as originally planned. Cash flow has often not covered debt service requirements, causing lenders (e.g., banks, insurance companies and savings and loans) to foreclose and become "inadvertent owners" who are motivated to sell these assets. In the Company's experience to date, these sellers have been primarily U.S. financial organizations. The Company believes that numerous international financial institutions are also inadvertent owners of lodging properties and expects there will be increased opportunities to acquire lodging properties from international financial institutions. While the interest of inadvertent owners to sell has created attractive acquisition opportunities with strong current yields, the lack of supply growth and increasing room night demand should contribute to higher long-term returns on invested capital. Given the relatively long lead time to develop urban, convention and resort hotels, as well as the lack of project financing, management believes the growth in room supply in this segment will be limited, at least until the year 2000.

#### HOTEL LODGING PROPERTIES

The Company's lodging portfolio consists of 104 upscale and luxury full-service hotels with over 50,000 rooms. The Company's hotel lodging properties represent quality assets in the upscale and luxury full-service lodging segments. All but three of the Company's hotel properties are currently operated under the Marriott or Ritz-Carlton brand names.

The following tables set forth certain information with respect to the operations of the Hotels to be owned by the Company following the REIT Conversion on a historical and pro forma basis for fiscal year 1997 and for the First Three Quarters 1998.

# FISCAL YEAR 1997

								AVERAGE	
PARTNERSHIP	NO.	OF HOTEL	S NO.	OF ROOMS	HOTEL	REVENUES	OCCUPANCY	DAILY RATE	REVPAR
	-								
					(IN	THOUSANDS	)		
Atlanta Marguis(1)(2)		4		1 671	\$	0F 207	60.0%	¢107 06	Ф 00 OF
Atlanta Marquis(1)(2)		1 1		1,671 256	Ф	85,397	69.8% 83.2	\$127.36 146.83	\$ 88.95 122.14
Chicago Suites(2)		1		884		6,568 33,369	73.0	169.55	123.77
Desert Springs(2)(3)		1		353		,	80.8	123.55	99.82
Hanover(2)		6				6,735 26,699	76.4	102.97	78.63
MDAH(2)		2		1,692		,	80.3	155.44	124.84
MHP(2)(4)		4		2,127		75,211 69,014	80.7	133.75	107.91
MHP2(2)(5)		4 8		3,411					
PHLP(2)(6)				3,181		50,323	78.5	105.21	82.63
Blackstone Hotels		12		5,520		147,524	72.8	166.72	121.33
Host (historical)(7)		95		45,718		946,726	78.4	133.74	104.84
Host (pro forma)(7)(8)		127		59,118	3,	601,866	77.7	132.73	103.09
	FIRST THREE QUARTERS 1998								
				111.5					
								AVERAGE	
PARTNERSHIP	NO.	OF HOTEL	S NO.	OF ROOMS	HOTEL	REVENUES	OCCUPANCY	DAILY RATE	REVPAR
	-								
					(IN	THOUSANDS	)		
		_						4101 15	<b>.</b>
Atlanta Marquis(1)(2)		1		1,671	\$	58,625	68.9	\$131.43	\$ 90.51
Chicago Suites(2)		1		256		5,120	83.5	159.34	132.98
Desert Springs(2)(3)		1		884		80,334	75.9	182.77	138.63
Hanover(2)		1		353		5,238	75.7	138.41	104.81
MDAH(2)		6		1,692		26,351	76.6	113.82	87.20
MHP(2)(4)		2		2,127		61,245	83.3	161.15	134.17
MHP2(2)(5)		4		3,411		29,513	83.7	150.16	125.76
PHLP(2)(6)		8		3,181		37,238	79.3	112.28	89.01
Blackstone Hotels		12		5,520		327,885	74.2	176.17	130.77
Host (historical)(7)		101		49,019		921,864	79.1	139.48	110.33
Host (pro forma)(7)(8)		127		59,118	2,	665,023	78.4	141.61	111.05

- (1) Atlanta Marquis has an 80% residual interest in the Atlanta Marriott Marquis Hotel. Revenues represents sales generated by the Hotel.
- (2) Partnership is participating in the "consent solicitation" relating to the Partnership Mergers.
- (3) Subsequent to November 25, 1997, revenues reflect gross hotel sales. Prior to that date, revenues reflected hotel rental income.
- (4) Includes Marriott's Harbor Beach Resort, in which MHP owns a 50.5% interest.
- (5) Includes the Santa Clara Marriott, in which MHP2 owns a 50% interest and Host owns the remaining 50% interest.
- (6) Includes the Tampa Westshore Marriott and the Raleigh Crabtree Marriott, which are currently consolidated by Host. A subsidiary of Host provided 100% nonrecourse financing totaling approximately \$35 million to PHLP, in which Host owns the sole general partner interest, for the acquisition of these two hotels.
- (7) Includes the hotels owned by Desert Springs, Hanover, MHP and MHP2 for both fiscal year 1997 and First Three Quarters 1998 and Atlanta Marquis for First Three Quarters 1998 because the Company owned a controlling interest in such Partnerships for the periods indicated. Such Partnerships are part of the Partnership Mergers (if and to the extent consummated).
- (8) Includes the hotels owned by all Partnerships and Private Partnerships and the Blackstone Hotels, assuming the Full Participation Scenario (i.e., the transactions comprising the REIT Conversion, including the Blackstone Acquisition, occur, all Partnerships participate in the Partnership Mergers and no shares of Host REIT Common Stock or Notes are issued in the Partnership Mergers).

One commonly used indicator of market performance for hotels is room revenue per available room, or REVPAR, which measures daily room revenues generated on a per room basis. This does not include food and beverage or other ancillary revenues generated by the property. REVPAR represents the combination of the average daily room rate charged and the average daily occupancy achieved. The Company has reported annual increases in REVPAR since 1993.

To maintain the overall quality of the Company's lodging properties, each property undergoes refurbishments and capital improvements on a regularly scheduled basis. Typically, refurbishing has been

provided at intervals of five years, based on an annual review of the condition of each property. For the First Three Quarters 1998, First Three Quarters 1997, fiscal years 1997, 1996 and 1995, the Company spent \$113 million, \$86 million, \$131 million, \$87 million and \$56 million, respectively, on capital improvements to existing properties. As a result of these expenditures, the Company will be able to maintain high quality rooms at its properties.

The Company's hotels average nearly 500 rooms. Twelve of the Company's hotels have more than 750 rooms. Hotel facilities typically include meeting and banquet facilities, a variety of restaurants and lounges, swimming pools, gift shops and parking facilities. The Company's hotels primarily serve business and pleasure travelers and group meetings at locations in downtown and suburban areas, near airports and at resort convention locations throughout the United States. The properties are generally well situated in locations where there are significant barriers to entry by competitors including downtown areas of major metropolitan cities at airports and resort/convention locations where there are limited or no development sites. Marriott International serves as the manager for 88 of the 104 hotels owned by the Company and all but three are part of Marriott International's fullservice hotel system. The average age of the properties is 15 years, although several of the properties have had substantial, more recent renovations or major additions. In 1997, for example, the Company substantially completed a two-year \$30 million capital improvement program at the New York Marriott Marquis which included renovations to all guestrooms, refurbishment of ballrooms, restaurant updates and retail additions. In early 1998, the Company completed a \$15 million capital improvement program at the Denver Marriott Tech Center. The program included replacement of guestroom interiors, remodeling of the lobby, ballroom, meeting rooms and corridors, as well as renovations to the exterior of the building.

The chart below sets forth performance information for the Company's comparable hotels:

	FIRS	ST THREE	QUA	RTERS	FISCAL	YEAR
	1	1998		1997	1997	1996
COMPARABLE FULL-SERVICE HOTELS(1) Number of properties	\$	78 38,589 141.68 79.9% 113.27 7.9%	\$	131.51 79.8%	54 27,074 \$134.49 79.4% \$106.76 12.6%	\$121.58 78.0%

(1) Consists of the 78 properties owned by the Company for the entire First Three Quarters 1998 and First Three Quarters 1997, respectively, and the 54 properties owned by the Company for the entire 1997 and 1996 fiscal years, respectively, except for the 85-room Sacramento property, which is operated as an independent hotel. These properties, for the respective periods, represent the "comparable properties." Properties held for less than all of the periods discussed above, respectively, are not considered comparable.

The chart below sets forth certain performance information for the Company's hotels:

	FIRST THREE QUARTERS			FISCAL YEAR			
		1998		1997	1997	1996	1995
Number of properties		104		86	95	79	55
Number of rooms		50,064		41,171	45,718	37,210	25,932
Average daily rate(1)	\$	139.48	\$	132.14	\$133.74	\$119.94	\$110.30
Occupancy percentage(1)		79.1%		79.9%	78.4%	77.3%	75.5%
REVPAR(1)	\$	110.33	\$	105.57	\$104.84	\$ 92.71	\$ 83.32

<sup>(1)</sup> Excludes the information related to the 85-room Sacramento property, which is operated as an independent hotel.

Revenues in 1997 for nearly all of the Company's hotels were improved or comparable to 1996. This improvement was achieved through steady increases in customer demand, as well as yield management techniques applied by the manager to maximize REVPAR on a property-by-property basis. REVPAR for

comparable properties increased 12.6% for fiscal year 1997 as average room rates increased almost 11% and average occupancy increased over one percentage  $\frac{1}{2}$ point. Overall, this resulted in outstanding sales growth. Sales expanded at a 9% rate for comparable hotels and house profit margins increased by over two percentage points. REVPAR in 1997 for all of the Company's properties (including both comparable and non-comparable properties) increased 12.9% as average room rates increased over 11% and average occupancy increased over one percentage point. For the First Three Quarters 1998, REVPAR for comparable properties increased 7.9% as average room rates increased nearly 8% and average occupancy had no change. Sales for the First Three Quarters 1998 expanded at an 8% rate for comparable hotels and the house profit margin increased by one percentage point. REVPAR for the First Three Quarters of 1998 for all of the Company's properties increased 4.5% as average room rates increased nearly 6% and average occupancy decreased over one percentage point. The Company believes that its hotels consistently outperform the industry's average REVPAR growth rates. The relatively high occupancy rates of the Company's hotels, along with increased demand for upscale and luxury fullservice hotel rooms, allowed the managers of the Company's hotels to increase average room rates by selectively raising room rates and replacing certain discounted group business with higher-rate group and transient business. The Company believes that these favorable REVPAR growth trends should continue due to the limited new construction of full-service properties and the expected improvements from the conversion of seven properties to the Marriott brand in 1996 and 1997.

A number of the Company's full-service hotel acquisitions were converted to the Marriott brand upon acquisition--most recently the Coronado Island Marriott Resort and the Manhattan Beach Marriott were converted in the second half of 1997. The conversion of these properties to the Marriott brand is intended to increase occupancy and room rates as a result of Marriott International's nationwide marketing and reservation systems, its Marriott Rewards program, group sales force, as well as customer recognition of the Marriott brand name. The Marriott brand name has consistently delivered occupancy and REVPAR premiums over other brands. Based upon data provided by Smith Travel Research, the Company's comparable properties have an eight percentage point occupancy premium and a 29% REVPAR premium over its competitive set for 1997. The Company actively manages the conversions and, in many cases, has worked closely with the manager to selectively invest in enhancements to the physical product to make the property more attractive to guests or more efficient to operate. The invested capital with respect to these properties is primarily used for the improvement of common areas, as well as upgrading soft and hard goods (i.e., carpets, drapes, paint, furniture and additional amenities). The conversion process typically causes periods of disruption to these properties as selected rooms and common areas are temporarily taken out of service. Historically, the conversion properties have shown improvements as the benefits of Marriott International's marketing and reservation programs, group sales force and customer service initiatives take hold. In addition, these properties have generally been integrated into Marriott International's systems covering purchasing and distribution, insurance, telecommunications and payroll processing.

Following the REIT Conversion, the Lessees and the Managers will continue to focus on cost control in an attempt to ensure that hotel sales increases serve to maximize house and operating profit. While control of fixed costs serves to improve profit margins as hotel sales increase, it also results in more properties reaching financial performance levels that allow the Managers to share in the growth of profits in the form of incentive management fees. The Company believes this is a positive development as it strengthens the alignment of the Company's, the Lessees' and the Managers' interests.

During 1996, the Company completed its divestiture of limited service properties through the sale and leaseback of 16 Courtyard and 18 Residence Inn properties. These properties, along with 37 Courtyard properties sold and leased back during 1995, continue to be reflected in the Company's revenues and are managed by Marriott International under long-term management agreements. Following the REIT Conversion, these properties will be subleased to a subsidiary of Crestline. During 1997, limited service properties represented 2% of the Company's hotel EBITDA, compared to 5% in 1996, and the Company expects this percentage to continue to decrease as the Company continues to acquire primarily full-service properties.

The following table presents full-service hotel information by geographic region for fiscal year 1997:

GEOGRAPHIC REGION	NUMBER OF HOTELS	AVERAGE NUMBER OF GUEST ROOMS	AVERAGE OCCUPANCY	AVERAGE DAILY RATE	REVPAR	AGGREGATE COMPLETED RENOVATION EXPENDITURES (IN THOUSANDS)	
Atlanta	7	441	76.5%	\$131.69	\$100.74	\$ 4,115	
Florida	11	511	80.9	131.78	106.64	14,007	
Mid-Atlantic	12	364	76.1	111.71	85.00	3,477	
Midwest	10	418	74.3	107.65	79.99	2,751	
New York	10	708	84.7	173.85	147.22	15,232	
Northeast	7	367	75.2	96.75	72.72	9,260	
South Central	15	525	76.5	120.81	92.39	15,190	
Western	21	519	79.5	140.07	111.39	19,806	
Latin America	2	436	62.7	129.54	81.17	290	
Average-all regions		485	78.4	133.74	104.84		

# HOTEL PROPERTIES

The following table sets forth, as of the date hereof, the location and number of rooms relating to each of the Company's hotels. All of the properties are operated under Marriott brands by Marriott International, unless otherwise indicated.

LOCATION	ROOMS
Alabama	
Grand Hotel Resort and Golf Club	306
Scottsdale Suites	251
The Ritz-Carlton, Phoenix (1)	281
California Coronado Island Resort (2)(6)	300
Costa Mesa Suites	253
Desert Springs Resort and Spa (3)(4)	884
Manhattan Beach (5)(6)	380
Marina Beach (6) Newport Beach	368 570
Newport Beach Suites	250
Ontario Airport (7)	299
Sacramento Airport (6)(8)	85
San Diego Mission Valley (9)	350
San Francisco Airport	684
San Francisco Fisherman's Wharf (10)	285
San Francisco Moscone Center (6)	368
Santa Clara (6)	754
The Ritz-Carlton, Marina del Rey (1)(6)(11)	306
The Ritz-Carlton, San Francisco (1)	336
TorranceColorado	487
Denver Southeast (6)(12)	595
Denver Tech Center	625
Denver West (6)	307 349
Connecticut	349
Hartford/Farmington	380
Hartford/Rocky Hill (6)	251
Florida Fort Lauderdale Marina	580
Harbor Beach Resort (3)(4)(6)	624
Jacksonville (6)(9)	256
Miami Airport (6)	782
Orlando World Center (3)(4)	279
Singer Island (Holiday Inn) (8)	222
Tampa Airport (6)	295
Tampa Westshore (6)(13) The Ritz-Carlton, Naples (1)	309 463
Georgia	
Atlanta Marriott Marquis (3)(4)	
Atlanta Midtown Suites (6)	254 222
Atlanta Northwest	400
Atlanta Perimeter (6)	400
JW Marriott Hotel at Lenox (6)	371
The Ritz-Carlton, Atlanta (1) The Ritz-Carlton, Buckhead (1)	447 553
Illinois	000
Chicago/Deerfield Suites	248
Chicago/Downers Grove Suites	254
LOGATTON	200110
LOCATION	ROOMS
Chicago/Downtown Courtyard	334
Chicago O'Hare (6)(12)	681
South Bend (6)	300
Louisiana	
New Orleans (4)	1,290
Maryland Bethesda (6)	407
Gaithersburg/Washingtonian Center	284
Massachusetts	
Boston/Newton (3)	430
Michigan Detroit Romulus	245
The Ritz Carlton, Dearborn (1)	306
Minnesota	
Minneapolis/Bloomington (12)	479 583
Minneapolis Southwest (9)	320

Missouri	
·	382
Kansas City Airport (6)St. Louis Pavilion (6)	302 672
	0/2
New Hampshire	254
Nashua	251
New Jersey	
Hanover (3)(4)	353
Newark Airport (6)	590
Park Ridge (6)	289
Saddle Brook (6)(12)	221
New York	
Albany (9)	359
New York Marriott Financial Center (14)	504
New York Marriott Marquis (6)	
Marriott World Trade Center (6)	820
North Carolina	
Charlotte Executive Park (10)	298
Raleigh Crabtree Valley (13)	375
Oklahoma	
Oklahoma City	354
Oklahoma City Waterford (5)	197
Oregon	
Portland	503
Pennsylvania	
Philadelphia (Convention Center) (6)	1,200
Philadelphia Airport (6)	419
Pittsburgh City Center (6)(10)	400
Tennessee	
Memphis (2)(6)	404
Texas	
Dallas/Fort Worth	492
Dallas Quorum (6)	547
El Paso (6)	296
Houston Airport (6)	566
JW Marriott Houston (6)	503
` '	

LOCATION	ROOMS
Plaza San Antonio (6)(10)	252
San Antonio Rivercenter (4)(6)	999
San Antonio Riverwalk (6)	500
Utah	000
Salt Lake City (6)	510
Virginia	010
Dulles Airport (6)	370
Key Bridge (6)(12)	588
Norfolk Waterside (6)(7)	404
Pentagon City Residence Inn	300
The Ritz-Carlton, Tysons Corner (6)	397
Washington Dulles Suites	254
Westfields	335
Williamsburg	295
Washington, D.C.	
Washington Metro Center	456
Canada	
Calgary	380
Toronto Airport (15)	423
Toronto Eaton Centré (6)	459
Toronto Delta Meadowvale (8)	374
Mexico	
Mexico City Airport (15)	600
JW Marriott Hotel, Mexico City (15)	314
TOTAL	50,067
	=====

Properties that are currently not consolidated by Host and are subject to the Partnership Mergers ("MDAH" refers to Marriott Diversified America Hotels, L.P., a Delaware limited partnership; "Chicago Suites" refers to Mutual Benefit Chicago Marriott Suite Hotel Partners, L.P., a Rhode Island limited partnership; and "PHLP" refers to Potomac Hotel Limited Partnership, a Delaware limited partnership):

HOTEL 	STATE	ROOMS
MDAH Fairview Park (6) Dayton Research Triangle Park Detroit Marriott Southfield	Ohio North Carolina	395 399 224 226
HOTEL	STATE	ROOMS
Detroit Marriott Livonia		224 224  1,692
Chicago Suites Marriott O'Hare Suites (6)	Illinois	256
PHLP Albuquerque (6) Greensboro-High Point (6) Houston Medical Center (6) Miami Biscayne Bay (6) Marriott Mountain Shadows Resort Seattle SeaTac Airport	North Carolina Texas Florida Arizona	386 605 337 459  2,497
TOTAL		4, 445 =====

Properties that are included in the Blackstone portfolio are as follows:

HOTEL	STATE	ROOMS
Four Seasons, Atlanta (8)	J	246
Four Seasons, Philadelphia (8)		365
Grand Hyatt, Atlanta (8)	Georgia	439
Hyatt Regency, Burlingame (8)	California	793
Hyatt Regency, Cambridge (8)	Massachusetts	469
Hyatt Regency, Reston (8)	Virginia	514
Swissotel, Atlanta (8)	Georgia	348

Swissotel, Boston (8)	Illinois New York	498 630 494 449
The Ritz-Carlton, Boston (1)		275
TOTAL		5,520 =====

. .......

- (1) Property is operated as a Ritz-Carlton. The Ritz-Carlton Hotel Company, L.L.C. manages the property and is wholly owned by Marriott International.
- (2) This property was acquired by the Company and converted to the Marriott brand in 1997 or 1998.
- (3) The Company acquired a controlling interest in the partnership that owns this property in 1997 or 1998. The Company previously owned a general partner interest in the partnership.
- (4) Property is held within a partnership and is currently consolidated by  $\operatorname{Host}$ .
- (5) The Company acquired a controlling interest in the newly-formed partnership that owns this property in 1997. The property was converted to the Marriott brand and is operated as a Marriott franchised property.
- (6) The land on which the hotel is built is leased under one or more longterm lease agreements.
- (7) The Company acquired a controlling interest in the newly-formed partnership that owns this property in 1997. The property is operated as a Marriott franchised property.
- (8) Property is not operated under the Marriott brand and is not managed by Marriott International.
- (9) The Company acquired a controlling interest in the partnership that owns this property in 1998. The property will be operated as a Marriott franchised property.
- (10) Property is operated as a Marriott franchised property.
- (11) Property was acquired by the Company in 1997.
- (12) The Company acquired the partnership that owns this property in 1997. The Company previously owned a general partner interest in the partnership.

- (13) Property is owned by PHLP. A subsidiary of the Company provided 100% nonrecourse financing totaling approximately \$35 million to PHLP, in which the Company owns the sole general partner interest, for the acquisition of these two hotels. The Company consolidates these properties in the accompanying financial statements.
- (14) The Company completed the acquisition of this property in early 1997. The Company previously had purchased the mortgage loan secured by the hotel in late 1996.
- (15) Property will be transferred to the Non-Controlled Subsidiary in conjunction with the REIT Conversion and no longer consolidated by the Company.

#### 1998 ACQUISITIONS

In January 1998, the Company acquired an additional interest in Atlanta Marriott Marquis II Limited Partnership, a Delaware limited partnership, which owns an interest in the 1,671-room Atlanta Marriott Marquis Hotel, for approximately \$239 million, including the assumption of approximately \$164 million of mortgage debt. The Company previously owned a 1.3% general and limited partnership interest. In March 1998, the Company acquired a controlling interest in the partnership that owns three hotels: the 359-room Albany Marriott, the 350-room San Diego Marriott Mission Valley and the 320-room Minneapolis Marriott Southwest for approximately \$50 million. In the second quarter of 1998, the Company acquired the partnership that owns the 289-room Park Ridge Marriott in Park Ridge, New Jersey for \$24 million. The Company previously owned a 1% managing general partner interest and a note receivable interest in such partnership. In addition, the Company acquired the 281-room Ritz-Carlton, Phoenix for \$75 million, the 397-room Ritz-Carlton in Tysons Corner, Virginia for \$96 million and the 487-room Torrance Marriott near Los Angeles, California for \$52 million. In the third quarter of 1998, the Company acquired the 308-room Ritz-Carlton, Dearborn for approximately \$65 million, the 336-room Ritz-Carlton, San Francisco for approximately \$161 million and the 404-room Memphis Crowne Plaza (which was converted to the Marriott brand upon acquisition) for approximately \$16 million. In April 1998, the Company, through the Operating Partnership, entered into an agreement to acquire certain assets from various affiliates of The Blackstone Group. See "--Blackstone Acquisition."

#### BLACKSTONE ACQUISITION

In April 1998, the Company reached a definitive agreement with the Blackstone Entities to acquire ownership of, or controlling interests in, twelve hotels and two mortgage loans, one secured by one of the acquired hotels and one secured by an additional hotel. In addition, the Company will acquire a 25% interest in Swissotel Management (USA) L.L.C., which operates five Swissotel hotels in the United States, from the Blackstone Entities, which the Company will transfer to Crestline in connection with the Initial E&P Distribution of Crestline common stock to the Company's stockholders and the Blackstone Entities. If the Blackstone Acquisition is consummated, the Operating Partnership expects to issue approximately 43.7 million OP Units (based upon a negotiated value of \$20.00 per OP Unit), assume debt and make cash payments totaling approximately \$862 million and distribute up to 18% of the shares of Crestline common stock and other consideration to the Blackstone Entities. The consideration received by the Blackstone Entities was determined through negotiations between the Company and Blackstone and was not based upon appraisals of the assets. Each OP Unit will be exchangeable for one share of Host REIT Common Stock (or its cash equivalent, at the Company's election). Upon completion of the Blackstone Acquisition and the REIT Conversion, the Blackstone Entities will own approximately 16% of the outstanding OP Units. John G. Schreiber, co-chairman of the Blackstone Real Estate Partners' investment committee, has joined the Board of Directors of the Company.

The Blackstone portfolio is one of the premier collections of hotel real estate properties. It includes: The Ritz-Carlton, Amelia Island (449 rooms); The Ritz-Carlton, Boston (275 rooms); Hyatt Regency Burlingame at San Francisco Airport (793 rooms); Hyatt Regency Cambridge, Boston (469 rooms); Hyatt Regency Reston, Virginia (514 rooms); Grand Hyatt Atlanta (439 rooms); Four Seasons Philadelphia (365 rooms); Four Seasons Atlanta (246 rooms); The Drake (Swissotel) New York (494 rooms); Swissotel Chicago (630 rooms); Swissotel Boston (498 rooms) and Swissotel Atlanta (348 rooms). Additionally, the transaction includes: the first mortgage loan on the Four Seasons Beverly Hills (285 rooms); two office buildings in Atlanta--the offices at The Grand (97,879 sq. ft.) and the offices at the Swissotel (67,110 sq. ft.); and a 25% interest in the Swissotel U.S. management company (which will be transferred to Crestline).

At the closing of the Blackstone Acquisition, the Blackstone portfolio will be contributed to the Company and its hotels will be leased to subsidiaries of Crestline and will continue to be managed on behalf of the Lessees under their existing management agreements. The Company's acquisition of the Blackstone portfolio is subject to certain conditions, including the REIT Conversion being consummated by March 31, 1999 and Host REIT qualifying as a REIT for 1999 (which condition may not be satisfied if the REIT Conversion is not completed prior to January 1, 1999).

#### INVESTMENTS IN AFFILIATED PARTNERSHIPS

The Company and certain of its subsidiaries also manage the Company's partnership investments and conduct the partnership services business. As such, as of the date hereof, the Company and/or its subsidiaries own an investment in, and generally serve as a general partner or managing general partner for, 18 unconsolidated partnerships which collectively own 20 Marriott full-service hotels, 120 Courtyard hotels, 50 Residence Inns and 50 Fairfield Inns. In addition, the Company holds notes receivable (net of reserves) from partnerships totaling approximately \$23 million at January 2, 1998. Thirteen of the 20 full-service hotels owned by the unconsolidated partnerships will be acquired by the Company in connection with the REIT Conversion.

As the managing general partner of these partnerships, the Company and its subsidiaries are responsible for the day-to-day management of partnership operations, which includes payment of partnership obligations from partnership funds, preparation of financial reports and tax returns and communications with lenders, limited partners and regulatory bodies. The Company or its subsidiaries are reimbursed for the cost of providing these services subject to limitations in certain cases.

Hotel properties owned by the unconsolidated partnerships generally were acquired from the Company or its subsidiaries in connection with limited partnership offerings. These hotel properties are currently operated under management agreements with Marriott International. As the managing general partner of such partnerships, the Company or its subsidiaries oversee and monitor Marriott International's performance pursuant to these agreements.

The Company's interests in these partnerships range from 1% to 50%. Cash distributions provided from these partnerships are tied to the overall performance of the underlying properties and the overall level of debt owed by the partnership. Partnership distributions to the Company were \$1 million for the First Three Quarters 1998, \$4 million for the First Three Quarters 1997, \$5 million in each of 1997 and 1996 and \$3 million in 1995. All partnership debt is nonrecourse to the Company and its subsidiaries, except that the Company is contingently liable under various guarantees of debt obligations of certain of these partnerships. Such commitments are limited in the aggregate to \$60 million at January 2, 1998. Subsequent to year-end, such maximum commitments were reduced to \$20 million in connection with the refinancing and acquisition of a controlling interest in the Atlanta Marriott Marquis. In most cases, fundings of such guarantees represent loans to the respective partnerships.

## MARKETING

As of the date hereof, 88 of the Company's 104 hotel properties are managed by Marriott International as Marriott or Ritz-Carlton brand hotels. Thirteen of the 16 remaining hotels are operated as Marriott brand hotels under franchise agreements with Marriott International. The Company believes that these Marriott-managed and franchised properties will continue to enjoy competitive advantages arising from their participation in the Marriott International hotel system. Marriott International's nationwide marketing programs and reservation systems as well as the advantage of the strong customer preference for Marriott brands should also help these properties to maintain or increase their premium over competitors in both occupancy and room rates. Repeat guest business in the Marriott hotel system is enhanced by the Marriott Rewards program, which expanded the previous Marriott Honored Guest Awards program. Marriott Rewards membership includes more than 7.5 million members

The Marriott reservation system provides Marriott reservation agents complete descriptions of the rooms available for sale and up-to-date rate information from the properties. The reservation system also features

connectivity to airline reservation systems, providing travel agents with access to available rooms inventory for all Marriott and Ritz-Carlton lodging properties. In addition, software at Marriott's centralized reservations centers enables agents to immediately identify the nearest Marriott or Ritz-Carlton brand property with available rooms when a caller's first choice is fully occupied.

#### COMPETITION

The Company's hotels compete with several other major lodging brands in each segment in which they operate. Competition in the industry is based primarily on the level of service, quality of accommodations, convenience of locations and room rates. Although the competitive position of each of the Company's hotel properties differs from market to market, the Company believes that its properties compare favorably to their competitive set in the markets in which they operate on the basis of these factors. The following table presents key participants in segments of the lodging industry in which the Company competes:

SEGMENT	REPRESENTATIVE PARTICIPANTS

#### RELATIONSHIP WITH HM SERVICES

On December 29, 1995, the Company distributed to its stockholders through a special dividend (the "HMS Special Dividend") all of the outstanding shares of common stock of Host Marriott Services Corporation ("HM Services"), formerly a direct, wholly-owned subsidiary of the Company which, as of the date of the HMS Special Dividend, owned and operated the food, beverage and merchandise concessions at airports, on tollroads and at stadiums and arenas and other tourist attractions. The HMS Special Dividend provided Company stockholders with one share of common stock of HM Services for every five shares of Company common stock held by such stockholders on the record date of December 22,

For the purpose of governing certain of the ongoing relationships between the Company and HM Services after the HMS Special Dividend, and to provide an orderly transition, the Company and HM Services have entered into various agreements, including agreements to (i) allocate certain responsibilities with respect to employee compensation, benefit and labor matters; (ii) define the respective parties' rights and obligations with respect to deficiencies and refunds of federal, state and other income or franchise taxes relating to the Company's businesses for tax years prior to the HMS Special Dividend and with respect to certain tax attributes of the Company after the HMS Special Dividend; (iii) provide certain administrative and other support services to each other for a transitional period on an as-needed basis; and (iv) to provide for the issuance of HM Services common stock in connection with the exercise of certain outstanding warrants to purchase shares of Company common stock.

# RELATIONSHIP WITH MARRIOTT INTERNATIONAL; MARRIOTT INTERNATIONAL DISTRIBUTION

Prior to October 8, 1993, the Company was named "Marriott Corporation." In addition to conducting its existing hotel ownership business and the business of HM Services (prior to its distribution to stockholders through the Special Dividend), Marriott Corporation engaged in lodging and senior living services management, timeshare resort development and operation, food service and facilities management and other contract services businesses (the "Management Business"). On October 8, 1993, the Company completed the Marriott International Distribution (as defined herein). Marriott International conducts the Management Business as a separate publicly traded company.

The Company and Marriott International have entered into agreements which provide, among other things, for Marriott International to (i) manage or franchise various hotel properties owned or leased by the Company, (ii) advance up to \$225 million to the Company under the Marriott International line of credit, which was

terminated in 1997, (iii) provide first mortgage financing of \$109 million for the Philadelphia Marriott Hotel, which was repaid in December 1996, (iv) provide financing for certain Company acquisitions, (v) guarantee the Company's performance in connection with certain loans or other obligations and (vi) provide certain limited administrative services. The Company views its relationship with Marriott International as providing various advantages, including access to high quality management services, strong brand names and superior marketing and reservation systems.

Marriott International has the right to purchase up to 20% of the voting stock of the Company if certain events involving a change of control (or potential change of control) of the Company occur, subject to certain limitations (including a limitation effective after the REIT Conversion intended to help protect the qualification of Host REIT as a REIT). See "Certain Relationships and Related Transactions--Relationship Between Host and Marriott International."

#### **EMPLOYEES**

Currently, the Company and its subsidiaries collectively have approximately 225 corporate employees, and approximately 300 other employees (primarily employed at one of its non-U.S. hotels) which are covered by collective bargaining agreements that are subject to review and renewal on a regular basis. The Company believes that it has good relations with its labor unions and has not experienced any material business interruptions as a result of labor disputes. Following the REIT Conversion, the Company expects to have approximately 175 employees. The balance of the Company's current employees are expected to become employees of Crestline following the REIT Conversion.

#### ENVIRONMENTAL AND REGULATORY MATTERS

Under various federal, state and local environmental laws, ordinances and regulations, a current or previous owner or operator of real property may be liable for the costs of removal or remediation of hazardous or toxic substances on, under or in such property. Such laws may impose liability whether or not the owner or operator knew of, or was responsible for, the presence of such hazardous or toxic substances. In addition, certain environmental laws and common law principles could be used to impose liability for release of asbestos-containing materials ("ACMs"), and third parties may seek recovery from owners or operators of real properties for personal injury associated with exposure to released ACMs. Environmental laws also may impose restrictions on the manner in which property may be used or business may be operated, and these restrictions may require expenditures. In connection with its current or prior ownership or operation of hotels, the Company may be potentially liable for any such costs or liabilities. Although the Company is currently not aware of any material environmental claims pending or threatened against it, no assurance can be given that a material environmental claim will not be asserted against the Company.

## LEGAL PROCEEDINGS

Following the Restructuring Transactions and the other transactions comprising the REIT Conversion, the Operating Partnership will assume all liability arising under legal proceedings filed against Host and will indemnify Host REIT as to all such matters. Host and the other defendants believe all of the lawsuits in which Host is a defendant, including the following lawsuits, are without merit and the defendants intend to defend vigorously against such claims. However, no assurance can be given as to the outcome of any of the lawsuits.

Texas Multi-Partnership Lawsuit. On March 16, 1998, limited partners in several limited partnerships sponsored by Host filed a lawsuit, Robert M. Haas, Sr. and Irwin Randolph Joint Tenants, et al. v. Marriott International, Inc., et al., Case No. 98-CI-04092, in the 57th Judicial District Court of Bexar County, Texas, alleging that the defendants conspired to sell hotels to the partnerships for inflated prices and that they charged the partnerships excessive management fees to operate the partnerships' hotels. The plaintiffs further allege that the defendants committed fraud, breached fiduciary duties and violated the provisions of various contracts. The plaintiffs are seeking unspecified damages. Although the partnerships have not been named as defendants, their

partnership agreements include provisions which require the partnerships to indemnify the general partners against losses, expenses and fees. The defendants filed answers and defenses to the petition.

Atlanta Marquis. Certain limited partners of Atlanta Marriott Marquis Limited Partnership ("AMMLP"), filed a putative class action lawsuit, Hiram and Ruth Sturm v. Marriott Marquis Corporation, et al., Case No. 97-CV-3706, in the U.S. District Court for the Northern District of Georgia, on December 12, 1997 against AMMLP's general partner, its directors and Host, regarding the merger of AMMLP into a new partnership (the "AMMLP Merger") as part of a refinancing of the partnership's debt. The plaintiffs allege that the defendants misled the limited partners in order to induce them to approve the AMMLP Merger, violated securities regulations and federal roll-up regulations and breached their fiduciary duties to the partners. The plaintiffs sought to enjoin, or in the alternative, rescind, the AMMLP Merger and damages. The partnership agreement includes provisions which require the partnership to indemnify the general partners against losses, expenses and fees. The defendants have filed a motion to dismiss.

Another limited partner of AMMLP sought similar relief and filed a separate lawsuit, styled Poorvu v. Marriott Marquis Corporation, et al., Civil Action No. 16095-NC, on December 19, 1997, in Delaware State Chancery Court. The defendants have filed an answer to the complaint.

Courtyard II. A group of partners in Courtyard by Marriott II Limited Partnership ("CBM II") filed a lawsuit, Whitey Ford, et al. v. Host Marriott Corporation, et al., Case No. 96-CI-08327, on June 7, 1996, in the 285th Judicial District Court of Bexar County, Texas, against Host, Marriott International and others alleging breach of fiduciary duty, breach of contract, fraud, negligent misrepresentation, tortious interference, violation of the Texas Free Enterprise and Antitrust Act of 1983 and conspiracy in connection with the formation, operation and management of CBM II and its hotels. The plaintiffs are seeking unspecified damages. On January 29, 1998, two other limited partners filed a petition in intervention seeking to convert the lawsuit into a class action. The defendants have filed an answer, the class has been certified, class counsel has been appointed and discovery is underway. Trial is presently scheduled for May 1999.

MHP2. Two groups of limited partners of Marriott Hotel Properties II Limited Partnership ("MHP2"), are each asserting putative class claims in lawsuits, filed in the United States District Court for the Southern District of Florida on May 10, 1996, Leonard Rosenblum, as Trustee of the Sylvia Bernice Rosenblum Trust, et al. v. Marriott MHP Two Corporation, et al., Case No. 96-8377-CIV-HURLEY, and, on December 18, 1997, Mackenzie Patterson Special Fund 2, L.P. et al. v. Marriott MHP Two Corporation, et al., Case No. 97-8989-CIV-HURLEY, respectively, against Host and certain of its affiliates alleging that the defendants violated their fiduciary duties and engaged in fraud and coercion in connection with a tender offer for MHP2 units. The District Court dismissed the Mackenzie Patterson case on August 4, 1998 and remanded the Rosenblum case to Palm Beach County Circuit Court on July 25, 1998. The defendants have moved to dismiss Rosenblum's fifth amended complaint in the case now styled Leonard Rosenblum, as Trustee of the Sylvia Bernice Rosenblum Trust, et al. v. Marriott MHP Two Corporation, et al., Case No. CL-96-4087-AD, or, in the alternative, to deny class certification.

PHLP. On July 15, 1998, one limited partner in PHLP filed a class action lawsuit styled Michael C. deBerardinis v. Host Marriott Corporation, Civil Action No. WMN 98-2263, in the United States District Court for the District of Maryland. The plaintiff alleges that Host misled the limited partners in order to induce them into approving the sale of one of the Partnership's hotels, violated the securities regulations by issuing a false and misleading consent solicitation and breached fiduciary duties and the partnership agreement. The complaint seeks unspecified damages. Host intends to vigorously defend against the claims asserted in the lawsuit.

# THE LEASES

Due to current federal income tax law restrictions on a REIT's ability to derive revenues directly from the operation of a hotel, Host recognized that it would be necessary to lease its hotels to one or more lessees just as other hotel REITs have done. Host desired to have a single lessee (or multiple lessees controlled by a single

person) in order to achieve substantial uniformity in its lease terms and avoid protracted negotiations with multiple parties over the terms of the lease arrangements, all of which would have been more complicated as a result of the existing long-term management agreements with Marriott International. Host also did not seriously attempt to restructure the existing Marriott International management agreements as leases (and Marriott International has not offered to do so in any of the negotiations with Host to date) because Host understands that Marriott International's general policy is to manage rather than lease hotels and Host also believed that Marriott International was unlikely to be an acceptable lessee of hotels operating under other brand names. Primarily for these reasons, and in order to give the economic benefit of the lessee's interest in the leases to Host's stockholders at the time of the REIT Conversion, Host decided to enter into leases with subsidiaries of Crestline and distribute the stock of Crestline to Host's stockholders. Host believed that Crestline was a more appropriate lessee than a newly formed company because Crestline already had an independent business and substantial assets and net worth and, thus, could perform well as a separate publicly traded company. In Host's judgment, these factors make it more likely that Crestline and its subsidiaries will have the financial stability, access to capital and wherewithal to perform on an ongoing basis the substantial obligations as lessee under the Leases (which is critical both to the Leases being respected for federal income tax purposes and to a viable long-term lessor-lessee relationship between Host REIT and Crestline, particularly in light of the fact that Host REIT remains secondarily liable to pay the management fees if Crestline defaults on its obligations). While Host recognized that, as with other REITs that own hotels, there would be additional administrative and operating complexities that would result from leasing its hotels to another party (including to Crestline and its subsidiaries) that has separate interests and economic objectives, Host believed that the advantages of the REIT Conversion substantially outweighed this disadvantage.

The following summary of the principal terms of the Leases is qualified in its entirety by reference to the Leases, a form of which has been filed as an exhibit to the Registration Statement of which this Proxy Statement/Prospectus is a part.

Lessees. There generally will be a separate Lessee for each Hotel or group of Hotels that is owned by a separate subsidiary of Host REIT. Each Lessee will be a Delaware limited liability company, whose purpose will be limited to acting as lessee under the applicable Lease(s). For those hotels where it is the Manager, Marriott International or a subsidiary will have a noneconomic membership interest in the Lessee entitling it to certain voting rights but no economic rights. The operating agreements for such Lessees will provide that the Crestline member of the Lessee will have full control over the management of the business of the Lessee, except with respect to certain decisions which will require the consent of both members. These decisions are: (i) dissolving, liquidating, consolidating, merging, selling or leasing all or substantially all of the assets of the Lessee; (ii) engaging in any other business or acquiring any assets or incurring any liabilities not reasonably related to the conduct of the Lessee's business; (iii) instituting voluntary bankruptcy or similar proceedings or consenting to involuntary bankruptcy or similar proceedings; (iv) terminating the Management Agreement relating to the Lessee's hotel, other than by reason of a breach by the Manager or upon exercise of express termination rights in the Management Agreement; (v) challenging the status of rights of the Manager or the enforceability of the membership rights; or (vi) incurring debt in excess of certain limits. Upon any termination of the applicable Management Agreement, these special voting rights of Marriott International (or its subsidiary) will cease.

Full-Service Lease Terms. Each full-service Hotel Lease will have a fixed term generally ranging from seven to ten years (depending upon the Lease), subject to earlier termination upon the occurrence of certain contingencies described in the Leases (including, particularly, the provisions described herein under "--Damage or Destruction," "--Termination of Leases upon Disposition of Full-Service Hotels" and "--Termination of the Leases upon Changes in Tax Laws").

Minimum Rent; Percentage Rent; Additional Charges. Each Lease will require the Lessee to pay (i) Minimum Rent (as defined below) in a fixed dollar amount per annum plus (ii) to the extent it exceeds Minimum Rent, Percentage Rent based upon specified percentages of aggregate sales from the applicable Hotel, including room sales, food and beverage sales, and other income ("Gross Revenues"), in excess of specified thresholds. "Minimum Rent" will be a fixed dollar amount specified in each Lease less the FF&E Adjustment

(which is described under "Personal Property Limitation" below). Any amounts other than Minimum Rent and Percentage Rent due to the Host REIT Lessor under the Hotel Leases are deemed to be "Additional Charges." The amount of Minimum Rent and the Percentage Rent thresholds will be adjusted each year (the "Annual Adjustment") based upon any increases in the Consumer Price Index ("CPI") and the Employment Cost Index during the previous twelve months. Neither Minimum Rent nor Percentage Rent thresholds will be decreased because of the Annual Adjustment.

Rental payments will be made on a Fiscal Year basis. The "Fiscal Year" shall mean the fiscal year used by the hotel Manager. Payments of Rent (defined herein) will be made within two business days after the required payment date under the Management Agreement for each Accounting Period. "Accounting Period" shall mean for those Hotels where Marriott International is the Manager, any of the thirteen four-week accounting periods which are used in the hotel Manager's accounting system. Rent payable for each Accounting Period will be the sum of (i) the excess (if any) of (x) the greater of cumulative Minimum Rent due and payable year-to-date or cumulative Percentage Rent due and payable year-to-date plus (ii) any Additional Charges due ("Rent"). If the total amount of Minimum Rent and Percentage Rent actually paid year-to-date, as of any rent payment date, is greater than both cumulative Minimum Rent due and payable year-to-date and cumulative Percentage Rent due and payable year-to-date, then the Lessor will remit the difference to the Lessee.

The full-service Hotel Leases will generally provide for a Rent adjustment in the event of damage, destruction, partial taking, certain capital expenditures, or an FF&E Adjustment.

Lessee Expenses. Each Lessee will be responsible for paying all of the expenses of operating the applicable Hotel(s), including all personnel costs, utility costs and general repair and maintenance of the Hotel(s). The Lessee also will be responsible for all fees payable to the applicable Manager, including base and incentive management fees, chain services payments and franchise or system fees, with respect to periods covered by the term of the Lease. The Lessee will not be obligated to bear the cost of any capital improvements or capital repairs to the Hotels or the other expenses borne by the Host REIT Lessor, as described below.

Host REIT Lessor Expenses. The Host REIT Lessor will be responsible for the following expenses: real estate taxes, personal property taxes (to the extent the Host REIT Lessor owns the personal property), casualty insurance on the structures, ground lease rent payments, required expenditures for FF&E (including maintaining the FF&E reserve, to the extent such is required by the applicable Management Agreement) and capital expenditures.

The consent of the Host REIT Lessor will be required for any capital expenditures funded by the Lessee (except in an emergency or where the owner's consent is not required under the Management Agreement) or a change in the amount of the FF&E Reserve payment.

Crestline Guarantees. Crestline and certain of its subsidiaries will enter into guarantees of the Lease obligations of each Lessee. For each of four identified "pools" of Hotels (determined on the basis of the term of the particular Lease with all leases having generally the same lease term placed in the same "pool"), the cumulative limit of Crestline's guaranty obligation will be the greater of 10% of the aggregate Rent payable for the immediately preceding Fiscal Year under all Leases in the pool or 10% of the aggregate Rent payable under all Leases in the pool for 1999 (with an agreed estimate of the 1999 Rent serving as the limit during 1999). For each pool, the subsidiary of Crestline that is the parent of the Lessees in the pool (a "Pool Parent") also will be a party to the guaranty of the Lease obligations for that pool. The obligations of the Pool Parent will not be limited in this manner.

The obligations of the Pool Parent under each guaranty will be secured by all funds received by the applicable Pool Parent from the Lessees in the pool, and the Lessees in the pool will be required to distribute their excess cash flow to the Pool Parent for each Accounting Period, in certain events. These events include a decline in Crestline's tangible net worth or consolidated debt coverage ratio below specified levels, a payment

default under any Lease in the same pool or any other Lease, with respect to which a guaranty by Crestline is in effect, or a default under the lien or debt covenants under any Leases in the same pool, or a reduction in Crestline's obligation under a pool guaranty to zero. Funds received from the Lessees will be deposited in a cash collateral account and applied, to the extent of available funds, to pay any shortfalls in payment of Rent under any Lease in the Pool. If certain conditions are satisfied, any remaining funds will then be released to the Pool Parent. Otherwise, the remaining funds will be applied to make a payment to the Pool Parent in a specified amount for overhead expenses and to maintain a reserve equal to the amount of Crestline's liability under the applicable guaranty (subject to a specified minimum amount), and, if certain conditions are satisfied, any remaining funds will be released to the Pool Parent.

In the event that Crestline's obligation under a guaranty is reduced to zero, the applicable Pool Parent can elect to terminate its guaranty and the pooling agreement for that pool by giving notice to the Operating Partnership. In that event, subject to certain conditions, the Pool Parent's guaranty will terminate six months after the effective date of such notice, subject to reinstatement in certain limited circumstances. Following termination of the Pool Parent's guaranty, the Host REIT Lessors in the applicable pool will have the right to terminate the Hotel Leases in the pool without payment of a termination fee, so long as the Hotel Leases are terminated within a year and all Hotel Leases in the pool are terminated.

Security. The obligations of the Lessee will be secured by a pledge of all personal property (tangible and intangible) of the Lessee related to or used in connection with the operation of the Hotels (including any cash and receivables from the Manager or others held by the Lessee as part of "working capital").

Working Capital. Each Host REIT Lessor will sell the existing working capital (including Inventory and fixed asset supplies (which principally consist of linen and similar items) and net receivables due from the Manager, net of accounts payable and accrued expenses) to the applicable Lessee upon the commencement of the Lease at a price equal to the fair market value of such assets (which shall be deemed to be book value after taking into account depreciation). The purchase price will be represented by a note evidencing a loan that bears interest at a rate per annum equal to the "long-term applicable federal rate" in effect on the commencement of the Lease. Interest accrued on the working capital loan will be due simultaneously with each periodic Rent payment, and the amount of each payment of interest will be credited against such Rent payment. The principal amount of the working capital loan will be payable upon termination of the Lease. At the termination or expiration of the Lease, the Lessee will sell to the Host REIT Lessor the then existing working capital at a price equal to the value of such assets at that time (which shall be deemed to be book value after taking into account depreciation). The Host REIT Lessor will pay the purchase price of the working capital by offsetting against the outstanding principal balance of the working capital loan. To the extent that the value of the working capital delivered to the Host REIT Lessor exceeds or is less than the value of the working capital delivered by the Host REIT Lessor to the Lessee at the commencement of the Lease, the Host REIT Lessor, or the Lessee, as appropriate, shall pay to the other party an amount equal to the difference in cash.

Termination of Leases upon Disposition of Full-Service Hotels. In the event the applicable Host REIT Lessor enters into an agreement to sell or otherwise transfer any full-service Hotel free and clear of the applicable Lease, the Host REIT Lessor must pay the Lessee a termination fee equal to the fair market value of the Lessee's leasehold interest in the remaining term of the Lease using a discount rate of 12%. Alternatively, the Host REIT Lessor will be entitled to (i) substitute a comparable Hotel or Hotels (in terms of economics and quality for the Host REIT Lessor and the Lessee as agreed to by the Lessee) for any Hotel that is sold or (ii) sell the Hotel subject to the Lease (subject to the Lessee's reasonable approval if the sale is to an entity that does not have sufficient financial resources and liquidity to fulfill the "owner's" obligations under the Management Agreement and the Host REIT Lessor's obligations under the Lease or is or controls or is controlled by a person convicted of a felony involving moral turpitude), without being required to pay a termination fee. Pursuant to the Lease, the Host REIT Lessor and the Lessee will each have the right to terminate the Lease without being required to pay any fee or other compensation as a result of such termination, provided that the termination rights of both the

Host REIT Lessor and the Lessee may only be exercised if the Host REIT Lessors and the Lessees under twelve of the other Hotel Leases have not already exercised their respective rights to terminate, and the Host REIT Lessor will only be permitted to exercise such right in connection with a sale of a Hotel to an unrelated third party or the transfer of a Hotel to a joint venture in which the Operating Partnership does not have a two-thirds or greater interest.

Termination of the Hotel Leases upon Changes in Tax Laws. In the event that changes in the federal income tax laws allow the Host REIT Lessors, or subsidiaries or affiliates of the Host REIT Lessors, to directly operate the Hotels without jeopardizing Host REIT's status as a REIT, the Host REIT Lessors will have the right to terminate all, but not less than all, of the full-service Hotel Leases (excluding Leases for Hotels that must continue to be leased following the tax law change) in return for paying the Lessees the fair market value of the

108--1

remaining terms of the full-service Hotel Leases, valued in the same manner as provided above under "Termination of Leases upon Disposition of Full-Service Hotels." The payment will be payable in cash or, subject to certain conditions, shares of Host REIT Common Stock, at the election of Host REIT and the Host REIT Lessor. Host has been pursuing the enactment of such legislation for more than one year, but there is no such bill pending in Congress, and there can be no prediction as to whether such legislation would be enacted in the future.

Damage or Destruction. If a Hotel is partially or totally destroyed and is no longer suitable for use as a hotel (as reasonably determined by the Host REIT Lessor), the Lease of such Hotel shall automatically terminate and the insurance proceeds shall be retained by the Host REIT Lessor, except to the extent of any personal property owned by the Lessee and proceeds from business interruption insurance. In this event, no termination fee shall be owed to the Lessee. If a Hotel is partially destroyed, but is still suitable for use as a hotel (as reasonably determined by the Host REIT Lessor), the Lessee, subject to the Host REIT Lessor agreeing to release the insurance proceeds to fund any shortfall in the insurance proceeds, shall apply the insurance proceeds to restore the Hotel to its preexisting condition. The Host REIT Lessor shall fund any shortfall in insurance proceeds less than or equal to 5% of the estimated cost of repair. The Host REIT Lessor may fund, in its sole discretion, any shortfall in insurance proceeds greater than 5% of the estimated cost of the repair, provided that if the Host REIT Lessor elects not to fund such shortfall, the Lessee may terminate the Hotel Lease and the Host REIT Lessor shall pay to the Lessee a termination fee equal to the Lessee's Operating Profit for the immediately preceding Fiscal Year.

Events of Default. Except as otherwise provided below, and subject to the notice and, in some cases, cure periods in the Hotel Lease, the Hotel Lease may be terminated without penalty by the applicable Host REIT Lessor if any of the following Events of Default (among others) occur:

- . Failure to pay Rent within ten days after the due date;
- . Failure to comply with, or observe any of, the terms of the Hotel Lease (other than failure to pay Rent) for 30 days after notice from the Host REIT Lessor, including failure to properly maintain the Hotel (other than by reason of the failure of the Host REIT Lessor to perform its obligations under the Hotel Lease), such period to be extended for up to an additional 90 days if such default cannot be cured with due diligence within 30 days;
- . Acceleration of maturity of certain indebtedness of the Lessee with a principal amount in excess of \$1,000,000;
- Failure of Crestline to maintain minimum net worth or debt service coverage ratio requirements;
- Filing of any petition for relief, bankruptcy or liquidation by or against the Lessee or any parent company of the Lessee;
- . The Lessee voluntarily ceases to operate the Hotel for 30 consecutive days, except as a result of a casualty, condemnation or emergency situation;
- . A change in control of Crestline, the Lessee or any subsidiary of Crestline that is a direct or indirect parent of the Lessee (provided, however, that if the change in control involves an "adverse party" which would include a competitor in the hotel business, a party without adequate financial resources, a party that has been convicted of a felony (or controlled by such a person), or a party who would jeopardize Host REIT's qualification as a REIT, the Host REIT Lessor must pay a termination fee equal to the Lessee's Operating Profit from the Hotel for the immediately preceding Fiscal Year if the Lease is terminated following such a change in control); or
- . The Lessee, Crestline or the Lessee's direct parent defaults under the assignment of the Management Agreement, the guarantees described above, the noncompetition agreement described below or certain other related agreements between the parties or their affiliates.

Assignment of the Lease. A Lessee will be permitted to assign its interest under its Hotel Lease, without the consent of the Host REIT Lessor, to any wholly-owned and controlled single-purpose subsidiary of Crestline,

provided that Crestline continues to meet the minimum net worth test and all other requirements of the Lease. Transfers to other parties will be permitted if approved by the Host REIT Lessor.

Subordination to Qualifying Mortgage Debt. The rights of each Lessee will be expressly subordinate to qualifying mortgage debt (which totaled approximately \$2.9 billion on a pro forma basis at September 11, 1998 assuming all Partnerships participate and the Blackstone Acquisition is consummated) and any refinancing thereof. A default under the loan documents may result in the termination of the Hotel Lease by the lender. The lender will not be required to provide a non-disturbance agreement to the Lessee.

The Host REIT Lessor will be obligated to compensate the Lessee, on a basis equal to the lease termination provision described in "--Termination of Leases upon Disposition of Full-Service Hotels" above, if the full-service Hotel Lease is terminated because of a non-monetary default under the terms of a loan that occurs because of an action or omission by the Host REIT Lessor (or its affiliates) or a monetary default where there is not an uncured monetary Event of Default of the Lessee. In addition, if any loan is not refinanced in a timely manner, and the loan amortization schedule is converted to a cash flow sweep structure, the Lessee has the right to terminate the Lease after a twelve-month cure period and the Host REIT Lessor will owe a termination fee as provided above. During any period of time that a cash flow sweep structure (or other similar cash management procedure) is in effect, the Host REIT Lessor will compensate the Lessee for any lost revenue resulting from such cash flow sweep. The Operating Pertnership will guarantee these obligations.

Personal Property Limitation. If a Host REIT Lessor reasonably anticipates that the average tax basis of the items of the Host REIT Lessor's FF&E and other personal property that are leased to the applicable Lessee will exceed 15% of the aggregate average tax basis of the real and personal property subject to the applicable Lease, the following procedures will apply, subject to obtaining lender consent where required:

- . The Host REIT Lessor will acquire any replacement FF&E that would cause the applicable limits to be exceeded (the "Excess FF&E"), and immediately thereafter the Lessee would be obligated either to acquire such Excess FF&E from the Host REIT Lessor or to cause a third party to purchase such FF&E.
- . The Lessee would agree to give a right of first opportunity to a Non-Controlled Subsidiary to acquire the Excess FF&E and to lease the Excess FF&E to the Lessee at an annual rental equal to the market leasing factor (as defined below) times the cost of the Excess FF&E. If such Non-Controlled Subsidiary does not agree to acquire the Excess FF&E and to such lease, then the Lessee may either acquire the Excess FF&E itself or arrange for another third party to acquire such Excess FF&E and to lease the same to Lessee.
- The annual Rent under the applicable Hotel Lease would be reduced in accordance with a formula based on market recovery rates.

Certain Actions under the Hotel Leases. The Leases prohibit the Lessee from taking the following actions with respect to the Management Agreement without notice to the Host REIT Lessor and, if the action would have a material adverse effect on the Host REIT Lessor, the consent of the Host REIT Lessor: (i) terminate the Management Agreement prior to the expiration of the term thereof; (ii) amend, modify or assign the Management Agreement; (iii) waive (or fail to enforce) any right of the "Owner" under the Management Agreement; (iv) waive any breach or default by the Manager under the Management Agreement (or fail to enforce any right of the "Owner" in connection therewith); (v) agree to any change in the Manager or consent to any assignment by the Manager; or (vi) take any other action which reasonably would be expected to materially adversely affect the Host REIT Lessor's rights or obligations under the Management Agreement for periods following the termination of the Hotel Lease (whether upon the expiration of its term or upon earlier termination as provided for therein).

Change in Manager. A Lessee will be permitted to change the Manager or the brand affiliation of a Hotel only with the approval of the applicable Host REIT Lessor, which approval may not be unreasonably withheld. The replacement manager must be a nationally recognized manager with substantial experience in managing hotels of comparable quality. No such replacement can extend beyond the term of the Lease without the consent of the Host REIT Lessor, which consent may be withheld in the Host REIT Lessor's sole discretion.

General. The Lessees will lease the Hotels from the Partnerships and the Private Partnerships under the Management Agreements between the Operating Partnership and the subsidiaries of Marriott International and other companies that currently manage the Hotels. Following the REIT Conversion and as a result of their assumptions of obligations under the Management Agreements, the Lessees will have substantially all of the rights and obligations of the "Owners" of the Hotels under the Management Agreements for the period during which the Leases are in effect (including the obligation to pay the management and other certain fees thereunder) and will hold the Company harmless with respect thereto. See "--Management Services Provided by Marriott International and Affiliates--Assignment of Management Agreements."

Relationship with Marriott International. Subsidiaries of Marriott International will serve as Managers for a substantial majority of the Company's Hotels which will be leased to the Lessees, pursuant to the Management Agreements. Marriott International and its subsidiaries also will provide various other services to Host REIT and its affiliates and to . Crestline and its affiliates. With respect to these contractual arrangements, the potential exists for disagreement as to contract compliance. Additionally, the possible desire of the Company to finance, refinance or effect a sale of any of the Hotels leased to the Lessees and managed by subsidiaries of Marriott International may, depending upon the structure of such transactions, result in a need to modify the Management Agreements with respect to such Hotel. Any such modification proposed by the Company may not be acceptable to Marriott International or the applicable Lessee, and the lack of consent from either Marriott International or the applicable Lessee that has assumed the Management Agreement could adversely affect the Company's ability to consummate such financing or sale. In addition, certain situations could arise where actions taken by Marriott International in its capacity as manager of competing lodging properties would not necessarily be in the best interests of the Company or the Lessees. Nevertheless, the Company believes that there is sufficient mutuality of interest between the Company, the Lessees and Marriott International to result in a mutually productive relationship.

Management Services Provided by Marriott International and Affiliates.

General. Under each Management Agreement related to a Marriott International-managed Hotel, the Manager will provide complete management services to the applicable Lessees in connection with its management of such Lessee's Hotels following the REIT Conversion. Except where specifically noted, these relationships are substantially identical to those that exist between the applicable Manager and Host or the applicable Partnership or Private Partnership currently, and that would exist between the Company's subsidiaries and the Manager in the event the Leases expire or otherwise terminate while the Management Agreements remain in effect. The services provided by each Manager to each Lessee will include the following:

Assignment of Management Agreements. The Management Agreements applicable to each Hotel will be assigned to the applicable Lessee for the term of the Lease of such Hotel. The Lessee will be obligated to perform all of the obligations of the Lessor under the Management Agreement during the term of its Lease including the payment of fees due under the Management Agreement, other than certain retained obligations, including, without limitation, payment of property taxes, property casualty insurance and ground lease rent, and maintaining a reserve fund for FF&E replacements and capital expenditures, for which the Lessor will retain responsibility. Although the Lessee will assume obligations of the Lessor under the Management Agreement, the Lessor will not be released from its obligations and, if a Lessee fails to perform any obligations, the Manager will be entitled to seek performance by or damages from the Lessor. The Lessees' obligation to pay the fees due under the Management Agreements, however, could adversely affect the ability of the Lessee to pay rent under the Leases, even though such amounts are otherwise due and owed to the Lessor. If the Lease is terminated for any reason, any new or successor Lessee must meet certain requirements for an "Approved Lessee" or otherwise be acceptable to Marriott International. The requirements for an "Approved Lessee" includes that the entity (i) has sufficient financial resources and liquidity to fill the obligations under the Management Agreement, (ii) is not in control of or controlled by persons who have been convicted of felonies, (iii) is not engaged, or affiliated with any person or entity engaged in the business of operating a branded hotel chain having 5,000 or more guest rooms in competition with Marriott International, and (iv) must be a single purpose entity in which Marriott

International has a noneconomic membership interest with the same rights as it has in Lessee. Any new lease must be in substantially the same form as the Lease or otherwise be acceptable to Marriott International.

Operational Services. The Managers have sole responsibility and exclusive authority for all activities necessary for the day-to-day operation of the Hotels, including establishment of all room rates, the processing of reservations, procurement of inventories, supplies and services, periodic inspection and consultation visits to the Hotels by the Managers' technical and operational experts and promotion and publicity of the Hotels. The Manager will receive compensation from the Lessee in the form of a base management fee and an incentive management fee, which are normally calculated as percentages of gross revenues and operating profits, respectively.

Executive Supervision and Management Services. The Managers generally provide all managerial and other employees for the Hotels; review the operation and maintenance of the Hotels; prepare reports, budgets and projections; provide other administrative and accounting support services, such as planning and policy services, financial planning, divisional financial services, risk planning services, product planning and development, employee planning, corporate executive management, legislative and governmental representation and certain in-house legal services; and protect the "Marriott" trademark and other tradenames and service marks. The Manager also will provide a national reservations system.

Chain Services. The Management Agreements require the Manager to furnish certain services (the "Chain Services") that are furnished generally on a central or regional basis to hotels in the Marriott hotel system. Such services include the following: (i) the development and operation of computer systems and reservation services, (ii) regional management and administrative services, regional marketing and sales services, regional training services, manpower development and relocation costs of regional personnel and (iii) such additional central or regional services as may from time to time be more efficiently performed on a regional or group level. Costs and expenses incurred in providing such services are allocated among all hotels in the Marriott hotel system managed by the Manager or its affiliates and each applicable Lessee will be required to reimburse the Manager for its allocable share of such costs and expenses.

Working Capital and Fixed Asset Supplies. The Lessee will be required to maintain working capital for each Hotel and fund the cost of fixed asset supplies, which principally consist of linen and similar items. The applicable Lessee will also be responsible for providing funds to meet the cash needs for the operations of the Hotels if at any time the funds available from operations are insufficient to meet the financial requirements of the Hotels.

Use of Affiliates. The Manager employs the services of its affiliates to provide certain services under the Management Agreements. Certain of the Management Agreements provide that the terms of any such employment must be no less favorable to the applicable Lessee, in the reasonable judgment of the Manager, than those that would be available from the Manager.

FF&E Replacements. The Management Agreements generally provide that once each year the Manager will prepare a list of FF&E to be acquired and certain routine repairs that are normally capitalized to be performed in the next year ("FF&E Replacements") and an estimate of the funds necessary therefor. Under the terms of the Leases, the Company, as lessor, is required to provide to the applicable Lessee, all necessary FF&E for the operation of the Hotels (including funding any required FF&E Replacements). Under each full-service Lease, Host REIT will be responsible for the costs of FF&E Replacements and for decisions with respect thereto (subject to its obligations to the Lessee under the Lease).

Building Alterations, Improvements and Renewals. The Management Agreements require the Manager to prepare an annual estimate of the expenditures necessary for major repairs, alterations, improvements, renewals and replacements to the structural, mechanical, electrical, heating, ventilating, air conditioning, plumbing and vertical transportation elements of each Hotel. Such estimate will be submitted to the Company and the Lessee

for their approval. In addition to the foregoing, the Management Agreements generally provide that the Manager may propose such changes, alterations and improvements to the Hotel as are required, in the Manager's reasonable judgment, to keep the Hotel in a competitive, efficient and economical operating condition or in accordance with Marriott standards. The cost of the foregoing shall be paid from the FF&E Reserve Account; to the extent that there are insufficient funds in such account, the Company is required to pay any shortfall.

Service Marks. During the term of the Management Agreements, the service mark "Marriott" and other symbols, logos and service marks currently used by the Manager and its affiliates may be used in the operation of the Hotels. Marriott International (or its applicable affiliates) intends to retain its legal ownership of these marks. Any right to use the service marks, logo and symbols and related trademarks at a Hotel will terminate with respect to that Hotel upon termination of the Management Agreement with respect to such Hotel.

Termination Fee. Certain of the Management Agreements provide that if the Management Agreement is terminated prior to its full term due to casualty, condemnation or the sale of the Hotel, the Manager will receive a termination fee as specified in the specific Management Agreement.

Termination for Failure to Perform. Substantially all of the Management Agreements may be terminated based upon a failure to meet certain financial performance criteria, subject to the Manager's right to prevent such termination by making certain payments to the Lessee based upon the shortfall in such criteria.

Events of Default. Events of default under the Management Agreements include, among others, the following: (i) the failure of either party to make payments pursuant to the Management Agreement within ten days after written notice of such non-payment has been made, (ii) the failure of either party to perform, keep or fulfill any of the covenants, undertakings, obligations or conditions set forth in the Management Agreement and the continuance of such default for a period of 30 days after notice of said failure or, if such default is not susceptible of being cured within 30 days, the failure to commence said cure within 30 days or thereafter the failure to diligently pursue such efforts to completion, (iii) if either party files a voluntary petition in bankruptcy or insolvency or a petition for reorganization under any bankruptcy law or admits that it is unable to pay its debts as they become due, (iv) if either party consents to an involuntary petition in bankruptcy or fails to vacate, within 90 days from the date of entry thereof, any order approving an involuntary petition by such party; or (v) if an order, judgment or decree by any court of competent jurisdiction, on the application of a creditor, adjudicating either party as bankrupt or insolvent or approving a petition seeking reorganization or appointing a receiver, trustee or liquidator of all or a substantial part of such party's assets is entered, and such order, judgment or decree continues unstayed and in effect for any period of 90 days.

As described above, all fees payable under the Management Agreements excluding the termination fee, if any, will become obligations of the Lessees, to be paid by the Lessees, as modified prior to the consummation of the REIT Conversion, for so long as the Leases remain in effect. The Lessees' obligations to pay these fees, however, could adversely affect the ability of one or more Lessees to pay Base Rent or Percentage Rent payable under the Leases, even though such amounts otherwise are due and owing to the Company.

Restrictions on Sales and Other Transfers of Interests in Hotels. The Management Agreements prohibit the hotel owner from selling, leasing or otherwise transferring the hotels unless the transferees assume the Management Agreements and satisfy certain criteria, including having sufficient financial resources and liquidity to satisfy the owner's obligations under the Management Agreements, not being in control or controlled by persons who have been convicted of felonies and not being engaged in operating a branded hotel chain having 5,000 or more guest rooms in competition with Marriott International. The Management Agreements also prohibit the Lessees from subleasing all or any portion of the Leases and from assigning the Leases except to entities which are wholly-owned by Crestline and have organizational documents which are the same as those of the Lessees, in each case without the prior written consent of Marriott International.

Restrictions on Lease Amendments. The Management Agreements prohibit the Lessees and the hotel lessors from entering in certain amendments of the Leases, including shortening the term of the Leases and

modifying the parties' respective rights and obligations with respect to FF&E, capital expenditures and approval of hotel budgets, in each case without the prior written consent of Marriott International.

Marriott International's Right to Cause Termination of Leases. The Management Agreements provide that if the hotel lessors do not elect to terminate the Lease upon a "Change in Control," Marriott International may require the hotel lessors to exercise their termination rights if, as a result of such Change in Control, the Lessee is or is controlled by a person who has been convicted of a felony or who is engaged (or affiliated with persons who are engaged) in operating a branded hotel chain having 5,000 or more guest rooms in competition with Marriott International.

#### NONCOMPETITION AGREEMENT

Crestline, Host and the Non-Controlled Subsidiary which will lease to Crestline any Excess FF&E existing at the commencement of the Leases (the "Initial FF&E Lessor") will enter into a non-competition agreement in connection with the Initial E&P Distribution. Pursuant to this non-competition agreement, Crestline will agree, among other things, that until the earlier of December 31, 2008 or the date on which it is no longer a Lessee of more than 25% of the number of hotels owned by Host REIT at the time of the Initial E&P Distribution, it will not (i) own, acquire, develop or construct for ownership any full-service hotel (except for (a) investments which represent an immaterial portion of a merger or similar transaction, (b) a minimal portfolio investment or (c) the provision of limited financing); (ii) without the consent of Host or Host REIT in its sole discretion, manage or operate (other than through a third party manager) any limited-service or full-service hotel properties owned by Host or Host REIT; or (iii) conduct, participate in, engage in or have a financial interest in any person that engages in the ownership or operation of any single or multiple full-service hotel franchise system operating under one or more common name brands. The restrictions described in (i) and (iii) above do not apply to any activities of Crestline related to limited-service hotels. Crestline generally is permitted to act as a manager with respect to full-service hotels, subject to certain restrictions intended to preclude Crestline from using the management agreement relationship to take a disguised equity ownership position in full-service hotels. Crestline also generally is permitted to act as a lessee with respect to full-service hotels owned (i) by any party under an arrangement where Crestline also manages the hotel or (ii) by a REIT, whether or not Crestline also manages the hotel, subject in each case to certain restrictions intended to preclude Crestline from using those leasing arrangements to take  $% \left( 1\right) =\left( 1\right) \left( 1\right)$ significant disguised equity ownership positions in full-service hotels

Until the earlier of December 31, 2008 or the date upon which Crestline is not the lessee of more than 25% of the number of hotels owned by Host at the time of the Initial E&P Distribution, Host or Host REIT and the FF&E Lessor have agreed that they will not conduct, participate in, engage in or have a financial interest in any person that engages in the business of leasing, operating or franchising limited-service or full-service hotel properties; provided, however, that this restriction does not prevent Host or Host REIT or the FF&E Lessor from (i) managing, operating or franchising limited-service or full-service hotels with respect to matters incident to the operation of such properties (e.g., management services with respect to food and beverages, plant and equipment operation and maintenance, reservations, sales and marketing) on behalf of third parties or (ii) leasing full-service or limited-service hotels to and from each other or (iii) in the case of Host or Host REIT, leasing full-service or limited-service hotels from certain other related parties. Until December 31, 2003, Host or Host REIT and the FF&E Lessor have also agreed that they will not conduct, participate in, engage in or have a financial interest in any person that engages in the ownership, acquisition or operation of senior living communities (except for (a) investments which represent an immaterial portion of a merger or similar transaction, (b) a minimal portfolio investment or (c) the provision of limited financing). In addition, both Crestline and Host REIT will agree not to hire or attempt to hire any of the other company's other senior employees at any time prior to December 31, 2000. See "Business and Properties-Noncompetition Agreement." Subject to the Marriott International hotel noncompetition agreement, which continues until October 2000, Crestline is free to pursue any and all activities with respect to limited service hotels (other than limited service hotels owned by Host), and it is permitted to operate full-service hotels as a third-party manager so long as it is not the owner of the brand or franchise under which such hotels are operated.

#### **INDEBTEDNESS**

Senior Note Refinancing. On August 5, 1998, HMH Properties, Inc. ("HMH Properties"), a subsidiary of Host that will merge into the Operating Partnership prior to the Effective Date, issued \$1.7 billion of 7 7/8% senior notes issued in two series, consisting of \$500 million due 2005 and \$1.2 billion due 2008 (the "New Senior Notes"). The New Senior Notes are guaranteed by Host, Host Marriott Hospitality, Inc. and certain subsidiaries of HMH Properties and are secured by pledges of equity interests in certain subsidiaries of HMH Properties. The Operating Partnership will assume the New Senior Notes in connection with the REIT Conversion and the guarantee by Host Marriott is expected to terminate on the Effective Date.

The indenture under which the New Senior Notes were issued contains covenants restricting the ability of HMH Properties and certain of its subsidiaries to incur indebtedness, acquire or sell assets or make investments in other entities, and make distributions to equityholders of HMH Properties and (following the REIT Conversion) the Operating Partnership. Following the REIT Conversion, the indenture permits the Operating Partnership to make distributions to holders of OP Units, including Host REIT, in amounts equal to the greater of (i) 95% of FFO plus net proceeds of equity offerings (provided that no event of default under the indenture has occurred and is continuing and the Operating Partnership is able to incur debt under the applicable indenture covenants) or (ii) an amount sufficient to permit Host REIT to maintain its status as a REIT and satisfy certain other requirements (provided that no event of default under the indenture has occurred and is continuing and the Operating Partnership has a consolidated debt to adjusted total assets ratio that is less than a specified level). The indenture also permits the Operating Partnership to make distributions to Host REIT sufficient to enable Host REIT to make the Initial E&P Distribution. The New Senior Notes also contain a financial covenant requiring the maintenance of a specified ratio of unencumbered assets to unsecured debt.

New Credit Facility. On August 5, 1998, HMH Properties entered into a \$1.25 billion credit facility (the "New Credit Facility") provided by a syndicate of financial institutions (the "Lenders") led by Bankers Trust Company. The New Credit Facility provides the Operating Partnership with (i) a \$350 million term loan facility (subject to increases as provided in the succeeding paragraph) and (ii) a \$900 million revolving credit facility. The New Credit Facility will have an initial term of three years with two one-year options to extend. The proceeds of the New Credit Facility, along with the proceeds from the New Senior Notes, were used to fund the purchase of \$1.55 billion of senior notes of HMH Properties at the initial closing on August 5, 1998, and repay \$22 million of outstanding borrowings under a line of credit provided by the Lenders to certain subsidiaries of Host and will be used (i) to acquire full-service hotels and other real estate assets including, under certain circumstances, senior living properties, (ii) under certain circumstances, to develop new full-service hotels and (iii) for general working capital purposes.

The term loan facility was funded on the closing date of the New Credit Facility. The \$350 million term loan facility may be increased by up to \$250 million after the initial closing and will be available, subject to terms and conditions thereof and to the commitment of sufficient Lenders, in up to two drawings to be made on or prior to the second anniversary of the closing of the New Credit Facility. The Lenders will advance funds under the revolving credit facility as requested by the Operating Partnership with minimum borrowing amounts and frequency limitations to be agreed upon, subject to customary conditions including, but not limited to, (i) no existing or resulting default or event of default under the New Credit Facility and (ii) continued accuracy of representations and warranties in all material respects. As of September 11, 1998, approximately \$350 million was outstanding under the New Credit Facility.

The interest rate applicable to the New Credit Facility and the unused commitment fee applicable to the revolving portion of the New Credit Facility are calculated based on a spread over LIBOR that will fluctuate based on the quarterly recalculation of a leverage ratio set forth in the New Credit Facility (7.5% at September 11, 1998). The New Credit Facility provides that in the event that the Operating Partnership achieves one of several investment grade long-term unsecured indebtedness ratings, the spread over LIBOR applicable to the New Credit Facility will be fixed based on the particular rating achieved. If the Operating Partnership elects to

exercise its one-year extensions, the Operating Partnership will be required to amortize approximately 22.5% per annum of the principal amount outstanding under the New Credit Facility at the end of the initial three-year term.

The Operating Partnership's obligations under the New Credit Facility are guaranteed, subject to certain conditions, on a senior basis by Host, Host Marriott Hospitality, Inc. and certain of HMH Properties' existing and future subsidiaries. The New Credit Facility will be assumed by the Operating Partnership in connection with the REIT Conversion and the guarantee of Host is expected to terminate on the Effective Date. While there are certain conditions to the termination of the guarantee of Host, the REIT Conversion has been structured so that such conditions will be satisfied. Termination does not otherwise require Lender approval. In addition, certain subsidiaries of Host other than HMH Properties and its subsidiaries may, under certain circumstances, guarantee the obligations under the New Credit Facility in the future. Borrowings under the New Credit Facility will rank pari passu with the New Senior Notes and other existing and future senior indebtedness of the Operating Partnership. The New Credit Facility is secured, on an equal and ratable basis, with the New Senior Notes by a pledge of the capital stock of certain direct and indirect subsidiaries of HMH Properties. In addition, the New Credit Facility may, under certain circumstances in the future, be secured by a pledge of capital stock of certain subsidiaries of Host other than HMH Properties and its subsidiaries.

The New Credit Facility includes financial and other covenants that require the maintenance of certain financial ratios and that restrict payment of distributions and investments, acquisitions and sales of assets by the Operating Partnership. The financial covenants impose the following requirements, among others, on the Operating Partnership:

- . the ratio of consolidated total debt to consolidated EBITDA must not exceed 5.5:1.0 through the end of the fiscal quarter ending on or about June 30, 1999, declining periodically to 4.5:1.0 at any time after the end of the fiscal quarter ending on or about June 30, 2000;
- . the ratio of consolidated secured debt to consolidated total debt must not exceed 0.57:1.0 through the end of the fiscal quarter ending on or about June 30, 1999, declining periodically to 0.40:1.0 at any time after the end of the fiscal quarter ending on or about December 31, 2001, with certain exceptions;
- . the ratio of EBITDA attributable to unencumbered properties of the Operating Partnership and its subsidiaries to consolidated EBITDA for any four fiscal quarters must not be less than 0.425:1.0 through the end of the fiscal quarter ending on or about March 31, 1999, increasing periodically to 0.60:1.0 for any four fiscal quarters ending on or after December 31, 2001;
- . the ratio of the consolidated EBITDA to consolidated interest expense for any four fiscal quarters must not be less than 2.25:1.0 through the end of the fiscal quarter ending on or about June 30, 1999, increasing periodically to 2.50:1.0 for any four fiscal quarters ending after the fiscal quarter that ends on or about June 30, 2000;
- . the ratio of EBITDA attributable to unencumbered properties of the Operating Partnership and its subsidiaries to consolidated unsecured debt for any four fiscal quarters must not be less than 2.0:1.0;
- . the ratio of consolidated EBITDA to total fixed charges (as defined in the New Credit Facility) for any four fiscal quarters must not be less than 1.5:1.0; and
- . the consolidated tangible net worth of the Operating Partnership must not be less than the sum of 75% of the tangible net worth on August 5, 1998 plus 75% of the aggregate net proceeds received from issuance of equity by the Operating Partnership after August 5, 1998.

Following the REIT Conversion, the New Credit Facility permits the Operating Partnership to make distributions to holders of OP Units, including Host REIT, in an aggregate amount for every four fiscal quarters equal to the greater of (i) 85% of adjusted funds from operations plus the net proceeds of equity offerings and (ii) the minimum amount necessary to permit Host REIT to maintain its status as a REIT and to satisfy certain other requirements, provided that no specified default or event of default has occurred under the New Credit Facility and is continuing. The New Credit Facility also permits the Operating Partnership to make distributions to Host REIT sufficient to enable Host REIT to make the Initial E&P Distribution.

#### DISTRIBUTION AND OTHER POLICIES

The following is a discussion of the anticipated policies with respect to distributions, investments, financing, lending, conflicts of interest and certain other activities of the Company. Upon consummation of the REIT Conversion, the Company's policies with respect to these activities will be determined by the Board of Directors of Host REIT and may be amended or revised from time to time at the discretion of the Board of Directors without notice to, or a vote of, the stockholders of Host REIT, except that changes in certain policies with respect to conflicts of interest must be consistent with legal and contractual requirements.

#### DISTRIBUTION POLICY

Host REIT and the Operating Partnership intend to pay regular quarterly distributions to holders of Host REIT Common Stock and OP Units. Host REIT and the Operating Partnership anticipate that distributions will be paid during January, April, July and October of each year, except that the first distribution in 1999 is expected to be paid at the end of February if the REIT Conversion is completed in 1998. The following discussion and the information set forth in the table and footnotes below should be read in conjunction with the Pro Forma Statements of Operations and notes thereto, "Summary--Forward-Looking Statements," "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Operations--Liquidity and Capital Resources."

Although the Code generally requires a REIT to distribute 95% of its taxable income for each year (within a certain period after the end of such year), the Operating Partnership will establish its initial distribution at a level that will enable Host REIT to distribute to its stockholders an amount equal to 100% of Host REIT's taxable income (other than capital gains, which will be addressed on a case-by-case basis) for each year no later than the end of January of the following year. Host REIT anticipates that distributions generally will be paid from cash available for distribution, but to the extent that cash available for distribution is insufficient, the Operating Partnership intends to borrow funds in order to make distributions consistent with such distribution policy. Based upon Host's preliminary estimates of Host REIT's taxable income for the twelve months ending December 31, 1999, Host and the Operating Partnership currently estimate that this policy will result in an initial annual distribution by the Operating Partnership of approximately \$0.84 per OP Unit (\$0.21 per quarter) during the twelve months ending December 31, 1999.

The Operating Partnership has estimated its pro forma cash available for distribution during the twelve months ending December 31, 1999 based upon the Operating Partnership's pro forma cash from operations during the fifty-two weeks ended September 11, 1998 (the "Last Twelve Months"), adjusted for certain known material events and/or contractual commitments that either have occurred or will occur prior to December 31, 1999. No effect was given to any changes in working capital resulting from changes in current assets and liabilities (which changes are not expected to be material) or to any immaterial changes in the net amount of cash estimated to be used for (or provided by) investing activities or financing activities. Rental income is recognized only for leases to be executed at or prior to completion of the REIT Conversion. The estimate of cash available for distribution is not intended to be a projection or forecast of the Operating Partnership's results of operations or its liquidity. The following table describes the calculation of the Operating Partnership's pro forma cash from operations during the Last Twelve Months and its estimated cash available for distribution, cash from contingent rents and borrowings to make estimated distributions during the twelve months ending December 31, 1999:

# (DOLLARS IN MILLIONS, EXCEPT PER OP UNIT AND SHARE AMOUNTS)

SHARE AMOUNTS)

Pro forma income before extraordinary items for the fiscal year ended January 2, 1998	\$ 26
Plus: Pro forma loss before extraordinary items for the	
First Three Quarters 1997 Less: Pro forma loss before extraordinary items for the	222
First Three Quarters 1998	(204)
Pro forma income before extraordinary items for the Last	4.4
Twelve Months Plus: Pro forma loss on sale of real estate for the Last	44
Twelve Months(1) Plus: Pro forma real estate related depreciation and	15
amortization for the Last Twelve Months(2)  Plus: Pro forma portion of cash from operations of unconsolidated equity investments for the Last Twelve Months, net of pro forma equity in earnings	333
of affiliates for the Last Twelve Months(3)	26
Less: One-time gain for the Last Twelve Months(4)	(10)
Less: Pro forma portion of cash from operations relating to minority owners for the Last Twelve Months, net of pro forma portion of minority interest relating	
to OP Units for the Last Twelve Months(5)	(4)
Pro forma cash from operations during the Last Twelve	
Months Adjustments:	404
FF&E reserves(6)	(191)
Principal repayments(7)	(69)
Estimated cash available for distribution of the Operating Partnership during the twelve months ending	
December 31, 1999	144
Estimated cash from contingent rents(8) Estimated borrowings to make estimated initial annual	64
cash distributions	23
Total estimated initial annual cash distributions of the Operating Partnership during the twelve months ending	
December 31, 1999(9)(13)	\$ 231 =====
Host REIT's share of total estimated initial annual cash distributions of the Operating Partnership during the	
twelve months ending December 31, 1999(10)(13)	\$ 172 =====
Estimated initial annual cash distributions per OP Unit during the twelve months ending December 31,	
1999(11)(13)	\$0.84 ====
Estimated initial annual cash distributions per share of Host REIT Common Stock during the twelve months ending	
December 31, 1999(12)(13)	\$0.84 ====

(1) Represents loss on sale of real estate for the last quarter 1997 of \$15  $\,$  million.

- (2) Represents pro forma real estate related depreciation and amortization for the fiscal year ended January 2, 1998 of \$342 million minus pro forma real estate related depreciation and amortization for the First Three Quarters 1997 of \$230 million plus pro forma real estate related depreciation and amortization for the First Three Quarters 1998 of \$221 million.
- (3) Represents pro forma portion of cash from operations of unconsolidated equity investments, net of pro forma equity in earnings of affiliates, for the fiscal year ended January 2, 1998 of \$30 million minus pro forma portion of cash from operations of unconsolidated equity investments, net of pro forma equity in earnings of affiliates, for the First Three Quarters 1997 of \$16 million plus pro forma portion of cash from operations of unconsolidated equity investments, net of pro forma equity in earnings of affiliates, for the First Three Quarters 1998 of \$12 million.
- (4) Represents pro forma one-time gain for the last quarter 1997 of \$10  $\,$  million.
- (5) Represents pro forma portion of cash from operations relating to minority owners, net of pro forma portion of minority interest relating to OP Units, for the fiscal year ended January 2, 1998 of \$10 million minus pro forma portion of cash from operations relating to minority owners, net of pro forma portion of minority interest relating to OP Units, for the First Three Quarters 1997 of \$8 million plus pro forma portion of cash from operations relating to minority owners, net of pro forma portion of minority interest relating to OP Units, for the First Three Quarters 1998 of \$3 million.

- (6) Represents FF&E reserves for the year ending December 31, 1999 of \$191 million based on pro forma FF&E for the Last Twelve Months. Any differences between such estimated amount and the Last Twelve Month pro forma amount are not expected to be material.
- (7) Represents principal repayments required for the year ending December 31, 1999 of \$69 million based on the terms of the pro forma indebtedness at September 11, 1998.
- (8) The amount of contingent rent received but deferred pursuant to EITF 98-9, "Accounting for Contingent Rents in Interim Financial Periods," at September 11, 1998 and September 12, 1997 was \$320 million and \$256 million, respectively. The difference of \$64 million represents the elimination of the net effect of these two deferred items, which has the effect of applying the applicable lease rental terms to the historical gross sales from the leased Hotels for the last Twelve Months to estimate rental revenues for the calendar year 1999. Interim and annual revenues will be impacted to the extent percentage rent thresholds under the leases are not met or exceeded. If the rental revenues represented by this adjustment were not realized for the twelve months ending December 31, 1999, then the Operating Partnership would be required to borrow the amount of the shortfall under the New Credit Facility or from other sources to make estimated initial annual cash distributions during 1999.
- (9) Based on a total of 274.6 million OP Units outstanding on a pro forma basis after the Partnership Mergers (based upon an assumed price of \$12.50 per OP Unit) and the preliminary estimated cash distributions during the twelve months ending December 31, 1999 of \$0.84 per OP Unit.
- (10) Based upon a total of 204.5 million OP Units to be owned by Host REIT on a pro forma basis after the Merger (based on a price of \$12.50 per OP Unit).
- (11) Based on a total of 274.6 million OP Units outstanding on a pro forma basis after the Partnership Mergers (based upon an assumed price of \$12.50 per OP Unit).
- (12) Based on a total of 204.5 million shares of Host REIT Common Stock outstanding on a pro forma basis after the Merger (based on an assumed price of \$12.50 per share).
- (13) The actual initial cash distributions per OP Unit and per share of Host REIT Common Stock during the twelve months ending December 31, 1999 will be dependent upon the number of OP Units issued in the Partnership Mergers, which will be a function of the price of an OP Unit for purposes of the Partnership Mergers and which Partnerships participate in the Partnership Mergers. Does not include 29.6 million shares (or OP Units) which are issuable upon conversion of the convertible debentures underlying the Convertible Preferred Securities (based on the current conversion price of \$18.604 per share), which does not take into account any anti-dilution adjustments which may result from certain effects of the REIT Conversion that cannot be determined at this time or any shares which are issuable upon exercise of outstanding stock options (6.8 million shares as of January 2, 1998), or any shares issued pursuant to the Special Dividend. Assuming a value of \$2.50 per share for the Initial  $\mathsf{E\&P}$  Distribution (consisting of the distribution of Crestline common stock in an estimated amount equal to approximately \$1.30 per share and the distribution of the Special Dividend in an estimated amount equal to approximately \$1.20 per share), and a pre-distribution Host stock price of \$15.00 per share (resulting in an assumed post distribution in Host stock price of \$12.50 per share), the Company estimates that such anti-dilution adjustments would result in a decrease in the conversion price from \$18.604 to approximately \$15.50 and approximately \$.9 million additional shares becoming issuable upon conversion of the debentures and Convertible Preferred Shares following the Initial E&P Distribution. The actual adjustments may differ from this estimate and the differences may be material.

If Host's preliminary estimate of \$231 million of cash distributions by the Operating Partnership during the twelve months ending December 31, 1999 proves accurate but the Operating Partnership's aggregate estimated cash available for distribution were less than \$144 million and estimated cash from contingent rents were only \$64 million during the twelve months ending December 31, 1999, then the Operating Partnership would be required to borrow approximately \$23 million (or \$0.08 per OP Unit) to make such distributions to enable Host REIT to distribute 100% of its estimated taxable income in accordance with its distribution policy. Moreover, if estimated cash from contingent rents were less than \$64 million, then the Operating Partnership also would be required to borrow any such shortfall in order to make such distributions. While the Operating Partnership does not believe this will be necessary, it believes it would be able to borrow the necessary amounts under the New Credit Facility or from other sources and that any such borrowing would not have a material adverse effect on its financial condition or results of operations.

The distributions to stockholders per share of Host REIT Common Stock are expected to be equal to the amount distributed by the Operating Partnership per OP Unit. However, if the REIT Conversion is not completed until after January 1, 1999, then Host REIT's distributions to stockholders in 1999 would be lower than the Operating Partnership's distributions per OP Unit (by the amount of Host REIT's 1999 corporate income tax payments) until its REIT election becomes effective, which would be no later than January 1, 2000. The Operating Partnership intends to make distributions during 1999 at the estimated level described above even if the REIT election of Host REIT were not effective until January 1, 2000. The following table describes the calculation of Host REIT's estimated initial cash distributions and estimated cash distributions per share of Host REIT Common Stock for the twelve months ending December 31, 1999, based on the Operating Partnership's estimated cash distributions of \$0.84 per OP Unit, if the REIT Conversion were to occur on January 1, 1999 but Host REIT's REIT election were not effective until January 1, 2000:

(	DOLLAF	RS IN	
MIL	LIONS	EXCEPT	
PER	SHARE	AMOUNT	)

Estimated cash distributions by the Operating Partnership for the twelve months ending December 31, 1999  Less: Estimated cash distributions to OP Unitholders (other than Host REIT)	\$ 231 (59)
Host REIT's share of estimated cash distributions by the Operating Partnership for the twelve months ending December 31, 1999	172 (65)
Estimated cash distributions by Host REIT (if Host REIT has not yet made a REIT election) for the twelve months ending December 31, 1999	\$107 =====
Estimated cash distributions per share of Host REIT Common Stock (if Host REIT has not yet made a REIT election) for the twelve months ending December 31, 1999(2)	\$0.53 ====

<sup>(1)</sup> Estimated cash tax payments based on applying Host REIT's blended statutory tax rate (assumed to be a federal rate of 35%, plus a blended state rate of 5% net of the federal benefit), taking into account utilization of Host REIT's estimated alternative minimum tax ("AMT") credit carryforwards (approximately \$21 million) and estimated AMT preferences (approximately \$55 million), and applying the resulting effective rate (25%) to estimated taxable income for the year.

Investors are cautioned that Host expects that its preliminary estimate of 1999 taxable income (and the resulting estimated distributions during 1999) may materially change as a result of issuances of additional common or preferred stock by Host either prior to or following the Partnership Mergers (which could reduce the distribution per OP Unit in accordance with its distribution policy), changes in operations, acquisitions or dispositions of assets, changes in the preliminary estimate of taxable income for 1999 and various other factors (some of which may be beyond the control of Host REIT and the Operating Partnership). Distributions will be made in the discretion of Host REIT's Board of Directors and will be affected by a number of factors, including the rental payments received by the Operating Partnership from the Lessees with respect to the Leases of the

<sup>(2)</sup> Based on a total of 204.5 million shares of Host REIT Common Stock outstanding.

Hotels, the operating expenses of the Operating Partnership, the level of borrowings and interest expense incurred in borrowing, the Operating Partnership's financial condition and cash available for distribution, the taxable income of Host REIT and the Operating Partnership, the effects of acquisitions and dispositions of assets, unanticipated capital expenditures and distributions required to be made on any preferred units issued by the Operating Partnership. Actual results may vary substantially from the estimates and no assurance can be given that the Operating Partnership's estimates will prove accurate or that any level of distributions will be made or sustained.

For a discussion of the tax treatment of distributions to the holders of Host REIT Common Stock, see "Federal Income Tax Consequences--Taxation of Taxable U.S. Stockholders Generally," "--Taxation of Tax-Exempt Stockholders of Host REIT" and "--Taxation of Non-U.S. Stockholders." For a discussion of the annual distribution requirements applicable to REITs, see "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Merger--Annual Distribution Requirements Applicable to REITs."

# INVESTMENT POLICIES

Investments in Real Estate or Interests in Real Estate. Host REIT is required to conduct all of its investment activities through the Operating Partnership. The Company's investment objectives are to (i) achieve long-term sustainable growth in Funds From Operations per share of Host REIT Common Stock, (ii) increase asset values by improving and expanding the initial Hotels, as appropriate, (iii) acquire additional existing and newly developed upscale and luxury full-service hotels in targeted markets, (iv) develop and construct upscale and luxury full-service hotels and (v) potentially pursue other real estate investments. The Company's business will be primarily focused on upscale and luxury full-service hotels. Where appropriate, and subject to REIT qualification rules and limitations contained in the Partnership Agreement, the Company may sell certain of its hotels.

The Company also may participate with other entities in property ownership through joint ventures or other types of co-ownership. Equity investments may be subject to existing mortgage financing and other indebtedness or such financing or indebtedness may be incurred in connection with acquiring investments. Any such financing or indebtedness will have priority over the Company's equity interest in such property.

Investments in Real Estate Mortgages. While the Company will emphasize equity real estate investments, it may, in its discretion, invest in mortgages and other similar interests. The Company does not intend to invest to a significant extent in mortgages or deeds of trust, but may acquire mortgages as a strategy for acquiring ownership of a property or the economic equivalent thereof, subject to the investment restrictions applicable to REITs. See "Business and Properties--Blackstone Acquisition," "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Mergers--Income Tests Applicable to REITs" and "--Asset Tests Applicable to REITs." As of June 19, 1998, the Company held two mortgages secured by hotels. In addition, the Company may invest in mortgage-related securities and/or may seek to issue securities representing interests in such mortgage-related securities as a method of raising additional funds.

Securities of or Interests in Persons Primarily Engaged in Real Estate Activities and Other Issuers. Subject to the percentage ownership limitations and gross and asset income tests necessary for REIT qualification, the Company also may invest in securities of other entities engaged in real estate activities or invest in securities of other issuers, including for the purpose of exercising control over such entities. The Company may acquire all or substantially all of the securities or assets of other REITs or similar entities where such investments would be consistent with the Company's investment policies. No such investments will be made, however, unless the Board of Directors determines that the proposed investment would not cause either Host REIT or the Operating Partnership to be an "investment company" within the meaning of the Investment Company Act of 1940, as amended.

120

## FINANCING POLICIES

The Operating Partnership's and Host REIT's organizational documents currently contain no restrictions on incurring debt. The Company, however, will have a policy of incurring debt only if upon such incurrence the debt-to-total market capitalization of Host REIT and the Operating Partnership would be 60% or less. In addition, the New Senior Notes indenture and the New Credit Facility impose limitations on the incurrence of indebtedness. The indenture for the Notes also limits the amount of debt that the Operating Partnership may incur if, immediately after giving effect to the incurrence of such additional debt, the aggregate principal amount of all outstanding debt of the Operating Partnership and its Subsidiaries (as defined in the Indenture) on a consolidated

basis is greater than 75% of the Operating Partnership's undepreciated total assets on the date of such incurrence. Indentures for debt issued to replace the public bonds may contain other restrictions. The Company may, from time to time, reduce its outstanding indebtedness by repurchasing a portion of such outstanding indebtedness, subject to certain restrictions contained in the Partnership Agreement and the terms of its outstanding indebtedness. The Company will from time to time reevaluate its borrowing policies in light of then current economic conditions, relative costs of debt and equity capital, market conditions, market values of properties, growth and acquisition opportunities and other factors. Consequently, the Company's financing policy is subject to modification and change. The Company may waive or modify its borrowing policy without any vote of the stockholders of Host REIT.

To the extent that the Board of Directors determines to seek additional capital, the Company may raise such capital through equity offerings, debt financing or retention of cash flow or a combination of these methods. As long as the Operating Partnership is in existence, the net proceeds of all equity capital raised by Host REIT will be contributed to the Operating Partnership in exchange for OP Units in the Operating Partnership, which will dilute the ownership interest of limited partners of the Operating Partnership.

In the future, the Company may seek to extend, expand, reduce or renew its New Credit Facility, or obtain new credit facilities or lines of credit, subject to its general policy relating to the ratio of debt-to-total market capitalization, for the purpose of making acquisitions or capital improvements or providing working capital or meeting the taxable income distribution requirements for REITs under the Code. In the future, the Company also may determine to issue securities senior to the Host REIT Common Stock or OP Units, including preferred shares and debt securities (either of which may be convertible into Host REIT Common Stock or OP Units or may be accompanied by warrants to purchase Host REIT Common Stock or OP Units).

The Company has not established any limit on the number or amount of mortgages that may be placed on any single hotel or on its portfolio as a whole, although the Company's objective is to reduce its reliance on secured indebtedness.

# LENDING POLICIES

The Company may consider offering purchase money financing in connection with the sale of a hotel where the provision of such financing will increase the value received by the Company for the hotel sold.

# CONFLICTS OF INTEREST POLICIES

Under the MGCL, no contract or transaction between a Maryland corporation and any of its directors or between a Maryland corporation and any other corporation, firm, or other entity in which any of its directors is a director, or has a material financial interest, shall be void or voidable solely for this reason, or solely because the director is present at the meeting of the board or committee of the board which authorizes, approves, or ratifies the contract or transaction, or solely because such director's or directors' votes are counted for such purpose, if (i) the fact of common directorship or interest is disclosed or known to the board of directors or the committee, and the board or committee authorizes, approves, or ratifies the contract or transaction by the affirmative vote of a majority of disinterested directors, even if the disinterested directors constitute less than a quorum, (ii) the fact of common directorship or interest is disclosed or known to the stockholders entitled to vote, and the contract or

transaction is authorized, approved, or ratified by a majority of the votes cast by the stockholders entitled to vote other than the votes of shares owned of record or beneficially by the interested corporation, firm or other entity, or (iii) the contract or transaction is fair and reasonable to the corporation. Common or interested directors or the stock owned by them or by an interested corporation, firm, or other entity may be counted in determining the presence of a quorum at a meeting of the board of directors or a committee of the board or at a meeting of the stockholders, as the case may be, at which the contract or transaction is authorized, approved or ratified.

Host REIT's Board of Directors also has adopted a policy to address conflicts of interest. In addition, Maryland and Delaware law impose certain duties on the Board of Directors and Host REIT, as general partner of the Operating Partnership (to the extent such duties have not been eliminated pursuant to the Host REIT Charter or the Partnership Agreement). There can be no assurance, however, that these policies always will be successful in eliminating the influence of such conflicts. If they are not successful, decisions could be made that may fail to reflect fully the interests of all holders of Host REIT Common Stock and limited partners of the Operating Partnership.

Host REIT has adopted a policy which would require that all material contracts and transactions between Host REIT, the Operating Partnership or any of its subsidiaries, on the one hand, and a director or executive officer of Host REIT or any entity in which such director or executive officer is a director or has a material financial interest, on the other hand, must be approved by the affirmative vote of a majority of the disinterested directors. Where appropriate in the judgment of the disinterested directors, the Board of Directors may obtain a fairness opinion or engage independent counsel to represent the interests of nonaffiliated security holders, although the Board of Directors will have no obligation to do so.

In addition, under Delaware law (where the Operating Partnership is formed), Host REIT, as general partner, has a fiduciary duty to the Operating Partnership and, consequently, such transactions are subject to the duties of care and loyalty that Host REIT, as general partner, owes to limited partners of the Operating Partnership (to the extent such duties have not been eliminated pursuant to the terms of the Partnership Agreement). The Partnership Agreement provides that (i) in considering to dispose of any of the assets of the Operating Partnership, Host REIT shall take into account the tax consequences to it of any such disposition and shall have no liability to the Operating Partnership or the limited partners for decisions based upon or influenced by such tax consequences (and the Operating Partnership generally is obligated to pay any taxes Host REIT incurs as result of such transactions), (ii) Host REIT, as general partner, is under no obligation to consider the separate interests of the limited partners (including, without limitation, tax consequences) in deciding whether to cause the Operating Partnership to take, or decline to take, any action and (iii) any act or omission by Host REIT, as a general partner, undertaken in the good faith belief that such action is necessary or desirable to protect the ability of Host REIT to continue to qualify as a REIT or to allow Host REIT to avoid incurring liability for taxes under Section 857 or 4981 of the Code (relating to required distributions) is deemed approved by all limited partners.

J.W. Marriott, Jr. and Richard E. Marriott, who are brothers, currently serve as directors of Host and directors (and, in the case of J.W. Marriott, Jr., also an officer) of Marriott International. After the REIT Conversion, J.W. Marriott, Jr. will serve as a director of Host REIT and will continue to serve as the Chairman of the Board and Chief Executive Officer of Marriott International, and Richard E. Marriott will serve as Chairman of the Board of Host REIT and continue to serve as a director of Marriott International. J.W. Marriott, Jr. and Richard E. Marriott also beneficially own (as determined for securities law purposes) approximately 10.6% and 10.2%, respectively, of the outstanding shares of common stock of Marriott International and will beneficially own approximately 5.33% and 5.31%, respectively, of the outstanding shares of Crestline (but neither will serve as an officer or director thereof). Because they will serve as directors of Host REIT, as well as directors (and in the case of J.W. Marriott, Jr., the Chief Executive Officer) of Marriott International they may be subject to certain potential conflicts of interest in fulfilling their responsibilities to Host REIT and its stockholders. See "Risk Factors--Conflicts of Interest--Relationships with Marriott International and Crestline."

# POLICIES WITH RESPECT TO OTHER ACTIVITIES

The Company may, but does not presently intend to, make investments other than as previously described. Host REIT will make investments only through the Operating Partnership. Host REIT and the Operating Partnership will have authority to offer their securities and to repurchase or otherwise reacquire their securities and may engage in such activities in the future. Host REIT and the Operating Partnership also may make loans to joint ventures in which they may participate in the future to meet working capital needs. Neither Host REIT nor the Operating Partnership will engage in trading, underwriting, agency distribution or sale of securities of other issuers. Host REIT's policies with respect to such activities may be reviewed and modified from time to time by Host REIT's directors without notice to, or the vote of, its stockholders.

## SELECTED FINANCIAL DATA

The following table presents certain selected historical financial data of Host which has been derived from Host's audited Consolidated Financial Statements for the five most recent fiscal years ended January 2, 1998 and the unaudited condensed consolidated financial statements for the First Three Quarters 1998 and First Three Quarters 1997. The income statement data for fiscal year 1993 does not reflect the Marriott International Distribution and related transactions and, accordingly, the table presents data for Host for 1993 that includes amounts attributable to Marriott International. As a result of the Marriott International Distribution and related transactions, the assets, liabilities and businesses of Host have changed substantially.

The information contained in the following table is not comparable to the operations of Host or the Operating Partnership on a going-forward basis because the historical information relates to an operating entity which owns and operates its hotels, while the Company will own the Hotels but will lease them to the Lessees and receive rental payments in connection therewith.

	FIRST THREE QUARTERS						
	1998(1)	1997(1)	1997(1)	1996(2)	1995(3)	1994(1)	1993(1)(4)
	(UNAUD	ITED)	(IN	MILLIONS	5)		
INCOME STATEMENT DATA:							
Revenues	\$1,040	\$ 768	\$1,147	\$ 732	\$ 484	\$ 380	\$ 659
Operating profit	493	304	449	233	114	152	92
Interest expense	245	198	302	237	178	165	164
Income (loss) from con-							
tinuing operations	100	38	47	(13)	(62)	(13)	56
Net income (loss)(5)	(48)	43	50	(13)	(143)	(25)	50
OTHER OPERATING DATA:							
Cash from operations	278	313	464	201	142	146	415
Cash used in investing							
activities	(273)	(497)	(1,046)	(504)	(208)	(178)	(262)
Cash provided by (used							
in) financing activi-							()
ties	23	391	389	806	200	26	(389)
Comparative FFO(6) (un-					400		
audited)	282	202	295	164	136	N/A	N/A
Depreciation and amor-	404	450	0.40	400	400	440	NI /A
tization	184	158	240	168	122	113	N/A
RATIO DATA (UNAUDITED): Ratio of earnings to							
fixed charges(7)	1.7x	1.4x	1.3x	1.0x			
Deficiency of earnings	1.78	1.48	1.38	1.00			
to fixed charges(7)					70	12	45
BALANCE SHEET DATA:					70	12	45
Cash, cash equivalents							
and short-term							
marketable							
securities	\$ 575	\$ 911	\$ 865	\$ 704	\$ 201	\$ 67	\$ 73
Total assets		6,362		5,152			3,362
Debt	•	3,634	3,783	2,647	2,178	1,871	2,113
	,	-,	-,	, -	, -	,	, ,

- (1) In the First Three Quarters of 1998, Host recognized a \$148 million extraordinary loss, net of taxes, on the extinguishment of certain debt. In the First Three Quarters 1997 and fiscal year 1997, Host recognized a \$5 million and a \$3 million, respectively, extraordinary gain, net of taxes, on the extinguishment of certain debt. In 1994, Host recognized a \$6 million extraordinary loss, net of taxes, on the required redemption of senior notes. In 1993, Host recognized a \$4 million extraordinary loss, net of taxes, on the completion of an exchange offer for its then outstanding bonds.
- (2) Fiscal year 1996 includes 53 weeks.
- (3) Operating results for 1995 include a \$10 million pre-tax charge to write down the carrying value of five limited service properties to their net realizable value and a \$60 million pre-tax charge to write down an undeveloped land parcel to its estimated sales value. In 1995, Host recognized a \$20 million extraordinary loss, net of taxes, on the extinguishment of debt.
  (4) Operating results for 1993 include the operations of Marriott
- (4) Operating results for 1993 include the operations of Marriott International through the Marriott International Distribution date of October 8, 1993. These operations had a net pre-tax effect on income of \$211 million for the year ended December 31, 1993 and are recorded as "Profit from operations distributed to Marriott International" on Host's consolidated statements of operations and are, therefore, not included in sales, operating profit before corporate expenses and interest, interest expense and interest income for the same period. The net pre-tax effect of these operations is, however, included in income before income taxes, extraordinary item and cumulative effect of changes in accounting principles and in net income for the same periods. Statement of Financial Accounting Standards ("SFAS") No. 109, "Accounting for Income Taxes," was adopted in the first quarter of 1993. In the second quarter of 1993, Host changed its accounting method for assets held for sale. During 1993, Host recorded a \$34 million credit to reflect the adoption of SFAS No. 109 and

- a \$32 million charge, net of taxes, to reflect the change in its accounting method for assets held for sale. Operating results in 1993 included pre-tax expenses related to the Marriott International Distribution totaling \$13 million.
- Distribution totaling \$13 million.

  (5) Host recorded a loss from discontinued operations, net of taxes, as a result of the Special Dividend (as defined herein) of \$61 million in 1995, \$6 million in 1994, and \$4 million in 1993. The 1995 loss from discontinued operations includes a pre-tax charge of \$47 million

for the adoption of SFAS No. 121, "Accounting For the Impairment of Long-Lived Assets and Long-Lived Assets to be Disposed Of," a pre-tax \$15 million restructuring charge and an extraordinary loss of \$10 million, net of taxes, on the extinguishment of debt.

- (6) Host considers Comparative Funds From Operations ("Comparative FFO," which represents Funds From Operations, as defined by NAREIT, plus deferred tax expense) a meaningful disclosure that will help the investment community to better understand the financial performance of Host, including enabling its stockholders and analysts to more easily compare Host's performance to REITS. FFO is defined by NAREIT as net income computed in accordance with GAAP, excluding gains or losses from debt restructurings and sales of properties, plus real estate related depreciation and amortization, and after adjustments for unconsolidated partnerships and joint ventures. FFO should not be considered as an alternative to net income, operating profit, cash flows from operations or any other operating or liquidity performance measure prescribed by GAAP. FFO is also not an indicator of funds available to fund Host's cash needs, including its ability to make distributions. Host's method of calculating FFO may be different from methods used by other REITs and, accordingly, may not be comparable to such other REITs.
- (7) The ratio of earnings to fixed charges is computed by dividing net income before interest expense and other fixed charges by total fixed charges, including interest expense, amortization of debt issuance costs and the portion of rent expense that is deemed to represent interest. The deficiency of earnings to fixed charges is largely the result of depreciation and amortization of \$122 million and \$113 million in 1995 and 1994, respectively.

# MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

## LACK OF COMPARABILITY FOLLOWING THE MERGER AND THE REIT CONVERSION

Because substantially all of the Company's Hotels will be leased following the Merger and the REIT Conversion, the Company does not believe that the historical results of operations will be comparable to the results of operations of Host following the Merger and the REIT Conversion. For pro forma information giving effect to the Merger and the REIT Conversion (including the Leases), see "Pro Forma Financial Statements."

## HISTORICAL RESULTS OF OPERATIONS

Revenues primarily represent house profit from hotel properties and senior living communities, net gains (losses) on property transactions and equity in the earnings (losses) of affiliates. House profit reflects the net revenues flowing to Host as property owner and represents gross hotel sales less property-level expenses (excluding depreciation, management fees, property taxes, ground and equipment rent, insurance and certain other costs which are classified as operating costs and expenses included in the accompanying financial statements). Other operating costs and expenses include idle land carrying costs and certain other costs.

Host's hotel operating costs and expenses are, to a great extent, fixed. Therefore, Host derives substantial operating leverage from increases in revenue. This operating leverage is somewhat diluted, however, by the impact of base management fees which are calculated as a percentage of sales, variable lease payments and incentive management fees tied to operating performance above certain established levels. Successful hotel performance resulted in certain of Host's properties reaching levels which allowed the manager to share in the growth of profits in the form of higher management fees. Host expects that this trend will continue in 1998 as the upscale and luxury full-service segments continue to strengthen. At these higher operating levels, Host's and the managers' interests are closely aligned, which helps to drive further increases in profitability, but moderates operating leverage.

For the periods discussed herein, Host's hotel properties have experienced substantial increases in REVPAR. REVPAR is a commonly used indicator of market performance for hotels which represents the combination of the average daily room rate charged and the average occupancy achieved. REVPAR does not include food and beverage or other ancillary revenues generated by the property. The REVPAR increase primarily represents strong percentage increases in room rates, while occupancy increases have been more moderate. Increases in average room rates have generally been achieved by the managers through shifting occupancies away from discounted group business to higher-rated group and transient business and by selectively increasing room rates. This has been made possible by increased travel due to improved economic conditions and by the favorable supply/demand characteristics existing in the upscale and luxury full-service segments of the lodging industry. Host expects this favorable relationship between supply growth and demand growth to continue in the upscale and luxury markets in which it operates, which should result in improved REVPAR and operating profits at its hotel properties in the near term. However, there can be no assurance that REVPAR will continue to increase in the future.

FIRST THREE QUARTERS 1998 COMPARED TO FIRST THREE QUARTERS 1997 (HISTORICAL)

Revenues. Revenues primarily represent house profit from Host's hotel properties, net gains (losses) on property transactions and equity in earnings (losses) of affiliates. Revenues increased \$272 million, or 35%, to \$1,040 million for the First Three Quarters 1998 from \$768 million for the First Three Quarters 1997. Host's revenue and operating profit were impacted by improved lodging results for comparable full-service hotel properties, the addition of 18 full-service hotel properties during 1997 and 11 full-service properties during the First Three Quarters 1998, the acquisition of 30 senior living communities in 1997 and one senior living community in 1998 and the gain on the sale of two hotel properties in the First Three Quarters 1998.

Hotel revenues increased \$186 million, or 25%, to \$922 million for the First Three Quarters 1998 due to growth in REVPAR and the addition of 29 full-service properties acquired in 1997 and through the first three quarters of 1998.

Hotel sales (gross hotel sales, including room sales, food and beverage sales, and other ancillary sales such as telephone sales) increased \$449 million, or 24%, to over \$2.3 billion in the First Three Quarters 1998, reflecting the REVPAR increases for comparable units and the addition of full-service properties in 1997 and 1998. Improved results for Host's full-service hotels were driven by strong increases in REVPAR for comparable units of 7.9% to \$113.27 for the First Three Quarters 1998. Results were further enhanced by a one percentage point increase in the house profit margin for comparable full-service properties. On a comparable basis for Host's full-service hotel properties, average room rates increased nearly eight percent with no change in occupancy.

Revenues generated from Host's 31 senior living communities totaled \$58 million for the First Three Quarters 1998 compared to \$16 million for the First Three Quarters 1997 (as the assets were purchased in the third quarter of 1997). For the First Three Quarters 1998, average occupancy was nearly 92% and the average per diem rate was \$88.19. Senior living communities' sales totaled \$166 million for the First Three Quarters 1998.

Revenues were also impacted by the gains on the sales of two hotel properties. The New York East Side Marriott was sold for \$191 million resulting in a pre-tax gain of approximately \$40 million. The Napa Valley Marriott was sold for \$21 million resulting in a pre-tax gain of approximately \$10 million.

Operating Costs and Expenses. Operating costs and expenses principally consist of depreciation, management fees, real and personal property taxes, ground, building and equipment rent, insurance and certain other costs. Operating costs and expenses increased \$83 million to \$547 million, primarily representing increased hotel and senior living communities operating costs. Hotel operating costs increased \$69 million to \$502 million for the First Three Quarters 1998, primarily due to the addition of 29 full-service hotel properties during 1997 and through the first three quarters of 1998 and increased management fees and rentals tied to improved property results. As a percentage of hotel revenues, hotel operating costs and expenses, from 63% and 59% of revenues for the First Three Quarters 1998, due to the significant increases in REVPAR discussed above, as well as the operating leverage as a result of a significant portion of the Company's hotel operating costs and expenses being fixed. The Company's senior living communities' operating costs and expenses were \$31 million for the First Three Quarters 1998, compared to \$9 million for the First Three Quarters 1998. As the senior living communities were purchased at the beginning of the 1997 third quarter, year-to-date operating costs and expenses are not comparable.

Operating Profit. As a result of the changes in revenues and operating costs and expenses discussed above, the Company's operating profit increased \$189 million, or 62%, to \$493 million for the First Three Quarters 1998. For the First Three Quarters 1998, hotel operating profit increased \$117 million, or 39%, to \$420 million, or 46% of hotel revenues, for 1998 compared to \$303 million, or 41% of hotel revenues, for 1997. Specifically, hotels in New York City, San Francisco, Toronto and Atlanta reported significant improvements for the 1998 third quarter over the third quarter of 1997.

Minority Interest. Minority interest expense increased \$12 million to \$36 million for the First Three Quarters 1998, primarily reflecting the impact of the consolidation of affiliated partnerships and the acquisition of controlling interests in newly-formed partnerships during 1997 and 1998.

127

Corporate Expenses. Corporate expenses increased \$6 million to \$33 million for the First Three Quarters 1998. As a percentage of revenues, corporate expenses decreased to 2.8% of revenues for the First Three Quarters 1998 from 1.1% in the First Three Quarters 1997, reflecting Host's efforts to control its corporate expenses in spite of the substantial growth in revenues.

REIT Conversion Expenses. REIT Conversion Expenses reflect the professional fees and other expenses associated with the Company's conversion to a REIT and totaled \$14 million for the First Three Quarters 1998.

Interest Expense. Interest expense increased 24% to \$245 million in the First Three Quarters 1998, primarily due to additional debt assumed in connection with the 1997 and 1998 full-service hotel and senior living community additions as well as the issuance of the New Senior Notes and the New Credit Facility.

Dividends on Convertible Preferred Securities. The dividends on Convertible Preferred Securities reflect the dividends accrued on the \$550 million in 6.75% Convertible Preferred Securities issued by Host in December 1996.

Interest Income. Interest income decreased \$1 million to \$36 million for the First Three Quarters 1998, primarily reflecting the lower level of cash and marketable securities held in 1998 compared to 1997.

Income before Extraordinary Item. Income before extraordinary item for the First Three Quarters 1998 was \$100 million, compared to \$38 million for the First Three Quarters 1997.

Extraordinary Gain (Loss). In connection with the purchase of the Old Senior Notes, Host recognized an extraordinary loss of \$148 million, which represents the bond premium and consent payments totaling approximately \$175 million and the write-off of deferred financing fees of approximately \$52 million related to the Old Senior Notes, net of taxes. In March 1997, Host purchased 100% of the outstanding bonds secured by a first mortgage on the San Francisco Marriott Hotel. Host purchased the bonds for \$219 million, which was an \$11 million discount to the face value of \$230 million. In connection with the redemption and defeasance of the bonds, Host recognized an extraordinary gain of \$5 million, which represents the \$11 million discount and the write-off of deferred financing fees, net of taxes.

Net Income (Loss). The significant net loss for the First Three Quarters 1998 was due to the \$148 million extraordinary loss on the extinguishment of debt. Net loss for the First Three Quarters 1998 was \$48 million compared to net income of \$43 million for the First Three Quarters 1997. Basic and diluted earnings (loss) per common share were \$(.24) and \$(.23), respectively, for 1998 and \$.21 for the First Three Quarters 1997.

# 1997 COMPARED TO 1996 (HISTORICAL)

Revenues. Revenues increased \$415 million, or 57%, to \$1.1 billion for 1997. Host's revenue and operating profit were impacted by:

- -- improved lodging results for comparable full-service hotel properties;
- -- the addition of 23 full-service hotel properties during 1996 and 18 full-service properties during 1997;
- -- the addition of 30 senior living communities in 1997;
- -- the 1996 sale and leaseback of 16 Courtyard properties and 18 Residence Inns; and
- -- the 1997 results including 52 weeks versus 53 weeks in 1996.

Hotel revenues increased \$376 million, or 52% to \$1.1 billion in 1997, as all three of the Company's lodging concepts reported growth in REVPAR. Hotel sales increased \$864 million, or 44%, to over \$2.8 billion in 1997, reflecting the REVPAR increases for comparable units and the addition of full-service properties during 1996 and 1997. Improved results for the Company's full-service hotels were driven by strong increases in REVPAR for comparable units of 12.6% in 1997. Results were further enhanced by a more than two percentage point

increase in the house profit margin for comparable full-service properties. On a comparable basis for Host's full-service properties, average room rates increased almost 11%, while average occupancy increased over one percentage point.

Revenues generated from Host's 1997 third quarter acquisition of 29 senior living communities totaled \$37 million. During 1997, average occupancy of the communities was 92% and the average per diem rate was \$84. Overall occupancies for 1997 were lower than the historical and anticipated future occupancies due to the significant number of expansion units added during the year, the overall disruption to the communities as a result of the construction and the time required to fill the expansion units. Senior living communities' sales totaled \$111 million for 1997.

Operating Costs and Expenses. Operating costs and expenses increased \$199 million to \$698 million for 1997, primarily representing increased hotel and senior living communities' operating costs, including depreciation and management fees. Hotel operating costs increased \$188 million to \$649 million, primarily due to the addition of 41 full-service properties during 1996 and 1997, and increased management fees and rentals tied to improved property results. As a percentage of hotel revenues, hotel operating costs and expenses decreased to 59% of revenues for 1997, from 64% of revenues for 1996, reflecting the impact of increased 1997 revenues on relatively fixed operating costs and expenses. Host's senior living communities operating costs and expenses were \$20 million (54% of revenues) for 1997.

Operating Profit. As a result of the changes in revenues and operating costs and expenses discussed above, Host's operating profit increased \$216 million, or 93%, to \$449 million in 1997. Hotel operating profit increased \$188 million, or 73%, to \$444 million, or 41% of hotel revenues, for 1997 compared to \$256 million, or 36% of hotel revenues, for 1996. In nearly all markets, Host's hotels recorded improvements in comparable operating results. In particular, Host's hotels in the Northeast, Mid-Atlantic and Pacific coast regions benefited from the upscale and luxury full-service room supply and demand imbalance. Hotels in New York City, Philadelphia, San Francisco/Silicon Valley and in Southern California performed particularly well. In 1998, Host expects results to be strong in these markets and other gateway cities in which the Company owns hotels. In 1997, Host's suburban Atlanta properties (three properties totaling 1,022 rooms) generally reported decreased results due to higher activity in 1996 related to the Summer Olympics and the impact of the additional supply added to the suburban areas. However, the majority of Host's hotel rooms in Atlanta are in the core business districts in downtown and Buckhead where they realized strong year-over-year results and were only marginally impacted by the additional supply. Host's senior living communities generated \$17 million (46% of revenues) of operating profit.

Minority Interest. Minority interest expense increased \$26 million to \$32 million for 1997, primarily reflecting the impact of the consolidation of affiliated partnerships and the acquisition of controlling interests in newlyformed partnerships during 1996 and 1997.

Corporate Expenses. Corporate expenses increased \$4 million to \$47 million in 1997. As a percentage of revenues, corporate expenses decreased to 4.1% of revenues in 1997 from 5.9% of revenues in 1996. This reflects Host's efforts to control its corporate expenses in spite of the substantial growth in revenues.

Interest Expense. Interest expense increased \$65 million to \$302 million in 1997, primarily due to the additional mortgage debt of approximately \$1.1 billion assumed in connection with the 1996 and 1997 full-service hotel additions, approximately \$315 million in debt incurred in conjunction with the acquisition of senior living communities, as well as the issuance of \$600 million of 8 7/8% senior notes in July 1997.

Dividends on Convertible Preferred Securities of Subsidiary Trust. The dividends on the Convertible Preferred Securities reflect the dividends on the \$550 million in 6.75% Convertible Preferred Securities issued by Host in December 1996.

Interest Income. Interest income increased \$4 million to \$52 million for 1997, primarily reflecting the interest income on the available proceeds generated by the December 1996 offering of Convertible Preferred Securities and the proceeds generated by the issuance of the 8 7/8% senior notes in July 1997.

Income (Loss) Before Extraordinary Items. Income before extraordinary items for 1997 was \$47 million, compared to a \$13 million loss before extraordinary items for 1996 as a result of the items discussed above.

Extraordinary Gain (Loss). In March 1997, Host purchased 100% of the outstanding bonds secured by a first mortgage on the San Francisco Marriott Hotel. Host purchased the bonds for \$219 million, which was an \$11 million discount to the face value of \$230 million. In connection with the redemption and defeasance of the bonds, Host recognized an extraordinary gain of \$5 million, which represents the \$11 million discount less the write-off of unamortized deferred financing fees, net of taxes. In December 1997, Host refinanced the mortgage debt secured by Marriott's Orlando World Center. In connection with the refinancing, Host recognized an extraordinary loss of \$2 million, which represents payment of a prepayment penalty and the write-off of unamortized deferred financing fees, net of taxes.

Net Income (Loss). The Company's net income in 1997 was \$50 million, compared to a net loss of \$13 million in 1996. Basic earnings per common share was \$.25 for 1997, compared to a basic loss per common share of \$.07 in 1996. Diluted earnings per common share was \$.24 for 1997 compared to a diluted loss per common share of \$.07 in 1996.

## 1996 COMPARED TO 1995 (HISTORICAL)

Revenues. Revenues increased \$248 million, or 51%, to \$732 million in 1996. Host's revenue and operating profit were impacted by:

- -- improved lodging results for comparable full-service hotel properties;
- -- the addition of nine full-service hotel properties during 1995 and 23 full-service properties during 1996;
- -- the 1996 and 1995 sale and leaseback of 53 of Host's Courtyard properties and 18 of Host's Residence Inns;
- -- the 1996 change in the estimated depreciable lives and salvage values for certain hotel properties which resulted in additional depreciation expense of \$15 million;
- -- the 1996 results including 53 weeks versus 52 weeks in 1995;
- -- the \$60 million pre-tax charge in 1995 to write down the carrying value of one undeveloped land parcel to its estimated sales value;
- -- a \$10 million pre-tax charge in 1995 to write down the carrying value of certain Courtyard and Residence Inn properties held for sale to their net realizable values included in "Net gains (losses) on property transactions"; and
- -- the 1995 sale of four Fairfield Inns.

Hotel revenues increased \$243 million, or 51%, to \$717 million in 1996, as all three of Host's lodging concepts reported growth in REVPAR. Hotel sales increased \$590 million, or 44%, to \$1.9 billion in 1996, reflecting the REVPAR increases for comparable units and the addition of full-service properties during 1995 and 1996.

Improved results for Host's full-service hotels were driven by strong increases in REVPAR for comparable units of 11% in 1996. Results were further enhanced by an almost two percentage point increase in the house profit margin for comparable full-service properties. On a comparable basis for Host's full-service properties, average room rates increased 8%, while average occupancy increased over two percentage points.

Operating Costs and Expenses. Operating costs and expenses increased \$129 million to \$499 million for 1996, primarily representing increased hotel operating costs, including depreciation, partially offset by the \$60 million pre-tax charge in 1995 to write down the carrying value of one undeveloped land parcel to its estimated

sales value. Hotel operating costs increased \$180 million to \$461 million, primarily due to the addition of 32 full-service properties during 1995 and 1996, increased management fees and rentals tied to improved property results and a change in the depreciable lives and salvage values of certain large hotel properties (\$15 million). As a percentage of hotel revenues, hotel operating costs and expenses increased to 64% of revenues for 1996, from 59% of revenues for 1995, reflecting the impact of the lease payments on the Courtyard and Residence Inn properties which have been sold and leased back, and the change in depreciable lives and salvage values for certain large hotel properties discussed above, as well as the shifting emphasis to full-service properties. Full-service hotel rooms accounted for 100% of Host's total hotel rooms on January 3, 1997, versus 84% on December 29, 1995.

Operating Profit. As a result of the changes in revenues and operating costs and expenses discussed above, Host's operating profit increased \$119 million, or 104%, to \$233 million in 1996. Hotel operating profit increased \$63 million, or 33%, to \$256 million, or 36% of hotel revenues, for 1996 compared to \$193 million, or 41% of hotel revenues, for 1995. Across the board, the Company's hotels recorded substantial improvements in comparable operating results. In addition, several hotels, including the New York Marriott Marquis, the New York Marriott East Side, the Philadelphia Marriott, the San Francisco Marriott and the Miami Airport Marriott posted particularly significant improvements in operating profit for the year. Host's Atlanta properties also posted outstanding results, primarily due to the 1996 Summer Olympics. Additionally, several hotels which recently converted to the Marriott brand, including the Denver Marriott Tech Center, the Marriott's Mountain Resort at Vail and the Williamsburg Marriott, recorded strong results compared to the prior year as they completed renovations and began to realize the benefit of their conversions.

Corporate Expenses. Corporate expenses increased \$7 million to \$43 million in 1996. As a percentage of revenues, corporate expenses decreased to 5.9% of revenues in 1996 from 7.4% of revenues in 1995. This reflects Host's efforts to control its corporate administrative expenses in spite of the substantial growth in revenues.

Interest Expense. Interest expense increased 33% to \$237 million in 1996, primarily due to the additional mortgage debt of approximately \$696 million incurred in connection with the 1996 full-service hotel additions and the issuance of \$350 million of notes issued by HMC Acquisition Properties, Inc., a wholly-owned subsidiary of Host, in December 1995, partially offset by the net impact of the 1995 redemptions of Host Marriott Hospitality, Inc. notes ("Hospitality Notes").

Loss from Continuing Operations. The loss from continuing operations for 1996 decreased \$49 million to \$13 million, as a result of the changes discussed above.

Net Loss. Host's net loss in 1996 was \$13 million, compared to a net loss of \$143 million in 1995, which included a \$61 million loss from discontinued operations and a \$20 million extraordinary loss primarily representing premiums paid on bond redemptions and the write-off of deferred financing fees and discounts on the debt. The basic and diluted loss per common share was \$.07 for 1996 and \$.90 for 1995.

# PRO FORMA RESULTS OF OPERATIONS

Because substantially all of the Company's Hotels will be leased to the Lessees following the REIT Conversion, the Company does not believe that the Company's historical results of operations will be comparable to the results of operations of the Company following the REIT Conversion. Accordingly, a comparison of the Company's pro forma results of operations for the First Three Quarters 1998 to the First Three Quarters 1997 and fiscal year 1997 to fiscal year 1996 have been included below. The following discussion and analysis should be read in conjunction with the Company's combined consolidated financial statements and the Company's unaudited pro forma financial statements and related notes thereto included elsewhere in this Proxy Statement/Prospectus. The following discussion and analysis has been prepared assuming the following two scenarios:

. All Partnerships participate in the Partnership Mergers and no Notes are issued ("100% Participation with No Notes Issued").

. All Partnerships participate in the Partnership Mergers and Notes are issued with respect to 100% of the OP Units allocable to each Partnership ("100% Participation with Notes Issued").

These presentations do not purport to represent what combination will result from the Merger and the REIT Conversion, but instead are designed to illustrate what the composition of the Company would have been like under the above scenarios. Furthermore, the unaudited pro forma financial statements do not purport to represent what the Company's results of operations or cash flows would actually have been if the Merger and the REIT Conversion had in fact occurred on such date or at the beginning of such period or to project the Company's results of operations or cash flows for any future date or period.

The following table presents the results of operations for the First Three Quarters 1998 and the First Three Quarters 1997 on a pro forma basis under the scenarios discussed above:

	WITH NO NOT	CIPATION ES ISSUED	100% PARTICIPATION WITH NOTES ISSUED			
	FIRST THREE	QUARTERS	FIRST THREE	QUARTERS		
	1998	1997	1998	1997		
		(IN MILL				
Rental revenues	542		542			
expense	(14) (30) (328) 24	(8) (29) (332) 20	(14)	(8) (29) (343) 20		
Loss before income taxes Benefit (provision) for income taxes	(215) 11	(234) 12	(226)	(244) 12		
Loss before extraordinary items	\$ (204)	\$ (222)		\$ (232)		

100% PARTICIPATION WITH NO NOTES ISSUED--FIRST THREE QUARTERS 1998 COMPARED TO FIRST THREE QUARTERS 1997 (PRO FORMA)

Revenues. Revenues primarily represent lease revenues, net gains (losses) on property transactions and equity in earnings (losses) of affiliates, including the Non-Controlled Subsidiaries. Revenues increased \$14 million, or 2.7%, to \$542 million for the First Three Quarters 1998 from \$528 million for the First Three Quarters 1997. EITF 98-9, "Accounting for Contingent Rents in Interim Financial Periods," requires a lessor to defer recognition of contingent rental income in interim periods until the specified target that triggers the contingent rental income is achieved. Based on the structure of the Company's leases, only minimum rent was recorded for the substantial majority of the Company's leases in the First Three Quarters 1998 and First Three Quarters 1997. On a pro forma basis, the Company would have received rental payments of \$860 million and \$777 million, respectively, resulting in deferred revenue of \$320 million and \$256 million, respectively, for the First Three Quarters 1998 and First Three Quarters 1997.

Hotel sales (gross hotel sales, including room sales, food and beverage sales, and other ancillary sales such as telephone sales) increased \$194 million, or 7.8%, to over \$2.6 billion in the First Three Quarters 1998, reflecting the REVPAR increases for the Company's hotels. Improved results for the Company's hotels were driven by strong increases in REVPAR of 8.1% to \$111.05 for the First Three Quarters 1998. Average room rates increased 8.4%, while average occupancy decreased slightly to 78.4%.

Operating Costs and Expenses. Operating costs and expenses principally consist of depreciation, property taxes, ground, rent, insurance and certain other costs. Operating costs and expenses decreased \$4 million to \$409 million in the First Three Quarters 1998. As a percentage of rental revenues, hotel operating costs and expenses decreased to 73% of rental revenues in the First Three Quarters 1998 from 77% of rental revenues in the First Three Quarters 1997 due to the increase in minimum rent under the Company's leases.

Operating Profit. As a result of the changes in rental revenues and operating costs and expenses discussed above, the Company's operating profit increased \$18 million, or 16%, to \$133 million for the First Three Quarters 1998. Hotel operating profit increased \$24 million, or 20%, to \$146 million, or 27% of rental revenues, for the First Three Quarters 1998 from \$122 million, or 23% of rental revenues, for the First Three Quarters 1997. The Company's hotels recorded significant improvements in comparable operating results, however, due to EITF 98-9, only minimum rent could be recorded. Specifically, hotels in New York City, Boston, Toronto and Atlanta reported significant improvements for the First Three Quarters 1998. Properties in Florida reported some temporary declines in operating results due to exceptionally poor weather in 1998.

Minority Interest. Minority interest expense increased \$6 million to \$14 million for the First Three Quarters 1998, primarily reflecting improved lodging results.

Corporate Expenses. Corporate expenses increased \$1 million to \$30 million for the First Three Quarters 1998 due to increased staffing levels and the impact of inflation.

Interest Expense. Interest expense decreased \$4 million to \$328 million in the First Three Quarters 1998, primarily due to the impact of principal amortization on the Company's mortgage debt.

Interest Income. Interest income increased \$4 million to \$24 million for the First Three Quarters 1998, primarily due to interest income from excess cash and marketable securities.

Income before Extraordinary Items. The loss before extraordinary items for the First Three Quarters 1998 was \$204 million, compared to \$222 million for the First Three Quarters 1997.

100% PARTICIPATION WITH NOTES ISSUED--FIRST THREE QUARTERS 1998 COMPARED TO FIRST THREE QUARTERS 1997 (PRO FORMA)

Revenues. Revenues increased \$14 million, or 2.7%, to \$542 million for the First Three Quarters 1998 from \$528 million for the First Three Quarters 1997. Based on the structure of the Company's leases, only minimum rent was recorded for the substantial majority of the Company's leases in the First Three Quarters 1998 and First Three Quarters 1997. On a pro forma basis, the Company would have received rental payments of \$860 million and \$777 million, respectively, resulting in deferred revenue of \$320 million and \$256 million, respectively, for the First Three Quarters 1998 and First Three Quarters 1997.

Hotel sales (gross hotel sales, including room sales, food and beverage sales, and other ancillary sales such as telephone sales) increased \$194 million, or 7.8%, to over \$2.6 billion in the First Three Quarters 1998, reflecting the REVPAR increases for the Company's hotels. Improved results for the Company's hotels were driven by strong increases in REVPAR of 8.1% to \$111.05 for the First Three Quarters 1998. Average room rates increased 8.4%, while average occupancy decreased slightly to 78.4%.

Operating Costs and Expenses. Operating costs and expenses decreased \$4 million to \$408 million in the First Three Quarters 1998. As a percentage of rental revenues, hotel operating costs and expenses decreased to 73% of revenues in the First Three Quarters 1998 from 76% of rental revenues in the First Three Quarters 1997 due to the increase in minimum rent under the Company's leases.

Operating Profit. As a result of the changes in rental revenues and operating costs and expenses discussed above, the Company's operating profit increased \$18 million, or 16%, to \$134 million for the First Three Quarters 1998. Hotel operating profit increased \$24 million, or 20%, to \$147 million, or 27% of rental revenues, for the First Three Quarters 1998 from \$123 million, or 24% of rental revenues, for the First Three Quarters 1997. Once again, the Company's hotels recorded significant improvements in comparable operating results;

however, due to EITF 98-9, only minimum rent could be recorded. Specifically, hotels in New York City, Boston, Toronto and Atlanta reported significant improvements for the First Three Quarters 1998. Properties in Florida reported some temporary declines in operating results due to exceptionally poor weather in 1998.

Minority Interest. Minority interest expense increased \$6 million to \$14 million for the First Three Quarters 1998, primarily reflecting improved lodging results.

Corporate Expenses. Corporate expenses increased \$1 million to \$30 million for the First Three Quarters 1998 due to increased staffing levels and the impact of inflation.

Interest Expense. Interest expense decreased \$3 million to \$340 million in the First Three Quarters 1998, primarily due to the impact of principal amortization on the Company's mortgage debt.

Interest Income. Interest income increased \$4 million to \$24 million for the First Three Quarters 1998, primarily due to interest income from excess cash and marketable securities.

Loss before Extraordinary Items. The loss before extraordinary items for the First Three Quarters 1998 was \$215 million compared to \$232 million for the First Three Ouarters 1997.

100% PARTICIPATION WITH NO NOTES ISSUED--1997 COMPARED TO 1996 (PRO FORMA)

The following table presents the results of operations for the Company for 1997 and 1996 on a pro forma basis under the two pro forma scenarios:

	NO NOTES	ISSUED	100% PARTICI	SSUED
	FISCAL	YEAR	FISCAL	YEAR
	1997	1996	1997	1996
		(IN MILL		
Rental revenues Total revenues Operating costs and	. ,	\$ 1,050 1,040	\$ 1,135 1,131	
expenses	611	600	609	599
interest expense Minority interest Corporate expenses Interest expense Interest income	(10) (44) (468)	(9) (39) (481) 29	(10) (44) (485) 29	(9) (39) (498) 29
Income (loss) before income taxes	(1)	(60)		(76) 4
Income (loss) before extraordinary items		\$ (57)		\$ (72)

Revenues. Revenues increased \$91 million, or 8.8%, to \$1,131 million for 1997. The Company's revenue and operating profit were principally impacted by improved lodging results for its hotel properties, which led to a substantial increase in rental revenues. The 1997 results also included 52 weeks versus 53 weeks in 1996.

Hotel sales increased \$263 million, or 7.9%, to nearly \$3.6 billion in 1997, reflecting the increases in REVPAR. Improved results for the Company's full-service hotels were driven by strong increases in REVPAR of 9.6% to \$103.09 in 1997. Average room rates increased nearly 9%, while average occupancy increased slightly to 77.7%.

Operating Costs and Expenses. Operating costs and expenses increased \$11 million to \$611 million for 1997. As a percentage of rental revenues, hotel operating costs and expenses decreased to 53% of rental revenues for 1997, from 57% of rental revenues for 1996, reflecting the impact of increased 1997 rental revenues on relatively fixed operating costs and expenses.

Operating Profit. As a result of the changes in rental revenues and operating costs and expenses discussed above, the Company's operating profit increased \$80 million, or 18%, to \$520 million in 1997. Hotel operating profit increased \$73 million, or 16%, to \$535 million, or 47% of rental revenues, for 1997 compared to \$462 million, or 44% of rental revenues, for 1996. In nearly all markets, the Company's hotels recorded improvements in comparable operating results. In particular, the Company's hotels in the Northeast, Mid-Atlantic and Pacific coast regions benefited from the upscale and luxury fullservice room supply and demand imbalance. Hotels in New York City, Philadelphia, San Francisco/Silicon Valley and in Southern California performed particularly well. In 1998, the Company expects results to be strong in these markets and other gateway cities in which the Company owns hotels. In 1997, the Company's suburban Atlanta properties (three properties totaling 1,022 rooms) generally reported decreased results due to higher activity in 1996 related to the Summer Olympics and the impact of the additional supply added to the suburban areas. However, the majority of the Company's hotel rooms in Atlanta are in the core business districts in downtown and Buckhead where they realized strong year-over-year results and were only marginally impacted by the additional supply.

Minority Interest. Minority interest expense increased 1 million to 10 million in 1997.

Corporate Expenses. Corporate expenses increased \$5 million to \$44 million in 1997 due to increased staffing levels and the impact of inflation.

Interest Expense. Interest expense decreased \$13\$ million to \$468\$ million in 1997, primarily due to the impact of principal amortization on the Company's mortgage debt.

Interest Income. Interest income remained unchanged at \$29 million for 1997, reflecting the interest income earned on the loan to the Non-Controlled Subsidiary for its acquisition of furniture and equipment, the working capital loan to Crestline, and a mortgage note on one property.

Income (Loss) Before Extraordinary Items. Income before extraordinary items for 1997 was \$26 million, compared to a \$57 million loss before extraordinary items for 1996 as a result of the items discussed above.

100% PARTICIPATION WITH NOTES ISSUED--1997 COMPARED TO 1996 (PRO FORMA)

Revenues. Revenues increased \$91 million, or 8.8%, to \$1,131 million for 1997. The Company's revenue and operating profit were principally impacted by improved lodging results for the Company's hotel properties, which led to a substantial increase in rental revenues. The 1997 results also included 52 weeks versus 53 weeks in 1996.

Hotel sales increased \$263 million, or 7.9%, to nearly \$3.6 billion in 1997, reflecting increases in REVPAR. Improved results for the Company's full-service hotels were driven by strong increases in REVPAR of 9.6% to \$103.09 in 1997. Average room rates increased nearly 9%, while average occupancy increased slightly to 77.7%.

Operating Costs and Expenses. Operating costs and expenses increased \$10 million to \$609 million for 1997. As a percentage of rental revenues, hotel operating costs and expenses decreased to 54% of rental revenues for 1997, from 57% of rental revenues for 1996, reflecting the impact of increased 1997 rental revenues on relatively fixed operating costs and expenses.

Operating Profit. As a result of the changes in rental revenues and operating costs and expenses discussed above, the Company's operating profit increased \$81 million, or 18%, to \$522 million in 1997. Hotel operating profit increased \$74 million, or 16%, to \$537 million, or 47% of rental revenues, for 1997 compared to \$463 million, or 44% of rental revenues, for 1996. In nearly all markets, the Company's hotels recorded improvements in comparable operating results. In particular, the Company's hotels in the Northeast, Mid-Atlantic and Pacific coast regions benefited from the upscale and luxury full-service room supply and demand imbalance. Hotels in New York City, Philadelphia, San Francisco/Silicon Valley and in Southern California performed particularly well. In 1998, the Company expects results to be strong in these markets and other gateway cities in which the

Company owns hotels. In 1997, the Company's suburban Atlanta properties (three properties totaling 1,022 rooms) generally reported decreased results due to higher activity in 1996 related to the Summer Olympics and the impact of the additional supply added to the suburban areas. However, the majority of the Company's hotel rooms in Atlanta are in the core business districts in downtown and Buckhead where they realized strong year-over-year results and were only marginally impacted by the additional supply.

Minority Interest. Minority interest increased \$1 million to \$10 million for 1997.

Corporate Expenses. Corporate expenses increased \$5 million to \$44 million in 1997 due to increased staffing levels and the impact of inflation.

Interest Expense. Interest expense decreased \$13\$ million to \$485\$ million in 1997, reflecting the impact of principal amortization on the Company's mortgage debt.

Interest Income. Interest income remained unchanged at \$29 million for 1997. Interest income includes the interest income earned on the loan to the Non-Controlled Subsidiary for its acquisition of furniture and equipment, the working capital loan to Crestline, and a mortgage note on one property.

Income (Loss) Before Extraordinary Items. Income before extraordinary items for 1997 was \$11 million, compared to a \$72 million loss before extraordinary items for 1996 as a result of the items discussed above.

## PRO FORMA RESULTS IF NO PARTNERSHIPS PARTICIPATE IN THE REIT CONVERSION

There is no requirement that any of the Partnerships participate in the REIT Conversion in order for the REIT Conversion to be consummated. Accordingly, the following discussion has been included assuming that no Partnerships participate in the REIT Conversion and compares the "No Partnership Participation" scenario to the 100% Participation with No Notes Issued scenario, both of which assume that the Blackstone Acquisition occurs.

Revenues for the First Three Quarters 1998 and fiscal year 1997 would have been higher by \$1 million and lower by \$69 million, respectively, as a result of the Company not reporting rental revenues related to the 24 hotel properties owned by the Partnerships, partially offset by the Company reporting gross hotel sales for four of the Partnerships (Desert Springs, MHP, MHP2 and Hanover) that it will continue to consolidate for financial reporting purposes since these partnerships are controlled by Host. Operating profit would have been higher by \$49 million for the First Three Quarters due to lower deferred rent under the No Partnership Participation scenario and lower by \$42 million for fiscal year 1997 due to the operating profit generated by the Partnerships on a full year basis including the substantial impact of no rental revenues from the 24 properties, partially offset by a significant decrease in depreciation expense. Interest expense would have been lower by \$24 million and \$29 million, respectively, due to the reduction in overall debt levels by \$426 million. The loss before extraordinary items for the First Three Quarters 1998 would have been lower by \$65 million principally due to the changes in operating profit and interest expense discussed above and the impact on deferred rent. Income before extraordinary items for fiscal year 1997 would have been lower by \$14 million, reflecting the net income generated by the Partnerships on a full year basis.

# PRO FORMA RESULTS IF HOST REIT'S ELECTION IS DELAYED UNTIL JANUARY 1, 2000

Host intends to use its best efforts to cause the REIT Conversion to be completed as soon as possible, but there is no assurance that it will be completed during 1998 in time for Host REIT to elect REIT status effective January 1, 1999, or at all. If the REIT Conversion does not occur in 1998, but the conditions to the REIT Conversion are otherwise satisfied, the effectiveness of Host REIT's election could be delayed to January 1, 2000, which could cause the Blackstone Acquisition (which is conditioned, among other things, on consummation of the REIT Conversion by March 31, 1999 and Host REIT qualifying as a REIT for 1999) not to be consummated. Accordingly, the following discussion compares the 100% Participation with No Notes Issued scenario with the "REIT 2000" scenario that assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur and Host does not become a REIT until January 1, 2000 and no Notes are issued.

Revenues for the First Three Quarters 1998 and fiscal year 1997 would have been lower by \$72 million and \$137 million, respectively, due to the decrease in rental revenues as a result of the Company not completing the Blackstone Acquisition. Operating profit for the First Three Quarters 1998 and fiscal year 1997 would have been lower by \$14 million and \$50 million, respectively, due to the impact on rental revenues, of not completing the Blackstone Acquisition, net of the related operating costs and expenses (principally depreciation). Interest expense would have been lower by \$36 million and \$48 million, respectively, due to the reduction in overall debt levels of \$600 million related to the Blackstone Acquisition. The loss before extraordinary items for the First Three Quarters 1998 would have been \$26 million less and income before extraordinary items for fiscal year 1997 \$4 million higher principally due to the changes in operating profit and interest expense discussed above. Note that the First Three Quarters 1998 is significantly impacted by \$43 million of deferred revenue related to the Blackstone properties.

## LIQUIDITY AND CAPITAL RESOURCES

Host funds its capital requirements with a combination of operating cash flow, debt and equity financing and proceeds from sales of selected properties and other assets. Host utilizes these sources of capital to acquire new properties, fund capital additions and improvements and make principal payments on debt.

Capital Transactions. Host has recently substantially changed its debt financing through the following series of transactions which were intended to facilitate the consummation of the REIT Conversion.

On April 20, 1998, Host and certain of its subsidiaries filed a shelf registration statement on Form S-3 (the "Shelf Registration") with the Commission for \$2.5 billion in securities, which may include debt, equity or any combination thereof. Host anticipates that any net proceeds from the sale of offered securities will be used for refinancing of Host's indebtedness, for acquisitions and general corporate purposes.

On August 5, 1998, HMH Properties, an indirect wholly-owned subsidiary of Host, which currently owns 72 of Host's hotels, purchased substantially all of its (i) \$600 million in 9 1/2% senior notes due 2005, (ii) \$350 million in 9% senior notes due 2007 and (iii) \$600 million in 8 7/8% senior notes due 2007 (collectively, the "Old Senior Notes"). Concurrently with each offer to purchase, HMH Properties solicited consents (the "1998 Consent Solicitations") from registered holders of the Old Senior Notes to certain amendments to eliminate or modify substantially all of the restrictive covenants and certain other provisions contained in the indentures pursuant to which the Old Senior Notes were issued. HMH Properties simultaneously utilized the Shelf Registration to issue an aggregate of \$1.7 billion in New Senior Notes. The New Senior Notes were issued in two series, \$500 million of 7 7/8% Series A notes due in 2005 and \$1.2 billion of 7 7/8% Series B notes due in 2008. The 1998 Consent Solicitations facilitated the merger of HMC Capital Resources Holdings Corporation ("Capital Resources"), a wholly-owned subsidiary of Host, with and into HMH Properties. Capital Resources, the owner of eight of Host's hotel properties, was the obligor under the \$500 million revolving credit facility (the "Old Credit Facility"). The Operating Partnership will assume the New Senior Notes in connection with the REIT Conversion and the guarantee of Host is expected to terminate on the Effective Date.

In conjunction with the issuance of the New Senior Notes, HMH Properties entered into the \$1.25 billion New Credit Facility with a group of commercial banks. The New Credit Facility has an initial three-year term with two one-year extension options. Borrowings under the New Credit Facility generally bear interest at the Eurodollar rate plus 1.75% (7.5% at September 11, 1998). The interest rate and commitment fee (0.35% at September 11, 1998) on the unused portion of the New Credit Facility fluctuate based on certain financial ratios. The New Senior Notes and the New Credit Facility are guaranteed by Host and its wholly-owned subsidiary, Host Marriott Hospitality, Inc., and certain subsidiaries of HMH Properties and are secured by pledges of equity interests in certain subsidiaries of HMH Properties. The New Credit Facility will be assumed by the Operating Partnership in connection with the REIT Conversion and the guarantee of Host is expected to terminate on the Effective Date. As of September 11, 1998, \$350 million was outstanding under the New Credit Facility.

The New Credit Facility and the indenture under which the New Senior Notes were issued contain covenants restricting the ability of HMH Properties and certain of its subsidiaries to incur indebtedness, grant liens on their assets, acquire or sell assets or make investments in other entities, and make distributions to equityholders of HMH Properties, Host, and (following the REIT Conversion) the Operating Partnership and Host REIT. The New Credit Facility and the New Senior Notes also contain certain financial covenants relating to, among other things, maintaining certain levels of tangible net worth and certain ratios of EBITDA to interest and fixed charges, total debt to EBITDA, unencumbered assets to unsecured debt, and secured debt to total

The New Credit Facility replaces Host's Old Credit Facility. The net proceeds from the offering and borrowings under the New Credit Facility were used by Host to purchase substantially all of the Existing Senior Notes, to repay amounts outstanding under the Existing Credit Facility and to make bond premium and consent payments totaling \$175 million. These costs, along with the write-off of deferred financing fees of approximately \$52 million related to the Old Senior Notes and the Old Credit Facility, were recorded as a pretax extraordinary loss on the extinguishment of debt in the First Three Quarters 1998.

Upon consummation of the REIT Conversion, the Company will be required to offer to repurchase the \$127.4 million of 10 3/4% Senior Notes due 2008 issued by Courtyard by Marriott II Limited Partnership ("CBM2"). A subsidiary of the Company is the general partner of CBM2. Consummation of the REIT Conversion will constitute a "change of control" under the indenture relating to the notes. The Company must purchase any notes that are tendered for cash at a price equal to 101% of the principal amount plus accrued interest. The Company expects to borrow under the New Credit Facility to obtain any funds required to repurchase any such notes that are tendered. The Company will continue to hold any notes that are repurchased.

Subject to market conditions, the Company is currently considering conducting a senior notes offering in the amount of approximately \$300 to \$500 million, the proceeds of which would be used to pay down other debt or pay other expenses of the REIT Conversion that would have been borrowed.

In June 1997, HMC Capital Resources Corporation ("Capital Resources"), a wholly-owned subsidiary of Host, entered into the Old Credit Facility with a group of commercial banks under which it may borrow up to \$500 million for certain permitted uses. As a result of this transaction, Host terminated its line of credit with Marriott International.

In July 1997, HMH Properties and HMC Acquisition Properties, Inc. ("Acquisitions"), indirect, wholly-owned subsidiaries of Host, completed consent solicitations with holders of their senior notes (the "1997 Consent Solicitations") to amend certain provisions of their senior notes indentures. The 1997 Consent Solicitations facilitated the merger of Acquisitions with and into HMH Properties (the "HMH Properties Merger"). The amendments to the indentures also increased the ability of HMH Properties to acquire, through certain subsidiaries, additional properties subject to non-recourse indebtedness and controlling interests in corporations, partnerships and other entities holding attractive properties and increased the threshold for distributions to affiliates to the excess of HMH Properties' earnings before interest expense, income taxes, depreciation and amortization and other non-cash items subsequent to the 1997 Consent Solicitations over 220% of HMH Properties' interest expense. HMH Properties paid dividends to Host of \$54 million, \$29 million and \$36 million in 1997, 1996 and 1995, respectively, as permitted under the indentures.

Concurrent with the 1997 Consent Solicitations and the HMH Properties Merger, HMH Properties issued an aggregate of \$600 million of 8 7/8% senior notes at par with a maturity of July 2007. HMH Properties received net proceeds of approximately \$570 million, net of the costs of the 1997 Consent Solicitations and the offering.

In addition to the capital resources provided by its new debt financings, Host Marriott Financial Trust (the "Issuer"), a wholly-owned subsidiary trust of Host, has outstanding 11 million shares of 6 3/4% Convertible Quarterly Income Preferred Securities (the "Convertible Preferred Securities"), with a liquidation preference of \$50 per share (for a total liquidation amount of \$550 million) issued in December 1996. The Convertible Preferred Securities represent an undivided beneficial interest in the assets of the Issuer and, pursuant to various agreements entered into in connection with the transaction, are fully, irrevocably and unconditionally guaranteed

by Host. Proceeds from the issuance of the Convertible Preferred Securities were invested in 6 3/4% Convertible Subordinated Debentures (the "Debentures") due December 2, 2026 issued by Host. The Issuer exists solely to issue the Convertible Preferred Securities and its own common securities (the "Common Securities") and invest the proceeds therefrom in the Debentures, which are its sole assets. Each of the Convertible Preferred Securities is convertible at the option of the holder into shares of Host common stock at the rate of 2.6876 shares per Convertible Preferred Security (equivalent to a conversion price of \$18.604 per share of Host common stock). The Debentures are convertible at the option of the holders into shares of Company common stock at a conversion rate of 2.6876 shares for each \$50 in principal amount of Debentures. The conversion rate is subject to adjustments in certain events, including (i) payment of dividends (and other distributions) on Host Common Stock by Host in shares of Host Common Stock; (ii) distributions to all holders of Host Common Stock of rights or warrants entitling such holders (for a period not to exceed 45 days) to subscribe for or purchase Host Common

Stock at an exercise price less than the market price of Host Common Stock; (iii) subdivisions and combinations of Host Common Stock; (iv) payment of dividends (and other distributions) on Host Common Stock consisting of indebtedness of Host, capital stock or other securities, assets or cash (other than certain cash dividends at an annualized rate of up to 12.5% of the market price of Host Common Stock); (v) payments for Host Common Stock by Host or any of its subsidiaries in respect of a tender or exchange offer (other than an odd-lot offer) at a price per share in excess of 110% of the market price of Host Common Stock; (vi) consummation by Host of certain mergers, a consolidation, a sale of all or substantially all of its assets, a recapitalization or certain reclassifications of Host Common Stock. The distribution of the capital stock of Crestline to all holders of Host REIT Common Stock would, and certain other elements of the REIT Conversion (such as other distributions of Host's accumulated earnings and profits) may, result in an adjustment to the conversion price of the Debentures. The Issuer will only convert Debentures pursuant to a notice of conversion by a holder of Convertible Preferred Securities. During 1997 and 1996, no shares were converted into common stock. Holders of the Convertible Preferred Securities are entitled to receive preferential cumulative cash distributions at an annual rate of 6 3/4% accruing from the original issue date, commencing March 1, 1997, and payable quarterly in arrears thereafter. The distribution rate and the distribution and other payment dates for the Convertible Preferred Securities will correspond to the interest rate and interest and other payment dates on the Debentures. Host may defer interest payments on the Debentures for a period not to exceed 20 consecutive quarters. If interest payments on the Debentures are deferred, so too are payments on the Convertible Preferred Securities. Under this circumstance, Host will not be permitted to declare or pay any cash distributions with respect to its capital stock or debt securities that rank pari passu with or junior to the Debentures. Subject to certain restrictions, the Convertible Preferred Securities are redeemable at the Issuer's option upon any redemption by Host of the Debentures after December 2, 1999. Upon repayment at maturity or as a result of the acceleration of the Debentures upon the occurrence of a default, the Debentures shall be subject to mandatory redemption, from which the proceeds will be applied to redeem Convertible Preferred Securities and Common Securities, together with accrued and unpaid distributions.

In connection with consummation of the REIT Conversion, the Operating Partnership will assume primary liability for repayment of the convertible debentures of Host underlying the Convertible Preferred Securities. Upon conversion by a Convertible Preferred Securities holder, Host REIT will issue shares of Host REIT Common Stock, which will be delivered to such holder. Upon the issuance of such shares by Host REIT, the Operating Partnership will issue to Host REIT a number of OP Units equal to the number of shares of Host REIT Common Stock issued in exchange for the Debentures.

In March 1996, Host completed the issuance of 31.6 million shares of common stock for net proceeds of nearly \$400 million.

In December 1995, Acquisitions issued \$350 million of 9% senior notes (the "Acquisitions Notes"). The Acquisitions Notes were issued at par and have a final maturity of December 2007. The net proceeds totaled \$340 million and were utilized to repay in full the outstanding borrowings of \$210 million under Acquisitions' \$230 million revolving credit facility (the "Revolver"), which was then terminated to acquire three full-service properties and to finance future acquisitions of full-service hotel properties with the remaining proceeds.

In May 1995, two wholly-owned subsidiaries of Host Marriott Hospitality, Inc. ("Hospitality"), a wholly-owned subsidiary of Host, issued an aggregate of \$1 billion of 9.5% senior secured notes in two concurrent offerings. HMH Properties, and Host Marriott Travel Plazas, Inc. ("HMTP"), the operator/manager of HM Services' food, beverage and merchandise concessions business, issued \$600 million and \$400 million, respectively, of senior notes. The net proceeds of approximately \$971 million were used to defease, and subsequently redeem, all of Hospitality's remaining bonds and to repay borrowings under the line of credit with Marriott International. The HMTP senior notes were included in the HM Services' special dividend.

During 1995, Host replaced its line of credit with a line of credit from Marriott International pursuant to which Host had the right to borrow up to \$225 million. The line of credit with Marriott International was terminated as a result of the Capital Resources transaction discussed above.

Asset Dispositions. Host historically has sold, and may from time to time in the future consider opportunities to sell, certain of its real estate properties at attractive prices when the proceeds could be redeployed into investments with more favorable returns. During the second quarter of 1998, Host sold the 662-room New York Marriott East Side for proceeds of \$191 million and recorded a pre-tax gain of approximately \$40 million and the Napa Valley Marriott for proceeds of \$21 million and recorded a pre-tax gain of approximately \$10 million. During 1997, Host sold the 255-room Sheraton Elk Grove Suites for proceeds of approximately \$16 million. Host also sold 90% of its 174-acre parcel of undeveloped land in Germantown, Maryland, for approximately \$11 million, which approximated its carrying value. During the first and second quarters of 1996, 16 of Host's Courtyard properties and 18 of Host's Residence Inn properties were sold (subject to a leaseback) to Hospitality Properties Trust for approximately \$314 million and Host will receive approximately \$35 million upon expiration of the leases. A gain on the transactions of approximately \$46 million was deferred and is being amortized over the initial term of the leases. During the first and third quarters of 1995, 37 of Host's Courtyard properties were sold to and leased back from Hospitality Properties Trust for approximately \$330 million. Host received net proceeds from the two 1995 transactions of approximately \$297 million and will receive approximately \$33 million upon expiration of the leases. A deferred gain of \$14 million on the sale/leaseback transactions is being amortized over the initial term of the leases. In 1995, Host also sold its four remaining Fairfield Inns for net cash proceeds of approximately \$6 million, which approximated their carrying value.

In cases where Host has made a decision to dispose of particular properties, Host assesses impairment of each individual property to be sold on the basis of expected sales price less estimated costs of disposal. Otherwise, Host assesses impairment of its real estate properties based on whether it is probable that undiscounted future cash flows from such properties will be less than their net book value. If a property is impaired, its basis is adjusted to its fair market value. In the second quarter of 1995, Host made a determination that its owned Courtyard and Residence Inn properties were held for sale and recorded a \$10 million charge to write down the carrying value of five individual Courtyard and Residence Inn properties to their estimated net sales values.

Capital Acquisitions, Additions and Improvements. Host seeks to grow primarily through opportunistic acquisitions of full-service hotels. Host believes that the upscale and luxury full-service hotel segments of the market offer opportunities to acquire assets at attractive multiples of cash flow and at discounts to replacement value, including under-performing hotels which can be improved by conversion to the Marriott or Ritz-Carlton brands. During 1997, Host acquired eight full-service hotels (3,600 rooms) and controlling interests in nine additional full-service hotels (5,024 rooms) for an aggregate purchase price of approximately \$766 million (including the assumption of approximately \$418 million of debt). Host also completed the acquisition of the 504-room New York Marriott Financial Center, after acquiring the mortgage on the hotel for \$101 million in late 1996. During 1996, Host acquired six full-service hotels (1,964 rooms) for an aggregate purchase price of \$189 million and controlling interests in 17 additional full-service properties (8,917 rooms) for an aggregate purchase price of approximately \$1.1 billion (including the assumption of \$696 million of debt). During 1995, Host acquired nine hotels totaling approximately 3,900 rooms in separate transactions for approximately \$390 million (\$141 million of which was financed through first mortgage financing on four of the hotels).

In the first quarter of 1998, Host acquired a controlling interest in the partnership that owns the 1,671-room Atlanta Marriott Marquis Hotel for \$239 million, including the assumption of \$164 million of mortgage debt. Host also acquired a controlling interest in the partnership that owns the 359-room Albany Marriott, the 350-room San Diego Marriott Mission Valley and the 320room Minneapolis Marriott Southwest for approximately \$50 million. In the second quarter of 1998, Host acquired the 289-room Park Ridge Marriott for \$24 million and acquired the 281-room Ritz-Carlton, Phoenix for \$75 million. Host is continually engaged in discussions with respect to other potential acquisition properties. In addition, Host acquired the 397-room Ritz-Carlton, Tysons Corner, Virginia and the 487-room Torrance Marriott near Los Angeles, California. In the third quarter of 1998, Host acquired the 308-room Ritz-Carlton, Dearborn for approximately \$65 million, the 336-room Ritz-Carlton, San Francisco for approximately \$161 million and the 404-room Memphis Marriott (which was converted to the Marriott brand upon acquisition) for approximately \$16 million.

On April 17, 1998, Host announced that it had reached a definitive agreement with the Blackstone Entities to acquire interests in twelve world-class luxury hotels and certain other assets. If the Blackstone Acquisition is consummated, the Operating Partnership expects to pay approximately \$862 million in cash and assumed debt and to issue approximately 43.7 million OP Units (based upon a negotiated value of \$20.00 per OP Unit) and other consideration. The Blackstone portfolio consists of two Ritz-Carltons, two Four Seasons, one Grand Hyatt, three Hyatt Regencies and four Swissotel properties and the mortgage on a third Four Seasons. These hotels are located in major urban and convention/resort markets with significant barriers to new competition. The Blackstone Acquisition is expected to close as part of, and is contingent upon, the REIT Conversion. At that time, the Blackstone hotels and other assets will be acquired by the Company. The hotels will be leased to Lessees and will be managed on behalf of the Lessees under their existing management contracts.

Under the terms of its hotel management agreements, Host is generally required to spend approximately 5% of gross hotel sales to cover the capital needs of the properties, including major guest room and common area renovations which occur every five to six years.

Host completed the construction of the 1,200-room Philadelphia Marriott, which opened on January 27, 1995. The construction costs of this hotel were funded 60% through a loan from Marriott International which was repaid in the fourth quarter of 1996. In March 1997, Host obtained a \$90 million mortgage which bears interest at a fixed rate of 8.49% and matures in 2009. Construction of a second hotel in Philadelphia, the 419-room Philadelphia Airport Marriott (the "Airport Hotel"), was completed and opened on November 1, 1995. The Airport Hotel was financed principally with \$40 million of proceeds from an industrial development bond financing. Host also completed construction of a 300-room Residence Inn in Arlington, Virginia, which opened in March 1996. Capital expenditures for these three hotels totaled \$11 million in 1996 and \$64 million in 1995.

In November 1997, Host announced that it had committed to develop and construct the 717-room Tampa Convention Center Marriott for a cost estimated at approximately \$88 million, net of an approximate \$16 million subsidy provided by the City of Tampa.

Host may also expand certain existing hotel properties where strong performance and market demand exists. Expansions to existing properties creates a lower risk to Host as the success of the market is generally known and development time is significantly shorter than new construction. Host recently committed to add approximately 500 rooms and an additional 15,000 square feet of meeting space to the 1,503-room Marriott's Orlando World Center.

In 1997, Host acquired the outstanding common stock of the Forum Group from Marriott Senior Living Services, Inc. ("MSLS"), a subsidiary of Marriott International. Host purchased the Forum Group portfolio of 29 senior living communities for approximately \$460 million, including approximately \$270 million in debt. The properties will continue to be operated by MSLS. In addition, Host plans to add approximately 875 units to these communities for approximately \$88 million through an expansion plan which will be completed in 1999. In 1997, approximately \$56 million (549 units) of the expansion plan had been completed (including \$33 million of debt financing provided by Marriott International). Host also acquired 49% of the remaining 50% interest in the venture which owned the 418-unit Leisure Park senior living community from Marriott International for approximately \$23 million, including approximately \$15 million of debt.

During the first quarter of 1998, Host acquired the Gables at Winchester in suburban Boston, a 124-unit senior living community, for \$21 million and entered into conditional purchase agreements to acquire two Marriott Brighton Gardens assisted living communities from the Summit Companies of Denver, Colorado. After the anticipated completion of construction in the first quarter of 1999, Host may acquire these two 160-unit properties located in Denver and Colorado Springs, Colorado, for \$35 million, if they achieve certain operating performance criteria. All three of these communities will be operated by MSLS under long-term operating agreements.

Under the terms of its senior living communities' management agreements, Host is generally required to spend an amount of gross revenues to cover certain routine repairs and maintenance and replacements and

renewals to the communities' property and improvements. The amount Host is required to spend will be 2.65% through fiscal year 2002, 2.85% for fiscal years 2003 through 2007, and 3.5% thereafter. Host anticipates spending approximately \$6 million in 1998.

As part of the Initial E&P Distribution, Host REIT and the Operating Partnership will distribute shares of Crestline common stock (which will own the Forum Group portfolio and other senior living and assisted living communities described above) to Host REIT's stockholders and the Blackstone Entities.

Debt Payments. At January 2, 1998, Host and its subsidiaries had \$1,585 million of senior notes, approximately \$2.0 billion of non-recourse mortgage debt secured by real estate assets and approximately \$219 million of unsecured and other debt.

Scheduled maturities over the next five years were \$942 million as of January 2, 1998, a significant portion of which represents the maturity of the mortgage on the New York Marriott Marquis of approximately \$270 million in December 1998. Management anticipates that the mortgage will be refinanced by the end of 1998 on comparable terms. Host's interest coverage, defined as EBITDA divided by cash interest expense, improved to nearly 2.5 times in 1997 from 2.0 times in 1996.

At January 2, 1998, Host was party to an interest rate exchange agreement with a financial institution (the contracting party) with an aggregate notional amount of \$100 million. Under this agreement, Host collects interest based on specified floating interest rates of one month LIBOR (rate of 6% at January 2, 1998) and pays interest at fixed rates (rate of 7.99% at January 2, 1998). This agreement expires in 1998, in conjunction with the maturity of the mortgage on the New York Marriott Marquis. Also in 1997, Host was party to two additional interest rate swap agreements with an aggregate notional amount of \$400 million. These agreements expired in May 1997. Host realized a net reduction of interest expense of \$1 million in 1997, \$6 million in 1996 and \$5 million in 1995 related to interest rate exchange agreements. Host monitors the creditworthiness of its contracting parties by evaluating credit exposure and referring to the ratings of widely accepted credit rating services. The Standard and Poors' long-term debt ratings for the contracting party is A- for its sole outstanding interest rate exchange agreement. Host is exposed to credit loss equal to the accrued interest receivable on the notional amount of interest rate exchange agreements outstanding (\$0 at January 2, 1998) in the event of non-performance by the contracting party to the interest rate swap agreement; however, Host does not anticipate non-performance by the contracting party.

Cash Flows. Host's cash flow from continuing operations in 1997, 1996 and 1995 totaled \$464 million, \$205 million and \$110 million, respectively. Cash flow from operations in the First Three Quarters 1998 and First Three Quarters 1997 totaled \$278 million and \$313 million, respectively.

Host's cash used in investing activities from continuing operations in 1997, 1996 and 1995 totaled \$1,046 million, \$504 million and \$156 million, respectively. Cash used in investing activities was \$273 million and \$497 million for the First Three Quarters 1998 and the First Three Quarters 1997, respectively. Cash from investing activities primarily consists of net proceeds from the sales of certain assets, offset by the acquisition of hotels and other capital expenditures previously discussed, as well as the purchases and sales of short-term marketable securities. Cash used in investing activities was significantly impacted by the purchase of \$354 million of short-term marketable securities in 1997 and the net sale of \$317 million of short-term marketable securities in the First Three Quarters 1998.

Host's cash from financing activities from continuing operations was \$389 million for 1997, \$806 million for 1996 and \$204 million for 1995. Cash from financing activities was \$23 million and \$391 million, respectively, for the First Three Quarters 1998 and First Three Quarters 1997. Host's cash from financing activities primarily consists of the proceeds from debt and equity offerings, the issuance of the Convertible Preferred Securities, mortgage financing on certain acquired hotels and borrowings under the Line of Credit, offset by redemptions and payments on senior notes, prepayments on certain hotel mortgages and other scheduled principal payments.

The ratio of earnings to fixed charges was 1.7 to 1.0, 1.4 to 1.0, 1.3 to 1.0, 1.0 to 1.0 and .7 to 1.0 for the First Three Quarters 1998, the First Three Quarters 1997, 1996 and 1995, respectively. The deficiency of earnings to fixed charges of \$70 million for 1995 is largely the result of depreciation and amortization of \$122 million. In addition, the deficiency for 1995 was impacted by the \$60 million pre-tax charge to write down the carrying value of one undeveloped land parcel to its estimated sales value.

Comparative FFO. Host believes that Comparative Funds From Operations ("Comparative FFO," which represents Funds From Operations, as defined by NAREIT, plus deferred tax expense) is a meaningful disclosure that will help the investment community to better understand the financial performance of Host, including enabling its stockholders and analysts to more easily compare Host's performance to REITs. FFO is defined by NAREIT as net income computed in accordance with GAAP, excluding gains or losses from debt restructurings and sales of properties, plus real estate related depreciation and amortization, and after adjustments for unconsolidated partnerships and joint ventures. FFO should not be considered as an alternative to net income, operating profit, cash flows from operations or any other operating or liquidity performance measure prescribed by GAAP. FFO is also not an indicator of funds available to fund Host's cash needs, including its ability to make distributions. Host's method of calculating FFO may be different from methods used by other REITs and, accordingly, is not comparable to such other REITs. Comparative FFO increased \$80 million, or 40%, to \$282 million in the First Three Quarters 1998. Comparative FFO increased \$131 million, or 80%, to \$295 million in 1997. The following is a reconciliation of Host's income (loss) before extraordinary items to Comparative FFO (in millions):

	FIRST THREE QUARTERS			F	ISCAL	YEAR		
	19	98	19	97 	19	997	19	996
Income before extraordinary items Real estate related depreciation and	\$	100	\$	38	\$	47	\$	(13)
amortization		184		158		240		168
Other real estate activities		(53)		1		6		7
Partnership adjustments		(9)		(6)		(13)		1
REIT Conversion expenses		14						
Deferred taxes		46		11		15		1
Comparative FFO	\$	282	\$	202	\$	295	\$	164
	===	====	===	====	=:	====	==	===

Host considers Comparative FFO to be an indicative measure of Host's operating performance due to the significance of Host's long-lived assets and because such data is considered useful by the investment community to better understand Host's results, and can be used to measure Host's ability to service debt, fund capital expenditures and expand its business; however, such information should not be considered as an alternative to net income, operating profit, cash from operations or any other operating or liquidity performance measure prescribed by generally accepted accounting principles. Cash expenditures for various long-term assets and income taxes have been, and will be, incurred which are not reflected in the Comparative FFO presentation.

On October 13, 1998, Host issued a press release reporting Comparative FFO  $\,$ and other financial information for the 1998 third quarter. The press release contained a statement to the effect that Host remained comfortable with industry analysts' consensus Comparative FFO per share estimate for 1998 (which implied diluted Comparative FFO of approximately \$425 million for 1998) and further stated that for 1999, given the broad range of forecasts of gross domestic product for such year, that Host believed that the range of Comparative FFO per share estimates for 1999 (which implied a diluted Comparative FFO range for 1999 of approximately \$700 million to \$730 million) was appropriate at such time. In making such statement, Host intended to indicate its comfort level with industry analysts' Comparative FFO estimates for 1999. Host's comments on its comfort with analysts' Comparative FFO estimates for 1999 (which reflect estimated activities of Host REIT and Crestline together on a combined basis) were based on a number of factors, including the following principal assumptions: (i) the REIT Conversion would be effective as of January 1, 1999 on terms expected at such time; (ii) REVPAR would increase by approximately 4% in 1999 compared to 1998; (iii) \$375

million of hotel properties would be acquired during 1999; (iv) construction of one new hotel property and expansion of three properties would be continued, which projects requiring 1999 outlays of funds of \$160 million would reach completion in 1999 and would not generate earnings in 1999; (v) approximately \$100 million of hotel dispositions would be consummated during 1999; (vi) net borrowings would be made in the third and fourth quarters of 1999 in the aggregate amount of \$235 million at an assumed average interest rate of 7.20%; (vii) new bond debt of \$1.2 billion would be issued at an assumed average interest rate of 8.10% in the first two quarters of 1999 to refinance existing mortgage debt with the remainder used for general corporate purposes; (viii) a limited number of OP Units would be issued in connection with hotel acquisitions; and (ix) conversion in full of the debentures and the Convertible Preferred Shares.

As indicated in the press release, certain matters discussed in the press  $% \left( 1\right) =\left( 1\right) \left( 1\right)$ release, as well as those set forth in the preceding paragraph, including the statements with respect to the range of analysts' estimates for 1999 Comparative FFO, constitute forward-looking statements within the meaning of the Private Litigation Reform Act of 1995. Host's actual Comparative FFO for 1999 could differ materially and adversely from that stated or implied by such forward-looking statements due to a number of factors, many of which are beyond Host's control. In particular, some of the assumptions relied on by Host in commenting upon the range of analysts' estimates for 1999 Comparative FFO inevitably will not be accurate and unanticipated events may occur subsequent to the date thereof which may cause the actual Comparative FFO for 1999 to vary materially from that set forth in such statements. The assumptions upon which Host based its comments on the range of analysts' estimates for 1999 Comparative FFO reflected conditions it expected to exist and courses of action it expected to take during 1999 and required Host to make estimates and assumptions as to significant matters affecting Comparative FFO, including certain matters beyond the control of Host and/or which are inherently difficult to predict. For example, in making comments on the range of analysts' estimates for 1999 Comparative FFO, Host assumed that a certain level of rates and occupancy could be achieved by its properties during 1999 and that financing would be available on terms consistent with anticipated market conditions; that certain acquisition opportunities would be available during 1999 consistent with those available during 1998; and that the costs of construction of new properties and of expansion of existing hotel properties would not change materially in 1999 from those experienced during 1998. There can be no assurance that events will not occur which will cause these or other factors affecting Comparative FFO to change, which events or factors may cause Comparative FFO for 1999 to be materially different from that presented in the press release and in the immediately preceding paragraph. Factors which might cause such a difference include, but are not limited to, those discussed under "Forward-Looking Statements" on page 1 of this Proxy Statement/Prospectus, "Risk Factors" commencing on page 32 of this Proxy Statement/Prospectus and under "Distribution and Other Policies" commencing on page 117 of this Proxy Statement/Prospectus. Moreover, if the REIT Conversion is consummated, Host REIT and Crestline will be operated as separated public companies, and their results will not be reported together. Because of these and other factors, there is no assurance that the consensus estimates for 1999 Comparative FFO will be realized or that actual Comparative FFO for 1999 will not vary materially from that presented above. Host and Host REIT disclaim any responsibility to update the information contained in the press release or the foregoing paragraphs.

Partnership Activities. Host has general and limited partner interests in numerous limited partnerships which own 240 hotels (including 20 full-service hotels) as of the date hereof, managed by Marriott International. Debt of the hotel limited partnerships is typically secured by first mortgages on the properties and is generally nonrecourse to the partnership and the partners. However, Host has committed to advance amounts to certain affiliated limited partnerships, if necessary, to cover certain future debt service requirements. Such commitments were limited, in the aggregate, to an additional \$60 million at January 2, 1998. Subsequent to year-end, this amount was reduced to \$20 million in connection with the refinancing and acquisition of a controlling interest in the partnership which owns the Atlanta Marriott Marquis. Amounts repaid to the Company under these guarantees totaled \$2 million and \$13 million in 1997 and 1996, respectively. Fundings by Host under these guarantees amounted to \$10 million in 1997 and \$8 million for 1995.

Leases. Host leases certain property and equipment under noncancelable operating leases, including the long-term ground leases for certain hotels, generally with multiple renewal options. The leases related to the 53 Courtyard properties and 18 Residence Inn properties sold during 1995 and 1996 are nonrecourse to Host and contain provisions for the payment of contingent rentals based on a percentage of sales in excess of stipulated amounts. Host remains contingently liable on certain leases related to divested non-lodging properties. Such contingent liabilities aggregated \$110 million at January 2, 1998. However, management considers the likelihood of any substantial funding related to these divested properties' leases to be remote.

Inflation. Host's hotel lodging properties are impacted by inflation through its effect on increasing costs and on the managers' ability to increase room rates. Unlike other real estate, hotels have the ability to change room rates on a daily basis, so the impact of higher inflation generally can be passed on to customers.

A substantial portion of Host's debt bears interest at fixed rates. This debt structure largely mitigates the impact of changes in the rate of inflation on future interest costs. However, Host currently is exposed to variable interest rates through an interest rate exchange agreement with a financial institution with an aggregate notional amount of \$100 million. Under this agreement, Host collects interest based on the specified floating rates of one month LIBOR (rate of 6% at January 2, 1998) and pays interest at fixed rates (rate of 7.99% at January 2, 1998). This agreement expires in 1998 in conjunction with the maturity of the mortgage on the New York Marriott Marquis. Host's Line of Credit and the mortgage on the San Diego Marriott Hotel and Marina (\$199 million at January 2, 1998) bears interest based on variable rates. Accordingly, the amount of Host's interest expense under the interest rate swap agreements and the floating rate debt for a particular year will be affected by changes in short-term interest rates.

#### YEAR 2000 PROBLEM

Year 2000 issues have arisen because many existing computer programs and chip-based embedded technology systems use only the last two digits to refer to a year, and therefore do not properly recognize a year that begins with "20" instead of the familiar "19". If not corrected, many computer applications could fail or create erroneous results. The following disclosure provides information regarding the current status of the Company's Year 2000 compliance program.

The Company has adopted the compliance program because it recognizes the importance of minimizing the number and seriousness of any disruptions that may occur as a result of the Year 2000 issue. The Company's compliance program includes an assessment of the Company's hardware and software computer systems and embedded systems, as well as an assessment of the Year 2000 issues relating to third parties with which the Company has a material relationship or whose systems are material to the operations of the Company's hotel properties. The Company's efforts to ensure that its computer systems are Year 2000 compliant have been segregated into two separate phases: in-house systems and third-party systems. Following the REIT Conversion, Crestline, as the Lessee of the Company's hotels, will deal directly with Year 2000 matters material to the operation of the Hotels, and Crestline has agreed to adopt and implement the program outlined below with respect to third-party systems for all Hotels for which it is Lessee.

In-House Systems. Since the distribution of Marriott International on October 8, 1993, the Company has invested in the implementation and maintenance of accounting and reporting systems and equipment that are intended to enable the Company to provide adequately for its information and reporting needs and which are also Year 2000 compliant. Substantially all of the Company's in-house systems have already been certified as Year 2000 compliant through testing and other mechanisms and the Company has not delayed any systems projects due to the Year 2000 issue. The Company is in the process of engaging a third party to review its Year 2000 in-house compliance. Management believes that future costs associated with Year 2000 issues for its in-house systems will be insignificant and therefore not impact the Company's business, financial condition and results of operations. The Company has not developed, and does not plan to develop, a separate contingency plan for its in-house systems due to their current Year 2000 compliance.

Third-Party Systems. The Company relies upon operational and accounting systems provided by third parties, primarily the managers and operators of its hotel and senior living properties, to provide the appropriate property-specific operating systems (including reservation, phone, elevator, security, HVAC and other systems) and to provide it with financial information. Based on discussion with the third parties that are critical to the Company's business, including the managers and operators of its hotels, the Company believes that these parties are in the process of studying their systems and the systems of their respective vendors and service providers and, in many cases, have begun to implement changes, to ensure that they are Year 2000 compliant. However, the Company has not received any oral or written assurances that these third parties will be Year 2000 compliant on time. To the extent these changes impact property-level systems, the Company may be required to fund capital expenditures for upgraded equipment and software. The Company does not expect these charges to be material, but is committed to making these investments as required. To the extent that these changes relate to a third party manager's centralized systems (including reservations, accounting, purchasing, inventory, personnel and other systems), the Company's management agreements generally provide for these costs to be charged to the Company's properties subject to annual limitations, which costs will be borne by Crestline following the REIT Conversion. The Company and Crestline expect that the third party managers will incur Year 2000 costs in lieu of costs for its centralized systems related to system projects that otherwise would have been pursued and therefore, its overall level of centralized systems charges allocated to the properties will not materially increase as a result of the Year 2000 compliance effort. The Company and Crestline believe that this deferral of certain system projects will not have a material impact on its future results of operations, although it may delay certain productivity enhancements at its properties. The Company (and, following the REIT Conversion, Crestline) will continue to monitor the efforts of these third parties to become Year 2000 compliant and will take appropriate steps to address any non-compliance issues. The Company believes that in the event of material Year 2000 noncompliance, the Company will have the right to seek recourse against the manager under its third party management agreements. The management agreements, however, generally do not specifically address the Year 2000 compliance issue. Therefore, the amount of any recovery in the event of Year 2000 non-compliance at a property, if any, is not determinable at this time (and if the REIT Conversion occurs, only a portion of such recovery would accrue to the Company, through increased lease rental payments from Crestline).

The Company (and, following the REIT Conversion, Crestline) will work with the third parties to ensure that appropriate contingency plans will be developed to address the most reasonably likely worst case Year 2000 scenarios, which may not have been identified fully. In particular, the Company has had extensive discussions regarding the Year 2000 problem with Marriott International, the manager of a substantial majority of its hotel properties and all of its senior living communities. Due to the significance of Marriott International to the Company's business, a detailed description of Marriott International's state of readiness follows.

Marriott International has adopted an eight-step process toward Year 2000 readiness, consisting of the following: (i) Awareness: fostering understanding of, and commitment to, the problem and its potential risks; (ii) Inventory: identifying and locating systems and technology components that may be affected; (iii) Assessment: reviewing these components for Year 2000 compliance, and assessing the scope of Year 2000 issues; (iv) Planning: defining the technical solutions and labor and work plans necessary for each particular system; (v) Remediation/Replacement: completing the programming to renovate or replace the problem software or hardware; (vi) Testing and Compliance Validation: conducting testing, followed by independent validation by a separate internal verification team; (vii) Implementation: placing the corrected systems and technology back into the business environment; and (viii) Quality Assurance: utilizing a dedicated audit team to review and test significant projects for adherence to quality standards and program methodology.

Marriott International has grouped its systems and technology into three categories for purposes of Year 2000 compliance: (i) information resource applications and technology (IT Applications)--enterprise-wide systems supported by Marriott International's centralized information technology organization ("IR"); (ii) Business-initiated Systems ("BIS")--systems that have been initiated by an individual business unit, and

that are not supported by Marriott International's IR organization; and (iii) Building Systems--non-IT equipment at properties that use embedded computer chips, such as elevators, automated room key systems and HVAC equipment. Marriott International is prioritizing its efforts based on how severe an effect noncompliance would have on customer service, core business processes or revenues, and whether there are viable, non-automated fallback procedures (System Criticality).

Marriott International measures the completion of each phase based on documented and quantified results, weighted for System Criticality. As of the end of the 1998 third quarter, the awareness and inventory phases were complete for IT Applications and nearly complete for BIS and Building Systems. For IT Applications, the Assessment, Planning and Remediation/Replacement phases were each over 80 percent complete, and Testing and Compliance Validation had been completed for a number of key systems, with most of the remaining work in its final stage. For BIS and Building Systems, Assessment and Planning were in the mid-to-upper-range of completion, with a substantial amount of work in process, while the progress level for Remediation/Replacement and Testing and Compliance Validation had not yet been documented and quantified. Quality Assurance is also in progress for IT Applications and is scheduled to begin for BIS and Building Systems in the near future. Marriott International's goal is to substantially complete the Remediation/Replacement and Testing phases for its System Critical IT Applications by the end of 1998, with 1999 reserved for unplanned contingencies and for Compliance Validation and Quality Assurance. For System Critical BIS and Building Systems, the same level of completion is targeted for June 1999 and September 1999, respectively.

Marriott International has initiated Year 2000 compliance communications with its significant third party suppliers, vendors and business partners, including its franchisees. Marriott International is focusing its efforts on the business interfaces most critical to its customer service and revenues, including those third parties that support the most critical enterprise-wide IT Applications, franchisees generating the most revenues, suppliers of the most widely used Building Systems and BIS, the top 100 suppliers, by dollar volume, of non-IT products, and financial institutions providing the most critical payment processing functions. Responses have been received from a majority of the firms in this group. A majority of these respondents have either given assurances of timely Year 2000 compliance or have identified the necessary actions to be taken by them or Marriott International to achieve timely Year 2000 compliance for their products.

Marriott International is also establishing a common approach for testing and addressing Year 2000 compliance issues for its managed and franchised properties. This includes a guidance protocol for operated properties, and a Year 2000 "Toolkit" for franchisees containing relevant Year 2000 compliance information. Marriott International is also utilizing a Year 2000 best-practices sharing system.

Risks. There can be no assurances that Year 2000 remediation by the Company or third parties will be properly and timely completed, and failure to do so could have a material adverse effect on the Company, its business and its financial condition. The Company cannot predict the actual effects to it of the Year 2000 problem, which depends on numerous uncertainties such as: (i) whether significant third parties properly and timely address the Year 2000 issue; and (ii) whether broad-based or systemic economic failures may occur. Moreover, following the REIT Conversion, the Company will be reliant upon Crestline to interface with third parties in addressing the Year 2000 issue at the hotels leased by Crestline. The Company is also unable to predict the severity and duration of any such failures, which could include disruptions in passenger transportation or transportation systems generally, loss of utility and/or telecommunications services, the loss or disruption of hotel reservations made on centralized reservation systems and errors or failures in financial transactions or payment processing systems such as credit cards. Due to the general uncertainty inherent in the Year 2000 problem and the Company's dependence on third parties (including Crestline following the REIT Conversion), the Company is unable to determine at this time whether the consequences of Year 2000 failures will have a material impact on the Company. The Company's Year 2000 compliance program, and Crestline's adoption thereof following the REIT Conversion are expected to significantly reduce the level of uncertainty about the Year 2000 problem and management believes that the possibility of significant interruptions of normal operations should be

Accounting Standards. Host adopted Statement of Financial Accounting Standard ("SFAS") No. 114, "Accounting by Creditors for Impairment of a Loan" and SFAS No. 121, "Accounting for the Impairment of Long-Lived Assets and for Long-Lived Assets to be Disposed Of" during 1995. Adoption of these statements

did not have a material effect on Host's continuing operations. See the discussion below for a discussion of the impact of the adoption of SFAS No. 121 on discontinued operations.

SFAS No. 121 requires that an impairment loss be recognized when the carrying amount of an asset exceeds the sum of the undiscounted estimated future cash flows associated with the asset. Under SFAS No. 121, Host reviewed the impairment of its assets employed in its operating group business lines (airport, toll plaza and sports and entertainment) on an individual operating unit basis. For each individual operating unit determined to be impaired, an impairment loss equal to the difference between the carrying value and the fair market value of the unit's assets was recognized. Fair market value was estimated to be the present value of expected future cash flows of the individual operating unit, as determined by management, after considering such factors as future air travel and toll-pay vehicle data and inflation. As a result of the adoption of SFAS No. 121, Host recognized a non-cash, pre-tax charge against earnings during the fourth quarter 1995 of \$47 million, which was reflected in discontinued operations.

In the fourth quarter of 1996, Host adopted SFAS No. 123, "Accounting for Stock Based Compensation." The adoption of SFAS No. 123 did not have a material effect on Host's financial statements.

During 1997, Host adopted SFAS No. 128, "Earnings Per Share," SFAS No. 129, "Disclosure of Information About Capital Structure" and SFAS No. 131, "Disclosures About Segments of an Enterprise and Related Information." The adoption of these statements did not have a material effect on Host's consolidated financial statements and the appropriate disclosures required by these statements have been incorporated herein.

In the First Quarter 1998, Host adopted SFAS No. 130, "Reporting Comprehensive Income." SFAS No. 130 establishes standards for reporting and display of comprehensive income and its components in financial statements. The objective of SFAS No. 130 is to report a measure of all changes in equity of an enterprise that result from transactions and other economic events of the period other than transactions with owners. Comprehensive income is the total of net income and all other nonowner changes in equity.

Host's only component of other comprehensive income is the right to receive up to 1.4 million shares of Host Marriott Services Corporation's common stock or an equivalent cash value subsequent to exercise of the options held by certain former and current employees of Marriott International. For the First Three Quarters 1998 and First Three Quarters 1997, Host's other comprehensive loss was \$3 million and \$8 million, respectively. As of September 11, 1998 and January 2, 1998, Host's accumulated other comprehensive income was approximately \$7 million and \$10 million, respectively.

On November 20, 1997, the Emerging Issues Task Force ("EITF") of the Financial Accounting Standards Board reached a consensus of EITF 97-2, "Application of FASB Statement No. 94 and APB Opinion No. 16 to Physician Practice Management Entities and Certain Other Entities with Contractual Management Arrangements." EITF 97-2 addresses the circumstances in which a management entity may include the revenues and expenses of a managed entity in its financial statements.

Host has considered the impact of EITF 97-2 on its financial statements and has determined that EITF 97-2 requires the Company to include property-level sales and operating expenses of its hotels and senior living communities in its statements of operations. Host will adopt EITF 97-2 in the fourth quarter of 1998, with retroactive effect in prior periods to conform to the new presentation. Application of EITF 97-2 to the consolidated financial statements for the First Three Quarters 1998, First Three Quarters 1997 and Fiscal Years 1997, 1996 and 1995 would have increased both revenues and operating expenses by approximately \$1,501 million, \$1,161 million, \$1,713 million, \$1,225 million and \$878 million, respectively, and would have had no impact on operating profit, net income or earnings per share.

EITF 98-9, "Accounting for Contingent Rent in Interim Financial Periods," was issued on May 21, 1998. EIFT 98-9 requires a lessor to defer recognition of contingent rental income in interim periods until the specified target that triggers the contingent rental income is achieved. EITF 98-9 has no impact on Host prior to the REIT Conversion, but will impact the revenue recognized under the Leases on a quarterly basis following the REIT Conversion.

## MANAGEMENT

## DIRECTORS AND EXECUTIVE OFFICERS OF HOST REIT

The following table sets forth certain information with respect to persons who will be Directors immediately after the completion of the Merger and the other transactions comprising the REIT Conversion, and the executive officers of Host REIT (or the Operating Partnership), all of whom are currently directors, executive officers or key employees of Host.

NAME 	AGE POSITION WITH HOST REIT (OR THE OPERATING PARTNERSHIP)
Richard E. Marriott(1) J.W. Marriott, Jr.(1) R. Theodore Ammon Robert M. Baylis Ann Dore McLaughlin Harry L. Vincent, Jr	59 Chairman of the Board of Directors 66 Director 48 Director 59 Director 56 Director 79 Director
John G. Schreiber Terence C. Golden Robert E. Parsons,	53 Director, President and Chief Executive Officer
Jr	42 Executive Vice President and Chief Financial Officer  36 Executive Vice President and Chief Operating Officer
Townsend Donald D. Olinger	51 Senior Vice President, General Counsel and Corporate Secretary 39 Senior Vice President and Corporate Controller

The following is a biographical summary of the experience of the persons who will be Directors and executive officers of Host REIT after the REIT Conversion:

Richard E. Marriott. Mr. Richard E. Marriott has been a Director of Host since 1979 and is a Director of Marriott International, Inc., Host Marriott Services Corporation, Potomac Electric Power Company and the Polynesian Cultural Center, and he is Chairman of the Board of First Media Corporation. He also serves as a Director of certain subsidiaries of Host and is a past President of the National Restaurant Association. In addition, Mr. Marriott is the President and a Trustee of the Marriott Foundation for People with Disabilities. Mr. Marriott's term as a Director of Host REIT will commence at or prior to the REIT Conversion and will expire at the 2001 annual meeting of stockholders. Mr. Marriott joined Host in 1965 and has served in various executive capacities. In 1984, he was elected Executive Vice President, and in 1986, he was elected Vice Chairman of the Board of Directors. In 1993, Mr. Marriott was elected Chairman of the Board. Mr. Marriott also has been responsible for management of Host's government affairs functions.

- J.W. Marriott, Jr. Mr. J.W. Marriott, Jr. has been a Director of Host since 1964 and is Chairman of the Board and Chief Executive Officer of Marriott International, Inc., and a Director of Host Marriott Services Corporation, General Motors Corporation and the U.S.-Russia Business Council. He also serves on the Boards of Trustees of the Mayo Foundation, Georgetown University and the National Geographic Society. He is on the President's Advisory Committee of the American Red Cross, the Executive Committee of the World Travel & Tourism Council and is a member of the Business Council and the Business Roundtable. Mr. Marriott's term as a Director of Host REIT will commence at or prior to the REIT Conversion and will expire at the 1999 annual meeting of stockholders.
- R. Theodore Ammon. Mr. Ammon has been a Director of Host since 1992 and is a private investor and Chairman of Big Flower Holdings, Inc. He was formerly a General Partner of Kohlberg Kravis Roberts & Company (a New York and San Francisco-based investment firm) from 1990 to 1992, and was an executive of such firm prior to 1990. Mr. Ammon is also a member of the Board of Directors of Samsonite Corporation and Culligan Water Technologies, Inc. In addition, he serves on the Board of Directors of the New York YMCA, Jazz @ Lincoln Center and the Institute of International Education and on the Board of Directors of Bucknell

<sup>(1)</sup> Richard E. Marriott and J.W. Marriott, Jr. are brothers.

University. Mr. Ammon's term as a Director of Host REIT will commence at or prior to the REIT Conversion and will expire at the 2001 annual meeting of stockholders.

Robert M. Baylis. Mr. Baylis has been a Director of Host since 1996 and is a Director of The International Forum, an executive education program of the Wharton School of the University of Pennsylvania. He was formerly Vice Chairman of CS First Boston. Mr. Baylis also serves as a Director of New York Life Insurance Company, Covance, Inc. and Gryphon Holdings, Inc. In addition, he is an overseer of the University of Pennsylvania Museum of Archeology and Anthropology. Mr. Baylis's term as a Director of Host REIT will commence at or prior to the REIT Conversion and will expire at the 2000 annual meeting of stockholders.

Ann Dore McLaughlin. Ms. McLaughlin has been a Director of Host since 1993 and currently is Chairman of the Aspen Institute. She formerly served as President of the Federal City Council from 1990 until 1995. Ms. McLaughlin has served with distinction in several U.S. Administrations in such positions as Secretary of Labor and Under Secretary of the Department of the Interior. She also serves as a Director of AMR Corporation, Fannie Mae, General Motors Corporation, Kellogg Company, Nordstrom, Potomac Electric Power Company, Union Camp Corporation, Donna Karan International, Inc., Vulcan Materials Company, Harman International Industries, Inc. and Sedgwick Group plc. Ms. McLaughlin's term as a Director of Host REIT will commence at or prior to the REIT Conversion and will expire at the 2000 annual meeting of stockholders.

Harry L. Vincent, Jr. Mr. Vincent has been a Director of Host since 1969 and is a retired Vice Chairman of Booz-Allen & Hamilton, Inc. He also served as a Director of Signet Banking Corporation from 1973 until 1989. Mr. Vincent's term as a Director of Host REIT will commence at or prior to the REIT Conversion and will expire at the 1999 annual meeting of stockholders.

John G. Schreiber. Mr. Schreiber has been a Director of Host since 1998 and is President of Schreiber Investments, Inc. and a Senior Advisor and Partner of Blackstone Real Estate Advisors, L.P. Mr. Schreiber serves as a Trustee of AMLI Residential Properties Trust and as a Director of Urban Shopping Centers, Inc., JMB Realty Corporation and a number of mutual funds advised by T. Rowe Price Associates, Inc. Prior to his retirement as an officer of JMB Realty Corporation in 1990, Mr. Schreiber was Chairman and CEO of JMB/Urban Development Company and an Executive Vice President of JMB Realty Corporation. Mr. Schreiber's term as a Director of Host REIT will commence at or prior to the REIT Conversion and will expire at the 1999 annual meeting of stockholders.

Terence C. Golden. Mr. Golden has been a Director of Host since 1995 and was named President and Chief Executive Officer of Host in 1995. Mr. Golden also serves as a Director of certain subsidiaries of Host. He also serves as Chairman of Bailey Realty Corporation and Bailey Capital Corporation and various affiliated companies. In addition, Mr. Golden is Chairman of the Washington Convention Center and a Director of Prime Retail, Inc., Cousins Properties, Inc., The Morris and Gwendolyn Cafritz Foundation and the District of Columbia Early Childhood Collaborative. He is also a member of the Executive Committee of the Federal City Council. Mr. Golden will be President and Chief Executive Officer of Host REIT commencing at or prior to the REIT Conversion and his term as a Director of Host REIT will commence at or prior to the REIT Conversion and will expire at the 2000 annual meeting of stockholders. Prior to joining Host, Mr. Golden was Chairman of Bailey Realty Corporation and prior to that had served as Chief Financial Officer of The Oliver Carr Company. Before joining The Oliver Carr Company, he served as Administrator of the General Services Administration and as Assistant Secretary of Treasury, and he was co-founder and national managing partner of Trammel Crow Residential Companies.

Robert E. Parsons, Jr. Mr. Parsons joined Host's Corporate Financial Planning staff in 1981 and was made Assistant Treasurer in 1988. In 1993, Mr. Parsons was elected Senior Vice President and Treasurer of Host, and in 1995, he was elected Executive Vice President and Chief Financial Officer of Host. Since September 1998, Mr. Parsons has been President and an initial Director of Host REIT but he will resign from such positions upon or prior to the REIT Conversion. Mr. Parsons will be Executive Vice President and Chief Financial Officer of Host REIT commencing at or prior to the REIT Conversion.

Christopher J. Nassetta. Mr. Nassetta joined Host in October 1995 as Executive Vice President and was elected Chief Operating Officer of Host in 1997. Mr. Nassetta will be Executive Vice President and Chief Operating Officer of Host REIT commencing at or prior to the REIT conversion. Prior to joining Host, Mr. Nassetta served as President of Bailey Realty Corporation from 1991 until 1995. He had previously served as Chief Development Officer and in various other positions with The Oliver Carr Company from 1984 through 1991.

Christopher G. Townsend. Mr. Townsend joined Host's Law Department in 1982 as a Senior Attorney. In 1984, Mr. Townsend was made Assistant Secretary of Host, and in 1986, he was made Assistant General Counsel. In 1993, Mr. Townsend was elected Senior Vice President, Corporate Secretary and Deputy General Counsel. In January 1997, he was elected General Counsel. Since September 1998, Mr. Townsend has been Vice President and an initial Director Of Host REIT but he will resign from such positions upon or prior to the REIT Conversion. Mr. Townsend will be Senior Vice President, General Counsel and Secretary of Host REIT commencing at or prior to the REIT Conversion.

Donald D. Olinger. Mr. Olinger joined Host in 1993 as Director--Corporate Accounting. Later in 1993, Mr. Olinger was promoted to Senior Director and Assistant Controller. He was promoted to Vice President--Corporate Accounting in 1995. In 1996, he was elected Senior Vice President and Corporate Controller. Since September 1998, Mr. Olinger has been Vice President of Host REIT but he will resign from such position at or prior to the REIT Conversion. Mr. Olinger will be Senior Vice President and Corporate Controller of Host REIT commencing at or prior to the REIT Conversion. Prior to joining Host, Mr. Olinger was with the public accounting firm of Deloitte & Touche.

# COMMITTEES OF THE BOARD OF DIRECTORS

Promptly following the consummation of the REIT Conversion, the Board of Directors of Host REIT will establish the following committees:

Audit Committee. The Audit Committee will be comprised of five Directors who are not employees of Host REIT, namely, R. Theodore Ammon (Chair), Harry L. Vincent, Jr., Ann Dore McLaughlin, John G. Schreiber and Robert M. Baylis. The Audit Committee will meet at least three times a year with the independent auditors, management representatives and internal auditors; recommend to the Board of Directors appointment of independent auditors; approve the scope of audits and other services to be performed by the independent and internal auditors; consider whether the performance of any professional service by the auditors other than services provided in connection with the audit function could impair the independence of the outside auditors; and review the results of internal and external audits, the accounting principles applied in financial reporting, and financial and operational controls.

Compensation Policy Committee. The Compensation Policy Committee will be comprised of six Directors who are not employees of Host REIT, namely, Harry L. Vincent, Jr. (Chair), R. Theodore Ammon, John G. Schreiber, Robert M. Baylis, J.W. Marriott, Jr. and Ann Dore McLaughlin. The Compensation Policy Committee's functions will include recommendations on policies and procedures relating to senior officers' compensation and various employee stock plans, and approval of individual salary adjustments and stock awards in those areas.

Nominating and Corporate Governance Committee. The Nominating and Corporate Governance Committee will be comprised of six Directors who are not employees of Host REIT, namely, Ann Dore McLaughlin (Chair), Harry L. Vincent, Jr., John G. Schreiber, R. Theodore Ammon, J.W. Marriott, Jr. and Robert M. Baylis. It will consider candidates for election as Directors and will be responsible for keeping abreast of and making recommendations with regard to corporate governance in general. In addition, the Nominating and Corporate Governance Committee will fulfill an advisory function with respect to a range of matters affecting the Board of Directors and its Committees, including the making of recommendations with respect to qualifications of Director candidates, compensation of Directors, the selection of committee chairs, committee assignments and related matters affecting the functioning of the Board.

Host REIT may from time to time form other committees as circumstances warrant. Such committees will have authority and responsibility as delegated by the Board of Directors.

# COMPENSATION OF DIRECTORS

Directors who are also officers of Host REIT will receive no additional compensation for their services as Directors. Directors elected by the holders of Host REIT Common Stock and who are not officers will receive an annual retainer fee of \$25,000 as well as an attendance fee of \$1,250 for each stockholders' meeting, meeting of the Board of Directors or meeting of a committee of the Board of Directors, regardless of the number of meetings held on a given day. The chair of each committee of the Board of Directors will receive an additional annual retainer fee of \$1,000, except for the chair of the Compensation Policy Committee, Mr. Vincent, who will receive an annual retainer fee of \$6,000. (The higher annual retainer fee paid to the chair of the Compensation Policy Committee relates to his additional duties which include, among other things, the annual performance appraisal of the chief executive officer on behalf of the Board, although the final appraisal is determined by the Board.) Any individual Director receiving these fees may elect to defer payment of all such fees or any portion thereof pursuant to the Plans (to the extent such Director is eligible) and/or Host REIT's Non-Employee Directors' Deferred Stock Compensation Plan. Directors also will be reimbursed for travel expenses and other out-of-pocket costs incurred in attending meetings or in visiting hotels or other properties controlled by Host REIT or by Marriott International.

In 1997, the following Directors of the Company received special one-time awards of Company common stock in the amounts indicated: Mr. Ammon, 4,000 shares; Mr. Baylis, 7,000 shares; Ms. McLaughlin, 7,000 shares and Mr. Vincent, 7,000 shares. The special one-time awards of Company common stock vest at the rate of 10% per year of a Directors service on the Board, with credit given for each year of service already completed, and will also become fully vested upon the death or disability of the Directors.

# **EXECUTIVE COMPENSATION**

The table below sets forth a summary of the compensation paid by Host for the last three fiscal years to the Chief Executive Officer and the four additional most highly compensated executive officers of Host for Host's fiscal year 1997 (the "Named Executive Officers").

		LONG-TERM COMPENSATION					
		ANNUA	L COMPENSA	ATION	AWARDS	PAYOUTS	
NAME AND PRINCIPAL POSITION	FISCAL YEAR	SALARY(1)(2)	BONUS(3)	OTHER ANNUAL COMPENSATION	RESTRICTED STOCK AWARDS(4)(5)	LTIP PAYOUTS	ALL OTHER COMPENSATION(6)
Richard E. Marriott	1997	\$271,449	\$108,580	\$110,789(7)	\$ 0	\$0	\$ 22,668(8)
Chairman of the Board	1996	262,951	105,180	, , ,	0	0	21, 439(8)
Taranaa C Caldan(0)	1995 1997	250,554	100,000		0 354 603	0 0	12,634
Terence C. Golden(9) President and Chief	1997	619,045 600,017	557,141 480,013	, , ,	354,693 10,476,603	0	66,105 560,827(11)
Executive Officer	1995	190,656	152,152		10,470,003	0	0
Robert E. Parsons, Jr	1997	338,889	254, 167		0	0	36,231
Executive Vice	1996	328,447	263,490		3,658,277	0	26,273
President and Chief Financial Officer	1995	213,767	123,649		0	0	10,951
Christopher J.							
Nassetta(9)	1997	338,889	254,167		0	0	36,231
Executive Vice	1996	328,447	263,490		3,647,513	0	119,168(11)
President and Chief Operating Officer Christopher G.	1995	78,000	50,700	Θ	0	0	0
Townsend	1997	202,962	111,629	0	1,015,800	0	18,405
Senior Vice President,	1996	186,232	102,428	0	Θ	0	15,891
General Counsel	1995	156,375	93,825	0	Θ	0	7,658

- (1) Fiscal year 1996 base salary earnings were for 53 weeks.
- (2) Salary amounts include base salary earned and paid in cash during the fiscal year, the amount of base salary deferred at the election of the executive officer under the Host Marriott Corporation Executive Deferred Compensation Plan and the increase in base salary for the period October 1, 1997 through the end of the fiscal year which was paid in 1998.
- (3) Bonus includes the amount of cash bonus earned pursuant to Host's Performance-Based Annual Incentive Bonus Plan (which was approved by the stockholders in 1996) and to the named individual's performance-based bonus plan during the fiscal year, which is either paid subsequent to the end of each fiscal year or deferred under the Host Marriott Corporation Executive Deferred Compensation Plan.
- (4) During 1997, the Compensation Policy Committee (the "Committee") of the Board of Directors approved the grant of restricted stock to certain key employees of Host, including Mr. Townsend. In 1996, the Committee approved similar grants of restricted stock to certain key employees of Host, including Messrs. Golden, Parsons and Nassetta. Mr. Golden also received grants of restricted stock on November 6, 1997 and on August 1, 1996 which were pursuant to the terms of his restricted stock agreement with Host. Messrs. Golden, Parsons and Nassetta each received awards which vest over a five-year period, and Mr. Townsend received an award which vests over a three-year period. All such awards consist of shares subject to restrictions relating primarily to continued employment ("General Restrictions") and shares subject to annual performance objectives such as financial performance of Host ("Performance Restrictions"). Performance objectives are established by the Committee and are subject to annual review and revision. Sixty percent of the shares awarded to each executive officer have annual Performance Restrictions, and forty percent of the shares awarded have General Restrictions conditioned upon continued employment. In addition, Messrs. Parsons and Nassetta each received an award of restricted stock which vests sixty percent on December 31, 1998 and forty percent on December 31, 2000, subject to the attainment of certain performance criteria and  $\frac{1}{2}$ to the named individual's continued employment ("Special Team Awards"). All Special Team Awards are presented above as "Restricted Stock Awards," and the value stated above is the fair market value on the date of the grant. At Mr. Golden's request and in order to motivate the management team to enhance stockholder value, the Committee issued these Special Team Awards of the shares of restricted stock to key executives of Host in connection with Mr. Golden's joining Host. The dollar value of those awards has been reflected in the Restricted Stock Awards column of the table for the Named Executive Officers. In the event that the executives to whom restricted stock was granted do not continue in the employ of Host or do not meet the performance criteria set by the Committee, those shares will be forfeited, and the Committee has retained the right to grant any forfeited restricted shares to Mr. Golden.

  (5) The aggregate number and value of shares of deferred stock and restricted
- stock subject to "General Restrictions" and "Performance Restrictions' (see footnote 4 above) held by each Named Executive Officer as of the end of fiscal year 1997 are as follows: Mr. R.E. Marriott, 264,000 shares valued at \$5,071,440; Mr. Golden, 655,231 shares valued at \$12,586,987; Mr. Nassetta, 240,267 shares valued at \$4,615,529; Mr. Parsons, 261,531 shares valued at \$5,073,335; and Mr. Townsend, 56,321 shares valued at \$1,078,485. During the period in which any restrictions apply, holders of restricted stock are entitled to receive all dividends or other distributions paid with respect to such stock. Under the terms of certain restricted stock award agreements granted under the long-term incentive plan, each share of restricted stock vests upon a change in control of Host. The stock bonus awards granted by Host are generally derived based on dividing 20% of each individual's annual cash bonus award by the average of the high and low trading prices for a share of common stock on the last trading day of the fiscal year. No voting rights or dividends are attributed to award shares until such award shares are distributed. Stock bonus awards may be denominated as current awards or deferred awards. A current award is distributed in 10 annual installments commencing one year after the award is granted. A deferred award is distributed in a lump sum or in up to 10 annual installments following termination of employment. Deferred award shares contingently vest pro rata in annual installments commencing one year after the stock bonus award is granted to the employee. Awards are not subject to forfeiture once the employee reaches age 55 with 10 years of service with Host or upon (i) retirement after 20 years of service, (ii) disability or (iii) death.
- (6) Amounts included in "All Other Compensation" represent total matching Host contribution amounts received under the Host Marriott Corporation (HMC) Retirement and Savings Plan and the Host Marriott Corporation Executive Deferred Compensation Plan. In 1997, the amounts attributable to the Host Marriott Corporation (HMC) Retirement and Savings Plan account for each Named Executive Officer were as follows: Mr. R.E. Marriott, \$9,024; Mr. Golden, \$7,939; Mr. Nassetta, \$9,024; Mr. Parsons, \$9,500; and Mr. Townsend, \$8,448. The amounts attributable to the Host Marriott Corporation Executive Deferred Compensation Plan for each named executive officer were as follows: Mr. R.E. Marriott, \$13,644; Mr. Golden, \$58,166; Mr. Nassetta, \$27,207; Mr. Parsons, \$26,731; and Mr. Townsend, \$9,957.
- (7) Amount includes \$92,000 in 1997, \$86,700 in 1996, and \$86,200 in 1995 for the allocation of Host personnel for non-Host business.
- (8) Effective beginning in 1996, Mr. R.E. Marriott waived (i) payments due to be made to him under the Host Marriott Corporation Executive Deferred Compensation Plan following his retirement and (ii) common stock due to

be distributed to him under the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan (formerly called the Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan) following his retirement. In connection with this waiver, Host entered into an arrangement to purchase life insurance policies for the benefit of a trust established by Mr. R.E. Marriott. The cost of the life insurance policies to Host has been actuarially determined and will not exceed the projected after-tax cost Host expected to incur in connection with the payments under the Host Marriott Corporation Executive Deferred Compensation Plan and the stock distributions under the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan (formerly called the Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan) that were waived by Mr. R.E. Marriott.

- (9) Mr. Golden joined Host as President and Chief Executive Officer on September 1, 1995. Mr. Nassetta joined Host as Executive Vice President on October 1, 1995.
- (10) Amount represents reimbursement of travel expenses of Mr. Golden's spouse when she accompanies him on Host business trips.
- (11) As part of their restricted stock agreements with Host, Messrs. Golden and Nassetta were awarded 44,910 and 8,421 shares of Host common stock, respectively, on February 1, 1996. The value of the shares was \$516,465 for Mr. Golden and \$96,842 for Mr. Nassetta.

The table below sets forth, on an aggregated basis, (i) information regarding the exercise during fiscal year 1997 of options to purchase Host Common Stock (and shares of the common stock of other companies which Host has previously spun off) by each of the executive officers listed on the Executive Compensation table above, and (ii) the value on January 2, 1998 of all unexercised options held by such individuals. Host did not grant any options to the executive officers listed on the following table in fiscal year 1997. Terence C. Golden and Christopher J. Nassetta do not have any options to purchase stock in any of the companies listed in the following table.

AGGREGATED STOCK OPTION EXERCISES IN LAST FISCAL YEAR AND FISCAL YEAR-END OPTION VALUES

NUMBER OF SHARES
UNDERLYING UNEXERCISED
OPTIONS AT FISCAL YEAR END
(#)

VALUE OF UNEXERCISED INTHE
MONEY OPTIONS AT
FISCAL YEAR END (2)
(\$)

						-		
NAME	COMPANY(1)	SHARES ACQUIRED ON EXERCISE (#)	VALUE REALIZED (\$)	EXERCISABLE	UNEXERCISABLE	E	EXERCISABLE	UNEXERCISABLE
						-		
R. E. Marriott	HM HMS MI	35,000 0 0	421,379 0 0	55,700(3) 11,140 55,700		0 0 0	944,307 145,741 3,198,557	0 0 0
	TOTAL	35,000	421,379	122,540		0	4,288,605	0
R. E. Parsons, Jr	HM	2,500	40,825	20, 225		0	311,837	0
·	HMS	500	5,993	4,045		0	49,212	0
	MI	0	Θ	1,625		0	85,423	0
	TOTAL	3,000	46,819	25,895		0	446,472	0
C. G. Townsend	HM	0	Θ	6,975		0	110,745	0
	HMS	0	Θ	1,395		0	17,354	0
	MI	0	0	0		0	Θ	0
	TOTAL	0	0	8,370		0	128,100	0

<sup>(1) &</sup>quot;HM" represents options to purchase Host Common Stock ("Host Options").
"HMS" represents options to purchase HM Services common stock. "MI" represents options to purchase Marriott International common stock. In connection with Host's issuance on December 29, 1995 of a special dividend (the "HMS Special Dividend") of HM Services common stock to Host's stockholders, and pursuant to the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan (formerly called the Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan), all Host Options held by employees of Host were adjusted to reflect the HMS Special Dividend by providing each option holder with the option to purchase one share of HM Services common stock for every option to purchase five shares of Host Common Stock held as of the close of business on December 29, 1995. The exercise price of the HM Services options was set, and the price of Host Options was adjusted, so that the economic value of Host Options prior to the HMS Special Dividend was preserved and not increased or decreased as a result of the HMS Special Dividend. In addition, in connection with Host's issuance on October 8, 1993 of a special dividend (the "MI Special Dividend") of Marriott International common stock to Host's stockholders, and pursuant to the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan (formerly called the Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan), all Host Options held by employees of Host were adjusted to reflect the MI Special Dividend by providing each option holder with the option to purchase one share of Marriott International common stock for every option to purchase one share of Host Common Stock held as of the close of business on October 8, 1993. The exercise price of the Marriott International options was set, and the price of Host Options was adjusted, so that the economic value of Host Options prior to the MI Special Dividend was preserved and not increased or decreased as a result of the MI Special Dividend.

<sup>(2)</sup> Based on a per share price for Host Common Stock of \$19.21, a per share price for HM Services common stock of \$14.43, and a per share price for Marriott International common stock of \$68.56. These prices reflect the average of the high and low trading prices on the New York Stock Exchange on January 2, 1998.

<sup>(3)</sup> In order to facilitate the REIT Conversion, Mr. Marriott has agreed to the cancellation of such options upon consummation of the Merger in exchange for a grant of an equal number of stock appreciation rights payable in cash that are economically equivalent to the options that are canceled.

#### LONG-TERM INCENTIVE PLAN

The table below sets forth the number of shares of Host Common Stock awarded under a long-term incentive plan on February 1, 1996 to Messrs. Parsons and Nassetta and on January 22, 1997 to Mr. Townsend.

Richard E. Marriott and Terence C. Golden did not receive any of the type of awards reported in the following table. These awards represent the number of restricted shares of Host Common Stock that may vest during or at the end of a three-year period, subject to the satisfaction of certain time and performance restrictions established by the Compensation Policy Committee of the Board of Directors. The vesting provisions governing these awards are subject to review and revision by the Compensation Policy Committee. The performance criteria are set in advance of the completion of the performance year, and if the time and performance criteria are not achieved, the full number of shares will be forfeited.

The shares may be paid in full if either of the following two formulas is net:

- . Prior to November 1, 1998, the average price of Host Common Stock traded on the NYSE during any consecutive 60-day period shall increase to 172.8% of the price of Host Common Stock on November 2, 1995; or
- . The average of the high and low prices of Host Common Stock traded on the NYSE for each of the first five days of trading prior to November 1, 1998 is 172.8% of the price of Host Common Stock on November 2, 1995.

The price of Host Common Stock on November 2, 1995 was determined to be \$11.08 (which reflects an adjustment for the distribution of the common stock of Host Marriott Services Corporation to Host's stockholders on December 29, 1995), and therefore the target price under the two formulas is \$19.146 (i.e., 172.8% of \$11.08). This increase represents a 20% compounded annual growth rate in the price of Host Common Stock.

#### LONG-TERM INCENTIVE PLAN AWARDS IN LAST FISCAL YEAR

NAME	NUMBER OF SHARES	PERFORMANCE OR OTHER PERIOD UNTIL MATURITY OR PAYOUT			
Robert E. Parsons, Jr	84,206	3 years 3 years 3 years			

# **EMPLOYMENT AGREEMENTS**

The Operating Partnership expects to have employment agreements with certain of its executive officers but there is no assurance that this will be the case. The terms of such agreements currently are under negotiation and are not expected to be finalized until the Effective Date.

# 1998 EMPLOYEE BENEFITS ALLOCATION AGREEMENT

As part of the REIT Conversion, Host, the Operating Partnership and Crestline expect to enter into the 1998 Employee Benefits Allocation Agreement which is expected to govern the allocation of responsibilities with respect to various compensation, benefits and labor matters. Under the 1998 Employee Benefits Allocation Agreement, Crestline is expected to assume from Host certain liabilities relating to covered benefits and labor matters with respect to individuals who are employed by Host REIT or its affiliates on or before the Effective Date who will be employed by Crestline or its affiliates ("Transferred Employees") and the Operating Partnership is expected to assume from Host certain other liabilities relating to employee benefits and labor matters. The 1998 Employee Benefits Allocation Agreement also is expected to govern the treatment of awards under the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan, formerly called the Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan (the "Comprehensive Stock Incentive Plan"),

as part of the REIT Conversion. The 1998 Employee Benefits Allocation Agreement is expected to require Crestline to establish the Crestline Capital Corporation 1998 Comprehensive Stock Incentive Plan. Additionally, the 1998 Employee Benefits Allocation Agreement is expected to provide that the Operating Partnership will adopt the Comprehensive Stock Incentive Plan.

#### COMPREHENSIVE STOCK INCENTIVE PLAN

Host sponsors the Comprehensive Stock Incentive Plan for purposes of attracting and retaining highly qualified employees. Host has reserved 44,442,911 shares of Host Common Stock for issuance pursuant to the Comprehensive Stock Incentive Plan. As part of the REIT Conversion, the Comprehensive Stock Incentive Plan is expected to be adopted by the Operating Partnership. Outstanding awards of Host Common Stock issued or reserved under the Comprehensive Stock Incentive Plan are expected to be exchanged for Host REIT Common Stock and Crestline common stock, according to the terms of the 1998 Employee Benefits Allocation Agreement.

Under the terms of the Comprehensive Stock Incentive Plan, an eligible full-time employee may receive an award of (i) options to purchase Host Common Stock, (ii) deferred shares of Host Common Stock, (iii) restricted shares of Host Common Stock, (iv) stock appreciation rights, (v) special recognition awards or (vi) other equity-based awards, including but not limited to, phantom shares of Host Common Stock, performance shares of Host Common Stock, bonus shares of Host Common Stock, dividend equivalent units or similar securities or rights. After the REIT Conversion, all awards under the Comprehensive Stock Incentive Plan will be for Host REIT Common Stock.

Options granted to officers and key employees will have an exercise price of not less than the fair market value on the date of grant. Incentive stock options granted under the Comprehensive Stock Incentive Plan expire no later than 10 years after the date of grant and non-qualified stock options expire no later than 15 years after the date of grant.

Under the terms of the Comprehensive Stock Incentive Plan, an eligible full-time employee may receive an award of deferred shares of Host Common Stock. Deferred shares may be granted as part of a bonus award or deferred stock agreement. After the REIT Conversion, an award of deferred shares under the Comprehensive Stock Incentive Plan will be for Host REIT Common Stock. Deferred shares generally vest over ten years in annual installments commencing one year after the date of grant.

The Comprehensive Stock Incentive Plan also provides for the issuance of restricted shares of Host Common Stock to officers and key executives to be distributed over the next three or five years in annual installments based on continued employment and the attainment of certain performance criteria. After the REIT Conversion, an award of restricted shares under the Comprehensive Stock Incentive Plan will be for Host REIT Common Stock.

Under the terms of the Comprehensive Stock Incentive Plan, an eligible full-time employee may receive a bonus award. Bonus awards may be part of a management incentive program which pays part of the annual performance bonus awarded to managers and other key employees in shares of Host Common Stock. A bonus award entitles the holder to receive a distribution of Host's Common Stock in accordance with the underlying agreement. Holders of bonus awards vest in the shares covered by their award over ten years in annual installments commencing one year after grant. Unless the holder of a bonus award elects otherwise, vested shares are distributed in 10 consecutive, approximately equal, annual installments. After the REIT Conversion, all bonus awards will be for shares of Host REIT Common Stock.

The Comprehensive Stock Incentive Plan authorizes the grant of stock appreciation rights ("SARs") to eligible full-time employees. SARs awarded under the Comprehensive Stock Incentive Plan give the holder the right to an amount equal to the appreciation in the value of the Host Common Stock over a specified price. SARs may be paid on the Host Common Stock, cash or other form or combination form of payout. After the REIT Conversion, SARs awarded under the Comprehensive Stock Incentive Plan will be tied to the appreciation in the value of Host REIT Common Stock.

Under the Comprehensive Stock Incentive Plan, an eligible full-time employee may receive a Special Recognition Award. Special Recognition Awards may be paid in the form of Host Common Stock or an option to purchase Host Common Stock at an amount not less than fair market value on the date of grant. After the REIT Conversion, Special Recognition Awards will be for Host REIT Common Stock.

# STOCK PURCHASE PLAN

Host sponsors the Host Marriott Corporation Employee Stock Purchase Plan (the "Stock Purchase Plan"). Under the terms of the Stock Purchase Plan, an individual who is: (i) an active eligible employee on the last day of the prior plan year, (ii) working more than 20 hours per week and (iii) customarily employed more than five months in a calendar year may, on the first day of the plan year, purchase Host Common Stock through contributions or payroll deductions at the lower of the fair market value on the first or last day of such plan year. The Stock Purchase Plan is expected to be adopted by the Operating Partnership as part of the REIT Conversion.

## 401(K) PLAN

Host sponsors the Host Marriott Corporation (HMC) Retirement and Saving Plan (the "401(k) Plan"). The 401(k) Plan has received a favorable ruling from the Internal Revenue Service ("IRS") as to its tax-qualified status. The 401(k) Plan is expected to be adopted by the Operating Partnership as part of the REIT Conversion. The 401(k) Plan is available to all eligible employees immediately upon their date of hire. A participant may elect to contribute from 1% to 15% of his compensation to the 401(k) Plan. Each year, Host makes a fixed matching contribution equal to 50% of the first 6% of the compensation contributed to the 401(k) Plan by employees. In addition, Host may make a discretionary contribution, in an amount, if any, determined annually by the Board, to the 401(k) Plan for the benefit of eligible employees.

Under the terms of the 401(k) Plan, participants may elect to invest part or all of their plan benefits in Host Common Stock. As part of the Merger, all shares of Host Common Stock held under the 401(k) Plan are expected to be converted to Host REIT Common Stock and shares of Crestline common stock. It is expected that after the REIT Conversion, 401(k) Plan participants will be able to elect to invest all or part of their plan benefits in Host REIT Common Stock.

#### NON-EMPLOYEE DIRECTOR PLAN

Host sponsors the Host Marriott Corporation Non-Employee Directors' Deferred Stock Compensation Plan (the "Non-Employee Director Plan") for purposes of attracting and retaining qualified non-employee Directors. Under the terms of the Non-Employee Director Plan, a non-employee Director may elect to defer payment of part or all of his Directors' fees from Host until such individual is no longer a member of the Board. In addition, the Non-Employee Director Plan provides for: (i) a special one time grant of Host Common Stock to participants who were directors of Host on May 1, 1997; and (ii) annual grants of 750 shares of Host Common Stock effective on May 14, 1997 and at each annual meeting thereafter. Currently, fees that are deferred under the Non-Employee Director Plan are treated as if invested in shares of Host Common Stock using the fair market value of such shares on the date of deferral. After the REIT Conversion, Host REIT intends to treat Directors' fees deferred under the Non-Employee Director Plan as if invested in Host REIT Common Stock.

Non-Employee Directors may elect to receive payment of their benefits under the Non-Employee Director Plan in Host Common Stock in lump sum or installment payments. After the REIT Conversion, Host REIT expects to allow participants of the Non-Employee Director Plan to elect to receive their benefits in Host REIT Common Stock in lump sum or installment payments.

# DEFERRED COMPENSATION PLAN

Host maintains the Host Marriott Corporation Executive Deferred Compensation Plan (the "Deferred Compensation Plan") for the benefit of select executives and directors. A participant may elect to defer part of

his compensation or director's fees in the case of non-employee directors pursuant to the Deferred Compensation Plan. The Deferred Compensation Plan will be adopted by the Operating Partnership as part of the REIT Conversion.

# LIMITATION OF LIABILITY AND INDEMNIFICATION

The Host REIT Charter and Host REIT Bylaws contain provisions limiting the liability of Host REIT's present and former directors and officers to the corporation and its stockholders and obligating Host REIT to indemnify present and former directors and officers all in accordance with Maryland law. See "The Restructuring Transactions--Limitation of Liability and Indemnification of Directors and Officers."

## INDEMNIFICATION AGREEMENTS

Host REIT intends to enter into indemnification agreements with each of its directors and officers. The indemnification agreements will require, among other things, that Host REIT indemnify its directors and officers to the fullest extent permitted by law and advance to its directors and officers all related expenses, subject to reimbursement if it is subsequently determined that indemnification is not permitted.

#### CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

#### RELATIONSHIP BETWEEN HOST AND MARRIOTT INTERNATIONAL

Host and Marriott International, prior to October 8, 1993, were operated as a single consolidated company. On October 8, 1993 in connection with the issuance of a special dividend (the "Marriott International Distribution"), the consolidated company's businesses were split between Host and Marriott International. Thereafter, Host retained the capital intensive lodging real estate business (the "Ownership Business") and the airport/tollroad concessions business (the "Host/Travel Plazas Business"), while Marriott International took over the management of the lodging and service management businesses (the "Lodging/Service Management Business"). (On December 29, 1995, Host distributed the Host/Travel Plazas Business to the stockholders of HM Services; see "--Relationship between Host and Host Marriott Services Corporation" below.) On the date of the Marriott International Distribution, Host and its subsidiaries and Marriott International and its subsidiaries entered into certain contractual arrangements governing their relationship following the Marriott International Distribution.

J.W. Marriott, Jr. and Richard E. Marriott beneficially own approximately 10.6% and 10.2%, respectively, of the outstanding shares of common stock of Marriott International. By reason of their ownership of such shares of common stock of Marriott International and their positions as Chairman and a Director, respectively, of Marriott International, J.W. Marriott, Jr. and Richard E. Marriott, who will also be a Director and Chairman, respectively, of Host REIT, could be deemed in control of Marriott International within the meaning of the federal securities laws. Other members of the Marriott family might also be deemed control persons of Marriott International by reason of their ownership of shares of Marriott International and/or their relationship to other family members.

Prior to the Marriott International Distribution, Host and Marriott International entered into a Distribution Agreement (the "Marriott International Distribution Agreement"), which provided for, among other things, (i) the division between Host and Marriott International of certain liabilities and (ii) certain other agreements governing the relationship between Host and Marriott International following the Marriott International Distribution.

Subject to certain exceptions, the Marriott International Distribution Agreement provided for, among other things, assumptions of liabilities and cross-indemnities designed to allocate, effective as of the Marriott International Distribution, financial responsibility for the liabilities arising out of or in connection with the Lodging/Service Management Business to Marriott International and its subsidiaries, and financial responsibility for the liabilities arising out of or in connection with the Ownership Business and Host/Travel Plazas Business, along with the consolidated company's liabilities under a substantial portion of its pre-existing financing and long-term debt obligations, to Host and its retained subsidiaries. The agreements executed in connection with the Marriott International Distribution Agreement also set forth certain specific allocations of liabilities between Host and Marriott International.

Under the Marriott International Distribution Agreement, Marriott International obtained the Marriott International Purchase Right which provided Marriott International with the right, until June 2017, to purchase up to 20% of each class of Host's voting stock (determined after assuming full exercise of the right) at its then fair market value (based on an average of trading prices during a specified period), upon the occurrence of certain specified events generally involving a change in control of Host. The Marriott International Purchase Right could be exercised for a 30-day period following the date a person or group of affiliated persons has (i) become the beneficial owner of 20% or more of the total voting power of the then outstanding shares of Host's voting stock or (ii) announced a tender offer for 30% more of the total voting power of the then outstanding shares of Host's common stock. The purchase price for the common stock to be purchased upon the exercise of the Marriott International Purchase Right is determined by taking the average of the closing sale price of the common stock during the 30 consecutive trading days preceding the date the Marriott International Purchase Right becomes exercisable. The Marriott International Purchase Right will continue in effect with respect to Host REIT after the

Merger as to the Host REIT Common Stock, subject to the following limitations intended to protect the REIT status of Host REIT. The Marriott International Purchase Right will be exercisable only to the extent that neither (i) Marriott International, or any entity in which it has a direct or indirect interest (and which would be deemed, under the applicable attribution rules, to own the shares of Host REIT owned by Marriott International), would, as a result of such exercise, own, taking into account the applicable attribution rules, more than 9.8% of both Host REIT and Crestline, any subsidiary of Crestline or any other tenant of Host REIT nor (ii) any owners of direct or indirect interests in Marriott International would, as a result of such exercise, own, taking into account the applicable attribution rules, more than 9.8% of both Host REIT and Crestline, any subsidiary of Crestline or any other tenant of Host REIT. In addition to the foregoing limitation, in the event the Operating Partnership is or would be considered a "publicly traded partnership" within the meaning of the Code, the Marriott International Purchase Right will be exercisable only if such acquisition and ownership of Host REIT Common Stock would not cause the Operating Partnership to be considered to own, directly or by attribution, 10% or more of Crestline, any subsidiary of Crestline or any other tenant of Host REIT (taking into account the applicable attribution rules and any stock of Crestline that the Operating Partnership is deemed to own under the attribution rules by reason of the ownership of an interest in the Operating Partnership by the Blackstone Entities).

The Marriott International Purchase Right will have an antitakeover effect to the extent that any person considering acquiring a substantial or controlling block of Host REIT Common Stock will face the possibility that its ability to exercise control would be impaired by the exercise of the Marriott International Purchase Right. In addition, the exercise price of the Marriott International Purchase Right could be lower than the price at which a potential acquiror might be willing to purchase a 20% block of Host REIT Common Stock because the purchase price for the Marriott International Purchase Right is based on the average trading price during a 30-day period which may be prior to the announcement of a takeover event. This potential price differential might have a further antitakeover effect by discouraging potential acquirors of Host REIT.

For the purpose of governing certain of the ongoing relationships between Host and Marriott International after the Marriott International Distribution, Host and Marriott International have entered into other agreements. Host believes that the agreements are fair to both parties and contain terms which are generally comparable to those which would have been reached in arm'slength negotiations with unaffiliated parties. Among such other agreements between Host and Marriott International are:

(i) Lodging Management and Franchise Agreements. Marriott International and certain of its subsidiaries have entered into management agreements with Host and certain of its subsidiaries to manage for fees the Marriott Hotels, Resorts and Suites, Ritz-Carlton hotels, Courtyard hotels and Residence Inns owned or leased by Host and its subsidiaries. Marriott International has also entered into franchise agreements with Host and certain of its subsidiaries to allow Host to use the Marriott brand, associated trademarks, reservation systems and other related items in connection with Host's operation of ten Marriott hotels not managed by Marriott International.

Each of those management and franchise agreements reflects market terms and conditions and is substantially similar to the terms of management and franchise agreements with other third-party owners regarding lodging facilities of a similar type. In 1997, Host paid to Marriott International fees of \$166 million from the managed and franchised lodging properties owned or leased by Host.

In addition, Host or one of its subsidiaries is a partner in several unconsolidated partnerships (some of which will be consolidated in connection with the REIT Conversion) that, at the end of 1997, owned 241 lodging properties operated by Marriott International or certain of its subsidiaries under long-term agreements. In such cases, Host or its subsidiary typically serves as the general partner. In 1997, these unconsolidated partnerships paid to Marriott International fees of \$119 million pursuant to such agreements. The partnerships also paid \$23 million in rent to Marriott International in 1997 for land leased from Marriott International upon which certain of the limited service partnerships' hotels are located.

In connection with the REIT Conversion, these management and franchise agreements will be assigned to the Lessees for the term of the applicable Leases (but the Operating Partnership will remain obligated in the event the Lessees fail to perform their obligations).

- (ii) Credit Agreement. In 1995, Marriott International and a subsidiary of Host entered into a Credit Agreement pursuant to which the subsidiary had the right to borrow up to \$225 million from Marriott International. In 1997, however, Host entered into a revolving line of credit agreement with third parties, and as a result, Host terminated the revolving line of credit under the Credit Agreement with Marriott International. Host remains subject to various covenants and guaranty reimbursement obligations under the Credit Agreement.
- (iii) Tax Sharing Agreement. Host and Marriott International have entered into a tax sharing agreement that defines the parties' rights and obligations with respect to deficiencies and refunds of federal, state and other income or franchise taxes relating to Host's businesses for tax years prior to the Marriott International Distribution and with respect to certain tax attributes of Host after the Marriott International Distribution. Host and Marriott International have agreed to cooperate with each other and to share information in preparing tax returns and in dealing with other tax matters.
- (iv) Noncompetition Agreement. Host and Marriott International entered into a noncompetition agreement that defines the parties' rights and obligations with respect to certain businesses operated by Marriott International and Host. In general, under the noncompetition agreement, Host and its subsidiaries are prohibited from entering into or acquiring any business that competes with the hotel management business as conducted by Marriott International until October 8, 2000, subject to certain exceptions. See "--Senior Living Communities Acquisitions."
- (v) Administrative Services Agreements. Marriott International and Host have entered into a number of agreements pursuant to which Marriott International has agreed to provide certain continuing administrative services to Host and its subsidiaries. Such services are provided on market terms and conditions. In general, the administrative services agreements can be kept in place at least through the end of 1998.
- (vi) Marriott International Guarantees. In connection with the Marriott International Distribution, Host and Marriott International entered into agreements pursuant to which Marriott International has agreed to guarantee Host's performance in connection with certain partnership, real estate and project loans and other Host obligations. Such guarantees are limited in an aggregate principal amount of up to \$107 million at June 19, 1998. Marriott International has not been required to make any payments pursuant to the guarantees.

In addition to the foregoing agreements, Host and Marriott International have had occasion to enter into other agreements in the ordinary course of business. Host believes that such agreements are fair to both parties and contain terms which are generally comparable to those which would have been reached in arm's-length negotiations with unaffiliated parties. Among such other agreements between Host and Marriott International are:

(a) Hotel Acquisitions. Marriott International has provided, and Host expects that Marriott International in the future will provide, financing to Host for a portion of the cost of acquiring properties to be operated or franchised by Marriott International. In 1997, Marriott International did not provide any new acquisition financing, although Host remained indebted to Marriott International for acquisition financing from prior years. Marriott International provided Host with \$70 million of mortgage financing in 1995 for the acquisition of three full-service hotels at an average interest rate of 8.5%. Marriott International subsequently sold one of the loans in 1996. In 1996, Marriott International and Host formed a joint venture (which will be owned by a Non-Controlled Subsidiary) and Marriott International provided Host with \$29 million in debt financing at an average interest rate of 12.7% and with \$28 million in preferred equity, for the acquisition of two full-service hotels in Mexico City.

(b) Senior Living Communities Acquisitions. On June 21, 1997, Host acquired the outstanding common stock of Forum Group, Inc. (the "Forum Group") from Marriott Senior Living Services, Inc., a subsidiary of Marriott International. Host purchased the Forum Group portfolio of 29 premier senior living

communities for approximately \$460 million, including approximately \$270 million in debt (\$59 million of which was provided by Marriott International). In 1997, Host had completed \$56 million of the approximately \$107 million expansion plan to add approximately 1,060 units to these communities. As a result, an additional \$33 million of debt financing has been provided by Marriott International and Marriott International may provide additional financing as the expansion plan is completed. The properties will continue to be managed by Marriott International. From the date of acquisition through the end of 1997, Host paid to Marriott International management fees of \$6 million from the senior living properties owned by Host. In connection with the acquisition, Host and Marriott International entered into a noncompetition agreement that defines the parties' rights and obligations with respect to the operation of senior living services by Marriott International and Host. In general, under the noncompetition agreement, Host and its subsidiaries are prohibited from entering into or acquiring any business that competes with the senior living management business as conducted by Marriott International until 2007. In 1997, Host also acquired all but 1% of the remaining 50% interest in the joint venture which owned the 418-unit Leisure Park senior living community from Marriott International for approximately \$23 million, including approximately \$15 million of mortgage debt assumed by Host. Shares of Crestline, which will own the senior living communities business, will be distributed to Host's stockholders as part of the Initial E&P Distribution in connection with the REIT Conversion.

(c) 1993 Employee Benefits Allocation Agreement. Host and Marriott International have entered into an Employee Benefits and Other Employment Matters Allocation Agreement ("1993 Employee Benefits Allocation Agreement") that provides for the allocation of certain responsibilities with respect to employment compensation, benefit and labor matters. The 1993 Employee Benefits Allocation Agreement was amended as of March 27, 1998 to: (i) reflect various conversions and redenominations that were necessary as a result of the spin-off and acquisitions described in Marriott International's February 12, 1998 Proxy, and to add New Marriott MI, Inc. (renamed Marriott International, Inc.) as a party to the 1993 Employee Benefits Allocation Agreement. In general, the 1993 Employee Benefits Allocation Agreement provides that Host retained all employee liabilities for employees who on or after the Marriott International Distribution were employees of Host, and that old Marriott International, Inc., which was renamed Sodexho Marriott Services, Inc., in 1998, retained all liabilities for employees who on or after the Marriott Distribution were employees of Marriott International. Pursuant to the 1993 Employee Benefits Allocation Agreement, and in connection with the Marriott Distribution, Host also adjusted outstanding awards under the Host employee benefit plans. The 1993 Employee Benefits Allocation Agreement is expected to be amended as part of the REIT Conversion to add the Operating Partnership and Crestline as parties to the agreement and to reflect the 1998 Employee Benefits Allocation Agreement.

# RELATIONSHIP BETWEEN HOST AND HOST MARRIOTT SERVICES CORPORATION

On December 29, 1995, Host issued a special dividend (the "HMSC Distribution") which split Host's businesses between Host and HM Services. Prior to December 29, 1995, HM Services was a wholly-owned subsidiary of Host. Thereafter, Host retained the capital intensive lodging real estate business (the "Ownership Business"), while HM Services took over the airport/tollroad concessions business (the "Host/Travel Plazas Business"). Host and its subsidiaries and HM Services and its subsidiaries have entered into certain relationships following the HMSC Distribution.

Richard E. Marriott and J.W. Marriott, Jr. beneficially own approximately 6.75% and 6.88%, respectively, of the outstanding shares of common stock of HM Services. By reason of their ownership of such shares of common stock of HM Services and their positions as Directors of HM Services, Richard E. Marriott and J.W. Marriott, Jr., who are also Chairman and a Director, respectively, of Host, could be deemed in control of HM Services within the meaning of the federal securities laws. Other members of the Marriott family might also be deemed control persons of HM Services by reason of their ownership of shares of HM Services and/or their relationship to other family members.

161

Prior to the HMSC Distribution, Host and HM Services entered into a Distribution Agreement (the "HMSC Distribution Agreement"), which provided for, among other things, (i) certain asset transfers to occur prior to the HMSC Distribution, (ii) the HMSC Distribution, (iii) the division between Host and HM Services of certain liabilities and (iv) certain other agreements governing the relationship between Host and HM Services following the HMSC Distribution.

Subject to certain exceptions, the HMSC Distribution Agreement provides for, among other things, assumptions of liabilities and cross-indemnities designed to allocate, effective as of the HMSC Distribution, financial responsibility for the liabilities arising out of or in connection with the Host/Travel Plazas Business to HM Services and its subsidiaries and financial responsibility for the liabilities arising out of or in connection with the Ownership Business to Host and its retained subsidiaries. The agreements executed in connection with the HMSC Distribution Agreement also set forth certain specific allocations of liabilities between Host and HM Services. The HMSC Distribution Agreement also provides that HM Services will assume its proportionate share of Host's current obligation for certain employee benefit awards denominated in Host common stock currently held by employees of Marriott International.

For the purpose of governing certain of the ongoing relationships between Host and HM Services after the HMSC Distribution, Host and HM Services have entered into other agreements. Host believes that the agreements are fair to both parties and contain terms which are generally comparable to those which would have been reached in arm's-length negotiations with unaffiliated parties. Among such other agreements between Host and HM Services are:

- (i) Tax Sharing Agreement. Host and HM Services have entered into a tax sharing agreement that defines the parties' rights and obligations with respect to deficiencies and refunds of federal, state and other income or franchise taxes relating to Host's businesses for tax years prior to the HMSC Distribution and with respect to certain tax attributes of Host after the HMSC Distribution. Host and HM Services have agreed to cooperate with each other and to share information in preparing tax returns and in dealing with other tax matters.
- (ii) Guarantees of Concession Agreements. Host and HM Services have entered into agreements pursuant to which Host has agreed to guarantee HM Services' performance in connection with certain tollroad concessions operated by HM Services. Host has not been required to make any payment pursuant to the guarantees and does not anticipate making any such payment in 1998.
- (iii) 1995 Employee Benefits Allocation Agreement. Host and HM Services have entered into an Employee Benefits and Other Employment Matters Allocation Agreement (the "1995 Employee Benefits Allocation Agreement") that provides for the allocation of certain responsibilities with respect to employee compensation, benefits and labor matters. In general, the 1995 Employee Benefits Allocation Agreement provides that Host retain all employee liabilities for employees who on or after the HMSC Distribution were employees of Host, and that HM Services retain all employee liabilities for employees who on or after the HMSC Distribution were employees of HM Services. Pursuant to the 1995 Employee Benefits Allocation Agreement, and in connection with the HMSC Distribution, Host also adjusted outstanding awards under Host employee benefit plans. The 1995 Employee Benefits Allocation Agreement is expected to be amended as part of the REIT Conversion to add the Operating Partnership and Crestline as parties to the agreement and to reflect the 1998 Employee Benefits Allocation Agreement.

RELATIONSHIP BETWEEN HOST AND CRESTLINE AFTER THE INITIAL E&P DISTRIBUTION

For the purposes of governing certain of the ongoing relationships between Crestline and Host after the Initial E&P Distribution and to provide mechanisms for an orderly transition, Crestline and Host will enter into various agreements, in addition to the Leases as described below. The descriptions of such agreements are qualified in their entirety by reference to the agreements, the forms of which (other than the Distribution Agreement and the Corporate Transitional Services Agreement) are included as exhibits to the Registration Statement of which this Proxy Statement/Prospectus is a part.

#### Distribution Agreement

Prior to the Initial E&P Distribution, Crestline and Host will enter into a distribution agreement (the "Distribution Agreement"), which will provide for, among other things, (i) the distribution of shares of Crestline in connection with the Initial E&P Distribution; (ii) the division between Crestline and Host of certain assets and liabilities; (iii) the contribution to Crestline of Host's 3% general partnership interest in Boynton Beach Limited Partnership, which owns a senior living community located in Boynton Beach; (iv) the transfer to Crestline of the 25% interest in the Swissotel management company to be acquired in the Blackstone Acquisition; (v) the return to Crestline of those shares of Crestline common stock held for delivery to the Blackstone Entities in the Blackstone Acquisition if that transaction does not occur prior to January 1, 1999, subject, however, to the right of Host to cause Crestline to sell such 18% of the common stock of Crestline to the Blackstone Entities at fair market value if the Blackstone Acquisition is consummated at a later date; and (vi) certain other agreements governing the relationship between Crestline and Host following the Initial E&P Distribution. Crestline also will grant Host REIT a contingent right to purchase Crestline's interest in Swissotel Management (USA) L.L.C. at fair market value in the event the tax laws are changed so that Host REIT could own such interest without jeopardizing its status as a REIT.

Subject to certain exceptions, the Distribution Agreement will provide for, among other things, assumptions of liabilities and cross-indemnities designed to allocate to Crestline, effective as of the date of the Initial E&P Distribution, financial responsibilities for liabilities arising out of or in connection with the business of the senior living communities.

## Tax Sharing Agreement

Crestline and Host will enter into a tax sharing agreement (the "Tax Sharing Agreement") which will define each party's rights and obligations with respect to deficiencies and refunds of federal, state and other income or franchise taxes relating to Crestline's business for taxable years prior to the Initial E&P Distribution and with respect to certain tax attributes of Crestline after the Initial E&P Distribution. Generally, Host will be responsible for filing consolidated returns and paying taxes for periods through the date of the Initial E&P Distribution, and Crestline will be responsible for filing returns and paying taxes for subsequent periods.

#### Asset Management Agreement

Host will enter into an asset management agreement (the "Asset Management Agreement") with Crestline and at least one Non-Controlled Subsidiary, pursuant to which Crestline will agree to provide review and advice on the management and operation of the hotels. Generally, Crestline will provide the following consulting services: (i) review of operating and financial results (including site visits) and meet with Host and the Non-Controlled Subsidiaries, as applicable, at least quarterly, to review such results of the hotels; (ii) review of financial statements and budgets, including periodic accounting statements, annual operating budgets, FF&E budgets and management analysis reports; (iii) review of revenue and capital spending projections; (iv) administration of approvals relating to the operation of the hotel required under any related loan documents; (v) advice relating to any changes to the hotel management agreements; (vi) review of market conditions and competition for each of the hotels; (vii) monitoring and negotiating with governmental agencies in connection with any condemnation proceedings against the hotels; and (viii) monitoring and negotiating with insurance companies and contractors following a casualty at a hotel. Crestline will be paid a fee not to exceed \$4.5 million for each calendar year for its consulting services under the Asset Management Agreements, which will be allocated between Host REIT and the Non-Controlled Subsidiary. The Asset Management Agreements will have a term of two years with an automatic one year renewal, unless earlier terminated by either party in accordance with the terms thereof.

## Corporate Transitional Services Agreement

Crestline and Host will, prior to the date of the Initial E&P Distribution, enter into a transitional services agreement (the "Corporate Transitional Services Agreement") pursuant to which Crestline and Host will provide certain limited services to each other for a fee. Among other things, Host will provide centralized administrative and computer systems services to Crestline. Such services will be provided, as needed, at cost (including a reasonable overhead allocation) on a time and materials basis. The charges associated with such services are not expected to be material.

## Non-Competition Agreement

Crestline and Host will enter into a non-competition agreement that limits the respective parties' future business opportunities. See "Business and Properties--Noncompetition Agreements."

1998 Employee Benefits and Other Employment Matters Allocation Agreement

As part of the REIT Conversion, Host, the Operating Partnership and Crestline expect to enter into the 1998 Employee Benefits Allocation Agreement relating to various compensation, benefits and labor matters. See "Management--1998 Employee Benefits Allocation Agreement."

## Guarantee and Pooling Agreements

Crestline and certain of its subsidiaries will enter into guarantees of the Lease obligations of each Lessee. See "Business and Properties--The Leases."

#### PRINCIPAL STOCKHOLDERS

The following table sets forth, as of July 31, 1998, the beneficial ownership of Host REIT Common Stock and OP Units of (i) each person who is expected to hold more than a 5% interest in Host REIT or the Operating Partnership, (ii) directors of Host REIT, (iii) the Chief Executive Officer and the four most highly compensated executive officers of Host REIT and (iv) the directors and executive officers of Host REIT as a group. Unless otherwise indicated in the footnotes, all of such interests are owned directly and the indicated person or entity has sole voting and investment power.

The "Percent of All Common Stock and OP Units" represents the number of shares of Host REIT Common Stock and OP Units the person is expected to hold immediately after the REIT Conversion, as a percentage of the total number of shares of Host REIT Common Stock and OP Units expected to be outstanding immediately after the REIT Conversion (excluding OP Units held by Host REIT and its subsidiaries). The information in this table assumes that all transactions comprising the REIT Conversion are consummated as currently expected. The address of each beneficial owner is 10400 Fernwood Road, Bethesda, Maryland 20817 unless otherwise indicated.

NAME	NUMBER OF OP UNITS	PERCENT OF ALL OP UNITS(1)		PERCENT OF COMMON STOCK(3)	COMMON	PERCENT OF ALL COMMON STOCK AND OP UNITS(5)
R. Theodore Ammon	0	*	15,500	*	*	*
Robert M. Baylis	0	*	13,500	*	*	*
Terence C. Golden(6)	0	*	781,684	*	*	*
J.W. Marriott,			,			
Jr.(6)(7)(8)	18,135	*	13,275,014	6.50%	6.51%	4.85%
Richard E.	,					
Marriott(6)(8)(9)	15,314	*	13,203,209	6.47	6.47	4.82
Ann Dore McLaughlin	, O	*	9,500	*	*	*
John G. Schreiber(11)	875,000	1.25%	, O	*	*	*
Harry L. Vincent, Jr	. 0	*	25,100	*	*	*
Christopher J.			•			
Nassetta(6)	0	*	356,201	*	*	*
Robert E. Parsons,			•			
Jr.(6)	0	*	404,244	*	*	*
Christopher G.						
Townsend(6)	0	*	109,417	*	*	*
Blackstone						
Entities(12)	43,700,000	62.31	0	*	17.63	15.93
Dresdner RCM Global						
Investors LLC(13)	0	*	13,595,975	6.66	6.66	4.96
FMR Corp.(14)	0	*	22,532,574	11.03	11.03	8.21
Southeastern Asset						
Management, Inc.(15)	0	*	36,758,000	18.00	18.00	13.40
ALL DIRECTORS AND						
EXECUTIVE OFFICERS AS A						
GROUP (11						
PERSONS)(6)(10)	908,449	1.30%	24,209,204	11.86%	12.25%	9.16%

- \* less than 1%
- (1) Represents the number of OP Units held by the person as a percentage of the total number of OP Units to be issued to persons other than Host REIT and its subsidiaries in the REIT Conversion (70.1 million OP Units), assuming a value of \$12.50 per OP Unit.
- (2) Consists of Host REIT Common Stock received in the REIT Conversion as a result of ownership of Host.
- (3) Represents the number of shares of Host REIT Common Stock held by the person as a percentage of the total number of shares of Host REIT Common Stock expected to be outstanding immediately following the REIT Conversion (204.2 million shares of Host REIT Common Stock).
- (4) Assumes that all OP Units held by the person are redeemed for Host REIT Common Stock. The total number of shares of Host REIT Common Stock outstanding used in calculating this percentage (204.2 million shares of Host REIT Common Stock plus the number of OP Units beneficially owned by the person) assumes that none of the OP Units held by other persons are redeemed for Host REIT Common Stock.
- (5) Assumes that all OP Units held by the person are redeemed for Host REIT Common Stock. The total number of shares of Host REIT Common Stock and OP Units outstanding used in calculating this percentage (274.3 million) assumes that all of the OP Units held by other persons also are redeemed for Host REIT Common Stock.
- (6) Includes (i) the shares of unvested restricted stock granted under Host's Comprehensive Stock Incentive Plan, which are voted by the holder thereof and (ii) the following number of shares which could be acquired by the named persons through the exercise of stock options within 60 days of July 31, 1998: for J.W. Marriott, Jr., 810,447 shares; for Richard E. Marriott, 55,700 shares; for Mr. Parsons, 15,225 shares; for Mr. Townsend, 6,975 shares; and for all directors and executive officers as a group, 913,147 shares. Does not include any other shares reserved, contingently vested or awarded under the above-named Plan. In order to facilitate the REIT Conversion, J.W. Marriott, Jr. and Richard E. Marriott have agreed to the cancellation of 326,568 options and 55,700 options held by them, respectively, upon consummation of the Merger in

exchange for a grant of an equal number of stock appreciation rights payable in cash that are economically equivalent to the options that are canceled.

- (7) Host REIT Common Stock includes: (i) 1,977,450 shares held in trust for which J.W. Marriott, Jr. is the trustee or a co-trustee; (ii) 68,426 shares held by the wife of J.W. Marriott, Jr.; (iii) 704,555 shares held in trust for which the wife of J.W. Marriott, Jr. is the trustee or a co-trustee; (iv) 2,451,787 shares held by the J. Willard Marriott Foundation of which J.W. Marriott, Jr. is a co-trustee; (v) 2,707,590 shares held by a limited partnership whose general partner is a corporation of which J.W. Marriott, Jr. is the controlling stockholder; and (vi) 80,000 shares held by a limited partnership whose general partner is J.W. Marriott, Jr.; does not include shares held by the adult children of J.W. Marriott, Jr.; J.W. Marriott, Jr. disclaims beneficial ownership of all such shares.
- (8) By virtue of their ownership of shares of Host Common Stock and their positions as Chairman and Director, respectively, Richard E. Marriott and J.W. Marriott, Jr. could be deemed in control of Host within the meaning of the federal securities laws. Other members of the Marriott family might also be deemed control persons by reason of their ownership of shares and/or their relationship to other family members. J.W. Marriott, Jr., Richard E. Marriott, their mother Alice S. Marriott and other members of the Marriott family and various trusts established by members of the Marriott family owned beneficially an aggregate of 25,179,933 shares, or 12.31% of the total shares outstanding of Host Common Stock as of July 31, 1998.
- (9) Host REIT Common Stock includes: (i) 1,874,709 shares held in trust for which Richard E. Marriott is the trustee or a co-trustee; (ii) 68,219 shares held by the wife of Richard E. Marriott; (iii) 603,828 shares held in trust for which the wife of Richard E. Marriott is the trustee or a co-trustee; (iv) 2,451,787 shares held by the J. Willard Marriott Foundation of Which Richard E. Marriott is a co-trustee; and (v) 2,302,729 shares held by a corporation of Which Richard E. Marriott is the controlling stockholder; does not include shares held by the adult children of Richard E. Marriott; Richard E. Marriott disclaims beneficial ownership of all such shares.
- (10) Host REIT Common Stock includes the total number of shares held by trusts for which both J.W. Marriott, Jr. and Richard E. Marriott are cotrustees. Beneficial ownership of such shares is attributable to each of J.W. Marriott, Jr. and Richard E. Marriott in the table above under the Director subheading, but such shares are included only once in reporting the total number of shares owned by all directors and executive officers as a group. All directors and executive officers as a group (other than members of the Marriott family) owned beneficially an aggregate of 1,757,788 shares, or 0.86%, of the total shares outstanding as of July 31, 1998. In addition, the Host Marriott Corporation (HMC) Retirement and Savings Plan owned 65,257 shares, or 0.03% of the total shares outstanding as of July 31, 1998.
- (11) OP Units include only John G. Schreiber's proportionate share of OP Units to be received by the Blackstone Entities in the Blackstone Acquisition; John G. Schreiber disclaims beneficial ownership of all other OP Units to be acquired by the Blackstone Entities.
- be acquired by the Blackstone Entities.

  (12) The Blackstone Entities constitute a series of affiliated partnerships.

  Initially, a majority of the OP Units received pursuant to the Blackstone Acquisition will be held by such affiliated partnerships, but eventually will be distributed by such affiliated partnerships to their partners.
- (13) Represents shares of Host Common Stock held by Dresdner RCM Global Investors LLC ("Dresdner RCM") and its affiliates, RCM Limited L.P. ("RCM Limited") and RCM General Corporation ("RCM General"), and by Dresdner Bank AG, of which Dresdner RCM is a wholly owned subsidiary. Dresdner RCM has reported in a Schedule 13G under the Exchange Act, filed with the Commission, sole dispositive power over 12,943,675 shares and shared dispositive power over 282,000 shares. Of these shares, Dresdner RCM has reported sole voting power over 8,854,200 shares and does not share voting power with respect to any shares. In addition, Dresdner Bank AG has reported in a separate Schedule 13G under the Exchange Act, filed with the Commission, sole dispositive and voting power over 370,300 shares of Host Common Stock, and such shares are included in the number reported in this table. The principal business address of Dresdner RCM, RCM Limited and RCM General is Four Embarcadero Center, San Francisco, California 94111. The principal business address of Dresdner Bank AG is Jurgen Ponto-Platz 1, 60301 Frankfurt, Germany.
- (14) Represents shares of Host Common Stock held by FMR Corp. ("FMR") and its subsidiaries, Fidelity Management Trust Company ("FMT") and Fidelity Management & Research Company ("FM&R"). FMR has reported in a Schedule 13G under the Exchange Act, filed with the Commission, that FMR, through its control of FM&R and certain investment funds for which FM&R acts as an investment adviser, has sole power to dispose of 22,474,835 shares of Host Common Stock owned by such investment funds, including the 15,610,500 shares of Host Common Stock (or 7.64% of the total shares outstanding of Host Common Stock as of July 31, 1998) held by the Fidelity Magellan Fund. FMR has no power to vote or direct the voting of the shares of Host Common Stock owned by the investment funds, which power resides with the Board of Directors of such investment funds. FMR, through its control of FMT and certain institutional accounts for which FMT serves as investment manager, has sole dispositive power over 57,739 shares, the sole power to vote or direct the voting of 44,301 shares, and no power to vote or direct the voting of 13,438 shares of Host Common Stock owned by the institutional accounts. The principal business address for FMR, FMT and FM&R is 82 Devonshire Street, Boston, Massachusetts 02109.
- (15) Represents shares of Host Common Stock held by Southeastern Asset Management, Inc. ("SAM"). SAM has reported in a Schedule 13G under the Exchange Act, filed with the Commission, sole dispositive power over

21,730,700 shares and shared dispositive power over 14,968,300 shares. Of these shares, SAM has reported sole voting power over 18,338,100 shares, shared voting power over 14,968,300 shares and no power to vote 3,451,600 shares. The principal business address of SAM is 6075 Poplar Avenue, Suite 900, Memphis, Tennessee 38119.

#### DESCRIPTION OF HOST REIT CAPITAL STOCK

The summary description of the capital stock of Host REIT set forth below does not purport to be complete and is subject to and qualified in its entirety by reference to the Host REIT Charter and Host REIT Bylaws, to be effective upon completion of the Merger, attached as Exhibits B and C, respectively, to the Agreement, which is attached to this Proxy Statement/Prospectus as Appendix A.

#### GENERAL

The Host REIT Charter provides that the total number of shares of stock of all classes which Host REIT has authority to issue is 800,000,000 shares of stock, initially consisting of 750,000,000 shares of Host REIT Common Stock and 50,000,000 shares of Host REIT Preferred Stock. The Board of Directors is authorized, without a vote of stockholders, to classify or reclassify any unissued shares of capital stock and to establish the preferences and rights of any preferred or other class or series of capital stock to be issued. At November 13, 1998, 100 shares of Host REIT Common Stock were issued and outstanding.

## HOST REIT COMMON STOCK

Subject to the preferential rights of any other classes or series of shares of capital stock and to the provisions of the Host REIT Charter regarding restrictions on transfers of shares of capital stock, holders of Host REIT Common Stock are entitled to receive distributions if, as and when authorized and declared by the Board of Directors, out of assets legally available therefor and to share ratably in the assets of Host REIT legally available for distribution to its stockholders in the event of its liquidation, dissolution or winding-up after payment of, or adequate provision for, all known debts and liabilities of Host REIT. Host REIT currently intends to pay regular quarterly distributions. See "Distribution and Other Policies."

Subject to the provisions to be contained in the Host REIT Charter regarding restrictions on the transfer of shares of capital stock, each outstanding share of Host REIT Common Stock will entitle the holder to one vote on all matters submitted to a vote of stockholders, including the election of directors, and, except as provided with respect to any other class or series of shares of Host REIT capital stock, the holders of shares of Host REIT Common Stock will possess the exclusive voting power. There is no cumulative voting in the election of directors, which means that the holders of a majority of the outstanding Host REIT Common Stock can elect all of the directors then standing for election.

Holders of shares of Host REIT Common Stock will have no preferences, conversion, sinking fund, redemption rights or preemptive rights to subscribe for any securities of Host REIT. Subject to the provisions to be contained in the Host REIT Charter regarding restrictions on transfer of capital stock, shares of Host REIT Common Stock will have equal distribution, liquidation and other rights.

Under the MGCL, a Maryland corporation generally cannot dissolve, amend its charter, merge, consolidate, effect a share exchange or transfer its assets within the meaning of the MGCL unless approved by the Board of Directors and by stockholders holding at least two-thirds of the shares entitled to vote on the matter (unless a greater or lesser percentage (but not less than a majority of all the votes entitled to be cast) is set forth in the corporation's charter. Under the Host REIT Charter, any merger, consolidation, share exchange or transfer of its assets will need to be approved (i) by the Board of Directors in the manner provided in the MGCL and (ii) by stockholders to the extent required under the MGCL. The Host REIT Charter generally will provide for stockholder approval of such transactions by a two-thirds vote of all the votes entitled to be cast, except that any merger of Host REIT with or into a trust organized for the purpose of changing Host REIT's form of organization from a corporation to a trust will require the approval of stockholders of Host REIT by the affirmative vote only of a majority of all the votes entitled to be cast on the matter. In addition, under the MGCL, certain mergers may be accomplished without a vote of stockholders. For example, no stockholder vote is required for a merger of a subsidiary of a Maryland corporation into its parent, provided the parent owns at least 90 percent of the subsidiary. In addition, a merger will not need to be approved by stockholders of a Maryland successor corporation if the merger does not reclassify or change the outstanding shares or otherwise amend the charter, and the number

of shares to be issued or delivered in the merger is not more than 20 percent of the number of its shares of the same class or series outstanding immediately before the merger becomes effective. A share exchange will need to be approved by a Maryland successor only by its Board of Directors. Any amendments to the provisions contained in the Host REIT Charter relating to restrictions on transferability of stock, the classified Board and fixing the size of the Board within the range set forth in the Host REIT Charter, as well as the provisions relating to removal of directors, the filling of Board vacancies and the exclusive authority of the Board of Directors to amend the Bylaws will require the approval of the Board of Directors and stockholders by the affirmative vote of the holders of not less than two-thirds of the votes entitled to be cast on the matter. Other amendments to the Host REIT Charter may be effected by requisite action of the Board of Directors and approval by stockholders by the affirmative vote of not less than a majority of the votes entitled to be cast on the matter.

The Host REIT Charter will authorize the Board of Directors to reclassify any unissued shares of Host REIT Common Stock into other classes or series of capital stock, including preferred stock, and to establish the number of shares in each class or series and to set the preferences, conversion and other rights, voting powers, restrictions, limitations as to dividends or other distributions, qualifications or terms or conditions of redemption for each such class or series.

#### PREFERRED STOCK

The Host REIT Charter initially will authorize the Board of Directors to issue 50 million shares of Host REIT Preferred Stock and to classify or reclassify any unissued preferred shares into one or more classes or series of capital stock, including Host REIT Common Stock. Prior to issuance of shares of any class or series of stock other than Host REIT Common Stock, the Board of Directors is required, under the MGCL, to set, subject to the provisions of the Host REIT Charter regarding the restriction on transfer of capital stock, the terms, preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends or other distributions, qualifications and terms or conditions of redemption for each such class or series. Thus, the Board of Directors could authorize the issuance of preferred shares or other capital stock with terms and conditions which could have the effect of delaying, deferring or preventing a transaction or a change in control of Host REIT that might involve a premium price for holders of shares of Host REIT Common Stock or otherwise be in their best interest. As of the date hereof, no shares other than Host REIT Common Stock are outstanding, but Host REIT may issue preferred shares or other capital stock in the future, including as a result of the issuance of preferred stock by Host prior to the REIT Conversion. Although the Board of Directors has no intention at the present time of doing so (other than in connection with the proposed Stockholders Rights Plan), it could authorize Host REIT to issue a class or series of shares that could, depending upon the terms of such class or series, delay, defer or prevent a transaction or a change in control of Host REIT that might involve a premium price for holders of shares of Host REIT Common Stock or otherwise be in their best interest.

# POWER TO ISSUE ADDITIONAL HOST REIT COMMON STOCK AND PREFERRED STOCK

Host REIT believes that the power of the Board of Directors to issue additional authorized but unissued shares of Host REIT Common Stock or Host REIT Preferred Stock and to classify or reclassify unissued Host REIT Common Stock or Host REIT Preferred Stock and thereafter to cause Host REIT to issue such classified or reclassified shares of capital stock in one or more classes or series will provide Host REIT with increased flexibility in structuring possible future financings and acquisitions and in meeting other needs which might arise. The additional classes or series, as well as the Host REIT Common Stock, will be available for issuance without further action by Host REIT's stockholders, unless such action is required by applicable law or the rules of any stock exchange or automated quotation system on which Host REIT's securities may be listed or traded.

## RESTRICTIONS ON OWNERSHIP AND TRANSFER

For Host REIT to qualify as a REIT under the Code, no more than 50% in value of its outstanding shares of stock may be owned, actually or constructively, by five or fewer individuals (as defined in the Code to include certain entities) during the last half of a taxable year (other than the first year for which an election to be treated

as a REIT has been made) or during a proportionate part of a shorter taxable year. In addition, if Host REIT, or one or more owners (actually or constructively) of 10% or more of Host REIT, actually or constructively owns 10% or more of a tenant of Host REIT (or a tenant of any partnership in which Host REIT is a partner), the rent received by Host REIT (either directly or through any such partnership) from such tenant will not be qualifying income for purposes of the REIT gross income tests of the Code. A REIT's shares also must be beneficially owned by 100 or more persons during at least 335 days of a taxable year of twelve months or during a proportionate part of a shorter taxable year (other than the first year for which an election to be treated as a REIT has been made).

Primarily because the Board of Directors believes it is desirable for Host REIT to qualify as a REIT, the Host REIT Charter will provide that, subject to certain exceptions, no person or persons acting as a group may own, or be deemed to own by virtue of the attribution provisions of the Code, more than (i) 9.8% of the lesser of the number or value of shares of Host REIT Common Stock outstanding or (ii) 9.8% of the lesser of the number or value of the issued and outstanding preferred or other shares of any class or series of Host REIT stock, subject to (a) an exception for a holder of shares of Host REIT Common Stock solely by reason of the Merger in excess of the Ownership Limit so long as such holder would not own, directly or by attribution under the Code, more than 9.9% by value of the outstanding capital stock of Host REIT as of the Special Merger Ownership Limit Effective Time, and (b) a limitation on the application of the "group" limitation (but no other element of the Ownership Limit) to any "group" that otherwise would exceed the Ownership Limit at the Effective Time solely by reason of its status as a "group." The Ownership Limit will prohibit Marriott International and its subsidiaries and affiliates (including members of the Marriott family) from collectively owning shares of capital stock in excess of the Ownership Limit, but Host REIT's Board of Directors will grant an exception (pursuant to the applicable provisions of the Host REIT Charter) that will permit Marriott International to exercise its right to purchase up to 20% of each class of Host REIT's voting stock in connection with a change in control of Host REIT (but only in the event that (i) Marriott International and its subsidiaries and affiliates (including members of the Marriott family) do not own at such time or thereafter, directly and by attribution, 10% or more of Crestline or any of the Lessees and (ii) such ownership of Host REIT shares would not cause the Operating Partnership to be considered to own, directly or by attribution, 10% or more of Crestline or any of the Lessees). See "Certain Relationships and Related Transactions--Relationship Between Host and Marriott International." The ownership attribution rules under the Code are complex and may cause Host REIT Common Stock owned actually or constructively by a group of related individuals and/or entities to be owned constructively by one individual or entity. As a result, the acquisition of less than 9.8% of the Host REIT Common Stock (or the acquisition or ownership of an interest in an entity that owns, actually or constructively, Host REIT Common Stock) by an individual or entity, could, nevertheless cause that individual or entity, or another individual or entity, to own constructively in excess of 9.8% of the outstanding Host REIT Common Stock and thus subject such Host REIT Common Stock to the Ownership Limit. The Board of Directors may grant an exemption from the Ownership Limit with respect to one or more persons who would not be treated as "individuals" for purposes of the Code if it is satisfied, based upon an opinion of counsel and such other evidence as is satisfactory to the Board of Directors in its sole discretion, that such ownership will not cause a person who is an individual to be treated as owning Host REIT Common Stock in excess of the Ownership Limit, applying the applicable constructive Ownership rules, and will not otherwise jeopardize Host REIT's status as a REIT (for example, by causing any tenant of the Operating Partnership or the Partnerships (including but not limited to Crestline and the Lessees) to be considered a "related party tenant" for purposes of the REIT qualification rules). As a condition of such waiver, the Board of Directors may require undertakings or representations from the applicant with respect to preserving the REIT status of Host REIT. PURSUANT TO THE TERMS OF THE HOST REIT CHARTER, THE OWNERSHIP LIMIT WILL BECOME EFFECTIVE TO ALL HOST REIT STOCKHOLDERS AS OF THE EFFECTIVE TIME (SUBJECT TO THE LIMITED EXCEPTIONS DESCRIBED ABOVE).

The Board of Directors of Host REIT will have the authority to increase the Ownership Limit from time to time, but will not have the authority to do so to the extent that after giving effect to such increase, five beneficial owners of Host REIT Common Stock could beneficially own in the aggregate more than 49.5% of the outstanding Host REIT Common Stock.

The Host REIT Charter will further prohibit (i) any person from actually or constructively owning shares of Host REIT Common Stock that would result in Host REIT being "closely held" under Section 856(h) of the Code or otherwise cause Host REIT to fail to qualify as a REIT and (ii) any person from transferring shares of Host REIT capital stock if such transfer would result in shares Host REIT capital stock being owned by fewer than 100 persons.

Any person who acquires or attempts or intends to acquire actual or constructive ownership of shares of Host REIT capital stock that will or may violate any of the foregoing restrictions on transferability and ownership is required to give notice immediately to Host REIT and provide Host REIT with such other information as Host REIT may request in order to determine the effect of such transfer on Host REIT's status as a REIT.

If any purported transfer of shares of Host REIT capital stock or any other event would otherwise result in any person violating the Ownership Limit or the other restrictions in the Host REIT Charter, then any such purported transfer will be void and of no force or effect with respect to the purported transferee (the "Prohibited Transferee") as to that number of shares that exceeds the Ownership Limit (referred to as "excess shares") and the Prohibited Transferee shall acquire no right or interest (or, in the case of any event other than a purported transfer, the person or entity holding record title to any such shares in excess of the Ownership Limit (the "Prohibited ') shall cease to own any right or interest) in such excess shares. Any such excess shares described above will be transferred automatically, by operation of law, to a trust, the beneficiary of which will be a qualified charitable organization selected by Host REIT (the "Beneficiary"). Such automatic transfer shall be deemed to be effective as of the close of business on the Business Day (as defined in the Host REIT Charter) prior to the date of such violating transfer. Within 20 days of receiving notice from Host REIT of the transfer of shares to the trust, the trustee of the trust (who shall be designated by Host REIT and be unaffiliated with Host REIT and any Prohibited Transferee or Prohibited Owner) will be required to sell such excess shares to a person or entity who could own such shares without violating the Ownership Limit, and distribute to the Prohibited Transferee an amount equal to the lesser of the price paid by the Prohibited Transferee for such excess shares or the sales proceeds received by the trust for such excess shares. In the case of any excess shares resulting from any event other than a transfer, or from a transfer for no consideration (such as a gift), the trustee will be required to sell such excess shares to a qualified person or entity and distribute to the Prohibited Owner an amount equal to the lesser of the fair market value of such excess shares as of the date of such event or the sales proceeds received by the trust for such excess shares. In either case, any proceeds in excess of the amount distributable to the Prohibited Transferee or Prohibited Owner, as applicable, will be distributed to the Beneficiary. Prior to a sale of any such excess shares by the trust, the trustee will be entitled to receive, in trust for the Beneficiary, all dividends and other distributions paid by Host REIT with respect to such excess shares, and also will be entitled to exercise all voting rights with respect to such excess shares. Subject to Maryland law, effective as of the date that such shares have been transferred to the trust, the trustee shall have the authority (at the trustee's sole discretion and subject to applicable law) (i) to rescind as void any vote cast by a Prohibited Transferee prior to the discovery by Host REIT that such shares have been transferred to the trust and (ii) to recast such vote in accordance with the desires of the trustee acting for the benefit of the Beneficiary. However, if Host REIT has already taken irreversible corporate action, then the trustee shall not have the authority to rescind and recast such vote. Any dividend or other distribution paid to the Prohibited Transferee or Prohibited Owner (prior to the discovery by Host REIT that such shares had been automatically transferred to a trust as described above) will be required to be repaid to the trustee upon demand for distribution to the Beneficiary. The Host REIT Charter will provide that, if the transfer to the trust as described above is not automatically effective (for any reason) to prevent violation of the Ownership Limit, then the transfer of the excess shares will be void.

In addition, shares of Host REIT stock held in the trust shall be deemed to have been offered for sale to Host REIT, or its designee, at a price per share equal to the lesser of (i) the price per share in the transaction that resulted in such transfer to the trust (or, in the case of a devise or gift, the market value at the time of such devise or gift) and (ii) the market value of such shares on the date Host REIT, or its designee, accepts such offer. Host REIT will have the right to accept such offer until the trustee has sold the shares held in the trust.

Upon such a sale to Host REIT, the interest of the Beneficiary in the shares sold will terminate and the trustee will distribute the net proceeds of the sale to the Prohibited Owner.

TO AVOID THE ADVERSE EFFECTS OF THE OWNERSHIP LIMIT, ANY HOLDER OF HOST COMMON STOCK WHO WOULD OWN SHARES OF HOST REIT COMMON STOCK IN EXCESS OF THE OWNERSHIP LIMIT AT THE EFFECTIVE TIME SHOULD DISPOSE OF SUCH EXCESS SHARES PRIOR TO THE EFFECTIVE TIME. SEE "THE RESTRUCTURING TRANSACTIONS--TERMS OF THE MERGER." ANY SHARES OF HOST REIT COMMON STOCK HELD AT THE EFFECTIVE TIME AND NOT SUBSEQUENTLY TRANSFERRED PRIOR TO THE SPECIAL MERGER OWNERSHIP LIMIT EFFECTIVE TIME, OR ACQUIRED OR OTHERWISE HELD AT ANY TIME AFTER THE EFFECTIVE TIME IN VIOLATION OF THE OWNERSHIP LIMIT WILL BE TRANSFERRED AUTOMATICALLY TO A TRUST FOR THE BENEFIT OF A DESIGNATED CHARITABLE BENEFICIARY, AND THE PERSON WHO ACQUIRED SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL NOT BE ENTITLED TO ANY DISTRIBUTIONS THEREON OR TO VOTE SUCH EXCESS SHARES OF HOST REIT COMMON STOCK. THE HOLDER OF ANY SUCH EXCESS SHARES OF HOST REIT COMMON STOCK WILL RECEIVE THE LESSER OF THE VALUE OF SUCH EXCESS SHARES AS OF THE EFFECTIVE TIME OR THE CASH PROCEEDS OF THE SALE OF SUCH EXCESS SHARES OF HOST REIT COMMON STOCK BY THE TRUSTEE OF THE TRUST.

The foregoing restrictions on transferability and ownership will not apply if the Board of Directors determines that it is no longer in the best interests of Host REIT to attempt to qualify, or to continue to qualify, as a REIT

The Host REIT Charter also will provide that "Benefit Plan Investors" (as defined in the Host REIT Charter) may not hold, individually or in the aggregate, 25% or more of the value of any class or series of shares of capital stock of Host REIT to the extent such class or series does not constitute "Publicly Offered Securities" (as defined in the Host REIT Charter).

All certificates representing shares of Host REIT capital stock will bear a legend referring to the restrictions described above.

All persons who own, directly or by virtue of the attribution provisions of the Code, more than 5% (or such other percentage between 1/2 of 1% and 5% as provided in the rules and regulations promulgated under the Code) of the lesser of the number or value of the outstanding shares of Host REIT capital stock must give a written notice to the Company within 30 days after the end of each taxable year. In addition, each stockholder will, upon demand, be required to disclose to Host REIT in writing such information with respect to the direct, indirect and constructive ownership of shares of Host REIT capital stock as the Board of Directors deems reasonably necessary to comply with the provisions of the Code applicable to a REIT, to comply with the requirements of any taxing authority or governmental agency or to determine any such compliance.

These ownership limitations could have the effect of delaying, deferring or preventing a takeover or other transaction in which holders of some, or a majority, of Host REIT Common Stock might receive a premium for their Host REIT Common Stock over the then prevailing market price or which such holders might believe to be otherwise in their best interest.

# TRANSFER AGENT AND REGISTRAR

The transfer agent and registrar for Host REIT Common Stock will be First Chicago Trust Company of New York.

# CERTAIN PROVISIONS OF MARYLAND LAW AND THE HOST REIT CHARTER AND BYLAWS

The following summary of certain provisions of Maryland law and of the Host REIT Charter and Host REIT Bylaws does not purport to be complete and is subject to and qualified in its entirety by reference to Maryland law and the forms of the Host REIT Charter and Host REIT Bylaws to be effective upon completion of the Merger copies of which are attached as Exhibits B and C, respectively, to the Agreement attached to this Proxy Statement/Prospectus as Appendix A.

The Host REIT Charter and Host REIT Bylaws will contain certain provisions that could make more difficult an acquisition or change in control of the Company by means of a tender offer, a proxy contest or otherwise. These provisions are expected to discourage certain types of coercive takeover practices and inadequate takeover bids and to encourage persons seeking to acquire control of Host REIT to negotiate first with the Board of Directors. Host REIT believes that the benefits of these provisions outweigh the potential disadvantages of discouraging such proposals because, among other things, negotiation of such proposals might result in an improvement of their terms. See also "--Anti-Takeover Effect of Certain Provisions of Maryland Law and the Host REIT Charter and Bylaws."

NUMBER OF DIRECTORS; CLASSIFICATION AND REMOVAL OF BOARD OF DIRECTORS; OTHER PROVISIONS

The Host REIT Charter will provide that the Board of Directors initially will consist of eight members and may thereafter be increased or decreased in accordance with the Host REIT Bylaws, provided that the total number of directors may not be fewer than three nor more than thirteen. Pursuant to the Host REIT Bylaws, the number of directors shall be fixed by the Board of Directors within the limits set forth in the Host REIT Charter. Further, the Host REIT Charter will provide that the Board of Directors will be divided into three classes of directors, with each class to consist as nearly as possible of an equal number of directors. The term of office of the first class of directors will expire at the 1999 annual meeting of stockholders; the term of the second class of directors will expire at the 2000 annual meeting of stockholders; and the term of the third class of directors will expire at the 2001 annual meeting of stockholders. At each annual meeting of stockholders, the class of directors to be elected at such meeting will be elected for a three-year term, and the directors in the other two classes will continue in office. Because stockholders will have no right to cumulative voting for the election of directors, at each annual meeting of stockholders the holders of a majority of the outstanding Host REIT Common Stock will be able to elect all of the successors to the class of directors whose term expires at that meeting.

The Host REIT Charter also will provide that, except for any directors who may be elected by holders of a class or series of capital stock other than the Host REIT Common Stock, directors may be removed only for cause only by the affirmative vote of stockholders holding at least two-thirds of all the votes entitled to be cast for the election of directors. Vacancies on the Board of Directors may be filled by the concurring vote of a majority of the remaining directors (except in the event of a vacancy on the Board of Directors among the directors elected by a class or series of stock other than Host REIT Common Stock, in which case such vacancy may be filled by a majority of the remaining directors elected by that class or series, or by the stockholders of that class or series unless otherwise provided in the articles supplementary for that series) and, in the case of a vacancy resulting from the removal of a director by the stockholders, by the stockholders by at least two-thirds of all the votes entitled to be cast in the election of directors. Under Maryland law, directors may fill any vacancy only until the next annual meeting of stockholders. A vote of stockholders holding at least two-thirds of all the votes entitled to be cast thereon is required to amend, alter, change, repeal or adopt any provisions inconsistent with the foregoing classified board and director removal provisions. These provisions may make it more difficult and time-consuming to change majority control of the Board of Directors of Host REIT and, thus, may reduce the vulnerability of Host REIT to an unsolicited proposal for the takeover of Host REIT or the removal of incumbent management.

Because the Board of Directors will have the power, without a vote of stockholders, to classify or reclassify any unissued shares of capital stock and to establish the preferences and rights of any preferred or other class or series of stock to be issued, the Board of Directors may afford the holders of any class or series of senior shares of capital stock preferences, powers and rights, voting or otherwise, senior to the rights of holders of Host REIT

Common Stock. The issuance of any such senior shares of capital stock could have the effect of delaying, deferring or preventing a change in control of Host REIT.

See "The Restructuring Transactions--Limitation of Liability and Indemnification of Directors and Officers" for a description of the limitations on liability of directors and officers of Host REIT and the provisions for indemnification of directors and officers provided for under applicable Maryland law and the Host REIT Charter.

#### CHANGES IN CONTROL PURSUANT TO MARYLAND LAW

Maryland Business Combination Law. Under the MGCL, unless an exemption is available, certain "business combinations" (including certain issuances of equity securities) between a Maryland corporation and any Interested Stockholder or an affiliate of the Interested Stockholder, are prohibited for five years after the most recent date on which the Interested Stockholder becomes an Interested Stockholder. Thereafter, any such business combination must be recommended by the board of directors and approved by the affirmative vote of at least (i) 80% of all the votes entitled to be cast by holders of the outstanding shares of voting stock and (ii) two-thirds of the votes entitled to be cast by holders of voting stock other than voting stock held by the Interested Stockholder who will (or whose affiliate will) be a party to the business combination or any affiliate or associate of the Interested Stockholder, voting together as a single voting group, unless, among other conditions, the corporation's common stockholders receive a minimum price (as defined in the MGCL) for their shares and the consideration is received in cash or in the same form as previously paid by the Interested Stockholder for its shares. A business combination that is approved by the board of directors of a Maryland corporation at any time before an Interested Stockholder first becomes an Interested Stockholder is not subject to the special voting requirements. The Board of Directors of Host REIT has not opted out of the business combination provisions of the MGCL. Consequently, the five-year prohibition and the super-majority vote requirements will apply to a business combination involving Host REIT; however, as permitted by the MGCL, Host REIT's Board of Directors may elect to opt out of these provisions in the future. The Board of Directors of Host REIT has adopted a resolution exempting from the operation of the "business combination" statute the acquisition of shares by Marriott International (together with its successors and permitted assignees, if any) pursuant to the terms of the Marriott International Purchase Right as well as any other transactions involving Host REIT, Marriott International, J.W. Marriott, Jr. and Richard E. Marriott; provided that any such other transaction with Marriott International or its affiliates or associates that is not in the ordinary course of business or with J.W. Marriott, Jr. or Richard E. Marriott must be approved by a majority of the directors of Host REIT present at a meeting at which a quorum is present, including a majority of disinterested directors, in addition to any vote of stockholders required by other provisions of the MGCL.

Maryland Control Share Acquisition Law. Under the MGCL, "control shares" acquired in a "control share acquisition" have no voting rights except to the extent approved by a vote of two-thirds of the votes entitled to be cast on the matter, excluding shares owned by the acquiror, by officers or by directors who are employees of the corporation. "Control shares" are voting shares which, if aggregated with all other such shares previously acquired by the acquiror or in respect of which the acquiror is able to exercise or direct the exercise of voting power (except solely by virtue of a revocable proxy), would entitle the acquiror to exercise voting power in electing directors within one of the following ranges of voting power: (i) one-fifth or more but less than one-third, (ii) one-third or more but less than a majority or (iii) a majority or more of all voting power. Control shares do not include shares the acquiring person is then entitled to vote as a result of having previously obtained stockholder approval. A "control share acquisition" means the acquisition of control shares, subject to certain exceptions.

A person who has made or proposes to make a control share acquisition, upon satisfaction of certain conditions (including an undertaking to pay expenses), may compel the board of directors of the corporation to call a special meeting of stockholders to be held within 50 days of demand to consider the voting rights of the shares. If no request for a meeting is made, the corporation may itself present the question at any stockholders meeting.

If voting rights are not approved at the meeting or if the acquiring person does not deliver an acquiring person statement as required by the statute, then, subject to certain conditions and limitations, the corporation may redeem any or all of the control shares (except those for which voting rights have previously been approved)

for fair value determined, without regard to the absence of voting rights for the control shares, as of the date of the last control share acquisition by the acquiror or of any meeting of stockholders at which the voting rights of such shares are considered and not approved. If voting rights for control shares are approved at a stockholders meeting and the acquiror becomes entitled to vote a majority of the shares entitled to vote, all other stockholders may exercise appraisal rights. The fair value of the shares as determined for purposes of such appraisal rights may not be less than the highest price per share paid by the acquiror in the control share acquisition.

The control share acquisition statute does not apply to (i) shares acquired in a merger, consolidation or share exchange if the corporation is a party to the transaction or (ii) acquisitions approved or exempted by the charter or bylaws of the corporation. The Board of Directors of Host REIT has not opted out of the control share provisions of the MGCL but, as permitted by the MGCL, may elect to opt out of these provisions in the future. The Host REIT Bylaws will contain an exemption from the control share acquisition provisions for any shares acquired by Marriott International (together with its successors and permitted assignees, if any) pursuant to the Marriott International Purchase Right, to the extent it is exercised.

## ADVANCE NOTICE OF DIRECTOR NOMINATIONS AND NEW BUSINESS

The Host REIT Bylaws provide that (i) with respect to an annual meeting of stockholders, subject to the rights of holders of any class or series of stock having a preference over the Host REIT Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, nominations of persons for election to the Board of Directors and the proposal of business to be considered by stockholders may be made only (A) pursuant to Host REIT's notice of meeting, (B) by or at the direction of the Board of Directors or (C) by a stockholder who was a stockholder of record both at the time of giving notice provided for in the Host REIT Bylaws and at the time of the annual meeting, and who is entitled to vote at the meeting and has complied with the advance notice procedures set forth in the Host REIT Bylaws and (ii) with respect to special meetings of the stockholders, only the business specified in Host REIT's notice of meeting may be brought before the meeting of stockholders and except as otherwise provided for or fixed by or pursuant to the provisions of the Host REIT Charter relating to the rights of the holders of any class or series of stock having a preference over the Host REIT Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, nominations of persons for election to the Board of Directors may be made only (X) pursuant to Host REIT's notice of the meeting, (Y) by or at the direction of the Board of Directors or (Z) provided that the Board of Directors has determined that directors shall be elected at such meeting, by a stockholder who was a stockholder of record both at the time of giving notice provided for in the Host REIT Bylaws and at the time of the special meeting, and who is entitled to vote at the meeting and has complied with the advance notice provisions set forth in the Host REIT Bylaws. The advance notice provisions contained in the Host REIT Bylaws generally require nominations and new business proposals by stockholders to be delivered to the Secretary of Host REIT not later than the close of business on the 60th day nor earlier than the close of business on the 90th day before the date on which Host REIT first mailed its proxy materials for the prior year's annual meeting of stockholders.

MEETINGS OF STOCKHOLDERS; CALL OF SPECIAL MEETINGS; STOCKHOLDER ACTION IN LIEU OF MEETING BY UNANIMOUS CONSENT

The Host REIT Bylaws provide that annual meetings of stockholders shall be held on a date and at the time set by the Board of Directors during the month of May each year (commencing in May 1999). Special meetings of the stockholders may be called by the President or the Board of Directors. The Secretary of Host REIT also is required to call a special meeting of the stockholders on the written request of stockholders entitled to cast a majority of all the votes entitled to be cast at the meeting. The Host REIT Bylaws further provide that special stockholder meetings may be called by the holders of any class or series of stock having a preference over the Host REIT Common Stock as to dividends or upon liquidation in the manner specified in articles supplementary filed as part of the Host REIT Charter. Pursuant to the MGCL and the Host REIT Bylaws, any action required or permitted to be taken by the stockholders must be effected at a duly called annual or special meeting of stockholders and may not be effected by any consent in writing by stockholders, unless such consent is unanimous.

Pursuant to the Host REIT Charter, subject to the terms of any class or series of stock at the time outstanding, Host REIT may merge with or into another entity, may consolidate with one or more other entities, may participate in a share exchange or may transfer its assets within the meaning of the MGCL, but any such merger, consolidation, share exchange or transfer of assets must be approved (i) by the Board of Directors in the manner provided in the MGCL. In general, such transactions by a Maryland corporation, such as Host REIT, must first be approved by a majority of the entire Board of Directors and thereafter approved by stockholders by the affirmative vote of two-thirds of all the votes entitled to be cast on the matter (unless the charter provides for a greater or lesser stockholder vote but not less than a majority of the number of votes entitled to be cast on the matter). The Host REIT Charter generally will provide for stockholder approval of such transactions by a two-thirds vote of all votes entitled to be cast, except that any merger of Host REIT with or into a trust organized for the purpose of changing Host REIT's form of organization from a corporation to a trust will require the approval of stockholders of Host REIT by the affirmative vote only of a majority of all the votes entitled to be cast on the matter provided that (i) the stockholders of the trust immediately following the merger are the same as the stockholders of Host REIT immediately prior to the merger and (ii) the trust's declaration of trust contains amendment provisions substantially equivalent to those contained in specified provisions of the Host REIT Bylaws. Under the MGCL, certain mergers may be accomplished without a vote of stockholders. For example, no stockholder vote is required for a merger of a subsidiary of a Maryland corporation into its parent, provided the parent owns at least 90 percent of the subsidiary. In addition, a merger need not be approved by stockholders of a Maryland successor corporation if the merger does not reclassify or change the outstanding shares or otherwise amend the charter, and the number of shares to be issued or delivered in the merger is not more than 20% of the number of its shares of the same class or series outstanding immediately before the merger becomes effective. A share exchange need be approved by a Maryland successor only by its Board of Directors. Under the MGCL, a "transfer of assets" is defined to mean any sale, lease, exchange or other transfer of all or substantially all of the assets of the corporation but does not include (i) a transfer of assets by a corporation in the ordinary course of business actually conducted by it, (ii) a mortgage, pledge or creation of any other security interest in any or all of the assets of the corporation, whether or not in the ordinary course of its business, (iii) an exchange of shares of stock through voluntary action under any agreement with the stockholders, or (iv) a transfer of assets to one or more persons if all the equity interests of the person or persons are owned, directly or indirectly, by the corporation. Pursuant to the MGCL, a voluntary dissolution of Host REIT also would require the affirmative vote of two-thirds of all the votes entitled to be cast on the matter.

# AMENDMENTS TO THE HOST REIT CHARTER AND BYLAWS

Under the MGCL, in order to amend the charter, the board of directors first must adopt a resolution setting forth the proposed amendment and declaring its advisability and direct that the proposed amendment be submitted to stockholders for their consideration either at an annual or special meeting of stockholders. Thereafter, the proposed amendment must be approved by stockholders by the affirmative vote of two-thirds of all the votes entitled to be cast on the matter, unless a greater or lesser proportion of votes (but not less than a majority of all votes entitled to be cast) is specified in the charter. The provisions to be contained in the Host REIT Charter relating to restrictions on transferability of Host REIT Common Stock, the classified Board and fixing the size of the Board within the range set forth in the Host REIT Charter, as well as the provisions relating to removal of directors and the filling of Board vacancies may be amended only by a resolution adopted by the Board of Directors and approved at an annual or special meeting of the stockholders by the affirmative vote of the holders of not less than twothirds of the votes entitled to be cast on the matter. Other amendments to the Host REIT Charter generally may be effected by requisite action of the Board of Directors and approval by stockholders by the affirmative vote of not less than a majority of the votes entitled to be cast on the matter. As permitted under the MGCL, the Host REIT Bylaws provide that directors have the exclusive right to amend the Host REIT Bylaws. Amendment of this provision of the Charter also would require Board action and approval by holders of not less than two-thirds of all votes entitled to be cast on the matter.

ANTI-TAKEOVER EFFECT OF CERTAIN PROVISIONS OF MARYLAND LAW AND THE HOST REIT CHARTER AND BYLAWS

The business combination and control share provisions of the MGCL, the provisions of the Host REIT Charter on the classification of the Board of Directors, fixing the size of the Board of Directors within a specified

range and removal of directors, the provisions authorizing the Board of Directors, without a vote of stockholders, to classify or reclassify any unissued shares of stock into one or more classes or series of stock, the provisions relating to mergers, consolidations, share exchanges and transfers of assets, the provisions for amending certain provisions of the Host REIT Charter and for amending the Host REIT Bylaws, the advance notice provisions of the Host REIT Bylaws and the limitations on the ability of stockholders to call special meetings could delay, defer or prevent a transaction or a change of control of Host REIT that might involve a premium price for holders of Host REIT Common Stock or otherwise be in their best interests. The stock transfer restrictions that will be contained in the Host REIT Charter, which are intended to help Host REIT satisfy certain requirements under the Code to qualify as a REIT for federal income tax purposes, could also delay, defer or prevent a transaction or a change of control of Host REIT that might involve a premium price for holders of Host REIT Common Stock or otherwise be in their best interests.

#### MARRIOTT INTERNATIONAL PURCHASE RIGHT

In connection with Host's spin-off of Marriott International in 1993, Marriott International obtained the Marriott International Purchase Right, which entitles Marriott to purchase up to 20% of each class of Host's outstanding voting shares at the then fair market value upon the occurrence of certain change of control events involving Host. The Marriott International Purchase Right will continue in effect after the Merger (until June 2017), subject to certain limitations intended to help protect the REIT status of Host REIT. The Marriott International Purchase Right may have the effect of discouraging a takeover of Host REIT because any person considering acquiring a substantial or controlling block of Host REIT Common Stock will face the possibility that its ability to obtain or exercise control would be impaired or made more expensive by the exercise of the Marriott International Purchase Right.

## STOCKHOLDER RIGHTS PLAN

Host currently has in effect a stockholder rights plan pursuant to the Host Rights Agreement, and it has preferred stock purchase rights attached to its common stock pursuant to such rights plan. Prior to the completion of the Merger, the Board of Directors intends to adopt a Stockholder Rights Plan pursuant to a rights agreement (the "Host REIT Rights Agreement") to replace the existing Host Rights Agreement and declare a dividend of one Host REIT Right, consisting of one preferred stock purchase right for each outstanding share of Host REIT Common Stock. All shares of Host REIT Common Stock issued by Host REIT between the date of adoption of the Host REIT Rights Agreement and the Rights Distribution Date (as defined below), or the date, if any, on which the Host REIT Rights are redeemed would have Host REIT Rights attached to them. It is expected that the Host REIT Rights will expire ten years after adoption of the Host REIT Rights Agreement, unless earlier redeemed or exchanged. Each Host REIT Right, when exercisable, would entitle the holder to purchase upon payment of the purchase price a fraction of a share of a newly created series of junior participating preferred stock. Until a Host REIT Right is exercised, the holder thereof, as such, would have no rights as a stockholder of Host REIT including, without limitation, the right to vote or to receive dividends.

The Host REIT Rights Agreement is expected to provide that the Host REIT Rights initially attach to all certificates representing Host REIT Common Stock then outstanding. The Host REIT Rights would separate from the Host REIT Common Stock and a distribution of Host REIT Rights certificates would occur (a "Rights Distribution Date") upon the earlier to occur of (i) ten days following a public announcement that a person or group of affiliated or associated persons (an "Acquiring Person") has acquired, or obtained the right to acquire, beneficial ownership of 20% or more of the outstanding Host REIT Common Stock (the "Stock Acquisition Date") or (ii) ten business days (or such later date as the Board of Directors may determine) following the commencement of a tender offer or exchange offer, the consummation of which would result in the beneficial ownership by a person of 20% or more of the outstanding Host REIT Common Stock. For the purposes of determining the 20% threshold amount, shares of Host REIT Common Stock that can be acquired by Marriott International pursuant to the Marriott International Purchase Right will be deemed exempt shares under the Host REIT Rights Agreement and will not be included in any determination of the number of shares of Host REIT

Common Stock beneficially owned by Marriott International. The following additional shares of Host REIT Common Stock also will be deemed exempt shares under the Host REIT Rights Agreement: (i) shares of Host REIT Common Stock that are received pursuant to the Merger in exchange for shares of Host Common Stock which such person beneficially owned on February 3, 1989 and owned continuously thereafter until the Merger; (ii) certain shares of Host REIT Common Stock which were acquired by a person pursuant to a gift, bequest, inheritance or distribution from a trust or from a corporation controlled by such person where such shares of Host REIT Common Stock were exempt shares immediately prior to such acquisition; and (iii) shares of Host REIT Common Stock acquired by a person as a result of a stock dividend, stock distribution or other recapitalization, in respect to exempt shares only. Until the Rights Distribution Date, the Host REIT Rights would be evidenced by the Host REIT Common Stock certificates, and would be transferred with, and only with, the Host REIT Common Stock certificates.

It is expected that, if a Person becomes the beneficial owner of 20% or more of the then outstanding Host REIT Common Stock (except pursuant to an offer for all outstanding Host REIT Common Stock which the directors by a two-thirds vote determine to be fair to and otherwise in the best interests of Host REIT and its stockholders), each holder of a Host REIT Right would, after the end of a redemption period, have the right (subject to the Ownership Limit and the other ownership restrictions to be contained in the Host REIT Charter) to exercise the Host REIT Right by purchasing shares of Host REIT Common Stock (or, in certain circumstances, cash, property or other securities of Host REIT) having a value equal to two times the purchase price of the Host REIT Right, subject to the Ownership Limit.

If at any time following the Stock Acquisition Date, (i) Host REIT is acquired in a merger or other business combination transaction in which it is not the surviving corporation (other than a merger which follows an offer described in the preceding paragraph) or (ii) 50% or more of Host REIT's assets or earning power is sold or transferred, each holder of a Host REIT Right would have the right to receive, upon exercise, common shares of the acquiring company having a value equal to two times the purchase price of the Host REIT Right.

It is expected that, in general, the Board of Directors of Host REIT may redeem the Host REIT Rights at a nominal price per Host REIT Right at any time until ten days after an Acquiring Person has been identified as such. If the decision to redeem the Host REIT Rights occurs after a person becomes an Acquiring Person, the decision will require the concurrence of directors by a two-thirds vote.

The Host REIT Rights would have certain anti-takeover effects. The Host REIT Rights would cause substantial dilution to a person or group that attempts to acquire Host REIT. The Host REIT Rights, however, would not interfere with any merger or other business combination approved by the Board of Directors since the Board may, at its option, at any time prior to any person becoming an Acquiring Person, redeem all rights or amend the Host REIT Rights Agreement to exempt the person from the Host REIT Rights Agreement.

#### DESCRIPTION OF THE PARTNERSHIP AGREEMENT AND OP UNITS

The following is a summary of the material terms of the Partnership Agreement and the OP Units to be issued in connection with the Partnership Mergers and the REIT Conversion. Following the REIT Conversion, OP Units will be owned by the following groups: Host REIT; the Blackstone Entities; the Limited Partners who participate in the Partnership Mergers; and partners in four Private Partnerships participating in the Private Partnership Transactions. Following the REIT Conversion, Host REIT will be the sole general partner of the Operating Partnership, and following consummation of the other transactions comprising the REIT Conversion, the Operating Partnership will carry on the business formerly conducted by Host. See "The REIT Conversion."

#### **GENERAL**

Holders of OP Units (other than Host REIT in its capacity as general partner) will hold a limited partnership interest in the Operating Partnership, and all holders of OP Units (including Host REIT in its capacity as general partner) will be entitled to share in cash distributions from, and in the profits and losses of, the Operating Partnership. Because Host REIT will hold a number of OP Units equal to the number of shares of Host REIT Common Stock outstanding, each OP Unit generally will receive distributions in the same amount paid on each share of Host REIT Common Stock. See "Distribution and Other Policies--Distribution Policy."

Holders of OP Units will have the rights to which limited partners are entitled under the Partnership Agreement and the Delaware Revised Uniform Limited Partnership Act (the "Delaware Act"). The OP Units will not be listed on any exchange or quoted on any national market system. The Partnership Agreement imposes certain restrictions on the transfer of OP Units, as described below.

#### FORMATTON

The Operating Partnership was formed as a Delaware limited partnership under the Delaware Act on April 15, 1998. Upon the consummation of the REIT Conversion, Host REIT will be admitted to the Operating Partnership as the sole general partner of the Operating Partnership. Following the REIT Conversion, Host REIT is expected to hold a substantial amount of the interests in the Operating Partnership. Of the interests in the Operating Partnership allocated to Host REIT, a 0.1% interest in the Operating Partnership will be held by Host REIT as the general partner of the Operating Partnership, and the remaining OP Units allocated to Host REIT will be held by Host REIT as a limited partner in the Operating Partnership.

# PURPOSES, BUSINESS AND MANAGEMENT

The purpose of the Operating Partnership includes the conduct of any business that may be lawfully conducted by a limited partnership formed under the Delaware Act, except that the Partnership Agreement requires the business of the Operating Partnership to be conducted in such a manner that will permit Host REIT to qualify as a REIT under Section 856 of the Code, unless Host REIT ceases to qualify as a REIT for reasons other than the conduct of the business of the Operating Partnership. Subject to the foregoing limitation, the Operating Partnership may enter into partnerships, joint ventures or similar arrangements and may own interests directly or indirectly in any other entity.

Host REIT, as general partner of the Operating Partnership, has the exclusive power and authority to conduct the business of the Operating Partnership subject to the consent of the limited partners in certain limited circumstances discussed below. No limited partner may take part in the operation, management or control of the business of the Operating Partnership by virtue of being a holder of OP Units.

In particular, the limited partners expressly acknowledge in the Partnership Agreement that Host REIT is acting on behalf of the Operating Partnership's limited partners and Host REIT's stockholders collectively, and is under no obligation to consider the tax consequences to limited partners when making decisions for the benefit of the Operating Partnership. Host REIT intends to make decisions in its capacity as general partner of the

Operating Partnership so as to maximize the profitability of Host REIT and the Operating Partnership as a whole, independent of the tax effects on the limited partners. Host REIT and the Operating Partnership will have no liability to a limited partner as a result of any liabilities or damages incurred or suffered by, or benefits not derived by, a limited partner as a result of the act or omission of Host REIT as general partner of the Operating Partnership unless Host REIT acted, or failed to act, in bad faith and the act or omission was material to the loss, liability or benefit not derived.

# HOST REIT MAY NOT ENGAGE IN OTHER BUSINESSES; CONFLICTS OF INTEREST

Host REIT, as general partner, may not conduct any business other than the business of the Operating Partnership without the consent of limited partners holding Percentage Interests that are more than 50% of the aggregate Percentage Interests of the outstanding limited partnership interests entitled to vote thereon, excluding any such interests held by Host REIT. Other persons (including officers, directors, employees, agents and other affiliates of Host REIT) will not be prohibited under the Partnership Agreement from engaging in other business activities. However, Host REIT, on behalf of the Operating Partnership, has adopted certain policies regarding noncompetition provisions and avoidance of conflicts of interest. See "Distribution and Other Policies--Conflicts of Interest Policies." In addition, the Partnership Agreement does not prevent another person or entity that acquires control of Host REIT in the future from conducting other businesses or owning other assets, even though such businesses or assets may be ones that it would be in the best interests of the limited partners for the Operating Partnership to own.

#### DISTRIBUTIONS; ALLOCATIONS OF INCOME AND LOSS

The Partnership Agreement provides for the quarterly distribution of Available Cash (as determined in the manner provided in the Partnership Agreement), to Host REIT and the limited partners as holders of OP Units in proportion to their Percentage Interests. "Available Cash" is generally defined as net income plus depreciation and amortization and any reduction in reserves and minus interest and principal payments on debt, capital expenditures and any additions to reserves and other adjustments. At the time of the REIT Conversion, neither Host REIT nor the limited partners will be entitled to any preferential or disproportionate distributions of Available Cash (except to the extent that Host REIT receives preferred units in the Operating Partnership with economic rights that mirror the economic rights of any preferred stock that Host has outstanding at the time of the REIT Conversion).

## BORROWING BY THE OPERATING PARTNERSHIP

Host REIT is authorized to cause the Operating Partnership to borrow money and to issue and guarantee debt as it deems necessary for the conduct of the activities of the Operating Partnership, including financing and refinancing the assets of the Operating Partnership. Such debt may be secured by mortgages, deeds of trust, liens or encumbrances on properties of the Operating Partnership. Host REIT also may cause the Operating Partnership to borrow money to enable the Operating Partnership to make distributions, including distributions to holders of OP Units, including Host REIT, in an amount sufficient to permit Host REIT, as long as it qualifies as a REIT, to avoid the payment of any federal income tax. See "Distribution and Other Policies--Financing Policies."

## REIMBURSEMENT OF HOST REIT; TRANSACTIONS WITH HOST REIT AND ITS AFFILIATES

Host REIT will not receive any compensation for its services as general partner of the Operating Partnership. Host REIT, however, as a partner in the Operating Partnership, has the same right to allocations and distributions as other partners in the Operating Partnership. In addition, the Operating Partnership will pay all expenses relating to the Operating Partnership's organization, the REIT Conversion, the acquisition and ownership of its assets and its operations. The Operating Partnership will be responsible for and will pay (or reimburse) all expenses and liabilities of any nature that Host REIT may incur (including expenses and liabilities arising out of the REIT Conversion and expenses related to the ongoing operations of Host REIT and to the management and administration of any subsidiaries of Host REIT permitted under the Partnership Agreement).

The Operating Partnership also will be responsible for paying any and all taxes incurred by Host REIT, except that the Operating Partnership will not be responsible for any taxes that Host REIT would not have been required to pay if it qualified as a REIT for federal income tax purposes or any taxes imposed on Host REIT by reason of its failure to distribute to its stockholders an amount equal to its taxable income. The Operating Partnership, however, will not be responsible for expenses or liabilities incurred by Host REIT that are excluded from the scope of the indemnification provisions of the Partnership Agreement.

Except as expressly permitted by the Partnership Agreement, Host REIT and its affiliates will not engage in any transactions with the Operating Partnership, except on terms that are determined in good faith by the general partner to be fair and reasonable and no less favorable to the Operating Partnership than would be obtained from an unaffiliated third party.

#### LIABILITY OF HOST REIT AND LIMITED PARTNERS

Host REIT, as general partner of the Operating Partnership, will be liable for all general recourse obligations of the Operating Partnership to the extent not paid by the Operating Partnership. Host REIT will not be liable for the nonrecourse obligations of the Operating Partnership.

The limited partners of the Operating Partnership will not be required to make additional capital contributions to the Operating Partnership. Assuming that a limited partner does not take part in the control of the business of the Operating Partnership and otherwise acts in conformity with the provisions of the Partnership Agreement, the liability of a limited partner for obligations of the Operating Partnership under the Partnership Agreement and the Delaware Act will be limited, subject to certain exceptions, generally to the loss of such limited partner's investment in the Operating Partnership represented by his OP Units. Under the Delaware Act, a limited partner may not receive a distribution from the Operating Partnership if, at the time of the distribution and after giving effect thereto, the liabilities of the Operating Partnership, other than liabilities to parties on account of their interests in the Operating Partnership and liabilities for which recourse is limited to specified property of the Operating Partnership, exceed the fair value of the Operating Partnership's assets, other than the fair value of any property subject to nonrecourse liabilities of the Operating Partnership, but only to the extent of such liabilities. The Delaware Act provides that a limited partner who receives a distribution knowing at the time that it violates the foregoing prohibition is liable to the Operating Partnership for the amount of the distribution. Unless otherwise agreed, such a limited partner will not be liable for the return of such distribution after the expiration of three years from the date of such distribution.

The Operating Partnership expects to qualify to conduct business in various states in which the conduct of its business requires such qualification. Maintenance of limited liability may require compliance with certain legal requirements of those jurisdictions and certain other jurisdictions. Limitations on the liability of a limited partner for the obligations of a limited partnership have not been clearly established in many jurisdictions. Accordingly, if it were determined that the right, or exercise of the right by the limited partners, to make certain amendments to the Partnership Agreement or to take other action pursuant to the Partnership Agreement constituted "control" of the Operating Partnership's business for the purposes of the statutes of any relevant jurisdiction, the limited partners might be held personally liable for the Operating Partnership's obligations. The Operating Partnership will operate in a manner Host REIT deems reasonable, necessary and appropriate to preserve the limited liability of the limited partners.

# EXCULPATION AND INDEMNIFICATION OF HOST REIT

The Partnership Agreement generally provides that Host REIT, as general partner of the Operating Partnership, will incur no liability to the Operating Partnership or any limited partner for losses sustained, liabilities incurred or benefits not derived as a result of errors in judgment or mistakes of fact or law or of any act or omission, unless Host REIT acted, or failed to act, in bad faith and the act or omission was material to the loss, liability or benefit not derived. In addition, Host REIT is not responsible for any misconduct or negligence on the part of its agents, provided Host REIT appointed such agents in good faith. Host REIT may consult with

legal counsel, accountants, appraisers, management consultants, investment bankers and other consultants and advisors, and any action it takes or omits to take in reliance upon the opinion of such persons, as to matters that Host REIT reasonably believes to be within their professional or expert competence, shall be conclusively presumed to have been done or omitted in good faith and in accordance with such opinion.

The Partnership Agreement also provides for indemnification of Host REIT, the directors and officers of Host REIT and such other persons as Host REIT may from time to time designate against any judgments, penalties, fines, settlements and reasonable expenses actually incurred by such person in connection with the proceeding unless it is established that: (i) the act or omission of the indemnified person was material to the matter giving rise to the proceeding and either was committed in bad faith or was the result of active and deliberate dishonesty; (ii) the indemnified person actually received an improper personal benefit in money, property or services; or (iii) in the case of any criminal proceeding, the indemnified person had reasonable cause to believe that the act or omission was unlawful. The Operating Partnership is obligated to advance to an indemnified person reasonable expenses incurred or expected to be incurred by such indemnified person if such indemnified person certifies to the Operating Partnership that his conduct has met the standards for indemnification and that he will repay any amounts received if it is determined subsequently that his conduct did not meet such standards. To the extent that the indemnification provisions purport to include indemnification for liabilities arising under the Securities Act, in the opinion of the Commission, such indemnification is contrary to public policy and therefore unenforceable.

## SALES OF ASSETS

Under the Partnership Agreement, Host REIT generally has the exclusive authority to determine whether, when and on what terms the assets of the Operating Partnership (including the Hotels) will be sold. In addition, Host REIT is not required to take into account the tax consequences to the limited partners in deciding whether to cause the Operating Partnership to undertake a specific transaction. A sale of all or substantially all of the assets of the Operating Partnership (or a merger of the Operating Partnership with another entity) requires an affirmative vote of limited partners holding Percentage Interests that are more than 50% of the aggregate Percentage Interests of the outstanding limited partnership interests entitled to vote thereon (including Percentage Interests held by Host REIT).

# REMOVAL OR WITHDRAWAL OF HOST REIT; TRANSFER OF HOST REIT'S INTERESTS

The Partnership Agreement provides that the limited partners may not remove Host REIT as general partner of the Operating Partnership with or without cause (unless neither Host REIT nor its parent entity is a "public company," in which case Host REIT may be removed with or without cause by limited partners holding Percentage Interests that are more than 50% of the aggregate Percentage Interests of the outstanding limited partnership interests entitled to vote thereon, including any such interests held by the general partner). In addition, Host REIT may not transfer any of its interests as general or limited partner of the Operating Partnership or withdraw as a general partner, except, in each case, in connection with a merger or sale of all or substantially all of its assets, provided that (i) the limited partners of the Operating Partnership either will receive, or will have the right to receive, substantially the same consideration as holders of Host REIT Common Stock, (ii) following such merger or other consolidation, substantially all of the assets of the surviving entity consist of OP Units and (iii) such transaction has been approved by partners holding Percentage Interests that are more than 50% of the aggregate Percentage Interests of the outstanding interests in the Operating Partnership entitled to vote thereon (including any such interests held by Host REIT). Host REIT initially will hold a majority of the OP Units and thus would control the outcome of this vote. See "--Sales of Assets."

Although Host REIT cannot transfer its partnership interests except in a transaction in which substantially all of the assets of the surviving entity consist of OP Units, the Partnership Agreement does not prevent a transaction in which another entity acquires control (or all of the shares of capital stock) of Host REIT and that other entity owns assets and conducts businesses outside of the Operating Partnership.

181

CERTAIN VOTING RIGHTS OF HOLDERS OF OP UNITS DURING THE FIRST YEAR FOLLOWING THE EFFECTIVE DATE OF THE PARTNERSHIP MERGERS

During the first year following the effective date of the Partnership Mergers, if a vote of the stockholders of Host REIT is required, then (i) a sale of all or substantially all of the assets of the Operating Partnership, (ii) a merger involving the Operating Partnership and (iii) any issuance of OP Units in connection with an issuance of Host REIT Common Stock representing 20% or more of the outstanding Host REIT Common Stock which would require stockholder approval under the rules of the NYSE, would require the approval of a majority of all outstanding OP Units (or, in the case of clause (iii), a majority of the OP Units that are voted, provided that at least a majority of the OP Units are voted), including OP Units held by Host REIT, voting as a single class with Host REIT voting its OP Units in the same proportion as its stockholders vote. In addition, during the one-year period following the effective date of the Partnership Mergers, any taxable sale or sales of Hotels representing more than 10% of the aggregate appraised value (as determined by an independent hotel valuation and financial advisory firm) of the Hotels previously owned by any Partnership would require, in addition to any other approval requirements, the approval of a majority of all outstanding OP Units held by persons who formerly were Limited Partners of such Partnership, voting as a separate class.

## RESTRICTIONS ON TRANSFERS OF INTERESTS BY LIMITED PARTNERS

The Partnership Agreement provides that no limited partner shall, without the prior written consent of Host REIT (which consent may be withheld in Host REIT's sole and absolute discretion), sell, assign, distribute or otherwise transfer all or any portion of his interest in the Operating Partnership, except that a limited partner may transfer, without the consent of Host REIT, all or a portion of its limited partnership interest (i) in the case of a limited partner who is an individual, to a member of his immediate family, any trust formed for the benefit of himself and/or members of his immediate family, or any partnership, limited liability company, joint venture, corporation or other business entity comprised only of himself and/or members of his immediate family and entities the ownership interests in which are owned by or for the benefit of himself and/or members of his immediate family, (ii) in the case of a limited partner which is a trust, to the beneficiaries of such trust, (iii) in the case of a limited partner which is a partnership, limited liability company, joint venture, corporation or other business entity to which OP Units were transferred pursuant to (i) above, to its partners, owners, or stockholders, as the case may be, who are members of the immediate family of or are actually the person(s) who transferred OP Units to it pursuant to (i) above, (iv) in the case of a limited partner which acquired OP Units as of the closing of the Partnership Mergers and which is a partnership, limited liability company, joint venture, corporation or other business entity, to its partners, owners, stockholders or Affiliates thereof, as the case may be, or the Persons owning the beneficial interests in any of its partners, owners or stockholders or Affiliates thereof (it being understood that this clause (iv) will apply to all of each Person's partnership interests whether the OP Units relating thereto were acquired on the date hereof or hereafter), (v) in the case of a limited partner which is a partnership, limited liability company, joint venture, corporation or other business entity other than any of the foregoing described in clause (iii) or (iv), in accordance with the terms of any agreement between such limited partner and the Operating Partnership pursuant to which such partnership interest was issued, (vi) pursuant to a gift or other transfer without consideration, (vii) pursuant to applicable laws of descent or distribution, (viii) to another limited partner and (ix) pursuant to a grant of security interest or other encumbrance effected in a bona fide transaction or as a result of the exercise of remedies related thereto. All of the foregoing transfers are subject to the provisions of the Partnership Agreement which require compliance with securities laws, prohibit transfers affecting the tax status of the Operating Partnership or the qualification of Host REIT as a REIT for tax purposes, prohibit transfers to holders of nonrecourse liabilities of the Operating Partnership and are also subject to the rules on substitution of limited partners. In addition, Limited Partners will be permitted to dispose of their OP Units following the first anniversary of the effective date of the Partnership Mergers by exercising their Unit Redemption Right. See "--Unit Redemption Right" below.

The right of any permitted transferee of OP Units to become a substitute limited partner is subject to the consent of Host REIT, which consent Host REIT may withhold in its sole and absolute discretion. If Host REIT does not consent to the admission of a transferee of OP Units as a substitute limited partner, the transferee will

succeed to all economic rights and benefits attributable to such OP Units (including the Unit Redemption Right), but will not become a limited partner or possess any other rights of limited partners (including the right to vote).

Transfers of OP Units may be effected only by means of entries in the record of the Operating Partnership, and Host REIT will require evidence satisfactory to it of compliance with all transfer restrictions prior to recording any transfer.

#### UNIT REDEMPTION RIGHT

Subject to certain limitations, holders of OP Units (other than Host REIT) may exercise the Unit Redemption Right by providing notice to the Operating Partnership at any time commencing one year after the effective date of the Partnership Mergers. Unless Host REIT elects to assume and perform the Operating Partnership's obligation with respect to the Unit Redemption Right, as described below, the redeeming holder of OP Units will receive cash from the Operating Partnership in an amount equal to the market value of the OP Units to be redeemed. The market value of an OP Unit for this purpose will be equal to the average of the daily market price of a share of Host REIT Common Stock on the NYSE for the ten consecutive trading days before the day on which the redemption notice was given. The market price for each such trading day shall be the closing price, regular way, on such day, or if no such sales take place on such day, the average of the closing bid and asked prices on such day. In lieu of the Operating Partnership's acquiring the OP Units for cash, Host REIT will have the right (except as described below, if the Host REIT Common Stock are not publicly traded) to elect to acquire the OP Units directly from a holder of OP Units exercising the Unit Redemption Right, in exchange for either cash or Host REIT Common Stock, and, upon such acquisition, Host REIT will become the owner of such OP Units. In either case, acquisition of such OP Units by Host REIT will be treated as a sale of the OP Units to Host REIT for federal income tax purposes. Upon exercise of the Unit Redemption Right, the right of the holder of OP Units to receive distributions for the OP Units so redeemed or exchanged will cease. At least 1,000 OP Units (or all remaining OP Units owned by the holder of OP Units if less than 1,000 OP Units) must be redeemed each time the Unit Redemption Right is exercised. The redemption generally will occur on the tenth business day after notice of the exercise of the Unit Redemption Right by a holder of OP Units is given to the Operating Partnership, except that no redemption or exchange can occur if delivery of Host REIT Common Stock would be prohibited either under the provisions of the Host REIT Charter relating to restrictions on ownership and transfer of Host REIT Common Stock or under applicable federal or state securities laws as long as the Host REIT Common Stock are publicly traded. See "Description of Host REIT Capital Stock--Restrictions on Ownership and Transfer.'

In the event that the Host REIT Common Stock are not publicly traded but another entity whose stock is publicly traded owns more than 50% of the shares of capital stock of Host REIT (referred to as the "Parent Entity"), the Unit Redemption Right will be determined by reference to the publicly traded shares of the Parent Entity and the general partner will have the right to elect to acquire the OP Units to be redeemed for publicly traded stock of the Parent Entity. In the event that the Host REIT Common Stock are not publicly traded and there is no Parent Entity with publicly traded stock, the Unit Redemption Right would be based upon the fair market value of the Operating Partnership's assets at the time the Unit Redemption Right is exercised (as determined in good faith by Host REIT), and, unless otherwise agreed by the redeeming limited partner, Host REIT and the Operating Partnership would be obligated to satisfy the Unit Redemption Right in cash, payable on the thirtieth business day after notice to the Operating Partnership of exercise of the Unit Redemption Right.

# NO WITHDRAWAL BY LIMITED PARTNERS

No limited partner has the right to withdraw from or reduce his capital contribution to the Operating Partnership, except as a result of the redemption, exchange or transfer of OP Units pursuant to the terms of the Partnership Agreement.

## ISSUANCE OF LIMITED PARTNERSHIP INTERESTS

Host REIT is authorized, without the consent of the limited partners, to cause the Operating Partnership to issue additional OP Units to Host REIT, to the limited partners or to other persons for such consideration and

upon such terms and conditions as Host REIT deems appropriate. The Operating Partnership also may issue to any of the foregoing persons or entities partnership interests in different series or classes, which may be senior to the OP Units, including with respect to distributions and upon liquidation. If additional OP Units or partnership interests in different series or classes of equity securities are issued to Host REIT, then Host REIT must issue additional Host REIT Common Stock or securities having substantially similar rights to such partnership interests, and must contribute the proceeds received by Host REIT from such issuance to the Operating Partnership. Consideration for additional partnership interests may be cash or any property or other assets permitted by the Delaware Act. Except under circumstances involving the purchase of OP Units by Host REIT with proceeds from the exercise of Host REIT Rights, limited partner has preemptive, preferential or similar rights with respect to capital contributions to the Operating Partnership or the issuance or sale of any partnership interests therein.

### MEETINGS; VOTING

Meetings of the limited partners may be called only by Host REIT, on its own motion or upon written request of limited partners owning at least 25% of the then outstanding OP Units (including those held by Host REIT). Limited partners may vote either in person or by proxy at meetings. Any action that is required or permitted to be taken by the limited partners may be taken either at a meeting of the limited partners or without a meeting if consents in writing setting forth the action so taken are signed by limited partners holding Percentage Interests which are not less than the minimum Percentage Interest that would be necessary to authorize or take such action at a meeting of the limited partners at which all limited partners entitled to vote on such action were present. On matters as to which limited partners are entitled to vote, each limited partner (including Host REIT to the extent it holds limited partnership interests) will have a vote equal to its Percentage Interest. A transferee of OP Units who has not been admitted as a substituted limited partner with respect to such OP Units will have no voting rights with respect to such OP Units, even if such transferee holds other OP Units as to which it has been admitted as a limited partner. The Partnership Agreement does not provide for annual meetings of the limited partners, and Host REIT does not anticipate calling such meetings.

# AMENDMENT OF THE PARTNERSHIP AGREEMENT

Amendments to the Partnership Agreement may be proposed by Host REIT or by limited partners owning at least 25% of the then outstanding OP Units. Generally, the Partnership Agreement may be amended with the approval of Host REIT, as general partner, and limited partners (including Host REIT) holding Percentage Interests that are more than 50% of the aggregate Percentage Interests of the outstanding limited partnership interests entitled to vote thereon. Certain provisions regarding, among other things, the rights and duties of Host REIT as general partner (e.g., restrictions on Host REIT's power to conduct businesses other than owning OP Units, the dissolution of the Operating Partnership or the rights of limited partners), may not be amended without the approval of limited partners (excluding Host REIT) holding Percentage Interests that are more than 50% of the aggregate Percentage Interests of the outstanding limited partnership interests entitled to vote thereon. Notwithstanding the foregoing, Host REIT, as general partner, will have the power, without the consent of the limited partners, to amend the Partnership Agreement as may be required to (i) add to the obligations of Host REIT as general partner or surrender any right or power granted to Host REIT as general partner, (ii) reflect the admission, substitution, termination or withdrawal of partners in accordance with the terms of the Partnership Agreement, (iii) establish the rights, powers, duties and preferences of any additional partnership interests issued in accordance with the terms of the Partnership Agreement, (iv) reflect a change that does not materially adversely affect any limited partner, or cure any ambiguity, correct or supplement any provisions of the Partnership Agreement not inconsistent with law or with other provisions of the Partnership Agreement, or make other changes concerning matters under the Partnership Agreement that are not otherwise inconsistent with the Partnership Agreement or applicable law or (v) satisfy any requirements of federal, state or local law.

Certain amendments that would, among other things, (i) convert a limited partner's interest into a general partner's interest, (ii) modify the limited liability of a limited partner, (iii) alter the interest of a partner in profits or losses, or the rights to receive any distributions (except as permitted under the Partnership Agreement with respect to the admission of new partners or the issuance of additional OP Units (including partnership interests in a different class or series to the extent otherwise authorized under the Partnership Agreement), which actions

will have the effect of changing the percentage interests of the partners and thus altering their interests in profits, losses and distributions), (iv) amend the limited partners' right to transfer or (v) alter the Unit Redemption Right, must be approved by Host REIT and each limited partner that would be adversely affected by such amendment.

#### BOOKS AND REPORTS

Host REIT is required to keep the Operating Partnership's books and records at the principal office of the Operating Partnership. The books of the Operating Partnership are required to be maintained for financial and tax reporting purposes on an accrual basis. The limited partners will have the right to receive copies of the most recent Commission fillings by Host REIT and the Operating Partnership, the Operating Partnership's federal, state and local income tax returns, a list of limited partners, the Partnership Agreement, the partnership certificate and all amendments thereto and certain information about the capital contributions of the partners. Host REIT may keep confidential from the limited partners any information that Host REIT believes to be in the nature of trade secrets or other information the disclosure of which Host REIT in good faith believes is not in the best interests of the Operating Partnership or which the Operating Partnership is required by law or by agreements with unaffiliated third parties to keep confidential.

Host REIT will furnish to each limited partner, no later than the date on which Host REIT mails its annual report to its stockholders, an annual report containing financial statements of the Operating Partnership (or Host REIT, if it prepares consolidated financial statements including the Operating Partnership) for each fiscal year, including a balance sheet and statements of operations, cash flow, partners' equity and changes in financial position. The financial statements will be audited by a nationally recognized firm of independent public accountants selected by Host REIT. In addition, if and to the extent that Host REIT mails quarterly reports to its stockholders, Host REIT will furnish to each limited partner, no later than the date on which Host REIT mails such reports to its stockholders, a report containing unaudited financial statements of the Operating Partnership (or Host REIT, if it prepares consolidated financial statements including the Operating Partnership) as of the last day of the calendar quarter and such other information as may be required by applicable law or regulation or as Host REIT deems appropriate.

Host REIT will use reasonable efforts to furnish to each limited partner, within 90 days after the close of each taxable year, the tax information reasonably required by the limited partners for federal and state income tax reporting purposes.

# POWER OF ATTORNEY

Pursuant to the terms of the Partnership Agreement, each limited partner and each assignee appoints Host REIT, any liquidator and the authorized officers and attorneys-in-fact of each, as such limited partner's or assignee's attorney-in-fact to do the following: execute, swear to, acknowledge, deliver, file and record in the appropriate public offices various certificates, documents and other instruments (including, among other things, the Partnership Agreement and the certificate of limited partnership and all amendments or restatements thereof) that Host REIT deems appropriate or necessary to effectuate the terms or intent of the Partnership Agreement. The Partnership Agreement provides that such power of attorney is irrevocable, will survive the subsequent incapacity of any limited partner and the transfer of all or any portion of such limited partner's or assignee's OP Units and will extend to such limited partner's or assignee's heirs, successors, assigns and personal representatives.

# DISSOLUTION, WINDING UP AND TERMINATION

The Operating Partnership will continue until December 31, 2098, unless sooner dissolved and terminated. The Operating Partnership will be dissolved prior to the expiration of its term and its affairs wound up upon the occurrence of the earliest of: (i) the withdrawal of Host REIT as general partner without the permitted transfer of Host REIT's interest to a successor general partner (except in certain limited circumstances); (ii) an election to dissolve the Operating Partnership prior to December 31, 2058 made by Host REIT with the consent of the

limited partners who hold 90% of the OP Units (including OP Units held by Host REIT), (iii) the sale of all or substantially all of the Operating Partnership's assets and properties for cash or for marketable securities; (iv) the entry of a decree of judicial dissolution of the Operating Partnership pursuant to the provisions of the Delaware Act; (v) the entry of a final non-appealable order for relief in a bankruptcy proceeding of the general partner, or the entry of a final non-appealable judgment ruling that the general partner is bankrupt or insolvent (except that, in either such case, in certain circumstances the limited partners (other than Host REIT) may vote to continue the Operating Partnership and substitute a new general partner in place of Host REIT); or (vi) an election by Host REIT in its sole and absolute discretion on or after December 31, 2058. Upon dissolution, Host REIT, as general partner, or any liquidator will proceed to liquidate the assets of the Operating Partnership and apply the proceeds therefrom in the order of priority set forth in the Partnership Agreement.

### OWNERSHIP LIMITATION

In order to help the Operating Partnership avoid being treated as a corporation for federal income tax purposes, the Partnership Agreement expressly provides that no Person (other than Host REIT and the wholly owned subsidiaries (direct and indirect) thereof) or Persons acting as a group may own, actually or constructively, more than 4.9% by value of any class of interests in the Operating Partnership. The Partnership Agreement contains self-executing mechanisms intended to enforce this prohibition. For a description of the consequences of the Operating Partnership being treated as a corporation for federal income tax purposes, see "Federal Income Tax Consequences--Tax Aspects of Host REIT's Ownership of OP Units." As general partner of the Operating Partnership, Host REIT, in its sole and absolute discretion, may waive or modify this ownership limitation if it is satisfied that ownership in excess of this limit will not cause the Operating Partnership to be treated as a corporation for federal income tax purposes. Host has agreed to grant The Blackstone Group an exception to this prohibition subject to the condition that neither The Blackstone Group, nor any person or entity that would be considered to own OP Units owned by The Blackstone Group, may own, directly or by attribution, 9.8% or more of the stock of Crestline or the equity of any of the Lessees.

100

### ERISA CONSIDERATIONS

### STATUS OF HOST REIT AND THE OPERATING PARTNERSHIP UNDER ERISA

This section discusses the extent to which the fiduciary requirements of ERISA and the prohibited transaction provisions of ERISA and the Code would apply to Host REIT or the Operating Partnership because one or more investors in Host REIT or the Operating Partnership is an ERISA Plan or Other Plan.

If the underlying assets of Host REIT are deemed to be assets of an investing ERISA Plan and Other Plan ("Plan Assets"), (i) the prudence standards and other provisions of Part 4 of Title I of ERISA and the prohibited transaction provisions of ERISA and the Code would be applicable to any transactions involving Host REIT's assets, and (ii) persons who exercise any authority or control over Host REIT's assets, or who provide investment advice to Host REIT for a fee or other compensation, would be (for purposes of ERISA and the Code) fiduciaries of ERISA Plans and Other Plans that acquire Host REIT Common Stock of Host REIT. Similarly, if the underlying assets of the Operating Partnership are deemed to be Plan Assets, (i) the prudence standards and other provisions of Part 4 of Title I of ERISA and the prohibited transaction provisions of ERISA and the Code would be applicable to any transactions involving the Operating Partnership's assets, and (ii) persons who exercise any authority or control over the Operating Partnership's assets, or who provide investment advice to the Operating Partnership for a fee or other compensation, would be (for purposes of ERISA and the Code) fiduciaries of ERISA Plans and Other Plans that acquire Host REIT Common Stock of Host REIT.

The United States Department of Labor ("DOL"), which has certain administrative responsibility over ERISA Plans and certain Other Plans, has issued a regulation defining plan assets for certain purposes ("DOL Regulation"). The DOL Regulation generally provides that when an ERISA Plan or Other Plan acquires a security that is an equity interest in an entity and that security is neither a "publicly offered security" nor a security issued by an investment company registered under the 1940 Act, the assets of the ERISA Plan or Other Plan include both the equity interest and an undivided interest in each of the underlying assets of the entity, unless it is established either that the entity is an "operating company" (as defined in the DOL Regulation) or that equity participation in the entity by "benefit plan investors" is not "significant."

The DOL Regulation defines a "publicly offered security" as a security that is "widely-held," "freely transferable," and either part of a class of securities registered under the Exchange Act, or sold pursuant to an effective registration statement under the Securities Act (provided the securities are registered under the Exchange Act within 120 days, or such later time as may be allowed by the Commission (the "Registration Period"), after the end of the fiscal year of the issuer during which the offering occurred). Host REIT anticipates that the Host REIT Common Stock will be considered "publicly offered securities," and therefore the underlying assets of Host REIT would not be deemed to be Plan Assets of any ERISA Plan or Other Plan that invests in the Host REIT Common Stock.

The DOL Regulation defines "benefit plan investors" to consist of any employee benefit plan as defined in section 3(3) of ERISA, any Other Plan, or any entity whose underlying assets include Plan Assets by reason of an employee benefit plan's investment in the entity. Equity participation in an entity by "benefit plan investors" is deemed "significant" if, immediately after the most recent acquisition of any equity interest in the entity, 25% or more of the value of any class of equity interest is held by "benefit plan investors." Furthermore, for purposes of determining the percentage interest in a class of equity held by "benefit plan investors," the value of interests held by persons who either have discretionary authority or control over the entity's assets, or who provide investment advice for a fee, or are affiliates of such persons, is disregarded.

Based upon the value of the interests in the Partnerships owned by "benefit plan investors" relative to the value of the interests in the Partnerships owned by other Partnership investors, the percentage of the Operating Partnership that the Blackstone Entities will own immediately following the REIT Conversion, and certain other factors, the Operating Partnership believes that immediately following the REIT Conversion "benefit plan investors" will not own a "significant" percentage of OP Units, and, thus, the underlying assets of the Operating Partnership will not constitute Plan Assets of any ERISA Plan or Other Plan that owns OP Units. Furthermore, the Partnership Agreement will restrict issuance or transfer of OP Units if such issuance or transfer would result in benefit plan investors owning more than 25%.

### INTRODUCTION

The following discussion summarizes the federal income tax consequences reasonably anticipated to be material to a Host stockholder in connection with the Merger, the Initial E&P Distribution, the OP Contribution, and the ownership and disposition of Host REIT Common Stock. The following discussion is intended to address only those federal income tax consequences that are generally relevant to all Host stockholders. Accordingly, it does not discuss all aspects of federal income taxation that might be relevant to a specific Host stockholder in light of his particular investment or tax circumstances. Therefore, it is imperative that a Host stockholder review the following discussion and consult with his own tax advisors to determine the interaction of his individual tax situation with the anticipated tax consequences of the Merger, the Initial E&P Distribution, the OP Contribution, and the subsequent ownership and disposition of Host REIT Common Stock.

The following discussion provides general information only, is not exhaustive of all possible tax considerations and is not intended to be (and should not be construed as) tax advice. For example, this summary does not give a detailed description of any state, local or foreign tax considerations. In addition, the discussion does not purport to deal with all aspects of taxation that may be relevant to a Host stockholder subject to special treatment under the federal income tax laws, including, without limitation, insurance companies, financial institutions or broker-dealers, tax-exempt organizations (except to the extent discussed under the headings "Federal Income Tax Consequences of the Initial E&P Distribution--Tax Consequences of the Initial E&P Distribution to Tax-Exempt Host Stockholders" and "Taxation of Tax-Exempt Stockholders of Host REIT") or foreign corporations and persons who are not citizens or residents of the United States (except to the extent discussed under the headings "Federal Income Tax Consequences of the Initial E&P Distribution--Tax Consequences of the Initial E&P Distribution to Non-U.S. Stockholders" and "Taxation of Non-U.S. Stockholders").

The information in this section is based on the Code, current, temporary and proposed Treasury Regulations thereunder, the legislative history of the Code, current administrative interpretations and practices of the IRS (including its practices and policies as endorsed in private letter rulings, which are not binding on the IRS), and court decisions, all as of the date hereof. No assurance can be given that future legislation, Treasury Regulations, administrative interpretations and court decisions will not significantly change the current law or adversely affect existing interpretations of current law. Any such change could apply retroactively to transactions preceding the date of the change. No assurance can be provided that the statements set forth herein (which do not bind the IRS or the courts) will not be challenged by the IRS or will be sustained by a court if so challenged. Except as described under "--Federal Income Tax Consequences of the OP Contribution," neither Host nor Host REIT has requested or plans to request any rulings from the IRS concerning the tax consequences of the OP Contribution, the Merger, the Initial E&P Distribution, and the subsequent ownership and disposition of Host REIT Common Stock.

Hogan & Hartson L.L.P. ("Hogan & Hartson"), counsel to Host, Host REIT and the Operating Partnership, has delivered to Host and Host REIT an opinion to the effect that the discussion herein under the heading "Federal Income Tax Consequences," to the extent that it contains descriptions of applicable federal income tax law, is correct in all material respects (see Appendix C for this opinion). The opinion, however, does not purport to address the tax consequences of the Merger, the Initial E&P Distribution, the OP Contribution, and the subsequent ownership and disposition of Host REIT Common Stock to any particular Host stockholder in light of his particular circumstances, nor does it purport to predict whether, and the extent to which, future events and transactions, only some of which may be within the control of Host REIT will have a material adverse impact on the income tax positions of Host REIT stockholders. The opinion is based on the Code and Treasury Regulations in effect on the date hereof, current administrative interpretations and positions of the IRS and existing court decisions. No assurance can be given that future legislation, Treasury Regulations, administrative interpretations and court decisions will not significantly change the law or the above conclusions reached by counsel. In addition, any such change could apply retroactively to transactions preceding the date of change.

Moreover, opinions of counsel merely represent counsel's best judgment with respect to the probable outcome on the merits and are not binding on the IRS or the courts. Accordingly, even if there is no change in applicable law, no assurance can be provided that such opinion (which does not bind the IRS or the courts) will not be challenged by the IRS or will be sustained by a court if so challenged.

The following discussion is not intended to be, and should not be construed by a Host stockholder as, tax advice. THE SPECIFIC TAX ATTRIBUTES OF A PARTICULAR HOST STOCKHOLDER COULD HAVE A MATERIAL IMPACT ON THE TAX CONSEQUENCES OF THE MERGER, THE INITIAL E&P DISTRIBUTION, THE OP CONTRIBUTION AND/OR THE SUBSEQUENT OWNERSHIP AND DISPOSITION OF HOST REIT COMMON STOCK. THEREFORE, IT IS ESSENTIAL THAT EACH HOST STOCKHOLDER CONSULT WITH HIS OWN TAX ADVISORS WITH REGARD TO THE APPLICATION OF THE FEDERAL INCOME TAX LAWS TO SUCH STOCKHOLDER'S PERSONAL TAX SITUATION, AS WELL AS ANY TAX CONSEQUENCES ARISING UNDER THE LAWS OF ANY STATE, LOCAL OR FOREIGN TAXING JURISDICTION. THE FOLLOWING DISCUSSION IS NOT INTENDED AS A SUBSTITUTE FOR CAREFUL TAX PLANNING.

### FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER

The Merger is intended to qualify as a reorganization under Section 368(a) of the Code, and the federal income tax consequences summarized below are based on the assumption that the Merger will so qualify. Hogan & Hartson, counsel for Host and Host REIT, has delivered to Host and Host REIT an opinion to the effect that the Merger will be treated for federal income tax purposes as a reorganization under Section 368(a) of the Code (see Appendix C for this opinion). The opinion, however, does not purport to address the tax consequences of the Merger to any particular Host REIT stockholder in light of his particular circumstances. It is based on (i) certain factual assumptions and representations made by Host and Host REIT, and (ii) the Code and Treasury Regulations in effect on the date hereof, current administrative interpretations and positions of the IRS and existing court decisions. No assurance can be given that future legislation, Treasury Regulations, administrative interpretations and court decisions will not significantly change the law or the above conclusions reached by counsel. In addition, any such change could apply retroactively to transactions preceding the date of change. Moreover, opinions of counsel merely represent counsel's best judgment with respect to the probable outcome on the merits and are not binding on the IRS or the courts. Accordingly, even if there is no change in applicable law, no assurance can be provided that such opinion (which does not bind the IRS or the courts) will not be challenged by the IRS or will be sustained by a court if so challenged. No ruling will be sought from the IRS regarding the Merger.

Because the Merger will qualify as a reorganization under the Code, no gain or loss will be recognized by Host or Host REIT as a result of the Merger. In addition, no gain or loss will be recognized by a Host stockholder (other than certain Host stockholders who are not considered "U.S. persons" for purposes of the Code and who own (or have owned) in excess of 5% of Host Common Stock) upon the conversion of such holder's shares of Host Common Stock into Host REIT Common Stock pursuant to the Merger. The initial tax basis of the Host REIT Common Stock received by such a former Host stockholder pursuant to the Merger should be the same as the stockholder's adjusted tax basis in the shares of Host Common Stock being converted pursuant to the Merger (subject to any adjustment resulting from the Initial E&P Distribution). The holding period of the Host REIT Common Stock received by such a former Host stockholder pursuant to the Merger should include the holder's holding period with respect to the shares of Host Common Stock being converted pursuant to the Merger, provided that such Host Common Stock is held as a capital asset as of the Effective Time. See also "Federal Income Tax Consequences of the Initial E&P Distribution.'

Certain Host stockholders that are nonresident alien individuals, foreign corporations, foreign partnerships or foreign estates or trusts (collectively, "Non-U.S. Stockholders") could recognize gain or loss pursuant to the Merger. In particular, because Host is a "U.S. real property holding company" for the purposes of the Foreign Investment in Real Property Tax Act of 1980 ("FIRPTA"), but Host REIT anticipates qualifying as a "domestically controlled REIT" for purposes of FIRPTA, a Non-U.S. Stockholder who at any time during the

189

five-year period ending on the Effective Date owned more than 5% of the Host Common Stock would recognize gain to the extent that the fair market value of the Host REIT Common Stock he receives in the Merger exceeds his adjusted tax basis in his Host Common Stock. Any such Non-U.S. Stockholder would be subject to regular United States income tax with respect to such gain in the same manner as a "U.S. person" for purposes of the Code (subject to any applicable alternative minimum tax, a special alternative minimum tax in the case of nonresident alien individuals and the possible application of a 30% branch profits tax in the case of foreign corporations). See "Taxation of Non-U.S. Stockholders--Sales of Host REIT Common Stock" for a related discussion of the impact of the application of FIRPTA.

# FEDERAL INCOME TAX CONSEQUENCES OF THE INITIAL E&P DISTRIBUTION

As described more fully above in "The REIT Conversion--The Initial E&P Distribution," Host stockholders will receive in the Initial E&P Distribution Crestline common stock and cash or other consideration. Host currently contemplates that such other consideration will be in the form of the Special Dividend, which is payable, at each Host stockholder's election, in cash or Host Common Stock (or, if the Merger has occurred, Host REIT Common Stock). Although no assurance can be provided that such other consideration will take this form, the following discussion is based on the assumption that it will. The Initial E&P Distribution will assist Host REIT to meet the requirement that it distribute to its stockholders its accumulated pre-REIT E&P. For a detailed discussion of this requirement, see "--Federal Income Taxation of Host REIT Following the Merger--Requirements for Qualification," below.

Tax Consequences of the Initial E&P Distribution to U.S. Stockholders. As used herein, the term "U.S. Stockholder" means a Host stockholder who (for United States federal income tax purposes) is (i) a citizen or resident of the United States, (ii) a corporation, partnership or other entity created or organized in or under the laws of the United States or of any political subdivision thereof, (iii) an estate or trust the income of which is subject to United States federal income taxation regardless of its source, or (iv) a trust whose administration is subject to the primary supervision of a United States court and which has one or more United States persons who have the authority to control all substantial decisions of the trust.

A U.S. Stockholder will include the fair market value of his share of the Initial E&P Distribution in gross income as ordinary income to the extent that the Initial E&P Distribution is made out of the U.S. Stockholder's share of the portion of the current and accumulated E&P of Host and Host REIT allocable to the Initial E&P Distribution. The income attributable to the distribution of the Crestline common stock will be recognized when the Host stockholders become the owners of that stock, which is currently anticipated to occur in 1998. In addition, although the law is not entirely clear, Host intends to take the position that the income attributable to the Special Dividend will be recognized in 1999 when it is determined what a Host stockholder will receive (i.e., cash or stock, and if stock, the value thereof) and such cash or stock is payable to the Host stockholders. Under this approach, the fair market value of a U.S. Stockholder's share of the Initial E&P Distribution would equal the sum of (i) the fair market value (on the distribution date, which currently is expected to be in 1998) of the Crestline common stock that he receives in the Initial E&P Distribution, (ii) the amount of cash he receives in lieu of fractional shares of Crestline common stock, (iii) the amount of cash or the fair market value of the stock he receives pursuant to the Special Dividend, plus (iv) the amount of any other cash and fair market value of any other consideration he receives in the Initial E&P Distribution. Thus, insofar as a U.S. Stockholder receives cash pursuant to the Special Dividend, the income recognized would include the amount of the cash he so receives. With respect to a U.S. Stockholder that receives Host (or Host REIT) stock pursuant to the Special Dividend, the law is not entirely clear, but Host intends to take the position that the fair market value of the consideration he receives will equal the sum of (i) fair market value of the stock as of the end of the period during which the election under the Special Dividend can be made, plus (ii) the amount of any cash he receives in lieu of fractional shares of such stock. No assurance can be provided, however, that the IRS would not challenge this position regarding the taxation of the cash or stock received pursuant to the Special Dividend (for example, contending that the ability to make the election provided for under the Special Dividend is property that must be valued as of the record date of the Special Dividend or possibly the date on which notice of the Special Dividend is distributed to

the Host stockholders) and included in income on such date, which currently is expected to be in 1998), or that such an IRS challenge would not be sustained by a court. In determining whether the Initial E&P Distribution is made out of the current and accumulated E&P of Host and Host REIT, such E&P will be allocated pursuant to complex rules under the Code in the following order: first, to the portion of the Initial E&P Distribution that is distributed on the distribution date (which Host intends to take the position consists only of the Crestline common stock), and second, to the portion of the Initial E&P Distribution that is distributed after the distribution date (which, although the law is not entirely clear, Host intends to take the position will include the cash and stock distributed pursuant to the Special Dividend).

Host currently estimates that the Initial E&P Distribution will be approximately \$2.10 to \$2.50 per share of Host Common Stock (of which approximately \$0.80 to \$1.20 per share would be attributable to the Special Dividend, assuming that a Host stockholder were to elect to receive cash). No assurance can be provided, however, that this estimate will be accurate. If the value of the Initial E&P Distribution is greater than Host's current estimate (for example, because the value of the Crestline common stock received is higher than currently estimated or because a stockholder receives stock pursuant to the Special Dividend and the value of such stock is higher than the dollar amount of the cash that could have been received pursuant thereto), then a U.S. Stockholder will include such additional amount in gross income as ordinary income to the extent of the U.S. Stockholder's share of the portion of the current and accumulated E&P of Host and Host REIT allocable to the Initial E&P Distribution.

A U.S. Stockholder will have a basis in the Crestline common stock he receives in the Initial E&P Distribution equal to the fair market value of such stock on the distribution date and his holding period in such stock will begin on the day after the distribution date. Although the law is not entirely clear, consistent with the approach outlined in the preceding paragraph, Host intends to take the position that a U.S. Stockholder who elects to receive Host REIT Common Stock pursuant to the Special Dividend would have a basis in such stock equal to the fair market value of such stock on the day after the final day of the election period under the Special Dividend, and his holding period in such stock would begin on that day.

Host currently believes that the substantial majority, if not all, of the Initial E&P Distribution will be considered made out of the portion of Host and Host REIT's current and accumulated E&P allocable to the Initial E&P Distribution. The calculation of such E&P, however, will be very complex. The amount will include (i) the allocated consolidated E&P of Host (including each of its predecessors) accumulated from 1929, the first year that the predecessor of Host was a "C" corporation, through and including Host's 1998 taxable year, (ii) the current E&P of Host and Host REIT in 1998, and (iii) to the extent the Initial E&P Distribution occurs in 1999, the current E&P of Host and Host REIT in 1999. In addition, the calculation depends upon a number of factual and legal interpretations related to the activities and operations of Host and its corporate affiliates during its entire corporate existence and is subject to review and challenge by the IRS. There can be no assurance that Host and Host REIT's calculation of this E&P will be respected by the IRS or that a challenge to such calculations by the IRS would not be sustained by a court. Hogan & Hartson will express no opinion as to the amount of the current and accumulated E&P of Host and Host REIT.

To the extent that the sum of the amount of the Initial E&P Distribution received by a U.S. Stockholder exceeds his share of the portion of Host and Host REIT's current and accumulated E&P allocable to the Initial E&P Distribution, it will be treated first as a tax-free return of capital to such U.S. Stockholder, reducing the adjusted basis in his Host Common Stock by the amount of such excess (but not below zero) and then, if such basis is reduced to zero and there is remaining excess, as capital gain to the extent of such remaining excess (provided that such Host Stockholder has held the Host Common Stock as a capital asset).

Backup Withholding for the Initial E&P Distribution. Under the backup withholding rules, a U.S. Stockholder may be subject to backup withholding at the rate of 31% with respect to the Initial E&P Distribution paid unless such holder (a) is a corporation or comes within certain other exempt categories and, when required, demonstrates this fact or (b) provides a taxpayer identification number, certifies as to no loss of exemption from backup withholding and otherwise complies with applicable requirements of the backup withholding rules. See "--Taxation of Non-U.S. Stockholders."

Tax Consequences of the Initial E&P Distribution to Tax-Exempt Host Stockholders. Provided that a tax-exempt Host stockholder (except certain tax-exempt Host stockholders described in the following sentence) has not held its Host Common Stock as "debt financed property" within the meaning of the Code and such Host Common Stock is not otherwise used in a trade or business, the Initial E&P Distribution will not constitute unrelated business taxable income ("UBTI"). For a tax-exempt Host stockholder that is a social club, voluntary employee benefit association, supplemental unemployment benefit trust or qualified group legal services plan exempt from federal income taxation under Code Sections 501 (c)(7), (c)(9), (c)(17) or (c)(20), respectively, the Initial E&P Distribution will constitute UBTI unless the organization is properly able to deduct amounts set aside or placed in reserve for certain purposes so as to offset the income generated by its investment in Host. Such Host stockholders should consult their own tax advisors concerning these "set aside" and reserve requirements.

Tax Consequences of the Initial E&P Distribution to Non-U.S. Stockholders. A Non-U.S. Stockholder will treat the fair market value of the Initial E&P Distribution as ordinary income to the extent of the Non-U.S. Stockholder's share of the current and accumulated E&P of Host REIT allocable to the Initial E&P Distribution. For a discussion regarding the fair market value of the Initial E&P Distribution and the extent to which the Initial E&P Distribution will be attributable to such E&P, see "--Tax Consequences of the Initial E&P Distribution to U.S. Stockholders" above.

For Non-U.S. Stockholders, the Initial E&P Distribution will be subject to withholding of United States federal income tax on a gross basis (that is, without allowance of deductions) at a 30% rate or such lower rate as may be specified by an applicable income tax treaty, unless it is treated as effectively connected with the conduct by the Non-U.S. Stockholder of a United States trade or business. Certain certification and disclosure requirements must be satisfied to be exempt from withholding under the effectively connected income exemption. If the Initial E&P Distribution is effectively connected with such a trade or business, a Non-U.S. Stockholder will be subject to tax on the Initial E&P Distribution on a net basis (that is, after allowance of deductions) at graduated rates, in the same manner as U.S. Stockholders are taxed with respect to the Initial E&P Distribution, and generally will not be subject to withholding. A Non-U.S. Stockholder that is a corporation may also be subject to an additional branch profits tax on the Initial E&P Distribution at a 30% rate or such lower rate as may be specified by an applicable income tax treaty. See "--Taxation of Non-U.S. Stockholders," below.

To the extent that the sum of the fair market value of the Initial E&P Distribution received by a Non-U.S. Stockholder exceeds his share of the portion of Host and Host REIT's current and accumulated E&P allocable to the Initial E&P Distribution, it will be treated first as a tax-free return of capital to such Non-U.S. Stockholder, reducing the adjusted basis in his Host Common Stock (or Host REIT Common Stock, to the extent the Initial E&P Distribution occurs after the Merger) by the amount of such excess (but not below zero). If such basis is reduced to zero and there is remaining excess, the Initial E&P Distribution will give rise to gain from the sale or exchange of such stock. For a discussion of the federal income tax consequences of a sale or exchange of Host REIT Common Stock by a Non-U.S. Stockholder, see "--Taxation of Non-U.S. Stockholders--Sales of Host REIT Common Stock," below.

As a result of a legislative change made by the Small Business Job Protection Act of 1996, it appears that Host would be required to withhold 10% of any distribution in excess of Host and Host REIT's current and accumulated E&P. Consequently, although Host currently intends to withhold at a rate of 30% on the entire amount of any distribution (or a lower applicable treaty rate), to the extent that Host does not do so, any portion of the Initial E&P Distribution not subject to withholding at a rate of 30% (or a lower applicable treaty rate) will be subject to withholding at a rate of 10%. However, a Non-U.S. Stockholder may seek a refund of such amounts from the IRS if it subsequently determined that the Initial E&P Distribution was, in fact, in excess of such E&P, and the amount withheld exceeded the Non-U.S. Stockholder's United States tax liability, if any, with respect to the Initial E&P Distribution.

To the extent that withholding tax is owed with respect to any Non-U.S. Stockholder with respect to Crestline common stock received in the Initial E&P Distribution and any Host (or Host REIT) stock received pursuant to the Special Dividend, Host anticipates that the Transfer and Exchange Agent would sell in the market

a portion of the shares of stock distributable to such Non-U.S. Stockholder to pay such withholding taxes and the actual number of shares of Crestline common stock and any Host (or Host REIT) stock received pursuant to the Special Dividend exercise of the Election Rights would be net of any shares sold to pay such withholding taxes.

Tax Consequences to Host of the Distribution of Crestline Common Stock. Generally, a corporation that distributes property (other than its own stock) to its shareholders will recognize any gain realized on that distribution, but will not recognize any loss realized. Accordingly, Host would recognize gain on the distribution of the Crestline common stock to the extent that the fair market value of the Crestline common stock were to exceed Host's adjusted basis in the stock. Conversely, Host would not be able to recognize any loss if the fair market value of the Crestline common stock were to be less than Host's adjusted tax basis in that stock. Based on its current estimate of the fair market value of the Crestline common stock, which estimate is described above, Host currently believes that it will not recognize gain on the distribution of the Crestline common stock, but that it may realize a loss, which loss cannot be recognized for federal income tax purposes. Similar considerations apply with respect to the delivery of Crestline common stock to the Blackstone Entities. Host will not recognize any gain or loss in connection with any distribution of Host Common Stock, Host REIT Common Stock or cash as part of the Special Dividend.

## FEDERAL INCOME TAX CONSEQUENCES OF THE OP CONTRIBUTION

As described more fully above in "The Restructuring Transactions--The OP Contribution," Host will contribute certain of its assets (excluding the Crestline common stock, cash and other consideration to be distributed pursuant to the Initial E&P Distribution and certain other de minimis assets that cannot be contributed to the Operating Partnership) to the Operating Partnership in exchange for OP Units, preferred partnership interests in the Operating Partnership (if Host has outstanding any preferred stock), and the assumption by the Operating Partnership of all liabilities of Host (other than liabilities of Crestline). Host has obtained an opinion of Hogan & Hartson to the effect that the discussion in this section, to the extent it contains descriptions of federal income tax law, is correct in all material respects, but Host is not obtaining an opinion of outside counsel as to the actual tax consequences to it of the OP Contribution.

Section 721 of the Code provides that no gain or loss is recognized in the case of a contribution of property to a partnership in exchange for an interest in the partnership. Accordingly, Section 721 generally would apply to prevent the recognition of gain by Host in the OP Contribution. However, there are several potential exceptions to the availability of nonrecognition treatment under Section 721 that could result in Host recognizing gain in connection with the OP Contribution.

First, the assumption by the Operating Partnership of Host liabilities would result in a deemed distribution of cash to Host in an amount equal to the excess, if any, of such liabilities over Host's share of the liabilities of the Operating Partnership immediately after the OP Contribution and the Partnership Mergers. To the extent any such deemed distribution of cash exceeds Host's adjusted tax basis in the assets it transfers to the Operating Partnership, Host will recognize gain in the OP Contribution. However, based upon the amount of its liabilities outstanding and the adjusted tax basis of its various assets, Host currently believes that immediately after the OP Contribution and the Partnership Mergers it will not be deemed to have received a deemed cash distribution that would result in the recognition of gain in the OP Contribution.

Second, the OP Contribution would not be tax free to Host to the extent that it were treated as a "disguised sale" of all or a portion of the assets Host contributes to the Operating Partnership under the Code or Treasury Regulations. Section 707 of the Code and the Treasury Regulations thereunder (the "Disguised Sale Regulations") generally provide that, unless one of certain prescribed exceptions is applicable, a partner's contribution of property to a partnership and a simultaneous transfer of money or other consideration (other than an interest in the partnership) from the partnership to the partner will be treated as a sale, in whole or in part, of such property by the partner to the partnership. For purposes of these rules, assumptions of certain liabilities are treated as transfers of money or other property from the partnership to the partner which may give rise to a disguised sale. In addition, the Disguised Sale Regulations provide generally that transfers of money or other

consideration between a partnership and a partner that are made within two years of each other are presumed to be a sale unless the facts and circumstances clearly establish that either the transfers do not constitute a sale or an exception to disquised sale treatment applies.

One exception to "disguised sale" treatment in the Disguised Sale Regulations relates to the assumption of "qualified liabilities" in connection with a contribution of property to a partnership. A second exception to "disguised sale" treatment relates to distributions of "operating cash flow," which are presumed not to be a part of a sale of property to a partnership unless the facts and circumstances clearly establish that such distributions are part of a sale. A third exception to "disguised sale" treatment relates to distributions made to partners to reimburse them for certain "preformation" capital expenditures. The application of these and other exceptions to the Disguised Sale Regulations are highly complex and depend on a number of factual determinations and other outside events which may or may not occur, but Host currently believes that even if a portion of the OP Contribution were treated as a disguised sale under the Disguised Sale Regulations, Host would not recognize a material amount of gain in the OP Contribution.

Third, if the OP Contribution were considered to be a transfer to an "investment company," as defined in the Treasury Regulations, gain would be recognized on such transfer under Section 721 of the Code. In the case of a transfer of property to a partnership, the Code and the Treasury Regulations provide that such transfer would be treated as having been made to an investment company if the transfer results in a diversification of the interests of two or more transferors, and the transferee is a partnership more than 80% of the value of whose assets are "stock and securities." Although the OP Contribution (when considered together with the Partnership Mergers) will result in the diversification of the interests of two or more transferors, no significant portion of the Operating Partnership's assets will constitute "stock and securities" as defined in the Code. Accordingly, Host currently believes that the transfers to the Operating Partnership will not constitute transfers to an "investment company," as defined in the current Treasury Regulations. It should be noted in this regard, however, that the Operating Partnership might not meet the IRS's guidelines for obtaining an advance ruling with respect to this issue.

Although the application of the foregoing exceptions to nonrecognition treatment is extremely complex, Host currently believes that such application will not result in the recognition by Host of a material amount of gain in connection with the OP Contribution. However, even if Host does not recognize material gain in the OP Contribution, there are a variety of subsequent events and transactions including (i) the sale or other taxable disposition of appreciated assets contributed by Host to the Operating Partnership in the OP Contribution (including one or more of the Hotels contributed by Host), (ii) the refinancing or repayment of certain liabilities secured by one or more of the Hotels contributed to the Operating Partnership by Host in the OP Contribution, (iii) the issuance of additional OP Units, including in connection with the acquisition of additional properties by the Operating Partnership in exchange for OP Units or other equity interests in the Operating Partnership, (iv) an increase to the basis of one or more of the Hotels contributed to the Operating Partnership by Host in the OP Contribution resulting from capital expenditures and (v) the elimination over time of the disparity between the current tax basis of one or more of the Hotels contributed to the Operating Partnership by Host in the OP Contribution and the "book basis" of such assets (based upon their fair market value at the time of the OP Contribution) that could cause Host REIT to recognize part or all of the taxable gain that otherwise has been deferred pursuant to the OP Contribution. It should be noted that Host REIT, as general partner of the Operating Partnership, will have control over whether the Operating Partnership will undertake specific transactions.

Because of the numerous transactions that comprise the OP Contribution and the REIT Conversion and the inherently factual nature of a number of the critical determinations relevant to the tax consequences to Host of the OP Contribution and the REIT Conversion, Host has not sought (and likely could not obtain) an opinion of outside counsel as to the tax consequences to it (in terms of the actual amount of income and gain that would be recognized) of the OP Contribution and the other transactions comprising the REIT Conversion (other than the Merger). Host believes, based upon its knowledge of the relevant facts and advice that it has received from various professional advisers (including both attorneys and accountants on its staff and outside attorneys and

accountants) as to the applicable general principles of law, that it will not recognize a material amount of income and gain as a result of the OP Contribution and the other transactions comprising the REIT Conversion. Host, however, will recognize some income and gain in connection with these various transactions, and there can be no assurance that under the relevant facts and circumstances such income and gain would not be material. If Host were to recognize a material amount of income and gain as a result of these various transactions, the resulting income tax liability could be material to Host, and the increase in E&P resulting from such income and gain could materially increase the amount that Host must distribute to its stockholders in order to qualify as a REIT, all of which could have adverse consequences to Host REIT and its stockholders.

# FEDERAL INCOME TAXATION OF HOST REIT FOLLOWING THE MERGER

General. Host REIT plans to make an election to be taxed as a REIT under Sections 856 through 859 of the Code, effective for its first taxable year commencing after the REIT Conversion is completed (which Host currently expects to be the year beginning January 1, 1999, but which might not be until the year beginning January 1, 2000). (This election is made with the filing of Host REIT's federal income tax return for the first taxable year for which the election is effective.) Host REIT intends that, commencing with such first taxable year, it will be organized and will operate in such a manner as to qualify for taxation as a REIT under the Code, and Host REIT

intends to continue to operate in such a manner, but no assurance can be given that it in fact will continue to operate in such a manner so as to qualify or remain qualified.

The sections of the Code and the corresponding Treasury Regulations that govern the federal income tax treatment of a REIT and its stockholders are highly technical and complex. The following sets forth a summary of the material aspects of these rules, which summary, however, is qualified in its entirety by the applicable Code provisions, rules and regulations promulgated thereunder, and administrative and judicial interpretations thereof.

Hogan & Hartson has acted as counsel to Host REIT in connection with the REIT Conversion and Host REIT's election to be taxed as a REIT. Hogan & Hartson has provided to Host REIT and the Operating Partnership an opinion to the effect that, beginning with Host REIT's first full taxable year commencing after the REIT Conversion is completed, Host REIT will be organized in conformity with the requirements for qualification as a REIT, and its proposed method of operation will enable it to meet the requirements for qualification and taxation as a REIT under the Code (see Appendix D for this opinion). It must be emphasized that this opinion is conditioned upon completion of the REIT Conversion in the manner described in this Proxy Statement/Prospectus and upon certain assumptions and representations made by Host REIT and the Operating Partnership as to factual matters relating to the organization and operation of Host REIT, the Operating Partnership, the Partnerships and Private Partnerships, the Subsidiary Partnerships, the Non-Controlled Subsidiaries, the Host Employee/Charitable Trust and Crestline and the Lessees, including the economic and other terms of each Lease and the expectations of Host REIT and the Lessees with respect thereto. In addition, this opinion is based upon the factual representations of Host REIT concerning its business and properties as described in this Proxy Statement/Prospectus and assumes that all of the transactions and other actions described under "The Restructuring Transactions" and "The REIT Conversion" and related portions of this Proxy Statement/Prospectus and under "Federal Income Tax Consequences" are completed in a timely fashion (which means prior to the first day of the first taxable year for which Host REIT's election to be taxed as a REIT will be effective, except where the description thereof clearly contemplates completion of such action subsequent thereto). If Host REIT desires to be a REIT for the taxable year commencing January 1, 1999, that means all of the transactions and other actions referred to above would need to be completed prior to that date; otherwise, Host REIT might not qualify to be taxed as a REIT until its taxable year commencing January 1, 2000 (assuming that all such transactions and other actions are completed prior to that later date). Moreover, such qualification and taxation as a REIT depends upon Host REIT's ability to meet on an ongoing basis (through actual annual operating results, distribution levels and diversity of share ownership) the various qualification tests imposed under the Code discussed below, the results of which will not be reviewed by Hogan & Hartson. Accordingly, no assurance can be given that the actual results of Host REIT's operations for any particular taxable year will satisfy such requirements. Further, the anticipated income tax treatment described in this Proxy Statement/Prospectus may be changed, perhaps retroactively, by legislative, administrative or judicial action at any time. See "--Failure of Host REIT to Qualify as a REIT" below.

If Host REIT qualifies for taxation as a REIT, it generally will not be subject to federal corporate income taxes on its net income that is currently distributed to its stockholders. This treatment would substantially eliminate the "double taxation" (at the corporate and stockholder levels) that generally results from investment in a regular corporation. However, Host REIT will be subject to federal income tax as follows:

- 1. Host REIT will be taxed at regular corporate rates on any undistributed "REIT taxable income," including undistributed net capital gains (provided, however, that properly designated undistributed capital gains will effectively avoid taxation at the stockholder level). A REIT's "REIT taxable income" is the otherwise taxable income of the REIT subject to certain adjustments, including a deduction for dividends paid.
- 2. Under certain circumstances, Host REIT may be subject to the "alternative minimum tax" on its items of tax preference.
- 3. If Host REIT has (i) net income from the sale or other disposition of "foreclosure property" which is held primarily for sale to customers in the ordinary course of business or (ii) other nonqualifying income from foreclosure property, it will be subject to tax at the highest corporate rate on such income.

4. If Host REIT has net income from "prohibited transactions" (which are, in general, certain sales or other dispositions of property held primarily for sale to customers in the ordinary course of business other than foreclosure property), such income will be subject to a 100% tax.

- 5. If Host REIT should fail to satisfy the 75% gross income test or the 95% gross income test (as discussed below), but has nonetheless maintained its qualification as a REIT because certain other requirements have been met, it will be subject to a 100% tax on an amount equal to (a) the gross income attributable to the greater of the amount by which Host REIT fails the 75% or 95% test multiplied by (b) a fraction intended to reflect Host REIT's profitability.
- 6. If Host REIT should fail to distribute during each calendar year at least the sum of (i) 85% of its REIT ordinary income for such year, (ii) 95% of its REIT capital gain net income for such year and (iii) any undistributed taxable income from prior periods, Host REIT would be subject to a 4% excise tax on the excess of such required distribution over the sum of amounts actually distributed and amounts retained but with respect to which federal income tax was paid.
- 7. If Host REIT acquires any asset from a "C" corporation (i.e., generally a corporation subject to full corporate-level tax) in a transaction in which the basis of the asset in the hands of Host REIT is determined by reference to the basis of the asset in the hands of the "C" corporation (a "Built-In Gain Asset"), and Host REIT recognizes gain on the disposition of such asset during the ten-year period beginning on the date on which such asset was acquired by Host REIT (the "Recognition Period"), then, to the extent of the asset's "Built-In Gain" (i.e., the excess of (a) the fair market value of such asset over (b) Host REIT's adjusted basis in the asset, determined when Host REIT acquired the asset), such gain will be subject to tax at the highest regular corporate rate applicable.

Host REIT will own an indirect interest in appreciated assets that Host held before the REIT Conversion. Such appreciated assets will have a "carryover" basis and thus will be "Built-In Gain Assets" with respect to Host REIT. Under IRS Notice 88-19, unless Host REIT were to elect to be subject to corporate income tax on any Built-In Gain recognized with respect to such Built-In Gain Assets during the Recognition Period commencing on the first day of Host REIT's first taxable year as a REIT, Host would have to pay federal corporate income tax on the Built-In Gain at the time of the REIT Conversion. In connection with the REIT Conversion, Host REIT will make the election provided for in Notice 88-19 with respect to all of Host's assets that will be owned by the Operating Partnership subsequent to the REIT Conversion. As a result of this election, if such appreciated property is sold within the ten-year period following the REIT Conversion, Host REIT will generally be subject to regular corporate tax on that gain to the extent of the Built-In Gain in that property at the time of the REIT Conversion. The total amount of gain on which Host REIT can be taxed is limited to its net Built-In Gain (defined for these purposes as the excess of the aggregate fair market value of its assets at the time it became a REIT over the adjusted tax bases of those assets at that time) at the time of the REIT Conversion. This tax could be very material, however, and may result in the Operating Partnership and Host REIT seeking to avoid a taxable disposition of any significant assets owned by Host at the time of the REIT Conversion for the ten taxable years following the REIT Conversion (even though such disposition might otherwise be in the best interests of Host REIT).

Notwithstanding Host REIT's status as a REIT, it is likely that substantial deferred liabilities of Host will be recognized over the next ten years (including, but not limited to, tax liabilities attributable to Built-In Gain Assets and deferred tax liabilities attributable to taxable income for which neither Host REIT nor the Operating Partnership will receive corresponding cash). In addition, the IRS could assert substantial additional liabilities for taxes against Host for taxable years prior to the time Host REIT qualifies as a REIT. Under the terms of the REIT Conversion and the Partnership Agreement, the Operating Partnership will be responsible for paying (or reimbursing Host REIT for the payment of) all such tax liabilities as well as any other liabilities (including contingent liabilities and liabilities attributable to litigation that Host REIT may incur), whether such liabilities are incurred by reason of Host's activities prior to the REIT Conversion or the activities of Host REIT subsequent thereto.

The Operating Partnership will pay (or reimburse Host REIT for) all taxes incurred by Host REIT (except for taxes imposed on Host REIT by reason of its failure to qualify as a REIT or to distribute to its stockholders an amount equal to its "REIT taxable income," including net capital gains). This obligation by the Operating Partnership would include any federal corporate income tax imposed on Built-In Gain.

Requirements for Qualification. The Code defines a REIT as a corporation, trust or association (i) which is managed by one or more directors or trustees; (ii) the beneficial ownership of which is evidenced by transferable shares or by transferable certificates of beneficial interest; (iii) which would be taxable as a domestic corporation, but for Sections 856 through 859 of the Code; (iv) which is neither a financial institution nor an insurance company subject to certain provisions of the Code; (v) the beneficial ownership of which is held by 100 or more persons; (vi) during the last half of each taxable year not more than 50% in value of the outstanding stock of which is owned, actually or constructively, by five or fewer individuals (as defined in the Code to include certain entities) (the "not closely held" requirement); and (vii) which meets certain other tests, described below, regarding the nature of its income and assets.

The Code provides that conditions (i) to (iv), inclusive, must be met during the entire taxable year and that condition (v) must be met during at least 335  $\,$ days of a taxable year of twelve months, or during a proportionate part of a taxable year of less than twelve months. Conditions (v) and (vi) will not apply until after the first taxable year for which Host REIT makes the election to be taxed as a REIT. For purposes of conditions (v) and (vi), pension funds and certain other tax-exempt entities are treated as individuals, subject to a "look-through" exception in the case of condition (vi). Compliance with condition (v) shall be determined by disregarding the Ownership of Host REIT shares by any person(s) who: (1) acquired such Host REIT shares as a gift or bequest or pursuant to a legal separation or divorce; (2) is the estate of any person making such transfer to the estate; or (3) is a company established exclusively for the benefit of (or wholly-owned by) either the person making such transfer or a person described in (1) or (2). In connection with condition (vi), Host REIT is required to send annual letters to its stockholders requesting information regarding the actual ownership of its shares. If Host REIT complies with this requirement, and it does not know, or exercising reasonable diligence would not have known, whether it failed to meet condition (vi), then it will be treated as having met condition (vi). If Host REIT fails to send such annual letters, it will be required to pay either a \$25,000 penalty or, if the failure is intentional, a \$50,000 penalty. The IRS may require Host REIT, under those circumstances, to take further action to ascertain actual ownership of its shares, and failure to comply with such an additional requirement would result in an additional \$25,000 (or \$50,000) penalty. No penalty would be assessed in the first instance, however, if the failure to send the letters is due to reasonable cause and not to willful neglect.

Host REIT believes that it will meet conditions (i) through (iv). In addition, Host REIT believes that it will have outstanding (commencing with its first taxable year as a REIT) Host REIT Common Stock with sufficient diversity of ownership to allow it to satisfy conditions (v) and (vi). With respect to condition (vi), Host REIT intends to comply with the requirement that it send annual letters to its stockholders requesting information regarding the actual ownership of its shares. In addition, the Host REIT Charter will provide for restrictions regarding the transfer and ownership of Host REIT Common Stock, which restrictions are intended to assist Host REIT in continuing to satisfy the share ownership requirements described in (v) and (vi) above. Such ownership and transfer restrictions are described in "Description of Host REIT Capital Stock--Restrictions on Ownership and Transfer." These restrictions, together with compliance with the annual stockholder letter requirement described above, however, may not ensure that Host REIT will, in all cases, be able to satisfy the share ownership requirements described above. If Host REIT fails to satisfy such share ownership requirements, Host REIT's status as a REIT will terminate. See "--Failure of Host REIT to Qualify as a REIT."

A corporation may not elect to become a REIT unless its taxable year is the calendar year. Although Host currently has adopted a 52-53 week year ending on the Friday closest to January 1, Host REIT will adopt a calendar year taxable year in connection with the REIT Conversion.

Distribution of "Earnings and Profits" Attributable to "C" Corporation Taxable Years. In order to qualify as a REIT, Host REIT cannot have at the end of any taxable year any undistributed E&P that is attributable to a "C" corporation taxable year. A REIT has until the close of its first taxable year as a REIT in which it has non-REIT earnings and profits to distribute such E&P. In connection with the REIT Conversion, Host REIT intends to make the Initial E&P Distribution. The aggregate amount of the Initial E&P Distribution is currently expected to be in the range of \$525 to \$625 million and is intended to eliminate the substantial majority,

if not all, of Host's undistributed E&P. For a more detailed discussion of the Initial E&P Distribution, see "The REIT Conversion--The Initial E&P Distribution," above. For a discussion of the federal income tax consequences of the Initial E&P Distribution, see "--Federal Income Tax Consequences of the Initial E&P Distribution," above. To the extent, however, that Host has any such undistributed E&P at the time of the REIT Conversion (including E&P resulting from either transactions undertaken in contemplation of the REIT Conversion or the REIT Conversion itself), such E&P (the "Acquired Earnings") will carry over to Host REIT and will be treated as accumulated earnings and profits of a REIT attributable to non-REIT years. Host REIT will be required to distribute such E&P prior to the end of 1999 (the first taxable year for which the REIT election of Host REIT currently is expected to be effective, although such REIT election may not be effective until taxable year 2000). Failure to do so would result in disqualification of Host REIT as a REIT at least for taxable year 1999. If Host REIT should be so disqualified for taxable year 1999, subject to the satisfaction by Host REIT of certain "deficiency dividend" procedures described below in "--Annual Distribution Requirements Applicable to REITs" and assuming that Host REIT otherwise satisfies the requirements for qualification as a REIT, Host REIT should qualify as a REIT for taxable year 2000 and thereafter. Host REIT believes that the Initial E&P Distribution, together with any subsequent distributions of Acquired Earnings made prior to December 31, 1999, will be sufficient to distribute all of the Acquired Earnings as of December 31, 1999, but there are substantial uncertainties relating to the estimate of the Acquired Earnings, as described below, and the value of noncash consideration to be distributed as part of the Initial E&P Distribution, and, thus, there can be no assurance this requirement will be met.

The estimated amount of the Acquired Earnings will be based on the allocated consolidated earnings and profits of Host (including each of its predecessors) accumulated from 1929, the first year that the predecessor of Host was a "C" corporation, through and including Host's 1998 taxable year (and taking into account the allocation, as a matter of law, of 81% of Host's accumulated E&P to Marriott International on October 8, 1993 in connection with the Marriott International Distribution), determined based on the available tax returns and certain assumptions with respect to both such returns and other matters. The calculation of the Acquired Earnings, however, depends upon a number of factual and legal interpretations related to the activities and operations of Host and its corporate affiliates during its entire corporate existence and is subject to review and challenge by the IRS. There can be no assurance that the IRS will not examine the tax returns of Host and its affiliates for all years prior to and including the REIT Conversion and propose adjustments to increase their taxable income. The impact of such proposed adjustments, if any, may be material. If the IRS were to examine Host's calculation of its E&P (and thus the amount of Acquired Earnings, if any), the IRS can consider all taxable years of Host, its affiliates and its predecessors as open for review for purposes of such determination.

Hogan & Hartson will express no opinion as to the amount of E&P of Host and its predecessors and, accordingly, for purposes of its opinion as to the qualification of Host REIT as a REIT following the REIT Conversion, Hogan & Hartson is relying upon a representation from Host REIT that by the end of 1999 it will have eliminated all Acquired Earnings.

Qualified REIT Subsidiary. If Host REIT owns a corporate subsidiary that is a "qualified REIT subsidiary," that subsidiary will be disregarded for federal income tax purposes, and all assets, liabilities and items of income, deduction and credit of the subsidiary will be treated as assets, liabilities and items of Host REIT itself. Generally, a qualified REIT subsidiary is a corporation all of the capital stock of which is owned by one REIT. Host REIT anticipates owning one or more qualified REIT subsidiaries for purposes of holding de minimis indirect interests in the Hotel Partnerships. A "qualified REIT subsidiary" will not be subject to federal corporate income taxation, although it may be subject to state and local taxation in certain jurisdictions.

Ownership of Partnership Interests by a REIT. In the case of a REIT which is a partner in a partnership, Treasury Regulations provide that the REIT will be deemed to own its proportionate share of the assets of the partnership and will be deemed to be entitled to the income of the partnership attributable to such share. In addition, the character of the assets and gross income of the partnership shall retain the same character in the hands of the REIT for purposes of Section 856 of the Code (including satisfying the gross income tests and the

asset tests). Thus, Host REIT's proportionate share of the assets and items of income of the Operating Partnership (including the Operating Partnership's share of such items of the Partnerships, Private Partnerships and any other subsidiaries that are partnerships or LLCs) will be treated as assets and items of income of Host REIT for purposes of applying the requirements described herein. A summary of the rules governing the federal income taxation of partnerships and their partners is provided below in "--Tax Aspects of Host REIT's Ownership of OP Units." As the sole general partner of the Operating Partnership, Host REIT will have direct control over the Operating Partnership and indirect control over the Hotel Partnerships and the partnerships in which the Operating Partnership or the Hotel Partnerships have a controlling interest and intends to operate these entities consistent with the requirements for qualification of Host REIT as a REIT.

Income Tests Applicable to REITs. In order to maintain qualification as a REIT, Host REIT annually must satisfy two gross income requirements. First, at least 75% of Host REIT's gross income (excluding gross income from "prohibited transactions") for each taxable year must be derived directly or indirectly from investments relating to real property or mortgages on real property (including "rents from real property" and, in certain circumstances, interest) or from certain types of temporary investments. Second, at least 95% of Host REIT's gross income (excluding gross income from "prohibited transactions") for each taxable year must be derived from such real property investments, dividends, interest, certain hedging instruments and gain from the sale or disposition of stock or securities, including certain hedging instruments (or from any combination of the foregoing).

Rents paid pursuant to the Leases (together with dividends and interest received from the Non-Controlled Subsidiaries) will constitute substantially all of the gross income of Host REIT. Several conditions must be satisfied in order for rents received by Host REIT, including the rents received pursuant to the Leases, to qualify as "rents from real property" in satisfying the gross income requirements for a REIT described in the preceding paragraph. First, the amount of rent must not be based in whole or in part on the income or profits of any person. However, an amount received or accrued generally will not be excluded from the term "rents from real property" solely by reason of being based on a fixed percentage or percentages of receipts or sales. Second, rents received from a tenant will not qualify as "rents from real property" in satisfying the gross income tests if Host REIT, or an actual or constructive owner of 10% or more of Host REIT, actually or constructively owns 10% or more of such tenant (a "Related Party Tenant"). Third, if rent attributable to personal property, leased in connection with a lease of real property, is greater than 15% of the total rent received under the lease, then the portion of rent attributable to such personal property will not qualify as "rents from real property" (the "15% Personal Property Test").

Finally, if (i) Host REIT operates or manages a property or furnishes or renders services to the tenants at the property other than through an independent contractor from whom Host REIT derives no revenue, excluding for these purposes services "usually or customarily rendered" in connection with the rental of real property and not otherwise considered "rendered to the occupant," and (ii) the greater of (a) the income derived from such services or (b) 150% of the cost of providing such services (the "Impermissible Tenant Service Income") exceeds one percent of the total amount received by Host REIT with respect to the property (or, if such services are not available to all tenants at a property, possibly with respect to each tenant to whom the services are made available), then no amount received by Host REIT with respect to the property (or, where possibly applicable, such tenant) will qualify as "rents from real property." If the Impermissible Tenant Service Income is one percent or less of the total amount received by the REIT with respect to the property (or, where possibly applicable, such tenant), then only the Impermissible Tenant Service Income will not qualify as "rents from real property." To the extent that services other than those customarily furnished or rendered in connection with the rental of real property are rendered to the tenants of the property by an independent contractor, the cost of the services must be borne by the independent contractor.

The Operating Partnership and each Partnership and Private Partnership that owns Hotels (together with certain other subsidiaries of the Operating Partnership that may own Hotels) will enter into a Lease with a Lessee that is a Crestline subsidiary, pursuant to which the owner of such Hotels will lease the Hotels that it owns to the

199

Lessee for a term of years (ranging generally from seven to ten years, depending upon the particular Hotel) commencing on or before the January 1, 1999 (assuming the Merger occurs prior to that date, otherwise as soon as practicable following the distribution of the Crestline common stock as part of the Initial E&P Distribution). In addition, the Operating Partnership will lease to the Lessees, on similar terms, the Hotels contributed by the Blackstone Entities. Each Lease will provide for thirteen payments per annum of the specified Base Rent plus, to the extent that it would exceed the Base Rent, Percentage Rent, which Percentage Rent will be calculated based upon the gross sales of the Hotels subject to the particular Lease, plus certain other amounts. See "Business and Properties--The Leases."

Neither Host REIT nor the Operating Partnership intends to (i) provide any services to the Lessees with respect to the operation of the Hotels; (ii) charge rent for any Hotel that is based in whole or in part on the income or profits of any person (except by reason of being based on a percentage of receipts or sales, as described above); (iii) rent any Hotel to a Related Party Tenant (unless the Board of Directors determines in its discretion that the rent received from such Related Party Tenant is not material and will not jeopardize Host REIT's status as a REIT); or (iv) derive rental income attributable to personal property other than personal property leased in connection with the lease of real property, the amount of which is less than 15% of the total rent received under the lease (unless the Board of Directors determines in its discretion that the amount of such rent attributable to personal property is not material and will not jeopardize Host REIT's status as a REIT).

In order for the rent paid pursuant to the Leases to constitute "rents from real property," (i) the Lessees must not be regarded as Related Party Tenants; and (ii) the Leases must be respected as true leases for federal income tax purposes and not treated as service contracts, joint ventures or some other type of arrangement. A Lessee will be regarded as a Related Party Tenant only if Host REIT and/or one or more actual or constructive owners of 10% or more of Host REIT, actually or constructively, own 10% or more of such Lessee through an ownership interest in Crestline. In order to help preclude the Lessees from being regarded as Related Party Tenants, (i) the Articles of Incorporation of Crestline will expressly prohibit any person (or persons acting as a group), including Host REIT (and/or any 10% or greater stockholder of Host REIT), from owning more than 9.8% of the lesser of the number or value of the shares of capital stock of Crestline; (ii) the Host REIT Charter will expressly prohibit any person (or persons acting as a group) or entity from owning, actually and/or constructively, more than 9.8% of the lesser of the number or value of Host REIT Common Stock (subject to a limited exception for a holder of shares of Host REIT Common Stock solely by reason of the Merger in excess of the Ownership Limit so long as the holder thereof would not own, directly or by attribution under the Code, more than 9.9% in value of the outstanding shares of capital stock of Host REIT as of the Special Merger Ownership Limit Effective Time) or any other class or series of shares of Host REIT; and (iii) the Partnership Agreement will expressly prohibit any person (or persons acting as a group) or entity (other than Host REIT and Blackstone) from owning more than 4.9% by value of any class of interests in the Operating Partnership. Each of these prohibitions will contain self-executing enforcement mechanisms. Assuming that these prohibitions are enforced at all times (and no waivers thereto are granted), the Lessees should not be regarded as Related Party Tenants. There can be no assurance, however, that the ownership restrictions described herein will be enforced in accordance with their terms in all circumstances or otherwise will ensure, in all cases, that the Lessees will not be regarded as Related Party Tenants.

The determination of whether the Leases are true leases depends upon an analysis of all the surrounding facts and circumstances. In making such a determination, courts have considered a variety of factors, including the following: (i) the intent of the parties; (ii) the form of the agreement; (iii) the degree of control over the property that is retained by the property owner (e.g., whether the lessee has substantial control over the operation of the property or whether the lessee was required simply to use its best efforts to perform its obligations under the agreement); and (iv) the extent to which the property owner retains the risk of loss with respect to the property (e.g., whether the lessee bears the risk of increases in operating expenses or the risk of damage to the property) or the potential for economic gain (e.g., appreciation) with respect to the property.

In addition, Section 7701(e) of the Code provides that a contract that purports to be a service contract (or a partnership agreement) is treated instead as a lease of property if the contract is properly treated as such, taking

into account all relevant factors, including whether or not: (i) the service recipient is in physical possession of the property; (ii) the service recipient controls the property; (iii) the service recipient controls the property; (iii) the service recipient has a significant economic or possessory interest in the property (e.g., the property's use is likely to be dedicated to the service recipient for a substantial portion of the useful life of the property, the recipient shares the risk that the property will decline in value, the recipient shares in any appreciation in the value of the property, the recipient shares in savings in the property's operating costs or the recipient bears the risk of damage to or loss of the property); (iv) the service provider does not bear any risk of substantially diminished receipts or substantially increased expenditures if there is nonperformance under the contract; (v) the service provider does not use the property concurrently to provide significant services to entities unrelated to the service recipient; and (vi) the total contract price does not substantially exceed the rental value of the property for the contract period. Since the determination of whether a service contract should be treated as a lease is inherently factual, the presence or absence of any single factor may not be dispositive in every case.

The Leases have been structured with the intent to qualify as true leases for federal income tax purposes. For example, with respect to each Lease (i) the Operating Partnership (or, where appropriate, the applicable Partnership or Private Partnership or other lessor entity) and the Lessee intend for their relationship to be that of a lessor and lessee and such relationship is documented by a lease agreement, (ii) the Lessee has the right to exclusive possession and use and quiet enjoyment of the Hotels covered by the Lease during the term of the Lease, (iii) the Lessee bears the cost of, and will be responsible for, day-to-day maintenance and repair of the Hotels (other than the cost of certain capital expenditures), and will dictate (through the Managers, who work for the Lessees during the terms of the Leases) how the Hotels are operated and maintained, (iv) the Lessee bears all of the costs and expenses of operating the Hotels (including the cost of any inventory used in their operation) during the term of the Lease (other than the cost of certain furniture, fixtures and equipment, and certain capital expenditures), (v) the Lessee benefits from any savings (and bears the burdens of any increases) in the costs of operating the Hotels during the term of the Lease, (vi) in the event of damage or destruction to a Hotel, the Lessee is at economic risk because it will bear the economic burden of the loss in income from operation of the Hotels subject to the right, in certain circumstances, to terminate the Lease if the lessor does not restore the Hotel to its prior condition, (vii) the Lessee has indemnified the Operating Partnership (or, where appropriate, the applicable Partnership or Private Partnership or other lessor entity) against all liabilities imposed on the Operating Partnership (or, where appropriate, the applicable Partnership or Private Partnership or other lessor entity) during the term of the Lease by reason of (A) injury to persons or damage to property occurring at the Hotels or (B) the Lessee's use, management, maintenance or repair of the Hotels, (viii) the Lessee is obligated to pay, at a minimum, substantial Base Rent for the period of use of the Hotels under the Lease, (ix) the Lessee stands to incur substantial losses (or reap substantial gains) depending on how successfully it (through the Managers, who work for the Lessees during the terms of the Leases) operates the Hotels, and (x) Host REIT and the Operating Partnership believe that each Lessee reasonably expects to derive a meaningful profit, after expenses and taking into account the risks associated with the Lease, from the operation of the Hotels during the term of its Leases. Moreover, upon termination of a Lease, each Hotel is expected to have a remaining useful life equal to at least 20% of its expected useful life on the date of the consummation of the REIT Conversion, and a fair market value equal to at least 20% of its fair market value on the date of the consummation of the REIT Conversion.

Based upon representations made by Host REIT and the Operating Partnership (including representations as to the matters described in the previous paragraph and as to economic and other terms of the Leases), Hogan & Hartson, counsel to Host REIT, has provided to Host REIT an opinion letter to the effect that the Leases will be respected as leases for federal income tax purposes. As noted previously, however, an opinion of counsel does not bind the IRS or the courts. Moreover, there are no controlling Treasury Regulations, published IRS rulings or judicial decisions involving leases with terms substantially the same as the Leases that discuss whether such leases constitute true leases for federal income tax purposes and that this issue, more so than many others, is likely to turn on the surrounding facts and underlying economic circumstances. Hogan & Hartson is not expressing any opinion as to the economic terms of the Leases, the commercial reasonableness thereof, whether the actual economic relationships created thereby are such that the Leases will be respected for federal income

tax purposes or whether the rental and other terms of the Leases conform with normal business practice (and are not being used as a means to base the rent paid on the income or profits of the Lessees). Therefore, there can be no assurance that the IRS will not assert a contrary position or that such position will be sustained by a court if so challenged. If the Leases were recharacterized as service contracts or partnership agreements, rather than true leases, or disregarded altogether for tax purposes, all or part of the payments that the Operating Partnership receives from the Lessees would not be considered rent or would not otherwise satisfy the various requirements for qualification as "rents from real property." In that case, Host REIT very likely would not be able to satisfy either the 75% or 95% gross income tests and, as a result, would lose its REIT status.

As indicated above, "rents from real property" must not be based in whole or in part on the income or profits of any person. Payments made pursuant to the Leases should qualify as "rents from real property" since they will be based on either a fixed dollar amount (i.e., Base Rent) or specified percentages of gross sales (i.e., Percentage Rents) which percentages will be fixed at the time the Leases are entered into. The foregoing assumes that the Leases (i) are not renegotiated during their term in a manner that has the effect of basing either Percentage Rent or Base Rent on income or profits and (ii) are not in reality used as a means of basing rent on income or profits. More generally, the rent payable under the Leases would not qualify as "rents from real property" if, considering the Leases and all the surrounding circumstances, the arrangement does not conform with normal business practice, but is in reality used as a means of basing rent on income or profits. Because each of the Base Rent and the Percentage Rent will be based on fixed dollar amounts and fixed percentages of the gross sales of each Hotel that are established in the Leases. In addition, Host REIT has represented that (i) the percentages will not be renegotiated during the terms of the Leases in a manner that has the effect of basing rent on income or profits and (ii) the rental provisions and other terms of the Leases conform with normal business practice and are not intended to be used as a means of basing rent on income or profits, the rent payable under the Leases should not be considered based in whole or in part on the income or profits of any person. Furthermore, Host REIT has represented that, with respect to other properties that it acquires in the future, it will not charge rent for any property that is based in whole or in part on the income or profits of any person (except by reason of being based on a fixed percentage of gross revenues, as described above).

Host REIT may lease certain items of personal property to the Lessees in connection with the Leases. The 15% Personal Property Test provides that if a lease provides for the rental of both real and personal property and the portion of the rent attributable to personal property is 15% or less of the total rent due under the lease, then all rent paid pursuant to such lease qualifies as "rent from real property." If, however, a lease provides for the rental of both real and personal property, and the portion of the rent attributable to personal property exceeds 15% of the total rent due under the lease, then the portion of the rent that is attributable to personal property does not qualify as "rent from real property." The amount of rent attributable to personal property is that amount which bears the same ratio to total rent for the taxable year as the average of the adjusted tax bases of the personal property at the beginning and end of the year bears to the average of the aggregate adjusted tax bases of both the real and personal property at the beginning and end of such year. Host REIT has represented that, with respect to each Lease that includes a lease of items of personal property, the amount of rent attributable to personal property with respect to such Lease, determined as set forth above, will not exceed 15% of the total rent due under the Lease (except for several Leases where the rent attributable to personal property, which would constitute non-qualifying income for purposes of the 75% and 95% gross income tests, would not be material relative to the overall gross income of Host REIT). Each Lease permits the Operating Partnership to take certain measures, including requiring the Lessee to purchase certain furniture, fixtures and equipment or to lease such property from a third party (including a Non-Controlled Subsidiary), if necessary to ensure that all of the rent attributable to personal property with respect to such Lease will qualify as "rent from real property." In order to protect Host REIT's ability to qualify as a REIT, the Operating Partnership, in connection with the REIT Conversion, will sell to a Non-Controlled Subsidiary substantial personal property associated with a number of Hotels currently owned by Host or being acquired in connection with the REIT Conversion. The Non-Controlled Subsidiary will separately lease all such personal property directly to the applicable Lessee and will receive rental payments which Host REIT believes represent the fair rental value of such personal property directly from the Lessees. If such arrangements were not respected for tax purposes for any reason, Host REIT likely would not qualify as a REIT.

If any of the Hotels were to be operated directly by the Operating Partnership or a Partnership or Private Partnership as a result of a default by a Lessee under the applicable Lease, such Hotel would constitute foreclosure property until the close of the third tax year following the tax year in which it was acquired (or for up to an additional three years if an extension is granted by the IRS), provided that (i) the operating entity conducts operations through an independent contractor (which might, but would not necessarily in all circumstances, include Marriott International and its subsidiaries) within 90 days after the date the Hotel is acquired as the result of a default by a Lessee, (ii) the operating entity does not undertake any construction on the foreclosed property other than completion of improvements that were more than 10% complete before default became imminent, and (iii) foreclosure was not regarded as foreseeable at the time the applicable Partnership or Private Partnership entered into such Leases. For as long as any of these Hotels constitute foreclosure property, the income from the Hotels would be subject to tax at the maximum corporate rates, but it would qualify under the 75% and 95% gross income tests. However, if any of these Hotels does not constitute foreclosure property at any time in the future, income earned from the disposition or operation of such property will not qualify under the 75% and 95% gross income tests.

"Interest" generally will not qualify under the 75% or 95% gross income tests if it depends in whole or in part on the income or profits of any person. However, interest will not fail to so qualify solely by reason of being based upon a fixed percentage or percentages of receipts or sales. Host REIT does not expect to derive significant amounts of interest that will not qualify under the 75% and 95% gross income tests.

The Non-Controlled Subsidiaries will hold various assets contributed by Host and its subsidiaries to the Operating Partnership, the ownership of which by the Operating Partnership might jeopardize Host REIT's status as a REIT. These assets primarily will consist of partnership or other interests in Hotels that are not leased and certain foreign hotels in which Host owns interests. In addition, as described above, the Operating Partnership will sell to a Non-Controlled Subsidiary approximately \$180 million in value of personal property associated with certain Hotels, in order to facilitate Host REIT's compliance with the 15% Personal Property Test. The Operating Partnership will own 100% of the nonvoting stock of each Non-Controlled Subsidiary but none of the voting stock (or control) of that Non-Controlled Subsidiary. Each Non-Controlled Subsidiary is taxable as a regular "C" corporation. The Operating Partnership's share of any dividends received from a Non-Controlled Subsidiary should qualify for purposes of the 95% gross income test, but not for purposes of the 75% gross income test. The Operating Partnership does not anticipate that it will receive sufficient dividends from the Non-Controlled Subsidiaries to cause it to exceed the limit on non-qualifying income under the 75% gross income test.

Given the magnitude and scope of Host's existing operations, Host REIT inevitably will have some gross income from various sources (including, but not limited to, "safe harbor" leases, the operation of the Hotel in Sacramento, minority partnership interests in partnerships that own hotels that are not leased under leases that produce rents qualifying as "rents from real property" and rent attributable to personal property at a few Hotels that does not satisfy the 15% Personal Property Test) that fails to constitute qualifying income for purposes of one or both of the 75% or 95% gross income tests. Host REIT, however, believes that, even taking into account the anticipated sources of non-qualifying income, its aggregate gross income from all sources will satisfy the 75% and 95% gross income tests applicable to REITs for each taxable year commencing subsequent to the date of the REIT Conversion.

If Host REIT fails to satisfy one or both of the 75% or 95% gross income tests for any taxable year, it may nevertheless qualify as a REIT for such year if it is entitled to relief under certain provisions of the Code. These relief provisions will be generally available if Host REIT's failure to meet such tests was due to reasonable cause and not due to willful neglect, Host REIT attaches a schedule of the sources of its income to its federal income tax return and any incorrect information on the schedule was not due to fraud with intent to evade tax. It is not possible, however, to state whether in all circumstances Host REIT would be entitled to the benefit of these relief provisions. For example, if Host REIT fails to satisfy the gross income tests because nonqualifying income that Host REIT intentionally incurs exceeds the limits on such income, the IRS could conclude that Host REIT's failure to satisfy the tests was not due to reasonable cause. If these relief provisions are inapplicable to a

particular set of circumstances involving Host REIT, Host REIT will not qualify as a REIT. As discussed above in "--General," even if these relief provisions apply, a tax would be imposed with respect to the excess net income.

Any gain realized by Host REIT on the sale of any property held as inventory or other property held primarily for sale to customers in the ordinary course of business (including Host REIT's share of any such gain realized by the Operating Partnership) will be treated as income from a "prohibited transaction" that is subject to a 100% penalty tax. Such prohibited transaction income may also have an adverse effect upon Host REIT's ability to satisfy the income tests for qualification as a REIT. Under existing law, whether property is held as inventory or primarily for sale to customers in the ordinary course of a trade or business is a question of fact that depends upon all the facts and circumstances with respect to the particular transaction. The Operating Partnership intends that both it and the Partnerships and Private Partnerships will hold the Hotels for investment with a view to long-term appreciation, to engage in the business of acquiring and owning the Hotels (and other hotels) and to make such occasional sales of the Hotels as are consistent with the Operating Partnership's investment objectives. There can be no assurance, however, that the IRS might not contend that one or more of such sales is subject to the 100% penalty tax.

Asset Tests Applicable to REITs. Host REIT, at the close of each quarter of its taxable year, must also satisfy three tests relating to the nature of its assets. First, at least 75% of the value of Host REIT's total assets must be represented by real estate assets, including for this purpose (i) its allocable share of real estate assets held by partnerships in which Host REIT owns an interest (including its allocable share of the assets held through the Operating Partnership) and (ii) stock or debt instruments held for not more than one year purchased with the proceeds of a stock offering or long-term (at least five years) debt offering of Host REIT, cash, cash items and government securities. Second, not more than 25% of Host REIT's total assets may be represented by securities other than those in the 75% asset class. Third, of the investments included in the 25% asset class, the value of any one issuer's securities owned by Host REIT may not exceed 5% of the value of Host REIT's total assets and Host REIT may not own more than 10% of any one issuer's outstanding voting securities.

The Operating Partnership does not own any of the voting stock of a Non-Controlled Subsidiary but it will own 100% of the nonvoting stock of each Non-Controlled Subsidiary. The Operating Partnership may also own nonvoting stock, representing substantially all of the equity, in other corporate entities that serve as partners or members in the various entities that hold title to the Hotels. Host REIT will represent, however, that neither Host REIT, the Operating Partnership nor any of the Hotel Partnerships or Subsidiary Partnerships will own more than 10% of the voting securities of any entity that would be treated as a corporation for federal income tax purposes. In addition, Host REIT and its senior management believe, and Host REIT will represent, that the securities of any one issuer owned by Host REIT, the Operating Partnership, the Partnerships and Private Partnerships and any Subsidiary Partnerships (including Host REIT's pro rata share of the value of the securities of each Non-Controlled Subsidiary) will not exceed 5% of the total value of Host REIT's assets. There can be no assurance, however, that the IRS might not contend that the value of such securities exceeds the 5% value limitation or that nonvoting stock of a Non-Controlled Subsidiary or another corporate entity owned by the Operating Partnership should be considered "voting stock" for this purpose.

After initially meeting the asset tests at the close of any quarter, Host REIT will not lose its status as a REIT for failure to satisfy the asset tests at the end of a later quarter solely by reason of changes in asset values. If the failure to satisfy the asset tests results from an acquisition of securities or other property during a quarter (including, for example, as a result of Host REIT increasing its interest in the Operating Partnership as a result of the exercise of the Unit Redemption Right or an additional capital contribution of proceeds from an offering of Host REIT Common Stock by Host REIT), the failure can be cured by disposition of sufficient nonqualifying assets within 30 days after the close of that quarter. Host REIT intends to maintain adequate records of the value of its assets to ensure compliance with the asset tests and to take such other actions within 30 days after the close of any quarter as may be required to cure any noncompliance. If Host REIT fails to cure noncompliance with the asset tests within such time period, Host REIT would cease to qualify as a REIT.

Clinton Administration's Proposed Changes to REIT Asset Test. The Clinton Administration's fiscal year 1999 budget proposal, announced on February 2, 1998, includes a proposal to amend the 10% voting securities test. The proposal would require a REIT to own no more than 10% of the vote or value of all classes of stock of any corporation (except for qualified REIT subsidiaries or corporations that qualify as REITs). Corporations (referred to herein as "subsidiary corporations") existing prior to the effective date of the proposal generally would be "grandfathered"; i.e., the REIT would be subject to the existing 10% voting securities test (described above) with respect to grandfathered subsidiary corporations. However, such "grandfathered" status would terminate with respect to a subsidiary corporation if the subsidiary corporation engaged in a new trade or business or acquired substantially new assets.

Because the Operating Partnership will own 100% of the nonvoting stock of each Non-Controlled Subsidiary, and Host REIT will be deemed to own an interest in each Non-Controlled Subsidiary equal to its proportionate interest in the Operating Partnership, Host REIT would not satisfy the proposed 10% value limitation with respect to any of the Non-Controlled Subsidiaries. Whether any of the Non-Controlled Subsidiaries would qualify as a grandfathered subsidiary corporation as the proposal is currently drafted would depend upon the effective date of the proposal (which is not yet known). If a Non-Controlled Subsidiary otherwise eligible for "grandfathered" status were to engage in a new trade or business or were to acquire substantial new assets, or if Host REIT were to make a capital contribution to a Non-Controlled Subsidiary otherwise eligible for "grandfathered" status, its "grandfathered" status would terminate and Host REIT would fail to qualify as a REIT. Moreover, Host REIT would not be able to own, directly or indirectly, more than 10% of the vote or value of any subsidiary corporation formed or acquired after the effective date of the proposal. Thus, the proposal, if enacted, would materially impede Host REIT's ability to engage in new thirdparty management or similar activities (and, if enacted prior to the REIT Conversion, might materially impair Host's ability to complete the REIT Conversion.)

Annual Distribution Requirements Applicable to REITs. Host REIT, in order to qualify as a REIT, is required to distribute dividends (other than capital gain dividends) to its stockholders in an amount at least equal to (i) the sum of (a) 95% of Host REIT's REIT taxable income (computed without regard to the dividends paid deduction and Host REIT's net capital gain) and (b) 95% of the net income (after tax), if any, from foreclosure property, minus (ii) the sum of certain items of noncash income. In addition, if Host REIT disposes of any Built-In Gain Asset during its Recognition Period, Host REIT will be required, pursuant to Treasury Regulations which have not yet been promulgated, to distribute at least 95% of the Built-In Gain (after tax), if any, recognized on the disposition of such asset. See "--General" above for a discussion of "Built-In Gain Assets." Such distributions must be paid in the taxable year to which they relate, or in the following taxable year if declared before Host REIT timely files its tax return for such year and if paid on or before the first regular dividend payment date after such declaration. Host REIT intends to make timely distributions sufficient to satisfy these annual distribution requirements. In this regard, the Partnership Agreement authorizes Host REIT, as general partner, to take such steps as may be necessary to cause the Operating Partnership to distribute to its partners an amount sufficient to permit Host REIT to meet these distribution requirements.

To the extent that Host REIT does not distribute all of its net capital gain or distributes at least 95%, but less than 100%, of its REIT taxable income, as adjusted, it will be subject to tax thereon at regular ordinary and capital gain corporate tax rates. Host REIT, however, may designate some or all of its retained net capital gain, so that, although the designated amount will not be treated as distributed for purposes of this tax, a stockholder would include its proportionate share of such amount in income, as capital gain, and would be treated as having paid its proportionate share of the tax paid by Host REIT with respect to such amount. The stockholder's basis in its Host REIT Common Stock would be increased by the amount the stockholder included in income and decreased by the amount of the tax the stockholder is treated as having paid. Host REIT would make an appropriate adjustment to its earnings and profits. For a more detailed description of the federal income tax consequences to a stockholder of such a designation, see "--Taxation of Taxable U.S. Stockholders Generally."

There is a significant possibility that Host REIT's REIT taxable income will exceed its cash flow (due in part to certain "non-cash" or "phantom" income expected to be taken into account in computing Host REIT's

REIT taxable income). Host REIT anticipates, however, that it will generally have sufficient cash or liquid assets to enable it to satisfy the distribution requirements described above. It is possible, however, that Host REIT, from time to time, may not have sufficient cash or other liquid assets to meet these distribution requirements. In such event, in order to meet the distribution requirements, Host REIT may find it necessary to arrange for short-term, or possibly long-term, borrowings to fund required distributions and/or to pay dividends in the form of taxable stock dividends.

Host REIT intends to calculate its REIT taxable income based upon the conclusion that the Hotel Partnerships, the Subsidiary Partnerships or the Operating Partnership, as applicable, is the owner of the Hotels for federal income tax purposes. As a result, Host REIT expects that the depreciation deductions with respect to the Hotels will reduce its REIT taxable income. This conclusion is consistent with the conclusion above that the Leases will be treated as true leases for federal income tax purposes. If the IRS were to challenge successfully this position, in addition to failing in all likelihood the 75% and 95% gross income tests described above, Host REIT also might be deemed retroactively to have failed to meet the REIT distribution requirements and would have to rely on the payment of a "deficiency dividend" in order to retain its REIT status.

Under certain circumstances, Host REIT may be able to rectify a failure to meet the distribution requirement for a year by paying "deficiency dividends" to stockholders in a later year, which may be included in Host REIT's deduction for dividends paid for the earlier year. Thus, Host REIT may be able to avoid being taxed on amounts distributed as deficiency dividends; however, Host REIT would be required to pay interest based upon the amount of any deduction taken for deficiency dividends.

Furthermore, if Host REIT should fail to distribute during each calendar year at least the sum of (i) 85% of its REIT ordinary income for such year, (ii) 95% of its REIT capital gain income for such year, and (iii) any undistributed taxable income from prior periods, Host REIT would be subject to a 4% excise tax on the excess of such required distribution over the sum of amounts actually distributed and amounts retained with respect to which the REIT pays federal income tax.

Failure of Host REIT to Qualify as a REIT. If Host REIT fails to qualify for taxation as a REIT in any taxable year, and if the relief provisions do not apply, Host REIT will be subject to tax (including any applicable alternative minimum tax) on its taxable income at regular corporate rates. Distributions to stockholders in any year in which Host REIT fails to qualify will not be deductible by Host REIT nor will they be required to be made. As a result, Host REIT's failure to qualify as a REIT would significantly reduce the cash available for distribution by Host REIT to its stockholders and could materially reduce the value of the Host REIT Common Stock. In addition, if Host REIT fails to qualify as a REIT, all distributions to stockholders will be taxable as ordinary income, to the extent of Host REIT's current and accumulated E&P, although subject to certain limitations of the Code, corporate distributees may be eligible for the dividends received deduction with respect to these distributions. Unless entitled to relief under specific statutory provisions, Host REIT also will be disqualified from taxation as a REIT for the four taxable years following the year during which qualification was lost. It is not possible to state whether in all circumstances Host REIT would be entitled to such statutory relief.

## TAXATION OF TAXABLE U.S. STOCKHOLDERS GENERALLY

Distributions by Host REIT. As long as Host REIT qualifies as a REIT, distributions made by Host REIT out of its current or accumulated E&P (and not designated as capital gain dividends) will constitute dividends taxable to its taxable U.S. Stockholders as ordinary income. Such distributions will not be eligible for the dividends received deduction in the case of U.S. Stockholders that are corporations. To the extent that Host REIT makes distributions (not designated as capital gain dividends) in excess of its current and accumulated E&P, such distributions will be treated first as a tax-free return of capital to each U.S. Stockholder, reducing the adjusted basis which such U.S. Stockholder has in its Host REIT Common Stock for tax purposes by the amount of such distribution (but not below zero), with distributions in excess of a U.S. Stockholder's adjusted basis in its Host REIT Common Stock taxable as capital gains (provided that the Host REIT Common Stock has been held as a capital asset). Dividends declared by Host REIT in October, November or December of any year and payable to

a stockholder of record on a specified date in any such month shall be treated as both paid by Host REIT and received by the stockholder on December 31 of such year, provided that the dividend is actually paid by Host REIT on or before January 31 of the following calendar year.

Distributions made by Host REIT that are properly designated by Host REIT as capital gain dividends will be taxable to taxable non-corporate U.S. Stockholders (i.e., individuals, estates or trusts) as gain from the sale or exchange of a capital asset held for more than one year (to the extent that they do not exceed Host REIT's actual net capital gain for the taxable year) without regard to the period for which such non-corporate U.S. Stockholder has held his Host REIT Common Stock. In the event that Host REIT designates any portion of a dividend as a "capital gain dividend," a U.S. Stockholder's share of such capital gain dividend would be an amount which bears the same ratio to the total amount of dividends paid to such U.S. Stockholder for the year as the aggregate amount designated as a capital gain dividend bears to the aggregate amount of all dividends paid on all classes of shares for the year. On November 10, 1997, the IRS issued Notice 97-64, which provides generally that Host REIT may classify portions of its designated capital gain dividend as (i) a 20% gain distribution (which would be taxable to non-corporate U.S. Stockholders at a maximum rate of 20%), (ii) an unrecaptured Section 1250 gain distribution (which would be taxable to non-corporate U.S. Stockholders at a maximum rate of 25%) or (iii) a 28% rate gain distribution (which would be taxable to non-corporate U.S. Stockholders at a maximum rate of 28%). If no designation is made, the entire designated capital gain dividend will be treated as a 28% rate gain distribution. Notice 97-64 provides that a REIT must determine the maximum amounts that it may designate as 20% and 25% rate capital gain dividends by performing the computation required by the Code as if the REIT were an individual whose ordinary income were subject to a marginal tax rate of at least 28%. Notice 97-64 further provides that designations made by the REIT only will be effective to the extent that they comply with Revenue Ruling 89-81, which requires that distributions made to different classes of shares be composed proportionately of dividends of a particular type. On July 22, 1998, as part of the IRS Restructuring Act, the holding period requirement for the application of the 20% and 25% capital gain tax rates was reduced to 12 months from 18 months for sales of capital gain assets on or after January 1, 1998. Although Notice 97-64 will apply to sales of capital gain assets after July 28, 1997 and before January 1, 1998, it is expected that the IRS will issue clarifying guidance (most likely applying the same principles set forth in Notice 97-64) regarding a REIT's designation of capital gain dividends in light of the new holding period requirements. For a discussion of the capital gain tax rates applicable to non-corporate U.S. Stockholders, see "--1997 Act and IRS Restructuring Act Changes to Capital Gain Taxation" below.

Distributions made by Host REIT that are properly designated by Host REIT as capital gain dividends will be taxable to taxable corporate U.S. Stockholders as long-term gain (to the extent that they do not exceed Host REIT's actual net capital gain for the taxable year) at a maximum rate of 35% without regard to the period for which such corporate U.S. Stockholder has held its Host REIT Common Stock. Such U.S. Stockholders may, however, be required to treat up to 20% of certain capital gain dividends as ordinary income.

U.S. Stockholders may not include in their individual income tax returns any net operating losses or capital losses of Host REIT. Instead, such losses would be carried over by Host REIT for potential offset against future income (subject to certain limitations). Distributions made by Host REIT and gain arising from the sale or exchange by a U.S. Stockholder of Host REIT Common Stock will not be treated as passive activity income, and, as a result, U.S. Stockholders generally will not be able to apply any "passive losses" against such income or gain. In addition, taxable distributions from Host REIT generally will be treated as investment income for purposes of the investment interest limitation. Capital gain dividends and capital gains from the disposition of shares (including distributions treated as such), however, will be treated as investment income only if the U.S. Stockholder so elects, in which case such capital gains will be taxed at ordinary income rates.

Host REIT will notify stockholders after the close of its taxable year as to the portions of distributions attributable to that year that constitute ordinary income, return of capital and capital gain. Host REIT may designate (by written notice to its stockholders) its net capital gain so that with respect to retained net capital gains, a U.S. Stockholder would include its proportionate share of such gain in income, as long-term capital gain,

and would be treated as having paid its proportionate share of the tax paid by Host REIT with respect to the gain. The U.S. Stockholder's basis in its Host REIT Common Stock would be increased by its share of such gain and decreased by its share of such tax. With respect to such long-term capital gain of a U.S. Stockholder that is an individual or an estate or trust, the IRS, as described above in this section, has authority to issue regulations that could apply the special tax rate applicable generally to the portion of the long-term capital gains of an individual or an estate or trust attributable to deductions for depreciation taken with respect to depreciable real property. IRS Notice 97-64, described above in this section, did not address the taxation of non-corporate REIT stockholders with respect to retained net capital gains.

Sales of Host REIT Common Stock. Upon any sale or other disposition of Host REIT Common Stock, a U.S. Stockholder will recognize gain or loss for federal income tax purposes in an amount equal to the difference between (i) the amount of cash and the fair market value of any property received on such sale or other disposition and (ii) the holder's adjusted basis in such Host REIT Common Stock for tax purposes. Such gain or loss will be capital gain or loss if the Host REIT Common Stock have been held by the U.S. Stockholder as a capital asset. In the case of a U.S. Stockholder who is an individual or an estate or trust, such gain or loss will be long-term capital gain or loss (and any such long-term capital gain shall be subject to the maximum capital gain rate of 20%). In the case of a U.S. Stockholder that is a corporation, such gain or loss will be long-term capital gain or loss if such shares have been held for more than one year (and any such capital gain shall be subject to the maximum capital gain rate of 35%). In general, any loss recognized by a U.S. Stockholder upon the sale or other disposition of Host REIT Common Stock that have been held for six months or less (after applying certain holding period rules) will be treated as a long-term capital loss, to the extent of distributions received by such U.S. Stockholder from Host REIT that were required to be treated as long-term capital gains.

1997 Act and IRS Restructuring Act Changes to Capital Gain Taxation. The 1997 Act altered the taxation of capital gain income. Under the 1997 Act, individuals, trusts and estates that hold certain investments for more than 18 months may be taxed at a maximum long-term capital gain rate of 20% on the sale or exchange of those investments. Individuals, trusts and estates that hold certain assets for more than one year but not more than 18 months may be taxed at a maximum long-term capital gain rate of 28% on the sale or exchange of those investments. The 1997 Act also provides a maximum rate of 25% for "unrecaptured Section 1250 gain" for individuals, trusts and estates, special rules for "qualified 5-year gain" and other changes to prior law. The recently enacted IRS Restructuring Act, however, reduced the holding period requirement established by the 1997 Act for the application of the 20% and 25% capital gain tax rates to 12 months from 18 months for sales of capital gain assets after December 31, 1997. The 1997 Act allows the IRS to prescribe regulations on how the 1997 Act's capital gain rates will apply to sales of capital assets by "pass-through entities" (including REITs, such as Host REIT) and to sales of interests in "pass-through entities." For a discussion of the rules under the 1997 Act that apply to the taxation of distributions by Host REIT to its stockholders that are designated by Host REIT as "capital gain dividends," see "--Distributions by Host REIT" above. Stockholders are urged to consult with their own tax advisors with respect to the rules contained in the 1997 Act and the IRS Restructuring Act.

# BACKUP WITHHOLDING FOR HOST REIT DISTRIBUTIONS

Host REIT will report to its U.S. Stockholders and the IRS the amount of dividends paid during each calendar year and the amount of tax withheld, if any. Under the backup withholding rules, a U.S. Stockholder may be subject to backup withholding at the rate of 31% with respect to dividends paid unless such holder (a) is a corporation or comes within certain other exempt categories and, when required, demonstrates this fact or (b) provides a taxpayer identification number, certifies as to no loss of exemption from backup withholding and otherwise complies with applicable requirements of the backup withholding rules. A U.S. Stockholder that does not provide Host REIT with a correct taxpayer identification number may also be subject to penalties imposed by the IRS. Any amount paid as backup withholding will be creditable against the stockholder's income tax liability. In addition, Host REIT may be required to withhold a portion of its capital gain distributions to any U.S. Stockholders who fail to certify their non-foreign status to Host REIT. See "--Taxation of Non-U.S. Stockholders."

208

The IRS has ruled that amounts distributed as dividends by a qualified REIT do not constitute unrelated business taxable income ("UBTI") when received by a tax-exempt entity. Based on that ruling, provided that a tax-exempt stockholder (except certain tax-exempt stockholders described below) has not held its Host REIT Common Stock as "debt financed property" within the meaning of the Code and such Host REIT Common Stock are not otherwise used in a trade or business, the dividend income from Host REIT will not be UBTI to a tax-exempt stockholder. Similarly, income from the sale of Host REIT Common Stock will not constitute UBTI unless such tax-exempt stockholder has held such Host REIT Common Stock as "debt financed property" within the meaning of the Code or has used the Host REIT Common Stock in a trade or business.

For tax-exempt stockholders that are social clubs, voluntary employee benefit associations, supplemental unemployment benefit trusts and qualified group legal services plans exempt from federal income taxation under Code Sections 501 (c)(7), (c)(9), (c)(17) and (c)(20), respectively, income from an investment in Host REIT will constitute UBTI unless the organization is properly able to deduct amounts set aside or placed in reserve for certain purposes so as to offset the income generated by its investment in Host REIT. Such prospective stockholders should consult their own tax advisors concerning these "set aside" and reserve requirements.

Notwithstanding the above, however, the Omnibus Budget Reconciliation Act of 1993 (the "1993 Act") provides that, effective for taxable years beginning in 1994, a portion of the dividends paid by a "pension held REIT" shall be treated as UBTI as to any trust which (i) is described in Section 401(a) of the Code, (ii) is tax-exempt under Section 501(a) of the Code and (iii) holds more than 10% (by value) of the interests in the REIT. Tax-exempt pension funds that are described in Section 401(a) of the Code are referred to below as "qualified trusts."

A REIT is a "pension held REIT" if (i) it would not have qualified as a REIT but for the fact that Section 856(h)(3) of the Code (added by the 1993 Act) provides that stock owned by qualified trusts shall be treated, for purposes of the "not closely held" requirement, as owned by the beneficiaries of the trust (rather than by the trust itself) and (ii) either (a) at least one such qualified trust holds more than 25% (by value) of the interests in the REIT or (b) one or more such qualified trusts, each of which owns more than 10% (by value) of the interests in the REIT, hold in the aggregate more than 50% (by value) of the interests in the REIT. The percentage of any REIT dividend treated as UBTI is equal to the ratio of (i) the UBTI earned by the REIT (treating the REIT as if it were a qualified trust and therefore subject to tax on UBTI) to (ii) the total gross income of the REIT. A de minimis exception applies where the percentage is less than 5% for any year. The provisions requiring qualified trusts to treat a portion of REIT distributions as UBTI will not apply if the REIT is able to satisfy the "not closely held" requirement without relying upon the "look-through" exception with respect to qualified trusts.

Based on the anticipated ownership of Host REIT Common Stock immediately following the REIT Conversion and as a result of certain limitations on transfer and ownership of Host REIT Common Stock contained in the Host REIT Charter, Host REIT does not expect to be classified as a "pension held REIT."

# TAXATION OF NON-U.S. STOCKHOLDERS

The rules governing federal income taxation of the ownership and disposition of Host REIT Common Stock by Non-U.S. Stockholders are complex and no attempt is made herein to provide more than a brief summary of such rules. Accordingly, the discussion does not address all aspects of federal income tax and does not address state, local or foreign tax consequences that may be relevant to a Non-U.S. Stockholder in light of its particular circumstances. In addition, this discussion is based on current law, which is subject to change, and assumes that Host REIT qualifies for taxation as a REIT. Prospective Non-U.S. Stockholders should consult with their own tax advisers to determine the impact of federal, state, local and foreign income tax laws with regard to an investment in Host REIT Common Stock, including any reporting requirements.

Distributions by Host REIT. Distributions by Host REIT to a Non-U.S. Stockholder that are neither attributable to gain from sales or exchanges by Host REIT of United States real property interests nor designated

by Host REIT as capital gains dividends will be treated as dividends of ordinary income to the extent that they are made out of current or accumulated E&P of Host REIT. Such distributions ordinarily will be subject to withholding of United States federal income tax on a gross basis (that is, without allowance of deductions) at a 30% rate or such lower rate as may be specified by an applicable income tax treaty, unless the dividends are treated as effectively connected with the conduct by the Non-U.S. Stockholder of a United States trade or business. Under certain treaties, however, lower withholding rates generally applicable to dividends do not apply to dividends from a REIT, such as Host REIT. Certain certification and disclosure requirements must be satisfied to be exempt from withholding under the effectively connected income exemption. Dividends that are effectively connected with such a trade or business will be subject to tax on a net basis (that is, after allowance of deductions) at graduated rates, in the same manner as U.S. Stockholders are taxed with respect to such dividends and are generally not subject to withholding. Any such dividends received by a Non-U.S. Stockholder that is a corporation may also be subject to an additional branch profits tax at a 30% rate or such lower rate as may be specified by an applicable income tax treaty. Host REIT expects to withhold United States income tax at the rate of 30% on any distribution made to a Non-U.S. Stockholder unless (i) a lower treaty rate applies and any required form or certification evidencing eligibility for that lower rate is filed with Host REIT or (ii) a Non-U.S. Stockholder files an IRS Form 4224 with Host REIT claiming that the distribution is effectively connected income.

Distributions in excess of current or accumulated E&P of Host REIT will not be taxable to a Non-U.S. Stockholder to the extent that they do not exceed the adjusted basis of the stockholder's Host REIT Common Stock, but rather will reduce the adjusted basis of such Host REIT Common Stock. To the extent that such distributions exceed the adjusted basis of a Non-U.S. Stockholder's Host REIT Common Stock, they will give rise to gain from the sale or exchange of its Host REIT Common Stock, the tax treatment of which is described below. As a result of a legislative change made by the Small Business Job Protection Act of 1996, it appears that Host REIT will be required to withhold 10% of any distribution in excess of Host REIT's current and accumulated earnings and profits. Consequently, although Host REIT intends to withhold at a rate of 30% on the entire amount of any distribution (or a lower applicable treaty rate), to the extent that Host REIT does not do so, any portion of a distribution not subject to withholding at a rate of 30% (or a lower applicable treaty rate) would be subject to withholding at a rate of 10%. However, the Non-U.S. Stockholder may seek a refund of such amounts from the IRS if it subsequently determined that such distribution was, in fact, in excess of current or accumulated E&P of Host REIT, and the amount withheld exceeded the Non-U.S. Stockholder's United States tax liability, if any, with respect to the distribution.

Distributions to a Non-U.S. Stockholder that are designated by Host REIT at the time of distribution as capital gain dividends (other than those arising from the disposition of a United States real property interest) generally will not be subject to United States federal income taxation, unless (i) the investment in the Host REIT Common Stock is effectively connected with the Non-U.S. Stockholder's United States trade or business, in which case the Non-U.S. Stockholder will be subject to the same treatment as U.S. Stockholders with respect to such gain (except that a stockholder that is a foreign corporation may also be subject to the 30% branch profits tax, as discussed above) or (ii) the Non-U.S. Stockholder is a nonresident alien individual who is present in the United States for 183 days or more during the taxable year and has a "tax home" in the United States, in which case the nonresident alien individual will be subject to a 30% tax on the individual's capital gains.

Pursuant to FIRPTA, distributions to a Non-U.S. Stockholder that are attributable to gain from sales or exchanges by Host REIT of United States real property interests (whether or not designated as capital gain dividends) will cause the Non-U.S. Stockholder to be treated as recognizing such gain as income effectively connected with a United States trade or business. Non-U.S. Stockholders would thus generally be taxed at the same rates applicable to U.S. Stockholders (subject to a special alternative minimum tax in the case of nonresident alien individuals). Also, such gain may be subject to a 30% branch profits tax in the hands of a Non-U.S. Stockholder that is a corporation, as discussed above. Host REIT is required to withhold 35% of any such distribution. That amount is creditable against the Non-U.S. Stockholder's federal income tax liability.

210

Although the law is not entirely clear on the matter, it appears that amounts designated by Host REIT pursuant to the 1997 Act as undistributed capital gains in respect of the Host REIT Common Stock held by U.S. Stockholders (see "--Annual Distribution Requirements Applicable to REITs" above) would be treated with respect to Non-U.S. Stockholders in the manner outlined in the preceding two paragraphs for actual distributions by Host REIT of capital gain dividends. Under that approach, the Non-U.S. Stockholders would be able to offset as a credit against their United States federal income tax liability resulting therefrom their proportionate share of the tax paid by Host REIT on such undistributed capital gains (and to receive from the IRS a refund to the extent their proportionate share of such tax paid by Host REIT were to exceed their actual United States federal income tax liability).

Sales of Host REIT Common Stock. Gain recognized by a Non-U.S. Stockholder upon the sale or exchange of Host REIT Common Stock generally will not be subject to United States taxation unless such shares constitute a "United States real property interest" within the meaning of FIRPTA. The Host REIT Common Stock will not constitute a "United States real property interest" so long as Host REIT is a "domestically controlled REIT." A "domestically controlled REIT" is a REIT in which at all times during a specified testing period less than 50% in value of its stock is held directly or indirectly by Non-U.S. Stockholders. Host REIT is unable at this time to predict whether it will be a "domestically controlled REIT," and therefore whether the sale of Host REIT Common Stock will be subject to taxation under FIRPTA. Moreover, even if Host REIT initially qualifies as a "domestically controlled REIT, because the Host REIT Common Stock is expected to be publicly traded, no assurance can be given that Host REIT would continue to be a "domestically controlled REIT." Notwithstanding the foregoing, gain from the sale or exchange of Host REIT Common Stock not otherwise subject to FIRPTA will be taxable to a Non-U.S. Stockholder if the Non-U.S. Stockholder is a nonresident alien individual who is present in the United States for 183 days or more during the taxable year and has a "tax home" in the United States. In such case, the nonresident alien individual will be subject to a 30% United States withholding tax on the amount of such individual's gain.

Even if Host REIT does not qualify as or ceases to be a "domestically controlled REIT," gain arising from the sale or exchange by a Non-U.S. Stockholder of Host REIT Common Stock would not be subject to United States taxation under FIRPTA as a sale of a "United States real property interest" if (i) the Host REIT Common Stock are "regularly traded" (as defined by applicable Treasury Regulations) on an established securities market (e.g. the NYSE) and (ii) such Non-U.S. Stockholder owned 5% or less of the Host REIT Common Stock throughout the five-year period ending on the date of the sale or exchange. If gain on the sale or exchange of Host REIT Common Stock were subject to taxation under FIRPTA, the Non-U.S. Stockholder would be subject to regular United States income tax with respect to such gain in the same manner as a taxable U.S. Stockholder (subject to any applicable alternative minimum tax, a special alternative minimum tax in the case of nonresident alien individuals and the possible application of the 30% branch profits tax in the case of foreign corporations) and the purchaser of the Host REIT Common Stock would be required to withhold and remit to the IRS 10% of the purchase price.

Backup Withholding Tax and Information Reporting. Backup withholding tax (which generally is a withholding tax imposed at the rate of 31% on certain payments to persons that fail to furnish certain information under the United States information reporting requirements) and information reporting will generally not apply to distributions paid to Non-U.S. Stockholders outside the United States that are treated as (i) dividends subject to the 30% (or lower treaty rate) withholding tax discussed above, (ii) capital gain dividends or (iii) distributions attributable to gain from the sale or exchange by Host REIT of United States real property interests. As a general matter, backup withholding and information reporting will not apply to a payment of the proceeds of a sale of Host REIT Common Stock by or through a foreign office of a foreign broker. Information reporting (but not backup withholding) will apply, however, to a payment of the proceeds of a sale of Host REIT Common Stock by a foreign office of a broker that (a) is a United States person, (b) derives 50% or more of its gross income for certain periods from the conduct of a trade or business in the United States or (c) is a "controlled foreign corporation" (generally, a foreign corporation controlled by United States stockholders) for United States tax purposes, unless the broker has documentary evidence in its records that the holder is a Non-U.S. Stockholder

211

and certain other conditions are met or the stockholder otherwise establishes an exemption. Payment to or through a United States office of a broker of the proceeds of a sale of Host REIT Common Stock is subject to both backup withholding and information reporting unless the stockholder certifies under penalty of perjury that the stockholder is a Non-U.S. Stockholder, or otherwise establishes an exemption. A Non-U.S. Stockholder may obtain a refund of any amounts withheld under the backup withholding rules by filing the appropriate claim for refund with the IRS.

The IRS has recently finalized regulations regarding the withholding and information reporting rules discussed above. In general, these regulations do not alter the substantive withholding and information reporting requirements but unify certification procedures and forms and clarify and modify reliance standards. These regulations generally are effective for payments made after December 31, 2000, subject to certain transition rules. Valid withholding certificates that are held on December 31, 1999, will remain valid until the earlier of December 31, 2000 or the date of expiration of the certificate under rules currently in effect (unless otherwise invalidated due to changes in the circumstances of the person whose name is on such certificate). A Non-U.S. Stockholder should consult its own advisor regarding the effect of the new Treasury Regulations.

### TAX ASPECTS OF HOST REIT'S OWNERSHIP OF OP UNITS

General. Substantially all of Host REIT's investments will be held through the Operating Partnership, which will hold the Hotels either directly or through the Partnerships or Private Partnerships (which, in turn, may hold Hotels through certain Subsidiary Partnerships (as defined in the following paragraph)). In general, partnerships are "pass-through" entities that are not subject to federal income tax. Rather, partners are allocated their proportionate shares of the items of income, gain, loss, deduction and credit of a partnership, and are potentially subject to tax thereon, without regard to whether the partners receive a distribution from the partnership. Host REIT will include in its income its proportionate share of the foregoing partnership items for purposes of the various REIT income tests and in the computation of its REIT taxable income. Moreover, for purposes of the REIT asset tests, Host REIT will include its proportionate share of assets held through the Operating Partnership, the Partnerships and Private Partnerships, and any Subsidiary Partnerships. See "--Federal Income Taxation of Host REIT Following the Merger--Ownership of Partnership Interests by a REIT."

Entity Classification. If the Operating Partnership or any of the partnerships or limited liability companies in which the Operating Partnership or any of the Partnerships or Private Partnerships will have a direct or indirect interest (other than an indirect interest held through an entity treated for federal income tax purposes as a corporation) following the REIT Conversion (the "Subsidiary Partnerships") were treated as an association, the entity would be taxable as a corporation and therefore would be subject to an entity level tax on its income. In such a situation, the character of Host REIT's assets and items of gross income would change and could preclude Host REIT from qualifying as a REIT (see "--Federal Income Taxation of Host REIT Following the Merger--Asset Tests Applicable to REITs" and "--Income Tests Applicable to REITs").

The entire discussion of the federal income tax consequences of the ownership of Host REIT Common Stock is based on the Operating Partnership and all of the Subsidiary Partnerships being classified as partnerships for federal income tax purposes. Pursuant to Treasury Regulations under Section 7701 of the Code, a partnership will be treated as a partnership for federal income tax purposes unless it elects to be treated as a corporation or would be treated as a corporation because it is a "publicly traded partnership." Neither the Operating Partnership nor any of the Subsidiary Partnerships will elect to be treated as a corporation, and therefore, subject to the disclosure below, each will be treated as a partnership for federal income tax purposes (or if it has only one partner or member, disregarded entirely for federal income tax purposes).

Pursuant to Section 7704 of the Code, however, a partnership that does not elect to be treated as a corporation nevertheless will be treated as a corporation for federal income tax purposes if it is a "publicly traded partnership," unless at least ninety percent (90%) of its income consists of "qualifying income" within the meaning of that section. A "publicly traded partnership" is any partnership (i) the interests in which are traded

on an established securities market or (ii) the interests in which are readily tradable on a "secondary market (or the substantial equivalent thereof)." OP Units will not be traded on an established securities market. There is a significant risk, however, that after the Unit Redemption Right becomes exercisable, such interests would be considered readily tradable on the substantial equivalent of a secondary market.

Hogan & Hartson has delivered to Host REIT and the Operating Partnership an opinion stating that, based on certain factual assumptions and representations described in the opinion (including as to the proposed method of operation of the Operating Partnership after the REIT Conversion), the Operating Partnership and each of the Subsidiary Partnerships will be treated as a partnership (or disregarded) for federal income tax purposes (and not treated as an association taxable as a corporation). Hogan & Hartson is of the opinion that, based upon representations by Host REIT and the Operating Partnership as to the expected ownership and operations of the Operating Partnership, even if the Operating Partnership were considered to be a "publicly traded partnership" under the Code because OP Units were considered to be readily tradable on the substantial equivalent of a secondary market, the proposed method of operation of the Operating Partnership is such that it will qualify as a partnership for federal income tax purposes because it will have sufficient "qualifying income." In this regard, the income requirements generally applicable to REITs and the definition of "qualifying income" under Section 7704 of the Code are similar in most key respects. There is one significant difference, however, that is relevant to the Operating Partnership. For a REIT, rent from a tenant does not qualify as "rents from real property" if the REIT and/or one or more actual or constructive owners of 10% or more of the REIT actually or constructively own 10% or more of the tenant; under Section 7704 of the Code, however, rent from a tenant is not qualifying income if a partnership and/or one or more actual or constructive owners of 5% or more of the partnership actually or constructively own 10% or more of the tenant.

A substantial majority of the Operating Partnership's income will come from rent payments by the Lessees, which will be indirectly controlled subsidiaries of Crestline. Accordingly, because The Blackstone Group, Host REIT and any owner of 10% or more of Host REIT will own (or be deemed to own) 5% or more of the Operating Partnership, if The Blackstone Group, Host REIT and/or any owner of 10% or more of Host REIT were to own (or be deemed to own) collectively 10% or more of Crestline, none of the rent from the Lessees would be qualifying income for purposes of determining whether the Operating Partnership should be taxed as a corporation. In order to avoid this result, the Crestline Articles of Incorporation will expressly provide that no person (or persons acting as a group), including The Blackstone Group, Host REIT and any owner of 10% or more of Host REIT, may own, actually and/or constructively, more than 9.8% by value of the equity in Crestline and the Crestline Articles of Incorporation will contain self-executing mechanisms intended to enforce this prohibition. In addition, the Partnership Agreement prohibits any person (or persons acting as a group) or entity (other than Blackstone and Host REIT) from owning, actually and/or constructively, more than 4.9% of the value of the Operating Partnership, and the Host REIT Charter will prohibit any person (or persons acting as a group) or entity (including Blackstone and the Marriott family and their affiliated entities as a group) from, subject to certain limited exceptions, owning, actually and/or constructively, more than 9.8% of the lesser of the number or value of the total outstanding shares of Host REIT. Assuming that all of these prohibitions are enforced at all times in accordance with their terms (as to which there can be no assurance), then so long as the Operating Partnership's income is such that Host REIT could meet the gross income tests applicable to REITs (see "--Federal Income Taxation of Host REIT Following the Merger--Income Tests Applicable to REITs" and "-- Ownership of Partnership Interests by a REIT"), the Operating Partnership's "qualifying income" should be sufficient for it to avoid being classified as a corporation even if it were considered a publicly traded partnership.

If the Operating Partnership were taxable as a corporation, most, if not all, of the tax consequences described herein would be inapplicable. In particular, Host REIT would not qualify as a REIT because the value of Host REIT's ownership interest in the Operating Partnership would exceed 5% of Host REIT's assets and Host REIT would be considered to hold more than 10% of the voting securities of another corporation (see "--Federal Income Taxation of Host REIT Following the Merger--Asset Tests Applicable to REITs"), which would adversely affect the value of the Host REIT Common Stock (see "--Federal Income Taxation of Host

213

REIT Following the Merger--Failure of Host REIT to Qualify as a REIT"). However, as described above, Hogan & Hartson, counsel to the Operating Partnership, is of the opinion that the proposed method of operation of the Operating Partnership (as represented by the Operating Partnership and Host) is such that it will be treated as a partnership for federal income tax purposes and will not be subject to tax as a corporation or an association taxable as a corporation. As noted previously, an opinion of counsel does not bind the courts and no assurance can be provided that such opinion will not be challenged by the IRS or will be sustained by a court if so challenged.

Allocations of Operating Partnership Income, Gain, Loss and Deduction. The Partnership Agreement provides that if the Operating Partnership operates at a net loss, net losses shall be allocated to Host REIT and the limited partners in proportion to their respective percentage ownership interests in the Operating Partnership, provided that net losses that would have the effect of creating a deficit balance in a limited partner's capital account (as specially adjusted for such purpose) ("Excess Losses") will be reallocated to Host REIT, as general partner of the Operating Partnership. The Partnership Agreement also provides that, if the Operating Partnership operates at a net profit, net income shall be allocated first to Host REIT to the extent of Excess Losses with respect to which Host REIT has not previously been allocated net income and any remaining net income shall be allocated in proportion to the respective percentage ownership interests of Host REIT and the limited partners. Finally, the Partnership Agreement provides that if the Operating Partnership has preferred units outstanding, income will first be allocated to such preferred units to the extent necessary to reflect and preserve the economic rights associated with such preferred units.

Although a partnership agreement will generally determine the allocation of income and loss among partners, such allocations will be disregarded for tax purposes if they do not comply with the provisions of Section 704(b) of the Code and the Treasury Regulations promulgated thereunder. Generally, Section 704(b) and the Treasury Regulations promulgated thereunder require that partnership allocations respect the economic arrangement of the partners.

If an allocation is not recognized for federal income tax purposes, the item subject to the allocation will be reallocated in accordance with the partners' interests in the partnership, which will be determined by taking into account all of the facts and circumstances relating to the economic arrangement of the partners with respect to such item. The allocations of taxable income and loss provided for in the Partnership Agreement and the partnership agreements for the Hotel Partnerships (and any Subsidiary Partnerships) are intended to comply with the requirements of Section 704(b) of the Code and the Treasury Regulations promulgated thereunder.

Tax Allocations with Respect to the Hotels. Pursuant to Section 704(c) of the Code, income, gain, loss and deduction attributable to appreciated or depreciated property (such as the Hotels) that is contributed to a partnership in exchange for an interest in the partnership must be allocated in a manner such that the contributing partner is charged with, or benefits from, respectively, the difference between the adjusted tax basis and the fair market value of such property at the time of contribution (the "Book-Tax Difference") associated with the property at the time of the contribution. The Partnership Agreement requires that such allocations be made in a manner consistent with Section 704(c) of the Code.

In general, the partners of the Operating Partnership (including Host REIT) who contributed depreciated assets having a Book-Tax Difference will be allocated depreciation deductions for tax purposes that are lower than such deductions would be if determined on a pro rata basis. Thus, the carryover basis of the contributed assets in the hands of the Operating Partnership may cause Host REIT to be allocated lower depreciation and other deductions, and therefore to be effectively allocated more income, which might adversely affect Host REIT's ability to comply with the REIT distribution requirements. See "--Federal Income Taxation of Host REIT Following the Merger--Annual Distribution Requirements Applicable to REITs". However, in light of the complexity of the governing rules affecting the calculation and allocation of depreciation with respect to properties contributed to a partnership, particularly when a number of those properties are subject to the separate adjustments required in connection with a technical termination under Section 708 of the Code, the number of Hotels that the Operating Partnership will be acquiring in connection with the OP Contribution, the Partnership

Mergers, and the Blackstone Acquisition, and the impact on these calculations of other outside events, including equity offerings by Host or Host REIT and other acquisitions undertaken by Host, Host REIT or the Operating Partnership prior to or in connection with the REIT Conversion, Host, Host REIT and the Operating Partnership believe that it is impossible to predict with any degree of precision the impact that the REIT Conversion will have on the future depreciation allocable by the Operating Partnership to Host REIT.

In addition, in the event of the disposition of any of the contributed assets which have such a Book-Tax Difference, all income attributable to such Book-Tax Difference generally will be allocated to such partners, even though the proceeds of such sale would be allocated proportionately among all the partners in the Operating Partnership (and likely would be retained by the Operating Partnership, rather than distributed to holders of OP Units). Thus, if the Operating Partnership were to sell a Hotel with a Book-Tax Difference that was contributed to the Operating Partnership by Host or Host REIT (including in connection with the OP Contribution), Host REIT generally would be allocated all of the income attributable to such Book-Tax Difference, which could exceed the economic or book income allocated to it as a result of such sale. Such an allocation might cause Host REIT to recognize taxable income in excess of cash proceeds, which might adversely affect Host REIT's ability to comply with the REIT distribution requirements (and could subject Host REIT to a corporate level tax on such gain to the extent the gain is recognized within the 10-year period after the first day of Host REIT's first taxable year as a REIT). See "--Federal Income Taxation of Host REIT Following the Merger--Annual Distribution Requirements Applicable to REITs" and "--Federal Income Taxation of Host REIT Following the Merger--General." It should be noted in this regard that as the general partner of the Operating Partnership, Host REIT will determine whether or not to sell a Hotel contributed to the Operating Partnership by Host REIT.

The Operating Partnership and Host REIT have determined to use generally the traditional method, with a provision for a curative allocation of gain on sale to the extent prior allocations of depreciation with respect to a specific Hotel were limited by the "ceiling rule" applicable under the traditional method, to account for Book-Tax Differences with respect to the Hotels contributed to the Operating Partnership in connection with the OP Contribution or the Partnership Mergers (although there may be certain exceptions). This method is generally a more favorable method for accounting for Book-Tax Differences from the perspective of those partners (including Host REIT) receiving OP Units in exchange for property with a low basis relative to value at the time of the OP Contribution and the Partnership Mergers and is a less favorable method from the perspective of those partners contributing cash (or "high basis" assets) to the Operating Partnership (including Host REIT, to the extent it contributes cash to the Operating Partnership).

With respect to any property purchased by the Operating Partnership subsequent to the REIT Conversion, such property will initially have a tax basis equal to its fair market value, and Section 704(c) of the Code will not apply.

# OTHER TAX CONSEQUENCES FOR HOST REIT AND ITS STOCKHOLDERS

Host REIT and its stockholders may be subject to state or local taxation in various state or local jurisdictions, including those in which it (through the Operating Partnership) or they transact business or reside. The state and local tax treatment of Host REIT and its stockholders may not conform to the federal income tax consequences discussed above. Consequently, prospective stockholders of Host REIT should consult their own tax advisors regarding the effect of state and local tax laws on an investment in Host REIT.

A portion of the cash to be used by Host REIT to fund distributions is expected to come from each Non-Controlled Subsidiary through payments of dividends on the shares of such corporation held by the Operating Partnership (and, in some cases, interest on notes held by the Operating Partnership). Each Non-Controlled Subsidiary will pay federal and state income tax at the full applicable corporate rates on its taxable income computed without regard to any deduction for dividends. To the extent that a Non-Controlled Subsidiary is required to pay federal, state or local taxes, the cash otherwise available for distribution by Host REIT to its stockholders will be reduced accordingly.

### LEGAL MATTERS

Certain legal matters, including certain tax matters, will be passed upon for Host and Host REIT by Hogan & Hartson. L.L.P., Washington, D.C.

#### **EXPERTS**

The financial statements and schedule of Host Marriott Corporation and Host Marriott Hotels as of January 2, 1998 and January 3, 1997 and for each of the three years in the period ended January 2, 1998, the financial statements and schedule of HMC Senior Communities, Inc. as of January 2, 1998 and for the period June 21, 1997 (inception) through January 2, 1998, the balance sheet of Host Marriott, L.P. as of June 19, 1998 and the balance sheet of HMC Merger Corporation as of September 28, 1998 included in this Proxy Statement/Prospectus have been audited by Arthur Andersen LLP, independent public accountants, as indicated in their reports with respect thereto, and are included herein in reliance upon the authority of said firm as experts in giving said reports.

## OTHER MATTERS

The Board of Directors of Host knows of no other business to be presented at the Special Meeting. In the event that other matters properly come before the Special Meeting, the persons named as proxies will vote on them in accordance with their best judgment.

### STOCKHOLDER PROPOSALS

Each of the Host Bylaws and Host REIT Bylaws requires the 1999 Annual Meeting to be held in May 1999. The Host Annual Meeting for 1999 (or, if the Merger has been consummated, the Host REIT Annual Meeting for 1999) is tentatively scheduled to be held on Thursday, May 13, 1999. Accordingly stockholder proposals submitted for presentation at the 1999 Annual Meeting of Stockholders of Host REIT, in the event of the approval of the Agreement and the consummation of the Merger, or the 1999 Annual Meeting of Stockholders of Host, in the event that the Agreement is not approved by Host's stockholders or the Merger is not otherwise consummated because other conditions have not been satisfied, must be submitted in writing by any stockholder who meets the requirements of the proxy rules under the Exchange Act by notice delivered or mailed by first-class United States mail, postage prepaid, to the Corporate Secretary, Host Marriott Corporation or HMC Merger Corporation, as the case may be, 10400 Fernwood Road, Bethesda, Maryland 20817-1109 and must be received no later than December 4, 1998 to be eligible for inclusion in the proxy materials for that meeting. Any such notice shall set forth: (a) the name and address of the stockholder and the text of the proposal to be introduced; (b) the number of shares of Host Common Stock held of record, owned beneficially and represented by proxy by such stockholder as of the date of such notice; and (c) a representation that the stockholder intends to appear in person or by proxy at the meeting to introduce the proposal specified in the notice. Any stockholder proposal received after February 22, 1999 shall be considered untimely, regardless of whether it is submitted for inclusion in the Company's proxy materials pursuant to Rule 14a-8 under the Exchange Act or simply to notify the Company of the stockholder's intention to introduce it for action at the meeting. The chairman of the meeting may refuse to acknowledge the introduction of any stockholder proposals for which notice is not provided in accordance with the foregoing procedures.

AVAILABLE INFORMATION; INCORPORATION OF CERTAIN DOCUMENTS BY REFERENCE

Host is subject to the informational requirements of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and in accordance therewith files reports, proxy statements and other information with the

Securities and Exchange Commission (the "Commission"). Such reports, proxy statements and other information can be inspected and copied at the public reference facilities maintained by the Commission at 450 Fifth Street, N.W., Washington, D.C. 20549, and at the Commission's regional offices located at 75 Park Place, New York, New York 10007 and 500 West Madison Street, Suite 1400, Chicago, Illinois 60661-2511. Copies of such material may be obtained at prescribed rates by writing the Commission, Public Reference Section, 450 Fifth Street, N.W., Washington, D.C. 20549. The Commission maintains a Web site at http://www.sec.gov that contains reports, proxy and information statements and other information regarding registrants, including Host, that file electronically with the Commission. Host Common Stock is listed on the NYSE, and reports, proxy statements and other information concerning Host may also be inspected at the offices of the New York Stock Exchange, Inc., 20 Broad Street, New York, New York 10005.

Host REIT has filed with the Commission a registration statement (the "Registration Statement," which term shall include any amendments thereto) on Form S-4 under the Securities Act of 1933, as amended (the "Securities Act"), with respect to Host REIT Common Stock to be issued upon the consummation of the Merger pursuant to the Agreement described herein, and this Proxy Statement/Prospectus constitutes the Prospectus of Host REIT with respect to such Host REIT Common Stock. This Proxy Statement/Prospectus does not contain all the information set forth in the Registration Statement and the exhibits thereto, certain parts of which are omitted in accordance with the rules and regulations of the Commission, and to which reference is hereby made. For further information, reference is hereby made to the Registration Statement and the exhibits thereto.

Statements contained in this Proxy Statement/Prospectus as to the contents of any contract or other document which is filed as an exhibit to the Registration Statement are not necessarily complete, and each such statement is qualified in its entirety by reference to the full text of such contract or document.

Upon consummation of the REIT Conversion, Host REIT and the Operating Partnership will be required to file reports and other information with the Commission pursuant to the Exchange Act. In addition to applicable legal or NYSE requirements, if any, holders of the Host REIT Common Stock will receive annual reports containing audited financial statements with a report thereon by Host REIT's and the Operating Partnership's independent public accountants, and quarterly reports containing unaudited financial information for each of the first three quarters of each fiscal year.

All documents filed with the Commission by Host pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act subsequent to the date hereof and prior to the date of the Special Meeting shall be deemed to be incorporated by reference in this Proxy Statement/Prospectus and to be a part hereof from the date any such document is filed. Any statement contained herein or in any document incorporated or deemed to be incorporated herein by reference shall be deemed to be modified or superseded for all purposes of this Proxy Statement/Prospectus to the extent a statement contained in this Proxy Statement/Prospectus or in any subsequently filed document that also is deemed to be incorporated herein by reference modifies or supersedes such statement. Any statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this Proxy Statement/Prospectus. All information appearing in this Proxy Statement/Prospectus is qualified in its entirety by the information and financial statements (including notes thereto) appearing in the documents incorporated herein by reference, except to the extent set forth in the immediately preceding statement.

No persons have been authorized to give any information or to make any representation with respect to the matters described in this Proxy Statement/Prospectus other than those contained herein or in the documents incorporated by reference herein and, if given or made, such information or representation must not be relied upon as having been authorized by Host or Host REIT. This Proxy Statement/Prospectus does not constitute an offer to sell, or a solicitation of an offer to buy, any securities in any jurisdiction to or from any person to whom it is not lawful to make any such offer or solicitation in such jurisdiction. Neither the delivery of this Proxy Statement/Prospectus nor any distribution of securities made hereunder shall under any circumstances create an implication that there has been no change in the facts set forth in this Proxy Statement/Prospectus or the affairs of Host or Host REIT since the date hereof or that the information herein is correct as of any time subsequent to the date hereof.

# GLOSSARY

"100% Participation with No Notes Issued" means all Partnerships participate in the Partnership Mergers and the REIT Conversion and no Notes are issued.

"100% Participation with Notes Issued" means all Partnerships participate in the Partnership Mergers and the REIT Conversion and every Limited Partner elects to receive Notes.

"15% Personal Property Test" means the test applied to determine whether a REIT satisfies the requirement under the Code that, in order for rent attributable to the lease of personal property to qualify as "rents from real property," such rent must not account for more than 15% of the total rent received under the lease of real and personal property.

"1993 Act" means the Omnibus Budget Reconciliation Act of 1993.

"1993 Employee Benefits Allocation Agreement" means the Employee Benefits and Other Employment Matters Allocation Agreement between Host and Marriott International, as amended.

"1995 Employee Benefits Allocation Agreement" means the Employee Benefits and Other Employment Matters Allocation Agreement entered into by Host and HM Services that provides for the allocation of certain responsibilities with respect to employee compensation, benefits and labor matters.

"1997 Act" means the Taxpayer Relief Act of 1997.

"1998 Employee Benefits Allocation Agreement" means the Employee Benefits and Other Employment Matters Allocation Agreement among Host, the Operating Partnership and Crestline, to be entered into in connection with the REIT Conversion.

"Acquired Earnings" means undistributed earnings and profits of Host REIT attributable to a "C" corporation taxable year (including accumulated undistributed earnings and profits acquired from Host, some of which may have resulted from either transactions undertaken in contemplation of the REIT Conversion or the REIT Conversion itself).

"Acquiring Person" means a person or group of affiliated or associated persons.  $% \begin{center} \end{center} \begin{center} \begin{center}$ 

"ADA" means the Americans with Disabilities Act.

"Agreement" means the Agreement and Plan of Merger dated November , 1998 by and among Host, Host REIT and the Operating Partnership, a copy of which is attached hereto as Appendix A.

"Atlanta Marquis" means Atlanta Marriott Marquis II Limited Partnership, a Delaware limited partnership, or, as the context may require, such entity together with is subsidiaries, or any of such subsidiaries.

"Available Cash" means net income plus depreciation and amortization and any reduction in reserves and minus interest and principal payments on debt, capital expenditures, any additions to reserves and other adjustments.

"Beneficiary" means a qualified charitable organization selected by Host REIT to be the beneficiary of a trust into which excess shares are to be transferred pursuant to the Host REIT Charter.

"Blackstone" means Blackstone Real Estate Acquisitions L.L.C.

"Blackstone Acquisition" means the expected acquisition from The Blackstone Entities of ownership of, or controlling interests in, twelve hotels and a mortgage loan secured by a thirteenth hotel in exchange for OP

Units, the assumption of certain liabilities and other consideration, including up to 18% of the shares of Crestline common stock, to the extent such acquisition is consummated.

"Blackstone Entities" means The Blackstone Group, a Delaware limited partnership and a series of funds controlled by Blackstone Real Estate Partners, a Delaware limited partnership.

"Blackstone Hotels" means the twelve upscale and luxury full-service hotel properties expected to be acquired by Host from the Blackstone Entities in the Blackstone Acquisition.

"Book-Tax Difference" means the difference between the fair market value of property contributed to a partnership at the time of contribution and the adjusted tax basis of such property at the time of such contribution.

"BT Wolfensohn Opinion" means the written opinion of BT Wolfensohn, which has acted as financial advisor to Host, dated November , 1998.

"Built-In Gain Asset" means an asset which has been acquired from a C corporation in a transaction in which the basis of the asset in the hands of Host REIT is determined by reference to the basis of the asset in the hands of the C corporation.

"Chain Services" mean services generally furnished on a central or regional basis to the Hotels. Such services include the following: (i) the development and operation of computer systems and reservation services, (ii) regional management and administrative services, regional marketing and sales services, regional training services, manpower development and relocation costs of regional personnel and (iii) such additional central or regional services as may from time to time be more efficiently performed on a regional or group level.

"Chicago Suites" means Mutual Benefit Chicago Marriott Suite Hotel Partners, L.P., a Rhode Island limited partnership.

"Code" means the Internal Revenue Code of 1986, as amended.

"Commission" means the U.S. Securities and Exchange Commission.

"Company" means Host (to the extent of its business and assets to be contributed to the Operating Partnership) with respect to the period prior to the REIT Conversion, and Host REIT and the Operating Partnership collectively with respect to the period after the REIT Conversion.

"Comprehensive Stock Incentive Plan" means the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan, formerly called the Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan.

"Control share acquisition" means the acquisition of control shares, subject to certain exceptions.

"Control shares" means voting shares which, if aggregated with all other such voting shares previously acquired by the acquiror or in respect of which the acquiror is able to exercise or direct the exercise of voting power (except solely by virtue of a revocable proxy), would entitle the acquiror to exercise voting power in electing directors within one of the following ranges of voting power: (i) one-fifth or more but less than one-third, (ii) one-third or more but less than a majority or (iii) a majority or more of the voting power. Control shares do not include shares the acquiring person is then entitled to vote as a result of having previously obtained stockholder approval.

"Convertible Preferred Securities" means the 6 3/4% Convertible Quarterly Income Preferred Securities issued by Host Marriott Financial Trust, with an aggregate liquidation amount of \$550 million, which are guaranteed on a subordinated basis by, and convertible into the common stock of, Host.

"CPI" means the Consumer Price Index.

"Crestline" means Crestline Capital Corporation (formerly HMC Senior Communities, Inc.), a Maryland corporation, or, as the context may require, such entity together with the Lessees and its other subsidiaries or any of them, which currently is a wholly-owned subsidiary of Host but will become a separate public company as part of the REIT Conversion when Host or Host REIT distributes the common stock of Crestline to its existing stockholders as part of the Initial E&P Distribution.

"Debentures" means the 6 3/4% Convertible Subordinated Debentures due December 2, 2026 issued by Host.

"Delaware Act" means the Delaware Revised Uniform Limited Partnership Act.

"Desert Springs" means Desert Springs Marriott Limited Partnership, a Delaware limited partnership, or, as the context may require, such entity together with its subsidiaries, or any of such subsidiaries.

"DGCL" means the Delaware General Corporation Law, as amended.

"E&P" means earnings and profits.

"Effective Date" means the date upon which the Reincorporation by means of the Merger is consummated.

"Effective Time" means the time at which the Articles of Merger with respect to the Merger of Host with and into Host REIT are filed with the SDAT and the Certificate of Merger is filed with the Delaware Secretary of State (or such later time as may be specified in the Articles of Merger).

"Excess FF&E" means replacement FF&E that could cause the average tax basis of the items of the Lessor's FF&E and other personal property that are leased to the applicable Lessee to exceed 15% of the aggregate average tax basis of the real and personal property subject to the applicable Lease.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"ERISA Plan" means an employee benefit plan subject to ERISA.

"FF&E" means furniture, fixtures and equipment.

"FF&E Adjustment" means the amount by which the annual Minimum Rent would be reduced in the event that the average tax basis of the items of the Lessor's FF&E and other personal property that are leased to the applicable Lessee could exceed 15% of the aggregate average tax basis of the real and personal property subject to the applicable Lease.

"FF&E Replacements" means FF&E to be acquired and certain routine repairs that are normally capitalized to be performed in the next year.

"FIRPTA" means the Foreign Investment in Real Property Tax Act of 1980.

"Forum Group" means the Forum Group, Inc., which was acquired by Host on June 21, 1997 from Marriott Senior Living Services, Inc., a subsidiary of Marriott International. The Forum Group holds interests in 31 senior living communities

"Full Participation Scenario" means the REIT Conversion, including the Blackstone Acquisition, occurs, all the Partnerships participate and no Host REIT Common Stock or Notes are issued.

"Funds From Operations" or "FFO" as defined by NAREIT means net income computed in accordance with GAAP, excluding gains or losses from debt restructuring and sales of properties, plus real estate related depreciation and amortization, and after adjustments for unconsolidated partnerships and joint ventures. FFO should not be considered as an alternative to net income, operating profit, cash flows from operations or any other operating or liquidity performance measure prescribed by GAAP. FFO is also not an indicator of funds

available to fund the Company's cash needs, including its ability to make distributions. The Company's method of calculating FFO may be different from methods used by other REITs and, accordingly, is not comparable to such other REITs.

"GAAP" means generally accepted accounting principles.

"Gross Revenues" means proceeds from aggregate sales from a Hotel, including room sales, food and beverage sales and telephone and other income.

"Hanover" means Hanover Marriott Limited Partnership, a Delaware limited partnership.

"HM Services" means Host Marriott Services Corporation.

"HMH Properties" means HMH Properties, Inc.

"Hogan & Hartson" means Hogan & Hartson L.L.P.

"Host" means Host Marriott Corporation, a Delaware corporation, and either the general partner or an affiliate of the general partner of each Partnership, or, as the context may require, Host Marriott Corporation together with its subsidiaries or any of such subsidiaries.

"Host Bylaws" means Host's Amended Bylaws.

"Host Certificate" means Host's Restated Certificate of Incorporation.

"Host Common Stock" means shares of the common stock, par value \$1.00 per share, of Host.

"Host Employee/Charitable Trust" means the Host Marriott Statutory Employee/Charitable Trust, a Delaware statutory business trust, the indirect beneficiaries of which with respect to distributions of income are employees of Host REIT eligible to participate in the Comprehensive Stock Incentive Plan (excluding Directors of Host REIT and certain other highly compensated employees) and the beneficiary of which with respect to capital is one or more charitable organizations specified therein. The Host Employee/Charitable Trust and possibly certain other investors will be the owners of the voting stock of the Non-Controlled Subsidiaries.

"Host REIT" means HMC Merger Corporation, a wholly-owned, newly formed Maryland subsidiary of Host (to be renamed "Host Marriott Corporation" in connection with the Merger), which was formed for the sole purpose of effecting the Reincorporation by means of the Merger, and which will be the sole general partner of the Operating Partnership, and the successor to Host, or, as the context may require, HMC Merger Corporation and the Operating Partnership collectively together with their subsidiaries or any of such subsidiaries.

"Host REIT Bylaws" means the bylaws of Host REIT attached to the Agreement as  $\mathsf{Exhibit}\ \mathsf{C}.$ 

"Host REIT Charter" means the Articles of Incorporation of Host REIT that will be in effect upon consummation of the Merger and the other transactions comprising the REIT Conversion, attached to the Agreement as Exhibit B.

"Host REIT Common Stock" means shares of the common stock, par value \$.01 per share, of Host REIT.

"Host REIT Preferred Stock" means shares of the preferred stock, par value \$.01 per share, of Host REIT.

"Host REIT Right" means one preferred stock purchase right issued under the Host REIT Rights Agreement.

"Host REIT Rights Agreement" means the rights agreement to be adopted by Host REIT substantially as described herein.

"Host Right" means the right issued under the Host Rights Agreement.

"Host Rights Agreement" means the rights agreement dated as of February 3, 1989 between Host and the Bank of New York.

"Hotels" means the approximately 125 full-service hotels operating primarily under the Marriott, Ritz-Carlton, Four Seasons, Swissotel and Hyatt brand names in which the Company and its subsidiaries are expected initially to have controlling interests or own outright following the REIT Conversion and the Blackstone Acquisition.

"Impermissible Tenant Service Income" means any amount charged to a tenant for services rendered by Host REIT or its affiliates other than through an independent contractor from whom Host REIT derives no revenue excluding for these purposes services "usually or customarily rendered" in connection with the rental of real property and not otherwise considered "rendered to the occupant."

"Initial E&P Distribution" means one or more taxable distributions by Host or Host REIT to its stockholders in connection with the REIT Conversion consisting of shares of common stock of Crestline and cash or other consideration in an amount to be determined.

"Interested Stockholder" means a person who owns 10% or more of the voting power of a corporation's then outstanding shares of capital stock or his affiliate.

"IRS" means the Internal Revenue Service.

"IRS Restructuring Act" means the Internal Revenue Service Restructuring and Reform Act of 1998, which was signed into law on July 22, 1998.

"Leases" means the lease agreements under which the Lessees will lease the Hotels from the Operating Partnership.

"Lessees" means the entities to which the Operating Partnership will lease the Hotels and who will operate the Hotels under the existing long-term Management Agreements and pay rent to the Operating Partnership.

"LIBOR" means the London Interbank Offered Rate.

"Limited Partners" means the limited partners, excluding those affiliated with Host, of the Partnerships that are participating in the Partnership Mergers.

"Management Agreements" means the existing long-term management agreements pursuant to which the Managers manage the Hotels.

"Managers" means the subsidiaries of Marriott International and other companies who manage the Hotels on behalf of Host, the Partnerships or the Private Partnerships (and following the REIT Conversion, on behalf of the Lessees) pursuant to the existing Management Agreements.

"Market Leasing Factor" means the amount used to determine the FF&E Adjustment to Minimum Rent in the event that the average tax basis of the items of the Lessor's FF&E and other personal property that are leased to the applicable Lessee could exceed 15% of the aggregate average tax basis of the real and personal property subject to the applicable Lease. The Market Leasing Factor will be determined for the first two years under a Lease at the time the Lease is executed. Each year thereafter, the Market Leasing Factor will be determined by an independent valuation expert based upon the median of the leasing rates of at least three nationally recognized companies engaged in the business of leasing similar personal property.

"Marriott International" means Marriott International, Inc., a Delaware corporation.

"Marriott International Purchase Right" means the right of Marriott International to purchase up to 20% of each class of Host's outstanding voting shares at the then fair market value upon the occurrence of certain change of control events involving Host, which will continue in effect after the Merger until June 2017, subject to certain limitations intended to protect the REIT status of Host REIT.

"MDAH" means Marriott Diversified American Hotels, L.P., a Delaware limited partnership.

"Merger" means the proposed merger of Host with and into Host REIT, as described in this Proxy Statement/Prospectus.

"MGCL" means the Maryland General Corporation Law, as amended.

"MHP" means Marriott Hotel Properties Limited Partnership, a Delaware limited partnership, or, as the context may require, such entity together with its subsidiaries, or any of such subsidiaries.

"MHP2" means Marriott Hotel Properties II Limited Partnership, a Delaware limited partnership, or, as the context may require, such entity together with its subsidiaries, or any of such subsidiaries.

"Minimum Rent" means a fixed dollar amount specified in each lease, less the FF&E Adjustment.

"NAREIT" means the National Association of Real Estate Investment Trusts,

"New Credit Facility" means the new \$1.25 billion credit facility of the Operating Partnership.

"Non-Controlled Subsidiaries" means the taxable corporations in which the Operating Partnership will own 95% of the economic interest but no voting stock and which will hold various assets contributed by Host and its subsidiaries to the Operating Partnership, which assets, if owned directly by the Operating Partnership, could jeopardize Host REIT's status as a REIT.

"Non-U.S. Stockholders" means persons that are, for purposes of federal income taxation, nonresident alien individuals, foreign corporations, foreign partnerships or foreign estates or trusts.

"Note" means an unsecured note of the Operating Partnership which a Limited Partner may elect to receive in connection with the Partnership Mergers in exchange for OP Units.

"NYSE" means the New York Stock Exchange, Inc.

"OP Contribution" means the contribution by Host of its wholly-owned full-service hotels, its interests in certain hotel partnerships and certain other assets to the Operating Partnership in exchange for units of limited and general partnership interest in the Operating Partnership and the assumption of liabilities.

"OP Units" means the limited partnership interests in the Operating Partnership.

"Operating Partnership" means Host Marriott, L.P., a Delaware limited partnership, the sole general partner of which will be Host REIT, or, as the context may require, such entity together with its subsidiaries, including the Non-Controlled Subsidiaries, or any of them.

"Other Plans" means an IRA, qualified pension, profit sharing or stock bonus plan, or any other plan that is not an ERISA Plan but is described in Section 4795(e)(1) of the Code.

"Ownership Limit" means the prohibition in the Host REIT Charter against ownership, directly or indirectly or by virtue of the attribution provisions of the Code, by any person or persons acting as a group of more than (i) 9.8% of the lesser of the number or value of shares of Host REIT Common Stock outstanding or (ii) 9.8% of the lesser of the number or value of the issued and outstanding shares of any class or series of Host REIT preferred stock, subject to waiver or modification by Host REIT in certain limited circumstances and to certain limited exceptions for a holder of shares of Host REIT Common Stock solely by reason of the Merger in excess of the Ownership Limit so long as such holder would not own, directly or by attribution under the Code, more than 9.9% by value of the outstanding capital stock of Host REIT as of the Special Merger Ownership Limit Effective Time.

"Partnership" means any of the eight limited partnerships that own full-service hotels in which Host or its subsidiaries are general partners (Atlanta Marquis, Chicago Suites, Desert Springs, Hanover, MDAH, MHP, MHP2 or PHLP), or, as the context may require, any such entity together with its subsidiaries, or any of such subsidiaries.

"Partnership Agreement" means the amended and restated agreement of limited partnership of the Operating Partnership.

"Partnership Mergers" means the acquisition by merger, if and to the extent consummated, of up to eight of the Partnerships that own full-service hotels in which Host or its subsidiaries are general partners in exchange for OP Units (which, in turn, may be exchanged for Host REIT Common Stock or Notes).

"Percentage Rent" means an amount of rent based upon specified percentages of aggregate sales (including room sales, food and beverage sales and telephone and other sales) at each Hotel which is subject to a Lease.

"Person" means an individual, corporation, partnership, limited liability company, trust or other entity.

"PHLP" means Potomac Hotel Limited Partnership, a Delaware limited partnership.

"Plan Assets" means the underlying assets of Host REIT which are deemed to be assets of an investing ERISA Plan.

"Plans" means, collectively, the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan (formerly called the Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan), the Host Marriott Corporation Employee Stock Purchase Plan, the Host Marriott Corporation (HMC) Retirement and Savings Plan and the Host Marriott Corporation Executive Deferred Compensation Plan.

"Private Partnership" means a partnership (other than a Partnership) or limited liability company that owns one or more full-service Hotels and that, prior to the REIT Conversion, is partially but not wholly-owned by Host or one of its subsidiaries. The Private Partnerships are not participating in the Partnership Mergers.

"Private Partnership Transactions" means the acquisition of the partnership interests from four Private Partnerships that own one or more full-service hotels and that, prior to the REIT Conversion, are partially but not whollyowned by Host or one of its subsidiaries in exchange for OP Units, if and to the extent such acquisitions are consummated.

"Prohibited Owner" means a Person holding record title to any shares in excess of the Ownership Limit.

"Prohibited Transferee" means a Person who would violate the Ownership Limit or any other restriction in the Host REIT Charter because of a transfer of shares of beneficial interest of Host REIT to such Person or any other event.

"Proxy Statement/Prospectus" means this Proxy Statement/Prospectus, dated November , 1998, including the appendices attached hereto.

"Recognition Period" means the ten-year period beginning on the date on which a Built-In Gain Asset is acquired by Host REIT.

"Record Date" means November 13, 1998.

"Reincorporation" means the reincorporation of Host from the State of Delaware to the State of Maryland by means of the Merger.

"REIT" means a real estate investment trust.

"REIT Conversion" means (i) the OP Contribution, (ii) the Reincorporation by means of the Merger, (iii) the Initial E&P Distribution, (iv) the recently completed refinancing and amendment of the debt securities and certain credit facilities of Host substantially in the manner described herein, (v) the Partnership Mergers (if and to the extent consummated), (vi) the Private Partnership Transactions (if and to the extent consummated), (vii) the Blackstone Acquisition (if and to the extent consummated), (vii) the creation and capitalization of the Non-Controlled Subsidiaries, (ix) the leasing of virtually all of the full-service Hotels owned or controlled by the Operating Partnership to the Lessees and (x) such other related transactions described in "The REIT Conversion" and such other steps occurring prior to, concurrent with or within a reasonable time after the Effective Date, as Host may determine in its sole discretion to be necessary or desirable to complete or facilitate the transactions contemplated herein or otherwise to permit Host REIT to elect to be treated as a REIT for federal income tax purposes for the first full taxable year commencing after the REIT Conversion.

"Related Party Tenant" means a tenant in which Host REIT, or an actual or constructive owner of 10% or more of Host REIT, actually or constructively owns 10% or more of the interests.

"REVPAR" means revenue per available room. REVPAR measures daily room revenues generated on a per room basis by combining the average daily room rate charged and the average daily occupancy achieved. REVPAR excludes food and beverage and other ancillary revenues generated by the hotel.

"Rights Distribution Date" means the earlier to occur of (i) 10 days following the Stock Acquisition Date or (ii) 10 business days (or such later date as the Board of Directors may determine) following the commencement of a tender offer or exchange offer, the consummation of which would result in the beneficial ownership by a person of 20% or more of the outstanding Host REIT Common Stock.

"SAR" means stock appreciation rights.

"SDAT" means the Maryland State Department of Assessments and Taxation.

"Securities Act" means the Securities Act of 1933, as amended.

"Senior Note Refinancing" means the recently completed refinancing of 1.55 billion of outstanding senior notes through offers to purchase such debt securities for cash and a concurrent solicitation of consents to amend the terms of the debt securities to facilitate the transactions constituting the REIT Conversion.

"SFAS" means Statement of Financial Accounting Standard.

"Special Dividend" means the currently contemplated form of the "other" consideration comprising the Initial E&P Distribution, consisting of a special dividend payable, at each stockholder's election, in cash or Host Common Stock (or Host REIT Common Stock if the Merger has occurred).

"Special Meeting" means the special meeting of stockholders of Host to be held at the Ritz-Carlton Hotel, 1700 Tysons Boulevard, McLean, Virginia, 22102, on December 15, 1998 at 10:00 a.m., local time, and any adjournments or postponements thereof.

"Special Merger Ownership Limit Effective Time" means 12:01 a.m. (Eastern Time) on the first day after the Effective Date.

"Stock Acquisition Date" means the date of a public announcement that an Acquiring Person has acquired, or obtained the right to acquire, beneficial ownership of 20% or more of the outstanding Host REIT Common Stock.

"Subsidiary Partnerships" means partnerships or limited liability companies in which either the Operating Partnership, the Partnerships or the Private Partnerships have an interest.

"Transfer and Exchange Agent" means First Chicago Trust Company of New York, the transfer agent, the exchange agent and registrar for Host REIT Common Stock

"Treasury Regulations" means the regulations promulgated by the IRS under the Code.  $\label{eq:code_state}$ 

"U.S. Stockholder" means a holder of Host REIT Common Stock who (for United States federal income tax purposes) is (i) a citizen or resident of the United States, (ii) a corporation, partnership, or other entity created or organized in or under the laws of the United States or any political subdivision thereof, (iii) an estate or trust the income of which is subject to United States federal income taxation regardless of its source, or (iv) a trust whose administration is subject to the primary supervision of a United States court and which has one or more United States persons who have authority to control all substantial decisions of the trust.

"UBTI" means unrelated business taxable income.

"Unit Redemption Right" means the right of Limited Partners who retain OP Units to redeem, beginning one year following the effective date of the Partnership Mergers, their OP Units and receive, at Host REIT's election, either Host REIT Common Stock on a one-for-one basis (subject to adjustment) or cash in an amount equal to the market value of such shares.

"UPREIT" means an umbrella partnership real estate investment trust.

226

# INDEX TO FINANCIAL STATEMENTS

The following financial information is included on the pages indicated:

# HISTORICAL FINANCIAL STATEMENTS

	PAGE
HOST MARRIOTT CORPORATION	
Report of Independent Public Accountants	F-3
Consolidated Balance Sheets as of January 2, 1998 and January 3, 1997 Consolidated Statements of Operations for the Fiscal Years Ended January	F-4
2, 1998, January 3, 1997 and December 29, 1995	F-5
Ended January 2, 1998, January 3, 1997 and December 29, 1995	F-6
2, 1998, January 3, 1997 and December 29, 1995	F-7
Notes to Consolidated Financial Statements	F-8
(unaudited)	F-31
Ended September 11, 1998 and September 12, 1997 (unaudited)	F-32
Ended September 11, 1998 and September 12, 1997 (unaudited)	F-33
Notes to Condensed Consolidated Financial Statements (unaudited) HOST MARRIOTT HOTELS	F-34
Report of Independent Public Accountants	F-43
Combined Consolidated Balance Sheets as of January 2, 1998 and January	
3, 1997  Combined Consolidated Statements of Operations for the Fiscal Years	F-44
Ended January 2, 1998, January 3, 1997 and December 29, 1995 Combined Consolidated Statements of Cash Flows for the Fiscal Years	F-45
Ended January 2, 1998, January 3, 1997 and December 29, 1995	F-46
Notes to Combined Consolidated Financial Statements	F-47
(unaudited)	F-71
Condensed Combined Consolidated Statements of Operations for the Thirty- Six Weeks Ended September 11, 1998 and September 12, 1997 (unaudited)	F-72
Condensed Combined Consolidated Statements of Cash Flows for the Thirty- Six Weeks Ended September 11, 1998 and September 12, 1997 (unaudited)	F-73
Notes to Condensed Combined Consolidated Financial Statements	
(unaudited)	F-74
OF HOST MARRIOTT CORPORATION (TENANT FINANCIAL STATEMENTS)	LINESS
Report of Independent Public Accountants	F-81
Consolidated Balance Sheet as of January 2, 1998	F-82
Consolidated Statement of Operations for the Period from June 21, 1997 (inception) through January 2, 1998	F-83
Consolidated Statement of Shareholder's Equity for the Period from June	
21, 1997 (inception) through January 2, 1998	F-84
(inception) through January 2, 1998	F-85
Notes to Consolidated Financial Statements	F-86
(unaudited)	F-96
Condensed Consolidated Statement of Operations for the Thirty-Six Weeks Ended September 11, 1998 and the Twelve Weeks Ended September 12, 1997	
(unaudited)	F-97
Condensed Consolidated Statement of Cash Flows for the Thirty-Six Weeks	
Ended September 11, 1998 and the Twelve Weeks Ended September 12, 1997 (unaudited)	F-98
Notes to Condensed Consolidated Financial Statements	F-90
HOST MARRIOTT L.P.	
Report of Independent Public Accountants	
Balance Sheet as of September 11, 1998 (unaudited) and June 19, 1998	
Notes to Balance Sheet	L-T03

HMC MERGER CORPORATION Report of Independent Public Accountants	
Notes to Balance Sheet	
UNAUDITED PRO FORMA FINANCIAL STATEMENTS	
OPERATING PARTNERSHIP	
Introduction to Unaudited Pro Forma Financial Statements of the Operating	F 100
Partnership	F-108
100% Participation with No Notes Issued Pro Forma Balance Sheet as of September 11, 1998(1)	г 110
Pro Forma Statement of Operations for the First Three Quarters 1998(1)	
Pro Forma Statement of Operations for Fiscal Year 1997(1)	
Pro Forma Statement of Cash Flows for the First Three Quarters 1998	
Pro Forma Statement of Cash Flows for Fiscal Year 1997	F-121
100% Participation with Notes Issued	. 121
Pro Forma Balance Sheet as of June 19, 1998(1)	F-123
Pro Forma Statement of Operations for the First Three Quarters 1998(1)	
Pro Forma Statement of Operations for Fiscal Year 1997(1)	
Pro Forma Statement of Cash Flows for the First Three Quarters 1998	
Pro Forma Statement of Cash Flows for Fiscal Year 1997	F-135
No Partnership Participation	
Pro Forma Balance Sheet as of September 11, 1998(1)	F-137
Pro Forma Statement of Operations for the First Three Quarters 1998(1)	
Pro Forma Statement of Operations for Fiscal Year 1997(1)	
Pro Forma Statement of Cash Flows for the First Three Quarters 1998	
Pro Forma Statement of Cash Flows for Fiscal Year 1997	F-147
HOST REIT	
Introduction to Unaudited Pro Forma Financial Statements of Host REIT	F-149
100% Participation with No Notes Issued	- 4-0
Pro Forma Balance Sheet as of September 11, 1998(1)	
Pro Forma Statement of Operations for the First Three Quarters 1998(1)	
Pro Forma Statement of Operations for Fiscal Year 1997(1)	F-152
Pro Forma Balance Sheet as of September 11, 1998(1)	C-155
Pro Forma Statement of Operations for the First Three Quarters 1998(1)	
Pro Forma Statement of Operations for Fiscal Year 1997(1)	
No Partnership Participation	1 137
Pro Forma Balance Sheet as of September 11, 1998(1)	F-160
Pro Forma Statement of Operations for the First Three Quarters 1998(1)	
Pro Forma Statement of Operations for Fiscal Year 1997(1)	
CRESTLINE	
Introduction to Unaudited Pro Forma Financial Statements of Crestline	F-164
All Partnerships Participate	
Pro Forma Balance Sheet as of September 11, 1998(1)	F-166
Pro Forma Statement of Operations for the First Three Quarters 1998(1)	
Pro Forma Statement of Operations for Fiscal Year 1997(1)	F-168
No Partnership Participation	
Pro Forma Balance Sheet as of September 11, 1998(1)	
Pro Forma Statement of Operations for the First Three Quarters 1998(1)	
Pro Forma Statement of Operations for Fiscal Year 1997(1)	⊦-175
	. +

PAGE

(1) Pro forma statement includes a final column to show the pro forma effect of a delay in the effectiveness of Host's REIT Conversion until January 1, 2000 (the "REIT 2000 Pro Forma").

#### REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To Host Marriott Corporation:

We have audited the accompanying consolidated balance sheets of Host Marriott Corporation and subsidiaries as of January 2, 1998 and January 3, 1997, and the related consolidated statements of operations, shareholders' equity and cash flows for each of the three fiscal years in the period ended January 2, 1998. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform an audit to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Host Marriott Corporation and subsidiaries as of January 2, 1998 and January 3, 1997, and the results of their operations and their cash flows for each of the three fiscal years in the period ended January 2, 1998, in conformity with generally accepted accounting principles.

As discussed in Notes 1 and 2 to the consolidated financial statements, in 1995 the Company changed its method of accounting for the impairment of long-lived assets.

Arthur Andersen LLP

Washington, D.C. February 27, 1998

# CONSOLIDATED BALANCE SHEETS

# JANUARY 2, 1998 AND JANUARY 3, 1997 (IN MILLIONS)

	1997	1996
ASSETS		
Property and Equipment, net	\$5,217	\$3,805
affiliates of \$23 million and \$156 million, respectively)	54	297
Due from Managers	93 13	89 11
Other Assets	284	246
Short-term Marketable Securities	354 511	
Casii aliu Casii Equivalents	311	
	\$6,526 =====	. ,
LIABILITIES AND SHAREHOLDERS' EQUITY Debt		
Senior Notes Issued by the Company or its Subsidiaries	\$1,585	\$1,021
Mortgage Debt	1,979	
Other	219	97
		2,647
Accounts Payable and Accrued Expenses  Deferred Income Taxes	97 508	74 464
Other Liabilities	388	290
Total Liabilities	4,776	
Company-obligated Mandatorily Redeemable Convertible Preferred Securities of a Subsidiary Trust Holding Company Substantially All of Whose Assets are the Convertible Subordinated Debentures Due 2026 ("Convertible Preferred Securities")	550	
Shareholders' Equity		
Common Stock, 600 million shares authorized; 203.8 million shares in 1997 and 202.0 million shares in 1996 issued and		
outstanding		202
Additional Paid-in Capital	947	926 (1)
<b>5</b>		
Total Shareholders' Equity	1,200	1,127
	\$6,526 =====	\$5,152

# CONSOLIDATED STATEMENTS OF OPERATIONS

FISCAL YEARS ENDED JANUARY 2, 1998, JANUARY 3, 1997 AND DECEMBER 29, 1995 (IN MILLIONS, EXCEPT PER COMMON SHARE AMOUNTS)

	1997	1996	
REVENUES Hotels Senior living communities	37	\$ 717 	\$ 474  (3)
Net gains (losses) on property transactions Equity in earnings of affiliates	5	3 11	
Other			
Total revenues	1,147		
OPERATING COSTS AND EXPENSES  Hotels (including Marriott International management fees of \$162 million, \$101 million and \$67 million, respectively)	649	461	291
Senior living communities (including Marriott			
International management fees of \$6 million in 1997) Other (including a \$60 million write-down of undeveloped			
land in 1995)			
Total operating costs and expenses	698		
OPERATING PROFIT BEFORE MINORITY INTEREST, CORPORATE EXPENSES AND INTEREST	449	233	114
Minority interest	(32)	(6)	(2)
Corporate expenses		(43) (237)	
Dividends on Convertible Preferred Securities of subsidiary trust	(37)	, ,	, ,
Interest income	52	48	27
INCOME (LOSS) FROM CONTINUING OPERATIONS BEFORE INCOME			
TAXES Benefit (provision) for income taxes	(36)	(8) (5)	13
INCOME (LOSS) FROM CONTINUING OPERATIONS DISCONTINUED OPERATIONS Loss from discontinued operations (net of income tax	47	(13)	
benefit of \$3 million in 1995)  Provision for loss on disposal (net of income tax			
benefit of \$23 million in 1995)			(53)
INCOME (LOSS) BEFORE EXTRAORDINARY ITEMS  Extraordinary itemsGain (loss) on extinguishment of debt (net of income tax expense (benefit) of \$1 million	47	(13)	(123)
in 1997 and (\$10) million in 1995)	3		
NET INCOME (LOSS)		\$ (13)	\$(143)
BASIC EARNINGS (LOSS) PER COMMON SHARE:			
CONTINUING OPERATIONS			(.39)
debt (net of income taxes)	.02		(.12)
BASIC EARNINGS (LOSS) PER COMMON SHARE		\$(.07)	\$(.90)
DILUTED EARNINGS (LOSS) PER COMMON SHARE:			
CONTINUING OPERATIONS	\$ .23		(.39)
debt (net of income taxes)	.01		,
DILUTED EARNINGS (LOSS) PER COMMON SHARE		\$(.07)	\$(.90)

# CONSOLIDATED STATEMENTS OF SHAREHOLDERS' EQUITY

FISCAL YEARS ENDED JANUARY 2, 1998, JANUARY 3, 1997 AND DECEMBER 29, 1995

COMMON SHARES OUTSTANDING		CONVERTIBLE PREFERRED STOCK	COMMON STOCK	ADDITIONAL PAID-IN CAPITAL	RETAINED EARNINGS
(IN MILLIONS)	1		(IN MIL		
153.6 	Balance, December 30, 1994 Net loss Distribution of stock of Host Marriott Services Corporation	\$ 13 	\$154 	\$479  (4)	\$ 64 (143)
1.3	Common stock issued for the comprehensive stock and employee stock purchase plans		1	16	
4.8	stock to common stock	(13)	5 	8	
159.7  3.9 6.8	Balance, December 29, 1995 Net loss Adjustment to Host Marriott Services Dividend Common stock issued for the comprehensive stock and employee stock purchase plans Common stock issued for warrants exercised Common stock issued in	  	160   3 7	499   17 42	16 (13) (4)
31.6	stock offering		32	368 	
202.0	Balance, January 3, 1997 Net income Common stock issued for the comprehensive stock and employee stock purchase plans		202	926  21	(1) 50
203.8	Balance, January 2, 1998	\$	\$204	\$947	\$ 49

# CONSOLIDATED STATEMENTS OF CASH FLOWS

FISCAL YEARS ENDED JANUARY 2, 1998, JANUARY 3, 1997 AND DECEMBER 29, 1995

	1997		1995
	(IN MILLIONS)		
OPERATING ACTIVITIES Income (loss) from continuing operations	\$ 47	\$ (13)	\$ (62)
Depreciation and amortization	240 (20) (4)	168 (35) (6)	122 (35) (7)
Net (gains) losses on property transactions Equity in earnings of affiliates Other	19 (5) 60	(3) 49	70  33
Changes in operating accounts: Other assetsOther liabilities		9 32	(2) (9)
Cash from continuing operations	464	205 (4)	110 32
Cash from operations		201	142
INVESTING ACTIVITIES Proceeds from sales of assets	51	373 (35)	358
Cash received from sales of assets	51	338	325
Capital expenditures: Capital expenditures for renewals and replacements Lodging construction funded by project financing	(131)	(87)	(56) (40)
New investment capital expenditures  Purchases of short-term marketable securities  Notes receivable collections	(29) (354) 6	(69)	(64)  43
Affiliate notes receivable and collections, net Other	(6)	21	2
Cash used in investing activities from continuing operations	(1,046)	(504)	(156)
operations			()
Cash used in investing activities	(1,046)		
FINANCING ACTIVITIES Issuances of debt			1,251
Issuances of common stock	6 (93) (403) 22	454 (82) (173) 28	13 (100) (960)
Cash from financing activities from continuing			
operations		806	204
operations  Cash from financing activities	389	806	(4)  200
INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS CASH AND CASH EQUIVALENTS, beginning of year	704	201	134 67
CASH AND CASH EQUIVALENTS, end of year		\$ 704	
Non-cash financing activities: Assumption of mortgage debt for the acquisition of, or purchase of controlling interests in, certain hotel properties and senior living communities	\$ 733 ======		\$ 141

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

# 1. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

#### Description of Business

As of January 2, 1998, Host Marriott Corporation (the "Company") owned, or had controlling interests in, 95 upscale and luxury full-service hotel lodging properties generally located throughout the United States and operated under the Marriott and Ritz-Carlton brand names. Most of these properties are managed by Marriott International, Inc. ("Marriott International"). At that date, the Company also held minority interests in various partnerships that own 242 additional properties, including 22 full-service hotel properties, managed by Marriott International. The Company also owned a portfolio consisting of 30 premier senior living communities as of January 2, 1998, all of which are managed by Marriott Senior Living Services, Inc. ("MSLS"), a subsidiary of Marriott International.

On December 29, 1995, the Company distributed to its shareholders through a special tax-free dividend (the "Special Dividend") its food, beverage, and merchandise concessions business at airports, on tollroads, and at arenas and other attractions (the "Operating Group"). See Note 2 for a discussion of the Special Dividend. The 1995 consolidated financial statements were restated to reflect the Operating Group as discontinued operations.

The structure of the Company was substantially altered on October 8, 1993 (the "Marriott International Distribution Date") when the Company distributed the stock of a wholly-owned subsidiary, Marriott International, Inc., in a special dividend (the "Marriott International Distribution"). See Note 14 for a description of the Marriott International Distribution and related transactions.

#### Principles of Consolidation

The consolidated financial statements include the accounts of the Company and its subsidiaries and controlled affiliates. Investments in affiliates over which the Company has the ability to exercise significant influence, but does not control, are accounted for using the equity method. All material intercompany transactions and balances have been eliminated.

#### Fiscal Year

The Company's fiscal year ends on the Friday nearest to December 31. Fiscal years 1997 and 1995 included 52 weeks compared to 53 weeks for fiscal year 1996

## Revenues and Expenses

Revenues primarily represent house profit from the Company's hotel properties and senior living communities because the Company has delegated substantially all of the operating decisions related to the generation of house profit from its hotel properties and senior living communities to the manager. Revenues also include net gains (losses) on property transactions and equity in the earnings of affiliates. House profit reflects the net revenues flowing to the Company as property owner and represents hotel properties' and senior living communities' operating results, less property-level expenses, excluding depreciation, management fees, real and personal property taxes, ground and equipment rent, insurance and certain other costs, which are classified as operating costs and expenses in the accompanying consolidated financial statements. See Note 18.

On November 20, 1997, the Emerging Issues Task Force ("EITF") of the Financial Accounting Standards Board reached a consensus on EITF 97-2, "Application of FASB Statement No. 94 and APB Opinion No. 16 to Physician Practice Management Entities and Certain Other Entities with Contractual Management Arrangements." EITF 97-2 addresses the circumstances in which a management entity may include the revenues and expenses of a managed entity in its financial statements.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

The Company has considered the impact of EITF 97-2 on its financial statements and has determined that EITF 97-2 requires the Company to include property-level sales and operating expenses of its hotels and senior living communities in its statements of operations (see Note 18). The Company will adopt EITF 97-2 in the fourth quarter of 1998 with retroactive effect in prior periods to conform to the new presentation. Application of EITF 97-2 to the consolidated financial statements for the fiscal years ended January 2, 1998, January 3, 1997 and December 29, 1995 would have increased both revenues and operating expenses by approximately \$1.7 billion, \$1.2 billion and \$.9 billion, respectively, and would have had no impact on operating profit, net income or earnings per share.

#### Earnings (Loss) Per Common Share

Basic earnings per common share are computed by dividing net income by the weighted average number of shares of common stock outstanding. Diluted earnings per common share are computed by dividing net income by the weighted average number of shares of common stock outstanding plus other dilutive securities. Diluted earnings per common share has not been adjusted for the impact of the Convertible Preferred Securities for 1997 and 1996 and for the comprehensive stock plan and warrants for 1996 and 1995 as they are antidilutive.

A reconciliation of the number of shares utilized for the calculation of dilutive earnings per common share follows:

	1997 	1996 	1995
Weighted average number of common shares outstanding Assuming distribution of common shares granted under comprehensive stock plan, less shares assumed purchased at average	203.1	188.7	158.3
market price	4.8		
Assuming distribution of common shares issuable for warrants,	•		
less shares assumed purchased at average market price	. 3		
Shares utilized for the calculation of diluted earnings per			
share	208.2	188.7	158.3
	=====	=====	=====

# International Operations

The consolidated statements of operations include the following amounts related to non-U.S. subsidiaries and affiliates: revenues of \$39 million and \$18 million and loss before income taxes of \$9 million and \$2 million in 1997 and 1996, respectively. International revenues and income before income taxes in 1995 were not material.

#### Property and Equipment

Property and equipment is recorded at cost. For newly developed properties, cost includes interest, rent and real estate taxes incurred during development and construction. Replacements and improvements are capitalized.

Depreciation is computed using the straight-line method over the estimated useful lives of the assets, generally 40 years for buildings and three to ten years for furniture and equipment. Leasehold improvements are amortized over the shorter of the lease term or the useful lives of the related assets.

Gains on sales of properties are recognized at the time of sale or deferred to the extent required by generally accepted accounting principles. Deferred gains are recognized as income in subsequent periods as conditions requiring deferral are satisfied or expire without further cost to the Company.

In cases where management is holding for sale particular hotel properties or senior living communities, the Company assesses impairment based on whether the estimated sales price less costs of disposal of each

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

individual property to be sold is less than the net book value. A property is considered to be held for sale when the Company has made the decision to dispose of the property. Otherwise, the Company assesses impairment of its real estate properties based on whether it is probable that undiscounted future cash flows from each individual property will be less than its net book value. If a property is impaired, its basis is adjusted to its fair market value.

#### Deferred Charges

Deferred financing costs related to long-term debt are deferred and amortized over the remaining life of the debt.

#### Cash, Cash Equivalents and Short-term Marketable Securities

The Company considers all highly liquid investments with a maturity of 90 days or less at the date of purchase to be cash equivalents. Cash and cash equivalents includes approximately \$115 million and \$67 million at January 2, 1998 and January 3, 1997, respectively, of cash related to certain consolidated partnerships, the use of which is restricted generally for partnership purposes to the extent it is not distributed to the partners. Short-term marketable securities include investments with a maturity of 91 days to one year at the date of purchase. The Company's short-term marketable securities represent investments in U.S. government agency notes and high quality commercial paper. The short-term marketable securities are categorized as available for sale and, as a result, are stated at fair market value. Unrealized holding gains and losses are included as a separate component of shareholders' equity until realized.

#### Concentrations of Credit Risk

Financial instruments that potentially subject the Company to significant concentrations of credit risk consist principally of cash, cash equivalents and short-term marketable securities. The Company maintains cash and cash equivalents and short-term marketable securities with various high credit-quality financial institutions. The Company performs periodic evaluations of the relative credit standing of these financial institutions and limits the amount of credit exposure with any institution.

#### Use of Estimates in the Preparation of Financial Statements

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

#### Self-Insurance Programs

Prior to the Marriott International Distribution Date, the Company was self-insured for certain levels of general liability, workers' compensation and employee medical coverage. Estimated costs of these self-insurance programs were accrued at present values of projected settlements for known and anticipated claims. The Company discontinued its self-insurance programs for claims arising subsequent to the Marriott International Distribution Date.

#### Interest Rate Swap Agreements

The Company has entered into a limited number of interest rate swap agreements to diversify certain of its debt to a variable rate or fixed rate basis. The interest rate differential to be paid or received on interest rate swap agreements is accrued as interest rates change and is recognized as an adjustment to interest expense.

# NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

New Statements of Financial Accounting Standards

The Company adopted Statements of Financial Accounting Standards ("SFAS") No. 114, "Accounting by Creditors for Impairment of a Loan" and SFAS No. 121, "Accounting for the Impairment of Long-Lived Assets and for Long-Lived Assets to be Disposed Of" during 1995. Adoption of these statements did not have a material effect on the Company's continuing operations. See Note 2 for a discussion of the adoption of SFAS No. 121 on discontinued operations.

During 1996, the Company adopted SFAS No. 123, "Accounting for Stock-Based Compensation." The adoption of SFAS No. 123 did not have a material effect on the Company's consolidated financial statements. See Note 10.

During 1997, the Company adopted SFAS No. 128, "Earnings Per Share;" SFAS No. 129, "Disclosure of Information About Capital Structure" and SFAS No. 131, "Disclosures About Segments of an Enterprise and Related Information." The adoption of these statements did not have a material effect on the Company's consolidated financial statements and the appropriate disclosures required by these statements have been incorporated herein. The Company will adopt SFAS No. 130, "Reporting Comprehensive Income," in 1998 and does not expect it to have a material effect on the Company's consolidated financial statements.

#### 2. HM SERVICES SPECIAL DIVIDEND

On December 29, 1995, the Company distributed to its shareholders through the Special Dividend all of the outstanding shares of common stock of Host Marriott Services Corporation ("HM Services"), formerly a wholly-owned subsidiary of the Company, which, as of the date of the Special Dividend, owned and operated food, beverage and merchandise concessions at airports, on tollroads and at stadiums and arenas and other tourist attractions. The Special Dividend provided Company shareholders with one share of common stock of HM Services for every five shares of Company common stock held by such shareholders on the record date of December 22, 1995. The Company recorded approximately \$9 million of expenses related to the consummation of the Special Dividend in 1995. Revenues for the Company's discontinued operations totaled \$1,158 million in 1995. The provision for loss on disposal includes the operating loss from discontinued operations from August 9, 1995 (measurement date) through December 29, 1995 of \$44 million, net of taxes, and estimated expenses related to the Special Dividend of \$9 million.

Effective September 9, 1995, the Company adopted SFAS No. 121, which requires that an impairment loss be recognized when the carrying amount of an asset exceeds the sum of the undiscounted estimated future cash flows associated with the asset. As a result of the adoption of SFAS No. 121, the Company recognized a non-cash, pre-tax charge during the fourth quarter of 1995 of \$47 million. Such charge has been reflected in discontinued operations for fiscal year 1995.

For purposes of governing certain of the ongoing relationships between the Company and HM Services after the Special Dividend and to provide for an orderly transition, the Company and HM Services entered into various agreements including a Distribution Agreement, an Employee Benefits Allocation Agreement, a Tax Sharing Agreement and a Transitional Services Agreement. Effective as of December 29, 1995, these agreements provide, among other things, for the division between the Company and HM Services of certain assets and liabilities, including but not limited to liabilities related to employee stock and other benefit plans and the establishment of certain obligations for HM Services to issue shares upon exercise of warrants (see Note 7) and to issue shares or pay cash to the Company upon exercise of stock options held by certain former employees of the Company (see Note 10).

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

#### 3. PROPERTY AND EQUIPMENT

Property and equipment consists of the following:

	1997	1996
	(IN MIL	LIONS)
Land and land improvements	4,796	\$ 349 3,507 548
Construction in progress	35	82
Less accumulated depreciation and amortization	,	4,486 (681)
	\$5,217 =====	\$3,805 =====

Interest cost capitalized in connection with the Company's development and construction activities totaled \$1 million in 1997, \$3 million in 1996 and \$5 million in 1995.

In 1997, the Company, through an agreement with the ground lessor of one of its properties terminated its ground lease and recorded a \$15 million loss on the write-off of its investment, including certain transaction costs, which has been included in net gains (losses) on property transactions in the accompanying consolidated financial statements.

In 1996, the Company recorded additional depreciation expense of \$15 million as a result of a change in the estimated depreciable lives and salvage values for certain hotel properties. Also, in 1996, the Company recorded a \$4 million charge to write down an undeveloped land parcel to its net realizable value based on its expected sales value.

In 1995, the Company made a determination that its owned Courtyard and Residence Inn properties were held for sale and recorded a \$10 million charge to write down the carrying value of five of these individual properties to their estimated net realizable values. In the fourth quarter of 1995, management instituted a program to liquidate certain non-income producing assets and to reinvest the proceeds in the acquisition of full-service hotels. As part of this program, management determined that a 174-acre parcel of undeveloped land in Germantown, Maryland that was to be developed into an office project over an extended period of time would no longer be developed and instead decided to attempt to sell the property. Accordingly, the Company recorded a pre-tax charge of \$60 million in the fourth quarter of 1995 to reduce the asset to its estimated sales value. In 1997, the Company sold a portion of the land parcel at its approximate net book value of \$11 million.

# 4. INVESTMENTS IN AND RECEIVABLES FROM AFFILIATES

Investments in and receivables from affiliates consist of the following:

	OWNERSHIP INTERESTS		97  MIL		996  IS)
Equity investments Hotel partnerships which own 22 full-service Marriott Hotels, 120 Courtyard hotels, 50 Residence Inns and 50 Fairfield Inns operated by Marriott International, as of January 2,					
1998 Notes and other receivables, net	1%-50% 	\$	13 23	\$	11 156
		\$	36	\$	167
		===	===	===	:====

Hotel properties owned by affiliates generally were acquired from the Company in connection with limited partnership offerings. The Company or one of its subsidiaries typically serve as a general partner of each partnership and the hotels are operated by Marriott International under long-term agreements.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

In 1997, the Company acquired all of the outstanding interests in the Chesapeake Hotel Limited Partnership ("CHLP") that owns six hotels and acquired controlling interests in four affiliated partnerships for approximately \$550 million, including the assumption of approximately \$410 million of debt. These affiliated partnerships included the partnerships that own the 353-room Hanover Marriott; the 884-room Marriott's Desert Springs Resort and Spa; the Marriott Hotel Properties Limited Partnership ("MHPLP") that owns the 1,503-room Marriott Orlando World Center and a 50.5% interest in the 624-room Marriott Harbor Beach Resort; and the partnership that owns the 418-unit Leisure Park retirement community. Subsequent to year-end, the Company obtained a controlling interest in the partnership that owns the 1,671-room Atlanta Marriott Marquis for approximately \$239 million, including the assumption of \$164 million of mortgage debt.

In 1996, the Company purchased controlling interests in four affiliated partnerships for \$640 million, including \$429 million of existing debt. These affiliated partnerships included the partnership that owns the 1,355-room San Diego Marriott Hotel and Marina; the Marriott Hotel Properties II Limited Partnership that owns the 1,290-room New Orleans Marriott, the 999-room San Antonio Marriott Rivercenter, the 368-room San Ramon Marriott, and a 50% limited partner interest in the 754-room Santa Clara Marriott; the Marriott Suites Limited Partnership that owns four hotels; and the partnership that owns the 510-room Salt Lake City Marriott.

Receivables from affiliates are reported net of reserves of \$144 million at January 2, 1998 and \$227 million at January 3, 1997. Receivables from affiliates at January 2, 1998 include a \$10 million debt service guarantee for the partnership that owns the Atlanta Marriott Marquis, which was repaid in early 1998. Receivables from affiliates at January 3, 1997 included a \$140 million mortgage note at 9% that amortizes through 2003, which was eliminated in the consolidation of CHLP in 1997. The Company has committed to advance additional amounts to affiliates, if necessary, to cover certain debt service requirements. Such commitments are limited, in the aggregate, to an additional \$60 million at January 2, 1998. Subsequent to January 2, 1998, this amount was reduced to \$20 million in connection with the refinancing and acquisition of a controlling interest in the Atlanta Marriott Marquis. Net amounts repaid to the Company under these commitments totaled \$2 million and \$13 million in 1997 and 1996, respectively. Net amounts funded by the Company totaled \$10 million in 1997 and \$8 million in 1995. There were no fundings in 1996.

The Company's pre-tax income from affiliates includes the following:

	1997	1996	1995
	(IN	MILLI	ONS)
Interest income			
Equity in net income	5	3	
	\$16	\$20	\$16
	===	===	===

Combined summarized balance sheet information for the Company's affiliates follows:

	1997	
	(IN MIL	
Property and equipment		334
Total assets		\$2,970
Debt, principally mortgages	412 (322)	672 (557)
Total liabilities and partners' deficit	\$2,275 =====	

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Combined summarized operating results for the Company's affiliates follow:

	1997	1996	1995
	(IN	MILLION	S)
Revenues Operating expenses:	\$ 610	\$ 737	\$ 770
Cash charges (including interest)  Depreciation and other non-cash charges			,
Income before extraordinary items Extraordinary itemsforgiveness of debt		42 12	24 181
Net income	\$ 77 =====	\$ 54 =====	\$ 205 =====

In December 1997, the Company, on behalf of six of its subsidiaries, filed a preliminary Prospectus/Consent Solicitation with the Securities and Exchange Commission, which describes the potential consolidation of six limited partnerships, including 219 limited-service hotel properties, into a single operating partnership and the formation of a new general partner which would intend to qualify as a real estate investment trust ("REIT"). Completion of this transaction is subject to several major contingencies, including a vote by the partners in each partnership, and no assurance can be given that the transaction will be consummated.

#### 5. DEBT

Debt consists of the following:

		1996  LIONS)
Properties Notes, with a rate of 9 1/2% due May 2005  New Properties Notes, with a rate of 8 7/8% due July 2007  Acquisitions Notes, with a rate of 9% due December 2007  Senior Notes, with an average rate of 9 3/4% at January 2,	\$ 600 600 350	
1998, maturing through 2012		71
Total Senior Notes	1,585	1,021
Mortgage debt (non-recourse) secured by \$3.0 billion of real estate assets, with an average rate of 8.6% at January 2, 1998, maturing through 2022	1,957	1,529
(7.6% at January 2, 1998) due March 1998	22	
Total Mortgage Debt	1,979	1,529
Other notes, with an average rate of 8% at January 2, 1998, maturing through 2027	19	86 11
Total Other	219	
	\$3,783	\$2,647 =====

In May 1995, HMH Properties, Inc. ("Properties"), a wholly-owned subsidiary of Host Marriott Hospitality, Inc., issued an aggregate of \$600 million of 9 1/2% senior secured notes (the "Properties Notes"). The bonds were issued in conjunction with a concurrent \$400 million offering by a subsidiary of the Company's discontinued HM Services' business at par, and have a final maturity of May 2005. The net proceeds were used

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

to defease, and subsequently redeem, all of the senior notes issued by Host Marriott Hospitality, Inc. and to repay borrowings under the line of credit with Marriott International. In connection with the redemptions and defeasance, the Company recognized an extraordinary loss in 1995 of \$17 million, net of taxes, related to continuing operations.

In December 1995, HMC Acquisition Properties, Inc. ("Acquisitions"), an indirect, wholly-owned subsidiary of the Company, issued \$350 million of 9% senior notes (the "Acquisitions Notes"). The Acquisitions Notes were issued at par and have a final maturity of December 2007. A portion of the net proceeds were utilized to repay in full the outstanding borrowings under the \$230 million revolving line of credit (the "Acquisition Revolver"), which was then terminated. In connection with the termination of the Acquisition Revolver, the Company recognized an extraordinary loss in 1995 of \$3 million, net of taxes.

On July 10, 1997, Properties and Acquisitions completed consent solicitations (the "Consent Solicitations") with holders of their senior notes to amend certain provisions of their senior notes' indentures. The Consent Solicitations facilitated the merger of Acquisitions with and into Properties (the "Merger"). The amendments to the indentures also increased the ability of Properties to acquire, through certain subsidiaries, additional properties subject to non-recourse indebtedness and controlling interests in corporations, partnerships and other entities holding attractive properties and increased the threshold required to permit Properties to make distributions to affiliates.

Concurrent with the Consent Solicitations and the Merger, Properties issued an aggregate of \$600 million of 8 7/8% senior notes (the "New Properties Notes") at par with a maturity of July 2007. Properties received net proceeds of approximately \$570 million, net of the costs of the Consent Solicitations and the Offering, which will be used to fund future acquisitions of, or the purchase of interests in, full-service hotels and other lodging-related properties, which may include senior living communities, as well as for general corporate purposes.

The Properties Notes, the Acquisitions Notes and the New Properties Notes are guaranteed on a joint and several basis by certain of Properties' subsidiaries and rank pari passu in right of payment with all other existing future senior indebtedness of Properties. Properties was the owner of 58 of the Company's 95 lodging properties at January 2, 1998.

The net assets of Properties at January 2, 1998 were approximately \$518 million, substantially all of which were restricted. The indentures governing the Properties Notes, the Acquisitions Notes and the New Properties Notes contain covenants that, among other things, limit the ability to incur additional indebtedness and issue preferred stock, pay dividends or make other distributions, repurchase capital stock or subordinated indebtedness, create certain liens, enter into certain transactions with affiliates, sell certain assets, issue or sell stock of subsidiaries, and enter into certain mergers and consolidations.

During 1997, the Company, through a newly-created, wholly-owned subsidiary, HMC Capital Resources Corporation ("Capital Resources"), entered into a revolving line of credit agreement (the "Line of Credit") with a group of commercial banks under which it may borrow up to \$500 million for the acquisition of lodging real estate and for the Company's working capital purposes. On June 19, 2000, any outstanding borrowings on the Line of Credit convert to a term loan arrangement with all unpaid advances due June 19, 2004. Borrowings under the Line of Credit bear interest at either the Eurodollar rate plus 1.7% or the Base Rate (as defined in the agreement) plus 0.7%, at the option of the Company. An annual fee of 0.35% is charged on the unused portion of the commitment. The Line of Credit was originally secured by six hotel properties contributed to Capital Resources, with a carrying value of approximately \$500 million as of January 2, 1998, and is guaranteed by the Company. As a result of this transaction, the Company terminated its line of credit with Marriott International. As of January 2, 1998, outstanding borrowings on the Line of Credit were approximately \$22 million as a result of a borrowing to fund the acquisition of the Ontario Airport Marriott.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

The Company also purchased 100% of the outstanding bonds secured by a first mortgage on the San Francisco Marriott in 1997. The Company purchased the bonds for \$219 million, an \$11 million discount to the face value of \$230 million. In connection with the redemption and defeasance of the bonds, the Company recognized an extraordinary gain of \$5 million, which represents the \$11 million discount less the write-off of unamortized deferred financing fees, net of taxes.

In 1997, the Company incurred approximately \$418 million of mortgage debt in conjunction with the acquisition of 11 hotels. In connection with the acquisition of the outstanding common stock of Forum Group, Inc. (the "Forum Group") in June 1997 (see Note 12), the Company also assumed debt of approximately \$270 million. The \$270 million of debt is comprised of secured debt of approximately \$198 million and unsecured debt of approximately \$72 million (\$59 million of which was provided by Marriott International). In 1997, the Company completed \$56 million of the \$107 million expansion plan for the Forum Group properties. As a result, an additional \$33 million of debt financing has been provided by Marriott International, and Marriott International may provide additional financing as the expansion plan is completed. The Company also assumed approximately \$15 million of debt in conjunction with the acquisition of the Leisure Park retirement community.

In conjunction with the construction of the Philadelphia Marriott, which was completed and opened in January 1995, the Company obtained first mortgage financing from Marriott International for 60% of the construction and development costs of the hotel. In the fourth quarter of 1996, the Company repaid the \$109 million mortgage, prior to the rate increasing to 10% per annum with an additional 2% deferred, with the proceeds from the convertible preferred securities offering discussed in Note 6. In the first quarter of 1997, the Company obtained \$90 million in first mortgage financing from two insurance companies secured by the Philadelphia Marriott. The mortgage bears interest at a fixed rate of 8.49% and matures in April 2009.

In December 1997, the Company successfully completed the refinancing of the MHPLP mortgage debt for approximately \$152 million. The new mortgage bears interest at 7.48% and matures in January 2008. In connection with the refinancing, the Company recognized an extraordinary loss of \$2 million, which represents payment of a prepayment penalty and the write-off of unamortized deferred financing fees, net of taxes.

At January 2, 1998, the Company was party to an interest rate exchange agreement with a financial institution (the contracting party) with an aggregate notional amount of \$100 million. Under this agreement, the Company collects interest based on specified floating interest rates of one month LIBOR (rate of 6% at January 2, 1998) and pays interest at fixed rates (rate of 7.99% at January 2, 1998). This agreement expires in 1998 in conjunction with the maturity of the mortgage on the New York Marriott Marquis. Also in 1997, the Company was party to two additional interest rate swap agreements with an aggregate notional amount of \$400 million which expired in May 1997. The Company realized a net reduction of interest expense of \$1 million in 1997, \$6 million in 1996 and \$5 million in 1995 related to interest rate exchange agreements. The Company monitors the creditworthiness of its contracting parties by evaluating credit exposure and referring to the ratings of widely accepted credit rating services. The Standard and Poors' long-term debt rating for the contracting party is A-. The Company is exposed to credit loss equal to the accrued interest receivable on the notional amount of interest rate exchange agreements outstanding (\$0 at January 2, 1998), in the event of non-performance by the contracting party to the interest rate swap agreements; however, the Company does not anticipate non-performance by the contracting party.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

The Company's debt balance at January 2, 1998, includes \$232 million of debt that is recourse to the parent company. Aggregate debt maturities at January 2, 1998, excluding capital lease obligations, are (in millions):

1998	\$ 371
1999	
2000	135
2001	220
2002	
Thereafter	2,822
	\$3,764
	=====

Cash paid for interest for continuing operations, net of amounts capitalized, was \$286 million in 1997, \$220 million in 1996 and \$177 million in 1995. Deferred financing costs, which are included in other assets, amounted to \$97 million and \$61 million, net of accumulated amortization, as of January 2, 1998 and January 3, 1997, respectively. Amortization of deferred financing costs totaled \$7 million, \$5 million and \$4 million in 1997, 1996 and 1995, respectively.

6. COMPANY-OBLIGATED MANDATORILY REDEEMABLE CONVERTIBLE PREFERRED SECURITIES
OF A SUBSIDIARY TRUST HOLDING COMPANY SUBSTANTIALLY ALL OF WHOSE ASSETS ARE
THE CONVERTIBLE SUBORDINATED DEBENTURES DUE 2026

In December 1996, Host Marriott Financial Trust (the "Issuer"), a whollyowned subsidiary trust of the Company, issued 11 million shares of 6 3/4% convertible quarterly income preferred securities (the "Convertible Preferred Securities"), with a liquidation preference of \$50 per share (for a total liquidation amount of \$550 million). The Convertible Preferred Securities represent an undivided beneficial interest in the assets of the Issuer. The payment of distributions out of moneys held by the Issuer and payments on liquidation of the Issuer or the redemption of the Convertible Preferred Securities are guaranteed by the Company to the extent the Issuer has funds available therefor. This guarantee, when taken together with the Company's obligations under the indenture pursuant to which the Debentures were issued, the Debentures, the Company's obligations under the Trust Agreement and its obligations under the indenture to pay costs, expenses, debts and liabilities of the Issuer (other than with respect to the Convertible Preferred Securities) provides a full and unconditional guarantee of amounts due on the Convertible Preferred Securities. Proceeds from the issuance of the Convertible Preferred Securities were invested in 6 3/4% Convertible Subordinated Debentures (the "Debentures") due December 2, 2026 issued by the Company. The Issuer exists solely to issue the Convertible Preferred Securities and its own common securities (the "Common Securities") and invest the proceeds therefrom in the Debentures, which is its sole asset. Separate financial statements of the Issuer are not presented because of the Company's guarantee described above; the Company's management has concluded that such financial statements are not material to investors and the Issuer is whollyowned and essentially has no independent operations.

Each of the Convertible Preferred Securities is convertible at the option of the holder into shares of Company common stock at the rate of 2.6876 shares per Convertible Preferred Security (equivalent to a conversion price of \$18.604 per share of Company common stock). The Debentures are convertible at the option of the holders into shares of Company common stock at a conversion rate of 2.6876 shares for each \$50 in principal amount of Debentures. The Issuer will only convert Debentures pursuant to a notice of conversion by a holder of Convertible Preferred Securities. During 1997 and 1996, no shares were converted into common stock.

Holders of the Convertible Preferred Securities are entitled to receive preferential cumulative cash distributions at an annual rate of 6 3/4% accruing from the original issue date, commencing March 1, 1997, and payable quarterly in arrears thereafter. The distribution rate and the distribution and other payment dates for the

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

Convertible Preferred Securities will correspond to the interest rate and interest and other payment dates on the Debentures. The Company may defer interest payments on the Debentures for a period not to exceed 20 consecutive quarters. If interest payments on the Debentures are deferred, so too are payments on the Convertible Preferred Securities. Under this circumstance, the Company will not be permitted to declare or pay any cash distributions with respect to its capital stock or debt securities that rank pari passu with or junior to the Debentures.

Subject to certain restrictions, the Convertible Preferred Securities are redeemable at the Issuer's option upon any redemption by the Company of the Debentures after December 2, 1999. Upon repayment at maturity or as a result of the acceleration of the Debentures upon the occurrence of a default, the Debentures shall be subject to mandatory redemption, from which the proceeds will be applied to redeem Convertible Preferred Securities and Common Securities, together with accrued and unpaid distributions.

#### 7. SHAREHOLDERS' EQUITY

Six hundred million shares of common stock, with a par value of \$1 per share, are authorized, of which 203.8 million and 202.0 million were issued and outstanding as of January 2, 1998 and January 3, 1997, respectively. One million shares of no par value preferred stock are authorized with none outstanding. During 1995, substantially all outstanding shares of such preferred stock were converted into approximately five million shares of common stock with the remainder defeased.

On March 27, 1996, the Company completed the issuance of 31.6 million shares of common stock for net proceeds of nearly \$400 million.

In connection with a class action settlement, the Company issued warrants to purchase up to 7.7 million shares of the Company's common stock at \$8.00 per share through October 8, 1996 and \$10.00 per share thereafter. During 1996, 6.8 million warrants were exercised at \$8.00 per share and an equivalent number of shares of Company common stock were issued. During 1997, approximately 60,000 warrants were exercised at \$10.00 per share and an equivalent number of shares of Company common stock were issued. As of January 2, 1998, there were approximately 550,000 warrants outstanding.

In February 1989, the Board of Directors adopted a shareholder rights plan under which a dividend of one preferred stock purchase right was distributed for each outstanding share of the Company's common stock. Each right entitles the holder to buy 1/1,000th of a share of a newly issued series of junior participating preferred stock of the Company at an exercise price of \$150 per share. The rights will be exercisable 10 days after a person or group acquires beneficial ownership of at least 20%, or begins a tender or exchange offer for at least 30%, of the Company's common stock. Shares owned by a person or group on February 3, 1989 and held continuously thereafter are exempt for purposes of determining beneficial ownership under the rights plan. The rights are non-voting and will expire on February 2, 1999, unless exercised or previously redeemed by the Company for \$.01 each. If the Company is involved in a merger or certain other business combinations not approved by the Board of Directors, each right entitles its holder, other than the acquiring person or group, to purchase common stock of either the Company or the acquiror having a value of twice the exercise price of the right.

#### 8. INCOME TAXES

Total deferred tax assets and liabilities at January 2, 1998 and January 3, 1997 were as follows:

	1997	1996
	(IN MILI	LIONS)
Deferred tax assets Deferred tax liabilities		
Net deferred income tax liability	\$ (508) =====	\$ (464) ======

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

The tax effect of each type of temporary difference and carryforward that gives rise to a significant portion of deferred tax assets and liabilities as of January 2, 1998 and January 3, 1997 follows:

	1997	1996
	(IN MILI	LIONS)
Investments in affiliates	(200) (65)	\$ (303) (135) (73) (92) 97 26 16
Net deferred income tax liability	\$ (508) ======	\$ (464) ======

The provision (benefit) for income taxes consists of:

	1997	1996	1995
	(IN	MILLIO	NS)
CurrentFederal	\$19	\$(2)	\$ 7
State	4		3
Foreign	3	3	
	26	4	10
Deferred Federal			(23)
State	2	(1)	
	10	1	(23)
	\$36	\$ 5	\$(13)
	===	===	====

At January 2, 1998, the Company had approximately \$41 million of alternative minimum tax credit carryforwards available which do not expire.

Through 1997, the Company settled with the Internal Revenue Service ("IRS") substantially all issues for tax years 1979 through 1993. The Company expects to resolve any remaining issues with no material impact on the consolidated financial statements. The Company made net payments to the IRS of approximately \$10 million and \$45 million in 1997 and 1996, respectively, related to these settlements. Certain adjustments totaling approximately \$2 million and \$11 million in 1996 and 1995, respectively, were made to the tax provision related to those settlements.

A reconciliation of the statutory Federal tax rate to the Company's effective income tax rate follows:

	1997	1996	1995
Statutory Federal tax rate State income taxes, net of Federal tax benefit Tax credits	4.9	21.7	` ,
Additional tax on foreign source income	`6.0´	40.8	
Tax contingencies			14.6
Permanent itemsOther, net			 0.7
other, net		1.0	•
Effective income tax rate	43.4%	62.5 %	(17.3)% =====

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

As part of the Marriott International Distribution and the Special Dividend, the Company, Marriott International and HM Services entered into tax-sharing agreements which reflect each party's rights and obligations with respect to deficiencies and refunds, if any, of Federal, state or other taxes relating to the businesses of the Company, Marriott International and HM Services prior to the Marriott International Distribution and the Special Dividend.

Cash paid for income taxes, including IRS settlements, net of refunds received, was \$56 million in 1997, \$40 million in 1996 and \$22 million in 1995.

#### 9. LEASES

The Company leases certain property and equipment under non-cancelable operating and capital leases. Future minimum annual rental commitments for all non-cancelable leases are as follows:

	CAPITAL OPERATING LEASES LEASES	
	(IN MILLIONS)	
1998	\$ 3	\$ 115
1999	3	112
2000	3	109
2001	3	106
2002	3	103
Thereafter	16	1,361
Total minimum lease payments	31	\$1,906
• •		======
Less amount representing interest	12	
Present value of minimum lease payments	\$19	
	===	

As discussed in Note 12, the Company sold and leased back 37 of its Courtyard properties in 1995 and an additional 16 Courtyard properties in 1996 to a REIT. Additionally, in 1996, the Company sold and leased back 18 of its Residence Inns to the same REIT. These leases, which are accounted for as operating leases and are included above, have initial terms expiring through 2012 for the Courtyard properties and 2010 for the Residence Inn properties, and are renewable at the option of the Company. Minimum rent payments are \$51 million annually for the Courtyard properties and \$17 million annually for the Residence Inn properties, and additional rent based upon sales levels are payable to the owner under the terms of the leases.

Leases also include long-term ground leases for certain hotels, generally with multiple renewal options. Certain leases contain provision for the payment of contingent rentals based on a percentage of sales in excess of stipulated amounts.

Certain of the lease payments included in the table above relate to facilities used in the Company's former restaurant business. Most leases contain one or more renewal options, generally for five or 10-year periods. Future rentals on leases have not been reduced by aggregate minimum sublease rentals of \$124 million payable to the Company under non-cancelable subleases.

The Company remains contingently liable at January 2, 1998 on certain leases relating to divested non-lodging properties. Such contingent liabilities aggregated \$110 million at January 2, 1998. However, management considers the likelihood of any substantial funding related to these leases to be remote.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Rent expense consists of:

	1997	1996	1995
	(IN I	MILLI	ONS)
Minimum rentals on operating leases			\$34 17
	\$118	\$99	\$51

#### 10. EMPLOYEE STOCK PLANS

At January 2, 1998, the Company has two stock-based compensation plans which are described below. Under the comprehensive stock plan (the "Comprehensive Plan"), the Company may award to participating employees (i) options to purchase the Company's common stock, (ii) deferred shares of the Company's common stock and (iii) restricted shares of the Company's common stock. In addition, the Company has an employee stock purchase plan (the "Employee Stock Purchase Plan"). The principal terms and conditions of the two plans are summarized below.

Total shares of common stock reserved and available for issuance under employee stock plans at January 2, 1998 are:

	(IN MILLIONS)
Comprehensive Plan	28
Employee Stock Purchase Plan	3
	31
	===

Employee stock options may be granted to officers and key employees with an exercise price not less than the fair market value of the common stock on the date of grant. Options granted before May 11, 1990 expire 10 years after the date of grant and nonqualified options granted on or after May 11, 1990 expire up to 15 years after the date of grant. Most options vest ratably over each of the first four years following the date of the grant. In connection with the Marriott International Distribution, the Company issued an equivalent number of Marriott International options and adjusted the exercise prices of its options then outstanding based on the relative trading prices of shares of the common stock of the two companies.

The Company continues to account for expense under its plans under the provisions of Accounting Principle Board Opinion 25 and related interpretations as permitted under SFAS No. 123. Accordingly, no compensation cost has been recognized for its fixed stock options under the Comprehensive Plan and its Employee Stock Purchase Plan.

For purposes of the following disclosures required by SFAS No. 123, the fair value of each option granted has been estimated on the date of grant using an option-pricing model with the following weighted average assumptions used for grants in 1997, 1996 and 1995, respectively: risk-free interest rate of 6.2%, 6.6% and 6.8%, respectively, volatility of 35%, 36% and 37%, respectively, expected lives of 12 years and no dividend yield. The weighted average fair value per option granted during the year was \$13.13 in 1997, \$8.68 in 1996 and \$5.76 in 1995.

Pro forma compensation cost for 1997, 1996 and 1995 would have reduced (increased) net income (loss) by approximately \$330,000, (\$150,000) and (\$5,000), respectively. Basic and diluted earnings per share on a pro forma basis were not impacted by the pro forma compensation cost in 1997, 1996 and 1995.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

The effects of the implementation of SFAS No. 123 are not representative of the effects on reported net income in future years because only the effects of stock option awards granted in 1995, 1996 and 1997 have been considered.

In connection with the Special Dividend, the then outstanding options held by current and former employees of the Company were redenominated in both Company and HM Services stock and the exercise prices of the options were adjusted based on the relative trading prices of shares of the common stock of the two companies. For all options held by certain current and former employees of Marriott International, the number and exercise price of the options were adjusted based on the trading prices of shares of the Company's common stock immediately before and after the Special Dividend. Therefore, the options outstanding reflect these revised exercise prices. Pursuant to the Distribution Agreement between the Company and HM Services, the Company has the right to receive up to 1.4 million shares of HM Services' common stock or an equivalent cash value subsequent to exercise of the options held by certain former and current employees of Marriott International. As of January 2, 1998, the Company valued this right at approximately \$20 million, which is included in other assets. A summary of the status of the Company's stock option plan for 1997, 1996 and 1995 follows:

	1997		1996		1995	
	SHARES (IN MILLIONS)	WEIGHTED AVERAGE EXERCISE PRICE	SHARES (IN MILLIONS)	WEIGHTED AVERAGE EXERCISE PRICE	SHARES (IN MILLIONS)	WEIGHTED AVERAGE EXERCISE PRICE
Balance, at beginning of						
year	8.3	\$ 4	10.0	\$ 4	11.7	\$ 4
Granted	.1	20	. 2	13		
Exercised	(1.6)	4	(1.9)	4	(2.3)	4
Forfeited/Expired Adjustment for Special	` ´		`'		`(.3)	4
Dividend					. 9	4
Balance, at end of						
year	6.8	4	8.3	4	10.0	4
	====		====		====	
Options exercisable at year-end	6.4		7.6		8.5	

The following table summarizes information about stock options outstanding at January 2, 1998:

OPTIONS OUTSTANDING				OPTIONS EXERC	ISABLE
RANGE OF EXERCISE PRICES	SHARES OUTSTANDING AT JANUARY 2, 1998	WEIGHTED AVERAGE REMAINING CONTRACTUAL LIFE	WEIGHTED AVERAGE EXERCISE PRICE	SHARES EXERCISABLE AT JANUARY 2, 1998	WEIGHTED AVERAGE EXERCISE PRICE
1-3 4-6	4.4 1.7	9	\$ 2	4.4 1.7	\$ 2
7-9 10-12 13-15	.4 .1 .1	12 14 14	9 12 15	.3  	9  
19-22	.1  6.8	15	20	  6.4	
	===			===	

Deferred stock incentive plan shares granted to officers and key employees after 1990 generally vest over 10 years in annual installments commencing one year after the date of grant. Certain employees may elect to defer payments until termination or retirement. Deferred stock incentive plan shares granted in 1990 and prior years generally vest in annual installments commencing one year after the date of grant and continuing for 10 years. Employees also could elect to forfeit one-fourth of their deferred stock incentive plan award in exchange for accelerated vesting over a 10-year period. The Company accrues compensation expense for the fair market

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

value of the shares on the date of grant, less estimated forfeitures. In 1997, 1996 and 1995, 14,000, 13,000 and 158,000 shares were granted, respectively, under this plan. The compensation cost that has been charged against income for deferred stock was \$1 million in 1995 and was not material in 1996 and 1997. The weighted average fair value per share granted during each year was \$15.81 in 1997, \$11.81 in 1996 and \$8.49 in 1995.

In 1993, 3,537,000 restricted stock plan shares under the Comprehensive Plan were issued to officers and key executives to be distributed over the next three to 10 years in annual installments based on continued employment and the attainment of certain performance criteria. The Company recognizes compensation expense over the restriction period equal to the fair market value of the shares on the date of issuance adjusted for forfeitures, and where appropriate, the level of attainment of performance criteria and fluctuations in the fair market value of the Company's common stock. In 1997 and 1996, 198,000 and 2,511,000 shares of additional restricted stock plan shares were granted to certain key employees under terms and conditions similar to the 1993 grants. Approximately 161,000 and 500,000 shares were forfeited in 1996 and 1995, respectively. There were no shares forfeited in 1997. The Company recorded compensation expense of \$13 million, \$11 million and \$5 million in 1997, 1996 and 1995, respectively, related to these awards. The weighted average fair value per share granted during each year was \$16.88 in 1997 and \$14.01 in 1996. There were no restricted stock plan shares granted in 1995.

Under the terms of the Employee Stock Purchase Plan, eligible employees may purchase common stock through payroll deductions at the lower of market value at the beginning or end of the plan year.

#### 11. PROFIT SHARING AND POSTEMPLOYMENT BENEFIT PLANS

The Company contributes to profit sharing and other defined contribution plans for the benefit of employees meeting certain eligibility requirements and electing participation in the plans. The amount to be matched by the Company is determined annually by the Board of Directors. The Company provides medical benefits to a limited number of retired employees meeting restrictive eligibility requirements. Amounts for these items were not material in 1995 through 1997.

#### 12. ACQUISITIONS AND DISPOSITIONS

In 1997, the Company acquired eight full-service hotels totaling 3,600 rooms for approximately \$145 million. In addition, the Company acquired controlling interests in nine full-service hotels totaling 5,024 rooms for approximately \$621 million, including the assumption of approximately \$418 million of debt. The Company also completed the acquisition of the 504-room New York Marriott Financial Center, after acquiring the mortgage on the hotel for \$101 million in late 1996.

Also in 1997, the Company acquired the outstanding common stock of the Forum Group from MSLS (29 senior living communities) for approximately \$460 million, including approximately \$270 million in debt, as described in Note 5. In addition, the Company plans to add approximately 1,060 units to these communities for approximately \$107 million through an expansion plan which will be completed in 1999. In 1997, approximately \$56 million of the expansion plan had been completed (including approximately \$33 million of debt financing provided by Marriott International). The Company also acquired 49% of the remaining 50% interest in the partnership which owned the 418-unit Leisure Park retirement community for approximately \$23 million, including the assumption of approximately \$15 million of debt.

In 1996, the Company acquired six full-service hotels totaling 1,964 rooms for an aggregate purchase price of approximately \$189 million. In addition, the Company acquired controlling interests in 17 full-service hotels totaling 8,917 rooms for an aggregate purchase price of approximately \$1.1 billion, including the assumption of approximately \$696 million of debt. The Company also purchased the first mortgage of the 504-room New York Marriott Financial Center for approximately \$101 million.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

In 1995, the Company acquired nine full-service hotels totaling approximately 3,900 rooms in separate transactions for approximately \$390 million.

During the first and third quarters of 1995, 37 of the Company's Courtyard properties were sold and leased back from a REIT for approximately \$330 million. The Company received net proceeds from the two transactions of approximately \$297 million and will receive approximately \$33 million upon expiration of the leases. A deferred gain of \$14 million on the sale/leaseback transactions is being amortized over the initial term of the leases.

In the first and second quarters of 1996, the Company completed the sale and leaseback of 16 of its Courtyard properties and 18 of its Residence Inn properties for \$349 million. The Company received net proceeds of approximately \$314 million and will receive approximately \$35 million upon expiration of the leases. A deferred gain of \$45 million on the sale/leaseback transactions is being amortized over the initial term of the leases.

The Company's summarized, unaudited consolidated pro forma results of operations, assuming the above transactions and the refinancings and new debt activity discussed in Note 5 occurred on December 30, 1995, are as follows (in millions, except per share amounts):

		1996
Revenues	\$1,274	\$1,121
<pre>Income (loss) before extraordinary items</pre>	42	(24)
Net income (loss)	45	(24)
Basic earnings (loss) per common share:		
<pre>Income (loss) before extraordinary items</pre>	.21	(.13)
Basic earnings (loss) per common share	.22	(.13)
Diluted earnings (loss) per common share:		
<pre>Income (loss) before extraordinary items</pre>	.20	(.13)
Diluted earnings (loss) per common share	.22	(.13)

# 13. FAIR VALUE OF FINANCIAL INSTRUMENTS

The fair values of certain financial assets and liabilities and other financial instruments are shown below:

	1331			-
	CARRYING	FAIR	CARRYING	
	(IN MILLIONS)			
Financial assets				
Short-term marketable securities	\$ 354	\$ 354	\$	\$
Receivables from affiliates	23	26	156	174
Notes receivable	31	48	141	155
Other	20	20	13	13
Financial liabilities				
Debt, net of capital leases	3,764	3,815	2,636	2,654
Other financial instruments				
Convertible Preferred Securities	550	638	550	595
Interest rate swap agreements				1
Affiliate debt service commitments				

1997

1996

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Short-term marketable securities and Convertible Preferred Securities are valued based on quoted market prices. Receivables from affiliates, notes and other financial assets are valued based on the expected future cash flows discounted at risk-adjusted rates. Valuations for secured debt are determined based on the expected future payments discounted at risk-adjusted rates. The fair values of the Line of Credit and other notes are estimated to be equal to their carrying value. Senior Notes are valued based on quoted market prices.

The Company is contingently liable under various guarantees of obligations of certain affiliates (affiliate debt service commitments) with a maximum commitment of \$60 million at January 2, 1998 and \$117 million at January 3, 1997. A fair value is assigned to commitments with expected future fundings. The fair value of the commitments represents the net expected future payments discounted at risk-adjusted rates. Such payments are accrued on an undiscounted basis.

The fair value of interest rate swap agreements is based on the estimated amount the Company would pay or receive to terminate the swap agreements. The aggregate notional amount of the agreements was \$100 million at January 2, 1998 and \$525 million at January 3, 1997.

# 14. MARRIOTT INTERNATIONAL DISTRIBUTION AND RELATIONSHIP WITH MARRIOTT INTERNATIONAL

On October 8, 1993 (the "Marriott International Distribution Date"), Marriott Corporation distributed, through a special tax-free dividend (the "Marriott International Distribution"), to holders of Marriott Corporation's common stock (on a share-for-share basis), approximately 116.4 million outstanding shares of common stock of an existing wholly-owned subsidiary, Marriott International, resulting in the division of Marriott Corporation's operations into two separate companies. The distributed operations included the former Marriott Corporation's lodging management, franchising and resort timesharing operations, senior living service operations, and the institutional food service and facilities management business. The Company retained the former Marriott Corporation's airport and tollroad food, beverage and merchandise concessions operations, as well as most of its real estate properties. Effective at the Marriott International Distribution Date, Marriott Corporation changed its name to Host Marriott Corporation.

The Company and Marriott International have entered into various agreements in connection with the Marriott International Distribution and thereafter which provide, among other things, that (i) the majority of the Company's hotel lodging properties are managed by Marriott International under agreements with initial terms of 15 to 20 years and which are subject to renewal at the option of Marriott International for up to an additional 16 to 30 years (see Note 15); (ii) 10 of the Company's full-service properties are operated under franchise agreements with Marriott International with terms of 15 to 30 years; (iii) all of the Company's senior living communities are managed by MSLS under agreements with initial terms of 25 to 30 years and which are subject to renewal at the option of Marriott International for an additional five to ten years (see Note 16); (iv) Marriott International provided the Company with \$92 million of financing at an average rate of 9% in 1997 (and may provide additional financing as the expansion plan is completed) in conjunction with the acquisition of senior living communities from Marriott International (see Notes 5 and 12); (v) the Company acquired 49% of Marriott International's 50% interest in the Leisure Park retirement community in 1997 for \$23 million, including approximately \$15 million of assumed debt; (vi) Marriott International guarantees the Company's performance in connection with certain loans and other obligations (\$107 million at January 2, 1998); (vii) the Company borrowed and repaid \$109 million of first mortgage financing for construction of the Philadelphia Marriott (see Note 5); (viii) Marriott International provided the Company with \$70 million of mortgage financing in 1995 for the acquisition of three full-service properties by the Company at an average interest rate of 8.5% (Marriott International subsequently sold one of the loans in November 1996); (ix) Marriott International and the Company formed a joint venture and Marriott International provided the Company with \$29 million in debt financing at an average interest rate of 12.7% and \$28 million in preferred equity in 1996 for the acquisition of two full-service properties in Mexico City, Mexico; (x) in 1995, the

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

Company also acquired a full-service property from a partnership in which Marriott International owned a 50% interest; and (xi) Marriott International provides certain limited administrative services.

In 1997, 1996 and 1995, the Company paid to Marriott International \$162 million, \$101 million and \$67 million, respectively, in hotel management fees; \$13 million, \$18 million and \$21 million, respectively, in interest and commitment fees under the debt financing and line of credit provided by Marriott International, \$3 million, \$4 million and \$12 million, respectively, for limited administrative services. The Company also paid Marriott International \$4 million, \$2 million and \$1 million, respectively, of franchise fees in 1997, 1996 and 1995. In connection with the acquisition of the Forum Group, the Company paid Marriott International \$6 million in senior living community management fees during 1997.

Additionally, Marriott International has the right to purchase up to 20% of the voting stock of the Company if certain events involving a change in control of the Company occur.

#### 15. HOTEL MANAGEMENT AGREEMENTS

Most of the Company's hotels are subject to management agreements (the "Agreements") under which Marriott International manages most of the Company's hotels, generally for an initial term of 15 to 20 years with renewal terms at the option of Marriott International of up to an additional 16 to 30 years. The Agreements generally provide for payment of base management fees equal to one to four percent of sales and incentive management fees generally equal to 20% to 50% of Operating Profit (as defined in the Agreements) over a priority return (as defined) to the Company, with total incentive management fees not to exceed 20% of cumulative Operating Profit, or 20% of current year Operating Profit. In the event of early termination of the Agreements, Marriott International will receive additional fees based on the unexpired term and expected future base and incentive management fees. The Company has the option to terminate certain management agreements if specified performance thresholds are not satisfied. No agreement with respect to a single lodging facility is cross-collateralized or cross-defaulted to any other agreement and a single agreement may be canceled under certain conditions, although such cancellation will not trigger the cancellation of any other agreement.

Pursuant to the terms of the Agreements, Marriott International is required to furnish the hotels with certain services ("Chain Services") which are generally provided on a central or regional basis to all hotels in the Marriott International hotel system. Chain Services include central training, advertising and promotion, a national reservation system, computerized payroll and accounting services, and such additional services as needed which may be more efficiently performed on a centralized basis. Costs and expenses incurred in providing such services are allocated among all domestic hotels managed, owned or leased by Marriott International or its subsidiaries. In addition, the Company's hotels also participate in the Marriott Rewards program. The cost of this program is charged to all hotels in the Marriott hotel system.

The Company is obligated to provide the manager with sufficient funds to cover the cost of (a) certain non-routine repairs and maintenance to the hotels which are normally capitalized; and (b) replacements and renewals to the hotels' property and improvements. Under certain circumstances, the Company will be required to establish escrow accounts for such purposes under terms outlined in the Agreements.

The Company has entered into franchise agreements with Marriott International for ten hotels. Pursuant to these franchise agreements, the Company generally pays a franchise fee based on a percentage of room sales and food and beverage sales as well as certain other fees for advertising and reservations. Franchise fees for room sales vary from four to six percent of sales, while fees for food and beverage sales vary from two to three percent of sales. The terms of the franchise agreements are from 15 to 30 years.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

The Company has entered into management agreements with The Ritz-Carlton Hotel Company, LLC ("Ritz-Carlton"), an affiliate of Marriott International, to manage four of the Company's hotels. These agreements have an initial term of 15 to 25 years with renewal terms at the option of Ritz-Carlton of up to an additional 10 to 40 years. Base management fees vary from two to four percent of sales and incentive management fees are generally equal to 20% of available cash flow or operating profit, as defined in the agreements.

The Company has also entered into management agreements with hotel management companies other than Marriott International and Ritz-Carlton for 12 of its hotels (10 of which are franchised under the Marriott brand). These agreement generally provide for an initial term of 10 to 20 years with renewal terms at the option of either party of up to an additional one to 15 years. The agreement generally provide for payment of base management fees equal to one to three percent of sales. Seven of the 12 agreements also provide for incentive management fees generally equal to 15 to 20 percent of available cash flow, as defined in the agreements.

At January 2, 1998 and January 3, 1997, \$75 million and \$76 million, respectively, have been advanced to the hotel managers for working capital and are included in "Due From Managers" in the accompanying consolidated balance sheets.

#### 16. SENIOR LIVING COMMUNITY OPERATING AGREEMENTS

The Company's senior living communities (the "Communities") are subject to operating agreements (the "SLC Agreements") which provide for MSLS to operate the communities, generally for an initial term of 25 to 30 years with renewal terms subject to certain performance criteria at the option of MSLS of up to an additional five to ten years. The SLC Agreements provide for payment of base management fees generally equal to five to eight percent of gross revenues and incentive management fees generally equal to zero to 20% of Operating Profit (as defined in the SLC Agreements) over a priority return to the Company. In the event of early termination of the SLC Agreements, MSLS will receive additional fees based on the unexpired term and expected future base and incentive management fees. The Company has the option to terminate certain, but not all, management agreements if specified performance thresholds are not satisfied. No SLC Agreement with respect to a single community is cross-collateralized or cross-defaulted to any other SLC Agreement and any single SLC Agreement may be terminated following a default by the Company or MSLS, although such termination will not trigger the cancellation of any other SLC Agreement.

Pursuant to the terms of the SLC Agreements, MSLS is required to furnish the Communities with certain services ("Central Administrative Services") which are provided on a central or regional basis to all properties in the Marriott Retirement Community System. These services include the development and operation of computer systems, computer payroll and accounting services, marketing and public relations services, and such additional services as may from time-to-time be performed more efficiently on a central or regional level. The SLC Agreements establish payment of Central Administrative Services fees generally equal to 0% of Gross Revenues for the first year and 2% thereafter.

MSLS is required under the SLC Agreements to deduct an amount from gross revenues and place the funds into an interest bearing reserve account to cover the cost of (a) certain routine repairs and maintenance to the Communities which are normally capitalized; and (b) replacements and renewals to the Communities' property and improvements. The annual payment amount (expressed as a percentage of gross revenues) generally will be 2.65% through fiscal year 2002, 2.85% for fiscal years 2003 through 2007, and 3.5% thereafter. The SLC Agreements provide that the Company shall provide MSLS with sufficient funds to cover the cost of certain major or non-routine repairs, alterations, improvements, renewals and replacements to the Communities which are required to maintain a competitive, efficient and economical operating condition in accordance with Marriott standards or for the continued safe and orderly operation of the Communities.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

At January 2, 1998, approximately \$6 million has been advanced to MSLS for working capital for the senior living communities.

#### 17. LITIGATION

The Company is from time-to-time the subject of, or involved in, judicial proceedings. Management believes that any liability or loss resulting from such matters will not have a material adverse effect on the financial position or results of operations of the Company.

In the fourth quarter of 1997, the Company reached a settlement in a lawsuit against Trinity Industries and others for claims related to construction of the New York Marriott Marquis. In settlement of the lawsuit, the Company and its affiliate received a cash settlement of approximately \$70 million, the majority of which was considered a recovery of construction costs and \$10 million of which has been recorded as other revenues in the accompanying financial statements.

#### 18. HOTEL AND SENIOR LIVING COMMUNITY OPERATIONS

As discussed in Note 1, revenues recorded in the consolidated financial statements reflect house profit from the Company's hotel properties and senior living communities. House profit reflects the net revenues flowing to the Company as property owner and represents all gross hotel and senior living communities' operating revenues, less all gross property-level expenses, excluding depreciation, management fees, real and personal property taxes, ground and equipment rent, insurance and certain other costs, which are classified as operating costs and expenses.

Accordingly, the following table presents sales and the details of house profit for the Company's hotels for 1997, 1996 and 1995:

		1996	
		MILLIONS	
Sales Rooms Food and Beverage	776	515	363
Other		125	
Total Hotel Sales		1,942	
Department Costs			
Rooms Food and Beverage Other	592 189	406 63	43
Total Department Costs			553
Department Profit	(504)		(325)
House Profit	\$1,093		\$ 474

The following table presents the sales and details of house profit for the Company's senior living communities for 1997 (in millions):

Senior Living Communities Sales			
Department Costs		7	'4
House Profit	\$	3	37
	_		

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

#### 19. GEOGRAPHIC AND BUSINESS SEGMENT INFORMATION

The Company operates in two business segments in the lodging industry: hotels and senior living communities. The Company's hotels are primarily operated under the Marriott or Ritz-Carlton brands, contain an average of nearly 500 rooms, as well as supply other amenities such as meeting space and banquet facilities; a variety of restaurants and lounges; gift shops; and swimming pools. They are typically located in downtown, airport, suburban and resort areas throughout the United States. Senior living communities generally contain 100 to 300 units and offer a variety of senior care services that include independent living, assisted living and health care. The communities provide security, meals, housekeeping, linen service and 24-hour emergency health care.

The Company evaluates the performance of its segments based primarily on operating profit before depreciation, corporate expenses, and interest expense. The Company's income taxes are included in the consolidated Federal income tax return of the Company and its affiliates and is allocated based upon the relative contribution to the Company's consolidated taxable income/loss and changes in temporary differences. The allocation of taxes is not evaluated at the segment level and, therefore, the Company does not believe the information is material to the consolidated financial statements.

The following table presents revenues and other financial information by business segment (in millions):

			1997	
		SENIOR	CORPORATE	
			& OTHER	
Revenues Operating profit (loss)	\$1,093 444	\$ 37 16	\$ 17 (11)	\$1,147 449
Interest expense	281	14	7	302
Interest income	40		12	52
Depreciation and amortization	226	11	3	240
Capital expenditures	153	3	4	160
Total assets	5,787	623	116	6,526
			1996	
		SENTOR	CORPORATE	
	H0TELS			CONSOLIDATED
Revenues	\$ 717	\$	\$ 15	\$ 732
Operating profit (loss)	256		(23)	233
Interest expense	228		` 9´	237
Interest income	31		17	48
Depreciation and amortization	165		3	168
Capital expenditures	156		3	159
Total assets	4,770		382	5,152
			1995	
		SENIOR	CORPORATE	
	H0TELS			CONSOLIDATED
Revenues	\$ 474	\$	\$ 10	\$ 484
Operating profit (loss)	193		(79)	114
Interest expense	161		17	178
Interest income	11		16	27
Depreciation and amortization	117		5	122
Capital expenditures	154		6	160
Total assets	3,175		382	3,557

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

As of January 2, 1998, the Company's foreign operations consist of four full-service hotel properties located in Canada and two full-service hotel properties located in Mexico. There were no intercompany sales between the properties and the Company. The following table presents revenues and long-lived assets for each of the geographical areas in which the Company operates (in millions):

	19	1997		1996		995
	REVENUES	LONG-LIVED ASSETS	REVENUES	LONG-LIVED ASSETS	REVENUES	LONG-LIVED ASSETS
United States		\$4,995	\$714	\$3,587	\$482	\$2,842
International	39	222	18	218	2	40
Total	\$1,147	\$5,217	\$732 	\$3,805	\$484 	\$2,882

#### 20. QUARTERLY FINANCIAL DATA (UNAUDITED)

	1997									
		RST	SI QI	ECOND JARTER	Ql	THIRD JARTER	FOI QU	JRTH ARTER	FIS YE	SCAL EAR
	(IN	MILLION		EXCEPT				AMOUNTS		
Revenues Operating profit before minority interest, corporate expenses and	\$	252	\$	270	\$	246	\$	379	\$	1,147
interestIncome before		91		124		89		145		449
extraordinary items		6		26		6		9		47
Net income  Basic earnings per common share: Income before		11		26		6		7		50
extraordinary items		.03		.13		.03		.04		. 23
Net income Diluted earnings per common share: Income before		.05		.13		.03		.03		. 25
extraordinary items		.03		.13		.03		.04		. 23
Net income		.05		.13		.03		.03		. 24
					1996					
	QUAF	RST RTER	Q١	ECOND JARTER	Ql	JARTER	QUA	JRTH ARTER	YE	SCAL EAR
								AMOUNTS		
Revenues Operating profit before minority interest, corporate expenses and	\$	130	\$	167	\$	167	\$	268	\$	732
<pre>interest Income (loss) before</pre>		38		62		49		84		233
extraordinary items Net income (loss) Basic earnings (loss) per common share:		(12) (12)		7 7		(2) (2)		(6) (6)		(13) (13)
Income (loss) before extraordinary items Net income (loss) Diluted earnings (loss) per common share:		(.07) (.07)		. 04 . 04		(.01) (.01)		(.03) (.03)		(.07) (.07)
Income (loss) before extraordinary items Net income (loss)		(.07) (.07)		.03		(.01) (.01)		(.03) (.03)		(.07) (.07)

The first three quarters consist of 12 weeks each in both 1997 and 1996, and the fourth quarter includes 16 weeks in 1997 and 17 weeks in 1996. The sum of the basic and diluted earnings (loss) per common share for the four quarters in 1997 and 1996 differs from the annual earnings per common share due to the required method of computing the weighted average number of shares in the respective periods.

#### HOST MARRIOTT CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS SEPTEMBER 11, 1998 (UNAUDITED, IN MILLIONS)

#### ASSETS

Property and Equipment, net	32 88 18 319 36
Debt Senior Notes Issued by the Company or its Subsidiaries Mortgage Debt Other	2,003
Accounts Payable and Accrued Expenses  Deferred Income Taxes Other Liabilities	70 526 447
Total Liabilities	5,267
Company-obligated Mandatorily Redeemable Convertible Preferred Securities of a Subsidiary Trust Whose Assets are the Convertible Subordinated Debentures Due 2026 ("Convertible Preferred Securities")	
Shareholders' Equity Common Stock, 600 million shares authorized; 204.5 million shares issued and outstanding	7

See Notes to Condensed Consolidated Financial Statements.

# HOST MARRIOTT CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS THIRTY-SIX WEEKS ENDED SEPTEMBER 11, 1998 AND SEPTEMBER 12, 1997 (UNAUDITED, IN MILLIONS, EXCEPT PER COMMON SHARE AMOUNTS)

	1998	
REVENUES Hotels	\$ 922 58 53 1 6	\$736 16 3 3
Total revenues	1,040	768
OPERATING COSTS AND EXPENSES  Hotels (including Marriott International management fees of \$138 million and \$111 million in 1998 and 1997, respectively)	502	433
Senior living communities (including Marriott International management fees of \$9 million and \$3 million in 1998 and 1997, respectively)	31 14	9 22
Total operating costs and expenses		464
OPERATING PROFIT. Minority interest. Corporate expenses. REIT Conversion expenses. Interest expense. Dividends on Convertible Preferred Securities of a subsidiary trust.	(14)	304 (24) (27)  (198)
Interest income	`36´	`37 <sup>′</sup>
INCOME BEFORE INCOME TAXES AND EXTRAORDINARY ITEM  Provision for income taxes	175 (75)	(28)
INCOME BEFORE EXTRAORDINARY ITEM Extraordinary item gain (loss) on extinguishment of debt (net	100	38
of income taxes of \$80 million in 1998 and \$3 million in 1997)	(148)	
NET INCOME (LOSS)	\$ (48) =====	
BASIC EARNINGS (LOSS) PER COMMON SHARE: Income before extraordinary item	\$ .49 (.73)	.02
NET INCOME (LOSS)	\$ (.24) =====	
DILUTED EARNINGS (LOSS) PER COMMON SHARE: Income before extraordinary item		\$.19 .02
NET INCOME (LOSS)		\$.21

See Notes to Condensed Consolidated Financial Statements.

# HOST MARRIOTT CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS THIRTY-SIX WEEKS ENDED SEPTEMBER 11, 1998 AND SEPTEMBER 12, 1997 (UNAUDITED, IN MILLIONS)

	1998	1997
OPERATING ACTIVITIES Income before extraordinary item	\$ 100	\$ 38
Depreciation and amortizationIncome taxes	184 50	158 
Gains on sales of hotel properties  Equity in (earnings) losses of affiliates  Changes in operating accounts	(50)  (33)	(3) 78
Other	27	42
Cash from operations	278	313
INVESTING ACTIVITIES Proceeds from sales of assets	211	35
Acquisitions	(636)	(441)
Renewals and replacements  New development projects	(113) (36)	(86) 
New investment capital expenditures  Purchases of short-term marketable securities	(19) (134)	(22) 
Sales of short-term marketable securities  Note receivable collections	451 3	 5
Affiliate collections, net	13 (13)	12
Cash used in investing activities		(497)
FINANCING ACTIVITIES Issuances of debt, net of related expenses	2,004	682
Issuances of common stock	, (42)	4 (77)
Debt prepayments	(1,750) (175)	(241) 
Other	`(16)	
Cash from financing activities	23	391
INCREASE IN CASH AND CASH EQUIVALENTS	\$ 28 =====	
Non-cash financing activities: Assumption of mortgage debt for the acquisition of, or purchase of controlling interests in, certain hotel		
properties and senior living communities	\$ 164 =====	

See Notes to Condensed Consolidated Financial Statements.

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

1. The accompanying condensed consolidated financial statements of Host Marriott Corporation and subsidiaries (the "Company" or "Host Marriott") have been prepared by the Company without audit. Certain information and footnote disclosures normally included in financial statements presented in accordance with generally accepted accounting principles have been condensed or omitted. The Company believes the disclosures made are adequate to make the information presented not misleading. However, the condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's Annual Report on Form 10-K for the fiscal year ended January 2, 1998.

In the opinion of the Company, the accompanying unaudited condensed consolidated financial statements reflect all adjustments necessary to present fairly the financial position of the Company as of September 11, 1998 and the results of operations for the twelve and thirty-six weeks ended September 11, 1998 and September 12, 1997 and cash flows for the thirty-six weeks ended September 11, 1998 and September 12, 1997. Interim results are not necessarily indicative of fiscal year performance because of the impact of seasonal and short-term variations.

2. In April 1998, the Company reached a definitive agreement with various affiliates of The Blackstone Group and Blackstone Real Estate Partners (collectively, "Blackstone") to acquire controlling interests in 12 luxury hotels and a first mortgage interest in another hotel in the United States and certain other assets. The Company expects to pay approximately \$862 million in cash and assumed debt and to issue approximately 43.7 million operating partnership units ("OP Units") of the new operating partnership (the "Operating Partnership") (based upon a negotiated value of \$20.00 per OP Unit) and distribute up to 18% of the shares of Crestline (defined below), to be formed as part of the Company's reorganization, described below. Each OP Unit will be exchangeable for one share of Host Marriott common stock (or its cash equivalent). Upon completion of the acquisition, Blackstone will own approximately 16% of the outstanding shares of Host Marriott common stock on a fully converted basis. The Blackstone portfolio consists of two Ritz-Carlton, two Four Seasons, one Grand Hyatt, three Hyatt Regency and four Swissotel properties, and a mortgage on a third Four Seasons property.

The Blackstone transaction is expected to close immediately after the REIT Conversion, as described below. At that time, Blackstone's hotels and other assets will be contributed into the Operating Partnership, and the hotels leased to subsidiaries of Crestline (defined below). The hotels will continue to be managed on behalf of the lessees under the existing management contracts. Consummation of the Blackstone transaction is also subject to certain conditions, including consummation of the REIT Conversion no later than March 31, 1999 and Host Marriott qualifying as a REIT for 1999.

3. The Company's board of directors (the "Board") has authorized the Company to restructure its business operations to qualify as a real estate investment trust ("REIT"), currently expected to be effective as of January 1, 1999, and to spin-off its senior living communities business (Crestline Capital Corporation, "Crestline") through a taxable stock dividend to its shareholders (collectively, the "REIT Conversion"). After the REIT Conversion, which is subject to shareholder and final Board approval, the Company intends to operate as an "UPREIT," with substantially all of its assets and operations conducted through the newly formed Operating Partnership of which Host Marriott will be the general partner.

Host Marriott will distribute shares in Crestline to its shareholders at the time of the REIT Conversion and Host Marriott expects to make a cash distribution at that time which is expected to be treated as a taxable dividend. The aggregate value of the Crestline common stock and the cash to be distributed to shareholders and Blackstone, if the Blackstone acquisition is consummated, is currently estimated to be approximately \$525 million to \$625 million. The actual amount of the distribution will be based, in part, upon the estimated amount of accumulated earnings and profits of Host Marriott as of the last day of its taxable year in which the REIT

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) (UNAUDITED)

Conversion is consummated. To the extent that the distributions made are not sufficient to eliminate Host Marriott's accumulated earnings and profits, one or more additional taxable distributions will be made prior to the last day of the first full taxable year as a REIT (currently expected to be December 31, 1999). The distribution satisfies the requirement that a "C" corporation converting to a REIT distribute all of its accumulated earnings and profits at the time of conversion to a REIT. Crestline is expected to own Host Marriott's portfolio of senior living properties. This portfolio currently consists of 31 retirement communities, totaling 7,259 units in 13 states. The communities will continue to be managed by Marriott International. In addition, Crestline will lease substantially all of the hotels currently owned by the Operating Partnership. Crestline will operate independently of Host Marriott.

Following the REIT Conversion, Host Marriott will own OP Units equal to the number of outstanding shares of Host Marriott common stock at the time of the REIT Conversion. The UPREIT structure will not affect the ownership by shareholders of their existing Host Marriott shares.

As part of the REIT Conversion, the Company filed a Prospectus/Consent Solicitation with the Securities and Exchange Commission. This Prospectus/Consent Solicitation describes a proposal whereby the Operating Partnership will acquire by merger (the "Mergers") eight limited partnerships (the "Partnerships") that own full-service hotels in which the Company or its subsidiaries are general partners. As more fully described in the Prospectus/Consent Solicitation, limited partners of those Partnerships that participate in the Mergers will receive OP Units in exchange for their partnership interests in such Partnerships and, at their election, may exchange such OP Units for unsecured notes due December 15, 2005 issued by the Operating Partnership ("Notes") or common stock in the REIT. The Solicitation Period expires on December 12, 1998, unless extended.

The REIT expects to qualify as a real estate investment trust under federal income tax law, beginning January 1, 1999. However, consummation of the REIT Conversion is subject to significant contingencies that are outside the control of the Company, including final Board approval, consent of shareholders, partners, bondholders, lenders and ground lessors of Host Marriott, its affiliates and other third parties. Accordingly, there can be no assurance that the REIT Conversion will be completed or that it will be effective as of January 1, 1999. If the REIT Conversion is not completed on January 1, 1999, the effectiveness of REIT election could be delayed until January 1, 2000, which would result in the Company continuing to pay a substantial amount of corporate-level income taxes in 1999.

On April 20, 1998, the Company and certain of its subsidiaries filed a shelf registration on Form S-3 (the "Shelf Registration") with the Securities and Exchange Commission for the issuance of up to \$2.5 billion in securities, which may include debt, equity or a combination thereof. The Company utilized \$1.7 billion of the capacity under this Shelf Registration to issue the New Senior Notes (defined below) and anticipates that any net proceeds from the sale of additional offered securities will be used for refinancing of the Company's indebtedness, acquisitions and general corporate purposes.

HMH Properties, Inc. ("HMH Properties") an indirect wholly owned subsidiary of the Company, which currently owns 72 of Host Marriott's hotels, utilized the Shelf Registration to issue an aggregate of \$1.7 billion in new senior notes (the "New Senior Notes"). The New Senior Notes were issued in two series, \$500 million of 7 7/8% Series A notes due in 2005 and \$1.2 billion of 7 7/8% Series B notes due in 2008. Approximately \$21 million of the Old Senior Notes remain outstanding. The 1998 Consent Solicitations facilitated the merger of HMC Capital Resources Holdings Corporation ("Capital Resources"), a wholly owned subsidiary of the Company, with and into HMH Properties. Capital Resources, the then owner of eight of Host Marriott's hotel properties, was the obligor under the \$500 million revolving credit facility (the "Old Credit Facility").

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) (UNAUDITED)

In conjunction with the issuance of the New Senior Notes, HMH Properties entered into a \$1.25 billion credit facility (the "New Credit Facility") with a group of commercial banks. The New Credit Facility has an initial three-year term with two one-year extension options. Borrowings under the New Credit Facility generally bear interest at the Eurodollar rate plus 1.75% (7.5% at September 11, 1998). The interest rate and commitment fee (0.35% at September 11, 1998) on the unused portion of the New Credit Facility fluctuate based on certain financial ratios. The New Senior Notes and the New Credit Facility are guaranteed by the Company and its wholly owned subsidiary, Host Marriott Hospitality, Inc., and certain subsidiaries of HMH Properties and are secured by pledges of equity interests in certain subsidiaries of HMH Properties. The New Senior Notes and the New Credit Facility will be assumed by the Operating Partnership in connection with the REIT Conversion and the guarantee of the Company is expected to terminate upon consumation of the REIT Conversion. As of September 11, 1998, \$350 million was outstanding under the New Credit Facility.

The New Credit Facility and the indenture under which the New Senior Notes were issued contain covenants restricting the ability of HMH Properties and certain of its subsidiaries to incur indebtedness, grant liens on their assets, acquire or sell assets or make investments in other entities, and make certain distributions to equityholders of HMH Properties, the Company, and (following the REIT Conversion) the Operating Partnership and Host REIT. The New Credit Facility and the New Senior Notes also contain certain financial covenants relating to, among other things, maintaining certain levels of tangible net worth and certain ratios of EBITDA to interest and fixed charges, total debt to EBITDA, unencumbered assets to unsecured debt, and secured debt to total debt.

On August 5, 1998, simultaneously with the issuance of the New Senior Notes and the New Credit Facility, HMH Properties, Inc. purchased substantially all of its (i) \$600 million in 9 1/2% senior notes due 2005, (ii) \$350 million in 9% senior notes due 2007 and (iii) \$600 million in 8 7/8% senior notes due 2007 (collectively, the "Old Senior Notes"). Concurrently with each offer to purchase, HMH Properties successfully solicited consents (the "1998 Consent Solicitations") from registered holders of the Old Senior Notes to certain amendments to eliminate or modify substantially all of the restrictive covenants and certain other provisions contained in the indentures pursuant to which the Old Senior Notes were issued.

The net proceeds from the offering and borrowings under the New Credit Facility were used by Host Marriott to purchase substantially all of the Old Senior Notes, to repay amounts outstanding under the Old Credit Facility and to make bond premium and consent payments totaling \$175 million. These costs, along with the write-off of deferred financing fees of approximately \$52 million related to the Old Senior Notes and the Old Credit Facility, were recorded as a pre-tax extraordinary loss on the extinguishment of debt in the third quarter of 1998.

On September 30, 1998, the Company filed a preliminary Proxy Statement/Prospectus with the Securities and Exchange Commission. The Proxy Statement/Prospectus describes the proposed merger by and among Host Marriott, HMC Merger Corporation ("Host REIT") and Host Marriott, L.P., a recently formed limited partnership, pursuant to the REIT Conversion. The Prospectus proposes the restructuring of Host Marriott by contribution of its wholly owned full-service hotels and its interests in certain hotel partnerships and other assets to the Operating Partnership and reincorporation by means of the merger of Host Marriott with and into Host REIT. As a result of the restructuring, shares of Host Marriott common stock will be converted to shares of Host REIT common stock.

4. Revenues primarily represent house profit from the Company's hotel properties and senior living communities, net gains (losses) on property transactions and equity in earnings (losses) of affiliates. House profit reflects the net revenues flowing to the Company as property owner and represents gross hotel and senior living communities' operating revenues, less all gross property-level expenses, excluding depreciation, management

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) (UNAUDITED)

fees, real and personal property taxes, ground and equipment rent, insurance and certain other costs, which are classified as operating costs and expenses.

House profit generated by the Company's hotels for 1998 and 1997 consists of:

	THIRTY-SIX WEEKS ENDED		
	1998	SEPTEMBER 12, 1997	
	(IN MI	LLIONS)	
Sales Rooms	\$1,514	\$1,246	
Food & Beverage Other	642 159	500 120	
Total Hotel Sales		1,866	
Department Costs			
Rooms	348	285	
Food & Beverage	477	381	
Other	80	63	
Total Department Costs	905	729	
Department Profit	1,410	1,137	
Other Deductions	488	401	
House Profit	\$ 922	\$ 736	
	======	======	

House profit generated by the Company's senior living communities for 1998 consists of:

	THIRTY-SIX	WEEKS ENDED
	SEPTEMBER 11, 1998	SEPTEMBER 12, 1997
	(IN MI	LLIONS)
Sales	\$166	\$47
Department Costs	108	31
House Profit	\$ 58	\$16
	====	===

On November 20, 1997, the Emerging Issues Task Force ("EITF") of the Financial Accounting Standards Board reached a consensus on EITF 97-2, "Application of FASB Statement No. 94 and APB Opinion No. 16 to Physician Practice Management Entities and Certain Other Entities with Contractual Management Arrangements." EITF 97-2 addresses the circumstances in which a management entity may include the revenues and expenses of a managed entity in its financial statements. The Company has considered the impact of EITF 97-2 on its financial statements and has determined that EITF 97-2 requires the Company to include property-level sales and operating expenses of its hotels  $% \left( 1\right) =\left( 1\right) \left( 1$ and senior living communities in its statements of operations. The Company will adopt EITF 97-2 in the fourth quarter of 1998 with retroactive effect in prior periods to conform to the new presentation. Application of EITF 97-2 to the consolidated financial statements would have increased both revenues and operating expenses by approximately \$1,501 million and \$1,161 million, respectively, for the thirty-six weeks ended September 11, 1998 and September 12, 1997, and would have had no impact on operating profit, net income or earnings per share.

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) (UNAUDITED)

5. Basic earnings per common share is computed by dividing net income by the weighted average number of shares of common stock outstanding. Diluted earnings per common share is computed by dividing net income plus dividends by the weighted average number of shares of common stock outstanding plus other potentially dilutive securities. Diluted earnings per common share was not adjusted for the impact of the Convertible Preferred Securities as they were anti-dilutive for all periods presented.

A reconciliation of the number of shares utilized for the calculation of diluted earnings per common share follows:

	THIRTY-SIX WEEKS ENDER		
	SEPTEMBER 11, 1998	1997	
		LLIONS)	
Weighted average number of common shares outstanding	204.1	202.8	
assumed purchased at average market price Assuming distribution of common shares upon redemption of Convertible Preferred	4.2	4.8	
Securities			
average market price	.1	.2	
Shares utilized for the calculation of diluted			
earnings per share	208.4 =====	207.8 =====	

6. As of September 11, 1998, the Company had minority interests in 18 affiliates that own an aggregate of 240 properties, 20 of which are full-service properties, managed primarily by Marriott International, Inc. The Company's equity in earnings of affiliates was \$1 million and \$3 million, respectively, for the thirty-six weeks ended September 11, 1998 and September 11, 1997.

 ${\hbox{\tt Combined summarized operating results reported by affiliates follows:} \\$ 

THIRTY-SIX WEEKS ENDE		
SEPTEMBER 11,	SEPTEMBER 12,	
1998	1997	
(IN MI	LLIONS)	
\$ 369	\$ 447	
(223)	(278)	
(101)	(140)	
45	29	
4	12	
\$ 49	\$ 41	
=====	=====	
	SEPTEMBER 11, 1998 (IN MI \$ 369 (223) (101)  45 4	

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) (UNAUDITED)

In the first quarter of 1998, the Company obtained a controlling interest in the partnership that owns the 1,671-room Atlanta Marriott Marquis for approximately \$239 million, including \$164 million in assumed mortgage debt. The Company previously owned a 1.3% general and limited partnership interest.

In the second quarter of 1998, the Company acquired the partnership that owns the 289-room Park Ridge Marriott in Park Ridge, New Jersey for \$24 million. The Company previously owned a 1% managing general partner interest and held a note receivable interest of approximately \$5 million.

7. In the first quarter of 1998, the Company acquired a controlling interest in, and became the managing general partner for, the partnership that owns the 359-room Albany Marriott, the 350-room San Diego Marriott Mission Valley and the 320-room Minneapolis Marriott Southwest for approximately \$50 million. Also, during the first quarter of 1998, the Company acquired the Gables at Winchester in suburban Boston, a 124-unit senior living community, for \$21 million and entered into conditional purchase agreements to acquire two Marriott Brighton Gardens assisted living communities in Denver and Colorado Springs, Colorado, for \$35 million in 1999 after the anticipated completion of construction, if they achieve certain operating performance criteria. All three of these communities will be operated by Marriott Senior Living Services, Inc. ("MSLS") under long-term operating agreements.

In the second quarter of 1998, the Company sold the 662-room New York Marriott East Side for approximately \$191 million and recorded a pre-tax gain of approximately \$40 million. The Company also sold the 191-room Napa Valley Marriott for approximately \$21 million and recorded a pre-tax gain of approximately \$10 million. Also, during the second quarter of 1998, the Company acquired the 397-room Ritz-Carlton, Tysons Corner for approximately \$96 million and the 281-room Ritz-Carlton, Phoenix for approximately \$75 million. In addition, the Company acquired the 487-room Torrance Marriott near Los Angeles, California for approximately \$52 million.

During the third quarter of 1998, the Company acquired the 308-room Ritz-Carlton, Dearborn, Michigan for approximately \$65 million, the 336-room Ritz-Carlton, San Francisco for approximately \$161 million and the 404-room Memphis Marriott (which was converted to the Marriott brand upon acquisition) for approximately \$16 million.

- 8. In March 1997, the Company purchased 100% of the outstanding bonds secured by a first mortgage on the San Francisco Marriott. The Company purchased the bonds for \$219 million, an \$11 million discount to the face value of \$230 million. In connection with the redemption and defeasance of the bonds, the Company recognized an extraordinary gain of \$5 million, which represents the \$11 million discount and the write-off of deferred financing fees, net of taxes.
- 9. The Company operates in two business segments in the lodging industry: hotels and senior living communities. The Company's hotels are primarily operated under the Marriott or Ritz-Carlton brands. The Company's senior living communities are operated under Marriott brands.

The Company evaluates the performance of its segments based primarily on operating profit before depreciation, corporate expenses, and interest expense. The Company's income taxes are included in the consolidated Federal income tax return of the Company and its affiliates and is allocated based upon the relative contribution to the Company's consolidated taxable income or loss and changes in temporary differences. The allocation of income taxes is not evaluated at the segment level and, therefore, the Company does not believe the information is material to the condensed consolidated financial statements.

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) (UNAUDITED)

	THIRTY-SIX	WEEKS	ENDED	SEPTEMBER	11.	1998
--	------------	-------	-------	-----------	-----	------

(52)

66

	HOTELS	SENIOR LIVING	CORPORATE & OTHER	CONSOLIDATED
Revenues	\$ 922	\$ 58	\$ 60	\$1,040
Operating profit		30	43	493
Interest income		1		36
Interest expense		(17)		(245)
Other	` ,		(73)	(109)
Income (loss) before	(00)		(10)	(100)
income taxes	191	9	(25)	175
Income taxes	101	J	(23)	175
	ТН	TRTY-STX WEEKS	ENDED SEPTEMBER 12	2 1997
	<b>HOTELS</b>	SENIOR LIVING	CORPORATE & OTHER	CONSOLIDATED
Revenues	\$ 736	\$ 16	\$ 16	\$ 768
Operating profit	202	•	(7)	204
(loss)		6	(7)	304
Interest income			12	37
Interest expense	` ,	(6)	(6)	(198)
Other	(24)		(53)	(77)
Income (loss) before				
incomo tavas	110		(E2)	66

As of September 11, 1998 and September 12, 1997, the Company's foreign operations consist of four full-service hotel properties located in Canada and two full-service hotel properties located in Mexico. There were no intercompany sales between the properties and the Company. The following table presents revenues for each of the geographical areas in which the Company operates (in millions):

income taxes..... 118

	THIRTY-SIX	WEEKS ENDED
	SEPTEMBER 11, 1998	1997
United States	\$1,009	\$743
International	31	25
Total	\$1,040	\$768
	=====	====

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) (UNAUDITED)

10. In the first quarter of 1998, the Company adopted SFAS No. 130, "Reporting Comprehensive Income," ("SFAS 130"). SFAS 130 establishes standards for reporting and display of comprehensive income and its components in financial statements. The objective of SFAS 130 is to report a measure of all changes in equity of an enterprise that result from transactions and other economic events of the period other than transactions with owners. Comprehensive income is the total of net income and all other nonowner changes in equity.

The Company's only component of other comprehensive income is the right to receive up to 1.4 million shares of Host Marriott Services Corporation's ("HMSC") common stock or an equivalent cash value subsequent to exercise of the options held by certain former and current employees of Marriott International. For the thirty-six weeks ended September 11, 1998, the other comprehensive loss was \$3 million. For the thirty-six weeks ended September 11, 1998, the comprehensive loss was \$51 million. For the thirty-six weeks ended September 12, 1997, other comprehensive income was \$8 million. For the thirty-six weeks ended September 12, 1997, comprehensive income was \$51 million. As of September 11, 1998, the Company's accumulated other comprehensive income was approximately \$7 million.

- 11. In the second quarter of 1998, the Company prepaid 92 million of 98 unsecured debt provided by Marriott International related to the Company's senior living communities.
- 12. In December 1996, Host Marriott Financial Trust (the "Issuer"), a wholly-owned subsidiary trust of the Company, issued 11 million shares of 6 3/4% convertible quarterly income preferred securities (the "Convertible Preferred Securities"), with a liquidation preference of \$50 per share (for a total liquidation amount of \$550 million). The Convertible Preferred Securities represent an undivided beneficial interest in the assets of the Issuer. The payment of distributions out of moneys held by the Issuer and payments on liquidation of the Issuer or the redemption of the Convertible Preferred Securities are guaranteed by the Company to the extent the Issuer has funds available therefor. This guarantee, when taken together with the Company's obligations under the indenture pursuant to which the Debentures were issued, the Debentures, the Company's obligations under the Trust Agreement and its obligations under the indenture to pay costs, expenses, debts and liabilities of the Issuer (other than with respect to the Convertible Preferred Securities) provides a full and unconditional guarantee of amounts due on the Convertible Preferred Securities. Proceeds from the issuance of the Convertible Preferred Securities were invested in 6 3/4% Convertible Subordinated Debentures (the "Debentures") due December 2, 2026 issued by the Company. The Issuer exists solely to issue the Convertible Preferred Securities and its own common securities (the "Common Securities") and invest the proceeds therefrom in the Debentures, which is its sole asset. Separate financial statements of the Issuer are not presented because of the Company's guarantee described above; the Company's management has concluded that such financial statements are not material to investors and the Issuer is wholly-owned and essentially has no independent operations.

Each of the Convertible Preferred Securities is convertible at the option of the holder into shares of Company common stock at the rate of 2.6876 shares per Convertible Preferred Security (equivalent to a conversion price of \$18.604 per share of Company common stock). The Debentures are convertible at the option of the holders into shares of Company common stock at a conversion rate of 2.6876 shares for each \$50 in principal amount of Debentures. The Issuer will only convert Debentures pursuant to a notice of conversion by a holder of Convertible Preferred Securities.

Holders of the Convertible Preferred Securities are entitled to receive preferential cumulative cash distributions at an annual rate of 6 3/4% accruing from the original issue date, commencing March 1, 1997, and

## NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) (UNAUDITED)

payable quarterly in arrears thereafter. The distribution rate and the distribution and other payment dates for the Convertible Preferred Securities will correspond to the interest rate and interest and other payment dates on the Debentures. The Company may defer interest payments on the Debentures for a period not to exceed 20 consecutive quarters. If interest payments on the Debentures are deferred, so too are payments on the Convertible Preferred Securities. Under this circumstance, the Company will not be permitted to declare or pay any cash distributions with respect to its capital stock or debt securities that rank pari passu with or junior to the Debentures.

Subject to certain restrictions, the Convertible Preferred Securities are redeemable at the Issuer's option upon any redemption by the Company of the Debentures after December 2, 1999. Upon repayment at maturity or as a result of the acceleration of the Debentures upon the occurrence of a default, the Debentures shall be subject to mandatory redemption, from which the proceeds will be applied to redeem Convertible Preferred Securities and Common Securities, together with accrued and unpaid distributions.

#### REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To Host Marriott Corporation:

We have audited the accompanying combined consolidated balance sheets of Host Marriott Hotels (as defined in Note 1) as of January 2, 1998 and January 3, 1997, and the related combined consolidated statements of operations and cash flows for each of the three fiscal years in the period ended January 2, 1998. These financial statements are the responsibility of Host Marriott Corporation's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform an audit to obtain reasonable assurance about whether the combined consolidated financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the combined consolidated financial statements referred to above present fairly, in all material respects, the financial position of Host Marriott Hotels as of January 2, 1998 and January 3, 1997, and the results of their operations and their cash flows for each of the three fiscal years in the period ended January 2, 1998, in conformity with generally accepted accounting principles.

As discussed in Notes 1 and 2 to the combined consolidated financial statements, in 1995 Host Marriott Hotels changed its method of accounting for the impairment of long-lived assets.

Arthur Andersen LLP

Washington, D.C. May 22, 1998

#### COMBINED CONSOLIDATED BALANCE SHEETS JANUARY 2, 1998 AND JANUARY 3, 1997 (IN MILLIONS)

	1997	1996
ASSETS  Property and Equipment, net	\$4,634	\$3,805
affiliates of \$23 million and \$156 million, respectively)  Due from Managers	54 87 13 271 354	
Cash and Cash Equivalents	\$5,907	704  \$5,152 =====
LIABILITIES AND EQUITY		
Debt Senior Notes Mortgage Debt Other Convertible Debt Obligation to Host Marriott Corporation	1,784 97 567	1,529 97
Accounts Payable and Accrued Expenses  Deferred Income Taxes	4,033 59 487 371	3,214 74
Total Liabilities		
Equity Investments and Advances from Host Marriott Corporation	957	1,110
	. ,	\$5,152 =====

See Notes to Combined Consolidated Financial Statements.

## COMBINED CONSOLIDATED STATEMENTS OF OPERATIONS FISCAL YEARS ENDED JANUARY 2, 1998, JANUARY 3, 1997 AND DECEMBER 29, 1995 (IN MILLIONS)

	1997	1996	1995
REVENUES			
Rooms	776 180	\$1,302 515 125	363 81
Total hotel revenues	2,806 (11) 5	1,942 1	1,352 (3)
Total revenues	2,823		1,362
OPERATING COSTS AND EXPENSES			
Hotel property-level costs and expenses Rooms  Food and beverage Other department costs and deductions Other hotel operating costs and expenses (including	428 592 693	313 406 506	226 284 368
Marriott International management fees of \$162 million, \$101 million, and \$67 million, respectively) Other (including a \$60 million write-down of undevel-	649	461	281
oped land in 1995)		38	
Total operating costs and expenses	2,391	1,724	
OPERATING PROFIT BEFORE MINORITY INTEREST, CORPORATE EXPENSES AND INTEREST	432		114
million in 1997 and 1996, respectively)		(240) 48	
INCOME (LOSS) FROM CONTINUING OPERATIONS BEFORE INCOME			
TAXES  Benefit (provision) for income taxes	82 (35)	(8) (5)	13
INCOME (LOSS) FROM CONTINUING OPERATIONS DISCONTINUED OPERATIONS		(13)	
Loss from discontinued operations (net of income tax benefit of \$3 million in 1995)			(8)
benefit of \$23 million in 1995)			(53)
INCOME (LOSS) BEFORE EXTRAORDINARY ITEMS Extraordinary itemsGain (loss) on extinguishment of debt (net of income tax expense (benefit) of \$1 million			(123)
in 1997 and (\$10) million in 1995)	3		(20)
NET INCOME (LOSS)		\$ (13)	

See Notes to Combined Consolidated Financial Statements.

## COMBINED CONSOLIDATED STATEMENTS OF CASH FLOWS FISCAL YEARS ENDED JANUARY 2, 1998, JANUARY 3, 1997 AND DECEMBER 29, 1995 (IN MILLIONS)

	1997	1996	1995
OPERATING ACTIVITIES Income (loss) from continuing operations	\$ 47	\$ (13)	\$ (62)
Depreciation and amortization	231 (20) (4)	168 (35) (6)	122 (35) (7)
Net (gains) losses on property transactions	19 (4) 62	(3) 49	70´  33
Changes in operating accounts: Other assets Other liabilities	57 44	9 32	(2) (9)
Cash from continuing operations	432	205 (4)	110
Cash from operations		201	142
INVESTING ACTIVITIES Proceeds from sales of assets	51	373	358
Cash received from sales of assets			325
Capital expenditures: Capital expenditures for renewals and replacements Lodging construction funded by project financing New investment capital expenditures	(29)	(87) (3) (69)	(56) (40) (64)
Purchases of short-term marketable securities  Notes receivable collections  Affiliate notes receivable and collections, net  Other	(354) 6 (6) 13		2
Cash used in investing activities from continuing			
operations Cash used in investing activities from discontinued	, ,	, ,	` ,
operations			
Cash used in investing activities	(807)		
FINANCING ACTIVITIES  Issuances of debt			1,251
Issuances of common stock by Host Marriott	6 (90)	454 (82)	13 (100)
Debt prepayments		(173) (17) 28	(960)  
Cash from financing activities from continuing			
operationsCash used in financing activities from discontinued operations	165 	806	204
Cash from financing activities	165	806	200
INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS CASH AND CASH EQUIVALENTS, beginning of year	(210)	503 201	134 67
CASH AND CASH EQUIVALENTS, end of year	\$ 494	\$ 704	
Non-cash financing activities: Assumption of mortgage debt for the acquisition of, or purchase of controlling interests in, certain hotel	====	====	
properties		\$ 696 =====	

See Notes to Combined Consolidated Financial Statements.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS

#### 1. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

#### BASIS OF PRESENTATION

On April 16, 1998, the Board of Directors of Host Marriott Corporation ("Host Marriott") approved a plan to reorganize Host Marriott's current business operations by spin-off of Host Marriott's senior living business ("Senior Living") and contribution of Host Marriott's hotels and certain other assets and liabilities to a newly formed Delaware limited partnership, Host Marriott, L.P. (the "Operating Partnership"). Host Marriott will merge into HMC Merger Corporation, a newly formed Maryland corporation (to be renamed "Host Marriott Corporation" also referred to herein as "the REIT"), which will qualify as a real estate investment trust ("REIT") and be the sole general partner of the Operating Partnership. Host Marriott's contribution of its hotels and certain assets and liabilities to the Operating Partnership (the "Contribution") in exchange for units of limited partnership interests in the Operating Partnership will be accounted for at Host Marriott's historical basis.

The accompanying combined consolidated financial statements include the accounts of the Host Marriott hotels and the assets and liabilities expected to be included in the Contribution by Host Marriott to the Operating Partnership upon its planned conversion to a REIT (the "REIT Conversion") . In these combined consolidated financial statements, Host Marriott Hotels is referred to as "Host Marriott Hotels" or the "Company." The combined consolidated financial statements exclude the assets, liabilities, equity, operations and cash flows related to Host Marriott's portfolio of 31 senior living communities. After the REIT Conversion, Senior Living will own these assets and lease the existing hotels from the Company.

In connection with the REIT Conversion, the Operating Partnership is proposing to acquire the remaining interests in eight public limited partnerships in which Host Marriott or its subsidiaries are general partners that own or control 24 full-service hotels in exchange for units in the Operating Partnership ("OP Units"), which could be exchangeable for notes of the Operating Partnership or common stock in the REIT in certain circumstances. Five of the partnerships (nine hotels) are already controlled and consolidated by Host Marriott as are two of the hotels in another of the partnerships for which a subsidiary of Host Marriott provided 100% non-recourse financing for the acquisition of these two hotels. The Operating Partnership is also proposing to acquire certain private partnerships in which Host Marriott or its subsidiaries are general partners in exchange for OP Units. Each OP Unit will be redeemable for cash in an amount equal to the market value of a share of Host Marriott common stock or, at the election of Host Marriott Corporation (the general partner), a share of Host Marriott common stock, beginning one year after the issuance of the OP Unit.

However, consummation of the REIT Conversion is subject to significant contingencies that are outside the control of the Company, including final Board approval, consent of shareholders, partners, bondholders, lenders and ground lessors of Host Marriott, its affiliates and other third parties. Accordingly, there can be no assurance that the REIT Conversion will be completed.

#### DESCRIPTION OF BUSINESS

As of January 2, 1998, the Company owned, or had controlling interests in, 95 upscale and luxury full-service hotel lodging properties generally located throughout the United States and operated under the Marriott and Ritz-Carlton brand names. Most of these properties are managed by Marriott International, Inc. ("Marriott International"). At that date, the Company also held minority interests in various partnerships that own 242 additional properties, including 22 full-service hotel properties, managed by Marriott International.

On December 29, 1995, Host Marriott distributed to its shareholders through a special tax-free dividend (the "Special Dividend") its food, beverage, and merchandise concessions business at airports, on tollroads, and at arenas and other attractions (the "Operating Group"). See Note 2 for a discussion of the Special Dividend. The 1995 financial statements were restated to reflect the Operating Group as discontinued operations.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

The structure of Host Marriott was substantially altered on October 8, 1993 (the "Marriott International Distribution Date") when the Company distributed the stock of a wholly-owned subsidiary, Marriott International, Inc., in a special dividend (the "Marriott International Distribution"). See Note 14 for a description of the Marriott International Distribution and related transactions.

An analysis of the activity in the "Investments and Advances from Host Marriott Corporation" follows (in millions):

Balance, December 30, 1994  Net loss  Distribution of Host Marriott Services Corporation  Issuances of common stock and other activity of Host Marriott	
Balance, December 29, 1995	675 (13) (4) (17) 469
Balance, January 3, 1997	50 (226) 23

The average balance in the "Investment and Advances from Host Marriott Corporation" was \$692 million for 1995, \$893 million for 1996 and \$1,034 million for 1997. The "Cash transfers to Host Marriott" reflects cash transfers to Host Marriott for the purchase of the Senior Living assets which, as contemplated, will be spun-off in conjunction with the REIT Conversion.

#### PRINCIPLES OF CONSOLIDATION

The combined consolidated financial statements include the accounts of the Company and its subsidiaries and controlled affiliates. Investments in affiliates over which the Company has the ability to exercise significant influence, but does not control, are accounted for using the equity method. All material intercompany transactions and balances have been eliminated.

#### FISCAL YEAR

The Company's fiscal year ends on the Friday nearest to December 31. Fiscal years 1997 and 1995 included 52 weeks compared to 53 weeks for fiscal year 1996.

#### REVENUES AND EXPENSES

Revenues primarily represent sales generated by the Company's hotels. Other hotel operating costs and expenses include depreciation, management fees, real and personal property taxes, ground and equipment rent, insurance and certain other costs.

#### EARNINGS (LOSS) PER OP UNIT

Basic and diluted earnings per OP Unit have been calculated based on the number of Host Marriott common shares outstanding for all periods presented because it is expected that upon the REIT Conversion the Operating Partnership will issue OP Units to Host Marriott in exchange for the Contribution equal to the number of shares

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

of outstanding Host Marriott common stock, accordingly, the following discussion of earnings (loss) per OP Unit is on a pro forma basis as if the REIT Conversion and the Contribution had occurred.

Basic earnings (loss) per OP Unit are computed by dividing net income (loss) by the weighted average number of shares of common stock outstanding of Host Marriott. Diluted earnings (loss) per OP Unit are computed by dividing net income (loss) by the weighted average number of shares of common stock outstanding plus other dilutive securities of Host Marriott. Diluted earnings (loss) per OP Unit has not been adjusted for the impact of the Convertible Debt Obligation to Host Marriott Corporation for 1997 and 1996 and for the comprehensive stock plan and warrants for 1996 and 1995 as they are antidilutive.

Basic and diluted (loss) earnings per OP Unit on a pro forma basis are as follows:

		1996	
Basic earnings (loss) per OP Unit: Continuing operations		\$(.07) 	
debt (net of income taxes)			. ,
Basic earnings (loss) per OP Unit	\$ .25		\$(.90)
Diluted earnings (loss) per OP Unit:			
Continuing operations		\$(.07) 	
debt (net of income taxes)			
Diluted earnings (loss) per OP Unit			

A reconciliation of the number of shares utilized (based on Host Marriott shares) for the calculation of dilutive earnings per OP Unit follows (in millions):

	1997	1996	1995
Weighted average number of common shares outstanding Assuming distribution of common shares granted under comprehensive stock plan, less shares assumed purchased	203.1	188.7	158.3
at average market price	4.8		
price	.3		
Shares utilized for the calculation of diluted earnings			
per OP Unit	208.2 =====	188.7 =====	158.3 =====

#### INTERNATIONAL OPERATIONS

The combined consolidated statements of operations include the following amounts related to non-U.S. subsidiaries and affiliates of Host Marriott: revenues of \$39 million and \$18 million and loss before income taxes of \$9 million and \$2 million in 1997 and 1996, respectively. International revenues and income before income taxes in 1995 were not material.

#### PROPERTY AND EQUIPMENT

Property and equipment is recorded at cost. For newly developed properties, cost includes interest, rent and real estate taxes incurred during development and construction. Replacements and improvements are capitalized.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

Depreciation is computed using the straight-line method over the estimated useful lives of the assets, generally 40 years for buildings and three to ten years for furniture and equipment. Leasehold improvements are amortized over the shorter of the lease term or the useful lives of the related assets.

Gains on sales of properties are recognized at the time of sale or deferred to the extent required by generally accepted accounting principles. Deferred gains are recognized as income in subsequent periods as conditions requiring deferral are satisfied or expire without further cost to the Company.

In cases where management is holding for sale particular hotel properties, the Company assesses impairment based on whether the estimated sales price less costs of disposal of each individual property to be sold is less than the net book value. A property is considered to be held for sale when the Company has made the decision to dispose of the property. Otherwise, the Company assesses impairment of its real estate properties based on whether it is probable that undiscounted future cash flows from each individual property will be less than its net book value. If a property is impaired, its basis is adjusted to its fair market value.

#### **DEFERRED CHARGES**

Deferred financing costs related to long-term debt are deferred and amortized over the remaining life of the debt.

#### CASH, CASH EQUIVALENTS AND SHORT-TERM MARKETABLE SECURITIES

The Company considers all highly liquid investments with a maturity of 90 days or less at the date of purchase to be cash equivalents. Cash and cash equivalents includes approximately \$103 million and \$67 million at January 2, 1998 and January 3, 1997, respectively, of cash related to certain consolidated partnerships, the use of which is restricted generally for partnership purposes to the extent it is not distributed to the partners. Short-term marketable securities include investments with a maturity of 91 days to one year at the date of purchase. The Company's short-term marketable securities represent investments in U.S. government agency notes and high quality commercial paper. The short-term marketable securities are categorized as available for sale and, as a result, are stated at fair market value.

#### CONCENTRATIONS OF CREDIT RISK

Financial instruments that potentially subject the Company to significant concentrations of credit risk consist principally of cash, cash equivalents and short-term marketable securities. The Company maintains cash and cash equivalents and short-term marketable securities with various high credit-quality financial institutions. The Company performs periodic evaluations of the relative credit standing of these financial institutions and limits the amount of credit exposure with any institution.

#### USE OF ESTIMATES IN THE PREPARATION OF FINANCIAL STATEMENTS

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

#### SELF-INSURANCE PROGRAMS

Prior to the Marriott International Distribution Date, the Company was self-insured for certain levels of general liability, workers' compensation and employee medical coverage. Estimated costs of these self-insurance programs were accrued at present values of projected settlements for known and anticipated claims. The

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Company discontinued its self-insurance programs for claims arising subsequent to the Marriott International Distribution Date.

#### INTEREST RATE SWAP AGREEMENTS

The Company has entered into a limited number of interest rate swap agreements to diversify certain of its debt to a variable rate or fixed rate basis. The interest rate differential to be paid or received on interest rate swap agreements is accrued as interest rates change and is recognized as an adjustment to interest expense.

#### NEW STATEMENTS OF FINANCIAL ACCOUNTING STANDARDS

The Company adopted Statements of Financial Accounting Standards ("SFAS") No. 114, "Accounting by Creditors for Impairment of a Loan" and SFAS No. 121, "Accounting for the Impairment of Long-Lived Assets and for Long-Lived Assets to be Disposed Of" during 1995. Adoption of these statements did not have a material effect on the Company's continuing operations. See Note 2 for a discussion of the adoption of SFAS No. 121 on discontinued operations.

During 1996, the Company adopted SFAS No. 123, "Accounting for Stock-Based Compensation." The adoption of SFAS No. 123 did not have a material effect on the Company's combined consolidated financial statements. (See Note 10.)

During 1997, the Company adopted SFAS No. 128, "Earnings Per Share," SFAS No. 129, "Disclosure of Information About Capital Structure" and SFAS No. 131, "Disclosures About Segments of an Enterprise and Related Information." The adoption of these statements did not have a material effect on the Company's combined consolidated financial statements and the appropriate disclosures required by these statements have been incorporated herein. The Company will adopt SFAS No. 130, "Reporting Comprehensive Income," in 1998 and does not expect it to have a material effect on the Company's combined consolidated financial statements.

#### 2. HM SERVICES SPECIAL DIVIDEND

On December 29, 1995, Host Marriott distributed to its shareholders through the Special Dividend all of the outstanding shares of common stock of Host Marriott Services Corporation ("HM Services"), formerly a wholly-owned subsidiary of Host Marriott, which, as of the date of the Special Dividend, owned and operated food, beverage and merchandise concessions at airports, on tollroads and at stadiums and arenas and other tourist attractions. The Special Dividend provided Host Marriott shareholders with one share of common stock of HM Services for every five shares of Host Marriott common stock held by such shareholders on the record date of December 22, 1995. Host Marriott recorded approximately \$9 million of expenses related to the consummation of the Special Dividend in 1995. Revenues for Host Marriott's discontinued operations totaled \$1,158 million in 1995. The provision for loss on disposal includes the operating loss from discontinued operations from August 9, 1995 (measurement date) through December 29, 1995 of \$44 million, net of taxes, and estimated expenses related to the Special Dividend of \$9 million.

Effective September 9, 1995, the Company adopted SFAS No. 121, which requires that an impairment loss be recognized when the carrying amount of an asset exceeds the sum of the undiscounted estimated future cash flows associated with the asset. As a result of the adoption of SFAS No. 121, the Company recognized a non-cash, pre-tax charge during the fourth quarter of 1995 of \$47 million. Such charge has been reflected in discontinued operations for fiscal year 1995.

For purposes of governing certain of the ongoing relationships between Host Marriott and HM Services after the Special Dividend and to provide for an orderly transition, Host Marriott and HM Services entered into various

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

agreements including a Distribution Agreement, an Employee Benefits Allocation Agreement, a Tax Sharing Agreement and a Transitional Services Agreement. Effective as of December 29, 1995, these agreements provide, among other things, for the division between Host Marriott and HM Services of certain assets and liabilities, including but not limited to liabilities related to employee stock and other benefit plans and the establishment of certain obligations for HM Services to issue shares upon exercise of warrants (see Note 7) and to issue shares or pay cash to Host Marriott upon exercise of stock options held by certain former employees of Host Marriott (see Note 10).

#### 3. PROPERTY AND EQUIPMENT

Property and equipment consists of the following:

	1997	1996
	(IN MIL	LIONS)
Land and land improvements  Buildings and leasehold improvements  Furniture and equipment  Construction in progress		\$ 349 3,507 548 82
Less accumulated depreciation and amortization	5,469 (835)  \$4,634	

Interest cost capitalized in connection with the Company's development and construction activities totaled \$1 million in 1997, \$3 million in 1996 and \$5 million in 1995.

In 1997, the Company, through an agreement with the ground lessor of one of its properties terminated its ground lease and recorded a \$15 million loss on the write-off of its investment, including certain transaction costs, which has been included in net gains (losses) on property transactions in the accompanying combined consolidated financial statements.

In 1996, the Company recorded additional depreciation expense of \$15 million as a result of a change in the estimated depreciable lives and salvage values for certain hotel properties. Also, in 1996, the Company recorded a \$4 million charge to write down an undeveloped land parcel to its net realizable value based on its expected sales value.

In 1995, the Company made a determination that its owned Courtyard and Residence Inn properties were held for sale and recorded a \$10 million charge to write down the carrying value of five of these individual properties to their estimated net realizable values. In the fourth quarter of 1995, management instituted a program to liquidate certain non-income producing assets and to reinvest the proceeds in the acquisition of full-service hotels. As part of this program, management determined that a 174-acre parcel of undeveloped land in Germantown, Maryland that was to be developed into an office project over an extended period of time would no longer be developed and instead decided to attempt to sell the property. Accordingly, the Company recorded a pre-tax charge of \$60 million in the fourth quarter of 1995 to reduce the asset to its estimated sales value. In 1997, the Company sold a portion of the land parcel at its approximate net book value of \$11 million.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

#### 4. INVESTMENTS IN AND RECEIVABLES FROM AFFILIATES

Investments in and receivables from affiliates consist of the following:

OWNERSHIP INTERESTS 1997 1996 (IN MILLIONS)

Equity investments Hotel partnerships which own 22 full-service Marriott Hotels, 120 Courtyard hotels, 50 Residence Inns and 50 Fairfield Inns operated by Marriott International, as of January 2, 1998.....

Notes and other receivables, net.....

1%-50% \$13 \$ 11 23 156 \$36 \$167 === ====

Hotel properties owned by affiliates generally were acquired from the Company in connection with limited partnership offerings. The Company or one of its subsidiaries typically serve as a general partner of each partnership and the hotels are operated by Marriott International under long-term agreements.

In 1997, the Company acquired all of the outstanding interests in the Chesapeake Hotel Limited Partnership ("CHLP") that owns six hotels and acquired controlling interests in three affiliated partnerships for approximately \$510 million, including the assumption of approximately \$395 million of debt. These affiliated partnerships included the partnerships that own the 353-room Hanover Marriott and the 884-room Marriott's Desert Springs Resort and Spa and the Marriott Hotel Properties Limited Partnership ("MHPLP") that owns the 1,503-room Marriott Orlando World Center and a 50.5% interest in the 624-room Marriott Harbor Beach Resort. Subsequent to year-end, the Company obtained a controlling interest in the partnership that owns the 1,671-room Atlanta Marriott Marquis for approximately \$239 million, including the assumption of \$164 million of mortgage debt.

In 1996, the Company purchased controlling interests in four affiliated partnerships for \$640 million, including \$429 million of existing debt. These affiliated partnerships included the partnership that owns the 1,355-room San Diego Marriott Hotel and Marina; the Marriott Hotel Properties II Limited Partnership that owns the 1,290-room New Orleans Marriott, the 999-room San Antonio Marriott Rivercenter, the 368-room San Ramon Marriott, and a 50% limited partner interest in the 754-room Santa Clara Marriott; the Marriott Suites Limited Partnership that owns four hotels; and the partnership that owns the 510-room Salt Lake City Marriott.

Receivables from affiliates are reported net of reserves of \$144 million at January 2, 1998 and \$227 million at January 3, 1997. Receivables from affiliates at January 2, 1998 include a \$10 million debt service guarantee for the partnership that owns the Atlanta Marriott Marquis, which was repaid in early 1998. Receivables from affiliates at January 3, 1997 included a \$140 million mortgage note at 9% that amortizes through 2003, which is eliminated in the consolidated financial statements in 1997. The Company has committed to advance additional amounts to affiliates, if necessary, to cover certain debt service requirements. Such commitments are limited, in the aggregate, to an additional \$60 million at January 2, 1998. Subsequent to January 2, 1998, this amount was reduced to \$20 million in connection with the refinancing and acquisition of a controlling interest in the Atlanta Marriott Marquis. Net amounts repaid to the Company under these commitments totaled \$2 million and \$13 million in 1997 and 1996, respectively. Net amounts funded by the Company totaled \$10 million in 1997 and \$8 million in 1995. There were no fundings in 1996.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

The Company's pre-tax income from affiliates includes the following:

	1997	1996	1995
	(IN	MILLI	ONS)
Interest income			
Equity in net income	5	3	
	\$16	\$20	\$16
	===	===	===

Combined summarized balance sheet information for the Company's affiliates follows:

	1997	
	(IN MIL	
	(IN MILL	LIUNS)
Property and equipment	. ,	\$2,605
Other assets	283	331
Total assets		
10001 000000111111111111111111111111111	=====	=====
Debt, principally mortgages		•
Other liabilities		672
Partners' deficit	(328)	(570)
Total liabilities and partners' deficit	\$2 263	\$2 936
Total liabilities and partners deflett	=====	======

Combined summarized operating results for the Company's affiliates follow:

	1997	1996	1995
	(IN	MILLION	S)
Revenues Operating expenses:	\$ 603	\$ 731	\$ 759
Cash charges (including interest)  Depreciation and other non-cash charges			
Income before extraordinary items  Extraordinary itemsforgiveness of debt		42 12	24 181
Net income	\$ 77 =====	\$ 54 =====	\$ 205 =====

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

#### 5. DEBT

Debt consists of senior notes, mortgage notes, a convertible debt obligation to Host Marriott and other debt, all of which are included in these financial statements because the debt or replacement debt is expected to be contributed to the Company upon the REIT conversion. In August 1998 the Company repaid the Properties Notes, New Properties Note and Acquisition Notes described below. Debt consists of the following:

		1996
		LIONS)
Properties Notes, with a rate of 9 1/2% due May 2005  New Properties Notes, with a rate of 8 7/8% due July 2007  Acquisitions Notes, with a rate of 9% due December 2007  Senior Notes, with an average rate of 9 3/4% at January 2,	\$ 600 600 350	350
1998, maturing through 2012		71
Total Senior Notes	1,585	•
Mortgage debt (non-recourse) secured by \$2.4 billion of real estate assets, with an average rate of 8.5% at January 2, 1998, maturing through 2022		1,529
January 2, 1998) due June 2004	22	
Total Mortgage Debt	1,784	
Convertible debt obligation to Host Marriott, with a rate of 6.75%, due 2026 (see Note 6)	567	567
maturing through 2017		86
Capital lease obligations	8	11
Total Other	664	664
	\$4,033	\$3,214 =====

In May 1995, HMH Properties, Inc. ("Properties"), a wholly-owned subsidiary of Host Marriott Hospitality, Inc. ("Hospitality"), issued an aggregate of \$600 million of 9 1/2% senior secured notes (the "Properties Notes"). The bonds were issued in conjunction with a concurrent \$400 million offering by a subsidiary of the discontinued HM Services' business at par, and have a final maturity of May 2005. The net proceeds were used to defease, and subsequently redeem, all of the senior notes issued by Host Marriott Hospitality, Inc. and to repay borrowings under the line of credit with Marriott International. In connection with the redemptions and defeasance, the Company recognized an extraordinary loss in 1995 of \$17 million, net of taxes, related to continuing operations.

In December 1995, HMC Acquisition Properties, Inc. ("Acquisitions"), an indirect, wholly-owned subsidiary of Host Marriott, issued \$350 million of 9% senior notes (the "Acquisitions Notes"). The Acquisitions Notes were issued at par and have a final maturity of December 2007. A portion of the net proceeds were utilized to repay in full the outstanding borrowings under the \$230 million revolving line of credit (the "Acquisition Revolver"), which was then terminated. In connection with the termination of the Acquisition Revolver, the Company recognized an extraordinary loss in 1995 of \$3 million, net of taxes.

On July 10, 1997, Properties and Acquisitions completed consent solicitations (the "Consent Solicitations") with holders of their senior notes to amend certain provisions of their senior notes' indentures. The Consent

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Solicitations facilitated the merger of Acquisitions with and into Properties (the "Properties Merger"). The amendments to the indentures also increased the ability of Properties to acquire, through certain subsidiaries, additional properties subject to non-recourse indebtedness and controlling interests in corporations, partnerships and other entities holding attractive properties and increased the threshold required to permit Properties to make distributions to affiliates.

Concurrent with the Consent Solicitations and the Properties Merger, Properties issued an aggregate of \$600 million of 8 7/8% senior notes (the "New Properties Notes") at par with a maturity of July 2007. Properties received net proceeds of approximately \$570 million, net of the costs of the Consent Solicitations and the Offering, which will be used to fund future acquisitions of, or the purchase of interests in, full-service hotels and other lodging-related properties, as well as for general corporate purposes.

The Properties Notes, the Acquisitions Notes and the New Properties Notes are guaranteed on a joint and several basis by certain of Properties' subsidiaries and rank pari passu in right of payment with all other existing future senior indebtedness of Properties. Properties was the owner of 58 of the Company's 95 lodging properties at January 2, 1998.

The net assets of Properties at January 2, 1998 were approximately \$518 million, substantially all of which were restricted to the payment of dividends. The payment of dividends to the Company from net income is restricted to the extent that cumulative EBITDA (Earnings Before Interest Expense, Taxes, Depreciation and Amortization) as defined in the Indenture exceeds 220% of cash interest expense (as defined). The indentures governing the Properties Notes, the Acquisitions Notes and the New Properties Notes contain covenants that, among other things, limit the ability to incur additional indebtedness and issue preferred stock, pay dividends or make other distributions, repurchase capital stock or subordinated indebtedness, create certain liens, enter into certain transactions with affiliates, sell certain assets, issue or sell stock of subsidiaries, and enter into certain mergers and consolidations.

During 1997, Host Marriott, through a newly-created, wholly-owned subsidiary, HMC Capital Resources Corporation ("Capital Resources"), entered into a revolving line of credit agreement (the "Line of Credit") with a group of commercial banks under which it may borrow up to \$500 million for the acquisition of lodging real estate and for Host Marriott's working capital purposes. On June 19, 2000, any outstanding borrowings on the Line of Credit convert to a term loan arrangement with all unpaid advances due June 19, 2004. Borrowings under the Line of Credit bear interest at either the Eurodollar rate plus 1.7% or the Base Rate (as defined in the agreement) plus 0.7%, at the option of Host Marriott. An annual fee of 0.35% is charged on the unused portion of the commitment. The Line of Credit was originally secured by six hotel properties contributed to Capital Resources, with a carrying value of approximately \$500 million as of January 2, 1998, and is guaranteed by the Company. As a result of this transaction, Host Marriott terminated its line of credit with Marriott International. As of January 2, 1998, outstanding borrowings on the Line of Credit were approximately \$22 million as a result of a borrowing to fund the acquisition of the Ontario Airport Marriott.

Host Marriott also purchased 100% of the outstanding bonds secured by a first mortgage on the San Francisco Marriott in 1997. Host Marriott purchased the bonds for \$219 million, an \$11 million discount to the face value of \$230 million. In connection with the redemption and defeasance of the bonds, the Company recognized an extraordinary gain of \$5 million, which represents the \$11 million discount less the write-off of unamortized deferred financing fees, net of taxes.

In 1997, Host Marriott incurred approximately \$418 million of mortgage debt in conjunction with the acquisition of 11 hotels.

In conjunction with the construction of the Philadelphia Marriott, which was completed and opened in January 1995, the Company obtained first mortgage financing from Marriott International for 60% of the construction and development costs of the hotel. In the fourth quarter of 1996, Host Marriott repaid the \$109

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

million mortgage, prior to the rate increasing to 10% per annum with an additional 2% deferred, with the proceeds from the convertible preferred securities offering discussed in Note 6. In the first quarter of 1997, Host Marriott obtained \$90 million in first mortgage financing from two insurance companies secured by the Philadelphia Marriott. The mortgage bears interest at a fixed rate of 8.49% and matures in April 2009.

In December 1997, Host Marriott successfully completed the refinancing of the MHPLP mortgage debt for approximately \$152 million. The new mortgage bears interest at 7.48% and matures in January 2008. In connection with the refinancing, the Company recognized an extraordinary loss of \$2 million which represents payment of a prepayment penalty and the write-off of unamortized deferred financing fees, net of taxes.

At January 2, 1998, the Company was party to an interest rate exchange agreement with a financial institution (the contracting party) with an aggregate notional amount of \$100 million. Under this agreement, the Company collects interest based on specified floating interest rates of one month LIBOR (rate of 6% at January 2, 1998) and pays interest at fixed rates (rate of 7.99% at January 2, 1998). This agreement expires in 1998 in conjunction with the maturity of the mortgage on the New York Marriott Marquis. Also in 1997, the Company was party to two additional interest rate swap agreements with an aggregate notional amount of \$400 million which expired in May 1997. The Company realized a net reduction of interest expense of \$1 million in 1997, \$6 million in 1996 and \$5 million in 1995 related to interest rate exchange agreements. The Company monitors the creditworthiness of its contracting parties by evaluating credit exposure and referring to the ratings of widely accepted credit rating services. The Standard and Poors' long-term debt rating for the contracting party is A-. The Company is exposed to credit loss equal to the accrued interest receivable on the notional amount of interest rate exchange agreements outstanding (\$0 at January 2, 1998) in the event of non-performance by the contracting party to the interest rate swap agreements; however, the Company does not anticipate non-performance by the contracting party.

Aggregate debt maturities at January 2, 1998, excluding capital lease obligations, are (in millions):

1998	\$ 316
1999	28
2000	131
2001	132
2002	156
Thereafter	3,262
	\$4,025

Cash paid for interest for continuing operations, net of amounts capitalized was \$316 million in 1997, \$220 million in 1996 and \$177 million in 1995. Deferred financing costs, which are included in other assets, amounted to \$96 million and \$61 million, net of accumulated amortization, as of January 2, 1998 and January 3, 1997, respectively. Amortization of deferred financing costs totaled \$7 million, \$5 million and \$4 million in 1997, 1996 and 1995, respectively.

#### 6. CONVERTIBLE DEBT OBLIGATION TO HOST MARRIOTT CORPORATION

The obligation for the \$567 million of 6 3/4% Convertible Subordinated Debentures (the "Debentures") has been pushed down to these financial statements because it is expected that upon the REIT Conversion the Operating Partnership will assume primary liability for repayment of the Debentures of Host Marriott underlying the Convertible Preferred Securities of the Host Marriott Financial Trust (the "Issuer"), a wholly-owned subsidiary trust of Host Marriott. The common securities of Host Marriott Financial Trust will not be contributed to the Operating Partnership and therefore Host Marriott Financial Trust will not be consolidated by the Operating Partnership. Upon conversion by a Convertible Preferred Securities holder, the REIT will issue shares of its common stock which will be delivered to such holder. Upon the issuance of such shares by the REIT, the Operating Partnership will issue to the REIT the number of OP Units equal to the number of shares of the REIT common stock in exchange for the Debentures.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

In December 1996, the Issuer issued 11 million shares of 6 3/4% convertible quarterly income preferred securities (the "Convertible Preferred Securities"), with a liquidation preference of \$50 per share (for a total liquidation amount of \$550 million). The Convertible Preferred Securities represent an undivided beneficial interest in the assets of the Issuer. The payment of distributions out of moneys held by the Issuer and payments on liquidation of the Issuer or the redemption of the Convertible Preferred Securities are guaranteed by Host Marriott to the extent the Issuer has funds available therefor. This guarantee, when taken together with Host Marriott obligations under the indenture pursuant to which the Debentures were issued, the Debentures, Host Marriott's obligations under the Trust Agreement and its obligations under the indenture to pay costs, expenses, debts and liabilities  $% \left( \frac{1}{2}\right) =\left( \frac{1}{2}\right) \left( \frac{1}{2}\right)$ of the Issuer (other than with respect to the Convertible Preferred Securities) provides a full and unconditional guarantee of amounts due on the Convertible Preferred Securities. Proceeds from the issuance of the Convertible Preferred Securities were invested in the Debentures due December 2, 2026 issued by Host Marriott. The Issuer exists solely to issue the Convertible Preferred Securities and its own common securities (the "Common Securities") and invest the proceeds therefrom in the Debentures. The note receivable from Host Marriott for the Debentures is the Issuer's sole asset. Separate financial statements of the Issuer are not presented because of Host Marriott's guarantee described above; Host Marriott's management has concluded that such financial statements are not material to investors and the Issuer is wholly-owned and essentially has no independent operations.

Each of the Convertible Preferred Securities is convertible at the option of the holder into shares of Host Marriott common stock at the rate of 2.6876 shares per Convertible Preferred Security (equivalent to a conversion price of \$18.604 per share of Company common stock). The Debentures are convertible at the option of the holders into shares of Host Marriott common stock at a conversion rate of 2.6876 shares for each \$50 in principal amount of Debentures. The conversion rate is subject to adjustments in certain events, including (i) payment of dividends (and other distributions) on Host common stock by Host in shares of Host common stock; (ii) distributions to all holders of Host common stock of rights or warrants entitling such holders (for a period not to exceed 45 days) to subscribe for or purchase Host common stock at an exercise price less than the market price of Host common stock; (iii) subdivisions and combinations of Host common stock; (iv) payment of dividends (and other distributions) on Host common stock consisting of indebtedness of Host, capital stock or other securities, assets or cash (other than certain cash dividends at an annualized rate of up to 12.5% of the market price of Host common stock); (v) payments for Host common stock by Host or any of its subsidiaries in respect of a tender or exchange offer (other than an odd-lot offer) at a price per share in excess of 110% of the market price of Host common stock; (vi) consummation by Host of certain mergers, a consolidation, a sale of all or substantially all of its assets, a recapitalization or certain reclassifications of Host common stock. The distribution of the capital stock of Crestline to all holders of Host REIT common stock would, and certain other elements of the REIT Conversion (such as other distributions of Host's accumulated earnings and profits) will, result in an adjustment to the conversion price of the Debentures. The REIT Conversion could result in a material adjustment to the conversion price of the Debentures and the Convertible Preferred Securities depending upon the ultimate amount of the E&P Distribution and the value of Host REIT Common Stock. The Issuer will only convert Debentures pursuant to a notice of conversion by a holder of Convertible Preferred Securities. During 1997 and 1996, no shares were converted into common stock.

Holders of the Convertible Preferred Securities are entitled to receive preferential cumulative cash distributions at an annual rate of 6 3/4% accruing from the original issue date, commencing March 1, 1997, and payable quarterly in arrears thereafter. The distribution rate and the distribution and other payment dates for the Convertible Preferred Securities will correspond to the interest rate and interest and other payment dates on the Debentures. Host Marriott may defer interest payments on the Debentures for a period not to exceed 20 consecutive quarters. If interest payments on the Debentures are deferred, so too are payments on the Convertible Preferred Securities. Under this circumstance, Host Marriott will not be permitted to declare or pay any cash distributions with respect to its capital stock or debt securities that rank pari passu with or junior to the Debentures.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Subject to certain restrictions, the Convertible Preferred Securities are redeemable at the Issuer's option upon any redemption by Host Marriott of the Debentures after December 2, 1999. Upon repayment at maturity or as a result of the acceleration of the Debentures upon the occurrence of a default, the Debentures shall be subject to mandatory redemption, from which the proceeds will be applied to redeem Convertible Preferred Securities and Common Securities, together with accrued and unpaid distributions.

#### 7. SHAREHOLDERS' EQUITY OF HOST MARRIOTT

It is expected that upon the REIT Conversion that the Company will issue OP Units to Host Marriott in exchange for the Contribution equal to the number of shares of outstanding Host Marriott common stock. Additionally, limited partnership units issued to partners of the eight public limited partnerships and five private limited partnerships will be convertible on a one for one basis into a share of stock of Host Marriott for each OP Unit owned or at the election of Host Marriott Trust, in an amount equal to the market value of such shares beginning one year after the issuance of the OP Unit.

Six hundred million shares of common stock of Host Marriott, with a par value of \$1 per share, are authorized, of which 203.8 million and 202.0 million were issued and outstanding as of January 2, 1998 and January 3, 1997, respectively. One million shares of no par value preferred stock are authorized with none outstanding. During 1995, substantially all outstanding shares of such preferred stock were converted into approximately five million shares of Host Marriott common stock with the remainder defeased.

On March 27, 1996, Host Marriott completed the issuance of 31.6 million shares of common stock for net proceeds of nearly \$400 million.

In connection with a class action settlement, Host Marriott issued warrants to purchase up to 7.7 million shares of Host Marriott's common stock at \$8.00 per share through October 8, 1996 and \$10.00 per share thereafter. During 1996, 6.8 million warrants were exercised at \$8.00 per share and an equivalent number of shares of Host Marriott common stock were issued. During 1997, approximately 60,000 warrants were exercised at \$10.00 per share and an equivalent number of shares of Host Marriott common stock were issued. As of January 2, 1998, there were approximately 550,000 warrants outstanding.

In February 1989, the Board of Directors of Host Marriott adopted a shareholder rights plan under which a dividend of one preferred stock purchase right was distributed for each outstanding share of Host Marriott's common stock. Each right entitles the holder to buy 1/1,000th of a share of a newly issued series of junior participating preferred stock of Host Marriott at an exercise price of \$150 per share. The rights will be exercisable 10 days after a person or group acquires beneficial ownership of at least 20%, or begins a tender or exchange offer for at least 30%, of Host Marriott's common stock. Shares owned by a person or group on February 3, 1989 and held continuously thereafter are exempt for purposes of determining beneficial ownership under the rights plan. The rights are non-voting and will expire on February 2, 1999, unless exercised or previously redeemed by Host Marriott for \$.01 each. If Host Marriott is involved in a merger or certain other business combinations not approved by the Board of Directors, each right entitles its holder, other than the acquiring person or group, to purchase common stock of either Host Marriott or the acquiror having a value of twice the exercise price of the right.

#### 8. INCOME TAXES

The accompanying financial statements reflect the deferred income taxes related to the expected future tax consequences of those temporary differences specifically allocable to the Company based on the Contribution. Upon the REIT Conversion and the Contribution it is expected that the Company will be a limited partnership and taxable income or loss will be allocated among its partners. Further, Host Marriott expects to qualify as a REIT and will allocate its taxable income or loss to its shareholders. Accordingly, upon the REIT Conversion

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

and the Contribution, the Company will not have a Federal tax provision or a state tax provision in many states and in accordance with Statement of Financial Accounting Standards No. 109 the Company will record an adjustment to the tax provision in the fiscal year during which the REIT Conversion takes place for the tax effect of the reversal of certain of the Company's deferred taxes.

Total deferred tax assets and liabilities at January 2, 1998 and January 3, 1997 were as follows:

	1997	1996
	(IN MIL	LIONS)
Deferred tax assets  Deferred tax liabilities		
Net deferred income tax liability	\$ (487)	\$ (464)

The tax effect of each type of temporary difference and carryforward that gives rise to a significant portion of deferred tax assets and liabilities as of January 2, 1998 and January 3, 1997 follows:

	1997	1996
	(IN MIL	LIONS)
Investments in affiliates	\$ (310)	\$ (303)
Property and equipment	(179)	(135)
Safe harbor lease investments	(65)	(73)
Deferred tax gain	(92)	(92)
Reserves	103	97
Alternative minimum tax credit carryforwards	41	26
Other, net		16
Net deferred income tax liability	\$ (487)	\$ (464)
	=====	=====

The provision (benefit) for income taxes consists of:

	1997	1996	1995
	(IN	MILLIO	NS)
Current Federal		\$(2)	\$ 7
State	4	3	3
Foreign		3	
	25	4	10
Deferred Federal			(23)
State	2	(1)	
	10	1	(23)
	\$35	\$ 5	\$(13)
	===	===	====

At January 2, 1998, Host Marriott had approximately \$41 million of alternative minimum tax credit carryforwards available which do not expire.

Through 1997, Host Marriott settled with the Internal Revenue Service ("IRS") substantially all issues for tax years 1979 through 1993. Host Marriott expects to resolve any remaining issues with no material impact on the combined consolidated financial statements. Host Marriott made net payments to the IRS of approximately \$10 million and \$45 million in 1997 and 1996, respectively, related to these settlements. Certain adjustments totaling approximately \$2 million and \$11 million in 1996 and 1995, respectively, were made to the tax provision related to those settlements.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

A reconciliation of the statutory Federal tax rate to the Company's effective income tax rate follows:

	1997	1996	1995
Statutory Federal tax rate State income taxes, net of Federal tax benefit Tax credits Additional tax on foreign source income Tax contingencies Permanent items Other, net	4.9 (2.7) 6.0  0.1 (0.6)	21.7  40.8 25.0 9.0 1.0	2.5 (0.1)  14.6  0.7
21100tive inound can receive the interest in t	====	=====	=====

As part of the Marriott International Distribution and the Special Dividend, Host Marriott, Marriott International and HM Services entered into tax-sharing agreements which reflect each party's rights and obligations with respect to deficiencies and refunds, if any, of Federal, state or other taxes relating to the businesses of Host Marriott, Marriott International and HM Services prior to the Marriott International Distribution and the Special Dividend.

Cash paid for income taxes, including IRS settlements, net of refunds received, was \$56 million in 1997, \$40 million in 1996 and \$22 million in 1995.

#### 9. LEASES

The Company leases certain property and equipment under non-cancelable operating and capital leases. Future minimum annual rental commitments for all non-cancelable leases are as follows:

	CAPITAL LEASES	OPERATING LEASES
	(IN MI	ILLIONS)
1998	\$ 2	\$ 115
1999	2	111
2000	1	108
2001	1	106
2002		103
Thereafter	5	1,358
Total minimum lease payments	12	\$1,901
		=====
Less amount representing interest	(4)	
Present value of minimum lease payments	\$ 8	
	===	

As discussed in Note 12, Host Marriott sold and leased back 37 of its Courtyard properties in 1995 and an additional 16 Courtyard properties in 1996 to Hospitality Properties Trust. Additionally, in 1996, Host Marriott sold and leased back 18 of its Residence Inns to Hospitality Properties Trust. These leases, which are accounted for as operating leases and are included above, have initial terms expiring through 2012 for the Courtyard properties and 2010 for the Residence Inn properties, and are renewable at the option of the Company. Minimum rent payments are \$51 million annually for the Courtyard properties and \$17 million annually for the Residence Inn properties, and additional rent based upon sales levels are payable to the owner under the terms of the leases.

Leases also include long-term ground leases for certain hotels, generally with multiple renewal options. Certain leases contain provision for the payment of contingent rentals based on a percentage of sales in excess of stipulated amounts.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Certain of the lease payments included in the table above relate to facilities used in the Company's former restaurant business. Most leases contain one or more renewal options, generally for five or 10-year periods. Future rentals on leases have not been reduced by aggregate minimum sublease rentals of \$124 million payable to the Company under non-cancelable subleases.

The Company remains contingently liable at January 2, 1998 on certain leases relating to divested non-lodging properties. Such contingent liabilities aggregated \$110 million at January 2, 1998. However, management considers the likelihood of any substantial funding related to these leases to be remote.

Rent expense consists of:

		1996  MILLI	3 1995  IONS)
Minimum rentals on operating leases		\$83 16	\$34 17
	\$118	\$99	\$51
	====	===	===

#### 10. EMPLOYEE STOCK PLANS

It is expected that upon the REIT Conversion the Company will issue OP Units to Host Marriott in exchange for the Contribution equal to the number of shares of outstanding Host Marriott common stock. Additionally, OP Units issued to partners of the eight public limited partnerships and five private limited partnerships will be convertible on a one for one basis into shares of Host Marriott stock for each OP Unit owned or, at the election of Host Marriott Trust, in an amount equal to the market value of such shares beginning one year after the issuance of the OP Unit.

At January 2, 1998, Host Marriott has two stock-based compensation plans which are described below. Under the comprehensive stock plan (the "Comprehensive Plan"), Host Marriott may award to participating employees (i) options to purchase Host Marriott common stock, (ii) deferred shares of Host Marriott's common stock and (iii) restricted shares of Host Marriott's common stock. In addition, Host Marriott has an employee stock purchase plan (the "Employee Stock Purchase Plan"). The principal terms and conditions of the two plans are summarized below.

Total shares of common stock reserved and available for issuance under employee stock plans at January 2, 1998 are:

	(IN MILLIONS)
Comprehensive Plan	
Employee Stock Purchase Plan	3
	31
	===

Employee stock options may be granted to officers and key employees with an exercise price not less than the fair market value of the common stock on the date of grant. Options granted before May 11, 1990 expire 10 years after the date of grant and nonqualified options granted on or after May 11, 1990 expire up to 15 years after the date of grant. Most options vest ratably over each of the first four years following the date of the grant. In connection with the Marriott International Distribution, Host Marriott issued an equivalent number of Marriott International options and adjusted the exercise prices of its options, then outstanding, based on the relative trading prices of shares of the common stock of the two companies.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Host Marriott continues to account for expense under its plans under the provisions of Accounting Principle Board Opinion 25 and related interpretations as permitted under SFAS No. 123. Accordingly, no compensation cost has been recognized for its fixed stock options under the Comprehensive Plan and its Employee Stock Purchase Plan.

For purposes of the following disclosures required by SFAS No. 123, the fair value of each option granted has been estimated on the date of grant using an option-pricing model with the following weighted average assumptions used for grants in 1997, 1996 and 1995, respectively: risk-free interest rate of 6.2%, 6.6% and 6.8%, respectively, volatility of 35%, 36% and 37%, respectively, expected lives of 12 years and no dividend yield. The weighted average fair value per option granted during the year was \$13.13 in 1997, \$8.68 in 1996 and \$5.76 in 1995.

Pro forma compensation cost for 1997, 1996 and 1995 would have reduced (increased) net income (loss) by approximately \$330,000, (\$150,000) and (\$5,000), respectively. Basic and diluted earnings per share on a pro forma basis for Host Marriott were not impacted by the pro forma compensation cost in 1997, 1996 and 1995.

The effects of the implementation of SFAS No. 123 are not representative of the effects on reported net income in future years because only the effects of stock option awards granted in 1995, 1996 and 1997 have been considered.

In connection with the Special Dividend, the then outstanding options held by current and former employees of Host Marriott were redenominated in both Host Marriott and HM Services stock and the exercise prices of the options were adjusted based on the relative trading prices of shares of the common stock of the two companies. For all options held by certain current and former employees of Marriott International, the number and exercise price of the options were adjusted based on the trading prices of shares of the Host Marriott's common stock immediately before and after the Special Dividend. Therefore, the options outstanding reflect these revised exercise prices. Pursuant to the Distribution Agreement between the Company and HM Services, Host Marriott has the right to receive up to 1.4 million shares of HM Services' common stock or an equivalent cash value subsequent to exercise of the options held by certain former and current employees of Marriott International. As of January 2, 1998, Host Marriott valued this right at approximately \$20 million, which is included in other assets. A summary of the status of Host Marriott's stock option plan for 1997, 1996 and 1995 follows:

	1997		1	1996		1995	
	SHARES	WEIGHTED AVERAGE EXERCISE PRICE	WEIGHTED SHARES	WEIGHTED AVERAGE EXERCISE PRICE	SHARES	AVERAGE EXERCISE PRICE	
	(IN MILLIONS)		(IN MILLIONS)		(IN MILLIONS)		
Balance, at beginning of							
year	8.3	\$ 4	10.0	\$ 4	11.7	\$ 4	
Granted	.1	20	. 2	13			
Exercised	(1.6)	4	(1.9)	4	(2.3)	4	
Forfeited/Expired Adjustment for Special					(.3)	4	
Dividend					. 9	4	
Balance, at end of year	6.8	4	8.3	4	10.0	4	
Options exercisable at year-end	6.4		7.6		8.5		

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

The following table summarizes information about stock options outstanding at January 2, 1998:

	OPTIONS OUTSTANDING			OPTIONS EXERCISABLE		
RANGE OF EXERCISE PRICES	OUTSTANDING AT	CONTRACTUAL	WEIGHTED AVERAGE EXERCISE PRICE	SHARES EXERCISABLE AT JANUARY 2, 1998	WEIGHTED AVERAGE EXERCISE PRICE	
1-3	4.4	9	\$ 2	4.4	\$ 2	
4-6	1.7	4	6	1.7	6	
7-9	. 4	12	9	.3	9	
10-12	.1	14	12			
13-15	.1	14	15			
19-22	.1	15	20			
	6.8			6.4		
	===			===		

Deferred stock incentive plan shares granted to officers and key employees after 1990 generally vest over 10 years in annual installments commencing one year after the date of grant. Certain employees may elect to defer payments until termination or retirement. Deferred stock incentive plan shares granted in 1990 and prior years generally vest in annual installments commencing one year after the date of grant and continuing for 10 years. Employees also could elect to forfeit one-fourth of their deferred stock incentive plan award in exchange for accelerated vesting over a 10-year period. Host Marriott accrues compensation expense for the fair market value of the shares on the date of grant, less estimated forfeitures. In 1997, 1996 and 1995, 14,000, 13,000 and 158,000 shares were granted, respectively, under this plan. The compensation cost that has been charged against income for deferred stock was \$1 million in 1995 and was not material in 1996 and 1997. The weighted average fair value per share granted during each year was \$15.81 in 1997, \$11.81 in 1996 and \$8.49 in 1995.

In 1993, 3,537,000 restricted stock plan shares under the Comprehensive Plan were issued to officers and key executives to be distributed over the next three to 10 years in annual installments based on continued employment and the attainment of certain performance criteria. Host Marriott recognizes compensation expense over the restriction period equal to the fair market value of the shares on the date of issuance adjusted for forfeitures, and where appropriate, the level of attainment of performance criteria and fluctuations in the fair market value of Host Marriott's common stock. In 1997 and 1996, 198,000 and 2,511,000 shares of additional restricted stock plan shares were granted to certain key employees under terms and conditions similar to the 1993 grants. Approximately 161,000 and 500,000 shares were forfeited in 1996 and 1995, respectively. There were no shares forfeited in 1997. Host Marriott recorded compensation expense of \$13 million, \$11 million and \$5 million in 1997, 1996 and 1995, respectively, related to these awards. The weighted average fair value per share granted during each year was \$16.88 in 1997 and \$14.01 in 1996. There were no restricted stock plan shares granted in 1995.

Under the terms of the Employee Stock Purchase Plan, eligible employees may purchase common stock through payroll deductions at the lower of market value at the beginning or end of the plan year.

#### 11. PROFIT SHARING AND POSTEMPLOYMENT BENEFIT PLANS

Host Marriott contributes to profit sharing and other defined contribution plans for the benefit of employees meeting certain eligibility requirements and electing participation in the plans. The amount to be matched by Host Marriott is determined annually by the Board of Directors. Host Marriott provides medical benefits to a limited number of retired employees meeting restrictive eligibility requirements. Amounts for these items were not material in 1995 through 1997.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

#### 12. ACQUISITIONS AND DISPOSITIONS

In 1998, the Company acquired, or purchased controlling interests in six full-service hotels totaling 3,270 rooms for an aggregate purchase price of approximately \$388 million and entered into an agreement to acquire a controlling interest in the 397-room Ritz-Carlton in Tysons Corner, Virginia.

In April 1998, Host Marriott reached a definitive agreement with various affiliates of The Blackstone Group and Blackstone Real Estate Partners (collectively, "Blackstone") to acquire interests in 12 world-class luxury hotels in the U.S. and certain other assets, including the assumption of two mortgages, one of which is secured by a thirteenth hotel. The Company expects to pay approximately \$862 million in cash and assumed debt and to issue approximately 43.7 million Operating Partnership units. Each OP Unit will be exchangeable for one share of Host Marriott common stock (or its cash equivalent). Upon completion of the acquisition, Blackstone will own approximately 18% of the outstanding shares of Host Marriott common stock on a fully converted basis. The Blackstone portfolio consists of two Ritz-Carlton, three Four Seasons, one Grand Hyatt, three Hyatt Regencies and four Swissotel properties and a mortgage on a third Four Seasons.

In 1998, the Company sold two hotels totaling 854 rooms for approximately \$212 million.

In 1997, the Company acquired eight full-service hotels totaling 3,600 rooms for approximately \$145 million. In addition, the Company acquired controlling interests in nine full-service hotels totaling 5,024 rooms for approximately \$621 million, including the assumption of approximately \$418 million of debt. The Company also completed the acquisition of the 504-room New York Marriott Financial Center, after acquiring the mortgage on the hotel for \$101 million in late 1996.

In 1996, the Company acquired six full-service hotels totaling 1,964 rooms for an aggregate purchase price of approximately \$189 million. In addition, the Company acquired controlling interests in 17 full-service hotels totaling 8,917 rooms for an aggregate purchase price of approximately \$1.1 billion, including the assumption of approximately \$696 million of debt. The Company also purchased the first mortgage of the 504-room New York Marriott Financial Center for approximately \$101 million.

In 1995, the Company acquired nine full-service hotels totaling approximately 3,900 rooms in separate transactions for approximately \$390 million.

During the first and third quarters of 1995, 37 of the Company's Courtyard properties were sold and leased back from Hospitality Properties Trust for approximately \$330 million. The Company received net proceeds from the two transactions of approximately \$297 million and will receive approximately \$33 million upon expiration of the leases. A deferred gain of \$14 million on the sale/leaseback transactions is being amortized over the initial term of the leases.

In the first and second quarters of 1996, the Company completed the sale and leaseback of 16 of its Courtyard properties and 18 of its Residence Inn properties for \$349 million. The Company received net proceeds of approximately \$314 million and will receive approximately \$35 million upon expiration of the leases. A deferred gain of \$45 million on the sale/leaseback transactions is being amortized over the initial term of the leases.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

The Company's summarized, unaudited combined consolidated pro forma results of operations, assuming the above transactions, the refinancings and new debt activity discussed in Note 5 occurred, along with the purchase of the remaining interests in the eight public partnerships and five private partnerships, the contribution and the REIT Conversion, on December 30, 1995, are as follows (in millions):

	1997	1996
Revenues	\$3,512	\$3,267
Operating profit	529	454
<pre>Income (loss) before extraordinary items</pre>	37	(42)

#### 13. FAIR VALUE OF FINANCIAL INSTRUMENTS

The fair values of certain financial assets and liabilities and other financial instruments are shown below:

	19	97	199	6
	CARRYING AMOUNT		CARRYING AMOUNT	
		(IN MI	LLIONS)	
Financial assets				
Short-term marketable securities	\$ 354	\$ 354	\$	\$
Receivables from affiliates	23	26	156	174
Notes receivable	31	48	141	155
Other	20	20	13	13
Financial liabilities				
Debt, net of capital leases	3,458	3,492	2,636	2,654
Other financial instruments				
Obligation to Host Marriott for				
Convertible Preferred Securities	550	638	550	595
Interest rate swap agreements				1
Affiliate debt service commitments				

Short-term marketable securities and the obligation to Host Marriott for Convertible Preferred Securities are valued based on quoted market prices. Receivables from affiliates, notes and other financial assets are valued based on the expected future cash flows discounted at risk-adjusted rates. Valuations for secured debt are determined based on the expected future payments discounted at risk-adjusted rates. The fair values of the Line of Credit and other notes are estimated to be equal to their carrying value. Senior Notes are valued based on quoted market prices.

The Company is contingently liable under various guarantees of obligations of certain affiliates (affiliate debt service commitments) with a maximum commitment of \$60 million at January 2, 1998 and \$117 million at January 3, 1997. A fair value is assigned to commitments with expected future fundings. The fair value of the commitments represents the net expected future payments discounted at risk-adjusted rates. Such payments are accrued on an undiscounted basis.

The fair value of interest rate swap agreements is based on the estimated amount the Company would pay or receive to terminate the swap agreements. The aggregate notional amount of the agreements was \$100 million at January 2, 1998 and \$525 million at January 3, 1997.

#### 14. MARRIOTT INTERNATIONAL DISTRIBUTION AND RELATIONSHIP WITH MARRIOTT INTERNATIONAL

On October 8, 1993 (the "Marriott International Distribution Date"), Marriott Corporation distributed, through a special tax-free dividend (the "Marriott International Distribution"), to holders of Marriott

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Corporation's common stock (on a share-for-share basis), approximately 116.4 million outstanding shares of common stock of an existing wholly-owned subsidiary, Marriott International, resulting in the division of Marriott Corporation's operations into two separate companies. The distributed operations included the former Marriott Corporation's lodging management, franchising and resort timesharing operations, senior living service operations, and the institutional food service and facilities management business. Host Marriott retained the former Marriott Corporation's airport and tollroad food, beverage and merchandise concessions operations, as well as most of its real estate properties. Effective at the Marriott International Distribution Date, Marriott Corporation changed its name to Host Marriott Corporation.

Host Marriott and Marriott International have entered into various agreements in connection with the Marriott International Distribution and thereafter which provide, among other things, that (i) the majority of the Company's hotel lodging properties are managed by Marriott International under agreements with initial terms of 15 to 20 years and which are subject to renewal at the option of Marriott International for up to an additional 16 to 30 years (see Note 15); (ii) 10 of the Company's full-service properties are operated under franchise agreements with Marriott International with terms of 15 to 30 years; (iii) Marriott International guarantees the Company's performance in connection with certain loans and other obligations (\$107 million at January 2, 1998); (iv) the Company borrowed and repaid \$109 million of first mortgage financing for construction of the Philadelphia Marriott (see Note 5); (v) Marriott International provided the Company with \$70 million of mortgage financing in 1995 for the acquisition of three full-service properties by the Company at an average interest rate of 8.5% (Marriott International subsequently sold one of the loans in November 1996); (vi)
Marriott International and the Company formed a joint venture and Marriott International provided the Company with \$29 million in debt financing at an average interest rate of 12.7% and \$28 million in preferred equity in 1996 for the acquisition of two full-service properties in Mexico City, Mexico; (vii) in 1995, the Company also acquired a full-service property from a partnership in which Marriott International owned a 50% interest; and (viii) Marriott International provides certain limited administrative services.

In 1997, 1996 and 1995, Host Marriott paid to Marriott International \$162 million, \$101 million and \$67 million, respectively, in hotel management fees; \$19 million, \$18 million and \$21 million, respectively, in interest and commitment fees under the debt financing and line of credit provided by Marriott International and \$3 million, \$4 million and \$12 million, respectively, for limited administrative services. Host Marriott also paid Marriott International \$4 million, \$2 million and \$1 million, respectively, in franchise fees in 1997, 1996 and 1995.

Additionally, Marriott International has the right to purchase up to 20% of the voting stock of Host Marriott if certain events involving a change in control of Host Marriott occur.

#### 15. HOTEL MANAGEMENT AGREEMENTS

Most of the Company's hotels are subject to management agreements (the "Agreements") under which Marriott International manages most of the Company's hotels, generally for an initial term of 15 to 20 years with renewal terms at the option of Marriott International of up to an additional 16 to 30 years. The Agreements generally provide for payment of base management fees equal to one to four percent of sales and incentive management fees generally equal to 20% to 50% of Operating Profit (as defined in the Agreements) over a priority return (as defined) to the Company, with total incentive management fees not to exceed 20% of cumulative Operating Profit, or 20% of current year Operating Profit. In the event of early termination of the Agreements, Marriott International will receive additional fees based on the unexpired term and expected future base and incentive management fees. The Company has the option to terminate certain management agreements if specified performance thresholds are not satisfied. No agreement with respect to a single lodging facility is cross-collateralized or cross-defaulted to any other agreement and a single agreement may be canceled under certain conditions, although such cancellation will not trigger the cancellation of any other agreement.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

Pursuant to the terms of the Agreements, Marriott International is required to furnish the hotels with certain services ("Chain Services") which are generally provided on a central or regional basis to all hotels in the Marriott International hotel system. Chain Services include central training, advertising and promotion, a national reservation system, computerized payroll and accounting services, and such additional services as needed which may be more efficiently performed on a centralized basis. Costs and expenses incurred in providing such services are allocated among all domestic hotels managed, owned or leased by Marriott International or its subsidiaries. In addition, the Company's hotels also participate in the Marriott Rewards program. The cost of this program is charged to all hotels in the Marriott hotel system.

The Company is obligated to provide the manager with sufficient funds to cover the cost of (a) certain non-routine repairs and maintenance to the hotels which are normally capitalized; and (b) replacements and renewals to the hotels' property and improvements. Under certain circumstances, the Company will be required to establish escrow accounts for such purposes under terms outlined in the Agreements.

The Company has entered into franchise agreements with Marriott International for ten hotels. Pursuant to these franchise agreements, the Company generally pays a franchise fee based on a percentage of room sales and food and beverage sales as well as certain other fees for advertising and reservations. Franchise fees for room sales vary from four to six percent of sales, while fees for food and beverage sales vary from two to three percent of sales. The terms of the franchise agreements are from 15 to 30 years.

The Company has entered into management agreements with The Ritz-Carlton Hotel Company, LLC ("Ritz-Carlton"), an affiliate of Marriott International, to manage four of the Company's hotels. These agreements have an initial term of 15 to 25 years with renewal terms at the option of Ritz-Carlton of up to an additional 10 to 40 years. Base management fees vary from two to four percent of sales and incentive management fees are generally equal to 20% of available cash flow or operating profit, as defined in the agreements.

The Company has also entered into management agreements with hotel management companies other than Marriott International and Ritz-Carlton for 12 of its hotels (10 of which are franchised under the Marriott brand). These agreements generally provide for an initial term of 10 to 20 years with renewal terms at the option of either party of up to an additional one to 15 years. These agreements generally provide for payment of base management fees equal to one to three percent of sales. Seven of the 12 agreements also provide for incentive management fees generally equal to 15 to 20 percent of available cash flow, as defined in the agreements.

At January 2, 1998 and January 3, 1997, \$75 million and \$76 million, respectively, have been advanced to the hotel managers for working capital and are included in "Due From Managers" in the accompanying combined consolidated balance sheets.

#### 16. LITIGATION

The Company is from time-to-time the subject of, or involved in, judicial proceedings. Management believes that any liability or loss resulting from such matters will not have a material adverse effect on the financial position or results of operations of the Company.

In the fourth quarter of 1997, the Company reached a settlement in a lawsuit against Trinity Industries and others for claims related to construction of the New York Marriott Marquis. In settlement of the lawsuit, the Company and its affiliate received a cash settlement of approximately \$70 million, the majority of which was considered a recovery of construction costs and \$10 million of which has been recorded as other revenues in the accompanying combined consolidated financial statements.

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

#### 17. GEOGRAPHIC AND BUSINESS SEGMENT INFORMATION

The Company operates in the full-service hotel segment of the lodging industry. The Company's hotels are primarily operated under the Marriott or Ritz-Carlton brands, contain an average of nearly 500 rooms, as well as supply other amenities such as meeting space and banquet facilities; a variety of restaurants and lounges; gift shops; and swimming pools. They are typically located in downtown, airport, suburban and resort areas throughout the United States.

As of January 2, 1998, the Company's foreign operations consist of four full-service hotel properties located in Canada and two full-service hotel properties located in Mexico. There were no intercompany sales between the properties and the Company. The following table presents revenues and long-lived assets for each of the geographical areas in which the Company operates (in millions):

	1997		1996		1995	
	REVENUES	LONG-LIVED ASSETS	REVENUES	LONG-LIVED ASSETS	REVENUES	LONG-LIVED ASSETS
United States International	\$2,732	\$4,412	\$1,908	\$3,587	\$1,359	\$2,842
	91	222	49	218	3	40
Total	\$2,823	\$4,634	\$1,957	\$3,805	\$1,362	\$2,882
	=====	=====	=====	=====	=====	=====

#### 18. SUBSEQUENT EVENTS (UNAUDITED)

On August 5, 1998, Properties issued \$1.7 billion of 7 7/8% senior notes issued in two series, consisting of \$500 million due 2005 and \$1.2 billion due 2008 (the "New Senior Notes"). The New Senior Notes are guaranteed by Host Marriott and Hospitality and certain subsidiaries of Properties and are secured by pledges of equity interests in certain subsidiaries of Properties. The Operating Partnership will assume the New Senior Notes in connection with the REIT Conversion and the guarantee by Host Marriott will terminate on the Effective Date.

The indenture under which the New Senior Notes were issued contains covenants restricting the ability of Properties and certain of its subsidiaries to incur indebtedness, acquire or sell assets or make investments in other entities, and make distributions to equityholders of Properties and (following the REIT Conversion) the Operating Partnership. The New Senior Notes also contain a financial covenant requiring the maintenance of a specified ratio of unencumbered assets to unsecured debt.

On August 5, 1998, Properties entered into a \$1.25 billion credit facility (the "New Credit Facility") provided by a syndicate of financial institutions (the "Lenders") led by Bankers Trust Company. The New Credit Facility provides the Company with (i) a \$350 million term loan facility (subject to increases as provided in the succeeding paragraph) and (ii) a \$900 million revolving credit facility. The New Credit Facility will have an initial term of three years with two one-year options to extend. The proceeds of the New Credit Facility, along with the proceeds from the New Senior Notes, were used to fund the purchase of \$1.55 billion of senior notes of Properties at the initial closing on August 5, 1998, and repay \$22 million of outstanding borrowings under a line of credit provided by the Lenders to certain subsidiaries of the Company.

The term loan facility was funded on the closing date of the New Credit Facility. The \$350 million term loan facility may be increased by up to \$250 million after the initial closing and will be available, subject to terms and conditions thereof and to the commitment of sufficient Lenders, in up to two drawings to be made on or prior to the second anniversary of the closing of the New Credit Facility. The Lenders will advance funds under the revolving credit facility as requested by the Company with minimum borrowing amounts and frequency limitations to be agreed upon, subject to customary conditions including, but not limited to, (i) no

#### NOTES TO COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

existing or resulting default or event of default under the New Credit Facility and (ii) continued accuracy of representations and warranties in all material respects.

The interest rate applicable to the New Credit Facility and the unused commitment fee applicable to the revolving portion of the New Credit Facility are calculated based on a spread over LIBOR that will fluctuate based on the quarterly recalculation of a leverage ratio set forth in the New Credit Facility. The New Credit Facility provides that in the event that the Company achieves one of several investment grade long-term unsecured indebtedness ratings, the spread over LIBOR applicable to the New Credit Facility will be fixed based on the particular rating achieved. If the Company elects to exercise its one-year extensions, the Company will be required to amortize approximately 22.5% per annum of the principal amount outstanding under the New Credit Facility at the end of the initial three-year term.

The Company's obligations under the New Credit Facility are guaranteed, subject to certain conditions, on a senior basis by Host Marriottt Hospitality and certain of Properties' existing and future subsidiaries. The New Credit Facility will be assumed by the Operating Partnership in connection with the REIT Conversion and the guarantee of Host Marriott will then terminate. In addition, certain subsidiaries of Host Marriott other than Properties and its subsidiaries may, under certain circumstances, guarantee the obligations under the New Credit Facility in the future. Borrowings under the New Credit Facility will rank pari passu with the New Senior Notes and other existing and future senior indebtedness of the Company. The New Credit Facility is secured, on an equal and ratable basis, with the New Senior Notes by a pledge of the capital stock of certain direct and indirect subsidiaries of Properties. In addition, the New Credit Facility may, under certain circumstances in the future, be secured by a pledge of capital stock of certain subsidiaries of Host Marriott other than Properties and its subsidiaries.

The New Credit Facility includes financial and other covenants that require the maintenance of certain ratios with respect to, among other things, maximum leverage, limitations on indebtedness, minimum net worth and interest and fixed charge coverage and restrict payment of distributions and investments, acquisitions and sales of assets by the Company.

#### CONDENSED COMBINED CONSOLIDATED BALANCE SHEET SEPTEMBER 11, 1998 (UNAUDITED, IN MILLIONS)

	HISTORICAL	PRO FORMA (SEE NOTE 12)
ASSETS		
Property and Equipment, net	\$5,288	\$5,288
due from affiliates of \$4 million)	29	29
Due from Managers	83	83
Investments in Affiliates	18	18
Other Assets	321	321
Short-term Marketable Securities	36	36
Cash and Cash Equivalents	512	287
	\$6,287	\$6,062
	=====	=====
LIABILITIES AND EQUITY		
Debt		
Senior Notes	\$1,748	\$1,748
Mortgage Debt	1,818	1,818
Other Convertible Debt Obligation to Host Marriott	445	445
Corporation	567	567
	4,578	4,578
Accounts Payable and Accrued Expenses	47	47
Deferred Income Taxes	464	464
Other Liabilities	437	437
Total Liabilities Equity	5,526	5,526
Investments and Advances from Host Marriott Corpo-		
ration	761	536
	\$6,287	\$6,062
	=====	=====

See Notes to Condensed Combined Consolidated Financial Statements.

# CONDENSED COMBINED CONSOLIDATED STATEMENTS OF OPERATIONS THIRTY-SIX WEEKS ENDED SEPTEMBER 11, 1998 AND SEPTEMBER 12, 1997 (UNAUDITED, IN MILLIONS)

	1998	1997
REVENUES Rooms Food and beverage Other	642	\$1,246 500 120
Total hotel revenues	2,315 53 1	1,866 3 3 10
Total revenues		1,882
OPERATING COSTS AND EXPENSES  Hotel property-level costs and expenses  Rooms	348 477 80	285 381 63
Other hotel operating costs and expenses (including Marriott International management fees of \$138 million and \$111 million, respectively)		
Total hotel operating costs and expenses Other		1,563 21
Total operating costs and expenses		1,584
OPERATING PROFIT BEFORE MINORITY INTEREST, CORPORATE EXPENSES, REIT CONVERSION EXPENSES AND INTEREST	465 (36) (30) (14)	297 (24) (26)
Host Marriott Corporation of \$26 million in 1998 and 1997)  Interest income	(253) 38	
INCOME BEFORE INCOME TAXES AND EXTRAORDINARY ITEM	170	66 (28)
INCOME BEFORE EXTRAORDINARY ITEM  Extraordinary itemGain (loss) on extinguishment of debt (net of		
taxes of \$80 million in 1998 and \$3 million in 1997)	(148)	
NET INCOME		\$ 43

See Notes to Condensed Combined Consolidated Financial Statements.

# CONDENSED COMBINED CONSOLIDATED STATEMENTS OF CASH FLOWS THIRTY-SIX WEEKS ENDED SEPTEMBER 11, 1998 AND SEPTEMBER 12, 1997 (UNAUDITED, IN MILLIONS)

	1998	1997
OPERATING ACTIVITIES Income from continuing operations	\$ 98	\$ 38
Depreciation and amortization	167 50	153 
Gains on sales of hotel properties	(50) (1) (35)	(4) 74
Other	31	43
Cash from operations	260	304
INVESTING ACTIVITIES Proceeds from sales of assets	211 (607)	35 (236)
Capital expenditures: Renewals and replacements New development projects	(108) (32)	(86)
New investment capital expenditures  Purchases of short-term marketable securities  Sales of short-term marketable securities	(19) (134) 451	(22)  
Notes receivable collections	4 (78) (12)	5 (1) 13
Cash used in investing activities		(292)
FINANCING ACTIVITIES Cash transferred to Host Marriott	(63) 2,004 2	(210) 652 4
Scheduled principal repayments  Debt prepayments  Costs of extinguishment of debt  Other	(39) (1,631) (175) (16)	24
Cash from financing activities	82	179
INCREASE IN CASH AND CASH EQUIVALENTS		\$ 191
Non-cash financing activities: Assumption of mortgage debt for the acquisition of, or pur- chase of	=	
controlling interests in, certain hotel properties	\$ 164 ======	\$ 287 =====

See Notes to Condensed Combined Consolidated Financial Statements.

#### NOTES TO CONDENSED COMBINED CONSOLIDATED FINANCIAL STATEMENTS

1. On April 16, 1998, the Board of Directors of Host Marriott Corporation ("Host Marriott") approved a plan to restructure Host Marriott's current business operations by the spin-off of Host Marriott's senior living business (Crestline Capital Corporation, "Crestline") and the contribution of Host Marriott's hotels and certain other assets and liabilities to a newly formed Delaware limited partnership, Host Marriott, L.P. (the "Operating Partnership") whose sole general partner will be HMC Merger Corporation (to be renamed "Host Marriott Corporation"), a newly formed Maryland corporation that will merge into Host Marriott Corporation, a Delaware corporation and will qualify as a real estate investment trust ("REIT"). Host Marriott's contribution of its hotels and certain assets and liabilities to the Operating Partnership (the "Contribution") in exchange for units of limited partnership interests in the Operating Partnership will be accounted for at Host Marriott's historical basis.

The accompanying condensed combined consolidated financial statements include the accounts of the Host Marriott hotels and the assets and liabilities expected to be included in the Contribution by Host Marriott to the Operating Partnership upon its planned conversion to a REIT (the "REIT Conversion"). In these condensed combined consolidated financial statements, Host Marriott Hotels is referred to as "Host Marriott Hotels" or the "Company." The condensed combined consolidated financial statements exclude the assets, liabilities, equity, operations and cash flows related to Host Marriott's portfolio of 31 senior living communities. After the REIT Conversion, Crestline will own these assets and lease the existing hotels from the Company.

In June 1998, as part of the REIT Conversion, Host Marriott and the Operating Partnership filed a preliminary Prospectus/Consent Solicitation with the Securities and Exchange Commission. This Prospectus/Consent Solicitation Statement describes a proposal whereby the Operating Partnership will acquire by merger (the "Mergers") eight limited partnerships (the "Partnerships") that own full-service hotels in which Host Marriott or its subsidiaries are general partners. As more fully described in the Prospectus/Consent Solicitation Statement, limited partners of those Partnerships that participate in the Mergers will receive OP Units in exchange for their partnership interests in such Partnerships and, at their election, they may exchange such OP Units for unsecured notes due December 15, 2005 issued by the Operating Partnership ("Notes") or common stock in the REIT. The Solicitation Period expires on December 12, 1998, unless extended.

The REIT expects to qualify as a real estate investment trust under federal income tax law beginning January 1, 1999. However, consummation of the REIT Conversion is subject to significant contingencies that are outside the control of Host Marriott, including final Board of Directors approval, consents of shareholders, partners, bondholders, lenders and ground lessors of Host Marriott, its affiliates and other third parties. Accordingly, there can be no assurance that the REIT Conversion will be completed or that it will be effective as of January 1, 1999. If the REIT Conversion is not completed on January 1, 1999, the effectiveness of REIT election could be delayed until January 1, 2000, which would result in the Company continuing to pay a substantial amount of corporate-level income taxes in

On April 20, 1998, Host Marriott and certain of its subsidiaries filed a shelf registration on Form S-3 (the "Shelf Registration") with the Securities and Exchange Commission for the issuance of up to \$2.5 billion in securities, which may include debt, equity or a combination thereof. Host Marriott utilized \$1.7 billion of the capacity under this Shelf Registration to issue the New Senior Notes (defined below) and anticipates that any net proceeds from the sale of additional offered securities will be used for refinancing of Host Marriott's indebtedness, acquisitions and general corporate purposes.

On August 5, 1998, HMH Properties, Inc. ("HMH Properties"), an indirect wholly-owned subsidiary of Host Marriott, which owns 71 of Host Marriott's hotels, purchased substantially all of its (i) \$600 million in 9 1/2% senior notes due 2005, (ii) \$350 million in 9% senior notes due 2007 and (iii) \$600 million in 8 7/8% senior notes due 2007 (collectively, the "Old Senior Notes"). Concurrently with each offer to

NOTES TO CONDENSED COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

purchase, HMH Properties successfully solicited consents (the "1998 Consent Solicitations") from registered holders of the Old Senior Notes to certain amendments to eliminate or modify substantially all of the restrictive covenants and certain other provisions contained in the indentures pursuant to which the Old Senior Notes were issued. HMH Properties simultaneously utilized the Shelf Registration to issue an aggregate of \$1.7 billion in new senior notes (the "New Senior Notes"). The New Senior Notes were issued in two series, \$500 million of 7 7/8% Series A notes due in 2005 and \$1.2 billion of 7 7/8% Series B notes due in 2008. Approximately \$21 million of the Old Senior Notes remain outstanding. The 1998 Consent Solicitations facilitated the merger of HMC Capital Resources Holdings Corporation ("Capital Resources"), a wholly-owned subsidiary of the Company, with and into HMH Properties. Capital Resources, the then owner of eight of Host Marriott's hotel properties, was the obligor under the \$500 million revolving credit facility (the "Old Credit Facility").

In conjunction with the issuance of the New Senior Notes, HMH Properties entered into a \$1.25 billion credit facility (the "New Credit Facility") with a group of commercial banks. The New Credit Facility has an initial three year term with two one-year extension options. Borrowings under the New Credit Facility generally bear interest at the Eurodollar rate plus 1.75% (7.5% at September 11, 1998). The interest rate and commitment fee (0.35% at September 11, 1998) on the unused portion of the Credit Facility fluctuates based on certain financial ratios. The New Senior Notes and the New Credit Facility are guaranteed by Host Marriott and its wholly-owned subsidiary, Host Marriott Hospitality, Inc. and certain subsidiaries of HMH Properties and are secured by pledges of equity interests in certain subsidiaries of HMH Properties. The New Senior Notes and the New Credit Facility will be assumed by the Operating Partnership in connection with the REIT Conversion and the guarantee of the Company is expected to terminate upon the consummation of the REIT Conversion. As of September 11, 1998, \$350 million was outstanding under the New Credit Facility.

The New Credit Facility replaces the Company's Old Credit Facility. The net proceeds from the offering and borrowings under the New Credit Facility were used by Host Marriott to purchase substantially all of the Old Senior Notes, to make repayments outstanding under the Old Credit Facility and to make bond premium and consent payments totaling approximately \$175 million. These costs, along with the write-off of deferred financing fees of approximately \$52 million related to the Old Senior Notes and the Old Credit Facility, will be recorded as a pre-tax extraordinary loss on the extinguishment of debt in the third quarter of 1998. The New Credit Facility and the indenture under which the New Senior Notes were issued contain covenants restricting the ability of HMH Properties and certain of its subsidiaries to incur indebtedness, grant liens on their assets, acquire or sell assets or make investments in other entities, and make distributions to equityholders of HMH Properties, Host Marriott, and (following the REIT Conversion) the Operating Partnership and Host REIT. The New Credit Facility and the New Senior Notes also contain certain financial covenants relating to, among other things, maintaining certain levels of tangible net worth and certain ratios of EBITDA to interest and fixed charges, total debt to EBITDA, unencumbered assets to unsecured debt, and secured debt to total debt.

The accompanying condensed combined consolidated financial statements have been prepared by the Company without audit. Certain information and footnote disclosures normally included in financial statements presented in accordance with generally accepted accounting principles have been condensed or omitted. The Company believes the disclosures made are adequate to make the information presented not misleading. However, the condensed combined consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's audited financial statements for the three fiscal years in the period ended January 2, 1998.

In the opinion of the Company, the accompanying unaudited condensed combined consolidated financial statements reflect all adjustments necessary to present fairly the financial position of the Company as of September 11, 1998 and the results of operations and cash flows for the thirty-six weeks ended

NOTES TO CONDENSED COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

September 11, 1998 and September 12, 1997. Interim results are not necessarily indicative of fiscal year performance because of the impact of seasonal and short-term variations.

2. In April 1998, Host Marriott reached a definitive agreement with various affiliates of The Blackstone Group and Blackstone Real Estate Partners (collectively, "Blackstone") to acquire controlling interests in 12 luxury hotels and a first mortgage interest in another hotel in the U.S. and certain other assets. The Company expects to pay approximately \$862 million in cash and assumed debt and to issue approximately 43.7

million Operating Partnership units ("OP Units") of the Operating Partnership (based upon a negotiated rate of \$20.00 per OP Unit) and distribute up to 18% of the shares of Crestline to be formed as part of the Company's reorganization. Each OP Unit will be exchangeable for one share of Host Marriott common stock (or its cash equivalent). Upon completion of the acquisition, Blackstone will own approximately 16% of the outstanding shares of Host Marriott common stock on a fully converted basis. The Blackstone portfolio consists of two Ritz-Carlton, two Four Seasons, one Grand Hyatt, three Hyatt Regency, four Swissotel properties, and a mortgage note on a third Four Seasons property.

The Blackstone transaction is expected to close immediately after the REIT Conversion. At that time, Blackstone's hotels and other assets will be contributed to the Operating Partnership and the hotels leased to subsidiaries of Crestline. The hotels will continue to be managed on behalf of the lessees under the existing management contracts. Consummation of the Blackstone transaction is also subject to certain conditions, including consummation of the REIT Conversion no later than March 31, 1999, and Host Marriott qualifying as a REIT for 1999.

- Revenues primarily represent sales generated by the Company's hotels. Other hotel operating costs and expenses include depreciation, management fees, real and personal property taxes, ground and equipment rent, insurance and certain other costs.
- 4. Basic and diluted earnings per OP Unit have been calculated based on the number of Host Marriott common shares outstanding for all periods presented because it is expected that upon the REIT Conversion the Operating Partnership will issue OP Units to Host Marriott in exchange for the Contribution equal to the number of shares of outstanding Host Marriott common stock. Accordingly, the following discussion of earnings per OP Unit is on a pro forma basis as if the REIT Conversion and Contribution had occurred.

Basic earnings per OP Unit is computed by dividing net income by the weighted average number of shares of common stock outstanding of Host Marriott. Diluted earnings per OP Unit is computed by dividing net income plus dividends by the weighted average number of shares of common stock outstanding plus other potentially dilutive securities of Host Marriott. Diluted earnings per OP Unit was not adjusted for the impact of the Convertible Debt Obligation to Host Marriott Corporation in 1997 as they were anti-dilutive.

Basic and diluted earnings per OP Unit on a pro forma basis are as follows:

	THIRTY-SIX WEEKS ENDED		
	SEPTEMBER 11, 1998		
Basic earnings per OP Unit: Income before extraordinary item Extraordinary itemGain (loss) on extinguishment of debt (net of income	\$ .48	\$.19	
taxes)	(.73)	.02	
Basis earnings per OP Unit	\$(.25) =====	\$.21 ====	
Diluted earnings per OP Unit:			
Income before extraordinary item Extraordinary itemGain (loss) on extinguishment of debt (net of income	\$ .47	\$.18	
taxes)	(.71)	.02	
Diluted earnings per OP Unit	\$(.24) =====	\$.20 ====	

NOTES TO CONDENSED COMBINED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED) A reconciliation of the number of shares utilized for the calculation of diluted earnings per OP Unit follows:

	THIRTY-SIX	WEEKS ENDED
	1998	SEPTEMBER 12, 1997
		LLIONS)
Weighted average number of common shares outstanding	204.1	202.8
assumed purchased at average market price Assuming distribution of common shares upon redemption of Convertible Debt to Host Marriott	4.2	4.8
Corporation Assuming distribution of common shares issuable for warrants, less shares assumed purchased at		
average market price	.1	.2
Shares utilized for the calculation of diluted		
earnings per OP Unit	208.4 =====	207.8 =====

5. As of September 11, 1998, the Company had minority interests in 18 affiliates that own an aggregate of 240 properties, 20 of which are full-service properties, managed primarily by Marriott International, Inc. The Company's equity in earnings of affiliates was \$1 million and \$3 million for the thirty-six weeks ended September 11, 1998 and September 12, 1997, respectively.

Combined summarized operating results reported by affiliates follows:

	THIRTY-SIX	WEEKS ENDED
	SEPTEMBER 11, 1998	SEPTEMBER 12, 1997
	(IN MI	LLIONS)
Revenues Operating expenses:	\$369	\$447
Cash charges (including interest)	(223)	(278)
Depreciation and other non-cash charges	(101)	(140)
<pre>Income (loss) before extraordinary item</pre>	45	29
Extraordinary itemforgiveness of debt	4	12
, c		
Net income	\$ 49	\$ 41
	====	====

In the first quarter of 1998, the Company obtained a controlling interest in the partnership that owns the 1,671-room Atlanta Marriott Marquis for approximately \$239 million, including \$164 million in assumed mortgage debt. The Company previously owned a 1.3% general and limited partnership interest.

In second quarter of 1998, the Company acquired the partnership that owns the 289-room Park Ridge Marriott in Park Ridge, New Jersey for \$24 million. The Company previously owned a 1% managing general partner interest and held a note receivable interest.

6. In the first quarter of 1998, the Company acquired a controlling interest in, and became the managing general partner for, the partnership that owns the 359-room Albany Marriott, the 350-room San Diego Marriott Mission Valley and the 320-room Minneapolis Marriott Southwest for approximately \$50 million.

In the second quarter of 1998, the Company acquired the 397-room Ritz-Carlton, Tysons Corner for \$96 million and the 281-room Ritz-Carlton, Phoenix for \$75 million. In addition, the Company acquired the 487- room Torrance Marriott near Los Angeles, California for \$52 million. Also in the second quarter of 1998, the Company sold the 662-room New York Marriott East Side for approximately \$191 million and recorded a pre-tax gain of approximately \$40 million. The Company also sold the 191-room Napa Valley Marriott for approximately \$21 million and recorded a pre-tax gain of approximately \$10 million.

NOTES TO CONDENSED COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

In the third quarter of 1998, the Company acquired the 308-room Ritz-Carlton, Dearborn for approximately \$65 million, the 336-room Ritz-Carlton, San Francisco for approximately \$161 million and the 404-room Memphis Marriott (which was converted to the Marriott brand upon acquisition) for approximately \$16 million.

- 7. In March 1997, Host Marriott purchased 100% of the outstanding bonds secured by a first mortgage on the San Francisco Marriott Hotel. Host Marriott purchased the bonds for \$219 million, an \$11 million discount to the face value of \$230 million. In connection with the redemption and defeasance of the bonds, the Company recognized an extraordinary gain of \$5 million, which represents the \$11 million discount and the write-off of deferred financing fees, net of taxes.
- 8. The Company operates in the full-service hotel segment of the lodging industry. The Company's hotels are primarily operated under the Marriott or Ritz-Carlton brands.

As of September 11, 1998 and September 12, 1997, the Company's foreign operations consist of four full-service hotel properties located in Canada and two full-service hotel properties located in Mexico. There were no intercompany sales between the properties and the Company. The following table presents revenues for each of the geographical areas in which the Company operates (in millions):

	THIRTY-SIX WEEKS ENDED		
	SEPTEMBER 11, 1998	3 SEPTEMBER 12, 1997	
United States		\$743 25	
Total	\$1,040 =====	\$768 ====	

9. In the first quarter of 1998, the Company adopted SFAS No. 130, "Reporting Comprehensive Income," ("SFAS 130"). SFAS 130 establishes standards for reporting and display of comprehensive income and its components in financial statements. The objective of SFAS 130 is to report a measure of all changes in equity of an enterprise that result from transactions and other economic events of the period other than transactions with owners. Comprehensive income is the total of net income and all other nonowner changes in equity.

The Company's only component of other comprehensive income is the right to receive up to 1.4 million shares of Host Marriott Services Corporation's ("HMSC") common stock or an equivalent cash value subsequent to exercise of the options held by certain former and current employees of Marriott International. For the thirty-six weeks ended September 11, 1998, other comprehensive loss was \$3 million and consisted of the unrealized loss on the appreciation of the HMSC common stock. For the thirty-six weeks ended September 11, 1998, comprehensive loss was \$51 million. For the thirty-six weeks ended September 12, 1997, other comprehensive income was \$8 million. For thirty-six weeks ended September 12, 1997, comprehensive income was \$51 million. As of September 11, 1998 and January 2, 1998, the Company's accumulated other comprehensive income of approximately \$7 million and \$10 million, respectively, was included in Investments and Advances from Host Marriott Corporation.

10. The obligation for the Convertible Subordinated Debentures (the "Debentures") has been pushed down to these financial statements because it is expected that upon the REIT Conversion the Operating Partnership will assume primary liability for repayment of the Debentures of Host Marriott underlying the Convertible Preferred Securities of Host Marriott Trust (the "Issuer"), a wholly-owned subsidiary trust of Host Marriott. The common securities of Host Marriott Financial Trust will not be contributed to the Operating Partnership and therefore Host Marriott Financial Trust will not be consolidated by the Operating Partnership. Upon conversion by a Convertible Preferred Securities holder, the REIT will issue shares of its

NOTES TO CONDENSED COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

common stock which will be delivered to such holder. Upon the issuance of such shares by the REIT, the Operating Partnership will issue to the REIT the number of OP Units equal to the number of shares of the REIT common stock in exchange for the Debentures.

In December 1996, the Issuer issued 11 million shares of 6 3/4% convertible quarterly income preferred securities (the "Convertible Preferred Securities"), with a liquidation preference of \$50 per share (for a total liquidation amount of \$550 million). The Convertible Preferred Securities represent an undivided beneficial interest in the assets of the Issuer. The payment of distributions out of moneys held by the Issuer and payments on liquidation of the Issuer or the redemption of the Convertible Preferred Securities are guaranteed by Host Marriott to the extent the Issuer has funds available therefor. This guarantee, when taken together with Host Marriott obligations under the indenture pursuant to which the Debentures were issued, the Debentures, Host Marriott's obligations under the Trust Agreement and its obligations under the indenture to pay costs, expenses, debts and liabilities of the Issuer (other than with respect to the Convertible Preferred Securities) provides a full and unconditional guarantee of amounts due on the Convertible Preferred Securities. Proceeds from the issuance of the Convertible Preferred Securities were invested in the Debentures due December 2, 2026 issued by Host Marriott. The Issuer exists solely to issue the Convertible Preferred Securities and its own common securities (the "Common Securities") and invest the proceeds therefrom in the Debentures. The note receivable from Host Marriott for the Debentures is the Issuer's sole asset. Separate financial statements of the Issuer are not presented because of Host Marriott's guarantee described above; Host Marriott's management has concluded that such financial statements are not material to investors and the Issuer is wholly-owned and essentially has no independent operations.

Each of the Convertible Preferred Securities is convertible at the option of the holder into shares of Host Marriott common stock at the rate of 2.6876 shares per Convertible Preferred Security (equivalent to a conversion price of \$18.604 per share of Company common stock). The Debentures are convertible at the option of the holders into shares of Host Marriott common stock at the conversion rate of 2.6876 shares for each \$50 in principal amount of Debentures. The conversion rate is subject to adjustments in certain events, including (i) payment of dividends (and other distributions) on Host common stock by Host in shares of Host common stock; (ii) distributions to all holders of Host common stock of rights or warrants entitling such holders (for a period not to exceed 45 days) to subscribe for or purchase Host common stock at an exercise price less than the market price of Host common stock; (iii) subdivisions and combinations of Host common stock; (iv) payment of dividends (and other distributions) on Host common stock consisting of indebtedness of Host, capital stock or other securities, assets or cash (other than certain cash dividends at an annualized rate of up to 12.5% of the market price of Host common stock); (v) payments for Host common stock by Host or any of its subsidiaries in respect of a tender or exchange offer (other than an odd-lot offer) at a price per share in excess of 110% of the market price of Host common stock; (vi) consummation by Host of certain mergers, a consolidation, a sale of all or substantially all of its assets, a recapitalization or certain reclassifications of Host common stock. The distribution of the capital stock of Crestline to all holders of Host REIT common stock would, and certain other elements of the REIT Conversion (such as other distributions of Host's accumulated earnings and profits) will, result in an adjustment to the conversion price of the Debentures. The REIT Conversion could result in a material adjustment to the conversion price of the Debentures and the Convertible Preferred Securities dependent upon the ultimate amount of the E&P Distribution and the value of Host REIT common stock. The Issuer will only convert Debentures pursuant to a notice of conversion by a holder of Convertible Preferred Securities. During 1998 and 1997, no shares were converted into common stock.

Holders of the Convertible Preferred Securities are entitled to receive preferential cumulative cash distributions at an annual rate of 6 3/4% accruing from the original issue date, commencing March 1, 1997,

NOTES TO CONDENSED COMBINED CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

and payable quarterly in arrears thereafter. The distribution rate and the distribution and other payment dates for the Convertible Preferred Securities will correspond to the interest rate and interest and other payment dates on the Debentures. Host Marriott may defer interest payments on the Debentures for a period not to exceed 20 consecutive quarters. If interest payments on the Debentures are deferred, so too are payments on the Convertible Preferred Securities. Under this circumstance, Host Marriott will not be permitted to declare or pay any cash distributions with respect to its capital stock or debt securities that rank pari passu with or junior to the Debentures.

Subject to certain restrictions, the Convertible Preferred Securities are redeemable at the Issuer's option upon any redemption by Host Marriott of the Debentures after December 2, 1999. Upon repayment at maturity or as a result of the acceleration of the Debentures upon the occurrence of a default, the Debentures shall be subject to mandatory redemption, from which the proceeds will be applied to redeem Convertible Preferred Securities and Common Securities, together with accrued and unpaid distributions.

- 11. In the second quarter of 1998, on behalf of Crestline, Host Marriott prepaid \$92 million of 9% unsecured debt provided by Marriott International. In the First Three Quarters 1998, the \$92 million note and an additional \$14.8 million intercompany note were forgiven and treated as a capital contribution to Crestline.
- 12. The pro forma September 11, 1998 balance sheet reflects the \$225 million earnings and profit distribution which is expected to be made by the end of the fiscal year in connection with the REIT Conversion. The amount of the earnings and profits distribution will be based upon Host Marriott's accumulated earnings and profit for tax purposes at the time of the REIT Conversion.

#### REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To Host Marriott Corporation:

We have audited the accompanying consolidated balance sheet of HMC Senior Communities, Inc. ("HMCSC"), which is the senior living communities' business of Host Marriott Corporation, as defined in Note 1 to the consolidated financial statements, as of January 2, 1998, and the related consolidated statements of operations, shareholder's equity and cash flows for the period from June 21, 1997 (inception) through January 2, 1998. These consolidated financial statements are the responsibility of Host Marriott Corporation's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audit.

We conducted our audit in accordance with generally accepted auditing standards. Those standards require that we plan and perform an audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of HMCSC as of January 2, 1998 and the results of its operations and its cash flows for the period from June 21, 1997 (inception) through January 2, 1998 in conformity with generally accepted accounting principles.

Arthur Andersen LLP

Washington, D.C. May 1, 1998

#### CONSOLIDATED BALANCE SHEET JANUARY 2, 1998 (IN THOUSANDS, EXCEPT SHARE DATA)

#### **ASSETS**

Property and equipment, net	1,332 10,686
Total assets	\$663,502 ======
LIABILITIES AND SHAREHOLDER'S EQUITY	
Debt Deferred income taxes Accounts payable and other accrued liabilities Amounts due to Marriott International, net Accrued interest Due to Host Marriott Corporation. Deferred revenue.	58,705 15,543 3,172 4,906 2,151 2,027
Total liabilities	436,438
Shareholder's equity: Common stock, 100 shares authorized, issued and outstanding, no par value	
Additional paid-in capital	226,706
Total shareholder's equity	227,064
Total liabilities and shareholder's equity	\$663,502 ======

See Notes to Consolidated Financial Statements.

#### CONSOLIDATED STATEMENT OF OPERATIONS

FOR THE PERIOD FROM JUNE 21, 1997 (INCEPTION) THROUGH JANUARY 2, 1998 (IN THOUSANDS)

REVENUES	\$ 36,900
OPERATING COSTS AND EXPENSES Depreciation and amortization	6.481
Total operating costs and expenses	20,929
OPERATING PROFIT BEFORE CORPORATE EXPENSES AND INTEREST	(2,304) (13,396)
INCOME BEFORE INCOME TAXES	
NET INCOME	

See Notes to Consolidated Financial Statements.

F-83

#### CONSOLIDATED STATEMENT OF SHAREHOLDER'S EQUITY

FOR THE PERIOD FROM JUNE 21, 1997 (INCEPTION) THROUGH JANUARY 2, 1998 (IN THOUSANDS)

		ADDITIONAL	
	COMMON	PAID-IN	RETAINED
	ST0CK	CAPITAL	EARNINGS
Balance, June 21, 1997	\$	\$	\$
Common stock issued			
tion		226,706	
Net income			358
Balance, January 2, 1998	\$	\$226,706	\$358
	====	=======	====

See Notes to Consolidated Financial Statements.

#### CONSOLIDATED STATEMENT OF CASH FLOWS

FOR THE PERIOD FROM JUNE 21, 1997 (INCEPTION) THROUGH JANUARY 2, 1998 (IN THOUSANDS)

OPERATING ACTIVITIES  Net income	\$	358
Depreciation and amortization. Change in amounts due to Marriott International. Change in amounts due to Host Marriott. Equity in earnings of affiliate. Change in other operating accounts.	1	.0,635 .0,073 2,151 (997) 3,156
Cash provided by operating activities	2	
INVESTING ACTIVITIES Capital expenditures	(3	(67)
Cash used in investing activities		
FINANCING ACTIVITIES Contribution of cash	( 2	7,319 2,142) 20,407 96
Cash provided by financing activities	2	
Increase in cash and cash equivalents	1	
Cash and cash equivalents, end of period	\$ 1	
SUPPLEMENTAL INFORMATIONNON-CASH ACTIVITY: Contributions from Host Marriott Corporation: Property and equipment	(33 (	9,892 31,669) 9,479) 2,054)
Deferred income taxes  Expansion costs paid by Host Marriott Corporation, which have been included in additional paid-in capital		68,435 .0,099

See Notes to Consolidated Financial Statements.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### 1. BASIS OF PRESENTATION

On June 21, 1997, Host Marriott Corporation ("Host Marriott") acquired all of the outstanding stock of Forum Group Inc. ("Forum Group") from Marriott Senior Living Services, Inc. ("MSLS"), a subsidiary of Marriott International, Inc. ("Marriott International") and concurrently contributed all of the assets and liabilities of Forum Group to HMC Senior Communities, Inc. ("HMCSC"). In connection with the acquisition, Forum Group assigned to Marriott International its interest as manager under long-term operating agreements (See Note 6).

On April 16, 1998, the Board of Directors of Host Marriott approved a plan to reorganize Host Marriott's current business operations by spinning-off HMCSC to the shareholders of Host Marriott, and contributing Host Marriott's hotels and certain other assets and liabilities to a newly formed Delaware limited partnership, Host Marriott, L.P., whose sole general partner would be Host Marriott Trust, a newly formed Maryland real estate investment trust. Host Marriott subsequently determined that Host Marriott would merge into HMC Merger Corporation, a Maryland corporation (to be renamed "Host Marriott Corporation"), which would qualify as a REIT and be the general partner of the Operating Partnership. After the proposed reorganization, HMCSC will lease hotels from Host Marriott, L.P. and Marriott International will continue to manage the hotels under long- term management agreements.

Consummation of the reorganization is subject to significant contingencies, including final Board approval and consent of shareholders, partners, bondholders, lenders and ground lessors of Host Marriott, its affiliates and other third parties. Accordingly, there can be no assurance that the reorganization will be completed.

The accompanying consolidated financial statements include the historical accounts of HMCSC, representing 31 senior living communities (the "Communities") located in 13 states, expected to be spun-off as part of the reorganization described above.

HMCSC operates as a unit of Host Marriott utilizing Host Marriott's employees, insurance and administrative services. HMCSC has no employees. Periodically, certain operating expenses, capital expenditures and other cash requirements of HMCSC are paid by Host Marriott and charged directly or allocated to HMCSC. Certain general and administrative costs of Host Marriott are allocated to HMCSC using a variety of methods, principally including Host Marriott's specific identification of individual cost items and otherwise through allocations based upon estimated levels of effort devoted by its general and administrative departments to individual entities or relative measures of size of the entities based on assets. In the opinion of management, the methods for allocating corporate, general and administrative expenses and other direct costs are reasonable. It is not practical to estimate the costs that would have been incurred by HMCSC if it had been operated on a stand-alone basis.

The consolidated financial statements present the financial position, results of operations and cash flows of HMCSC beginning on June 21, 1997 (the date Host Marriott acquired the stock of the Forum Group) through January 2, 1998. Host Marriott's basis in the assets and liabilities of HMCSC has been carried over to these financial statements. All material intercompany transactions and balances between HMCSC and its subsidiaries have been eliminated.

#### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

#### Principles of Consolidation

The consolidated financial statements include the accounts of HMCSC and its subsidiaries and controlled affiliates. Investments in affiliates over which HMCSC has the ability to exercise significant influence, but does not control, are accounted for using the equity method. All material intercompany transactions and balances have been eliminated.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

Fiscal Year

HMCSC's fiscal year ends on the Friday nearest to December 31.

#### Revenues

Revenues represent house profit from the Communities. House profit reflects the net revenues flowing to HMCSC as property owner and represents gross community operating sales less property-level expenses excluding depreciation and amortization, real and personal property taxes, insurance, management fees and certain other costs which are classified as operating costs and expenses in the accompanying statement of operations.

Resident fees and health care service revenues are generated primarily from monthly charges for independent living units and daily charges for assisted living suites and nursing beds, and are recognized monthly based on the terms of the residents' agreements. Advance payments received for services are deferred until the services are provided. Included in resident fees revenue is ancillary revenue, which is generated on a "fee for service" basis for supplemental items requested by residents and is recognized as the services are provided.

A portion of revenues from health care services were attributable to patients whose bills are paid by Medicare or Medicaid under contractual arrangements. Reimbursements under these contractual arrangements are subject to retroactive adjustments based on agency reviews. Revenues from health care services are recorded net of estimated contractual allowances in the accompanying consolidated financial statements. Management believes that reserves recorded are adequate to cover any retroactive adjustments arising from such reviews.

HMCSC has considered the impact of EITF 97-2 on its financial statements and has determined that it requires HMCSC to include property-level revenues and operating expenses of its senior living communities in its statements of operations. HMCSC will adopt EITF 97-2 in the fourth quarter of 1998 with retroactive effect in prior periods to conform to the new presentation. The effect of this change will be to increase 1997 revenues and operating costs and expenses by approximately \$74.1 million and will have no impact on operating profit or net income. See Note 3.

#### Cash and Cash Equivalents

All highly liquid investments with a maturity of three months or less at date of purchase are considered cash equivalents.

#### Property and Equipment

Property and equipment is recorded at cost, or if contributed by Host Marriott, is recorded at Host Marriott's basis. Replacements and improvements that extend the useful life of property and equipment are capitalized.

Depreciation is computed using the straight-line method over the estimated useful lives of the assets, generally 40 years for buildings and three to 10 years for furniture and equipment. Leasehold improvements are amortized over the shorter of the lease term or the useful lives of the related assets.

In cases where management is holding for sale a particular Community, HMCSC assesses impairment based on whether the estimated sales price less cost of disposal of each individual property to be sold is less than the net book value. A property is considered to be held for sale when a decision is made to dispose of the Community. Otherwise, impairment is assessed based on whether it is probable that undiscounted future cash flows from each Community will be less than its net book value. If a Community is impaired, its basis is adjusted to its fair value.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

#### Concentration of Credit Risk

Financial instruments that potentially subject HMCSC to significant concentration of credit risk consist principally of cash and cash equivalents. HMCSC maintains cash and cash equivalents with various high credit-quality financial institutions and limits the amount of credit exposure with any institution.

#### Working Capital

Pursuant to the terms of HMCSC's Operating Agreements (see Note 6), HMCSC is required to provide Marriott International with working capital and supplies to meet the operating needs of the Communities. Marriott International converts cash advanced by HMCSC into other forms of working capital consisting primarily of operating cash, inventories, resident deposits and trade receivables and payables which are maintained and controlled by Marriott International. Upon the termination of the Operating Agreements, Marriott International is required to convert working capital and supplies into cash and return it to HMCSC. As a result of these conditions, the individual components of working capital and supplies controlled by Marriott International are not reflected in the accompanying consolidated balance

#### Deferred Revenue

Monthly fees deferred for the non-refundable portion of the entry fees are included as deferred revenue in the accompanying balance sheet. These amounts are recognized as revenue as health care services are performed over the expected term of the residents' contracts.

#### Use of Estimates in the Preparation of Financial Statements

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

#### New Accounting Standards

In 1997, the Company adopted Statement of Financial Accounting Standards No. 129 "Disclosure of Information About Capital Structure." The adoption of this statement did not have a material effect on these consolidated financial statements.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

#### 3. REVENUES

House profit generated by the Communities consist of the following for the period from June 21, 1997 (inception) through January 2, 1998 (in thousands):

Community Sales RoutineAncillary	10,980
Total Community Sales	110,969
Department Costs RoutineAncillary	64,516 9,553
Total Department Costs	74,069
Department Profit RoutineAncillary	35,473 1,427
Revenues	\$ 36,900

Community sales consist of routine and ancillary sales. Routine sales are generated from monthly charges for independent living units and daily charges for assisted living suites and nursing beds, and are recognized monthly based on the terms of the residents' agreements. Advance payments received for services are deferred until the services are provided. Ancillary sales are generated on a "fee for service" basis for supplementary items requested by residents, and are recognized as the services are provided.

Total sales include amounts estimated by management to be reimbursable through Medicare, Medicaid and other third party payor agreements. Medicare and Medicaid represented 11% and 3%, respectively, of sales for the period from June 21, 1997 (inception) through January 2, 1998. Reimbursement arrangements are subject to audit and retroactive adjustment. Provisions are made for potential adjustments that may result. To the extent those provisions vary from settlements, sales are charged or credited when the adjustments become final. There were no adjustments in the period presented related to reimbursement arrangements. In management's opinion, any adjustments related to current and prior years' operations will be immaterial to current and future financial statements. Audits under the reimbursement agreements have been completed through fiscal year 1996 and there were no material audit adjustments.

#### 4. PROPERTY AND EQUIPMENT

Property and equipment consists of the following at January 2, 1998 (in thousands):

Land and land improvements  Buildings and leasehold improvements  Furniture and equipment	518,056
Less accumulated depreciation and amortization	644,475 (10,635)
	\$633,840

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

In December 1997, LTJ Senior Communities Corporation ("LTJ"), a wholly owned subsidiary of HMCSC, acquired 49% of the remaining 50% interest in Leisure Park Venture Limited Partnership (the "Partnership") which owns a 418-unit retirement community in New Jersey for approximately \$23 million, including the assumption of approximately \$15 million of debt. Subsequent to this acquisition, HMCSC indirectly owns a 99% interest in the Partnership. Marriott International owns the remaining 1% limited partner interest.

In the first quarter of 1998, LTJ also acquired the Gables of Winchester in suburban Boston, a 124-unit upscale senior living community, for \$21 million and entered into conditional purchase agreements for two Marriott Brighton Gardens assisted living communities with the Summit Companies of Denver, Colorado. After the anticipated completion of construction in the first quarter of 1999, HMCSC may acquire these two 160-unit properties located in Denver and Colorado Springs, Colorado, for approximately \$35 million, if they achieve certain operating performance criteria. All three of these communities will be operated by Marriott International under long-term operating agreements.

#### 5. RESTRICTED CASH

Restricted cash consists of the following at January 2, 1998 (in thousands):

Debt service reserve fund	\$ 1,528
Fixed asset reserve fund	4,300
Real estate tax reserve fund	3,590
Insurance reserve fund	1,268
	\$10,686

The debt service, fixed asset, real estate tax and insurance reserve funds consist of cash transferred into segregated escrow accounts out of sales generated by the Communities, pursuant to HMCSC's secured debt agreements. These funds are periodically disbursed by the collateral agent to pay for debt service, capital expenditures, insurance premiums and real estate taxes relating to the secured properties. In some cases, to ensure prompt payment, HMCSC utilizes its unrestricted cash to pay for capital expenditures, insurance premiums and real estate taxes and is subsequently reimbursed for such payments out of funds held in the appropriate escrow account.

#### 6. OPERATING AGREEMENTS

The Communities are subject to operating agreements (the "Operating Agreements") which provide for Marriott International to operate the Communities, generally for an initial term of 25 to 30 years with renewal terms subject to certain performance criteria at the option of Marriott International of up to an additional five to ten years. The Operating Agreements provide for payment of base management fees generally equal to five to eight percent of gross sales and incentive management fees generally equal to zero to 20% of Operating Profit (as defined in the Operating Agreements) over a priority return to HMCSC. In the event of early termination of the Operating Agreements, Marriott International will receive additional fees based on the unexpired term and expected future base and incentive management fees. HMCSC has the option to terminate certain, but not all, management agreements if specified performance thresholds are not satisfied. No Operating Agreement with respect to a single Community is cross-collateralized or crossdefaulted to any other Operating Agreement, and any single Operating Agreement may be terminated following a default by HMCSC or Marriott International, although such termination will not trigger the cancellation of any other Operating Agreement.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

Pursuant to the terms of the Operating Agreements, Marriott International is required to furnish the Communities with certain services ("Central Administrative Services") which are provided on a central or regional basis to all properties in the Marriott Retirement Community System. These services include the development and operation of computer systems, computer payroll and accounting services, marketing and public relations services, and such additional services as may from time-to-time be performed more efficiently on a central or regional level. The Operating Agreements require payment of Central Administrative Services fees equal to 2% of gross sales beginning in the third guarter of 1998.

Marriott International is required under the Operating Agreements to deduct an amount from gross sales and place the funds into an interest-bearing reserve account to cover the cost of (a) certain routine repairs and maintenance to the Communities which are normally capitalized and (b) replacements and renewals to the Communities' property and improvements. The annual payment amount (expressed as a percentage of gross sales) generally will be 2.65% through fiscal year 2002, 2.85% for fiscal years 2003 through 2007, and 3.5% thereafter. The amount contributed for the period June 21, 1997 (inception) through January 2, 1998 was \$2,025,000. The Operating Agreements provide that HMCSC shall provide Marriott International with sufficient funds to cover the cost of certain major or non-routine repairs, alterations, improvements, renewals and replacements to the Communities which are required to maintain a competitive, efficient and economical operating condition in accordance with Marriott standards or for the continued safe and orderly operation of the Communities.

#### 7. AMOUNTS DUE TO MARRIOTT INTERNATIONAL

The components of the amounts due to Marriott International, net, at January 2, 1998 are as follows (in thousands):

	======
Total	\$ 3,172
Other, net	355
Community working capital due to HMCSC	(6,093)
Management fees payable to Marriott International	
Community operating expenses payable to Marriott International	

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

#### 8. DEBT

Debt consists of the following at January 2, 1998 (in thousands):

Mortgage Debt: Secured by eight Communities with \$232 million of assets, with an interest rate of 10.01%, maturing through 2020 (balance includes fair value adjustment of \$15.5 million)	\$137,713
fair value adjustment of \$2.6 million)	49,353
rate of 7.45%, maturing through 1999 (repaid in 1998)	26,403
	213,469
Notes payable to Marriott International, with a rate of 9%, maturing through 2001 (repaid in 1998)	92,195
Other notes: Revenue Bonds with a rate of 5.875%, maturing through 2027 Other notes, with an average rate of 6.6%, maturing through December	14,700
2027 Capital lease obligations	18,943 10,627
	44,270
Total debt	\$349,934 ======

Debt maturities at January 2, 1998, excluding the unamortized fair value adjustments of approximately \$18 million resulting from recording the mortgages at their fair value on June 21, 1997, are as follows (in thousands):

1998	 \$ 54,515
1999	
2000	 4,503
2001	 88,043
2002	 2,504
Thereafter	 152,046
	\$331,808

In conjunction with the acquisition of Forum Group Inc., HMCSC recorded the debt assumed at its fair value, which exceeded the face value by approximately \$19 million. HMCSC is amortizing this adjustment to interest expense over the remaining life of the related debt. The amortization for the period from June 21, 1997 (inception) through January 2, 1998 totaled \$834,000. Cash paid for interest for the period from June 21, 1997 (inception) through January 2, 1998 totaled \$8,183,000.

In conjunction with the June 21, 1997 acquisition of Forum Group Inc., HMCSC assumed approximately \$197 million in mortgage debt, \$11 million in capital lease obligations (see Note 9), as well as issued \$72 million in notes payable to Marriott International. Subsequent to the acquisition, HMCSC issued additional notes payable to Marriott International for additional expansion units totaling approximately \$20 million. These notes were guaranteed by Host Marriott. In the second quarter of 1998, Host Marriott repaid the \$92 million in notes payable to Marriott International.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

In December 1997, in connection with the acquisition of the remaining 50% interest in the Leisure Park Venture Limited Partnership (see Note 4), HMCSC assumed approximately \$15 million of debt.

The net assets of seventeen of the Communities are subject to mortgage debt which places restrictions on their assets. The net assets of the Communities totaled approximately \$150 million at January 2, 1998. The indentures governing these mortgages contain covenants that, among other things, require maintenance of segregated cash collection of all rents, separate cash reserves for debt service, property improvements, real estate taxes and insurance, limit the ability to incur additional indebtedness, issue stock or admit additional partners, pay dividends or make certain distributions, enter into or cancel leases, enter into certain transactions with affiliates or sell certain assets.

During the first quarter of 1998, Host Marriott prepaid \$26.4 million in mortgage debt. Host Marriott's prepayment of the debt was recorded as a capital contribution to HMCSC, there was no gain or loss on the prepayment.

#### 9. LEASES

HMCSC leases certain property under non-cancelable capital and operating leases. Future minimum annual rental commitments for all non-cancelable leases are as follows:

	CAPITAL LEASES 	
1998. 1999. 2000. 2001. 2002. Thereafter.	1,287 1,300 1,320 1,338	\$ 278 278 278 278 278 278 3,062
Total minimum lease payments	,	\$4,452 =====
Less amount representing interest  Present value of minimum lease payments		

HMCSC leases two communities under capital leases expiring in 2016. Upon the expiration of the lease or anytime prior to lease expiration, HMCSC has the first right of refusal (the "Option") to submit a counter offer to any acceptable bona fide offer from a third party within 30 days of notice from the lessor. If HMCSC fails to exercise its Option, then the lessor may proceed with the sale of the leased property and all assets therein.

HMCSC also has one long-term operating ground lease which expires in 2013. The operating lease includes three renewal options exercisable in 5 year increments through the year 2028.

Rent expense for the period from June 21, 1997 (inception) through January 2, 1998 was \$141,000.

#### 10. INCOME TAXES

Total deferred tax assets and liabilities as of January 2, 1998 were as follows (in thousands):

Deferred tax assets Deferred tax liabilities	
Net deferred income tax liability	\$(58,705)

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

The tax effect of each type of temporary difference and carryforward that gives rise to a significant portion of deferred tax assets and liabilities as of January 2, 1998 was as follows (in thousands):

Property and equipment	\$(68,687)
Debt adjustment to fair value at acquisition	7,591
Other, net	2,391
Net deferred income tax liability	\$(58,705)

The provision for income taxes consists of the following for the period from June 21, 1997 (inception) through January 2, 1998 (in thousands):

CurrentFederal	
State	(5)
	(30)
DeferredFederal	238
State	41
	279
	\$249
	=====

A reconciliation of the statutory Federal tax rate to HMCSC's effective income tax rate for the period from June 21, 1997 (inception) through January 2, 1998 follows:

·	====
	41.0%
State income taxes, net of federal tax benefit	6.0
Statutory rederal tax rate	35.0%

HMCSC is included in the consolidated federal income tax return of Host Marriott and its affiliates (the "Group") for the period from June 21, 1997 (inception) through January 2, 1998. Tax expense is allocated to HMCSC as a member of the Group based upon the relative contribution to the Group's consolidated taxable income/loss and changes in temporary differences. This allocation method results in federal and net state tax expense allocated for all periods presented that is substantially equal to the expense that would have been recognized if HMCSC had filed separate tax returns. HMCSC reimburses Host Marriott for the allocable share of current taxes payable relating to the period that HMCSC has been included in Host Marriott's consolidated federal income tax return.

#### 11. COMMITMENTS AND CONTINGENCIES

On June 15, 1995, The Russell F. Knapp Revocable Trust (the "Plaintiff") filed a complaint in the United States District Court for the Southern District of Indiana (the "Indiana Court") against the general partner of one of HMCSC's subsidiary partnerships, Forum Retirement Partners, L.P. alleging breach of the partnership agreement, breach of fiduciary duty, fraud, insider trading and civil conspiracy/aiding and abetting. On February 4, 1998, the Plaintiff, MSLS, the general partner, Forum Group and HMCSC entered into a Settlement and Release Agreement (the "Settlement Agreement"), pursuant to which Host Marriott agreed to purchase, at a price of \$4.50 per unit, the partnership units of each limited partner electing to join in the Settlement Agreement.

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

HMCSC held 79% of the outstanding limited partner units in the partnership at that time. HMCSC also agreed to pay as much as an additional \$1.25 per unit to the settling limited partners, under certain conditions, in the event that HMCSC within three years following the date of settlement initiates a tender offer for the purchase of units not presently held by HMCSC or the settling limited partners. On February 5, 1998, the Indiana Court entered an order approving the dismissal of the Plaintiff's case.

In connection with the Settlement Agreement on March 25, 1998, HMCSC acquired 1,000,894 limited partner unit shares for approximately \$4,504,000. The purchase price of the shares approximated fair value and accordingly, no portion of the purchase price has been expensed. As a result of this purchase, HMCSC's ownership interest in the partnership was increased to approximately 86%

#### 12. FAIR VALUE OF FINANCIAL INSTRUMENTS

HMCSC believes the carrying amount of its financial instruments (excluding property indebtedness) approximates their fair value due to the relatively short maturity of these instruments. There is no quoted market value available for any of HMCSC's financial instruments.

Valuations of debt are determined based on expected future payments discounted at risk-adjusted rates. The debt was adjusted to its fair value in conjunction with Host Marriott's acquisition of the Communities on June 21, 1997. As of January 2, 1998, the fair value of debt approximated its carrying value.

# CONDENSED CONSOLIDATED BALANCE SHEET SEPTEMBER 11, 1998 (UNAUDITED, IN THOUSANDS, EXCEPT SHARE DATA)

#### **ASSETS**

Property and equipment, net	4,097 14,290
Total assets	\$694,419 ======
LIABILITIES AND SHAREHOLDER'S EQUITY	
Debt Deferred income taxes Due to Host Marriott Corporation, net Accounts payable and other accrued liabilities Deferred revenue  Total liabilities	61,376 12,989 13,639 1,310
Shareholder's equity:	
Common stock, 100 shares authorized, issued and outstanding, no	
par valueAdditional paid-in capitalRetained earnings	
Total shareholder's equity	392,071
Total liabilities and shareholder's equity	

See Notes to Condensed Consolidated Financial Statements.

# CONDENSED CONSOLIDATED STATEMENT OF OPERATIONS FOR THE THIRTY-SIX WEEKS ENDED SEPTEMBER 11, 1998 AND THE TWELVE WEEKS ENDED SEPTEMBER 12, 1997 (UNAUDITED, IN THOUSANDS)

	SEPTEMBER 11, 1998	WEEKS ENDED SEPTEMBER 12,
REVENUES	\$ 57,800	\$16,036
OPERATING COSTS AND EXPENSES		
Depreciation and amortization	14,759 9,408 4,773 863	4,866 3,159 1,462 251
Total operating costs and expenses	29,803	9,738
OPERATING PROFIT BEFORE CORPORATE EXPENSES AND IN-		
TEREST	27,997 (2,937) (17,560) 1,120	
INCOME (LOSS) BEFORE INCOME TAXES	8,620 (3,534)	(451) 185
NET INCOME (LOSS)	\$ 5,086 =====	\$ (266) ======

See Notes to Condensed Consolidated Financial Statements.

# CONSOLIDATED STATEMENT OF CASH FLOWS FOR THE THIRTY-SIX WEEKS ENDED SEPTEMBER 11, 1998 AND THE TWELVE WEEKS ENDED SEPTEMBER 12, 1997 (UNAUDITED, IN THOUSANDS)

	1998	TWELVE WEEKS ENDED SEPTEMBER 12, 1997
OPERATING ACTIVITIES  Net income (loss)	\$ 5,086	\$ (266)
Depreciation and amortization	14,759	4,866
al	(5,353)	9,259
tion Equity in earnings of affiliate	10,507 51	5,096 (19)
Change in other operating accounts	(6,026)	2,551
Cash provided by operating activities	19,024	21,487
INVESTING ACTIVITIES Capital expenditures Increase in capital improvement reserve	(8,007) 478	(18,409) 386
Cash used in investing activities	(7,529)	(18,022)
FINANCING ACTIVITIES Repayments (issuances) of debt	(3,502) 867	19,191 2,205
Cash used in financing activities	(2,635)	21,396
Increase in cash and cash equivalents	8,860 17,644	9,571 6,401
Cash and cash equivalents, end of period	\$26,504 ======	\$15,972 ======
SUPPLEMENTAL INFORMATIONNON-CASH ACTIVITY: Contributions from Host Marriott Corporation: Property and equipment Other Mortgage debt paid by Host Marriott	\$22,439 4,084 26,403	  

See Notes to Condensed Consolidated Financial Statements.

# HMC SENIOR COMMUNITIES, INC., WHICH IS THE SENIOR LIVING COMMUNITIES' BUSINESS OF HOST MARRIOTT CORPORATION NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

1. On June 21, 1997, Host Marriott Corporation ("Host Marriott") acquired all of the outstanding stock of Forum Group Inc. ("Forum Group"), from Marriott Senior Living Services, Inc. ("MSLS"), a subsidiary of Marriott International, Inc. ("Marriott International") and concurrently contributed all of the assets and liabilities of Forum Group, Inc. to HMC Senior Communities, Inc. ("HMCSC"). In connection with the acquisition, Forum Group assigned to Marriott International its interest as manager under long-term operating agreements.

On April 16, 1998, the Board of Directors of Host Marriott approved a plan to reorganize Host Marriott's current business operations by spinning-off Host Marriott's senior living business ("Senior Living") into a separate corporation, the Senior Living Communities Company and contributing Host Marriott's hotels and certain other assets and liabilities to a newly formed Delaware limited partnership, Host Marriott, L.P., whose sole general partner will be HMC Merger Corporation (to be renamed "Host Marriott Corporation"), a newly formed Maryland corporation into which Host Marriott, a Delaware corporation, will merge and will qualify as a real estate investment trust ("REIT") (the "REIT Conversion"). After the proposed REIT Conversion, HMCSC will lease hotels from Host Marriott, L.P. and Marriott International will continue to manage the hotels under long term management agreements.

Consummation of the REIT Conversion is subject to significant contingencies, including final Board approval, consent of shareholders, partners, bondholders, lenders and ground lessors of Host Marriott, its affiliates and other third parties. Accordingly, there can be no assurance that the REIT Conversion will be completed.

The accompanying consolidated financial statements include the historical accounts of HMCSC, representing 31 senior living communities (the "Communities") located in 13 states, expected to be spun-off as part of the REIT Conversion described above.

The accompanying condensed consolidated financial statements have been prepared by HMCSC without audit. Certain information and footnote disclosures normally included in financial statements presented in accordance with generally accepted accounting principles have been condensed or omitted. HMCSC believes the disclosures made are adequate to make the information presented not misleading. However, the condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's audited financial statements for the period from June 21, 1997 (inception) through January 2, 1998.

In the opinion of the Company, the accompanying unaudited condensed consolidated financial statements reflect all adjustments necessary to present fairly the financial position of the Company as of September 11, 1998 and the results of operations and cash flows for the thirty-six weeks ended September 11, 1998 and the twelve weeks ended September 12, 1997. Interim results are not necessarily indicative of fiscal year performance because of the impact of seasonal and short-term variations.

2. Revenues represent house profit from the Communities. House profit reflects the net revenues flowing to HMCSC as property owner and represents gross community operating sales less property-level expenses excluding depreciation and amortization, real and personal property taxes, insurance, management fees and certain other costs which are classified as operating costs and expenses.

Resident fees and health care service revenues are generated primarily from monthly charges for independent living units and daily charges for assisted living suites and nursing beds, and are recognized monthly based on the terms of the residents' agreements. Advance payments received for services are deferred until the services are provided. Included in resident fees revenue is ancillary revenue, which is generated on a "fee for service" basis for supplemental items requested by residents and is recognized as the services are provided.

# HMC SENIOR COMMUNITIES, INC., WHICH IS THE SENIOR LIVING COMMUNITIES' BUSINESS OF HOST MARRIOTT CORPORATION

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

A portion of revenues from health care services were attributable to patients whose bills are paid by Medicare or Medicaid under contractual arrangements. Reimbursements under these contractual arrangements are subject to adjustments based on agency reviews. Revenues and receivables from health care services are recorded net of estimated contractual allowances in the accompanying consolidated financial statements. Management believes that reserves recorded are adequate to cover any retroactive adjustments arising from such reviews.

House profit generated by the Communities consist of the following for the thirty-six weeks ended September 11, 1998 and the twelve weeks ended September 12, 1997 (in thousands):

	1998	ENDED SEPTEMBER 12,
Community Sales		
Routiné	\$149,581	\$42,249
Ancillary	16,410	4,265
Total Community Sales	165,991	46,514
Department Costs		
Routine		26,821
Ancillary	13,482	3,658
Total Department Costs	108,191	30,479
Described to Describe		
Department Profit	F.4. 070	45 400
Routine		15,428
Ancillary	2,928	607
Dovonuos	¢ 57 000	
Revenues	\$ 57,800	\$16,035 

HMCSC has considered the impact of EITF 97-2 on its financial statements and has determined that it requires HMCSC to include property-level revenues and operating expenses of its senior living communities in its statements of operations. HMCSC will adopt EITF 97-2 in the fourth quarter of 1998 with retroactive effect in prior periods to conform to the new presentation. The effect of this change will be to increase revenues and operating costs and expenses for the thirty-six weeks ended September 11, 1998 and the twelve weeks ended September 12, 1997 by approximately \$108 million and \$30 million, respectively, and will have no impact on operating profit or net income.

- 3. In the first quarter of 1998, HMCSC also acquired the Gables of Winchester in suburban Boston, a 124-unit upscale senior living community, for \$21 million and entered into conditional purchase agreements for two Marriott Brighton Gardens assisted living communities from the Summit Companies of Denver, Colorado. After the anticipated completion of construction in the first quarter of 1999, HMCSC may acquire these two 160-unit properties located in Denver and Colorado Springs, Colorado, for approximately \$35 million, if they achieve certain operating performance criteria. All three of these communities will be operated by Marriott International under long-term operating agreements.
- 4. During the first quarter of 1998, Host Marriott prepaid \$26.4 million in mortgage debt. Host Marriott's prepayment of debt was recorded as a capital contribution to HMCSC. In the second quarter of 1998, Host Marriott prepaid \$92 million of 9% unsecured debt provided by Marriott International related to the Communities. Host Marriott then held a \$92 million note which was forgiven and treated as a capital contribution to HMCSC, along with an additional \$14.8 million in intercompany notes, in the third quarter of 1998.
- 5. During the first quarter of 1998, Host Marriott prepaid \$26.4 million in mortgage debt. Host Marriott's prepayment of debt was recorded as a capital contribution to HMCSC.

#### REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To the Partners of Host Marriott L.P.

We have audited the accompanying balance sheet of Host Marriott, L.P. (the "Partnership"), a Delaware limited partnership as of June 19, 1998. This balance sheet is the responsibility of the Partnership's management. Our responsibility is to express an opinion on the balance sheet based on our audit.

We conducted our audit in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the balance sheet is free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the balance sheet. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall balance sheet presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, the balance sheet referred to above presents fairly, in all material respects, the financial position of the Partnership as of June 19, 1998, in conformity with generally accepted accounting principles.

Arthur Andersen LLP

Washington, D.C. August 5, 1998

## HOST MARRIOTT, L.P.

### BALANCE SHEET

### ASSETS

	SEPTEMBER 11, 1998	1998
	(UNAUDITED)	
Cash	\$ ====	\$ =====
PARTNERS' CAPITAL		
General partnerLimited partner	\$ 1 99	\$ 1 99
Less: subscription receivable	100 (100)	100 (100)
	\$	\$
	=====	=====

The accompanying notes are an integral part of this balance sheet.

F-102

### NOTES TO BALANCE SHEET SEPTEMBER 11, AND JUNE 19, 1998

#### NOTE 1. ORGANIZATION

On April 16, 1998, the Board of Directors of Host Marriott Corporation ("Host Marriott") approved a plan to restructure Host Marriott's current business operations through the spin-off of Host Marriott's senior living business ("SLC") and the contribution of Host Marriott's hotels and certain other assets and liabilities to a newly formed Delaware limited partnership, Host Marriott, L.P. (the "Operating Partnership"), whose sole general partner will be HMC Merger Corporation, a newly formed Maryland corporation into which Host Marriott Corporation, a Delaware corporation will be merged subsequent to the contribution. HMC Merger Corporation (to be renamed "Host Marriott Corporation"), expects to qualify as a REIT and will be the general partner of the Operating Partnership. Host Marriott's contribution of its hotels and certain assets and liabilities to the Operating Partnership (the "Contribution") in exchange for units of limited partnership interests in the Operating Partnership will be accounted for at Host Marriott's historical basis.

The accompanying balance sheets of the Operating Partnership includes its accounts as of September 11, and June 19, 1998. The assets and liabilities of the Host Marriott Hotels will be included in the Contribution by Host Marriott to the Operating Partnership in connection with its planned conversion to a REIT (the "REIT Conversion"), anticipated to become effective January 1, 1999.

In June 1998, as part of the REIT Conversion, the Operating Partnership and HMC Merger Corporation filed a preliminary Prospectus/Consent Solicitation with the Securities and Exchange Commission. This Prospectus/Consent Solicitation Statement describes a proposal whereby the Operating Partnership will acquire by merger (the "Mergers") eight public limited partnerships (the "Partnerships") that own or control 24 full-service hotels in which Host Marriott or its subsidiaries are general partners. As more fully described in the Prospectus/Consent Solicitation Statement, limited partners of those Partnerships that participate in the Mergers will receive OP Units, and may at their election, exchange such OP Units for unsecured notes due December 15, 2005 issued by the Operating Partnership ("Notes") or common shares of the REIT.

However, the consummation of the REIT Conversion is subject to significant contingencies that are outside the control of Host Marriott, including final Board of Directors approval, consents of shareholders, partners, bondholders, lenders and ground lessors of Host Marriott, its affiliates and other third parties. Accordingly, there can be no assurance that the REIT Conversion or the Contribution will be completed.

On April 20, 1998, Host Marriott and certain of its subsidiaries filed a shelf registration on Form S-3 (the "Shelf Registration") with the Securities and Exchange Commission for \$2.5 billion in securities, which may include debt, equity or a combination thereof. Host Marriott anticipates that any net proceeds from the sale of offered securities will be used for refinancing of Host Marriott's indebtedness, potential future acquisitions and general corporate purposes.

On August 5, 1998, HMH Properties, Inc. ("HMH Properties"), an indirect wholly-owned subsidiary of Host Marriott, which owns 61 of Host Marriott's hotels, purchased substantially all of its (i) \$600 million in 9 1/2% senior notes due 2005, (ii) \$350 million in 9% senior notes due 2007 and (iii) \$600 million in 8 7/8% senior notes due 2007 (collectively, the "Old Senior Notes"). Concurrently with each offer to purchase, HMH Properties solicited consents (the "1998 Consent Solicitations") from registered holders of the Old Senior Notes to certain amendments to eliminate or modify substantially all of the restrictive covenants and certain other provisions contained in the indentures pursuant to which the Old Senior Notes were issued. HMH Properties simultaneously utilized the Shelf Registration to issue an aggregate of \$1.7 billion in senior notes (the "New Senior Notes"). The New Senior Notes were issued in two series, \$500 million of 7 7/8 Series A notes due in 2005 and \$1.2 billion of 7 7/8 Series B notes due in 2008. The 1998 Consent Solicitations facilitated the merger

of HMC Capital Resources Holdings Corporation ("Capital Resources"), a whollyowned subsidiary of the Company, with and into HMH Properties. Capital Resources, the owner of eight of Host Marriott's hotel properties, was the obligor under the \$500 million credit facility (the "Old Credit Facility").

In conjunction with the issuance of the New Senior Notes, HMH Properties entered into a \$1.25 billion credit facility (the "New Credit Facility") with a group of commercial banks. The New Credit Facility will initially have a three year term with two one year extension options. Borrowings under the New Credit Facility generally bear interest at the Eurodollar rate plus 1.75%. The interest rate and commitment fee (currently 0.35% on the unused portion of the New Credit Facility) fluctuate based on certain financial ratios.

The New Credit Facility and the indenture under which the New Senior Notes were issued contain covenants restricting the ability of HMH Properties and certain of its subsidiaries to incur indebtedness, grant liens on their assets, acquire or sell assets or make investments in other entities, and make distributions to equityholders of HMH Properties, Host Marriott, and (following the REIT Conversion) the Operating Partnership and Host REIT.

Following the REIT Conversion, the New Credit Facility permits the Operating Partnership to make distributions to holders of OP Units, including Host REIT, in an aggregate amount for every four fiscal quarters equal to the greater of (i) 85% of adjusted funds from operations plus the net proceeds of equity offerings and (ii) the minimum amount necessary to permit Host REIT to maintain its status as a REIT and to satisfy certain other requirements, provided that no specified default or event of default has occurred under the New Credit Facility and is continuing. The New Credit Facility also permits the Operating Partnership to make distributions to Host REIT sufficient to enable Host REIT to make the E&P Distribution.

Following the REIT Conversion, the indenture permits the Operating Partnership to make distributions to holders of OP Units, including Host REIT, in amounts equal to the greater of (i) 95% of FFO plus net proceeds of equity offerings (provided that no event of default under the indenture has occurred and is continuing and the Operating Partnership is able to incur debt under the applicable indenture covenants) or (ii) an amount sufficient to permit Host REIT to maintain its status as a REIT and satisfy certain other requirements (provided that no event of default under the indenture has occurred and is continuing and the Operating Partnership has a consolidated debt to adjusted total assets ratio that is less than a specified level). The indenture also permits the Operating Partnership to make distributions to Host REIT sufficient to enable Host REIT to make the E&P Distribution.

The New Credit Facility and the New Senior Notes also contain certain financial covenants relating to, among other things, maintaining certain levels of tangible net worth and certain ratios of EBITDA to interest and fixed charges, total debt to EBITDA, unencumbered assets to unsecured debt, and secured debt to total debt.

The New Credit Facility replaces the Company's Old Credit Facility. The net proceeds from the offering and borrowings under the New Credit Facility were used by Host Marriott to purchase substantially all of the Old Senior Notes to repay amounts outstanding under the Old Credit Facility and to make bond premium and consent payments totaling approximately \$175 million. These costs, along with the write-off of deferred financing fees of approximately \$52 million related to the Old Senior Notes and the Old Credit Facility, were recorded as a pre-tax extraordinary loss on the extinguishment of debt in the third quarter of 1998. The New Senior Notes and the New Credit Facility are guaranteed by Host Marriott and its wholly owned subsidiary, Host Marriott Hospitality, Inc. and certain subsidiaries of HMH Properties and are secured by pledges of equity interests in certain subsidiaries of HMH Properties.

#### REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To the Board of Directors of Host Marriott Corporation

We have audited the accompanying balance sheet of HMC Merger Corporation (the "Company"), a Maryland corporation, as of September 28, 1998. This balance sheet is the responsibility of the Company's officers. Our responsibility is to express an opinion on the balance sheet based on our audit.

We conducted our audit in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the balance sheet is free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the balance sheet. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall balance sheet presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, the balance sheet referred to above presents fairly, in all material respects, the financial position of the Company as of September 28, 1998, in conformity with generally accepted accounting principles.

Arthur Andersen LLP

Washington, D.C. September 29, 1998

### HMC MERGER CORPORATION

## BALANCE SHEET AS OF SEPTEMBER 28, 1998

### ASSETS

Cash	\$
EQUITY	
Common shares, no par value, 100 shares authorized, issued and outstanding	100

The accompanying notes are an integral part of this balance sheet.

F-106

#### HMC MERGER CORPORATION

### NOTES TO BALANCE SHEET SEPTEMBER 28, 1998

#### NOTE 1. ORGANIZATION

On April 16, 1998, the Board of Directors of Host Marriott Corporation ("Host Marriott") approved a plan to restructure Host Marriott's current business operations through the spin-off of Host Marriott's senior living business ("Crestline") and the contribution of Host Marriott's hotels and certain other assets and liabilities to a newly formed Delaware limited partnership, Host Marriott, L.P. (the "Operating Partnership") whose sole general partner will be HMC Merger Corporation (the "Company"), a newly formed Maryland corporation that will merge with Host Marriott Corporation, a Delaware corporation, subsequent to the Contribution. Host Marriott's contribution of its hotels and certain assets and liabilities to the Operating Partnership (the "Contribution") in exchange for units of limited partnership interests in the Operating Partnership will be accounted for at Host Marriott's historical basis. The accompanying balance sheet of the Company includes its accounts as of September 28, 1998, its formation date. Subsequent to the REIT Conversion, it is expected that the sole asset of the Company will be its 74% ownership of the Operating Partnership.

In June 1998, as part of the REIT Conversion, the Operating Partnership and the Company filed a preliminary Prospectus/Consent Solicitation with the Securities and Exchange Commission. This Prospectus/Consent Solicitation Statement describes a proposal whereby the Operating Partnership will acquire by merger (the "Mergers") eight public limited partnerships (the "Partnerships") that own or control 24 full-service hotels in which Host Marriott or its subsidiaries are general partners. As more fully described in the Prospectus/Consent Solicitation Statement, limited partners of those Partnerships that participate in the Mergers will receive OP Units and may, at their election, exchange such OP Units received for unsecured notes due December 15, 2005 issued by the Operating Partnership ("Notes") or common stock of the Company.

The consummation of the REIT Conversion is subject to significant contingencies that are outside the control of Host Marriott, including final Board of Directors approval, consents of shareholders, partners, bondholders, lenders and ground lessors of Host Marriott, its affiliates and other third parties. Accordingly, there can be no assurance that the REIT Conversion or the Contribution will be completed.

#### PRO FORMA FINANCIAL INFORMATION OF THE OPERATING PARTNERSHIP

Given the structure of Host Marriott's Consent Solicitation, the Mergers and the REIT Conversion may take a variety of different forms. The variations are dependent in part on the number and identity of the Partnerships that elect to merge and whether limited partners elect to tender their Partnership Interests for OP Units or Notes in connection with the REIT Conversion.

In light of the number of possible variations, the Company is not able to describe all possible combinations of Hotel Partnerships that could compose the Operating Partnership. However, to assist shareholders in analyzing the Mergers and the REIT Conversion, the Company has prepared four separate sets of unaudited pro forma financial statements to show the impact of the Mergers and the REIT Conversion assuming the following four scenarios:

- All Partnerships participate and no Notes are issued ("100% Participation with No Notes Issued")
- . All Partnerships participate with Notes issued with respect to 100% of the OP Units allocable to each Partnership ("100% Participation with Notes Issued")
- . No Partnerships participate ("No Partnership Participation")
- . REIT 2000 Scenario (see below)

There is no minimum condition to participation in the Mergers and the Operating Partnership does not believe that the presentation of additional scenarios is relevant to investors or required. These presentations do not purport to represent what combination will result from the Mergers and the REIT Conversion, but instead are designed to illustrate what the composition would have been under the above scenarios. Furthermore, the unaudited proforma financial statements do not purport to represent what the results of operations or cash flows would actually have been if the Mergers and the REIT Conversion had in fact occurred on such date or at the beginning of such period or to project the results of operations or cash flows for any future date or period.

Host intends to use its best efforts to cause the REIT Conversion to be completed as soon as possible, but there is no assurance that it will be completed during 1998 in time for Host REIT to elect REIT status effective January 1, 1999, or at all. If the REIT Conversion does not occur in 1998, but the conditions to the REIT Conversion are otherwise satisfied, the effectiveness of Host REIT's election could be delayed to January 1, 2000, which could cause the Blackstone Acquisition (which is conditioned, among other things, on consummation of the REIT Conversion by March 31, 1999 and Host REIT qualifying as a REIT for 1999) not to be consummated. Accordingly, a separate column (the "REIT 2000 Pro Forma") has been included for each presentation that assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur and Host does not become a REIT until January 1, 2000.

The unaudited pro forma financial statements are based upon available information and upon certain assumptions, as set forth in the notes to the unaudited pro forma financial statements, that the Operating Partnership believes are reasonable under the circumstances. Rental revenue is recognized only for Leases to be entered at or prior to completion of the REIT Conversion.

The unaudited pro forma statements of operations of the Operating Partnership reflect the following transactions for the First Three Quarters 1998 and the fiscal year ended January 2, 1998 as if such transactions had been completed at the beginning of the fiscal year:

Acquisitions, Dispositions and Other Activities

- . Blackstone Acquisition
- . 1998 Senior Note Refinancing
- 1998 acquisition of, or purchase of controlling interests in, eleven full-service properties

- . 1998 purchase of minority interests in two full-service hotels
- . 1998 disposition of two full-service properties
- . 1997 acquisition of, or purchase of controlling interests in, 18 full-service properties  $\,$
- . 1997 refinancing or repayment of mortgage debt for three full-service properties

#### **REIT Conversion Activities**

- 1998 deconsolidation of the assets and liabilities contributed to the Non-Controlled Subsidiary, including the sale of certain furniture and equipment to the Non-Controlled Subsidiary
- . 1998 Mergers
- . 1998 acquisition of minority interests in four private Partnerships in exchange for OP Units
- . 1998 lease of certain hotel properties to Crestline and conversion of revenues and certain operating expenses to rental income
- . 1998 adjustment to remove deferred taxes resulting from the change in tax status related to the REIT Conversion
- . 1998 Special Dividend
- . 1998 sale of an investment in a joint venture to Crestline

The unaudited pro forma balance sheet as of September 11, 1998 reflects all of the above 1998 transactions except for the acquisition of, or purchase of controlling interests in, 11 full-service properties the 1998 Senior Note Refinancing, the contribution of notes receivable to Crestline and the disposition of two full-service properties which occurred prior to September 11, 1998, and were already reflected in the historical balance sheet. The unaudited pro forma balance sheet as of September 11, 1998 also reflects (a) the distribution of the Initial E&P Distribution (which includes the Crestline common stock) to Host stockholders, which Host intends to declare if the Agreement is approved by Host stockholders at the Special Meeting and the Board of Directors of Host determines that the conditions to the Merger have been or likely will be satisfied or waived (and, in particular, that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur), and (b) the Leases even though the Merger and other transactions comprising the REIT Conversion may be delayed or possibly never consummated.

The assumptions regarding the number and identity of participating Partnerships, the number of OP Units to be issued and price per OP Unit are outside the control of Host Marriott and have been made for illustrative purposes only. The unaudited pro forma financial statements and accompanying notes should be read in conjunction with the historical consolidated financial statements of the Host Marriott and "Management's Discussion and Analysis of Financial Condition and Results of Operations" contained elsewhere herein.

The Emerging Issues Task Force (EITF) reached a concensus in May 1998 on Issue 98-9, "Accounting for Contingent Rents in Interim Financial Periods" ("EITF 98-9"). EITF 98-9 requires a lessor to defer recognition of contingent rental income in interim periods until the specified target that triggers the contingent rental income is achieved. The accompanying pro forma financial statements reflect the application of EITF 98-9 to the interim periods. EITF 98-9 will have no impact on the full-year rental income recorded by the Operating Partnership.

## UNAUDITED PRO FORMA BALANCE SHEET SEPTEMBER 11, 1998 100% PARTICIPATION WITH NO NOTES ISSUED (IN MILLIONS, EXCEPT OP UNITS AMOUNTS)

# ACQUISITIONS

DISPOSITIONS AND OTHER
ACTIVITIES
В

	HOST MARRIOTT CORPORATION HISTORICAL	A DISTRI- BUTION ADJUST- MENTS	HOST MARRIOTT HOTELS HISTORICAL	B BLACK- STONE ACQUI- SITION	ACQUIS- ITIONS	DEBT REPAYMENT & REFINANCING
ASSETS						
Property and equipment,						
net Notes and other	\$5,937	\$(649)	\$5,288	\$1,450	\$	\$
receivables, net	32	(3)	29	63		
Due from managers Investments in	88	(5)	83	5		
affiliates	18		18			
Other assets	319	2	321			
for working capital Cash, cash equivalents and short-term						
marketable		>				
securities	575 	(27)	548	(262)		
	\$6,969	\$(682)	\$6,287	\$1,256	\$	\$
LIABILITIES AND EQUITY	=====	=====	=====	=====	====	====
Debt(I) Convertible debt	\$4,224	\$(213)	\$4,011	\$ 600	\$	\$
obligation to Host Marriott Corporation Accounts payable and		567	567			
accrued expenses	70	(23)	47			
Deferred income taxes Other liabilities	526 447	(62) (10)	464 437			
Total liabilities Convertible Preferred	5,267	259	5,526	600		
Securities Limited Partner interests of third parties at redemption value (on a pro forma basis 70.1 million OP Units	550	(550)				
outstanding)(J) Equity General Partner (on a				656		
pro forma basis .2 million OP Units outstanding)(J) Limited Partner interests of Host REIT (on a pro forma basis 204.3 million OP Units outstanding)(J)		(391)	761			
	\$6,969	\$(682)	\$6,287	\$1,256	\$	\$
	=====	=====	=====	=====	====	====
Book value per OP Unit						

## MERGERS AND REIT CONVERSION ACTIVITIES

	С	D	E	F	K	G	Н		L
				EARNINGS			DEFERRED		REIT
	NON-		PRIVATE	& PROFITS	CONTRI-	LEASE	TAX		2000
	CONTROLLED		PARTNER-	DISTRIBU-	BUTION TO	CONVER-	ADJUST-	PR0	PR0
	SUBSIDIARIES	MERGERS	SHIPS	TION(1)	CRESTLINE	SION	MENT	FORMA	FORMA
ASSETS									
Property and equipment,									
net	\$(290)	\$560	\$ 61	\$	\$	\$	\$	\$7,069	\$5,619
Notes and other									
receivables, net	214	(3)						303	240
Due from managers	(1)	12				(85)		14	14
Investments in									
affiliates	9							27	27
Other assets	5	23	(11)					338	338

Receivable from Lessee for working capital Cash, cash equivalents and short-term						85		85	80
marketable securities	(8)	7	(11)	(150)	(15) 7			116	378
	\$ (71)	\$599	\$ 39	\$(150)	\$(8)	\$	\$	\$7,952	. ,
LIABILITIES AND EQUITY									
Debt(I) Convertible debt obligation to Host	\$ (39)	\$323	\$	\$ 75	\$	\$	\$	\$4,970	\$4,370
Marriott Corporation Accounts payable and								567	567
accrued expenses	(1)	9						55	55
Deferred income taxes	6						(195)	275	470
Other liabilities	(37)	(21)	(6)			320		693	650
Total liabilities Convertible Preferred	(71)	311	(6)	75		320	(195)	6,560	6,112
Securities Limited Partner interests of third parties at redemption value (on a pro forma basis 70.1 million OP Units									
outstanding)(J) Equity General Partner (on a pro forma basis .2 million OP Units outstanding)(J) Limited Partner interests of Host REIT (on a pro forma basis 204.3 million OP Units		288	45					989	333
outstanding)(J)				(225)	(8)	(320)	195	403	251
=	\$ (71)	\$599 = ======	\$ 39 =======	\$(150) ======	\$ (8)	\$ ======	\$ =======	. ,	\$6,696 =====
Book value per OP Unit		<b></b>	<b>_</b>	<b></b>		<b></b>			\$ 2.53
								=====	=====

See Notes to the Unaudited Pro Forma Balance Sheet.

# NOTES TO UNAUDITED PRO FORMA BALANCE SHEET 100% PARTICIPATION WITH NO NOTES ISSUED

- A. Represents the adjustment to record the spin-off of Crestline and:
- . Reduce property and equipment by \$649 million
- . Reduce receivables by \$3 million related to certain Crestline notes held by the Operating Partnership
- Reduce due from managers by \$5 million
- . Increase other assets by \$2 million
- . Reduce cash, cash equivalents and short-term marketable securities by \$27 million
- Reduce debt by \$213 million
- Reduce accounts payable and accrued expenses by \$23 million
- . Reduce deferred income taxes by \$62 million
- . Reduce other liabilities by \$10 million
- . Reduce equity by \$391 million
- . Eliminate the \$550 million Convertible Preferred Securities of Host Marriott which remain an obligation of Host REIT
- . Record the \$567 million of Convertible Debt Obligation to Host Marriott which is eliminated in consolidation on the historical financial statements of Host Marriott Corporation
- B. Represents the adjustment to record the Blackstone Acquisition of 12 full-service properties (5,520 rooms) and a mortgage note secured by a thirteenth full-service property including the issuance of 43.7 million OP Units as determined through negotiations between the Operating Partnership and Blackstone:
  - . Record property and equipment of \$1,450 million
  - . Record mortgage note receivable of \$63 million
  - . Record increase in due from managers of \$5 million
  - Record the use of cash of \$262 million
  - . Record the assumption of mortgage debt of \$600 million
  - . Record the issuance of 43.7 million OP Units with an estimated fair value of \$656 million using an assumed Host Marriott stock price of \$15.00 (which includes the value of Crestline since Blackstone will receive shares of Crestline in addition to OP Units)

The purchase price of the Blackstone properties and mortgage note was determined based on the estimated fair value of the 43.7 million OP Units to be issued. The number of units to be issued will not increase or decrease depending on the stock price of Host Marriott at the time of closing of the acquisition.

- C. Represents the adjustment to record the investment in the Non-Controlled Subsidiaries and to reflect the sale of certain hotel furniture and equipment to the Non-Controlled Subsidiary:
  - Record decrease in property and equipment of \$290 million, including \$180 million of hotel furniture and equipment sold to the Non-Controlled Subsidiaries
  - . Record receivable from Non-Controlled Subsidiaries for the furniture and equipment loan of \$180 million, and transfer of other notes totaling \$34 million
  - . Record decrease in due from managers of \$1 million
  - . Record investment in the Non-Controlled Subsidiaries of \$9 million
  - . Record increase in other assets of \$5 million
  - . Record decrease in cash of \$8 million
  - . Record decrease in debt of \$39 million of debt transferred to the Non-Controlled Subsidiaries.
  - . Record decrease in accounts payable and accrued expenses of \$1 million
  - . Record increase in deferred taxes of \$6 million
  - . Record decrease in other liabilities of \$37 million
  - D. Represents the adjustment to record the Mergers:
  - . Record property and equipment of \$560 million

- Record decrease in notes receivable of \$3 million
- Record increase in due from managers of \$12 million
- Record other assets of \$23 million
- Record cash of \$7 million Record debt of \$323 million
- Record accounts payable and accrued expenses of \$9 million
- Record decrease in other liabilities of \$21 million
- Record the issuance of 23.0 million OP Units totaling approximately \$288 million

The purchase price and number of OP Units expected to be issued to the limited partners of each Partnership is (in millions, except OP Units in thousands):

			INCREASE TO PROPERTY AND EQUIPMENT
			AND EQUITIENT
Atlanta Marquis		1,921	\$ 24
Desert Springs	37	2,943	36
Hanover	5	434	5
MHP	73	5,832	54
MHP II	84	6,684	78
Chicago Suites	11	888	38
MDAH	45	3,595	162
PHLP	9	725	163
	\$288	23,022	\$560
	====	=====	====

The number of OP Units was determined based on the purchase price and an estimated price of an OP Unit of \$12.50 which is based upon the recent trading range of Host Marriott Corporation's stock as adjusted for the proposed dividend of Crestline to its shareholders. The purchase price was determined based on the fair market value of the net assets to be acquired.

The purchase price for minority interests (Atlanta Marquis, Desert Springs, Hanover, MHP and MHP2) was allocated to property to the extent that the purchase price exceeded the minority interest liability recorded. The purchase price for the other three partnerships that are presently not consolidated was allocated in accordance with APB Opinion No. 16 with the debt of each partnership recorded at estimated fair value, all assets and liabilities, except for property being recorded at historical carrying values of each partnership with the residual allocated to property. The amounts allocated to property are in all cases less than estimated current replacement cost.

- E. Represents the adjustment to record the purchase of the remaining minority interests in four Private Partnerships:
  - Record property and equipment of \$61 million
  - Record decrease in other assets of \$11 million
  - Record use of cash of \$11 million
  - Record decrease in minority interest liabilities of \$6 million
  - Record the issuance of 3.6 million OP Units totaling approximately \$45 million
- F. Represents the estimated \$225 million cash payment of the Special Dividend to shareholders of Host Marriott, which when combined with the value of the Crestline common stock (estimated to be approximately \$1.30 per share for a total Initial E&P Distribution of approximately \$2.20 per share) will represent the Initial E&P Distribution, by a draw on the New Credit Facility of \$75 million and available cash of \$150 million. The aggregate value of the Crestline common stock and the cash or other consideration to be distributed to Host stockholders (and the Blackstone Entities) in connection with the Initial E&P Distribution is currently estimated to be approximately \$525 million to \$625 million of which approximately \$200 to \$300 million is expected to be represented by the Special Dividend.(/1/)
- G. Represents the adjustment to record the transfer of working capital to Crestline related to the leasing of the Operating Partnership's hotels by decreasing working capital and recording a receivable from the lessee of \$85 million and the adjustment to record deferred revenue of \$320 million in connection with the application of EITF 98-9 to the Operating Partnership's rental income.

- H. Represents the adjustment to record the effect on deferred taxes for the change in tax status resulting from the REIT Conversion by decreasing deferred taxes and increasing equity by \$195 million.
- I. The Operating Partnership's pro forma aggregate debt maturities at September 11, 1998, excluding \$8 million of capital lease obligations and the \$8 million debt discount recorded in conjunction with the Senior Notes Refinancing, are (in millions):

1998 1999																									
2000																									
2001	 																								99
2002	 ٠.		 		٠.							٠.												1	.50
Thereafter	 ٠.	٠.	 																				4	, 5	84
																							\$5	, 5	37

J. The number of OP Units includes the following (in millions):

Limited	Partner	interests	of	Host	REIT			 	 	 	 204.3
General	Partner	interests	of	Host	REIT			 	 	 	 0.2
Limited	Partner	interests	of	Partn	ership	s		 	 	 	 23.0
Limited	Partner	interests	of	Priva	te Par	tners	hips	 	 	 	 3.4
Limited	Partner	interests	of	Black	stone	Group		 	 		
Total	OP Units	3						 	 	 	 274.6
											=====

- K. Represents the adjustment to record the contribution of \$15 million in cash to Crestline as a reduction in equity and to record the sale of an investment of approximately \$7 million in a joint venture which holds a mortgage note of approximately \$130 million from a consolidated subsidiary of Host.
- L. The "REIT 2000 Pro Forma" assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur, Host does not become a REIT until January 1, 2000 and no Notes issued. The amounts reflect the "Pro Forma" column less the amounts in column B--"Blackstone Acquisition" and the reduction in other liabilities and increase in equity for the deferred revenue of \$43 million related to the application of EITF 98-9 to the Blackstone properties.
- (1) The amount of earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

# UNAUDITED PRO FORMA STATEMENT OF OPERATIONS FIRST THREE QUARTERS 1998 100% PARTICIPATION WITH NO NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AMOUNTS AND RATIOS)

ACQUISITIONS, DISPOSITIONS AND OTHER ACTIVITIES

B C E G

		Α		В		С	E		G	
	HOST MARRIOTT CORPORATION HISTORICAL	DISTRIBUTION ADJUSTMENT	HOST MARRIOTT HOTELS HISTORICAL	BLACKSTONE ACQUISITION	AC	1998 QUISI- TIONS	DISPOSITION	S R	BOND EFINANCING	-
REVENUE Rental revenues	\$	\$	\$	\$	\$		\$		\$	
Hotel revenues Equity in earnings (losses) of	922	 	922		122	43		(6)	Ф	
affiliates Other revenues	1 117	 (58)	1 59					 (50)		
Total revenues	1,040	(58) 	982		122	43		(56)		
OPERATING COSTS AND EXPENSES										
Hotels Other	502 45	(30)	502 15		70 	23 		(3)		 
Total operating costs and										
expenses	547 	(30)	517 		70 	23		(3)		
OPERATING PROFIT	493	(28)	465		52	20		(53)		
Minority interest Corporate	(36)		(36)			(1	)	1		
expenses REIT Conversion	(33)	3	(30)							
expenses Interest	(14)		(14)		(26)		`			(17)
expense Dividends on Convertible Preferred	(245)	(8)	(253)		(36)	(1	)	1	1	(17)
Securities Interest	(26)	26								
income	36 	2	38 		(6) 	(16	)	(1)		
Income (loss) before income taxes Benefit	175	(5)	170		10	2		(52)		(17)
(provision) for income taxes	(75)	3	(72)		(4)	(1	)	21		7
Income (loss) before extraordinary items	\$ 100	\$ (2)	\$ 98	\$	6 \$	1	\$	(31)	\$	(10)
Basic loss per OP Unit	====	====	====	======		=======			======	` '
Ratio of earnings to fixed charges	1.7x =====		1.8x =====							
Deficiency of earnings to fixed charges										
		MERGEF	RS AND REIT	CONVERSION						
	Н	J F	(	L	P/N	I/M	0		Q REIT	
		PRI\ MERGERS PARTNE	/ATE ERSHIPS DIS			T CONVE	R- TAX	PRO FORMA	2000 PRO FORMA	
REVENUE Rental revenues	\$	\$ \$ -	- <b>-</b>	\$	\$	\$ 5	40 \$	\$ 540	\$ 468	
Hotel revenues Equity in earnings	(14)	56 -			φ					

(losses) of affiliates Other revenues	(3) (5)	 		 		 	 	(2) 4	(2)
Total revenues	(22)	56				(583)		542	470
OPERATING COSTS AND EXPENSES Hotels	(12)	36	2			(224)	  	394 15	336 15
Total operating costs and expenses	(12)	36	2			(224)		409	351
OPERATING PROFIT Minority	(10)	20	(2)			(359)		133	119
interest Corporate	3	18	1					(14)	(14)
expenses REIT Conversion								(30)	(30)
expenses Interest					14				
expense Dividends on Convertible Preferred	3	(21)		(4)				(328)	(292)
Securities Interest									
income	2	1		(5)	(2)	13		24	30
Income (loss) before income taxes Benefit (provision) for income taxes	(2)	18	(1)	(9)	12	(346)	 (71)	(215)	(187)
Income (loss) before extraordinary items	\$ (1)	\$ 11	\$ (1)	\$ (5)	\$ 7	\$ (208)	\$ (71)	\$(204)	
Basic loss per OP Unit	=======	====== =	:======== :		=======	=======	=======	\$(.74) ======	\$(.77)
Ratio of earnings to fixed charges								N/A	N/A
Deficiency of earnings to fixed charges								\$(196) =====	\$(170)

See Notes to the Unaudited Pro Forma Statements of Operations.

# UNAUDITED PRO FORMA STATEMENT OF OPERATIONS FISCAL YEAR 1997 100% PARTICIPATION WITH NO NOTES ISSUED

(IN MILLIONS, EXCEPT PER OP UNIT AMOUNTS AND RATIOS)

				ACQUIS	ITIONS,	DISPOSITIO	NS AND OTHER	R ACTIVITIE	ES	
	HOST MARRIOTT CORPORATION HISTORICAL	A DISTRIBUTION ADJUST- MENT	HOST MARRIOTT HOTELS HISTORICAL	B BLACKSTO		C 1998 DUISITIONS	D 1997 ACQUISITIONS	E S DISPOSIT:	IONS	
REVENUE Rental revenues	\$	\$	\$	\$		\$	\$	\$		
Hotel revenues Equity in earnings of affiliates	1,093		1,093	148	3	112	89	(23)	)	
Other revenues	5 49	(37)	5 12							
Total revenues	1,147	(37)	1,110	148		112	89	(23)	)	
OPERATING COSTS AND EXPENSES										
Hotels Other	649 49	(20)	649 29	101 	L	62 	42 	(10)	)	
Total operating										
costs and expenses	698	(20)	678	101		62	42	(10)	)	
OPERATING PROFIT Minority	449	(17)	432	47	7	50	47	(13)	)	
interest	(32)		(32)			(4)	5	(1)	)	
Corporate expenses	(47)	2	(45)							
Interest expense Dividends on Convertible	(302)	(23)	(325)	(48	3)	(12)	(12)	3		
Preferred Securities Interest	(37)	37								
income	52		52	(7	7)	(14)	(14)			
Income (loss) before income taxes Benefit	83	(1)	82	(8	3)	20	26	(11)	)	
(provision) for income taxes	(36)	1	(35)	3	3	(8)	(10)	4		
Income (loss) before extraordinary					•					
items	\$ 47 =====	\$ =====	\$ 47 =====	\$ (! ====	•	\$ 12 =====	\$ 16 =====	\$ (7) =====	)	
Basic earnings per OP Unit Ratio of earnings to										
fixed charges	1.3x =====		1.3x =====							
		MERO	GERS AND RE	IT CONVER	RSION AC	CTIVITIES				
	F/G	н	 J	 К			I/M	0		Q
	DEBT	NON-			ARNINGS	3	LEASE	INCOME		REIT 2000
	REPAYMENT	CONTROLLED NG SUBSIDIARY	PA MERGERS S	RTNER- [	DISTRI- JTION(1)	OTHER REI ACTIVITIE	T CONVER-	TAX ADJUSTMENT	PRO FORMA	PRO FORMA
REVENUE										
Rental revenues Hotel revenues Equity in	\$ 	\$ (17)			\$	\$ 	\$ 1,135 (1,476)		\$1,135 	\$998 
earnings of affiliates Other revenues		(12) (9)							(7) 3	(7) 3
Total revenues		(38)					(341)		1,131	994
OPERATING COSTS AND EXPENSES										

ACQUISITIONS, DISPOSITIONS AND OTHER ACTIVITIES

Hotels Other	 	(9) (18)	51 	2	 		(288)		600 11	513 11
Total operating costs and expenses		(27)	51	2			(288)		611	524
OPERATING PROFIT Minority		(11)	23	(2)			(53)		520	470
interest Corporate		4	17	1					(10)	(10)
expenses Interest		1							(44)	(44)
expense Dividends on Convertible Preferred	(48)	5	(25)		(6)				(468)	(420)
Securities Interest										
income	(3)	4	1		(8)		18		29	36
Income (loss) before income taxes Benefit (provision) for	(51)	3	16	(1)	(14)		(35)		27	32
income taxes	20	(3)	(6)		6		14	14	(1)	(2)
Income (loss) before extraordinary items	\$ (31)	\$	\$ 10	\$ (1)	\$ (8)	\$	\$ (21)	\$ 14	\$ 26	\$ 30
Basic earnings per OP Unit	=========	=======	====== :	======	=======================================	======	= ====== =	======	\$ .09	\$.13 ======
Ratio of earnings to fixed charges									1.1x ======	1.1x

See Notes to the Unaudited Pro Forma Statements of Operations.

# NOTES TO UNAUDITED PRO FORMA STATEMENTS OF OPERATIONS ASSUMING 100% PARTICIPATION WITH NO NOTES ISSUED

- A. Represents the adjustment to reduce revenues, operating expenses, corporate expenses, interest expense, interest income and income taxes for the spin-off of Crestline.
- B. Represents the adjustment to record the historical revenues, operating expenses, interest expense, income taxes and to reduce interest income associated with the acquisition of the equity and debt interests for the Blackstone Acquisition.
- C. Represents the adjustment to record the historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income associated with the 1998 acquisition of, or purchase of controlling interests in 11 full-service properties.
- D. Represents the adjustment to record the historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income associated with the 1997 acquisition of, or purchase of controlling interests in, 18 full-service properties.
- E. Represents the adjustment to record historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income for the 1998 sale of the New York Marriott East Side and the Napa Valley Marriott, including the elimination of the non-recurring gains on the sales totalling \$50 million and related taxes of \$20 million in 1998.
- F. Represents the adjustment to reduce the interest expense, interest income and to record income taxes associated with the refinancing or payoff of mortgage debt for three full-service properties (Marriott's Orlando World Center, the Philadelphia Marriott and the San Francisco Marriott).
- G. Represents the adjustment to record interest expense and related amortization of deferred financing fees, reduce interest income, and to record income taxes as a result of the Senior Note Refinancing. The adjustment excludes the extraordinary loss of \$148 million, net of taxes, related to the Senior Note Refinancing resulting from the write-off of deferred financing fees and the payment of bond tender and consent fees.
- H. Represents the adjustment for revenues, operating expenses, minority interest, interest expense, corporate expenses, income taxes and interest income to deconsolidate the Non-Controlled Subsidiary and reflect the Operating Partnership's share of income as equity in earnings of affiliate.
- I. Represents the adjustment to reduce depreciation expense of \$18 million and \$26 million for First Three Quarters 1998 and fiscal year 1997 related to certain furniture and equipment sold to the Non-Controlled Subsidiary, record interest income of \$9 million and \$13 million for First Three Quarters 1998 and fiscal year 1997 earned on the 7%, \$180 million in notes from the Non-Controlled Subsidiaries and reduce the lease payment to the Operating Partnership from the Lessee.
- J. Represents the adjustment to record the historical revenues, operating expenses, minority interest, interest expense, interest income and income taxes associated with the Mergers, including three partnerships not previously consolidated by the Operating Partnership.
- K. Represents the adjustment to record additional depreciation expense and the decrease in minority interest expense related to the purchase of the remaining minority interests in the Private Partnerships.
- L. Represents the adjustment to reduce interest income, record interest expense and income taxes for the estimated \$225 million cash payment of the Special Dividend to shareholders of Host Marriott including a draw on the New Credit Facility of \$75 million.(1)

M. Represents the adjustment to remove hotel revenues, management fees and other expenses of \$224 million and \$288 million, respectively, for First Three Quarters 1998 and fiscal year 1997, and to record rental revenues associated with the leasing of certain hotel properties to Crestline and other lessees and interest income of \$3 million and \$5 million for First Three Quarters 1998 and fiscal year 1997 earned on the 6%, \$85 million in notes from Crestline. First Three Quarters 1998 included a \$320 million reduction to rental income to record deferred revenue for percentage rents in accordance with EITF 98-9. Management believes the change to the lease structure described above will not impact hotel operating results because the hotel manager and asset management function will remain unchanged. Rental revenues under the Leases are based on the greater of Percentage Rent or Minimum Rent. Total rent in the pro forma statement of operations is calculated based on the historical gross sales of the property and the negotiated rental rates and thresholds by property as if the leases were entered into on the first day of fiscal year 1997. There are generally three sales categories utilized in the rent calculation: rooms, food and beverage and other. For rooms and food and beverage, there are three tiers of rent with two thresholds, while the other category generally has two tiers of rent and one threshold. The percentage rent thresholds are increased annually on the first day of each year after the initial lease year based on a blended increase of the Consumer Price Index ("CPI") and a wage and benefit index. For purposes of the pro formas, 1997 is the assumed initial lease year and the blended increase applied to the thresholds at January 3, 1998 is assumed to be 3%. Minimum rent is expressed as a fixed dollar amount that increases annually on the first day of each year after the initial lease year as 50% of the CPI increase. Accordingly, the 1998 rent thresholds and minimum rent included in the pro formas were adjusted as of January 3, 1998 for the 1997 increases in the indices. Rental revenues is recognized only for leases to be executed with Crestline at or prior to completion of the REIT Conversion. The execution of the leases is dependent upon the distribution of the Crestline common stock to the Host's stockholders, which is subject to contingencies that are outside the control of the Operating Partnership, including consent of shareholders, lenders, debt holders, partners and ground lessors of Host. The Operating Partnership believes that negotiations with third parties to complete the REIT Conversion will not result in any material change to the leases. The table below details gross sales, minimum rent and total rent for all full-service properties to be leased and summarized amounts for the limited-service properties to be subleased:

FISCAL YEAR 1997 FIRST THREE QUARTERS 1998

PROPERTY	GROSS SALES	RENT	TOTAL RENT	GROSS SALES	MINIMUM RENT	RENT	TOTAL RENT RECOGNIZED
			(IN MI	LLIONS)			
Grand Hotel Resort and Golf							
Club	\$23.4	\$2.8	\$4.2	\$18.0	\$2.0	\$3.7	\$2.0
Scottsdale Suites	11.9	3.0	5.0	8.2	2.1	3.4	2.1
The Ritz-Carlton, Phoenix	23.3	4.6	7.2	17.3	3.2	5.5	3.2
Coronado Island Resort	22.0	2.1	2.1	16.2	1.5	3.6	1.5
Costa Mesa Suites	9.7	1.9	3.3	7.2	1.3	2.3	1.3
Desert Springs Resort and							
Spa	103.3	21.3	30.3	80.3	15.0	22.6	15.0
Manhattan Beach	16.3	2.4	4.8	12.2	1.7	3.6	1.7
Marina Beach	21.1	4.6	7.1	16.9	3.2	6.2	3.2
Newport Beach	33.5	5.5	8.7	24.0	3.9	6.8	3.9
Newport Beach Suites	11.0	2.6	4.1	8.0	1.8	3.0	1.8
Ontario Airport San Diego Marriott Hotel	12.1	1.8	3.4	8.3	1.3	2.2	1.3
and Marina	103.3	38.0	39.6	78.6	26.7	31.1	26.7
San Diego Mission Valley	16.7	3.4	5.1	12.6	2.4	5.6	2.5
San Francisco Airport	43.8	8.2	13.2	32.2	5.8	9.5	5.8
San Francisco Fisherman's							
Wharf	17.8	4.0	6.4	12.1	2.8	4.3	2.8
San Francisco Moscone							
Center	120.2	20.7	37.9	90.5	14.6	28.5	14.6
San Ramon	19.7	4.4	5.1	14.4	3.1	4.0	3.1
Santa Clara	47.3	7.8	16.5	37.2	5.5	13.5	8.8
The Ritz-Carlton, Marina							
del Rey	32.4	5.5	10.8	23.4	3.9	7.9	3.9
The Ritz-Carlton, San							
Francisco	50.1	9.6	14.7	34.2	6.7	10.3	6.7
Torrance	20.5	2.3	3.5	15.0	1.6	5.1	1.6
Denver Southeast	21.5	3.0	6.2	14.9	2.1	4.1	2.1
Denver Tech Center	26.8	5.1	8.3	20.1	3.6	6.0	3.6
Denver West	13.7	1.8	4.0	9.6	1.2	2.7	1.7
Marriott's Mountain Resort	47.0	0.0			0.4	4 -	0.4
at Vail	17.6	3.0	5.1	14.1	2.1	4.5	2.1
Hartford/Farmington	18.4	3.5	4.7	13.4	2.4	3.5	2.4
Hartford/Rocky Hill	11.6	1.5	2.7	8.5	1.1	2.0	1.1
Fort Lauderdale Marina	28.5	4.3	7.9	20.4	3.0	5.7	3.2
Harbor Beach Resort	58.1	16.5	19.3	43.2	11.6	14.0	11.6
Jacksonville	11.8	1.8	3.6	8.0	1.2	2.4	1.2
Miami Airport	29.7	3.9	8.4	21.6	2.8 16.5	6.1	2.9
Orlando World Center  Palm Beach Gardens	128.2	23.5 1.9	39.6 3.7	98.7 8.5	16.5	30.4	18.3 1.4
Singer Island (Holiday	11.8	1.9	3.1	0.5	1.4	3.0	1.4
Strider Istalia (Hottada							

Inn)	6.6	1 4	2 5	5 2	1 0	2 1	1.0
Tampa Airport							
Tampa Westshore							

				1113			
PROPERTY	GROSS SALES	MINIMUM RENT	RENT	GROSS SALES	MINIMUM RENT	TOTAL RENT	TOTAL RENT RECOGNIZED
			(IN MILI	_IONS)			
The Ritz-Carlton, Naples			\$ 23.3			\$ 18.0	
Atlanta Marriott Marquis	85.4		33.3	58.6	15.0	25.6	
Atlanta Midtown Suites	10.5		3.5	7.8	1.3	2.6	1.3
Atlanta Norcross	7.6	1.0	1.7	5.6	0.7	1.2	0.7
Atlanta Northwest	14.9	2.7	4.3	11.3	1.9	3.3	1.9
Atlanta Perimeter	16.6	2.5	4.5	12.6	1.7	3.5	1.7
JW Marriott Hotel at							
LenoxThe Ritz-Carlton,	24.8	3.7	6.8	17.7	2.6	5.0	2.6
Atlanta The Ritz-Carlton,	30.2	5.8	8.8	21.7	4.1	6.8	4.1
Buckhead	49.3	13.1	16.3	35.8	9.2	11.7	9.2
Chicago/Deerfield Suites			3.1	7.4	1.3	2.3	
Chicago/Downers Grove	10.2						
Suites Chicago/Downtown	9.0	1.8	2.9	6.7	1.3	2.2	1.3
Courtyard	16.3	3.1	4.9	12.2	2.2	3.9	2.2
Chicago O'Hare	40.0		11.5	28.8	3.9	8.2	
South Bend	9.9		2.1	7.0	0.8	1.5	
New Orleans	66.4		21.8	47.6	12.3	15.8	
Bethesda	23.2		5.6	17.3	2.2	4.1	
Gaithersburg/Washingtonian							
Center	13.2		3.8	9.7	1.7	2.8	
Boston/Newton	27.4		7.8	19.1		5.5	
Detroit Romulus The Ritz-Carlton,	8.8	1.1	1.8	6.6	0.8	1.4	0.8
Dearborn	25.7	3.6	5.5	17.7	2.5	4.0	2.5
Minneapolis/Bloomington	20.2	3.3	6.5	13.8	2.3	4.7	3.1
Minneapolis City Center	27.5		7.5	20.4	2.4	5.2	
Minneapolis Southwest	14.9		4.8	10.1		4.0	
Kansas City Airport	14.3		3.7	9.9	1.2	2.5	
St. Louis Pavilion	27.5		6.5	18.5	4.3	4.3	
Nashua	7.5		1.3	5.3		0.9	
Hanover	22.5		6.6	15.1		4.3	
Newark Airport	39.4		11.8	29.2		8.6	
Park Ridge	16.0		4.0	11.9	1.7	4.2	
Saddle Brook	10.7		2.1	7.8	0.9	1.7	
Albany New York Marriott	18.5	3.5	6.1	12.4	2.5	5.2	
Financial Center New York Marriott	39.6	7.7	13.2	29.1	5.4	10.1	5.4
Marquis Marriott World Trade	210.3	40.0	60.8	155.4	29.7	47.6	29.7
Center	65.4	12.2	19.4	49.1	8.6	14.9	8.6
Charlotte Executive Park	14.0	2.3	3.7	9.8	1.6	2.6	1.6
Raleigh Crabtree Valley	14.9		3.9	10.9	1.7	2.8	
Oklahoma City	15.6		3.8	10.4	1.4	2.4	
Oklahoma City Waterford	9.1		2.7	6.1		1.7	
Portland	26.4		7.5	17.6	2.9	4.8	
Philadelphia (Convention							
Center)	80.7	14.2	25.0	58.2	10.0	17.8	10.0
Philadelphia Airport	25.0	4.1	7.6	18.6	2.9	5.6	2.9
Pittsburgh City Center	16.4	1.9	3.0	11.1	1.3	2.2	1.3
Memphis	10.6		3.2	5.7	1.0	1.8	
Dallas/Fort Worth	28.9		9.3	21.9	4.1	7.0	
Dallas Quorum	25.7		8.2	18.3	3.0	5.8	
El Paso	11.6		2.3	7.8	0.6	1.4	
Houston Airport	21.6		6.0	16.9	2.0	4.6	
JW Marriott Houston	27.2		8.0	20.1	3.5	5.9	
Plaza San Antonio							
	13.8		4.6	9.7	2.0	3.3	
San Antonio Rivercenter	68.9		24.5	49.3	12.3	17.8	
San Antonio Riverwalk	29.3		10.3	21.7	4.3	7.6	
Salt Lake City	28.5		9.5	21.1	3.9	7.2	
Dulles Airport	14.6		4.0	10.9	1.2	3.0	
Key Bridge	29.4	5.6	10.2	21.2	3.9	7.4	3.9
Norfolk Waterside Pentagon City Residence	18.1	3.3	5.4	12.8	2.4	3.8	2.4
Inn The Ritz-Carlton, Tysons	11.7	3.5	5.5	8.7	2.5	4.2	2.5
Corner	34.4	5.9	9.8	24.9	4.1	7.3	4.1
Washington Dulles Suites	10.3		4.0	7.8	1.8	3.0	

FISCAL YEAR 1997 FIRST THREE QUARTERS 1998

	FISCAL YEAR 1997			FIRST THREE QUARTERS 1998			
PROPERTY	GROSS SALES	MINIMUM RENT	TOTAL RENT	GROSS SALES	MINIMUM RENT	TOTAL RENT	TOTAL RENT RECOGNIZED
				MILLIONS	 )		
			(=		,		
Westfields Williamsburg Washington Metro	\$ 28.0 12.6	1.8	\$ 7.4 2.8	\$ 20.3 9.3	1.3	\$ 5.4 2.1	\$ 3.3 1.3
Center	25.2		7.3	19.2		5.3	3.2
Calgary Toronto Airport	13.4 17.1		1.7 5.6	9.8 13.0		2.3 4.2	1.2 2.0
Toronto Eaton Centre Toronto Delta	21.1	6.1	7.1	16.0	4.3	5.6	4.3
Meadowvale	16.1		4.9	10.6		3.1	1.9
Fairview Park	22.5 18.2		7.3 6.0	16.3 13.4		5.2 4.3	2.8 2.3
Dayton Research Triangle Park	9.1		2.9	6.8		2.3	1.0
Detroit Marriott Southfield	8.8		2.1	6.9		1.7	0.9
Detroit Marriott							
Livonia	10.0		2.6	7.4		1.9	1.0
Fullerton	6.8		1.8	5.0		1.3	0.8
Marriott O'Hare Suites Albuquerque	14.4 16.4		4.9 3.6	10.8 11.1		4.0 2.6	1.9 2.5
Greensboro-High Point	13.6		3.3	10.2		2.4	2.3
Houston Medical Center	16.5	4.0	4.0	12.2		2.9	2.8
Miami Biscayne Bay Marriott Mountain	26.8		6.6	20.5		5.1	4.5
Shadows Resort	24.1		4.5	16.9		3.1	3.1
Seattle SeaTac Airport Four Seasons, Atlanta(4)	23.1 15.6		6.7 5.9	17.5 14.2		5.1 4.5	4.7 4.1
Four Seasons,	15.0	3.0	5.9	14.2	4.1	4.5	4.1
Philadelphia(4) Grand Hyatt,	41.1	7.9	12.4	30.6	5.6	10.1	5.6
Atlanta(4) Hyatt Regency,	25.3		10.0	22.6		8.2	7.0
Burlingame(4) Hyatt Regency,	47.9	8.8	17.6	39.5	6.2	15.1	9.0
Cambridge(4) Hyatt Regency,	32.4	6.7	11.9	26.8	4.7	10.4	6.1
Reston(4)	30.5		11.3	24.2		9.2	4.8
Swissotel, Atlanta(4)	22.2		6.3	17.2		5.8	3.5
Swissotel, Boston(4)	26.8		8.5	20.5		6.9	4.5
Swissotel, Chicago(4) The Drake (Swissotel), New York(4)	38.1		15.1 13.6	28.9		12.0	7.7 8.2
The Ritz-Carlton, Amelia	45.7		13.4	37.4		11.1	7.2
Island(4) The Ritz-Carlton,							
Boston(4) Non-Controlled	40.1		10.5	31.4		8.8	4.8
Subsidiary Rent		(39.7)	(39.7)		(27.5)	(27.5)	(27.5)
Total Full-service		070.0	4 055 0	0 074 0	477.0	242.2	404 0
Properties Total Courtyard	3,600.8		1,055.9	2,671.0		810.8	491.2
Properties Total Residence Inns	212.0 69.9	17.2	59.2 20.3	159.2 50.6	12.0	36.8 12.3	36.8 12.3
Total	\$3,882.7 ======	\$744.0	\$1,135.4 ======	\$2,880.8 ======	\$524.0	859.9	\$540.3 =====
Less: Deferred rent under EITF 98-9	<b></b>	<b></b>	<b>_</b>	<b></b>	<b></b>	(319.6)	
Total rant							
Total rent recognized						\$ 540.3	

N. Represents the adjustment to eliminate interest income recorded for the \$92 million note receivable contributed to Crestline for the First Three Quarters 1998.

======

- O. Represents the adjustment to the income tax provision to reflect the REIT Conversion.
- P. Represents the adjustment to eliminate non-recurring expenses incurred in connection with the REIT Conversion. Management expects that the total estimated non-recurring expenses to be incurred will be approximately \$50 million.
- Q. The "REIT 2000 Pro Forma" reflects the adjustment to eliminate the revenues, operating expenses, interest expense and interest income which assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur, Host does not become REIT until January 1, 2000

and no Notes are issued.

- -----
- (1) The amount of earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

### UNAUDITED PRO FORMA STATEMENTS OF CASH FLOWS

# FIRST THREE QUARTERS 1998 100% PARTICIPATION WITH NO NOTES ISSUED (IN MILLIONS)

				ACQUISITION	NS, DISPOSITIO	ONS AND OTHER	ACTIVITIES
	HOST	Α	HOST	В	С	E	G
	MARRIOTT CORPORATION HISTORICAL	DISTRIBUTION ADJUST- MENT	MARRIOTT HOTELS	BLACKSTONE ACQUISITION	1998 ACQUISITIONS	DISPOSITIONS	BOND REFINANCING
OPERATING ACTIVI- TIES							
Income before extraordinary items	\$ 100	\$ (2)	\$ 98	\$ 6	\$ 1	\$ (31)	\$ (10)
Adjustments to reconcile to cash provided by operations:							
Depreciation and amortization Income taxes Gains on sales	184 50	(17) 	167 50	42 	11 		
of hotel properties Equity (earnings)	(50)		(50)			50	
losses of affiliates Changes in operating		(1)	(1)				
accounts	(33)	(2)	(35)				
Other assets	27 	4	31				
Cash provided by (used in) operations	278	(18)	260	48	12	19	(10)
INVESTING							
ACTIVITIES Acquisitions Cash received	(636)	29	(607)		607		
from sale of assets	211		211			(211)	
Capital expenditures Purchases of short-term	(168)	9	(159)	(16)	(5)	2	
marketable securities Sales of short-term marketable	(134)		(134)				
securities Other	451 3	(89)	451 (86)	<del></del>		21	
Cash provided by (used in) investing							
activities	(273) 	(51) 	(324)	(16) 	602	(188) 	
FINANCING ACTIVITIES Issuances of							
debt Scheduled principal	2,004		2,004				1,967
repayments Debt	(42)	3	(39)				
prepayments Transfers to Host	(1,750)	119	(1,631)			35	(1,250)
Marriott Other	(189)	(63) 	(63) (189)				
Cash provided by (used in) financing							
activities	23	59 	82			35 	717
INCREASE (DECREASE) IN CASH AND CASH							
EQUIVALENTS	\$ 28 =====	\$(10) ====	\$ 18 ======	\$ 32 ====	\$614 ====	\$(134) =====	\$ 707 =====

#### MERGERS AND REIT CONVERSION ACTIVITIES Κ 0/P **EARNINGS** & PROFITS LEASE NON-PRTVATE DISTRI- CONVER- OTHER REIT TAX CONTROLLED SUBSIDIARY MERGERS PARTNERSHIPS BUTION(1) SION ACTIVITIES ADJUSTMENT FORMA OPERATING ACTIVI-TIES Income before extraordinary \$(1) \$ (204) \$(1) \$ (5) \$(208) \$ 7 \$(71) items ..... \$ 11 Adjustments to reconcile to cash provided by operations: Depreciation and amortization.... (4) 17 2 (18)217 Income taxes.... 138 188 Gains on sales of hotel properties..... Equity (earnings) losses of affiliates..... (1) Changes in operating accounts..... 320 285 Other assets.... 31 Cash provided by (used in) operations..... (5) 28 1 (5) 232 7 (71)516 INVESTING **ACTIVITIES** Acquisitions.... Cash received from sale of assets..... - -Capital expenditures.... (9) (186)Purchases of short-term marketable securities..... (134)Sales of shortterm marketable securities..... 451 Other..... (65)Cash provided by (used in) investing (9) 66 activities..... 1 FINANCING **ACTIVITIES** Issuances of (75) 3,896 debt..... Scheduled principal repayments..... (2) (15)(56)Debt (2,846)prepayments..... - -Transfers to Host Marriott..... (63)Other.....

See Notes to the Unaudited Pro Forma Statements of Cash Flows.

(75)

\$(80)

(15)

\$ 217

\$ 7

742

\$1,324

\$(71)

Cash provided by (used in) financing activities.....

INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS..... (2)

# UNAUDITED PRO FORMA STATEMENT OF CASH FLOWS FISCAL YEAR 1997 100% PARTICIPATION WITH NO NOTES ISSUED (IN MILLIONS)

				ACQU	ISITIONS, DIS	POSITIONS AND	OTHER ACTIVI	TIES
	HOST	Α	HOST	В	D	С	E	F/G
	MARRIOTT CORPORATION	DISTRIBUTION ADJUST- MENT	MARRIOTT HOTELS	BLACKSTONE ACQUISITION	1997 ACQUISITIONS	1998 ACQUISITIONS	DISPOSITIONS	DEBT REPAYMENT & REFINANCING
OPERATING ACTIV-								
ITIES Income before extraordinary Adjustments to	\$ 47	\$	\$ 47	\$ (5)	\$ 16	\$ 12	\$ (7)	\$ (31)
reconcile to cash provided by operations:								
Depreciation and amortization	240	(9)	231	61	17	31	(3)	
Income taxes	(20)		(20)					
Other Changes in	70	3	73				(2)	
operating accounts	127	(26)	101					
Cash provided by								
(used in) operations	464	(32)	432	56	33	43	(12)	(31)
operacions	404	(32)	432			43	(12)	(31)
INVESTING								
ACTIVITIES	()		()					
Acquisitions Cash received from sale of	(596)	237	(359)		359			
assets Purchase of	51		51					
short-term								
marketable securities Capital	(354)		(354)					
expenditures	(160)	2	(158)	(20)	(13)	(13)	4	
Other	13		13				1	
Cash provided by (used in) investing								
activities	(1,046)	239	(807)	(20)	346	(13)	5	
FINANCING ACTIVITIES Issuances of								
debt Scheduled principal	857		857					1,188
payments Debt	(93)	3	(90)					
prepayments Transfers to	(403)		(403)					(1,147)
Host Marriott		(226)	(226)					
Other	28	(1)	27 					
Cash provided by (used in)		_	-	-		-		
financing activities	389	(224)	165					41
INCREASE (DECREASE) IN CASH AND CASH								
EQUIVALENTS	\$ (193) ======	\$ (17) =====	\$(210) =====	\$ 36 ====	\$379 ====	\$ 30 ====	\$ (7) ====	\$ 10 =====
		MERGERS ANI	D REIT CONVE	ERSION ACTIVI	TIES			
	Н	I J	K EARNINGS			V		
	NON- CONTROLLED SUBSIDIARY	PARTNI				AX PRO STMENT FORMA		

extraordinary Adjustments to reconcile to cash provided by	\$	\$10	\$(1)	\$ (8)	\$	\$(21)	\$14	\$ 26
operations: Depreciation and amortization Income taxes	(5) 	26	2			(26)		334 (20)
Other Changes in operating accounts								71
Cash provided by (used in)								
operations	(5)	36	1	(8)		(47)	14	512
INVESTING ACTIVITIES Acquisitions								
Cash received from sale of								
assets Purchase of short-term marketable	(35)							16
securities Capital								(354)
expenditures Other	2 33	(11)						(209) 47
Cash provided by (used in) investing								4
activities		(11)						(500)
FINANCING ACTIVITIES Issuances of								
debt Scheduled	(3)			(75)				1,967
principal payments Debt	(6)	(7)				(30)		(133)
prepayments Transfers to								(1,550)
Host Marriott Other								(226) 27
Cash provided by (used in) financing								
activities	(9)	(7)		(75)		(30)		85
INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS	¢(14)	¢10	¢ 1	¢(92)	œ.	¢(77)	14	¢ 07
LYUIVALLNI3	Φ(14) ========			\$(83) ====================================				

See Notes to the Unaudited Pro Forma Statements of Cash Flows.

# NOTES TO PRO FORMA STATEMENTS OF CASH FLOWS 100% PARTICIPATION WITH NO NOTES ISSUED

- A) Represents the adjustment for cash flow amounts related to Crestline.
- B) Represents the adjustment to record depreciation expense and estimated capital expenditures for the Blackstone Acquisition.
- C) Represents the adjustment to record depreciation expense and estimated capital expenditures for the 1998 acquisition of, or purchase of controlling interests in, 11 full-service properties.
- D) Represents the adjustment to record depreciation expense and capital expenditures for the 1997 acquisition of, or purchase of controlling interests in, 18 full-service properties. Cash from investing activities has also been adjusted as if the period's historical acquisitions occurred immediately prior to the period presented.
- E) Represent the adjustment to record the decrease in depreciation expense, capital expenditures and other investing activities for the sale of the New York East Side Marriott and the Napa Valley Marriott, including the elimination of the non-recurring gains on the sales totaling \$50 million.
- F) Represents the adjustment to reflect the decrease in interest expense associated with the refinancing or payoff of mortgage debt for three full-service properties (Marriott Orlando World Center, the Philadelphia Marriott, and the San Francisco Marriott). Cash from financing activities has also been adjusted as if the period's historical prepayments and issuances of debt occurred immediately prior to the period presented.
- G) Represents the adjustment to reflect the issuance of the \$1.7 billion of New Senior Notes net of the discount of \$8 million, the retirement of the Old Senior Notes of \$1.55 billion and the initial draw on the New Credit Facility of \$372 million, including interest expense and commitment fees.
- H) Represents the adjustment to record the removal of depreciation, capital expenditures, the sale of certain assets and the scheduled principal amortization of notes to reflect the deconsolidation of the Non-Controlled Subsidiary.
- I) Represents the adjustment to record depreciation expense and capital expenditures related to the Mergers, including the scheduled principal amortization of notes.
- J) Represents the adjustment to record depreciation expense related to the acquisition of the Private Partnerships.
- K) Represents the adjustment to reflect the decrease in interest income, net of tax, as a result of the estimated \$225 million Special Dividend and interest expense for the \$75 million draw on the New Credit Facility.
- L) Represents the adjustment to revenues to reflect lease income and remove hotel revenues and management fees.
- M) Represents the adjustment to reduce depreciation expense related to the sale of certain furniture and equipment to the Non-Controlled Subsidiary, including the scheduled principal amortization of notes.
- N) Represents the adjustment to the income tax provision to reflect the REIT Conversion.
- 0) Represents the adjustment to eliminate non-recurring expenses incurred in connection with the REIT Conversion.
- (1) The amount of earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

## UNAUDITED PRO FORMA BALANCE SHEET SEPTEMBER 11, 1998 100% PARTICIPATION WITH NOTES ISSUED (IN MILLIONS, EXCEPT OP UNIT AMOUNTS)

# ACQUISITIONS, DISPOSITIONS AND OTHER ACTIVITIES

		A DISTRIBUTION	HOST MARRIOTT HOTELS	B BLACKSTONE		DEBT REPAYMENT
	HISTORICAL	ADJUSTMENT	HISTORICAL	ACQUISITION	ACQUISITIONS	& REFINANCING
ASSETS Property and equipment,						
net Notes and other receivables,	\$5,937	\$(649)	\$5,288	\$1,450	\$	\$
net Due from	32	(3)	29	63		
managers Investments in	88	(5)	83	5		
affiliates Other assets	18 319	2	18 321			
Receivable from Lessee for working						
capital Cash, cash equivalents and short-term marketable						
securities	575	(27)	548	(262)		
	\$6,969	\$(682)	\$6,287	\$1,256	\$	\$ \$
LIABILITIES AND	=====	=====	=====	=====	====	====
EQUITY						
Debt(I)	\$4,224	\$(213)	\$4,011	\$ 600	\$	\$ 
Convertible debt obligation to Host Marriott						
Corporation Accounts payable and accrued		567	567			
expenses Deferred income	70	(23)	47			
taxes	526	(62)	464			
liabilities	447	(10)	437			
Total						
liabilities Convertible Preferred	5,267	259	5,526	600		
Securities Limited Partner interests of third parties at redemption value (on a pro forma basis 47.1 million OP Units	550	(550)				
outstanding)(J) Equity General Partner (on a proforma basis .2 million OP Units outstanding)(J). Limited Partner interests of Host REIT (on a pro forma basis 204.3 million OP Units				656		
outstanding)(J).	. 1,152	(391) 	761 			
	\$6,969 =====	\$(682) =====	\$6,287 =====	\$1,256 =====	\$ ====	\$ ====

MERGERS AND REIT CONVERSION ACTIVITIES

ASSETS Property and equipment, net \$(290) \$520 \$ 61 \$ \$ \$ \$7,029 Notes and other	L REIT 2000 PRO FORMA  \$5,579 240 14 27
NON- CONTROLLED PRIVATE DISTRIBU- CONTRIBUTION CONVER- SUBSIDIARY MERGERS PARTNERSHIPS TION(1) TO CRESTLINE SION ADJUSTMENT FORMA  ASSETS Property and equipment, net	2000 PRO FORMA  \$5,579 240 14
ASSETS Property and equipment, net \$(290) \$520 \$ 61 \$ \$ \$ \$7,029 Notes and other	\$5,579 240 14
Property and equipment, net \$(290) \$520 \$ 61 \$ \$ \$ \$7,029 Notes and other	240
Notes and other	240
receivables,	14
net 214 (3) 303 Due from	
managers (1) 12 (85) 14 Investments in	27
affiliates       9           27         Other assets       5       23       (11)           338	338
Receivable from Lessee for working capital 85 85 Cash, cash	80
equivalents and short-term marketable securities (8) 7 (11) (150) (15) 116	378
7	
\$ (71) \$559 \$ 39 \$ (150) (8) \$ \$ \$7,912	•
LIABILITIES AND     EQUITY Debt(I) \$ (39) \$571 \$ \$ 75 \$ \$ \$5,218 Convertible debt     obligation to	\$4,618
Host Marriott Corporation 567 Accounts payable	567
and accrued expenses (1) 8 54 Deferred income	54
taxes 6 (195) 275	470
Other liabilities (37) (20) (6) 320 694	651
Total liabilities (71) 559 (6) 75 320 (195) 6,808 Convertible Preferred	6,360
Securities Limited Partner interests of third parties at redemption value (on a pro	
forma basis 47.1 million OP Units outstanding)(J) 45 701	45
Equity  General Partner  (on a proforma basis .2 million OP Units outstanding)(J)  Limited Partner interests of Host REIT (on a pro forma basis 204.3 million OP Units	
outstanding)(J) (225) (8) (320) 195 403	251
\$ (71) \$559 \$ 39 \$ (150) \$ (8) \$ \$ \$7,912 ====================================	•
Book value per OP Unit \$ 4.39	\$ 1.42

#### NOTES TO UNAUDITED PRO FORMA BALANCE SHEET 100% PARTICIPATION WITH NOTES ISSUED

- A. Represents the adjustment to record the spin-off of Crestline:
- . Reduce property and equipment by \$649 million
- . Reduce receivables by \$3 million related to certain Crestline notes held by the Operating Partnership
- . Reduce due from managers by \$5 million
- . Increase other assets by \$2 million
- . Reduce cash, cash equivalents and short-term marketable securities by \$27 million
- . Reduce debt by \$213 million
- . Reduce accounts payable and accrued expenses by \$23 million
- . Reduce deferred income taxes by \$62 million
- . Reduce other liabilities by \$10 million
- . Reduce equity by \$391 million
- Eliminate the \$550 million Convertible Preferred Securities of Host Marriott which remain an obligation of Host REIT
- Record the \$567 million of Convertible Debt Obligation to Host Marriott which is eliminated in consolidation on the historical financial statements of Host Marriott Corporation
- B. Represents the adjustment to record the Blackstone Acquisition of 12 full-service properties (5,520 rooms) and a mortgage note secured by a thirteenth full-service property including the issuance of 43.7 million OP Units as determined through negotiations between the Operating Partnership and Blackstone:
  - . Record property and equipment of \$1,450 million . Record mortgage note receivable of \$63 million

  - . Record increase in due from managers of \$5 million
  - Record the use of cash of \$262 million
  - . Record the assumption of mortgage debt of \$600 million
  - . Record the issuance of 43.7 million OP Units with an estimated fair value of \$656 million using an assumed Host Marriott stock price of \$15.00 (which includes the value of Crestline since Blackstone will receive shares of Crestline in addition to OP Units).

The purchase price of the Blackstone properties and mortgage note was determined based on the estimated fair value of the 43.7 million OP Units to be issued. The number of units to be issued will not increase or decrease depending on the stock price of Host Marriott at the time of closing of the

- C. Represents the adjustment to deconsolidate the assets and liabilities of the Non-Controlled Subsidiaries and to reflect the sale of certain hotel furniture and equipment to the Non-Controlled Subsidiary:
  - Record decrease in property and equipment of \$290 million, including \$180 million of hotel furniture and equipment sold to the Non-Controlled
  - Record receivable from Non-Controlled Subsidiary for the furniture and equipment loan of \$180 million and other notes totaling \$34 million
  - Record decrease in due from managers of \$1 million
  - . Record investment in subsidiary of \$9 million
  - Record increase in other assets of \$5 million
  - Record decrease in cash of \$8 million
  - Record decrease in debt of \$39 million of debt transferred to the Non-Controlled Subsidiaries.
  - . Record decrease in accounts payable and accrued expenses of \$1 million
  - . Record increase in deferred taxes of \$6 million
  - . Record decrease in other liabilities of \$37 million

- D. Represents the adjustment to record the Mergers and issuance of Notes at the Note Election Amount (the greater of Liquidation Value or 80% of Exchange Value) to the Limited Partners:
  - . Record property and equipment of \$520 million
  - Record decrease in notes receivable of \$3 million
  - . Record increase in due from managers of \$12 million
  - Record other assets of \$23 million
  - Record cash of \$7 million
  - . Record debt of \$571 million including \$248 million of Notes to the
  - Limited Partners at the Note Election Amount . Record accounts payable and accrued expenses of \$8 million

. Record decrease in other liabilities of \$20 million

The value of 6.56% Notes expected to be issued to the limited partners of each Partnership is (in millions):

	PURCHASE PRICE VALUE OF NOTES ISSUED	PROPERTY AND EQUIPMENT
Atlanta Marquis	\$ 19	\$ 19
Desert Springs	29	29
Hanover	4	4
MHP	64	45
MHPII	73	67
Chicago Suites	11	37
MDAH		157
PHLP	7	162
	\$248	\$520
	====	====

The purchase price for minority interests (Atlanta Marquis, Desert Springs, Hanover, MHP and MHP2) was allocated to property to the extent that the purchase price exceeded the minority interest liability recorded. The purchase price for the three partnerships that are presently not consolidated was allocated in accordance with APB Opinion Number 16 with the debt of each partnership recorded at estimated fair value, all assets and liabilities, except for property being recorded at historical carrying values of each partnership with the residual allocated to property. The amounts allocated to property are in all cases less than estimated current replacement cost.

- E. Represents the adjustment to record the purchase of the remaining minority interests in four Private Partnerships:
  - . Record property and equipment of \$61 million
  - . Record decrease in other assets of \$11 million
  - Record use of cash of \$11 million
  - Record decrease in minority interest liabilities of \$6 million
  - . Record the issuance of 3.6 million OP Units totaling approximately \$45
- F. Represents the estimated \$225 million cash payment of the Special Dividend to shareholders of Host Marriott, which when combined with the value of the Crestline common stock (estimated to be approximately \$1.30 per share for a total Initial E&P Distribution of approximately \$2.20 per share) will represent the Initial E&P Distribution, by a draw on the New Credit Facility of \$75 million and available cash of \$150 million. The aggregate value of the Crestline common stock and the cash or other consideration to be distributed to Host stockholders (and the Blackstone Entities) in connection with the Initial E&P Distribution is currently estimated to be approximately \$525 million to \$625 million of which approximately \$200 to \$300 million is expected to be represented by the Special Dividend.(/1/)
- G. Represents the adjustment to record the transfer of working capital to Crestline related to the leasing of the Operating Partnership's hotels by decreasing working capital and recording a receivable from the lessee of \$85 million and the adjustment to record \$320 million in deferred revenue in connection with the application of EITF 98-9 to the Operating Partnership's rental revenue.

- H. Represents the adjustment to record the effect on deferred taxes for the change in tax status resulting from the REIT Conversion by decreasing deferred taxes and increasing equity by \$195 million.
- I. The Operating Partnership's pro forma aggregate debt maturities at September 11, 1998, excluding \$8 million of capital lease obligations and the \$8 million debt discount recorded in conjunction with the Senior Note Refinancing, are (in millions):

1998 1999.	
2000	
2001	
2002	 150
Thereafter	 4,832
	\$5,785

J. The number of OP Units includes the following (in millions):

General Partner interests of Host REIT	0.2
Limited Partner interests of Host REIT	204.3
Limited Partner interests of Private Partnerships	3.4
Limited Partner interests of Blackstone Group	43.7
Total OP Units	251.6
	=====

- K. Represents the adjustment to record the contribution of \$15 million in cash to Crestline as a reduction in equity and to record the sale of an investment of approximately \$7 million in a joint venture which holds a mortgage note of approximately \$130 million from a consolidated subsidiary of Host.
- L. The "REIT 2000 Pro Forma" assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur, Host does not become a REIT until January 1, 2000 and no Notes issued. The amounts reflect the "Pro Forma" column less the amounts in column B--"Blackstone Acquisition" and the reduction in other liabilities and increase in equity related to the deferred revenue of \$43 million related to the application of EITF 98-9 to the Blackstone properties.
- . . .
- (1) The amount of earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

# UNAUDITED PRO FORMA STATEMENT OF OPERATIONS FIRST THREE QUARTERS 1998 100% PARTICIPATION WITH NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AMOUNTS AND RATIOS)

				ACQUISTI10	NO, DIOPUOIII	UNS AND UTHER	WCITATITE2
		Α		В	С	E	G
	HOST MARRIOTT CORPORATION HISTORICAL	DISTRIBUTION ADJUSTMENT	HOST MARRIOTT HOTELS HISTORICAL		1998 ACQUISITIONS	DISPOSITIONS	BOND REFINANCING
REVENUE							
Rental revenues Hotel revenues Equity in	\$ 922	\$ 	\$ 922	\$ 122	\$ 43	\$ (6)	\$ 
earnings of affiliates	1		1				
Other revenues	117 	(58) 	59 			(50) 	
Total revenues	1,040	(58) 	982	122 	43 	(56) 	
OPERATING COSTS AND EXPENSES	500		500	70	00	(0)	
Hotels Other	502 45	(30)	502 15	70  	23 	(3)  	
Total operating							
costs and expenses	547	(30)	517 	70 	23	(3)	
OPERATING PROFIT	493	(28)	465	52	20	(53)	
Minority interest	(36)		(36)		(1)	1	
corporate expenses	(33)	3	(30)				
REIT Conversion expenses Interest	(14)		(14)				
expense Dividends on Convertible	(245)	(8)	(253)	(36)	(1)	1	(17)
Preferred Securities Interest	(26)	26					
income	36	2	38	(6)	(16)	(1)	
Income (loss) before income taxes	175	(5)	170	10	2	(52)	(17)
Benefit (provision) for	175	(5)	170	10	2	(32)	(17)
income taxes	(75)	3	(72)	(4)	(1)	21	7
Income (loss) before extraordinary							
items	\$ 100 =====	\$ (2) ====	\$ 98 =====	\$ 6 ====	\$ 1 ====	\$(31) ====	\$(10) ====
Basic loss per OP Unit Ratio of							
earnings to fixed charges	1.7x =====		1.8x =====				
Deficiency of earnings to fixed changes	=====		====				
				T CONVERSION			
	Н	J	К	L	P/N		0 Q
	CONTROLLED	ISSUANCE PART		STRIBUTION(1)		CONVER- T.	REI COME 200 AX PRO PR STMENT FORMA FOR
REVENUE							
REVENUE Rental revenues Hotel revenues Equity in	\$ (14)			\$ 	\$		\$ 540 \$ 4 

ACQUISITIONS, DISPOSITIONS AND OTHER ACTIVITIES

Other revenues	(5)	<b></b>	<b></b>				<b></b>	4	4
Total revenues	(22)	56				(583)		542	470
OPERATING COSTS AND EXPENSES Hotels	(12)	35 	2	 		(224)		393 15	335 15
Total operating costs and expenses	(12)	35	2			(224)		408	350
OPERATING PROFIT Minority	(10)	21	(2)			(359)		134	120
interest Corporate expenses	3	18	1					(14)	(14) (30)
REIT Conversion expenses Interest					14				
expense Dividends on Convertible Preferred	3	(33)		(4)				(340)	(304)
Securities Interest income	2	1		 (5)	(2)	13		24	30
Income (loss) before income taxes Benefit (provision) for income taxes	(2)	7	(1)	(9)	12	(346)	(75)	(226)	(198)
Income (loss) before extraordinary items	\$ (1)	\$ 4	\$ (1)	\$ (5)		\$ (208)	\$(75)	\$(215)	
Basic loss per OP Unit	==========	====== :		=======================================	=======================================	=======================================	:=======	\$(.85)	\$(.90)
Ratio of earnings to fixed charges								N/A	N/A
Deficiency of earnings to fixed changes								\$(208) =====	\$(180)

See Notes to the Unaudited Pro Forma Statements of Operations.

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS FISCAL YEAR 1997 100% PARTICIPATION WITH NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AMOUNTS AND RATIOS)

AND EXPENSES

	,	SPOSITIONS AN	
12011		D	 

		Α		В	С	D	E	F/G
	HOST MARRIOTT CORPORATION HISTORICAL	I DISTRIBUTI ADJUSTMEN		BLACKSTONE ACQUISITION	1998 ACQUISITIONS	1997 S ACQUISITIONS	DISPOSITIONS	DEBT REPAYMENT & REFINANCING
DEVENUE.								
REVENUE Rental								
revenues	\$	\$	\$	\$	\$	\$	\$	\$
Hotel revenues	1,093		1,093	148	112	89	(23)	
Equity in								
earnings of affiliates	5		5					
Other revenues	49	(37)	12					
Total mayanyas		(27)		140			(22)	
Total revenues	1,147	(37)	1,110	148	112	89 	(23)	
OPERATING COSTS AND EXPENSES								
Hotels	649		649	101	62	42	(10)	
Other	49	(20)	29					
Total operating								
costs and								
expenses	698	(20)	678	101	62	42	(10)	
OPERATING								
PROFIT	449	(17)	432	47	50	47	(13)	
Minority	(00)		(00)		(4)	_	(4)	
interest Corporate	(32)		(32)		(4)	5	(1)	
expenses	(47)	2	(45)					
Interest	()	()	()	( .= )	()	()	_	( )
expense Dividends on	(302)	(23)	(325)	(48)	(12)	(12)	3	(48)
Convertible								
Preferred								
Securities	(37)	37						
Interest income	52 		52 	(7) 	(14)	(14)		(3)
Income (loss)								
before income		(4)	20	(0)			(44)	(54)
taxes Benefit	83	(1)	82	(8)	20	26	(11)	(51)
(provision) for								
income taxes	(36)	1	(35)	3	(8)	(10)	4	20
Income (loss) before extraordinary items	\$ 47	\$	\$ 47	\$ (5)	\$ 12	\$ 16	\$ (7)	\$(31)
Basic earnings	=====	====	=====	====	====	====	====	====
per OP Unit Ratio of earnings to fixed								
charges	1.3x		1.3x					
	=====		=====					
		MERGERS	AND REIT CON	VERSION ACTI	VITIES			
	Н	J	K L		I/M	0	Q	
			EARNIN IVATE & PROF RTNER- DISTR	ITS OTHER	LEASE CONVER-	INCOME TAX PR	REIT 0 2000	
	SUBSIDIARY	ISSUANCE S	HIPS BUTION	(1) ACTIVITI	ES SION A	ADJUSTMENT FOR		A.
REVENUE								
Rental	,							
revenues Hotel revenues	\$ (17)	\$ 74	\$ 	\$ 	\$ 1,135 (1,476)	•	135 \$998 	
Equity in	(11)	74			(1,4/0)			
earnings of								
affiliates Other revenues	(12) (9)						(7) (7) 3 3	
other revenues	(9)						3 3 	
Total revenues	(38)	74			(341)	1,	131 994 	
OPERATING COSTS								

Hotels Other	(9) (18)	49 	2	 	 	(288) 		598 11	511 11
Total operating costs and expenses	(27)	49	2			(288)		609	522
OPERATING PROFIT Minority	(11)	25	(2)			(53)		522	472
interest Corporate	4	17	1					(10)	(10)
expenses Interest	1							(44)	(44)
expense Dividends on Convertible	5	(42)		(6)				(485)	(437)
Preferred Securities Interest income	4	 1		 (8)		 18		 29	36
Income (loss) before income taxes	3	1	(1)	(14)		(35)		12	17
(provision) for income taxes	(3)			6		14	8	(1)	(1)
Income (loss) before extraordinary items	\$	\$ 1	\$ (1)	\$ (8)	\$	\$ (21)	\$ 8	\$ 11	\$ 16
Basic earnings per OP Unit	=======	======= :	======	=======	=======	= ====== =:	=======	\$ .04	\$.08
Ratio of earnings to fixed charges								1.1x	1.1x

See Notes to the Unaudited Pro Forma Statements of Operations.

## NOTES TO UNAUDITED PRO FORMA STATEMENTS OF OPERATIONS ASSUMING 100% PARTICIPATION WITH NOTES ISSUED

- A. Represents the adjustment to reduce revenues, operating expenses, corporate expenses, interest expense, interest income and income taxes for the spin-off of Crestline.
- B. Represents the adjustment to record the historical revenues, operating expenses, interest expense, income taxes and to reduce interest income associated with the acquisition of the equity and debt interests for the Blackstone Acquisition.
- C. Represents the adjustment to record the historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income associated with the 1998 acquisition of, or purchase of controlling interests in, 11 full-service properties.
- D. Represents the adjustment to record historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income associated with the 1997 acquisition of, or purchase of controlling interests in, 18 full-service properties.
- E. Represents the adjustment to record historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income for the 1998 sale of the New York Marriott East Side and the Napa Valley Marriott, including the elimination of the non-recurring gains on the sales totalling \$50 million and related taxes of \$20 million in 1998.
- F. Represents the adjustment to reduce the interest expense, interest income and income taxes associated with the refinancing or payoff of mortgage debt for three full-service properties (Marriott's Orlando World Center, the Philadelphia Marriott and the San Francisco Marriott).
- G. Represents the adjustment to record interest expense and related amortization of deferred financing fees and reduce interest income and income taxes as a result of the Senior Note Refinancing. The adjustment excludes the extraordinary loss of \$148 million, net of taxes, related to the Senior Note Refinancing resulting from the write-off of deferred financing fees and the payment of bond tender and consent fees.
- H. Represents the adjustment for revenues, operating expenses, minority interest, interest expense, corporate expenses, income taxes and interest income to deconsolidate the Non-Controlled Subsidiaries and reflect the Operating Partnership's share of income as equity in earnings of affiliate.
- I. Represents the adjustment to reduce depreciation expense of \$18 million and \$26 million for First Three Quarters 1998 and fiscal year 1997 related to certain furniture and equipment sold to the Non-Controlled Subsidiary, record interest income of \$9 million and \$13 million for First Three Quarters 1998 and fiscal year 1997 earned on the 7%, \$180 million in notes from the Non-Controlled Subsidiary and reduce the lease payment to the Operating Partnership from the Lessee.
- J. Represents the adjustment to record the historical revenues, operating expenses, minority interest, interest expense, interest income and income taxes associated with the Mergers, including three partnerships not previously consolidated by the Operating Partnership. Interest expense reflects interest on various mortgage notes and the estimated \$248 million in 6.56% Notes issued in lieu of OP Units.
- K. Represents the adjustment to record additional depreciation expense and the decrease in minority interest expense related to the purchase of the remaining minority interests in the Private Partnerships.
- L. Represents the adjustment to reduce interest income, income taxes and record interest expense for the estimated \$225 million cash payment of the Special Dividend to shareholders of Host Marriott including a draw on the New Credit Facility of \$75 million.(/1/)

M. Represents the adjustment to remove hotel revenues and management fees and other expenses of \$224 million and \$288 million, respectively, for First Three Quarters 1998 and Fiscal 1997 and record rental revenues associated with the leasing of certain hotel properties to Crestline and other lessees and interest income of \$3 million and \$5 million for First Three Quarters 1998 and fiscal year 1997 earned on the 6%, \$85 million in notes from Crestline. First Three Quarters 1998 included a \$320 million reduction to rental income to record deferred revenue for percentage rents in accordance with EITF 98-9 Management believes the change to the lease structure described above will not impact hotel operating results because the hotel manager and asset management function will remain unchanged. Rental revenues under the Leases are based on the greater of Percentage Rent or Minimum Rent. Total Rent in the pro forma statement of operations is calculated based on the historical gross sales of the property and the negotiated pay rates and thresholds by property as if the leases were entered into on the first day of fiscal year 1997. There are generally three sales categories utilized in the rent calculation: rooms, food and beverage and other. For rooms and food and beverage, there are three tiers of rent with two thresholds, while the other category generally has two tiers of rent and one threshold. The percentage rent thresholds are increased annually on the first day of each year after the initial lease year based on a blended increase of the Consumer Price Index ("CPI") and a wage and benefit index. For purposes of the pro formas, 1997 is the assumed initial lease year and the blended increase applied to the thresholds at January 3, 1998 is assumed to be 3%. Minimum rent is expressed as a fixed dollar amount that increases annually on the first day of each year after the initial lease year as 50% of the CPI increase. Accordingly, the 1998 rent thresholds and minimum rent included in the pro formas were adjusted as of January 3, 1998 for the 1997 increases in the indices. Rental revenue is recognized only for leases to be executed with Crestline at or prior to the completion of the REIT Conversion. The execution of the leases is dependent upon distribution of the Crestline common stock to the Host's stockholders, which is subject to contingencies that are outside the control of the Operating Partnership, including consent of shareholders, lenders, debt holders, partners and ground lessors of Host. The Operating Partnership believes that negotiations with third parties to complete the REIT Conversion will not result in any material change to the leases. See page F-120 for a table detailing gross sales, minimum rent and total rent for all full-service properties to be leased and summarized amounts for the limited-service properties to be subleased.

FISCAL YEAR 1997

FIRST THREE OUARTERS 1998

	FISC	AL IEAR .	1991	LIKS	OI INKEE	I AAU	EK2 1990
PROPERTY		MINIMUM RENT	RENT	SALES		RENT	RECOGNIZED
				LLIONS			
			( 114 111	LLIONO	'		
Grand Hotel Resort and Golf	<b>#</b> 00 4	<b>#2.0</b>	<b>#4</b> 0	<b>#</b> 40 0	<b>#2.0</b>	<b>#</b> 0 7	<b>#2.0</b>
Club Scottsdale Suites	\$23.4 11.9	\$2.8 3.0	\$4.2 5.0	\$18.0 8.2	\$2.0 2.1	\$3.7 3.4	\$2.0 2.1
The Ritz-Carlton, Phoenix	23.3	4.6	7.2	17.3	3.2	5.5	3.2
Coronado Island Resort	22.0	2.1	2.1	16.2	3.2 1.5	3.6	1.5
Costa Mesa Suites	9.7	1.9	3.3	7.2	1.3	2.3	1.3
Desert Springs Resort and	9.7	1.9	3.3	1.2	1.3	2.3	1.3
Spa	103.3	21.3	30.3	80.3	15.0	22.6	15.0
Manhattan Beach	16.3	2.4	4.8	12.2	1.7	3.6	1.7
Marina Beach	21.1	4.6	7.1	16.9	3.2	6.2	3.2
Newport Beach	33.5	5.5	8.7	24.0	3.9	6.8	3.9
Newport Beach Suites	11.0	2.6	4.1	8.0	1.8	3.0	1.8
Ontario Airport San Diego Marriott Hotel	12.1	1.8	3.4	8.3	1.3	2.2	1.3
and Marina	103.3	38.0	39.6	78.6	26.7	31.1	26.7
San Diego Mission Valley	16.7	3.4	5.1	12.6	2.4	5.6	2.5
San Francisco Airport	43.8	8.2	13.2	32.2	5.8	9.5	5.8
San Francisco Fisherman's							
Wharf San Francisco Moscone	17.8	4.0	6.4	12.1	2.8	4.3	2.8
Center	120.2	20.7	37.9	90.5	14.6	28.5	14.6
San Ramon	19.7	4.4	5.1	14.4	3.1	4.0	3.1
Santa Clara	47.3	7.8	16.5	37.2	5.5	13.5	8.8
The Ritz-Carlton, Marina							
del Rey The Ritz-Carlton, San	32.4	5.5	10.8	23.4	3.9	7.9	3.9
Francisco	50.1	9.6	14.7	34.2	6.7	10.3	6.7
Torrance	20.5	2.3	3.5	15.0	1.6	5.1	1.6
Denver Southeast	21.5	3.0	6.2	14.9	2.1	4.1	2.1
Denver Tech Center	26.8	5.1	8.3	20.1	3.6	6.0	3.6
Denver West	13.7	1.8	4.0	9.6	1.2	2.7	1.7
Marriott's Mountain Resort at Vail	17.6	3.0	5.1	14.1	2.1	4.5	2.1
Hartford/Farmington	18.4	3.5	4.7	13.4	2.4	3.5	2.4
Hartford/Rocky Hill	11.6	1.5	2.7	8.5	1.1	2.0	1.1
Fort Lauderdale Marina	28.5	4.3	7.9	20.4	3.0	5.7	3.2
Harbor Beach Resort	58.1	16.5	19.3	43.2	11.6	14.0	11.6
Jacksonville	11.8	1.8	3.6	8.0	1.2	2.4	1.2
	0		0.0	0.0		'	

	FISCAL YEAR 1997			FIRST THREE QUARTERS 1998				
PROPERTY	GROSS SALES	MINIMUM RENT		GROSS SALES	MINIMUM RENT		TOTAL RENT RECOGNIZED	
			(IN MILL	.10NS)				
Miami Airport Orlando World Center	\$ 29.7 128.2	\$ 3.9 23.5	\$ 8.4 39.6	\$ 21.6 98.7	\$ 2.8 16.5	\$ 6.1		
Palm Beach Gardens	11.8		3.7	8.5	1.4	30.4		
Singer Island (Holiday								
Inn) Tampa Airport	6.6 17.1		2.5 3.5	5.2 13.1	1.0 1.1	2.1 2.7		
Tampa Westshore	15.0	1.8	3.6	10.8	1.3	2.6		
The Ritz-Carlton, Naples	66.4		23.3	53.1	12.7	18.0		
Atlanta Marriott Marquis Atlanta Midtown Suites	85.4 10.5	21.3 1.8	33.3 3.5	58.6 7.8	15.0 1.3	25.6 2.6		
Atlanta Norcross	7.6	1.0	1.7	5.6	0.7	1.2		
Atlanta Northwest	14.9		4.3	11.3	1.9	3.3		
Atlanta Perimeter  JW Marriott Hotel at	16.6	2.5	4.5	12.6	1.7	3.5	1.7	
Lenox	24.8	3.7	6.8	17.7	2.6	5.0	2.6	
The Ritz-Carlton,	20.0	FO	0.0	01 7	4 4	6.0	4 1	
Atlanta The Ritz-Carlton,	30.2	5.8	8.8	21.7	4.1	6.8	4.1	
Buckhead	49.3		16.3	35.8	9.2			
Chicago/Deerfield Suites Chicago/Downers Grove	10.2	1.8	3.1	7.4	1.3	2.3	1.3	
Suites	9.0	1.8	2.9	6.7	1.3	2.2	1.3	
Chicago/Downtown								
Chicago O'Haro	16.3 40.0	3.1 5.5	4.9 11.5	12.2 28.8	2.2 3.9	3.9 8.2		
Chicago O'Hare South Bend	9.9	1.1	2.1	7.0	0.8	1.5		
New Orleans	66.4		21.8	47.6	12.3			
Bethesda	23.2	3.2	5.6	17.3	2.2	4.1	2.2	
Gaithersburg/Washingtonian Center	13.2	2.4	3.8	9.7	1.7	2.8	1.7	
Boston/Newton	27.4		7.8	19.1	3.4	5.5		
Detroit Romulus	8.8	1.1	1.8	6.6	0.8	1.4	0.8	
The Ritz-Carlton, Dearborn	25.7	3.6	5.5	17.7	2.5	4.0	2.5	
Minneapolis/Bloomington	20.2	3.3	6.5	13.8	2.3	4.7		
Minneapolis City Center	27.5	3.7	7.5	20.4	2.4	5.2		
Minneapolis Southwest Kansas City Airport	14.9 14.3	2.7 1.7	4.8 3.7	10.1 9.9	1.9 1.2	4.0 2.5		
St. Louis Pavilion	27.5	6.1	6.5	18.5	4.3	4.3		
Nashua	7.5		1.3	5.3	0.5	0.9		
Hanover	22.5		6.6	15.1 29.2	3.3 4.6	4.3 8.6	3.9 4.6	
Newark Airport Park Ridge	39.4 16.0		11.8 4.0	11.9	1.7	4.2		
Saddle Brook	10.7	1.3	2.1	7.8	0.9	1.7	0.9	
Albany	18.5	3.5	6.1	12.4	2.5	5.2	2.7	
New York Marriott Financial Center	39.6	7.7	13.2	29.1	5.4	10.1	5.4	
New York Marriott Marquis	210.3	40.0	60.8	155.4	29.7	47.6	29.7	
Marriott World Trade Center	65.4	12.2	19.4	49.1	8.6	14.9	8.6	
Charlotte Executive Park	14.0	2.3	3.7	9.8	1.6	2.6		
Raleigh Crabtree Valley	14.9	2.4	3.9	10.9	1.7	2.8		
Oklahoma City	15.6		3.8	10.4	1.4	2.4		
Oklahoma City Waterford Portland	9.1 26.4		2.7 7.5	6.1 17.6	1.0 2.9	1.7 4.8		
Philadelphia (Convention								
Center)Philadelphia Airport	80.7 25.0	14.2 4.1	25.0 7.6	58.2 18.6	10.0 2.9	17.8 5.6		
Pittsburgh City Center	16.4	1.9	3.0	11.1	1.3	2.2	1.3	
Memphis	10.6	1.5	3.2	5.7	1.0	1.8		
Dallas/Fort Worth Dallas Quorum	28.9 25.7	5.9 4.2	9.3 8.2	21.9 18.3	4.1 3.0	7.0 5.8		
El Paso	11.6		2.3	7.8	0.6	1.4		
Houston Airport	21.6		6.0	16.9	2.0	4.6	2.7	
JW Marriott Houston	27.2		8.0	20.1	3.5	5.9	3.5	
Plaza San Antonio San Antonio Rivercenter	13.8 68.9	2.9 17.5	4.6 24.5	9.7 49.3	2.0 12.3	3.3 17.8		
San Antonio Riverwalk	29.3	6.1	10.3	21.7	4.3	7.6	4.4	
Salt Lake City	28.5	5.6	9.5	21.1	3.9	7.2	4.2	

FIRST THREE QUARTERS
FISCAL YEAR 1997 1998

							T0T41
PROPERTY	GROSS SALES	MINIMUM RENT	TOTAL RENT	GROSS SALES	MINIMUM RENT	TOTAL RENT	TOTAL RENT RECOGNIZED
			(IN MILLI				
Dulles Airport	\$ 14.6	\$ 1.8	\$ 4.0	\$ 10.9	\$ 1.2	\$ 3.0	\$ 1.7
Key Bridge	29.4		10.2	21.2		7.4	3.9
Norfolk Waterside	18.1		5.4	12.8		3.8	2.4
Pentagon City Residence							
Inn	11.7	3.5	5.5	8.7	2.5	4.2	2.5
The Ritz-Carlton, Tysons							
CornerWashington Dulles	34.4	5.9	9.8	24.9	4.1	7.3	4.1
Suites	10.3	2.5	4.0	7.8	1.8	3.0	1.8
Westfields	28.0		7.4	20.3		5.4	3.3
Williamsburg	12.6		2.8	9.3		2.1	1.3
Washington Metro							
Center	25.2	4.5	7.3	19.2	3.2	5.3	3.2
Calgary	13.4		1.7	9.8		2.3	1.2
Toronto Airport	17.1		5.6	13.0		4.2	2.0
Toronto Eaton Centre	21.1		7.1	16.0		5.6	4.3
Toronto Delta		0.1		20.0		0.0	
Meadowvale	16.1	2.6	4.9	10.6	1.9	3.1	1.9
Fairview Park	22.5		7.3	16.3		5.2	2.8
Dayton	18.2		6.0	13.4		4.3	2.3
Research Triangle Park	9.1		2.9	6.8		2.3	1.0
Detroit Marriott	0.1			0.0			2.0
Southfield	8.8	1.2	2.1	6.9	0.9	1.7	0.9
Detroit Marriott							
Livonia	10.0	1.4	2.6	7.4	1.0	1.9	1.0
Fullerton	6.8	1.2	1.8	5.0	0.8	1.3	0.8
Marriott O'Hare Suites	14.4	2.7	4.9	10.8	1.9	4.0	1.9
Albuquerque	16.4	3.6	3.6	11.1	2.5	2.6	2.5
Greensboro-High Point	13.6	3.3	3.3	10.2	2.3	2.4	2.3
Houston Medical Center	16.5	4.0	4.0	12.2	2.8	2.9	2.8
Miami Biscayne Bay	26.8	6.5	6.6	20.5	4.5	5.1	4.5
Marriott Mountain							
Shadows Resort	24.1	4.4	4.5	16.9	3.1	3.1	3.1
Seattle SeaTac Airport	23.1	6.7	6.7	17.5	4.7	5.1	4.7
Four Seasons,							
Atlanta(4)	15.6	5.8	5.9	14.2	4.1	4.5	4.1
Four Seasons,							
Philadelphia(4)	41.1	7.9	12.4	30.6	5.6	10.1	5.6
Grand Hyatt,	0= 0	10.0	40.0				
Atlanta(4)	25.3	10.0	10.0	22.6	7.0	8.2	7.0
Burlingame(4)	47.9	8.8	17.6	39.5	6.2	15.1	9.0
Hyatt Regency,							
Cambridge(4)	32.4	6.7	11.9	26.8	4.7	10.4	6.1
Hyatt Regency,							
Reston(4)	30.5	6.5	11.3	24.2	4.5	9.2	4.8
Swissotel, Atlanta(4)	22.2		6.3	17.2		5.8	3.5
Swissotel, Boston(4)	26.8		8.5	20.5		6.9	4.5
Swissotel, Chicago(4)	38.1	10.9	15.1	28.9	7.7	12.0	7.7
The Drake (Swissotel),	20.0	11 6	12.6	24.2	0.0	10 4	0.0
New York(4)	38.8	11.6	13.6	34.2	8.2	13.4	8.2
The Ritz-Carlton, Amelia	45.7	10.3	13.4	37.4	7.2	11.1	7.2
Island(4)	45.7	10.5	13.4	37.4	1.2	11.1	1.2
The Ritz-Carlton, Boston(4)	40.1	6.9	10.5	31.4	4.8	8.8	4.8
Non-Controlled	40.1	0.9	10.5	31.4	4.0	0.0	4.0
Subsidiary Rent		(39.7)	(39.7)		(27.5)	(27.5)	(27.5)
Substitutary Rent		(39.7)	(39.7)		(27.3)	(27.5)	(27.3)
Total Full-service							
Properties	3,600.8	676.2	1,055.9	2,671.0	477.0	810.8	491.2
Total Courtyard	,		,	, -			
Properties	212.0	50.6	59.2	159.2	35.0	36.8	36.8
Total Residence Inns	69.9		20.3	50.6		12.3	12.3
Total	\$3,882.7	\$744.0	\$1,135.4	\$2,880.8	\$524.0	859.9	\$540.3
	======		======	======			=====
Less: Deferred rent							
under EITF 98-9						(319.6)	
Total rent							
recognized						\$ 540.3	

N. Represents the adjustment to eliminate interest income recorded for the \$92 million note receivable contributed to Crestline for First Three Quarters 1998.

 $<sup>{\</sup>tt O.}$  Represents the adjustment to the income tax provision to reflect the REIT Conversion.

- P. Represents the adjustment to eliminate non-recurring expenses incurred in connection with the REIT Conversion. Management expects that the total estimated nonrecurring expenses to be incurred will be approximately \$50 million.
- Q. The "REIT 2000 Pro Forma" reflects the adjustment to eliminate the revenues, operating expenses, interest expense and interest income which assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur and Host does not become a REIT until January 1, 2000.

(1) The amount of earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

## UNAUDITED PRO FORMA STATEMENT OF CASH FLOWS FIRST THREE QUARTERS 1998 100% PARTICIPATION WITH NOTES ISSUED (IN MILLIONS)

				ACQUISITIONS	S, DISPOSITION	NS AND OTHER	ACTIVITIES
		Α		В	С	E	G
	HOST MARRIOTT CORPORATION HISTORICAL	DISTRIBUTION ADJUST- MENTS	HOST MARRIOT HOTELS HISTORICAL	BLACKSTONE ACQUISITIONS	1998 ACQUISITIONS	DISPOSITIONS	BOND REFINANCING
OPERATING ACTIVITIES							
Income before extraordinary items Adjustment to reconcile to cash	\$ 100	\$ (2)	\$ 98	\$ 6	\$ 1	\$ (31)	\$ (10)
provided by operations: Depreciation and	101	(47)	407				
amortization Income taxes Gains on sales	184 50	(17) 	167 50	42 	11 		
of hotel properties Equity	(50)		(50)			50	
<pre>(earnings) losses of affiliates Changes in</pre>		(1)	(1)				
operating accounts	(33) 27	(2) 4	(35) 31	<del>-</del> -	<del>-</del> -	<del>-</del> -	
Cash provided by							
(used in) operations	278	(18)	260	48	12 	19 	(10)
INVESTING ACTIVI- TIES Acquisitions	(636)	29	(607)		607		
Cash received from sale of assets	211		211			(211)	
Capital expenditures Purchases of short-term	(168)	9	(159)	(16)	(5)	2	
marketable securities Sales of short- term marketable	(134)		(134)				
securities Other	451 3	 (89)	451 (86)			 21	
Cash provided by (used in) investing							
activities	(273)	(51) 	(324)	(16)	602 	(188) 	
TIES Issuances of debt Scheduled	2,004		2,004				1,967
principal repayments Debt repayments	(42) (1,750)	3 119	(39) (1,631)			 35	 (1,250)
Transfers to Host Marriott Other	 (189)	(63) 	(63) (189)				
Cash provided by (used in) financing	22		02			25	747
activities  INCREASE	23	59 	82 			35 	717 
(DECREASE) IN CASH AND CASH EQUIVALENTS	\$ 28 =====	\$(10) ====	\$ 18 =====	\$ 32 ====	\$614 ====	\$(134) =====	\$ 707 =====

MERGERS AND REIT CONVERSION ACTIVITIES

	H NON- CONTROLLED SUBSIDIARY		J PRIVATE PARTNERSHIPS	K EARNINGS & PROFITS DISTRIBUTION(1)		O OTHER REIT ACTIVITIES	N TAX ADJUSTMENT	PRO FORMA
OPERATING ACTIVITIES Income before extraordinary items	\$(1)	\$ 4	\$(1)	\$ (5)	\$(208)	\$ 7	\$(75)	\$ (215)
Depreciation and amortization Income taxes Gains on sales of hotel	(4)	16 	2		(18) 138			216 188
properties Equity (earnings)								
losses of affiliates Changes in operating								(1)
accounts Other assets					320			285 31
Cash provided by (used in) operations	(5)	20	1	(5)	232	7	(75)	504
INVESTING ACTIVI-								
TIES Acquisitions Cash received from sale of								
assets								
Capital expenditures Purchases of short-term marketable	1	(9)						(186)
securities Sales of short- term marketable								(134)
securities Other								451 (65)
Cash provided by (used in) investing activities	4	(0)						66
activities		(9)						66 
FINANCING ACTIVI- TIES Issuances of				(75)				
debt Scheduled principal				(75)				3,896
repayments Debt repayments	(2)				(15) 			(56) (2,846)
Transfers to Host								
Marriott Other								(63) (189)
Cash provided by								
(used in) financing activities	(2)			(75)	(15)			742
INCREASE (DECREASE) IN CASH AND CASH								
EQUIVALENTS				\$(80) =======	\$ 217 ======		` ,	

See Notes to the Unaudited Pro Forma Statements of Cash Flows.

## UNAUDITED PRO FORMA STATEMENT OF CASH FLOWS FISCAL YEAR 1997 100% PARTICIPATION WITH NOTES ISSUED (IN MILLIONS)

		Α		В	D	C	E	F/G
	HOST		HOST	Ь	D	C	<b>E</b>	
	MARRIOTT CORPORATION HISTORICAL	DISTRIBUTION ADJUST- MENT	MARRIOTT HOTELS HISTORICAL	BLACKSTONE ACQUISITION	1997 ACQUISITIONS	1998 ACQUISITIONS	DISPOSITIONS	DEBT REPAYMENTS & REFINANCING
OPERATING ACTIVITIES Income before extraordinary items Adjustment to reconcile to cash	\$ 47	\$	\$ 47	\$ (5)	\$ 16	\$ 12	\$ (7)	\$ (31)
provided by operations: Depreciation and amortization Income taxes Other Changes in operating	240 (20) 70	(9)  3	231 (20) 73	61  	17  	31  	(3)  (2)	::
accounts	127 	(26) 	101					
Cash provided by (used in) operations	464	(32)	432	56	33	43	(12)	(31)
INVESTING ACTIVI- TIES Acquisitions Cash received from sale of	(596)	237	(359)		359			
assets Purchase of short-term mar-	51		51					
ketable securi- ties Capital expendi-	(354)		(354)					
tures	(160) 13	2	(158) 13	(20)	(13) 	(13) 	4 1	
Cash provided by (used in) investing activities	(1,046)	239	(807)	(20)	346	(13)	5	
FINANCING ACTIVI-								
TIES Issuances of debt Scheduled principal	857		857					1,188
repayments Debt prepay-	(93)	3	(90)					
ments Transfers to Host	(403)		(403)					(1,147)
Marriott Other	 28	(226) (1)	(226) 27					
Cash provided by (used in)								
financing activities	389	(224)	165					41
INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS	\$ (193) ======	\$ (17) =====	\$(210) =====	\$ 36 ====	\$379 ====	\$ 30 ====	\$ (7) ====	\$ 10 =====
		MERGERS AI	ND REIT CONV	ERSION ACTIV	ITIES			
	Н	I J			L/M	N		
	NON- NON- NON- NON- NON- NON- NON- NON-		NER- DISTRI	TS		TAX PRO USTMENT FORMA		

ACQUISITIONS, DISPOSITIONS AND OTHER ACTIVITIES

Income before extraordinary items	\$	\$ 1	\$(1)	\$ (8)	\$	\$(21)	\$ 8	\$ 11
Depreciation and amortization	(5)	23	2			(26)		331
Income taxes								(20)
Other Changes in								71
operating accounts		15						116
Cash provided by (used in)								
operations	(5)	39	1	(8)		(47)	8	509
INVESTING ACTIVI- TIES								
Acquisitions Cash received from sale of								
assets Purchase of short-term mar- ketable securi-	(35)							16
ties								(354)
Capital expendi- tures	2	(11)						(209)
Other	33	′						47
Cash provided by (used in) investing activities		(11)						(500)
FINANCING ACTIVI- TIES								
Issuances of debt	(3)			(75)				1,967
Scheduled principal								
repayments Debt prepay-	(6)	(7)				(30)		(133)
ments Transfers to Host								(1,550)
Marriott								(226)
Other							 	27
Cash provided by (used in) financing	(0)	(7)		(75)		(00)		0.5
activities	(9)	(/)		(75)		(७७)	 	85 
INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS	\$(14)	\$ 21	\$ 1	\$(83)	\$	\$(77)	\$ 8	\$ 94
	========	=======	======	=======	=======	=======================================		= ======

See Notes to the Unaudited Pro Forma Statements of Cash Flows.

#### NOTES TO UNAUDITED CASH FLOW STATEMENTS 100% PARTICIPATION WITH NOTES ISSUED

- A) Represents the adjustment for cash flow amounts related to the distribution of Crestline.
- B) Represents the adjustment to record the depreciation expense and estimated capital expenditures for the Blackstone Acquisition.
- C) Represents the adjustment to record depreciation expense and estimated capital expenditures for the 1998 acquisition of, or purchase of controlling interests in, 11 full-service properties.
- D) Represents the adjustment to record depreciation expense and capital expenditures for the 1997 acquisition of, or purchase of controlling interests in, 18 full service properties. Cash flows from investing activities has also been adjusted as if the period's historical acquisitions occurred immediately prior to the period presented.
- E) Represent the adjustment to record the decrease in depreciation expense, capital expenditures and other investing activities for the sale of the New York East Side Marriott and the Napa Valley Marriott, including the elimination of the non-recurring gains on the sales totaling \$50 million.
- F) Represents the adjustment to reflect the decrease in interest expense associated with the refinancing or payoff of mortgage debt for three full-service properties (Marriott Orlando World Center, the Philadelphia Marriott, and the San Francisco Marriott). Cash from financing activities has also been adjusted as if the periods historical prepayments and issuances of debt occurred immediately prior to the period presented.
- G) Represents the adjustment to reflect the issuance of the \$1.7 billion of New Senior Notes net of the discount of \$8 million, the retirement of the Old Senior Notes of \$1.55 billion and the initial draw on the New Credit Facility of \$350 million, including interest expense and commitment fees.
- H) Represents the adjustment to record the removal of depreciation, capital expenditures, the sale of certain assets and the scheduled principal amortization of notes to reflect the deconsolidation of the Non-Controlled Subsidiaries.
- I) Represents the adjustment to record depreciation expense and capital expenditures related to the Merger, including the scheduled principal amortization of notes.
- J) Represents the adjustment to record depreciation expense related to the acquisition of the Private Partnerships.
- K) Represents the adjustment to reflect the decrease in interest income, net of tax, as a result of the estimated \$225 million Special Dividend and the interest expense for the \$75 million draw on the New Credit Facility.
- L) Represent the adjustment to revenues to reflect lease income and remove hotel revenues and management fees.(/1/)
- M) Represents the adjustment to reduce depreciation expense related to the sale of certain furniture and equipment to the Non-Controlled Subsidiary, including the scheduled principal amortization of notes.
- N) Represents the adjustment to the income tax provision to reflect the REIT Conversion.
- 0) Represents the adjustment to eliminate non-recurring expenses incurred in connection with the REIT Conversion.
- (1) The amount of the earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

# UNAUDITED PRO FORMA BALANCE SHEET SEPTEMBER 11, 1998 NO PARTNERSHIP PARTICIPATION (IN MILLIONS, EXCEPT OP UNITS AMOUNTS)

ACQUISITIONS
DISPOSITIONS AND OTHER
ACTIVITIES

	HOST MARRIOTT CORPORATION HISTORICAL	MENTS	MARRIOTT HOTELS	ACQUI- SITION	ITIONS	
ASSETS						
Property and equipment,						
net Notes and other	\$5,937	\$(649)	\$5,288	\$1,450	\$	\$
receivables, net	32	(3)	29	63		
Due from managers Investments in	88	(5)	83	5		
affiliates	18		18			
Other assets	319	2	321			
Receivable from Lessee for working capital Cash, cash equivalents and short-term marketable						
securities	575	(27)	548	(262)		
		 Φ(693)	 ФС 207	т. обе	\$	
	\$6,969 =====	\$(682) =====	\$6,287 =====	\$1,256 =====	===== Φ	\$ ====
LIABILITIES AND EQUITY	<b>44</b> 004	<b>(040)</b>	<b>0</b> 4 044	Ф 000	•	•
Debt(I) Convertible debt obligation to Host	\$4,224	\$(213)	\$4,011	\$ 600	\$	\$
Marriott Corporation Accounts payable and		567	567			
accrued expenses Deferred income taxes	70 526	(23) (62)	47 464			
Other liabilities	447	(10)	437			
Total lightlities	 F 267	250	 F F26	600		
Total liabilities Convertible Preferred	5,267	259	5,526	600		
Securities Limited Partner interests of third parties at redemption value (on a pro forma basis 47.1 million OP Units	550	(550)				
outstanding)(J) Equity General Partner (on a pro forma basis .2 million OP Units outstanding)(J) Limited Partner interests of Host REIT (on a pro forma basis 204.3 million OP Units		(004)		656		
outstanding)(J)	. 1,152	(391)	761 			
	\$6,969	\$(682)	\$6,287	\$1,256	\$	\$
Book value per OP Unit	=====	====	=====	=====	=====	====

## MERGERS AND REIT CONVERSION ACTIVITIES

	С		D	E	J	F	G		K
				EARNINGS			DEFERRED		REIT
	NON-		PRIVATE	& PROFITS	CONTRI-	LEASE	TAX		2000
	CONTROLLED		PARTNER-	DISTRIBU-	BUTION TO	CONVER-	ADJUST-	PR0	PR0
	SUBSIDIARIES	MERGERS	SHIPS	TION(1)	CRESTLINE	SION	MENT	FORMA	FORMA
ASSETS									
Property and equipment,									
net	\$(501)	\$	\$ 61	\$	\$	\$	\$	\$6,298	\$4,848
Notes and other									
receivables, net	234							326	263
Due from managers	(8)					(67)		13	13
Investments in									
affiliates	204							222	222
Other assets	(15)		(11)					295	295

marketable	(11)						
securities (16)	(11)	(150)	(15) 7			101	363
\$(102) \$ ==================================			. ( - )		\$		
LIABILITIES AND EQUITY						\$4,544	
Marriott Corporation Accounts payable and						567	567
accrued expenses (2) Deferred income taxes (16) Other liabilities 58	 (6)	  		  261	(173) 	45 275 750	45 448 707
Total liabilities (102) Convertible Preferred	(6)	75		261	(173)	6,181	5,711
Securities							
outstanding)(J) Equity General Partner (on a pro forma basis .2 million OP Units outstanding)(J) Limited Partner interests of Host REIT (on a pro forma basis 204.3 million OP Units	45					701	45
outstanding)(J)		(225)	(8)	(261)	173	440	310
		(150) S	\$ (8) <b>\$</b>			\$7,322 =====	•
Book value per OP Unit			<b></b>	<del>-</del>	<b>-</b>	\$ 4.53	\$ 1.71

See Notes to the Unaudited Pro Forma Balance Sheet.

#### NOTES TO UNAUDITED PRO FORMA BALANCE SHEET NO PARTNERSHIP PARTICIPATION

- A. Represents the adjustment to record the spin-off of Crestline and:
- Reduce property and equipment by \$649 million
- Reduce receivables by \$3 million related to certain Crestline notes held by the Operating Partnership Reduce due from managers by \$5 million
- . Increase other assets by \$2 million
- Reduce cash, cash equivalents and short-term marketable securities by \$27 million
- Reduce debt by \$213 million
- Reduce accounts payable and accrued expenses by \$23 million
- Reduce deferred income taxes by \$62 million
- Reduce other liabilities by \$10 million
- Reduce equity by \$391 million
- Eliminate the \$550 million Convertible Preferred Securities of Host Marriott which remain an obligation of Host REIT
- Record the \$567 million of Convertible Debt Obligation to Host Marriott which is eliminated in consolidation on the historical financial statements of Host Marriott Corporation
- B. Represents the adjustment to record the Blackstone Acquisition of 12 full-service properties (5,520 rooms) and a mortgage note secured by a thirteenth full-service property including the issuance of 43.7 million OP Units as determined through negotiations between the Operating Partnership and Blackstone:
  - Record property and equipment of \$1,450 million
  - Record mortgage note receivable of \$63 million
  - Record increase in due from managers of \$5 million
  - Record the use of cash of \$262 million
  - Record the assumption of mortgage debt of \$600 million
  - Record the issuance of 43.7 million OP Units with an estimated fair value of \$656 million using an assumed Host Marriott stock price of \$15.00 (which includes the value of Crestline since Blackstone will receive shares of Crestline in addition to OP Units)

The purchase price of the Blackstone properties and mortgage note was determined based on the estimated fair value of the 43.7 million OP Units to be issued. The number of units to be issued will not increase or decrease depending on the stock price of Host Marriott at the time of closing of the acquisition.

- C. Represents the adjustment to record the investment in the Non-Controlled Subsidiaries and to reflect the sale of certain hotel furniture and equipment to the Non-Controlled Subsidiary:
  - Record decrease in property and equipment of \$501 million, including \$180 million of hotel furniture and equipment sold to the Non-Controlled
  - Record receivable from Non-Controlled Subsidiaries for the furniture and equipment loan of \$180 million, and transfer of other notes totaling \$54 million
  - Record decrease in due from managers of \$8 million
  - Record investment in the Non-Controlled Subsidiaries of \$204 million
  - Record decrease in other assets of \$15 million
  - Record decrease in cash of \$16 million
  - Record decrease in debt of \$142 million of debt transferred to the Non-Controlled Subsidiaries.
  - Record decrease in accounts payable and accrued expenses of \$2 million
  - Record decrease in deferred taxes of \$16 million
  - Record increase in other liabilities of \$58 million
- D. Represents the adjustment to record the purchase of the remaining minority interests in four Private Partnerships:
  - Record property and equipment of \$61 million
  - Record decrease in other assets of \$11 million

- . Record use of cash of \$11 million
- Record decrease in minority interest liabilities of \$6 million
- . Record the issuance of 3.6 million OP Units totaling approximately \$45
- E. Represents the estimated \$225 million cash payment of the Special Dividend to shareholders of Host Marriott, which when combined with the value of the Crestline common stock (estimated to be approximately \$1.30 per share for a total Initial E&P Distribution of approximately \$2.20 per share) will represent the Initial E&P Distribution, by a draw on the New Credit Facility of \$75 million and available cash of \$150 million. The aggregate value of the Crestline common stock and the cash or other consideration to be distributed to Host stockholders (and the Blackstone Entities) in connection with the Initial E&P Distribution is currently estimated to be approximately \$525 million to \$625 million of which approximately \$200 to \$300 million is expected to be represented by the Special Dividend.(/1/)
- F. Represents the adjustment to record the transfer of working capital to Crestline related to the leasing of the Operating Partnership's hotels by decreasing working capital and recording a receivable from the lessee of \$67 million and the adjustment to record deferred revenue of \$261 million in connection with the application of EITF 98-9 to the Operating Partnership's rental income.
- G. Represents the adjustment to record the effect on deferred taxes for the change in tax status resulting from the REIT Conversion by decreasing deferred taxes and increasing equity by \$173 million.
- H. The Company's pro forma aggregate debt maturities at June 19, 1998, excluding \$8 million of capital lease obligations and the \$8 million debt discount recorded in conjunction with the Senior Note Refinancing, are (in millions):

1998	 \$ 272
1999	 30
2000	
2001	
2002	 150
Thereafter	 4,450
	\$5,111
	======

I. The number of OP Units includes the following (in millions):

Limited	Partner	interests	of	Host REIT	204.3
General	Partner	interests	of	Host REIT	0.2
Limited	Partner	interests	of	Private Partnerships	3.4
Limited	Partner	interests	of	Blackstone Group	43.7
Total	OP Units	8			251.6

- J. Represents the adjustment to record the contribution of \$15 million in cash to Crestline as a reduction in equity and to record the sale of an investment of approximately \$7 million in a joint venture which holds a mortgage note of approximately \$130 million from a consolidated subsidiary of Host.
- K. The "REIT 2000 Pro Forma" assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur, Host does not become a REIT until January 1, 2000 and no Notes issued. The amounts reflect the "Pro Forma" column less the amounts in column B--"Blackstone Acquisition" and the reduction in other liabilities and increase in equity for the deferred revenue of \$43 million related to the application of EITF 98-9 to the Blackstone properties.
- (4) ----
- (1) The amount of earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

# UNAUDITED PRO FORMA STATEMENT OF OPERATIONS FIRST THREE QUARTERS 1998 NO PARTNERSHIP PARTICIPATION (IN MILLIONS, EXCEPT PER OP UNIT AMOUNTS AND RATIOS)

				ACQUISITIO	NS, DISPO	SITIONS AN	D OTHER ACT	IVITIES	6	
		Α		В		C	E			
	HOST MARRIOTT CORPORATION HISTORICAL	DISTRIBUTION ADJUSTMENT	HOST MARRIOTT HOTELS HISTORICAL	BLACKSTONE ACQUISITION	ACQL	998 JISI- IONS	DISPOSITIO	NS	BOND REFINANCI	NG
REVENUE Rental										
revenues Hotel revenues Equity in earnings	\$ 922	\$ 	\$ 922	Ψ	\$ 122	43	\$	(6)	\$	
(losses) of affiliates	1		1							
Other revenues	117 	(58)	59 		 			(50)		
Total revenues	1,040	(58)	982		122	43		(56)		
OPERATING COSTS AND EXPENSES										
Hotels Other	502 45	(30)	502 15		70 	23		(3)		
Total operating										
costs and	E 4.7	(20)	517		70	23		(2)		
expenses	547 	(30)						(3)		
OPERATING PROFIT	493	(28)	465		52	20		(53)		
Minority interest	(36)		(36)			(1)		1		
Corporate expenses	(33)	3	(30)							
REIT Conversion expenses	(14)		(14)							
Interest expense	(245)	(8)	(253)		(36)	(1)		1		(17)
Dividends on Convertible										
Preferred Securities Interest	(26)	26								
income	36	2	38		(6)	(18)		(1)		
Income (loss) before income										
taxes Benefit	175	(5)	170		10			(52)		(17)
(provision) for income taxes	(75)	3	(72)		(4)			21		7
Income (loss) before										
extraordinary items	\$ 100	\$ (2)	\$ 98	\$	6 \$		\$	(31)	\$	(10)
Basic loss per	====	====	====	======			======	` '	=====	` ,
OP Unit Ratio of										
earnings to fixed charges	1.7x		1.8x							
Deficiency of earnings to	====		====							
fixed charges										
				CONVERSION						
	Н	•	]	K	0/M	I/L	N		P REIT	
	NON - CONTROLLED SUBSIDIARY	PRI\ MERGERS PARTNE	/ATE	EARNINGS & PROFITS TRIBUTION(1)	OTHER REIT ACTIVITI	LEASE CONVER- IES SION	INCOME TAX ADJUSTMENT	PRO FORMA	2000 PRO FORMA	
DEVENUE										
REVENUE Rental	•	<b>.</b>		•	<b>*</b>		•	<b>.</b>	ф 225	
revenues Hotel revenues Equity in	\$ (40)	\$		\$ 	\$ 	\$ 410 (896)	\$ 	\$ 410 145	\$ 338 145	
earnings										

(losses) of affiliates Other revenues	(17) (5)	 	 	 	 			(16) 4	(16)
Total revenues	(62)					(486)		543	471
OPERATING COSTS AND EXPENSES Hotels	(27)		2			(221)		346 15	288 15
Total operating costs and expenses	(27)		2			(221)		361	303
OPERATING PROFIT Minority	(35)		(2)			(265)		182	168
interest Corporate	22		1					(13)	(13)
expenses REIT Conversion								(30)	(30)
expenses Interest					14				
expense Dividends on Convertible Preferred	6			(4)				(304)	(268)
Securities Interest									
income	1			(5)	(2)	12		19	25
Income (loss) before income taxes Benefit (provision) for income taxes	(6) 6		(1)	(9) 4	12 (5)	(253) 101	 (51)	(146) 7	(118) 6
Income (loss) before extraordinary items	\$ 0	\$	\$ (1)	\$ (5)	\$ 7	\$(152)	\$ (51)	\$(139)	` ,
Basic loss per OP Unit	=======	=======================================	=======================================	==============	=======	====== :	======	\$(.55) ======	\$(.54)
Ratio of earnings to fixed charges								N/A	N/A
Deficiency of earnings to fixed charges								\$(129)	\$(101)

See Notes to the Unaudited Pro Forma Statements of Operations.

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS FISCAL YEAR 1997 NO PARTNERSHIP PARTICIPATION (IN MILLIONS, EXCEPT PER OP UNIT AMOUNTS AND RATIOS)

AND EXPENSES

	ACQUISITIONS,	DISPOSITIONS	AND OTHER	ACTIVITIES
ност	В	С	D	Е

		Α			В	С	D	Е		
	HOST MARRIOTT   CORPORATION HISTORICAL	DISTRIBUTION ADJUST- MENT	HOTELS	BLACK		1998 ACQUISITIONS	1997 ACQUISITIONS	S DISPOSITI	IONS	
REVENUE Rental revenues Hotel revenues Equity in	\$ 1,093	\$ 	\$ 1,093	\$	 148	\$ 112	\$ 89	\$ (23)	)	
earnings of										
affiliates Other revenues	5 49 	(37)	5 12	2						
Total revenues	1,147	(37)	1,110		 148	112	89	(23)	)	
OPERATING COSTS AND EXPENSES										
Hotels	649		649		101	62	42	(10)	)	
Other	49	(20)	29							
Total operating costs and	600	(20)	670		101	62	42	(10)		
expenses	698	(20)	678		101 	62	42	(10)	)	
OPERATING PROFIT	449	(17)	432	2	47	50	47	(13)	)	
Minority interest Corporate	(32)		(32	2)		(4)	5	(1)	)	
expenses Interest	(47)	2	(45	5)						
expense Dividends on Convertible	(302)	(23)	(325	5)	(48)	(12)	(12)	3		
Preferred Securities Interest	(37)	37								
income	52		52		(7)	(14)	(12)			
<pre>Income (loss) before income taxes Benefit</pre>	83	(1)	82		(8)	20	28	(11)	)	
(provision) for income taxes	(36)	1	(35	-	3	(8)	(10)	4		
Income (loss) before extraordinary items	\$ 47	\$	\$ 47		(5)	\$ 12	\$ 18	\$ (7)	)	
Basic earnings per OP Unit Ratio of earnings to	=====	====	=====	==	===	====	====	====		
fixed charges	1.3x =====		1.3x							
		MER	GERS AND	REIT CON	VERSION	I ACTIVITIES				
	F/G	Н		J	K		I/L	N		Р
		G SUBSIDIARY	MERGERS	SHIPS	DISTR BUTION		S SION A	INCOME TAX ADJUSTMENT	PRO FORMA	REIT 2000 PRO FORMA
REVENUE										
Rental revenues Hotel revenues Equity in	\$ 	\$ (53)	\$	\$	\$		\$ 882 (1,173)	\$	\$ 882 193	\$745 193
earnings of affiliates Other revenues		(21) (9)							(16) 3	3
Total revenues		(83)					(291)		1,062	925
OPERATING COSTS									-,	

Hotels		(26) (18)		2		 	(247)	 	573 11	486 11
Total operating costs and expenses		(44)		2			(247)		584	497
OPERATING PROFIT Minority		(39)		(2)			(44)		478	428
interest Corporate		23		1					(8)	(8)
expenses Interest		1							(44)	(44)
expense Dividends on Convertible Preferred	(48)	9			(6)				(439)	(391)
Securities										
<pre>Interest income</pre>	(3)	1			(8)		17		26	33
Income (loss) before income taxes Benefit	(51)	(5)		(1)	(14)		(27)		13	18
<pre>(provision) for income taxes</pre>	20	5			6		11	3	(1)	(1)
Income (loss) before extraordinary items	\$ (31)	\$	\$	\$ (1)	\$ (8)	\$	\$ (16)	\$ 3	\$ 12	\$ 17
=	Ψ (01)	========	======	=======	=======	=======	=======================================	:======	======	=====
Basic earnings per OP Unit									\$ .05	\$.08
Ratio of earnings to fixed charges									1.1x	
. Inca onar goot in									======	

See Notes to the Unaudited Pro Forma Statements of Operations.

## NOTES TO UNAUDITED PRO FORMA STATEMENTS OF OPERATIONS NO PARTNERSHIP PARTICIPATION

- A. Represents the adjustment to reduce revenues, operating expenses, corporate expenses, interest expense, interest income and income taxes for the spin-off of Crestline.
- B. Represents the adjustment to record the historical revenues, operating expenses, interest expense, income taxes and to reduce interest income associated with the acquisition of the equity and debt interests for the Blackstone Acquisition.
- C. Represents the adjustment to record the historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income associated with the 1998 acquisition of, or purchase of controlling interests in, 11 full-service properties.
- D. Represents the adjustment to record historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income associated with the 1997 acquisition of, or purchase of controlling interests in, 18 full-service properties.
- E. Represents the adjustment to record historical revenues, operating expenses, minority interest, interest expense, income taxes and to reduce interest income for the 1998 sale of the New York Marriott East Side and the Napa Valley Marriott, including the elimination of the non-recurring gains on the sales totalling \$50 million and related taxes of \$20 million in 1998.
- F. Represents the adjustment to reduce the interest expense, interest income and income taxes associated with the refinancing or payoff of mortgage debt for three full-service properties (Marriott's Orlando World Center, the Philadelphia Marriott and the San Francisco Marriott).
- G. Represents the adjustment to record interest expense and related amortization of deferred financing fees and reduce interest income and income taxes as a result of the Senior Note Refinancing. The adjustment excludes the extraordinary loss of \$148 million, net of taxes, related to the Senior Note Refinancing resulting from the write-off of deferred financing fees and the payment of bond tender and consent fees.
- H. Represents the adjustment for revenues, operating expenses, minority interest, interest expense, corporate expenses, income taxes and interest income to deconsolidate the Non-Controlled Subsidiaries and reflect the Operating Partnership's share of income as equity in earnings of affiliate.
- I. Represents the adjustment to reduce depreciation expense of \$18 million and \$26 million for First Three Quarters 1998 and fiscal year 1997 related to certain furniture and equipment sold to the Non-Controlled Subsidiary, record interest income of \$9 million and \$13 million for First Three Quarters 1998 and fiscal year 1997 earned on the 7%, \$180 million in notes from the Non-Controlled Subsidiary and reduce the lease payment to the Operating Partnership from the Lessee.
- J. Represents the adjustment to record additional depreciation expense and the decrease in minority interest expense related to the purchase of the remaining minority interests in the Private Partnerships.
- K. Represents the adjustment to reduce interest income, income taxes and record interest expense for the estimated \$225 million cash payment of the Special Dividend to shareholders of Host Marriott including a draw on the New Credit Facility of \$75 million.(/1/)
- L. Represents the adjustment to remove hotel revenues and management fees and other expenses of \$221 million and \$247 million, respectively, for First Three Quarters 1998 and Fiscal 1997 and record rental revenues associated with the leasing of certain hotel properties to Crestline and other lessees and interest income of \$3 million and \$5 million for First Three Quarters 1998 and fiscal year 1997 earned on the 6%, \$85 million in notes from Crestline. First Three Quarters 1998 included a \$261 million reduction to rental income to record deferred

revenue for percentage rents in accordance with EITF 98-9. Management believes the change to the lease structure described above will not impact hotel operating results because the hotel manager and asset management function will remain unchanged. Rental revenues under the Leases are based on the greater of Percentage Rent or Minimum Rent. Total Rent in the pro forma statement of operations is calculated based on the historical gross sales of the property and the negotiated pay rates and thresholds by property as if the leases were entered into on the first day of fiscal year 1997. There are generally three sales categories utilized in the rent calculation: rooms, food and beverage and other. For rooms and food and beverage, there are three tiers of rent with two thresholds, while the other category generally has two tiers of rent and one threshold. The percentage rent thresholds are increased annually on the first day of each year after the initial lease year based on a blended increase of the Consumer Price Index ("CPI") and a wage and benefit index. For purposes of the pro formas, 1997 is the assumed initial lease year and the blended increase applied to the thresholds at January 3, 1998 is assumed to be 3%. Minimum rent is expressed as a fixed dollar amount that increases annually on the first day of each year after the initial lease year as 50% of the CPI increase. Accordingly, the 1998 rent thresholds and minimum rent included in the pro formas were adjusted as of January 3, 1998 for the 1997 increases in the indices. Rental revenue is recognized only for leases to be executed with Crestline at or prior to the completion of the REIT Conversion. The execution of the leases is dependent upon the distribution of the Crestline common stock to the Host's stockholders, which is subject to contingencies that are outside the control of the Operating Partnership, including consent of shareholders, lenders, debt holders, partners and ground lessors of Host. The Operating Partnership believes that negotiations with third parties to complete the REIT Conversion will not result in any material change to the leases. See page F-120 for a table detailing gross sales, minimum rent and total rent for all full-service properties to be leased and summarized amounts for the limitedservice properties to be subleased.

	FISC	AL YEAR 1	1997	FIRST THREE QUARTERS 1998				
PROPERTY	GROSS SALES	MINIMUM RENT	TOTAL RENT	GROSS SALES	MINIMUM RENT	RENT	TOTAL RENT RECOGNIZED	
				N MILLIONS				
Grand Hotel Resort and								
Golf Club	\$ 23.4		\$ 4.2			\$ 3.7		
Scottsdale Suites The Ritz-Carlton,	11.9		5.0	8.2	2.1	3.4		
Phoenix	23.3		7.2	17.3	3.2	5.5		
Coronado Island Resort	22.0		2.1	16.2	1.5	3.6		
Costa Mesa Suites	9.7		3.3	7.2	1.3	2.3		
Manhattan Beach	16.3		4.8	12.2	1.7	3.6		
Marina Beach	21.1		7.1	16.9	3.2	6.2		
Newport Beach	33.5		8.7	24.0	3.9	6.8		
Newport Beach Suites	11.0		4.1	8.0	1.8	3.0		
Ontario Airport San Diego Marriott Hotel	12.1	1.8	3.4	8.3	1.3	2.2	1.3	
and Marina	103.3	38.0	39.6	78.6	26.7	31.1	26.7	
Valley	16.7	3.4	5.1	12.6	2.4	5.6	2.5	
San Francisco Airport	43.8		13.2	32.2	5.8	9.5		
San Francisco								
Fisherman's Wharf	17.8	4.0	6.4	12.1	2.8	4.3	2.8	
San Francisco Moscone								
Center	120.2	20.7	37.9	90.5	14.6	28.5	14.6	
The Ritz-Carlton, Marina								
del Rey The Ritz-Carlton, San	32.4	5.5	10.8	23.4	3.9	7.9	3.9	
Francisco	50.1	9.6	14.7	34.2	6.7	10.3	6.7	
Torrance	20.5	2.3	3.5	15.0	1.6	5.1	1.6	
Denver Southeast	21.5	3.0	6.2	14.9	2.1	4.1	2.1	
Denver Tech Center	26.8	5.1	8.3	20.1	3.6	6.0	3.6	
Denver West	13.7	1.8	4.0	9.6	1.2	2.7	1.7	
Marriott's Mountain								
Resort at Vail	17.6		5.1	14.1	2.1	4.5		
Hartford/Farmington	18.4		4.7	13.4	2.4	3.5		
Hartford/Rocky Hill	11.6		2.7	8.5	1.1	2.0		
Fort Lauderdale Marina	28.5		7.9	20.4	3.0	5.7		
Jacksonville	11.8		3.6	8.0	1.2	2.4		
Miami Airport	29.7	3.9	8.4	21.6	2.8	6.1	2.9	

	FISCAL YEAR 1997			FIRST THREE QUARTERS 1998			
PROPERTY	GROSS SALES	MINIMUM RENT	TOTAL RENT	GROSS SALES	MINIMUM RENT	RENT	TOTAL RENT RECOGNIZED
				MILLIONS			
Palm Beach Gardens Singer Island (Holiday			\$ 3.7			\$ 3.0	\$ 1.4
Inn)	6.6	1.4	2.5	5.2	1.0	2.1	1.0
Tampa Airport	17.1	1.6	3.5	13.1	1.1	2.7	1.1
Tampa Westshore	15.0	1.8	3.6	10.8	1.3	2.6	1.3
The Ritz-Carlton, Naples	66.4		23.3	53.1		18.0	12.7
Atlanta Midtown Suites	10.5	1.8	3.5	7.8	1.3	2.6	1.3
Atlanta Norcross	7.6		1.7	5.6	0.7	1.2	0.7
Atlanta Northwest Atlanta Perimeter JW Marriott Hotel at	14.9 16.6		4.3 4.5	11.3 12.6	1.9 1.7	3.3 3.5	1.9 1.7
LenoxThe Ritz-Carlton,	24.8	3.7	6.8	17.7	2.6	5.0	2.6
Atlanta	30.2	5.8	8.8	21.7	4.1	6.8	4.1
Buckhead	49.3	13.1	16.3	35.8	9.2	11.7	9.2
Chicago/Deerfield Suites	10.2		3.1	7.4	1.3	2.3	1.3
Chicago/Downers Grove Suites	9.0	1.8	2.9	6.7	1.3	2.2	1.3
Chicago/Downtown							
Courtyard	16.3	3.1	4.9	12.2	2.2	3.9	2.2
Chicago O'Hare	40.0		11.5	28.8	3.9	8.2	
South Bend	9.9	1.1	2.1	7.0	0.8	1.5	0.8
Bethesda Gaithersburg/Washingtonian	23.2		5.6	17.3	2.2	4.1	2.2
Center	13.2	2.4	3.8	9.7	1.7	2.8	1.7
Boston/Newton	27.4		7.8	19.1	3.4	5.5	3.4
Detroit Romulus	8.8		1.8	6.6	0.8	1.4	0.8
Dearborn	25.7	3.6	5.5	17.7		4.0	2.5
Minneapolis/Bloomington Minneapolis City Center	20.2 27.5	3.3 3.7	6.5 7.5	13.8 20.4	2.3 2.4	4.7 5.2	3.1 2.4
Minneapolis Southwest	14.9	2.7	4.8	10.1	1.9	4.0	1.9
Kansas City Airport	14.9	1.7	3.7	9.9		2.5	1.9
St. Louis Pavilion	27.5	6.1	6.5	18.5	4.3	4.3	4.3
Nashua	7.5	0.8	1.3	5.3	0.5	0.9	0.5
Newark Airport	39.4		11.8	29.2		8.6	4.6
Park Ridge	16.0	2.5	4.0	11.9	1.7	4.2	1.7
Saddle Brook	10.7		2.1	7.8	0.9	1.7	0.9
Albany New York Marriott	18.5	3.5	6.1	12.4	2.5	5.2	2.7
Financial Center New York Marriott	39.6	7.7	13.2	29.1	5.4	10.1	5.4
Marquis Marriott World Trade	210.3	40.0	60.8	155.4	29.7	47.6	29.7
Center	65.4		19.4	49.1	8.6	14.9	8.6
Charlotte Executive Park	14.0	2.3	3.7	9.8	1.6	2.6	1.6
Raleigh Crabtree Valley	14.9		3.9	10.9	1.7	2.8	1.7
Oklahoma City	15.6		3.8	10.4	1.4	2.4	1.4
Oklahoma City Waterford Portland Philadelphia (Convention	9.1 26.4		2.7 7.5	6.1 17.6	1.0 2.9	1.7 4.8	1.0 2.9
Center)	80.7	14.2	25.0	58.2	10.0	17.8	10.0
Philadelphia Airport	25.0	4.1	7.6	18.6	2.9	5.6	2.9
Pittsburgh City Center	16.4	1.9	3.0	11.1	1.3	2.2	1.3
Memphis	10.6	1.5	3.2	5.7	1.0	1.8	1.0
Dallas/Fort Worth	28.9	5.9	9.3	21.9	4.1	7.0	4.1
Dallas Quorum	25.7	4.2	8.2	18.3	3.0	5.8	3.7
El Paso	11.6	0.9	2.3	7.8	0.6	1.4	0.6
Houston Airport	21.6	2.8	6.0	16.9	2.0	4.6	2.7
JW Marriott Houston	27.2	5.0	8.0	20.1	3.5	5.9	3.5
Plaza San Antonio	13.8	2.9	4.6	9.7	2.0	3.3	2.0
San Antonio Riverwalk	29.3		10.3	21.7	4.3	7.6	4.4
Salt Lake City	28.5	5.6	9.5	21.1	3.9	7.2	4.2
Dulles Airport	14.6	1.8	4.0	10.9	1.2	3.0	1.7
Key Bridge	29.4		10.2	21.2		7.4	3.9

	FISCAL YEAR 1997			FIRST THREE QUARTERS 1998			
PROPERTY	GROSS SALES	MINIMUM RENT	TOTAL RENT	GROSS SALES	MINIMUM RENT	TOTAL RENT	
				 MILLIONS)			
			`	,			
Norfolk Waterside Pentagon City Residence	\$ 18.1		\$ 5.4	,		\$ 3.8	\$ 2.4
Inn The Ritz-Carlton, Tysons	11.7		5.5	8.7		4.2	2.5
CornerWashington Dulles	34.4	5.9	9.8	24.9	4.1	7.3	4.1
Suites	10.3	2.5	4.0	7.8	1.8	3.0	1.8
Westfields	28.0	4.7	7.4	20.3	3.3	5.4	3.3
Williamsburg Washington Metro	12.6	1.8	2.8	9.3	1.3	2.1	1.3
Center	25.2	4.5	7.3	19.2	3.2	5.3	3.2
Calgary	13.4	1.7	1.7	9.8	1.2	2.3	1.2
Toronto Airport	17.1	2.9	5.6	13.0	2.0	4.2	2.0
Toronto Eaton Centre Toronto Delta	21.1	6.1	7.1	16.0	4.3	5.6	4.3
Meadowvale Four Seasons,	16.1	2.6	4.9	10.6	1.9	3.1	1.9
Atlanta(4) Four Seasons,	15.6	5.8	5.9	14.2	4.1	4.5	4.1
Philadelphia(4) Grand Hyatt,	41.1	7.9	12.4	30.6	5.6	10.1	5.6
Atlanta(4)	25.3	10.0	10.0	22.6	7.0	8.2	7.0
Burlingame(4) Hyatt Regency,	47.9	8.8	17.6	39.5	6.2	15.1	9.0
Cambridge(4)	32.4	6.7	11.9	26.8	4.7	10.4	6.1
Reston(4)	30.5	6.5	11.3	24.2	4.5	9.2	4.8
Swissotel, Atlanta(4)	22.2		6.3	17.2		5.8	3.5
Swissotel, Boston(4)	26.8		8.5	20.5		6.9	4.5
Swissotel, Chicago(4)	38.1		15.1	28.9		12.0	7.7
The Drake (Swissotel),	30.1	10.9	13.1	20.9	7.7	12.0	7.7
New York(4) The Ritz-Carlton, Amelia	38.8	11.6	13.6	34.2	8.2	13.4	8.2
Island(4) The Ritz-Carlton,	45.7	10.3	13.4	37.4	7.2	11.1	7.2
Boston(4) Non-Controlled	40.1	6.9	10.5	31.4	4.8	8.8	4.8
Subsidiary Rent		(39.7)	(39.7)		(27.5)	(27.5)	(27.5)
Total Full-service							
Properties	2,790.7	498.2	802.6	2,071.7	351.9	620.9	360.4
Total Courtyard Properties	212.0		59.2			36.8	36.8
Total Residence Inns							
_			20.3	50.6		12.3	12.3
Total	\$3,072.6 ======			\$2,281.6 ======		670.0	\$409.5 =====
Less: Deferred rent under EITF 98-9						260.5	
Total rent recognized						\$409.5	
. 55592250111111111111							

- M. Represents the adjustment to eliminate interest income recorded for the \$92 million note receivable contributed to Crestline for First Three Quarters 1998
- N. Represents the adjustment to the income tax provision to reflect the REIT Conversion.
- O. Represents the adjustment to eliminate non-recurring expenses incurred in connection with the REIT Conversion. Management expects that the total estimated nonrecurring expenses to be incurred will be approximately \$50 million
- P. The "REIT 2000 Pro Forma" reflects the adjustment to eliminate the revenues, operating expenses, interest expense and interest income which assumes that the REIT Conversion occurs on January 1, 1999, the Blackstone Acquisition does not occur, Host does not become a REIT until January 1, 2000 and no Notes issued.
- (1) The amount of earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its

shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

### UNAUDITED PRO FORMA STATEMENT OF CASH FLOWS FIRST THREE QUARTERS 1998 NO PARTNERSHIP PARTICIPATION (IN MILLIONS)

				ACQUISITIONS	S, DISPOSITION	NS AND OTHER	ACTIVITIES
	HOST	Α	H0ST	В	С	E	G
	MARRIOTT CORPORATION HISTORICAL	DISTRIBUTION ADJUST- MENTS	MARRIOT HOTELS HISTORICAL		1998 ACQUISITIONS	DISPOSITIONS	BOND REFINANCING
OPERATING ACTIVI-							
TIES Income before extraordinary items Adjustment to	\$ 100	\$ (2)	\$ 98	\$ 6	\$	\$ (31)	\$ (10)
reconcile to cash provided by operations: Depreciation and							
amortization Income taxes Gains on sales of hotel	184 50	(17) 	167 50	42 	11 		
properties Equity (earnings) losses of	(50)		(50)			50	
affiliates Changes in operating		(1)	(1)				
accounts	(33)	(2)	(35)				
Other assets	27	4	31				
Cash provided by							
(used in) operations	278	(18)	260	48	11	19	(10)
INVESTING ACTIVI-							
Acquisitions Cash received from sale of	(636)	29	(607)		607		
assets Capital	211		211			(211)	
expenditures Purchases of short-term	(168)	9	(159)	(16)	(5)	2	
marketable securities Sales of short- term marketable	(134)		(134)				
securities	451		451				
Other	3	(89)	(86)			21	
Cash provided by (used in) investing							
activities	(273)	(51)	(324)	(16)	602	(188)	
FINANCING ACTIVI- TIES Issuances of							
debt Scheduled principal	2,004		2,004				1,967
repayments Debt repayments Transfers to Host	(42) (1,750)	3 119	(39) (1,631)			35	(1,250)
Marriott Other	(189)	(63) 	(63) (189)				
OCHOI	(189)		(109)				
Cash provided by (used in) financing							
activities	23	59 	82			35 	717
INCREASE (DECREASE) IN CASH AND CASH							
EQUIVALENTS	\$ 28	\$(10)	\$ 18	\$ 32	\$613	\$(134)	\$ 707
	======	====	=====	====	====	====	======

MERGERS AND REIT CONVERSION ACTIVITIES

	NON-	MERGERS	1	EARNINGS	LEASE	IN	rı	
	CONTROLLED		PRIVATE	& PROFITS DISTRIBUTION(1)		OTHER REIT	TAX ADJUSTMENT	PRO EORMA
OPERATING ACTIVI- TIES Income before								
extraordinary items	\$	\$	\$ (1)	\$ (5)	\$(152)	\$ 7	\$(51)	\$ (139)
Adjustment to reconcile to cash provided by operations: Depreciation and								
amortization Income taxes	(4)		1	 	(18) 112			199 162
Gains on sales of hotel								
properties Equity (earnings)								
losses of affiliates								(1)
operating accounts Other assets					320			285 31
Cash provided by								
(used in) operations	(4)			(5)	262	7	(51)	537
INVESTING ACTIVI- TIES								
Acquisitions Cash received from sale of								
assets Capital								
expenditures Purchases of short-term	1							(177)
marketable securities Sales of short- term marketable								(134)
securities Other								451 (65)
Cash provided by (used in) investing								
activities	1							75
FINANCING ACTIVI-								
Issuances of debtScheduled				(75)				3,896
principal repayments	(2)				(15)			(56)
Debt repayments Transfers to Host								(2,846)
Marriott								(63) (189)
Cash provided by (used in)								
financing activities	(2)			(75)	(15)			742
INCREASE (DECREASE) IN CASH AND CASH								
EQUIVALENTS	\$ (5) ======	\$ ======	\$ ========	\$(80) =======	\$ 247 ======	\$ 7 ======	\$(51) ======	\$1,354 ======

See Notes to the Unaudited Pro Forma Statements of Cash Flows.

## UNAUDITED PRO FORMA STATEMENT OF CASH FLOWS FISCAL YEAR 1997 NO PARTNERSHIP PARTICIPATION (IN MILLIONS)

				ACQU:	IES			
	A HOST		HOST	В	D	С	E	F/G
	MARRIOTT CORPORATION HISTORICAL	DISTRIBUTION ADJUST- MENT	MARRIOTT HOTELS HISTORICAL	BLACKSTONE ACQUISITION	1997 ACQUISITIONS	1998 ACQUISITIONS	DISPOSITIONS	DEBT REPAYMENTS & REFINANCING
OPERATING ACTIVITIES Income before ex-								
traordinary items Adjustment to reconcile to cash provided by	\$ 47	\$	\$ 47	\$ (5)	\$ 18	\$ 12	\$ (7)	\$ (31)
operations: Depreciation and	240	(0)	221	61	17	21	(2)	
amortization	240	(9)	231	61	17	31	(3)	
Income taxes Other	(20) 70	 3	(20) 73				(2)	
Changes in oper-	70	3	13				(2)	
ating accounts	127	(26)	101					
acting accounts		(20)	101					
Cash provided by (used in)								
operations	464	(32)	432	56	35	43	(12)	(31)
INVESTING ACTIVI-								
Acquisitions Cash received from sale of	(596)	237	(359)		359			
assets Purchase of short-term mar-	51		51					
ketable securi- ties	(354)		(354)					
Capital expendi- tures	(160)	2	(158)	(20)	(13)	(13)	4	
Other	13		13				1	
Cash provided by (used in) investing								
activities	(1,046)	239	(807)	(20)	346	(13)	5	
FINANCING ACTIVI-								
TIES Issuances of								
debt Scheduled princi-	857		857					1,188
pal repayments Debt prepay-	(93)	3	(90)					
ments Transfers to Host	(403)		(403)					(1,147)
Marriott		(226)	(226)					
Other	28	(1)	27					
Cash provided by (used in)								
financing activities	389	(224)	165					41
INCREASE (DECREASE) IN CASH AND CASH	ф (102)		φ(210)		****			
EQUIVALENTS	\$ (193) =====	\$ (17) =====	\$(210) =====	\$ 36 ====	\$381 ====	\$ 30 ====	\$ (7) ====	\$ 10 =====
				==== /ERSION ACTIV				
	Н	I	 Ј		K/L	 М		
		1	J		10/ L	• •		

**EARNINGS** 

NON- MERGERS PRIVATE & PROFITS LEASE

CONTROLLED & NOTES PARTNER- DISTRI- OTHER REIT CONVER- TAX PRO
SUBSIDIARY ISSUANCE SHIPS BUTION(1) ACTIVITIES SION ADJUSTMENT FORMA

OPERATING ACTIVITIES Income before extraordinary

itemsAdjustment to reconcile to cash provided by	\$	\$	\$(1)	\$ (8)	\$	\$(16)	\$ 3	\$ 12
operations: Depreciation and amortization Income taxes Other Changes in oper-	(5)  	  	2  	  	  	(26) 11 	  	308 (9) 71
ating accounts  Cash provided by (used in) operations	(5)		1	(8)		(31)	3	101  483
INVESTING ACTIVI- TIES Acquisitions Cash received								
from sale of assets Purchase of short-term mar- ketable securi-	(35)							16
ties	 2 33							(354) (198) 47
Cash provided by (used in) investing activities								(489)
FINANCING ACTIVI-								(409)
TIES Issuances of debt	(3)			(75)				1,967
Scheduled princi- pal repayments Debt prepay-	(6)					(30)		(126)
ments Transfers to Host								(1,550)
Marriott Other								(226) 27
Cash provided by (used in) financing activities	(9)			(75)		(30)		92
INCREASE (DECREASE) IN CASH AND CASH								
EQUIVALENTS	\$(14) ======	\$ =======	\$ 1 =======	\$(83) =====	\$ =======	\$(61) = ====== =	\$ 3 	\$ 86 == ======

See Notes to the Unaudited Pro Forma Statements of Cash Flows.

## NOTES TO UNAUDITED CASH FLOW STATEMENTS NO PARTNERSHIP PARTICIPATION

- A) Represents the adjustment for cash flow amounts related to the distribution of Crestline.
- B) Represents the adjustment to record the depreciation expense and estimated capital expenditures for the Blackstone Acquisition.
- C) Represents the adjustment to record depreciation expense and estimated capital expenditures for the 1998 acquisition of, or purchase of controlling interests in, 11 full-service properties.
- D) Represents the adjustment to record depreciation expense and capital expenditures for the 1997 acquisition of, or purchase of controlling interests in, 18 full service properties. Cash flows from investing activities has also been adjusted as if the period's historical acquisitions occurred immediately prior to the period presented.
- E) Represent the adjustment to record the decrease in depreciation expense, capital expenditures and other investing activities for the sale of the New York East Side Marriott and the Napa Valley Marriott, including the elimination of the non-recurring gains on the sales totaling \$50 million.
- F) Represents the adjustment to reflect the decrease in interest expense associated with the refinancing or payoff of mortgage debt for three full-service properties (Marriott Orlando World Center, the Philadelphia Marriott, and the San Francisco Marriott). Cash from financing activities has also been adjusted as if the periods historical prepayments and issuances of debt occurred immediately prior to the period presented.
- G) Represents the adjustment to reflect the issuance of the \$1.7 billion of New Senior Notes net of the discount of \$8 million, the retirement of the Old Senior Notes of \$1.55 billion and the initial draw on the New Credit Facility of \$350 million, including interest expense and commitment fees.
- H) Represents the adjustment to record the removal of depreciation, capital expenditures, the sale of certain assets and the scheduled principal amortization of notes to reflect the deconsolidation of the Non-Controlled Subsidiaries.
- I) Represents the adjustment to record depreciation expense related to the acquisition of the Private Partnerships.
- J) Represents the adjustment to reflect the decrease in interest income, net of tax, as a result of the estimated \$225 million Special Dividend and the interest expense for the \$75 million draw on the New Credit Facility.
- K) Represent the adjustment to revenues to reflect lease income and remove hotel revenues and management fees.(/1/)
- L) Represents the adjustment to reduce depreciation expense related to the sale of certain furniture and equipment to the Non-Controlled Subsidiary, including the scheduled principal amortization of notes.
- M) Represents the adjustment to the income tax provision to reflect the REIT Conversion.
- N) Represents the adjustment to eliminate non-recurring expenses incurred in connection with the REIT Conversion.
- (1) The amount of the earnings and profit distribution shown reflects only the estimated distribution to be made in connection with the REIT Conversion, and could consist of a combination of cash and other consideration, including possibly securities of Host REIT. The actual amount of the distribution will be based in part upon the estimated amount of Host's accumulated earnings and profits for tax purposes. To the extent that the distributions made in connection with the Initial E&P Distribution are not sufficient to eliminate Host's estimated accumulated earnings and profits, Host REIT will make one or more additional taxable distributions to its shareholders (in the form of cash or securities) prior to the last day of its first full taxable year as a REIT (currently expected to be December 31, 1999) in an amount intended to be sufficient to eliminate such earnings and profits, and the Operating Partnership will make corresponding distributions to all holders of OP Units (including Host REIT) in an amount sufficient to permit Host REIT to make such additional distributions.

#### PRO FORMA FINANCIAL INFORMATION OF HOST REIT

Given the structure of Host Marriott's Consent Solicitation, the Mergers and the REIT Conversion may take a variety of different forms. The variations are dependent in part on the number and identity of the Partnerships that elect to merge and whether limited partners elect to tender their Partnership Interests for OP Units, REIT shares, or Notes in connection with the REIT Conversion.

In light of the number of possible variations, Host REIT and the Company is not able to describe all possible combinations of Hotel Partnerships that could compose Host REIT. However, to assist Shareholders and current Host shareholders in analyzing the Mergers and the REIT Conversion, Host REIT and the Company have prepared unaudited pro forma financial statements of the Operating Partnership to show the impact of the Mergers and the REIT Conversion under various scenarios (see pro forma financial information of the Company--F-108).

Additionally, Host REIT and the Company have prepared pro forma financial statements of Host REIT in order to present the differences between the Operating Partnership and Host REIT. See pro forma financial information of the Company--F-108 for the adjustments to Host's historical financial statements necessary to arrive at Operating Partnership--Pro Forma in the accompanying pro forma financial statements of Host REIT.

The unaudited pro forma balance sheet and statements of operations of Host REIT reflect the following adjustments to the pro forma financial statements of the Operating Partnership.

- . The elimination in consolidation of the convertible debt obligation to Host Marriott of the Operating Partnership and the presentation of Company-obligated Mandatorily Redeemable Convertible Preferred Securities of a Subsidiary Trust Holding Company Substantially All of Whose Assets are the Convertible Subordinated Debentures Due 2026 ("Convertible Preferred Securities") on the balance sheet of Host REIT. Interest expense paid by the Operating Partnership related to the convertible debt obligation to Host Marriott (on a pro forma basis) is eliminated and dividend expense for the Convertible Preferred Securities is reflected on the pro forma statement of operations of Host REIT.
- . The presentation of the Limited Partner interests of third parties in the Operating Partnership as minority interest in Host REIT and the reflection of Operating Partnership income allocable to the third party Limited Partners as minority interest expense of Host REIT.

In addition, the Host REIT pro forma financial information has been adjusted to the "REIT 2000" scenario. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, Host REIT would be subject to federal income taxes and the Blackstone Acquisition may not occur. The adjustments between Host REIT Pro Forma and REIT 2000 Pro Forma reflect adjustments to:

- . Record the tax provision (benefit) and related deferred tax amounts eliminated in the Operating Partnership Pro Forma.
- . Remove amounts related to the Blackstone Acquisition.

The unaudited pro forma financial statements are based upon available information and upon certain assumptions as set forth in the notes to the unaudited pro forma financial statements, that Host REIT believes are reasonable under the circumstances and should be read in conjunction with the unaudited pro forma financial statements of the Operating Partnership and the consolidated financial statements and notes thereto for Host.

The execution of the leases is dependent upon the successful consummation of the REIT Conversion which is subject to contingencies that are outside the control of Host REIT, including consent of shareholders, lenders, debt holders, partners and ground lessors of Host. Host REIT believes that negotiations with third parties to complete the REIT Conversion will not result in any material change to the leases.

### HOST REIT

### UNAUDITED PRO FORMA BALANCE SHEET

## SEPTEMBER 11, 1998 100% PARTICIPATION WITH NO NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	OPERATING PARTNERSHIP PRO FORMA	PRO FORMA ADJUSTMENTS	PRO FORMA	ADDITIONAL PRO FORMA ADJUSTMENTS	REIT 2000 PRO FORMA
ASSETS					
Property and equipment,					
net	\$7,069	\$	\$7,069	\$(1,450)(E)	\$5,619
Notes and other receiv-	. ,		, ,	, ( ) / ( )	, .
ables, net	303		303	(63)(E)	240
Due from managers	14		14		14
Investments in affili-	0.7		0.7		07
ates Other assets	27 338		27 338		27 338
Receivable from Lessee	330		330		330
for working capital	85		85	(5)(E)	80
Cash, cash equivalents				( )( )	
and short-term market-					
able securities	116		116	262 (E)	378
T-4-14-	47.0F0		 #7 050	Φ(4, OF C)	<b></b>
Total assets	\$7,952 =====	\$ ====	\$7,952 =====	\$(1,256) ======	\$6,696 =====
LIABILITIES AND EQUITY					
Debt	\$4,970	\$	\$4,970	\$ (600)(E)	\$4,370
Convertible debt obliga-	+ 1/ - 1 -	•	+ .,	+ ()(-)	+ .,
tion to Host Marriott					
Corporation	567	(567)(A)			
Accounts payable and ac-					
crued expenses	55 275		55	 100 (D)	55
Deferred income taxes Other liabilities	275 700		275 700	189 (D) (43)(E)	464 657
other madification				(43)(L)	
Total liabilities	6,567	(567)	6,000	(454)	5,546
Minority interest Convertible Preferred		354 (B)	354	(221)(E)	133
Securities Limited Partner		550 (A)	550		550
interests of third					
parties at redemption					
value (on a pro forma					
<pre>basis 70.1 million OP units outstanding)</pre>	989	(989)(C)			
Equity	303	(303)(0)			
General Partner (on a					
pro forma basis 0.2					
million OP Units					
outstanding)					
Limited Partner interests of Host					
REIT (on a pro forma					
basis 204.3 million					
OP Units					
outstanding)	396	(396)(C)			
Shareholders' Equity					
(on a pro forma basis					
800 million shares authorized; 204.5					
million issued and					
outstanding)		17 (A)	1,048	(392)(E)	467
		(354)(B)	,	(189)(D)	
		`989´(C)		. , , ,	
		396 (C)			
	47 0E2	 c	47 OF2	e(1 2E6)	46 606
	\$7,952 =====	\$ ====	\$7,952 =====	\$(1,256) ======	\$6,696 =====

### HOST REIT

### UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

## FIRST THREE QUARTERS 1998 100% PARTICIPATION WITH NO NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	OPERATING PARTNERSHIP PRO FORMA			ADDITIONAL PRO FORMA ADJUSTMENTS	REIT 2000 PRO FORMA
REVENUE					
Rental revenues Other revenues	\$ 540 2	\$ 	\$ 540 2	\$(72)(F) 	\$ 468 2
Total revenues	542		542	(72)	470
OPERATING COSTS AND EX- PENSE					
Hotels Other	394 15		394 15	(58)(F) 	336 15
Total operating costs and ex-					
penses	409		409	(58)	351
OPERATING PROFIT Minority interest Corporate expenses Interest expense	133 (14) (30) (328)	 52  26	133 38 (30) (302)	(14) (39)(F)  36 (F)	119 (1) (30) (266)
Dividends on Convertible Preferred Securities Interest income	24	(26)	(26)	6 (F)	(26) 30
Income (loss) before income taxes Benefit (provision) for	(215)	52	(163)	(11)	(174)
income taxes	11	(3)	8	63 (D)	71
<pre>Income (loss) before ex- traordinary items</pre>	\$(204)	\$ 49	\$(155) =====		\$(103)
Diluted loss per share	====	====	===== \$(.76) =====	====	===== \$(.50) =====

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

## FISCAL YEAR 1997 100% PARTICIPATION WITH NO NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	OPERATING PARTNERSHIP PRO FORMA	PRO FORMA ADJUSTMENTS	HOST REIT PRO FORMA	ADDITIONAL PRO FORMA ADJUSTMENTS	REIT 2000 PRO FORMA
REVENUE  Rental revenues  Equity in earnings  (losses) of affili-	\$1,135	\$	\$1,135	\$(137)(F)	\$ 998
atesOther revenues	(7) 3		(7) 3		(7) 3
Total revenues	1,131		1,131	(137)	994
OPERATING COSTS AND EX- PENSE					
HotelsOther	600 11		600 11	(87)(F) 	513 11
Total operating costs and ex-penses	611		611	(87)	524
<b>P</b>					
OPERATING PROFIT Minority interest Corporate expenses Interest expense Dividends on Convertible	520 (10) (44) (468)	(7)(B)  38 (A)	520 (17) (44) (430)	(50) 5 (F)  48 (F)	(44)
Preferred Securities Interest income	 29	(37)(A) 	(37) 29	7 (F)	(37) 36
Income (loss) before in-					
come taxes	27	(6)	21	10	31
income taxes	(1)	(D)	(1)	(12)(D)	(13)
Income (loss) before extraordinary items	\$ 26 =====	\$ (6) ====	\$ 20 =====	\$ (2) =====	\$ 18 =====
Diluted earnings per share			\$ .09 =====		\$ .08 =====

#### NOTES TO UNAUDITED PRO FORMA FINANCIAL STATEMENTS

#### 100% PARTICIPATION WITH NO NOTES ISSUED

#### A. Represents the following adjustments:

- . Adjustment to remove \$567 million of convertible debt obligation to Host Marriott which eliminates in consolidation.
- . Record the \$550 million of convertible preferred securities held by Host RFIT.
- . Record \$17 million adjustment to equity for the elimination in consolidation of the convertible debt obligation to Host Marriott.
- . Remove interest of \$26 million and \$38 million, respectively, on \$567 million of convertible debt obligation to Host Marriott.
- . Record dividends of \$26 million and \$37 million, respectively, on the \$550 million of convertible preferred securities.
- B. Represents the adjustment to present the Limited Partner interests of third parties in the Operating Partnership as minority interest in Host REIT and the reflection of Operating Partnership income allocable to the third party Limited Partners as minority interest expense of Host REIT as follows:
  - . Record \$354 million minority interest liability representing Limited Partner interests of third parties (70.1 million OP Units out of total OP Units of 274.6 million) pro rata share of total combined Operating Partnership equity and Limited Partner interests of third parties at redemption value.
  - . Record adjustment to shareholders' equity of \$354 million to record the Limited Partner interests of third parties as minority interest.
  - . Record minority expense based on the Limited Partner interests of third parties pro rata share of Operating Partnership net income (70.1 million OP Units out of total OP Units of 274.6 million).
- C. Represents the adjustment to eliminate the Partner's Capital of the Operating Partnership and record the Common Stock, Additional Paid-in Capital, and Retained Earnings of Host REIT.
- D. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, Host REIT would be subject to federal income taxes. This adjustment records the change to Host REIT's tax (benefit) provision assuming the REIT Conversion did not occur in time for Host REIT to elect REIT status to result in an estimated overall effective tax rate for Host REIT in such circumstance of approximately 41%.
- E. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the Blackstone Acquisition may not occur. The following adjustments remove the Blackstone Acquisition from the REIT pro forma financial information:
  - . Remove property and equipment of \$1,450 million.
  - . Remove mortgage note receivable of \$63 million.
  - . Remove receivable from Lessee for working capital of \$5 million.
  - . Add back \$262 million of cash used to make the Blackstone Acquisition.
  - . Remove the assumption of mortgage debt of \$600 million.
  - . Remove the 43.7 million OP Units with an estimated fair value of \$656 million using an assumed Host Marriott stock price of \$15.00 (which includes the value of Crestline since Blackstone will receive shares of Crestline in addition to OP Units).
  - . Remove minority interest liability of \$221 million associated with the Blackstone limited partner interest (43.7 million OP Units) in the Operating Partnership.

- . Net effect on equity of the removal of the minority interest liability, Blackstone assets and liabilities, and recording of deferred income taxes.
- F. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the Blackstone Acquisition may not occur. The following adjustments remove the Blackstone Acquisition from the REIT pro forma financial information:
  - . Remove rental revenues and hotel operating expenses applicable to the Blackstone Acquisition.
  - . Remove the minority interest associated with the 43.7 million OP units owned by the Blackstone limited partners.
  - . Remove the interest expense on the \$600 million of debt assumed in the Blackstone Acquisition.
  - . Record interest income that was eliminated in connection with the Blackstone Acquisition.

## HOST REIT UNAUDITED PRO FORMA BALANCE SHEET

## SEPTEMBER 11, 1998 100% PARTICIPATION WITH NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	OPERATING PARTNERSHIP PRO FORMA	ADJUSTMENTS	PRO FORMA	ADDITIONAL PRO FORMA ADJUSTMENTS	REIT 2000 PRO FORMA ADJUSTMENTS
ASSETS					
Property and equipment,					
net Notes and other	\$7,029	\$	\$7,029	\$(1,450)(E)	\$5,579
receivables, net	303		303	(63)(E)	240
Due from managers Investments in	14		14	( )( )	14
affiliates	27		27		27
Other assets Receivable from Lessee	338		338		338
for working capital	85		85	(5)(E)	80
Cash, cash equivalents				. , . ,	
and short-term marketable securities	116		116	262 (E)	378
marketable securities				202 (L)	
Total assets	\$7,912	\$	\$7,912	\$(1,256)	\$6,656
LIABILITIES AND EQUITY	=====	====	=====	======	=====
Debt	\$5,218		5,218	\$(600)(E)	\$4,618
Convertible debt					
obligation to Host Marriott Corporation	567	(567)(A)			
Accounts payable and	307	(301)(A)			
accrued expenses	54		54		54
Deferred income taxes Other liabilities	275 701		275 701	189 (D) (43)(E)	464 658
00.101 1140111111001111111					
Total liabilities	6,815	(567)	6,248	(454)	5,794
Convertible Preferred					
Securities		550 (A)	550		550
Minority interest Limited Partner		205 (B)	205	(190)(E)	15
interests of third					
parties at redemption					
value (on a pro forma basis 47.1 million OP					
units outstanding)	701	(701)(C)			
Equity					
General Partner (on a pro forma basis 0.2					
million OP Units					
outstanding)					
Limited Partner interests of Host					
REIT (on a pro forma					
basis 204.3 million OP Units					
outstanding)	396	(396)(C)			
Shareholders' Equity					
(on a pro forma basis 800 million shares					
authorized; 204.5					
million issued and		17 (1)	000	(422)(5)	207
outstanding)		17 (A) (205)(B)	909	(423)(E) (189)(D)	297
		701 (C)		(200)(0)	
		396 (C)			
	\$7,912	\$	\$7,912	\$(1,256)	\$6,656
	=====	====	=====	======	=====

## HOST REIT UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

# FIRST THREE QUARTERS 1998 100% PARTICIPATION WITH NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	OPERATING PARTNERSHIP PRO FORMA			ADDITIONAL PRO FORMA ADJUSTMENTS	REIT 2000 PRO FORMA
DEVENUE					
REVENUE Rental revenues	\$ 540	\$	\$ 540	\$(72)(F)	\$ 468
Other revenues	2		2	Ψ(12)(1) 	2
Total revenues	542		542	(72)	470
OPERATING COSTS AND EX- PENSE					
Hotels	393		393	(58)(F)	335
Other	15		15	/ /	15
Total operating costs and ex-				()	
penses	408		408	(58)	350
OPERATING PROFIT Minority interest	134 (14)	  40 (B)	134 26	(14) (38)(F)	120
Corporate expenses	(30)		(30)		(30)
Interest expense Dividends on Convertible	(340)	26 (A)	(314)	36 (F)	(278)
Preferred Securities		(26)(A)	(26)		(26)
Interest income	24		24	6 (F)	30
Income (less) before in					
Income (loss) before income taxes  Benefit (provision) for	(226)	40	(186)	(10)	(196)
income taxes	11	(2)(D)	9	71 (D)	80
<pre>Income (loss) before ex- traordinary items</pre>	` ,	\$ 38 ====	\$ (177) =====		\$(116) =====
Diluted loss per share			\$ (.87) =====		\$(.57) =====

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

## FISCAL YEAR 1997 100% PARTICIPATION WITH NOTES ISSUED (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	OPERATING PARTNERSHIP PRO FORMA	PRO FORMA ADJUSTMENTS	PRO FORMA	ADDITIONAL PRO FORMA ADJUSTMENTS	REIT 2000 PRO FORMA
REVENUE Rental revenues Equity in earnings (losses) of affili-	\$1,135	\$	\$1,135	\$(137)(F)	\$ 998
atesOther revenues	(7) 3		(7) 3	 	(7) 3
Total revenues	1,131		1,131	(137)	994
OPERATING COSTS AND EX- PENSE					
Hotels Other	598 11		598 11	(87)(F) 	511 11
Total operating costs and ex-penses	609		609	(87)	522
pensestititititi					
OPERATING PROFIT Minority interest Corporate expenses	522 (10) (44)	(2)(B)	522 (12) (44)	(50) 2 (F) 	(44)
Interest expense Dividends on Convertible	(485)	38 (A)	(447)	48 (F)	(399)
Preferred Securities Interest income	 29	(37)(A) 	(37) 29	7 (F)	(37) 36
Income (loss) before income taxes Benefit (provision) for	12	(1)	11	7	18
income taxes	(1)	(D)	(1)	(6)(D)	(7)
<pre>Income (loss) before ex- traordinary items</pre>	\$ 11 =====	\$ (1) =====	\$ 10 =====	\$ 1 =====	\$ 11 =====
Diluted earnings per share			\$ .04 =====		\$ .05 =====

#### NOTES TO UNAUDITED PRO FORMA FINANCIAL STATEMENTS

#### 100% PARTICIPATION WITH NOTES ISSUED

#### A. Represents the following adjustments:

- . Adjustment to remove \$567 million of convertible debt obligations to Host Marriott which eliminates in consolidation.
- . Record the \$550 million of convertible preferred securities held by Host REIT.
- . Record \$17 million adjustment to equity for the elimination in consolidation of the convertible debt obligation to Host Marriott.
- . Remove Interest of \$26 million and \$38 million, respectively, on \$567 million of convertible debt obligation to Host Marriott.
- . Record dividends of \$26 million and \$37 million, respectively, on the \$550 million of convertible preferred securities.
- B. Represents the adjustment to present the Limited Partner interests of third parties in the Operating Partnership as minority interest in Host REIT and the reflection of Operating Partnership income allocable to the third party Limited Partners as minority interest expense of Host REIT as follows:
  - Record \$205 million minority interest liability representing Limited Partner interests of third parties (47.1 million OP Units out of total OP Units of 251.6 million) pro rata share of total combined Operating Partnership equity and Limited Partner interests of third parties at redemption value.
  - . Record adjustment to shareholders' equity of \$205 million to record the Limited Partner interests of third parties as minority interest.
  - . Record minority expense based on the Limited Partners interests of third parties pro rata share of Operating Partnership net income (47.1 million OP Units out of total OP Units of 251.6 million).
- C. Represents the adjustment to eliminate the Partner's Capital of the Operating Partnership and record the Common Stock, Additional Paid-in Capital, and Retained Earnings of Host REIT.
- D. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, Host REIT would be subject to federal income taxes. This adjustment records the change to Host REIT's tax (benefit) provision, assuming the REIT Conversion did not occur in time for Host REIT to elect REIT status to result in an estimated overall effective tax rate for Host REIT in such circumstance of approximately 41%.
- E. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the Blackstone Acquisition may not occur. The following adjustments remove the Blackstone Acquisition from the REIT pro forma financial information:
  - . Remove property and equipment of \$1,450 million.
  - . Remove mortgage note receivable of \$63 million.
  - . Remove receivable from Lessee for working capital of \$5 million.
  - . Add back \$262 million of cash used to make the Blackstone Acquisition.
  - . Remove the assumption of mortgage debt of \$600 million.
  - . Remove the 43.7 million OP Units with an estimated fair value of \$656 million using an assumed Host Marriott stock price of \$15.00 (which includes the value of Crestline since Blackstone will receive shares of Crestline in addition to OP Units).
  - . Remove minority interest liability of \$190 million associated with the Blackstone limited partner interest (43.7 million OP Units) in the Operating Partnership.
  - Net effect on equity on the removal of the minority interest liability, Blackstone assets and liabilities, and recording of deferred income taxes.

- F. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the Blackstone Acquisition may not occur. The following adjustments remove the Blackstone Acquisition from the REIT pro forma financial information:
  - . Remove rental revenues and hotel operation expenses applicable to the Blackstone Acquisition.
  - . Remove the minority interest associated with the  $43.7\ \text{million}$  OP Units owned by the Blackstone limited partners.
  - . Remove the interest expense on the \$600 million of debt assumed in the Blackstone Acquisition.
  - . Record interest income that was eliminated in connection with the Blackstone Acquisition.

## UNAUDITED PRO FORMA BALANCE SHEET

## SEPTEMBER 11, 1998 NO PARTNERSHIP PARTICIPATION (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	PRO FORMA	PRO FORMA ADJUSTMENTS	PRO FORMA	ADDITIONAL PRO FORMA ADJUSTMENTS	REIT 2000 PRO FORMA
ASSETS					
Property and equipment, net	\$6,298	\$	\$6,298	\$(1,450)(E)	\$4,848
Notes and other receiv-	•	<b>-</b>	•		
ables, net Due from managers Investments in affili-	326 13		326 13	(63)(E) 	263 13
ates	222		222		222
Other assets Receivable from Lessee	295		295		295
for working capital Cash, cash equivalents and short-term market-	67		67	(5)(E)	62
able securities	101		101	262 (E)	363
Total assets	\$7,322	\$	\$7,322	\$(1,256)	\$6,066
	=====	=====	=====	======	=====
LIABILITIES AND EQUITY Debt Convertible debt obliga-	\$4,544	\$	\$4,544	\$ (600)(E)	\$3,944
tion to Host Marriott Corporation	567	(567)(A)			
Accounts payable and ac-					
crued expenses Deferred income taxes	45 275		45 275	 189 (D)	45 464
Other liabilities	757		757	(43)(E)	714
Total liabilities	6,188	(567)	5,621	(454)	5,167
Minority interest		212 (B)	212	(197)(E)	15
Convertible Preferred		` ,		(197)(L)	
Securities Limited Partner interests of third parties at redemption value (on a pro forma basis 47.1 million OP		550 (A)	550		550
units outstanding) Equity General Partner (on a pro forma basis 0.2 million OP Units	701	(701)(C)			
outstanding) Limited Partner interests of Host REIT (on a pro forma basis 204.3 million OP Units					
outstanding) Shareholders' Equity (on a pro forma basis 800 million shares authorized; 204.5	433	(433)(C)			
million issued and outstanding)		17 (A) (212)(B) 701 (C) 433 (C)	939	(416)(E) (189)(D)	334
	\$7,322	\$	\$7,322	\$(1,256)	\$6,066
	=====	=====	=====	======	=====

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

# FIRST THREE QUARTERS 1998 NO PARTNERSHIP PARTICIPATION (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	OPERATING PARTNERSHIP PRO FORMA	PRO FORMA ADJUSTMENTS	HOST REIT PRO FORMA	ADDITIONAL PRO FORMA ADJUSTMENTS	REIT 2000 PRO FORMA
REVENUE					
Rental revenues	\$ 410	\$	\$ 410	\$(72)(F)	\$ 338
Hotel revenues	145		145		145
Other revenues	(12)		(12)		(12)
Total revenues	543		543	(72)	471
OPERATING COSTS AND EX- PENSE					
Hotels	346		346	(58)(F)	288
Other	15		15		15
Total operating costs and expenses	361		361	(58)	303
ODEDATING DOCET	100		100		100
OPERATING PROFIT	182		182 13	(14)	168
Minority interest Corporate expenses	(13) (30)	26(B)	(30)	(25)(F)	(12) (30)
Interest expense Dividends on Convertible	(304)	26 (A)	(278)	36 (F)	(242)
Preferred Securities		(26)(A)	(26)		(26)
Interest income	19		19	4 (F)	23
Income (loss) before in-					
come taxes Benefit (provision) for	(146)	26	(120)	1	(119)
income taxes	7	(1)(D)	6	43(D)	49
Income (loss) before ex-					
traordinary items	\$(139) =====	\$ 25 ====	\$(114) =====	\$ 44 ====	\$ (70) =====
Diluted loss per share	<b></b>	<b>_</b>	\$(.56) =====	<b>_</b>	\$(.34) =====

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

## FISCAL YEAR 1997 NO PARTNERSHIP PARTICIPATION (IN MILLIONS, EXCEPT PER OP UNIT AND PER SHARE AMOUNTS)

	OPERATING PARTNERSHIP PRO FORMA	PRO FORMA ADJUSTMENTS	HOST REIT PRO FORMA	ADJUSTMENTS	REIT 2000 PRO FORMA
REVENUE Rental revenues Hotel revenues	\$ 882 193	\$ 	\$ 882 193	\$(137)(F) 	\$ 745 193
Equity in earnings (losses) of affiliates	(16) 3		(16) 3		(16) 3
Total revenues	1,062		1,062	(137)	925
OPERATING COSTS AND EX- PENSE					
Hotels Other	573 11		573 11	(87)(F) 	11
Total operating costs and ex-penses	584		584	(87)	497
OPERATING PROFIT Minority interest Corporate expenses	478 (8) (44)	(2)(B)	478 (10) (44)	(50) 2 (F)	(44)
Interest expense Dividends on Convertible Preferred Securities	(439)	38 (A) (37)(A)	(401) (37)	48 (F)	(353)
Interest income	26	(37)(A) 	26	7 (F)	
Income (loss) before income taxes Benefit (provision) for	13	(1)	12	7	19
income taxes	(1)	(D)	(1)	(7)(D)	(8)
<pre>Income (loss) before ex- traordinary items</pre>	\$ 12	\$ (1)	\$ 11	\$	\$ 11
Diluted earnings per share	=====	====	\$ .05 =====	====	\$ .05 ====

#### NOTES TO UNAUDITED PRO FORMA FINANCIAL STATEMENTS

#### NO PARTNERSHIP PARTICIPATE

## A. Represents the following adjustments:

- . Adjustment to remove \$567 million of convertible debt obligation to Host Marriott which eliminates in consolidation.
- . Record the \$550 million of convertible preferred securities held by Host REIT.
- . Record \$17 million adjustment to equity for the elimination in consolidation of the convertible debt obligation to Host Marriott.
- . Remove interest of \$26 million and \$38 million, respectively, on \$567 million of convertible debt obligation to Host Marriott.
- . Record dividends of \$26 million and \$37 million, respectively, on the \$550 million of convertible preferred securities.
- B. Represents the adjustment to present the Limited Partner interests of third parties in the Operating Partnership as minority interest in Host REIT and the reflection of Operating Partnership income allocable to the third party Limited Partners as minority interest expense of Host REIT as follows:
  - Record \$212 million minority interest liability representing Limited Partner interests of third parties (47.1 million OP Units out of total OP Units of 251.6 million) pro rata share of total combined Operating Partnership equity and Limited Partner interests of third parties at redemption value.
  - . Record adjustment to shareholders' equity of \$212 million to record the Limited Partner interests of third parties as minority interest.
  - . Record minority expense based on the Limited Partner interests of third parties pro rata share of Operating Partnership net income (47.1 million OP Units out of total OP Units of 251.6 million).
- C. Represents the adjustment to eliminate the Partner's Capital of the Operating Partnership and record the Common Stock, Additional Paid-in Capital, and Retained Earnings of Host REIT.
- D. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, Host REIT would be subject to federal income taxes. This adjustment records the change to Host REIT's tax (benefit) provision assuming the REIT Conversion did not occur in time for Host REIT to elect REIT status to result in an estimated overall effective tax rate for Host REIT in such circumstance of approximately 41%.
- E. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the Blackstone Acquisition may not occur. The following adjustments remove the Blackstone Acquisition from the REIT pro forma financial information:
  - . Remove property and equipment of \$1,450 million.
  - . Remove mortgage note receivable of \$63 million.
  - . Remove receivable from Lessee for working capital of \$5 million.
  - . Add back \$262 million of cash used to make the Blackstone Acquisition.
  - . Remove the assumption of mortgage debt of \$600 million.
  - . Remove the 43.7 million OP Units with an estimated fair value of \$656 million using an assumed Host Marriott stock price of \$15.00 (which includes the value of Crestline since Blackstone will receive shares of Crestline in addition to OP Units).
  - . Remove minority interest liability of \$197 million associated with the Blackstone limited partner interest (43.7 million OP Units) in the Operating Partnership.
- F. If the REIT Conversion does not occur in time for Host REIT to elect REIT status effective January 1, 1999, the Blackstone Acquisition may not occur. The following adjustments remove the Blackstone Acquisition from the REIT pro forma financial information:
  - . Remove rental revenues and hotel operation expenses applicable to the Blackstone Acquisition.
  - . Remove the minority interest associated with the 43.7 million OP Units owned by the Blackstone limited partners.
  - . Remove the interest expense on the \$600 million of debt assumed in the Blackstone Acquisition.
  - . Record interest income that was eliminated in connection with the

#### PRO FORMA FINANCIAL STATEMENTS OF CRESTLINE

The REIT Conversion involves a complex series of transactions not all of which necessarily have to occur in order for the REIT Conversion to be consummated. For example, there is no requirement that any of the Partnerships participate in the REIT Conversion in order for the REIT Conversion to be consummated (but if a Partnership does not participate in the REIT Conversion, its hotel properties would not be leased to the Company), nor is there a requirement that the Blackstone Acquisition occur in order for the REIT Conversion to be consummated. In addition, the consent of a number of lenders and several outside partners in certain key Partnerships are required in order for certain hotel properties owned by Host to be leased to the Company, and there can be no assurance that all such consents will be obtained. Accordingly, the number of hotel properties that will be leased by Crestline may vary, perhaps substantially.

In light of the number of possible variations, the Company is not able to describe all possible combinations of Partnerships that could be leased. However, to assist in analyzing the REIT Conversion, the Company has prepared two separate sets of unaudited pro forma financial statements to show the impact of the REIT Conversion assuming that all Partnerships participate in the REIT Conversion, the Blackstone Acquisition is consummated, and that all required lender and outside partner consents are obtained ("All Partnerships Participate") and a second scenario showing no Partnerships participate in the REIT Conversion (but that the Blackstone Acquisition is consummated, and that all required lender and outside partner consents (other than for the Partnerships) are obtained) ("No Partnership Participation").

The unaudited pro forma condensed consolidated statements of operations of Crestline reflect the following transactions for the First Three Quarters 1998 and for the fiscal year ended January 2, 1998, as if such transactions had been completed at the beginning of each of the periods:

- . 1997 acquisition of Forum Group, Inc. (the "Forum Acquisition") and one additional senior living community
- . 1998 retirement of \$26 million of debt through a capital contribution from Host Marriott
- . 1998 repayment and forgiveness of \$92 million of unsecured debt and \$14.8 million intercompany note treated as a capital contribution by Host
- . 1998 acquisition of one senior living community
- . 1998 acquisition of minority interests in certain consolidated subsidiaries of Crestline through contributions from Host Marriott
- . 1998 spin off of Crestline by Host Marriott and the concurrent lease or sublease of hotels from Host REIT
- . 1998 adoption of EITF 97-2 to reflect the change in presentation to present property-level sales and operating expenses
- . Acquisition from Host of a 5% interest in a joint venture which holds an approximate \$130 million mortgage note from a consolidated subsidiary of Host in connection with the REIT Conversion
- . Adjustment to corporate expenses as if Crestline were operated on a stand-alone basis, partially offset by the asset management fee to be charged to Host REIT.

The adjustments to the unaudited pro forma balance sheet of Crestline reflect the lease and sublease of substantially all of Host Marriott's owned or leased hotels and certain other transactions as described herein in conjunction with the REIT Conversion.

In 1998, Crestline acquired one senior living community for \$21 million. Also, during 1998, Host Marriott prepaid approximately \$26 million of Crestline's mortgage debt and repaid \$92 million of unsecured debt to Marriott International. The prepayment was recorded as a capital contribution to Crestline and the \$92 million was repaid in exchange for a \$92 million note due to Host Marriott with similar terms. The \$92 million note and an additional \$14.8 million intercompany note were forgiven by Host and treated as a capital contribution in the First Three Quarters 1998.

In 1997, Host Marriott Corporation acquired 29 senior living communities from Marriott International and concurrently contributed all of the assets and liabilities obtained in the Forum Acquisition to Crestline. In addition, during 1997, Crestline acquired 49% of the remaining 50% interest in Leisure Park Venture Limited Partnership which owns a 418-unit retirement community in New Jersey for approximately \$23 million, including the assumption of approximately \$15 million in debt. Crestline currently owns 99% of the partnership.

If all components of Host's conversion to a REIT are not completed by January 1, 1999, Host REIT may be precluded from electing REIT status until January 1, 2000 and the Blackstone Acquisition may not be consummated. Accordingly a separate column ("Pro Forma without Blackstone") has been included in the pro forma financials that assumes that the Blackstone Acquisition does not occur.

The unaudited pro forma financial statements present the financial position and the results of operations of Crestline as if the transactions described above were completed. These presentations do not purport to represent what Crestline's results of operations would actually have been if the transactions described above had in fact occurred on such date or at the beginning of such period or to project Crestline's results of operations for any future date or period.

The unaudited pro forma financial statements are based upon certain assumptions, as set forth in the notes to the unaudited pro forma financial statements, that Crestline believes are reasonable under the circumstances and should be read in conjunction with the Consolidated Financial Statements and Notes thereto for HMC Senior Communities, Inc included elsewhere herein.

## UNAUDITED PRO FORMA BALANCE SHEET

AS OF SEPTEMBER 11, 1998 ALL PARTNERSHIPS PARTICIPATE (IN THOUSANDS, EXCEPT SHARE DATA)

		PRO FORMA ADJUSTMENTS		PRO FORMA WITHOUT BLACKSTONE (I)
ASSETS				
Property and equipment, net	\$649,528	\$	\$649,528	\$649,528
International	4,097	 SE 000(A)	4,097	,
Other assets	14,290	85,000(A) 6,482(B)	105,772	100,772
Cash and cash equivalents	26,504	15,000(B) (6,482)(B)	35,022	35,022
Total assets	\$694,419 ======	\$100,000 ======	\$794,419 ======	\$789,419 ======
LIABILITIES AND STOCKHOLDER'S EQUITY				
Debt  Deferred income taxes	\$213,034 61,376	\$	\$213,034 61,376	\$213,034 61,376
Due to Host Marriott Corpo-	,		,	•
ration, net	12,989	85,000(A)	97,989	92,989
accrued liabilities Deferred revenue	13,639	 	13,639	13,639
Deferred revenue	1,310		1,310	1,310
Total liabilities	302,348	85,000 	387,348	382,348
Stockholder's equity Common stock, 100 shares authorized, issued and outstanding (on a pro forma basis 75 million shares of common stock authorized; 24.8 million issued and				
outstanding)(/1/) Additional paid-in capi-				
_tal	386,627	15,000(B)	401,627	401,627
Retained earnings	5,444		5,444	5,444
Total stockholder's equi-				
ty	392,071	15,000	407,071	407,071
Total liabilities and				
stockholder's equity	\$694,419 ======	\$100,000 =====	\$794,419 ======	\$789,419 ======

See Notes to Unaudited Pro Forma Financial Statements.

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

## FIRST THREE QUARTERS 1998 ALL PARTNERSHIPS PARTICIPATE (IN THOUSANDS, EXCEPT PER SHARE DATA)

	HISTORICAL	D DEBT REFINANCING/ REPAYMENTS	E COMMUNITY ACQUISITIONS	F CORPORATE	G HOTEL LEASES	H ADOPTION OF EITF 97-2	PRO FORMA	I PRO FORMA WITHOUT BLACKSTONE
		REPATHENTS	ACQUISTITIONS	EXPENSES	LEASES			
REVENUES Hotels								
RoomsFood and beverage	\$	\$	\$ 	\$ 	\$	\$1,847,650 863,953	\$1,847,650 863,953	\$1,636,922 765,417
Other						163,268	163,268	144,647
House profit					1,123,100	(1,123,100)		
Total hotels					1,123,100	1,751,771	2,874,871	2,546,986
Senior living communities Routine	54,872		84			94,792	149,748	149,748
Ancillary	2,928		1			13,483	16,412	16,412
T-4-1 1::								
Total senior living communities	57,800		85			108,275	166,160	166,160
_								
Total revenues	57,800		85		1,123,100	1,860,046	3,041,031	2,713,146
OPERATING COSTS AND EXPENSES Hotels								
Property-level costs and expenses Rooms						729,475	729,475	647,105
Food and beverage						766,688	766,688	680,116
Other department costs and								
deductions Management fees and						255,608	255,608	226,746
other Lease expense					205,169 887,400		205,169 887,400	193,888 771,900
Total hotels					1,092,569	1,751,771	2,844,340	2,519,755
Senior living communities Property-level costs and expenses								
Routine						94,792	94,792	94,792
Ancillary						13,483	13,483	13,483
Other operating costs and expenses	29,803		49				29,852	29,852
•								
Total senior living communities	29,803		49			108,275	138,127	138,127
Communities	29,003					100,273		
Total operating costs and	20, 002		40		1 000 500	1 000 040	2 002 467	2 657 002
expenses	29,803 		49 		1,092,569	1,860,046	2,982,467	2,657,882
Operating profit	27,997		36		30,531		58,564	55,264
Corporate expenses	(2,937) (17,560)	 4,789		(7,948) 	(3,531)		(10,885) (16,302)	(10,885) (16,094)
Interest expense Interest and dividend	(17,300)	4,709			(3,331)		(10,302)	(10,094)
income	1,120		6		392		1,518	1,518
Income (loss) before								
income taxes	8,620	4,789	42	(7,948)	27,392		32,895	29,803
Benefit (provision)	•		(40)		•			
for income taxes	(3,534)	(1,963)	(18) 	3,259	(11,231)		(13,487)	(12,219)
Income (loss) before extraordinary item	\$ 5,086 ======	\$ 2,826 ======	\$ 24 ====		\$ 16,161 ======	\$ =======	\$ 19,408 ======	\$ 17,584 =======
Pro forma earnings per share	\$ .21(2) ======						\$ .78(2 ======	) \$ .86(3)

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

FISCAL YEAR ENDED JANUARY 2, 1998 ALL PARTNERSHIPS PARTICIPATE (IN THOUSANDS, EXCEPT PER SHARE DATA)

		С	D	E	F	G	Н	
	HISTORICAL	FORUM ACQUISITION	DEBT REFINANCING/ REPAYMENT	COMMUNITY ACQUISITIONS	CORPORATE EXPENSES	HOTEL LEASES	ADOPTION OF EITF 97-2	PRO FORMA
REVENUES Hotels								
Room Food and bever-	\$	\$	\$	\$	\$	\$	\$ 2,460,816	\$2,460,816
age Other							1,156,017 248,471	1,156,017 248,471
House profit						1,475,300	(1,475,300)	
Total hotels						1,475,300	2,390,004	3,865,304
Senior living								
communities Routine	35,473	30,859		7,031			127,135	200,498
Ancillary	1,427	1,983		188			18,693	22,291
Total senior living								
communities	36,900	32,842		7,219			145,828	222,789
Total revenues	36,900	32,842		7,219		1,475,300	2,535,832	4,088,093
OPERATING COSTS AND								
Hotels Property-level cos	ts and expen	ses						
Rooms							993,374	993,374
Food and bever- age							1,047,903	1,047,903
Other department costs and								
deductions Management fees							348,727	348,727
and other Lease expense						260,500 1,175,100		260,500 1,175,100
Ecase expense								
Total hotels						1,435,600	2,390,004	3,825,604
Senior living com- munities								
Property-level cos Routine	ts and expen	ses 					127,135	127,135
Ancillary Other operating							18,693	18,693
costs and expenses	20,929	17,977		4,733				43,639
Total senior								
living communities	20,929	17,977		4,733			145,828	189,467
Total aparating								
Total operating costs and								
expenses	20,929	17,977	<b></b>	4,733	<b></b>	1,435,600	2,535,832	4,015,071
Operating profit	15,971	14,865		2,486		39,700		73,022
Corporate expenses	(2,304)	(5,115)		745	(6,826)			(13,500)
Interest expense Interest and	(13, 396)	(9,630)	7,312	(2,118)	(0,820)	(5,100)		(22,932)
dividend income	336	598				567		1,501
Income (loss)								
before income								
taxes Benefit	607	718	7,312	1,113	(6,826)	35,167		38,091
<pre>(provision) for income taxes</pre>	(249)	(294)	(2,998)	(456)	2,799	(14,419)		(15,617)
Income (loss) before								
extraordinary								
item	\$ 358 ======	\$ 424 ======	\$ 4,314 ======	\$ 657 ======	\$(4,027) ======		\$ ========	\$ 22,474 ========
Pro forma earnings per share	\$ .01(2		<b>_</b>	<b>_</b>	<b></b>	<b></b>	<b></b>	\$ .91(2)

I ppn

PRO FORMA WITHOUT BLACKSTONE

REVENUES

Hotels Room..... \$2,203,294

Food and bever-

Total hotels.... 3,460,804

Saniar living

Senior living communities.....

Routine...... 200,498 Ancillary..... 22,291

Total senior

living communities.... 222,789

Total revenues.. 3,683,593

OPERATING COSTS AND EXPENSES
Hotels

Property-level costs and expenses Rooms........... 886,763

Other department costs and deductions..... 311,301

deductions..... 311,301

Management fees
and other..... 246,500

Lease expense.... 1,038,600

Total hotels.... 3,418,604

Senior living com-

munities
Property-level costs and expenses
Routine 127 135

Routine...... 127,135 Ancillary...... 18,693

Other operating costs

and expenses.... 43,639

Total senior

living communities.... 189,467

Total operating costs and expenses.....

expenses...... 3,608,071

Operating profit ..... 75,522

Corporate
expenses.......... (13,500)
Interest expense... (22,633)
Interest and

dividend income.. 1,501

Income (loss)
before income
taxes 40 890

income taxes..... (16,765)

Income (loss)
before
extraordinary
item..........\$ 24,125

Pro forma earnings
per share..... \$ 1.18(3)

#### NOTES TO UNAUDITED PRO FORMA FINANCIAL STATEMENTS

#### ALL PARTNERSHIPS PARTICIPATE

- A. Represents the adjustment to increase working capital and record a loan payable to Host of \$85 million to record the transfer of hotel working capital to the Company related to the leasing of Host's hotels.
- B. Represents the following transactions in connection with the Distribution:
  - . Host's contribution of \$15 million in cash to the Company.
  - . Acquisition from Host of a 5% limited partner interest in a joint venture with Host that owns an approximate \$130 million note receivable from a consolidated subsidiary of Host.
- C. Represents the adjustment to reflect the historical revenues, operating expenses, corporate expenses, interest expense and interest income for the Forum Acquisition as if such acquisition occurred at the beginning of 1997 (actual acquisition date was June 21, 1997).
- D. Represents the adjustment to eliminate interest expense on \$133 million of debt repaid during 1998 by Host on behalf of the Company and treated as a capital contribution by Host.
- E. Represents the adjustment to record the historical revenues, operating expenses, corporate expenses and interest income related to the acquisition of one senior living community in 1998 and the acquisition of one senior living community in 1997. The adjustment also includes the elimination of \$745,000 of minority interest expense included in corporate expenses related to the purchase of minority interests in certain consolidated subsidiaries of the Company in 1997.
- F. Represents the adjustment to record additional corporate expenses anticipated to be incurred when the Company is operated on a stand-alone basis subsequent to the Distribution, net of the asset management contract of \$4.5 million per annum. The adjustment includes the following (in thousands):

	FIRST THREE QUARTERS 1998	1997
Payroll costs	\$ 6,748	\$ 6,894
Rent and insurance	1,218	1,267
Other general and administrative costs	3,097	3,165
	11,063	11,326
Less: asset management fee	(3,115)	(4,500)
Net corporate expense adjustment	\$ 7,948	\$ 6,826
	<u>·</u>	

G. Represents the adjustment to record the historical hotel revenues and hotel expenses and pro forma lease expense associated with the leasing of certain hotel properties from Host, interest expense on the \$85 million working capital loan at 6%, and dividend income from the 5% investment in the joint venture with Host that owns a \$130 million note receivable from a consolidated subsidiary of Host.

Rental revenues under the leases are based on the greater of Percentage Rent or Minimum Rent. Total rent in the pro forma statement of operations is calculated based on the historical gross sales of the property and the negotiated rental rates and thresholds by property as if the leases were entered into on the first day of fiscal year 1997. There are generally three sales categories utilized in the rent calculation: rooms, food and beverage and other. For rooms and food and beverage, there are three tiers of rent with two thresholds, while the other category generally has two tiers of rent and one threshold. The percentage rent thresholds are increased annually on the first day of each year after the initial lease year based on a blended increase of the Consumer Price Index ("CPI") and a wage and benefit index. For purposes of the pro formas, 1997 is the assumed initial lease year

and the blended increase applied to the thresholds at January 3, 1998 is assumed to be 3%. Minimum rent is expressed as a fixed dollar amount that increases annually on the first day of each year after the initial lease year at 50% of the CPI increase. Accordingly, the 1998 rent thresholds and minimum rent included in the pro formas were adjusted as of January 3, 1998 for the 1997 increases in the indices. Rental revenues are recognized only for leases to be executed with Host REIT at or prior to completion of the Distribution. The execution of the leases is dependent upon the consummation of the Distribution, which is subject to contingencies that are outside the control of the Company, including consent of shareholders, lenders, debt holders, partners and ground lessors of Host. The Company believes that negotiations with third parties to complete the Distribution will not result in any material change to the leases. The table below details gross sales, minimum rent and total rent for all full-service properties to be leased and summarized amounts for the limited-service properties to be subleased:

	FISCAL YEAR 1997			FIRST THREE QUARTERS 1998		
PROPERTY	SALES	RENT	RENT	GROSS SALES	MINIMUM RENT	TOTAL RENT
				LLIONS		
Grand Hotel Resort and Golf Club Scottsdale Suites The Ritz-Carlton, Phoenix Coronado Island Resort Costa Mesa Suites	\$23.4	\$2.8	\$4.2	\$18.0	\$2.0	\$3.7
	11.9	3.0	5.0	8.2	2.1	3.4
	23.3	4.6	7.2	17.3	3.2	5.5
	22.0	2.1	2.1	16.2	1.5	3.6
	9.7	1.9	3.3	7.2	1.3	2.3
Desert Springs Resort and Spa  Manhattan Beach  Marina Beach	103.3	21.3	30.3	80.3	15.0	22.6
	16.3	2.4	4.8	12.2	1.7	3.6
	21.1	4.6	7.1	16.9	3.2	6.2
	33.5	5.5	8.7	24.0	3.9	6.8
Newport Beach Suites Ontario Airport San Diego Marriott Hotel and Marina		2.6 1.8 38.0	4.1 3.4 39.6	8.0 8.3 78.6	1.8 1.3 26.7	3.0 2.2 31.1 5.6
San Diego Mission Valley	16.7 43.8 17.8 120.2	3.4 8.2 4.0 20.7	5.1 13.2 6.4 37.9	12.6 32.2 12.1 90.5	2.4 5.8 2.8 14.6	9.5 4.3 28.5
San Ramon	19.7	4.4	5.1	14.4	3.1	4.0
	47.3	7.8	16.5	37.2	5.5	13.5
	32.4	5.5	10.8	23.4	3.9	7.9
The Ritz-Carlton, San Francisco  Torrance  Denver Southeast  Denver Tech Center	50.1	9.6	14.7	34.2	6.7	10.3
	20.5	2.3	3.5	15.0	1.6	5.1
	21.5	3.0	6.2	14.9	2.1	4.1
	26.8	5.1	8.3	20.1	3.6	6.0
Denver West	13.7	1.8	4.0	9.6	1.2	2.7
Marriott's Mountain Resort at Vail	17.6	3.0	5.1	14.1	2.1	4.5
Hartford/Farmington	18.4	3.5	4.7	13.4	2.4	3.5
Hartford/Rocky Hill  Fort Lauderdale Marina  Harbor Beach Resort  Jacksonville	11.6	1.5	2.7	8.5	1.1	2.0
	28.5	4.3	7.9	20.4	3.0	5.7
	58.1	16.5	19.3	43.2	11.6	14.0
	11.8	1.8	3.6	8.0	1.2	2.4
Miami Airport	29.7	3.9	8.4	21.6	2.8	6.1
Orlando World Center	128.2	23.5	39.6	98.7	16.5	30.4
Palm Beach Gardens	11.8	1.9	3.7	8.5	1.4	3.0
Singer Island (Holiday Inn)	6.6	1.4	2.5	5.2	1.0	2.1
	17.1	1.6	3.5	13.1	1.1	2.7
	15.0	1.8	3.6	10.8	1.3	2.6
The Ritz-Carlton, Naples	66.4	18.1	23.3	53.1	12.7	18.0
	85.4	21.3	33.3	58.6	15.0	25.6
	10.5	1.8	3.5	7.8	1.3	2.6
	7.6	1.0	1.7	5.6	0.7	1.2
Atlanta Northwest Atlanta Perimeter JW Marriott Hotel at Lenox	14.9	2.7	4.3	11.3	1.9	3.3
	16.6	2.5	4.5	12.6	1.7	3.5
	24.8	3.7	6.8	17.7	2.6	5.0
The Ritz-Carlton, Atlanta	30.2	5.8	8.8	21.7	4.1	6.8

FISCAL YEAR 1997

DDODEDTV	GROSS	MINIMUM RENT		GROSS	MINIMUM	TOTAL
PROPERTY 	SALES		RENT	SALES	RENT	RENT
			(IN MILL	LIONS)		
The Ritz-Carlton, Buckhead		\$ 13.1	\$ 16.3			\$ 11.7
Chicago/Deerfield Suites Chicago/Downers Grove	10.2	1.8	3.1	7.4	1.3	2.3
Suites	9.0	1.8	2.9	6.7	1.3	2.2
Chicago/Downtown Courtyard	16.3		4.9	12.2		3.9
Chicago O'Hare	40.0		11.5	28.8		8.2
South Bend	9.9 66.4		2.1 21.8	7.0 47.6		1.5 15.8
Bethesda	23.2		5.6	17.3		4.1
Gaithersburg/Washingtonian						
Center	13.2		3.8	9.7		2.8
Boston/Newton  Detroit Romulus	27.4 8.8	4.8 1.1	7.8 1.8	19.1 6.6		5.5 1.4
The Ritz-Carlton, Dearborn	25.7		5.5	17.7		4.0
Minneapolis/Bloomington	20.2		6.5	13.8	2.3	4.7
Minneapolis City Center	27.5		7.5	20.4		5.2
Minneapolis Southwest Kansas City Airport	14.9 14.3		4.8 3.7	10.1 9.9		4.0 2.5
St. Louis Pavilion	27.5		6.5	18.5		4.3
Nashua	7.5		1.3	5.3		0.9
Hanover	22.5		6.6	15.1		4.3
Newark Airport	39.4 16.0		11.8 4.0	29.2 11.9		8.6 4.2
Park Ridge Saddle Brook	10.0		2.1	7.8		1.7
Albany	18.5	3.5	6.1	12.4		5.2
New York Marriott Financial						
Center	39.6	7.7 40.0	13.2 60.8	29.1		10.1
New York Marriott Marquis Marriott World Trade	210.3	40.0	00.0	155.4	29.7	47.6
Center	65.4	12.2	19.4	49.1	8.6	14.9
Charlotte Executive Park	14.0		3.7	9.8		2.6
Raleigh Crabtree Valley	14.9		3.9	10.9		2.8
Oklahoma CityOklahoma City Waterford	15.6 9.1		3.8 2.7	10.4 6.1		2.4 1.7
Portland	26.4		7.5	17.6		4.8
Philadelphia (Convention						
Center)	80.7		25.0	58.2	10.0	17.8
Philadelphia Airport Pittsburgh City Center	25.0 16.4		7.6 3.0	18.6 11.1		5.6 2.2
Memphis	10.4		3.2	5.7		1.8
Dallas/Fort Worth	28.9		9.3	21.9		7.0
Dallas Quorum	25.7		8.2	18.3		5.8
El Paso Houston Airport	11.6 21.6		2.3 6.0	7.8 16.9		1.4 4.6
JW Marriott Houston	27.2		8.0	20.1		5.9
Plaza San Antonio	13.8		4.6	9.7		3.3
San Antonio Rivercenter	68.9	17.5	24.5	49.3	12.3	17.8
San Antonio Riverwalk	29.3	6.1	10.3	21.7	4.3	7.6
Salt Lake City Dulles Airport	28.5 14.6	5.6 1.8	9.5 4.0	21.1 10.9	3.9 1.2	7.2 3.0
Key Bridge	29.4		10.2	21.2	3.9	7.4
Norfolk Waterside	18.1	3.3	5.4	12.8	2.4	3.8
Pentagon City Residence	44 7	2 5		0.7	2 5	4.0
Inn The Ritz-Carlton, Tysons	11.7	3.5	5.5	8.7	2.5	4.2
Corner	34.4	5.9	9.8	24.9	4.1	7.3
Washington Dulles Suites	10.3	2.5	4.0	7.8	1.8	3.0
Westfields	28.0		7.4	20.3	3.3	5.4
Williamsburg Washington Metro Center	12.6 25.2		2.8 7.3	9.3 19.2	1.3 3.2	2.1 5.3
Calgary	13.4		1.7	9.8	1.2	2.3
Toronto Airport	17.1		5.6	13.0	2.0	4.2
Toronto Eaton Centre	21.1		7.1	16.0	4.3	5.6
Toronto Delta Meadowvale Fairview Park	16.1 22.5		4.9 7.3	10.6 16.3	1.9 2.8	3.1 5.2
I WTI ATEM LUIV	22.5	3.9	1.3	10.3	2.0	3.2

FISCAL YEAR 1997 1998 GROSS MINIMUM TOTAL SALES RENT RENT GROSS MINIMUM TOTAL **PROPERTY** SALES RENT RENT ----------(IN MILLIONS) Dayton..... \$ 18.2 \$ 3.2 \$ 6.0 \$ 13.4 \$ 2.3 \$ 4.3 1.4 Research Triangle Park..... 9.1 2.9 6.8 1.0 2.3 Detroit Marriott Southfield... 8.8 1.2 2.1 6.9 0.9 1.7 Detroit Marriott Livonia.... 10.0 1.4 2.6 7.4 1.9 1.0 Fullerton.....
Marriott O'Hare Suites..... 1.8 5.0 0.8 6.8 1.2 1.3 14.4 2.7 4.9 10.8 1.9 4.0 Albuquerque..... 16.4 3.6 3.6 11.1 2.5 2.6 2.4 Greensboro-High Point..... 13.6 3.3 3.3 10.2 2.3 Houston Medical Center..... 4.0 4.0 12.2 2.8 2.9 16.5 Miami Biscayne Bay..... 26.8 6.5 6.6 20.5 4.5 Marriott Mountain Shadows Resort..... 24.1 4.4 4.5 16.9 3.1 3.1 Seattle SeaTac Airport..... 17.5 23.1 6.7 6.7 4.7 5.1 Four Seasons, Atlanta(4)..... 15.6 5.8 5.9 14.2 4.1 4.5 Four Seasons Philadelphia(4)..... 7 9 12 4 30 6 41 1 5.6 10.1 Grand Hyatt, Atlanta(4)..... 25.3 10.0 10.0 22.6 7.0 8.2 Hyatt Regency, Burlingame(4)..... 47.9 8.8 17.6 39.5 15.1 Hyatt Regency, Cambridge(4).. 32.4 6.7 11.9 26.8 4.7 10.4 Hyatt Regency, Reston(4)..... Swissotel, Atlanta(4)..... 24.2 30.5 6.5 11.3 4.5 9.2 22.2 5.0 17.2 6.3 3.5 5.8 Swissotel, Boston(4)...... Swissotel, Chicago(4)..... 26.8 6 4 8.5 20 5 4.5 6 9 38.1 10.9 15.1 28.9 7.7 12.0 The Drake (Swissotel), New York(4)..... 38.8 11.6 13.6 34.2 8.2 13.4 The Ritz-Carlton, Amelia Island(4)..... 45.7 10.3 13.4 37.4 11.1 The Ritz-Carlton, Boston(4).. 40.1 6.9 31.4 10.5 4.8 8.8 Total Full-service 3,600.8 715.9 1,095.6 2,671.0 504.5 838.3 Properties..... 59.2 Total Courtyard Properties... 212.0 50.6 159.2 35.0 36.8

FIRST THREE QUARTERS

H. Represents the adjustment to reflect the Company's anticipated adoption of EITF 97-2 in the fourth quarter of 1998 by recording property-level sales and operating expenses. The adjustment has no impact on operating profit or net income.

69.9

. . . . . . . . . . . . . . . .

Total...... \$3,882.7 \$783.7 \$1,175.1 \$2,880.8 \$551.5 \$887.4

17.2

20.3

50.6

\_\_\_\_\_

12.0

12.3

- I. The "Pro Forma without Blackstone" column reflects the adjustment to eliminate the revenues, operating expenses and working capital relating to the Blackstone hotel properties assuming that the Blackstone Acquisition does not occur. Amounts related to Crestline's 25% interest in Swissotel Management (USA) L.L.C. have not been included in the pro forma financial statements due to immateriality.
- (1) On a pro forma basis as of September 11, 1998, the Company had 75 million shares of common stock, \$.01 par value authorized with 24.8 million shares issued and outstanding. In addition, on a pro forma basis, 10 million shares of preferred stock, \$.01 par value are authorized with none issued or outstanding.
- (2) Reflects the pro forma earnings per share based on 24.8 million weighted average shares outstanding subsequent to the Distribution. Pro forma weighted average shares are based on Host's weighted average shares outstanding, adjusted for a one-for-ten share distribution, and the issuance of shares to the Blackstone Entities.
- (3) Reflects the pro forma earnings per share based on 20.4 million weighted average shares outstanding subsequent to the Distribution. Pro forma weighted average shares are based on Host's weighted average shares outstanding, adjusted for a one-for-ten share distribution, and assumes that the Blackstone acquisition does not occur.
  - (4) Represent the Blackstone hotel properties.

Total Residence Inns.....

## UNAUDITED PRO FORMA BALANCE SHEET

## AS OF SEPTEMBER 11, 1998 NO PARTNERSHIP PARTICIPATION (IN THOUSANDS, EXCEPT SHARE DATA)

	HISTORICAL	PRO FORMA ADJUSTMENTS	PRO FORMA	PRO FORMA WITHOUT BLACKSTONE
ASSETS				
Property and equipment, net Amounts due from Marriott Inter-	\$649,528	\$	\$649,528	\$649,528
national Other assets	4,097 14,290	 67 000(A)	4,097 87,772	4,097 82,772
	•	67,000(A) 6,482(B)	•	•
Cash and cash equivalents	26,504	15,000(B) (6,482)	•	35,022
Tatal accets	<b></b>	<b>****</b>	 ф700 004	 ф777 004
Total assets	\$694,419 ======	\$82,000 =====	\$782,901 ======	\$777,901 ======
LIABILITIES AND STOCKHOLDER'S EQ- UITY				
Debt	\$213,034	\$	\$213,034	\$213,034
Deferred income taxes  Due to Host Marriott Corporation,	61,376	Ψ 	61,376	61,376
net	12,989	67,000(A)	79,989	74,989
crued liabilities	13,639		13,639	13,639
Deferred revenue	1,310		1,310	1,310
Total liabilities	302,348	67,000	369,348	364,348
Stockholder's equity Common stock, 100 shares authorized, issued and outstanding (on a pro forma basis 75 million shares of common stock authorized; 24.8 million issued and				
outstanding)(/1/)	386,627	 15,000(B)	 401 627	401,627
Retained earnings	5,444		5,444	5,444
Total stockholder's equity	392,071	15,000	407,071	407,071
Total liabilities and stock-holder's equity	\$694,419 ======	\$82,000 =====	\$776,419 ======	\$771,419 ======

See Notes to Unaudited Pro Forma Financial Statements.

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

## FIRST THREE QUARTERS 1998 NO PARTNERSHIP PARTICIPATION (IN THOUSANDS, EXCEPT PER SHARE DATA)

	HISTORICAL	D DEBT REFINANCING/ REPAYMENTS	E COMMUNITY ACQUISITIONS	F CORPORATE EXPENSES		H ADOPTION OF EITF 97-2		I PRO FORMA WITHOUT BLACKSTONE
REVENUES Hotels								
RoomsFood and beverage	\$ 	\$	\$ 	\$ 	\$ 	1,507,687 675,831	1,507,687 675,831	1,292,185 579,231
Other						110,423	110,423	94,640
House profit					896,321	(896,321)		
Total hotels					896,321	1,397,620		1,966,056
Senior living communities								
RoutineAncillary	54,872 2,928		84 1			94,792 13,483	149,748 16,412	149,748 16,412
Total senior living								
communities	57,800		85			108,275	166,160	166,160
Total revenues	57,800		85		896,321	1,505,895	2,460,101	2,132,216
OPERATING COSTS AND EXPENSES								
Hotels Property-level costs and expenses								
Rooms Food and beverage						651,662 632,750	651,662 632,750	569,292 546,178
Other department costs and deductions Management fees and						113,208	113,208	105,846
other					177,521		177,521	166,240
Lease expense					697,500		697,500	554,500 
Total hotels					875,021	1,397,620	2,272,641	1,942,056
Senior living communities Property-level costs and expenses								
Routine						94,792	94,792	94,792
Ancillary Other operating costs						13,483	13,483	13,483
and expenses	29,803		49				29,852	29,852
Total senior living								
communities	29,803		49		<b></b>	108,275	138,127	138,127
Total operating costs								
and expenses	29,803		49		875,021	1,505,895	2,410,768	2,080,183
Operating profit	27,997		36		21,300		49,333	52,033
Corporate expenses Interest expense	(2,937) (17,560)	 4,789		(7,948) 	(2,949)		(10,885) (15,720)	(10,885) (15,513)
Interest and dividend	(17,300)	4,709			(2,343)		(13,720)	(13,313)
income	1,120		6		392		1,518	1,518
<pre>Income (loss) before   income taxes Benefit (provision) for</pre>	8,620	4,789	42	(7,948)	18,743		24,246	27,153
income taxes	(3,534)	(1,963)	(18)	3,259	(7,684)		(9,941)	(11,133)
Income (loss) before extraordinary item	\$ 5,086 ======	\$ 2,826 ======	\$ 24 =====	\$(4,689) ======			14,305 ======	16,020 ======
Pro forma earnings per share	\$ .21(2) ======						\$ .58(2)	

See Notes to Unaudited Pro Forma Financial Statements.

## UNAUDITED PRO FORMA STATEMENT OF OPERATIONS

FISCAL YEAR ENDED JANUARY 2, 1998 NO PARTNERSHIP PARTICIPATION (IN THOUSANDS, EXCEPT PER SHARE DATA)

		С	D	E	F	G	Н	
	HISTORICAL	FORUM ACQUISITION	DEBT REFINANCING/ REPAYMENT	COMMUNITY ACQUISITIONS	CORPORATE EXPENSES	HOTEL LEASES	ADOPTION OF EITF 97-2	PRO FORMA
REVENUES Hotels								
Room Food and bever-	\$	\$	\$	\$	\$	\$	\$1,984,659	\$1,984,659
age Other							874,761 167,322	874,761 167,322
House profit						1,163,628	(1,163,628)	
Total hotels						1,163,628	1,863,114	3,026,742
TOTAL HOTCIS								
Senior living communities Routine	35,473	30,859		7,031			127,135	200,498
Ancillary	1,427	1,983		188			18,693	22,291
Total senior								
living communities	36,900	32,842		7,219			145,828	222,789
Total revenues	36,900	32,842		7,219		1,163,628	2,008,942	3,249,531
		32,642		7,219			2,008,942	
OPERATING COSTS AND Hotels	EXPENSES							
Property-level cos	ts and expense	es						
Rooms Food and bever-							889,525	889,525
age							852,701	852,701
Other department costs and								·
deductions Management fees							120,888	120,888
and other						214,128		214,128
Lease expense						921,800		921,800
Total hotels						1,135,928	1,863,114	2,999,042
Senior living com- munities Property-level cos	ts and expense	es						
Routine							127,135	127,135
Ancillary Other operating costs and							18,693	18,693
expenses	20,929	17,977		4,733				43,639
Total conjer								
Total senior living								
communities	20,929	17,977		4,733			145,828	189,467
Total operating								
costs and	_							
expenses	20,929	17,977 		4,733		1,135,928	2,008,942	3,188,509
Operating profit			<b></b>		<del>-</del>	<b></b>	<b></b>	<del></del>
Corporate	15,971	14,865		2,486		27,700		61,022
expenses Interest expense	(2,304) (13,396)	(5,115) (9,630)	 7,312	745 (2,118)	(6,826) 	 (4,261)		(13,500) (22,093)
Interest and dividend income	336	598				567		1,501
dividend income		596				307		1,501
Income (loss)								
before income taxes Benefit	607	718	7,312	1,113	(6,826)	24,006		26,930
(provision) for								
income taxes	(249)	(294)	(2,998)	(456)	2,799	(9,843)		(11,041)
Income (loss) before								
extraordinary	Ф 050	<b>.</b> 40.4	Ф 4 04 4	Φ 057	Φ(4 007 <b>)</b>	ф 14 100	Φ.	Ф 45.000
item	\$ 358 ======	\$ 424 ======	\$ 4,314 ======	\$ 657 =====		\$ 14,163 =======	\$ =======	\$ 15,889 =======
Pro forma earnings per share	\$ .01(2)							\$ .64(2)

Ι PR0 **FORMA** WITHOUT **BLACKSTONE** 

**REVENUES** 

Hotels Room..... \$1,719,425

Food and bever-

757,02 144,961 age..... Other..... House profit.....

Total hotels.... 2,622,242

Senior living

communities..... Routine..... 200,498 22,291 Ancillary.....

Total senior

living

communities.... 222,789 Total revenues.. 2,845,031

OPERATING COSTS AND EXPENSES

Hotels

Property-level costs and expenses Rooms..... 782,914

Food and bever-

740,238

age..... Other department costs and

83,462

deductions..... Management fees

200,128

and other..... Lease expense.... 785,300 Total hotels....

2,592,042

Senior living com-

munities

Property-level costs and e-- xpenses

Ancillary..... 18,693

Other operating costs and

expenses..... 43,639

Total senior

living

communities.... 189,467

Total operating

costs and

expenses..... 2,781,509

Operating profit 63,522

. . . . . . . . . . . . . . . . . .

Corporate (13,500) (21,793) expenses.....

Interest expense.. Interest and

dividend income..

1,501

Income (loss) before income

29,730

(12, 189)

taxes.....

Benefit (provision) for

income taxes....

Income (loss) before

extraordinary item..... \$ 17,541

Pro forma earnings

per share..... \$ .86(3)

#### NOTES TO UNAUDITED PRO FORMA FINANCIAL STATEMENTS

#### NO PARTNERSHIP PARTICIPATION

- A. Represents the adjustment to increase working capital and record a loan payable to Host of \$67 million to record the transfer of hotel working capital to the Company related to the leasing of Host's hotels.
- B. Represents the following transactions in connection with the Distribution:
  - . Host's contribution of \$15 million in cash to the Company.
  - . Acquisition from Host of a 5% limited partner interest in a joint venture with Host that owns an approximate \$130 million note receivable from a consolidated subsidiary of Host.
- C. Represents the adjustment to reflect the historical revenues, operating expenses, corporate expenses, interest expense and interest income for the Forum Acquisition as if such acquisition occurred at the beginning of 1997 (actual acquisition date was June 21, 1997).
- D. Represents the adjustment to eliminate interest expense on \$133 million of debt repaid during 1998 by Host on behalf of the Company and treated as a capital contribution by Host.
- E. Represents the adjustment to record the historical revenues, operating expenses, corporate expenses and interest income related to the acquisition of one senior living community in 1998 and the acquisition of one senior living community in 1997. The adjustment also includes the elimination of \$745,000 of minority interest expense included in corporate expenses related to the purchase of minority interests in certain consolidated subsidiaries of the Company in 1997.
- F. Represents the adjustment to record additional corporate expenses anticipated to be incurred when the Company is operated on a stand-alone basis subsequent to the Distribution, net of the asset management contract of \$4.5 million per annum. The adjustment includes the following (in thousands):

	FIRST THREE QUARTERS 1998	1997
Payroll costs	\$6,748 1,217 3,097	\$6,894 1,267 3,165
Less: asset management fee	11,062 (3,115)	11,326 (4,500)
Net corporate expense adjustment	\$7,947 	\$6,826

G. Represents the adjustment to record the historical hotel revenues and hotel expenses and pro forma lease expense associated with the leasing of certain hotel properties from Host, interest expense on the \$67 million working capital loan at 6%, and dividend income from the 5% investment in the joint venture with Host that owns a \$130 million note receivable from a consolidated subsidiary of Host.

Rental revenues under the leases are based on the greater of Percentage Rent or Minimum Rent. Total rent in the pro forma statement of operations is calculated based on the historical gross sales of the property and the negotiated rental rates and thresholds by property as if the leases were entered into on the first day of fiscal year 1997. There are generally three sales categories utilized in the rent calculation: rooms, food and beverage and other. For rooms and food and beverage, there are three tiers of rent with two thresholds, while the other category generally has two tiers of rent and one threshold. The percentage rent thresholds are increased annually on the first day of each year after the initial lease year based on a blended increase of the CPI and a wage and benefit index. For purposes of the pro formas, 1997 is the assumed initial lease year and the blended increase applied to the thresholds at January 3, 1998 is assumed to be 3%. Minimum rent is expressed as a fixed dollar amount that increases annually on the first day of each year after the initial lease year at 50% of the CPI increase. Accordingly, the 1998 rent thresholds and minimum rent included in the pro formas were adjusted as of

January 3, 1998 for the 1997 increases in the indices. Rental revenues are recognized only for leases to be executed with Host REIT at or prior to completion of the Distribution. The execution of the leases is dependent upon the consummation of the Distribution, which is subject to contingencies that are outside the control of the Company, including consent of shareholders, lenders, debt holders, partners and ground lessors of Host. The Company believes that negotiations with third parties to complete the Distribution will not result in any material change to the leases. The table below details gross sales, minimum rent and total rent for all full-service properties to be leased and summarized amounts for the limited-service properties to be subleased:

		_ YEAR 1		FIRST THREE QUARTERS 1998			
PROPERTY	GROSS N SALES	MINIMUM RENT	TOTAL RENT	GROSS SALES	MINIMUM RENT	TOTAL RENT	
			(IN MILL				
Grand Hotel Resort and Golf							
Club	\$ 23.4 9		\$ 4.2			\$ 3.7	
Scottsdale Suites The Ritz-Carlton, Phoenix	11.9 23.3	3.0 4.6	5.0 7.2	8.2 17.3	2.1 3.2	3.4 5.5	
Coronado Island Resort	22.0	2.1	2.1	16.2	1.5	3.6	
Costa Mesa Suites	9.7	1.9	3.3	7.2	1.3	2.3	
Manhattan Beach	16.3	2.4	4.8	12.2	1.7	3.6	
Marina Beach	21.1	4.6	7.1	16.9	3.2	6.2	
Newport Beach	33.5	5.5	8.7	24.0	3.9	6.8	
Newport Beach Suites	11.0	2.6	4.1	8.0	1.8	3.0	
Ontario AirportSan Diego Marriott Hotel and	12.1	1.8	3.4	8.3	1.3	2.2	
Marina	103.3	38.0	39.6	78.6	26.7	31.1	
San Diego Mission Valley	16.7	3.4	5.1	12.6	2.4	5.6	
San Francisco Airport San Francisco Fisherman's	43.8	8.2	13.2	32.2	5.8	9.5	
WharfSan Francisco Moscone	17.8	4.0	6.4	12.1	2.8	4.3	
Center The Ritz-Carlton, Marina del	120.2	20.7	37.9	90.5	14.6	28.5	
Rey The Ritz-Carlton, San	32.4	5.5	10.8	23.4	3.9	7.9	
Francisco	50.1	9.6	14.7	34.2	6.7	10.3	
Torrance	20.5	2.3	3.5	15.0	1.6	5.1	
Denver Southeast	21.5	3.0	6.2	14.9	2.1	4.1	
Denver Tech Center	26.8	5.1	8.3	20.1	3.6	6.0	
Denver West Marriott's Mountain Resort	13.7	1.8	4.0	9.6	1.2	2.7	
at Vail	17.6	3.0	5.1	14.1	2.1	4.5	
Hartford/Farmington	18.4	3.5	4.7	13.4	2.4	3.5	
Hartford/Rocky Hill	11.6	1.5	2.7	8.5	1.1	2.0	
Fort Lauderdale Marina Jacksonville	28.5 11.8	4.3 1.8	7.9 3.6	20.4 8.0	3.0 1.2	5.7 2.4	
Miami Airport	29.7	3.9	8.4	21.6	2.8	6.1	
Palm Beach Gardens	11.8	1.9	3.7	8.5	1.4	3.0	
Singer Island (Holiday							
Inn)	6.6	1.4	2.5	5.2	1.0	2.1	
Tampa Airport	17.1	1.6	3.5	13.1	1.1	2.7	
Tampa Westshore	15.0	1.8	3.6	10.8	1.3	2.6	
The Ritz-Carlton, Naples	66.4	18.1	23.3	53.1	12.7	18.0	
Atlanta Midtown Suites Atlanta Norcross	10.5 7.6	1.8 1.0	3.5 1.7	7.8 5.6	1.3 0.7	2.6 1.2	
Atlanta Northwest	14.9	2.7			1.9	3.3	
Atlanta Perimeter	16.6	2.5	4.5	12.6	1.7	3.5	
JW Marriott Hotel at Lenox	24.8	3.7	6.8	17.7	2.6	5.0	
The Ritz-Carlton, Atlanta	30.2	5.8	8.8	21.7	4.1	6.8	
The Ritz-Carlton, Buckhead	49.3	13.1	16.3	35.8	9.2	11.7	
Chicago/Deerfield Suites Chicago/Downers Grove	10.2	1.8	3.1	7.4	1.3	2.3	
Suites	9.0	1.8	2.9	6.7	1.3	2.2	
Chicago/Downtown Courtyard	16.3	3.1	4.9	12.2	2.2	3.9	
Chicago O'HareSouth Bend	40.0 9.9	5.5 1.1	11.5 2.1	28.8 7.0	3.9 0.8	8.2 1.5	
Bethesda	23.2	3.2	5.6	17.3	2.2	4.1	

FIRST THREE QUARTERS
FISCAL YEAR 1997 1998

PROPERTY	G	ROSS ALES	MINIMUM RENT	TOTAL RENT	GROSS SALES	MINIMUM RENT	TOTAL RENT
				(IN MILL			
Gaithersburg/Washingtonian							
Center	\$	13.2	\$ 2.4	\$ 3.8	\$ 9.7	\$ 1.7	\$ 2.8
Boston/Newton		27.4	4.8	7.8		3.4	5.5
Detroit Romulus		8.8	1.1	1.8		0.8	1.4
The Ritz-Carlton, Dearborn		25.7	3.6	5.5			4.0
Minneapolis/Bloomington		20.2	3.3	6.5		2.3	4.7
Minneapolis City Center		27.5	3.7	7.5			5.2
Minneapolis Southwest Kansas City Airport		14.9 14.3	2.7 1.7	4.8 3.7		1.9 1.2	4.0 2.5
St. Louis Pavilion		27.5	6.1	6.5		4.3	4.3
Nashua		7.5	0.8	1.3		0.5	0.9
Newark Airport		39.4	6.5	11.8			8.6
Park Ridge		16.0	2.5	4.0	11.9	1.7	4.2
Saddle Brook		10.7	1.3	2.1	7.8	0.9	1.7
Albany		18.5	3.5	6.1	12.4	2.5	5.2
New York Marriott Financial		00.0		40.0	00.4	- 4	40.4
Center New York Marriott Marquis		39.6 210.3	7.7 40.0	13.2 60.8		5.4 29.7	10.1 47.6
Marriott World Trade Center		65.4	12.2	19.4		8.6	14.9
Charlotte Executive Park		14.0	2.3	3.7		1.6	2.6
Raleigh Crabtree Valley		14.9	2.4	3.9		1.7	2.8
Oklahoma City		15.6	2.0	3.8	10.4	1.4	2.4
Oklahoma City Waterford		9.1	1.5	2.7	6.1	1.0	1.7
Portland		26.4	4.1	7.5	17.6	2.9	4.8
Philadelphia (Convention							
Center)		80.7	14.2	25.0		10.0	17.8
Philadelphia Airport Pittsburgh City Center		25.0 16.4	4.1 1.9	7.6 3.0		2.9 1.3	5.6 2.2
Memphis		10.4	1.5	3.0		1.0	1.8
Dallas/Fort Worth		28.9	5.9	9.3		4.1	7.0
Dallas Quorum		25.7	4.2	8.2		3.0	5.8
El Paso		11.6	0.9	2.3	7.8	0.6	1.4
Houston Airport		21.6	2.8	6.0		2.0	4.6
JW Marriott Houston		27.2	5.0	8.0		3.5	5.9
Plaza San Antonio		13.8	2.9	4.6			3.3
San Antonio Riverwalk		29.3 28.5	6.1	10.3 9.5			7.6 7.2
Salt Lake City Dulles Airport		14.6	5.6 1.8	4.0			3.0
Key Bridge		29.4	5.6	10.2			7.4
Norfolk Waterside		18.1	3.3	5.4			3.8
Pentagon City Residence Inn		11.7	3.5	5.5	8.7	2.5	4.2
The Ritz-Carlton, Tysons							
Corner		34.4	5.9	9.8		4.1	7.3
Washington Dulles Suites Westfields		10.3 28.0	2.5 4.7	4.0 7.4		1.8 3.3	3.0 5.4
Williamsburg		12.6	1.8	2.8			
Washington Metro Center		25.2	4.5	7.3		3.2	5.3
Calgary		13.4	1.7	1.7		1.2	2.3
Toronto Airport		17.1	2.9	5.6	13.0	2.0	4.2
Toronto Eaton Centre		21.1	6.1	7.1		4.3	5.6
Toronto Delta Meadowvale		16.1	2.6	4.9		1.9	3.1
Four Seasons, Atlanta(4)		15.6	5.8	5.9	14.2	4.1	4.5
Four Seasons, Philadelphia(4)		41.1	7.9	12.4	30.6	5.6	10.1
Grand Hyatt, Atlanta(4)		25.3	10.0	10.0		7.0	8.2
Hyatt Regency,		20.0	1010	10.0	22.0	1.0	0.2
Burlingame(4)		47.9	8.8	17.6	39.5	6.2	15.1
Hyatt Regency, Cambridge(4)		32.4	6.7	11.9		4.7	10.4
Hyatt Regency, Reston(4)		30.5	6.5	11.3		4.5	9.2
Swissotel, Atlanta(4)		22.2	5.0	6.3		3.5	5.8
Swissotel, Boston(4)		26.8	6.4	8.5		4.5	6.9
Swissotel, Chicago(4) The Drake (Swissotel), New		38.1	10.9	15.1	28.9	7.7	12.0
York(4)		38.8	11.6	13.6	34.2	8.2	13.4
The Ritz-Carlton, Amelia				•			
Island(4)		45.7	10.3	13.4	37.4	7.2	11.1

	FISCAL YEAR 1997				1998		
PROPERTY	GROSS SALES	MINIMUM RENT		GROSS SALES	MINIMUM RENT	TOTAL RENT	
			(IN MII	LLIONS)			
The Ritz-Carlton, Boston(4)	\$ 40.1	\$ 6.9	\$ 10.5	\$ 31.4	\$ 4.8	\$ 8.8	
Total Full-service Properties	2,790.7	537.9	842.3	2,071.7	379.4	648.4	
Total Courtyard Properties	212.0	50.6	59.2	159.2	35.0	36.8	
Total Residence Inns	69.9	17.2	20.3	50.6	12.0	12.3	
Total	\$3,072.6	\$605.7	\$921.8	\$2,281.6	\$426.4	\$697.5	

FIRST THREE QUARTERS

- H. Represents the adjustment to reflect the Company's anticipated adoption of EITF 97-2 in the fourth quarter of 1998 by recording property-level sales and operating expenses. The adjustment has no impact on operating profit or net income.
- I. The "Pro Forma without Blackstone" column reflects the adjustment to eliminate the revenues and operating expenses and working capital relating to the Blackstone hotel properties assuming that the Blackstone Acquisition does not occur. Amounts related to Crestline's 25% interest in Swissotel Management (USA) L.L.C. have not been included in the pro forma financial statements due to immateriality.
- (1) On a pro forma basis as of September 11, 1998, the Company had 75 million shares of common stock, \$.01 par value authorized with 24.8 million shares issued and outstanding. In addition, on a pro forma basis, 10 million shares of preferred stock, \$.01 par value authorized with none issued or outstanding.
- (2) Reflects the pro forma earnings per share based on 24.8 million weighted average shares outstanding subsequent to the Distribution. Pro forma weighted average shares are based on Host's weighted average shares outstanding, adjusted for a one-for-ten share distribution, and the issuance of shares to the Blackstone Entities.
- (3) Reflects the pro forma earnings per share based on 20.4 million weighted average shares outstanding subsequent to the Distribution. Pro forma weighted average shares are based on Host's weighted average shares outstanding, adjusted for a one-for-ten share distribution, and assumes that the Blackstone acquisition does not occur.
  - (4) Represent the Blackstone hotel properties.

#### AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER (this "Agreement"), dated as of November , 1998, among Host Marriott Corporation, a Delaware corporation ("Host"), HMC Merger Corporation, a Maryland corporation ("Host REIT") and Host Marriott, L.P., a Delaware limited partnership (the "Operating Partnership").

#### **RECITALS**

WHEREAS, Host has adopted an overall plan (the "REIT Conversion") to restructure its business operations so that it will qualify as a real estate investment trust for federal income tax purposes; and

WHEREAS, the transactions comprising the REIT Conversion include, among others, (a) the contribution by Host of its wholly-owned full-service hotels, its interests in certain hotel partnerships and certain other assets to the Operating Partnership, which was organized and is wholly-owned by Host and of which Host will be the sole general partner, in accordance with the terms of a contribution agreement to be entered into in connection therewith (the "OP Contribution") and (b) the reincorporation (the "Reincorporation") of Host from the State of Delaware to the State of Maryland by means of a merger of Host with and into Host REIT, which was formed as a direct wholly-owned subsidiary of Host for the sole purpose of effecting the Reincorporation; and

WHEREAS, the Board of Directors of Host and the Board of Directors of Host REIT each has determined that the OP Contribution and the Reincorporation by means of the Merger (as defined below) are advisable for their respective companies and stockholders, and accordingly each has agreed to effect the OP Contribution and the Reincorporation by means of the Merger provided for herein upon the terms and subject to the conditions set forth herein;

NOW, THEREFORE, in consideration of the foregoing, and of the representations, warranties, covenants and agreements contained herein, the parties hereto hereby adopt the plan of restructuring encompassed by this Agreement and agree as follows:

#### I. THE OP CONTRIBUTION

Subject to the applicable terms and conditions of this Agreement and the terms and conditions of a contribution agreement to be entered into prior to the Merger by Host and the Operating Partnership, Host will effectuate the OP Contribution consisting of the contribution of (a) the wholly-owned full-service hotel assets held by Host, (b) certain interests in various partnerships that own full-service hotels in which Host is the general partner (the "Partnerships"), (c) certain other partnerships or limited liability companies (the "Private Partnerships") that own one or more full-service hotels and that are partially but not wholly-owned by Host (other than its interests in the general partners of the Partnerships, each of which is Host or a wholly-owned, direct or indirect subsidiary of Host, and certain other subsidiaries of Host which will remain in existence as subsidiaries of Host REIT following the REIT Conversion) and (d) certain of its other assets (excluding Host's senior living assets and the cash or other consideration to be distributed to stockholders of Host or Host REIT in connection with the REIT Conversion and certain other de minimis assets that cannot be contributed to the Operating Partnership) to the Operating Partnership in exchange for (i) a number of units of limited partnership interest in the Operating Partnership equal to the number of outstanding shares of Host Common Stock (as defined below) at the completion of the REIT Conversion (and preferred partnership interests in the Operating Partnership corresponding to any shares of preferred stock of Host, if any, outstanding at the time of the REIT Conversion), including a 0.1% general partnership interest and (ii) the assumption by the Operating Partnership of liabilities of Host, including past and future contingent liabilities, other than liabilities of Crestline Capital Corporation, a Maryland corporation ("Crestline"), which currently is a wholly-owned subsidiary of Host but will become a separate public company as part of the REIT Conversion, all as set forth in the contribution agreement. Following the above-described OP Contribution, as well as certain other contributions

to the Operating Partnership by various subsidiaries of Host in connection with the REIT Conversion, the Operating Partnership and its subsidiaries will directly or indirectly own all of Host's and its subsidiaries' wholly-owned hotels, substantially all of Host's direct and indirect interests in both the Partnerships and the Private Partnerships and all of Host's other assets (excluding its senior living assets and the cash or other consideration to be distributed to the stockholders of Host or Host REIT in connection with the REIT Conversion and certain other de minimis assets that cannot be contributed to the Operating Partnership).

#### II. THE MERGER

#### ARTICLE 1

## THE MERGER; CLOSING; EFFECTIVE TIME

- 1.1. THE MERGER. Subject to the terms and conditions of this Agreement, at the Effective Time (as defined in Section 1.3), Host shall be merged with and into Host REIT in accordance with this Agreement and the separate corporate existence of Host shall thereupon cease (the "Merger"). The parties intend that the Merger qualify as a reorganization within the meaning of Section 368(a)(1)(F) of the Internal Revenue Code of 1986, as amended. Host REIT shall be the surviving entity in the Merger (sometimes hereinafter referred to as the "Surviving Corporation") and shall continue to be governed by the laws of the State of Maryland, and the separate existence of Host REIT with all its rights, privileges, immunities, powers and franchises shall continue unaffected by the Merger. The Merger shall have the effects specified in the Delaware General Corporation Law ("DGCL") and the Maryland General Corporation Law ("MGCL").
- 1.2. THE CLOSING. Subject to the terms and conditions of this Agreement, the closing of the Merger (the "Closing") shall take place (a) at the offices of Hogan & Hartson L.L.P., 555 Thirteenth Street, N.W., Washington, D.C., 20004, at 9:00 a.m., local time, on or (b) at such other time, date or place as Host and Host REIT may agree. The date on which the Closing occurs is hereinafter referred to as the "Closing Date."
- 1.3. EFFECTIVE TIME. If all the conditions to the Merger set forth in Article 5 shall have been satisfied or waived in accordance herewith and this Agreement shall not have been terminated as provided in Article 6, following the Closing, the parties hereto shall, at such time as they deem advisable, cause (i) a Certificate of Merger (the "Certificate of Merger") to be executed, acknowledged and filed with the Secretary of State of Delaware, as provided in Section 252 of the DGCL and (ii) Articles of Merger substantially in the form attached hereto as Exhibit A (the "Articles of Merger") to be filed with the State Department of Assessments and Taxation of Maryland (the "SDAT"), as provided in Section 3-107 of the MGCL. The Merger shall become effective upon the filing of the Certificate of Merger with the Secretary of State of Delaware and the acceptance for record of the Articles of Merger by the SDAT, or such other time as specified in the Articles of Merger and the Certificate of Merger (the "Effective Time").

#### ARTICLE 2

#### NAME, ARTICLES OF INCORPORATION AND BYLAWS OF THE SURVIVING CORPORATION

- 2.1. NAME OF SURVIVING CORPORATION. The name of the Surviving Corporation shall be changed at the Effective Time to "Host Marriott Corporation", as provided in the Restated Articles (as defined in Section 2.2 hereof).
- 2.2. ARTICLES OF INCORPORATION. The Articles of Amendment and Restatement of Articles of Incorporation of Host REIT, substantially in the form attached hereto as Exhibit B (the "Restated Articles"), shall be filed with the SDAT as part of the Merger and shall be the Articles of Incorporation of the Surviving Corporation until duly amended in accordance with the terms thereof and the MGCL.

2.3. BYLAWS. The Bylaws of Host REIT in effect immediately prior to the Effective Time (a copy of which is attached hereto as Exhibit C) shall be the Bylaws of the Surviving Corporation until duly amended in accordance with the terms thereof and the MGCL.

#### ARTICLE 3

#### DIRECTORS AND OFFICERS OF THE SURVIVING CORPORATION

- 3.1. DIRECTORS. As set forth in the Restated Articles, the directors of Host immediately prior to the Effective Time shall be the directors of the Surviving Corporation as of and following the Effective Time until their successors have been duly elected and qualified or until their earlier death, resignation or removal in accordance with the Articles of Incorporation and Bylaws of the Surviving Corporation.
- 3.2. OFFICERS. The officers of Host immediately prior to the Effective Time shall be the officers of the Surviving Corporation as of and following the Effective Time until their successors have been duly appointed and qualified or until their earlier death, resignation or removal in accordance with the Articles of Incorporation and Bylaws of the Surviving Corporation.

#### ARTICLE 4

## EFFECT ON CAPITAL STOCK; EXCHANGE OF CERTIFICATES

- 4.1 EFFECT ON CAPITAL STOCK. At the Effective Time, by virtue of the Merger and without any further action on the part of any holder of any capital stock of Host:
  - (a) Each share of common stock, par value \$1.00 per share, of Host ("Host Common Stock") issued and outstanding immediately prior to the Effective Time, together with the associated right ("Host Right") issued under the Rights Agreement dated as of February 3, 1989, as amended, between Marriott Corporation (the former name of Host) and the Bank of New York (the "Host Rights Agreement"), shall be converted into one validly issued, fully paid and nonassessable share of common stock, par value \$0.01 per share, of Host REIT ("Host REIT Common Stock"), together with the associated right issued under a stockholder rights plan to be adopted by Host REIT prior to the Merger (the "Host REIT Stockholder Rights Plan"). Any shares of preferred stock of Host outstanding immediately prior to the Effective Time would be converted into an equal number of shares of preferred stock of Host REIT having substantially the same rights and preferences.
  - (b) All shares of Host Common Stock shall no longer be outstanding and shall be canceled and retired and shall cease to exist. At the Effective Time, each certificate representing shares of Host Common Stock will be deemed for all purposes to evidence the same number of shares of Host REIT Common Stock until such certificate is exchanged for a certificate representing shares of Host REIT Common Stock in accordance with Article 4 hereof.
  - (c) Each share of Host Common Stock held in Host's treasury at the Effective Time shall, by virtue of the Merger and without any action on the part of the holder thereof, cease to be outstanding, shall be canceled and retired without payment of any consideration therefor and shall cease to exist.
  - (d) Each share of Host REIT Common Stock issued and outstanding immediately prior to the Effective Time shall, by virtue of the Merger and without any action on the part of Host REIT or the holder of such shares, be canceled and retired without payment of any consideration therefor.
  - (e) Host's obligations with respect to any dividends or other distributions to the stockholders of Host that have been declared by Host but not paid prior to the Effective Time will be assumed by Host REIT in accordance with the terms thereof.

4.2 BENEFIT PLANS. At the Effective Time and by virtue of the Merger, the rights and obligations of Host under the Host Marriott Corporation 1997 Comprehensive Stock Incentive Plan (formerly called the Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan), the Host Marriott Corporation Employee Stock Purchase Plan, the Host Marriott Corporation (HMC) Retirement and Savings Plan, the Host Marriott Corporation Executive Deferred Compensation Plan and the Host Marriott Corporation Medical/Dental/Vision Care Plan (collectively, the "Plans") and related and other agreements will be transferred to the Operating Partnership, and all rights of the parties thereto and the participants therein to acquire shares of Host Common Stock on the terms and conditions of the Plans and such agreements will be converted into rights to acquire shares of Host REIT Common Stock, in each case, to the extent set forth in, and in accordance with, the terms of such Plans and related and other agreements and the Employee Benefits and Other Employment Matters Allocation Agreement to be entered into by Host in connection with the REIT Conversion. The Host Marriott Corporation Non-Employee Director's Deferred Stock Compensation Plan will be continued by Host REIT.

#### 4.3 EXCHANGE OF CERTIFICATES.

- (a) As of the Effective Time, Host REIT shall deposit, or shall cause to be deposited, with First Chicago Trust Company of New York, the transfer agent and registrar for the shares of Host REIT Common Stock and the exchange agent for purposes of the Merger (the "Transfer and Exchange Agent"), for exchange in accordance with this Article 4, certificates evidencing shares of Host REIT Common Stock (the "Exchange Fund") to be issued pursuant to Section 4.1 and delivered pursuant to this Section 4.3 in exchange for outstanding securities of Host.
- (b) As soon as practicable after the Effective Time, Host REIT shall cause the Transfer and Exchange Agent to mail to each holder of record of a certificate or certificates which immediately prior to the Effective Time represented outstanding shares of Host Common Stock ("Certificates"), (i) a letter of transmittal which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon delivery of the Certificates to the Transfer and Exchange Agent and shall be in such form and have such other provisions as Host REIT may reasonably specify and (ii) instructions for use in effecting the surrender of the Certificates in exchange for certificates evidencing shares of Host REIT Common Stock. Upon surrender of a Certificate for cancellation to the Transfer and Exchange Agent together with such letter of transmittal, duly executed and completed in accordance with the instructions thereto, the holder of such Certificate shall be entitled to receive in exchange therefor a certificate evidencing the number of shares of Host REIT Common Stock which such holder has the right to receive in respect of the Certificate surrendered pursuant to the provisions of this Article 4, and the Certificate so surrendered shall forthwith be canceled. In the event of a surrender of a Certificate representing shares of Host Common Stock in exchange for a certificate evidencing shares of Host REIT Common Stock in the name of a person other than the person in whose name such shares of Host Common Stock are registered, a certificate evidencing the proper number of shares of Host REIT Common Stock that may be issued to such a transferee if the Certificate evidencing such securities is presented to the Transfer and Exchange Agent, accompanied by all documents required by the Transfer and Exchange Agent or Host REIT to evidence and effect such transfer and to evidence that any applicable transfer taxes have been paid.
- (c) All shares of Host REIT Common Stock issued upon the surrender for exchange of shares of Host Common Stock in accordance with the terms hereof shall be deemed to have been issued in full satisfaction of all rights pertaining to such shares of Host Common Stock, subject, however, to the Surviving Corporation's obligation to pay any dividends or make any other distributions with a record date prior to the Effective Time which may have been declared or made by Host on such Host Common Stock and which remain unpaid at the Effective Time. Any such dividends or distributions that would have entitled stockholders of Host to receive shares of Host Common Stock shall be deemed, after the Effective Time, to entitle stockholders of Host REIT to receive the same number of shares of Host REIT Common Stock.
- (d) At and after the Effective Time, there shall be no transfers on the stock transfer books of Host of shares of Host Common Stock which were outstanding immediately prior to the Effective Time. If, after the Effective Time, Certificates are presented to the Surviving Corporation, they shall be canceled and exchanged for certificates representing shares of Host REIT Common Stock in accordance with the procedures set forth in this Article 4.

- (e) No fractional shares of Host REIT Common Stock shall be issued pursuant
- (f) Any portion of the Exchange Fund that remains unclaimed by the former stockholders of Host one year after the Effective Time shall be delivered to the Surviving Corporation. Any former stockholders of Host who have not theretofore complied with this Article 4 shall thereafter look only to the Surviving Corporation for payment of their shares of Host REIT Common Stock deliverable in respect of each share of Host Common Stock such stockholder holds as determined pursuant to this Agreement, without any interest thereon. None of Host, Host REIT, the Transfer and Exchange Agent or any other person shall be liable to any former holder of shares or securities of Host for any amount properly delivered to a public official pursuant to applicable abandoned property, escheat or similar laws.
- (g) In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by the Surviving Corporation, the posting by such person of a bond in such reasonable amount as the Surviving Corporation may direct as indemnity against any claim that may be made against it with respect to such Certificate, the Transfer and Exchange Agent or the Surviving Corporation will issue in exchange for such lost, stolen or destroyed Certificate the shares of Host REIT Common Stock deliverable in respect thereof pursuant to this Agreement.

## ARTICLE 5

#### CONDITIONS

- 5.1 CONDITIONS AS TO EACH PARTY'S OBLIGATION TO EFFECT THE MERGER. The respective obligations of Host and Host REIT to consummate the Merger are subject to the satisfaction or, with respect to Sections 5.1(c) and (d) only, waiver, of the following conditions:
  - (a) The Merger shall have been duly approved by Host, as the sole stockholder of Host REIT, and this Agreement shall have been duly approved by the holders of Host Common Stock representing not less than two-thirds (66 2/3%) of the outstanding shares of Host Common Stock.
  - (b) Host's Board of Directors shall have determined (i) that the transactions constituting the REIT Conversion which impact Host REIT's status as a REIT for federal income tax purposes have occurred or are reasonably likely to occur, and (ii) based upon the advice of counsel, that Host REIT can elect to be treated as a REIT for federal income tax purposes effective no later than the first full taxable year commencing after the REIT Conversion is completed.
  - (c) Host REIT shall have adopted the Host REIT Stockholder Rights Plan, which Plan shall have been approved by the Board of Directors of Host prior to the Effective Time.
  - (d) Host shall amend the Host Rights Agreement to provide that each Host Right issued pursuant to the Host Rights Agreement will be converted, as of the Effective Time, into a right issued under the Host REIT Stockholder Rights Plan.
  - (e) Host shall have received all governmental and third party consents to the Merger, including consents of lenders and Marriott International, Inc., except for consents as would not reasonably be expected to have a material adverse effect on the business, financial condition or results of operations of Host REIT, the Operating Partnership and their subsidiaries taken as a whole.
  - (f) Host shall have determined, in its sole discretion, that The United States Congress shall not have enacted legislation, or proposed legislation with a reasonable possibility of being enacted, that would have the effect of (i) substantially impairing the ability of Host REIT to qualify as a REIT or the Operating Partnership to qualify as a partnership, (ii) substantially increasing the federal tax liabilities of Host REIT resulting from the REIT Conversion or (iii) substantially reducing the expected benefits to Host REIT resulting from the REIT Conversion.

# ARTICLE 6

# **TERMINATION**

- 6.1 TERMINATION OF AGREEMENT. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, before or after the approval of this Agreement by the stockholders of Host, by either (i) the mutual written consent of the Board of Directors of Host and the Board of Directors of Host REIT or (ii) the Board of Directors of Host in its sole discretion.
- 6.2 EFFECT OF TERMINATION AND ABANDONMENT. In the event of termination of this Agreement and abandonment of the Merger pursuant to this Article 6, no party hereto (or any of its directors or officers) shall have any liability or further obligation to any other party to this Agreement.

#### ARTICLE 7

#### GENERAL PROVISIONS

- 7.1 ENTIRE AGREEMENT. This Agreement, the Exhibits and any documents delivered by the parties in connection herewith constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings among the parties with respect thereto. No addition to or modification of any provision of this Agreement shall be binding upon any party hereto unless made in writing and signed by all parties hereto.
- 7.2 AMENDMENT. This Agreement may be amended by the parties hereto, by action taken by their respective Boards of Directors at any time before or after approval of the Agreement by the stockholders of Host, but after such stockholder approval, no amendment shall be made which by law requires the further approval of such stockholders without obtaining such further approval. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- 7.3 GOVERNING LAW. This Agreement shall be governed by and construed in accordance with the laws of the States of Delaware and Maryland.
- 7.4 COUNTERPARTS. This Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument. Each counterpart may consist of a number of copies hereof, each signed by less than all, but together signed by all of the parties hereto.
- 7.5 HEADINGS. Headings of the Articles and Sections of this Agreement are for the convenience of the parties only, and shall be given no substantive or interpretive effect whatsoever.
- 7.6 INCORPORATION. All Exhibits attached hereto and referred to herein are hereby incorporated herein and made a part hereof for all purposes as if fully set forth herein.
- 7.7 SEVERABILITY. Any term or provision of this Agreement which is invalid or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.
- 7.8 WAIVER OF CONDITIONS. The conditions to each of the parties' obligations to consummate the Merger are for the sole benefit of such party and may be waived by such party in whole or in part to the extent permitted by applicable law.
- 7.9 NO THIRD-PARTY BENEFICIARIES. This Agreement is not intended to confer upon any person other than the parties hereto any rights or remedies hereunder.

ATTEST:	
	HOST MARRIOTT CORPORATION
By:	Ву:
[Title]	(SEAL) [Title]
ATTEST:	
	HMC MERGER CORPORATION
By:	By: (SEAL)
[Title]	[Title]
ATTEST:	
	HOST MARRIOTT, L.P. (for purposes of Section I only)
By:	By:
[Title]	(SEAL) [Title]
A	u-7

IN WITNESS WHEREOF, the parties have executed this Agreement and caused the same to be duly delivered on their behalf on the day and year first written above.

# ARTICLES OF MERGER

BETWEEN

HMC MERGER CORPORATION (a Marvland Corporation)

AND

HOST MARRIOTT CORPORATION (a Delaware Corporation)

HMC Merger Corporation, a corporation duly organized and existing under the laws of the State of Maryland ("Host-MD"), and Host Marriott Corporation, a corporation duly organized and existing under the laws of the State of Delaware ("Host-DE"), do hereby certify that:

FIRST: Host-MD and Host-DE agree to merge.

SECOND: The name and place of incorporation of each party to these Articles are HMC Merger Corporation, a Maryland corporation, and Host Marriott Corporation, a Delaware corporation. Host-MD shall be the successor corporation in the merger. Upon consummation of the merger, the current name of Host-MD shall be changed to "Host Marriott Corporation" and Host-MD shall continue as a corporation of the State of Maryland.

THIRD: Host-DE was incorporated on July 10, 1929, under the Delaware General Corporation Law under the name Marriott Corporation, which name was changed to Host Marriott Corporation on October 8, 1993 pursuant to an amendment to the Certificate of Incorporation of Host-DE. Host-DE has been registered and qualified to do business in Maryland since May 7, 1937 and owns an interest in land in Montgomery County, Maryland.

FOURTH: Both Host-MD and Host-DE have their principal offices in Bethesda, Maryland, Montgomery County.

FIFTH: The terms and conditions of the transaction set forth in these Articles were advised, authorized, and approved by each corporation party to these Articles in the manner and by the vote required by its Articles or Certificate of Incorporation, as the case may be, and the laws of its state of incorporation. The manner of approval was as follows:

- (a) The directors of Host-DE determined the proposed agreement and plan of merger to be advisable for Host-DE and its stockholders, approved the agreement and plan of merger at a special meeting of the board of directors on November , 1998 and directed that the agreement and plan of merger be submitted for consideration by the stockholders of Host-DE at a special meeting for the purpose of acting on the agreement and plan of merger. At a special meeting of Host-DE stockholders held on December , 1998, the stockholders of Host-DE adopted a resolution approving the agreement and plan of merger by the affirmative vote of at least two-thirds (66 2/3%) of the shares of capital stock of Host-DE outstanding on the record date for such special meeting and entitled to vote thereon.
- (b) The Board of Directors of Host-MD adopted a resolution at a special meeting of the board of directors on November , 1998 which declared that the proposed merger was advisable on substantially the terms and conditions set forth or referred to in the resolution and directed that such resolution be submitted for consideration by the sole stockholder of Host-MD. In accordance with Sections 2-504(d) and 2-505 of the Maryland General Corporation Law, the sole stockholder of Host-MD adopted a resolution by unanimous written consent dated as of December , 1998, approving the proposed merger.

SIXTH: The Articles of Incorporation of Host-MD, the successor corporation, shall be amended and restated as part of the merger to read as set forth in Exhibit A to these Articles.

SEVENTH: (a) The total number of shares of stock of all classes which Host-DE has authority to issue is 601,000,000 shares, of which 600,000,000 shares are classified as common stock, par value \$1.00 per share ("Host-DE Common Stock"), and 1,000,000 shares are classified as preferred stock, without par value.

- (b) The aggregate par value of all the shares of stock of all classes of Host-DE is \$600,000,000.
- EIGHTH: (a) As of immediately before the merger, the total number of shares of stock of all classes which Host-MD has authority to issue is 1,000 shares of common stock, par value \$0.01 per share.
- (b) As changed by the merger, the total number of shares of stock of all classes which Host-MD has authority to issue is 800,000,000 shares, of which 750,000,000 shares are classified as common stock, par value \$0.01 per share ("Host-MD Common Stock"), and 50,000,000 shares are classified as preferred stock, par value \$0.01 per share.
- (c) The aggregate par value of all the shares of stock of all classes of Host-MD is \$10\$ before the merger and \$8,000,000 as changed by the merger.

NINTH: The merger shall become effective at p.m. (Eastern Time) on December , 1998 (the "Effective Time").

TENTH: The manner and basis of converting or exchanging issued stock of the merging corporations into different stock of a corporation, or other consideration, and the treatment of any issued stock of the merging corporations not to be converted or exchanged, are as follows:

- (a) Each share of Host-DE Common Stock issued and outstanding immediately prior to the Effective Time, together with the associated right ("Host Right") issued under the Rights Agreement dated as of February 3, 1989, as amended, between Marriott Corporation (the former name of Host) and the Bank of New York (the "Host Rights Agreement"), shall, upon effectiveness of the merger and without further act, be converted into one validly issued, fully paid and nonassessable share of Host-MD Common Stock, together with the associated right issued under a stockholder rights plan to be adopted by Host-MD prior to the Merger (the "Host-MD Stockholder Rights Plan").
- (b) After the Effective Time, all shares of Host-DE Common Stock shall no longer be outstanding and shall be canceled and retired and shall cease to exist. At the Effective Time, each certificate representing shares of Host-DE Common Stock will be deemed for all purposes to evidence the same number of shares of Host-MD Common Stock until exchanged for a certificate representing shares of Host-MD Common Stock in accordance with the terms of the Agreement and Plan of Merger.
- (c) Each share of Host-DE Common Stock held in Host-DE's treasury at the Effective Time shall, by virtue of the merger and without any action on the part of the holder thereof, cease to be outstanding, shall be canceled and retired without payment of any consideration therefor and shall cease to exist.
- (d) Each share of Host-MD Common Stock issued and outstanding immediately prior to the Effective Time shall, by virtue of the merger and without any action on the part of Host-MD or the holder of such shares, be canceled and retired without payment of any consideration therefor.
- (e) Host's obligations with respect to any dividends or other distributions to the stockholders of Host-DE that have been declared by Host-DE but not paid prior to the Effective Time will be paid by Host-MD in accordance with the terms thereof.
- (f) As soon as practicable after the Effective Time, Host-MD shall cause First Chicago Trust Company of New York, the transfer agent and registrar for the Host-MD Common Stock and the exchange agent for purposes of the merger (the "Transfer and Exchange Agent"), to mail to each holder of record of a

certificate which immediately prior to the Effective Time represented outstanding shares of Host-DE Common Stock ("Certificates"), (i) a letter of transmittal and (ii) instructions for use in effecting the surrender of such Certificates in exchange for certificates evidencing Host-MD Common Stock. Upon surrender of a Certificate for cancellation to the Transfer and Exchange Agent together with such letter of transmittal, duly executed and completed in accordance with the instructions thereto, the holder of such Certificate shall be entitled to receive in exchange therefor a certificate evidencing the number of shares of Host-MD Common Stock which such holder has the right to receive in respect of the Certificate surrendered pursuant to the terms of the Agreement and Plan of Merger and the Certificate so surrendered shall forthwith be canceled.

IN WITNESS WHEREOF, HMC Merger Corporation and Host Marriott Corporation have caused these presents to be signed in their respective names and on their respective behalves by their respective presidents and witnessed by their respective secretaries on December  $\,$  , 1998.

WITNESS:

	HMC MERGER CORPORATION
Christopher G. Townsend Secretary	Terence C. Golden President
WITNESS:	HOST MARRIOTT CORPORATION
Christopher G. Townsend Secretary	Terence C. Golden President and Chief Executive Officer

THE UNDERSIGNED, President of HMC Merger Corporation, a Maryland corporation, who executed on behalf of said corporation the foregoing Articles of Merger of which this certificate is made a part, hereby acknowledges in the name and on behalf of said corporation the foregoing Articles of Merger to be the corporate act of said corporation and hereby certifies that to the best of his knowledge, information and belief the matters and facts set forth therein with respect to the authorization and approval thereof are true in all material respects under the penalties of perjury.

- ------

Terence C. Golden President

THE UNDERSIGNED, President and Chief Executive Officer of Host Marriott Corporation, a Delaware corporation, who executed on behalf of said corporation the foregoing Articles of Merger of which this certificate is made a part, hereby acknowledges in the name and on behalf of said corporation the foregoing Articles of Merger to be the corporate act of said corporation and hereby certifies that to the best of his knowledge, information and belief the matters and facts set forth therein with respect to the authorization and approval thereof are true in all material respects under the penalties of perjury.

Terence C. Golden
President and Chief Executive Officer

FORM OF ARTICLES OF AMENDMENT AND RESTATEMENT

0F

ARTICLES OF INCORPORATION OF

HMC MERGER CORPORATION

HMC Merger Corporation, a Maryland corporation having its principal office in Maryland in Bethesda, Maryland, and having CSC-Lawyers Incorporating Service Company as its resident agent, located at 11 East Chase Street, Baltimore, Maryland 21202, hereby certifies to the State Department of Assessments and Taxation of Maryland that:

FIRST: The Articles of Incorporation of the Corporation, filed with the State Department of Assessments and Taxation of Maryland on September 28, 1998, are hereby amended and restated in full as follows:

ARTICLE I NAME

The name of the corporation (which is hereinafter called the "Corporation") is Host Marriott Corporation.

ARTICLE II

The purposes for which the Corporation is formed are to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Maryland (the "MGCL").

ARTICLE III PRINCIPAL OFFICE

The present address of the principal office of the Corporation in the State of Maryland is 10400 Fernwood Road, Bethesda, Maryland 20817.

ARTICLE IV REGISTERED AGENT

The name and address of the resident agent of the Corporation in the State of Maryland is CSC-Lawyers Incorporating Service Company, 11 East Chase Street, Baltimore, Maryland 21202. Said resident agent is a Maryland corporation.

ARTICLE V

The duration of the Corporation shall be perpetual.

ARTICLE VI CAPITALIZATION

Section 6(a) Shares and Par Value. The total number of shares of stock of all classes ("Capital Stock") which the Corporation has authority to issue is 800,000,000 shares, 750,000,000 of which initially are classified

as common stock, par value of \$.01 per share ("Common Stock"), and 50,000,000 of which initially are classified as preferred stock, par value \$.01 per share ("Preferred Stock"). The aggregate par value of all classes of stock that the Corporation shall have authority to issue is \$8,000,000. The Board of Directors may, by adopting a resolution and filing articles supplementary with the State Department of Assessments and Taxation of Maryland, classify and reclassify any unissued shares of Capital Stock by setting or changing in any one or more respects the preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends, qualifications, or terms or conditions of redemption of such shares of stock. The power of the Board of Directors under this Section 6(a) to classify and reclassify any of the shares of Capital Stock shall include, without limitation, authority to classify or reclassify any unissued shares of such stock (including shares initially designated as Common Stock or Preferred Stock above) into Common Stock, Preferred Stock, a class or classes of preferred stock, preference stock, special stock or other stock (including non-voting common stock), and to divide and classify shares of any class into one or more series of such class. Unless otherwise specifically provided for in the terms of any class or series of stock now or hereafter created, the amount that would be needed, if the Corporation were to be dissolved at the time of a distribution, to satisfy the preferential rights on dissolution of stockholders whose preferential rights are superior to those receiving the distribution, shall not limit the ability of the Corporation to make any distribution or the amount thereof.

Section 6(b) Common Stock. The following is a description of the preferences, conversion and other rights, voting powers, restrictions, limitations as to dividends, qualifications, and terms and conditions of redemption of the Common Stock of the Corporation:

- (1) Voting Rights. Each share of Common Stock shall have one vote on all actions to be taken by the stockholders of the Corporation, and, except as otherwise provided in respect of any class of stock at any time classified or reclassified, the exclusive voting power for all purposes shall be vested in the holders of the Common Stock.
- (2) Dividends. Subject to the provisions of law and any preferences of any class of Capital Stock, including any shares of Preferred Stock, hereafter classified or reclassified, dividends, including dividends payable in shares of another class of the Corporation's stock, may be paid on the Common Stock of the Corporation at such time and in such amounts as the Board of Directors may deem advisable and the holders of the Common Stock shall share ratably in any such dividends, in proportion to the number of shares of Common Stock held by them respectively, on a share for share basis.
- (3) Liquidation Rights. In the event of any liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, the holders of the Common Stock shall be entitled, after payment or provision for payment of the debts and other liabilities of the Corporation and the amount to which the holders of any class of Capital Stock at any time classified or reclassified having a preference on distributions in the liquidation, dissolution or winding up of the Corporation are entitled, including any shares of Preferred Stock, together with the holders of any other class of Capital Stock hereafter classified or reclassified not having a preference on distributions in the liquidation, dissolution or winding up of the Corporation, to share ratably in the remaining net assets of the Corporation.

Section 6(c) Preferred Stock. The Board of Directors shall have the authority to classify and reclassify any unissued shares of Preferred Stock from time to time by setting or changing in any one or more respects the preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends, qualifications, or terms or conditions of redemption of such shares of the Preferred Stock. Subject to the foregoing, the power of the Board of Directors to classify and reclassify any of the shares of Preferred Stock shall include, without limitation, subject to the provisions of the charter, authority to classify or reclassify any of the shares of such stock into Common Stock, a class or classes of preferred stock, preference stock, special stock or other stock, and to divide and classify shares of any class into one or more series of such class, by determining, fixing, or altering one or more of the following:

(1) The distinctive designation of such class or series and the number of shares to constitute such class or series; provided that, unless otherwise prohibited by the terms of such or any other class or series, the

number of shares of any class or series may be decreased by the Board of Directors in connection with any classification or reclassification of unissued shares and the number of shares of such class or series may be increased by the Board of Directors in connection with any such classification or reclassification, and any shares of any class or series which have been redeemed, purchased, otherwise acquired or converted into shares of Common Stock or any other class or series shall become part of the authorized class of stock so redeemed, purchased, otherwise acquired or converted into shares of Common Stock and be subject to classification and reclassification as provided in this Article VI.

- (2) Whether or not and, if so, the rates, amounts and times at which, and the conditions under which, dividends shall be payable on shares of such class or series, whether any such dividends shall rank senior or junior to or on a parity with the dividends payable on any other class or series of stock, and the status of any such dividends as cumulative, cumulative to a limited extent or non-cumulative and as participating or non-participating.
- (3) Whether or not shares of such class or series shall have voting rights, in addition to any voting rights provided by law and, if so, the terms of such voting rights.
- (4) Whether or not shares of such class or series shall have conversion or exchange privileges and, if so, the terms and conditions thereof, including provision for adjustment of the conversion or exchange rate in such events or at such times as the Board of Directors shall determine.
- (5) Whether or not shares of such class or series shall be subject to redemption and, if so, the terms and conditions of such redemption, including the date or dates upon or after which they shall be redeemable and the amount per share payable in case of redemption, which amount may vary under different conditions and at different redemption dates; and whether or not there shall be any sinking fund or purchase account in respect thereof, and if so, the terms thereof.
- (6) The rights of the holders of shares of such class or series upon the liquidation, dissolution or winding up of the affairs of, or upon any distribution of the assets of, the Corporation, which rights may vary depending upon whether such liquidation, dissolution or winding up is voluntary or involuntary and, if voluntary, may vary at different dates, and whether such rights shall rank senior or junior to or on a parity with such rights of any other class or series of stock.
- (7) Whether or not there shall be any limitations applicable, while shares of such class or series are outstanding, upon the payment of dividends or making of distributions on, or the acquisition of, or the use of moneys for purchase or redemption of, any stock of the Corporation, or upon any other action of the Corporation, including action under this Section 6(c), and, if so, the terms and conditions thereof.
- (8) Any other preferences, rights, restrictions, including restrictions on transferability, and qualifications of shares of such class or series, not inconsistent with law and the charter of the Corporation.

Section 6(d) Ranking of Classes or Series of Capital Stock. For the purposes hereof and of any articles supplementary to the charter providing for the classification or reclassification of any shares of Capital Stock or of any other charter document of the Corporation (unless otherwise provided in any such articles or document), any class or series of stock of the Corporation shall be deemed to rank:

- (1) prior to another class or series either as to dividends or upon liquidation, if the holders of such class or series shall be entitled to the receipt of dividends or of amounts distributable on liquidation, dissolution or winding up, as the case may be, in preference or priority to holders of such other class or series;
- (2) on a parity with another class or series either as to dividends or upon liquidation, whether or not the dividend rates, dividend payment dates or redemption or liquidation price per share thereof be different from those of such others, if the holders of such class or series of stock shall be entitled to receipt of dividends or amounts distributable upon liquidation, dissolution or winding up, as the case may be, in proportion to their respective dividend rates or redemption or liquidation prices, without preference or priority over the holders of such other class or series; and

(3) junior to another class or series either as to dividends or upon liquidation, if the rights of the holders of such class or series shall be subject or subordinate to the rights of the holders of such other class or series in respect of the receipt of dividends or the amounts distributable upon liquidation, dissolution or winding up, as the case may be.

# ARTICLE VII BOARD OF DIRECTORS

Section 7(a) Number of Directors; Classification. Effective upon the filing of these Articles of Amendment and Restatement with the State Department of Assessments and Taxation of Maryland, the number of directors shall be increased from two (2) to eight (8). Except as otherwise fixed by or pursuant to the provisions of Article VI hereof relating to the rights of the holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation to elect additional directors under specified circumstances, the number of directors may thereafter be increased or decreased pursuant to the Bylaws of the Corporation; provided such number established in accordance with the Bylaws is not decreased to less than three (3) nor increased to more than thirteen (13). The directors, other than those who may be elected by the holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation, shall be divided into three classes as nearly equal in number as possible, with the term of office of one class expiring each year. One class of directors, consisting initially of three members, shall hold office initially for a term expiring at the annual meeting of stockholders in 1999 (Class I); another class, consisting initially of three members, shall hold office initially for a term expiring at the annual meeting of stockholders in 2000 (Class II); and the third class, consisting initially of two members, shall hold office initially for a term expiring at the annual meeting of shareholders in 2001 (Class III). In the event of any increase or decrease in the number of directors, other than resulting from the rights of the holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation to elect additional directors pursuant to the provisions of Article VI hereof, the newly created or eliminated directorships resulting from such increase or decrease shall be apportioned by the Board of Directors among the three classes of directors so as to maintain such classes as nearly equal in number as possible. The names and classes of the directors upon the filing of these Articles of Amendment and Restatement with the State Department of Assessments and Taxation of Maryland who will serve until their successors are elected and qualify are:

CLASS OF DIRECTOR NAME

Class I: J.W. Marriott, Jr.

Harry L. Vincent, Jr.

John G. Schreiber Class II: Robert M. Baylis

Ann Dore McLaughlin

Terence C. Golden Class III: R. Theodore Ammon

Class III: R. Theodore Ammon Richard E. Marriott

Section 7(b) Removal of Directors. Subject to the rights of holders of one or more classes or series of Capital Stock other than Common Stock to elect one or more directors, any director may be removed only for cause and only by the affirmative vote of stockholders holding at least two thirds of all the votes entitled to be cast for the election of directors.

Section 7(c) Vacancies. Except in the case of a vacancy on the Board of Directors among the directors elected by a class or series of Capital Stock other than Common Stock, any vacancy on the Board of Directors may be filled by the affirmative vote of the remaining directors (except that a vacancy which results from an increase in the number of directors may be filled by a majority of the entire Board of Directors) and, in the case of a vacancy resulting from the removal of a director, by the stockholders by the affirmative vote of two-thirds

of the votes entitled to be cast for the election of directors. Any vacancy on the Board of Directors among the directors elected by a class or series of Capital Stock other than Common Stock may be filled by a majority of the remaining directors elected by that class or series or by the sole remaining director elected by that class or series, or by the stockholders of that class or series unless otherwise provided in the articles supplementary for that class or series.

Section 7(d) Amendments. Notwithstanding any other provisions of the charter or Bylaws of the Corporation, the affirmative vote of stockholders holding at least two-thirds of all of the votes entitled to be cast thereon shall be required to amend, alter, change, repeal, or adopt any provisions inconsistent with, the provisions of this ARTICLE VII.

#### ARTICLE VIII

RESTRICTION ON TRANSFER AND OWNERSHIP OF SHARES OF CAPITAL STOCK

Section 8.1 Definitions. For the purpose of this Article VIII, the following terms shall have the following meanings:

Beneficial Ownership. The term "Beneficial Ownership" shall mean ownership of shares of Capital Stock by a Person, whether the interest in the shares of Capital Stock is held directly or indirectly (including by a nominee), and shall include (in addition to direct ownership and indirect ownership through a nominee or similar arrangement) interests that would be treated as owned through the application of Section 544 of the Code, as modified by Section 856(h)(1)(B) of the Code. The terms "Beneficial Owner," "Beneficially Owns" and "Beneficially Owned" shall have the correlative meanings.

Benefit Plan Investor. The term "Benefit Plan Investor" shall have the meaning provided in 29 C.F.R. (S) 2510.3-101(f)(2), or any successor regulation thereto.

Business Day. The term "Business Day" shall mean any day, other than a Saturday or Sunday, that is neither a legal holiday nor a day on which banking institutions in the state of Maryland or in the state of New York are authorized or required by law, regulation or executive order to close.

Charitable Beneficiary. The term "Charitable Beneficiary" shall mean one or more beneficiaries of the Charitable Trust as determined pursuant to Section 8.3.7, provided that each such organization must be described in Sections 501(c)(3), 170(b)(1)(A) (other than clause (vii) or (viii) thereof) and 170(c)(2) of the Code and contributions to each such organization must be eligible for deduction under each of Sections 170(b)(1)(A), 2055 and 2522 of the Code.

Charitable Trust. The term "Charitable Trust" shall mean any trust provided for in Section 8.2.1(b)(i) and Section 8.3.1.

Charitable Trustee. The term "Charitable Trustee" shall mean the Person, unaffiliated with the Corporation and a Prohibited Owner, that is appointed by the Corporation from time to time to serve as trustee of the Charitable Trust. In the absence of such designation, the Charitable Trustee shall be

Closing Price. The "Closing Price" on any date shall mean the last sale price for such shares of Capital Stock, regular way, or, in case no such sale takes place on such day, the average of the closing bid and asked prices, regular way, for such shares of Capital Stock, in either case as reported on the principal consolidated transaction reporting system with respect to securities listed or admitted to trading on the NYSE or, if such shares of Capital Stock are not listed or admitted to trading on the NYSE, as reported on the principal consolidated transaction reporting system with respect to securities listed on the principal national securities exchange on which such shares of Capital Stock are listed or admitted to trading or, if such shares of Capital Stock are not listed or admitted to trading on any national securities exchange, the last quoted price, or, if not so quoted, the average of the high bid and low asked prices, in the over-thecounter market, as reported by the NASDAQ Stock Market or, if such system is no longer in use, the principal other automated quotation system that may then be in use or, if such shares of Capital Stock are not quoted by any such organization, the average of the closing bid and asked prices as furnished by a

professional market maker making a market in such shares of Capital Stock selected by the Board of Directors or, in the event that no trading price is available for such shares of Capital Stock, the fair market value of such shares, as determined in good faith by the Board of Directors.

Code. The term "Code" means the Internal Revenue Code of 1986, as amended

Constructive Ownership. The term "Constructive Ownership" shall mean ownership of shares of Capital Stock by a Person, whether the interest in shares of Capital Stock is held directly or indirectly (including by a nominee), and shall include any interests that would be treated as owned through the application of Section 318(a) of the Code, as modified by Section 856(d)(5) of the Code. The terms "Constructive Owner," "Constructively Owns" and "Constructively Owned" shall have the correlative meanings.

Effective Date. The term "Effective Date" shall mean the date on which the Effective Time occurs.

Effective Time. The term "Effective Time" shall mean the later of (i) the time the State Department of Assessments and Taxation of Maryland accepts the articles of merger for the Merger for record or (ii) the time established under the articles of merger for the Merger.

ERISA Investor. The term "ERISA Investor" shall mean any holder of shares of Capital Stock that is (i) an employee benefit plan subject to Title I of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), (ii) a plan as defined in Section 4975(e) of the Code (any such employee benefit plan or "plan" described in clause (i) or this clause (ii) being referred to herein as a "Plan"), (iii) a trust which was established pursuant to a Plan, or a nominee for such trust or Plan, or (iv) an entity whose underlying assets include assets of a Plan by reason of such Plan's investment in such entity.

Excepted Holder. The term "Excepted Holder" shall mean a stockholder of the Corporation for whom an Excepted Holder Limit is created by the Board of Directors pursuant to Section 8.2.7.

Excepted Holder Limit. The term "Excepted Holder Limit" shall mean, provided that (and only so long as) the affected Excepted Holder complies with all of the requirements established by the Board of Directors pursuant to Section 8.2.7, and subject to adjustment pursuant to Section 8.2.8, the percentage limit established by the Board of Directors pursuant to Section 8.2.7.

Excluded Holder. The term "Excluded Holder" shall mean any Person who acquires Constructive Ownership or Beneficial Ownership of shares of Common Stock solely by reason of the Transfer of Common Stock in the Merger and who, immediately following the Merger, either Beneficially Owns or Constructively Owns shares of Common Stock in excess of the Ownership Limit solely by reason of such Transfer of Common Stock in the Merger.

Excluded Holder Limit. The term "Excluded Holder Limit" shall mean, with respect to any Excluded Holder, the lesser of (i) shares of Capital Stock representing 9.9% (in value) of the outstanding shares of Capital Stock of the Corporation or (ii) the shares of Capital Stock that such Excluded Holder was considered to Constructively Own immediately following the Merger solely by reason of the Merger (taking into account only such shares of Capital Stock and no other shares as to which such Person may thereafter become, for any reason, the Constructive Owner or the Beneficial Owner), provided, however, that if at any time the Excluded Holder Limit for any Excluded Holder would be less than the Ownership Limit, such Excluded Holder shall cease to be an Excluded Holder and the Ownership Limit shall thereafter apply to such Person.

Market Price. The term "Market Price" on any date shall mean, with respect to any class or series of outstanding shares of Capital Stock, the Closing Price for such shares of Capital Stock on such date.

Merger. The term "Merger" shall mean the merger of Host Marriott Corporation, a Delaware corporation, with and into the Corporation, with the Corporation being the surviving entity and with the outstanding shares of capital stock of Host Marriott Corporation being converted into shares of Capital Stock as set forth in the articles of merger for the Merger.

NYSE. The term "NYSE" shall mean the New York Stock Exchange, Inc.

Ownership Limit. The term "Ownership Limit" shall mean (i) with respect to shares of Common Stock, 9.8% (in value or number of shares, whichever is more restrictive) of the outstanding Common Stock of the Corporation; and (ii) with respect to any class or series of shares of Preferred Stock or other stock, 9.8% (in value or number of shares, whichever is more restrictive) of the outstanding shares of such class or series of Preferred Stock or other stock of the Corporation.

Person. The term "Person" shall mean an individual, corporation, partnership, estate, trust (including a trust qualified under Sections 401(a) or 501(c)(17) of the Code), portion of a trust permanently set aside for or to be used exclusively for the purposes described in Section 642(c) of the Code, association, private foundation within the meaning of Section 509(a) of the Code, joint stock company, limited liability company, or other entity and also includes a group as that term is used for purposes of Section 13(d)(3) of the Securities Exchange Act of 1934, as amended; provided, however, that the term "Person" shall not include any group as that term is used for purposes of Section 13(d)(3) of the Securities Exchange Act of 1934, as amended, if such "group" would be an Excluded Holder (but any Person that is a member of such "group" shall still be considered to be a "Person" for purposes hereof).

Prohibited Owner. The term "Prohibited Owner" shall mean any Person who, but for the provisions of Section 8.2.1, would Beneficially Own or Constructively Own shares of Capital Stock, and if appropriate in the context, shall also mean any Person who would have been the record owner of shares of Capital Stock that the Prohibited Owner would have so owned.

Publicly Offered Securities. The term "Publicly Offered Securities" shall have the meaning provided in 29 C.F.R Section 2510.3-101(b)(2), or any successor regulation thereto.

REIT. The term "REIT" shall mean a real estate investment trust within the meaning of Section 856 of the Code.

Restriction Date. The term "Restriction Date" shall mean the first day after the Effective Date.

Restriction Termination Date. The term "Restriction Termination Date" shall mean the first day after the Effective Date on which the Board of Directors determines that it is no longer in the best interests of the Corporation to attempt to, or continue to, qualify as a REIT or that compliance with the restrictions and limitations on Beneficial Ownership, Constructive Ownership and Transfers of shares of Capital Stock set forth herein is no longer required in order for the Corporation to qualify as a REIT

Transfer. The term "Transfer" shall mean any issuance, sale, transfer, gift, assignment, devise or other disposition, as well as any other event (or any agreement to take any such actions or cause any such events) that causes any Person to acquire Beneficial Ownership or Constructive Ownership of shares of Capital Stock or the right to vote or receive dividends on shares of Capital Stock, including without limitation, (a) the issuance and transfer to holders of shares of Host Marriott Corporation of shares of Capital Stock in the Merger, (b) a change in the capital structure of the Corporation, (c) a change in the relationship between two or more Persons which causes a change in ownership of shares of Capital Stock by application of either Section 544 of the Code, as modified by Section 856(h) or Section 318(a) of the Code, as modified by Section 856(d)(5), (d) the grant or exercise of any option or warrant (or any disposition of any option or warrant, or any event that causes any option or warrant not theretofore exercisable to become exercisable), pledge, security interest or similar right to acquire shares of Capital Stock, (e) any disposition of any securities or rights convertible into or exchangeable for shares of Capital Stock or any interest in shares of Capital Stock or any exercise of any such conversion or exchange right, (f) Transfers of interests in other entities that result in changes in Beneficial Ownership or Constructive Ownership of shares of Capital Stock, and (g) solely for purposes of applying the Excluded Holder Limit, any change in the value of one class or series of shares of Capital Stock relative to the value of any other class or series of shares of Capital Stock, in each case, whether voluntary or involuntary, whether owned of record, Constructively Owned or Beneficially Owned, and whether by operation of law or otherwise. (For purposes of this Article VIII, the right of a limited partner in Host Marriott, L.P., a Delaware limited partnership (or any successor thereto), to require the partnership to redeem such limited partner's units of

limited partnership interest pursuant to Section 8.6 of the Amended and Restated Agreement of Limited Partnership of Host Marriott, L.P shall not be considered to be an option or similar right to acquire shares of Capital Stock of the Corporation so long as such Section 8.6 is not amended in a manner that would grant to a limited partner a legal right to require that either Host Marriott, L.P. (or any successor thereto) or the Corporation issue to such limited partner shares of Capital Stock and so long as the restrictions in Section 8.6.C of such Agreement apply to the exercise of the rights set forth in such Section 8.6.) The terms "Transferring" and "Transferred" shall have the correlative meanings.

Section 8.2 Restrictions on Ownership and Transfer of Shares.

Section 8.2.1 Ownership Limitations. During the period commencing at the Effective Time and ending at the close of business on the Restriction Termination Date:

# (a) Basic Restrictions.

- (i) (1) No Person, other than an Excepted Holder or an Excluded Holder, shall Beneficially Own or Constructively Own shares of Capital Stock in excess of the Ownership Limit, (2) no Excepted Holder shall Beneficially Own or Constructively Own shares of Capital Stock in excess of the Excepted Holder Limit for such Excepted Holder, and (3) commencing at 12:01 a.m. (Eastern Standard Time) on the Restriction Date, no Excluded Holder shall Beneficially Own or Constructively Own shares of Capital Stock in excess of the Excluded Holder Limit for such Excluded Holder.
- (ii) No Person shall Beneficially Own or Constructively Own shares of Capital Stock to the extent that (1) such Beneficial Ownership of shares of Capital Stock would result in the Corporation being "closely held" within the meaning of Section 856(h) of the Code (without regard to whether the ownership interest is held during the last half of a taxable year), (2) such Constructive Ownership would cause either the Corporation to be considered to Constructively Own an interest in a tenant that is described in Section 856(d)(2)(B) of the Code for purposes of applying Section 856(c) of the Code or Host Marriott, L.P. (or any successor thereto) to be considered to Constructively Own an interest in a tenant that is described in Section 856(d)(2)(B) of the Code for purposes of applying Section 7704(d) of the Code, or (3) such Beneficial Ownership or Constructive Ownership of shares of Capital Stock would result in the Corporation otherwise failing to qualify as a REIT or Host Marriott, L.P. (or any successor thereto) to fail to qualify as a partnership for federal income tax purposes.
- (iii) No Person shall Transfer any shares of Capital Stock if, as a result of the Transfer, the outstanding shares of all classes and series of Capital Stock would be Beneficially Owned by less than 100 Persons (determined without reference to the rules of attribution under Section 544 of the Code). Subject to Section 8.5 and notwithstanding any other provisions contained herein, any Transfer of shares of Capital Stock (whether or not such Transfer is the result of a transaction entered into through the facilities of the NYSE or any other national securities exchange or automated interdealer quotation system) that, if effective, would result in outstanding shares of all classes and series of Capital Stock being Beneficially Owned by less than 100 Persons (determined under the principles of Section 856(a)(5) of the Code) shall be void ab initio, and the intended transferee shall acquire no rights in such shares of Capital Stock.
- (b) Transfer in Trust. If, as a result of the Merger and the Transfer of shares of Capital Stock to holders of shares of Host Marriott Corporation, a Delaware corporation, any Person Beneficially Owns or Constructively Owns shares of Capital Stock in violation of Section 8.2.1(a)(i) or 8.2.1(a)(ii) or 8.2.1(a)(iii) as of 12:01 a.m. (Eastern Standard Time) on the Restriction Date, then that number of shares of Capital Stock the Beneficial Ownership or Constructive Ownership of Which otherwise would cause such Person to violate Section 8.2.1(a)(i) or 8.2.1(a)(ii) or 8.2.1(a)(iii) (whichever is applicable) (rounded upward to the nearest whole share) shall be automatically transferred to a Charitable Trust for the benefit of a Charitable Beneficiary, as described in Section 8.3, effective as of 12:01 a.m.

(Eastern Standard Time) on the Restriction Date, and such Person shall have no rights in such shares of Capital Stock (except as otherwise provided in Section 8.3.5(b)). If any Transfer of shares of Capital Stock other than as a result of the Merger and Transfer of shares of Capital Stock to holders of shares of Host Marriott Corporation (whether or not such Transfer is the result of a transaction entered into through the facilities of the NYSE or any other national securities exchange or automated inter-dealer quotation system) occurs which, if effective, would result in any Person Beneficially Owning or Constructively Owning shares of Capital Stock in violation of Section 8.2.1(a)(i) or 8.2.1(a)(ii) or 8.2.1(a)(iii), as applicable,

- (i) then that number of shares of Capital Stock the Beneficial Ownership or Constructive Ownership of which otherwise would cause such Person to violate Section 8.2.1(a)(i) or 8.2.1(a)(ii) or 8.2.1(a)(iii) (rounded upward to the nearest whole share) shall be automatically transferred to a Charitable Trust for the benefit of a Charitable Beneficiary, as described in Section 8.3, effective as of the close of business on the Business Day prior to the date of such Transfer (or as of the close of business on the Effective Date as to any such Transfer that occurs on the Effective Date), and such Person shall acquire no rights in such shares of Capital Stock; or
- (ii) if the transfer to the Charitable Trust described in clause (i) of this sentence would not be effective for any reason to prevent the violation of Section 8.2.1(a)(i) or 8.2.1(a)(ii) or 8.2.1(a)(iii), as applicable, then the Transfer of that number of shares of Capital Stock that otherwise would cause any Person to violate Section 8.2.1(a)(i) or 8.2.1(a)(i) or 8.2.1(a)(ii), as applicable, shall be void ab initio, and the intended transferee shall acquire no rights in such shares of Capital Stock.

Section 8.2.2 Remedies for Breach. If the Board of Directors or any duly authorized committee thereof shall at any time determine in good faith that a Transfer or other event has taken place that results in a violation of Section 8.2.1 or that a Person intends to acquire or has attempted to acquire Beneficial Ownership or Constructive Ownership of any shares of Capital Stock in violation of Section 8.2.1 (whether or not such violation is intended), the Board of Directors or a committee thereof shall take such action as it deems advisable to refuse to give effect to or to prevent such Transfer or other event, including, without limitation, causing the Corporation to redeem shares of Capital Stock, refusing to give effect to such Transfer on the books of the Corporation or instituting proceedings to enjoin such Transfer or other event; provided, however, that any Transfer or attempted Transfer or other event in violation of Section 8.2.1 shall automatically result in the transfer to the Charitable Trust described above, and, where applicable under Section 8.2.1(b)(ii), such Transfer (or other event) shall be void ab initio as provided above irrespective of any action (or non-action) by the Board of Directors or a committee thereof.

Section 8.2.3 Notice of Restricted Transfer. Any Person who acquires or attempts or intends to acquire Beneficial Ownership or Constructive Ownership of shares of Capital Stock that will or may violate Section 8.2.1(a), or any Person who would have owned shares of Capital Stock that resulted in a transfer to the Charitable Trust pursuant to the provisions of Section 8.2.1(b), shall immediately give written notice to the Corporation of such event, or in the case of such a proposed or attempted transaction, give at least 15 days prior written notice, and shall provide to the Corporation such other information as the Corporation may request in order to determine the effect, if any, of such acquisition or ownership on the Corporation's status as a REIT.

Section 8.2.4 Owners Required To Provide Information. During the period commencing at the Effective Time and ending at the close of business on the Restriction Termination Date:

(a) Every stockholder of record of more than five percent (or such lower percentage as required by the Code or the Treasury Regulations promulgated thereunder) of the outstanding shares of Capital Stock, within 30 days after the end of each taxable year, shall give written notice to the Corporation stating the name and address of such owner, the number of shares Beneficially Owned, and a

description of the manner in which such shares of Capital Stock are held; provided that a stockholder of record who holds outstanding shares of Capital Stock as nominee for another Person, which other Person is required to include in gross income the dividends received on such shares (an "Actual Owner"), shall give written notice to the Corporation stating the name and address of such Actual Owner and the number of shares of Capital Stock of such Actual Owner with respect to which the stockholder of record is nominee. Each such stockholder of record and each Actual Owner shall provide to the Corporation such additional information as the Corporation may request in order to determine the effect, if any, of such Beneficial Ownership on the Corporation's status as a REIT and to ensure compliance with the Ownership Limit.

(b) Each Person who is a Beneficial Owner or Constructive Owner of shares of Capital Stock and each Person (including the stockholder of record) who is holding shares of Capital Stock for a Beneficial Owner or Constructive Owner shall provide to the Corporation such information as the Corporation may request, in good faith, in order to determine the Corporation's status as a REIT and to comply with requirements of any taxing authority or governmental authority or to determine such compliance.

Section 8.2.5 Remedies Not Limited. Subject to Section 8.5, nothing contained in this Section 8.2 shall limit the authority of the Board of Directors to take such other action as it deems necessary or advisable to protect the Corporation and the interests of its stockholders in preserving the Corporation's status as a REIT.

Section 8.2.6 Ambiguity. In the case of an ambiguity in the application of any of the provisions of this Section 8.2, Section 8.3 or any definition contained in Section 8.1, the Board of Directors shall have the power to determine the application of the provisions of this Section 8.2 or Section 8.3 with respect to any situation based upon the facts known to it. If Section 8.2 or 8.3 requires an action by the Board of Directors and the charter of the Corporation fails to provide specific guidance with respect to such action, the Board of Directors shall have the power to determine the action to be taken so long as such action is not contrary to the provisions of Sections 8.1, 8.2 or 8.3.

# Section 8.2.7 Exceptions.

(a) The Board of Directors, in its sole and absolute discretion, may grant to any Person who makes a request therefor an exception to the Ownership Limit (or one or more elements thereof) with respect to the ownership of any series or class of Capital Stock of the Corporation, subject to the following conditions and limitations: (A) the Board of Directors shall have determined that (x) assuming such Person would Beneficially Own or Constructively Own the maximum amount of shares of Common Stock and stock of the Corporation (other than Common Stock) permitted as a result of the exception to be granted and (y) assuming that all other Persons who would be treated as "individuals" for purposes of Section 542(a)(2) of the Code (determined taking into account Section 856(h)(3)(A) of the Code) would Beneficially Own or Constructively Own the maximum amount of shares of Common Stock and stock of the Corporation (other than Common Stock) permitted under this Article VIII (taking into account any exception, waiver or exemption granted under this Section 8.2.7 to (or with respect to) such Persons), the Corporation would not be "closely held" within the meaning of Section 856(h) of the Code (assuming that the ownership of shares of Capital Stock is determined during the second half of a taxable year) and would not otherwise fail to qualify as a REIT; and (B) such Person provides to the Board of Directors such representations and undertakings, if any, as the Board of Directors may, in its sole and absolute discretion, determine to be necessary in order for it to make the determination that the conditions set forth in clause (A) above of this Section 8.2.7(a) have been and/or will continue to be satisfied (including, without limitation, an agreement as to a reduced Ownership Limit or Excepted Holder Limit for such Person with respect to the Beneficial Ownership or Constructive Ownership of one or more other classes or series of shares of Capital Stock not subject to the exception), and such Person agrees that any violation of such representations and undertakings or any attempted violation thereof will result in the application of the remedies set forth in Section 8.2 with respect to shares of Capital Stock held in excess of the Ownership Limit or the Excepted Holder Limit (as

may be applicable) with respect to such Person (determined without regard to the exception granted such Person under this subparagraph (a)). If a member of the Board of Directors requests that the Board of Directors grant an exception pursuant to this subparagraph (a) with respect to such member, or with respect to any other Person if such Board member would be considered to be the Beneficial Owner or Constructive Owner of shares of Capital Stock owned by such other Person, such member of the Board of Directors shall not participate in the decision of the Board of Directors as to whether to grant any such exception.

- (b) In addition to exceptions permitted under subparagraph (a) above, the Board of Directors, in its sole and absolute discretion, may grant to any Person who makes a request therefor (a "Requesting Person") an exception from the Ownership Limit (or one or more elements thereof) if: (i) such Person submits to the Board of Directors information satisfactory to the Board of Directors, in its reasonable discretion, demonstrating that such Requesting Person is not an individual for purposes of Section 542(a)(2) of the Code (determined taking into account Section 856(h)(3)(A) of the Code); (ii) such Requesting Person submits to the Board of Directors information satisfactory to the Board of Directors, in its reasonable discretion, demonstrating that no Person who is an individual for purposes of Section 542(a)(2) of the Code (determined taking into account Section 856(h)(3)(A) of the Code) would be considered to Beneficially Own shares of Capital Stock in excess of the Ownership Limit by reason of the Requesting Person's ownership of shares of Capital Stock in excess of the Ownership Limit pursuant to the exception granted under this subparagraph (b); (iii) such Requesting Person submits to the Board of Directors information satisfactory to the Board of Directors, in its reasonable discretion, demonstrating that neither clause (2) nor clause (3) of subparagraph (a)(ii) of Section 8.2.1 will be violated by reason of the Requesting Person's ownership of shares of Capital Stock in excess of the Ownership Limit pursuant to the exception granted under this subparagraph (b); and (iv) such Requesting Person provides to the Board of Directors such representations and undertakings, if any, as the Board of Directors may, in its sole and absolute discretion, require to ensure that the conditions in clauses (i), (ii) and (iii) hereof are satisfied and will continue to be satisfied throughout the period during which such Requesting Person owns shares of Capital Stock in excess of the Ownership Limit pursuant to any exception thereto granted under this subparagraph (b), and such Requesting Person agrees that any violation of such representations and undertakings or any attempted violation thereof will result in the application of the remedies set forth in Section 8.2 with respect to shares of Capital Stock held in excess of the Ownership Limit with respect to such Requesting Person (determined without regard to the exception granted such Requesting Person under this subparagraph (b)).
- (c) Prior to granting any exception or exemption pursuant to subparagraph (a) or (b), the Board of Directors may require a ruling from the IRS or an opinion of counsel, in either case in form and substance satisfactory to the Board of Directors, in its sole and absolute discretion as it may deem necessary or advisable in order to determine or ensure the Corporation's status as a REIT; provided, however, that the Board of Directors shall not be obligated to require obtaining a favorable ruling or opinion in order to grant an exception hereunder.
- (d) Subject to Section 8.2.1(a)(ii), an underwriter that participates in a public offering or a private placement of shares of Capital Stock (or securities convertible into or exchangeable for shares of Capital Stock) may Beneficially Own or Constructively Own shares of Capital Stock (or securities convertible into or exchangeable for shares of Capital Stock) in excess of the Ownership Limit, but only to the extent necessary to facilitate such public offering or private placement; and provided, that the ownership of shares of Capital Stock by such underwriter would not result in the Corporation being "closely held" within the meaning of Section 856(h) of the Code, or otherwise result in the Corporation's failing to qualify as a REIT. In this regard, at no time may either (x) an underwriter, or (y) any Person who would Constructively Own shares of Capital Stock owned by an underwriter Constructively Own, concurrently, 10% or more of the outstanding securities of any class or series of (i) the Corporation and any tenant or lessee of the Corporation (which, as of the Effective Date, includes Crestline Capital Corporation and its subsidiaries), or (ii) the Corporation and any Person that would be considered to Constructively Own or Beneficially Own 10% or more of any tenant or lessee of the Corporation (which, as of the Effective Date, includes Crestline Capital Corporation and its subsidiaries).

(e) The Board of Directors may only reduce the Excepted Holder Limit for an Excepted Holder: (1) with the written consent of such Excepted Holder at any time or (2) pursuant to the terms and conditions of the agreements and undertakings entered into with such Excepted Holder in connection with the establishment of the Excepted Holder Limit for that Excepted Holder. No Excepted Holder Limit shall be reduced to a percentage that is less than the Ownership Limit.

Section 8.2.8 Increase or Decrease in Ownership Limit. The Board of Directors may from time to time increase or decrease the Ownership Limit, subject to the limitations provided in this Section 8.2.8.

- (a) Any decrease may be made only prospectively as to subsequent holders (other than a decrease as a result of a retroactive change in existing law, in which case such change shall be effective immediately).
- (b) The Ownership Limit may not be increased if, after giving effect to such increase, five Persons who are considered individuals pursuant to Section 542 of the Code, as modified by Section 856(h)(3) of the Code (taking into account all of the Excepted Holders), could Beneficially Own, in the aggregate, more than 49.5% of the value of the outstanding shares of Capital Stock.
- (c) Prior to the modification of the Ownership Limit pursuant to this Section 8.2.8, the Board of Directors may require such opinions of counsel, affidavits, undertakings or agreements as it may deem necessary or advisable in order to determine or ensure the Corporation's status as a REIT if the modification in the Ownership Limit were to be made.

Section 8.2.9 Legend. Each certificate for shares of Capital Stock (or securities exercisable for or convertible into shares of Capital Stock) shall bear substantially the following legend:

The shares of Capital Stock represented by this certificate are subject  $% \left( 1\right) =\left( 1\right) \left( 1\right) \left($ to restrictions on Beneficial Ownership and Constructive Ownership and Transfer primarily for the purpose of the Corporation's maintenance of its status as a real estate investment trust (a "REIT") under the Internal Revenue Code of 1986, as amended (the "Code"). Except as expressly provided in the Corporation's charter, (i) no Person may Beneficially Own or Constructively Own shares of Common Stock of the Corporation in excess of 9.8 percent (in value or number of shares, whichever is more restrictive) of the outstanding Common Stock of the Corporation unless such Person is an Excepted Holder (in which case the Excepted Holder Limit shall be applicable) or an Excluded Holder (in which case the Excluded Holder Limit shall be applicable); (ii) with respect to any class or series of shares of Capital Stock other than Common Stock, no Person may Beneficially Own or Constructively Own more than 9.8 percent (in value or number of shares, whichever is more restrictive) of the outstanding shares of such class or series of such stock of the Corporation (collectively, (i) and (ii) are referred to herein as the "Ownership Limit"), unless such Person is an Excepted Holder (in which case the Excepted Holder Limit shall be applicable) or an Excluded Holder (in which case the Excluded Holder Limit shall be applicable); (iii) no Person may Beneficially Own or Constructively Own shares of Capital Stock that would result in the Corporation being "closely held" under Section 856(h) of the Code, would cause either the Corporation to be considered to Constructively Own an interest in a tenant that is described in Section 856(d)(2)(B) of the Code for purposes of applying Section 856(c) of the Code or Host Marriott, L.P. (or any successor thereto) to be considered to Constructively Own an interest in a tenant that is described in Section 856(d)(2)(B) of the Code for purposes of applying Section 7704(d) of the Code, or otherwise would cause the Corporation to fail to qualify as a REIT under the Code; and (iv) no Person may Transfer shares of Capital Stock if such Transfer would result in shares of Capital Stock of the Corporation being owned by fewer than 100 Persons. Notwithstanding the foregoing, commencing at 12:01 a.m. (Eastern Standard Time) on the day immediately following the effective date of the merger (the "Merger") of Host Marriott Corporation, a Delaware corporation, with and into the Corporation, no Excluded Holder shall Beneficially Own or Constructively Own shares of Capital Stock in excess of the Excluded Holder Limit for such Excluded Holder. An "Excepted Holder means a stockholder of the Corporation for whom an Excepted Holder Limit is created by the Board of Directors. An "Excluded Holder" means any Person who acquires Constructive Ownership or Beneficial Ownership of shares of Common Stock solely be reason of the Transfer of Common Stock in the Merger and who, immediately following the Merger, either Beneficially Owns or Constructively Owns shares of Common

Stock in excess of the Ownership Limit solely by reason of the Transfer of Common Stock in the Merger. The "Excluded Holder Limit" means, with respect to any Excluded Holder, the lesser of (i) shares of Capital Stock representing 9.9% (in value) of the outstanding shares of Capital Stock of the Corporation or (ii) the shares of Capital Stock that such Excluded Holder was considered to Constructively Own immediately following the Merger solely by reason of the Merger (taking into account only such shares of Capital Stock and no other shares as to which such Person may thereafter become, for any reason, the Constructive Owner or the Beneficial Owner), provided, however, that if at any time the Excluded Holder Limit for any Excluded Holder would be less than the Ownership Limit, such Excluded Holder shall cease to be an Excluded Holder and the Ownership Limit shall thereafter apply to such Person. Any Person who Beneficially Owns or Constructively Owns or attempts to Beneficially Own or Constructively Own shares of Capital Stock which cause or will cause a Person to Beneficially Own or Constructively Own shares of Capital Stock in excess or in violation of the above limitations must immediately notify the Corporation. If any of the restrictions on Transfer are violated, the shares of Capital Stock represented hereby will be automatically transferred to a Charitable Trustee of a Charitable Trust for the benefit (except as otherwise provided in the charter of the Corporation) of one or more Charitable Beneficiaries. In addition, upon the occurrence of certain events, attempted Transfers in violation of the restrictions described above may be void ab initio. A Person who attempts to Beneficially Own or Constructively Own shares of Capital Stock in violation of the Transfer restrictions described above shall have no claim, cause of action or any recourse whatsoever against a transferor of such shares of Capital Stock. All capitalized terms in this legend have the meanings defined in the Corporation's charter, as the same may be amended from time to time, a copy of which, including the restrictions on Transfer, will be furnished to each holder of shares of Capital Stock of the Corporation on request and without charge.

Instead of the foregoing legend, the certificate may state that the Corporation will furnish a full statement about certain restrictions on transferability to a stockholder on request and without charge.

Section 8.3 Transfer of Shares of Capital Stock in the Corporation.

Section 8.3.1 Ownership in Trust. Upon any purported Transfer or other event described in Section 8.2.1(b) that would result in a transfer of shares of Capital Stock to a Charitable Trust, such shares of Capital Stock shall be deemed to have been transferred to the Charitable Trustee as trustee of a Charitable Trust for the exclusive benefit of one or more Charitable Beneficiaries (except to the extent otherwise provided in Section 8.3.5). Such transfer to the Charitable Trustee shall be deemed to be effective (i) as of 12:01 a.m. (Eastern Standard Time) on the Restriction Date, as to any Person who Beneficially Owns or Constructively Owns shares of Capital Stock in violation of Section 8.2.1(a)(i) or 8.2.1(a)(ii) or 8.2.1(a)(iii) as of such time and date as a result of the Merger and the Transfer of shares of Capital Stock to holders of shares of Host Marriott Corporation or (ii) as of the close of business on the Business Day prior to any other purported Transfer or other event that otherwise results in the transfer to the Charitable Trust pursuant to Section 8.2.1(b) (or as of the close of business on the Effective Date if such other purported Transfer or other event occurs on that date). The Charitable Trustee shall be appointed by the Corporation and shall be a Person unaffiliated with the Corporation and any Prohibited Owner. Each Charitable Beneficiary shall be designated by the Corporation as provided in Section 8.3.7.

Section 8.3.2 Status of Shares of Capital Stock Held by the Charitable Trustee. Shares of Capital Stock held by the Charitable Trustee shall be issued and outstanding shares of Capital Stock of the Corporation. The Prohibited Owner shall have no rights in the shares of Capital Stock held by the Charitable Trustee. The Prohibited Owner shall not benefit economically from ownership of any shares of Capital Stock held in trust by the Charitable Trustee (except to the extent otherwise provided in Section 8.3.5), shall have no rights to dividends or other distributions, and shall not possess any rights to vote or other rights attributable to the shares of Capital Stock held in the Charitable Trust. The Prohibited Owner shall have no claim, cause of action or other recourse whatsoever against the purported transferor of such shares of Capital Stock.

Section 8.3.3 Dividend and Voting Rights. The Charitable Trustee shall have all voting rights and rights to dividends or other distributions with respect to shares of Capital Stock held in the Charitable Trust, which rights shall be exercised for the exclusive benefit of the Charitable Beneficiary (except to the extent otherwise provided in Section 8.3.5). Any dividend or other distribution paid prior to the discovery by the Corporation that shares of Capital Stock have been transferred to the Charitable Trustee shall be paid with respect to such shares of Capital Stock to the Charitable Trustee upon demand and any dividend or other distribution authorized but unpaid shall be paid when due to the Charitable Trustee. Any dividends or distributions so paid over to the Charitable Trustee shall be held in trust for the Charitable Beneficiary. The Prohibited Owner shall have no voting rights with respect to shares of Capital Stock held in the Charitable Trust and, subject to Maryland law, effective as of the date that shares of Capital Stock have been transferred to the Charitable Trustee, the Charitable Trustee shall have the authority (at the Charitable Trustee's sole discretion) (i) to rescind as void any vote cast by a Prohibited Owner prior to the discovery by the Corporation that shares of Capital Stock have been transferred to the Charitable Trustee and (ii) to recast such vote in accordance with the desires of the Charitable Trustee acting for the benefit of the Charitable Beneficiary; provided, however, that if the Corporation has already taken irreversible action, then the Charitable Trustee shall not have the power to rescind and recast such vote. Notwithstanding the provisions of this Article VIII, until the Corporation has received notification that shares of Capital Stock have been transferred into a Charitable Trust, the Corporation shall be entitled to rely on its share transfer and other shareholder records for purposes of preparing lists of stockholders entitled to vote at meetings, determining the validity and authority of proxies, and otherwise conducting votes of stockholders.

Section 8.3.4 Rights Upon Liquidation. Upon any voluntary or involuntary liquidation, dissolution or winding up of or any distribution of the assets of the Corporation, the Charitable Trustee shall be entitled to receive, ratably with each other holder of shares of Capital Stock of the class or series of shares of Capital Stock that is held in the Charitable Trust, that portion of the assets of the Corporation available for distribution to the holders of such class or series (determined based upon the ratio that the number of shares of such class or series of shares of Capital Stock held by the Charitable Trustee bears to the total number of shares of Capital Stock then outstanding). The Charitable Trustee shall distribute any such assets received in respect of the shares of Capital Stock held in the Charitable Trust in any liquidation, dissolution or winding up or distribution of the assets of the Corporation, in accordance with Section 8.3.5.

# Section 8.3.5 Sale of Shares by Charitable Trustee.

- (a) Within 20 days of receiving notice from the Corporation that shares of Capital Stock have been transferred to the Charitable Trust, the Charitable Trustee of the Charitable Trust shall sell the shares of Capital Stock held in the Charitable Trust (together with the right to receive dividends or other distributions with respect to such shares of Capital Stock as to any shares of Capital Stock transferred to the Charitable Trustee as a result of the operation of Section 8.2.1(b)) to a person, designated by the Charitable Trustee, whose ownership of the shares of Capital Stock will not violate the ownership limitations set forth in Section 8.2.1(a). Upon such sale, the interest of the Charitable Beneficiary in the shares of Capital Stock sold shall terminate and the Charitable Trustee shall distribute the net proceeds of the sale to the Prohibited Owner and to the Charitable Beneficiary as provided in this Section 8.3.5.
- (b) A person who becomes a Prohibited Owner by virtue of the first sentence of Section 8.2.1(b) shall receive the lesser of (i) all of the net sales proceeds per share received by the Charitable Trustee from the sale or other disposition of such excess shares of Capital Stock held in the Charitable Trust, or (ii) an amount equal to the Closing Price of such excess shares of Capital Stock on the Effective Date, with the balance, if any, to be paid to the Charitable Beneficiary.
- (c) Any other Prohibited Owner shall receive the lesser of (1) the net price paid by the Prohibited Owner for the shares of Capital Stock or, if the Prohibited Owner did not give value for the shares of Capital Stock in connection with the event causing the shares of Capital Stock to be held in the

Charitable Trust (e.g., in the case of a gift, devise or other such transaction), the Market Price of the shares of Capital Stock on the day of the event causing the shares of Capital Stock to be held in the Charitable Trust, and (2) the net sales proceeds per share received by the Charitable Trustee from the sale or other disposition of the shares of Capital Stock held in the Charitable Trust. Any net sales proceeds in excess of the amount payable to the Prohibited Owner shall be immediately paid to the Charitable Beneficiary. If, prior to the discovery by the Corporation that shares of Capital Stock have been transferred to the Charitable Trustee, such shares of Capital Stock are sold by a Prohibited Owner, then (i) such shares of Capital Stock shall be deemed to have been sold on behalf of the Charitable Trust and (ii) to the extent that the Prohibited Owner received an amount for such shares of Capital Stock that exceeds the amount that such Prohibited Owner was entitled to receive pursuant to this Section 8.3.5, such excess shall be paid to the Charitable Trustee upon demand.

Section 8.3.6 Purchase Right in Shares of Capital Stock Transferred to the Charitable Trustee. Shares of Capital Stock transferred to the Charitable Trustee shall be deemed to have been offered for sale to the Corporation, or its designee, at a price per share equal to the lesser of (i) the price per share in the transaction that resulted in such transfer to the Charitable Trust (or, in the case of a devise, gift or other such transaction, the Market Price of the shares of Capital Stock on the day of the event causing the shares of Capital Stock to be held in the Charitable Trust) and (ii) the Market Price on the date the Corporation, or its designee, accepts such offer. The Corporation shall have the right to accept such offer until the Charitable Trustee has sold the shares of Capital Stock held in the Charitable Trust pursuant to Section 8.3.5. Upon such a sale to the Corporation, the interest of the Charitable Beneficiary in the shares of Capital Stock sold shall terminate and the Charitable Trustee shall distribute the net proceeds of the sale to the Prohibited Owner.

Section 8.3.7 Designation of Charitable Beneficiaries. By written notice to the Charitable Trustee, the Corporation shall designate from time to time one or more nonprofit organizations to be the Charitable Beneficiary of the interest in the Charitable Trust such that (i) shares of Capital Stock held in the Charitable Trust would not violate the restrictions set forth in Section 8.2.1(a) in the hands of such Charitable Beneficiary and (ii) each such organization must be described in Sections 501(c)(3), 170(b)(1)(A) or 170(c)(2) of the Code and contributions to each such organization must be eligible for deduction under each of Sections 170(b)(1)(A), 2055 and 2522 of the Code. In the absence of any such determination by the Corporation, the Charitable Beneficiary shall be

Section 8.4. Restrictions on Ownership and Transfer of Shares of Capital Stock by Benefit Plans.

Section 8.4.1 Ownership Limitations. Notwithstanding any other provisions herein, if and to the extent that any class or series of shares of Capital Stock do not constitute Publicly Offered Securities, then Benefit Plan Investors may not, on any date, hold, individually or in the aggregate, 25 percent or more of the value of such class or series of shares of Capital Stock. For purposes of determining whether Benefit Plan Investors hold, individually or in the aggregate, 25 percent or more of the value of such class or series of shares of Capital Stock, the value of shares of Capital Stock of such class held by any director or officer of the Corporation, or any other Person who has discretionary authority or control with respect to the assets of the Corporation, or any Person who provides investment advice for a fee to the Corporation in connection with its assets, or an "affiliate" of such person, as defined in 29 C.F.R. Section 2510.3-101(f)(3), or any successor regulation thereto, shall be disregarded.

Section 8.4.2 Remedies for Violations by Benefit Plan Investors. If the Board of Directors or any duly authorized committee thereof shall at any time determine in good faith that (i) a Transfer or other event has taken place that results in a violation of Section 8.4.1 or will otherwise result in the underlying assets and property of the Corporation becoming assets of any ERISA Investor or (ii) that a Person intends to acquire or has attempted to acquire or hold shares of Capital Stock in a manner that will result in a violation of Section 8.4.1 or will otherwise result in the underlying assets and property of the Corporation becoming assets of any ERISA Investor, the Board of Directors or a committee thereof shall take such action as it

deems advisable to mitigate, prevent or cure the consequences that might result to the Corporation from such Transfer or other event, including without limitation, refusing to give effect to or preventing such Transfer or event through redemption of such shares of Capital Stock or refusal to give effect to the Transfer or event on the books of the Corporation or instituting proceedings to enjoin such Transfer or other event.

Section 8.4.3 Information on Benefit Plan Status. Any Person who acquires or attempts or intends to acquire or hold shares of Capital Stock shall provide to the Corporation such information as the Corporation may request in order to determine whether such acquisition or holding has resulted or will result in a violation of Section 8.4.1 or otherwise has resulted or will result in the underlying assets and property of the Corporation becoming assets of any ERISA Investor, including the name and address of any Person for whom a nominee holds shares of Capital Stock and whether the underlying assets of such Person include assets of any Benefit Plan Investor.

Section 8.5 NYSE Transactions. Nothing in this Article VIII shall preclude the settlement of any transaction entered into through the facilities of the NYSE or any other national securities exchange or automated inter-dealer quotation system. The fact that the settlement of any transaction takes place shall not negate the effect of any other provision of this Article VIII and any transferee in such a transaction shall be subject to all of the provisions and limitations set forth in this Article VIII.

Section 8.6 Enforcement. The Corporation is authorized specifically to seek equitable relief, including injunctive relief, to enforce the provisions of this Article VIII.

Section 8.7 Non-Waiver. No delay or failure on the part of the Corporation or the Board of Directors in exercising any right hereunder shall operate as a waiver of any right of the Corporation or the Board of Directors, as the case may be, except to the extent specifically waived in writing.

Section 8.8 Enforceability. If any of the restrictions on transfer of shares of Capital Stock contained in this Article VIII are determined to be void, invalid or unenforceable by any court of competent jurisdiction, then the Prohibited Owner may be deemed, at the option of the Corporation, to have acted as an agent of the Corporation in acquiring such shares and to hold such shares on behalf of the Corporation.

Section 8.9 Amendments. Notwithstanding any other provisions of the charter or Bylaws of the Corporation, the affirmative vote of stockholders holding at least two-thirds of all of the votes entitled to be cast thereon shall be required to amend, alter, change, repeal, or adopt any provisions inconsistent with, the provisions of this ARTICLE VIII.

# ARTICLE IX MERGER, CONSOLIDATION, SHARE EXCHANGE OR TRANSFER OF ASSETS

Subject to the terms of any class or series of Capital Stock at the time outstanding, the Corporation may merge with or into another entity, may consolidate with one or more other entities, may participate in a share exchange or may transfer its assets within the meaning of the MGCL, but any such merger, consolidation, share exchange or transfer of its assets must be approved (i) by the Board of Directors in the manner provided in the MGCL and (ii) by the stockholders by the affirmative vote of two-thirds of all votes entitled to be cast thereon to the extent a stockholder vote is required under the MGCL to effect any such transaction. Notwithstanding the foregoing, any merger of the Corporation with or into a trust organized for the purpose of changing the Corporation's form of organization from a corporation to a trust shall require the approval of stockholders of the Corporation by the affirmative vote only of a majority of all the votes entitled to be cast on the matter, provided that (i) the shareholders of the trust immediately following the merger are the same as the stockholders of the Corporation immediately prior to the merger and (ii) the trust's declaration of trust contains amendment provisions substantially equivalent to those contained in Section 7(d), Section 8.9, the last sentence of this Article IX and Section 10(a)(7) hereof. Notwithstanding any other provisions of the charter or Bylaws of the Corporation, the affirmative vote of stockholders holding at least twothirds of all of the votes entitled to be cast thereon shall be required to amend, alter, change, repeal, or adopt any provisions inconsistent with, the provisions of this ARTICLE IX.

# ARTICLE X MISCELLANEOUS PROVISIONS

Section 10(a) Additional Provisions. The following provisions are hereby adopted for the purpose of defining, limiting, and regulating the powers of the Corporation and of the directors and stockholders of the Corporation:

- (1) Authority to Issue Stock. The Board of Directors is hereby empowered to authorize the issuance from time to time of shares of stock of the Corporation of any class or series, whether now or hereafter authorized, or securities convertible into shares of its stock of any class or classes, whether now or hereafter authorized, for such consideration as may be deemed advisable by the Board of Directors and without any action by the stockholders.
- (2) No Preemptive Rights. No stockholder of the Corporation shall have preemptive rights to purchase, subscribe for, or otherwise acquire any stock or other securities of the Corporation, and any and all preemptive rights are hereby denied; other than such, if any, as the Board of Directors, in its sole discretion, may determine and at such price or prices and upon such other terms as the Board of Directors, in its sole discretion, may fix; and any stock or other securities which the Board of Directors may determine to offer for subscription may, as the Board of Directors in its sole discretion shall determine, be offered to the holders of any class, series or type of stock or other securities at the time outstanding to the exclusion of the holders of any or all other classes, series or types of stock or other securities at the time outstanding.
- (3) Indemnification. The Corporation shall indemnify (A) its directors and officers, whether serving the Corporation or at its request any other entity, to the full extent required or permitted by the general laws of the State of Maryland now or hereafter in force, including the advance of expenses under the procedures and to the full extent permitted by law and (B) other employees and agents to such extent as shall be authorized by the Board of Directors or the Bylaws and be permitted by law. The foregoing rights of indemnification shall not be exclusive of any other rights to which those seeking indemnification may be entitled. The Board of Directors may take such action as is necessary to carry out these indemnification provisions and is expressly empowered to adopt, approve and amend from time to time such by-laws, resolutions or contracts implementing such provisions or such further indemnification arrangements as may be permitted by law. No amendment of the charter of the Corporation or repeal of any of its provisions shall limit or eliminate the right to indemnification provided hereunder with respect to any act or omission occurring prior to such amendment or repeal.
- (4) Liability of Directors and Officers. To the fullest extent permitted by Maryland statutory or decisional law, as amended or interpreted, no director or officer of this Corporation shall be personally liable to the Corporation or its stockholders for money damages. No amendment of the charter of the Corporation or repeal of any of its provisions shall limit or eliminate the benefits provided to directors and officers under this provision with respect to any act or omission which occurred prior to such amendment or repeal.
- (5) Call of Special Meetings of Stockholders. A special meeting of the stockholders of the Corporation may be called by the President, the Board of Directors or any other person specified in the Bylaws. The Secretary of the Corporation shall also call a special meeting of the stockholders on the written request of stockholders entitled to cast a majority of all the votes entitled to be cast at the meeting.
- (6) Bylaws. The power to adopt, alter and repeal the Bylaws of the Corporation is vested exclusively in the Board of Directors.
- (7) Amendments. The Corporation reserves the right from time to time to make any amendments of its charter which may now or hereafter be authorized by law, including without limitation any amendments changing the terms or contract rights, as expressly set forth in the charter, of any of its outstanding stock by classification, reclassification or otherwise. Except as otherwise provided in the charter of the Corporation, any amendment to the charter shall be valid only if approved by the affirmative vote of stockholders of the Corporation holding not less than a majority of all the votes entitled to be cast on the matter. Notwithstanding any other provisions of the charter or Bylaws of the Corporation, the affirmative vote of

stockholders holding at least two-thirds of all of the votes entitled to be cast thereon shall be required to amend, alter, change, repeal, or adopt any provisions inconsistent with, the provisions of Section 10(a) of this ARTICLE X.

Section 10(b) No Limitation of Powers. The enumeration and definition of particular powers of the Board of Directors included herein shall in no way be limited or restricted by reference to or inference from the terms of any other clause of this or any other Article or the charter of the Corporation, or construed as or deemed by inference or otherwise in any manner to exclude or limit any powers conferred upon the Board of Directors under the general laws of the State of Maryland now or hereinafter in force.

SECOND: Immediately before this amendment and restatement, the total number of shares of stock of all classes which the Corporation had authority to issue was 1,000 shares, all of which were classified as shares of Common Stock, par value \$.01 per share, amounting to an aggregate par value of \$10. The foregoing amendment and restatement increases the authorized capital stock of the Corporation to 800,000,000 shares, of which 750,000,000 shares initially are classified as Common Stock, par value \$.01 per share, amounting to an aggregate par value of \$7,500,000, and 50,000,000 shares initially are classified as Preferred Stock, par value \$.01 per share, amounting to an aggregate par value of \$500,000. A description of each class of stock, including preferences, conversion and other rights, voting powers, restrictions, limitations as to dividends and qualifications is set forth in Article VI hereof.

THIRD: The board of directors of the Corporation, by a unanimous consent in writing in lieu of a meeting under Section 2-408 of the MGCL, dated , 1998, adopted a resolution which sets forth the foregoing amendment and restatement of the Articles of Incorporation, declaring that the said amendment and restatement of the Articles of Incorporation was advisable and directing that it be submitted for action thereon by the stockholders by a unanimous consent in writing in lieu of a meeting under Section 2-505 of the MGCL.

FOURTH: Notice of a meeting of stockholders to take action on the amendment and restatement of the Articles of Incorporation was waived by all of the stockholders of the Corporation.

FIFTH: The amendment and restatement of the Articles of Incorporation of the Corporation as hereinabove set forth was approved by the unanimous consent in writing of the stockholders on , 1998.

IN WITNESS WHEREOF, HMC Merger Corporation has caused these Articles of Amendment and Restatement to be signed in its name and on its behalf by its President and attested by its Secretary on , 1998.

HMC MERGER	CORPORATION
------------	-------------

	Бу.	
	President	
ttest:		
Secretary		

I, , President of HMC Merger Corporation hereby acknowledge the foregoing Articles of Amendment and Restatement of Articles of Incorporation of HMC Merger Corporation to be the corporate act of HMC Merger Corporation, and to the best of my knowledge, information and belief, these matters and facts are true in all material respects, and my statement is made under penalties of perjury.

President of HMC Merger Corporation

# HMC MERGER CORPORATION

BYLAWS

#### ARTICLE I

#### **OFFICES**

Section 1. Principal Office. The principal office of HMC Merger Corporation (the "Corporation") shall be located at such place or places as the directors may designate.

Section 2. Additional Offices. The Corporation may have additional offices at such places as the directors may from time to time determine or the business of the Corporation may require.

# ARTICLE II

# MEETINGS OF STOCKHOLDERS

Section 1. Place. All meetings of stockholders shall be held at the principal office of the Corporation or at such other place within the United States as shall be stated in the notice of the meeting.

Section 2. Annual Meeting. An annual meeting of the stockholders for the election of directors and the transaction of any business within the powers of the Corporation shall be held during the month of May of each year, after the delivery of the annual report referred to in Section 12 of this Article II, at a convenient location and on proper notice, on a date and at the time set by the directors, beginning with the year 1999. Failure to hold an annual meeting does not invalidate the Corporation's existence or affect any otherwise valid acts of the Corporation.

Section 3. Special Meetings. A special meeting of the stockholders of the Corporation may be called by the President or the Board of Directors or by holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation in the manner specified in articles supplementary filed as part of the charter of the Corporation (the "Charter"). The Secretary of the Corporation shall also call a special meeting of the stockholders on the written request of stockholders entitled to cast a majority of all the votes entitled to be cast at the meeting. A special meeting need not be called to consider any matter which is substantially the same as a matter voted on at any meeting of the stockholders held during the preceding twelve months.

Section 4. Notice. Not less than ten nor more than 90 days before each meeting of stockholders, the Secretary shall give to each stockholder entitled to vote at such meeting and to each stockholder not entitled to vote who is entitled to notice of the meeting written or printed notice stating the time and place of the meeting and, in the case of a special meeting or as otherwise may be required by any statute, the purpose for which the meeting is called, either by mail or by presenting it to such stockholder personally or by leaving it at his residence or usual place of business. If mailed, such notice shall be deemed to be given when deposited in the United States mail addressed to the stockholder at his post office address as it appears on the records of the Corporation, with postage thereon prepaid.

Section 5. Scope of Notice. Any business of the Corporation may be transacted at an annual meeting of stockholders without being specifically designated in the notice, except such business as is required by any statute to be stated in such notice. No business shall be transacted at a special meeting of stockholders except as specifically designated in the notice.

Section 6. Organization. At every meeting of the stockholders, the Chairman of the Board, if there is one, shall conduct the meeting or, in the case of vacancy in office or absence of the Chairman of the Board, one of the following officers present shall conduct the meeting in the order stated: the Vice Chairman of the Board, if there is one, the Chief Executive Officer, if there is one, the President, the Vice Presidents in their order of rank and seniority, or a Chairman chosen by the stockholders entitled to cast a majority of the votes which all

stockholders present in person or by proxy are entitled to cast, shall act as Chairman, and the Secretary, or, in his absence, an Assistant Secretary, or in the absence of both the Secretary and Assistant Secretaries, a person appointed by the Chairman shall act as Secretary.

Section 7. Quorum. At any meeting of stockholders, the presence in person or by proxy of stockholders entitled to cast a majority of all the votes entitled to be cast at such meeting shall constitute a quorum; but this section shall not affect any requirement under any statute or the Charter for the vote necessary for the adoption of any measure. If, however, such quorum shall not be present at any meeting of the stockholders, the stockholders entitled to vote at such meeting, present in person or by proxy, shall have the power to adjourn the meeting from time to time to a date not more than 120 days after the original record date without notice other than announcement at the meeting. At such adjourned meeting at which a quorum shall be present, any business may be transacted which might have been transacted at the meeting as originally notified.

Section 8. Voting. Subject to the rights of the holders of any class or series of stock (other than Common Stock) to elect additional directors under specified circumstances, a plurality of all the votes cast at a meeting of stockholders duly called and at which a quorum is present shall be sufficient to elect a director. Each share may be voted for as many individuals as there are directors to be elected and for whose election the share is entitled to be voted. Unless otherwise provided in the Charter, a majority of the votes cast at a meeting of stockholders duly called and at which a quorum is present shall be sufficient to approve any other matter which may properly come before the meeting, unless more than a majority of the votes cast is required herein or by statute or by the Charter. Unless otherwise provided in the Charter, each outstanding share, regardless of class, shall be entitled to one vote on each matter submitted to a vote at a meeting of stockholders.

Section 9. Proxies. A stockholder may cast the votes entitled to be cast by the shares owned of record by him either in person or by proxy executed in writing by the stockholder or by his duly authorized attorney in fact. Such proxy shall be filed with the Secretary of the Corporation before or at the time of the meeting. No proxy shall be valid after 11 months from the date of its execution, unless otherwise provided in the proxy.

Section 10. Voting of Shares by Certain Holders. Shares of the Corporation registered in the name of a corporation, partnership, trust or other entity, if entitled to be voted, may be voted by the president or a vice president, a general partner, limited liability company, trustee or member manager thereof, as the case may be, or a proxy appointed by any of the foregoing individuals, unless some other person who has been appointed to vote such shares pursuant to a bylaw or a resolution of the governing board of such corporation or other entity or agreement of the partners of the partnership presents a certified copy of such bylaw, resolution or agreement, in which case such person may vote such shares. Any trustee or other fiduciary may vote shares registered in his name as such fiduciary, either in person or by proxy.

Shares of the Corporation directly or indirectly owned by it shall not be voted at any meeting and shall not be counted in determining the total number of outstanding shares entitled to be voted at any given time, unless they are held by it in a fiduciary capacity, in which case they may be voted and shall be counted in determining the total number of outstanding shares at any given time.

The directors may adopt by resolution a procedure by which a stockholder may certify in writing to the Corporation that any shares registered in the name of the stockholder are held for the account of a specified person other than the stockholder. The resolution shall set forth the class of stockholders who may make the certification, the purpose for which the certification may be made, the form of certification and the information to be contained in it; if the certification is with respect to a record date or closing of the share transfer books, the time after the record date or closing of the share transfer books within which the certification must be received by the Corporation; and any other provisions with respect to the procedure which the directors consider necessary or desirable. On receipt of such certification, the person specified in the certification, the stockholder of record of the specified shares in place of the stockholder who makes the certification.

Section 11. Inspectors. At any meeting of stockholders, the chairman of the meeting may appoint one or more persons as inspectors for such meeting. Such inspectors shall ascertain and report the number of shares represented at the meeting based upon their determination of the validity and effect of proxies, count all votes, report the results and perform such other acts as are proper to conduct the election and voting with impartiality and fairness to all the stockholders.

Each report of an inspector shall be in writing and signed by him or by a majority of them if there is more than one inspector acting at such meeting. If there is more than one inspector, the report of a majority shall be the report of the inspectors. The report of the inspector or inspectors on the number of shares represented at the meeting and the results of the voting shall be prima facie evidence thereof.

Section 12. Reports to Stockholders. The directors shall submit to the stockholders at or before the annual meeting of stockholders a report of the business and operations of the Corporation during the prior fiscal year, containing a balance sheet and a statement of income and surplus of the Corporation, accompanied by the certification of an independent certified public accountant, and such further information as the directors may determine is required pursuant to any law or regulation to which the Corporation is subject. Within the earlier of 20 days after the annual meeting of stockholders or 120 days after the end of the fiscal year of the Corporation, the directors shall place the annual report on file at the principal office of the Corporation and with any governmental agencies as may be required by law and as the directors may deem appropriate.

# Section 13. Nominations and Proposals by Stockholders.

- (a) Annual Meetings of Stockholders. (1) Subject to the rights of holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, nominations of persons for election to the Board of Directors and the proposal of business to be considered by the stockholders may be made at an annual meeting of stockholders (i) pursuant to the Corporation's notice of meeting, (ii) by or at the direction of the directors or (iii) by any stockholder of the Corporation who was a stockholder of record both at the time of giving of notice provided for in this Section 13(a) and at the time of the annual meeting, who is entitled to vote at the meeting and who complied with the notice procedures set forth in this Section 13(a).
- (2) For nominations or other business to be properly brought before an annual meeting by a stockholder pursuant to clause (iii) of paragraph (a) (1) of this Section 13, the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation and such other business must otherwise be a proper matter for action by stockholders. To be timely, a stockholder's notice shall be delivered to the Secretary at the principal executive offices of the Corporation not later than the close of business on the 60th day nor earlier than the close of business on the 90th day before the date on which the Corporation first mailed its proxy materials for the prior year's annual meeting of stockholders. In no event shall the public announcement of a postponement or adjournment of an annual meeting to a later date or time commence a new time period for the giving of a stockholder's notice as described above. Such stockholder's notice shall set forth as to each person whom the stockholder proposes to nominate for election or reelection as a director all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors in an election contest, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (the "Exchange Act") (including such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected); (ii) as to any other business that the stockholder proposes to bring before the meeting, a brief description of the business desired to be brought before the meeting, the reasons for conducting such business at the meeting and any material interest in such business of such stockholder and of the beneficial owner, if any, on whose behalf the proposal is made; and (iii) as to the stockholder giving the notice and the beneficial owner, if any, on whose behalf the nomination or proposal is made, (x) the name and address of such stockholder, as they appear on the Corporation's books, and of such beneficial owner and (y) the number of each class of shares of the Corporation which are owned beneficially and of record by such stockholder and such beneficial owner.

- (b) Special Meetings of Stockholders. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting. Except as otherwise provided for or fixed by or pursuant to the provisions of the Charter of the Corporation relating to the rights of the holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, nominations of persons for election to the Board of Directors may be made at a special meeting of stockholders at which directors are to be elected (i) pursuant to the Corporation's notice of meeting, (ii) by or at the direction of the Board of Directors or (iii) provided that the Board of Directors has determined that directors shall be elected at such special meeting, by any stockholder of the Corporation who was a stockholder of record both at the time of giving of notice provided for in this Section 13(b) and at the time of the special meeting, who is entitled to vote at the meeting and who complied with the notice procedures set forth in this Section 13(b). Except as otherwise provided for or fixed by or pursuant to the provisions of the Charter of the Corporation relating to the rights of the holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, in addition to the foregoing requirements, for nominations or other business to be properly brought before a special meeting by a stockholder, such stockholder's notice containing the information required by paragraph (a)(2) of this Section 13 must be delivered to the Secretary at the principal executive offices of the Corporation not earlier than the close of business on the 90th day prior to such special meeting and not later than the close of business on the later of the 60th day prior to such special meeting or the tenth day following the day on which public announcement is first made of the date of the special meeting. In no event shall the public announcement of a postponement or adjournment of a special meeting to a later date or time commence a new time period for the giving of a stockholder's notice as described above.
- (c) General. (1) Except as otherwise provided for or fixed by or pursuant to the provisions of the Charter of the Corporation relating to the rights of the holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, only such persons who are nominated in accordance with the procedures set forth in this Section 13 shall be eligible to serve as directors and only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with the procedures set forth in this Section 13. The chairman of the meeting shall have the power and duty to determine whether a nomination or any business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with the procedures set forth in this Section 13 and, if any proposed nomination or business is not in compliance with this Section 13, to declare that such nomination or proposal shall be disregarded.
- (2) For purposes of this Section 13, "public announcement" shall mean disclosure in a press release reported by the Dow Jones News Service, Associated Press or comparable news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the Exchange Act.
- (3) Notwithstanding the foregoing provisions of this Section 13, a stockholder shall also comply with all applicable requirements of state law and of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 13.
- Section 14. Informal Action by Stockholders. Subject to the rights of the holders of any class or series of stock (other than Common Stock) to elect additional directors under specified circumstances and notwithstanding the provisions of Section 13 of this Article II, any action required or permitted to be taken at a meeting of stockholders may be taken without a meeting if a consent in writing, setting forth such action, is signed by all stockholders entitled to vote on such matter.

Section 15. Voting by Ballot. Voting on any question or in any election at a meeting of stockholders may be viva voce unless the presiding officer shall order or any stockholder present at such meeting in person or by proxy shall demand that voting be by ballot.

# ARTICLE III

# **DIRECTORS**

Section 1. General Powers; Qualifications; Trustees Holding Over. The business and affairs of the Corporation shall be managed under the direction of its Board of Directors. A director shall be an individual at least 21 years of age who is not under legal disability. In case of failure to elect directors at an annual meeting of the stockholders, the directors holding over shall continue to direct the management of the business and affairs of the Corporation until their successors are elected and qualify.

Section 2. Number. At any regular meeting or at any special meeting called for that purpose, a majority of the entire Board of Directors may establish, increase or decrease the number of directors, subject to any limitations on the number of directors set forth in the Charter.

Section 3. Annual and Regular Meetings. An annual meeting of the directors shall be held immediately after and at the same place as the annual meeting of stockholders, no notice other than this Bylaw being necessary. The directors may provide, by resolution, the time and place, either within or without the State of Maryland, for the holding of regular meetings of the directors without other notice than such resolution.

Section 4. Special Meetings. Special meetings of the directors may be called by or at the request of the Chairman of the Board or the President or by a majority of the directors then in office. The person or persons authorized to call special meetings of the directors may fix any place, either within or without the State of Maryland, as the place for holding any special meeting of the directors called by them.

Section 5. Notice. Notice of any special meeting shall be given by written notice delivered personally, telegraphed, facsimile-transmitted or mailed to each director at his business or residence address. Personally delivered or telegraphed notices shall be given at least two days prior to the meeting. Notice by mail shall be given at least five days prior to the meeting. Telephone or facsimile-transmission notice shall be given at least 24 hours prior to the meeting. If mailed, such notice shall be deemed to be given when deposited in the United States mail properly addressed, with postage thereon prepaid. If given by telegram, such notice shall be deemed to be given when the telegram is delivered to the telegraph company. Telephone notice shall be deemed given when the director is personally given such notice in a telephone call to which he is a party. Facsimile-transmission notice shall be deemed given upon completion of the transmission of the message to the number given to the Corporation by the director and receipt of a completed answer-back indicating receipt. Neither the business to be transacted at, nor the purpose of, any annual, regular or special meeting of the directors need be stated in the notice, unless specifically required by statute or these Bylaws.

Section 6. Quorum. A majority of the directors shall constitute a quorum for convening any meeting of the directors, provided that, if less than a majority of such directors are present at said meeting, a majority of the directors present may adjourn the meeting from time to time without further notice, and provided further that if, pursuant to the Charter or these Bylaws, the vote of a majority of a particular group of directors is required for action, a quorum must also include a majority of such group.

The directors present at a meeting which has been duly called and convened may continue to transact business until adjournment, notwithstanding the withdrawal of enough directors to leave less than a quorum.

Section 7. Voting. The action of the majority of the directors present at a meeting at which a quorum is present when such meeting is convened shall be the action of the directors, unless the concurrence of a greater proportion is required for such action by applicable statute, the Charter or these Bylaws.

Section 8. Telephone Meetings. The directors may participate in a meeting by means of a conference telephone or similar communications equipment if all persons participating in the meeting can hear each other at the same time. Participation in a meeting by these means shall constitute presence in person at the meeting.

Section 9. Informal Action by Directors. Any action required or permitted to be taken at any meeting of the directors may be taken without a meeting, if a consent in writing to such action is signed by each director and such written consent is filed with the minutes of proceedings of the directors.

Section 10. Vacancies. If for any reason any or all of the directors cease to be directors, such event shall not terminate the Corporation or affect these Bylaws or the powers of the remaining directors hereunder. Subject to the rights of holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation to elect directors under specified circumstances, any vacancy (including a vacancy created by an increase in the number of directors) shall be filled, at any regular meeting or at any special meeting called for that purpose, by a majority of the directors. Any individual so elected as director shall hold office until the next annual meeting of stockholders and until his successor is elected and qualified.

Section 11. Compensation. (a) The directors shall not receive any stated salary for their services as directors but, by resolution of the directors, may receive fixed sums per year and/or per meeting and/or per visit to real property owned or to be acquired by the Corporation and for any service or activity they perform or engage in as directors. Such fixed sums may be paid either in cash or in shares of the Corporation. Directors may be reimbursed for expenses of attendance, if any, at each annual, regular or special meeting of the directors or of any committee thereof; and for their expenses, if any, in connection with each property visit and any other service or activity performed or engaged in as directors; but nothing herein contained shall be construed to preclude any directors from serving the Corporation in any other capacity and receiving compensation therefor.

(b) The Corporation may lend money to, guarantee an obligation of or otherwise assist a director or a trustee or director of a direct or indirect subsidiary of the Corporation; provided, however, that such director or other person is also an executive officer of the Corporation or of such subsidiary, or the loan, guarantee or other assistance is in connection with the purchase of Shares. The loan, guarantee or other assistance may be with or without interest, unsecured, or secured in any manner that the Board of Directors approves, including a pledge of shares.

Section 12. Removal of Directors. Subject to the rights of holders of one or more classes or series of stock other than Common Stock to elect one or more directors, any director may be removed only for cause and only by the affirmative vote of stockholders holding at least two thirds of all the votes entitled to be cast for the election of directors generally.

Section 13. Loss of Deposits. No director shall be liable for any loss which may occur by reason of the failure of the bank, trust company, savings and loan association, or other institution with whom moneys or shares have been deposited.

Section 14. Surety Bonds. Unless required by law, no director shall be obligated to give any bond or surety or other security for the performance of any of his duties.

Section 15. Reliance. Each director, officer, employee and agent of the Corporation shall, in the performance of his duties with respect to the Corporation, be fully justified and protected with regard to any act or failure to act in reliance in good faith upon the books of account or other records of the Corporation, upon an opinion of counsel or upon reports made to the Corporation by any of its officers or employees or by the advisers, accountants, appraisers or other experts or consultants selected by the directors or officers of the Corporation, regardless of whether such counsel or expert may also be a director.

Section 16. Certain Rights of Directors, Officers, Employees and Agents. The directors shall have no responsibility to devote their full time to the affairs of the Corporation. Any director or officer, employee or agent of the Corporation (other than a full-time officer, employee or agent of the Corporation), in his personal capacity or in a capacity as an affiliate, employee or agent of any other person, or otherwise, may have business interests and engage in business activities similar or in addition to those of or relating to the Corporation.

# ARTICLE IV

# COMMITTEES

Section 1. Number, Tenure and Qualification. The directors may appoint from among its members an Executive Committee, an Audit Committee and a Compensation Committee, each composed of at least three directors, and other committees, each composed of one or more directors, to serve at the pleasure of the directors; provided, that the membership of the Compensation Committee shall consist of a majority of Independent Directors and the membership of the Audit Committee shall consist only of Independent Directors. An individual shall be deemed to be an "Independent Director" hereunder if such individual is not an affiliate of the Corporation and is not an employee of the Corporation.

Section 2. Powers. The directors may delegate to committees appointed under Section 1 of this Article IV any of the powers of the directors, except as prohibited by law.

Section 3. Meetings. Notice of committee meetings shall be given in the same manner as notice for special meetings of the Board of Directors. One-third, but not less than two (except for one-member committees), of the members of any committee shall be present in person at any meeting of such committee in order to constitute a quorum for the transaction of business at such meeting, and the act of a majority present shall be the act of such committee. The Board of Directors may designate a chairman of any committee, and such chairman or any two members of any committee (except for one-member committees) may fix the time and place of its meetings unless the Board shall otherwise provide. In the absence or disqualification of any member of any such committee, the members thereof present at any meeting and not disqualified from voting, whether or not they constitute a quorum, may unanimously appoint another director to act at the meeting in the place of such absent or disqualified members.

Each committee shall keep minutes of its proceedings and shall report the same to the Board of Directors at the next succeeding meeting, and any action by the committee shall be subject to revision and alteration by the Board of Directors, provided that no rights of third persons shall be affected by any such revision or alteration.

Section 4. Telephone Meetings. Members of a committee of the directors may participate in a meeting by means of a conference telephone or similar communications equipment if all persons participating in the meeting can hear each other at the same time. Participation in a meeting by these means shall constitute presence in person at the meeting.

Section 5. Informal Action by Committees. Any action required or permitted to be taken at any meeting of a committee of the directors may be taken without a meeting, if a consent in writing to such action is signed by each member of the committee and such written consent is filed with the minutes of proceedings of such committee.

Section 6. Vacancies. Subject to the provisions hereof, the Board of Directors shall have the power at any time to change the membership of any committee, to fill all vacancies, to designate alternate members to replace any absent or disqualified member or to dissolve any such committee.

Section 7. Emergency. In the event of a state of disaster of sufficient severity to prevent the conduct and management of the affairs and business of the Corporation by its directors and officers as contemplated by the Charter and these Bylaws, any two or more available members of the then incumbent Executive Committee shall constitute a quorum of that Committee for the full conduct and management of the affairs and business of the Corporation in accordance with the provisions of this Article IV. In the event of the unavailability, at such time, of a minimum of two members of the then incumbent Executive Committee, the available directors shall elect an Executive Committee composed of any two members of the Board of Directors, whether or not they be officers of the Corporation, which two members shall constitute the Executive Committee for the full conduct and management of the affairs of the Corporation in accordance with the foregoing provisions of this Section 7. This Section 7 shall be subject to implementation by resolution of the Board of Directors passed from time to time for that purpose, and any provisions of the Bylaws (other than this Section 7) and any resolutions which are contrary to the provisions of this Section 7 or to the provisions of any such implementing resolutions shall be

suspended until it shall be determined by any interim Executive Committee acting under this Section 7 that it shall be to the advantage of the Corporation to resume the conduct and management of its affairs and business under all the other provisions of these Bylaws.

#### ARTICLE V

# **OFFICERS**

Section 1. General Provisions. The officers of the Corporation shall include a President, a Secretary and a Treasurer and may include a Chairman of the Board, a Vice Chairman of the Board, a Chief Executive Officer, a Chief Operating Officer, a Chief Financial Officer, a Corporate Counsel, one or more Vice Presidents, one or more Assistant Secretaries and one or more Assistant Treasurers. In addition, the directors may from time to time appoint such other officers with such powers and duties as they shall deem necessary or desirable. The officers of the Corporation shall be elected annually by the directors at the first meeting of the directors held after each annual meeting of stockholders. If the election of officers shall not be held at such meeting, such election shall be held as soon thereafter as may be convenient. Each officer shall hold office until his successor is elected and qualifies or until his death, resignation or removal in the manner hereinafter provided. Any two or more offices except President and Vice President may be held by the same person. In their discretion, the directors may leave unfilled any office except that of President and Secretary. Election of an officer or agent shall not of itself create contract rights between the Corporation and such officer or agent.

Section 2. Removal and Resignation. Any officer or agent of the Corporation may be removed at any time by the directors if in their judgment the best interests of the Corporation would be served thereby, but such removal shall be without prejudice to the contract rights, if any, of the person so removed. Any officer of the Corporation may resign at any time by giving written notice of his resignation to the directors, the Chairman of the Board, the President or the Secretary. Any resignation shall take effect at any time subsequent to the time specified therein or, if the time when it shall become effective is not specified therein, immediately upon its receipt. The acceptance of a resignation shall not be necessary to make it effective unless otherwise stated in the resignation. Such resignation shall be without prejudice to the contract rights, if any, of the Corporation.

Section 3. Vacancies. A vacancy in any office may be filled by the directors for the balance of the term.

Section 4. Chairman and Vice Chairman of the Board. The directors may from time to time appoint a Chairman of the Board and a Vice Chairman of the Board. The Chairman of the Board shall preside over the meetings of the directors and of the stockholders at which he shall be present and shall in general oversee all of the business and affairs of the Corporation. In the absence of the Chairman of the Board, the Vice Chairman of the Board shall preside at such meetings at which he shall be present. The Chairman and the Vice Chairman of the Board may execute any deed, mortgage, bond, contract or other instrument, except in cases where the execution thereof shall be expressly delegated by the directors or by these Bylaws to an officer or some other agent of the Corporation or shall be required by law to be otherwise executed. The Chairman of the Board and the Vice Chairman of the Board shall perform such other duties as may be assigned to him or them by the directors.

Section 5. Chief Executive Officer. The directors may designate a Chief Executive Officer from among the elected officers. The Chief Executive Officer shall have responsibility for implementation of the policies of the Corporation, as determined by the directors, and for the administration of the business affairs of the Corporation. In the absence of both the Chairman and Vice Chairman of the board, the Chief Executive Officer shall preside over the meetings of the directors and of the stockholders at which he shall be present.

Section 6. Chief Operating Officer. The directors may designate a Chief Operating Officer from among the elected officers. Said officer will have the responsibilities and duties as set forth by the Chief Executive Officer, the President or the directors.

Section 7. Chief Financial Officer. The directors may designate a Chief Financial Officer from among the elected officers. Said officer will have the responsibilities and duties as set forth by the Chief Executive Officer, the President or the directors.

Section 8. Corporate Counsel. The directors may designate a Corporate Counsel from among the elected officers. Said officer will have the responsibilities and duties as set forth by the Chief Executive Officer, the President or the directors.

Section 9. President. In the absence of the Chairman, the Vice Chairman of the Board and the Chief Executive Officer, the President shall preside over the meetings of the directors and of the stockholders at which he shall be present. In the absence of a designation of a Chief Executive Officer by the directors, the President shall be the Chief Executive Officer and shall be ex officio a member of all committees that may, from time to time, be constituted by the directors. The President may execute any deed, mortgage, bond, contract or other instrument, except in cases where the execution thereof shall be expressly delegated by the directors or by these Bylaws to some other officer or agent of the Corporation or shall be required by law to be otherwise executed; and in general shall perform all duties incident to the office of president and such other duties as may be prescribed by the Chief Executive Officer or the directors from time to time.

Section 10. Vice Presidents. In the absence of the President or in the event of a vacancy in such office, the Vice President (or in the event there be more than one Vice President, the Vice Presidents in the order designated at the time of their election or, in the absence of any designation, then in the order of their election) shall perform the duties of the President and when so acting shall have all the powers of and be subject to all the restrictions upon the President; and shall perform such other duties as from time to time may be assigned to him by the Chief Executive Officer, the President or the directors. The directors may designate one or more Vice Presidents as Executive Vice President, Senior Vice President or as Vice President for particular areas of responsibility.

Section 11. Treasurer. The Treasurer shall have the custody of the funds and securities of the Corporation and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and shall deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the directors.

The Treasurer shall disburse the funds of the Corporation as may be ordered by the directors, taking proper vouchers for such disbursements, and shall render to the Chief Executive Officer, the President and the directors, at the regular meetings of the directors or whenever they may require it, an account of all his transactions as Treasurer and of the financial condition of the Corporation.

If required by the directors, the Treasurer shall give the Corporation a bond in such sum and with such surety or sureties as shall be satisfactory to the directors for the faithful performance of the duties of his office and for the restoration to the Corporation, in case of his death, resignation, retirement or removal from office, of all books, papers, vouchers, moneys and other property of whatever kind in his possession or under his control belonging to the Corporation.

Section 12. Secretary. The Secretary shall (a) keep the minutes of the proceedings of the stockholders, the directors and committees of the directors in one or more books provided for that purpose; (b) see that all notices are duly given in accordance with the provisions of these Bylaws or as required by law; (c) be custodian of the trust records and of the seal of the Corporation; (d) keep a register of the post office address of each stockholder which shall be furnished to the Secretary by such stockholder; (e) have general charge of the share transfer books of the Corporation; and (f) in general perform such other duties as from time to time may be assigned to him by the Chief Executive Officer, the President or the directors.

Section 13. Assistant Secretaries and Assistant Treasurers. The Assistant Secretaries and Assistant Treasurers, in general, shall perform such duties as shall be assigned to them by the Secretary or Treasurer, respectively, or by the Chief Executive Officer, the President or the directors. The Assistant Treasurers shall, if required by the directors, give bonds for the faithful performance of their duties in such sums and with such surety or sureties as shall be satisfactory to the directors.

Section 14. Salaries. The salaries and other compensation of the officers shall be fixed from time to time by the directors and no officer shall be prevented from receiving such salary or other compensation by reason of the fact that he is also a director.

#### ARTICLE VI

# CONTRACTS, LOANS, CHECKS AND DEPOSITS

Section 1. Contracts. The directors may authorize any officer or agent to enter into any contract or to execute and deliver any instrument in the name of and on behalf of the Corporation and such authority may be general or confined to specific instances. Any agreement, deed, mortgage, lease or other document executed by one or more of the directors or by an authorized person shall be valid and binding upon the directors and upon the Corporation when authorized or ratified by action of the directors.

Section 2. Checks and Drafts. All checks, drafts or other orders for the payment of money, notes or other evidences of indebtedness issued in the name of the Corporation shall be signed by such officer or agent of the Corporation in such manner as shall from time to time be determined by the directors.

Section 3. Deposits. All funds of the Corporation not otherwise employed shall be deposited from time to time to the credit of the Corporation in such banks, trust companies or other depositories as the directors may designate.

# ARTICLE VII

# SHARES

Section 1. Certificates. Each stockholder shall be entitled to a certificate or certificates which shall represent and certify the number of shares of each class of stock held by him in the Corporation. Each certificate shall be signed by the Chief Executive Officer, the President or a Vice President and countersigned by the Secretary or an Assistant Secretary or the Treasurer or an Assistant Treasurer and may be sealed with the seal, if any, of the Corporation. The signatures may be either manual or facsimile. Certificates shall be consecutively numbered; and if the Corporation shall, from time to time, issue several classes or series of shares, each class or series may have its own number series. A certificate is valid and may be issued whether or not an officer who signed it is still an officer when it is issued. Each certificate representing shares which are restricted as to their transferability or voting powers, which are preferred or limited as to their dividends or as to their allocable portion of the assets of the Corporation upon liquidation or which are redeemable at the option of the Corporation, shall have a statement of such restriction, limitation, preference or redemption provision, or a summary thereof, plainly stated on the certificate. In lieu of such statement or summary, the Corporation may set forth upon the face or back of the certificate a statement that the Corporation will furnish to any stockholder, upon request and without charge, a full statement of such information.

Section 2. Transfers. Certificates shall be treated as negotiable and title thereto and to the shares they represent shall be transferred by delivery thereof. Upon surrender to the Corporation or the transfer agent of the Corporation of a share certificate duly endorsed or accompanied by proper evidence of succession, assignment or authority to transfer, the Corporation shall issue a new certificate to the person entitled thereto, cancel the old certificate and record the transaction upon its books.

The Corporation shall be entitled to treat the holder of record of any share or shares as the holder in fact thereof and, accordingly, shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of the State of Maryland.

Notwithstanding the foregoing, transfers of shares of stock of the Corporation will be subject in all respects to the Charter and all of the terms and conditions contained therein.

Section 3. Replacement Certificate. Any officer designated by the directors may direct a new certificate to be issued in place of any certificate previously issued by the Corporation alleged to have been lost, stolen or destroyed upon the making of an affidavit of that fact by the person claiming the certificate to be lost, stolen or destroyed. When authorizing the issuance of a new certificate, an officer designated by the directors may, in his discretion and as a condition precedent to the issuance thereof, require the owner of such lost, stolen or destroyed certificate or the owner's legal representative to advertise the same in such manner as he shall require and/or to give bond, with sufficient surety, to the Corporation to indemnify it against any loss or claim which may arise as a result of the issuance of a new certificate.

Section 4. Closing of Transfer Books or Fixing of Record Date. The directors may set, in advance, a record date for the purpose of determining stockholders entitled to notice of or to vote at any meeting of stockholders or determining stockholders entitled to receive payment of any dividend or the allotment of any other rights, or in order to make a determination of stockholders for any other proper purpose. Such date, in any case, shall not be prior to the close of business on the day the record date is fixed and shall be not more than 90 days and, in the case of a meeting of stockholders not less than ten days, before the date on which the meeting or particular action requiring such determination of stockholders of record is to be held or taken.

In lieu of fixing a record date, the directors may provide that the share transfer books shall be closed for a stated period but not longer than 20 days. If the share transfer books are closed for the purpose of determining stockholders entitled to notice of or to vote at a meeting of stockholders, such books shall be closed for at least ten days before the date of such meeting.

If no record date is fixed and the share transfer books are not closed for the determination of stockholders, (a) the record date for the determination of stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day on which the notice of meeting is mailed or the 30th day before the meeting, whichever is the closer date to the meeting; and (b) the record date for the determination of stockholders entitled to receive payment of a dividend or an allotment of any other rights shall be the close of business on the day on which the resolution of the directors, declaring the dividend or allotment of rights, is adopted.

When a determination of stockholders entitled to vote at any meeting of stockholders has been made as provided in this section, such determination shall apply to any adjournment thereof, except when (i) the determination has been made through the closing of the transfer books and the stated period of closing has expired or (ii) the meeting is adjourned to a date more than 120 days after the record date fixed for the original meeting, in either of which case a new record date shall be determined as set forth herein.

Section 5. Share Ledger. The Corporation shall maintain at its principal office or at the office of its counsel, accountants or transfer agent, an original or duplicate share ledger containing the name and address of each stockholder and the number of shares of each class held by such stockholder.

Section 6. Fractional Shares; Issuance of Notes. The directors may issue fractional shares or provide for the issuance of scrip, all on such terms and under such conditions as they may determine. Notwithstanding any other provision of the Charter or these Bylaws, the directors may issue units consisting of different securities of the Corporation. Any security issued in a unit shall have the same characteristics as any identical securities issued by the Corporation, except that the directors may provide that for a specified period securities of the Corporation issued in such unit may be transferred on the books of the Corporation only in such unit.

# ARTICLE VIII

# FISCAL YEAR

The directors shall have the power, from time to time, to fix the fiscal year of the Corporation by a duly adopted resolution.

## ARTICLE IX

## **DISTRIBUTIONS**

Section 1. Authorization. Dividends and other distributions upon the shares of stock of the Corporation may be authorized and declared by the directors, subject to the provisions of law and the Charter. Dividends and other distributions may be paid in cash, property or shares of the Corporation, subject to the provisions of law and the Charter.

Section 2. Contingencies. Before payment of any dividends or other distributions, there may be set aside out of any funds of the Corporation available for dividends or other distributions such sum or sums as the directors may from time to time, in their absolute discretion, think proper as a reserve fund for contingencies, for equalizing dividends or other distributions, for repairing or maintaining any property of the Corporation or for such other purpose as the directors shall determine to be in the best interest of the Corporation, and the directors may modify or abolish any such reserve in the manner in which it was created.

#### ARTICLE X

## INVESTMENT POLICIES

Subject to the provisions of the Charter, the Board of Directors may from time to time adopt, amend, revise or terminate any policy or policies with respect to investments by the Corporation as it shall deem appropriate in its sole discretion.

## ARTICLE XI

## SEAL

Section 1. Seal. The directors may authorize the adoption of a seal by the Corporation. The seal shall have inscribed thereon the name of the Corporation and the year of its formation. The directors may authorize one or more duplicate seals and provide for the custody thereof.

Section 2. Affixing Seal. Whenever the Corporation is permitted or required to affix its seal to a document, it shall be sufficient to meet the requirements of any law, rule or regulation relating to a seal to place the word "(SEAL)" adjacent to the signature of the person authorized to execute the document on behalf of the Corporation.

## ARTICLE XII

## INDEMNIFICATION AND ADVANCE OF EXPENSES

To the maximum extent permitted by Maryland law in effect from time to time, the Corporation shall indemnify (a) any director or officer or any former director or officer (including among the foregoing, for all purposes of this Article XII and without limitation, any individual who, while a director or officer and at the express request of the Corporation, serves or has served another corporation, partnership, joint venture, trust, employee benefit plan or any other enterprise as a director, officer, partner or trustee of such corporation, partnership, joint venture, trust, employee benefit plan or other enterprise) who has been successful, on the merits or otherwise, in the defense of a proceeding to which he was made a party by reason of service in such capacity, against reasonable expenses incurred by him in connection with the proceeding, and (b) any director or officer or any former director or officer against any claim or liability to which he may become subject by reason of such status unless it is established that (i) his act or omission was material to the matter giving rise to the proceeding and was committed in bad faith or was the result of active and deliberate dishonesty, (ii) he actually received an

improper personal benefit in money, property or services or (iii) in the case of a criminal proceeding, he had reasonable cause to believe that his act or omission was unlawful. In addition, the Corporation shall, without requiring a preliminary determination of the ultimate entitlement to indemnification, pay or reimburse, in advance of final disposition of a proceeding, reasonable expenses incurred by a director or officer or former director or officer made a party to a proceeding by reason of such status, provided that, in the case of a director or officer, the Corporation shall have received (i) a written affirmation by the director or officer of his good faith belief that he has met the applicable standard of conduct necessary for indemnification by the Corporation as authorized by these Bylaws and (ii) a written undertaking by or on his behalf to repay the amount paid or reimbursed by the Corporation if it shall ultimately be determined that the applicable standard of conduct was not met. The Corporation may, with the approval of its directors, provide such indemnification or payment or reimbursement of expenses to any director or officer or any former director or officer who served a predecessor of the Corporation and to any employee or agent of the Corporation or a predecessor of the Corporation. Neither the amendment nor repeal of this Article, nor the adoption or amendment of any other provision of the Charter or these Bylaws inconsistent with this Article, shall apply to or affect in any respect the applicability of this Article with respect to any act or failure to act which occurred prior to such amendment, repeal or adoption.

Any indemnification or payment or reimbursement of the expenses permitted by these Bylaws shall be furnished in accordance with the procedures provided for indemnification or payment or reimbursement of expenses, as the case may be, under Section 2-418 of the MGCL for directors of Maryland corporations. The Corporation may provide to directors and officers such other and further indemnification or payment or reimbursement of expenses, as the case may be, to the fullest extent permitted by the MGCL, as in effect from time to time, for directors of Maryland corporations.

#### ARTICLE XIII

## WAIVER OF NOTICE

Whenever any notice is required to be given pursuant to the Charter or Bylaws or pursuant to applicable law, a waiver thereof in writing, signed by the person or persons entitled to such notice, whether before or after the time stated therein, shall be deemed equivalent to the giving of such notice. Neither the business to be transacted at nor the purpose of any meeting need be set forth in the waiver of notice, unless specifically required by statute. The attendance of any person at any meeting shall constitute a waiver of notice of such meeting, except where such person attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened.

# ARTICLE XIV

## AMENDMENT OF BYLAWS

The directors shall have the exclusive power to adopt, alter or repeal any provision of these Bylaws and to make new Bylaws.

## ARTICLE XV

# MISCELLANEOUS

All references to the Charter shall include any amendments thereto. In these Bylaws, unless the context otherwise requires, words used in the singular or in the plural include both the plural and singular and words denoting any gender include all genders.

\* \* \* \*

November [ ], 1998

The Board of Directors Host Marriott Corporation 10400 Fernwood Road

Bethesda, Maryland 20817

Lady and Gentlemen:

BT Wolfensohn has acted as financial advisor to Host Marriott Corporation, a Delaware corporation ("Host"), in connection with its proposed conversion to a real estate investment trust ("REIT").

You have requested BT Wolfensohn's opinion, as investment bankers, as to the fairness, from a financial point of view, to the holders of Host common stock of certain transactions, described in the appendix attached hereto (the "REIT Transactions"), to be effected by Host in connection with the conversion to a REIT.

In connection with BT Wolfensohn's role as financial advisor to Host, and in arriving at its opinion, BT Wolfensohn has reviewed certain publicly available financial and other information concerning Host and Crestline, as defined in the appendix attached hereto, and certain internal analyses and other information furnished to it by Host and Crestline. BT Wolfensohn has also held discussions with members of the senior management of Host and Crestline regarding the businesses and prospects of Host and of Crestline. In addition, BT Wolfensohn has (i) reviewed the reported prices and trading activity for Host common stock, (ii) compared certain financial and stock market information for Host with similar information for certain companies whose securities are publicly traded, (iii) reviewed certain public information of certain companies it deemed appropriate in analyzing Host and Crestline, and (iv) performed such other studies and analyses and considered such other factors as it deemed appropriate.

BT Wolfensohn has not assumed responsibility for independent verification of, and has not independently verified, any information, whether publicly available or furnished to it, concerning Host and Crestline, including, without limitation, any financial information, forecasts or projections considered in connection with the rendering of its opinion. Accordingly, for purposes of its opinion, BT Wolfensohn has assumed and relied upon the accuracy and completeness in all material respects of all such information and BT Wolfensohn has not conducted a physical inspection of any of the properties or assets, and has not prepared or obtained any independent evaluation or appraisal of any of the assets or liabilities, of Host and Crestline. With respect to the financial forecasts and projections made available to BT Wolfensohn and used in its analyses, BT Wolfensohn has assumed that they have been reasonably prepared on bases reflecting the best currently available estimates and judgments of the management of Host and Crestline, as the case may be, as to the matters covered thereby. In rendering its opinion, BT Wolfensohn expresses no view as to the reasonableness of such forecasts and projections or the assumptions on which they are based. BT Wolfensohn's opinion is necessarily based upon economic, market and other conditions as in effect on, and the information made available to it as of, the date hereof.

BT Wolfensohn has assumed that the REIT Transactions will take place prior to or in connection with Host's election of REIT status in accordance with Sections 856 through 860 of the Internal Revenue Code of 1986, as amended (the "REIT Rules"). For purposes of rendering its opinion, BT Wolfensohn has analyzed the REIT Transactions assuming alternatively that the Blackstone Acquisition (as defined in the appendix attached hereto) is not consummated or is consummated in accordance with the terms of such transaction. BT Wolfensohn has also assumed that all material federal, state, local and other approvals and consents required in connection with the REIT Transactions will be obtained and that in connection with obtaining any necessary federal, state,

local and other approvals and consents, or any amendments, modifications or waivers to any agreements, instruments or orders to which Host is a party or is subject or by which it is bound, no limitations, restrictions or conditions will be imposed or amendments, modifications or waivers made that would have a material adverse effect on Host or Crestline. In addition, BT Wolfensohn has been advised by Host, and accordingly has assumed for purposes of its opinion, that the E&P Distribution (as defined in the appendix attached hereto) will be taxable to the Host stockholders.

BT Wolfensohn has also assumed that upon consummation of the REIT Transactions, Host will qualify for treatment as a REIT under the REIT Rules no later than the full year beginning January 1, 2000 (and BT Wolfensohn understands that Host will obtain an opinion, which will be based on certain assumptions and representations of Host and subject to certain qualifications, to that effect from its outside counsel prior to consummation of the REIT Transactions). At your request, we have not solicited any proposals from any third parties for the acquisition of any of the assets of Host or Crestline nor made any determination as to whether any such proposal could be obtained if solicited.

This opinion is addressed to, and for the use and benefit of, the Board of Directors of Host and is not a recommendation to holders of Host common stock to approve the merger of Host into HMC Merger Corporation in order to give effect to the REIT Transactions. This opinion is limited to the fairness, from a financial point of view, to the holders of Host common stock of the REIT Transactions, and BT Wolfensohn expresses no opinion as to the merits of the underlying decision by Host to engage in the REIT Transactions.

BT Wolfensohn is engaged in the merger and acquisition and client advisory business of Bankers Trust (together with its affiliates, the "BT Group") and, for legal and regulatory purposes, is a division of BT Alex. Brown Incorporated, a registered broker-dealer and member of the New York Stock Exchange. BT Wolfensohn will be paid a fee for its services as financial advisor to Host in connection with the REIT Transactions, a substantial portion of which is contingent upon consummation of the REIT Transactions. BT Wolfensohn regularly performs advisory services for Host, which pays an annual retainer fee for such services. One or more members of the BT Group have, from time to time, provided investment banking, commercial banking (including extension of credit) and other financial services to Host or its affiliates for which it has received compensation. A member of the BT Group is the lead administrative agent for Host's \$1.25 billion bank facility and was the coadministrator agent for Host's recent bond tender and subsequent \$1.7 billion re-offering. In the ordinary course of business, members of the BT Group may actively trade in the securities and other instruments and obligations of Host for their own accounts and for the accounts of their customers. Accordingly, the BT Group may at any time hold a long or short position in such securities, instruments and obligations.

Based upon and subject to the foregoing, it is BT Wolfensohn's opinion as investment bankers that the REIT Transactions, taken together, are fair, from a financial point of view, to the holders of Host common stock.

Very truly yours,

BT WOLFENSOHN

## **APPENDIX**

For the purposes of this opinion, the REIT Transactions mean (i) the contribution by Host of its wholly-owned full-service hotels, certain interests in hotel partnerships and certain other assets to Host Marriott L.P. in exchange for units of limited and general partnership interest in Host Marriott L.P. and the assumption of liabilities, (ii) the reincorporation of Host from the State of Delaware to the State of Maryland by merging Host into HMC Merger Corporation ("Host REIT"), a wholly-owned subsidiary of Host, (iii) the taxable distribution by Host to its stockholders which Host has advised BT Wolfensohn will consist of shares of common stock of Crestline (as defined below) and approximately \$250 million in cash or, at the election of Host common stockholders, in Host REIT common stock (the "E&P Distribution"), (iv) the assumption by Host Marriott L.P. of the debentures of Host underlying the Convertible Quarterly Income Preferred Securities and other indebtedness of Host and the related conversion price adjustment of the such debentures, (v) the acquisition by merger of eight limited partnerships that own full service hotels in which Host or its subsidiaries are general partners in exchange for limited partnership interests in Host Marriott L.P., (vi) the acquisition of partnership interests from unaffiliated partners of four partnerships that own one or more full-service hotels and that are partially but not wholly-owned by Host or one of its subsidiaries in exchange for limited partnership interests in Host Marriott L.P., (vii) the acquisition from The Blackstone Group and a series of funds controlled by Blackstone Real Estate Partners of ownership of, or controlling interests in, twelve hotels and a mortgage loan secured by a thirteenth hotel in exchange for limited partnership interests in Host Marriott L.P., the assumption of certain liabilities and other consideration, including up to 18% of the shares of Crestline (as defined below) common stock (the "Blackstone Acquisition"), (viii) the creation and capitalization of the taxable corporations in which Host Marriott L.P. will own 95% of the economic interest but not voting stock and which will hold various assets contributed by Host and its subsidiaries to Host Marriott L.P., which assets, if owned directly by Host Marriott L.P., could jeopardize Host's status as a REIT, and (ix) the leasing of all of virtually the full-service hotels owned or controlled by Host Marriott L.P. for initial terms ranging generally from seven to ten years to lessees, which will be indirect wholly-owned subsidiaries of Crestline (the "Lessees"), whereby the Lessees will operate the hotels under the existing brand names and pursuant to their existing management agreements.

The terms and conditions of the transactions comprising the REIT Transactions are more fully described in the Proxy Statement/Prospectus dated November [ ], 1998.

Crestline means Crestline Capital Corporation (formerly HMC Senior Communities, Inc.), a Maryland corporation, or, as the context may require, such entity together with the Lessees and its other subsidiaries or any of them, which currently is a wholly-owned subsidiary of Host but will become a separate public company as part of the REIT Transactions when Host distributes the common stock of Crestline to its existing stockholders, as part of the E&P Distribution.

HOGAN & HARTSON L.L.P.

COLUMBIA SQUARE

555 13TH STREET, N.W.

WASHINGTON, D.C. 20004

NOVEMBER 20, 1998

HMC Merger Corporation Host Marriott, L.P. 10400 Fernwood Road Bethesda, MD 20817

#### Ladies and Gentlemen:

We have acted as tax counsel to HMC Merger Corporation, a Maryland corporation ("Host REIT"), Host Marriott Corporation, a Delaware corporation ("Host"), and Host Marriott, L.P., a Delaware limited partnership (the "Operating Partnership"), in connection with the following series of related transactions (which collectively are referred to as the "REIT Conversion"), each of which is described more fully in the Proxy Statement/Prospectus which is part of the Registration Statement filed with the Securities and Exchange Commission by Host REIT on Form S-4 (File No. 333-64793):

- (1) the contribution by Host of its wholly-owned full-service hotel assets, interests in certain hotel partnerships, and certain other assets to the Operating Partnership in exchange for OP Units (and possibly preferred partnership interests) and the assumption of all of Host's liabilities other than the liabilities of Crestline Capital Corporation, a Maryland corporation ("Crestline"), (the "OP Contribution");
- (2) the distribution by Host of Crestline common stock to Host's shareholders and the one or more distributions of cash or other consideration made in connection with the REIT Conversion by Host or Host REIT to their respective shareholders (collectively, the "Initial E&P Distribution");
- (3) the reincorporation of Host from the State of Delaware to the State of Maryland by means of the merger of Host with and into Host REIT (the "Merger");
- (4) the recently completed refinancing and amendment of the debt securities and certain credit facilities of Host;
- (5) the acquisition by merger of up to eight limited partnerships that own full-service hotels in which Host or its subsidiaries are general partners (the "Partnerships") in exchange for OP Units (which may be exchanged for Host REIT Common Stock or unsecured notes of the Operating Partnership), if and to the extent consummated (the "Partnership Mergers");
- (6) the acquisition of the partnership or membership interests from unaffiliated partners or members of four partnerships or limited liability companies that own one or more full-service hotels and that are partially but not wholly-owned by Host or one of its subsidiaries (the "Private Partnerships") in exchange for OP Units, if and to the extent consummated;
- (7) the expected acquisition by the Operating Partnership from The Blackstone Group and a series of funds controlled by Blackstone Real Estate Partners of ownership of, or controlling interests in, twelve hotels and a mortgage loan secured by a thirteenth hotel in exchange for OP Units, the assumption of certain liabilities and other consideration, including up to approximately 18% of the shares of Crestline common stock (the "Blackstone Acquisition"), to the extent such acquisition is consummated;
- (8) the creation and capitalization of the taxable corporations (the "Non-Controlled Subsidiaries") in which the Operating Partnership will own 95% of the economic interest but no voting stock and which will hold various assets contributed by Host and its subsidiaries to the Operating Partnership, with all of the voting stock, representing 5% of the economic interest, to be owned by the Host Marriott Statutory

Employee/Charitable Trust, the beneficiaries of which will be a separate trust benefiting certain employees of Host REIT, a designated public charity, and possibly other outside investors (the "Host Employee/Charitable Trust");

- (9) the leasing of virtually all of the full-service Hotels owned or controlled by the Operating Partnership for initial terms ranging generally from seven to ten years (the "Leases") to lessees, which will be indirect wholly-owned subsidiaries of Crestline (the "Lessees"); and
- (10) the related transactions described in the Proxy Statement/Prospectus and the other steps occurring prior to, substantially concurrent with or within a reasonable time after the effective date of the Merger (the "Effective Date"), as Host may determine to be necessary or desirable to complete or facilitate the transactions contemplated in the Proxy Statement/Prospectus or otherwise to permit Host REIT to elect to be treated as a REIT for federal income tax purposes for its first taxable year commencing after the completion of the REIT Conversion.

In connection with the REIT Conversion, we have been asked to provide you with the opinions on certain federal income tax matters set forth in this letter. Capitalized terms used in this letter and not otherwise defined herein have the meaning set forth in the Proxy Statement/Prospectus.

## BASES FOR OPINIONS

The opinions set forth in this letter are based on our best judgment regarding the application of relevant current provisions of the Internal Revenue Code of 1986, as amended (the "Code"), Treasury Regulations thereunder (including proposed and temporary Treasury Regulations), and interpretations of the foregoing as expressed in court decisions, applicable legislative history, and the administrative rulings and practices of the IRS, including its practices and policies in issuing private letter rulings, which are not binding on the IRS except with respect to a taxpayer that receives such a ruling, all as of the date hereof. These provisions and interpretations are subject to change, which may or may not be retroactive in effect, that might result in material modifications of our opinions. Nevertheless, we undertake no responsibility to advise you of any new developments in the law or in the application or interpretation of the federal income tax laws.

Our opinion does not foreclose the possibility of a contrary determination by the IRS or a court of competent jurisdiction, or of a contrary position taken by the IRS or the Treasury Department in regulations or rulings issued in the future. In this regard, an opinion of counsel with respect to an issue merely represents counsel's best judgment with respect to the probable outcome on the merits with respect to such issue, is not binding on the IRS or the courts, and is not a guarantee that the IRS will not assert a contrary position with respect to such issue or that a court will not sustain such a position asserted by the IRS.

In rendering the following opinions, we have examined such statutes, regulations, records, agreements, certificates and other documents as we have considered necessary or appropriate as a basis for such opinions, including the following:

- (1) the form of Agreement and Plan of Merger attached to the Proxy Statement/Prospectus at Appendix A and to be executed by and among Host, Host REIT and the Operating Partnership (the "Merger Agreement");
  - (2) the Proxy Statement/Prospectus;
- (3) the Prospectus/Consent Solicitation Statement which is part of the Registration Statement filed with the SEC by the Operating Partnership and Host REIT on Form S-4 (File No. 333-55807) and which relates to the Partnership Mergers;
- (4) the Prospectus which is part of the Registration Statement filed with the SEC by Crestline on Form S-1 (File No. 333-64657);
- (5) the Restated Certificate of Incorporation of Host, dated November 11, 1997 and the Bylaws of Host, as certified by the Secretary of Host as being true, complete, correct and in effect as of the date hereof;

- (6) the Articles of Incorporation of Host REIT dated September 28, 1998 and the Bylaws of Host REIT dated September 28, 1998;
- (7) the form of Articles of Amendment and Restatement of Articles of Incorporation of Host REIT;
- (8) the form of Amended Articles of Incorporation of Crestline and the form of Bylaws of Crestline;
- (9) the form of the contribution agreement relating to the OP Contribution:
- (10) the form of Second Amended and Restated Agreement of Limited Partnership of the Operating Partnership, proposed to be entered into at or about the time of the Partnership Mergers;
- (11) the form of agreement of merger relating to the Partnership Mergers (including various exhibits thereto) by and among Host REIT, the Operating Partnership, a Partnership, and its corresponding Merger Partnership, as amended to the date hereof;
- (12) the partnership agreements (or form thereof), each as amended to the date hereof and as proposed to be amended in connection with the REIT Conversion, of each Partnership, each Merger Partnership and each Private Partnership that will remain in existence after the REIT Conversion;
- (13) the form of lease relating to the Leases and the executed term sheets regarding each Lease;
- (14) the form of organizational documents relating to the formation and capitalization of the Host Employee/Charitable Trust;
- (15) the contribution agreement relating to the Blackstone Acquisition, dated as of April 16, 1998, as amended; and
  - (16) any other documents as we deem necessary or appropriate.

The opinions set forth in this letter also are premised on certain written factual representations of Host and Host REIT contained in a letter to us dated as of this date, regarding the Merger, the OP Contribution and the Initial E&P Distribution (the "Representation Letter").

For purposes of rendering our opinions, we have not made an independent investigation or audit of the facts set forth in any of the above-referenced documents, including the Proxy Statement/Prospectus and the Representation Letter. We consequently have relied upon representations in the Representation Letter that the information presented in such documents or otherwise furnished to us is accurate and complete in all material respects. We are not aware, however, of any material facts or circumstances inconsistent with such representations or the assumptions set forth herein. In this regard, we have assumed with your consent the following:

(1) All of the representations and statements set forth in the documents that we reviewed in connection with the preparation of this opinion letter, including the Representation Letter (the "Reviewed Documents") are true and correct and will be true and correct at the time of the Merger, the OP Contribution, and/or the Initial E&P Distribution (as appropriate), and that any such representation or

statement made as a belief or made "to the knowledge of" or similarly qualified is correct and accurate without such qualification.

- (2) All of the information set forth in the Reviewed Documents is accurate and completely describes all material facts relevant to our opinion.
- (3) All of the obligations imposed by the Reviewed Documents on the parties thereto have been and will continue to be performed or satisfied in accordance with their terms.
- (4) All signatures to the Reviewed Documents are genuine, all documents submitted to us as originals are authentic, all documents submitted to us as copies conform to the originals, and all original documents from which any copies were made are authentic.
- (5) All of the Reviewed Documents have been (or will be by the Effective Date or by the time described in the Proxy Statement/Prospectus) duly executed and delivered, where due execution and delivery are prerequisites to the effectiveness thereof. As to Reviewed Documents which we have reviewed only a form, all such documents will be duly executed and delivered without material changes from the form reviewed by us.
- (6) The Merger will be consummated in accordance with the laws of the States of Maryland and Delaware and will qualify as a statutory merger under the laws of the States of Maryland and Delaware.
- (7) The Merger will be consummated in accordance with the Merger Agreement and as described in the Proxy Statement/Prospectus (including satisfaction of all covenants and conditions to the obligations of the parties without amendment or waiver thereof).
- (8) Host and Host REIT will comply with all reporting obligations with respect to the Merger required under the Code and the Treasury Regulations thereunder.
- (9) The Merger Agreement and all other documents and instruments referred to therein or in the Proxy Statement/Prospectus are valid and binding in accordance with their terms.
- (10) Host and Host REIT have been and will continue to be operated in the manner described in their organizational documents and in the Proxy Statement/Prospectus.

Any variation or difference in the facts from those set forth in the Reviewed Documents (including, in particular but without limitation of the foregoing, the Proxy Statement/Prospectus and the Representation Letter) may affect the conclusions stated herein. In addition, if any one of the statements, representations, warranties or assumptions upon which we have relied to issue this opinion letter is incorrect, our opinions might be adversely affected and may not be relied upon.

## OPINIONS

Based upon, subject to, and limited by the assumptions and qualifications set forth in this letter, and subject to the condition that the REIT Conversion be completed in the manner set forth in the Proxy Statement/Prospectus, we are of the opinion that for federal income tax purposes:

- 1. The Merger will qualify as a reorganization within the meaning of Section 368(a) of the Code.
- 2. No gain or loss will be recognized by Host or Host REIT solely by reason of the Merger.
- 3. Except with respect to holders of Host Common Stock who are nonresident alien individuals, foreign corporations, foreign partnerships or foreign estates or trusts and who at any time during the five-year period ending on the Effective Date owned more than 5% of the Host Common Stock ("Excluded Stockholders"), no gain or loss will be recognized by holders of Host Common Stock upon the conversion of such stock into Host REIT Common Stock.
- 4. Except with respect to Excluded Stockholders, the tax basis of shares of Host REIT Common Stock received by a holder of Host Common Stock pursuant to the Merger in the aggregate will be the same as the holder's adjusted tax basis in the shares of Host Common Stock being converted in the Merger.

- 5. Except with respect to Excluded Stockholders, the holding period of shares of Host REIT Common Stock received by a holder of Host Common Stock pursuant to the Merger will include the holder's holding period with respect to the shares of Host Common Stock being converted in the Merger, assuming that the Host Common Stock was held as a capital asset at the time of the Merger.
- 6. The discussion in the Proxy Statement/Prospectus under the heading "Federal Income Tax Consequences," to the extent it contains descriptions of applicable federal income tax law, is correct in all material respects.

\* \* \* \* \*

This opinion letter addresses only the specific federal income tax matters set forth above and does not address any other federal, state, local or foreign tax consequences that may result from the Merger or any other transaction undertaken in connection with the Merger or that may result from the REIT Conversion. In particular, but not in limitation of the foregoing, this opinion letter does not address (i) the tax consequences of the Merger as applied to any particular holder of Host Common Stock in light of his particular circumstances, such as dealers in securities, corporate shareholders subject to the alternative minimum tax, foreign persons, and holders of shares of Host Common Stock who acquired such shares upon exercise of stock options or in other compensatory transactions; (ii) the tax consequences to Host and its subsidiaries of the OP Contribution and the contributions by such subsidiaries to the Operating Partnership; or (iii) the tax consequences to Host of the distribution of Crestline common stock pursuant to the Initial E&P Distribution. With respect to the qualification of Host REIT as a REIT, we refer you to our opinion letter addressing this issue, which we have delivered to you and which can be found at Appendix D to the Proxy Statement/Prospectus and to the discussion in the Proxy Statement/Prospectus under the heading "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Merger.'

We assume no obligation to advise you of any changes in our opinion subsequent to the delivery of this opinion letter.

This opinion letter has been prepared for your use in connection with the Proxy Statement/Prospectus and the REIT Conversion and should not be quoted in whole or in part or otherwise be referred to, or filed with or furnished to any governmental agency or other person or entity, without the prior written consent of this firm. We do, however, consent to the references to this opinion letter and to Hogan & Hartson L.L.P. under the captions, "Legal Matters" and "Federal Income Tax Consequences," in the Proxy Statement/Prospectus and to the inclusion of this opinion letter as an appendix to the Proxy Statement/Prospectus. In giving this consent, we do not thereby admit that we are an "expert" within the meaning of the Securities Act of 1933.

Very truly yours,

/s/ Hogan & Hartson L.L.P.

Hogan & Hartson L.L.P.

HOGAN & HARTSON L.L.P.

COLUMBIA SOUARE

555 13TH STREET, N.W.

WASHINGTON, DC 20004

NOVEMBER 20, 1998

HMC Merger Corporation Host Marriott, L.P. 10400 Fernwood Road Bethesda, MD 20817

## Ladies and Gentlemen:

We have acted as tax counsel to HMC Merger Corporation, a Maryland corporation ("Host REIT"), Host Marriott Corporation, a Delaware corporation ("Host"), and Host Marriott, L.P., a Delaware limited partnership (the "Operating Partnership"), in connection with the following series of related transactions (which collectively are referred to as the "REIT Conversion"), each of which is described more fully in the Prospectus/Consent Solicitation Statement which is part of the Registration Statement filed with the Securities and Exchange Commission by the Operating Partnership and Host REIT on Form S-4 (File No. 333-55807) and which includes the Supplement for each Partnership (as defined in (i) below) attached thereto (the "Consent Solicitation") and in the Proxy Statement/Prospectus which is part of the Registration Statement filed with the Securities and Exchange Commission by Host REIT on Form S-4 (File No. 333-64793) (the "Prospectus"):

- (i) the contribution of the following assets by Host and its subsidiaries to the Operating Partnership, in exchange for a number of units of limited partnership interest ("OP Units") and units of general partnership interest of the Operating Partnership equal to the number of shares of Host common stock outstanding at the time of the REIT Conversion, preferred partnership interests in the Operating Partnership corresponding to any shares of Host preferred stock outstanding at the time of the REIT Conversion, and the assumption of certain liabilities of Host and its subsidiaries: (a) its wholly owned full-service hotel assets; (b) its interests in Atlanta  $\,$ Marriott Marquis II Limited Partnership, a Delaware limited partnership ("Atlanta Marquis"); Desert Springs Marriott Limited Partnership, a Delaware limited partnership ("Desert Springs"); Hanover Marriott Limited Partnership, a Delaware limited partnership ("Hanover"); Marriott Diversified American Hotels, L.P., a Delaware limited partnership ("MDAH"); Marriott Hotel Properties Limited Partnership, a Delaware limited partnership ("MHP"); Marriott Hotel Properties II Limited Partnership, a Delaware limited partnership ("MHP2"); Mutual Benefit Chicago Marriott Suite Hotel Partners, L.P., a Rhode Island limited partnership ("Chicago Suites"); and Potomac Hotel Limited Partnership, a Delaware limited partnership ("PHLP") (collectively, the "Partnerships"); (c) its interests in partnerships (other than the Partnerships) or limited liability companies that own one or more full-service hotels and are not wholly owned by Host or one of its subsidiaries (the "Private Partnerships" and together with the Partnerships, the "Hotel Partnerships"); and (d) certain other businesses and assets (excluding that portion of its shares of common stock of Crestline Capital Corporation, a Delaware corporation ("Crestline"), that Host will distribute to its existing shareholders, the common stock of Crestline that Host or Host REIT will distribute to the Blackstone Entities (as defined in (v) below) and the cash or other consideration that Host or Host REIT will distribute to Host's or Host REIT's shareholders, and certain other de minimis assets);
- (ii) the recently completed refinancing and amendment of the debt securities and certain credit facilities of Host;
- (iii) the proposed mergers of subsidiaries of the Operating Partnership (the "Merger Partnerships") into the Partnerships, in which mergers the Partnerships will be the surviving entities (the "Mergers");

- (iv) the acquisition (whether by merger or otherwise) by the Operating Partnership of certain Private Partnerships or interests therein;
- (v) the acquisition by the Operating Partnership of ownership of, or controlling interests in, twelve upscale and luxury full-service hotel properties (the "Blackstone Hotels") and certain other related assets (including a mortgage loan secured by an additional hotel) from The Blackstone Group and a series of funds controlled by Blackstone Real Estate Partners (collectively, the "Blackstone Entities") in exchange for the assumption or repayment of debt, OP Units and shares of capital stock of Crestline, and cash (the "Blackstone Acquisition");
- (vi) the creation and capitalization of the one or more taxable corporations in which the Operating Partnership will own 95% of the economic interest but no voting stock and which will hold various assets contributed by Host and its subsidiaries to the Operating Partnership (the "Non-Controlled Subsidiaries"), with all of the voting stock, representing 5% of the economic interest, to be owned by the Host Marriott Statutory Employee/Charitable Trust, the beneficiaries of which will be certain employees of Host REIT, and a designated public charity, and possibly other outside investors (the "Host Employee/Charitable Trust");
- (vii) the distribution by Host of Crestline common stock to Host's shareholders and by Host or Host REIT of cash or other consideration to Host's or Host REIT's shareholders;
  - (viii) the merger of Host into Host REIT;
- (ix) the leasing of the hotels in which the Operating Partnership has a direct or indirect interest (the "Hotels") to subsidiaries of Crestline; and  ${\sf C}$
- $(\mbox{\tt x})$  the related transactions described in the Consent Solicitation and the Prospectus and the other steps necessary or desirable to complete the REIT Conversion.

In connection with the REIT Conversion, we have been asked to provide you with the opinions on certain federal income tax matters set forth in this letter. Capitalized terms used in this letter and not otherwise defined herein have the meaning set forth in the Consent Solicitation.

## BASES FOR OPINIONS

The opinions set forth in this letter are based on relevant current provisions of the Internal Revenue Code of 1986, as amended (the "Code"), Treasury Regulations thereunder (including proposed and temporary Treasury Regulations), and interpretations of the foregoing as expressed in court decisions, applicable legislative history, and the administrative rulings and practices of the Internal Revenue Service (the "IRS"), including its practices and policies in issuing private letter rulings, which are not binding on the IRS except with respect to a taxpayer that receives such a ruling, all as of the date hereof. These provisions and interpretations are subject to change, which may or may not be retroactive in effect, that might result in material modifications of our opinions. Our opinions do not foreclose the possibility of a contrary determination by the IRS or a court of competent jurisdiction, or of a contrary position taken by the IRS or the Treasury Department in regulations or rulings issued in the future. In this regard, an opinion of counsel with respect to an issue merely represents counsel's best judgment with respect to the probable outcome on the merits with respect to such issue, is not binding on the IRS or the courts, and is not a guarantee that the IRS will not assert a contrary position with respect to such issue or that a court will not sustain such a position asserted by the IRS.

In rendering the following opinions, we have examined such statutes, regulations, records, agreements, certificates and other documents as we have considered necessary or appropriate as a basis for such opinions, including the following: (1) the Consent Solicitation and the Prospectus; (2) the form of Second Amended and Restated Agreement of Limited Partnership of the Operating Partnership, proposed to be entered into at or about the time of the Mergers; (3) the form of the Articles of Amendment and Restatement of Articles of Incorporation of Host REIT and the Bylaws of Host REIT dated September 28, 1998; (4) the form of Articles of Incorporation of Crestline and the form of Bylaws of Crestline; (5) the partnership agreements (or form thereof), each as

amended to the date hereof and as proposed to be amended in connection with the REIT Conversion, of each Partnership, each Merger Partnership and each Private Partnership that will remain in existence after the REIT Conversion; (6) the form of agreement of merger relating to the Mergers (including various exhibits thereto) by and among Host REIT, the Operating Partnership, a Partnership, and its corresponding Merger Partnership, as amended to the date hereof; (7) each contribution agreement (or form thereof) relating to the acquisition by the Operating Partnership of the non-Host interests in the Private Partnerships; (8) the contribution agreement relating to the Blackstone Acquisition, dated as of April 16, 1998, as amended, and the form of the contribution agreement relating to the acquisition by the Operating Partnership of Host's assets; (9) the form of lease pursuant to which the Operating Partnership, its subsidiaries and its controlled partnerships will lease virtually all of the Hotels to the Lessees and the term sheets regarding each such lease setting forth the rental provisions of each such lease (the "Leases"); (10) the form of organizational documents relating to the formation and capitalization of the Host Employee/Charitable Trust; and (11) any other documents as we deem necessary or appropriate. The opinions set forth in this letter also are premised on certain written factual representations of Host REIT, Host and the Operating Partnership contained in a letter to us dated as of this date, regarding the organization, ownership and operations (including the income, assets, businesses, liabilities, properties and accumulated undistributed earnings and profits) of Host REIT, the Operating Partnership, the Hotel Partnerships, the Subsidiary Partnerships, the Non-Controlled Subsidiaries, the Host Employee/Charitable Trust, and Crestline and the Lessees following the REIT Conversion (the "Representation Letter").

For purposes of rendering our opinions, we have not made an independent investigation or audit of the facts set forth in any of the above-referenced documents, including the Consent Solicitation, the Prospectus and the Representation Letter. We consequently have relied upon representations in the Representation Letter that the information presented in such documents or otherwise furnished to us is accurate and complete in all material respects. We are not aware, however, of any material facts or circumstances contrary to, or inconsistent with, the representations we have relied upon as described herein, or other assumptions set forth herein.

In this regard, we have assumed with your consent the following: (i) that all of the representations and statements set forth in the documents that we reviewed (including the Representation Letter) are true and correct and will be true and correct at the time of the Mergers, that any representation or statement made as a belief or made "to the knowledge of" or similarly qualified is correct and accurate without such qualification, and that all of the obligations imposed by any such documents on the parties thereto have been and will continue to be performed or satisfied in accordance with their terms; (ii) the genuineness of all signatures, the proper execution of all documents, the authenticity of all documents submitted to us as originals, the conformity to originals of documents submitted to us as copies, the authenticity of the originals from which any copies were made and that any documents as to which we have reviewed only a form will be duly executed at the time of the Mergers without changes from the form reviewed by us; (iii) that each of Host REIT, the Operating Partnership, the Hotel Partnerships, Crestline, the Non-Controlled Subsidiaries, the Host Employee/Charitable Trust, the Lessees, and the Subsidiary Partnerships have been (where applicable) and will continue to be operated in the manner described in the relevant partnership agreement, limited liability company operating agreement, articles of incorporation, or other organizational documents and in the Consent Solicitation and the Prospectus; (iv) that each of Host REIT, the Operating Partnership, the Hotel Partnerships, Crestline, the Non-Controlled Subsidiaries, the Host Employee/Charitable Trust, the Lessees, and the Subsidiary Partnerships will be duly incorporated or organized and validly existing under the laws of the state in which it was created at the time of the REIT Conversion; (v) as represented by Host REIT and the Operating Partnership, that each of the Leases will be enforced in accordance with its terms, and that each of the lessors and the Lessees will act at all times in accordance with the terms thereof; (vi) as represented by Host REIT and the Operating Partnership, that there will be no agreements or understandings between Host REIT or the Operating Partnership, on the one hand, and the Host Employee/Charitable Trust, which owns 100% of the voting stock of each Non-Controlled Subsidiary, or any of the Non-Controlled Subsidiaries themselves, on the other hand, that are inconsistent with the Host Employee/Charitable Trust being considered to be both the record and beneficial owner of more than 90% of the outstanding voting stock of each of the Non-Controlled Subsidiaries; and (vii) as represented by Host REIT and the Operating Partnership, no member of the Marriott family, or any entity in which any member of the Marriott family owns an interest, nor any other shareholder of Host REIT will own (determined by taking into account

the attribution rules under Section 318(a) of the Code, as modified by Section 856(d)(5) of the Code) at the time of the REIT Conversion or at any time thereafter more than 9.8% by value of Host REIT. Any variation or difference in the facts from those set forth in the documents that we have reviewed and upon which we have relied (including, in particular, the Consent Solicitation, the Prospectus and the Representation Letter) may adversely affect the conclusions stated herein.

#### OPTNTONS

Based upon, subject to, and limited by the assumptions and qualifications set forth herein (including those set forth below), and subject to the condition that the REIT Conversion be completed in the manner set forth in the Consent Solicitation and the Prospectus, we are of the opinion that:

- 1. Host REIT, beginning with its first taxable year commencing after consummation of the REIT Conversion, will be organized in conformity with the requirements for qualification as a REIT, and its proposed method of operation will enable it to meet the requirements for qualification and taxation as a REIT under the Code.
- 2. The Leases will be respected as leases for federal income tax purposes.
- 3. If the Operating Partnership were a "publicly traded partnership" within the meaning of Section 7704 of the Code because OP Units were readily tradable on the substantial equivalent of a secondary market after the Mergers and the REIT Conversion, it would qualify as a partnership for federal income tax purposes because, based upon factual representations made by Host, Host REIT and the Operating Partnership as to the proposed method of operation of the Operating Partnership after the Mergers and the REIT Conversion, at least ninety percent (90%) of its income will consist of "qualifying income," as defined in Section 7704(d) of the Code.

\* \* \* \* \*

Host REIT's ability to qualify as a REIT will depend in particular upon whether each of the Leases is respected as a lease for federal income tax purposes. If any one of such Leases is not respected as a lease for federal income tax purposes, the Company likely will fail to qualify as a REIT. The determination of whether a lease is a lease for federal income tax purposes is highly dependent on specific facts and circumstances. In addition, the rental provisions of the Leases and the other terms thereof must conform with normal business practice and not be used as a means to base the rent paid on the income or profits of the Lessees. In delivering the opinion set forth above that each of the Leases will be respected as a lease for federal income tax purposes, and the opinion set forth above that Host REIT's proposed method of operation (as described in the Representation Letter) should enable Host REIT to meet the requirements for qualification and taxation as a REIT for its first taxable year commencing following consummation of the REIT Conversion and subsequent taxable years, we expressly rely upon, among other things, Host REIT's representations as to various factual matters with respect to the Leases, including representations as to the commercial reasonableness of the economic and other terms of the Leases, the intent and economic expectations of the parties to the Leases, the allocation of various economic risks between the parties to the Leases, taking into account all surrounding facts and circumstances, the conformity of the rental provisions and other terms of the Leases with normal business practice, and the conclusion that such terms are not being used as a means to base the rent paid on the income or profits of the Lessees. We express no opinion as to any of the economic terms of the Leases, the commercial reasonableness thereof, or whether the actual economic relationships created thereby are such that the Leases will be respected for federal income tax purposes or whether the rental and other terms of the Leases conform with normal business practice (and are not being used as a means to base the rent paid on the income or profits of the Lessees).

Host REIT's ability to qualify as a REIT also will depend upon Host REIT not having at the close of its first taxable year for which its REIT election is effective any "earnings and profits" accumulated in any prior taxable year of Host REIT, Host, or any of its predecessors or subsidiaries (which would be based on the

consolidated earnings and profits of Host (including each of its predecessors) accumulated from 1929, the first year that the predecessor of Host was a "C" corporation, through and including Host's 1998 taxable year). The calculation "earnings and profits" depends upon a number of factual and legal interpretations related to the activities and operations of Host and its corporate affiliates during its entire corporate existence and is subject to review and challenge by the IRS. Host and Host REIT have represented to us for purposes of our opinion that Host REIT will have distributed by the close of its first taxable year for which its REIT election is effective any "earnings and profits" accumulated in any prior taxable year of Host REIT, Host, or any of its predecessors or subsidiaries. There can be no assurance, however, that the IRS will not examine the tax returns of Host and its affiliates for all years prior to and including the REIT Conversion and propose adjustments to increase their taxable income, which could result in Host REIT being considered to have undistributed "earnings and profits" at the close of its first taxable year for which its REIT election is effective, in which event Host REIT would not qualify as a REIT for such year. We express no opinion as to Host's current and accumulated "earnings and profits" or whether Host REIT will be considered to have undistributed "earnings and profits" at the close of its first taxable year for which its REIT election is effective.

Host REIT's qualification and taxation as a REIT depend upon Host REIT's ability to meet on an ongoing basis (through actual annual operating results, distribution levels, diversity of share ownership and otherwise) the various qualification tests imposed under the Code and described in the Consent Solicitation and the Prospectus. We have relied upon representations of Host REIT and the Operating Partnership with respect to these matters (including those set forth in the Representation Letter and in the Consent Solicitation and the Prospectus) and will not review Host REIT's compliance with these requirements on a continuing basis. Accordingly, no assurance can be given that the actual results of Host REIT's operations, the sources of its income, the nature of its assets, the level of its distributions to shareholders and the diversity of its share ownership for any given taxable year will satisfy the requirements under the Code for qualification and taxation as a REIT.

For a discussion relating the law to the facts, and the legal analysis underlying the opinions set forth in this letter, we incorporate by reference the discussions of federal income tax issues in the Section of the Consent Solicitation under the heading "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Mergers" and in the Section of the Prospectus under the heading "Federal Income Tax Consequences--Federal Income Taxation of Host REIT Following the Merger."

We assume no obligation to advise you of any changes in our opinion subsequent to the delivery of this opinion letter.

This opinion letter addresses only the specific federal income tax matters set forth above and does not address any other federal, state, local or foreign tax consequences that may result from the REIT Conversion or any other transaction undertaken in connection therewith. This opinion letter has been prepared for your use in connection with the Consent Solicitation, the Prospectus and the REIT Conversion and should not be quoted in whole or in part or otherwise be referred to, or filed with or furnished to any governmental agency or other person or entity, without the prior written consent of this firm. We do, however, consent to the references to this opinion letter and to Hogan & Hartson L.L.P. under the captions, "Legal Matters" and "Federal Income Tax Consequences," in the Consent Solicitation and the Prospectus (and under the caption, "Federal Income Tax Consequences," in the Supplements) and to the inclusion of (i) this opinion letter as an exhibit to the Prospectus and (ii) the form of this opinion letter as an exhibit to the Consent Solicitation. In giving this consent, we do not thereby admit that we are an "expert" within the meaning of the Securities Act of 1933.

Very truly yours,

/s/ Hogan & Hartson L.L.P.

Hogan & Hartson L.L.P.

ITEM 20. INDEMNIFICATION OF DIRECTORS AND OFFICERS.

The Host REIT Charter authorizes it, to the maximum extent permitted by Maryland law, to obligate itself to indemnify and to pay or reimburse reasonable expenses in advance of final disposition of a proceeding to (i) any present or former director or officer or (ii) any individual who, while a director of Host REIT and at the request of Host REIT, serves or has served another corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or any other enterprise from and against any claim or liability to which such person may become subject or which such person may incur by reason of his or her status as a present or former Director or officer of Host REIT. The Host REIT Bylaws obligate it, to the maximum extent permitted by Maryland law, to indemnify and to pay or reimburse reasonable expenses in advance of final disposition of a proceeding to (a) any present or former director or officer who is made party to the proceeding by reason of his service in that capacity or (b) any individual who, while a director or officer of Host REIT and at the request of Host REIT, serves or has served another corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or any other enterprise as a director, trustee, officer or partner of such corporation, real estate investment trust, partnership, joint venture, trust, employee benefit plan or other enterprise and who is made a party to the proceeding by reason of his service in that capacity, against any claim or liability to which he may become subject by reason of such status. The Host REIT Charter and Host REIT Bylaws also permit Host REIT to indemnify and advance expenses to any person who served a predecessor of Host REIT in any of the capacities described above and to any employee or agent of Host REIT or a predecessor of Host REIT. The Host REIT Bylaws require Host REIT to indemnify a director or officer who has been successful, on the merits or otherwise, in the defense of any proceeding to which he is made a party by reason of his service in that capacity.

The MGCL permits a Maryland corporation to indemnify and advance expenses to its directors, officers, employees and agents, and permits a corporation to indemnify its present and former directors and officers, among others, against judgments, penalties, fines, settlements and reasonable expenses actually incurred by them in connection with any proceeding to which they may be made a party by reason of their service in those or other capacities unless it is established that (a) the act or omission of the director or officer was material to the matter giving rise to the proceeding and (i) was committed in bad faith or (ii) was the result of active and deliberate dishonesty, (b) the director or officer actually received an improper personal benefit in money, property or services or (c) in the case of any criminal proceeding, the director or officer had reasonable cause to believe that the act or omission was unlawful. However, under the MGCL, a Maryland corporation may not indemnify a director or officer in a suit by or in the right of the corporation if such director or officer has been adjudged to be liable to the corporation. In accordance with the MGCL, the Host REIT Bylaws require it, as a condition to advancing expenses, to obtain (1) a written affirmation by the director or officer of his good faith belief that he has met the standard of conduct necessary for indemnification by Host REIT as authorized by the Host REIT Bylaws and (2) a written statement by or on his behalf to repay the amount paid or reimbursed by Host REIT if it shall ultimately be determined that the standard of conduct was not met.

Host REIT intends to enter into indemnification agreements with each of its directors and officers. The indemnification agreements will require, among other things, that Host REIT indemnify its directors and officers to the fullest extent permitted by law and advance to its directors or officers all related expenses, subject to reimbursement if it is subsequently determined that indemnification is not permitted.

The Partnership Agreement also provides for indemnification of Host REIT and its officers and directors to the same extent that indemnification is provided to officers and directors of Host REIT in the Host REIT Charter, and limits the liability of Host REIT and its officers and directors to the Operating Partnership and its respective partners to the same extent that the liability of the officers and directors of Host REIT to Host REIT and its stockholders is limited under the Host REIT Charter.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers or persons controlling the registrant pursuant to the foregoing provisions, Host REIT has been informed that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Securities Act and is therefore unenforceable.

## FINANCIAL STATEMENT SCHEDULES

	PAGE
Report of Independent Public Accountants on Financial Statement Schedules	

## EXHIBITS

-----

- 2.1 Form of Agreement and Plan of Merger dated November , 1998 (the "Merger Agreement") by and among Host Marriott Corporation, HMC Merger Corporation and Host Marriott L.P. (included as Appendix A to the Proxy Statement/Prospectus that is part of this Registration Statement).
- 3.1 Articles of Incorporation of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) dated September 28, 1998 (incorporated by reference to Exhibit 3.12 of HMC Merger Corporation Registration Statement No. 333-55807).
- 3.2 Articles of Amendment of Articles of Incorporation of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) dated November 19, 1998.
   3.3 Bylaws of HMC Merger Corporation (to be renamed Host Marriott
- 3.3 Bylaws of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) dated September 28, 1998 (included as Exhibit C to the Merger Agreement which is attached as Appendix A to the Proxy Statement/Prospectus that is part of this Registration Statement).
- 3.4 Form of Articles of Amendment and Restatement of Articles of Incorporation of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) (included as Exhibit B to the Merger Agreement which is attached as Appendix A to the Proxy Statement/Prospectus that is part of this Registration Statement).
- 4.1 Form of Common Stock Certificate of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) (incorporated by reference to Exhibit 4.7 of HMC Merger Corporation Registration Statement No. 333-55807).
- 4.2 Guarantee Agreement, dated December 2, 1996, between Host Marriott Corporation and IBJ Schroeder Bank & Trust Company, as Guarantee Trustee (incorporated by reference to Exhibit 4.6 of Host Marriott Corporation Registration Statement No. 333-19923).
- 4.3(i) Rights Agreement between Marriott Corporation and The Bank of New York as Rights Agent dated February 3, 1989 (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-62444).
- 4.3(ii) First Amendment to Rights Agreement between Marriott Corporation and Bank of New York as Rights Agent dated as of October 8, 1993 (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-51707).
- 4.4 Indenture by and among HMC Acquisition Properties, Inc., as Issuer, HMC SFO, Inc., as Subsidiary Guarantors, and Marine Midland Bank, as Trustee (incorporated by reference to Host Marriott Corporation Registration Statement No. 333-00768).
   4.5 Indenture by and among HMH Properties, Inc., as Issuer, HMH Courtyard
- 4.5 Indenture by and among HMH Properties, Inc., as Issuer, HMH Courtyard Properties, Inc., HMC Retirement Properties, Inc., Marriott Financial Services, Inc., Marriott SBM Two Corporation, HMH Pentagon Corporation and Host Airport Hotels, Inc., as Subsidiary Guarantors, and Marine Midland Bank, as Trustee (incorporated by reference to Host Marriott Corporation Registration Statement 33-95058).
- 4.6 Indenture by and among HMH Properties, Inc., as Issuer, and the Subsidiary Guarantors named therein, and Marine Midland Bank, as Trustee (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated August 6, 1998).
- 4.7 Indenture for the 6 3/4% Convertible Debentures, dated December 2, 1996, between Host Marriott Corporation and IBJ Schroeder Bank & Trust Company, as Indenture Trustee (incorporated by reference to Exhibit 4.3 of Host Marriott Corporation Registration Statement No. 333-19923).

#### **EXHIBITS**

\_\_\_\_

- Amended and Restated Trust Agreement, dated December 2, 1996, among Host Marriott Corporation, IBJ Schroeder Bank & Trust Company, as Property Trustee, Delaware Trust Capital Management, Inc., as Delaware Trustee, and Robert E. Parsons, Jr., Bruce D. Wardinski and Christopher G. Townsend, as Administrative Trustees (incorporated by reference to Exhibit 4.2 of Host Marriott Corporation Registration Statement No. 333-19923).
- 5.1 Opinion of Hogan & Hartson L.L.P. regarding legality of the securities being registered.
- 8.1 Opinion of Hogan & Hartson L.L.P. regarding certain tax matters (included as Appendix C to the Proxy Statement/Prospectus that is part of this Registration Statement).
- 8.2 Opinion of Hogan & Hartson L.L.P. regarding the qualification of Host REIT as a REIT (included as Appendix D to the Proxy Statement/Prospectus that is part of this Registration Statement).
- 10.1 Form of Second Amended and Restated Agreement of Limited Partnership of Host Marriott, L.P. (incorporated by reference to Exhibit 3.1 of HMC Merger Corporation Registration Statement No. 333-55807).
- 10.2 Form of Indenture between Host Marriott L.P., as Issuer, and Marine Midland Bank, as Indenture Trustee, and Form of 6.56% Callable Note due December 15, 2005 (incorporated by reference to Exhibit 4.1 of HMC Merger Corporation Registration Statement No. 333-55807).
- Amended and Restated Credit Agreement dated as of June 19, 1997 and Amended and Restated as of August 5, 1998 among Host Marriott Corporation, Host Marriott Hospitality, Inc., HMH Properties, Inc., Host Marriott, L.P., HMC Capital Resources Corp., Various Banks, Wells Fargo Bank, National Association, The Bank of Nova Scotia and Credit Lyonnais New York Branch, as Co-Arrangers, and Bankers Trust Company as Arranger and Administrative Agent (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated September 11, 1998).
- Marriott Corporation Executive Deferred Compensation Plan dated as of December 6, 1990 (incorporated by reference to Exhibit 19(i) of the Host Marriott Corporation Annual Report on Form 10-K for the fiscal year ended December 28, 1991).
- 10.5 Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan effective as of October 8, 1993 (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- 10.6 Distribution Agreement dated as of September 15, 1993 between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- Amendment No. 1 to the Distribution Agreement dated December 29, 1995 by and among Host Marriott Corporation, Host Marriott Services Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated January 16, 1996).
- 10.8 Amendment No. 2 to the Distribution Agreement dated June 21, 1997 by and among Host Marriott Corporation, Host Marriott Services Corporation and Marriott International, Inc.
- 10.9 Amendment No. 3 to the Distribution Agreement dated March 3, 1998 by and among Host Marriott Corporation, Host Marriott Services Corporation and Marriott International, Inc.
- 10.10 Form of Amendment No. 4 to the Distribution Agreement by and among Host Marriott Corporation and Marriott International, Inc.
- 10.11 Distribution Agreement dated December 22, 1995 by and between Host Marriott Corporation and Host Marriott Services Corporation (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated January 16, 1996).
- Tax Sharing Agreement dated as of October 5, 1993 by and between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- 10.13 Assignment and License Agreement dated as of October 8, 1993 by and between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).

- 10.14 Amendment No. 1 to the Assignment and License Agreement dated as of October 8, 1993 by and between Marriott International, Inc. and Host Marriott Corporation (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated January 16, 1996).
- 10.15 Form of Noncompetition Agreement between Host Marriott Corporation, Host Marriott, L.P. and Crestline Capital Corporation and other parties named therein.
- Tax Administration Agreement dated as of October 8, 1993 by and between Marriott 10.16 Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- Restated Noncompetition Agreement dated March , 1998 by and among Host Marriott Corporation, Marriott International, Inc. and Sodexho Marriott Services, Inc. Form of First Amendment to Restated Noncompetition Agreement by and among Host 10.17
- 10.18 Marriott Corporation, Marriott International, Inc. and Sodexho Marriott Services,
- 10.19 Host Marriott Lodging Management Agreement--Marriott Hotels, Resorts and Hotels dated September 25, 1993 by and between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-51707).
- Employee Benefits and Other Employment Matters Allocation Agreement dated as of December 29, 1995 by and between Host Marriott Corporation and Host Marriott 10.20 Services Corporation (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated January 16, 1996).
- Tax Sharing Agreement dated as of December 29, 1995 by and between Host Marriott 10.21 Corporation and Host Marriott Services Corporation (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated January 16, 1996).
- Marriott/Host Marriott Employees' Profit Sharing Retirement and Savings Plan and Trust (incorporated by reference to Host Marriott Corporation Registration 10.22 Statement No. 33-62444).
- Contribution Agreement dated as of April 16, 1998 among Host Marriott Corporation, 10.23 Host Marriott, L.P. and the contributors named therein, together with Exhibit B (incorporated by reference to Exhibit 10.18 of HMC Merger Corporation Registration Statement No. 333-55807).
- 10.24 Amendment No. 1 to Contribution Agreement dated May 8, 1998 among Marriott Corporation, Host Marriott, L.P. and the contributors named therein (incorporated by reference to Exhibit 10.19 of HMC Merger Corporation Registration Statement No. 333-55807).
- Amendment No. 2 to Contribution Agreement dated May 18, 1998 among Host Marriott 10.25 Corporation, Host Marriott, L.P. and the contributors named therein (incorporated by reference to Exhibit 10.20 of HMC Merger Corporation Registration Statement No. 333-55807).
- #10.26 Form of Lease Agreement.
- #10.27 Form of Management Agreement for Full-Service Hotels (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-51707)
- 10.28\*\* Form of Employee Benefits and Other Employment Matters Allocation Agreement between Host Marriott Corporation, Host Marriott, L.P. and Crestline Capital Corporation. Form of Pool Guarantee Agreement between HMC Merger Corporation, the lessees
- 10.29
  - referred to therein and Crestline Capital Corporation.
- 10.30 Form of Pooling and Security Agreement by and among HMC Merger Corporation and Crestline Capital Corporation.
- 10.31 Form of Amended and Restated Communities Noncompetition Agreement.
- 21.1\*\* List of Subsidiaries of HMC Merger Corporation.
- Consent of Hogan & Hartson L.L.P. (included in Exhibits 5.1 and 8.1). 23.1
- Consent of Arthur Andersen LLP. 23.2
- Consents of Director nominees of HMC Merger Corporation (to be renamed Host 23.3 Marriott Corporation in connection with the REIT Conversion).
- Consent of BT Wolfensohn. 23.4
- 27.1\*\* Financial Data Schedule.
- 99.1\*\* Form of Proxy Card.
- Form of Opinion of BT Wolfensohn (included as Appendix B to the Proxy 99.2 Statement/Prospectus that is part of this Registration Statement).

<sup>\*\*</sup> Previously filed.

Agreement filed is illustrative of numerous other agreements to which the Company is a party.

## ITEM 22. UNDERTAKINGS.

The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of any employee benefit plan's annual report pursuant to Section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in this Registration Statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof

The undersigned registrant hereby undertakes as follows: that prior to any public reoffering of the securities registered hereunder through use of a prospectus that is a part of this registration statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the issuer undertakes that such offering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other items of the applicable form.

The undersigned registrant undertakes that every prospectus (i) that is filed pursuant to paragraph (c) immediately preceding, or (ii) that purports to meet the requirements of Section 10(a)(3) of the Act and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to the registration statement and will not be used until such amendment is effective, and that, for purposes of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

Insofar as indemnification for liabilities arising under the Securities Act of 1933, as amended, may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Securities Act of 1933 and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act of 1933 and will be governed by the final adjudication of such issue.

The undersigned registrant hereby undertakes to respond to requests for information that is incorporated by reference into the prospectus pursuant to Item 4, 10(b), 11 or 13 of this form, within one business day of receipt of such request, and to send the incorporated documents by first class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the registration statements through the date of responding to the request.

The undersigned registrant hereby undertakes to supply by means of a post-effective amendment all information concerning a transaction, and being acquired involved therein, that was not the subject of and included in the registration statement when it became effective.

## SIGNATURES

PURSUANT TO THE REQUIREMENTS OF THE SECURITIES ACT, THE REGISTRANT HAS DULY CAUSED THIS AMENDMENT NO. 3 TO THE REGISTRATION STATEMENT TO BE SIGNED ON ITS BEHALF BY THE UNDERSIGNED, THEREUNTO DULY AUTHORIZED, IN THE CITY OF BETHESDA, STATE OF MARYLAND ON NOVEMBER 20, 1998.

HMC Merger Corporation

PURSUANT TO THE REQUIREMENTS OF THE SECURITIES ACT, THIS REGISTRATION STATEMENT HAS BEEN SIGNED BELOW BY THE FOLLOWING PERSONS IN THE CAPACITIES AND ON THE DATES INDICATED.

SIGNATURE	TITLE 	DATE 
/s/ Robert E. Parsons, Jr.  ROBERT E. PARSONS, JR.	President and Initial Director (Chief Executive Officer and Chief	November 20, 1998
/s/ Donald D. Olinger	Financial Officer)  Vice President (Chief Accounting Officer)	November 20, 1998
DONALD D. OLINGER  /s/ Christopher G. Townsend	Vice President and Initial _ Director	November 20, 1998
CHRISTOPHER G. TOWNSEND		

## EXHIBIT INDEX

- 2.1 Form of Agreement and Plan of Merger dated November , 1998 (the "Merger Agreement") by and among Host Marriott Corporation, HMC Merger Corporation and Host Marriott L.P. (included as Appendix A to the Proxy Statement/Prospectus that is part of this Registration Statement).
- 3.1 Articles of Incorporation of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) dated September 28, 1998 (incorporated by reference to Exhibit 3.12 of HMC Merger Corporation Registration Statement No. 333-55807).
- 3.2 Articles of Amendment of Articles of Incorporation of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) dated November 19, 1998.
- 3.3 Bylaws of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) dated September 28, 1998 (included as Exhibit C to the Merger Agreement which is attached as Appendix A to the Proxy Statement/Prospectus that is part of this Registration Statement).
- 3.4 Form of Articles of Amendment and Restatement of Articles of Incorporation of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) (included as Exhibit B to the Merger Agreement which is attached as Appendix A to the Proxy Statement/Prospectus that is part of this Registration Statement).
- 4.1 Form of Common Stock Certificate of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion) (incorporated by reference to Exhibit 4.7 of HMC Merger Corporation Registration Statement No. 333-55807).
- 4.2 Guarantee Agreement, dated December 2, 1996, between Host Marriott Corporation and IBJ Schroeder Bank & Trust Company, as Guarantee Trustee (incorporated by reference to Exhibit 4.6 of Host Marriott Corporation Registration Statement No. 333-19923).
- 4.3(i) Rights Agreement between Marriott Corporation and The Bank of New York as Rights Agent dated February 3, 1989 (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-62444).
- 4.3(ii) First Amendment to Rights Agreement between Marriott Corporation and Bank of New York as Rights Agent dated as of October 8, 1993 (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-51707).
- Indenture by and among HMC Acquisition Properties, Inc., as Issuer, HMC SFO, Inc., as Subsidiary Guarantors, and Marine Midland Bank, as Trustee (incorporated by reference to Host Marriott Corporation Registration Statement No. 333-00768).
- Indenture by and among HMH Properties, Inc., as Issuer, HMH Courtyard Properties, Inc., HMC Retirement Properties, Inc., Marriott Financial Services, Inc., Marriott SBM Two Corporation, HMH Pentagon Corporation and Host Airport Hotels, Inc., as Subsidiary Guarantors, and Marine Midland Bank, as Trustee (incorporated by reference to Host Marriott Corporation Registration Statement 33-95058).
- 4.6 Indenture by and among HMH Properties, Inc., as Issuer, and the Subsidiary Guarantors named therein, and Marine Midland Bank, as Trustee (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated August 6, 1998).
- 4.7 Indenture for the 6 3/4% Convertible Debentures, dated December 2, 1996, between Host Marriott Corporation and IBJ Schroeder Bank & Trust Company, as Indenture Trustee (incorporated by reference to Exhibit 4.3 of Host Marriott Corporation Registration Statement No. 333-19923).
- 4.8 Amended and Restated Trust Agreement, dated December 2, 1996, among Host Marriott Corporation, IBJ Schroeder Bank & Trust Company, as Property Trustee, Delaware Trust Capital Management, Inc., as Delaware Trustee, and Robert E. Parsons, Jr., Bruce D. Wardinski and Christopher G. Townsend, as Administrative Trustees (incorporated by reference to Exhibit 4.2 of Host Marriott Corporation Registration Statement No. 333-19923).

- 5.1 Opinion of Hogan & Hartson L.L.P. regarding legality of the securities being registered.
- 8.1 Opinion of Hogan & Hartson L.L.P. regarding certain tax matters (included as Appendix C to the Proxy Statement/Prospectus that is part of the Registration Statement).
- 8.2 Opinion of Hogan & Hartson L.L.P. regarding the qualification of Host REIT as a REIT (included as Appendix D to the Proxy Statement/Prospectus that is part of the Registration Statement).
- 10.1 Form of Second Amended and Restated Agreement of Limited Partnership of Host Marriott, L.P. (incorporated by reference to Exhibit 3.1 of HMC Merger Corporation Registration Statement No. 333-55807).
- 10.2 Form of Indenture between Host Marriott L.P., as Issuer, and Marine Midland Bank, as Indenture Trustee, and Form of 6.56% Callable Note due December 15, 2005 (incorporated by reference to Exhibit 4.1 of HMC Merger Corporation Registration Statement No. 333-55807).
- Amended and Restated Credit Agreement dated as of June 19, 1997 and Amended and Restated as of August 5, 1998 among Host Marriott Corporation, Host Marriott Hospitality, Inc., HMH Properties, Inc., Host Marriott, L.P., HMC Capital Resources Corp., Various Banks, Wells Fargo Bank, National Association, The Bank of Nova Scotia and Credit Lyonnais New York Branch, as Co-Arrangers, and Bankers Trust Company as Arranger and Administrative Agent (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated September 11, 1998).
- 10.4 Marriott Corporation Executive Deferred Compensation Plan dated as of December 6, 1990 (incorporated by reference to Exhibit 19(i) of the Host Marriott Corporation Annual Report on Form 10-K for the fiscal year ended December 28, 1991).
- 10.5 Host Marriott Corporation 1993 Comprehensive Stock Incentive Plan effective as of October 8, 1993 (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- 10.6 Distribution Agreement dated as of September 15, 1993 between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- 10.7 Amendment No. 1 to the Distribution Agreement dated December 29, 1995 by and among Host Marriott Corporation, Host Marriott Services Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated January 16, 1996).
- 10.8 Amendment No. 2 to the Distribution Agreement dated June 21, 1997 by and among Host Marriott Corporation, Host Marriott Services Corporation and Marriott International, Inc.
- 10.9 Amendment No. 3 to the Distribution Agreement dated March 3, 1998 by and among Host Marriott Corporation, Host Marriott Services Corporation and Marriott International, Inc.
- 10.10 Form of Amendment No. 4 to the Distribution Agreement by and among Host Marriott Corporation and Marriott International, Inc.
- 10.11 Distribution Agreement dated December 22, 1995 by and between Host
  Marriott Corporation and Host Marriott Services Corporation
  (incorporated by reference to Host Marriott Corporation Current Report
  on Form 8-K dated January 16, 1996).
- 10.12 Tax Sharing Agreement dated as of October 5, 1993 by and between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- 10.13 Assignment and License Agreement dated as of October 8, 1993 by and between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- 10.14 Amendment No. 1 to the Assignment and License Agreement dated as of October 8, 1993 by and between Marriott International, Inc. and Host Marriott Corporation (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated January 16, 1996).

- 10.15 Form of Noncompetition Agreement between Host Marriott Corporation, Host Marriott, L.P. and Crestline Capital Corporation and other parties named therein.
- 10.16 Tax Administration Agreement dated as of October 8, 1993 by and between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated October 23, 1993).
- 10.17 Restated Noncompetition Agreement dated March , 1998 by and among Host Marriott Corporation, Marriott International, Inc. and Sodexho Marriott Services, Inc.
- 10.18 Form of First Amendment to Restated Noncompetition Agreement by and among Host Marriott Corporation, Marriott International, Inc. and Sodexho Marriott Services, Inc.
- 10.19 Host Marriott Lodging Management Agreement--Marriott Hotels, Resorts and Hotels dated September 25, 1993 by and between Marriott Corporation and Marriott International, Inc. (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-51707).
- 10.20 Employee Benefits and Other Employment Matters Allocation Agreement dated as of December 29, 1995 by and between Host Marriott Corporation and Host Marriott Services Corporation (incorporated by reference to Current Report on Host Marriott Corporation Form 8-K dated January 16, 1996).
- 10.21 Tax Sharing Agreement dated as of December 29, 1995 by and between Host Marriott Corporation and Host Marriott Services Corporation (incorporated by reference to Host Marriott Corporation Current Report on Form 8-K dated January 16, 1996).
- 10.22 Marriott/Host Marriott Employees' Profit Sharing Retirement and Savings Plan and Trust (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-62444).
- 10.23 Contribution Agreement dated as of April 16, 1998 among Host Marriott Corporation, Host Marriott, L.P. and the contributors named therein, together with Exhibit B (incorporated by reference to Exhibit 10.18 of HMC Merger Corporation Registration Statement No. 333-55807).
- 10.24 Amendment No. 1 to Contribution Agreement dated May 8, 1998 among Marriott Corporation, Host Marriott, L.P. and the contributors named therein (incorporated by reference to Exhibit 10.19 of HMC Merger Corporation Registration Statement No. 333-55807).
- 10.25 Amendment No. 2 to Contribution Agreement dated May 18, 1998 among Host Marriott Corporation, Host Marriott, L.P. and the contributors named therein (incorporated by reference to Exhibit 10.20 of HMC Merger Corporation Registration Statement No. 333-55807).
- #10.26 Form of Lease Agreement.
- #10.27 Form of Management Agreement for Full-Service Hotels (incorporated by reference to Host Marriott Corporation Registration Statement No. 33-51707).
- 10.28\*\* Form of Employee Benefits and Other Employment Matters Allocation Agreement between Host Marriott Corporation, Host Marriott, L.P. and Crestline Capital Corporation.
- 10.29 Form of Pool Guarantee Agreement between HMC Merger Corporation, the lessees referred to therein and Crestline Capital Corporation.
- 10.30 Form of Pooling and Security Agreement by and among HMC Merger Corporation and Crestline Capital Corporation.
- 10.31 Form of Amended and Restated Communities Noncompetition Agreement.
- 21.1\*\* List of Subsidiaries of HMC Merger Corporation.
- 23.1 Consent of Hogan & Hartson L.L.P. (included in Exhibits 5.1 and 8.1).
- 23.2 Consent of Arthur Andersen LLP.

- Consents of Director nominees of HMC Merger Corporation (to be renamed Host Marriott Corporation in connection with the REIT Conversion). 23.3
- 23.4 Consent of BT Wolfensohn.
- 27.1\*\* Financial Data Schedule.
- 99.1\*\* Form of Proxy Card.
- 99.2 Form of Opinion of BT Wolfensohn (included as Appendix B to the Proxy Statement/Prospectus that is part of this Registration Statement).

<sup>\*\*</sup> Previously filed.

# Agreement filed is illustrative of numerous other agreements to which the Company is a party.

ARTICLES OF AMENDMENT
OF
ARTICLES OF INCORPORATION
OF
HMC MERGER CORPORATION

 $\,$  HMC Merger Corporation, a Maryland corporation having its principal office in Bethesda, Maryland (the "Corporation"):

#### DOES HEREBY CERTIFY THAT:

 $\mbox{\it FIRST:}\ \mbox{\it The Corporation}$  desires to amend its Articles of Incorporation as currently in effect.

 ${\tt SECOND:}\ \ \, {\tt The}\ \, {\tt amendments}\ \, {\tt to}\ \, {\tt the}\ \, {\tt Articles}\ \, {\tt of}\ \, {\tt Incorporation}\ \, {\tt of}\ \, {\tt the}\ \, {\tt Corporation}\ \, {\tt are}\ \, {\tt as}\ \, {\tt follows:}$ 

Article FIFTH of the Articles of Incorporation of the Corporation shall be amended by deleting such Article FIFTH and replacing it with the following so that, as amended, Article FIFTH shall read in its entirety as follows:

"FIFTH: Section 5(a) Shares and Par Value. The total number of

shares of stock of all classes ("Capital Stock") which the Corporation has authority to issue is 800,000,000 shares, 750,000,000 of which initially are classified as common stock, par value of \$.01 per share ("Common Stock"), and 50,000,000 of which initially are classified as preferred stock, par value \$.01 per share ("Preferred Stock"). The aggregate par value of all classes of stock that the Corporation shall have authority to issue is \$8,000,000. The Board of Directors may, by adopting a resolution and filing articles supplementary with the State Department of Assessments and Taxation of Maryland, classify and reclassify any unissued shares of Capital Stock by setting or changing in any one or more respects the preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends, qualifications, or terms or conditions of redemption of such shares of stock. The power of the Board of Directors under this Section 5(a) to classify and reclassify any of the shares of Capital Stock shall include, without limitation, authority to classify or reclassify any unissued shares of such stock (including shares initially designated as Common Stock or Preferred Stock above)

into Common Stock, Preferred Stock, a class or classes of preferred stock, preference stock, special stock or other stock (including non-voting common stock), and to divide and classify shares of any class into one or more series of such class. Unless otherwise specifically provided for in the terms of any class or series of stock now or hereafter created, the amount that would be needed, if the Corporation were to be dissolved at the time of a distribution, to satisfy the preferential rights on dissolution of stockholders whose preferential rights are superior to those receiving the distribution, shall not limit the ability of the Corporation to make any distribution or the amount thereof.

Section 5(b) Common Stock. The following is a description of

the preferences, conversion and other rights, voting powers, restrictions, limitations as to dividends, qualifications, and terms and conditions of redemption of the Common Stock of the Corporation:

- (1) Voting Rights. Each share of Common Stock shall have one vote on all actions to be taken by the stockholders of the Corporation, and, except as otherwise provided in respect of any class of stock at any time classified or reclassified, the exclusive voting power for all purposes shall be vested in the holders of the Common Stock.
- preferences of any class of Capital Stock, including any shares of Preferred Stock, hereafter classified or reclassified, dividends, including dividends payable in shares of another class of the Corporation's stock, may be paid on the Common Stock of the Corporation at such time and in such amounts as the Board of Directors may deem advisable and the holders of the Common Stock shall share ratably in any such dividends, in proportion to the number of shares of Common Stock held by them respectively, on a share for share basis.
- (3) Liquidation Rights. In the event of any liquidation,
  dissolution or winding up of the Corporation, whether voluntary or
  involuntary, the holders of the Common Stock shall be entitled, after
  payment or provision for payment of the debts and other liabilities of the
  Corporation and the amount to which the holders of any class of Capital
  Stock at any time classified or reclassified having a preference on
  distributions in the liquidation, dissolution or winding up of the
  Corporation are entitled, including any shares of Preferred Stock, together
  with the holders of any other class of Capital Stock hereafter

classified or reclassified not having a preference on distributions in the liquidation, dissolution or winding up of the Corporation, to share ratably in the remaining net assets of the Corporation.

Section 5(c) Preferred Stock. The Board of Directors shall have

the authority to classify and reclassify any unissued shares of Preferred Stock from time to time by setting or changing in any one or more respects the preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends, qualifications, or terms or conditions of redemption of such shares of the Preferred Stock. Subject to the foregoing, the power of the Board of Directors to classify and reclassify any of the shares of Preferred Stock shall include, without limitation, subject to the provisions of the charter, authority to classify or reclassify any of the shares of such stock into Common Stock, a class or classes of preferred stock, preference stock, special stock or other stock, and to divide and classify shares of any class into one or more series of such class, by determining, fixing, or altering one or more of the following:

- (1) The distinctive designation of such class or series and the number of shares to constitute such class or series; provided that, unless otherwise prohibited by the terms of such or any other class or series, the number of shares of any class or series may be decreased by the Board of Directors in connection with any classification or reclassification of unissued shares and the number of shares of such class or series may be increased by the Board of Directors in connection with any such classification or reclassification, and any shares of any class or series which have been redeemed, purchased, otherwise acquired or converted into shares of Common Stock or any other class or series shall become part of the authorized class of stock so redeemed, purchased, otherwise acquired or converted into shares of Common Stock and be subject to classification and reclassification as provided in this Article FIFTH.
- (2) Whether or not and, if so, the rates, amounts and times at which, and the conditions under which, dividends shall be payable on shares of such class or series, whether any such dividends shall rank senior or junior to or on a parity with the dividends payable on any other class or series of stock, and the status of any such dividends as cumulative, cumulative to a limited extent or non- cumulative and as participating or non-participating.
- (3) Whether or not shares of such class or series shall have voting rights, in addition to any voting rights provided by law and, if so, the terms of such voting rights.

- (4) Whether or not shares of such class or series shall have conversion or exchange privileges and, if so, the terms and conditions thereof, including provision for adjustment of the conversion or exchange rate in such events or at such times as the Board of Directors shall determine.
- (5) Whether or not shares of such class or series shall be subject to redemption and, if so, the terms and conditions of such redemption, including the date or dates upon or after which they shall be redeemable and the amount per share payable in case of redemption, which amount may vary under different conditions and at different redemption dates; and whether or not there shall be any sinking fund or purchase account in respect thereof, and if so, the terms thereof.
- (6) The rights of the holders of shares of such class or series upon the liquidation, dissolution or winding up of the affairs of, or upon any distribution of the assets of, the Corporation, which rights may vary depending upon whether such liquidation, dissolution or winding up is voluntary or involuntary and, if voluntary, may vary at different dates, and whether such rights shall rank senior or junior to or on a parity with such rights of any other class or series of stock.
- (7) Whether or not there shall be any limitations applicable, while shares of such class or series are outstanding, upon the payment of dividends or making of distributions on, or the acquisition of, or the use of moneys for purchase or redemption of, any stock of the Corporation, or upon any other action of the Corporation, including action under this Section 5(c), and, if so, the terms and conditions thereof.
- (8) Any other preferences, rights, restrictions, including restrictions on transferability, and qualifications of shares of such class or series, not inconsistent with law and the charter of the Corporation.

Section 5(d) Ranking of Classes or Series of Capital Stock. For the purposes hereof and of any articles supplementary to the charter providing for the classification or reclassification of any shares of Capital Stock or of any other charter document of the Corporation (unless otherwise provided in any such articles or document), any class or series of stock of the Corporation shall be deemed to rank:

(1) prior to another class or series either as to dividends or upon liquidation, if the holders of such class or series  ${\bf r}$ 

shall be entitled to the receipt of dividends or of amounts distributable on liquidation, dissolution or winding up, as the case may be, in preference or priority to holders of such other class or series;

- (2) on a parity with another class or series either as to dividends or upon liquidation, whether or not the dividend rates, dividend payment dates or redemption or liquidation price per share thereof be different from those of such others, if the holders of such class or series of stock shall be entitled to receipt of dividends or amounts distributable upon liquidation, dissolution or winding up, as the case may be, in proportion to their respective dividend rates or redemption or liquidation prices, without preference or priority over the holders of such other class or series; and
- (3) junior to another class or series either as to dividends or upon liquidation, if the rights of the holders of such class or series shall be subject or subordinate to the rights of the holders of such other class or series in respect of the receipt of dividends or the amounts distributable upon liquidation, dissolution or winding up, as the case may be "

THIRD: In accordance with the requirements of Section 2-604 of the Maryland General Corporation Law, as amended, (i) the Board of Directors of the Corporation, acting by unanimous written consent, duly adopted resolutions proposing and declaring advisable the amendments to the Articles of Incorporation of the Corporation as set forth above and (ii) the sole stockholder of the Corporation, acting by written consent of the sole holder of all of the outstanding shares of capital stock of the Corporation, duly approved and adopted the amendments to the Articles of Incorporation of the Corporation as set forth above.

FOURTH: Immediately before the effectiveness of these Articles of Amendment of Articles of Incorporation of HMC Merger Corporation (the "Effective Time"), the total number of shares of stock the Corporation had authority to issue was one thousand (1,000) shares of common stock, having a par value of \$0.01 per share, all of one class, with an aggregate par value of \$10. At the Effective Time, the total number of shares of stock the Corporation will have authority to issue is 800,000,000 shares, 750,000,000 of which initially will be classified as common stock, having a par value of \$0.01 per share, and 50,000,000 of which initially will be classified as preferred stock, par value \$.01 per share. The aggregate par value of all classes of stock that the Corporation shall have authority to issue is \$8,000,000.

FIFTH: The preferences, conversion and other rights, voting powers, restrictions, limitations as to dividends, qualifications, and terms and conditions of redemption of each class of capital stock of the Corporation shall be as set forth in Article SECOND of these Articles of Amendment of Articles of Incorporation.

IN WITNESS WHEREOF, the Corporation has caused these Articles of Amendment of Articles of Incorporation of HMC Merger Corporation to be signed in its name and on its behalf by its President and attested by its Secretary on this 19th day of November, 1998, and its President acknowledges that these Articles of Amendment of Articles of Incorporation of HMC Merger Corporation are the act of the Corporation, and under the penalties of perjury, that the matters and facts set forth herein with respect to authorization and approval are true in all material respects to the best of his knowledge, information and belief.

By: /s/ Robert E. Parsons, Jr. (seal)
Robert E. Parsons, Jr.
President

ATTEST:

## [LETTERHEAD OF HOGAN & HARTSON L.L.P.]

November 20, 1998

Board of Directors HMC Merger Corporation 10400 Fernwood Road Bethesda, Maryland 20817

Ladies and Gentlemen:

We are acting as counsel to HMC Merger Corporation, a Maryland corporation (the "Company"), in connection with its registration statement on Form S-4, as amended (the "Registration Statement") filed with the Securities and Exchange Commission relating to the proposed offering of up to 208,000,000 shares of the Company's common stock, par value \$.01 per share (the "Common Stock"), in connection with the proposed merger of Host Marriott Corporation, a Delaware corporation, into the Company. This opinion letter is furnished to you at your request to enable you to fulfill the requirements of Item 601(b)(5) of Regulation S-K, 17 C.F.R. (S) 229.601(b)(5), in connection with the Registration Statement.

- 1. An executed copy of the Registration Statement.
- The Articles of Incorporation of the Company, as certified by the Maryland State Department of Assessments and Taxation (the "SDAT") on September 29, 1998, and by the Secretary of the Company on the date hereof as then being complete, accurate and in effect.
- 3. The Articles of Amendment to Articles of Incorporation of the Company, as filed with the SDAT on November 19, 1998, and as certified by the Company on the date hereof as then being complete, accurate and in effect.
- The Form of Articles of Amendment and Restatement of Articles of Incorporation of the Company (the "Restated Articles").

- The Bylaws of the Company, as certified by the Secretary of the Company on the date hereof as then being complete, accurate and in effect.
- 6. Resolutions of the Board of Directors of the Company adopted on November 18, 1998, as certified by the Secretary of the Company on the date hereof as then being complete, accurate and in effect, relating to the issuance and sale of the Common Stock and arrangements in connection therewith.
- 7. The Form of Agreement and Plan of Merger to be entered into by and among Host, the Company and Host Marriott, L.P., a Delaware limited partnership (the "Agreement and Plan of Merger").

In our examination of the aforesaid documents, we have assumed the genuineness of all signatures, the legal capacity of natural persons, the accuracy and completeness of all documents submitted to us, and the authenticity of all original documents and the conformity to original documents of all documents submitted to us as certified, telecopied, photostatic, or reproduced copies. In rendering the opinion below, we have assumed that none of the Common Stock will be issued in violation of the ownership limits set forth in the Restated Articles. This opinion letter is given, and all statements herein are made, in the context of the foregoing.

This opinion letter is based as to matters of law solely on applicable provisions of Maryland law. We express no opinion herein as to any other laws, statutes, ordinances, rules or regulations or as to compliance with the securities (or "blue sky") laws of Maryland.

Based upon, subject to and limited by the foregoing, we are of the opinion that following (i) effectiveness of the Registration Statement, (ii) due execution and delivery by the parties thereto of the Agreement and Plan of Merger, (iii) effectiveness of the Merger under Delaware and Maryland law and (iv) issuance of the Common Stock pursuant to the terms of the Merger Agreement and as described in the Registration Statement, the Common Stock will be validly issued, fully paid and nonassessable.

This opinion letter has been prepared for your use in connection with the filing of the Registration Statement on the date of this opinion letter and speaks as of the date hereof. We assume no obligation to advise you of any changes in the foregoing subsequent to the delivery of this opinion letter.

HMC Merger Corporation November 20, 1998 Page 3

We hereby consent to the filing of this opinion letter as Exhibit 5.1 to the Registration Statement and to the reference to this firm under the caption "Legal Matters" in the prospectus constituting a part of the Registration Statement. In giving this consent, we do not thereby admit that we are an "expert" within the meaning of the Securities Act of 1933, as amended.

Very truly yours,
/s/ Hogan & Hartson L.L.P.
HOGAN & HARTSON L.L.P.

# AMENDMENT NO. 2 TO DISTRIBUTION AGREEMENT

THIS AMENDMENT NO. 2 TO DISTRIBUTION AGREEMENT (this "Agreement") is made and entered into as of the 21st day of June, 1997, by and between HOST MARRIOTT CORPORATION, a Delaware corporation ("Host Marriott"; successor by change of name to Marriott Corporation), HOST MARRIOTT SERVICES CORPORATION, a Delaware corporation ("HMSC"), and MARRIOTT INTERNATIONAL, INC., a Delaware corporation ("MII").

# RECITALS

WHEREAS, Host Marriott and MII are parties to that certain Distribution Agreement, dated as of September 15, 1993, as amended by Amendment No. 1 dated as of December 29, 1995 among Host Marriott, MII, and (as successor by spin-off to Host Marriott's the Host/Travel Plazas Business), HMSC (as so amended, the "Distribution Agreement"); and

WHEREAS, Host Marriott and MII are parties to that certain Stock Purchase Agreement, dated as of June 21, 1997 (the "Stock Purchase Agreement"); and

WHEREAS, the execution and delivery of this Agreement is a condition precedent to MII's obligation to consummate the transactions contemplated by the Stock Purchase Agreement; and

WHEREAS, Host Marriott and MII desire to amend the Distribution Agreement as set forth herein, and HMSC has agreed to execute this amendment for the avoidance of doubt as to whether its signature is necessary for an amendment to the Distribution Agreement to be effective.

NOW, THEREFORE, in consideration of the foregoing recitals and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties, intending to be legally bound, hereby agree as follows:

1. The last sentence of the first paragraph of Section 6.07(a) of the Distribution Agreement is hereby deleted in its entirety and replaced with the following:

"The Right shall terminate on June 21, 2017."

2. Except as specifically amended hereby, the Distribution Agreement shall remain unchanged and in full force and effect.

## SIGNATURE PAGE FOLLOWS

IN WITNESS WHEREOF, the parties have caused this Agreement to be signed by their duly authorized representatives as of the day and year first above written.

HOST MARRIOTT CORPORATION

By: /s/ Robert E. Parsons, Jr.

Name: Robert E. Parsons, Jr.

Title: Executive Vice President

MARRIOTT INTERNATIONAL, INC.

By: /s/ Paul E. Johnson, Jr.

Name: Paul E. Johnson, Jr.

Title: Vice President

HOST MARRIOTT SERVICES CORPORATION

By: /s/Joe Martin
.....
Name: Joe Martin

Name: Joe Martin Title: Vice President and General Counsel

# AMENDMENT NO. 3 TO DISTRIBUTION AGREEMENT

Amendment No. 3 dated as of March 3, 1998 (this "Amendment") between Marriott International, Inc. ("Old MII") (the name of which will be changed to Sodexho Marriott Services, Inc. on or about the Amendment Effective Date), Host Marriott Corporation ("Host Marriott"), Host Marriott Services Corporation ("HMSC") and New Marriott MI, Inc. ("New MII") (the name of which will be changed to Marriott International, Inc. on or about the Amendment Effective Date) to Distribution Agreement dated as of September 15, 1993, among Old MII and Host Marriott (as successor by change of name to Marriott Corporation), as amended by Amendment No. 1 dated as of December 29, 1995 and Amendment No. 2 dated as of June 21, 1997, each among Old MII, Host Marriott, and HMSC (the Distribution Agreement, as amended by Amendments No. 1 and No. 2, the "Present Agreement"). All capitalized terms used herein, unless otherwise defined herein, shall have the same meanings as set forth in the Present Agreement.

#### Recitals:

Whereas, Pursuant to a Distribution Agreement dated as of September 30, 1997 between Old MII and its wholly-owned subsidiary, New MII (the "SMS Distribution Agreement"), Old MII intends to "spin off" to its shareholders New MII, which will directly or through subsidiaries own all or substantially all of Old MI's lodging, senior living and distribution services businesses; and

Whereas, immediately following the spin off, (i) New MII will be renamed "Marriott International, Inc." and (ii) Old MI's food service and facilities management division, which will be the only business remaining in Old MII, will enter into a business combination with Sodexho Alliance's North American operations, after which Old MII will be renamed "Sodexho Marriott Services, Inc."; and

Whereas, The spin-off and related transactions (the "1998 Spin Off") are more fully described in Old MII's definitive Proxy Statement dated February 12, 1998 which was filed with the Securities and Exchange Commission; and

Whereas, in connection with the 1998 Spin Off, Old MII and New MII have requested that New MII assume all of Old MI's rights and obligations under the Present Agreement, and that Old MII be released from all obligations under the Present Agreement other than those relating to the business which Old MII will retain after the Spin Off; and

Whereas, the parties hereto are, subject to the terms and conditions set forth below, amenable to amending the Present Agreement and agree to do so as hereafter provided.

#### AGREEMENT:

NOW, THEREFORE, it is agreed:

- I. Assignment, Assumption, and Release. In consideration of the mutual covenants and agreements set forth below, HMSC and Host Marriott hereby agree with Old MII and New MII that from and after the closing of the 1998 Spin Off (the "Amendment Effective Date"):
  - 1. New MII assumes and shall be liable for all liabilities and obligations of Old MII under the Present Agreement (including, without limitation, the indemnification under Section 5.02 for MII Liabilities, which include HMT's pre-distribution guaranty of certain [Inflight industrial development revenue bonds] and certain obligations with respect to the Athens, Greece and Cairo, Egypt Marriott hotels from which Host Marriott was not released at the time of the Distribution) and Old MII shall be released from any and all such liabilities and obligations, whether such liabilities or obligations arose before or may arise after the Closing, provided that any such liabilities or obligations which are also "Retained Liabilities" within the meaning of the SMS Distribution Agreement shall, with respect to HMSC and Host Marriott, remain joint and several obligations of New MII and Old MII.
  - 2. Except to the extent that they refer to historical events which took place prior to the Amendment Effective Date and subject to the proviso to paragraph 1 above, all references to "MII" in the Present Agreement shall be deemed to refer to New MII, and not Old MII, as if New MII, and not Old MII were originally the named entity.
  - 3. New MII shall succeed to all rights, title and interest of Old MII in and under the Present Agreement, provided that Old MII shall retain all rights, title and interest under the Present Agreement to the extent related to the "Retained Business" within the meaning of the SMS Distribution Agreement.
- A. Reference to and Effect on the Agreement. On and after the Amendment Effective Date each reference in the Present Agreement to "this Agreement", "hereunder", "hereof" or words of like import referring to the Agreement shall mean and be a reference to the Present Agreement as amended hereby. Except as specifically amended hereby, the Present Agreement is and shall continue to be in full force and effect and is hereby in all respects ratified and confirmed. The execution, delivery and effectiveness of this Amendment shall not, except as expressly provided herein, operate as a waiver of any provision of the Agreement.

- B. Termination Prior to Amendment Effective Date. This Amendment No. 2 will be automatically canceled and its provisions will be of no further force or effect if Old MII and New MII notify Host Marriott and HMSC in writing prior to the Amendment Effective Date that they no longer intend to go forward with the 1998 Spin Off.
- C. Execution in Counterparts. This Amendment may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed and delivered shall be deemed to be an original and all of which taken together shall constitute but one and the same agreement.
- D. Governing Law. This Amendment shall be governed by, and construed in accordance with, the laws of the State of Maryland.

In witness whereof, the parties hereto have caused this Amendment to be executed and delivered by their respective duly authorized officers as of the date first written above.

Old MII: Host Marriott:

MARRIOTT INTERNATIONAL, INC. (to be renamed Sodexho Marriott Services, Inc.)

HOST MARRIOTT CORPORATION

By: /s/ James L. Best
James L. Best

By: /s/ Christopher G. Townsend

Its: Senior Vice President

Authorized Signatory

New MII: HMSC:

NEW MARRIOTT MI, INC. (to be renamed Marriott International, Inc.)

HOST MARRIOTT SERVICES CORPORATION

THAT TOLL INCOME THAT INC. ) COM ONATIO

By: /s/ James L. Best

James L. Best

Authorized Signatory

By: /s/ Joseph Martin
Its: Senior VP & General Counsel

AMENDMENT NO. 4 DATED AS OF \_\_\_\_\_\_\_, 1998 TO DISTRIBUTION AGREEMENT DATED AS OF SEPTEMBER 15, 1993 BETWEEN HOST MARRIOTT CORPORATION AND MARRIOTT INTERNATIONAL, INC.

Host Marriott Corporation (f/k/a Marriott Corporation, "Host Marriott"), Marriott International, Inc. ("MII") and Host Marriott Services Corporation desire to adopt this Amendment to the Distribution Agreement between Host Marriott and MII dated as of September 15, 1993 (the "Original Agreement," and, as amended hereby and by that certain Amendment No. 1 to the Original Agreement dated as of December 29, 1995, that certain Amendment No. 2 to the Original Agreement dated as of June 21, 1997 and that certain Amendment No. 3 to the Original Agreement dated as of March 3, 1998, the "Distribution Agreement").

WHEREAS, on or about December 29, 1998, (i) Host Marriott will distribute approximately 82% of the outstanding common stock of Crestline Capital Corporation, a Maryland corporation ("CCC"), to the shareholders of Host Marriott and will contribute the remaining 18% of such CCC common stock to Host Marriott, L.P. for delivery to The Blackstone Group and certain affiliated entities thereof (or for return to CCC if not delivered to The Blackstone Group and its affiliated entities) and (ii) Host Marriott will merge (the "Merger") into HMC Merger Corporation, a Maryland corporation to be renamed "Host Marriott Corporation" ("Host REIT"); and

WHEREAS, the parties hereto desire to amend the Distribution Agreement in connection with such distribution and the Merger.

NOW, THEREFORE, in consideration of the foregoing and other good and valuable consideration, the parties hereto hereby agree as follows:

- 1. Upon the effectiveness of the Merger, Section 6.07 of the Distribution Agreement shall be amended by adding the following subsections (j) and (k):
- "(j) Notwithstanding anything to the contrary in this Agreement or otherwise, MII's Right will be limited to the purchase and subsequent ownership of only such number of shares, if any, as would not (i) cause MII, or any Person in which MII owns a direct or indirect interest, to own or be deemed (taking into account the attribution rules of Code Section 318(a), as modified by Code Section 856(d)(5)) to own more than 9.9% (the "Disqualification Threshold") of Host

Marriott if MII, or any such other Person in which MII owns a direct or indirect interest, also owns or would be deemed to own (taking into account the attribution rules of Code Section 318(a), as modified by Code Section 856(d)(5)), more than 9.9% of Crestline Capital Corporation, a Maryland corporation ("CCC") (or any Subsidiary of CCC) or any other tenant of real property leased by Host Marriott or any Subsidiary of Host Marriott [(other than the Lease Agreement, dated October 24, 1984, between and [for the Harbor Beach hotel]],  $(\overline{\text{ii}})$  cause any Person that owns (or is deemed to own (taking into account the attribution rules of Code Section 318(a), as modified by Code Section 856(d)(5))) a direct or indirect interest in MII to exceed the Disqualification Threshold with respect to Host Marriott if such Person owns or would be deemed to own (taking into account the attribution rules of Code Section 318(a), as modified by Code Section 856(d)(5)) more than 9.9% of CCC (or any Subsidiary of CCC) or any other tenant of real property leased by Host Marriott or any Subsidiary of Host Marriott, or (iii) in the event that Host Marriott L.P. is or could be considered a "publicly traded partnership" within the meaning of Code Section 7704, cause Host Marriott, L.P. to own more than 9.9% of CCC (determined by taking into account (A) the attribution rules of Code Section 318(a), as modified by Code Sections 856(d)(5) and 7704(d)(3)(B) and (B) any stock of CCC that Host Marriott, L.P. is deemed to own under these rules by reason of the ownership of an interest in Host Marriott, L.P. by [Blackstone Real Estate Acquisitions L.L.C.] or any of its

affiliated entities). In the event that the Right would not be exercisable in full solely by reason of clause (iii) of the preceding sentence (but not clause (i) or clause (ii) thereof), MII shall, subject to the conditions set forth below, have the right to assign that portion (but only such portion) of the Right that is not exercisable by reason of such clause (iii) (the "Blocked Portion of the Right"), subject to further reduction as set forth below, to a Person whose exercise of the assigned Blocked Portion of the Right would not be precluded by the preceding sentence (applied replacing the term "MII" each place it appears in clauses (i) and (ii) of such sentence with the following: "MII and/or any permitted assignee pursuant to the next sentence below"), subject to the following conditions: (A) Host Marriott and MII shall have obtained from the Internal Revenue Service, for the benefit of Host Marriott, a private letter ruling (the user fees and legal fees related to which shall be shared equally by Host Marriott and MII, provided, however, that in no event shall Host Marriott be required to pay more than a total of \$50,000 with respect to such fees) to the effect that neither the existence of such assignment right nor the exercise of such assignment right shall cause MII (or any Person that is considered pursuant to Code Section 318(a) to own any stock of Host Marriott considered owned by MII, actually or constructively pursuant to Code Section 318(a)) to be considered for purposes of Code Sections 318(a), 856)(d)(2), and 7704(d)(1)(C)and (d)(3)(B) to own all or any portion of the Host Marriott stock that is subject to the Blocked Portion of the Right, (B) the proposed assignee shall not be a Person who is engaged or affiliated with any Person who is engaged in the business of owning (as opposed to operating) or, with respect to any assignment made hereunder at any time following

October 8, 2000, owning or operating full-service hotels consisting of, in the aggregate, more than fifteen (15) full-service hotels, in competition with full-service hotels owned by Host Marriott and (C) if, at the time of such assignment, the proposed assignee or any of its Affiliates holds Voting Stock or options, warrants or other rights (including conversion rights) to acquire Voting Stock ("Acquisition Rights"), the portion of the Blocked Portion of the Right which may be assigned to such proposed assignee shall be reduced such that, immediately following such assignment, the aggregate number of shares of Voting Stock which the proposed assignee and its Affiliates would own (assuming for such purpose the exercise in full at such time of the assigned portion of the Blocked Portion of the Right and the Acquisition Rights) would not execeed twenty percent (20%) of the total outstanding shares of Voting Stock (assuming for such purposes the exercise in full at such time of all outstanding Acquisition Rights).

- (k) In the event that Host Marriott shall be advised by outside tax counsel in writing after December 29, 1998 that, as the result of a change in law (including published interpretations by the Internal Revenue Service) after such date, there is a significant risk to Host Marriott that the restrictions set forth in the subsection (j) above, would not be effective to protect the status of Host Marriott as a "real estate investment trust" (a "REIT") under the applicable provisions of the Code, the parties to agree to negotiate in good faith to develop a modification to subsection (j) in a manner that would protect the interests of MII in being able to exercise the Right substantially as and to the extent contemplated in subsection (j) while permitting Host Marriott to continue to qualify as a REIT under the applicable provisions of the Code.
- 2. Upon the effectiveness of the Merger, the term "Voting Stock" in Section 6.07(a) of the Distribution Agreement shall mean shares of common stock, par value \$.01 per share, of Host REIT and any other class of capital stock having, or capable of having, general voting rights to elect the Directors of Host REIT, whether or not now authorized or issued."

"The foregoing sentence is not intended to, and shall not, cause this Agreement to bind or to inure to the benefit of CCC or to grant MI any rights with respect to CCC."

4. Upon effectiveness of the Merger, the Distribution Agreement shall be binding upon and inure to the benefit of Host REIT and Host REIT shall assume all the rights and obligations of Host Marriott thereunder.

5. Host REIT represents that, prior to the effectiveness of the Merger, its Board of Directors shall have duly adopted a resolution and a bylaw provision exempting the exercise by MII of the Right as to Host REIT, as set forth in Paragraph 1 above, from the Maryland Business Combination Statute and the Maryland Control Share Acquisition Statute, respectively, as well as a resolution exempting certain other transactions between Host REIT and MII or their respective subsidiaries and a resolution granting MII an irrevocable exemption under Section 8.2.7 of the Amended and Restated Articles of Incorporation of Host REIT from the Ownership Limit (as defined in Section 8.1 of the Amended and Restated Articles of Incorporation of Host REIT) to permit MII to exercise the Right. A copy of such resolutions and bylaw provision are attached as Annex A hereto. Host REIT covenants that, for as long as MII has the

right to exercise the Right, Host REIT will not revoke or amend such resolutions or amend, alter or repeal such bylaw provision without the prior written consent of MII. Host REIT covenants that in the event it adopts a [shareholders' rights plan], such plan shall provisions which are consistent with Section 6.07 of the Distribution Agreement. Host REIT agrees that MII would suffer irreparable damage in the event any of the foregoing provisions of this Paragraph 5 were not to be performed in accordance with the terms hereof, and that, in such event, MII's remedy at law would be inadequate. Host REIT agrees and consents that temporary and permanent injunctive relief may be granted in favor of MII in any proceeding which may be brought to enforce any provision of this Paragraph 5 without the necessity of proof of actual damage.

- 6. Except as specifically amended hereby, the Distribution Agreement continues in full force and effect without modification and is hereby ratified and confirmed in all respects.
- 7. This Amendment may be executed in any number of counterparts, which, when taken together, shall constitute a single binding instrument.

[signatures appear on the following page]

IN WITNESS WHEREOF, the parties executed and delivered as of	have caused this Amendment No. 4 to be duly, 1998.
	MARRIOTT INTERNATIONAL, INC.
	ву:
	Name:
	Title:
	HOST MARRIOTT CORPORATION
	Ву:
	Name:
	Title:
	HOST MARRIOTT SERVICES CORPORATION
	ву:
	Name:
	Title:
The undersigned is executing this $\ensuremath{Am}$ acknowledging and consenting to the $\ensuremath{ }$	endment solely for the purpose of provisions of Paragraphs 4 and 5 hereof.
	HMC MERGER CORPORATION
	Ву:
	Name:
	Title:

#### NONCOMPETITION AGREEMENT

THIS NONCOMPETITION AGREEMENT ("Agreement") is made and entered into as of \_\_\_\_\_\_, 1998, between and among HOST MARRIOTT CORPORATION, a Delaware corporation ("HMC"), HOST MARRIOTT, L.P., a Delaware limited partnership (the "Operating Partnership," together with HMC "Host"), CRESTLINE CAPITAL CORPORATION, a Delaware corporation ("CCC"), \_\_\_\_\_\_\_\_, a Delaware corporation ("NCS1") and \_\_\_\_\_\_\_, a Delaware corporation ("NCS2," together with NCS1, "NCS"). As used in this Agreement, the terms "Host," "CCC" and "NCS" shall mean Host, CCC and NCS, as the case may be, and their respective Subsidiaries and Affiliates (as such terms are defined in Section 1).

WHEREAS, in connection with (i) the lease of substantially all of the full-service hotels owned by Host, and the sublease of certain limited-service hotels leased by Host from third parties, to CCC (each, a "Hotel Lease" and, together, the "Hotel Leases") and (ii) the lease by NCS to CCC of certain furniture, furnishing, fixtures, soft goods, case goods, equipment and other similar items for use in the hotels ("FF&E") under certain leases entered into in connection with the Hotel Leases (the "FF&E Leases"), in each case as part of the REIT Conversion (as defined in Section 1), Host, CCC and NCS have agreed to enter into this Agreement; and

WHEREAS, as of the date hereof, CCC's principal business consists of owning the Senior Living Community Business, the Hotel Leasing Business, the Asset Management Services Business and the Swissotel Management Company Interest (as such terms are defined in Section 1); and

WHEREAS, as of the date hereof, Host's principal business consists of owning the Host Business and the Non-Controlled Subsidiary Interests (as such terms are defined in Section 1).

NOW, THEREFORE, in consideration of the premises and the mutual covenants contained herein, in the Hotel Leases and the FF&E Leases, and in the related agreements entered into pursuant to or related to the Hotel Leases or the FF&E Leases, and for other valuable consideration, the receipt and sufficiency of which are hereby mutually acknowledged, Host, CCC, and NCS agree as follows:

#### ARTICLE ONE

#### **DEFINITIONS**

#### Definitions.

The following terms when used herein shall have the meanings set forth below:

"Affiliates" shall mean any Person directly or indirectly controlling or

controlled by, or under direct or indirect common control with, Host, NCS or CCC, as the case may be. For purposes of this definition, "control," when used with respect to any Person, means the power to direct the management and policies of such Person, directly or indirectly, through the ownership of voting securities, by contract, or otherwise. Notwithstanding the foregoing, (i) Host's Affiliates shall not include CCC, any Non-Controlled Subsidiary or Marriott International, Inc., or their respective Subsidiaries or Affiliates, (ii) CCC's Affiliates shall not include Host, any Non-Controlled Subsidiary or Marriott International, Inc., or their respective Subsidiaries or Affiliates, (iii) NCS1's Affiliates shall not include Host, CCC, Marriott International, Inc. or any other Non-Controlled Subsidiary, or their respective Subsidiaries or Affiliates and (iv) NCS2's Affiliates shall not include Host, CCC, Marriott International, Inc. or any other Non-Controlled Subsidiary, or their respective Subsidiaries or Affiliates.

"Asset Management Services Business" means the provision of asset

management services to owners of hotels, including without limitation, (i) administration of contracts, (ii) review of operating and financial results, financial statements, budgets, revenue projections and capital spending plans with hotel managers and owners, (iii) administration of facility loans, (iv) negotiation of third party management arrangements, (v) assessment of market conditions, (vi) negotiation of regulatory issues and (vii) provision of advice and information in connection with acquisitions or dispositions of hotels.

"Carried Interest" shall mean with respect to any Person, any right of

another Person (by reason of its status as a general partner, sponsor or otherwise) to receive a specific portion of the earnings or assets of such Person once other investors in such Person have received an agreed upon return on their investment in such Person.

"CCC" shall have the meaning set forth in the first paragraph of this  $\stackrel{--}{\mbox{---}}$  Agreement.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Compete" shall mean (i) to conduct or participate or engage in, or bid

for, or otherwise pursue, a business, whether as a principal, sole proprietor, partner, stockholder, or agent of, or consultant or lender to, or manager for, any Person or in any other capacity, or (ii) to have any ownership or financial interest in any Person or business which conducts, participates or engages in, or bids for, or otherwise pursues, a business, whether as a principal, sole proprietor, partner, stockholder, or agent of, investor in, or consultant or lender to or manager for, any Person or in any other capacity.

"FF&E" shall have the meaning set forth in the second paragraph of this  $\hdots$  . Agreement.

"FF&E Leases" shall have the meaning set forth in the second paragraph of this Agreement.

"Franchise Business" shall mean the ownership or operation of any single or

multiple full-service hotel management or franchise system operating under one or more common brand names. Without otherwise expanding the definition of "Franchise Business," the term "Franchise Business" shall not include (i) the operation of hotels, whether owned by CCC or otherwise, pursuant to a franchise or similar license agreement with the owner or operator of the brand name as long as such owner or operator is not CCC, including the operation of full-service hotels owned by owners other than CCC pursuant to leases or management agreements, (ii) any business or activity with respect to limited-service hotels or (iii) any asset management activities undertaken with respect to hotels for the owners of such hotels.

"Host" shall have the meaning set forth in the first paragraph of this  $\hdots$  Agreement.

"Host Business" shall mean the business of owning full-service hotels,

including without limitation the Initial Hotels, acquiring additional existing and newly developed full-service hotels, developing and constructing for ownership by Host full-service hotels, and improving and expanding the Initial Hotels and any additional full-service hotels in which Host acquires an interest. The term "Host Business" shall not include, without limitation, (i) any business or other activity with respect to limited-service hotels or (ii) any business or other activity with respect to full-service hotels other than the acquisition, development or ownership of full-service hotels or equity interests therein.

"Hotel Leasing Business" means the business of leasing, as the tenant or

subtenant, limited-service or full-service hotel properties and operating or franchising, as the operator or franchisee, such properties, either directly or through a contractual arrangement with a third party, provided, however, that a

lease (or sublease) of a full-service hotel by CCC shall not be considered a permitted part of the Hotel Leasing Business unless the following conditions shall apply:

- (a) If the lease is with respect to a full-service hotel and CCC will not be the operator or manager of such hotel (other than through a contractual arrangement with a third-party manager),
  - (i) the lessor must be a Publicly-Traded REIT or a Permitted Private REIT and the rental payments made by CCC to the lessor must qualify as "rents from real property" within the meaning of Section 856(d) of the Code; and
  - (ii) the estimated economic benefit to be derived by the tenant or subtenant, as applicable, from the lease during the period commencing on the lease commencement date and ending on the initial lease termination date, or, in the case of any extension or renewal of any lease which may be exercised solely at the lessee's option (other than renewals or extensions which provide for rent at then prevailing fair market rental rates), the termination of the extension or renewal period (the "Lease Term"), shall not exceed fifteen percent (15%) of the estimated total economic benefit to be generated by the leased property during the Lease Term, computed as described in clause (iii) below.
    - (iii) For purposes of clause (ii) above:
    - (A) the estimated economic benefit to be derived by the tenant or subtenant during the Lease Term shall equal the present value of the projected net operating income to be received by the tenant or subtenant, respectively, from the leased property for the Lease Term (after deduction for rental payments but before deduction for debt service and including any residual payments to be received by the tenant or subtenant at any time during the Lease Term (including on the last day thereof)), calculated by applying a discount factor which is appropriate in light of the leased property's condition and overall market conditions at the time of determination of such

economic value and the risk to the tenant or subtenant associated with the proposed lease structure; and

(B) the estimated total economic benefit to be derived from the leased property during the Lease Term shall equal the sum of (1) the present value (calculated by applying a discount factor which is appropriate in light of the leased property's condition and overall market conditions at the time of determination of such economic value) of the projected total net operating income from the leased property for the Lease Term (before any deductions for rental payments to be made under the lease, before any deductions for debt service, and in the event the hotel is a CCC Managed Hotel, before any deductions for management fees paid to CCC) and (2) the present value (calculated by applying a discount factor which is appropriate in light of the leased property's condition and overall market conditions at the time of determination of such economic value) of the projected residual value of the leased property at the expiration of the Lease Term, computed by applying a capitalization rate which is appropriate in light of the leased property's expected condition and overall market conditions as of such date to the projected annual net operating income (before any deductions for debt service, rent, and in the event the hotel is a CCC Managed Hotel, management fees paid to CCC) as of the last day of the Lease Term (the "Leased Hotel Present Value").

All of the various determinations of net operating income from the leased property provided for above, together with the various adjustments thereto, and the discount rates and capitalization rates to be applied in making the computations provided for above shall be as reasonably determined by CCC (subject to the review and approval of Host in its reasonable judgment and to Section 5.1 of this Agreement in the event that there is a disagreement between Host and CCC with respect thereto).

(b) If the lease is with respect to a full-service hotel and CCC will be the operator or manager of the hotel (other than through a contractual arrangement with a third-party manager) (a "CCC Managed Hotel"), the lessor may be any Person, but the estimated economic benefit to be derived by CCC from the lease and any related management contract (including both management fees, including any base, incentive and other similar management fees, and any operating profit to be retained by the tenant or subtenant in such capacity) during the Lease Term, computed as described in clause (i) below, shall not exceed either twenty-five percent (25%) of the estimated Leased Hotel Present Value, or fifty percent (50%) of

the Leased Hotel Equity Value of such hotel, computed as described in clause (ii) below:

- (i) For purposes of subparagraph (b) above, the estimated economic benefit to be derived by CCC during the Lease Term shall equal the present value of the sum of (1) the projected net operating income to be received by the tenant or subtenant, respectively, from the leased property for the Lease Term (after deduction for rental payments but before deduction for debt service and including any residual payments to be received by the tenant or subtenant at any time during the Lease Term (including on the last day thereof)) and (2) the projected management fees (including base management fees, incentive management fees, and other amounts payable to CCC), calculated by applying a discount factor which is appropriate in light of the leased property's condition and overall market conditions at the time of determination of such economic value and the risk to the tenant or subtenant associated with the proposed lease structure; and
- (ii) For purposes of subparagraph (b) above, the equity value of such hotel shall be the Hotel Present Value of such hotel less the outstanding debt with respect to such hotel at the commencement of the Lease Term (the "Leased Hotel Equity Value").

All of the various determinations of net operating income from the leased property provided for above, together with the various adjustments thereto, and the discount rates and capitalization rates to be applied in making the computations provided for above shall be as reasonably determined by CCC (subject to the review and approval of Host in its reasonable judgment and to Section 5.1 of this Agreement in the event that there is a disagreement between Host and CCC with respect thereto).

Without otherwise expanding the definition of "Hotel Leasing Business," the term "Hotel Leasing Business" shall not, as to the activities of any Person, include the Hotel Management Business when undertaken with respect to a hotel, whether as to management activities by such Person on behalf of itself or on behalf of others, where such Person has not entered into any lease relationship with respect to such hotel.

third parties with respect to matters incident to the operation of such properties including, without limitation, management services with respect to food, beverages, housekeeping, laundry, vending, plant and equipment operation and maintenance, grounds care, gift or merchandise shops within such properties, reservations, sales and marketing services, conference and meeting facilities, health rooms, swimming and other sports facilities and all other services related to the operation of such hotel properties.

"Initial Hotels" shall mean the full-service hotels operated primarily

under the Marriott, Ritz-Carlton, Four Seasons, Swissotel and Hyatt brand names in which Host will initially have controlling interests or own outright following the REIT Conversion, as set forth on Schedule A hereto.

-----

"Leased Hotel Equity Value" shall have the meaning given to it in

subsection (b)(ii) of the definition of "Hotel Leasing Business" (computed as of the time specified in the specific provision of this Agreement where the term is being used).

"Leased Hotel Present Value" shall have the meaning given to it in

subsection (a)(iii)(B) of the definition of "Hotel Leasing Business" (computed as of the time specified in the specific provision of this Agreement where the term is being used).

"Managed Hotel Equity Value" shall mean with respect to any CCC Managed

Hotel the Managed Hotel Present Value of such hotel less the outstanding debt with respect to such hotel at the commencement of the Management Agreement Term (computed as of the time specified in the specific provision of this Agreement where the term is being used).

"Managed Hotel Present Value" shall have the meaning given to it in

subsection (ii) of the definition of "Permitted Full-Service Management Agreement" (computed as of the time specified in the specific provision of the agreement where the term is being used).

"Merger" shall mean the merger of HMC with and into Host REIT as part of  $\overset{----}{\ldots}$  the REIT Conversion.

"NCS" shall have the meaning set forth in the first paragraph of this  $\stackrel{---}{\ldots}$  Agreement.

"NCS1" shall have the meaning set forth in the first paragraph of this  $\stackrel{---}{\mbox{---}}$  Agreement.

"NCS2" shall have the meaning set forth in the first paragraph of this  $\stackrel{---}{\ldots}$  Agreement.

"1998 Noncompetition Agreement" shall have the meaning set forth in Section 4.1.B. hereof.

"Non-Controlled Subsidiary" shall mean any taxable corporation, including

without limitation NCS, in which the Operating Partnership owns, directly or through a Subsidiary, more than fifty percent (50%) of the economic interest but which the Operating Partnership, either directly or through a Subsidiary, does not control. For purposes of this definition, "control," when used with respect to any Non-Controlled Subsidiary, means the power to direct the management and policies of such Non-Controlled Subsidiary, directly or indirectly, through the ownership of voting securities, by contract, or otherwise.

"Non-Controlled Subsidiary Interests" shall mean the economic interests
----held by the Operating Partnership, either directly or through a Subsidiary, in
the Non-Controlled Subsidiaries.

"Operating Partnership" shall have the meaning set forth in the first  $\_\_\_\_$  paragraph of this Agreement.

"Permitted Full-Service Lease" shall mean either a Permitted REIT Lease or a Permitted Operating Lease.

"Permitted Full-Service Management Agreement" shall mean a management

agreement entered into by CCC (or to be entered into by CCC) with respect to a full-service hotel where CCC will be the manager but CCC will not also be the lessee with respect to such full-service hotel and with respect to which the estimated economic benefit to be derived by CCC from such management contract (including any base, incentive and other similar management fees) during period commencing on the management agreement commencement date and ending on the initial management agreement termination date, or, in the case of any extension or renewal of any management agreement which may be exercised solely at the manager's option (other than renewals or extensions which provide for management fees at then prevailing fair market rates), the termination of the extension or renewal period (the "Management Agreement Term"), computed as described in clause (i) below, shall not exceed twenty percent (20%) of the estimated Managed Hotel Present Value (as defined below):

(i) The estimated economic benefit to be derived by CCC during the Management Agreement Term shall equal the present value of the projected management fees (including base management fees, incentive management fees, and other amounts payable to CCC), calculated by applying a discount factor which is appropriate in light of overall market conditions at the time of determination of such

economic value and the risk to the manager associated with the proposed management agreement fee structure; and

(ii) The term "Managed Hotel Present Value" means the estimated total economic benefit to be derived from the managed property during the Management Agreement Term, which shall equal the sum of (1) the present value (calculated by applying a discount factor which is appropriate in light of the managed property's condition and overall market conditions at the time of determination of such economic value) of the projected total net operating income from the managed property for the Management Agreement Term (before any deductions for rental payments to be made under any lease, before any deductions for debt service, and before any deduction for any management fees paid to CCC) and (2) the present value (calculated by applying a discount factor which is appropriate in light of the managed property's condition and overall market conditions at the time of determination of such economic value) of the projected residual value of the managed property at the expiration of the Management Agreement Term, computed by applying a capitalization rate which is appropriate in light of the managed property's expected condition and overall market conditions as of such date to the projected annual net operating income (before any deductions for debt service, rent, and any management fees paid to CCC) as of the last day of the Management Agreement Term (the "Managed Hotel Present Value").

All of the various determinations of net operating income from the managed property provided for above, together with the various adjustments thereto, and the discount rates and capitalization rates to be applied in making the computations provided for above shall be as reasonably determined by CCC (subject to the review and approval of Host in its reasonable judgment and to Section 5.1 of this Agreement in the event that there is a disagreement between Host and CCC with respect thereto).

"Permitted Operating Lease" shall mean a lease which satisfies the criteria
----set forth in subparagraph (b) to the proviso in the definition of "Hotel Leasing Business."

"Permitted Private REIT" shall mean an entity that would be a Publicly-

Traded REIT but for the fact that shares of capital stock or other units of equity interests of the REIT that are generally entitled to vote for the election of directors or similar managers are not listed or admitted to trading on the New York Stock Exchange or the American Stock Exchange or designated for quotation on the Nasdaq National Market, or any successor to any of the foregoing, so long as CCC's lease or sublease of a full-service hotel from such entity and any other investment

in or with respect to such full-service hotel would not cause CCC to violate Section 3.D.

"Permitted REIT Lease" shall mean a lease which satisfies the criteria set

forth in subparagraph (a) to the proviso in the definition of "Hotel Leasing Business" and that is not a Permitted Operating Lease.

"Person" shall mean any person, firm, corporation, general or limited -----partnership, association, or other entity.

"Primary Host Lessee" shall mean the lessee of more than 25% by number of the Initial Hotels.

"Publicly-Traded REIT" shall mean a REIT whose shares of capital stock or

other units of equity interests that are generally entitled to vote for the election of directors or similar managers are listed or admitted to trading on the New York Stock Exchange or the American Stock Exchange or designated for quotation on the Nasdaq National Market, or any successor to any of the foregoing. The term "Publicly-Traded REIT" shall include any Person whose operations are consolidated with those of a Publicly-Traded REIT under generally accepted accounting principles or in which the Publicly-Traded REIT otherwise owns, directly or indirectly, a twenty-five percent (25%) or greater equity interest so long as, in either case, the Publicly-Traded REIT must take the assets and operations of such Person into account in determining whether it satisfies the income and asset requirements of Section 856(c) of the Code.

"REIT" shall mean "real estate investment trust" within the meaning of  $\overline{\phantom{a}}$ . Sections 856 through 859 of the Code.

"REIT Conversion" shall mean the reorganization of Host's business

operations to permit Host REIT to qualify as a REIT, including the Merger and the other transactions described in the Prospectus/Consent Solicitation that is part of the Registration Statement filed with the Securities and Exchange Commission by Host REIT and the Operating Partnership on Form S-4 (File No. 333-55807).

"Senior Living Community Business" shall mean, as to any Person, the business of acquiring and owning existing and newly developed retirement community properties, improving and expanding the retirement community properties owned and acquired by such Person and or operating retirement.

community properties, improving and expanding the retirement community properties owned and acquired by such Person and/or operating retirement community properties for other owners thereof (whether pursuant to a management agreement, operating agreement, lease, license or otherwise).

"Subsidiaries" shall mean corporations or other entities which are more

than ten percent (10%) owned, directly or indirectly, by Host, CCC or NCS, as the case may be, and partnerships in which Host, CCC or NCS, as the case may be, or a Subsidiary thereof, is a general partner. Notwithstanding the foregoing, Host's

Subsidiaries shall not include NCS or any other Non-Controlled Subsidiary which becomes a party to this Agreement or otherwise agrees to be bound by terms which are substantially the same as those set forth in Section 2.

"Transfer" shall mean the sale, conveyance, disposal of or other transfer ------of ownership, title or other interest.

## ARTICLE TWO

# NONCOMPETITION WITH RESPECT TO THE SENIOR LIVING COMMUNITY BUSINESS

- Certain Restrictions on Host and NCS.
- A. Except as provided in Section 2.C., from the date hereof until December 31, 2003, neither Host nor NCS shall Compete in the Senior Living Community Business.
- B. Except as provided in Section 2.C., from the date hereof until the earlier of (i) December 31, 2008 and (ii) the date on which CCC is no longer the Primary Host Lessee, neither Host nor NCS shall Compete in the Hotel Leasing Business.
- C. Neither Section 2.A. nor Section 2.B. shall prohibit Host or NCS from engaging in the following activities:
  - (i) the ownership of any equity interest in any Person which Competes in the Senior Living Community Business or the Hotel Leasing Business if Host or NCS, as the case may be, directly or indirectly, is the beneficial owner of not more than five percent (5%) of such Person's outstanding equity interests, including for such purpose any Carried Interest in such Person, whether or not earned (based upon the maximum percentage applicable for such Carried Interest) and cannot, by reason of the ownership of such equity interest or otherwise, have any right to control such Person (including, but not limited to, control resulting from a general partner interest, special rights as a manager of a limited liability company or similar entity, contractual or other rights to representation on the board of such Person that are disproportionate to Host's or NCS', as the case may be, equity ownership in such Person, disproportionate voting rights with respect to Host's or NCS', as the case may be, equity position, or veto or approval rights as to major decisions);

- (ii) the acquisition (by merger, stock purchase or otherwise) of, or the purchase of assets from, any Person who Competes in the Senior Living Community Business or the Hotel Leasing Business if the fair market value, on the acquisition date, of the acquired assets which relate to activities which Compete with the Senior Living Community Business or the Hotel Leasing Business, as the case may be, do not constitute more than ten percent (10%) of the total purchase price for the transaction; or
- (iii) (A) the leasing, directly or indirectly, by Host from NCS or by NCS from Host of limited-service or full-service hotel properties, (B) the leasing, directly or indirectly, by Host of properties pursuant to the leases listed on Schedule B attached hereto, including any renewals or extensions thereof, or (C) the leasing, directly or indirectly, by Host from any other Person of limited-service or full-service hotel properties where Host has a direct or indirect equity interest in such Person sufficient for such Person to be consolidated with Host for financial accounting purposes.
- D. Each of Host and NCS agrees that, from the date hereof until December 31, 2000, it will not solicit, hire or induce the termination of employment of, a person who is employed by CCC at the time of, or was employed by CCC at any time within three months prior to, such solicitation, hiring or inducement and whose grade is, or, if applicable, was at the time of the termination of his employment with CCC, the equivalent of Host's current grade 56 or above.

#### ARTICLE THREE

#### NONCOMPETITION WITH RESPECT TO THE HOST BUSINESS

- . Certain Restrictions on CCC.
- A. Except as provided in Section 3.B., from the date hereof until the earlier of (i) December 31, 2008 and (ii) the date on which CCC is no longer the Primary Host Lessee, (a) CCC shall not Compete in the Host Business, and (b) CCC shall not, without the consent of Host, in its sole and absolute discretion, either enter into any leases with respect to full-service hotels other than Permitted Full-Service Leases or enter into any management agreements with respect to full-service hotels other than Permitted Full-Service Management Agreements.
- B. Section 3.A. shall not prohibit CCC from engaging in the following activities:
  - (i) any activity (including any investments) undertaken by CCC that is necessary to and reasonably connected with its business of acting as a lessee of full-service hotels, including acquisitions of property and assets used

in such hotels that are incidental to CCC's role as lessee (such as "hotel working capital" and "furniture, fixtures and equipment" in a manner similar to that contemplated under the Hotel Leases) but excluding loans to or equity investments in the lessor or any of its Affiliates except to the extent permitted under clause (v) below;

- (ii) any activity undertaken by CCC with respect to the Asset Management Services Business;
- (iii) the ownership of any equity interest in any Person which Competes in the Host Business if CCC, directly or indirectly, is the beneficial owner of not more than five percent (5%) or more of such Person's outstanding equity interests, including for such purpose any Carried Interest in such Person, whether or not earned (based upon the maximum percentage applicable for such Carried Interest) and cannot, by reason of the ownership of such equity interest or otherwise, have any right to control such Person (including, but not limited to, control resulting from a general partner interest, special rights as a manager of a limited liability company or similar entity, contractual or other rights to representation on the board of such Person that are disproportionate to CCC's equity ownership in such Person, disproportionate voting rights with respect to CCC's equity position, or veto or approval rights as to major decisions);
- (iv) the acquisition (by merger, stock purchase or otherwise) of, or the purchase of assets from, any Person who Competes in the Host Business if the fair market value, on the acquisition date, of the acquired assets which relate to activities which Compete with the Host Business do not constitute more than ten percent (10%) of the total purchase price for the transaction; or
- (v) the provision of financing for any full-service hotel (whether directly or by participation in a lender syndicate) so long as the following conditions are met:
  - (A) on the date on which CCC becomes contractually committed to provide such financing (1) CCC is (or in connection with such financing will become) the lessee of such hotel (pursuant to a lease which is a Permitted Full-Service Lease), (2) CCC is (or in connection with such financing will become) the manager of such hotel (pursuant to a management agreement which is a Permitted Full-Service Management Agreement), or (3) CCC has a bona fide contract to become either the lessee of such hotel (pursuant to a lease which would be a Permitted Full-Service Lease) or the manager of such hotel (pursuant to a management agreement which would be a Permitted Full-Service Management Agreement) upon completion of the

construction and development or stabilization of such hotel and upon satisfaction of reasonable conditions; and either

- (B) if the Permitted Full-Service Lease will not be a Permitted Operating Lease, the following conditions are met:
  - (1) if such financing is in the form of a loan, (x) the present value of CCC in such financing (determined applying the principles set forth in clause (E) below) does not exceed fifteen percent (15%) of the Leased Hotel Present Value of the hotel which is subject to such financing, each determined as of the date on which CCC becomes contractually committed to provide such financing, and (y) such loan does not include any equity participation feature (whether in the form of warrants, options, a conversion right, interest payments based upon profits, revenues, and/or appreciation, or otherwise) that would cause CCC to violate subclause (2) below at any time, assuming for purposes of such determination that CCC would exercise any and all options and other rights that it might have in connection with such loan (provided that the foregoing shall not prevent the exercise by CCC of its rights upon foreclosure of such indebtedness unless the default with respect to such indebtedness giving the right to such foreclosure had occurred or was imminent at the time CCC acquired such indebtedness), and/or
  - (2) if such financing is in the form of an equity investment, directly or indirectly, in the full-service hotel or the Person owning, directly or indirectly, such hotel, CCC will not beneficially own (and will not have any right to acquire beneficial ownership of) more than fifteen percent (15%) of the outstanding Leased Hotel Equity Value, computed on the date on which CCC becomes contractually committed to make such equity investment (including for such purpose any Carried Interest in the Person which owns such hotel, whether or not earned (based upon the maximum percentage applicable for such Carried Interest)), and CCC cannot, by reason of the ownership of such equity interest or otherwise, have any right to control the hotel or the Person owning such hotel (including, but not limited to, control resulting from a general partner interest, special rights as a manager of a limited liability company or similar entity, contractual or other rights to representation on the board of such Person that are disproportionate to CCC's equity ownership in such Person, disproportionate voting rights with respect to CCC's equity position, or veto or approval rights as to major decisions), and

- (3) the combined economic interest of CCC in the full-service hotel under the Permitted REIT Lease (calculated as set forth in the definition of "Hotel Leasing Business") and the present value of CCC's interest in any financing or equity interests described in subclauses (1) and (2) above (determined applying the principles set forth in clause (E) below), all determined as of the date CCC becomes contractually committed to make such investment, cannot exceed either (x) twenty percent (20%) of the Leased Hotel Present Value, on the date CCC becomes contractually committed to provide such financing, of the hotel which is subject to such financing, or (y) forty percent (40%) of the Leased Hotel Equity Value of such hotel on such date; or
- (C) if the Permitted Full-Service Lease will be a Permitted Operating Lease, the following conditions are met:
  - (1) if such financing is in the form of a loan, such loan does not include any equity participation feature (whether in the form of warrants, options, a conversion right, interest payments based upon profits, revenues, and/or appreciation, or otherwise) that would cause CCC to violate subclause (2) below at any time, assuming for purposes of such determination that CCC would exercise any and all options and other rights that it might have in connection with such loan (provided that the foregoing shall not prevent the exercise by CCC of its rights upon foreclosure of such indebtedness unless the default with respect to such indebtedness giving the right to such foreclosure had occurred or was imminent at the time CCC acquired such indebtedness),
  - (2) if such financing is in the form of an equity investment, directly or indirectly, in the full-service hotel or the Person owning, directly or indirectly, such hotel, CCC cannot, by reason of the ownership of such equity interest or otherwise, have any right to control the hotel or the Person owning such hotel (including, but not limited to, control resulting from a general partner interest, special rights as a manager of a limited liability company or similar entity, contractual or other rights to representation on the board of such Person that are disproportionate to CCC's equity ownership in such Person, disproportionate voting rights with respect to CCC's equity position, or veto or approval rights as to major decisions), and
  - (3) the combined economic interest of CCC in the full-service hotel under the Permitted Operating Lease (calculated as set forth in the definition of "Hotel Leasing Business") and the

present value of CCC's interest in any financing or equity interests described in subclauses (1) and (2) above (determined applying the principles set forth in clause (E) below), all determined as of the date CCC becomes contractually committed to make such investment, cannot exceed either (x) twenty five percent (25%) of the Leased Hotel Present Value, on the date CCC becomes contractually committed to provide such financing, of the hotel which is subject to such financing, or (y) fifty percent (50%)] of the Leased Hotel Equity Value of such hotel on such date; or

- (D) if CCC is a party to, or has a bona fide contract to become a party to, a Permitted Full-Service Management Agreement, the following conditions are met:
  - (1) if such financing is in the form of a loan, such loan does not include any equity participation feature (whether in the form of warrants, options, a conversion right, interest payments based upon profits, revenues, and/or appreciation, or otherwise) that would cause CCC to violate subclause (2) below at any time, assuming for purposes of such determination that CCC would exercise any and all options and other rights that it might have in connection with such loan (provided that the foregoing shall not prevent the exercise by CCC of its rights upon foreclosure of such indebtedness unless the default with respect to such indebtedness giving the right to such foreclosure had occurred or was imminent at the time CCC acquired such indebtedness),
  - (2) if such financing is in the form of an equity investment, directly or indirectly, in the full-service hotel or the Person owning, directly or indirectly, such hotel, CCC cannot, by reason of the ownership of such equity interest or otherwise, have any right to control the hotel or the Person owning such hotel (including, but not limited to, control resulting from a general partner interest, special rights as a manager of a limited liability company or similar entity, contractual or other rights to representation on the board of such Person that are disproportionate to CCC's equity ownership in such Person, disproportionate voting rights with respect to CCC's equity position, or veto or approval rights as to major decisions), and
  - (3) the combined economic interest of CCC in the full-service hotel under the Permitted Full-Service Management Agreement (calculated as set forth in the definition of "Permitted Full-Service Management Agreement") and the present value of

CCC's in any financing or equity interests described in subclauses (1) and (2) above (determined applying the principles set forth in clause (E) below), all determined as of the date CCC becomes contractually committed to make such investment, cannot exceed either (x) twenty percent (20%) of the Managed Hotel Present Value, on the date CCC becomes contractually committed to provide such financing, of the hotel which is subject to such financing, or (y) forty percent (40%) of the Managed Hotel Equity Value of such hotel on such date.

- (E) The present value of CCC's interest in any financing or equity interests purposes of this Section 3.B.(v) shall be determined by computing the present value of the cash flow projected to be received by CCC with respect to such financing or equity interest during the Lease Term or the Management Agreement Term, as applicable, assuming that such financing or equity interest is liquidated for its projected fair market value at the expiration of the Lease Term or Management Agreement Term, as applicable, calculated by applying a discount factor which is appropriate in light of overall market conditions at the time of determination of such economic value and the risk to CCC associated with the proposed financing or equity interest. All of the various determinations of cash flow projected to be received from the proposed financing or equity interest and the discount rates to be applied in making the computations provided for above (and any capitalization rates to be applied for determining projected fair market values at the expiration of the Lease Term or Management Agreement Term, as applicable) shall be as reasonably determined by CCC (subject to the review and approval of Host in its reasonable judgment and to Section 5.1 of this Agreement in the event that there is a disagreement between Host and CCC with respect thereto).
- C. CCC agrees that, from the date hereof until December 31, 2000, it will not solicit, hire, or induce the termination of employment of, a person who is employed by Host or NCS at the time of, or was employed by Host or NCS at any time within three months prior to, such solicitation, hiring or inducement and whose grade, is or, if applicable, was at the time of the termination of his employment with Host or NCS, the equivalent of Host's current grade 56 or above.
- D. Notwithstanding any other provision of this Agreement, until the expiration of the period set forth in Section 3.A., CCC shall not lease or sublease any full-service hotel from a REIT that is not a Publicly-Traded REIT unless the combined economic interest of CCC in the full-service hotel under such lease and any other leases of full-service hotels from REITs that are not Publicly-Traded REITs (calculated as set forth in the definition of "Hotel Leasing Business" but with such calculation being made as of the date on which the determination is being

made under this Paragraph D) and any financing or equity interests in hotels leased from such REITs (or in the Persons owning such hotels) (with the value thereof determined as of the date on which the determination is being made under this Paragraph D) does not exceed the greater of \$125,000,000 or fifteen percent (15%) of the book value of CCC's assets as of such date.

#### ARTICLE FOUR

#### LIMITATION ON ENGAGEMENT IN THE HOTEL MANAGEMENT BUSINESS

- Certain Restrictions on CCC.
- A. Except as provided in Sections 4.1.B. and 4.1.C., CCC shall be entitled to Compete in the Hotel Management Business.
- B. CCC acknowledges that the provisions of that certain Restated Noncompetition Agreement between and among Host and Marriott International, Inc., dated March 3, 1998 (the "1998 Noncompetition Agreement"), applies to it and that such 1998 Noncompetition Agreement has been amended effective the date hereof to include CCC as a party thereto.
- C. Notwithstanding the foregoing Section 4.1.A., from the date hereof until the earlier of (i) December 31, 2008 and (ii) the date on which CCC is no longer the Primary Host Lessee, CCC shall comply with the following restrictions:
  - (i) CCC shall not, without the consent of Host in its sole discretion, engage in the Hotel Management Business with regard to any hotels owned by Host, provided that, the foregoing shall not be deemed to prohibit CCC from acting in its capacity as a lessee of hotels owned by Host where CCC has engaged another Person who is not a Affiliate of CCC to manage or operate, within the meaning of the term "Hotel Management Business," the leased hotels.
  - (ii) CCC shall not engage in the Hotel Management Business with regard to any full-service hotels not owned by Host unless either (a) if CCC is not the lessee (or sublessee) with respect to such hotel, the management agreement with respect to such hotel is a Permitted Full-Service Management Agreement, or (b) if CCC is either the lessee or sublessee with respect to such hotel, the lease with respect to such hotel is a Permitted Full-Service Lease and CCC is compliance with the conditions set forth in Section 3.B(v) with respect to such hotel.
    - (iii) CCC shall not Compete in the Franchise Business.

D. Notwithstanding anything herein to the contrary, nothing in this Agreement shall prohibit CCC from owning the Swissotel Management Company Interest or any activities undertaken by Swissotel Management (U.S.A.) L.L.C.

#### ARTICLE FIVE

#### MISCELLANEOUS

#### 5.1 Arbitration of Certain Matters.

Host, CCC and NCS agree that any controversy or dispute concerning any calculation or determination of value, present values, net operating income, anticipated cash flow, capitalization rate or sales arising under the definition of "Host Leasing Business," "Leased Hotel Equity Value," "Leased Hotel Present Value," "Permitted Full-Service Management Agreement," "Managed Hotel Present Value," "Permitted Full-Service Management Agreement," "Managed Hotel Provalue," or "Managed Hotel Equity Value" in Section 1, or under Section value, or hands detailly value (i.i., Section 3.B.(iv), Section 3.B.(v), or Section 3.E hereof, including without limitation any dispute as to whether as determination of any of the foregoing by CCC is reasonable, shall be settled in arbitration in accordance with the Rules of the American Arbitration Association then in effect. Such arbitration shall take place in Washington, D.C. Any judgment upon the award rendered by the arbitrators may be entered in any court having jurisdiction thereof. The arbitrators shall not, under any circumstances, have any authority to award punitive, consequential, exemplary or similar damages, and may not, in any event, make any ruling, finding or award that does not conform to the terms and conditions of this Agreement. Nothing contained in this Section 5.1 shall limit or restrict in any way the right or power of a party at any time to seek injunctive relief in any court and to litigate the issues relevant to such request for injunctive relief before such court (i) to restrain the other party from breaching this Agreement, or (ii) for specific enforcement of this Section The parties agree that any legal remedy available to a party with respect to a breach of this Section 5.1 will not be adequate and that, in addition to all other legal remedies, each party is entitled to an order specifically enforcing this Section 5.1. Neither party nor the arbitrators may disclose the existence or results of any arbitration under this Agreement or any evidence presented during the course of the arbitration without the prior written consent of both parties, except as required to fulfill applicable disclosure and reporting obligations, or as otherwise required by agreements with third parties, or by law.

#### 5.2 Entire Agreement.

This Agreement, the Hotel Leases, the FF&E Leases and the 1998 Noncompetition Agreement constitute the entire agreement of the parties concerning the subject matter hereof.

#### 5.3 Modification.

This Agreement may only be amended, modified or supplemented in a written agreement signed by both parties hereto.

#### 5.4 Waiver.

No term or condition of this Agreement shall be deemed to have been waived, nor shall there be any estoppel against the enforcement of any provision hereof, except by written instrument of the party charged with such waiver or estoppel.

#### 5.5 Severability.

Host, CCC and NCS agree that the period of restriction and the lack of geographical area of restriction imposed upon the parties are fair and reasonable, are reasonably required for the protection of each of the parties hereto and have been specifically negotiated and carefully tailored with a view to preventing the serious and irreparable injury the other party will suffer in the event of competition by such party with the other party during the time periods set forth herein. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect as though the invalid portions were not a part If the provisions of this Agreement relating to the geographical area of restriction or the period of restriction shall be deemed to exceed the maximum geographical area or period which a court having jurisdiction over the matter would deem enforceable, such area or period shall, for purposes of this Agreement, be deemed to be the maximum geographical area or period which such court would deem valid and enforceable.

#### 5.6 Remedies.

CCC, Host and NCS agree that irreparable damage would occur in the event any of the provisions of this Agreement were not to be performed in accordance with the terms hereof, and that their remedy at law for any breach of the other party's obligations hereunder would be inadequate. CCC, Host and NCS agree and consent that temporary and permanent injunctive relief may be granted in any proceeding which may be brought to enforce any provision hereof without the necessity of proof of actual damage.

The parties hereby agree that the obligations of Host and NCS hereunder are independent and that neither Host nor NCS shall have any liability for the breach by the other of such other's obligations hereunder. CCC and Host agree that, in the event that any Non-Controlled Subsidiary which is not a party to this Agreement engages in any activity in which Host is prohibited from engaging under this Agreement, CCC shall not be entitled to terminate this Agreement but Host shall

indemnify and hold CCC harmless from any liabilities, damages, losses and reasonable expenses incurred by CCC as a result thereof.

#### 5.7 Enforceability.

The terms, conditions and promises contained in this Agreement shall be binding upon and shall inure to the benefit of each of the parties hereto, their heirs, personal representatives, or successors and assigns. Without limiting the generality of the foregoing, the parties agree that, following the Merger, Host REIT shall be deemed to be a successor of Host under this Agreement. Each of the parties hereto shall cause its Subsidiaries which are not Non-Controlled Subsidiaries to comply with such party's obligations hereunder. Nothing herein, expressed or implied, shall be construed to give any other Person any legal or equitable rights hereunder.

#### 5.8 Assignment and Successors and Assigns.

Neither party shall, without the prior written consent of the other, assign any rights or delegate any obligations under this Agreement. Notwithstanding anything herein to the contrary, the restrictions, rights and obligations set forth herein shall be treated as follows: in the event Host Transfers all or substantially all of the Host Business, the transferee thereof shall automatically be bound by the terms of this Agreement; in the event CCC Transfers all or substantially all of the Senior Living Community Business or all or substantially all of the Hotel Leasing Business or all or substantially all of the Asset Management Services Business, the transferee thereof shall automatically be bound by the terms of this Agreement; and, in the event either NCS1 or NCS2 Transfers all or substantially all of its business of leasing FF&E to lessees of full and limited-service hotels, the transferee thereof shall automatically be bound by the terms of this Agreement. The parties acknowledge that, upon the effectiveness of the Merger, Host REIT shall succeed to all of the rights and obligations of HMC under this Agreement.

#### 5.9 Consent to Jurisdiction.

Subject to Section 5.1 hereof, the parties irrevocably submit to the exclusive jurisdiction of (i) the Courts of the State of Maryland in Montgomery County, and (ii) if federal jurisdiction exists, the United States District Court for the State of Maryland for the purposes of any suit, action or other proceeding arising out of this Agreement.

#### 5.10 Interpretation.

When a reference is made to this Agreement to a Section, Article, or Schedule, such reference shall be to a Section, Article, or Schedule of this

Agreement unless otherwise indicated. The headings contained in this Agreement are for reference purposes only and shall neither affect the meaning or interpretation of this Agreement, nor define or limit the scope or intent of any provision or part hereof. Whenever the words "include," or "includes," or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation."

### 5.11 Notices.

All notices and other communications hereunder shall be in writing and shall be delivered by hand, by telecopier with computer generated acknowledgment of receipt, by mail or by Federal Express or similar expedited commercial carrier, to the parties at the following addresses (or at such other addresses for a party as shall be specified by like notice), postpaid and certified with return receipt requested (if by mail), or with all freight charges prepaid (if by Federal Express or similar carrier), and shall be deemed given on the date of acknowledged receipt, in the case of a notice by telecopier, and, in all other cases, on the date of receipt or refusal:

#### To Host:

Host Marriott Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attention: Christopher G. Townsend, Senior Vice President, General Counsel and Corporate Secretary

Fax No.: 301/380-3588

#### To CCC:

Crestline Capital Corporation
10400 Fernwood Road
Bethesda, Maryland 20817
Attention: Tracy M.J. Colden, Senior Vice President,
General Counsel and Corporate Secretary
Fax No.: 301/\_\_\_\_\_

To NCS1:

10400 Fernwood Road
Bethesda, Maryland 20817
Attention: \_\_\_\_\_\_
Fax: 301/\_\_\_\_\_

# 5.12 Governing Law.

This Agreement shall be governed by, and construed in accordance with, the laws of the State of Maryland, regardless of the laws that might be applied under applicable principles of conflicts of laws.

### 5.13 Relationship of Parties.

10400 Fernwood Road Bethesda, Maryland 20817

Attention: Fax: 301/\_

It is understood and agreed that nothing in this Agreement shall be deemed or construed by the parties or any third party as creating an employer-employee, principal/agent, partnership or joint venture relationship between or among the parties.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered, all as of the day and year first above written.

## CRESTLINE CAPITAL CORPORATION

By:
Name: Title:
HOST MARRIOTT CORPORATION
By:
Name: Title:
HOST MARRIOTT, L.P.
By: HMC REAL ESTATE LLC, General Partner
Ву:
Name: Title:
[NCS1]
Ву:
Name: Title:

By:
Name: Title:
executing this Agreement solely for the purpose of to the provisions of the last sentence of Section
HMC MERGER CORPORATION
By:
Name: Title:

-25-

[NCS2]

Schedule A
----Initial Hotels

Schedule B

Certain Permitted Leases

# RESTATED NONCOMPETITION AGREEMENT

THIS RESTATED NONCOMPETITION AGREEMENT ("Agreement") is made and entered into as of March \_\_, 1998, by and between HOST MARRIOTT CORPORATION, a Delaware corporation ("Host Marriott"), NEW MARRIOTT MI, INC., a Delaware corporation (to be renamed Marriott International, Inc.) ("New Marriott"), and, subject to Section 5.14 hereof, MARRIOTT INTERNATIONAL, INC., a Delaware corporation (to be renamed Sodexho Marriott Services, Inc.) ("SMS"). As used in this Agreement, and subject to the provisions of Section 5.14 hereof with respect to Marriott International, the terms "Host Marriott" and "Marriott International" shall mean Host Marriott and New Marriott, respectively, and their respective Subsidiaries and Affiliates, other than any Subsidiaries and Affiliates in which the respective direct or indirect ownership interest of Host Marriott or Marriott International is less than fifty percent (50%).

WHEREAS, SMS and Host Marriott entered into a Noncompetition Agreement dated as of October 8, 1993, as amended (the "Original Agreement"), in connection with and pursuant to that certain Distribution Agreement between them dated as of September 15, 1993, (as thereafter amended from time to time, "Distribution Agreement").

WHEREAS, on December 29, 1995, Host Marriott spun off certain of the businesses subject to the Original Agreement through a distribution of the stock of its then subsidiary, Host Marriott Services Corporation ("HMSC"), to its shareholders, and accordingly, SMS, Host Marriott and HMSC entered into an amendment dated as of December 29, 1995 to the Original Agreement which added HMSC as a party (the Original Agreement, as so amended, the "Existing Agreement").

WHEREAS, on October 1, 1997, SMS announced its intention to spin off to its shareholders a new company, New Marriott, which will directly or through subsidiaries own all or substantially all of Marriott International's lodging, senior living and distribution services businesses; and to rename SMS, the corporate entity which will retain its management services business, Sodexho Marriott Services, Inc.

WHEREAS, as a result of consummation the two spin off transactions described above, the businesses subject to the Existing Agreement will be owned by four separate companies, Host Marriott, HMSC, SMS, and Marriott International; with the result that four companies would need to participate in any and every future modification of or waiver under the Existing Agreement, even though any such waiver or modification would likely have no relevance to two of the four companies.

WHEREAS, New Marriott, SMS, Host Marriott and HMSC now wish to replace the Existing Agreement with two bilateral agreements, of which this Agreement is one, each covering only that subset of the businesses covered by the Existing Agreement which are germane to such parties and each of which is to be deemed by the parties thereto to be a continuation of the Original Agreement with respect to such parties; and accordingly are entering into an Acknowledgment and Release substantially in the form of Exhibit A attached hereto.

-----

NOW, THEREFORE, in consideration of the premises and the mutual convenants contained herein and for other valuable consideration, the receipt and sufficiency of which are hereby mutually acknowledged, Host Marriott and Marriott International agree as between themselves to Amend and Restate the Existing Agreement as follows:

# ARTICLE ONE

1. DEFINITIONS. The following terms when used herein shall have the meaning set forth below:

"AFFILIATES" shall mean any Person directly or indirectly controlling

or controlled by, or under direct or indirect common control with Host Marriott or Marriott International, as the case may be. For purposes of this definition "control", when used with respect to any Person, means the power to direct the management and policies of such person, directly or indirectly, through the ownership of voting securities, by contract, or otherwise. Notwithstanding the foregoing, Host Marriott's Affiliates shall not include Marriott International or its Subsidiaries or Affiliates, and Marriott International's Affiliates shall not include Host Marriott or its Subsidiaries of Affiliates.

"COMMON NAME" shall mean any single or multiple hotel management or

franchise system operating under one or more common brand names.

"COMPETE" shall mean (i) to conduct or participate or engage in, or

bid for or otherwise pursue a business, whether as a principal, sole proprietor, partner, stockholder, or agent of, or consultant to or manager for, any Person or in any other capacity, or (ii) have any ownership interest in any Person or business which conducts, participates or engages in, or bids for or otherwise pursues a business, whether as a principal, sole proprietor, partner, stockholder, or agent of, or consultant to or manager for, any Person or in any other capacity.

"CONFERENCE CENTERS" shall mean the facilities for conferences and

meetings of groups and associations (together with the lodging, food and other services related thereto), principally utilized by Persons belonging to or affiliated with educational, health care, governmental, corporate or other organizations, or

other facilities marketed primarily for such conference and group meeting business, such as the U.S. Postal Service Conference Center located in Norman, Oklahoma, substantially as it is being operated by Marriott International as of October 8, 1993.

"EFFECTIVE PERIOD" shall mean that period commencing on October 8,

1993 and automatically terminating without further documentation on October 8, 2000.

"HOTEL MANAGEMENT BUSINESS" means the business of managing, operating

or franchising limited service or full service hotel properties with respect to matters incident to the operation of such properties including, without limitation, management services with respect to food, beverages, housekeeping, laundry, vending, plant and equipment operation and maintenance, grounds care, gift or merchandise shops within such properties, reservations, sales and marketing services, conference and meeting facilities, health rooms, swimming and other sports facilities and all other services related to the operation of such hotel properties.

"HOST MARRIOTT" shall have the meaning set forth in the first
-----paragraph of this Agreement.

"MARRIOTT INTERNATIONAL" shall have the meaning set forth in the first
------paragraph of this Agreement.

"NEW MARRIOTT" shall have the meaning set forth in the first paragraph of this Agreement.

"NEW MARRIOTT SPINOFF" means the spinoff of New Marriott to the

shareholders of SMS which was announced on October 1, 1997 and is expected to be consummated on or about March 27, 1998.

"PERSON" shall mean any person, firm, corporation, general or limited -----partnership, association, or other entity.

"SMS" shall have the meaning set forth in the first paragraph of this
--Agreement.

"SUBSIDIARIES" shall mean corporation or other entities which are more

than fifty percent (50%) owned, directly or indirectly, by Host Marriott or Marriott International, as the case may be, and partnerships in which Host Marriott or Marriott International, as the case may be, or a subsidiary corporation, is a general partner.

"TERRITORY" shall mean the United States, Canada, and their respective ------territories and protectorates.

"TRANSFER" shall mean the sole, conveyance, disposal of or other

transfer of ownership, title or other interest.

Any capitalized terms defined in the Distribution Agreement and used herein shall have the meanings ascribed to them in the Distribution Agreement unless otherwise defined herein. By this reference, the Distribution Agreement is incorporated in this Agreement.

ARTICLE TWO NONCOMPETITION WITH RESPECT TO THE MMS BUSINESS

2. [INTENTIONALLY DELETED].

ARTICLE THREE
NONCOMPETITION WITH RESPECT TO THE HOST BUSINESS

3. [INTENTIONALLY DELETED].

ARTICLE FOUR
NONCOMPETITION WITH RESPECT TO THE HOTEL MANAGEMENT BUSINESS

4.1 CERTAIN RESTRICTIONS ON HOST MARRIOTT.

A. Except as provided in Sections 4.1.B., 4.1.C. and 4.1.D., Host Marriott shall be entitled to (i) own, lease, acquire and develop hotel properties, (ii) operate or manage (self-managed or managed by a third party) Host Marriott's hotel properties, and (iii) Compete in the Hotel Management Business

- B. Host Marriott shall not own, lease subsequently acquire any hotel properties using the names "Marriott", "Courtyard", "Residence Inn" or "Fairfield" unless Marriott International is operating, managing or franchising such properties pursuant to an agreement with Host Marriott.
- C. Notwithstanding the foregoing Section 4.1.A., during the Effective Period, but solely with respect to hotel properties acquired subsequent to the Effective Date, Host Marriott shall comply with the following restrictions:
  - (i) Host Marriott may operate an unlimited number of hotel properties so long as Host Marriott does not operate more than ten (10) such properties under a Common Name.
  - (ii) Host Marriott may contract with a third party manager for operation of an unlimited number of hotel properties so long as the number so operated in not more than the greater of (a) ten (10) such

properties operated under a common name or (b) twenty-five percent (25%) of the system operated by such manager or managers under a Common Name.

(iii) Host Marriott may franchise as franchisor an unlimited number of hotel properties so long as Host Marriott is not franchisor for more than ten (10) such properties under a Common Name.

The foregoing restrictions in this Section 4.1.C. shall not be binding upon a lender of Host Marriott nor on a purchaser of the aforementioned hotel properties.

D. Nothing in this Agreement shall give any party the right to be a franchisee or franchisor of Marriott International without the approval of Marriott International.

# ARTICLE FIVE MISCELLANEOUS

- 5.1 [INTENTIONALLY DELETED].
- 5.2 ENTIRE AGREEMENT. This Agreement, the Distribution Agreement and

the Restated Documents constitute the entire agreement of the parties concerning the subject matter hereof. Additional restrictions on competition may be contained in the Assignment and License Agreement between Host Marriott and Marriott International of even date herewith, or in certain hotel facility, management agreements, and restrictions continued therein shall be in addition to the restrictions contained herein.

- 5.3 MODIFICATION. This Agreement may only be amended, modified or supplemented in a written agreement signed by both parties hereto.
  - 5.4 WAIVER. No term or condition of this Agreement shall be deemed
    ----ve been waived, nor shall there be any estonnel against the enforcement of

to have been waived, nor shall there be any estoppel against the enforcement of any provision hereof, except by written instrument of the party charged with such waiver or estoppel.

5.5 SEVERABILITY. Host Marriott and Marriott International agree

that the period of restriction and the geographical area of restriction imposed upon the parties are fair and reasonable and are reasonably required for the protection of each of the parties hereto. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect as though the invalid portions were not a part hereof. If the provisions of this Agreement relating to the area of restriction or the period of restriction shall be deemed to exceed the maximum area or period which a court having jurisdiction over the matter would deem enforceable, such area or period

shall, for purposes of this Agreement, be deemed to be the maximum area or period which such court would deem valid and enforceable.

- 5.6 REMEDIES. Marriott International and Host Marriott agree that
- irreparable damage would occur in the event any of the provisions of this Agreement were not to be performed in accordance with the terms hereof, and that their remedy at law for any breach of the other party's obligations hereunder would be inadequate. Marriott International and Host Marriott agree and consent that temporary and permanent injunctive relief may be granted in any proceeding which may be brought to enforce any provision hereof without the necessity of proof of actual damage.
  - ${\tt 5.7}$   ${\tt ENFORCEABILITY}.$  The terms, conditions and promises contained in

this Agreement shall be binding upon and shall inure to the benefit of each of the parties hereto, their heirs, personal representatives, or successors and assigns. Each of the parties hereto shall cause its subsidiaries to comply with such party's obligations hereunder. Nothing herein, expressed or implied, shall be construed to give any other person any legal or equitable rights hereunder.

5.8 ASSIGNMENT AND SUCCESSORS AND ASSIGNS. Neither party shall,

without the prior written consent of the other, assign any rights or delegate any obligations under this Agreement.

5.9 CONSENT TO JURISDICTION. Subject to Section 5.1 hereof, the

parties irrevocably submit to the exclusive jurisdiction of (a) the Courts of the State of Maryland in Montgomery County, and (b) if federal jurisdiction exists, the United States District Court for the State of Maryland for the purposes of any suit, action or other proceeding arising out of this Agreement. Each party hereby irrevocably designates, appoints and empowers Prentice Hall Corporation System, Inc. as its true and lawful agent and attorney-in-fact in its name, place, and stead to receive on its behalf service of process in any action, suit, or proceeding with respect to any matters as to which it has submitted to jurisdiction as set forth in the immediate preceding sentence.

5.10 INTERPRETATION. When a reference is made in this Agreement to a

Section, Article, or Schedule, such reference shall be to a Section, Article, or Schedule of this Agreement unless otherwise indicated. The headings contained in this Agreement are for reference purposes only and shall neither affect the meaning or interpretation of this Agreement, nor define or limit the scope or intent of any provision or part hereof. Whenever the words "include," or "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation."

5.11 NOTICES. All notices and other communications hereunder shall

be in writing and shall be delivered by hand, by facsimile or mailed by registered or certified mail (return receipt requested) to the parties at the following addresses (or

at such other addresses for a party as shall be specified by like notice) and shall be deemed given on the date on which such notice is received:

### To Host Marriott:

Host Marriott Corporation 10400 Fernwood Road Washington, D.C. 20058 (registered or certified mail) Bethesda, Maryland 20817 (express mail or courier) Attention: General Counsel Dept. 72/923 FAX NO.\_\_\_\_\_

### To Marriott International:

Marriott International, Inc. One Marriott Drive Washington, D.C. 20058 (registered or certified mail) Attention: General Counsel Dept. 52/923 FAX NO. 301/380-6727

or

10400 Fernwood Road Bethesda, Maryland 20817 (express mail or courier) Attention: General Counsel Dept. 52/923 FAX NO. 301/380-6727

5.12 GOVERNING LAW. This Agreement shall be governed by, and

construed in accordance with, the laws of the State of Maryland, regardless of the laws that might be applied under applicable principles of conflicts of laws.

5.13 RELATIONSHIP OF PARTIES. It is understood and agreed that nothing in this Agreement shall be deemed or construed by the parties or any third party as creating an employer-employee, principal/agent, partnership or joint venture relationship between the parties.

5.14 NEW MARRIOTT SPINOFF; RELEASE OF SMS. Until such time as the

New Marriott Spinoff is consummated, the term "Marriott International" shall mean SMS and its Subsidiaries (including, without limitation, New Marriott) and Affiliates, other than any Subsidiaries and Affiliates in which the direct or indirect ownership interest of SMS is less than fifty percent (50%). Upon consummation of the New Marriott Spinoff, SMS shall automatically cease to be a party to this

Agreement, and shall have no further rights and be released from all duties hereunder.  $\,$ 

5.15 EFFECTIVENESS AND EFFECT ON EXISTING AGREEMENT. This Agreement

shall not become effective, and the Existing Agreement shall remain in effect, until such time as New Marriott, Host Marriott, SMS, and HMSC have executed and delivered an Acknowledgment and Release substantially in the form of Exhibit A

hereto. Thereafter, as between Marriott International and Host Marriott, this Agreement shall for all purposes be deemed to be an amendment, restatement and continuation of the Existing Agreement, which shall then have no further independent force or effect.

[SIGNATURES ON FOLLOWING PAGE]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered, all as of the day and year first above written.

Marriott International:

NEW MARRIOTT MI, INC. (to be renamed Marriott International, Inc.)

By: /s/ Raymond G. Murphy

Raymond G. Murphy

Senior Vice President and Treasurer

Host Marriott:

HOST MARRIOTT CORPORATION

By: /s/ C.G. Townsend

Name: C.G. Townsend Title: Senior Vice President

SMS [Subject to the provisions of Section 5.14]

MARRIOTT INTERNATIONAL, INC. (to be renamed Sodexho Marriott Services, Inc.)

By: /s/ Raymond G. Murphy

Raymond G. Murphy

Senior Vice President and Treasurer

## ACKNOWLEDGMENT AND RELEASE

This ACKNOWLEDGMENT AND RELEASE is made and entered into as of March \_\_\_\_, 1998, by and between (each a "Party") HOST MARRIOTT CORPORATION, a Delaware corporation ("Host Marriott"), NEW MARRIOTT MI, INC., a Delaware corporation (to be renamed Marriott International, Inc.) ("New Marriott"), MARRIOTT INTERNATIONAL, INC., a Delaware corporation (to be renamed Sodexho Marriott Services, Inc.) ("SMS"), and HOST MARRIOTT SERVICES CORPORATION, a Delaware corporation ("HMSC").

WHEREAS, SMS and Host Marriott entered into a Noncompetition Agreement dated as of October 8, 1993, as amended (the "Original Agreement").

WHEREAS, on December 29, 1995, Host Marriott spun off certain of the businesses subject to the Original Agreement through a distribution of the stock of its then subsidiary, HMSC, to its shareholders, and accordingly, SMS, Host Marriott and HMSC entered into an amendment dated as of December 29, 1995 to the Original Agreement which added HMSC as a party (the Original Agreement, as so amended, the "Existing Agreement").

WHEREAS, on October 1, 1997, SMS announced its intention to spin off to its shareholders a new company, New Marriott, which will directly or through subsidiaries own all or substantially all of Marriott International's lodging, senior living and distribution services businesses; and to rename SMS, the corporate entity which will retain its management services business, Sodexho Marriott Services, Inc.

WHEREAS, as a result of consummation the two spin off transactions described above, the businesses subject to the Existing Agreement will be owned by the four separate Parties, with the result that each Party would need to be participate in any and every future modification of one waiver under the Existing Agreement, even though any such waiver or modification would likely have no relevance to two of the four Parties.

WHEREAS, the Parties have therefore agreed to replace the Existing Agreement with the following bilateral agreements, each covering only that subset of the businesses covered by the Existing Agreement which is germane to such Parties and each of which is deemed by the Parties thereto to be a continuation of the Original Agreement with respect to such parties (each, a "Restated Agreement"): (1) that certain Restated Noncompetition Agreement dated as of March \_\_, 1998 between New Marriott, Host Marriott and (only until the consummation of the New Marriott spinoff) SMS, and (2) that certain Restated Noncompetition Agreement dated as of March \_\_, 1998 between SMS and HMSC.

NOW, THEREFORE, in consideration of the premises and mutual covenants contained in the Restated Agreements and for other valuable consideration, the receipt and sufficiency of which are hereby mutually acknowledged, each Party acknowledges that except as set forth in the Restated Agreement to which it is party, (1) no Party shall have any further right, duty, or obligation to any other Party under the Existing Agreement and releases each other party from any and all such other duties and obligations, and (2) the Existing Agreement has no further independent force or effect.

NEW MARRIOTT MI, INC. (to be renamed Marriott International, Inc.)	HOST MARRIOTT CORPORATION						
ву:	Ву:						
Raymond G. Murphy Senior Vice President and Treasurer	Name: Title:						
MARRIOTT INTERNATIONAL, INC. (to be renamed Sodexho Marriott Services, Inc.)	HOST MARRIOTT SERVICES CORPORATION						
Ву:	Ву:						
Raymond G. Murphy	Name:						
Senior Vice President and Treasurer	Title:						

### FIRST AMENDMENT TO RESTATED NONCOMPETITION AGREEMENT

THIS FIRST AMENDMENT TO RESTATED NONCOMPETITION AGREEMENT (this "Amendment") is made and entered into as of \_\_th day of \_\_\_\_\_\_, 1998 by and among Marriott International, Inc., a Delaware corporation ("Marriott International"), Host Marriott Corporation, a Delaware corporation ("Host Marriott"), and Crestline Capital Corporation, a Maryland corporation ("Crestline").

## RECITALS

WHEREAS, Host Marriott and Marriott International are parties that certain Restated Noncompetition Agreement dated March 3, 1998 (as amended, the "1998 Agreement"); and

WHEREAS, Host Marriott has determined that it will restructure its business operations so as to qualify as a real estate investment trust for federal income tax purposes, and, in connection therewith, on or about December 29, 1998, (i) Host Marriott will distribute approximately 82% of the outstanding common stock of Crestline to or on behalf of the stockholders of Host Marriott (the "Crestline Distribution") and will contribute the remaining 18% of such Crestline common stock to Host Marriott, L.P. for delivery to The Blackstone Group and certain affiliated entities thereof (or for return to Crestline if not delivered to The Blackstone Group and its affiliated entities) and (ii) Host Marriott will merge (the "Merger") into HMC Merger Corporation, a Maryland corporation ("Host REIT"); and

WHEREAS, the parties hereto desire to amend the 1998 Agreement in connection with the Crestline Distribution and the Merger.

NOW, THEREFORE, in consideration of the foregoing recitals and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties intending to be legally bound agree as follows:

1. Crestline is hereby added as a party to the 1998 Agreement and shall be subject to the same restrictions, obligations and benefits thereunder as Host Marriott, it being further understood that all references to "Host Marriott" in the 1998 Agreement (other than in the definition of "Host Marriott" in Article One and the third sentence of the definition of "Affiliates" under Article One and the last sentence of Section 4.1.C. as set forth in Paragraph 5 below, which, in each of the foregoing cases, shall mean only Host Marriott Corporation and its Subsidiaries and Affiliates) shall mean each of Host Marriott and Crestline as if each of the foregoing entities were parties to a separate agreement with MI having terms identical to the 1998 Agreement as amended by this Amendment, subject to the limitations set forth in Paragraph 5 below

2. The third sentence of the definition of "Affiliates" under Article One is hereby deleted and replaced in its entirety with the following:

Notwithstanding the foregoing, (i) Host Marriott's Affiliates shall not include Marriott International, Crestline or their respective Subsidiaries or Affiliates, (ii) Marriott International's Affiliates shall not include Host Marriott, Crestline or their respective Subsidiaries or Affiliates, and (iii) Crestline's Affiliates shall not include Marriott International, Host Marriott, or their respective Subsidiaries or Affiliates.

3. The following definition shall be added to Article One after the definition of "Conference Centers":

"Crestline" means Crestline Capital Corporation, a Maryland corporation.

4. The following sentence shall be added to the definition of "Hotel Management Business" at the end thereof."

For purposes of this definition, it is expressly understood and acknowledged that, with respect to the activities of any Person who is the lessee of property, the terms "operate" and "operating" do not include the terms "lease" and "leasing," respectively, where such Person has engaged another Person who is not an Affiliate of such Person to manage or operate, within the meaning of the term "Hotel Management Business" set forth herein, the leased property.

5. Section 4.1.C. of the 1998 Agreement is hereby amended by adding the following sentence at the end thereof:

It is hereby understood for purposes of this Section 4.1.C. that (i) in no event shall Host Marriott and Crestline be permitted, on an aggregate basis, to operate or franchise as franchisor more than ten (10) hotel properties under a Common Name or to contract with a third party manager for operation of the greater of (a) ten (10) hotel properties operated by such manager under a Common Name or (b) twenty-five percent (25%) of the system operated by such manager under a Common Name and (ii) each hotel owned by Host Marriott which is leased to Crestline shall count as one hotel property for purposes of determining compliance with the foregoing clause.

6. The following language shall be added to the end of Section 5.11:

To Crestline:

Crestline Capital Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attention: General Counsel FAX NO. 301/380-\_\_\_\_

7. The following sentence shall be added to the end of Section 5.13:

It is further understood and agreed that the obligations of each of Host Marriott and Crestline under this Agreement, and any liability arising therefrom, shall be several and not joint.

8. The parties acknowledge that upon the effectiveness of the Merger, Host REIT shall succeed to all of the rights and obligations of Host Marriott under the 1998 Agreement.

- 9. Except as specifically amended hereby, the 1998 Agreement shall remain unchanged and in full force and effect.
- 10. This Amendment shall be effective as of the effective date of the Crestline Distribution.
- 11. This Amendment may be executed in any number of counterparts, which, when taken together, shall constitute a single binding instrument.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered, all as of the day and year first above written.

By:

Name:
Title:

HOST MARRIOTT CORPORATION

By:

Name:
Title:

CRESTLINE CAPITAL CORPORATION

By:

Name:
Title:

MARRIOTT INTERNATIONAL, INC.

The undersigned is executing this Amendment solely for the purpose of acknowledging and consenting to the provisions of paragraph 8 above.

Ву:		 	 	 	 		 	
	Name:							
	Title:	 	 	 	 	-	 	

HMC MERGER CORPORATION

4

### LEASE AGREEMENT

DATED A	S 0F	_, 199
	BY AND BETWEEN	
	AS LANDLORD,	
	AND	

AS TENANT

## TABLE OF CONTENTS

-----

ARTICLE 1 DEFINITIONS1
ARTICLE 2 COLLECTIVE LEASED PROPERTIES AND TERM
2.1 Leased Property25
2.2 Assignment and Assumption of Contracts; Initial Transaction
2.3 Condition of Leased Property27
2.4 Term
ARTICLE 3 RENT
3.1 Rent
3.1.1 Payment of Rent28
3.1.2 Revenues Computation29
3.1.3 Confirmation of Percentage Rent31
3.1.4 Annual Adjustments32
3.1.5 Additional Charges33
3.1.6 Adjustments to/Abatements of Rent
3.2 Late Payment of Rent
3.3 Net Lease
3.4 No Termination, Abatement, Etc38
3.5 Lockbox Procedures39
ARTICLE 4 USE OF THE LEASED PROPERTY
4.1 Permitted Use41
4.1.1 Primary Intended Use41
4.1.2 Necessary Approvals41
4.1.3 Lawful Use, Etc41

4.2	Compliance with Legal and Insurance Requirements, Etc42
4.3	Environmental Matters42
	4.3.1 Restriction on Use, Etc42
	4.3.2 Environmental Indemnification of Tenant by Landlord
	4.3.3 Environmental Indemnification of Landlord by Tenant
	4.3.4 Survival44
4.4	FF&E Reserve
4.5	Working Capital45
ARTICLE 5	MAINTENANCE AND REPAIRS; SURRENDER
5.1	Maintenance and Repair
	5.1.1 Tenant's Obligations46
	5.1.2 Landlord Obligations47
5.2	Tenant's Personal Property48
5.3	Surrender
5.4	Encroachments, Restrictions, Etc
5.5	Landlord to Grant Easements, Etc
ARTICLE 6	CAPITAL EXPENDITURES, ETC
6.1	Capital Expenditures51
6.2	[Intentionally Omitted]
6.3	Cooperation by Tenant
6.4	Alterations
6.5	Salvage53
ARTICLE 7	LIENS
7.1	Prohibition on Liens
7.2	Landlord Lien

ARTICLE 8 PERMITTED CONTESTS54
ARTICLE 9 INSURANCE AND INDEMNIFICATION55
9.1 General Insurance Requirements55
9.2 Responsibility for Insurance57
9.3 Replacement Cost
9.4 Waiver of Subrogation58
9.5 Form Satisfactory, Etc58
9.6 Blanket Policy59
9.7 No Separate Insurance59
9.8 General Indemnification of Landlord by Tenant59
9.9 General Indemnification of Tenant by Landlord66
9.10 Independent Contractor62
ARTICLE 10 CASUALTY62
10.1 Insurance Proceeds62
10.2 Reconstruction in the Event of Casualty63
10.3 Reconstruction in the Event of Damage or Destruction Not Covered by Insurance64
10.4 Tenant's Property and Business Interruption Insurance65
10.5 Abatement of Rent65
ARTICLE 11 CONDEMNATION65
11.1 Total Condemnation, Etc65
11.2 Partial Condemnation66
11.3 Abatement of Rent67
11.4 Allocation of Award67
ARTICLE 12 TENANT DEFAULTS; REMEDIES67
12.1 Event of Default67

12.2	Remedies70
12.3	Waivers
12.4	Application of Funds
12.5	Landlord's Right to Cure Tenant's Default72
ARTICLE 13	HOLDING OVER
ARTICLE 14	LANDLORD NOTICE OBLIGATION; LANDLORD DEFAULT
14.1	Landlord Notice Obligation
14.2	Landlord Default
ARTICLE 15	ARBITRATION
15.1	Arbitration
15.2	Arbitration Procedures
ARTICLE 16	SUBLETTING AND ASSIGNMENT
16.1	Subletting and Assignment
16.2	Required Sublease Provisions
16.3	No Right of Tenant to Mortgage Its Leasehold
ARTICLE 17	ESTOPPEL CERTIFICATES AND FINANCIAL STATEMENTS
17.1	Estoppel Certificates
17.2	Financial Statements
17.3	Annual Budget80
ARTICLE 18	LANDLORD'S RIGHT TO INSPECT82
ARTICLE 19	APPRAISAL82
ARTICLE 20	FACILITY MORTGAGES
20.1	Landlord May Grant Liens84
20.2	Subordination of Lease
20.3	Notice to Mortgagee and Ground Landlord87

	20.4	Transfer of	Leased Property8	7
ARTICL	E 21	ADDITIONAL C	OVENANTS OF TENANT8	7
	21.1	Conduct of B	usiness8	7
	21.2	Maintenance	of Accounts and Records8	8
	21.3	Management o	f Leased Property8	8
		21.3.1 M	lanagement Agreement, Consent and Assignment, Etc8	8
		21.3.2 R	eversion upon Termination8	8
		21.3.3 A	batement of Rent8	9
		21.3.4 C	onsent Required for Certain Actions8	9
		21.3.5 R	eplacement of Manager9	0
	21.4	Facility Mor	tgagee Agreement9	1
	21.5	[Intentional	ly Omitted]9	2
	21.6	Single Purpo	se Entity Covenants9	2
		21.6.1 S	eparate Existence9	2
		21.6.2 I	ndependent Member9	2
		21.6.3 L	imitation on Indebtedness and Guarantees9	2
		21.6.4 D	ristributions9	3
		21.6.5 S	ingle Purpose9	3
			ertain Fundamental Changes9	
			mendments to Organizational Documents9	
		•	ualified Affiliate9	
ARTICL			9	
	22.1	REIT Complia	nce9	4
	22.2	FF&E Limitat	ion9	5
	22 2	Cublacca Ban	t Limitation 0	_

	22.4 Subtease Tenant Limitation95
	22.5 Tenant Ownership Limitation95
RTIC	LE 23 MISCELLANEOUS95
	23.1 No Waiver96
	23.2 Remedies Cumulative96
	23.3 Severability
	23.4 Acceptance of Surrender96
	23.5 No Merger of Title96
	23.6 Release of Landlord Following Conveyance97
	23.7 Quiet Enjoyment
	23.8 Landlord's Consent
	23.9 Memorandum of Agreement98
	23.10 Notices
	23.11 Construction
	23.12 Counterparts; Headings99
	23.13 Governing Law; Jurisdiction
	23.14 No Broker
	23.15 Related Agreements
	23.16 Legal Fees and Costs of Litigation
	23.17 Force Majeure
	23.18 Conflicts with Related Agreements
	23.19 Operating Lease
RTIC	LE 24 TERMINATION RIGHTS
	24.1 Landlord's Right to Terminate Lease upon Sale or Tax Law Change
	24.2 Tanant's Pight to Tarminata Lass upon Cartain Events

24.3	Termination	of	Lease	Following	Notice	of	Termination	of	Guarantee	 	 	 	 	 	 	1	.04

## EXHIBITS

-----

EXHIBIT B	Preliminary	Statement	of	Cash	and	Working	Capital

EXHIBIT C EBITDA Performance Requirements

The Land

EXHIBIT D Guarantee

EXHIBIT A

EXHIBIT E Working Capital Note

Schedule 2.2(a)(vi) Excluded Leases

## SCHEDULES

-----

Schedule 3.1.1	Minimum Rent
Schedule 3.1.2	Revenue Percentage and Break Points
Schedule 3.1.3(b)	Items of Gross Revenues
Schedule 3.1.5	Prepaid Expenses
Schedule 5.1.1	Existing Conditions Relating to Hazardous Materials
Schedule 17.2(f)	Period Report Format
Schedule 20.3	Superior Mortgagee(s) and Superior Landlord(s)
Schedule 21.5	Form of Facility Mortgagee Agreement
Schedule 22.2	Provisions Relating to Excess FF&E
Schedule 22.2-A	Form of Excess FF&E Lease
Schedule 22.2-B	Market Leasing Factor
Schedule 23.8	Deemed Consents or Approvals

-viii-

### LEASE AGREEMENT

THIS LEASE AGREEMENT (this "LEAS	SE") is entered into this day of
, 1998, but effective for	all purposes as of the Commencement Date,
by and between,	a Delaware limited liability company,
having its principal office at 10400	Fernwood Road, Bethesda, Maryland 20817
("LANDLORD"), and	, a Delaware limited liability company,
having its principal office at 10400	Fernwood Road, Bethesda, Maryland 20817
("TENANT").	

### WITNESSETH:

WHEREAS, Landlord owns [fee simple/leasehold] title to the Leased Property (this and other capitalized terms used and not otherwise defined herein having the meanings ascribed to such terms in Article 1);

WHEREAS, Landlord wishes to lease the Leased Property to Tenant and Tenant wishes to lease the Leased Property from Landlord, all subject to and upon the terms and conditions herein set forth; and

WHEREAS, as a condition to entering into this Lease, Landlord and Tenant have entered into that "Consent and Assignment" (as that term is defined hereafter), and have agreed that Tenant shall assume certain obligations of Landlord under the "Assigned Agreements" (as that term is defined hereafter); and

NOW, THEREFORE, in consideration of the mutual covenants herein contained and other good and valuable consideration, the mutual receipt and legal sufficiency of which are hereby acknowledged, Landlord and Tenant hereby agree as follows:

# ARTICLE 1

For all purposes of this Lease, except as otherwise expressly provided or unless the context otherwise requires, (i) the terms defined in this Article shall have the meanings assigned to them in this Article and include the plural as well as the singular, (ii) all accounting terms not otherwise defined herein shall have the meanings assigned to them in accordance with GAAP, (iii) all references in this Lease to designated "Articles," "Sections" and other subdivisions are to the designated Articles, Sections and other subdivisions of this Lease, and (iv) the words "herein," "hereof," "hereunder" and other words of similar import refer to this Lease as a whole and not to any particular Article, Section or other subdivision.

- 1.1. "AAA" shall have the meaning given such term in Section 15.1.
- 1.2. "ACCOUNTING PERIOD" shall mean (i) for so long as the Management Agreement is in effect each of the four (4) week accounting periods that are used in Manager's accounting system, except that an Accounting Period may occasionally contain five (5) weeks when necessary to conform Manager's accounting system to the calendar and (ii) during any period when the Management Agreement is not in effect, each calendar month during the Term.

- 1.3. "ACCOUNTING PERIOD STATEMENT" shall have the meaning given such term in Section 17.2(d).
- 1.4. "ADDITIONAL CHARGES" shall have the meaning given such term in Section 3.1.5.
- 1.5. "ADVERSE PARTY" shall mean any Person (i) who is engaged in the Hotel Business or is an Affiliate (other than an Affiliate who is a lessee under a lease of a hotel from a real estate investment trust or other passive owner which is managed on behalf of such Affiliate by an unaffiliated third party manager) of any Person engaged in the Hotel Business, (ii) who is, or is an Affiliate of, a Person who has been convicted of, or has pleaded nolo contendere to, a felony, (iii) who does not have a Tangible Net Worth of at least (a) two and one-half (2.5) times the amount of Guarantor's then maximum liability under the Guarantee or (b) \$50,000,000, whichever is greater, (iv) whose ownership of a direct or indirect interest in Tenant would violate the Management Agreement or (v) who, prior to the effective time of a Change in Control, fails to enter into a written agreement reasonably satisfactory in form and substance to Host REIT (1) to the effect that neither (A) such Person nor (B) any Person who would be considered to constructively own either any interest in such Person or any interest in Tenant owned directly or indirectly by such Person (as determined under Section 318(a) of the Code, as modified by Section 856(d) of the Code) (referred to as "CONSTRUCTIVE OWNERSHIP") would own (or have Constructive Ownership of) any interest in Host REIT or Host O.P. that would cause either Host REIT or Host O.P. to be considered to Constructively Own any interest in either Tenant or any other tenant of either Host REIT or Host O.P. that is described in Section 856(d)(2)(B) of the Code for purposes of applying either Section 856(c) of the Code or Section 7704(d) of the Code, or any similar or successor provisions thereto and (2) with respect to such other matters reasonably required to protect the status of Host REIT as a "real estate investment trust" for federal income tax purposes.
- 1.6. "AFFILIATE" shall mean any individual or entity directly or indirectly through one or more intermediaries controlling, controlled by or under common control with a party. The term "control," as used in the immediately preceding sentence, means, with respect to a corporation, the right to exercise, directly or indirectly, fifty percent (50%) or more of the voting rights of any class of the shares of the controlled corporation, and, with respect to an entity that is not a corporation, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of the controlled entity. Notwithstanding the foregoing, Manager shall not be deemed an Affiliate of either Landlord or Tenant.
  - 1.7. "ALTERATIONS" shall have the meaning given such term in Section 6.4.
- 1.8. "ANNUAL ADJUSTMENT" shall mean (a) with respect to Minimum Rent, the product of (i) the amount of the Minimum Rent then in effect and (ii) \_\_\_\_\_ percent (\_\_\_\_%) of any Percentage Increase in CPI; (b) with respect to each of the Annual Room Revenues First Break Point, the Annual Food and Beverages Sales First Break Point, and the Annual Other Income Break Point, the product of (i) the amount of each such break point then in effect and (ii) the weighted average resulting from the sum of \_\_\_\_ percent (\_\_\_\_%) of any Percentage Increase in CPI plus \_\_\_\_ percent (\_\_\_\_%) of any Percentage

Increase in the Labor Index; and (c) with respect to the Annual Food and Beverages Sales Second Break Point and the Annual Room Revenues Second Break Point, the product of (i) the amount of each such break point then in effect and (ii) \_\_\_\_\_\_% of the figure calculated in subsection (b)(ii) above.

- 1.9. "ANNUAL BUDGET" shall have the meaning given such term in Section 17.3.
- 1.10. "ANNUAL FOOD AND BEVERAGES SALES FIRST BREAK POINT" shall have the meaning given such term in Section 3.1.2(b)(3).
- 1.11. "ANNUAL FOOD AND BEVERAGES SALES SECOND BREAK POINT" shall have the meaning given such term in Section 3.1.2(b)(3).
- 1.12. "ANNUAL OPERATING STATEMENT" shall have the meaning given such term in Section 17.2(c).
- 1.13. "ANNUAL OTHER INCOME BREAK POINT" shall have the meaning given such term in Section 3.1.2(b)(3).
- 1.14. "ANNUAL ROOM REVENUES FIRST BREAK POINT" shall have the meaning given such term in Section 3.1.2(b)(3).
- 1.15. "ANNUAL ROOM REVENUES SECOND BREAK POINT" shall have the meaning given such term in Section 3.1.2(b)(3).
- 1.16. "APPLICABLE EXPECTED LIFE" shall have the meaning given such term in paragraph (e)(i) of Schedule 22.2.
- 1.17. "ASSET MANAGEMENT AGREEMENT" shall mean that certain agreement dated as of even date herewith as amended or restated from time to time, between CCC and Host O.P. relating to the provision of asset management services by CCC or a wholly owned Subsidiary thereof with respect to the Facility and the hotel facilities covered by the other Pool Leases or the Other Leases.
- 1.18. "ASSIGNED AGREEMENTS" shall have the meaning given such term in Section  $2.2(a)\,.$
- 1.19. "AWARD" shall mean all compensation, sums or other value awarded, paid or received by virtue of a total or partial Condemnation of the Leased Property (after deduction of all reasonable legal fees and other reasonable costs and expenses, including, without limitation, expert witness fees, incurred by Landlord in connection with obtaining any such award).
- 1.20. "BANKRUPTCY CODE" shall have the meaning given such term in Section 16.1.  $\,$
- 1.21. "BUDGET SPREAD" shall have the meaning given such term in Section 17.3.  $\,$

- 1.22. "BUSINESS DAY" shall mean any day other than Saturday, Sunday, or any other day on which banking institutions in the State are authorized by law or executive action to close.
- 1.23. "CAPITAL BUDGET" shall have the meaning given such term in Section 17.3.  $\,$
- 1.24. "CAPITAL EXPENDITURE" shall mean any expenditure for repairs, alterations, improvements, renewals or replacements to the structure or exterior facade of the Facility, or to the mechanical, electrical, heating, ventilating, air conditioning, plumbing, or vertical transportation elements of the Facility (together with all other repair and maintenance expenditures which are classified as capital expenditures under GAAP), or, during any period when the Management Agreement is in effect, to the extent in conflict with the foregoing, "Capital Expenditure" as defined in the Management Agreement. To the extent applicable as used herein, Capital Expenditure shall also refer to the actual repair, alteration, improvement, renewal or replacement to which the Capital Expenditure relates.
- 1.25. "CAPITAL PORTION" shall have the meaning given such term in Section 17.3.  $\,$
- 1.26. "CAPITALIZED INTEREST" shall mean interest that is capitalized and is not counted as interest expense in accordance with  $\sf GAAP$ .
- 1.27. "CAPITALIZED LEASE OBLIGATIONS" of any Person shall mean all rental obligations which, under GAAP, are or will be required to be capitalized on the books of such Person, in each case taken at the amount thereof accounted for as indebtedness in accordance with such principles.
- 1.28. "CASE GOODS" shall mean furniture and furnishings used in the Facility, including, without limitation, chairs, beds, chests, headboards, desks, lamps, tables, television sets, mirrors, pictures, wall decorations and similar items.
  - 1.29. "CASUALTY" shall have the meaning given such term in Section 10.1.
- 1.30. "CCC" shall mean Crestline Capital Corporation, a Maryland corporation.
- 1.31. "CCC CONSOLIDATED FINANCIAL STATEMENTS" shall mean for any Fiscal Year or quarter thereof, consolidated statements of operations, retained earnings and cash flow for such period and for the period from the beginning of the respective Fiscal Year to the end of such period and the related balance sheet as at the end of such period, together with the notes to any such statement, all in such detail as may be required by the SEC with respect to filings relating to periods within the Term made by Landlord, Host REIT or Host O.P., and, after the first full Fiscal Year of Tenant, setting forth in comparative form the corresponding figures for the corresponding period in the preceding Fiscal Year, and prepared in accordance with GAAP and (in the case of annual financial statements) audited by a "Big Five" firm of independent certified public accountants or such other firm of independent certified public accountants as may be reasonably approved by Landlord. CCC Consolidated Financial Statements shall be prepared on the basis of the Fiscal Year or quarter thereof, as applicable. Except as otherwise provided herein, the cost for one (1) consolidated audit with respect to CCC per Fiscal Year and one (1) quarterly unaudited

statement with respect to CCC for each of the first three fiscal quarters in the Fiscal Year as may be required by the SEC for regular reporting requirements of Landlord, Host REIT or Host O.P. under the Securities and Exchange Act of 1934 shall be borne by Tenant. Except as otherwise provided herein, the cost of any other audits or financial statements requested by Landlord shall be borne by Landlord. CCC Consolidated Financial Statements shall be accepted by Landlord if prepared on a combined basis along with all of the other hotels leased from Landlord or any of its Affiliates by Tenant, OpCo, CCC, or a wholly owned Subsidiary of Tenant, OpCo or CCC.

- 1.32. "CHANGE IN CONTROL" shall mean the (i) acquisition (after a registered offering of shares or otherwise) by any Person, or two or more Persons acting in concert, in a single or series of transactions (whether or not related), of:
- (a) with respect to Tenant or OpCo, beneficial ownership (within the meaning of Rule 13d-3 of the SEC) of, or rights (including conversion rights), options or warrants to acquire (whether absolute or conditional), all or any portion of the outstanding voting or economic interests in Tenant or OpCo (other than voting rights held by an Independent Member of Tenant as set forth in the Limited Liability Company Operating Agreement of Tenant as in effect on the date hereof); or
- (b) with respect to CCC, beneficial ownership of, or rights (including conversion rights), options or warrants to acquire, thirty-five percent (35%) or more of the outstanding shares of voting stock of CCC;
- or (ii) the merger or consolidation of Tenant, OpCo or CCC with or into any other Person (other than any one or more Qualified Affiliates) if, in the case of CCC, immediately following the effectiveness of such merger or consolidation, more than thirty-five percent (35%) of the beneficial ownership of the outstanding voting shares of the surviving entity of such merger or consolidation (including for such purpose in both the numerator and the denominator, shares of voting stock issuable upon the exercise of then outstanding rights (including conversion rights), options or warrants) is held by one or more Persons who were not holders of the outstanding voting shares of CCC (or outstanding rights (including conversion rights), options or warrants to acquire voting shares of CCC) immediately prior to the effectiveness of such merger or consolidation; or
- (iii) any one or a series of related sales or conveyances to any Person (other than any one or more Qualified Affiliates) of all or substantially all of the assets of Tenant, OpCo or CCC; or
- (iv) a change in the majority of the board of directors or similar managing body of CCC during the longer of (A) any twelve (12) month period and (B) the period during which two consecutive annual meetings of the equity holders of CCC at which directors or similar managers are elected have been held;

provided, however, that the acquisition by any one or more Qualified Affiliates of beneficial ownership of, or rights (including conversion rights),options or warrants to acquire, all or any portion of Tenant or OpCo shall not be a "Change in Control" hereunder.

- 1.33. "CLAIMS" shall have the meaning given such term in Article 8.
- 1.34. "CODE" shall mean the Internal Revenue Code of 1986 and, to the extent applicable, the Treasury Regulations promulgated thereunder, each as from time to time amended.
  - 1.35. "COMMENCEMENT DATE" shall mean January 1, 1999.
- 1.36. "COMPARABLE LEASE" shall have the meaning given such term in Section 24.1(b).
- 1.37. "CONDEMNATION" shall mean, with respect to the Leased Property, (a) the exercise of any governmental power with respect to all or part of the Leased Property, whether by legal proceedings or otherwise, by a Condemnor of its power of condemnation, (b) a voluntary sale or transfer of all or part of the Leased Property by Landlord to any Condemnor, either under threat of condemnation or while legal proceedings for condemnation are pending, or (c) a taking or voluntary conveyance of all or part of the Leased Property, or any interest therein, or right accruing thereto or use thereof, as the result or in settlement of any Condemnation or other eminent domain proceeding affecting the Leased Property, whether or not the same shall have actually been commenced.
- 1.38. "CONDEMNOR" shall mean any public or quasi-public authority, or private corporation or individual, having the power of Condemnation.
- 1.39. "CONSENT AND ASSIGNMENT" shall have the meaning given such term in Section 21.3.1.
- 1.40. "CONSOLIDATED COVERAGE RATIO" of any Person shall mean, for any period, the ratio of (x) Consolidated EBITDA of such Person for such period to (y) the sum of (i) Consolidated Interest Expense of such Person for such period, and (ii) the amount of all scheduled principal amortization payments (other than balloon payments at maturity of such Indebtedness) on all Indebtedness of such Person and its Subsidiaries made during such period.
- 1.41. "CONSOLIDATED EBITDA" of any Person shall mean, for any period, the Consolidated Net Income of such Person for such period adjusted (A) to add thereto (to the extent deducted in determining Consolidated Net Income for such period), without duplication, the sum of the amounts for such period of (i) Consolidated Interest Expense, (ii) provisions for taxes based on income, (iii) depreciation and amortization expense (provided that the depreciation and amortization expense of a non-wholly owned Subsidiary shall be included only to the extent of the equity interest of such Person in such non-wholly owned Subsidiary), (iv) any other noncash items reducing Consolidated Net Income of such Person for such period (except to the extent that such noncash items relate to the write-off or write-down of any item included in Consolidated Net Income in a prior period) and (v) any cash receipts of such Person or a Subsidiary thereof during such period that represent items included in Consolidated Net Income of such Person for a prior period which were excluded from Consolidated EBITDA in such prior period by virtue of clause (B) of this definition, and (B) to subtract therefrom, without duplication, the sum of the

amounts for such period of (i) all noncash items increasing Consolidated Net Income of such Person for such period and (ii) any cash expenditures of such Person and its Subsidiaries during such period to the extent that such cash expenditures (x) did not reduce Consolidated Net Income of such Person for such period and (y) were applied against reserves or accruals that constituted noncash items reducing Consolidated Net Income of such Person when reserved or accrued for in a prior period; provided, that Consolidated EBITDA shall be determined without giving effect to any extraordinary gains or losses (including any taxes attributable to any such extraordinary gains or losses) or gains or losses from sales of assets other than from sales of inventory in the ordinary course of business.

- 1.42. "CONSOLIDATED INTEREST EXPENSE" of any Person shall mean, for any period, the total consolidated cash interest expense of such Person and its Subsidiaries for such period (calculated without regard to any limitations on the payment thereof) plus, without duplication, that portion of the Capitalized Lease Obligations of such Person and its Subsidiaries representing the interest factor for such period, but excluding (i) any Capitalized Lease Obligations that relate to leases of FF&E, (ii) the interest factor (if any) in such period under [the Initial FF&E Lease,] any Excess FF&E Leases or this Lease with respect to the FF&E subject to this Lease, and (iii) the Working Capital Note. Notwithstanding anything to the contrary contained in the requirements of GAAP, the amount of Capitalized Interest incurred during any period shall be added as a component of Consolidated Interest Expense.
- 1.43. "CONSOLIDATED NET INCOME" of any Person shall mean, for any period, the net income (or loss) of such Person and its Subsidiaries for such period, determined on a consolidated basis; provided that (i) net income (or loss) of (x) any other Person that is accounted for by such specified Person by the equity method of accounting or (y) any Subsidiary of such Person that is restricted from declaring or paying dividends or other distributions, directly or indirectly, by operation of the terms of its charter, any applicable agreement, instrument, judgment, decree, order, statute, rule or governmental regulation or otherwise shall (in each case) be included only to the extent of the amount of cash dividends or distributions paid to the specified Person or a wholly owned Subsidiary of such Person, (ii) the net income (or loss) of any other Person acquired by such specified Person or a Subsidiary of such Person in a pooling of interests transaction for any period prior to the date of such acquisition shall be excluded, (iii) to the extent Consolidated Net Income reflects amounts attributable to minority interests in Subsidiaries that are not wholly owned Subsidiaries of such Person, Consolidated Net Income shall be reduced by the amounts attributable to such minority interests and (iv) any income, rental credit or expense associated with leases of Excess FF&E by Tenant or any Other Tenant shall not be taken into account in determining Consolidated Net Income.
- 1.44. "CONSTRUCTIVE OWNERSHIP" shall have the meaning given such term in Section 1.5.  $\,$
- 1.45. "CONSUMER PRICE INDEX" shall mean the "Consumer Price Index" published by the Bureau of Labor Statistics of the United States Department of Labor, All Items for Urban Wage Earners and Clerical Workers (1982-1984 = 100).

- 1.46. "CONTINUING OBLIGATIONS" shall have the meaning given such term in Section 21.3.1.
  - 1.47. "CONTRACTOR" shall have the meaning given such term in Section 9.10.
- 1.48. "CONTRACTOR'S INSURANCE CERTIFICATE" shall have the meaning given such term in Section 9.10.
- 1.49. "CONTRACTOR'S LIABILITY COVERAGE" shall have the meaning given such term in Section 9.10 .
- 1.50. "CUMULATIVE PORTION" shall have the meaning given such term in Section 3.1.2(b)(1).
- 1.51. "DEFAULT" shall mean any event, act or condition that with the giving of notice or lapse of time, or both, would constitute an Event of Default.
- 1.52. "EMERGENCY SITUATIONS" shall mean fire, any other Casualty, or any other events, circumstances or conditions that threaten the safety or physical well-being of the Facility's guests or employees or that involve the risk of material property damage or loss.
- 1.53. "ENTITY" shall mean any corporation, general or limited partnership, limited liability company or partnership, stock company or association, joint venture, association, company, trust, bank, trust company, land trust, business trust, cooperative, any government or agency or political subdivision thereof or any other entity.
- 1.54. "ENVIRONMENT" shall mean soil, surface water, groundwater, land, stream, sediment, surface or subsurface strata, ambient air, physical structures and equipment, and where radon gas is present, the interior air of buildings.
- 1.55. "ENVIRONMENTAL LAWS" shall mean any federal, state or local law, rule or regulation (both present and future) dealing with the use, generation, treatment, storage, disposal, or abatement of Hazardous Materials, including, but not limited to, (i) the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. Section 9601 et seq., as amended and (ii) the regulations promulgated thereunder, from time to time.
- 1.56. "ENVIRONMENTAL LIABILITIES" shall have the meaning given such term in Section 4.3.2.
- 1.57. "ENVIRONMENTAL NOTICE" shall have the meaning given such term in Section 4.3.1.  $\,$
- 1.58. "ENVIRONMENTAL OBLIGATION" shall have the meaning given such term in Section 4.3.1.

1.60. "EXCESS FF&E" shall have the meaning given such term in paragraph (b) of Schedule 22.2.

-----

- 1.61. "EXCESS FF&E COST" shall have the meaning given such term in paragraph (e) of Schedule 22.2.
- 1.62. "EXCESS FF&E LEASE" shall have the meaning given such term in paragraph (b) of Schedule 22.2.

-----

- 1.63. "EXCESS FF&E LEASEHOLD INTEREST" shall have the meaning given such term in paragraph (h) of Schedule 22.2.
- 1.64. "EXCESS FF&E LEASEHOLD INTEREST TRANSFER" shall have the meaning given such term in paragraph (h) of Schedule 22.2.
- 1.65. "EXCESS FF&E LEASEHOLD INTEREST TRANSFER PRICE" shall have the meaning given such term in paragraph (h) of Schedule 22.2.
- 1.66. "EXCESS FF&E NOTICE" shall have the meaning given such term in paragraph (b) of Schedule 22.2.  $\,$
- 1.67. "EXCESS FF&E REIMBURSEMENT AMOUNT" shall have the meaning given such term in paragraph (i) of Schedule 22.2.
- 1.68. "EXCESS FF&E REPURCHASE" shall have the meaning given such term in paragraph (g) of Schedule 22.2.
- 1.69. "EXCESS FF&E REPURCHASE PRICE" shall have the meaning given such term in paragraph (g) of Schedule 22.2.
- 1.70. "EXCESS FF&E VALUE" shall have the meaning given such term in paragraph (e) of Schedule 22.2.
- 1.71. "EXISTING CONDITION" shall mean any physical condition existing at the Leased Property as of the Commencement Date.
  - 1.72. "FACILITY" shall have the meaning given such term in Section 2.1(b).
- 1.73. "FACILITY MORTGAGE" shall mean, with respect to the Leased Property, any Lien placed by Landlord upon the Leased Property in accordance with Article 20.
  - 1.74. "FACILITY MORTGAGEE" shall mean the holder of a Facility Mortgage.
- 1.75. "FACILITY MORTGAGEE AGREEMENT" shall have the meaning given such term in Section 21.4.
- 1.76. "FACILITY TRADE NAME" shall mean, with respect to the Facility, any name under which Tenant has conducted the business of operating the Facility at any time during the Term.

- 1.77. "FAIR MARKET RENTAL" shall mean, with respect to the Leased Property, the rental which a willing tenant not compelled to rent would pay a willing landlord not compelled to lease for the use and occupancy of the Leased Property on the terms and conditions of this Lease for the term in question, assuming Tenant is not in default hereunder and determined by agreement between Landlord and Tenant or, failing agreement, in accordance with the appraisal procedures set forth in Article 19.
- 1.78. "FAIR MARKET VALUE" shall mean, with respect to the leasehold estate hereunder, the price that a willing transferee not compelled to accept a transfer would pay a willing transferor not compelled to transfer for the leasehold estate hereunder (without taking into account any reduction in value resulting from any indebtedness to which the Leased Property is subject), determined by agreement between Landlord and Tenant or, failing agreement, in accordance with the appraisal procedures set forth in Article 19.
- 1.79. "FF&E" shall mean furniture, furnishings, fixtures, Soft Goods, Case Goods, signage and equipment at the Facility (including, without limitation, facsimile machines, communication systems, audio visual equipment, and all computer and other equipment needed for the reservation system and the property management system, and all other electronic systems needed for the Facility, from time to time, as well as similar systems based on other technologies which may be developed in the future), or, during any period when the Management Agreement is in effect, to the extent in conflict with the foregoing, "FF&E" as defined in the Management Agreement; provided, however, that FF&E shall not include Tenant's Personal Property.
- 1.80. "FF&E ADJUSTMENT" shall have the meaning given such term in paragraph (e) of Schedule 22.2.

-----

- 1.81. "FF&E ESTIMATE" shall have the meaning given such term in Section 17.3(c).
- 1.82. "FF&E LIMITATION" shall have the meaning given such term in paragraph (a) of Schedule 22.2.
  - 1.83. "FF&E RESERVE" shall have the meaning given such term in Section 4.4.
- 1.84. "FINANCIAL OFFICER'S CERTIFICATE" shall mean, as to any Person, a certificate of the chief financial officer, controller, chief accounting officer or treasurer of such Person (or, in the case of any financial statements other than the CCC Consolidated Financial Statements, the designee of any such officer, duly authorized (including by providing Notice of such designee to the other party)), accompanying the financial statements required to be delivered by such Person pursuant to Section 17.2, in which such officer shall certify (a) that such statements have been properly prepared in accordance with GAAP and fairly present, in all material respects, the financial condition of such Person as of the dates thereof, and the results of its operations for the periods covered thereby (except that, in the case of financial statements delivered pursuant to Sections 17.2(a) and 17.2(c), the certificate shall state the extent to which such financial statements are not in accordance with GAAP), (b) that such officer has no knowledge of any monetary Default or Event of Default hereunder, and (c) the then-current Minimum Net Worth,

Consolidated Coverage Ratio, and the maximum amount of liability of the Guarantors under the Guarantee.

- 1.85. "FIRST CLASS OPERATING STANDARDS" shall mean both the operational standards (for example, staffing, amenities offered to guests, advertising, etc.) and the physical standards (for example, the quality, condition, utility and age of the personal property, etc.) of comparable full-service hotels in the hotel system of which the Facility is a part (e.g., the Marriott hotel system or the Ritz-Carlton hotel system), as such operational and physical standards may change from time to time (provided, however, that First Class Operating Standards shall in no event be lower than the operational and physical standards, as of the date in question, of comparable "quality segment" (as such term was being used as of the effective date of the Management Agreement) full-service hotels in other full-service hotel systems), or, during any period when the Management Agreement is in effect, to the extent in conflict with the foregoing, the operating standards required pursuant to the Management Agreement.
- 1.86. "FIRST TIER FOOD AND BEVERAGES SALES PERCENTAGE" shall have the meaning given such term in Section 3.1.2(b)(2).
- 1.87. "FIRST TIER OTHER INCOME PERCENTAGE" shall have the meaning given such term in Section 3.1.2(b)(2).
- 1.88. "FIRST TIER ROOM REVENUE PERCENTAGE" shall have the meaning given such term in Section 3.1.2(b)(2).
- 1.89. "FIRST YEAR FF&E ADJUSTMENT" shall have the meaning given such term in paragraph (e)(i) of Schedule 22.2.
- 1.90. "FISCAL YEAR" shall mean the fiscal year used by Manager, which now ends at midnight on the Friday closest to December 31 in each calendar year and begins on the Saturday immediately following said Friday. If the fiscal year used by Manager is changed in the future by Manager or if the Management Agreement is no longer in effect, appropriate adjustment to this Lease's reporting and accounting procedures shall be made; provided, however, that no such change or adjustment shall alter the Term of this Lease, or in any way reduce the amount of Rent or other payments due to or by Landlord hereunder, or otherwise adversely affect Landlord's or Tenant's rights or obligations under this Lease.
- 1.91. "FIXED ASSET SUPPLIES" shall mean supply items included within "Property and Equipment" under the Uniform System, including linen, china, glassware, silver, uniforms and similar items.
  - 1.92. "FIXTURES" shall have the meaning given such term in Section 2.1(d).
- 1.93. "FOOD AND BEVERAGES SALES" shall mean (a) gross revenue from the sale of food and beverages that are prepared at the Facility and sold or delivered on or off the Facility by or on behalf of Tenant (including, without limitation, revenues from mini-bars), whether for cash or for credit, including in respect of guest rooms, banquet rooms, meeting rooms and other similar rooms and (b) gross revenue from the rental of banquet, meeting

and other similar rooms. Food and Beverages Sales shall be determined in a manner consistent with the Uniform System. Food and Beverages Sales shall not include the following:

- (i) sales by Tenant's subtenants, licensees and concessionaires, the rent, license fees or concession payments from which are included in Gross Revenues, and any rent, license fees, concession payments or other amounts paid to Tenant by any such subtenants, licensees and concessionaires;
  - (ii) vending machine sales;
- (iii) any gratuities or service charges added to a customer's bill or statement in lieu of a gratuity that is paid directly to Facility employees, to the extent actually paid to such employees;
  - (iv) credits, rebates or refunds; or
- $\mbox{\ensuremath{(v)}}$  sales taxes or taxes of any other kind imposed on the sale of food or beverages.
- 1.94. "GAAP" shall mean generally accepted accounting principles consistently applied.
- 1.95. "GOVERNMENT AGENCIES" shall mean any court, agency, authority, board (including, without limitation, environmental protection, planning and zoning), bureau, commission, department, office or instrumentality of any nature whatsoever of any governmental unit of the United States, the State, any county, or any political subdivision of any of the foregoing, whether now or hereafter in existence, having jurisdiction over Tenant or the Leased Property or any portion thereof.
- 1.96. "GROSS REVENUES" shall mean Room Revenues, Food and Beverages Sales, and Other Income, and all revenues and receipts of every kind derived from operating the Facility and parts thereof, including, but not limited to:
- (i) revenues and receipts from both cash and credit transactions, before commissions and discounts for prompt or cash payments, from rental of stores, offices, meeting, exhibit or sales space of every kind;
- (ii) license, lease and concession fees and rentals (not including gross receipts of the licensees, lessees and concessionaires under licenses, leases and concessionaire agreements);
  - (iii) revenues and receipts from vending machines;
  - (iv) health club membership fees;
- (v) sales of merchandise (other than proceeds from the sale of FF&E no longer necessary to the operation of the Facility);

- (vi) service charges, to the extent not paid to the employees at the Facility as, or in lieu of, gratuities; and
- (vii) proceeds, if any, from business interruption or other loss of income insurance;

provided, however, that Gross Revenues shall not include the following:

- (i) banquet service charges to the extent actually paid to employees;
- (iii) federal, state or municipal excise, sales, use or similar taxes collected directly from patrons or guests or included as part of the sales price of any goods or services;
- (iv) insurance proceeds (other than proceeds from business interruption or other loss of income insurance);
  - (v) condemnation proceeds (other than for a temporary taking);
- (vi) any proceeds from any sale of the Facility or from the refinancing of any debt encumbering the Facility;
- (vii) proceeds from the disposition of FF&E no longer necessary for the operation of the Facility; or
- (viii) interest which accrues on amounts deposited in either the FF&E Reserve or any escrow accounts that are established for real estate and personal property taxes, levies, assessments and similar charges in accordance with the terms of the Management Agreement.
- 1.97. "GUARANTEE" shall have the meaning given such term in Section 21.3.4 (A).
- 1.98. "GUARANTORS" shall have the meaning given such term in Section 21.3.4(A).
- 1.99. "HAZARDOUS MATERIALS" shall mean any substance or material containing one or more of any of the following: "hazardous material," "hazardous waste," "regulated substance," "petroleum," "pollutant," "contaminant," or "asbestos," as such terms are defined in any applicable Environmental Law, in such concentration(s) or amount(s) as may impose cleanup, removal, monitoring or other responsibility under any applicable Environmental Law, or which may present a significant risk of harm to guests, invitees or employees of the Facility.
- 1.100. "HOST O.P." shall mean Host Marriott, L.P., a Delaware limited partnership and the operating partnership of Host REIT.

- 1.101. "HOST REIT" shall mean either Host Marriott Corporation, a Delaware corporation, or Host Marriott Corporation, a Maryland corporation, which is intended to be the successor by merger to Host Marriott Corporation, a Delaware corporation, or their successors or assigns.
- 1.102. "HOTEL BUSINESS" means the business of owning, developing, constructing, leasing, operating, managing or franchising, either directly or through a contractual arrangement with a third party, hotels having, in the aggregate, five thousand (5,000) or more rooms, but excluding hotels leased, as a lessee, from a real estate investment trust or other passive owner and operated on behalf of such lessee by an unaffiliated third party manager.
- 1.103. "IMPOSITIONS" shall mean, collectively, all taxes (including, without limitation, all ad valorem, sales and use, single business, gross receipts, transaction, privilege, rent or similar taxes as the same relate to or are imposed upon Tenant or the business conducted upon the Leased Property) not included in Landlord Obligations, water, sewer or other rents and charges, excises, tax levies, fees (including, without limitation, license, permit, inspection, authorization and similar fees) and all other governmental charges not included in Landlord Obligations, in each case whether general or special, ordinary or extraordinary, foreseen or unforeseen, of every character in respect of the Leased Property or the business conducted thereon by Tenant (including all interest and penalties thereon due to any failure in payment by Tenant), which at any time during or in respect of the Term hereof may be assessed or imposed on or in respect of or be a lien upon (a) Landlord's interest in the Leased Property; (b) the Leased Property, any part thereof, any rent therefrom, or any estate, right, title, or interest therein; or (c) any occupancy, operation, use or possession of, or sales from, or activity conducted on or in connection with the Leased Property or the leasing or use of the Leased Property or any part thereof by Tenant.
- 1.104. "INDEBTEDNESS" shall mean all obligations, contingent or otherwise, which in accordance with GAAP should be reflected on the obligor's balance sheet as debt.
- 1.105. "INDEPENDENT MEMBER" means a Person who is not, and has not within the past five (5) years been, (i) an officer, director, employee, partner, member, manager, stockholder or beneficial-interest holder of Tenant; (ii) an officer, director, employee, partner, member, beneficial-interest holder, or stockholder of any "Affiliate" (as defined below) of Tenant; or (iii) a spouse, parent, sibling, or child of any Person described in (i) or (ii) of this section; provided, however, that a Person shall not be deemed to be a director or member of an Affiliate solely by reason of such Person being a director, manager, or member of a single-purpose entity that would otherwise be deemed to be an Affiliate because they are under common control. For the purpose of this definition alone, "Affiliate" means any Person (x) which owns beneficially, directly or indirectly, more than ten percent (10%) of the outstanding equity interest in Tenant or which is otherwise in control of Tenant; (y) of which more than ten percent (10%) of the outstanding voting securities are owned beneficially, directly or indirectly, by any Person described in clause (x) above; or (z) which is controlled by, or under common control with, any Person or entity described in

- clause (x) above; the terms "control" and "controlled by" shall have the meanings assigned to them in Rule 405 under the Securities Act of 1933.
- 1.106. ["INITIAL FF&E LEASE" shall have the meaning given such term in paragraph (a) of Schedule 22.2.]
- 1.107. "INSURANCE REQUIREMENTS" shall mean all terms of any insurance policy required by this Lease and all requirements of the issuer of any such policy.
- 1.108. "INTANGIBLE ASSET" shall mean, with respect to any Person, a long-lived asset that is useful in the operations of such Person, that is not directly used in revenue generation and that is not held for sale, and is without physical qualities, including but not limited to patents, copyrights and goodwill, but excluding capitalized costs associated with the acquisition of brand names, franchises and trademarks, franchise agreements and management agreements.
- 1.109. "INVENTORIES" shall have the definition given it in the Uniform System, such as provisions in storerooms, refrigerators, pantries and kitchens; beverages in wine cellars and bars; other merchandise intended for sale; fuel; mechanical supplies; stationery; and other expensed supplies and similar items.
- 1.110. "LABOR INDEX" shall mean Employment Cost Index, Compensation, Private Industry, Services Industries, Not Seasonally Adjusted as published by the U.S. Department of Labor, Bureau of Labor Statistics.
  - 1.111. "LAND" shall have the meaning given such term in Section 2.1(a).
- 1.112. "LANDLORD" shall have the meaning given such term in the preamble to this Lease.
- 1.113. "LANDLORD DEFAULT" shall have the meaning given such term in Section 14.2.
- 1.114. "LANDLORD INDEMNITEE" shall have the meaning set forth in Section 4.3.3.
  - 1.115. "LANDLORD LIEN" shall have the meaning set forth in Section 7.2.
- 1.116. "LANDLORD OBLIGATIONS" shall mean (i) Real Estate Taxes, (ii) any tax based on net income imposed on Landlord, (iii) any net revenue tax of Landlord, (iv) any transfer fee or other tax imposed with respect to the sale, exchange, financing, mortgaging, or other disposition by Landlord of the Leased Property or the proceeds thereof, (v) any sales, use, gross receipts or other similar tax relating to the leasing of the FF&E by Landlord to Tenant pursuant to this Lease, (vi) any single business tax, gross receipts tax, transaction, privilege, rent or similar taxes as the same relate to or are imposed upon Landlord, including, without limitation, any tax imposed on Landlord as a result of this Lease, (vii) any interest or penalties imposed on Landlord as a result of the failure of Landlord to file any return or report timely and in the form prescribed by law or to pay any tax or imposition, except to the extent such failure is a result of a breach by Tenant of its obligations pursuant to Section 3.1.3, (viii) any taxes or fees imposed on Landlord that are

a result of Landlord not being considered a "United States person" as defined in Section 7701(a)(30) of the Code, (ix) any taxes or fees that are enacted or adopted as a substitute for any tax that would not have been payable by Tenant pursuant to the terms of this Lease, (x) any taxes, fees or amounts required to be paid to discharge Liens (a) imposed as a result of a breach of covenant or representation by Landlord in any agreement with Tenant governing Landlord's conduct or operation or (b) as a result of the gross negligence or willful misconduct of Landlord in connection with its ownership of the Leased Property or the performance of its obligations hereunder, or (c) securing any indebtedness or obligations of Landlord, or (d) imposed in connection with any other Landlord Obligations, (xi) any sales, use, occupancy or other similar tax imposed in connection with payments made pursuant to this Lease, (xii) all rent and other sums due from Landlord pursuant to any ground lease affecting the Leased Property or any portion thereof, (xiii) all premiums for insurance for which Landlord is responsible pursuant to Article 9 hereof, (xiv) all amounts payable by Landlord under or pursuant to the Management Agreement with respect to the Retained Obligations and/or the Continuing Obligations, (xv) any interest and/or penalties incurred by Tenant as a result of Landlord's failure to forward any invoices, assessment notices or other bills relating to Impositions to Tenant, and (xvi) equipment rental costs that are treated as a "Deduction" under the Management Agreement. "Landlord Obligations" shall not include any sales, use, gross receipts, occupancy, single business, transaction, privilege, rent, ad valorem or other tax not specifically enumerated herein.

- 1.117. "LEASE YEAR" shall mean any consecutive annual period starting on the Commencement Date and ending on the day prior to the anniversary thereof; provided that if the Commencement Date is not the first (1st) day of a calendar month, then the first (1st) Lease Year shall end on the last day of the calendar month in which the Commencement Date occurs.
- 1.118. "LEASED PROPERTY" shall have the meaning given such term in Article 2.
- 1.119. "LEGAL REQUIREMENTS" shall mean all applicable federal, state, county, municipal and other governmental statutes, laws, rules, orders, regulations, and ordinances, and all applicable judicial, administrative and regulatory judgments, decrees and injunctions, affecting Tenant or the Leased Property or the maintenance, construction, alteration or operation thereof, whether now or hereafter enacted or in existence, including, without limitation, (a) all permits, licenses, certificates of need, authorizations and regulations necessary to operate the Leased Property for its Primary Intended Use and (b) all covenants, agreements, restrictions and encumbrances contained in any instruments either of record or actually known to Tenant (other than encumbrances hereafter created by Landlord without the consent of Tenant) at any time in force affecting the Leased Property, including those which may (i) require material repairs, modifications or alterations in or to the Leased Property or (ii) in any way materially adversely affect the use and enjoyment thereof, but excluding any requirements arising as a result of Landlord's status as a real estate investment trust other than those set forth in Article 22 hereof.
- 1.120. "LENDING INSTITUTION" shall mean any insurance company, federally insured commercial or savings bank, national banking association, savings and loan

association, investment banking firm, employees' welfare, pension or retirement fund or system, syndicated lenders' group, commercial finance company, leasing company, corporate profit sharing or pension trust, college or university, or real estate investment trust, including any corporation qualified to be treated for federal tax purposes as a real estate investment trust, such trust having a Tangible Net Worth of at least \$100,000,000.

- 1.121. "LICENSES" shall have the meaning given such term in Section  $5.3(\mbox{d})$  (i).
- 1.122. "LIEN" shall mean any mortgage, security interest, pledge, collateral assignment, or other encumbrance, lien or charge of any kind (including, without limitation, any easements, covenants, conditions and restrictions), or any transfer of any property or assets for the purpose of subjecting the same to an encumbrance, lien or charge securing the payment of Indebtedness or performance of any other obligation in priority to payment of any Person's general creditors.
- 1.123. "LOCKBOX ACCOUNT" shall have the meaning given such term in Section 3.5(b).
- 1.124. "LOCKBOX NOTICE DATE" shall have the meaning given such term in Section 3.5(a).
- 1.125. "LOCKBOX PERIOD" shall have the meaning given such term in Section 3.5(a).
- 1.126. "MAJOR CAPITAL EXPENDITURE" shall mean (a) any Capital Expenditure expected to result in a significant change in the use of any portion of the Facility intended to materially alter or enhance an existing use (including, by way of example and not limitation, the conversion of a typical hotel dining facility to a nationally-known, premium-quality restaurant); (b) any and all Capital Expenditures with respect to one or more new revenue-generating buildings, rooms, facilities, amenities, or structures constituting any portion of the Facility (including, by way of example and not limitation, the construction of a new wing or new story), or the renovation, material expansion, construction, or conversion of existing improvements of the Facility in order to increase the room capacity of the Facility, or to provide a functionally new facility needed to provide services not previously offered; or (c) any and all Capital Expenditures which, individually or in a series of related expenditures, have a total cost of completion in excess of the lesser of (i) seven and one-half percent (7.5%) of Gross Revenues for the Fiscal Year ending immediately prior to such expenditure, or (ii) \$5,000,000. Notwithstanding the foregoing, in no event shall Major Capital Expenditures include any room, lobby, or ballroom refurbishments in the ordinary course or any other Routine Capital Expenditures.
- 1.127. "MANAGEMENT AGREEMENT" shall mean the Management Agreement described in Section 21.3.1.
  - 1.128. "MANAGER" shall have the meaning given such term in Section 21.3.1.
- 1.129. "MARKET LEASING FACTOR" shall have the meaning given such term in paragraph (e) of Schedule 22.2.

-----

- 1.130. "MEASUREMENT DATE" shall mean September 30th of each year during the
- 1.131. "MINIMUM NET WORTH" means, with respect to the last day of any Accounting Period, a Tangible Net Worth (calculated on a book value basis), excluding amounts held in any lockbox or cash collateral account under this Lease, any other Pool Lease, or any Other Lease, the Pooling Agreement or any pooling and security agreement relating to Other Leases, immediately after the "Distribution Date" (as defined in the Pooling Agreement) for such Accounting Period, equal to one hundred fifty percent (150%) of the aggregate of the "CCC Limits of Liability" (as defined in the Guarantee) under the Guarantee and the "Other Guarantees" (as defined in the Pooling Agreement) from time to time; provided that the Minimum Net Worth shall be reduced by one hundred fifty percent (150%) of the total amount of cash held in any lockbox or cash collateral account under this Lease, any other Pool Lease, or any Other Lease, the Pooling Agreement or any pooling and security agreement relating to Other Leases, immediately after the Distribution Date for such Accounting Period.
- 1.132. "MINIMUM RENT" shall mean for each Fiscal Year the amount set forth on Schedule 3.1.1.
- 1.133. "NONCOMPETITION AGREEMENT" shall have the meaning given such term in Section 23.15.
- 1.134. "NON-CONTROLLED SUBSIDIARY" shall mean any Person (other than a Subsidiary of Host O.P.) in which Host O.P. owns, directly or indirectly, at least 90% of the economic interest.
  - 1.135. "NOTICE" shall mean a notice given in accordance with Section 23.10.
- 1.136. "OFFICER'S CERTIFICATE" shall mean a certificate signed by an authorized officer of Tenant.
- 1.137. "OPCO" shall mean any wholly owned subsidiary of CCC that owns, directly or through one or more wholly owned subsidiaries, the entire beneficial equity interest in Tenant.
- 1.138. "OPERATING BUDGET" shall have the meaning given such term in Section 17.3.
- 1.139. "OPERATING BUDGET SUMMARY" shall have the meaning given such term in Section 17.3.
- 1.140. "OTHER INCOME" shall mean all revenues, receipts, and income of any kind derived directly or indirectly by Tenant from or in connection with the Facility and included in Gross Revenues other than Room Revenues and Food and Beverages Sales (including, without limitation, gross revenue from rental income, license fees, concession payments, or other amounts received from subtenants, licensees and concessionaires, but excluding all exclusions from Gross Revenues set forth in the definition thereof.

- 1.141. "OTHER LEASES" shall mean any lease of a full service hotel, other than this Lease or any other Pool Lease, under which Host O.P. or a Subsidiary of Host O.P. is the lessor and a Subsidiary of CCC is lessee and that is the subject of a guarantee by CCC and a pooling and security agreement in effect on the date hereof substantially similar to the Guarantee and the Pooling Agreement.
- 1.142. "OTHER TENANTS" shall mean the tenants under the Pool Leases (other than this Lease) and the Other Leases.
- 1.143. "OVERDUE RATE" shall mean, on any date, a per annum rate of interest equal to the lesser of the Prime Rate plus two (2) percentage points and the maximum rate then permitted under applicable law.
  - 1.144. "OWNER" shall have the meaning given such term in Section 21.3.1.
- 1.145. "PERCENTAGE INCREASE IN CPI" shall mean, measured as of each Measurement Date, the percentage increase, if any, in the Consumer Price Index over the twelve (12) months preceding the Measurement Date.
- 1.146. "PERCENTAGE INCREASE IN THE LABOR INDEX" shall mean, measured as of each Measurement Date, the percentage increase, if any, in the Labor Index over the twelve (12) months preceding the Measurement Date.
- 1.147. "PERCENTAGE RENT" shall mean the amount of Rent (excluding Additional Charges) payable pursuant to Section 3.1.1 with respect to one or more Accounting Periods or a Fiscal Year, as the case may be, in excess of Minimum Rent payable for such period.
- 1.148. "PERCENTAGE RENT SCHEDULE" shall have the meaning given such term in Section 3.1.3(a).
- 1.149. "PERIOD REPORT" shall have the meaning given such term in Section 17.2(f).
- 1.150. "PERMITTED DEBT" shall have the meaning given such term in Section 21.6.3.
- 1.151. "PERMITTED LIENS" shall mean (a) this Lease; (b) all rights, restrictions, and easements of record as of the Commencement Date; (c) the lessor's interest in any Excess FF&E leased to Tenant pursuant to [the Initial FF&E Lease or] an Excess FF&E Lease and the lessor's interest in any other furniture, fixtures or equipment leased to Tenant under any of the equipment leases included in the Assigned Agreements or entered into by Tenant during the Term; (d) security interests securing the purchase price of equipment or personal property acquired by Tenant before or after the Commencement Date (including, but not limited to, Excess FF&E owned by Tenant); provided, however, that (i) any such Lien described in this clause (d) shall at all times be confined solely to the asset in question, and (ii) the aggregate principal amount of Indebtedness secured by any such Lien described in this clause (d) shall not exceed the cost of acquisition of the property subject thereto; and (e) any other Liens as may have been consented to in writing by Landlord, and, if required pursuant to the Management Agreement, Manager, from time to time, including, without limitation, those granted in accordance with Section 5.5 hereof.

- 1.152. "PERSON" shall mean any individual or Entity, and the heirs, executors, administrators, legal representatives, successors and assigns of such Person where the context so requires.
- 1.153. "POOLING AGREEMENT" shall have the meaning given such term in Section 23.15.
  - 1.154. "POOL LEASE" shall mean any lease subject to the Pooling Agreement.
- 1.156. "PRIMARY INTENDED USE" shall have the meaning given such term in Section 4.1.1.
- 1.157. "PRIME RATE" shall mean the "prime rate" as published in the "Money Rates" section of The Wall Street Journal; however, if such rate is, at any time during the Term, no longer so published, the term "Prime Rate" shall mean the average of the prime interest rates that are announced, from time to time, by the three (3) largest banks (by assets) headquartered in the United States which publish a "prime rate."
- 1.158. "QUALIFIED AFFILIATE" shall mean any Person directly or indirectly wholly owned and controlled by CCC or  ${\tt OpCo.}$
- 1.159. "QUALIFIED APPRAISER" shall mean an appraiser who is not in control of, controlled by or under common control with either Landlord or Tenant or any Affiliate of Landlord or Tenant and has not been an employee of Landlord or Tenant or any Affiliate of Landlord or Tenant within the past two (2) years, who is qualified to appraise commercial real estate in the State and is a member of the American Institute of Real Estate Appraisers (or any successor association or body of comparable standing if such institute is not then in existence).
- 1.160. "REAL ESTATE TAXES" shall mean all real estate taxes, including general and special assessments and impact fees, if any, which are imposed upon or relate to the Land and any improvements thereon, including, without limitation, the following:
- (a) any franchise, corporate, estate, inheritance, succession, or capital levy imposed on Landlord;
- (b) special assessments (regardless of when due or whether they are paid as a lump sum or in installments over time) imposed because of facilities that are constructed by or on behalf of the assessing jurisdiction (i.e., roads, sidewalks, sewers, culverts, etc.), which directly benefit the Facility (regardless of whether or not they also benefit other buildings);
- (c) impact fees (regardless of when due or whether they are paid as a lump sum or in installments over time) that are required of Landlord as a condition to the issuance of site plan approval, zoning variances or building permits: and

- (d) tax-increment financing or similar financing whereby the municipality or other taxing authority has assisted in financing the construction of the Facility by temporarily reducing or abating normal impositions in return for substantially higher levels of impositions at later dates.
- 1.161. "RELATED AGREEMENTS" shall have the meaning given such term in Section 23.15.
- 1.162. "RENT" shall mean, collectively, Minimum Rent, Percentage Rent, and Additional Charges attributable to the Term of this Lease.
- 1.163. "REPLACEMENT COST" shall have the meaning given such term in Section 9.3.
- 1.164. "RETAINED OBLIGATIONS" shall have the meaning given such term in Section 21.3.1.
- 1.165. "REVENUE DATA PUBLICATION" shall mean Smith's STAR Report, a monthly publication distributed by Smith Travel Research, Inc., of Gallatin, Tennessee, or an alternative source, reasonably satisfactory to both parties, of data regarding the REVPAR of hotels in the general trade area of the Facility. If such Smith's STAR Report is discontinued in the future, or ceases to be a satisfactory source of data regarding the REVPAR of various hotels in the general trade area of the Facility (as provided in the Management Agreement), Tenant shall select, or shall cause Manager to select, an alternative source in accordance with the Management Agreement.
- 1.166. "REVENUES COMPUTATION" shall have the meaning given such term in Section 3.1.2(a).
- 1.167. "REVPAR" shall mean (i) the term "revenue per room" as defined by the Revenue Data Publication or (ii) if the Revenue Data Publication is no longer being used (as more particularly set forth in the definition of "Revenue Data Publication"), the aggregate gross room revenues of the hotel in question for a given period of time divided by the total room nights of such hotel for such period. If clause (ii) of the preceding sentence is being used, a "room" shall be a hotel guestroom which is keyed as a single unit and shall include rooms that are temporarily unavailable due to maintenance or ongoing renovation work.
- 1.168. "ROOM REVENUES" shall mean gross revenues from the rental of guest rooms, whether to individuals, groups, or transients at the Facility, determined in a manner consistent with GAAP, excluding the following:
- (a) the amount of all credits, rebates or refunds to customers, guests or patrons;
- (b) all sales taxes or any other taxes imposed on the rental of such guest rooms;
- (c) any fees collected for amenities including, but not limited to, telephone, room service, laundry, movies or concessions.

- 1.169. "ROUTINE CAPITAL EXPENDITURES" shall mean exterior and interior repainting, resurfacing of building walls, floors, roof and parking areas, replacing folding walls, and other on-going routine Capital Expenditures the cost of which is to be paid for by the Manager out of the FF&E Reserve pursuant to the Management Agreement.
  - 1.170. "SEC" shall mean the Securities and Exchange Commission.
- 1.171. "SECOND TIER FOOD AND BEVERAGES SALES PERCENTAGE" shall have the meaning given such term in Section 3.1.2(b)(2).
- 1.172. "SECOND TIER OTHER INCOME PERCENTAGE" shall have the meaning given such term in Section 3.1.2(b)(2).
- 1.173. "SECOND TIER ROOM REVENUE PERCENTAGE" shall have the meaning given such term in Section 3.1.2(b)(2).
- 1.174. "SECURITY AGREEMENT" shall have the meaning given such term in Section 7.2.
- "SINGLE PURPOSE" means, with respect to a Person, that such Person, (A) at all times since its formation, except as otherwise permitted in or contemplated by this Agreement (i) has been a duly formed and existing limited partnership, limited liability company, or corporation, as the case may be; (ii) has observed all customary formalities regarding its partnership, limited liability company or corporate existence; (iii) has maintained financial statements, accounting records, and other partnership, limited liability company, or corporate documents separate from those of any other Person (provided that nothing shall prohibit such Person from being included in the consolidated financial statements or tax group of another Person); (iv) has not commingled its assets with those of any other Person; (v) has paid its own liabilities out of its own funds, including funds contributed to its capital by its equity holders, and all such capital contributions have been reflected properly in its books and records; (vi) has allocated fairly and reasonably any overhead for shared office expenses; (vii) has identified itself in all dealings with the public, under its own name and as a separate and distinct entity (provided that nothing shall prohibit such Person from engaging an agent to represent such Person with respect to tenants, vendors, and other parties, in accordance with standard industry practice); (viii) has not identified itself as being a division or part of any other Person; (ix) has not identified any other Person as being a division or part of such Person; (x) has corrected any known misunderstanding regarding its separate identity; (xi) has been adequately capitalized in light of the nature of its business; (xii) has not assumed or guaranteed the obligations of any other Person (other than by virtue of being a general partner of such other Person but only if such other Person is Tenant and provided that this clause shall not be deemed to be violated by reason of joint and several liabilities arising as a matter of law); and (xiii) has not engaged in any other business other than as permitted by this Lease (provided that nothing shall prohibit a general partner from acquiring, owning, and disposing of limited partner interests in any limited partnership, non-managing membership interests in a limited liability company, or debt securities issued by any Person, or the stock of a corporation that is not engaged in any activities other than such activities, provided that no liabilities or other obligations on the part of the

holder of such investments will arise as a result of such investments and the only rights of the holder of such investment are the right to receive payments and the right to vote); (B) such Person agrees to covenants substantially to the effect of Sections 7.1 and 21.6 hereof; and (C) such Person's organizational documents contain restrictions similar to those contained in Sections 5.4 (only paragraphs a, b, c, d and g thereof), 5.5, 5.6, and 5.7 of the Limited Liability Company Operating Agreement of Tenant as in effect on the date hereof.

- 1.176. "SOFT GOODS" shall mean all fabric, textile and flexible plastic products (not including items that are classified as "Fixed Asset Supplies" under the Uniform System), which are used in furnishing the Facility, including, without limitation, carpeting, drapes, bedspreads, wall and floor coverings, mats, shower curtains and similar items.
- 1.177. "SPECIAL FORM" shall have the meaning given such term in Section 9.1(b).
- 1.178. "STATE" shall mean the state, commonwealth, district, or, if the Leased Property is not located in the United States, the country, in which the Leased Property is located.
- 1.179. "SUBSIDIARY" shall mean, as to any Person, (i) any corporation more than fifty percent (50%) of whose stock of any class or classes having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at the time stock of any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time owned by such Person or one or more Subsidiaries of such Person and (ii) any partnership, limited liability company, association, joint venture or other entity in which such Person or one or more Subsidiaries of such Person has more than a fifty percent (50%) equity interest at the time.
- 1.180. "SUCCESSOR LANDLORD" shall have the meaning given such term in Section 20.2.
- 1.181. "SUPERIOR LANDLORD" shall have the meaning given such term in Section 20.2.  $\,$
- 1.182. "SUPERIOR LEASE" shall have the meaning given such term in Section 20.2.  $\,$
- 1.183. "SUPERIOR MORTGAGE" shall have the meaning given such term in Section 20.2.
- 1.184. "SUPERIOR MORTGAGEE" shall have the meaning given such term in Section 20.2.  $\,$
- 1.185. "TANGIBLE NET WORTH" of any Person shall mean, on any date, without duplication, (a) the sum of the shareholders' equity of such Person on a consolidated basis in accordance with GAAP minus (b) the sum of all Intangible Assets (net of accumulated amortization) of such Person and its Subsidiaries, each as shown on the consolidated balance sheets of such Person.

- 1.187. "TENANT" shall have the meaning given such term in the preamble to this Lease.
- 1.188. "TENANT INDEMNITEE" shall have the meaning given such term in Section 4.3.2.
- 1.189. "TENANT'S OPERATING PROFIT" shall equal for any Fiscal Year the amount equal to revenues due to Tenant from the Leased Property after the payment of all expenses relating to the operation or leasing of the Leased Property less Rent paid to Landlord, but in any event not less than \$1.00.
- 1.190. "TENANT'S PERSONAL PROPERTY" shall mean all tangible and intangible personal property now owned, leased or hereafter acquired by Tenant on or after the date hereof, including, without limitation, Working Capital, Excess FF&E owned by Tenant, Excess FF&E Leasehold Interests, and all contracts and agreements to which Tenant is a party used in the operation of the Facility, but expressly excluding (a) FF&E leased hereunder, (b) Excess FF&E leased during the Term hereof by Tenant [and (c) FF&E leased pursuant to the Initial FF&E Lease, if any.]
- 1.191. "TERM" shall have the meaning given such term in Section 2.4, unless sooner terminated pursuant to the terms of this Lease.
- 1.192. "THIRD-PARTY PURCHASER" shall have the meaning given such term in paragraph (b) of Schedule 22.2.
- 1.193. "THIRD TIER FOOD AND BEVERAGES SALES PERCENTAGE" shall have the meaning given such term in Section 3.1.2(b)(2).
- 1.194. "THIRD TIER ROOM REVENUE PERCENTAGE" shall have the meaning given such term in Section 3.1.2(b)(2).
  - 1.195. "TRUSTEE" shall have the meaning given such term in Section 16.1.
- 1.196. "UNAVOIDABLE DELAY" shall mean delay due to strikes, acts of God, force majeure, governmental restrictions or actions (including the revocation or refusal to grant Licenses or permits, where such revocation or refusal is not due to the fault of the party whose performance is excused by reason of such Unavoidable Delay), enemy action, civil commotion, fire, unavoidable Casualty, condemnation or other comparable causes beyond the reasonable control of Landlord or Tenant; provided, however, that lack of funds shall not be deemed a cause beyond the reasonable control of a party, unless such lack of funds is caused by the breach of the other party's obligations under this Lease.
- 1.197. "UNIFORM SYSTEM" shall mean the Uniform System of Accounts for Hotels (9th Revised Edition 1996) as published by the Hotel Association of New York City, Inc., as the same may hereafter be revised.
- 1.198. "UNSUITABLE FOR ITS PRIMARY INTENDED USE" shall mean a state or condition of the Facility such that (a) following any damage or destruction involving the Leased Property, the Leased Property cannot reasonably be expected to be restored within

twelve (12) months following such damage or destruction to substantially the same condition as existed immediately before such damage or destruction or (b) as the result of a partial taking by Condemnation, the Facility cannot be operated on a commercially practicable basis for its Primary Intended Use taking into account, among other relevant factors, the number of usable rooms, the amount of square footage, or the revenues affected by such damage, destruction, or partial taking.

- 1.199. "WORKING CAPITAL" shall have the meaning given such term in Section 4.5.  $\,$
- 1.200. "WORKING CAPITAL NOTE" shall have the meaning given such term in Section 4.5.
- 1.201. "YEAR END PERCENTAGE RENT SCHEDULE" shall mean the Percentage Rent Schedule covering the immediately preceding full Fiscal Year delivered in accordance with Section 17.2(e).

# ARTICLE 2 COLLECTIVE LEASED PROPERTIES AND TERM

#### 2.1 Leased Property.

Upon and subject to the terms and conditions hereinafter set forth, including, but not limited to, the execution, delivery and performance of the terms and conditions of the Consent and Assignment, and subject to the rights of any Facility Mortgagee and any Superior Landlord, Landlord leases to Tenant and Tenant leases from Landlord, effective as of the Commencement Date, all of Landlord's right, title, and interest in and to the following (collectively, the "LEASED PROPERTY"):

(a) those certain tracts, pieces and parcels of land that are more particularly described in Exhibit A attached hereto (together with all and

singular the rights and appurtenances pertaining to such tracts and parcels, including any right, title and interest of Landlord in and to any easements benefiting the Leased Property, adjacent strips or gores, streets, alleys or rights-of-way, and all rights of ingress and egress thereto) (the foregoing are hereinafter referred to collectively as the "LAND");

- (b) all buildings, structures, and other improvements of every kind situated upon the Land, including, but not limited to, all Capital Expenditures, alleyways and connecting tunnels, sidewalks, utility pipes, conduits and lines (onsite and offsite), parking areas and appurtenant roadways, all luxury apartments, all swimming pools, restaurants, hotel rooms, lounges, and various other guest and spa facilities (collectively, the "FACILITY");
- (c) all easements, rights and appurtenances relating to the Land and the Facility;
- (d) all equipment, machinery, and other items of property, now or hereafter permanently affixed to or incorporated into the Facility, including, without limitation, all lift systems, furnaces, boilers, heaters, electrical equipment, heating, plumbing, lighting, ventilating, refrigerating, incineration, air and water pollution control,

waste disposal, air-cooling and air-conditioning systems and apparatus, sprinkler systems, and fire and theft protection equipment, all of which, to the maximum extent permitted by law, are hereby deemed by the parties hereto to constitute real estate, together with all replacements, modifications, alterations and additions thereto, but specifically excluding Tenant's Personal Property (collectively, the "FIXTURES"); and

- (e) the FF&E owned by Landlord located in or on the Facility or stored offsite and designated for use in connection with any present or future occupation or operation of the Facility, together with all replacements, modifications, alterations and additions thereto, but specifically excluding any Excess FF&E not owned by Landlord and Tenant's Personal Property.
  - 2.2 Assignment and Assumption of Contracts; Initial Transaction.
- (a) Effective upon the Commencement Date, Landlord hereby transfers and assigns to Tenant and Tenant assumes and covenants to perform all of Landlord's obligations under, the following agreements and contracts to which the Leased Property remains subject on the Commencement Date, to the maximum extent assignable pursuant to applicable law (the "ASSIGNED AGREEMENTS"):
- (i) All contracts for the use or occupancy of guest rooms and/or the meeting, dining, banquet, and spa and health facilities of the Facility;
- (ii) All service and maintenance contracts, equipment leases, purchase orders and other contracts pertaining to the ownership, maintenance, operation, provisioning or equipping of the Facility, including warranties and guaranties relating thereto;
- $\,$  (iii) All Licenses and permits used in or relating to the ownership, occupancy or operation of any part of the Facility;
- (iv) Any developer's, declarant's, or owner's interests under any operating agreements or reciprocal easement agreements or other similar agreements affecting and/or benefiting the Facility;
- (v) Landlord's interest as owner under the Management Agreement as and to the extent provided in the Consent and Assignment; and
- (vi) All leases of space (including any security deposits held by Landlord pursuant thereto, which will be paid over to Tenant by check on the Commencement Date, or credited to Tenant against the cost of the Working Capital sold to Tenant pursuant to Section 4.5) in the Facility to tenants thereof, excluding those leases set forth on Schedule 2.2(a)(vi) attached hereto.

This Lease is executed by Landlord and accepted by Tenant on the understanding that Tenant will and does hereby assume and agree to perform all of Landlord's obligations arising from and after the Commencement Date under all of the Assigned Agreements (except to the extent otherwise provided with respect to the Management Agreement in the Consent and Assignment). In connection with the Licenses

and permits described in clause (iii) above and the Management Agreement referred to in clause (v) above, Landlord shall be obligated (a) to take any action that can legally or contractually be taken only by it and not Tenant, (b) not to take any action that is not permitted by, or is contrary to the terms of, such Licenses or permits or the Management Agreement, and (c) to cooperate with Tenant as to actions that must be taken in Landlord's name. Notwithstanding the foregoing but subject to Section 5.3(d)(ii) hereof, such Assigned Agreements shall, without the necessity of further documentation, be deemed reassigned to (and reassumed by) Landlord upon the expiration or earlier termination of the Term. In connection with any reassignment thereof occurring following the expiration or earlier termination of the Term, such reassignment shall not release Tenant from any liability thereunder with respect to the period beginning on the Commencement Date and ending on the date of expiration or earlier termination of the Term.

(b) As between Landlord and Tenant, Landlord shall be entitled to all income and shall be responsible for the payment or settlement of all expenses of the Leased Property accruing prior to the Commencement Date and after the expiration or earlier termination of the Term. Tenant shall act as Landlord's agent for the collection of all such income and shall remit the same to Landlord promptly upon Tenant's receipt thereof. Tenant shall notify Landlord of all such expenses and shall act as Landlord's payment agent for such expenses using funds provided by Landlord from time to time.

## 2.3 Condition of Leased Property.

Tenant acknowledges receipt and delivery of possession of the Leased Subject to the warranty of title hereinafter set forth and the indemnifications by Landlord set forth in Sections 4.3.2, 9.9 and 21.3.3 hereof, Tenant accepts the Leased Property in its "as is" condition, subject to the rights of all occupants and parties in possession, the existing state of title, including all covenants, conditions, restrictions, reservations, mineral leases, easements and other matters of record or that are visible or apparent on the Leased Property, all applicable Legal Requirements, the lien of financing instruments, mortgages and deeds of trust, and such other matters which would be disclosed by an inspection of the Leased Property and the record title thereto or by an accurate survey thereof, and all other Permitted Liens. TENANT REPRESENTS THAT, EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS LEASE, IT IS NOT RELYING ON ANY REPRESENTATION OR WARRANTY OF LANDLORD OR LANDLORD'S AGENTS OR EMPLOYEES WITH RESPECT TO THE CONDITION OF THE LEASED PROPERTY, AND EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS LEASE TENANT WAIVES ANY CLAIM OR ACTION AGAINST LANDLORD IN RESPECT THEREOF. EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS LEASE, LANDLORD MAKES NO WARRANTY OR REPRESENTATION, EXPRESS OR IMPLIED, IN RESPECT OF THE LEASED PROPERTY OR ANY PART THEREOF, EITHER AS TO ITS FITNESS FOR USE, DESIGN OR CONDITION FOR ANY PARTICULAR USE OR PURPOSE OR OTHERWISE, OR AS TO THE QUALITY OF THE MATERIAL OR WORKMANSHIP THEREIN, LATENT OR PATENT, IT BEING AGREED THAT EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS LEASE, ALL SUCH RISKS ARE TO BE BORNE BY TENANT. NOTWITHSTANDING THE FOREGOING, LANDLORD HEREBY WARRANTS ITS [FEE/LEASEHOLD] TITLE TO THE LEASED PROPERTY TO TENANT, SUBJECT TO THE PERMITTED LIENS; PROVIDED,

HOWEVER, THAT IN NO EVENT SHALL THE AMOUNT RECOVERABLE BY TENANT FOR ANY BREACH OF SUCH WARRANTY EXCEED TENANT'S PROPORTIONATE SHARE OF THE NET AMOUNT ACTUALLY COLLECTED BY LANDLORD UNDER LANDLORD'S EXISTING TITLE INSURANCE POLICIES, AFTER TAKING INTO ACCOUNT THE AMOUNTS EXPENDED BY LANDLORD TO RECOVER THE AMOUNT COLLECTED AND THE AMOUNT OF ANY FEES OWED TO MANAGER RESULTING FROM BREACH OF SUCH WARRANTY. TENANT'S PROPORTIONATE SHARE FOR THE PURPOSES OF THE PRECEDING SENTENCE SHALL BE EQUAL TO THE FRACTION, EXPRESSED AS A PERCENTAGE, OBTAINED BY DIVIDING THE FAIR MARKET VALUE OF TENANT'S LEASEHOLD ESTATE AS DETERMINED IN ACCORDANCE WITH SECTION 24.1(b) BELOW BY THE FAIR MARKET VALUE OF THE LEASED PROPERTY AS DETERMINED BY ARTICLE 19. LANDLORD AGREES TO DILIGENTLY PURSUE THE TITLE COMPANY OR COMPANIES PROVIDING THE EXISTING TITLE INSURANCE IN THE EVENT IT IS DETERMINED THAT THERE IS A DEFECT IN LANDLORD'S TITLE TO THE LEASED PROPERTY. TENANT SHALL FULLY COOPERATE WITH LANDLORD IN THE PROSECUTION OF ANY SUCH CLAIMS, IN LANDLORD'S OR TENANT'S NAME, ALL AT LANDLORD'S SOLE COST AND EXPENSE.

#### 2.4 Term.

The term of this Lease (the "TERM") shall commence at 12:01 a.m. on the Commencement Date and shall expire at 11:59 p.m. on the last day of the [tenth (10th)] Lease Year, unless sooner terminated pursuant to the terms of this Lease; provided that, subject to the terms of Article 13, the Term shall also include any period of holding over by Tenant.

## ARTICLE 3

#### 3.1 Rent.

Tenant shall pay to Landlord, in lawful money of the United States of America which shall be legal tender for the payment of public and private debts, without offset, abatement, demand or deduction, Minimum Rent, Percentage Rent and Additional Charges, during the Term, except as hereinafter expressly provided. All payments to Landlord shall be made by wire transfer of immediately available federal funds or by other means reasonably acceptable to Landlord in its sole discretion.

## 3.1.1 Payment of Rent.

On or before the second (2nd) Business Day after the required payment date under the Management Agreement for amounts due from Manager thereunder relating to each Accounting Period (whether or not such payment under the Management Agreement is made on the required payment date), Tenant shall pay to Landlord an amount of Rent equal to the excess, if any, of (A) the greater of (i) the product of (x) the Minimum Rent for the current Fiscal Year (as set forth on Schedule 3.1.1) times (y) a fraction, the numerator of which is the

number of elapsed Accounting Periods in such Fiscal Year (including the

Accounting Period with respect to which payment of Rent is being made) and the denominator of which is 13 (or 12, if the Accounting Periods are then calendar months) or (ii) the Revenues Computation for such Fiscal Year (determined in accordance with Section 3.1.2), over (B) the aggregate amount of Rent previously paid pursuant to this Section 3.1.1 during such Fiscal Year (less the aggregate amount previously paid by Landlord to Tenant pursuant to the next sentence in this Section 3.1.1 during such Fiscal Year). In the event that the amount specified in clause (A) is less than the amount specified in clause (B) as of any payment date, then Landlord will promptly pay to Tenant the difference in cash. In the event of a change in the Accounting Period during the Term which results in payment dates (and/or number of payments) different from the payment dates (and/or number of payments) reflected on Schedule 3.1.1, the amount of

subsequent installments of Minimum Rent shall be appropriately adjusted in a fair and equitable manner. In the absence of agreement between Landlord and Tenant on the appropriate adjustments, the matter may be submitted by either party to arbitration under Article 15 for resolution.

#### 3.1.2 Revenues Computation.

(a) The "REVENUES COMPUTATION" for the applicable Fiscal Year shall be the sum of the following amounts:

- (1) an amount equal to the product of (a) the First Tier Room Revenue Percentage times (b) all year-to-date Room Revenues up to (but not exceeding) the Cumulative Portion of the Annual Room Revenues First Break Point,
- (2) an amount equal to the product of (a) the Second Tier Room Revenue Percentage times (b) all year-to-date Room Revenues in excess of the Cumulative Portion of the Annual Room Revenues First Break Point up to (but not exceeding) the Cumulative Portion of the Annual Room Revenues Second Break Point,
- (3) an amount equal to the product of (a) the Third Tier Room Revenue Percentage times (b) all year-to-date Room Revenues in excess of the Cumulative Portion of the Annual Room Revenues Second Break Point,
- (4) an amount equal to the product of (a) the First Tier Food and Beverages Sales Percentage times (b) all year-to-date Food and Beverages Sales up to (but not exceeding) the Cumulative Portion of the Annual Food and Beverages Sales First Break Point,
- (5) an amount equal to the product of (a) the Second Tier Food and Beverages Sales Percentage times (b) all year-to-date Food and Beverages Sales in excess of the Cumulative Portion of the Annual Food and Beverages Sales First Break Point up to (but

- not exceeding) the Cumulative Portion of the Annual Food and Beverages Sales Second Break Point,
- (6) an amount equal to the product of (a) the Third Tier Food and Beverages Sales Percentage times (b) all year-to-date Food and Beverages Sales in excess of the Cumulative Portion of the Annual Food and Beverages Sales Second Break Point,
- (7) an amount equal to the product of (a) the First Tier Other Income Percentage times (b) all year-to-date Other Income up to (but not exceeding) the Cumulative Portion of the Annual Other Income Break Point, and
- (8) an amount equal to the product of (a) the Second Tier Other Income Percentage times (b) all year-to-date Other Income in excess of the Cumulative Portion of the Annual Other Income Break Point.
- (b) The Revenues Computation shall be computed utilizing the following definitions:
  - 1) "CUMULATIVE PORTION" shall mean a fraction having as its numerator the total number of days in the Fiscal Year which have elapsed through the last day of the Accounting Period with respect to which a payment of Minimum Rent or Percentage Rent is due, and having as its denominator the total number of days in the Fiscal Year.
  - (2) "FIRST TIER ROOM REVENUE PERCENTAGE," "SECOND TIER ROOM REVENUE PERCENTAGE," "THIRD TIER ROOM REVENUE PERCENTAGE," "FIRST TIER FOOD AND BEVERAGES SALES PERCENTAGE," "SECOND TIER FOOD AND BEVERAGES SALES PERCENTAGE," "THIRD TIER FOOD AND BEVERAGES SALES PERCENTAGE," "FIRST TIER OTHER INCOME PERCENTAGE," AND "SECOND TIER OTHER INCOME PERCENTAGE," AND "SECOND TIER OTHER INCOME PERCENTAGE" shall mean the percentages set forth on Schedule 3.1.2.

------

(3) "ANNUAL ROOM REVENUES FIRST BREAK POINT," "ANNUAL ROOM REVENUES SECOND BREAK POINT," "ANNUAL FOOD AND BEVERAGES SALES FIRST BREAK POINT," "ANNUAL FOOD AND BEVERAGES SALES SECOND BREAK POINT," AND "ANNUAL OTHER INCOME BREAK POINT" shall mean the amounts set forth on Schedule 3.1.2.

-----

- 3.1.3 Confirmation of Percentage Rent.
- (a) Tenant shall submit to Landlord on each rental payment date pursuant to Section 3.1.1 a reasonably detailed schedule (the "PERCENTAGE"  $\,$

RENT SCHEDULE"), signed and certified by Tenant to be correct, showing the Revenues Computation with respect to the Facility for the preceding Accounting Period and the calculation of the amount of Rent owed on such date (other than Additional Charges) by Tenant or the amount Landlord is required to pay to Tenant for such Accounting Period. Tenant's Percentage Rent Schedule shall clearly indicate how much of the Gross Revenues is comprised by Room Revenues, Food and Beverages Sales and Other Income, and shall contain such detail and breakdown as Landlord may reasonably require. If, after notice from Landlord and the expiration of the cure period provided for in Section 12.1(c) of this Lease, Tenant fails to submit the Percentage Rent Schedule to Landlord when due, Landlord, in addition to any other remedies Landlord has, shall have the right (subject to compliance with the Management Agreement and the Consent and Assignment) to retain an independent certified public accountant, at Tenant's sole expense, to prepare such Percentage Rent Schedule and to perform all inspections and audits related thereto.

(b) Tenant shall maintain in accordance with the usual and customary practices of Tenant and in accordance with GAAP during the Term, for the current Fiscal Year and for the immediately preceding Fiscal Year, (i) complete and accurate general books of account, which shall reflect Gross Revenues and which shall include, without limitation, those items set forth and described on

Schedule 3.1.3(b) hereof, and (ii) all other original records and other  $\hdots$ 

pertinent papers that will enable Landlord to determine the Gross Revenues derived by Tenant during the relevant Fiscal Year from Room Revenues, Food and Beverages Sales and Other Income. Such records for the current and most recent Fiscal Year shall be maintained at Tenant's corporate headquarters. The provisions of this Section 3.1.3(b) shall survive the expiration or earlier termination of this Lease for a period of three (3) years thereafter, subject to extension upon Notice from Landlord, provided that all storage and related expenses of maintaining records beyond such three-year period shall be at Landlord's sole cost and expense. In addition to the audit rights set forth in Section 3.1.3(c) below, in the event Landlord, any Lending Institution, any Facility Mortgagee, or any potential purchaser of the Leased Property, or any of their respective representatives, desires to audit Tenant's financial records described in clauses (i) and (ii) above, Tenant shall, at Landlord's sole cost and expense, cooperate with such audit by making such records available at Tenant's corporate headquarters during normal business hours upon reasonable prior Notice.

(c) Landlord shall have fifty (50) days after the receipt of the Year End Percentage Rent Schedule to have an independent certified public accountant examine Tenant's records, during regular business hours upon reasonable prior Notice by Landlord, of Room Revenues, Food and Beverages Sales, and/or Other Income for the related Fiscal Year and all other relevant financial information. The acceptance by Landlord of each periodic payment of Percentage Rent shall not prejudice Landlord's right to proceed with such examination as described in the immediately preceding sentence. If Landlord raises no objections within such fifty (50)-day period, the Year End Percentage Rent Schedule shall be deemed to have been accepted by Landlord as true and correct, and Landlord shall have no further right to question its accuracy. If Landlord does raise such an objection, by Notice to Tenant, Landlord shall arrange for an independent certified public accountant to commence such examination within fifty (50) days after the date of such objection and shall cause such audit to be completed no later than one hundred eighty (180) days after

Landlord's receipt of the Year End Percentage Rent Schedule. Landlord shall pay all costs and expenses of such audit; provided, however:

- (i) if such audit discloses that the amount of Percentage Rent was underreported by Tenant by five percent (5%) or more for such Fiscal Year, Tenant shall pay to Landlord, within fifteen (15) days of its receipt of Notice from Landlord, the cost of the audit, as Additional Charges, in addition to any deficiency in Percentage Rent that may be due to Landlord; or
- (ii) if the audit discloses that the amount of Percentage Rent either was not underreported or was underreported by Tenant by less than five percent (5%), Landlord shall be responsible for all costs associated with the audit and Tenant shall pay, within fifteen (15) days of its receipt of Notice from Landlord, any deficiency in Percentage Rent that may be due to Landlord; or
- (iii) if the audit discloses that the Percentage Rent was overreported by Tenant for the related Fiscal Year, Landlord shall be responsible for all costs associated with the audit, shall give Tenant Notice of such overreporting within fifteen (15) days of Landlord's receipt of audit results, and shall reimburse the amount of such overpayment to Tenant in cash along with delivery of such Notice to Tenant. The provisions of this Section shall survive the expiration or the earlier termination of this Lease for a period of one (1) year thereafter.

#### 3.1.4 Annual Adjustments.

Upon the commencement of each Fiscal Year during the Term beginning with the second Fiscal Year, each of the Minimum Rent, the Annual Room Revenues First Break Point, the Annual Food and Beverages Sales First Break Point, the Annual Other Income Break Point, the Annual Food and Beverages Sales Second Break Point and the Annual Room Revenues Second Break Point shall be increased by the amount of the applicable Annual Adjustment.

In no event shall the Minimum Rent, the Annual Room Revenues First Break Point, Annual Food and Beverages Sales First Break Point, the Annual Room Revenues Second Break Point, the Annual Food and Beverages Sales Second Break Point, and the Annual Other Income Break Point be reduced as a result of the Annual Adjustment. The Annual Adjustment described above shall be effective on the first day of the new Fiscal Year. If Rent for the new Fiscal Year is paid prior to the determination of the amount of the Annual Adjustment, whether because of a delay in the publication of the Consumer Price Index or the Labor Index applicable for the Measurement Date or because of any other reason, payment adjustments for any shortfall in Rent paid shall be made with the first Minimum Rent and Percentage Rent payments due after the Annual Adjustment is determined.

If (1) a significant change is made in the number or nature (or both) of items used in determining the Consumer Price Index or the Labor Index or (2) the Consumer  ${\sf Consumer}$ 

Price Index or Labor Index shall be discontinued for any reason, the Bureau of Labor Statistics shall be requested to furnish a new index comparable to the Consumer Price Index or Labor Index, together with information that will make possible a conversion to the new index in computing the Annual Adjustment. If for any reason the Bureau of Labor Statistics does not furnish such an index and information, the parties will instead mutually select, accept and use such other indexes or comparable statistics on the cost of living and cost of labor in various United States cities that is computed and published by an agency of the United States or a responsible financial periodical of recognized authority.

## 3.1.5 Additional Charges.

In addition to the Minimum Rent and Percentage Rent payable hereunder, Tenant shall pay and discharge as and when due and payable the following (collectively, "ADDITIONAL CHARGES"):

(a) Impositions. Subject to Article 8 relating to Permitted

Contests and the right of any Facility Mortgagee to require tax escrows as described in the last sentence of this paragraph of subsection (a), Tenant shall pay or cause to be paid all Impositions attributable to any period during the Term before any fine, penalty, interest or cost (other than any opportunity cost as a result of a failure to take advantage of any discount for early payment) may be added for nonpayment, such payments to be made directly to the taxing authorities where feasible, and shall promptly, upon request, furnish to Landlord copies of official receipts or other satisfactory proof evidencing such payments. If any such Imposition may, at the option of the taxpayer, lawfully be paid in installments (whether or not interest shall accrue on the unpaid balance of such Imposition), Tenant may exercise the option to pay the same (and any accrued interest on the unpaid balance of such Imposition) in installments and, in such event, shall pay such installments during the Term as the same become due and before any fine, penalty, premium, further interest or cost may be added thereto. Landlord, at its expense, shall, to the extent required or permitted by Legal Requirements, prepare and file all tax returns required to be filed by Landlord, including, without limitation, returns in respect of Landlord's net income, gross receipts, sales and use, single business, transaction, privilege, rent, ad valorem, franchise taxes, Real Estate Taxes and other Landlord Obligations, and taxes on its capital stock, and Tenant, at its expense, to the extent required or permitted by Legal Requirements, prepare and file all tax returns and reports required to be filed by Tenant in respect of any Imposition as may be required by any Government Agency. Provided no monetary Default or Event of Default shall have occurred and be continuing, if any refund shall be due from any taxing authority in respect of any Imposition paid by Tenant, the same shall be paid over to or retained by Tenant. Landlord and Tenant shall each, upon request of the other, provide such data as is maintained by the party to whom the request is made with respect to the Leased Property as may be necessary to prepare any required returns and reports. In the event Government Agencies classify any property covered by this Lease as personal property, Tenant shall file all personal property tax returns in such jurisdictions where it may legally be required to so file. Each party shall, to the extent it possesses the same, provide the other, upon request, with cost and depreciation records necessary for filing returns for any property so classified as personal property. Where Landlord is legally required to file personal property tax returns, Landlord shall provide Tenant with copies of assessment notices in sufficient time for Tenant to file a protest. All

Impositions assessed against such personal property that comprises FF&E shall be (irrespective of whether Landlord or Tenant shall file the relevant return) paid by the party that owns the FF&E not later than the last date on which the same may be made without interest or penalty. All Impositions assessed against such personal property that comprises Tenant's Personal Property shall be (irrespective of whether Landlord or Tenant shall file the relevant return) paid by Tenant not later than the last date on which the same may be made without interest or penalty. If the provisions of any Facility Mortgage require deposits on account of Impositions to be made with the Facility Mortgagee, provided the Facility Mortgagee has not elected to waive such provision, Tenant shall either pay Landlord the monthly amounts required with respect to any such Impositions at the time and place that payments of Minimum Rent are required and Landlord shall transfer such amounts to the Facility Mortgagee or, pursuant to written direction by Landlord, Tenant shall make such deposits directly with the Facility Mortgagee, and such payment to Landlord or Facility Mortgagee shall be deemed to satisfy Tenant's obligation hereunder to pay the Impositions.

Landlord shall give prompt Notice to Tenant of all Impositions payable by Tenant hereunder of which Landlord at any time has knowledge; provided, however, that Landlord's failure to give any such Notice shall in no way diminish Tenant's obligation hereunder to pay such Impositions, except that Landlord shall be responsible for any interest and/or penalties incurred by Tenant as a result of Landlord's failure to forward any invoices, assessment notices or other bills to Tenant and no Default or Event of Default shall be deemed to have occurred hereunder if Tenant's failure to pay timely any Impositions is due to Landlord's failure to give Tenant such Notice at least ten (10) days before the amounts therein are due.

Tenant shall give prompt Notice to Landlord of all taxes included in Landlord Obligations payable by Landlord hereunder of which Tenant at any time has knowledge; provided, however, that Tenant's failure to give any such Notice shall in no way diminish Landlord's obligation hereunder to pay such Landlord Obligations, except that Tenant shall be responsible for any interest and/or penalties incurred by Landlord as a result of Tenant's failure to forward any invoices, assessment notices or other bills to Landlord and no Landlord Default shall be deemed to have occurred hereunder if Landlord's failure to pay timely any taxes included in Landlord Obligations is due to Tenant's failure to give Landlord such Notice at least ten (10) days before the amounts therein are due.

- (b) Utility Charges. Tenant shall pay or cause to be paid all charges for electricity, power, gas, oil, water and other utilities used in connection with the Leased Property.
- (d) Other Charges. Tenant shall pay or cause to be paid all other amounts, liabilities and obligations that Tenant assumes or agrees to pay under this Lease,

including, without limitation, all agreements to indemnify Landlord under Sections 4.3.3 and 9.8.

(f) Reimbursement for Additional Charges. If Tenant (as opposed to

Manager) pays or causes to be paid Additional Charges attributable to periods before the Commencement Date, Tenant shall provide Notice to Landlord of such amounts and, within fifteen (15) days thereafter, Landlord shall remit to Tenant the amount of such Additional Charges paid. If Tenant pays or causes to be paid Additional Charges attributable to periods after the end of the Term, Tenant may, within sixty (60) days after the end of the Term, provide Notice to Landlord of such amounts. Provided no uncured monetary Default or Event of Default then exists, Landlord shall reimburse Tenant for all payments of such Additional Charges that are attributable to any period after the expiration or earlier termination of the Term of this Lease within fifteen (15) days after its receipt of such Notice. Notwithstanding the foregoing, Tenant shall only be reimbursed to the extent such Additional Charges are not taken into account in the calculation and settlement of Working Capital accounts as set forth in Section 4.5. In the event Manager pays for Additional Charges attributable to periods before the Commencement Date, Tenant shall receive a payment as set forth in Section 3.1.6(g) below. Tenant shall reimburse Landlord within fifteen (15) days after receipt of Notice (which shall be given within sixty (60) days after the end of the Term) for any Additional Charges paid by Landlord attributable to periods during the Term for which Tenant received a credit in the settlement of Working Capital.

- 3.1.6 Adjustments to/Abatements of Rent.
  - (a) FF&E Adjustment to Rent. Pursuant to the terms of

Schedule 22.2, Rent shall be reduced by the amount of the FF&E Adjustments

then in effect. In the event there is an FF&E Adjustment which commences on a day other than the first day of an Accounting Period in which such FF&E Adjustment occurs, Landlord shall reimburse Tenant in cash, within fifteen (15) days after such FF&E Adjustment, an amount equal to the amount by which Rent paid for the Accounting Period in which the FF&E Adjustment commenced exceeded the amount of Rent (as reduced by the FF&E Adjustment) owed for such Accounting Period.

(b) Adjustments of Rent Following Owner-Funded Capital Expenditures

Under Management Agreement. Except as otherwise provided in Section 3.1.6(c) or

(e), effective on the date of any "Owner"-funded Capital Expenditures (as the term "Owner" is defined in the Management Agreement) paid by Landlord (other than Capital Expenditures paid out of the FF&E Reserve), the Rent due for each Fiscal Year thereafter in which Room Revenues for such Fiscal Year exceed the Annual Room Revenues Second Break Point shall be increased by an amount equal to [2%] of the amount of such contributions or expenditures.

(c) Adjustment of Rent for Major Capital Expenditures. In the event  $% \left( 1\right) =\left( 1\right) \left( 1\right)$ 

of either a reduction or increase in Gross Revenues or an adverse or beneficial impact on the Facility's operations, in either case resulting from any Major Capital Expenditure, Landlord and Tenant shall, in good faith, negotiate possible modifications to the Minimum Rent and Percentage Rent to determine an appropriate temporary or permanent adjustment of Rent and the effective date of such adjustment. If Landlord and Tenant are unable to agree that a reduction or increase in Gross Revenues or an adverse or beneficial impact on Facility operations, in either case resulting from any Major Capital Expenditure, has occurred, within thirty (30) days after the date of Notice from either party to the other that such event has occurred (accompanied by reasonably detailed computations and documentation to support such assertion), the matter may be submitted by either party to arbitration under Article 15 for resolution and for determination of the amount of the adjustment to Minimum Rent and Percentage Rent contemplated hereby. Tenant shall continue to pay Minimum Rent and Percentage Rent as required under Sections 3.1.1 through 3.1.5 of this Lease until such time as any adjustments to Percentage Rent and/or Minimum Rent are agreed upon or determined as set forth above.

(d) Adjustment of Rent for Increases or Decreases in the FF&E Reserve
Percentage. In the event that there is an increase or decrease, as the case may

be, in the percentage of Gross Revenues from the Leased Property required to be deposited by Landlord in the FF&E Reserve (the "FF&E RESERVE PERCENTAGE"), the Rent due for each Fiscal Year thereafter in which Room Revenues for such Fiscal Year exceed the Annual Room Revenues Second Break Point shall be increased (in the case of an increase in the FF&E Reserve Percentage) by an amount equal to the product of (i) the aggregate of the Room Revenues, the Food and Beverages Sales and the Other Income for such Fiscal Year times (ii) [19]% times (iii) the percentage point increase in the FF&E Reserve Percentage, or decreased (in the case of a decrease in the FF&E Reserve Percentage) by an amount equal to the product of (i) the aggregate of the Room Revenues, the Food and Beverages Sales and the Other Income for such Fiscal Year times (ii) [21]% times (iii) the percentage point decrease in the FF&E Reserve Percentage, in each case for the period during which the adjusted FF&E Reserve Percentage remains in effect.

- (e) Adjustment of Rent for Landlord Funded Increase in FF&E Reserve.

  In the event that Landlord is required by Manager to fund an increase in the FF&E Reserve (other than by reason of an increase in the FF&E Reserve Percentage), the Rent due for each of the five full Fiscal Years following the Fiscal Year in which the funding of such increase occurred shall be increased, for each Fiscal Year during such five (5)-year period that the Room Revenues for such Fiscal Year exceed the Annual Room Revenues Second Break Point, by an amount equal to [4.75]% of the dollar amount increase in the FF&E Reserve resulting from such funding.
- (f) Tenant Reimbursement for Landlord Obligations Paid by Manager or Tenant. Landlord shall reimburse Tenant in cash for the amount of any Landlord Obligations either (i) paid by Manager and deducted by Manager out of amounts owed to Tenant in accordance with the Management Agreement or (ii) paid by Tenant, such reimbursement to be paid within fifteen (15) days after receiving Notice thereof, provided no uncured monetary Default or Event of Default then exists.

Manager pays Additional Charges attributable to the period before the Commencement Date, and deducts the same from "Gross Revenues" (as defined in the Management Agreement) in accordance with the Management Agreement, and such Additional Charges were not taken into account in the calculation and settlement of Working Capital accounts as set forth in Section 4.5, such Additional Charges shall be the responsibility of Landlord, and Landlord shall remit to Tenant in cash within fifteen (15) days after receiving Notice thereof the amount of such Additional Charges paid, provided no uncured monetary Default or Event of Default then exists. In the event any item of Gross Revenues attributable to periods prior to the Commencement Date is not properly reflected in the initial calculation of Working Capital pursuant to Section 4.5, Tenant shall remit to Landlord in cash within fifteen (15) days after receiving Notice thereof the amount of such item not properly reflected.

- (i) Abatement of Rent for Partial Condemnation. In the event of a partial Condemnation as described in Section 11.2 which does not result in a termination of this Lease by Landlord pursuant to Section 11.2, the Rent shall be abated in the manner and to the extent provided in Section 11.3.

## 3.2 Late Payment of Rent.

If any installment of (i) Minimum Rent, (ii) Percentage Rent, or (iii) Additional Charges (but only as to those Additional Charges that are payable directly to Landlord) shall not be paid on its due date, Tenant shall pay Landlord, on demand, as Additional Charges, a late charge (to the extent permitted by law) computed at the Overdue Rate on the amount of such installment, from the due date of such installment to the date of payment thereof. To the extent that Tenant pays any Additional Charges directly to Landlord or the Facility Mortgagee pursuant to any requirement of this Lease, Tenant shall be relieved of its obligation to pay such Additional Charges to the Entity to which they would otherwise be due.

In the event of any failure by Tenant to pay any Additional Charges when due to any Entity other than Landlord, Tenant shall promptly pay and discharge, as Additional Charges, every fine, penalty, interest and cost that may be added by the Entity to which such Additional Charges are due (other than Landlord) for nonpayment or late payment of such items (subject to Landlord's obligation to pay or reimburse as provided in Section 3.1.5(a)).

## 3.3 Net Lease.

The Rent shall be absolutely net to Landlord so that this Lease shall yield to Landlord the full amount of the installments or amounts of Rent throughout the Term, subject to any other provisions of this Lease which expressly provide for adjustment or

abatement of such Rent or expressly provide that certain Landlord Obligations and Capital Expenditures are to be paid and/or performed by Landlord.

#### 3.4 No Termination, Abatement, Etc.

Except as otherwise specifically provided in this Lease, Tenant, to the maximum extent permitted by law, shall remain bound by this Lease in accordance with its terms and shall neither take any action without the consent of Landlord to modify, surrender or terminate this Lease, nor seek, nor be entitled to, any abatement, deduction, deferment or reduction of the Rent, or set-off against the Rent, nor shall the respective obligations of Landlord and Tenant be otherwise affected by reason of (a) any damage to or destruction of the Leased Property or any portion thereof from whatever cause or any Condemnation; (b) the lawful or unlawful prohibition of, or restriction upon, Tenant's use of the Leased Property, or any portion thereof, or the interference with such use by any Person, except to the extent that a court of competent jurisdiction has issued a final, nonappealable order determining that Tenant was constructively evicted from the Leased Property; (c) any claim that Tenant may have against Landlord by reason of any default or breach of any warranty by Landlord under this Lease or any other agreement between Landlord and Tenant, or to which Landlord and Tenant are parties (except for the Consent and Assignment); (d) any bankruptcy, insolvency, reorganization, composition, readjustment liquidation, dissolvency, winding up or other proceedings affecting Landlord or any assignee or transferee of Landlord; or (e) any other cause whether similar or dissimilar to any of the foregoing, other than a discharge of Tenant from any such obligations as a matter of law; provided, however, that the foregoing shall not apply or be construed to restrict any other rights Tenant may have as a result of any act or omission by Landlord constituting gross negligence or willful misconduct. Tenant hereby waives all rights arising from any occurrence whatsoever, which may now or hereafter be conferred upon it by law, to (i) modify, surrender or terminate this Lease or quit or surrender the Leased Property or any portion thereof, or (ii) entitle Tenant to any abatement, reduction, suspension or deferment of the Rent or other sums payable or other obligations to be performed by Tenant hereunder, except as otherwise specifically provided in this Lease and except to the extent that a court of competent jurisdiction has issued a final, nonappealable order determining that Tenant was constructively evicted from the Leased Property. The obligations of Tenant hereunder shall be separate and independent covenants and agreements, and the Rent and all other sums payable by Tenant hereunder shall continue to be payable in all events unless the obligations to pay the same shall be terminated pursuant to the express provisions of this Lease or this Lease shall be terminated. Notwithstanding anything set forth in this Lease to the contrary in any instance where, after the occurrence of a Default or an Event of Default, this Lease expressly permits Landlord to retain funds which, but for the Default or Event of Default, would be payable to Tenant, Landlord shall refund such funds to Tenant to the extent the amount exceeds the amount estimated by Landlord in good faith to be necessary to compensate Landlord for any cost, loss, or damage incurred or reasonably expected to be incurred in connection with such Default or Event of Default.

#### 3.5 Lockbox Procedures.

- (a) In the event that either (i) the Tangible Net Worth of CCC is less than two hundred percent (200%) of the amount of the maximum liability under the Guarantee from time to time as of the end of each of three (3) consecutive Accounting Periods, or (ii) the Consolidated Coverage Ratio of CCC is less than 2.3 to 1.0 for each of three (3) consecutive Accounting Periods, or (iii) there is an Event of Default under Section 12.1(a)(i), 12.1(c) (by reason of a breach of Section 7.1 or 21.6.3), 12.1(f), 12.1(g), 12.1(h), 12.1(i), 12.1(k), or 12.1(m) of this Lease, or (iv) there is a default continuing beyond any applicable cure period under Section 12.1(a)(i) or Section 12.1(c) (by reason of a breach of Section 7.1 or 21.6.3) of any Pool Lease (other than this Lease) or any Other Lease with respect to which CCC Limit of Liability under any CCC guarantee applicable to such Other Lease is greater than zero, then, at Landlord's option and upon Notice to Manager and Tenant (the date of such notice, the "LOCKBOX NOTICE DATE"), the procedures set forth in this Section 3.5 shall apply during the period beginning no later than two (2) weeks after the Lockbox Notice Date and continuing thereafter until the first (1st) day of the first (1st) full Accounting Period after (w) the Tangible Net Worth of CCC is at least equal two hundred percent (200%) of the maximum liability under the Guarantee from time to time as of the end of each of three (3) consecutive Accounting Periods, (x) the Consolidated Coverage Ratio of CCC is at least 2.3 to 1.0 for each of three (3) consecutive Accounting Periods, (y) no Event of Default under Section 12.1(a)(i), 12.1(c) (by reason of a breach of Section 7.1 or 21.6.3), 12.1(f), 12.1(g), 12.1(h), 12.1(i), 12.1(k), or 12.1(m) of this Lease has been in effect as of the end of each of three (3) consecutive Accounting Periods, or (z) there has been no default continuing beyond any applicable cure period under Section 12.1(a)(i) or Section 12.1(c) (by reason of a breach of Section 7.1 or 21.6.3) of any Pool Lease (other than this Lease) or any Other Lease with respect to which CCC Limit of Liability under any CCC guarantee applicable to such Other Lease is greater than zero, as of the end of each of three (3) consecutive Accounting Periods (such period, a "LOCKBOX" PERIOD"). At any time after the termination of the Guarantee pursuant to Section 10 thereof, a Lockbox Period shall commence pursuant to this subsection (a) only if there is an Event of Default under Section 12.1(a)(i), 12.1(c) (by reason of a breach of Section 7.1 or 21.6.3), 12.1(f), 12.1(g), 12.1(h), 12.1(i), or 12.1(k) of this Lease.
- (b) Following the Lockbox Notice Date and prior to commencement of the Lockbox Period, Landlord (or, if required by the terms of any Facility Mortgage, any Facility Mortgage or its agent) shall establish and maintain a cash collateral account at a financial institution mutually acceptable to Landlord and Tenant (the "LOCKBOX ACCOUNT"), as a segregated account in the name of Landlord (or, if applicable, the Facility Mortgagee or its agent), into which all revenues of Tenant due from Manager to Tenant or paid by Landlord to Tenant hereunder that are to be held as security for Tenant's obligations under this Lease shall be deposited.
- (c) During the Lockbox Period, Manager shall, in accordance with the Consent and Assignment, deposit into the Lockbox Account all amounts required to be paid to Tenant under the Management Agreement and Landlord shall deposit any funds that it is required to pay to Tenant under this Lease directly into the Lockbox Account. During the Lockbox Period, any funds received directly by Tenant and not yet deposited into the Lockbox Account shall irrevocably be deemed to be held in trust pursuant to the terms of this Section 3.5 and shall, immediately upon receipt and identification as belonging to Tenant (and in no event later than one full Business Day after receipt and identification as belonging to Tenant), be deposited by Tenant into the Lockbox Account.

(d) During the Lockbox Period, all funds that are received in cash in the Lockbox Account in any Accounting Period shall be disbursed in the following order of priority on the next date on which a scheduled payment of Rent is due:

First: for the account of Tenant, to pay to Landlord all Rent (including Additional Charges) due and payable by Tenant as of such payment date; and

Second: provided that no Event of Default under Section 12.1(a)(i) has occurred and is then continuing, all Distributable Cash (as defined in the Pooling Agreement) shall be remitted to the Collateral Agent under the Pooling Agreement for deposit in the Pool Cash Collateral Account (as defined in the Pooling Agreement); and

Third: provided that no Event of Default under Section 12.1(a)(i) or 12.1(c) (by reason of a breach of Section 7.1 or 21.6.3) hereof has occurred and is then continuing, any remaining balance will be transferred from the Lockbox Account to Tenant.

- (e) The funds on deposit in the Lockbox Account shall be the property of Tenant but shall be pledged to Landlord as further security for Tenant's obligations under this Lease pursuant to the Security Agreement.
- (f) Upon termination of the Lockbox Period, provided the Guarantors are not then in monetary default under the Guarantee, and unless otherwise specified in the Pooling Agreement, all amounts in the Lockbox Account shall be paid to Tenant.
- (g) Notwithstanding the foregoing, a Lockbox Period shall commence upon the giving of any notice of termination of the Guarantee pursuant to Section 10(a) of the Guarantee. Any Lockbox Period that commences under this subsection (g) shall continue until the earlier of (i) the date 12 months after the date on which the six-month period specified in Section 10(a)(iii) of the Guarantee commences or (ii) if no such six-month period commences, the date on which such notice is deemed to be void pursuant to Section 10(a)(ii)(F) of the Guarantee. During the portion (if any) of such Lockbox Period that commences on the day after the expiration of such six-month period, the Collateral Agent shall retain all funds not applied to payment of Rent under the Lease and, (x) on the "Distribution Date" (as such term is defined in the Pooling Agreement) relating to the third Accounting Period beginning after the commencement of such portion of the Lockbox Period and (y) at the termination of such Lockbox Period, distribute all funds held in the Lockbox Account to Tenant, provided that there are not then any unpaid Rents that are due under the Lease or that no other Event of Default under Section 7.1 or 21.6.3 of the Lease has occurred and is continuing (in which event such funds will only be disbursed in accordance with the provisions of Section 3.5(d) above).

#### ARTICLE 4

## USE OF THE LEASED PROPERTY

## 4.1 Permitted Use.

## 4.1.1 Primary Intended Use.

Tenant shall, at all times during the Term and at any other time that Tenant shall be in possession of any Leased Property, continuously use the Leased Property for the operation of a first class, full-service hotel and for such other uses as may be incidental or necessary thereto (such use being hereinafter referred to as the Leased Property's "PRIMARY INTENDED USE"). Tenant covenants and agrees to use its best efforts to operate the Facility according to First Class Operating Standards (which, during any period when the Management Agreement is in effect, Tenant shall be deemed to have done so long as Tenant uses its best efforts to cause the Manager to operate the Facility in accordance with the terms of the Management Agreement) and, except as otherwise expressly provided herein, agrees to pay all costs related thereto. Tenant shall not use the Leased Property or any portion thereof for any use inconsistent with the Primary Intended Use without the prior written consent of Landlord. Tenant shall not take or omit to take any action, the taking or omission of which materially impairs the value or the usefulness of the Leased Property or any part thereof for its Primary Intended Use.

## 4.1.2 Necessary Approvals.

Tenant shall proceed with all due diligence and exercise best efforts to obtain and maintain (or to cause Manager to procure and maintain) all Licenses necessary to use and operate, for its Primary Intended Use, the Leased Property and Facility located thereon under Legal Requirements and Landlord shall cooperate with respect to obtaining any such License, including joining in any application for Licenses to the extent required by Legal Requirements.

#### 4.1.3 Lawful Use, Etc.

Tenant shall not use, and shall use commercially reasonable efforts to prohibit third parties from using, the Leased Property or Tenant's Personal Property for any unlawful purpose. Tenant shall not commit, and shall use commercially reasonable efforts not to suffer to be committed, any waste on the Leased Property, or in the Facility, nor shall Tenant cause or permit any nuisance thereon or therein. Tenant shall not use, and shall use commercially reasonable efforts to prohibit third parties from using, the Leased Property or any portion thereof, including Tenant's Personal Property, in such a manner as (i) might reasonably tend to impair Landlord's (or Tenant's, as the case may be) title thereto or to any portion thereof or (ii) may reasonably make possible a claim or claims for adverse usage or adverse possession by the public, or of implied dedication of the Leased Property or any portion thereof. Tenant shall not use, and shall use commercially reasonable efforts to prohibit third parties from using, the Leased Property in any manner that will cause the cancellation of any insurance policy covering the Leased Property or any part thereof (unless another adequate policy is available), nor shall Tenant sell or otherwise provide to guests or residents therein, and shall use commercially reasonable

efforts to prohibit third parties from keeping, using or selling in or about the Leased Property, any article which may be prohibited by law or by the standard form of fire insurance policies, or any other insurance policies required to be carried hereunder, or fire underwriter's regulations.

#### 4.2 Compliance with Legal and Insurance Requirements, Etc.

Subject to the provisions of the Management Agreement and Article 8 hereof, and subject to compliance by Landlord with its obligations hereunder, Tenant, at its sole expense, shall (i) comply in all material respects with Legal Requirements and Insurance Requirements in respect of the use, operation, maintenance, repair, alteration and restoration of the Leased Property, including Legal Requirements regarding labor relations with respect to Manager's employees as required pursuant to the Management Agreement, and (ii) procure, maintain and comply in all material respects with all appropriate licenses, permits, and other authorizations and agreements required for any use of the Leased Property and Tenant's Personal Property then being made, and for the proper erection, installation, operation and maintenance of the Leased Property or any part thereof. Landlord shall comply in all material respects with Legal Requirements pertaining to Landlord regarding labor relations with respect to Manager's employees.

#### 4.3 Environmental Matters.

## 4.3.1 Restriction on Use, Etc.

Tenant shall not store, spill upon, dispose of or transfer to or from the Leased Property any Hazardous Materials, except that Tenant may store, transfer and dispose of Hazardous Materials in compliance with all Environmental Laws. Tenant shall use commercially reasonable efforts to cause Manager to maintain the Leased Property at all times free of any Hazardous Materials (except such Hazardous Materials as are maintained in compliance with all Environmental Laws). Tenant shall promptly (a) notify Landlord in writing of any material change in the nature or extent of Hazardous Materials at the Leased Property of which Tenant has notice or actual knowledge, (b) transmit to Landlord a copy of any Community Right to Know report which is required to be filed by Tenant with respect to the Leased Property pursuant to SARA Title III or any other Environmental Law, (c) transmit to Landlord copies of any demand letters, complaints or other documents initiating legal action, citations, orders, notices or other material communications asserting claims by private parties or government agencies with respect to Hazardous Materials received by Tenant or its agents or representatives (collectively, "ENVIRONMENTAL NOTICE"), which Environmental Notice requires a written response or any action to be taken and/or if such Environmental Notice gives notice of and/or could give rise to a material violation of any Environmental Law and/or could give rise to any material cost, expense, loss or damage (an "ENVIRONMENTAL OBLIGATION"), (d) use commercially reasonable efforts to cause Manager to observe and comply with all Environmental Laws relating to the use, maintenance and disposal of Hazardous Materials and all orders or directives from any official, court or agency of competent jurisdiction relating to the use or maintenance thereof or requiring the removal, treatment, containment or other disposition thereof, and (e) subject to Section 4.3.2, pay or otherwise dispose of any fine, charge or imposition related thereto, unless Tenant shall contest the

same in good faith and by appropriate proceedings and the right to use, and the value of, the Leased Property is not adversely affected thereby in any substantial manner.

If at any time Hazardous Materials are discovered in violation of Environmental Laws on the Leased Property, subject to Section 4.3.2 Tenant shall exercise commercially reasonable efforts to cause Manager to take all actions and incur any and all expenses (which actions and expenses shall be subject to Landlord's prior approval, not to be unreasonably withheld, conditioned or delayed except in Emergency Situations, in which case Landlord's prior approval shall not be required) as may be necessary or as may be required by any Government Agency (i) to clean up and remove from and about the Leased Property all Hazardous Materials thereon, (ii) to contain and prevent any further release or threat of release of Hazardous Materials on or about the Leased Property and (iii) to use good faith efforts to eliminate any further release or threat of release of Hazardous Materials on or about the Leased Property.

#### 4.3.2 Environmental Indemnification of Tenant by Landlord.

Landlord shall protect, indemnify and hold harmless Tenant, its Affiliates and their respective members, shareholders or other equity owners, directors, management committee, or similar persons, trustees, officers, and employees and any of their respective successors or assigns (hereafter the "TENANT INDEMNITEES," and when referred to singly, a "TENANT INDEMNITEE") for, from and against any and all debts, liens, claims, causes of action, administrative orders or notices, costs, fines, penalties or expenses (including, without limitation, reasonable attorneys' fees and expenses) imposed upon, incurred by or asserted against any Tenant Indemnitee resulting from, either directly or indirectly, the presence during the Term in the Environment of the Leased Property or any properties surrounding the Leased Property of any Hazardous Materials (collectively, "ENVIRONMENTAL LIABILITIES"), except to the extent that the same arise by reason of the negligence or willful misconduct of Tenant, any other Tenant Indemnitee, or their respective agents (including during the Term, the Manager). Landlord's duty herein includes, but is not limited to, indemnification for costs associated with personal injury or property damage claims as a result of the presence of Hazardous Materials in, upon or under the soil or ground water of the Leased Property in violation of any Environmental Law. Upon Notice from Tenant, Landlord shall undertake the defense, at Landlord's sole cost and expense, of any indemnification duties set forth herein, in which event Landlord shall not be responsible for any duplicative attorneys' fees incurred by any Tenant Indemnitee.

Landlord shall, upon demand, pay to Tenant any cost, expense, loss or damage (including, without limitation, reasonable attorneys' fees) incurred by Tenant in asserting any right under this Section 4.3.2, including, without limitation, any right of indemnity under this Section 4.3.2 or otherwise arising from a failure of Landlord strictly to observe and perform the foregoing requirements, which amounts shall bear interest from the date upon which demand is made therefor until paid by Landlord to Tenant at the Overdue Rate. Landlord agrees to waive any and all rights of contribution from Tenant which might arise under applicable law for any liability imposed upon Landlord pursuant to this Section 4.3.2.

#### 4.3.3 Environmental Indemnification of Landlord by Tenant.

Tenant shall protect, indemnify and hold harmless Landlord and each Facility Mortgagee, their Affiliates and their respective members, shareholders or other equity owners, directors, management committee, or similar persons, trustees, officers, and employees, and any of their respective successors or assigns (hereafter the "LANDLORD INDEMNITEES," and when referred to singly, a "LANDLORD INDEMNITEE") for, from and against any and all Environmental Liabilities to the extent that the same arise by reason of Tenant's, any other Tenant Indemnitee's, or their respective agents' (including the Manager's) gross negligence or willful misconduct. Upon Notice from Landlord, Tenant shall undertake the defense, at Tenant's sole cost and expense, of any indemnification duties set forth herein, in which event Tenant shall not be responsible for payment of any duplicative attorneys' fees incurred by any Landlord Indemnifee.

Tenant shall, upon demand, pay to Landlord, as an Additional Charge, any cost, expense, loss or damage (including, without limitation, reasonable attorneys' fees) incurred by Landlord in asserting any right under this Section 4.3.3, including, without limitation, any right of indemnity under this Section 4.3.3 or otherwise arising from a failure of Tenant strictly to observe and perform the foregoing requirements, which amounts shall bear interest from the date upon which demand is made therefor until paid by Tenant to Landlord at the Overdue Rate. The foregoing notwithstanding, with respect to Tenant's indemnification of Landlord for the acts or failure to act of Manager, Tenant shall only become obligated to pay any sums upon a final, non-appealable determination, in a judicial proceeding in which Manager is a party (or a binding written acknowledgment of Manager to such effect), that Manager is liable to Tenant for Environmental Liabilities in accordance with the terms of the Management Agreement.

#### 4.3.4 Survival.

 $\,$  The provisions of this Section 4.3 shall survive the expiration or sooner termination of this Lease.

#### 4.4 FF&E Reserve.

Tenant shall cause Manager to establish and maintain a reserve account (the "FF&E RESERVE") in accordance with the terms of the Management Agreement and to deposit into such account during each Fiscal Year monies for the account of Landlord (which amounts shall be funds of Landlord for all purposes and shall be credited against the Rent obligation of Tenant pursuant to Section 3.1.6(f)) in the amount required to be maintained pursuant to the Management Agreement and in accordance with established custom of Manager prior to the Commencement Date. Landlord shall be responsible for the payment of all amounts required to be paid by "Owner" (as defined in the Management Agreement) pursuant to the Management Agreement to fund the FF&E Reserve, including, without limitation, any deficits therein, and for paying for all FF&E necessary for the continued operation of the Facility in accordance with First Class Operating Standards, subject to the provisions of Schedule 22.2. Tenant shall

make no expenditure for replacement of FF&E in excess of the amounts in the FF&E Reserve without first obtaining the written approval of Landlord (unless Tenant makes such

expenditures from its own funds without any right to reimbursement hereunder). In the event Tenant funds any amounts required to be funded by Landlord for FF&E pursuant hereto or by "Owner" pursuant to the Management Agreement which Landlord has approved, Tenant shall receive a reimbursement in accordance with Section 3.1.6(f) hereof. Any additions to or replacements of furniture, fixtures, and equipment located at the Leased Property shall become part of the FF&E which is owned by Landlord, subject to the limitations set forth in Schedule 22.2 hereof. Throughout the Term of this Lease, Tenant shall, at its

sole cost and expense but subject to the terms hereof and of the Management Agreement, cause all of the items of FF&E to be in proper working order and in good condition (ordinary wear and tear excepted).

## 4.5 Working Capital.

Subject to the provisions of the Management Agreement, Landlord agrees to sell the existing working capital (including (a) Inventories, Fixed Asset Supplies, and net receivables due from Manager, and (b) accounts payable, accrued payroll expenses and other accrued expenses related to the Facility) (the "WORKING CAPITAL") to Tenant on the Commencement Date at a purchase price equal to the fair market value of such assets (which Landlord and Tenant agree is equal to the book value of such assets on the Commencement Date after taking into account any depreciation as of the Commencement Date). Except as provided in the next sentence, title to the Working Capital so assigned, transferred and delivered shall be free and clear of any Liens of any nature whatsoever created by Landlord or arising in respect of any obligation of Landlord or arising by reason of any act or omission of Landlord. Such sale shall be made expressly subject to the Facility Mortgagee's first-priority lien, if any, on some or all of such Working Capital, and Tenant shall acknowledge same in writing to Lender and sign UCC-1 financing statements confirming same (at Landlord's expense) in accordance with the Facility Mortgagee Agreement, which financing statements shall be prepared and filed by Landlord at Landlord's sole cost and expense. The purchase price for the Working Capital shall be paid by Tenant by the issuance to Landlord of a promissory note in the form set forth on Exhibit E hereto (the "WORKING CAPITAL NOTE"). The Working Capital Note shall be in the principal amount of said purchase price and shall provide for accrual of interest on such principal amount from the Commencement Date until payment in full thereof at the rate per annum equal to the "long-term applicable federal rate" (as defined in Section 1274 of the Code) in effect on the Commencement The Working Capital Note shall provide that such interest will be payable from time to time at such time as each payment of Rent is due under Section 3.1.1, and the amount of interest paid under the Working Capital Note on any interest payment date also shall be credited against Rent payable on such date. The Working Capital Note shall further provide that the principal amount thereof will be payable in full, together with accrued and unpaid interest thereon, upon the expiration or earlier termination of this Lease for any reason (including, without limitation, a termination by the Facility Mortgagee in accordance with Section 20.2 hereof) as follows: Upon such expiration or termination, Tenant will transfer to Landlord, in payment of the principal amount of the Working Capital Note and accrued and unpaid interest thereon, title to all Working Capital then owned by Tenant. To the extent that the fair market value of the Working Capital at such time (which Landlord and Tenant agree shall be equal to the book value of such assets at such time after taking into account any depreciation as of such date) exceeds the principal amount of the Working Capital Note

plus accrued and unpaid interest thereon, Landlord shall pay to Tenant an amount in cash equal to such excess; to the extent that such fair market value is less than the principal amount of the Working Capital Note plus accrued and unpaid interest thereon, Tenant shall pay to Landlord an amount in cash equal to such deficiency.

In the event that this Lease is terminated by the Facility Mortgagee pursuant to the terms of Section 20.2 hereof, Tenant agrees that it shall transfer title to the Working Capital then owned by it (and pay in cash any deficiency due to Landlord pursuant to the Working Capital Note) directly to the Facility Mortgagee or its designee and that it shall look only to Host O.P. for payment of the amount (if any) by which such payment exceeds the principal amount of the Working Capital Note, all in accordance with the terms of the Facility Mortgagee Agreement, and Host O.P. agrees to pay such amount.

Landlord and Tenant agree that, following the sale of the Working Capital on the Commencement Date as provided for above, all Working Capital during the Term of this Lease shall be the property of Tenant (and not Landlord) for all purposes (subject, however, to the Liens hereinafter referred to in this paragraph), and neither Landlord nor Tenant shall at any time take a position (in its books and records or otherwise) or make an assertion inconsistent therewith. Tenant has granted Landlord a security interest in all such Working Capital pursuant to Section 7.2 hereof. Tenant acknowledges that Landlord has pledged and assigned to the Facility Mortgagee, as additional security for Landlord's obligations under the loan secured by the Facility Mortgage, the Working Capital Note and Landlord's rights and interest in respect of this Lease, including all Landlord Liens with respect to any and all Tenant's Personal Property, including, without limitation, all Working Capital owned by Tenant during the Term of this Lease, securing Tenant's obligations hereunder.

#### ARTICLE 5

## MAINTENANCE AND REPAIRS; SURRENDER

- 5.1 Maintenance and Repair.
  - 5.1.1 Tenant's Obligations.

Except as otherwise expressly provided in this Lease, and except for conditions caused by the gross negligence or willful misconduct of Landlord, its employees, agents or independent contractors (which terms shall not be deemed to include Manager with respect to actions or inactions of Manager during the Term of this Lease), Tenant shall, at its sole cost and expense, keep the Leased Property and all private roadways, sidewalks and curbs appurtenant thereto (and Tenant's Personal Property) in good order and repair, ordinary wear and tear excepted (whether or not the need for such repairs occurs as a result of Tenant's use, any prior use, the elements or the age of the Leased Property or Tenant's Personal Property, or any portion thereof), and shall promptly make all necessary and appropriate repairs and replacements thereto of every kind and nature (excluding Capital Expenditures), whether interior or exterior, structural or nonstructural, ordinary or extraordinary, foreseen or unforeseen, necessary for the Primary Intended Use (concealed or otherwise). Notwithstanding anything set forth herein to the contrary, unless the need for compliance with this Section 5.1.1 is caused by Tenant's negligence or

willful misconduct or that of its employees or agents (including Manager, but only to the extent that Tenant is obligated to indemnify Landlord and Landlord Indemnitees for Manager's acts or omissions pursuant to Section 4.3.3) and is not otherwise covered by insurance, Tenant shall not be responsible for any necessary repair of Existing Conditions relating to Hazardous Materials, including, without limitation, those set forth and described on Schedule 5.1.1

Concadic Cilii

hereto, which are Landlord Obligations. All repairs shall be made in a good, workmanlike and first-class manner, in accordance with all Legal Requirements. Tenant shall not take or omit to take any action, the taking or omission of which materially impairs the value or the usefulness of the Leased Property or any part thereof for the Primary Intended Use. Tenant's obligations under this Section 5.1.1 shall be limited, in the event of any Casualty or Condemnation involving the Leased Property, as set forth in Articles 10 and 11. Tenant's obligations under this Section 5.1.1 with respect to FF&E are subject to the provisions of Section 4.4 hereof. Tenant shall have the non-exclusive right to prosecute claims against Landlord's predecessors-in-interest (other than any Affiliates of Landlord), contractors, subcontractors and suppliers for breach of any representation or warranty or for any latent defects in the Leased Property, unless Landlord is already diligently pursuing such claims.

## 5.1.2 Landlord Obligations.

(a) Landlord shall be obligated to pay the actual costs of any items that are classified as Capital Expenditures and are approved by Landlord as well as the actual costs of any necessary repairs (i) of Existing Conditions relating to Hazardous Materials or (ii) resulting from the gross negligence or willful misconduct of Landlord, its employees, agents or independent contractors (which terms shall not be deemed to include Manager with respect to acts or omissions of Manager during the Term of this Lease). Landlord shall not, however, except as expressly set forth herein or otherwise required under the Management Agreement or the Consent and Assignment (including, without limitation, as set forth in this Section 5.1.2 and in Sections 4.3.2, 4.4, 5.1.1, 9.9 and 21.3.3 and in Article 6 hereof), be required to build or rebuild any improvement on the Leased Property, or to make any repairs, replacements, alterations, restorations, or renewals of any nature or description to the Leased Property, whether ordinary or extraordinary, structural or nonstructural, foreseen or unforeseen, or to make any expenditure whatsoever with respect thereto, or to maintain the Leased Property in any way. Landlord shall have the right to give, record and post, as appropriate, notices of nonresponsibility under any mechanic's lien laws now or hereafter existing.

(b) If, at any time or from time to time, the Management Agreement requires that funds be disbursed for repairs, maintenance, renovations or replacements at or to the Leased Property, or, pursuant to the terms of this Lease (including, without limitation, Section 4.4 hereof), Tenant is required to make any expenditures in connection with any repair, maintenance or renovation with respect to the Leased Property which constitute Landlord Obligations, and the amount of such disbursements or expenditures exceeds the amount on deposit in the FF&E Reserve, Tenant shall furnish Landlord with reasonable detail regarding the nature of the required repair, renovation or replacement, the estimated cost thereof and such other information with respect thereto as Landlord may reasonably require. Landlord shall, within ten (10) Business Days after such Notice, subject to and in accordance with the applicable

provisions of Article 6, disburse such required funds to Tenant (or, if Tenant shall so elect, or if an Event of Default shall then exist hereunder, directly to Manager or any other Person performing the required work).

## 5.2 Tenant's Personal Property.

Tenant may (and shall as provided hereinbelow), at its expense, install, affix or assemble or place on the Facility any items of Tenant's Personal Property, and Tenant may, subject to Section 7.2 and the conditions set forth below, remove and replace the same at any time in the ordinary course of business. Tenant shall provide and maintain throughout the Term all such Tenant's Personal Property as shall be necessary in order to operate the Facility in compliance in all material respects with applicable Legal Requirements and Insurance Requirements and otherwise in accordance with customary practice in the industry for such Primary Intended Use.

Subject to Section 4.5 hereof, all Tenant's Personal Property (except that removed and replaced in the ordinary course of business as permitted above, but including supplies and inventory that are equivalent, on an aggregate basis, in amount and value similar to that reasonably established for use by the Facility in the immediately preceding Lease Year) shall remain at the Leased Property at the expiration or earlier termination of this Lease without the necessity of any payment by Landlord to Tenant and without any obligation to account therefor.

If Tenant acquires an interest in any material item of Tenant's Personal Property (other than motor vehicles) on, or in connection with, the Leased Property that belongs to anyone other than Tenant, Tenant shall use its commercially reasonable efforts to require the agreement permitting such use to provide that Landlord or its designee may assume Tenant's rights under such agreement upon management or operation of the applicable Facility by Landlord or its designee.

## 5.3 Surrender.

(a) Condition of Leased Property Upon Surrender. Upon the expiration

or sooner termination of this Lease, Tenant shall vacate and surrender the Leased Property to Landlord in the condition in which the Leased Property was on the Commencement Date, ordinary wear and tear excepted, except as repaired, rebuilt, restored, altered or added to as permitted or required by the provisions of this Lease, ordinary wear and tear excepted. Together therewith Tenant shall surrender to Landlord any and all records and documents related to the Leased Property and Tenant's Personal Property (but not, subject to Section 5.3(d) hereof, documents primarily related to Tenant's business operated therein).

- (b) [Intentionally omitted]
- (c) Membership Contracts. Upon expiration or earlier termination of

this Lease, Tenant shall assign to Landlord any membership contracts relating to golf courses, spas or other facilities in which there are nonequity interests entered into by Tenant during the Term, and Landlord agrees to assume the obligations of Tenant under any such

membership contracts arising from and after the date of expiration or earlier termination of this Lease.

- (d) Transition Procedures. Tenant shall cooperate in good faith  $% \left( 1\right) =\left( 1\right) \left( 1\right) \left($
- to provide access and information to any prospective purchaser or tenant of the Leased Property that may acquire the Leased Property or lease it upon the expiration or earlier termination of the Term, provided that Landlord shall use its reasonable efforts to cause such prospective purchaser or tenant to enter into a confidentiality agreement with respect thereto reasonably acceptable to Tenant. Upon any expiration or earlier termination of the Term, Landlord and Tenant shall do the following and, in general, shall cooperate in good faith to effect an orderly transition of the management or lease of the Facility (and the provisions of this Section 5.3(d) shall survive the expiration or earlier termination of this Lease until they have been fully performed, up to a maximum period of one (1) year after such expiration or termination, and nothing contained herein shall limit Landlord's rights and remedies under this Lease if such termination occurs as the result of an Event of Default):
  - (i) Upon the expiration or earlier termination of the Term, Tenant shall use its commercially reasonable efforts (A) to transfer to Landlord or Landlord's designee all licenses, operating permits and other governmental authorizations and all contracts with governmental or quasi-governmental entities, that may be necessary for the operation of the Facility (collectively, "LICENSES"), to the extent transferable or (B) if such transfer is prohibited by law or Landlord otherwise elects, to cooperate with Landlord or Landlord's designee in connection with the processing by Landlord or Landlord's designee of any applications for all Licenses; provided, in either case, that the costs and expenses of any such transfer or the processing of any such application shall be paid by Landlord or Landlord's designee.
  - (ii) Tenant shall assign or cause to be assigned to Landlord or Landlord's designee simultaneously with the termination of this Lease, and the assignee shall assume, all leases, contracts, concession agreements and other agreements in effect with respect to the Facility then in Tenant's name; provided, however, that any leases of Excess FF&E shall be subject to the provisions of Schedule 22.2 hereof, and

provided further, that Landlord shall not be obligated to assume and may reject any such contracts or agreements requiring the "Owner's" approval under the Management Agreement which were entered into subsequent to the date hereof and were not previously approved or deemed approved by Landlord. In the event Landlord declines to assume or reject any such contracts, agreements and/or leases, the contracts, agreements and/or leases so rejected shall not be assigned or shall be deemed reassigned and shall remain the property and responsibility of Tenant. In no event shall Landlord (or its designee) have any liability under such contracts for obligations or liabilities accruing under such contracts after the Commencement Date and prior to the date of such assumption by such party.

- (iii) To the extent that Landlord has not already received copies thereof, copies of all books and records (including computer records) for the Facility kept by (or available to) Tenant shall be promptly delivered or made available for inspection and copying to Landlord or Landlord's designee.
- (iv) Subject to Section 4.5 hereof, Tenant shall be entitled to retain all cash, bank accounts and house banks, and to collect all Gross Revenues and accounts receivable accrued through the termination date of this Lease. Tenant shall be responsible for the payment of Rent, all operating expenses of the Facility as provided in this Lease and all other obligations of Tenant accrued under this Lease as of the termination date, and Landlord shall be responsible for all operating expenses of the Facility accruing after the termination date.
- (v) So long as termination is not the result of an Event of Default, Landlord shall reimburse Tenant for its reasonable expenses in connection with any cooperation with a prospective purchaser or tenant hereunder.

#### 5.4 Encroachments, Restrictions, Etc.

If any portion of the Facility shall, at any time, encroach upon any property, street or right-of-way adjacent to the Leased Property, or shall violate the agreements or conditions contained in any lawful restrictive covenant or other agreement affecting the Leased Property, or any part thereof, or shall impair the rights of others under any easement or right-of-way to which the Leased Property is subject, upon the request of Landlord (but only as to any encroachment, violation or impairment that is not a Permitted Lien) or of any Person affected by any such encroachment, violation or impairment, Tenant shall, at Tenant's sole cost and expense (except to the extent that the encroachment, violation or impairment existed prior to the Commencement Date or was the result of the act or omission of Landlord or its employees, agents (excluding Manager except for Manager's acts or omissions in respect of Landlord Obligations) or independent contractors), subject to its right to contest the existence of any encroachment, violation or impairment in accordance with the provisions of Article 8, either (a) obtain valid and effective waivers or settlements of all claims, liabilities and damages resulting from each such encroachment, violation or impairment, whether the same shall affect Landlord or Tenant or (b), subject to Section 6.1, make such changes in the Facility and take such other actions as are reasonably practicable to remove such encroachment and to end such violation or impairment, including, if necessary, the alteration of the Facility and, in any event, take all such actions as may be necessary in order to enable the continued operation of the Facility for its Primary Intended Use substantially in the manner and to the extent the Facility was operated prior to the assertion of such violation, impairment or encroachment. Any such alteration shall be made in conformity with the applicable requirements of this Article 5. Tenant's obligations under this Section 5.4 shall be in addition to and shall in no way discharge or diminish any obligation of any insurer under any policy of title or other insurance.

#### 5.5 Landlord to Grant Easements, Etc.

Landlord shall from time to time, at the request of Tenant and at Landlord's sole cost and expense, (a) grant easements and other rights in the nature of easements with respect to the Leased Property to third parties, (b) release existing easements or other rights in the nature of easements which are for the benefit of the Leased Property, (c) dedicate or transfer unimproved portions of the Leased Property for road, highway or other public purposes, (d) execute petitions to have the Leased Property annexed to any municipal corporation or utility district, (e) execute amendments to any covenants and restrictions affecting the Leased Property and (f) execute and deliver to any Person any instrument appropriate to confirm or effect such grants, release, dedications, transfers, petitions and amendments (to the extent of its interests in the Leased Property); provided, however, that Landlord shall have first determined, in its reasonable discretion, that such grant, release, dedication, transfer, petition or amendment is not detrimental to the operation of the Leased Property for its Primary Intended Use and is beneficial to the value of the Leased Property and is not in violation of the Management Agreement, and Landlord shall have received within five (5) Business Days of its request an Officer's Certificate confirming such determination, together with such additional information as Landlord may request. Landlord agrees to obtain Tenant's prior written approval, which shall not be unreasonably withheld, conditioned or delayed, as to any recorded covenants, conditions and restrictions which would affect the use or operation of the Leased Property as it is then being used or operated by Tenant in accordance with this Lease (it being agreed that the provisions of a Facility Mortgage or a ground lease shall not be deemed to constitute covenants, conditions and restrictions subject to this sentence).

# ARTICLE 6 CAPITAL EXPENDITURES, ETC.

# 6.1 Capital Expenditures.

It shall be Landlord's obligation (and a Retained Obligation) to approve and fund the cost of any Capital Expenditures, including, without limitation, any Capital Expenditures required under the Management Agreement, in accordance with the terms of the Management Agreement. Except as otherwise hereinafter provided, Tenant shall not construct or install Capital Expenditures on the Leased Property without obtaining Landlord's prior written consent.
Landlord's approval of the Capital Portion of the Annual Budget shall constitute approval of the Capital Expenditures described therein. If Owner's consent is required under the Management Agreement with respect to any Capital Expenditure not covered in the Capital Portion of the Annual Budget, prior to commencing construction of any Capital Expenditure Tenant shall submit to Landlord, in writing, a proposal setting forth, in reasonable detail, any proposed Capital Expenditure and shall provide to Landlord such plans and specifications, permits, licenses, contracts and other information concerning the proposed Capital Expenditure as Landlord may reasonably request. Without limiting the generality of the foregoing, such proposal shall indicate the approximate projected cost of constructing such Capital Expenditure and the use or uses to which it will be put. Tenant shall not fund or otherwise finance the cost of any construction of any Capital Expenditure without the prior written consent of Landlord, which consent may be withheld by Landlord in Landlord's sole discretion (unless Landlord

has declined to fund such Capital Expenditure, Manager has consented to such Capital Expenditure and Tenant's financing of such Capital Expenditure would not result in the imposition of a Lien on any portion of the Leased Property or the Capital Expenditure, in which event Landlord shall not unreasonably withhold its consent). Any Capital Expenditures funded or financed by Tenant shall, upon the expiration or sooner termination of this Lease, pass to and become the property of Landlord, free and clear of all encumbrances other than Permitted Liens. Notwithstanding the foregoing, Landlord's consent shall not be required in connection with any Capital Expenditures (i) with respect to which immediate action is required in order to comply with Legal Requirements or to prevent or remedy Emergency Situations or (ii) not requiring the "Owner's" consent under the Management Agreement; it being the intention hereof that the consent requirements with respect to Capital Expenditures under the Management Agreement.

#### 6.2 [Intentionally Omitted.]

#### 6.3 Cooperation by Tenant.

If Landlord shall, in Landlord's sole discretion, elect to finance any proposed Capital Expenditure, or if Landlord is otherwise required to do so pursuant to the terms of the Management Agreement and the Retained Obligations, Tenant shall provide Landlord with such information as Landlord may from time to time reasonably request.

#### 6.4 Alterations.

Subject to the provisions of the Management Agreement, Tenant shall have the right, at Tenant's sole cost and expense, to make additions, modifications or improvements to the Leased Property which are not Capital Expenditures ("ALTERATIONS") and which have a total cost of completion of less than or equal to (a) \$10,000, as to any individual Alteration, or (b) \$50,000, as to all Alterations, in the aggregate, over a twelve-month period, from time to time as Tenant, in its discretion, may deem desirable for the Primary Intended Use, provided that any such Alteration will not materially alter the character or purpose or materially detract from the value, operating efficiency or revenue-producing capability of the Leased Property or adversely affect the ability of Tenant to comply with the provisions of this Lease, and, without limiting the foregoing, will not violate any Legal Requirement or Insurance Requirement applicable to the Leased Property. Any Alteration estimated to exceed the applicable limits set forth above shall be subject to Landlord's prior approval and the terms set forth in Section 6.1. All such Alterations shall, upon expiration or earlier termination of this Lease, pass to and become the property of Landlord, free and clear of all liens and encumbrances, other than Permitted Liens.

### 6.5 Salvage.

All materials which are scrapped or removed in connection with the making of either Capital Expenditures, Alterations, or repairs required by Article 5 shall be the property of the party that paid for such work.

#### 7.1 Prohibition on Liens.

Tenant shall not, directly or indirectly, create or allow to remain and shall promptly discharge, at its expense, any Lien on the Leased Property, Tenant's leasehold interest therein, any Tenant's Personal Property now or hereafter owned, any Excess FF&E owned by Tenant, any Excess FF&E Leasehold Interest, Working Capital, or the Rent, other than (a) Permitted Liens, (b) liens for Real Estate Taxes and other Landlord Obligations, (c) subleases permitted by Article 16, (d) liens for Impositions or for sums arising from the application of Legal Requirements so long as the same (i) are not yet delinquent or (ii) are being contested in accordance with Article 8, (e) liens of mechanics, laborers, materialmen, suppliers or vendors incurred in the ordinary course of business that are not yet delinquent or are for sums that are being contested in accordance with Article 8, (f) any Facility Mortgage or other liens which are the responsibility of Landlord pursuant to the provisions of Article 20, (g) liens first arising prior to the Commencement Date, and (h) Landlord Notwithstanding the foregoing, but subject to all other applicable terms and conditions of this Lease, including without limitation the provisions hereof regarding a Change in Control, a pledge of the ownership interests in CCC or OpCo to secure bona fide Indebtedness shall not be deemed a violation of this Section 7.1, provided that a subsequent foreclosure or other realization upon such pledge shall be subject to the Change in Control provisions of this Lease.

All materialmen, contractors, artisans, mechanics and laborers and other persons contracting with Tenant with respect to the Leased Property, or any part thereof, are hereby charged with notice that Liens are expressly prohibited and that they must look solely to Tenant to secure payment for any work done or material furnished for Alterations or Capital Expenditures by Tenant or for any other purpose during the Term.

Tenant hereby acknowledges and agrees that, at all times while a Facility Mortgage is in effect, all Excess FF&E acquired by Tenant, directly or indirectly, from Landlord or leased by Tenant in accordance with Schedule 22.2

hereof shall be and remain subject to a first priority Lien in favor of the Facility Mortgagee. Tenant further acknowledges that Landlord has assigned and pledged (or may assign or pledge in the future) to the Facility Mortgagee, as additional security for Landlord's obligations under the loan secured by the Facility Mortgage, Landlord's rights and interests under the Landlord Liens with respect to any and all Tenant's Personal Property, Working Capital and Excess FF&E Leasehold Interest now owned or hereafter acquired by Tenant at any time while the Facility Mortgage remains in effect.

### 7.2 Landlord Lien.

In addition to any statutory landlord's lien and in order to secure (a) payment of the Rent and all other sums payable to Landlord hereunder by Tenant, (b) payment of the amounts owed under the Working Capital Note, (c) payment of any loss, cost or damage that Landlord may suffer by reason of Tenant's breach of this Lease, and (d) the performance of all of Tenant's other obligations hereunder and under the Working

Capital Note, Tenant and Landlord have, simultaneously with the execution of this Lease, entered into a Security Agreement (the "SECURITY AGREEMENT") whereby Tenant grants unto Landlord a security interest in, and an express contractual lien upon, (i) Tenant's Personal Property (other than Working Capital), and all proceeds therefrom (subject to any Permitted Liens), (ii) any Excess FF&E owned by Tenant, (iii) any Excess FF&E Leasehold Interest of Tenant, and (iv) any Working Capital of Tenant (collectively, together with any statutory lien rights, "LANDLORD LIENS"). Tenant's Personal Property shall not be removed from the Leased Property by Tenant at any time when a Default (with respect to any monetary obligation of Tenant to Landlord hereunder and under the Working Capital Note) or an Event of Default has occurred and is continuing except as otherwise permitted pursuant to Section 5.2. In addition, Tenant hereby grants unto Landlord a security interest in those contracts described in Section 5.3(d)(ii) hereof.

Upon Landlord's request, Tenant shall execute and deliver to Landlord financing statements in form sufficient to perfect the security interest of Landlord in (subject to the Facility Mortgagee's first-priority Lien thereon) (x) Tenant's Personal Property and the proceeds thereof, (y) the contracts described in Section 5.3(d)(ii) hereof, in accordance with the provisions of the applicable laws of the State, and (z) any and all Working Capital, Excess FF&E and Excess FF&E Leasehold Interests, in each case owned by Tenant from time to time during the Term of this Lease.

# ARTICLE 8 PERMITTED CONTESTS

Tenant shall have the right to contest the amount or validity of any Imposition, Legal Requirement, Insurance Requirement, Environmental Obligation, lien, attachment, levy, encumbrance, charge or claim (collectively, "CLAIMS") as to the Leased Property, by appropriate legal proceedings, conducted in good faith and with due diligence, provided that (a) the foregoing shall in no way be construed as relieving, modifying or extending any obligation of Tenant provided for in this Lease to pay any Claims as finally determined, (b) such contest, or the maintenance of any Lien during such contest, shall not cause Landlord or Tenant to be in default under any mortgage, deed of trust or other agreement encumbering the Leased Property or any interest therein or result in a Lien attaching to the Leased Property unless Tenant shall within ten (10) days thereafter, have such Lien released of record or deliver to Landlord a bond or other security reasonably satisfactory to Landlord, which shall be in form, amount, and issued by a surety reasonably satisfactory to Landlord, indemnifying Landlord against all costs and liabilities resulting from such Lien and the foreclosure or attempted foreclosure thereof, (c) no part of the Leased Property nor any Rent therefrom shall be in any immediate danger of sale, forfeiture, attachment or loss, and (d) Tenant shall indemnify and hold harmless Landlord from and against any cost, claim, damage, penalty or reasonable expense, including reasonable attorneys' fees, incurred by Landlord in connection therewith or as a result thereof. Tenant shall be entitled to any refund of any Claims and such charges and penalties or interest thereon which have been paid by Tenant or paid by Landlord and for which Landlord has been fully reimbursed by Tenant. If Tenant shall fail (x) to pay any Claims when finally determined, (y) to provide reasonable security therefor, or (z) to prosecute any such contest diligently and in good faith, Landlord may, upon reasonable notice to Tenant (which notice may be oral and shall not be required if Landlord shall

reasonably determine that the same is not practicable), pay such charges, together with interest and penalties due with respect thereto, and Tenant shall reimburse Landlord therefor, upon demand, as Additional Charges. Landlord agrees to join in any such proceedings if required legally to prosecute such contest, provided that Landlord shall not thereby be subjected to any liability or cost therefor (including, without limitation, for the payment of any costs or expenses in connection therewith).

# ARTICLE 9 INSURANCE AND INDEMNIFICATION

#### 9.1 General Insurance Requirements.

At all times during the Term and at any other time Tenant shall be in possession of the Leased Property, the Leased Property shall be insured against the risks and in the amounts described below. This insurance shall be written by companies authorized to issue insurance in the State. The policies must name the party obtaining the policy as the insured and the other party, the Manager and any Facility Mortgagee(s) as additional named insureds and/or (in the case of any Facility Mortgagees) as loss payees. Losses shall be payable to Landlord or Tenant as provided in Article 10 of this Lease. Any loss adjustment for coverages insuring both parties shall require the written consent of Landlord and Tenant, each acting reasonably and in good faith. The policies on the Leased Property (including the Facility, Fixtures and FF&E owned by Landlord), and on Tenant's Personal Property, shall, subject to Section 9.1(k), satisfy the requirements of any ground lease, mortgage, security agreement or other financing lien affecting the Leased Property (provided, however, that any insurance coverage maintained with respect to the Leased Property by Manager or Landlord in accordance with the Management Agreement shall be deemed to satisfy such requirements so long as the Management Agreement is in effect) and at a minimum shall include:

- (a) "All Risk" property insurance, including insurance against loss or damage by fire, vandalism and malicious mischief, explosion of steamboilers, pressure vessels or other similar apparatus, now or hereafter installed in the Facility, extended coverage perils, earthquake (if available at commercially reasonable rates) and all physical loss perils insurance, including, but not limited to, sprinkler leakage, in an amount equal to one hundred percent (100%) of the then full Replacement Cost thereof (as defined in Section 9.2), with the usual extended coverage endorsements, including a Replacement Cost Endorsement and Builder's Risk Coverage during the continuance of any construction at the Leased Property;
- (b) Loss of rent insurance (on the "SPECIAL FORM") in the minimum amount of one (1) year of Minimum Rent and Percentage Rent (based on the last twelve (12) months in which the Facility was operated for its Primary Intended Use) for the benefit of Landlord, and business interruption insurance on the "Special Form" in the amount of one year of projected net profit of Tenant from the Facility (exclusive of collection costs and any operating expenses that are considered by the applicable insurance company to be non-continuing as a result of any Casualty), for the benefit of Tenant;

- (c) Flood (if the Facility is located in whole or in part within an area identified as an area having special flood hazards and in which flood insurance has been made available under the National Flood Insurance Act of 1968, as amended, or the Flood Disaster Protection Act of 1973, as amended (or any successor acts thereto)) and such other hazards and in such amounts as may be customary for comparable properties in the area, said coverage to be in an amount equal to the lesser of the full Replacement Cost of the Facility or the maximum amount available;
- (d) Comprehensive general liability insurance, including bodily injury, property damage, and innkeeper's liability (including broad form contractual liability, fire legal liability and completed operations coverage) having policy limits as to claims with respect to the Leased Property of at least One Million Dollars (\$1,000,000) per occurrence, Two Million Dollars (\$2,000,000) aggregate per location, provided that such limits shall be modified to conform to any required underlying statutory coverage, and Umbrella coverage shall be provided having limits of One Hundred Million Dollars (\$100,000,000) per occurrence and in the aggregate, and attaching in excess of policy limits as to general liability, where applicable, and employer's liability coverage, covering each of the following: bodily injury, death, or property damage liability per occurrence, personal injury, general aggregate, products and completed operations, and "all risk legal liability" (including liquor law or "dram shop" liability, if liquor or alcoholic beverages are served on the Leased Property) with respect to Landlord and Tenant;
- (e) To the extent applicable at the Facility, comprehensive form automobile liability insurance for owned, non-owned and hired vehicles in the amount of \$1,000,000;
- (f) Fidelity bonds or blanket crime policies with limits and deductibles as may be reasonably determined by Landlord, covering Tenant's employees in job classifications normally bonded under prudent hotel management practices in the United States or otherwise required by law;
- (g) To the extent applicable at the Facility, garagekeeper's legal liability insurance covering both comprehensive and collision-type losses with a limit of liability of \$1,000,000 for any one occurrence;
- (h) Insurance coverage for claims by employees of Manager for wrongful termination, discrimination or sexual harassment;
- (i) Worker's compensation insurance coverage for all persons employed by Tenant on the Leased Property with statutory limits and otherwise with limits of and provisions in accordance with Legal Requirements, and employer's liability insurance having a limit of \$500,000;
- (j) To the extent applicable at the Facility, safe deposit box legal liability insurance covering property or guests while in a safe deposit box on the Leased Property for which Tenant is legally responsible with a limit of not less than \$100,000 in any one occurrence; and

(k) Such additional insurance and endorsements (and/or increased amounts of insurance hereinabove required) as may be reasonably required, from time to time, by Landlord, any Facility Mortgagee or any rating agency, or any existing or future ground lessor, provided that any incremental cost incurred as a result of such additional coverage shall be borne by Landlord.

#### 9.2 Responsibility for Insurance.

Tenant shall obtain or cause to be obtained the insurance and pay the premiums for the coverages described in Sections 9.1(d) through (j) and for that portion of the premium for the coverage described in Section 9.1(b) that is attributable to business interruption insurance, and Landlord shall obtain the insurance and pay the premiums for the coverages described in Sections 9(a) through (c) and (k) (excluding that portion of the premium for the coverage described in Section 9.1(b) that is attributable to business interruption insurance that is for the benefit of Tenant). The party responsible for the premium for any insurance coverage shall also be responsible for any and all deductibles and self-insured retentions in connection with such coverages. In the event that either party can obtain comparable insurance coverage required to be carried by the other party from comparable insurers and at a cost significantly less than that at which such other party can obtain such coverage, the parties shall cooperate in good faith to obtain such coverage at the lower cost and shall allocate the premiums therefor in accordance with the provisions of the first sentence of this Section 9.2.

Notwithstanding anything in this Article 9 to the contrary, so long as the Management Agreement is in full force and effect and Manager is maintaining the insurance required thereunder, Landlord's and Tenant's obligations to maintain the insurance required under this Article 9 shall be deemed to have been met (provided, that the costs therefor shall be allocated between Landlord and Tenant in the manner contemplated by the first sentence of this Section 9.2).

## 9.3 Replacement Cost.

"REPLACEMENT COST" as used herein, shall mean the actual replacement cost of the Leased Property requiring replacement from time to time, including an increased cost of construction endorsement, if available, and the cost of debris removal less exclusions provided in the standard form of fire insurance policy. In the event either party believes that the then full Replacement Cost has increased or decreased at any time during the Term, such party, at its own cost, shall have the right to have such full Replacement Cost redetermined by an independent accredited appraiser approved by the other, which approval shall not be unreasonably withheld or delayed. The party desiring to have the full Replacement Cost so redetermined shall forthwith, on receipt of such determination by such appraiser, give written notice thereof to the other. The determination of such appraiser shall be final and binding on the parties hereto, and Landlord shall forthwith conform the amount of the insurance carried to the amount so determined by the appraiser.

#### 9.4 Waiver of Subrogation.

Landlord and Tenant agree that (insofar as and to the extent that such agreement may be effective without invalidating or making it impossible to secure insurance coverage from responsible insurance companies doing business in the State) with respect to any loss covered by insurance then being carried by Landlord or Tenant, respectively, the party carrying such insurance and suffering said loss releases the other of and from any and all claims with respect to such loss; and they further agree that their respective insurance companies shall have no right of subrogation against the other on account thereof, even though extra premium may result therefrom.

## 9.5 Form Satisfactory, Etc.

Subject to the second paragraph of Section 9.2, all insurance policies and endorsements required pursuant to this Article 9 shall be fully paid for, nonassessable and, except for umbrella and flood coverage, shall contain such provisions and expiration dates and be in such form and amounts and issued by insurance carriers authorized to do business in the State, having a general policy holder's rating of A/VI in Best's latest rating guide (or such other higher rating or such other customarily used rating agency as may be required by the Facility Mortgagee, provided that any additional expense associated with such higher ratings shall be borne by Landlord), and otherwise satisfactory to Landlord and Tenant. Without limiting the foregoing, such policies shall include only deductibles reasonably approved by Landlord. The party responsible for obtaining any policy shall deliver policies or certificates thereof to the other party prior to their effective date (and, with respect to any renewal policy, thirty (30) days prior to the expiration of the existing policy), and, in the event either party shall fail to effect such insurance as herein required, to pay the premiums therefor or to deliver such policies or certificates to the other party or the Facility Mortgagee at the times required, the other party shall have the right, but not the obligation, after ten (10) days' written notice to the responsible party, to acquire such insurance and pay the premiums therefor, which amounts shall be reimbursed by the responsible party, together with interest accrued thereon at the Overdue Rate from the date such payment is made until the date repaid. All such policies shall provide the non-responsible party (and the Facility Mortgagee and Manager, if required by the same) thirty (30) days' prior written notice of any material modification, expiration or cancellation of such policy.

#### 9.6 Blanket Policy.

Notwithstanding anything to the contrary contained in this Article 9, Tenant's and Landlord's obligations to maintain the insurance herein required may be brought within the coverage of a so-called blanket policy or policies of insurance, provided that, except as otherwise approved by the other party in writing, (a) the coverage thereby afforded will not be reduced or diminished from that which would exist under a separate policy meeting all other requirements of this Lease, and (b) the requirements of this Article 9 (including an appropriate allocation of the costs for such blanket policy between Landlord and Tenant in a manner consistent with the first sentence of Section 9.2) are otherwise satisfied.

#### 9.7 No Separate Insurance.

Neither Landlord nor Tenant shall take out separate insurance concurrent in form or contributing in the event of loss with that required by this Article 9, or increase the amount of any existing insurance by securing an additional policy or additional policies, unless all parties having an insurable interest in the subject matter of such insurance are included therein as additional insureds and the loss is payable under such insurance in the same manner as losses are payable under the insurance required to be carried pursuant to this Lease. In the event either shall take out any such separate insurance or increase any of the amounts of the then existing insurance, it shall give the other party prompt Notice thereof.

### 9.8 General Indemnification of Landlord by Tenant.

Except as otherwise provided in Sections 4.3.2, 5.1, 9.9, 14.1 and 21.3.3, and except for liabilities, obligations, claims, damages, penalties, causes of action, costs and expenses (including, without limitation, attorneys' fees) arising from Landlord's or any other Landlord Indemnitee's or their respective agents' (including Manager, but only with respect to Manager's acts or omissions with respect to Landlord Obligations) willful misconduct or gross negligence, Tenant shall protect, indemnify and hold harmless Landlord and each Landlord Indemnitee for, from and against all liabilities, obligations, claims, damages, penalties, causes of action, costs and reasonable expenses (including, without limitation, reasonable attorneys' fees), to the maximum extent permitted by law, and notwithstanding the existence of any insurance provided for herein, but subject to Section 9.4 hereof, and without regard to the policy limits of any such insurance, imposed upon or incurred by or asserted against Landlord or any Landlord Indemnitee by reason of:

- (i) liabilities arising after the Commencement Date and during the Term under any leases, contracts, concession agreements or other agreements either (A) entered into by Tenant or any Tenant Indemnitee, or their respective agents (including Manager, except with respect to Manager's acts or omissions with respect to Landlord Obligations), with respect to the Facility, or (B) assigned to Tenant pursuant to Section 2.2(a) (except liabilities arising out of agreements described in this clause (B) for any employee claims by employees of Manager for wrongful termination, discrimination or sexual harassment,
- (ii) the gross negligence or willful misconduct of Tenant or any other Tenant Indemnitee, or their respective agents (including Manager, except with respect to Manager's acts or omissions with respect to Landlord Obligations), including any such gross negligence or willful misconduct giving rise to any employee claims by employees of Manager for wrongful termination, discrimination or sexual harassment,
- (iii) any accident, injury to or death of persons or loss of or damage to property occurring on or about the Leased Property or adjoining sidewalks or rights of way on or after the Commencement Date and during the Term, including, without limitation, any claims under liquor liability, "dram shop" or similar laws,
- (iv) any litigation, proceeding (other than Condemnation proceedings) or claim by governmental entities or other third parties to which Landlord or any Landlord

Indemnitee is made a party or participant or which is otherwise asserted against Landlord or any Landlord Indemnitee, relating to the Leased Property or Tenant's Personal Property or any use, misuse, non-use, condition, management, maintenance, or repair thereof, the occurrence giving rise to which litigation, proceeding or claim occurs on or after the Commencement Date and during the Term.

- (v) any Impositions that are the obligations of Tenant to pay pursuant to the applicable provisions of this Lease, and
- (vi) any failure on the part of Tenant or anyone claiming under Tenant to perform or comply with any of the terms of this Lease.

Tenant shall pay all amounts payable under this Section 9.8 within ten (10) days after demand therefor and, if not timely paid, such amounts shall bear interest at the Overdue Rate from the date of determination to the date of payment. Tenant, at its expense, shall contest, resist and defend any such claim, action or proceeding asserted or instituted against Landlord and shall not be responsible for any duplicate attorneys' fees incurred by Landlord or may compromise or otherwise dispose of the same, with Landlord's prior written consent (which consent may not be unreasonably withheld or delayed). In the event Landlord shall unreasonably withhold or delay its consent, Tenant shall not be liable pursuant to this Section 9.8 for any incremental increase in costs or expenses resulting therefrom. The obligations of Tenant under this Section 9.8 are in addition to the obligations set forth in Section 4.3.3 and 21.3.3 and shall survive the termination or expiration of this Lease.

Nothing in this Section 9.8 is intended to limit Tenant's right to assert a claim or defense against Landlord based upon the obligations imposed upon Landlord pursuant to Sections 4.3, 5.1, 9.9, 14.1 and 21.3.3.

#### 9.9 General Indemnification of Tenant by Landlord.

Except for liabilities arising from Tenant's or any other Tenant Indemnitee's, or their respective agents' (including Manager, except with respect to Manager's acts or omissions with respect to Landlord Obligations) willful misconduct or gross negligence, Landlord shall protect, indemnify and hold harmless Tenant and each Tenant Indemnitee for, from and against all liabilities, obligations, claims, damages, penalties, causes of action, costs and reasonable expenses (including, without limitation, reasonable attorneys' fees), to the maximum extent permitted by law, and notwithstanding the existence of any insurance provided for herein, but subject to Section 9.4 hereof, and without regard to the policy limits of any such insurance, imposed upon or incurred by or asserted against Tenant or any Tenant Indemnitee by reason of:

(i) liabilities arising at any time under any leases, contracts, concession agreements or other agreements with respect to the Facility, except liabilities arising after the Commencement Date and during the Term under any leases, contracts, concession agreements or other agreements either (a) entered into during the Term by Tenant or any other Tenant Indemnitee, or their respective agents (including Manager, except with

respect to Manager's acts or omissions with respect to Landlord Obligations) with respect to the Facility, or (b) assigned to Tenant pursuant to Section 2.2(a),

- (ii) the gross negligence or willful misconduct of Landlord or any other Landlord Indemnitee, or their respective agents (including Manager, but only with respect to Manager's acts or omissions with respect to Landlord Obligations),
- (iii) any accident, injury to or death of persons or loss of or damage to property occurring on or about the Leased Property or adjoining sidewalks or rights of way prior to the Commencement Date or after expiration or earlier termination of the Term, including, without limitation any claims under liquor liability, "dram shop" or similar laws,
- (iv) any litigation, proceeding (other than Condemnation proceedings) or claim by governmental entities or other third parties to which Tenant or any Tenant Indemnitee is made a party or participant or which is otherwise asserted against Tenant or any Tenant Indemnitee, relating to the Leased Property or any use, misuse, non-use, condition, management, maintenance, or repair thereof, the occurrence giving rise to which litigation, proceeding or claim occurs prior to the Commencement Date or after the expiration or earlier termination of the Term,
  - (v) any Landlord Obligations, and
- (vi) any failure on the part of Landlord or anyone claiming under Landlord (including Manager, but only with respect to Manager's acts or omissions with respect to Landlord Obligations) to perform or comply with any of the terms of this Lease.

Landlord shall pay all amounts payable under this Section 9.9 within ten (10) days after demand therefor and, if not timely paid, such amounts shall bear interest at the Overdue Rate from the date of determination to the date of payment. Landlord, at its expense, shall contest, resist and defend any such claim, action or proceeding asserted or instituted against Tenant and shall not be responsible for any duplicate attorneys' fees incurred by Tenant, or may compromise or otherwise dispose of the same with Tenant's prior written consent (which consent may not be unreasonably withheld or delayed). In the event Tenant shall unreasonably withhold or delay its consent, Landlord shall not be liable pursuant to this Section 9.9 for any incremental increase in costs or expenses resulting therefrom. The obligations of Landlord under this Section 9.9 are in addition to the obligations set forth in Section 4.3.2 and 21.3.3 and shall survive the termination of this Lease.

Nothing in this Section 9.8 is intended to limit Landlord's right to assert a claim or defense against Tenant based upon the obligations imposed upon Tenant pursuant to Sections 4.3, 5.1, 9.8 and 21.3.3.

### 9.10 Independent Contractor.

Except as otherwise approved by Landlord in writing, Tenant shall cause any person or company entering upon the Leased Property to provide any installation, construction or repair (each a "CONTRACTOR"), which (x) constitutes a Capital Expenditure

or (y) has an anticipated cost in excess of [\$\_\_\_\_\_] to (a) have in full force and effect Contractor's Liability Coverage (hereafter defined) effective throughout the period said Contractor is upon the Leased Property and for one year thereafter and (b) deliver within five (5) Business Days of Landlord's request a certificate ("CONTRACTOR'S INSURANCE CERTIFICATE") evidencing compliance with subpart (a) to Tenant prior to the Contractor's first entry upon the Leased Property. As used herein, the term "CONTRACTOR'S LIABILITY COVERAGE" means a comprehensive general liability insurance policy meeting the requirements of this Article 9 (as if required to be provided by Tenant) except the minimum policy limit shall be \$1,000,000 per occurrence and \$3,000,000 in the aggregate. Within thirty (30) days after delivery of Landlord's written request, Tenant shall deliver copies of all Contractor's Insurance Certificates to Landlord.

# ARTICLE 10

#### 10.1 Insurance Proceeds.

All proceeds of the insurance contemplated by Section 9.1(a)-(c)payable by reason of any loss, damage or destruction to the Leased Property, or any portion thereof ("CASUALTY"), and insured under any policy of property or casualty insurance required by Article 9 (other than proceeds of business interruption insurance and any insurance proceeds paid with respect to Tenant's Personal Property, which shall be paid to Tenant) shall be paid directly to Landlord, and paid out by Landlord from time to time for the reasonable costs of reconstruction or repair of the Leased Property necessitated by such Casualty, if and to the extent required by the provisions of Section 10.2; provided, however, that so long as no Event of Default shall have occurred and be continuing, all such proceeds less than or equal to \$1,000,000 shall be paid directly to Tenant and applied to the reasonable costs of restoration and repair of the Leased Property necessitated by such Casualty, and such losses may be adjusted without Landlord's consent. Any excess proceeds of insurance remaining after the completion of the restoration shall be paid to Landlord. All salvage resulting from any risk covered by insurance shall belong to Landlord, except for any amount thereof paid with respect to Tenant's Personal Property. If Landlord is not required and elects not to repair and restore as permitted under this Lease, and this Lease is terminated pursuant to Section 10.2, all such insurance proceeds shall be retained by Landlord, except for any amount thereof paid with respect to Tenant's Personal Property and any amount attributable to business interruption insurance.

### 10.2 Reconstruction in the Event of Casualty.

10.2.1 Facility Rendered Unsuitable for Its Primary Intended Use. If

during the Term the Leased Property is totally or partially destroyed by a Casualty and the Facility is thereby rendered Unsuitable for Its Primary Intended Use, as reasonably determined by Landlord, this Lease shall terminate as of the date of the Casualty and neither Landlord nor Tenant shall have any further liability hereunder except for any liabilities which have arisen or occurred prior to such termination, and those which expressly survive termination of this Lease, and Landlord shall be entitled to retain all Casualty insurance proceeds (except for any amount thereof paid with respect to Tenant's Personal Property and any amount attributable to business interruption insurance). If the

Lease is terminated in accordance with this Section 10.2.1, Landlord shall not be permitted to commence restoration of the Leased Property for a period of one (1) year after the date of the Casualty.

# 10.2.2 Facility Not Rendered Unsuitable for Its Primary Intended Use.

If during the Term the Leased Property is totally or partially destroyed by a Casualty, but the Facility is not thereby rendered Unsuitable for Its Primary Intended Use, as reasonably determined by Landlord, Landlord or, at Landlord's election, Tenant shall, subject to Section 10.2.3, promptly restore the Facility as provided in Section 10.2.4. Except as provided in Section 10.2.3, such damage or destruction shall not terminate this Lease.

# 10.2.3 Deficiency in Insurance Proceeds. If the cost of the repair $% \left( 1\right) =\left( 1\right) \left( 1\right)$

or restoration exceeds the amount of proceeds received by Landlord from the insurance required under Article 9 and Tenant is obligated to restore pursuant to Section 10.2.2 hereof, Landlord agrees, subject to this Section 10.2.3, to contribute any excess amounts needed to restore the Facility prior to requiring Tenant to commence such work. Such difference shall be made available by Landlord, together with any insurance proceeds, for application to the cost of repair and restoration in accordance with the provisions of Section 10.2.4. the event the sum of (a) the insurance proceeds released to Landlord, and (b) that portion of the deductible, if any, which is greater than five percent (5%) of the cost of the repair, is equal to at least ninety-five percent (95%) of the cost of the repair or restoration, Landlord shall fund the deficiency. In the event the sum of (y) the insurance proceeds, and (z) that portion of the deductible, if any, which is greater than five percent (5%) of the cost of the repair, is less than ninety-five percent (95%) of the cost of the repair or restoration, Landlord shall fund such deficiency in its sole discretion; provided, however, that in the event Landlord does not agree to make such deficiency available for restoration, either Landlord or Tenant may terminate this Lease by written notice to the other, whereupon this Lease shall terminate as provided in Section 10.2.1 and Landlord shall pay to Tenant a termination fee equal to Tenant's Operating Profit for the immediately preceding Fiscal Year. If Landlord or Tenant elects to terminate this Lease in accordance with this Section 10.2.3, Landlord shall not be permitted to commence restoration of the Leased Property for a period of one (1) year after the date of the Casualty.

# 10.2.4 Disbursement of Proceeds. In the event Tenant is required to

restore the Leased Property pursuant to Section 10.2.2, Tenant shall (or shall direct Manager to) commence promptly and continue diligently to perform the repair and restoration of the Leased Property, so as to restore the Leased Property in compliance with all Legal Requirements to substantially the same condition, to the extent reasonably practicable, as existed immediately before the damage or destruction and otherwise in accordance with this Lease. Landlord shall advance the insurance proceeds and, subject to the terms hereof, any additional amounts payable by Landlord pursuant to Section 10.2.3 to Tenant regularly during the repair and restoration period so as to permit payment for the cost of any such restoration and repair. Any such advances shall be made not more frequently than monthly within ten (10) Business Days after Tenant submits to Landlord a written requisition and substantiation therefor on AIA Forms G702 and G703 (or on such other form or forms as may be reasonably acceptable to Landlord). Landlord may, at its option, condition advancement of said insurance proceeds and other amounts on (i) the absence of

any uncured Event of Default, (ii) its approval of plans and specifications of an architect reasonably satisfactory to Landlord (which approval shall not be unreasonably withheld or delayed), (iii) general contractor's estimates, (iv) architect's certificates, (v) unconditional lien waivers of general contractors (if available), (vi) evidence of approval by all governmental authorities and other regulatory bodies whose approval is required and (vii) such other certificates as Landlord may, from time to time, reasonably require.

Landlord's obligation to disburse insurance proceeds under this Article 10 shall be subject to the release of such proceeds by any Facility Mortgagee to Landlord, and Tenant's obligation to restore the Leased Property pursuant to this Article 10 shall be subject to the release of available insurance proceeds by any Facility Mortgagee to Landlord or directly to Tenant or Manager and, in the event such proceeds are insufficient, Landlord electing to make such deficiency available therefor (and disbursement of such deficiency); provided, however, that Landlord and Tenant shall each have the same termination rights in the event of any Facility Mortgagee's failure or refusal to disburse insurance proceeds as they have with respect to Landlord's failure to disburse any deficiency in insurance proceeds, as provided in Section 10.2.3. If Landlord elects to terminate this Lease in accordance with this Section 10.2.4, Landlord shall not be permitted to commence restoration of the Leased Property for a period of one (1) year after the date of the Casualty.

10.3 Reconstruction in the Event of Damage or Destruction Not Covered by Insurance.

If during the Term the Facility is totally or materially damaged or destroyed by a risk not covered by the insurance described in Article 9, or if the proceeds of such insurance are not available to Landlord for restoration of the Facility, whether or not in either event such damage or destruction renders the Facility Unsuitable for Its Primary Intended Use, Landlord at its option shall either, (a) direct Tenant to restore the Facility at Landlord's sole cost and expense to substantially the same condition it was in immediately before such damage or destruction, in which case such damage or destruction shall not terminate this Lease, or (b) terminate this Lease and neither Landlord nor Tenant shall have any further liability thereunder, except for any liabilities which have arisen or occurred prior to such termination and those which expressly survive termination of this Lease. If Landlord elects to terminate this Lease in accordance with this Section 10.3, Landlord shall not be permitted to commence restoration of the Leased Property for a period of one (1) year after the date of the Casualty.

10.4 Tenant's Property and Business Interruption Insurance.

All insurance proceeds payable by reason of any loss of or damage to any of Tenant's Personal Property and the business interruption insurance maintained for the benefit of Tenant shall be paid to Tenant.

## 10.5 Abatement of Rent.

Any damage or destruction due to Casualty notwithstanding, this Lease shall remain in full force and effect (except as otherwise expressly provided in this Article 10)

and Tenant's obligation to pay Rent required by this Lease shall remain unabated by any Casualty which does not result in a reduction of Gross Revenues. If and to the extent that any Casualty results in a reduction of Gross Revenues (including proceeds of any business interruption insurance actually received by Tenant or that would have been received by Tenant if Tenant were in full compliance with the insurance requirements of Article 9) which would otherwise be realizable from the operation of the Facility, then Landlord shall receive all loss of rental income insurance and Tenant shall have no obligation to pay Rent with respect to any Accounting Period during the continuation of such Casualty in excess of the greater of (i) one-thirteenth (or one-twelfth, if Accounting Periods are then the same as calendar months) of the aggregate amount of Rent (excluding Additional Charges) paid to Landlord with respect to the last full Fiscal Year prior to such Casualty or (ii) the amount of Rent calculated with respect to such Accounting Period under Section 3.1.1 without regard to clause (A)(i) of such Section (which relates to Minimum Rent); provided, however, that if such damage or destruction was caused by Tenant's gross negligence or willful misconduct, Tenant shall remain liable for the amount of Rent which would have been payable hereunder at a rate equal to the average Rent during the last three (3) preceding Lease Years (or if three (3) Lease Years shall not have elapsed, the average during the preceding Lease Years), as if such Casualty had not occurred.

# ARTICLE 11

### 11.1 Total Condemnation, Etc.

If either (i) the whole of the Leased Property shall be taken by Condemnation or (ii) a Condemnation of less than the whole of the Leased Property renders the Leased Property Unsuitable for Its Primary Intended Use, as reasonably determined by Landlord, this Lease shall terminate as of the day of the Condemnation, and Tenant and Landlord shall seek the Award for their respective interests in the Leased Property as provided in Section 11.4.

#### 11.2 Partial Condemnation.

In the event of a Condemnation of less than the whole of the Leased Property such that the Leased Property is still suitable for its Primary Intended Use, as reasonably determined by Landlord, Tenant shall, to the extent that the Award and additional amounts disbursed by Landlord are sufficient therefor, commence promptly and continue diligently to restore the untaken portion of the Facility so that the Facility shall constitute a complete architectural unit of the same general character and condition (as nearly as may be possible under the circumstances) as the Facility existing immediately prior to such Condemnation, in full compliance with all Legal Requirements. Subject to the terms hereof, Landlord shall contribute to the cost of restoration that part of the Award necessary to complete such repair or restoration, together with severance and other damages awarded for the taken portion of the Facility (provided, however, that the amount of such contribution shall not exceed such cost), and such amounts shall be advanced to Tenant regularly during the restoration period so as to permit payment for the cost of such repair or restoration. Landlord may, at its option, condition advancement of such Award and other amounts on (i) the absence of any continuing Event of Default, (ii) its approval of

plans and specifications of an architect reasonably satisfactory to Landlord (which approval shall not be unreasonably withheld or delayed), (iii) general contractors' estimates, (iv) architect's certificates, (v) unconditional lien waivers of general contractors (if available), (vi) evidence of approval by all governmental authorities and other regulatory bodies whose approval is required and (vii) such other certificates as Landlord may, from time to time, reasonably require. Landlord's obligation under this Section 11.2 to disburse the Award and such other amounts shall be subject to (x) the collection of the Award by Landlord and (y) the satisfaction of any applicable requirements of the Facility Mortgage, and the release of such Award by the Facility Mortgagee. Tenant's obligation to restore the Leased Property shall be subject to the release of the Award and any additional funds to be disbursed by Landlord pursuant hereto required for restoration. Subject to Section 21.3.3, if Landlord has received the Award, but elects not to make the Award available to Tenant for restoration, then Tenant shall have the right to terminate this Lease and Landlord shall pay to Tenant a termination fee equal to the amount of Tenant's Operating Profit for the immediately preceding Fiscal Year. Subject to Section 21.3.3, if Landlord has not received the Award, or the Award is insufficient to restore the untaken portion of the Facility as provided above, then Landlord, in its sole discretion, shall have the right to terminate this Lease and neither Landlord nor Tenant shall have any further liability hereunder, except for any liabilities which have arisen or occurred prior to such termination and those which expressly survive termination of this Lease, and Landlord shall be entitled to retain the entire Award; provided that Tenant shall be permitted to seek a separate award for the value of Tenant's Personal Property that was taken in such Condemnation; and provided further that if Landlord elects to terminate this Lease in accordance with this Section 11.2, then Landlord shall not be permitted to commence restoration of the untaken portion of the Leased Property for a period of one (1) year after the date of the Condemnation.

#### 11.3 Abatement of Rent.

In the event of a partial Condemnation as described in Section 11.2 which does not result in a termination of this Lease by Landlord, the Rent shall be abated in the manner and to the extent that is fair, just, and equitable to both Tenant and Landlord, taking into consideration, among other relevant factors, the number of usable rooms, the amount of square footage, or the effect upon revenues of such partial Condemnation. If Landlord and Tenant are unable to agree upon the amount of such abatement within thirty (30) days after such partial Condemnation, the matter shall be submitted to appraisal as provided for in Article 19 hereof.

### 11.4 Allocation of Award.

Except as provided in the second sentence of this Section 11.4, the total Award shall be solely the property of and payable to Landlord. Any portion of the Award made for the taking of Tenant's leasehold interest in the Leased Property (valued without regard to any right of termination in Landlord that otherwise exists under this Article 11), loss of business during the remainder of the Term (determined without regard to any provision for termination that might otherwise arise under this Article 11), the taking of Tenant's Personal Property or Tenant's removal and relocation expenses shall be the sole property of and payable to Tenant (subject to the provisions of Section 11.2). Subject to the

rights of the Manager under the Management Agreement and the rights of any Facility Mortgagee under a Facility Mortgage, in any Condemnation proceedings, Landlord and Tenant shall each seek its own Award in conformity herewith, at its own expense.

# ARTICLE 12 TENANT DEFAULTS; REMEDIES

#### 12.1 Event of Default.

The occurrence of any one or more of the following events shall constitute an "EVENT OF DEFAULT" hereunder:

- (a) Tenant fails (i) to make any payment of the Minimum Rent or Percentage Rent payable hereunder when due and such failure continues for a period of ten (10) days after the date due, or (ii) subject to the right to contest same pursuant to Article 8 hereof, to make any required payments of Additional Charges within ten (10) days following Notice from Landlord that such payment is due and owing and unpaid.
- (b) Tenant fails to maintain the insurance coverages that it is required to maintain under Article 9.
- (c) Except as otherwise expressly provided herein, Tenant defaults in the due observance or performance of any of the terms, covenants or agreements contained herein to be performed or observed by it (other than as specified in clauses (a) and (b) above), and, in either case, such default continues for a period of thirty (30) days after Notice thereof from Landlord to Tenant; provided, however, that if such default is curable but such cure cannot be accomplished with due diligence within such period of time and if, in addition, Tenant commences to cure such default within thirty (30) days after Notice thereof from Landlord and thereafter prosecutes the curing of such default with all due diligence, such period of time shall be extended to such period of time (not to exceed one hundred twenty (120) days in the aggregate, subject to Unavoidable Delay) as may be necessary to cure such default, provided further that the cure rights shall not apply to any breach of a Tenant covenant under Section 22.1, 22.3, 22.4, or 22.5.
- (d) Any obligation of Tenant in respect of any Indebtedness (other than Tenant's obligations under any Excess FF&E Lease that constitutes Indebtedness) in a principal amount in excess of (\$1,000,000) for money borrowed or for the deferred purchase price of any material property or services, is declared to be, or as a result of acceleration becomes, due and payable prior to the stated maturity thereof.
- (e) There occurs a final unappealable determination by applicable federal or State authorities of the revocation or limitation of any material license (including, but not limited to, any gaming license), permit, certification, or approval required for the material lawful operation of the Facility in accordance with its Primary Intended Use or the loss or limitation of any material license (including but not limited to any gaming license), permit, certification, or approval under any other circumstances under which Tenant is required to cease its operation of the Facility in accordance with its Primary Intended Use at the time of such loss or limitation, which revocation, limitation or loss is

not caused by actions of Landlord or its Affiliates or which is not beyond the reasonable control of Tenant.

- (f) Tenant is generally not paying its debts as they become due, or Tenant makes a general assignment for the benefit of creditors.
- (g) Any petition is filed by or against Tenant under the Federal bankruptcy laws, or any other proceeding is instituted by or against Tenant seeking to adjudicate it a bankrupt or insolvent, or seeking liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors, or seeking the entry of an order for relief or the appointment of a receiver, trustee, custodian or other similar official for Tenant or for any substantial part of the property of Tenant, and, in the case of any involuntary petition filed or proceeding instituted against Tenant only, such proceeding is not dismissed within sixty (60) days after institution thereof, or Tenant takes any action to authorize or effect any of the actions set forth above in this paragraph.
- (h) Tenant causes or institutes any proceeding for its dissolution or termination.
- (i) Tenant ceases operation of the Leased Property for its Primary Intended Use for a period in excess of thirty (30) consecutive days, except as a result of a Casualty, other Emergency Situations, the matters set forth in Section 23.17 or partial or complete Condemnation of or to the Facility or of or to the immediate surroundings so as to prohibit reasonable access by patrons to the Facility.
- (j) The estate or interest of Tenant in the Leased Property or any part thereof is levied upon or attached in any proceeding and the same is not vacated or discharged within the later of (i) one hundred and twenty (120) days after commencement thereof, unless the amount in dispute is less than \$250,000, in which case Tenant shall give Notice to Landlord of the dispute but Tenant may defend in any suitable way, and (ii) thirty (30) days after receipt by Tenant of Notice thereof from Landlord (unless Tenant shall be contesting such lien or attachment in good faith in accordance with Article 8).
  - (k) Any Change in Control occurs.
- (1) Tenant, OpCo or CCC defaults under the terms of any of the Related Agreements beyond the expiration of any applicable notice and cure periods.
- (m) CCC fails to maintain for three (3) consecutive Accounting Periods either (i) a Tangible Net Worth equal to or greater than (x) the maximum liability under the Guarantee from time to time less (y) amounts held in any lockbox or cash collateral account under this Lease, any other Pool Lease, any Other Lease, the Pooling Agreement or any pooling and security agreement relating to Other Leases immediately after payment of Rent under Section 3.1.1 with respect to the preceding Accounting Period, or (ii) a Consolidated Coverage Ratio of at least 1.4 to 1.0.

(n) For so long as this Lease is subject to the Pooling Agreement, there is a default that remains uncured beyond the expiration of applicable notice and cure periods by any of the Other Tenants under Section 12.1(a)(i) or Section 12.1(c) (by reason of a breach of Section 7.1 or Section 21.6.3) of any Pool Lease (other than this Lease); provided that, if a notice of termination has been given pursuant to Section 10(a) of the Guarantee and the six-month period specified in Section 10(a)(iii) of the Guarantee has commenced, then a default that remains uncured beyond the expiration of applicable notice and cure periods occurring under Section 12.1(a)(i) or 12.1(c) (by reason of a breach of Section 7.1 or 21.6.3) of any Pool Lease (other than this Lease) after commencement of such period shall not be a default under this paragraph (n).

In any such event, Landlord, in addition to all other remedies available to it, may terminate this Lease by giving Notice thereof to Tenant, and upon the expiration of the time, if any, fixed in such Notice, this Lease shall terminate and all rights of Tenant under this Lease shall cease. Subject to Section 23.11, Landlord shall have, and may exercise in its sole and absolute discretion, all, or none of the, rights and remedies available at law and in equity to Landlord as a result of Tenant's breach of this Lease; provided, however, that notwithstanding anything set forth herein to the contrary, (A) Landlord's sole remedy for an Event of Default under Section 12.1(k) and (m) shall be to terminate this Lease, and (B) Landlord's actual damages in the event of a breach by Tenant of any of its obligations pursuant to Article 22 and a resulting loss of REIT status by Host REIT shall include, without limitation, amounts equal to income taxes paid by Host REIT and (without duplication) loss of value of Host REIT, both to the extent attributable to the loss of REIT status; provided that any termination as a result of an Event of Default under Section 12.1(k) shall occur no later than 450 days after the applicable Change in Control (or such shorter period as may be expressly provided in the Management Agreement or Consent and Assignment) and upon not less than 30 days' Notice to Tenant. If Landlord terminates this Lease because of an Event of Default under Section 12.1(k) above, and the Change in Control does not involve an Adverse Party, then Landlord shall pay to Tenant a termination fee equal to Tenant's Operating Profit for the immediately preceding Fiscal Year.

Upon the occurrence of an Event of Default, Landlord may, in addition to any other remedies provided herein, enter upon the Leased Property and take possession thereof, and either (i) retain any and all of Tenant's Personal Property on the Leased Property, without liability for trespass or conversion (Tenant hereby waiving any right to Notice or hearing prior to such taking of possession by Landlord) or (ii) sell the same at public or private sale, after giving Tenant reasonable Notice of the time and place of any public or private sale, at which sale Tenant or its assigns may purchase all or any portion of Tenant's Personal Property. Unless otherwise provided by law and without intending to exclude any other manner of giving Tenant reasonable notice, the requirement of reasonable Notice shall be met if such Notice is given at least ten (10) Business Days before the date of sale. The proceeds from any such disposition, less all expenses incurred in connection with the taking of possession, holding and selling of such property (including reasonable attorneys' fees) shall belong to Landlord and shall be applied as a credit against the indebtedness which is secured by the security interest granted in Section 7.2. Any surplus shall be paid to Tenant or as otherwise required by law, and Tenant shall pay any deficiency to Landlord, as Additional Charges, upon demand.

The foregoing provisions of this Section 12.1 notwithstanding, in the event that Landlord exercises its right to terminate this Lease as the result of an Event of Default that first occurs after a notice of termination has been given pursuant to Section 10(a) of the Guarantee and the six-month period specified in Section 10(a)(iii) of the Guarantee has commenced, Landlord shall be entitled to all accrued Rent remaining unpaid as of the date of termination but shall not be entitled to an award based on lost future Rents under this Lease, Landlord's sole remedy in such event being termination of the Lease and payment of all accrued but unpaid Rent.

The foregoing provisions hereof notwithstanding, Landlord shall have no right to assert any remedy hereunder in respect of an Event of Default, and an Event of Default shall be deemed to no longer exist, if Tenant (or Guarantor, or its Affiliate, pursuant to the Guarantee or otherwise) cures an Event of Default (A) under Section 12.1(a) prior to the earlier of (x) the commencement by Landlord of the exercise of any remedy under this Lease by Landlord or (y) Landlord's Notice to Tenant stating that an Event of Default exists and further stating Landlord's intention to assert one or more remedies hereunder, and (B) under any of Section 12.1(b)-(m), prior to the commencement by Landlord of the exercise of any remedy under this Lease by Landlord.

#### 12.2 Remedies.

None of (a) the termination of this Lease pursuant to Section 12.1, (b) the repossession of the Leased Property, (c) the failure of Landlord to relet the Leased Property, or (d) the reletting of the Leased Property, shall relieve Tenant of its liability and obligations hereunder, all of which shall survive any such termination, repossession or re-letting. In the event of any such termination, Tenant shall forthwith pay to Landlord all Rent due and payable with respect to the Leased Property through and including the date of such termination. Thereafter, Tenant, until the earlier of what would have been the end of the Term of this Lease in the absence of such termination, or the date on which Tenant pays Landlord the liquidated final damages described in the next paragraph of this Section 12.2, and whether or not the Leased Property or any portion thereof shall have been re-let, shall be liable to Landlord for, and shall pay to Landlord, as current damages, the Rent and other charges which would be payable hereunder for the remainder of the Term had such termination not occurred, less the net proceeds, if any, of any re-letting or other operation by or on behalf of Landlord of the Leased Property, after deducting all expenses in connection with such reletting, including, without limitation, all repossession costs, brokerage commissions, legal expenses, attorneys' fees, advertising, expenses of employees, alteration costs and expenses of preparation for such reletting. Tenant shall pay such current damages to Landlord monthly on the days on which the Minimum Rent would have been payable hereunder if this Lease had not been so terminated.

At any time after such termination, at Landlord's election, whether or not Landlord shall have previously collected any such current damages, as liquidated final damages beyond the date of such termination, Tenant shall pay to Landlord an amount equal to the present value (discounted at a rate of twelve percent (12%) per annum) of the excess, if any, of the Rent and other charges which would be payable hereunder from the date of such termination (assuming that, for the purposes of this paragraph, annual payments by Tenant on account of Impositions would be the same as payments required for

the immediately preceding twelve calendar months, or if less than twelve calendar months have expired since the Commencement Date, the payments required for such lesser period projected to an annual amount) for what would be the then unexpired Term of this Lease if the same remained in effect, over the Fair Market Rental for the same period. Nothing contained in this Lease shall, however, limit or prejudice the right of Landlord to prove and obtain in proceedings for bankruptcy or insolvency an amount equal to the maximum allowed by any statute or rule of law in effect at the time when, and governing the proceedings in which, the damages are to be proved, whether or not the amount be greater than, equal to, or less than the amount of the loss or damages referred to above

Percentage Rent for the purposes of this Section 12.2 shall be a sum equal to (i) the average of the annual amounts of Percentage Rent for the three (3) Fiscal Years immediately preceding the Fiscal Year in which the termination, re-entry or repossession takes place, or (ii) if three (3) Fiscal Years shall not have elapsed, the Percentage Rent during the preceding Fiscal Year during which this Lease was in effect, or (iii) if one (1) Fiscal Year has not elapsed, the amount derived by annualizing the Percentage Rent from the Commencement Date of this Lease.

In case of any Event of Default, re-entry, expiration and dispossession by summary proceedings or otherwise, Landlord may, subject to the rights of Manager under the Management Agreement, (a) relet the Leased Property or any part or parts thereof, either in the name of Landlord or otherwise, for a term or terms which may, at Landlord's option, be equal to, less than or exceed the period which would otherwise have constituted the balance of the Term and may grant concessions or free rent to the extent that Landlord considers advisable and necessary to relet the same, and (b) make such reasonable alterations, repairs and decorations in the Leased Property or any portion thereof as Landlord, in its sole and absolute discretion, considers advisable and necessary for the purpose of reletting the Leased Property; and the making of such alterations, repairs and decorations shall not operate or be construed to release Tenant from liability hereunder as aforesaid. Subject to the last sentence of this paragraph, Landlord shall in no event be liable in any way whatsoever for any failure to relet all or any portion of the Leased Property, or, in the event that the Leased Property is relet, for failure to collect the rent under such reletting. To the maximum extent permitted by law, Tenant hereby expressly waives any and all rights of redemption granted under any present or future laws in the event of Tenant being evicted or dispossessed, or in the event of Landlord obtaining possession of the Leased Property, by reason of the occurrence and during the continuation of an Event of Default hereunder. Landlord covenants and agrees in the event of any termination of this Lease as a result of an Event of Default to use commercially reasonable efforts to mitigate its damages.

### 12.3 Waivers.

LANDLORD AND TENANT WAIVE, TO THE EXTENT PERMITTED BY LAW, ANY RIGHT TO A TRIAL BY JURY IN CONNECTION WITH ANY MATTER ARISING OUT OF OR IN ANY WAY CONNECTED WITH THIS LEASE, THE RELATIONSHIP OF LANDLORD AND TENANT AND/OR TENANT'S USE OR OCCUPANCY OF THE LEASED PROPERTY. TENANT WAIVES, TO THE EXTENT

PERMITTED BY LAW, THE BENEFIT OF ANY LAWS NOW OR HEREAFTER IN FORCE EXEMPTING PROPERTY FROM LIABILITY FOR RENT OR FOR DEBT.

### 12.4 Application of Funds.

Any payments received by Landlord under any of the provisions of this Lease during the existence or continuance of any Event of Default (and any payment made to Landlord rather than Tenant due to the existence of any Event of Default) shall be applied to Tenant's obligations under this Lease in such order as Landlord may determine or as may be prescribed by the laws of the State.

#### 12.5 Landlord's Right to Cure Tenant's Default.

If (i) a Default shall have occurred and is continuing which in the reasonable judgment of Landlord requires immediate action on the part of Landlord or (ii) an Event of Default shall have occurred and is continuing, Landlord, after Notice to Tenant (which Notice need not precede such action if Landlord shall reasonably determine immediate action is necessary to protect person or property), without waiving or releasing any obligation of Tenant and without waiving or releasing any Event of Default, may (but shall not be obligated to), at any time thereafter, make such payment or perform such act for the account and at the expense of Tenant, and may, to the maximum extent permitted by law, enter upon the Leased Property or any portion thereof for such purpose and take all such action thereon as, in Landlord's sole and absolute discretion, may be necessary or appropriate therefor, including the management of the Facility by Landlord or its designee, and Tenant hereby irrevocably appoints, in the event of such election by Landlord, Landlord or its designee as the operator of the Facility and its attorney in fact for such purpose, irrevocably and coupled with an interest, in the name, place and stead of Tenant. No such entry shall be deemed an eviction of Tenant. All reasonable costs and expenses (including, without limitation, reasonable attorneys' fees) incurred by Landlord in connection therewith, together with interest thereon (to the extent permitted by law) at the Overdue Rate from the date such sums are paid by Landlord until repaid, shall be paid by Tenant to Landlord, on demand.

#### ARTICLE 13 HOLDING OVER

Any holding over by Tenant after the expiration or sooner termination of this Lease (other than with the express written consent of Landlord) shall be treated as a tenancy at sufferance at a rate equal to (a) one and one-half (1.5) times one-twelfth of the aggregate Minimum Rent and Percentage Rent payable with respect to the last Fiscal Year of the Term, (b) all Additional Charges accruing during the applicable Accounting Period, and (c) all other sums, if any, payable by Tenant under this Lease with respect to the Leased Property during the applicable Accounting Period. Tenant shall also pay to Landlord all damages (direct or indirect) sustained by reason of any such holding over. Otherwise, such holding over shall be on the terms and conditions set forth in this Lease, to the extent applicable. Nothing contained herein shall constitute the consent, express or implied, of Landlord to the holding over of Tenant after the expiration or earlier termination of this Lease.

# ARTICLE 14 LANDLORD NOTICE OBLIGATION; LANDLORD DEFAULT

#### 14.1 Landlord Notice Obligation.

Landlord shall give prompt Notice to Tenant and Manager of any matters materially affecting the Leased Property of which Landlord receives written notice or actual knowledge and, to the extent Tenant otherwise has no notice or actual knowledge thereof, Landlord shall be liable for any liabilities arising from the failure to deliver such Notice to Tenant. Except as expressly set forth herein or in the Consent and Assignment, Landlord shall not amend the Management Agreement without Tenant's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed.

#### 14.2 Landlord Default.

The occurrence of the following events shall constitute a "LANDLORD DEFAULT":

- (a) Landlord fails to make any payment due hereunder when due and such failure continues for a period of ten (10) days following Notice from Tenant that such payment is due and owing and unpaid.
- (b) Landlord fails to maintain the insurance coverages that it is required to maintain under Article 9.
- (c) Landlord defaults in the due observance or performance of any of the terms, covenants or agreements contained herein to be performed or observed by it (other than as specified in clauses (a) and (b) above), and, in either case, such default continues for a period of thirty (30) days after Notice thereof from Tenant to Landlord; provided, however, that if such default is curable but such cure cannot be accomplished with due diligence within such period of time and if, in addition, Landlord commences to cure such default within thirty (30) days after Notice thereof from Tenant and thereafter prosecutes the curing of such default with all due diligence, such period of time shall be extended to such period of time (not to exceed one hundred twenty (120) days in the aggregate, subject to Unavoidable Delay) as may be necessary to cure such default.
- (d) Landlord is generally not paying its debts as they become due, or Landlord makes a general assignment for the benefit of creditors.
- (e) Any petition is filed by or against Landlord under the Federal bankruptcy laws, or any other proceeding is instituted by or against Landlord seeking to adjudicate it a bankrupt or insolvent, or seeking liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors, or seeking the entry of an order for relief or the appointment of a receiver, trustee, custodian or other similar official for Landlord or for any substantial part of the property of Landlord and, in the case of any involuntary petition filed or proceeding instituted against Landlord only, such proceeding is not

dismissed within sixty (60) days after institution thereof, or Landlord takes any action to authorize or effect any of the actions set forth above in this paragraph.

- (f) Landlord causes or institutes any proceeding for its dissolution or termination.

Subject to Section 23.11, in the event of a Landlord Default, Tenant shall have and may exercise all rights and remedies available at law and in equity, including, without limitation, the right to pursue an action for damages against Landlord; provided, however, that except as otherwise expressly provided in this Lease, Tenant shall have no right to terminate this Lease for any Landlord Default hereunder and no right to offset or counterclaim against any Rent or other charges due hereunder. If Landlord shall in good faith dispute the occurrence of any Landlord Default and Landlord shall give Notice thereof to Tenant, setting forth, in reasonable detail, the basis therefor, and Tenant and Landlord shall fail, in good faith, to resolve any such dispute within ten (10) days after Landlord's Notice of dispute, then either may submit the matter to arbitration under Article 15 for resolution, and no Landlord Default shall be deemed to have occurred and Landlord shall have no obligation with respect thereto until final adverse determination thereof in such arbitration. In the event of any such adverse determination, Landlord shall pay to Tenant interest on any disputed funds at the Overdue Rate, from the date demand was made until paid, and in the event Landlord fails to make such payment within fifteen (15) days after such adverse determination, Tenant shall be entitled to offset such amount against the payment or, if necessary, payments of Rent next coming due hereunder. If Tenant reasonably determines that immediate action is necessary to protect person or property or to comply with Legal Requirements, Tenant may forthwith cure the Landlord Default and invoice Landlord for costs and expenses (including reasonable attorneys' fees and court costs) incurred by Tenant in curing the same, together with interest thereon from the date Landlord receives Tenant's Notice, at the Overdue Rate.

# ARTICLE 15 ARBITRATION

#### 15.1 Arbitration.

In each case specified in this Lease in which it shall become necessary to resort to arbitration, such arbitration shall be determined as provided in this Article 15. The party desiring such arbitration shall give Notice to that effect to the other party, specifying the nature of the dispute, the amount involved (if any), and the remedy sought. An arbitrator shall be selected by mutual agreement of the parties, or if they cannot agree within thirty (30) days of such Notice, by appointment made by the American Arbitration Association ("AAA") from among the members of its panels who are qualified and who have experience in resolving matters of a nature similar to the matter to be resolved by arbitration.

#### 15.2 Arbitration Procedures.

Any arbitration commenced pursuant to Section 15.1 shall be conducted in accordance with the AAA's Rules of Commercial Arbitration in business disputes. A single arbitrator shall be designated and shall resolve the The arbitrator's decision shall be binding on all parties, shall not be subject to further review or appeal except as otherwise allowed by applicable law and may be filed in and enforced by a court of competent jurisdiction. The arbitrator shall award to the prevailing party, if any, as determined by the arbitrator, all of its costs and expenses, including attorneys' fees, arbitrator's fees, and reasonable out-of-pocket expenses of any kind, except as otherwise specified below in this Section 15.2. The term "prevailing party," as used in the preceding sentence, shall mean the party whose position is most nearly upheld in arbitration. Upon the failure of either party (the "noncomplying party") to comply with the arbitrator's decision, the arbitrator shall be empowered, at the request of the other party, to order such compliance by the non-complying party and to supervise or arrange for the supervision of the noncomplying party's obligation to comply with the arbitrator's decision, all at the expense of the non-complying party. To the maximum extent practicable, the arbitrator and the parties, and the AAA, if applicable, shall take any action necessary to insure that the arbitration shall be concluded within ninety (90) days of the filing of such dispute. Unless otherwise agreed in writing by the parties or required by the arbitrator or the AAA, if applicable, arbitration proceedings hereunder shall be conducted in the Washington, DC, metropolitan area. Notwithstanding formal rules of evidence, each party may submit such evidence as each party deems appropriate to support its position, and the arbitrator shall have access to and the right to examine all books and records of Landlord and Tenant and, subject to the Management Agreement, the Manager regarding the Leased Property during the arbitration The consideration of the parties to be bound by arbitration is not only the waiver of trial by jury but also the waiver of any rights to appeal the arbitration finding.

# ARTICLE 16 SUBLETTING AND ASSIGNMENT

#### 16.1 Subletting and Assignment.

Except as otherwise expressly provided in this Section 16.1, Tenant shall not, without the prior written consent of Landlord (which consent may be withheld in Landlord's sole and absolute discretion), assign, mortgage, pledge, hypothecate, encumber or otherwise transfer this Lease or sublease (which term shall be deemed to include the granting of concessions, licenses and the like) all or any part of the Leased Property, or suffer or permit this Lease or the leasehold estate created hereby or any other rights arising under this Lease to be assigned, transferred, mortgaged, pledged, hypothecated or encumbered, in whole or in part, whether voluntarily, involuntarily or by operation of law, or permit the use or occupancy of the Leased Property by anyone other than Tenant or Manager (except in accordance with Section 21.3.4), or permit the Leased Property to be offered or advertised for assignment or subletting. Notwithstanding the foregoing, (i) any sublease to a subtenant, licensee or concessionaire made in the ordinary course of operating the Facility (a) shall not require Landlord's consent if it is for a term of not more than five (5) years and involves not more than one thousand (1,000) square feet of rentable space, or (b) shall require Landlord's consent, but such consent shall not be unreasonably

withheld, conditioned or delayed, if it is for a term of more than five (5) years or involves more than one thousand (1,000) square feet of rentable space and (ii) Tenant may, in each instance, after Notice to Landlord, assign this Lease to any Qualified Affiliate in accordance with Section 21.6.6, so long as such assignment will not violate or affect any applicable Legal Requirements or Insurance Requirements. For purposes of this Section 16.1, an assignment of this Lease (other than to a Qualified Affiliate in accordance with Section 21.6.6) shall be deemed to constitute a Change in Control.

If this Lease is assigned or if the Leased Property or any part thereof is sublet in contravention of this Lease, Landlord may collect the rents from such assignee, subtenant or occupant, as the case may be, and apply the net amount collected to the Rent herein reserved, but no such collection shall be deemed (i) a waiver of the provisions set forth in the first paragraph of this Section 16.1, (ii) the acceptance by Landlord of such assignee, subtenant or occupant, as the case may be, as a tenant, or (iii) a release of Tenant from the future performance by Tenant of its covenants, agreements or obligations contained in this Lease.

Tenant, as the debtor in possession, or the trustee for Tenant (collectively, the "TRUSTEE") in any proceeding under Title 11 of the United States Bankruptcy Code relating to Bankruptcy, as amended (the "BANKRUPTCY CODE"), shall not have the right to assign this Lease or sublet the Leased Property to an assignee or sublessee that (i) is a competitor of Landlord or (ii) is not a capable, reliable, qualified Person of good reputation and character with the financial capacity to satisfy Tenant's obligations under this Lease. The Trustee shall not have the right to assign this Lease or sublet the Leased Property to a real estate investment trust that is, or intends to be, publicly traded.

In the event that Tenant becomes the subject of any proceeding under Title 11 of the Bankruptcy Code, Tenant covenants and agrees that: (i) it shall promptly upon demand therefor from Landlord, but in no event later than sixty (60) days (as such time may be extended by a bankruptcy court in such proceeding) after the commencement of such proceeding (a "TENANT BANKRUPTCY"), announce its decision whether to assume or reject this Lease and Tenant's obligations under the Consent and Assignment and the Assigned Agreements, and promptly take and diligently pursue such actions as may be necessary to authorize and implement such decision; and (ii) it shall either assume this Lease and all of Tenant's obligations under the Consent and Assignment and the Assigned Agreements to the extent such Assigned Agreements have not expired or terminated in accordance with their respective terms, or it shall reject this Lease and all of Tenant's obligations under the Consent and Assignment and the Assigned Agreements. In a Tenant Bankruptcy, Tenant covenants and agrees that it cannot cure any defaults under this Lease and cannot provide adequate assurances of future performance of this Lease without curing any and all monetary and non-monetary defaults of Tenant's obligations under the Consent and Assignment and the Assigned Agreements and providing adequate assurances of Tenant's future performance of its obligations under the Consent and Assignment and the Assigned Agreements.

No subletting or assignment shall in any way impair the continuing primary liability of Tenant hereunder, and no consent to any subletting or assignment in a

particular instance shall be deemed to be a waiver of the prohibition set forth in this Section 16.1. No assignment, subletting or occupancy shall affect any Primary Intended Use. Any subletting, assignment or other transfer of Tenant's interest under this Lease in contravention of this Section 16.1 shall be voidable at Landlord's option.

#### 16.2 Required Sublease Provisions.

Any sublease of all or any portion of the Leased Property entered into on or subsequent to the Commencement Date shall provide (a) that it is subject and subordinate to this Lease and to the matters to which this Lease is or shall be subject or subordinate; (b) that in the event of termination of this Lease or reentry or dispossession of Tenant by Landlord under this Lease, Landlord may, at its option, terminate such sublease or take over all of the right, title and interest of Tenant, as sublessor under such sublease, and such subtenant shall, at Landlord's option, attorn to Landlord pursuant to the then executory provisions of such sublease, except that neither Landlord nor the Facility Mortgagee, as holder of a mortgage or as Landlord under this Lease, if such mortgagee succeeds to that position, shall (i) be liable for any act or omission of Tenant under such sublease, (ii) be subject to any credit, counterclaim, offset or defense which theretofore accrued to such subtenant against Tenant, (iii) be bound by any previous modification of such sublease requiring Landlord's consent hereunder and not consented to in writing by Landlord or by any previous prepayment of more than one (1) month's rent, (iv) be bound by any covenant of Tenant to undertake or complete any construction of the Leased Property or any portion thereof, (v) be required to account for any security deposit of the subtenant other than any security deposit actually delivered to Landlord by Tenant, (vi) be bound by any obligation to make any payment to such subtenant or grant any credits, except for services, repairs, maintenance and restoration provided for under the sublease that are to be performed after the date of such attornment, (vii) be responsible for any monies owing by Tenant prior to the date of attornment to the credit of such subtenant, or (viii) be required to remove any Person occupying any portion of the Leased Property; and (c) in the event that such subtenant receives a written Notice from Landlord or the Facility Mortgagee stating that an Event of Default has occurred and is continuing, such subtenant shall thereafter be obligated to pay all rentals accruing under such sublease directly to the party giving such Notice or as such party may direct. All rentals received from such subtenant by Landlord or the Facility Mortgagee, as the case may be, shall be credited against the amounts owing by Tenant under this Lease, and such sublease shall provide that the subtenant thereunder shall, at the request of Landlord, execute a suitable instrument in confirmation of such agreement to attorn. An original counterpart of each such sublease and assignment and assumption, duly executed by Tenant and such subtenant or assignee, as the case may be, in form and substance reasonably satisfactory to Landlord, shall be delivered promptly to Landlord upon request and (a) in the case of an assignment, the assignee shall assume in writing and agree to keep and perform all of the terms of this Lease on the part of Tenant to be kept and performed and shall be, and become, jointly and severally liable with Tenant for the performance thereof and (b) in case of either an assignment or subletting, Tenant shall remain primarily liable, as principal rather than as surety, for the prompt payment of the Rent and for the performance and observance of all of the covenants and conditions to be performed by Tenant hereunder.

The provisions of this Section 16.2 shall not be deemed a waiver of the provisions set forth in the first paragraph of Section 16.1.

16.3 No Right of Tenant to Mortgage Its Leasehold.

Notwithstanding any other provision of this Lease to the contrary, Tenant shall not assign its interest in this Lease as collateral for Indebtedness

# ARTICLE 17 ESTOPPEL CERTIFICATES AND FINANCIAL STATEMENTS

### 17.1 Estoppel Certificates.

At any time and from time to time, upon not less than ten (10) Business Days prior Notice by either party, the non-requesting party shall furnish to the requesting party, or a designee thereof, an Officer's Certificate certifying that this Lease is unmodified and in full force and effect (or that this Lease is in full force and effect as modified and setting forth the modifications), the date to which the Rent has been paid, that to the knowledge of the certifying party, no Default or an Event of Default has occurred and is continuing or, if a Default or an Event of Default shall exist, specifying in reasonable detail the nature thereof and the steps being taken to remedy the same, and such additional information as the requesting party may reasonably request. Any such certificate furnished pursuant to this Section 17.1 may be relied upon by the requesting party, its lender, and any prospective purchaser or mortgagee of the Leased Property or the leasehold estate conveyed hereby.

### 17.2 Financial Statements.

Tenant shall furnish the following statements to Landlord:

- (a) within forty (40) days after each of the first three (3) quarters of any Fiscal Year, the most recent unaudited CCC Consolidated Financial Statements accompanied by the Financial Officer's Certificate;
- (b) within eighty (80) days after the end of each Fiscal Year, the CCC Consolidated Financial Statements for such Fiscal Year, certified by an independent certified public accountant reasonably satisfactory to Landlord and accompanied by a Financial Officer's Certificate;
- (c) within [ ] days after the end of each Fiscal Year, the results of operations of the Facility for the preceding Fiscal Year (the "ANNUAL OPERATING STATEMENT"), in the form received by Tenant from Manager in accordance with the Management Agreement, accompanied by a Financial Officer's Certificate;
- (d) simultaneously with the payment of Rent under Section 3.1.1, the results of operations of the Facility for the preceding Accounting Period (the "ACCOUNTING PERIOD STATEMENT"), in the form received by Tenant from Manager in accordance with the Management Agreement;

- (e) simultaneously with the payment of Rent under Section 3.1.1, the Percentage Rent Schedule and, simultaneously with the payment of Rent with respect to the final Accounting Period of each Fiscal Year, the Year End Percentage Rent Schedule, accompanied by a Financial Officer's Certificate;
- (f) to the extent such information has been provided by Manager to Tenant, not later than twenty-eight (28) days after the end of each Accounting Period, except as described below, the summary of operating results of the Facility (the "PERIOD REPORT"). The Period Report will provide REVPAR and EBITDA substantially in the format of the period report attached hereto as Schedule 17.2(f). Notwithstanding the foregoing, for the first Accounting

Period of each Fiscal Year, the Period Report will be provided to Landlord not later than thirty-five (35) days after the end of the prior Fiscal Year;

- (g) to the extent such information has been provided by Manager to Tenant, not later than twenty-eight (28) days after each of the first three (3) quarters of any Fiscal Year, Tenant will provide the forecast Gross Revenues, Room Revenues and EBITDA for the Facility. In addition, to the extent such information is available from Manager, Tenant will provide forecast Gross Revenues by department by Accounting Period and for the Fiscal Year;
- (h) promptly after the sending or filing thereof, copies of all periodic reports which CCC, OpCo or Tenant files with the SEC or any stock exchange on which any of their shares are listed or traded;
- (i) promptly after the delivery thereof to Tenant, a copy of any management letter or written report prepared by the independent certified public accountants with respect to the financial condition, operations, business or prospects of Tenant, as the case may be; and
- (j) at the expense of Landlord, at any time and from time to time upon not less than forty-five (45) days Notice from Landlord, any financial reporting information required to be filed by Landlord with any securities or exchange commission, the SEC or any successor agency, or any other governmental authority, or required pursuant to any order issued by any court, governmental authority or arbitrator in any litigation to which Landlord is a party, for purposes of compliance therewith.

Landlord may at any time, and from time to time, provide the Facility Mortgagee with copies of any of the foregoing statements, provided that Landlord has used commercially reasonable efforts to cause the Facility Mortgagee to execute and deliver a confidentiality agreement reasonably satisfactory to Tenant. Upon reasonable Notice from Landlord, Tenant agrees to cooperate with Landlord to provide to Landlord the data, forecasts, and reports used in the preparation of the foregoing statements within a reasonable time frame after such data and reports become available to Tenant, provided, however, that Tenant makes no representation as to the accuracy of the data, forecasts and reports provided.

#### 17.3 Annual Budget.

Not later than twenty-eight (28) days prior to the commencement of each Fiscal Year except for the first Fiscal Year, Tenant shall prepare and submit to Landlord a schedule of Tenant's reasonable estimate of Gross Revenues by department (the "OPERATING BUDGET SUMMARY"). Not later than fifteen (15) days prior to the commencement of each Fiscal Year except for the first Fiscal Year, Tenant shall prepare and submit to Landlord an operating budget (the "OPERATING BUDGET") and a capital budget (the "CAPITAL BUDGET") prepared in accordance with the requirements of this Section 17.3. The Operating Budget and the Capital Budget (together, the "ANNUAL BUDGET") shall be consistent with the format provided by Manager and show the following for the year as a whole:

- (a) Tenant's reasonable estimate of Gross Revenues (including room rates and anticipated Room Revenues) for the forthcoming Fiscal Year, together with a summary of the estimated operating results provided by the Manager;
- (b) An estimate of any amounts Landlord will be requested to provide for Capital Expenditures during the Fiscal Year;
- (c) An estimate of the expenditures necessary for replacements and renewals of FF&E (the "FF&E ESTIMATE"); and
- (d) Tenant's reasonable estimate of Percentage Rent payable with respect to Room Revenues, Food and Beverages Sales, and Other Income.

Upon reasonable Notice from Landlord, Tenant agrees to cooperate with Landlord to provide to Landlord the data, forecasts, and reports used in the preparation of the Capital Budget and the Operating Budget within a reasonable time frame after such data and reports become available to Tenant, provided, however, that Tenant makes no representation as to the accuracy of the data, forecasts and reports provided. Landlord shall have twenty-five (25) days after the date on which it receives the Annual Budget to review, disapprove or change the entries and information appearing in the Annual Budget relating to the Capital Budget or the FF&E Estimate, but with regard to the FF&E Estimate, but only to the extent the FF&E Estimate indicates amounts in excess of the FF&E Reserve (the Capital Budget and the FF&E Estimate, to the extent it proposes expenditures in excess of the FF&E Reserve, are collectively referred to herein as the "CAPITAL PORTION"). If the parties are not able to reach agreement on the Capital Portion for any Fiscal Year during Landlord's twenty-five (25)-day review period, the parties shall attempt in good faith during the subsequent twenty-five (25)-day period to resolve any disputes, which attempt shall include, if requested by either party, at least one (1) meeting of executive level officers of Landlord and Tenant. In the event the parties are still not able to reach agreement on the Capital Portion of the Annual Budget for any particular Fiscal Year after complying with the foregoing requirements of this Section 17.3.2, no Capital Expenditures or FF&E expenditures in excess of the FF&E Reserve shall be made unless the same are set forth in a previously approved Capital Budget or are specifically approved by Landlord or are otherwise required to comply with Legal

Requirements or to make emergency expenditures in connection with Emergency Situations or otherwise required pursuant to the Management Agreement.

Tenant shall operate the Leased Property consistent with the Annual Budget and shall promptly report to Landlord in writing any actual or anticipated deviation from the Operating Budget or Capital Budget of any material or long-term consequence. To the extent the budget estimates for the categories of revenues identified in (d) above are not made available on an Accounting Period or Fiscal Year basis from Manager, Tenant shall cooperate with Landlord, at Landlord's sole cost, in its efforts to obtain such budget information from Manager.

Not later than seventy-five (75) days after the beginning of the Fiscal Year, Tenant shall provide Landlord with itemized schedules on an Accounting Period basis detailing the Operating Budget and the anticipated Percentage Rent (collectively, the "BUDGET SPREAD").

# ARTICLE 18 LANDLORD'S RIGHT TO INSPECT

Tenant shall permit Landlord and its authorized representatives to inspect the Leased Property during normal business hours upon not less than twenty-four (24) hours Notice, and to make such repairs as are required by Legal Requirements and Insurance Requirements which Tenant fails to make and as Landlord is permitted or required to make pursuant to the terms of this Lease, provided that any inspection or repair by Landlord or its representatives will not unreasonably interfere with Tenant's use and operation of the Leased Property and provided further that in the event of an emergency, as determined by Landlord in its reasonable discretion, prior Notice shall not be necessary.

#### ARTICLE 19 APPRAISAL

In the event that it becomes necessary to determine the Fair Market Value or Fair Market Rental for any purpose of this Lease, or the amount of any adjustment to or abatement of Minimum Rent and Percentage Rent hereunder, and the parties cannot agree thereon, such Fair Market Value or Fair Market Rental, or reduction, adjustment or abatement, as the case may be, shall be determined upon the written request of either party in accordance with the following procedure.

The party requesting an appraisal, by Notice given to the other, shall propose and unilaterally appoint a Qualified Appraiser. The other party, by Notice given within fifteen (15) days after receipt of such Notice appointing the first Qualified Appraiser, may appoint a second Qualified Appraiser. If the other party fails to appoint the second Qualified Appraiser within such fifteen (15)-day period, such party shall have waived its right to appoint a Qualified Appraiser, the first Qualified Appraiser shall appoint a second Qualified Appraiser within fifteen (15) days thereafter, and the Fair Market Value or Fair Market Rental, or reduction, adjustment or abatement, as the case may be, shall be determined by the Qualified Appraisers as set forth below.

The two Qualified Appraisers shall thereupon endeavor to agree upon the Fair Market Value or Fair Market Rental, or reduction, adjustment or abatement, as the case may be. If the two Qualified Appraisers so named cannot agree upon such value or rental, or reduction, adjustment or abatement, as the case may be, within thirty (30) days after the designation of the second such appraiser, each such appraiser shall, within five (5) days after the expiration of such thirty (30)-day period, submit his appraisal to the other appraiser in writing, and if the Fair Market Values or Fair Market Rentals, or amounts of reduction, adjustment or abatement, as the case may be, set forth in such appraisals vary by five percent (5%) or less of the greater value, the Fair Market Value or Fair Market Rental, or reduction, adjustment or abatement, as the case may be, shall be determined by calculating the average of the two determinations of the two appraisers.

If the Fair Market Values or Fair Market Rentals, or amounts of reduction, adjustment or abatement, as the case may be, set forth in the two appraisals vary by more than five percent (5%) of the greater value, the two Qualified Appraisers shall select a third Qualified Appraiser within an additional fifteen (15) days following the expiration of the aforesaid five (5)-day period. If the two appraisers are unable to agree upon the appointment of a third appraiser within such fifteen (15)-day period, either party may, upon written notice to the other, request that such appointment be made by the then President (or equivalent officer) of the State's Chapter of the American Institute of Real Estate Appraisers, or his or her designee, or, if there is no such organization or if such individual declines to make such appointment, by any state or Federal court of competent jurisdiction for the State.

In the event that all three of the appraisers cannot agree upon Fair Market Value or Fair Market Rental, or reduction, adjustment or abatement, as the case may be, within twenty (20) days following the selection of the third appraiser, each appraiser shall, within ten (10) days thereafter, submit his appraisal to the other two appraisers in writing, and the Fair Market Value or Fair Market Rental, or reduction, adjustment or abatement, as the case may be, shall be determined by calculating the average of the two numerically closest values (or, if the values are equidistant, the average of all three values) determined by the three appraisers.

In the event that any appraiser appointed hereunder does not or is unable to perform his or her obligation hereunder, then the party or the appraiser(s) appointing such appraiser shall have the right to propose and approve unilaterally a substitute Qualified Appraiser, but if the party or the appraiser(s) who have the right to appoint a substitute Qualified Appraiser fail to do so within ten (10) days after written Notice from the other party (or either party in the event such appraiser was appointed by the other appraisers), either party may, upon written Notice to the party having the right to appoint a substitute Qualified Appraiser, request that such appointment be made by such officer of the American Institute of Real Estate Appraisers or court of competent jurisdiction as described above; provided, however, that a party who has the right to appoint an appraiser or a substitute appraiser shall have the right to make such appointment only up until the time such appointment is made by such officer or court.

In connection with the appraisal process, so long as Tenant receives reasonable prior Notice, Tenant shall provide the appraisers full access during normal business hours

to examine the Leased Property, the books, records and files of Tenant and all agreements, leases and other operating agreements relating to the Leased Property.

The costs of each such appraisal shall be borne by the party selecting the appraiser, provided the cost of the third appraiser shall be split equally between Landlord and Tenant. Upon determining such Fair Market Value or Fair Market Rental, or reduction, adjustment or abatement, as the case may be, the appraisers shall promptly notify Landlord and Tenant in writing of such determination. If any party shall fail to appear at the hearings appointed by the appraisers, the appraisers may act in the absence of such party.

The determination of the Qualified Appraisers made in accordance with the foregoing provisions shall be final and binding upon the parties, such determination may be entered as an award in arbitration in a court of competent jurisdiction, and judgment thereon may be entered.

# ARTICLE 20 FACILITY MORTGAGES

#### 20.1 Landlord May Grant Liens.

- (a) Without the consent of Tenant, Landlord may, subject to the terms and conditions set forth in this Section 20.1, from time to time, directly or indirectly, create or otherwise cause to exist any Lien or ground lease upon its interest in the Leased Property, or any portion thereof or interest therein, whether to secure any borrowing or other means of financing or refinancing, provided that any such Lien or ground lease shall be consistent with the requirements of the Management Agreement or otherwise approved by Manager, and shall not modify the terms of this Lease, except as expressly set forth in Section 20.2. Landlord agrees to provide to Tenant copies of all existing and future ground leases, and amendments thereto, which affect the Leased Property.
- (b) Tenant shall, upon the request of Landlord or any existing, potential or future Facility Mortgagee, and to the extent in Tenant's possession or obtainable from Manager pursuant to the Management Agreement, (i) provide Landlord or the Facility Mortgagee with copies of all licenses, permits, occupancy agreements, operating agreements, leases, contracts, inspection reports, studies, appraisals, assessments, default or other notices and similar materials reasonably requested in connection with any existing or proposed financing of the Leased Property, and (ii) execute such estoppel certificates and collateral assignments with respect to the Facility's licenses and any of the other aforementioned agreements as the Facility Mortgagee may reasonably request in connection with any such financing, provided that no such estoppel certificate or collateral assignment shall, except as expressly set forth in Section 20.2, modify the terms of this Lease.

## 20.2 Subordination of Lease.

Subject to Section 20.1 and the terms of Section 7.1, this Lease, and any and all rights of Tenant hereunder, are and shall be subject and subordinate to any Facility

Mortgage, any ground or master lease, and all renewals, extensions, modifications, consolidations and replacements thereof, and to each and every advance made or hereafter to be made under any such Facility Mortgage. This section shall be self-operative and no further instrument of subordination shall be required. In confirmation of such subordination, Tenant shall promptly execute, acknowledge and deliver any instrument that Landlord, the landlord under any such lease or the holder of any such mortgage or the trustee or beneficiary of any deed of trust or any of their respective successors in interest may reasonably request to evidence such subordination. Tenant shall not unreasonably withhold its consent to any amendment to this Lease reasonably required by such lender or ground lessor, provided that such amendment does not (i) increase Tenant's rental obligations or other financial obligations hereunder, or (ii) have a material adverse effect upon Tenant's rights hereunder, or (iii) materially increase Tenant's non-economic obligations hereunder, or (iv) decrease Landlord's obligations hereunder. Landlord shall exercise commercially reasonable efforts to require any future Facility Mortgage or landlord under a ground lease affecting the Leased Property to provide Tenant with notice and an opportunity to cure Landlord defaults under the respective Facility Mortgage or ground lease.

Any lease to which this Lease is, at the time referred to, subject and subordinate is herein called a "SUPERIOR LEASE," and the landlord of a Superior Lease or its successor in interest at the time referred to is herein called "SUPERIOR LANDLORD"; the Facility Mortgage and any other mortgage or deed of trust to which this Lease is, at the time referred to, subject and subordinate, is herein called a "SUPERIOR MORTGAGE," and the Facility Mortgagee and any other holder, trustee or beneficiary of a Superior Mortgage is herein called "SUPERIOR MORTGAGEE." Tenant shall have no obligations under any Superior Lease or Superior Mortgage other than those expressly set forth in this Section 20.2.

Notwithstanding the obligations of Tenant hereunder, neither any Superior Mortgagee nor any Superior Landlord shall have an obligation to provide a non-disturbance agreement to Tenant. Any Superior Mortgagee or Superior Landlord shall have the right to terminate this Lease upon the foreclosure, deed in lieu of foreclosure or exercise of the power of sale with respect to the Leased Property; provided that, if such right is exercised because of (a) a non-monetary default by Landlord under the terms of the relevant loan agreement or ground lease not caused by an Event of Default hereunder or (b) a monetary default by Landlord (including a misapplication of Rent paid by Tenant) where Tenant is not in Default in the payment of Rent hereunder beyond the expiration of applicable notice and cure periods, then Landlord shall pay to Tenant the Fair Market Value of Tenant's leasehold estate as of the termination date in accordance with Section 24.1(b); provided further that (i) such fee shall be paid first by offsetting any amounts owed by Tenant to Landlord at such time and the balance (if any) shall be paid to Tenant in cash, and (ii) Tenant agrees to seek payment of such cash balance (if any) solely from Host O.P. (which, by its execution of this Lease, agrees to be primarily liable for Landlord's obligation under this subparagraph of Section 20.2) pursuant to the terms of the Facility Mortgagee Agreement, and shall not make any demand or claim therefor against Landlord, the Facility Mortgagee, any purchaser in foreclosure or transferee by deed in lieu of foreclosure or other party claiming under any of the foregoing.

In the event a cash flow sweep structure is implemented by any Superior Mortgagee, and such structure remains in place for twelve (12) consecutive months, Tenant, at its election, upon not less than thirty (30) days Notice, may terminate this Lease effective as of the end of such twelve (12) month period. For any period prior to such termination date, (i) Tenant's obligation to pay Rent or any other amounts payable hereunder shall be reduced by any amounts received by any Superior Mortgagee and (ii) Landlord shall compensate Tenant on an Accounting Period basis for any Tenant Operating Profit not received because of the cash flow sweep structure (i.e., any amount swept in excess of the Rent and other amounts otherwise payable by Tenant under this Lease) and any other costs incurred or advanced by Tenant pursuant to this Lease, and Host O.P. agrees to be primarily liable for Landlord's obligation under this clause (ii). Likewise, for any period during which cash management procedures are implemented by or on behalf of any Superior Mortgagee, (a) Tenant's obligation to pay Rent or any other amounts payable hereunder shall be reduced by any amounts received by any Superior Mortgagee and (b) Landlord shall compensate Tenant on an Accounting Period basis for any Tenant Operating Profit not received because of the cash management procedures (i.e., any amount swept in excess of the Rent and other amounts otherwise payable by Tenant under this Lease) and any other costs incurred or advanced by Tenant pursuant to this Lease, and Host O.P. agrees to be primarily liable for Landlord's obligation under this clause (b). If Tenant elects to terminate this Lease pursuant to the termination right granted in the first sentence of this paragraph, Landlord shall pay to Tenant the Fair Market Value of Tenant's leasehold estate as of the termination date calculated in accordance with Section 24.1(b); provided, however, (x) such fee shall be paid first by offsetting any amounts owed by Tenant to Landlord at such time and the balance (if any) shall be paid to Tenant in cash, and (y) Tenant agrees to seek payment of such cash balance (if any) solely from Host O.P. (which, by its execution of this Lease, agrees to be primarily liable for Landlord's obligation to pay the cash balance referenced in this clause (y)) pursuant to the terms of the Facility Mortgagee Agreement, and shall not make any demand or claim therefor against Landlord, the Facility Mortgagee, any purchaser in foreclosure or transferee by deed in lieu of foreclosure or other party claiming under any of the foregoing.

Subject to the termination rights of any Superior Landlord or Superior Mortgagee, if any, in the event that any Superior Landlord or Superior Mortgagee or the nominee or designee of any Superior Landlord or Superior Mortgagee shall succeed to the rights of Landlord under this Lease (any such person, LANDLORD"), whether through possession or foreclosure action or delivery of a new lease or deed, or otherwise, such Successor Landlord shall recognize Tenant's rights under this Lease as herein provided and Tenant shall attorn to and recognize the Successor Landlord as Tenant's landlord under this Lease and Tenant shall promptly execute and deliver any instrument that such Successor Landlord may reasonably request to evidence such attornment (provided that such instrument does not alter the terms of this Lease), whereupon, this Lease shall continue in full force and effect as a direct lease between the Successor Landlord and Tenant upon all of the terms, conditions and covenants as are set forth in this Lease, except that the Successor Landlord (unless formerly the landlord under this Lease or its nominee or designee) shall not be (a) liable in any way to Tenant for any act or omission, neglect or default on the party of any prior Landlord under this Lease, (b) responsible for any monies owing by or on deposit with any prior Landlord to the credit of

Tenant (except to the extent actually paid or delivered to the Successor Landlord), (c) subject to any counterclaim or setoff which theretofore accrued to Tenant against any prior Landlord, (d) bound by any modification of this Lease subsequent to such Superior Lease or Superior Mortgage, or by any previous prepayment of Minimum Rent or Additional Rent for more than one (1) month in advance of the date due hereunder, which was not approved in writing by the Superior Landlord or the Superior Mortgagee, (e) liable to Tenant beyond the Successor Landlord's interest in the Leased Property and the rents, income, receipts, revenues, issues and profits issuing from the Leased Property, (f) responsible for the performance of any work to be done by Landlord under this Lease to render the Leased Property ready for occupancy by Tenant (subject to Landlord's obligations under Section 5.1.2(b) or with respect to any insurance or Condemnation proceeds), or (g) required to remove any Person occupying the Leased Property or any part thereof, except if such Person claims by, through or under the Successor Landlord. Tenant agrees at any time and from time to time to execute a suitable instrument in confirmation of Tenant's agreement to attorn, as aforesaid.

20.3 Notice to Mortgagee and Ground Landlord.

Attached hereto as Schedule 20.3 is a list of the Superior

Mortgagee(s) and Superior Landlord(s) as of the date hereof, including their addresses for notices. No default Notice from Tenant to Landlord as to the Leased Property shall be effective unless and until a copy of the same is given to such Superior Mortgagee(s) and Superior Landlord(s), and subsequent to the receipt by Tenant of Notice from Landlord as to the identity of any future Facility Mortgagee or ground landlord under a lease with Landlord, as ground tenant, which includes the Leased Property as part of the demised premises and which complies with Sections 20.1 and 20.2 (which Notice shall include a copy of the applicable mortgage or lease), no default Notice from Tenant to Landlord hereunder shall be effective unless and until a copy of the same is given to the Facility Mortgagee or ground landlord at the address set forth in such Notice. The curing of any Landlord Default by any Superior Mortgagee(s) or Superior Landlord(s) listed on Schedule 20.3, or by any Facility Mortgagee or ground

landlord of which Tenant receives Notice after the date hereof as provided above, shall be treated as performance by Landlord, provided any such cure shall be made within the time periods set forth herein.

20.4 Transfer of Leased Property.

Notwithstanding anything set forth herein to the contrary, but subject to the rights of a Facility Mortgagee as set forth herein, Landlord shall not, without the consent of Tenant, transfer the Leased Property, or any interest therein, to any Person (i) which does not have sufficient financial resources and liquidity to fulfill the Landlord's Retained Obligations and Continuing Obligations under the Management Agreement and Landlord's obligations under this Lease, or (ii) which has been, or is in control of or controlled by Persons who have been, convicted of felonies involving moral turpitude in any state or federal court, or otherwise would cause a breach of the Management Agreement.

ARTICLE 21
ADDITIONAL COVENANTS OF TENANT

#### 21.1 Conduct of Business.

Tenant shall do or cause to be done all things necessary to preserve, renew and keep in full force and effect and in good standing its existence and its rights and licenses necessary to conduct such business.

#### 21.2 Maintenance of Accounts and Records.

Tenant shall keep records and books of account in which full, true and correct entries in all material respects will be made of dealings and transactions in relation to the business and affairs of Tenant.

#### 21.3 Management of Leased Property.

# 21.3.1 Management Agreement, Consent and Assignment, Etc.

For and during the Term of this Lease, but not thereafter, Landlord has assigned unto Tenant Landlord's interest as "Owner" under that \_, \_\_\_\_, by and between certain Management Agreement dated Landlord and (the "MANAGER"), as amended prior to the date hereof and subsequently modified and amended pursuant to that certain [Consent, Assignment and Assumption and Amendment of Management Agreement] dated as of even date herewith by and among Landlord, Tenant and Manager (the "CONSENT AND ASSIGNMENT") (together with all past and future modifications and amendments thereto, the "MANAGEMENT AGREEMENT"). Pursuant to the Consent and Assignment, Tenant has accepted such assignment and assumed and agreed to perform all of Landlord's obligations as "Owner" under the Management Agreement (including certain obligations of "Owner" that the Consent and Assignment expressly provides shall both apply to and be binding upon Tenant, and continue to apply to and be binding upon Landlord (as, but only to the extent, such obligations apply to Landlord, the "CONTINUING OBLIGATIONS")), except for certain obligations of "Owner" that the Consent and Assignment expressly provides shall remain the sole obligations of Landlord (the "RETAINED OBLIGATIONS"). Landlord hereby covenants to perform the Retained Obligations and the Continuing Obligations. In addition, Landlord shall be entitled, together with Tenant, to exercise the "Continuing Rights," as defined in the Consent and Assignment, and Tenant shall not be entitled to exercise any of the "Reserved Rights," as defined in the Consent and Assignment, which are reserved exclusively to Landlord.

#### 21.3.2 Reversion upon Termination.

All of Landlord's rights, benefits and privileges with respect to the Management Agreement shall be vested in Tenant throughout the Term of this Lease; provided, however, that upon termination of this Lease, for whatever reason, all of Landlord's rights, benefits and privileges under the Management Agreement shall automatically revert to Landlord without the necessity of any action on the part of Landlord hereunder.

#### 21.3.3 Compliance with Management Agreement and Indemnification.

To the extent that any of the provisions of the Management Agreement impose a greater obligation on Tenant than the corresponding provisions of this Lease, then Tenant shall be obligated to comply with, and to take all reasonable actions necessary to prevent breaches or defaults under, the provisions of the Management Agreement. Notwithstanding anything contained herein to the contrary, Tenant shall perform and comply in every respect with the provisions of the Management Agreement, except for the Retained Obligations and Continuing Obligations, which shall remain the sole responsibility of Landlord, so as to avoid any default thereunder during the Term of this Lease. Tenant shall, at all times, direct and require Manager to perform all of Manager's obligations under the Management Agreement. Tenant shall protect, indemnify and hold harmless Landlord for, from and against all liabilities, obligations, claims, damages, penalties, causes of action, costs and reasonable expenses (including, without limitation, reasonable attorneys' fees), to the maximum permitted by law, imposed upon, incurred by, or asserted against Landlord by reason of a default by Tenant under the Management Agreement, including, without limitation, any default by Tenant under the Management Agreement attributable to a failure by Tenant to perform its obligations under this Lease. Likewise, Landlord shall protect, indemnify and hold harmless Tenant for, from and against all liabilities, obligations, claims, damages, penalties, causes of action, costs and reasonable expenses (including, without limitation, reasonable attorneys' fees), to the maximum permitted by law, imposed upon, incurred by, or asserted against Tenant by reason of a default by Landlord under the Management Agreement, including, without limitation, any default by Landlord under the Management Agreement attributable to a failure by Landlord to perform its obligations under this Lease. The obligations of Landlord and Tenant set forth in this Section 21.3.3 shall survive the termination of this Lease.

#### 21.3.4 Consent Required for Certain Actions.

Without at least thirty (30) days prior written Notice to Landlord in the case of subsections (1), (2), (5), and (6) below, and without at least fifteen (15) days prior written Notice in the case of subsections (3) and (4) below, Tenant shall not take any of the following actions:

- $\hspace{1cm}$  (1) terminate the Management Agreement prior to the expiration of the term thereof;
- (2) amend, modify or assign its interest in (except in connection with an assignment permitted pursuant to Section 16.1 hereof) the Management Agreement;
- (3) waive (or fail to enforce) any right of "Owner" under the Management Agreement;
- (4) waive any breach or default by Manager under the Management Agreement (or fail to enforce any right of "Owner" in connection therewith);

- $\mbox{(5)}$  agree to any change in Manager or consent to any assignment by Manager; or
- (6) take any other action which reasonably could be expected to materially adversely affect Landlord's rights or obligations under the Management Agreement for periods following termination of this Lease (whether upon the expiration of its term or upon earlier termination as provided for herein).

Notwithstanding the foregoing, Tenant shall not take any of the actions listed in clauses (1) through (6) above without Landlord's prior written consent if such action:

- (A) would materially impair the ability of Tenant to perform Tenant's obligations under this Lease (including, without limitation, make all payments of Rent as and when due under this Lease) (determined taking into account the guarantee attached hereto as Exhibit F (the "GUARANTEE") of CCC and OpCo (together, the "GUARANTORS") of Tenant's payment and performance of all of its obligations under this Lease);
- (B) would cause Tenant not to comply with the obligations of Tenant set forth in Section 4.1;
- (C) would materially adversely affect the economic value of the Leased Property to Landlord following the termination of this Lease (whether upon the expiration of the Term or upon earlier termination as provided for herein); or
- (D) would materially increase the legal exposure of Landlord to Manager under the Management Agreement during the Term, either by reason of Landlord's continuing liability to Manager pursuant to the Consent and Assignment (determined taking into account the Guarantee), or with respect to the Retained Obligations or the Continuing Obligations.

# 21.3.5 Replacement of Manager.

Notwithstanding Section 21.3.4 above, Tenant shall not agree to change the management or the brand affiliation of the Facility without the prior approval of Landlord, which approval shall not be unreasonably withheld, conditioned or delayed. In addition, the proposed replacement manager shall be nationally recognized and shall have substantial experience managing hotels of comparable quality. No such replacement or change in brand affiliation shall continue beyond the Term hereof without the prior written approval of Landlord, which approval may be granted or withheld in Landlord's sole discretion.

# 21.4 Facility Mortgagee Agreement.

Tenant agrees to enter into an agreement with Landlord, any Facility Mortgagee and Host O.P. (the "FACILITY MORTGAGEE AGREEMENT") pursuant to which, in consideration of Facility Mortgagee's consent to this Lease and the transactions contemplated hereby: (i) Tenant acknowledges to the Facility Mortgagee that the Working

Capital purchased by Tenant with the Working Capital Note and all Excess FF&E acquired by Tenant from Landlord while the Facility Mortgage is in effect is subject to the Facility Mortgagee's first priority Lien (without Tenant assuming any liability for Landlord's obligations that are secured by or arise under any Facility Mortgage) and that the Facility Mortgagee's rights and remedies with respect to such Working Capital and Excess FF&E shall survive and be enforceable with respect thereto; (ii) Tenant covenants to the Facility Mortgagee to sign, provided Landlord shall prepare and file at its sole cost, UCC-1 financing statements confirming the foregoing for notice purposes; (iii) Tenant covenants not to sell, lease, transfer or otherwise dispose of such Working Capital or Excess FF&E or any interest therein (other than in the ordinary course of business or other than to Landlord in accordance with the terms of this Lease upon the expiration or earlier termination of the Term), or grant or cause or permit to exist any lien, charge or encumbrance with respect thereto, other than the aforesaid Lien in favor of the Facility Mortgagee and any other Liens securing obligations which are the responsibility of Landlord; (iv) Landlord agrees that this Lease and the Working Capital Note will not be modified or amended in any manner that would adversely affect the Facility Mortgagee's collateral under the Facility Mortgage without the Facility Mortgagee's prior written consent; (v) Landlord and Tenant agree to confirm to the Facility Mortgagee that the Facility Mortgagee shall have the right to terminate this Lease in the event of a foreclosure by the Facility Mortgagee under the Facility Mortgage, pursuant to Article 20 hereof; (vi) Tenant agrees that, in the event of a foreclosure by the Facility Mortgagee under the Facility Mortgage, the Facility Mortgagee shall have the right to continue this Lease as lessor and succeed to all Landlord's rights hereunder and under the Security Agreement, provided that in such event the Facility Mortgagee recognizes Tenant's rights hereunder; (vii) if the Facility Mortgagee terminates this Lease in accordance with Article [20] [24] hereof, (A) Tenant agrees, at the Facility Mortgagee's request, to convey all Working Capital existing at such time directly to the Facility Mortgagee or its designee in payment of the Working Capital Note (together with any cash balance due from Tenant in respect thereto), (B) Tenant agrees to look solely to Host O.P. for payment of any cash balance payable to Tenant for Working Capital to the extent such Working Capital exceeds the balance due under the Working Capital Note, and (C) Host O.P. agrees to pay such cash balance directly to Tenant, without the necessity of notice or demand on Landlord; and (viii) Tenant agrees that if the amount specified in Section 24.1(b) of this Lease and/or the Excess FF&E Reimbursement Amount become payable in connection with a foreclosure by the Facility Mortgagee under the Facility Mortgage, then (A) such amount(s) shall be paid first by offsetting any amounts owed by Tenant to Landlord, and the balance (if any) shall be paid by Host O.P. to Tenant in cash, (B) Host O.P. agrees to pay such cash balance (if any) directly to Tenant without the need for notice or demand on Landlord, and (C) Tenant agrees to look only to Host O.P. for payment of such cash balance.

- 21.5 [Intentionally Omitted.]
- 21.6 Single Purpose Entity Covenants.
  - 21.6.1 Separate Existence

Tenant shall (i) maintain its books and records and bank accounts separate from any other person or entity (except that, for accounting and reporting

purposes, Tenant may be included in the CCC Consolidated Financial Statements in accordance with GAAP); (ii) maintain an arm's length relationship with its members, Affiliates and any other party furnishing services to it; (iii) maintain its books, records, resolutions and agreements as official records; (iv) conduct its business in its own name and through its own authorized officers and agents; (v) maintain its financial statements, accounting records and other limited liability company documents separate from those of any other Person (except for inclusion in the CCC Consolidated Financial Statements); (vi) pay its own liabilities out of its own funds and other assets, including funds contributed to its capital by its equity holders, and all such capital contributions shall be reflected properly in its books and records; (vii) observe all limited liability company formalities, as applicable, necessary to maintain its identity as an entity separate and distinct from its members, CCC, and all other Affiliates; (viii) participate in the fair and reasonable allocation of any and all overhead expenses and other common expenses for facilities, goods or services provided to multiple entities; (ix) use its own stationery, invoices and checks (except when acting in a representative capacity); (x) hold and identify itself as a separate and distinct entity under its own name and not as a division or part of any other Person (except for inclusion in the CCC Consolidated Financial Statements); and (xi) hold its assets in its own name.

#### 21.6.2 Independent Member

Upon Notice from Landlord, Tenant shall have an Independent Member (who shall be a non-equity member) at all times, or if the Independent Member has withdrawn, Tenant shall not take any action which may not be taken pursuant to the organizational documents of Tenant without the consent of the Independent Member until such time as a replacement Independent Member has been admitted to Tenant.

#### 21.6.3 Limitation on Indebtedness and Guarantees

Except pursuant to the Consent and Assignment or as otherwise expressly provided herein, Tenant shall not (i) incur, create or assume any Indebtedness of any kind; or (ii) guarantee or have any consensual contingent obligation for the obligations of any other Person; provided that, so long as no Event of Default has occurred and is continuing, Tenant may incur, create or assume any Permitted Debt (as defined below). As used herein, "PERMITTED DEBT" shall mean:

- (i) if no Default or Event of Default has occurred and is continuing, purchase money Indebtedness and capitalized lease obligations for the purchase or lease of FF&E in the ordinary course of business (and not inconsistent with customary industry practices), which Indebtedness may be secured by a first priority lien on the goods and equipment that have been so purchased or leased;
- (ii) if no Event of Default has occurred and is continuing, unsecured Indebtedness owing to CCC or any wholly owned Subsidiary thereof, with respect to which the lender shall have agreed in writing, in form and substance satisfactory to Landlord, that payment of such Indebtedness shall be subordinated in all respects to performance of Tenant's

obligations under this Lease and that no remedies may be exercised with respect to enforcement or collection of such Indebtedness until such time as this Lease shall have terminated and all obligations owed by Tenant hereunder shall have been discharged in full; and

(iii) if no Event of Default has occurred and is continuing, Indebtedness solely in respect of surety and appeal bonds, performance bonds and other obligations of a like nature (to the extent that such incurrence does not result in the incurrence of any obligation to repay any obligation relating to borrowed money of others), all in the ordinary course of business in accordance with customary industry practices.

# 21.6.4 Distributions

Tenant shall make no distributions of cash or other assets to any of its members if an Event of Default has occurred and is continuing under Section 12.1(a) hereof.

#### 21.6.5 Single Purpose

Tenant shall not have or create any Subsidiaries or hold any equity interest in any other Person. Tenant shall at all times be a Single Purpose entity.

Tenant shall not (i) engage in any business activity or operate for any purpose other than as stated in its Limited Liability Company Operating Agreement as in effect on the date hereof; (ii) without the consent of all its members, including the consent of an Independent Member, file a bankruptcy or insolvency petition or otherwise institute bankruptcy proceedings; or (iii) acquire any assets not reasonably related to the business and operation of the Facility.

# 21.6.6 Certain Fundamental Changes

Without the consent of Landlord and any Facility Mortgagee (if required), Tenant shall not (i) be a party to any merger or consolidation with any Person, or (ii) assign its rights under this Lease, or assign, transfer, or sell all or any substantial portion of its assets to any Person, in each case other than a Qualified Affiliate that is a Single Purpose entity, has no outstanding Indebtedness (other than Permitted Debt) and no Liens on any of its assets (other than Permitted Liens) at the time of such assignment, and assumes all of the obligations of Tenant hereunder. Without the consent of Landlord and any Facility Mortgagee, Tenant shall not adopt a plan of dissolution or liquidation or dissolve, wind up or liquidate. Without the approval of the Independent Member, Tenant shall not take any action for which the approval of the Independent Member is required under its organizational documents.

# 21.6.7 Amendments to Organizational Documents

Tenant shall not, in any manner, without the consent of Landlord and any Facility Mortgagee (if required), amend, modify or alter the terms of Sections 2.2, 2.3,

2.4, 5.4 (only paragraphs a, b, c, d and g thereof), 5.5, 5.6, 5.7, and 8.1 of the Limited Liability Company Operating Agreement of Tenant, as in effect on the date hereof.

# 21.6.8 Qualified Affiliate

#### ARTICLE 22 LIMITATIONS

# 22.1 REIT Compliance.

Tenant acknowledges that Host REIT intends to qualify as a real estate investment trust under the Code. Tenant agrees that it will not knowingly or intentionally take or omit to take any action, or permit any status or condition to exist at the Leased Property, which Tenant actually knows (acting in good faith) would or could result in (i) the Rent payable under this Lease not qualifying as "rents from real property" as defined in Section 856(d) of the Code or (ii) Host REIT being disqualified from treatment as a real estate investment trust under the Code as the provisions exist on the date hereof; provided, however, that notwithstanding anything herein to the contrary, (i) Tenant shall not be responsible for any act or omission of Landlord or Manager (unless Manager's action was with the express written consent or at the direction of Tenant), and (ii) any action by Tenant taken in compliance with the express terms of this Lease, the Consent and Assignment, or the Management Agreement shall not be deemed to create a Default or Event of Default under this Section 22.1.

#### 22.2 FF&E Limitation.

This Section 22.2 is intended to insure that all of the rent payable under this Lease qualifies as "rents from real property" within the meaning of Section 856(d) of the Code or any similar or successor provisions thereto. In furtherance of such purpose, the parties have agreed to the terms set forth in Schedule 22.2 attached hereto.

#### 22.3 Sublease Rent Limitation.

Anything contained in this Lease to the contrary notwithstanding, from and after the Commencement Date, Tenant shall not knowingly or intentionally (acting in good faith) enter into any sublease with respect to the Leased Property or any part thereof on any basis such that the rental to be paid by the sublessee thereunder would be based (or considered to be based), in whole or in part, on either (a) the income or profits derived by the business activities of the sublessee, or (b) any other formula such that any portion of the rent payable hereunder would or could, to Tenant's actual knowledge (acting in good faith), fail to qualify as "rents from real property" within the meaning of Section 856(d) of the Code, or any similar or successor provisions thereto.

#### 22.4 Sublease Tenant Limitation.

Anything contained in this Lease to the contrary notwithstanding, Tenant shall not knowingly or intentionally (acting in good faith) sublease any Leased Property or any part thereof to any Person or Entity in which Landlord, Host O.P., or Host REIT owns, directly or indirectly, a ten percent (10%) or greater interest, within the meaning of Section 856(d)(2)(B) of the Code, or any similar or successor provisions thereto. Tenant shall take reasonable precautions in connection with each sublease (including providing Landlord with prompt Notice of the same) to ensure that such sublease will not result in a violation of this Section 22.4.

#### 22.5 Tenant Ownership Limitation.

Anything contained in this Lease to the contrary notwithstanding, Tenant shall not knowingly or intentionally (acting in good faith), and shall use commercially reasonable efforts to cause its Affiliates not to knowingly or intentionally (acting in good faith), acquire, directly or indirectly, (a) a nine and 80/100 percent (9.8%) or greater interest in Landlord or Host REIT, or (b) a four and 90/100 percent (4.9%) or greater interest in Host 0.P., within the meaning of Section 856(d)(2)(B) of the Code, or any similar or successor provisions thereto.

# ARTICLE 23

### 23.1 No Waiver.

No failure by either Landlord or Tenant to insist upon the strict performance of any term hereof or to exercise any right, power or remedy consequent upon a breach thereof, and no acceptance of full or partial payment of Rent during the continuance of any such breach, shall constitute a waiver of any such breach or of any such term. To the maximum extent permitted by law, no waiver of any breach shall affect or alter this Lease, which shall continue in full force and effect with respect to any other then existing or subsequent breach.

#### 23.2 Remedies Cumulative.

To the maximum extent permitted by law, each legal, equitable or contractual right, power and remedy of either party now or hereafter provided either in this Lease or by statute, or otherwise, shall be cumulative and concurrent and shall be in addition to every other right, power and remedy, and the exercise or beginning of the exercise by either party of any one or more of such rights, powers and remedies shall not preclude the simultaneous or subsequent exercise by such party of any or all of such other rights, powers and remedies.

# 23.3 Severability.

Any clause, sentence, paragraph, section or provision of this Lease held by a court of competent jurisdiction to be invalid, illegal or ineffective shall not impair, invalidate or nullify the remainder of this Lease, but rather the effect thereof shall be

confined to the clause, sentence, paragraph, section or provision so held to be invalid, illegal or ineffective, and this Lease shall be construed as if such invalid, illegal or ineffective provisions had never been contained herein.

# 23.4 Acceptance of Surrender.

No surrender to Landlord of this Lease or of the Leased Property or any part thereof, or of any interest therein, shall be valid or effective unless agreed to and accepted in writing by Landlord, and no act by Landlord or any representative or agent of Landlord, other than such a written acceptance by Landlord, shall constitute an acceptance of any such surrender.

# 23.5 No Merger of Title.

It is expressly acknowledged and agreed that it is the intent of the parties that there shall be no merger of this Lease or of the leasehold estate created hereby by reason of the fact that the same Person may acquire, own or hold, directly or indirectly, this Lease or the leasehold estate created hereby and the fee estate or ground landlord's interest in the Leased Property.

# 23.6 Release of Landlord Following Conveyance.

If Landlord or any successor owner of all or any portion of the Leased Property shall convey all or any portion of the Leased Property in accordance with the terms hereof other than as security for a debt, and the grantee or transferee of the Leased Property shall expressly assume all obligations of Landlord hereunder arising or accruing from and after the date of such conveyance or transfer, Landlord or such successor owner, as the case may be, shall thereupon be released from all future liabilities and obligations of Landlord under this Lease with respect to such of the Leased Property arising or accruing from and after the date of such conveyance or other transfer, all such future liabilities and obligations shall thereupon be binding upon the new owner, and all references herein to Landlord thereafter shall be deemed to refer to the new owner.

# 23.7 Quiet Enjoyment.

Provided that no Event of Default shall have occurred and be continuing, Tenant shall peaceably and quietly have, hold and enjoy the Leased Property for the Term, free of hindrance or molestation by Landlord or anyone claiming by, through or under Landlord, but subject to (a) all Permitted Liens, (b) Liens as to obligations of Landlord that are either not yet due or which are being contested in good faith and by proper proceedings, provided the same do not materially interfere with Tenant's ability to operate the Facility, (c) Liens that have been consented to in writing by Tenant, and (d) Landlord's option to terminate this Lease pursuant to Article 24. Except as otherwise provided in this Lease, no failure by Landlord to comply with the foregoing covenant shall give Tenant any right to cancel or terminate this Lease or abate, reduce or make a deduction from or offset against the Rent or any other sum payable under this Lease, or to fail to perform any other obligation of Tenant hereunder.

#### 23.8 Landlord's Consent.

Where provision is made in this Lease for Landlord's consent and Landlord shall fail or refuse to give such consent, except to the extent expressly provided herein to the contrary Tenant shall not be entitled to any damages for any withholding by Landlord of its consent, it being intended that Tenant's sole remedy shall be an action for specific performance or injunction and that such remedy shall be available only in those cases where Landlord has expressly agreed in writing not unreasonably to withhold its consent. Whenever in this Lease the consent or approval of Landlord or Tenant is required, such consent or approval shall (except to the extent that such consent or approval is specifically designated as being "within the discretion" of a party, or words to that effect, in the applicable provision) not be unreasonably withheld, conditioned or delayed, shall be in writing and shall be executed by a duly authorized officer or agent of the party granting such consent or approval; provided, however, that Landlord shall be deemed to have reasonably withheld its consent in the event any Facility Mortgagee withholds its consent or otherwise objects to any proposed consent or approval. With respect only to the matters set forth on Schedule 23.8, if either Tenant or Landlord fails to respond within

fifteen (15) days (or such shorter or longer period of time as may be expressly specified in this Lease) to a request in the form of a Notice by the other party for a consent or approval, such consent or approval shall be deemed to have been given.

#### 23.9 Memorandum of Lease.

Unless required by Legal Requirements, neither Landlord nor Tenant shall record this Lease. However, Landlord and Tenant shall promptly, upon the request of the other, enter into a short form memorandum of this Lease, in form suitable for recording under the laws of the State in which reference to this Lease, and all options contained herein, shall be made. The requesting party shall bear the costs and expenses of recording such memorandum. If a memorandum of this Lease is required by Legal Requirements to be recorded, the parties shall share equally the costs and expenses of recording such memorandum.

#### 23.10 Notices.

- (a) Any and all notices, demands, consents, approvals, offers, elections and other communications required or permitted under this Lease shall be deemed adequately given if in writing and the same shall be delivered either in hand, by telecopier with computer generated acknowledgment of receipt, or by mail or Federal Express or similar expedited commercial carrier, addressed to the recipient of the notice, postpaid and certified with return receipt requested (if by mail), or with all freight charges prepaid (if by Federal Express or similar carrier).
- (b) All notices required or permitted to be sent hereunder shall be deemed to have been given for all purposes of this Lease upon the date of acknowledged receipt, in the case of a notice by telecopier, and, in all other cases, upon the date of receipt or refusal, except that whenever under this Lease a notice is either received on a day which is not a Business Day or is required to be delivered on or before a specific day which is not

a Business Day, the day of receipt or required delivery shall automatically be extended to the next Business Day.

(c) All such notices shall be addressed:

if to Landlord to:

a/a Hast Marriatt Corporation

c/o Host Marriott Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attn: Chris Nassetta

with a copy (which shall not constitute notice) to:

Host Marriott Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attn: General Counsel

If to Tenant to:

-----

c/o Crestline Capital Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attn: Bruce Stemerman

with a copy (which shall not constitute notice) to:

Crestline Capital Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attn: General Counsel

(d) By notice given as herein provided, the parties hereto and their respective successors and assigns shall have the right from time to time and at any time during the Term of this Lease to change their respective addresses effective upon receipt by the other parties of such notice and each shall have the right to specify as its address any other address within the United States of America.

# 23.11 Construction.

Anything contained in this Lease to the contrary notwithstanding, all claims against, and liabilities of, Tenant or Landlord arising prior to any date of termination or expiration of this Lease shall survive such termination or expiration. In no event shall either party be liable for any punitive or consequential damages as the result of a breach of this Lease by such other party. Neither this Lease nor any provision hereof may be changed, waived, discharged or terminated except by an instrument in writing signed by

the party to be charged. All the terms and provisions of this Lease shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns. Each term or provision of this Lease to be performed by Tenant or Landlord shall be construed as an independent covenant and condition. Time is of the essence with respect to the exercise of any rights of, and performance of any obligations by, Tenant or Landlord under this Lease. Except as otherwise set forth in this Lease, any obligations of Tenant and Landlord (including, without limitation, any monetary, repair and indemnification obligations) shall survive the expiration or sooner termination of this Lease.

# 23.12 Counterparts; Headings.

This Lease may be executed in two or more counterparts, each of which shall constitute an original, but which, when taken together, shall constitute but one instrument and shall become effective as of the date hereof when copies hereof, which, when taken together, bear the signatures of each of the parties hereto shall have been signed. Headings in this Lease are for purposes of reference only and shall not limit or affect the meaning of the provisions hereof.

# 23.13 Governing Law; Jurisdiction.

- (a) This Lease shall be interpreted, construed, applied and enforced in accordance with the laws of the State applicable to contracts between residents of the State, which are to be performed entirely within the State, and the laws of the State shall apply to the perfection and priority of liens upon and the disposition of and disposition with respect to the Leased Property and in any case regardless of (i) where this Lease is executed or delivered; or (ii) where any payment or other performance required by this Lease is made or required to be made; or (iii) where any breach of any provision of this Lease occurs, or any cause of action otherwise accrues; or (iv) where any action or other proceeding is instituted or pending; or (v) the nationality, citizenship, domicile, principal place of business, or jurisdiction of organization or domestication of any party; or (vi) whether the laws of the forum jurisdiction otherwise would apply the laws of a jurisdiction other than the State; or (vii) any combination of the foregoing.
- (b) To the maximum extent permitted by applicable law, any action to enforce, arising out of, or relating in any way to, any of the provisions of this Lease may be brought and prosecuted only in the court or courts located in the State of Maryland; and the parties consent to the jurisdiction of said court or courts located in the State and to service of process by certified mail, return receipt requested, or by any other manner provided by law.

# 23.14 No Broker.

Each party hereby represents and warrants to the other that it has not engaged, dealt with or otherwise discussed this transaction with any broker, agent or finder. Each party agrees to indemnify and hold the other harmless from and against any claim arising out of a breach of the foregoing agreement and representation and warranty.

#### 23.15 Related Agreements.

- [(a)] Anything herein to the contrary notwithstanding, neither Landlord nor Tenant shall take any action, or fail to take any action, which would constitute a breach or default under (i) that certain Noncompetition Agreement dated as of even date herewith by and between CCC and Host O.P., as the same may be subsequently amended or restated (the "NONCOMPETITION AGREEMENT"), (ii) that certain Pooling and Security Agreement dated as of even date herewith by and between CCC, OpCo, the Tenants and Landlords named therein and Host O.P., as the same may be subsequently amended or restated (the "POOLING AGREEMENT"), [(iii) that certain Facility Mortgagee Agreement dated as of even date herewith by and between Tenant Landlord, and [Landlord's lender], as the same may be amended or restated,] (iv) the Security Agreement, (v) the Consent and Assignment, or (vi) the Guarantee (collectively, the Noncompetition Agreement, the Pooling Agreement, the Facility Mortgagee Agreement, the Security Agreement, the Consent and Assignment and the Guarantee are sometimes referred to herein as the "RELATED AGREEMENTS."
- [(b) Anything herein to the contrary notwithstanding, Tenant shall not take any action or fail to take any action authorized or required to be taken hereunder that would constitute a breach or default of the Landlord under that certain ground lease dated as of \_\_\_\_\_\_\_ by and between \_\_\_\_\_ and \_\_\_\_\_ as the same may have been or may subsequently be amended or restated.]

# 23.16 Legal Fees and Costs of Litigation.

In the event either party to this Lease commences legal action of any kind to enforce the terms and conditions of this Lease, the prevailing party in such litigation will be entitled to collect from the other party all reasonable costs, expenses and attorneys' fees incurred in connection with such action.

# 23.17 Force Majeure.

If Landlord or Tenant is in any way delayed or prevented from performing any obligation, except any monetary obligation, hereunder due to acts of God, acts of war, civil disturbance, action of any Governmental Agency (including the revocation or refusal to grant licenses or permits, where such revocation or refusal is not due to the fault of the party whose performance is to be excused for reasons of force majeure), strikes, fire or other Casualty, or any other cause beyond the reasonable control of either party (as applicable), then the time for performance of such obligation shall be excused for the period of such delay or prevention and extended for a period equal to the period of such delay, interruption or prevention.

# 23.18 Conflicts with Related Agreements.

In the event of any conflict or inconsistency between this Lease and any of the Related Agreements or the Asset Management Agreement, the terms of any of the Related Agreements or the Asset Management Agreement shall govern. Notwithstanding the foregoing, in the event of any conflict or inconsistency between this Lease and the Consent and Assignment, as they relate to the respective rights and obligations of Landlord and Tenant, the terms of this Lease shall govern.

# 23.19 Operating Lease.

The parties hereto intend that this Lease shall be deemed for all purposes to be an operating lease and not a capital lease.

# ARTICLE 24 TERMINATION RIGHTS

24.1 Landlord's Right to Terminate Lease upon Sale or Tax Law Change.

(a) In the event (i) Landlord enters into a bona fide contract to sell the Leased Property to a non-Affiliate, or (ii) of a Tax Law Change resulting in Landlord's determination to terminate this Lease, then in either such event Landlord may terminate this Lease by giving not less than sixty (60) days prior Notice to Tenant of Landlord's election to terminate this Lease upon the closing under such contract or upon a date specified by Landlord which is on or after the effective date of the Tax Law Change. Landlord shall be entitled to terminate this Lease by reason of a Tax Law Change only if all other Pool Leases and Other Leases are concurrently terminated, other than any other Pool Lease or Other Lease that relates to a property that Host REIT, Host O.P. or any wholly owned Subsidiary thereof cannot own, following such Tax Law Change, without incurring potential adverse effects for Host REIT, Host O.P. or such Subsidiary, unless such property is subject to a lease. Effective upon such date, this Lease shall terminate and be of no further force and effect except as to any obligations of the parties existing as of such date that survive termination of this Lease, and all Rent, including Percentage Rent and Additional Charges, shall be adjusted as of the termination date. (The term "TAX LAW CHANGE" shall mean any change in the Code (including, without limitation, a change in the Treasury regulations promulgated thereunder), or in the judicial or administrative interpretations of the Code, which in Landlord's determination will permit Landlord, Host REIT, Host O.P. or another entity in which Host REIT or Host O.P. owns substantially all of the economic interests to operate the Facility as a hotel without adversely affecting Host REIT's qualification for taxation as a real estate investment trust under applicable code provisions). Landlord, in the event it enters into a bona fide contract to sell the Leased Property to a non-Affiliate, and subject to the restrictions set forth in Section 20.4, shall be permitted to transfer the Leased Property subject to the Lease, provided, however, that the parties shall make such reasonable modifications, if any, hereto as shall be necessary or appropriate in connection with such transfer, including, without limitation, termination or modification of the Related Agreements and/or the Asset Management Agreement as they relate to this Lease, but no amendment hereto shall (i) increase Tenant's rental obligations or other financial obligations hereunder, (ii) have a material adverse effect upon Tenant's rights hereunder, (iii) materially increase Tenant's non-economic obligations hereunder, or (iv) decrease Landlord's obligations hereunder; and provided further that in such event Landlord shall not, without the consent of Tenant, transfer the Leased Property or any interest therein to any Person which (A) does not have sufficient financial resources and liquidity to fulfill "Owner's" obligations under the Management Agreement and Landlord's obligations under this Lease, or (B) who has been, or is in control of, controlled by or under common control with Persons who have been, convicted of felonies involving moral turpitude in any state or federal court.

(b) As compensation for the early termination of its leasehold estate under this Article 24 because of a sale of the Leased Property (or as otherwise expressly provided in this Lease), and as a condition precedent to Landlord's right to terminate this Lease upon such a sale, Landlord shall, not more than one (1) year prior to the anticipated termination date of this Lease and in any event within sixty (60) days prior to the closing

of such sale, either (i) pay to Tenant the Fair Market Value of Tenant's leasehold estate hereunder as of the termination date of the Lease or (ii) offer to lease to Tenant, pursuant to one or more leases, one or more substitute hotel facilities (a "COMPARABLE LEASE") that (A) are comparable, in Tenant's commercially reasonable judgment, to the average quality of the properties leased pursuant to the other Pool Leases and the Other Leases, taking into consideration the age, physical condition, location and other relevant factors, and (B) would create for Tenant leasehold estates having an aggregate Fair Market Value as to that portion of its term equal to the remaining Term hereunder of no less than the Fair Market Value of the remaining Term hereunder, both such values to be determined as of the closing of the sale of the Leased Property. It is the intent of the parties that the Comparable Lease shall result in substantially the same ratio between Tenant's Operating Profit and Rent as then exists under this Lease for the Fiscal Year immediately preceding the sale. For the purposes of determining the Fair Market Value for purposes of this Section 24.1 or pursuant to any other Section of this Lease providing for such compensation of Tenant upon a Lease termination, a discount rate of twelve percent (12%) per annum will be used, and the annual income for the remainder of the Term will be assumed to be equal to the average Tenant Operating Profit generated during the three (3) Fiscal Years immediately preceding the termination date, or if three (3) Fiscal Years have not elapsed since the Commencement Date, the average during the preceding Fiscal Years that have elapsed (with the annual income for each of such Fiscal Years escalated from the end of each such Fiscal Year to the date of determination at the rate of inflation before such average is determined), provided that this amount shall be determined on a pro forma basis if the Leased Property has not operated as a hotel for at least the preceding twelve (12) months. In the event Landlord and Tenant are unable to agree upon the Fair Market Value of the original leasehold estate or the proposed Comparable Lease leasehold estate, it shall be determined by arbitration pursuant to the procedure set forth in Article 15. The parties agree that, if Landlord elects to offer enter into a Comparable Lease, to the extent that the Fair Market Value of the Comparable Lease is less than the Fair Market Value of the original leasehold estate, calculated as set forth above, then Landlord shall compensate Tenant in cash for the deficiency prior to the effective date of the termination.

- (c) Notwithstanding the provisions of Section 24.1(b), Landlord shall be entitled to terminate this Lease in connection with a sale or other transfer of the Leased Property to an unrelated Person or a Person in which Host O.P. owns, directly or indirectly, less than two-thirds of the equity interests, without payment of any termination fee, by giving not less than sixty (60) days prior written Notice to Tenant, provided that the landlords under the Other Leases and the other Pool Leases (excluding this Lease) relating to an aggregate of fewer than twelve (12) hotels have elected to terminate such Other Leases or other Pool Leases (excluding this Lease) without payment of a termination fee.
- (d) Host O.P. agrees to guarantee Landlord's obligation to pay to Tenant the compensation for (i) termination by a Superior Mortgagee or Superior Landlord under Section 20.2, (ii) termination of this Lease following a Casualty pursuant to Section 10.2.3, or (iii) termination of this Lease by Tenant by reason of Landlord's election not to make an Award available to Tenant for restoration following a Condemnation pursuant to Section 11.2; provided that at the time of any such termination Landlord is a wholly owned direct or indirect subsidiary of Host O.P., and if Landlord is then partially owned, directly or indirectly, by Host O.P., Host O.P. shall guaranty that portion of such compensation

that represents the same percentage of the total compensation payable as Host O.P.'s direct or indirect percentage ownership interest in Landlord. Landlord agrees to be exclusively responsible for (and shall reimburse Tenant for any payment Tenant otherwise might be required to make) the Manager's termination fee (if any) payable to Manager because of the termination of the Management Agreement in connection with the sale of the Leased Property.

(e) As compensation for the early termination of its leasehold estate under this Article 24 because of a Tax Law Change, Landlord shall, not more than one (1) year prior to the anticipated termination date of this Lease and in any event within sixty (60) days of such termination, pay to Tenant the Fair Market Value of Tenant's leasehold estate hereunder as of the termination date of this Lease, as determined under Section 24.1(b). The amount determined to be the Fair Market Value of Tenant's leasehold estate shall be payable, at the option of Landlord, in cash or in the form of stock in Host REIT.

# 24.2 Tenant's Right to Terminate Lease upon Certain Events.

Notwithstanding any provision of this Lease to the contrary, Tenant shall be entitled to terminate this Lease by giving not less than 180 days' prior Notice to Landlord, without penalty, provided that the Other Tenants have theretofore elected to terminate Other Leases and other Pool Leases (excluding this Lease) relating to an aggregate of fewer than twelve (12) hotels without penalty pursuant to a similar provision contained in such Other Leases or other Pool Leases (excluding this Lease).

 $24.3\ \mbox{Termination}$  of Lease Following Notice of Termination of Guarantee.

At any time after a notice of termination has been given pursuant to Section 10(a) of the Guarantee and the six-month period specified in Section 10(a)(iii) of the Guarantee has commenced, Landlord shall be entitled to terminate this Lease by giving Notice thereof to Tenant, and upon expiration of the time, if any, fixed in such Notice, this Lease shall terminate and except as otherwise expressly provided herein the rights and obligations of the parties under this Lease shall cease, subject to compliance with the provisions of this Section 24.3. Landlord shall not be required to pay any termination fee under this Article 24 if such termination pursuant to this Section 24.3 occurs no later than 12 months after the date on which the six-month period specified in Section 10(a)(iii) of the Guarantee commenced; provided, however, that if the termination occurs after such 12-month period, then Landlord shall be required to pay Tenant a termination fee calculated in accordance with Section 24.1(b). Notwithstanding the foregoing, in the event any other Pool Lease under which there does not then exist a default by the Other Tenant thereunder beyond the applicable notice and cure period, is terminated pursuant to a provision similar to the foregoing, Landlord's right to terminate this Lease under this Section 24.3 without payment of a termination fee shall be conditioned upon termination of this Lease and all other Pool Leases no later than 4 months after the earliest date on which termination of this Lease or any other Pool Lease pursuant to this Section 24.3 becomes effective. The provisions of this 24.3 shall be in addition to, and not in lieu of, the right of Landlord to terminate this Lease by reason of the occurrence of an Event of Default hereunder pursuant to the provisions of Article 12, without regard to the time limits or other provisions of this Section 24.3.

	LANDLORD:
Attest:	By:
Name:	Name:
Title:	Title:
Attest:	TENANT: By:
Name:	Name:
Title:	Title:
	olely to acknowledge and agree to the obligations, 20.2, 21.4 and 24.1(d) and Schedule 22.2.
	HOST MARRIOTT, L.P., a Delaware limited partnership
Attest:	Ву:
Name:	Name:
Title:	Title:

#### PROVISIONS RELATING TO EXCESS FF&E

(a) This Schedule 22.2 is intended to insure that all of the rent

payable under this Lease qualifies as "rents from real property" within the meaning of Section 856(d) of the Code or any similar or successor provisions thereto. In furtherance of such purpose, the parties have agreed to the terms set forth in the following paragraphs of this Schedule 22.2 with the objective

that, anything contained in this Lease to the contrary notwithstanding, the average of the adjusted tax basis of the items of "personal property" (within the meaning of Section 856(d)(i)(C) of the Code) that are leased to Tenant under this Lease at the beginning and at the end of any calendar year shall not exceed fifteen percent (15%) of the average of the aggregate adjusted tax bases of the Leased Property at the beginning and at the end of each such calendar year (the "FF&E LIMITATION"). The provisions contained in the following paragraphs shall be interpreted in a manner consistent with the intent and objective described above (it being understood that this paragraph constitutes a statement of the parties' mutual intent only and that the failure to achieve such objective, absent any Default or Event of Default under the other paragraphs of this Schedule 22.2 or any other provisions of this Lease, shall not constitute a

Default or an Event of Default hereunder). [In order to avoid exceeding the FF&E Limitation at the commencement of this Lease, Tenant has entered into an Excess FF&E Lease (as defined below) with Non-Controlled Subsidiary for certain Excess FF&E (as defined below) more specifically described therein that Landlord has sold to Non-Controlled Subsidiary and that is now owned by Non-Controlled Subsidiary (the "INITIAL FF&E LEASE").]

(b) If Landlord reasonably anticipates and gives Notice (an "EXCESS FF&E NOTICE") and reasonably satisfactory evidence to Tenant that the FF&E Limitation might be exceeded with respect to the Leased Property for any Fiscal Year, Tenant shall, in accordance with the provisions set forth below and within sixty (60) days following the delivery of such Excess FF&E Notice, either (a) purchase from Landlord those items or categories of FF&E to be acquired by Landlord during such Fiscal Year which are designated in such Excess FF&E Notice as anticipated to cause Landlord to exceed the FF&E Limitation ("EXCESS FF&E") or (b) arrange for Non-Controlled Subsidiary or another third party (in either case, a "THIRD-PARTY PURCHASER") to purchase such Excess FF&E from Landlord and to lease it to Tenant pursuant to a written lease agreement between such Third-Party Purchaser and Tenant (an "EXCESS FF&E LEASE") that shall include the terms specified for an Excess FF&E Lease in this Schedule 22.2 and in Schedule 22.2-A

hereto (it being understood that, without limiting the foregoing, Landlord and Tenant intend that each Excess FF&E Lease be structured in a manner intended to avoid the classification of Tenant's obligations thereunder as Capitalized Lease Obligations).

(c) Upon receiving an Excess FF&E Notice, Tenant shall first offer to Non-Controlled Subsidiary the opportunity to purchase from Landlord the Excess FF&E designated therein and to lease same to Tenant pursuant to an Excess FF&E Lease. Each Excess FF&E Lease with Non-Controlled Subsidiary shall provide for an annual rental in an amount equal to the mathematical product of (i) the applicable Market Leasing Factor (as defined below) for all Excess FF&E subject to such Excess FF&E Lease multiplied by (ii) the Excess FF&E Value (as defined below) of the Excess FF&E subject to such Excess FF&E Lease.

(d) If Non-Controlled Subsidiary does not agree to purchase and lease all the Excess FF&E which is the subject of an Excess FF&E Notice within fifteen (15) days after the date it receives Tenant's offer with respect thereto, then Tenant shall either purchase such Excess FF&E from Landlord for Tenant's own account or shall arrange for another Third-Party Purchaser that has satisfied the requirements of paragraph (j) of this Schedule 22.2 to purchase such Excess

FF&E from Landlord and lease it to Tenant pursuant to an Excess FF&E Lease. If a Third-Party Purchaser that has satisfied the requirements of paragraph (j) of this Schedule 22.2 shall not have purchased such Excess FF&E from Landlord and

leased it to Tenant under an Excess FF&E Lease within forty-five (45) days after Landlord's delivery of the Excess FF&E Notice relating thereto, then Tenant shall itself purchase such Excess FF&E from Landlord as and when (but only after) Landlord takes title to such Excess FF&E. Tenant shall purchase, or shall cause each Third-Party Purchaser to purchase, Excess FF&E with the purchaser's own funds.

- (e) With respect to any Excess FF&E first leased or purchased by Tenant pursuant to the terms of this Schedule 22.2 during a particular calendar
- year, Tenant's annual Rent obligations shall be reduced in the following manner (the "FF&E ADJUSTMENT"):
  - (i) For the calendar year in which such Excess FF&E is first placed in service by either Tenant or a Third-Party Purchaser, such reduction shall be in an amount (the "FIRST YEAR FF&E ADJUSTMENT") equal to the mathematical product of (A) the Market Leasing Factor (as defined below) for personal property with an average expected useful life corresponding to the weighted average expected useful life (as determined in accordance with GAAP and rounded to the nearest whole year) of all Excess FF&E first placed in service by Tenant or a Third-Party Purchaser during such calendar year (such weighted average, the "APPLICABLE EXPECTED LIFE") times (B) the Excess FF&E Cost (as defined below) of all Excess FF&E first placed in service by Tenant or a Third-Party Purchaser during such calendar year times (C) either (x) 100% if Tenant leases such Excess FF&E from Non-Controlled Subsidiary or (y) 110% if Tenant purchases such Excess FF&E or leases such Excess FF&E from a Third-Party Purchaser other than Non-Controlled Subsidiary times (D) 50%;
  - (ii) For each subsequent calendar year prior to the calendar year in which the Applicable Expected Life for such Excess FF&E expires, such reduction shall be in an amount equal to twice the First Year FF&E Adjustment; and
  - (iii) For the calendar year in which the Applicable Expected Life for such Excess FF&E expires, such reduction shall be in an amount equal to the First Year FF&E Adjustment.

It is contemplated that there would be a separate FF&E Adjustment for all Excess FF&E first placed in service during a single calendar year (with such FF&E Adjustment extending for a period equal to the lesser of the remaining Term or the Applicable Expected Life of the Excess FF&E acquired during such calendar year). The Rent payable by Tenant

for each Accounting Period in a calendar year to which one or more FF&E Adjustments apply shall be reduced by an amount equal to the mathematical product of (i) the amount of such applicable FF&E Adjustment (or if more than one FF&E Adjustment apply in such calendar year, the sum of such applicable FF&E Adjustments) times (ii) a fraction, the numerator of which is one and the denominator of which is the number of Accounting Periods in such calendar year. The "EXCESS FF&E VALUE" of any Excess FF&E shall be the fair market value of such Excess FF&E (which shall be the purchase price paid by the purchaser thereof from Landlord, whether such purchaser is Non-Controlled Subsidiary, another Third Party Purchaser or Tenant) plus the aggregate amount of out-of-pocket transactional costs (including, without limitation, reasonable attorneys' fees and any ad valorem, sales, transfer, transaction or similar tax, levy or other governmental charge) incurred by such purchaser in connection with its purchase of such Excess FF&E. The "MARKET LEASING FACTOR" for the first two (2) Lease Years shall be the amounts set forth in Schedule 22.2-B\* (with there to be

a separate Market Leasing Factor for each whole number of years of expected useful life of Excess FF&E). Thereafter the Market Leasing Factors shall be determined by an independent valuation expert, acceptable to both Landlord and Tenant, who shall determine the Market Leasing Factors based on the median of the leasing rates of at least three nationally recognized companies engaged in the business of leasing similar FF&E or personal property and equipment with average expected useful lives equal to the weighted average of the expected useful lives set forth on Schedule 22.2-B. The cost of such expert shall be

borne by Landlord. The Market Leasing Factors shall take into account any use taxes and similar Impositions payable by Tenant in connection with its leasing of Excess FF&E under the relevant Excess FF&E Leases, as well as the reasonably estimated anticipated out-of-pocket cost (including reasonable attorneys' and accountants' fees) to Tenant of administering such Excess FF&E Leases during the Term, so that the economic burden of such Impositions and administration costs will be borne by Landlord. The "EXCESS FF&E COST" of any Excess FF&E shall be the Excess FF&E Value of such Excess FF&E plus, if Tenant leases such Excess FF&E from a Third-Party Purchaser, the aggregate amount of out-of-pocket transactional costs (including, without limitation, reasonable attorneys' fees and any ad valorem, sales, transfer, transaction or similar tax, levy or other governmental charge) incurred by Tenant in connection with its entry into an Excess FF&E Lease of such Excess FF&E in accordance with this Schedule 22.2.

(f) Landlord and Tenant agree to cause Manager to purchase all Excess FF&E for Landlord's account with funds from the FF&E Reserve (or with funds otherwise made available by Landlord). The parties specifically intend (and Tenant hereby agrees to take such reasonable steps at Landlord's expense as Landlord may request to insure) that

As provided below with respect to the Market Leasing Factors for subsequent

years, the Market Leasing Factor for the first two years will take into account any use taxes and similar Impositions payable by Tenant in connection with its leasing of Excess FF&E under the relevant Excess FF&E Leases, as well as the reasonably estimated anticipated out-of-pocket cost (including reasonable attorneys' and accountants' fees) to Tenant of administering such Excess FF&E Leases during the Term, so that the economic burden of such Impositions will be borne by Landlord.

-3-

Landlord shall own all Excess FF&E for a period of time sufficient to permit such Excess FF&E to become subject to any then existing Liens in favor of the Facility Mortgagee that encumber FF&E acquired by Landlord; provided, however, that in no event shall Landlord own any Excess FF&E (i) for more than five (5) Business Days, or (ii) so long that the FF&E Limitation would be exceeded at the beginning or end of any calendar year. Without limiting Tenant's obligation under the immediately preceding sentence to take reasonable steps requested by Landlord (at Landlord's expense) to achieve the objective set forth therein, it is understood and agreed that the failure to achieve such objective, absent any Default or Event of Default under the other provisions of this Schedule 22.2 or

any other provisions of this Lease, shall not constitute a Default or an Event of Default hereunder. Every purchase of Excess FF&E from Landlord, whether by Tenant or a Third-Party Purchaser, shall be made expressly subject to any and all Liens encumbering such Excess FF&E in favor of any Facility Mortgagee. Following the purchase of any Excess FF&E by Tenant or a Third-Party Purchaser as contemplated by this Schedule 22.2, Landlord shall not be considered to own

or be the lessor of any of such Excess FF&E during the Term of this Lease for any purpose, nor shall any of such Excess FF&E be considered part of the Leased Property, and neither Landlord nor Tenant shall at any time take a position (in its books and records or otherwise) or make an assertion inconsistent therewith.

(g) In the event that Tenant owns any Excess FF&E at the expiration or earlier termination of this Lease (including, without limitation, a termination in connection with a transfer of ownership of the Leased Property), Landlord shall purchase from Tenant and Tenant shall sell to Landlord (the "EXCESS FF&E REPURCHASE"), on the effective date of such expiration or termination, all such Excess FF&E (except for any Excess FF&E to which the terms of paragraph (i) of this Schedule 22.2 apply) for a purchase price equal to the fair market value

(which the parties hereby agree shall not be less than the adjusted book value) of such Excess FF&E at such time (the "EXCESS FF&E REPURCHASE PRICE"). The Excess FF&E Repurchase Price shall be payable first by offset against any Rent owed by Tenant to Landlord as of such time and any amounts owed by Tenant to Landlord as of such time under the Working Capital Note, and the remainder (if any) shall be paid by Landlord to Tenant in cash within ten (10) days after the expiration or termination of this Lease.

(h) In the event that Tenant is leasing any Excess FF&E from a Third-Party Purchaser at the expiration or earlier termination of this Lease (including, without limitation, a termination resulting in connection with a transfer of ownership of the Leased Property), Landlord shall purchase and assume from Tenant, and Tenant shall sell, assign and delegate to Landlord, on the effective date of such expiration or termination all Tenant's right, title and interest in and its obligations under each Excess FF&E Lease between Tenant and such Third-Party Purchaser (an "EXCESS FF&E LEASEHOLD INTEREST"). In the aforesaid transaction (an "EXCESS FF&E LEASEHOLD INTEREST TRANSFER"), the transfer price for such Excess FF&E Leasehold Interest (the "EXCESS FF&E LEASEHOLD INTEREST TRANSFER PRICE") shall be an amount equal to the fair market value of such Excess FF&E Leasehold Interest at such time (as determined in accordance with the procedure provided below) and shall be payable (i) by Landlord if such fair market value is a positive number or (ii) by Tenant if such fair market value is a negative number. The fair market value of the Excess FF&E Leasehold Interest shall be

determined taking into account all relevant factors, including the remaining term thereof, the remaining expected useful life of the Excess FF&E (as determined in accordance with GAAP) subject to such leasehold, and the leasing rate that would apply, under market conditions at that time, if a new lease for such Excess FF&E were to be entered into with an unrelated party for a term equal to the term remaining for such Excess FF&E Leasehold Interest; provided, however, that in no event shall such fair market value be less than the adjusted book value of such Excess FF&E Leasehold Interest. Any amount payable pursuant to this paragraph shall be paid within ten (10) days after the expiration or termination of this Lease and, if due from Landlord, shall be paid first by offset against any Rent owed by Tenant to Landlord as of such time and any amounts owed by Tenant to Landlord as of such time under the Working Capital Note, and the remainder (if any) shall be paid in cash.

- (i) In the event that the Facility Mortgagee forecloses on its Lien on any Excess FF&E owned by Tenant in connection with, but not separate from, a foreclosure of the Leased Property, Landlord shall reimburse Tenant for the loss of such Excess FF&E in an amount equal to the Excess FF&E Repurchase Price plus any reasonable costs and expenses (including, without limitation, reasonable attorneys' fees) incurred by Tenant in complying with (but not contesting) such foreclosure (the "EXCESS FF&E REIMBURSEMENT AMOUNT"). In the event that the Excess FF&E Reimbursement Amount becomes payable to Tenant pursuant to the immediately preceding sentence, (i) it shall be paid first by offsetting any amounts owed by Tenant to Landlord as of such time and the balance (if any) shall be paid in cash, (ii) Tenant agrees to seek payment of such cash balance (if any) solely from Host O.P. pursuant to the terms of the Facility Mortgagee Agreement and shall not make any demand or claim therefor against Landlord, the Facility Mortgagee, any purchaser in foreclosure or transferee by deed in lieu of foreclosure or other party claiming under any of the foregoing, and (iii) Host O.P. agrees to pay any such cash balance.
- (j) No Third-Party Purchaser shall purchase any Excess FF&E from Landlord unless and until such Third-Party Purchaser shall have (i) agreed to purchase such Excess FF&E with cash from such Third-Party Purchaser's own funds; (ii) agreed to lease such Excess FF&E to Tenant under an Excess FF&E Lease; and (iii) entered into an agreement with Landlord, the Facility Mortgagee and Host O.P. pursuant to which (A) the Third-Party Purchaser acknowledges to the Facility Mortgagee that all Excess FF&E purchased by such Third-Party Purchaser while the Facility Mortgage is in effect is subject to the Facility Mortgagee's first-priority Lien (without Third-Party Purchaser assuming any liability for Landlord's obligations that are secured by or arise under any Facility Mortgage) and that the Facility Mortgagee's rights and remedies with respect to such Excess FF&E shall survive and be enforceable with respect thereto; (B) the Third-Party Purchaser covenants to the Facility Mortgagee to execute and deliver UCC-1 financing statements prepared by Landlord or any Facility Mortgagee confirming the foregoing for notice purposes, which UCC-1 financing statements may then be filed by Landlord or the Facility Mortgagee at Landlord's sole expense; (C) the Third-Party Purchaser covenants not to sell, lease, transfer or otherwise dispose of such Excess FF&E or any interest therein, or grant or cause or permit there to exist any lien, charge or encumbrance with respect thereto, other than the Lien in favor of the Facility Mortgagee, any other Liens which are the responsibility of Landlord and any Lien arising pursuant to such agreement

or the Excess FF&E Lease, (D) in the event that this Lease expires or is terminated prior to the expiration or termination of the Excess FF&E Lease, Landlord agrees to purchase from the Third-Party Purchaser, and the Third-Party Purchaser agrees to sell to Landlord, all Excess FF&E owned by the Third-Party Purchaser at that time (other than any Excess FF&E to which clause (E) below applies) on the same terms as those applicable to the Excess FF&E Repurchase; and (E) if the Facility Mortgagee forecloses on its Lien with respect to Excess FF&E owned by the Third-Party Purchaser: (x) Landlord and Host O.P. agree to reimburse the Third-Party Purchaser for the loss of such Excess FF&E (such reimbursement to be paid first by offsetting any amounts owed by the Third-Party Purchaser to Landlord and the balance (if any) to be paid in cash); (y) Host O.P. agrees to pay the cash balance (if any) of such reimbursement amount directly to the Third-Party Purchaser without the need for notice or demand on Landlord; and (z) the Third-Party Purchaser agrees to look only to Host O.P. for payment of such cash balance.

(k) It is the intent of Landlord and Tenant that the leases of FF&E pursuant to the [Initial FF&E Lease,] any Excess FF&E Lease[,] and this Lease shall be treated as operating leases and not Capitalized Lease Obligations under GAAP. Landlord and Tenant agree to cooperate to the extent feasible and consistent with the terms of this Lease to provide terms for such leases of FF&E that are so treated.

# DEEMED CONSENTS

- . Section 4.3.1 If Hazardous materials are discovered in violation of Environmental Laws on the Leased Property, Tenant will exercise all commercially reasonable efforts to cause Manager to take all actions and incur all expenses (which actions and expenses will be subject to Landlord's prior approval, except in Emergency Situations) as may be necessary or required by any Government Agency.
- . Section 9.5 All insurance policies shall include only deductibles reasonably approved by Landlord.
- . Section 10.2.4 Landlord may, at its option, condition advancement of insurance proceeds and other amounts on, among other things, its approval of plans and specifications of an architect reasonably satisfactory to Landlord.
- . Section 11.2 Landlord may, at its option, condition advancement of an Award and other amounts on, among other things, its approval of plans and specifications of an architect reasonably satisfactory to Landlord.
- . Section 16.1 For subleases made in the ordinary course of operating the Facility which are for a term of more than five years and involve more than 1,000 square feet, Landlord's consent is required, but such consent may not be unreasonably withheld.

#### POOL GUARANTEE

THIS POOL GUARANTEE (this "GUARANTEE") is made as of,
199 by Crestline Capital Corporation, a Maryland corporation ("CCC"),
and, a Delaware corporation ("POOL PARENT") (individually
a "GUARANTOR" and collectively the "GUARANTORS"), to and for the benefit of eac
of the entities listed on Schedule 1 attached hereto (as such Schedule 1 may be

amended from time to time to reflect any entities that are landlords under any Leases (as hereinafter defined) added or deleted from Schedule 3 attached

hereto, collectively, the "LANDLORDS").

WHEREAS, CCC is the owner of all of the capital stock of Pool Parent; and

WHEREAS, Pool Parent is the owner, directly or through wholly owned subsidiaries, of all of the equity interests of each of the entities listed on Schedule 2 attached hereto (as such Schedule 2 may be amended from time to time

to reflect any entities that are tenants under any Leases added or deleted from Schedule 3 attached hereto, collectively, the "TENANTS"); and  $\frac{1}{2} \left( \frac{1}{2} \right) \left($ 

WHEREAS, the Tenants and the Landlords have executed and delivered the leases (the "LEASES"), working capital notes, and other agreements and instruments listed on Schedule 3 attached hereto (as such Schedule 3 may be

amended from time to time to reflect any addition or deletion of Leases on Schedule 1 to the Pooling Agreement (as hereinafter defined)) and the Guarantors, Tenants, and Landlords have executed and delivered a Pooling and Security Agreement of even date herewith (as such agreement may be amended from time to time, the "POOLING AGREEMENT" and, together with the agreements and instruments listed on Schedule 3, the "AGREEMENTS"); and

WHEREAS, each Landlord is willing to enter into the Agreements to which it is a party only upon the condition that it receive this guarantee of the Tenants' payment and performance under all of the Agreements;

WHEREAS, the Guarantors desire to induce the Landlords to enter into the Leases, which may be reasonably expected to benefit, directly or indirectly, the Guarantors;

WHEREAS, certain capitalized terms used herein are defined in Section 11 of this  $\mbox{\it Guarantee}\,.$ 

NOW, THEREFORE, in consideration of the premises and in order to induce the Landlords to enter into the Agreements to which they are a party, the Guarantors jointly and severally agree as follows:

# 1. GUARANTEE OF OBLIGATIONS

The Guarantors unconditionally and absolutely guarantee the due and punctual payment of rent and all other the monies due or which may become due under the Agreements, and the due and punctual performance and observance by the Tenants or any

Guarantor of all of the other terms, covenants and conditions of the Agreements to be performed by any Tenant, whether according to the present terms thereof, at any earlier or accelerated date or dates as provided therein, or pursuant to any extension of time or to any change or changes in the terms, covenants and conditions thereof now or at any time hereafter made or granted, and including obligations that, but for any automatic stay under Section 362(a) of the Bankruptcy Code (11 U.S.C. (S)101 et seq.) (or other successor provision), would become due. All payments due hereunder shall be made in accordance with the procedures set forth in the Pooling Agreement, and payment in accordance with such procedures shall be deemed to satisfy the obligations of the Guarantors hereunder.

#### 2. LIMIT OF CCC'S LIABILITY

Notwithstanding any provisions of this Guarantee to the contrary (except as specifically stated), the obligations and liabilities of CCC under this Guarantee shall at all times be subject to the CCC Limit of Liability; provided that the CCC Limit of Liability shall not limit the obligation of CCC (i) to guarantee repayment of funds misappropriated or wrongfully applied by Pool Parent or any Tenant or Tenant Member (as defined in the Pooling Agreement) in violation of the provisions of the Pooling Agreement, but only to the extent of the funds misappropriated or wrongfully applied, or (ii) to pay for any losses of Landlords resulting from the fraud of Pool Parent or any Tenant or Tenant Member, or (iii) to discharge any liens on any property of Pool Parent or any Tenant existing in violation of the obligations of Pool Parent or any Tenant under the Pooling Agreement or any applicable Lease (the obligations described in clauses (i), (ii), and (iii), collectively, the "SPECIAL GUARANTEE OBLIGATIONS").

# 3. COLLATERAL

The obligations of Pool Parent hereunder are secured by the pledge of certain accounts and other collateral pursuant to the Pooling Agreement.

# 4. CERTAIN WAIVERS

The Guarantors waive diligence, presentment, protest, notice of dishonor, demand, extension of time for payment, notice of non-payment at maturity and indulgences and notices of every kind, and consent to any and all forbearances and extensions of time of payment of any monies due or which may become due under the Agreements, and to any and all changes in terms, covenants and conditions of the Agreements; it being the intention hereof that each Guarantor shall remain liable as a principal until the full amount of all sums payable under the Agreements shall have been fully paid, the terms, covenants and conditions of the Agreements shall have been performed and observed by the Tenants and each Guarantor and all other amounts payable under this Guarantee shall have been fully paid, notwithstanding any act, omission or thing which might otherwise operate as a legal or equitable discharge of the Guarantors, or any one or more of them.

# 5. OBLIGATIONS UNCONDITIONAL

(a) The Guarantors agree that the obligation of each Guarantor as a Guarantor shall not be impaired, modified, changed, released or limited in any manner

2

whatsoever by any impairment, modification, change, release or limitation of the liability of any one or more of the Tenants or their respective estates in bankruptcy, resulting from the operation of any present or future provision of the bankruptcy laws or other similar statute, or from the decision of any court.

(b) The Guarantors agree that any one or more of the Landlords shall have the full right, in its or their sole discretion and without any notice to or consent from the Guarantors, from time to time and at any time and without affecting, impairing or discharging, in whole or in part, the liability of the Guarantors hereunder: (i) to make any change, amendment or modification whatsoever of any of the terms or conditions of any one or more of the Agreements; (ii) to extend, in whole or in part, by renewal or otherwise, and on one or any number of occasions, the time for the payment of any amount pursuant to any one or more of the Agreements or for the performance of any term or condition of any one or more of the Agreements; (iii) to settle, compromise, release, substitute, surrender, modify or impair, to enforce and exercise, or to fail or refuse to enforce or exercise, any claims, rights or remedies, of any kind or nature, which any one or more of the Landlords may at any time have against any one or more of the Tenants or any Guarantor, or with respect to any security interest of any kind held by any one or more of the Landlords at any time, whether under any one or more of the Agreements or otherwise; (iv) to release, substitute, surrender or enforce any security interest of any kind held by any one or more of the Landlords at any time, and to collect and retain or liquidate any collateral subject to such security interest, whether under this Guarantee or otherwise; and (v) to apply any sums by whomsoever paid or howsoever realized to any liability or liabilities of any one or more of the Tenants to any one or more of the Landlords regardless of what liabilities of any one or more of the Tenants remain unpaid.

#### 6. OBLIGATIONS INDEPENDENT

The Guarantors agree that the Guarantors' obligations hereunder are irrevocable, are joint and several (except as the obligations of CCC may be limited under Section 2), and are independent of the obligations of any one or more of the Tenants; that a separate action or actions may be brought and prosecuted against the Guarantors or any one or more of them regardless of whether any action is brought against any one or more of the Tenants (or against any other Guarantor) or whether any one or more of the Tenants (or any other Guarantor) is joined in any such action or actions; and that the Guarantors waive the benefit of any statute of limitations affecting their liability hereunder or the enforcement hereof.

### 7. NO SUBROGATION

The Guarantors agree that the Guarantors shall have no right of subrogation whatever with respect to the payment and performance obligations of the Tenants under the Agreements or to any collateral securing such payment and performance obligations unless and until such payment and performance obligations have been paid and performed in full.

#### 8. REMEDIES

- (a) The Guarantors agree that this Guarantee is a continuing guaranty and that all liabilities which are guaranteed by the Guarantors hereunder shall be conclusively presumed to have been created in reliance hereon. No failure or delay on the part of any Landlord in exercising any right, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The rights and remedies herein expressly specified are cumulative and not exclusive of any rights or remedies which any Landlord would otherwise have. No notice to or demand on any one or more of the Guarantors in any case shall entitle any one or more of the Guarantors to any other further notice or demand in similar or other circumstances or constitute a waiver of the rights of any Landlord to any other or further action in any circumstances without notice or demand.
- (b) The Guarantors agree that in the event that any one or more of the Landlords retain or engage an attorney or attorneys to enforce this Guarantee and one or more of the Landlords prevail in such enforcement action, the Guarantors will reimburse the Landlords for all expenses incurred, including reasonable attorneys' fees and disbursements, without regard to the CCC Limit of Liability set forth in Section 2. In the event that the Guarantors prevail in any such enforcement action, the Landlords bringing such action will reimburse the Guarantors for all expenses incurred, including reasonable attorneys' fees and disbursements. If the Guarantors commence an action to enforce their rights hereunder, the parties prevailing in such action shall be reimbursed by the parties losing such action for all expenses incurred, including reasonable attorneys' fees and disbursements, without regard to the CCC Limit of Liability.

# 9. BENEFIT OF GUARANTEE

The Guarantors agree that this Guarantee shall inure to the benefit of and may be enforced by each of the Landlords and its respective successors, transferees and assigns, and shall be binding upon and enforceable against the Guarantors and the Guarantors' successors or assigns. Without limiting the generality of the foregoing, the Guarantors further agree that each Landlord may assign or otherwise transfer its rights under the Agreements or any interest in any of the foregoing to any person or entity (including any lender or other creditor of such Landlord), and such other person or entity shall thereupon become vested with all of the rights in respect thereof granted to the Landlords herein or otherwise; provided that no Landlord shall be permitted to assign its rights hereunder to any Person to which such Landlord sells or conveys any hotel subject to a Lease, other than a Subsidiary (as defined in the Pooling Agreement) of the Landlord Representative.

# 10. TERMINATION OF GUARANTEE; REINSTATEMENT

(a) At any time when the CCC Limit of Liability shall be equal to zero as of the most recent Distribution Date (as defined in the Pooling Agreement), after giving effect to any payments of Rent Shortfalls on such date, Pool Parent shall be entitled to cause this Guarantee and the Pooling Agreement to terminate (except to the extent provided in paragraph (b) below) by giving Landlord Representative (as defined in the Pooling Agreement) notice of termination, in which event the following provisions will apply:

- (i) If (x) the notice is given after the end of a Fiscal Year, (y) the CCC Limit of Liability as of the end of the Fiscal Year immediately preceding the date of such notice was zero, and (z) either the notice is given prior to the Distribution Date with respect to the first Accounting Period of a Fiscal Year or the Cumulative Guarantee Payments as of the end of the Fiscal Year immediately preceding the date of such notice are greater than or equal to the Cumulative Guarantee Ceiling for the Fiscal Year in which such notice is given, this Guarantee will terminate (except to the extent provided in paragraph (b) below) effective on the date 6 months after the date of such notice, and the provisions of paragraph (iii) below shall apply during such 6-month period.
- (ii) If the conditions set forth in paragraph (i) above do not apply, then the following provisions will be applicable.
  - (A) Pool Parent shall provide to Landlord Representative, together with the termination notice, a statement setting forth a calculation of the Allowable Projected Year-End Guarantee Payments as of the end of the then current Fiscal Year and the Allowable Projected Annual Operating Deficit for each Lease as to which a Guarantee Payment has been made in such Fiscal Year.
  - (B) If the statement provided by the Pool Parent under paragraph (A) shows that the Allowable Projected Year-End Guarantee Payments as of the end of the then current Fiscal Year are greater than or equal to the Cumulative Guarantee Ceiling for such Fiscal Year, and the Landlord Representative does not dispute such calculation, then the termination will be effective six months after the date of such notice, and the provisions of paragraph (iii) below shall apply during such 6-month period.
  - (C) If the statement provided by the Pool Parent does not show that the Allowable Projected Year-End Guarantee Payments as of the end of the then current Fiscal Year are greater than or equal to the Cumulative Guarantee Ceiling for such Fiscal Year, or the Landlord Representative informs the Pool Parent that it reasonably believes that the calculations on the statement incorrectly show that the Allowable Projected Year-End Guarantee Payments as of the end of the then current Fiscal Year are greater than or equal to the Cumulative Guarantee Ceiling for such Fiscal Year, then the notice will not be given effect at such time, and the provisions of paragraph (D) will apply.
  - (D) If the initial statement does not show that the Allowable Projected Year-End Guarantee Payments as of the end of the then current Fiscal Year are greater than or equal to the Cumulative Guarantee Ceiling for such Fiscal Year or the Landlord Representative reasonably disputes the conclusion set forth in the statement, then Pool Parent may provide additional statements with respect to subsequent Accounting Periods in such Fiscal Year, setting forth the calculations in the manner set forth in paragraph (A). If a subsequent statement shows that the Allowable Projected Year-End Guarantee Payments as of the end of the then current Fiscal Year are greater than or equal to the Cumulative Guarantee Ceiling for such Fiscal Year, and the Landlord

Representative does not dispute such calculation, then the termination will be effective six months after the end of the applicable Accounting Period, and the provisions of paragraph (iii) below shall apply during such 6-month period.

- (E) If, as of the end of the Fiscal Year in which the notice of termination is given, the CCC Limit of Liability, based on actual results, is zero, then the termination will be effective six months after the end of such Fiscal Year, and the provisions of paragraph (iii) below shall apply during such 6-month period.
- (F) If, as of the end of the Fiscal Year in which the notice of termination is given, the CCC Limit of Liability, based on actual results, is greater than zero, then the notice of termination will be deemed to be void and of no further effect.
- (G) If (i) a notice of termination becomes effective pursuant to paragraph (D) or (E) above, (ii) CCC and the Pool Parent made additional Guarantee Payments after the date of the initial notice and (iii) the CCC Limit of Liability as of the end of the Fiscal Year (disregarding the floor of zero) was less than the CCC Limit of Liability computed (disregarding the floor of zero) as of the Distribution Date on which the notice was initially given or immediately preceding the date of such notice, then, promptly after the date on which the relevant financial information becomes available that permits such determination, the Landlord Representative shall reimburse Pool Parent for the lesser of the amount of such additional Guarantee Payments after the date of the initial notice and the amount by which the CCC Limit of Liability as of the end of such Fiscal Year was less than the CCC Limit of Liability computed as of the time of such notice.
- (iii) During the six-month period after a notice of termination becomes effective pursuant to paragraph (i), (ii)(B), (ii)(D), or (ii)(E) above, CCC shall have no obligations hereunder other than with respect to Special Guarantee Obligations, which shall survive termination of this Guarantee, and the obligations of Pool Parent under this Guarantee shall remain in full force and effect, provided that the aggregate liability of Pool Parent for payment of Rent Shortfalls during such six-month period shall not exceed the aggregate amount of Distributable Cash with respect to all Leases during such period.
- (b) Upon expiration of the six-month period specified in paragraph (iii), neither CCC nor Pool Parent shall have any obligations hereunder other than with respect to Special Guarantee Obligations which shall survive such termination.
- (c) In the event that, on or before the 91st day after the expiration of the six-month period specified in paragraph (a)(iii) above, any Tenant or Pool Parent shall have become the debtor in any voluntary or involuntary bankruptcy proceeding, the Guarantee shall not be deemed to terminate as a result of such notice and shall be reinstated.

#### 11. CERTAIN DEFINITIONS

As used in this Guarantee, the following capitalized terms have the meanings set forth below:

"ANNUAL OPERATING DEFICIT" shall mean, with respect to any hotel for each full Fiscal Year in the Measurement Period (treating the period from the date of this Guarantee to the end of the Fiscal Year ending on or about December 31, 1999 as one full Fiscal Year), an amount equal to the excess (if any) of Tenant Operating Expenses for such Fiscal Year over Gross Revenues for such Fiscal Year. If Gross Revenues for such Fiscal Year are greater than or equal to Tenant Operating Expenses for such Fiscal Year with respect to the hotel, then there shall be no Annual Operating Deficit for such period with respect to such hotel.

"ALLOWABLE PROJECTED ANNUAL OPERATING DEFICIT" means, for any Lease with respect to the Fiscal Year in which the date of determination occurs, an amount equal to (i) the projected Annual Operating Deficit, if any, for such Lease as of the end of such Fiscal Year, based on the actual Gross Revenues and Tenant Operating Expenses through the end of the most recent Accounting Period and the projected Gross Revenues and the projected Tenant Operating Expenses for the remaining Accounting Periods in such Fiscal Year as set forth in the most recent projected operating results prepared by the applicable hotel manager, less (ii) an amount equal to 1% of the amount of Rent under such Lease for the remaining Accounting Periods in such Fiscal Year as reasonably projected by the Tenant on the basis of the most recent projected operating results prepared by the applicable hotel manager.

"ALLOWABLE PROJECTED YEAR-END GUARANTEE PAYMENTS" means, as of any date of determination, an amount equal to the sum of (i) the Cumulative Guarantee Payments as of the end of the prior Fiscal Year plus (ii) with respect to each Lease as to which any Guarantee Payment has been made in the then current Fiscal Year, the lesser of (x) the aggregate amount of such Guarantee Payments or (y) the Allowable Projected Annual Operating Deficit (as defined below) for such Lease for such Fiscal Year.

"CCC LIMIT OF LIABILITY" at any time during a Fiscal Year shall be an amount (not less than zero) equal to the excess, if any, of (i) the Cumulative Guarantee Ceiling for such Fiscal Year over (ii) the Cumulative Guarantee

Payments (as hereinafter defined) as of the date of determination.

"CUMULATIVE GUARANTEE CEILING" means, (i) for Fiscal Year 1999, \$\_\_\_\_\_\_/1/ and (ii) for any subsequent Fiscal Year, ten percent (10%) of the greater of (x) the aggregate Rent, prior to any FF&E Adjustments and excluding Additional Charges (as all such terms are defined in the Leases), payable with respect to the previous Fiscal Year pursuant to all of the Leases and (y) the aggregate actual Rent (prior to FF&E Adjustments and excluding Additional Charges) payable with respect to Fiscal Year 1999 pursuant to all of the Leases.

/1/ An agreed estimate of 10% of the aggregate amount of Rent (prior to FF&E Adjustments and excluding Additional Charges) payable under all Leases listed on Schedule 3 for Fiscal Year 1999.

- -----

"CUMULATIVE GUARANTEE PAYMENTS" shall mean, with respect to any date of determination, the lesser of (i) the aggregate amount of Annual Operating Deficits and Current Year Operating Deficits with respect to all Leases during the applicable Measurement Period and (ii) the aggregate amount of Guarantee Payments made by Pool Parent and CCC during the applicable Measurement Period.

"CURRENT YEAR OPERATING DEFICIT" shall mean, with respect to any hotel as of any date of determination, an amount equal to the excess (if any) of Tenant Operating Expenses for the period from the beginning of the Fiscal Year in which the most recently completed Accounting Period falls to the end of the most recently completed Accounting Period over Gross Revenues for such period. If Gross Revenues for such period are greater than or equal to Tenant Operating Expenses for such period with respect to the hotel, then there shall be no Current Year Operating Deficit for such period with respect to such hotel.

"GROSS REVENUES" shall have the meaning ascribed thereto for each hotel in the applicable Lease.

"GUARANTEE PAYMENT" shall mean a payment by Pool Parent or CCC in respect of a Rent Shortfall (as defined in the Pooling Agreement).

"MEASUREMENT PERIOD" means, as of any date of determination, the period beginning on the first day of the Accounting Period in which the date of this Guarantee occurs and ending on the last day of the Accounting Period ending immediately preceding such date of determination.

"TENANT OPERATING EXPENSES" shall mean, with respect to any hotel, all expenses incurred by Tenant or the applicable hotel manager on behalf of Tenant in connection with the operation of such hotel, including Rent and base and incentive management fees payable by Tenant, all determined in accordance with GAAP; provided that Tenant Operating Expenses shall not include (i) any amounts that are Landlord Obligations under the applicable Lease, (ii) any debt service payable by Tenant, or (iii) any administrative or overhead expenses of Tenant (other than such expenses reimbursable to the applicable hotel manager).

#### 12. MISCELLANEOUS

- (a) This Guarantee, the rights and obligations of the parties hereto, and any claims or disputes relating thereto, shall be governed by and construed in accordance with the laws of Maryland (but not including the choice of law rules thereof).
- (b) The Guarantors agree that service of process may be made upon any one or more of them by registered mail (return receipt requested) directed to the applicable Guarantors at their respective addresses designated for notice on the signature page hereof, and service so made upon a Guarantor shall be deemed to be completed upon receipt by such applicable Guarantor. Nothing in this Section 12(b) shall affect the right of any Landlord to serve legal process in any other manner permitted by law.
- (c) To the maximum extent permitted by law, the Guarantors hereby waive any right that they may have to a trial by jury of any dispute (whether a claim in tort,

contract, equity or otherwise) arising under or related to this Guarantee or any related matters, and agree that any such dispute shall be tried before a judge sitting without a jury.

- (d) For all purposes of this Guarantee, except as otherwise expressly provided or unless the context otherwise requires, (i) all accounting terms not otherwise defined herein shall have the meanings assigned to them in accordance with generally accepted accounting principles consistently applied, (ii) all references in this Guarantee to designated "Sections" and other subdivisions are to the designated Sections and other subdivisions of this Guarantee, (iii) the term "including" shall have the same meaning as "including, without limitation," and (v) the words "herein," "hereof," "hereunder" and other words of similar import refer to this Guarantee as a whole and not to any particular Section or other subdivision.
- (e) This Guarantee may be executed in two or more counterparts, each of which shall constitute an original, but which, when taken together, shall constitute but one instrument and shall become effective as of the date hereof when copies hereof, which, when taken together, bear the signatures of each of the parties hereto, shall have been signed.

IN WITNESS WHEREOF, the Guarantors have executed this Guarantee as of the day and year first above written.  $\,$ 

# CRESTLINE CAPITAL CORPORATION

By:
Bruce D. Wardinski President and CEO
Address for Notice: Crestline Capital Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attn: General Counsel
[POOL PARENT]
By:
James L. Francis President
Address for Notice: c/o Crestline Capital Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attn: General Counsel
HOST MARRIOTT, L.P.
Pv: Host Marriott Corporation its general

Host Marriott Corporation, its general partner

Ву:\_\_\_

Name:

Title:

Address for Notice: Host Marriott, L.P. 10400 Fernwood Road Bethesda, Maryland 20817 Attn: General Counsel

10

Schedule 1 to Guarantee

LIST OF LANDLORDS

[TO BE PROVIDED]

11

Schedule 2 to Guarantee

LIST OF TENANTS

[TO BE PROVIDED]

12

Schedule 3 to Guarantee

## LIST OF AGREEMENTS

[List will include the following agreements in connection with each lease pool: all Leases, all working capital notes, and all owner's agreements]

POOLING AND SECURITY AGREEMENT

BY AND AMONG

CRESTLINE CAPITAL CORPORATION,

[POOL PARENT], AND

THE PERSONS NAMED AS TENANTS AND TENANT MEMBERS HEREIN

AND

HOST MARRIOTT, L.P.

AND THE PERSONS NAMED AS LANDLORDS HEREIN

DATED AS OF \_\_\_\_\_\_, 199\_\_\_\_

## TABLE OF CONTENTS

		PAGE
1	CERTAIN DEFINITIONS	1
	CASH COLLATERAL ARRANGEMENTS	
	2.1 Security Agreement Relating to Account Collateral	11
	2.2 Establishment, Maintenance and Funding of Pool Cash Collateral Account	
	2.3 Disbursements from Pool Cash Collateral Account	14
2	ADDITION OF LEASES	15
٥.	ADDITION OF LEASES	15
	3.1 Addition of Leases, Tenants, or Tenant Members	15
4.	NEGATIVE COVENANTS APPLICABLE TO POOL PARENT	16
	4.1 Indebtedness	16
	4.2 Liens.	
	4.3 Investments.	
	4.4 Transfers of Interests in Tenants and Tenant Members	16
	4.5 Amendment of Organizational Documents	17
	4.6 Misapplication of Funds	17
	4.7 Place of Business	17
5.	SPE COVENANTS	17
	5.1 Business of Pool Parent	17
	5.2 Separate Existence.	
	5.3 Independent Director	
	5.4 Fundamental Changes	18
6	EVENTS OF DEFAULT; REMEDIES	18
о.	EVENTS OF DEPAULI, REMEDIES	10
	6.1 Events of Default	18
	6.2 Remedies in Case of an Event of Default	
	6.3 Remedies, Etc. Cumulative	
	6.4 Indemnity	21
7	MISCELLANEOUS PROVISIONS	22
٠.		
	7.1 Termination	
	7.2 Financial Reporting	
	7.3 Notices	
	7.4 Waiver; Amendment	
	7.5 Benefit	
	7.7 Headings	
	7.8 Counterparts	
	7.9 Assignment	
	7.10 Waiver of Jury Trial	
	7.11 Delay Not a Waiver	
	7.12 Severability	
	7.13 Preferences	
	7.14 Waiver of Notice	
	. C. L. D.	∠0

## SCHEDULES

Schedule 1 Leases, Tenants, Landlords, and Hotels

Schedule 2 Tenant Members

## POOLING AND SECURITY AGREEMENT

THIS POOLING AND SECURITY AGREEMENT (this "AGREEMENT") is made and entered into as of December [\_\_\_], 1998 by and among Crestline Capital Corporation, a Maryland corporation ("CCC"), [\_\_\_\_\_], a Delaware corporation ("POOL PARENT"), the Tenants party hereto named on Schedule 1 hereto from time to time,

and the Tenant Members party hereto named on Schedule 2 hereto from time to time

(collectively, the "CCC ENTITIES") for the benefit of (i) the Landlords named on Schedule 1 hereto from time to time, (ii) Host Marriott, L.P., a Delaware

limited partnership, in its capacity as Landlord Representative hereunder, and (iii) \_\_\_\_\_, as collateral agent for the benefit of the Landlords (together with its permitted successors and assigns, the "COLLATERAL AGENT").

#### **RECITALS**

- A. The Tenants and Landlords are party to leases (the "LEASES") of the respective hotels (the "HOTELS") set forth on Schedule 1 hereto.
- B. Pool Parent, the Tenants, and the Tenant Members are direct or indirect wholly owned Subsidiaries of CCC. The Tenants and the Tenant Members are direct or indirect wholly owned Subsidiaries of Pool Parent. The Tenant Members are the sole equity members of the respective Tenants set forth on Schedule 2 attached hereto.
- C. In order to induce the Landlords to enter into the Leases, CCC and Pool Parent (together, the "GUARANTORS") have executed and delivered a Pool Guarantee, of even date herewith (the "GUARANTEE"), pursuant to which the Guarantors have guaranteed certain obligations of Tenants under the Leases for the benefit of the respective Landlords, and the CCC Entities are executing and delivering this Agreement.

NOW, THEREFORE, in consideration of the covenants contained in this Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

## 1. CERTAIN DEFINITIONS

For all purposes of this Agreement, except as otherwise expressly provided or unless the context otherwise requires, (i) the terms defined in this Article shall have the meanings assigned to them in this Article and include the plural as well as the singular, (ii) all accounting terms not otherwise defined herein shall have the meanings assigned to them in accordance with GAAP, (iii) all references in this Agreement to designated "Articles," "Sections" and other subdivisions are to the designated Articles, Sections and other subdivisions of this Agreement, (iv) the term "including" shall have the same meaning as "including, without limitation," and (v) the words "herein," "hereof,"

"hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision.

"ACCOUNT COLLATERAL" has the meaning ascribed to such term in Section 2.1(a).

"ACCOUNTING PERIOD" means (i) for so long as CCC uses a fiscal year consisting of 52/53 weeks ending on the Friday closest to December 31 each year, each of the four (4) week accounting periods which are used in CCC's accounting system, except that an Accounting Period may occasionally contain five (5) weeks when necessary to conform CCC's accounting system to the calendar, and (ii) if CCC uses a fiscal year based on calendar quarters, each calendar month.

"AFFILIATE" means, with respect to any Person, any other Person directly or indirectly through one or more intermediaries controlling, controlled by or under common control with the subject Person. The term "control," as used in the immediately preceding sentence, means, with respect to a corporation, the right to exercise, directly or indirectly, fifty percent (50%) or more of the voting rights of any class of the shares of the controlled corporation, and, with respect to an entity that is not a corporation, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of the controlled entity. Notwithstanding the foregoing, Marriott International, Inc. and its Subsidiaries shall not be deemed to be Affiliates of CCC, Pool Parent, Landlord Representative, any Landlord or any Tenant.

"BUSINESS DAY" means any day other than Saturday, Sunday, or any other day on which banking institutions in the State of Maryland are authorized by law or executive action to close.

"CAPITALIZED INTEREST" shall mean interest that is capitalized and is not counted as interest expense in accordance with  ${\sf GAAP}.$ 

"CAPITALIZED LEASE OBLIGATIONS" of any Person shall mean all rental obligations which, under GAAP, are or will be required to be capitalized on the books of such Person, in each case taken at the amount thereof accounted for as indebtedness in accordance with such principles.

"CASH EXPENSES" means, with respect to any Tenant or Tenant Member, all expenses incurred by such Person and payable in cash, excluding (i) expenses of a Tenant paid by a manager on behalf of such Tenant pursuant to a hotel management agreement with respect to a hotel that is subject to a Lease, (ii) expenses relating to the performance of asset management services for the hotels subject to the Leases, including salaries or other employee costs of employees relating to the performance of asset management services, fees paid to independent contractors for provision of such services, or amounts paid to Affiliates for provision of such services, and (iii) expense reimbursements, overhead allocations and any other amount paid to an Affiliate of such Person.

"CCC" means Crestline Capital Corporation, and its successors and permitted assigns.

"CCC CONSOLIDATED FINANCIAL STATEMENTS" shall mean for any Fiscal Year or quarter thereof, consolidated statements of operations, retained earnings and cash flow for such period and for the period from the beginning of the respective Fiscal Year to the end of such period and the related balance sheet as at the end of such period, together with the notes to any such statement, all in such detail as may be required by the Securities and Exchange Commission with respect to filings made by Landlord Representative or Host REIT and, after the first full Fiscal Year of Pool Parent, setting forth in comparative form the corresponding figures for the corresponding period in the preceding Fiscal Year, and prepared in accordance with GAAP and (in the case of annual financial statements) audited by a "Big Five" firm of independent certified public accountants or such other firm of independent certified public accountants as may be reasonably approved by Landlord Representative. CCC Consolidated Financial Statements shall be prepared on the basis of the Fiscal Year or quarter thereof, as applicable.

"CCC LIMIT OF LIABILITY" means, with respect to the Guarantee or any Other Guarantee, the maximum liability of CCC under such Guarantee or Other Guarantee for obligations under the Leases or Other Leases, as the case may be, other than liability with respect to fraud or misapplication or misappropriation of funds or the existence of liens in violation of the Pooling Agreement or any Lease.

"CHANGE OF CONTROL" means the (i) acquisition (after a registered offering of shares or otherwise) by any Person, or two or more Persons acting in concert, in a single or series of transactions (whether or not related), of:

- (a) with respect to Pool Parent, beneficial ownership (within the meaning of Rule 13d-3 of the SEC) of, or rights (including conversion rights), options or warrants to acquire (whether absolute or conditional), all or any portion of the outstanding voting or economic interests in Pool Parent (other than voting rights held by an Independent Member of Tenant as set forth in the Limited Liability Company Operating Agreement of Tenant as in effect on the date hereof); or
- (b) with respect to CCC, beneficial ownership of, or rights (including conversion rights), options or warrants to acquire, thirty-five percent (35%) or more of the outstanding shares of voting stock of CCC;
- or (ii) the merger or consolidation of Pool Parent or CCC with or into any other Person if, in the case of CCC, immediately following the effectiveness of such merger or consolidation, more than thirty-five percent (35%) of the beneficial ownership of the outstanding voting shares of the surviving entity of such merger or consolidation (including for such purpose in both the numerator and the denominator, shares of voting stock issuable upon the exercise of then outstanding rights (including conversion rights), options or warrants) is held by one or more Persons who were not holders of the outstanding voting shares of CCC (or outstanding rights (including conversion rights), options or warrants to acquire voting shares of CCC) immediately prior to the effectiveness of such merger or consolidation; or
- (iii) any one or a series of related sales or conveyances to any Person of all or substantially all of the assets of Pool Parent or CCC; or

(iv) a change in the majority of the board of directors or similar managing body of CCC during the longer of (A) any twelve (12) month period and (B) the period during which two consecutive annual meetings of the equity holders of CCC at which directors or similar managers are elected have been held;

provided, however, that the acquisition by any one or more wholly owned direct or indirect Subsidiaries of CCC of beneficial ownership of, or rights, options or warrants to acquire, all or any portion of Pool Parent shall not be a "Change of Control" hereunder.

"COLLATERAL AGENT" means  $[\_\_\_]$ , or its successors and assigns.

"CONSOLIDATED COVERAGE RATIO" of any Person means, for any period, the ratio of (x) Consolidated EBITDA of such Person for such period to (y) the sum of (i) Consolidated Interest Expense of such Person for such period, and (ii) the amount of all scheduled principal amortization payments (other than balloon payments at maturity of such Indebtedness) on all Indebtedness of such Person and its Subsidiaries made during such period.

"CONSOLIDATED EBITDA" of any Person shall mean, for any period, the Consolidated Net Income of such Person for such period adjusted (A) to add thereto (to the extent deducted in determining Consolidated Net Income for such period), without duplication, the sum of the amounts for such period of (i) Consolidated Interest Expense, (ii) provisions for taxes based on income, (iii) depreciation and amortization expense (provided that the depreciation and amortization expense of a non-wholly owned Subsidiary shall be included only to the extent of the equity interest of such Person in such non-wholly owned Subsidiary), (iv) any other noncash items reducing Consolidated Net Income of such Person for such period (except to the extent that such noncash items relate to the write-off or write-down of any item included in Consolidated Net Income in a prior period) and (v) any cash receipts of such Person or a Subsidiary thereof during such period that represent items included in Consolidated Net Income of such Person for a prior period which were excluded from Consolidated EBITDA in such prior period by virtue of clause (B) of this definition, and (B) to subtract therefrom, without duplication, the sum of the amounts for such period of (i) all noncash items increasing Consolidated Net Income of such Person for such period and (ii) any cash expenditures of such Person and its Subsidiaries during such period to the extent that such cash expenditures (x) did not reduce Consolidated Net Income of such Person for such period and (y) were applied against reserves or accruals that constituted noncash items reducing Consolidated Net Income of such Person when reserved or accrued for in a prior period; provided, that Consolidated EBITDA shall be determined without giving effect to any extraordinary gains or losses (including any taxes attributable to any such extraordinary gains or losses) or gains or losses from sales of assets other than from sales of inventory in the ordinary course of

"CONSOLIDATED INTEREST EXPENSE" of any Person shall mean, for any period, the total consolidated cash interest expense of such Person and its Subsidiaries for such period (calculated without regard to any limitations on the payment thereof) plus, without duplication, that portion of the Capitalized Lease Obligations of such Person and its Subsidiaries representing the interest factor for such period, but excluding (i) any Capitalized Lease Obligations that relate to leases of furniture, fixtures and equipment,

(ii) the interest factor (if any) in such period under any leases of furniture, fixtures and equipment or any Lease or Other Lease with respect to the furniture, fixtures and equipment subject to such Lease or Other Lease, and (iii) any note issued by a tenant to a landlord pursuant to a Lease or Other Lease in connection with the purchase of working capital by such tenant. Notwithstanding anything to the contrary contained in the requirements of GAAP, the amount of Capitalized Interest incurred during any period shall be added as a component of Consolidated Interest Expense.

"CONSOLIDATED NET INCOME" of any Person shall mean, for any period, the net income (or loss) of such Person and its Subsidiaries for such period, determined on a consolidated basis; provided, that (i) net income (or loss) of (x) any other Person that is accounted for by such specified Person by the equity method of accounting or (y) any Subsidiary of such Person that is restricted from declaring or paying dividends or other distributions, directly or indirectly, by operation of the terms of its charter, any applicable agreement, instrument, judgment, decree, order, statute, rule or governmental regulation or otherwise shall (in each case) be included only to the extent of the amount of cash dividends or distributions paid to the specified Person or a wholly owned Subsidiary of such Person, (ii) the net income (or loss) of any other Person acquired by such specified Person or a Subsidiary of such Person in a pooling of interests transaction for any period prior to the date of such acquisition shall be excluded, (iii) to the extent Consolidated Net Income reflects amounts attributable to minority interests in Subsidiaries that are not wholly owned Subsidiaries of such Person, Consolidated Net Income shall be reduced by the amounts attributable to such minority interests, and (iv) any income, rental credit or expense associated with leases of Excess FF&E by any Tenant or any Other Tenant shall not be taken into account in determining Consolidated Net Income.

"DISTRIBUTABLE CASH" means, with respect to a Tenant as of any Rent Payment Date, an amount of cash equal to (i) the aggregate amount of cash in all lockbox accounts of such Tenant maintained pursuant to the applicable Lease as of such date, immediately after payment of all rent owing under the Leases to which such Tenant is party less (ii) the amount of Cash Expenses of such Tenant then

payable less (y) the amount of available cash reasonably required to be reserved

for payment of future Cash Expenses for which such Tenant reasonably believes, after taking into account existing cash reserves of such Tenant, that it will not have adequate cash after payment of rent in future Accounting Periods. "Distributable Cash" means, with respect to a Tenant Member as of any Rent Payment Date, an amount of cash equal to (x) the aggregate amount of cash distributed to such Tenant Member by the applicable Tenant on such date, less

(y) the amount of Cash Expenses of such Tenant Member then payable, less (z) the

amount of available cash reasonably required to be reserved for payment of future Cash Expenses for which such Tenant Member reasonably believes, after taking into account existing cash reserves of such Tenant Member, that it will not have adequate cash.

"DISTRIBUTION DATE" means, with respect to any Accounting Period, the first Business Day after the latest date on which payment of regularly scheduled Rent is due under any Lease.

"EVENT OF DEFAULT" has the meaning ascribed to such term in Section 6.1.

"EXCESS FF&E" means furniture, fixtures or equipment that any Tenant or Other Tenant is required to lease from a Person other than a Landlord or the landlord under an Other Lease, as the case may be, and receives a rental credit in respect of such lease, pursuant to the terms of any Lease or Other Lease.

"FINANCIAL OFFICER'S CERTIFICATE" shall mean a certificate of the chief financial officer, controller, chief accounting officer or treasurer of CCC, accompanying the financial statements required to be delivered by CCC pursuant to Section 7.2, in which such officer shall certify (a) that such statements have been properly prepared in accordance with GAAP and fairly present, in all material respects, the financial condition of CCC as of the dates thereof, and the results of its operations for the periods covered thereby (except that, in the case of financial statements delivered pursuant to clause Section 7.2(b)(i), the certificate shall state the extent to which such financial statements are not in accordance with GAAP), (b) that such officer has no knowledge of any Payment Default or Event of Default hereunder, and (c) the then-current Tangible Net Worth, Consolidated Coverage Ratio, Minimum Net Worth, and the CCC Limit of Liability under the Guarantee and each Other Guarantee.

"FISCAL YEAR" means the fiscal year of CCC.

"GAAP" means generally accepted accounting principles consistently applied.

"GUARANTEE" means that certain Pool Guarantee, of even date herewith, entered into by CCC and Pool Parent for the benefit of the Landlords, as it may be amended from time to time.

"GUARANTEED OBLIGATIONS" means any obligation of a Tenant or a Tenant Member that is guaranteed pursuant to the Guarantee.

"HOST REIT" shall mean Host Marriott Corporation, a Maryland corporation, or its successors or assigns.

"INDEBTEDNESS" shall mean all obligations, contingent or otherwise, which in accordance with GAAP should be reflected on the obligor's balance sheet as debt.

"INDEPENDENT DIRECTOR" means a Person who is not, and has not within the past five (5) years been, (i) an officer, director, employee, partner, member, manager, stockholder or beneficial-interest holder of Pool Parent; (ii) an officer, director, employee, partner, member, beneficial-interest holder, or stockholder of any "Affiliate" (as defined below) of Pool Parent; or (iii) a spouse, parent, sibling, or child of any Person described in (i) or (ii) of this section; provided, however, that a Person shall not be deemed to be a director or member of an Affiliate solely by reason of such Person being a director, manager, or member of a single-purpose entity that would otherwise be deemed to be an Affiliate because they are under common control. For the purpose of this definition alone, "Affiliate" means any Person (x) which owns beneficially, directly or indirectly, more than ten percent (10%) of the outstanding equity interest in Pool Parent or which is otherwise in control of Pool Parent; (y) of which more than ten percent (10%) of the outstanding voting securities are owned beneficially, directly or indirectly, by any Person described in clause (x) above; or (z) which is controlled by, or under common control with, any Person or entity

described in clause (x) above; the terms "control" and "controlled by" shall have the meanings assigned to them in Rule 405 under the Securities Act of 1933.

"INTANGIBLE ASSET" shall mean, with respect to any Person, a long-lived asset that is useful in the operations of such Person, that is not directly used in revenue generation and that is not held for sale, and is without physical qualities, including but not limited to patents, copyrights and goodwill, but excluding capitalized costs associated with the acquisition of brand names, franchises and trademarks, franchise agreements and management agreements.

"LANDLORD" means each Person named as the lessor under a Lease on Schedule

1 hereto, and each Person that becomes a "Landlord" hereunder pursuant to

--Section [\_\_].

"LANDLORD REPRESENTATIVE" means Host Marriott, L.P., a Delaware limited partnership, or its successor or permitted assigns.

"LEASE" means any lease listed on Schedule 1 hereto.

"LIEN" means any mortgage, security interest, pledge, collateral assignment, or other encumbrance, lien or charge of any kind (including, without limitation, any easements, covenants, conditions and restrictions), or any transfer of any property or assets for the purpose of subjecting the same to an encumbrance, lien or charge securing the payment of Indebtedness or performance of any other obligation in priority to payment of any Person's general creditors

"LOCKBOX NOTICE DATE" means, with respect to a Lockbox Trigger Event, the earliest date of a notice by a Landlord or the Landlord Representative to a Tenant and to CCC and the Pool Parent of the exercise by such Landlord of its right to implement the lockbox procedures under the Lease to which such Landlord and Tenant are parties.

"LOCKBOX PERIOD" means the period beginning no later than two weeks after a Lockbox Notice Date and continuing thereafter until the first day of the first full Accounting Period after (i) the Consolidated Coverage Ratio of CCC is at least 2.3 to 1.0 for each of three consecutive Accounting Periods, (ii) the Tangible Net Worth of CCC is at least equal to two hundred percent (200%) of the aggregate of the CCC Limits of Liability under the Guarantee and all Other Guarantees from time to time as of the end of each of three consecutive Accounting Periods, and (iii) no Payment Default under (A) any Lease or (B) any Other Lease with respect to which the CCC Limit of Liability under the applicable Other Guarantee is greater than zero and no "Event of Default" (as defined in the applicable Lease) under Section 12.1(c) of any Lease by reason of a breach of Section 7.1 or 21.6.3 of such Lease has been in effect as of three consecutive Distribution Dates; provided that a Lockbox Period shall not terminate by reason of an event described in clause (i), (ii), or (iii) if the CCC Limit of Liability under the Guarantee shall have been reduced to zero.

"LOCKBOX TRIGGER EVENT" means any of the following events:

(i) the Tangible Net Worth of CCC is less than two hundred percent (200%) of the amount of the aggregate of the CCC Limits of Liability under the Guarantee and all Other Guarantees from time to time as of the end of each of three consecutive Accounting Periods:

- (ii) the Consolidated Coverage Ratio of CCC is less than 2.3 to 1.0 for each of three consecutive Accounting Periods;
- (iii) the occurrence of a Payment Default under (A) any Lease or (B) any Other Lease with respect to which the CCC Limit of Liability under the applicable Other Guarantee is greater than zero, or an "Event of Default" (as defined in the applicable Lease) under Section 12.1(c) of any Lease by reason of a breach of Section 7.1 or 21.6.3 of such Lease; or
- (iv) the CCC Limit of Liability under the Guarantee shall have been reduced to zero.

"MINIMUM NET WORTH" means, with respect to the last day of any Accounting Period, a Tangible Net Worth (calculated on a book value basis), excluding amounts held in any lockbox or cash collateral account under any Lease, any Other Lease, this Agreement or any pooling and security agreement relating to Other Leases, immediately after the Distribution Date for such Accounting Period, equal to one hundred fifty percent (150%) of the aggregate of the CCC Limits of Liability under the Guarantee and the Other Guarantees from time to time; provided that the Minimum Net Worth shall be reduced by one hundred fifty percent (150%) of the total amount of cash held in any lockbox or cash collateral account under any Lease, any Other Lease, this Agreement or any pooling and security agreement relating to Other Leases, immediately after the Distribution Date for such Accounting Period.

"OTHER GUARANTEE" means a guarantee by CCC and any Subsidiary of CCC of the obligations under one or more Other Leases.

"OTHER LEASE" means any lease (other than a Lease) of a full service hotel under which Host Marriott, L.P. or a Subsidiary of Host Marriott, L.P. is the lessor and a Subsidiary of CCC is the lessee and that is the subject of a guarantee by CCC and a pooling and security agreement substantially similar to the Guarantee and this Agreement, as in effect on the date hereof.

"OTHER TENANT" means the lessee under an Other Lease.

"PAYMENT DEFAULT" means, with respect to a Lease or any Other Lease as to which the CCC Limit of Liability under the applicable Other Guarantee is greater than zero, a default by the applicable Tenant or Other Tenant in the payment of Rent after expiration of any applicable notice and cure periods and after giving effect to distributions from the Pool Cash Collateral Account or payment of such Rent by Pool Parent or CCC pursuant to the Guarantee; provided that no Payment Default will be deemed to have occurred with respect to any Other Lease under which the ownership interest of CCC (direct or indirect) in the lessee is less than 100%, so long as CCC and the applicable

Subsidiary of CCC that has entered into a guarantee of the obligations under such Other Lease shall have fulfilled their obligations under such guarantee./1/

"PERMITTED DEBT" has the meaning ascribed to such term in Section 4.1.

"PERMITTED INVESTMENTS" means any one or more of the following:

- (i) securities issued or directly and fully guaranteed or insured by the United States of America or any agency or instrumentality thereof (provided that the full faith and credit of the United
  - States of America is pledged in support thereof) having maturities of not more than one year from the date of acquisition (or such shorter period required by the terms of this Agreement);
- (ii) U.S. dollar denominated time deposits, certificates of deposit and bankers acceptances of any bank whose short-term commercial paper rating from Standard & Poor's Rating Services ("S&P") is at least A-2 or the equivalent thereof or from Moody's Investors Service, Inc. ("MOODY'S") is at least P-2 or the equivalent thereof (any such bank, an "APPROVED BANK"), in each case with maturities of not more than one year from the date of acquisition (or such shorter period required by the terms of this Agreement);
- (iii) commercial paper issued by any Approved Bank or by the parent company of any Approved Bank and commercial paper issued by, or guaranteed by, any industrial or financial company with a shortterm commercial paper rating of at least A-2 or the equivalent thereof by S&P or at least P-2 or the equivalent thereof by Moody's, or guaranteed by any industrial company with a long term unsecured debt rating of at least A or A-2, or the equivalent of each thereof, from S&P or Moody's, as the case may be, and in each case maturing within one year after the date of acquisition (or such shorter period required by the terms of this Agreement);
- (iv) marketable direct obligations issued by the District of Columbia or any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof maturing within one year from the date of acquisition thereof (or such shorter period required by the terms of this Agreement) and, at the time of acquisition, having one of the two highest ratings obtainable from either S&P or Moody's; and
- (v) investments in money market funds substantially all the assets of which are comprised of securities of the types described in clauses (i) through (iv) above.

<sup>/1/</sup> The liability under the guarantee of such a lease will be limited to the percentage ownership interest of CCC in the lessee.

"PERMITTED LIEN" has the meaning ascribed to such term in Section 4.2.

"PERSON" means any individual, partnership, limited liability company, joint venture, firm, corporation, association, trust or other enterprise or any government or political subdivision or any agency, department or instrumentality thereof.

"POOL CASH COLLATERAL ACCOUNT" has the meaning ascribed to such term in Section 2.2(a).

"RENT" means, with respect to any Lease or Other Lease, minimum rent, percentage rent or additional charges owing under such Lease or Other Lease.

"RENT PAYMENT DATE" means, with respect to any Lease, the date on which a regularly scheduled payment of Rent is due.

"RENT SHORTFALL" means, with respect to any Lease, the amount of any deficit in payment of Rent on any Rent Payment Date, prior to giving effect to any distribution from the Pool Cash Collateral Account or any payment of the amount owing by Pool Parent or CCC pursuant the Guarantee.

"REQUIRED RESERVE AMOUNT" means, as of any Distribution Date, an amount equal to the greater of (i) the CCC Limit of Liability under the Guarantee as of such Distribution Date, after giving effect to all payments on such date or (ii) 0.5% of the aggregate Rent (excluding Additional Charges (as defined in each Lease) and without deduction of FF&E Adjustments under any Lease) payable under all Leases for the preceding Fiscal Year (with the amount in this clause (ii) being for Fiscal Year 1999 0.5% of the aggregate Rent (excluding Additional Charges (as defined in each Lease) and without deduction of FF&E Adjustments under any Lease) payable under all Leases for Fiscal Year 1999).

"SECONDARY LOCKBOX PERIOD" means the period beginning upon the occurrence of a Secondary Lockbox Trigger Event and continuing thereafter until the first day of the first full Accounting Period after (i) the Tangible Net Worth of CCC is at least equal to the Minimum Net Worth as of the end of each of three consecutive Accounting Periods, (ii) the Consolidated Coverage Ratio of CCC is at least 1.8 to 1.0 for each of three consecutive Accounting Periods, and (iii) no Payment Default under (A) any Lease or (B) any Other Lease with respect to which the CCC Limit of Liability under the applicable Other Guarantee is greater than zero and no "Event of Default" (as defined in the applicable Lease) under Section 12.1(c) of any Lease by reason of a breach of Section 7.1 or 21.6.3 of such Lease has been in effect as of three consecutive Distribution Dates; provided that a Secondary Lockbox Period shall not terminate by reason of an event described in clause (i), (ii), or (iii) if the CCC Limit of Liability under the Guarantee shall have been reduced to zero.

"SECONDARY LOCKBOX TRIGGER EVENT" means any of the following events:

(i) the Tangible Net Worth of CCC is less than the Minimum Net Worth as of the end of each of three consecutive Accounting Periods;

- (ii) the Consolidated Coverage Ratio of CCC is less than 1.8 to 1.0 for each of three consecutive Accounting Periods;
- (iii) the occurrence of a Payment Default under (A) any Lease or (B) any Other Lease with respect to which the CCC Limit of Liability under the applicable Other Guarantee is greater than zero, or "Event of Default" (as defined in the applicable Lease) under Section 12.1(c) of any Lease by reason of a breach of Section 7.1 or 21.6.3 of such Lease; or
- (iv) the CCC Limit of Liability under the Guarantee shall have been reduced to zero.

"SUBSIDIARY" shall mean, as to any Person, (i) any corporation more than fifty percent (50%) of whose stock of any class or classes having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at the time stock of any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time owned by such Person or one or more Subsidiaries of such Person and (ii) any partnership, limited liability company, association, joint venture or other entity in which such Person or one or more Subsidiaries of such Person has more than a fifty percent (50%) equity interest at the time.

"TANGIBLE NET WORTH" means, with respect to any Person on any date, without duplication, (i) the sum of the shareholders' equity of such Person on a consolidated basis in accordance with GAAP minus (ii) the sum of all Intangible Assets (net of accumulated amortization) of such Person and its Subsidiaries, each as shown on the consolidated balance sheets of such Person.

"TENANT" means each Person named as the lessee under a Lease on Schedule 1
-----hereto, and each Person that becomes a "Tenant" hereunder pursuant to Section
3.1.

"TENANT MEMBER" means each wholly owned Subsidiary of Pool Parent named as the sole equity member of a Tenant on Schedule 2 hereto, and each Person that

becomes a "Tenant Member" hereunder pursuant to Section 3.1.

#### 2. CASH COLLATERAL ARRANGEMENTS

- 2.1 SECURITY AGREEMENT RELATING TO ACCOUNT COLLATERAL
- (a) Pledge of Account Collateral. To secure the full and punctual payment and performance when due of all of the Guaranteed Obligations, Pool Parent hereby pledges, and grants to the Collateral Agent for the benefit of the Landlords, a first and continuing security interest in and to, the following property, whether now owned or existing or hereafter acquired or arising and regardless of where located (such property, collectively, the "ACCOUNT COLLATERAL"):

- (i) all of Pool Parent's right, title and interest in and to any dividends, distributions or other amounts paid with respect to any equity interests in a Tenant or Tenant Member;
- (ii) all of Pool Parent's right, title and interest in and to the Pool Cash Collateral Account; and all moneys and Permitted Investments, if any, from time to time deposited or held in the aforementioned account; and
- (iii) to the extent not covered by clause (i) or (ii) above, all cash, instruments and other property received, receivable or otherwise distributed in respect of or in exchange for any or all of the foregoing.
- (b) Financing Statements; Further Assurances. Upon the request of Landlord

Representative, Pool Parent will execute and deliver to the Collateral Agent for filing a financing statement or statements in connection with the Account Collateral in the form required to properly perfect the Collateral Agent's security interest for the benefit of the Landlords in the Account Collateral to the extent that it may be perfected by such a filing. From time to time, at the expense of Pool Parent, Pool Parent shall promptly execute and deliver all further instruments prepared by the Landlord Representative, and take all further action, that the Collateral Agent and the Landlord Representative may reasonably request, in order to perfect and protect the pledge and security interest granted or purported to be granted hereby, or to enable the Collateral Agent to exercise and enforce the Collateral Agent's rights and remedies hereunder with respect to any Account Collateral. Neither the Collateral Agent nor the Pool Parent shall be responsible for the determination of the financing statements and other instruments necessary to perfect such security interest or for the filing of such financing statements and other instruments at the locations necessary to perfect and continue such security interest.

- (c) Transfers and Other Liens. Pool Parent shall not sell or otherwise dispose of any of the Account Collateral other than pursuant to the terms hereof, or create or permit to exist any Lien upon or with respect to all or any of the Account Collateral, except for the Lien granted to the Collateral Agent for the benefit of the Landlords under this Agreement.
- (d) Continuing Security Interest; Termination. This Section 2.1 shall create a continuing pledge of and security interest in the Account Collateral and shall remain in full force and effect so long as the Guarantee remains outstanding. Upon termination or expiration of the Guarantee and payment of all Guaranteed Obligations, subject to the limits specified in the Guarantee, Pool Parent shall be entitled to the return, upon its request and at its expense, of such of the Account Collateral as shall not have been sold or otherwise applied pursuant to the terms hereof, and, upon written notification by the Landlord Representative to the Collateral Agent that the Guarantee has terminated or expired and all outstanding Guaranteed Obligations have been paid in full, the Collateral Agent shall release any funds then held by it in the Pool Cash Collateral Account and execute such instruments and documents as may be reasonably requested by Pool Parent to evidence such termination and the release of the pledge and lien hereof.

- 2.2 ESTABLISHMENT, MAINTENANCE AND FUNDING OF POOL CASH COLLATERAL ACCOUNT
- (a) At any time after a Lockbox Notice Date, or in the discretion of Landlord Representative prior thereto, Landlord Representative shall establish and thereafter maintain with the Collateral Agent a master collection account (the "POOL CASH COLLATERAL ACCOUNT") in the name of Pool Parent and the Collateral Agent for the benefit of the Landlords as secured party.
- (b) During any Lockbox Period, Pool Parent shall cause to be promptly deposited (and in any event, Pool Parent will use its best efforts to cause such to occur within 24 hours) in the Pool Cash Collateral Account all revenues and other income that it receives from any source whatsoever. Pool Parent shall not permit any other funds so to be deposited or commingled in the Pool Cash Collateral Account, except as required pursuant to paragraph (d) below.
- (c) During any Lockbox Period, each Tenant shall distribute to the Pool Parent (or, if the equity member of such Tenant is a Tenant Member, to the applicable Tenant Member and such Tenant Member shall distribute to the Pool Parent) all Distributable Cash on each Rent Payment Date under each Lease to which such Tenant is a party and Pool Parent shall deposit such funds, immediately upon receipt, into the Pool Cash Collateral Account.
- (d) During any Lockbox Period, CCC and Pool Parent shall deposit all payments made by either of them under the Guarantee directly into the Pool Cash Collateral Account, and the deposit of all amounts owing under the Guarantee into the Pool Cash Collateral Account shall be deemed to constitute payment of their obligations thereunder.
- (e) Pool Parent shall direct the Collateral Agent in writing to invest and reinvest any balance in the Pool Cash Collateral Account from time to time in Permitted Investments; provided that (i) the maturities of the Permitted Investments on deposit in the Pool Cash Collateral Account shall be such that the amounts required to be withdrawn therefrom pursuant to Section 2.3 shall be available in cash as of the close of business on the Business Day immediately preceding the date on which such funds are required to be withdrawn, (ii) all such Permitted Investments other than during the continuance of an Event of Default shall be held in the name of Pool Parent and, during the continuance of an Event of Default, shall be held in the name of the Collateral Agent for the benefit of the Landlords, and (iii) no Permitted Investment shall be directed to be made unless the Collateral Agent for the benefit of the Landlords shall retain a perfected first priority Lien in such Permitted Investment and all filings and other actions necessary to ensure the validity, perfection, and priority of such Lien have been taken. Neither the Collateral Agent, the Landlord Representative nor any Landlord shall have any liability for any loss in investments of funds in the Pool Cash Collateral Account that are invested in Permitted Investments (unless, in the case of the Collateral Agent, such funds are invested contrary to Pool Parent's or the Landlord Representative's written direction, as applicable) and no such loss shall affect Pool Parent's obligation to fund, or liability for funding, the Pool Cash Collateral Account. Pool Parent shall include all earnings on the Pool Cash Collateral Account as income of Pool Parent for federal and applicable state tax purposes. In the

absence of any written direction, the Collateral Agent shall invest the balance of the Pool Cash Collateral Account in the Permitted Investment described in clause (v) of the definition thereof. Notwithstanding the foregoing, the Collateral Agent shall be permitted to keep uninvested funds in the Pool Cash Collateral Account from the day prior to each Rent Payment Date until such Rent Payment Date.

- (f) Pool Parent shall be the owner of the Pool Cash Collateral Account for all purposes, subject to the security interest therein created by this Agreement and to the application of all amounts held therein for the account of the Pool Parent as set forth in this Agreement.
  - 2.3 DISBURSEMENTS FROM POOL CASH COLLATERAL ACCOUNT.
  - (a) Disbursements Prior to Secondary Lockbox Event. On each Distribution

Date during any Lockbox Period, and provided that no Secondary Lockbox Trigger Event shall have occurred and be continuing (in which event the provisions of paragraphs (b) or (c) below shall apply), upon receipt of a certificate signed by Pool Parent and the Landlord Representative stating that all Rent owing for the immediately preceding Accounting Period and all prior Accounting Periods under each Lease and under each Other Lease as to which the CCC Limit of Liability under the applicable Other Guarantee is greater than zero has been paid in full, the Collateral Agent shall disburse all funds held in the Pool Cash Collateral Account to Pool Parent. If, on any such Distribution Date, a Rent Shortfall exists under any Lease for the immediately preceding Accounting Period or any prior Accounting Period, upon receipt of a certificate from the Landlord Representative setting forth the Leases as to which there is a Rent Shortfall and the amount of each such Rent Shortfall, the Collateral Agent shall pay to the applicable Landlord an amount equal to such Rent Shortfall; if the funds available in the Pool Cash Collateral Account are not sufficient to pay all Rent Shortfalls, then the Collateral Agent shall make payments with respect to each Rent Shortfall to the extent of available funds ratably based on the ratio of the amount of each Rent Shortfall to the aggregate amount of Rent Shortfalls under all Leases. Any funds remaining in the Pool Cash Collateral Account after satisfying any Rent Shortfall shall be distributed promptly upon receipt of the certificate referred to in the first sentence of this paragraph.

(b) Disbursements During Secondary Lockbox Period. On each Distribution

Date during a Secondary Lockbox Period, the Collateral Agent shall disburse from
the Pool Cash Collateral Account, to the extent available therein, funds in
amounts sufficient for the following purposes, in the following order of
priority:

First, to the Landlord under each Lease with respect to which there is a Rent Shortfall for the immediately preceding Accounting Period or any prior Accounting Period, an amount equal to such Rent Shortfall; if the funds available in the Pool Cash Collateral Account are not sufficient to pay all Rent Shortfalls, then the Collateral Agent shall make payments with respect to each Rent Shortfall to the extent of available funds ratably based on the ratio of the amount of each Rent Shortfall to the aggregate amount of Rent Shortfalls under all Leases;

Second, subject to the provisions of Section 2.3(c), to Pool Parent, an amount equal to (x) 3,077 times the number of Leases as to which there is no

Payment Default after application of funds disbursed pursuant to clause First above times the number of full Accounting Periods that have then elapsed since the beginning of the applicable Secondary Lockbox Period, less (y) the aggregate amount of Cash Expenses of all Tenants deducted in determining Distributable Cash for all Accounting Periods since the beginning of the applicable Secondary Lockbox Period, less (z) the cumulative amount of all distributions under this clause Second during such Secondary Lockbox Period; and

Third, subject to the provisions of Section 2.3(c), to Pool Parent any remaining funds in excess of the Required Reserve Amount, so long as there is no Payment Default under any Lease or under any Other Lease as to which the CCC Limit of Liability under the applicable Other Guarantee is greater than zero and no Event of Default hereunder shall have occurred and be continuing; if the remaining funds are less than or equal to the Required Reserve Amount, the Collateral Agent shall retain all funds remaining in the Pool Cash Collateral Account.

- (c) Disbursements after Default. If a Payment Default under any Lease or under any Other Lease as to which the CCC Limit of Liability under the applicable Other Guarantee is greater than zero or an Event of Default hereunder shall have occurred and be continuing, then the Collateral Agent shall not disburse any funds from the Pool Cash Collateral Account pursuant to paragraph Second or Third of Section 2.3(b).
- (d) Disbursements at End of Lockbox Period. Promptly after the termination of any Lockbox Period, the Collateral Agent shall disburse to the Pool Parent any funds then held in the Pool Cash Collateral Account that are not required to cure any existing Rent Shortfalls.
- (e) Reliance on Certificates. During a Secondary Lockbox Period, the
  Collateral Agent shall make distributions from the Pool Cash Collateral Account
  on the basis of, and it shall be entitled to rely on, certificates furnished by
  Landlord Representative.

## 3. ADDITION OF LEASES

## 3.1 ADDITION OF LEASES, TENANTS, OR TENANT MEMBERS

From time to time, the Landlord Representative and the Pool Parent shall be entitled to cause an additional lease of a lodging facility under which Host Marriott, L.P. or a Subsidiary of Host Marriott, L.P. is the lessor and a Subsidiary of CCC is the lessee to become subject to this Agreement, without the consent of the Landlords or Tenants, by amending Schedule 1 hereto to add such

lease, provided that the obligations under such lease are guaranteed pursuant to the Guarantee and the lessee and lessor under such lease agree in writing to be bound by the respective obligations hereunder of a "Tenant" or "Landlord," as the case may be. Upon such amendment of Schedule 1 and such agreement by such

the lessor and lessee under such lease, such lease shall be a "Lease" and the lessee and lessor thereunder shall be a "Tenant" or "Landlord," as the case may be, under this Agreement.

## 4. NEGATIVE COVENANTS APPLICABLE TO POOL PARENT

## 4.1 INDEBTEDNESS

Pool Parent shall not (i) incur, create or assume any Indebtedness; or (ii) guarantee or have any consensual contingent obligation for the obligations of any other Person, other than pursuant to the Guarantee or this Agreement; provided that, so long as no Event of Default has occurred and is continuing, Pool Parent may incur, create or assume any Permitted Debt (as defined below). As used herein, "PERMITTED DEBT" shall mean:

- (i) if no Event of Default has occurred and is continuing, unsecured Indebtedness owing to CCC, with respect to which CCC shall have agreed in writing, in form and substance satisfactory to Landlord Representative, that payment of such Indebtedness shall be subordinated in all respects to performance of Pool Parent's obligations under this Agreement and that no remedies may be exercised with respect to enforcement or collection of such Indebtedness until such time as this Agreement shall have terminated and all obligations owed by Pool Parent hereunder shall have been discharged in full; and
- (ii) if no Event of Default has occurred and is continuing, Indebtedness solely in respect of surety and appeal bonds, performance bonds and other obligations of a like nature (to the extent that such incurrence does not result in the incurrence of any obligation to repay any obligation relating to borrowed money of others), all in the ordinary course of business in accordance with customary industry practices.

## 4.2 LIENS

Pool Parent shall not create or suffer to exist any Liens on its assets other than Permitted Liens. As used herein, "PERMITTED LIEN" shall mean the Liens created by this Agreement and any other Lien securing obligations under the Guarantee.

## 4.3 INVESTMENTS

Pool Parent shall not make any investment in, or advance, loan or extend credit to, make any capital contribution to, or acquire or own stock or other equity interests in, any other Person other than advances, loans or extensions of credit to, capital contributions to, or equity interests in Tenants and Tenant Members.

## 4.4 TRANSFERS OF INTERESTS IN TENANTS AND TENANT MEMBERS

Pool Parent shall not (i) transfer or otherwise dispose of any equity interest in any Tenant other than to a wholly owned Subsidiary of Pool Parent that agrees in writing  $\frac{1}{2} \left( \frac{1}{2} \right) = \frac{1}{2} \left( \frac{1}{2} \right) \left( \frac{1$ 

to be bound by the obligations of a Tenant Member hereunder or (ii) transfer or otherwise dispose of any equity interest in any Tenant Member.

## 4.5 AMENDMENT OF ORGANIZATIONAL DOCUMENTS

CCC and Pool Parent shall not modify Sections \_\_\_\_ [SPECIFY SECTIONS RELATING TO SPE PROVISIONS] of the certificate of incorporation of Pool Parent and shall not modify, or permit or fail to prevent any modification of, the limited liability company operating agreement, articles of incorporation, partnership agreement or similar organizational document of any Tenant in violation of the applicable Lease or of any Tenant Member without the consent of the Landlord Representative (such consent not to be unreasonably withheld). Pool Parent shall not (i) issue any capital stock to any Person other than CCC and shall not permit any Tenant or Tenant Member to admit any new member or issue any capital stock to any Person other than Pool Parent or (ii) permit or suffer to occur any change in the identity of the owner of its capital stock or the capital stock or membership interests of any Tenant or Tenant Member without the prior approval of the Landlord Representative (such approval not to be unreasonably withheld).

## 4.6 MISAPPLICATION OF FUNDS

Pool Parent shall not make any distribution to or for the benefit of any of its equity owners while a Event of Default is pending; or fail to remit or cause to be remitted to the Pool Cash Collateral Account all amounts required so to be remitted pursuant to this Agreement.

#### 4.7 PLACE OF BUSINESS

Pool Parent shall not change its chief executive office or its principal places of business without giving the Landlord Representative at least 20 days' prior written notice thereof and promptly providing the Landlord Representative such information as the Landlord Representative may reasonably request in connection therewith.

## 5. SPE COVENANTS

## 5.1 BUSINESS OF POOL PARENT

Pool Parent shall not engage in any business other than the ownership of equity interests in the Tenants and Tenant Members.

## 5.2 SEPARATE EXISTENCE

Pool Parent shall (i) maintain its books and records and bank accounts separate from any other Person (except that, for accounting and reporting purposes, Pool Parent may be included in the CCC Consolidated Financial Statements in accordance with GAAP);

(ii) maintain an arm's length relationship with its stockholders, Affiliates and any other party furnishing services to it; (iii) maintain its books, records, resolutions and agreements as official records; (iv) conduct its business in its own name and through its own authorized officers and agents; (v) maintain its financial statements, accounting records and other corporate documents separate from those of any other Person (except for inclusion in the CCC Consolidated Financial Statements); (vi) pay its own liabilities out of its own funds and other assets, including funds contributed to its capital by its equity holders, and all such capital contributions shall be reflected properly in its books and records; (vii) observe all corporate formalities, as applicable, necessary to maintain its identity as an entity separate and distinct from its stockholders, CCC, and all other Affiliates; (viii) participate in the fair and reasonable allocation of any and all overhead expenses and other common expenses for facilities, goods or services provided to multiple entities; (ix) use its own stationery, invoices and checks (except when acting in a representative capacity); (x) hold and identify itself as a separate and distinct entity under its own name and not as a division or part of any other Person (except for inclusion in the CCC Consolidated Financial Statements); and (xi) hold its assets in its own name.

## 5.3 INDEPENDENT DIRECTOR

Pool Parent shall have an Independent Director at all times, or if the Independent Director has resigned, Pool Parent shall not take any action which may not be taken pursuant to the organizational documents of Pool Parent without the consent of the Independent Director until such time as a replacement Independent Director has been appointed to the board of directors of Pool Parent.

## 5.4 FUNDAMENTAL CHANGES

Pool Parent shall not merge or consolidate with any other Person, liquidate or dissolve, or permit any Tenant or Tenant Member to merge or consolidate with any other Person or liquidate or dissolve.

## 6. EVENTS OF DEFAULT; REMEDIES

## 6.1 EVENTS OF DEFAULT

The occurrence of any one or more of the following events shall be an "EVENT OF DEFAULT" hereunder:

- (a) A failure by CCC or Pool Parent to make any payment under the Guarantee when  $\mbox{due}$ ;
- (b) Pool Parent or any Tenant is in default of any of its obligations under Article 2 hereof and such default is not cured within (i), with respect to their respective obligations under Section 2.2, 5 Business Days after the occurrence thereof, or (ii), with

respect to any other obligations under Article 2, 10 Business Days after receipt of notice of such default;

- (c) Pool Parent is in default of any of its obligations under Article 4 or 5 hereof;
  - (d) Any Change of Control occurs;
- (e) Any obligation of Pool Parent in respect of any Indebtedness in a principal amount in excess of \$1,000,000 for money borrowed or for the deferred purchase price of any material property or services, is declared to be, or as a result of acceleration becomes, due and payable prior to the stated maturity thereof;
- (f) Pool Parent or CCC is generally not paying its debts as they become due, or Pool Parent or CCC makes a general assignment for the benefit of creditors;
- (g) A petition is filed by or against Pool Parent or CCC under the federal bankruptcy laws, or any other proceeding is instituted by or against Pool Parent or CCC seeking to adjudicate it a bankrupt or insolvent, or seeking liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors, or seeking the entry of an order for relief or the appointment of a receiver, trustee, custodian or other similar official for Pool Parent or CCC or for any substantial part of the property of Pool Parent or CCC, and, in the case of any involuntary petition filed or proceeding instituted against Pool Parent or CCC only, such proceeding is not dismissed within sixty (60) days after institution thereof, or Pool Parent or CCC takes any action to authorize or effect any of the actions set forth above in this paragraph;
- (h) Pool Parent or CCC causes or institutes any proceeding for its dissolution or termination; or
- (i) CCC fails to maintain for each of three (3) consecutive Accounting Periods either (i) a Tangible Net Worth equal to or greater than (x) the CCC Limit of Liability under the Guarantee from time to time less (y) amounts held in any lockbox or cash collateral account under any Lease, any Other Lease, this Agreement or any pooling and security agreement relating to Other Leases, immediately after the Distribution Date for the preceding Accounting Period, or (ii) a Consolidated Coverage Ratio of at least 1.4 to 1.0.

## 6.2 REMEDIES IN CASE OF AN EVENT OF DEFAULT

(a) If an Event of Default shall have occurred and be continuing, in addition to such other rights and remedies as may be available at law or in equity, the Collateral Agent shall be entitled to exercise all of the rights, powers and remedies (whether vested in it by this Agreement, the Guarantee or by law) for the protection and enforcement of its rights in respect of the Account Collateral, including, without limitation, all the rights and remedies of a secured party upon default under the Uniform Commercial Code of the State of Maryland, and the Collateral Agent shall be entitled, without limitation, to exercise any

or all of the following rights, which Pool Parent hereby agrees to be commercially reasonable:

- (i) to receive all amounts payable in respect of the Account Collateral otherwise payable to Pool Parent; and
- (ii) to transfer all or any part of the Account Collateral into the Collateral Agent's name or the name of its nominee or nominees.
- (b) Upon the occurrence and during the continuance of an Event of Default, the Collateral Agent may, upon prior written notice from the Landlord Representative and without notice or liability to Pool Parent, apply any or all amounts in the Pool Cash Collateral Account for any of the following purposes relating to the Guaranteed Obligations, in the following order:
  - (i) reimbursement of the Landlord Representative, the Collateral Agent and the Landlords for all losses and reasonable expenses (including without limitation reasonable legal fees) actually suffered or incurred by such Persons as a result of such Event of Default;
  - (ii) payment of any reasonable amount expended in exercising rights and remedies available to the Collateral Agent, the Landlord Representative or the Landlords at law or in equity or under this Agreement or under the Guarantee; and
  - (iii) payment of any Guaranteed Obligations then due and payable (and if the amount available in the Pool Cash Collateral Account is insufficient to pay all such amounts in full, then the amounts then due and payable to each Landlord in respect of Guaranteed Obligations will be paid pro rata based on the respective amounts of Guaranteed Obligations then due to each Landlord).
- (c) Every right and remedy granted to the Collateral Agent under this Agreement or by law may be exercised by the Collateral Agent at any time and from time to time as and when provided hereby, and as often as the Collateral Agent may deem it expedient. Any and all of the Collateral Agent's rights with respect to the pledge of and security interest in the Account Collateral granted hereunder shall continue unimpaired, and Pool Parent shall be and remain obligated in accordance with the terms hereof, notwithstanding (i) any proceeding of Pool Parent, CCC, or any Tenant under the United States Bankruptcy Code or any bankruptcy, insolvency or reorganization laws or statutes of any state, (ii) the release or substitution of Account Collateral at any time, or of any rights or interests therein pursuant to this Agreement, or (iii) any delay, extension of time, renewal, compromise or other indulgence granted by the Collateral Agent at the written direction of the Landlord Representative in the event of any Event of Default with respect to the Account Collateral or otherwise hereunder.
- (d) Collateral Agent Appointed Attorney-In-Fact. Pool Parent hereby
  irrevocably constitutes and appoints the Collateral Agent as Pool Parent's true
  and lawful attorney-in-fact, with full power of substitution, at any time after
  the occurrence and

during the continuation of an Event of Default, to execute, acknowledge and deliver any instruments and to exercise and enforce every right, power, remedy, option and privilege of Pool Parent with respect to the Account Collateral, and do in the name, place and stead of Pool Parent, all such acts, things and deeds for and on behalf of and in the name of Pool Parent with respect to the Account Collateral, which Pool Parent could or might do in the absence of such Event of Default or which the Landlord Representative may deem necessary or desirable to more fully vest in the Collateral Agent the rights and remedies provided for herein with respect to the Account Collateral and to accomplish the purposes of this Agreement. For so long as the Guarantee remains outstanding, the foregoing powers of attorney are irrevocable and coupled with an interest.

## 6.3 REMEDIES, ETC. CUMULATIVE

Each right, power and remedy of the Collateral Agent provided for in this Agreement or the Guarantee now or hereafter existing at law or in equity or by statute shall be cumulative and concurrent and shall be in addition to every other such right, power or remedy. The exercise or beginning of the exercise by the Collateral Agent of any one or more of the rights, powers or remedies provided for in this Agreement or the Guarantee or now or hereafter existing at law or in equity or by statute or otherwise shall not preclude the simultaneous or later exercise by the Collateral Agent of all such other rights, powers or remedies, and no failure or delay on the part of the Collateral Agent to exercise any such right, power or remedy shall operate as a waiver thereof.

## 6.4 INDEMNITY

Pool Parent agrees (i) to indemnify and hold harmless the Collateral Agent from and against any and all claims, demands, losses, judgments and liabilities (including liabilities for penalties) of whatsoever kind or nature arising out of the performance of its duties hereunder, and (ii) to reimburse the Collateral Agent, the Landlord Representative and the Landlords for all reasonable costs and expenses, including reasonable attorneys' fees, growing out of or resulting from the exercise by the Collateral Agent of any right or remedy granted to it hereunder or under the Guarantee except, with respect to clauses (i) and (ii) above, for those claims, demands, losses, judgments and liabilities arising from the Collateral Agent's gross negligence, bad faith, fraud or willful misconduct. In no event shall the Collateral Agent be liable, in the absence of gross negligence, bad faith, fraud or willful misconduct on its part, for any matter or thing in connection with this Agreement other than to account for moneys or other property actually received by it in accordance with the terms hereof or thereof. If and to the extent that the obligations of Pool Parent under this Section are unenforceable for any reason, Pool Parent hereby agrees to make the maximum contribution to the payment and satisfaction of such obligations which is permissible under applicable law.

## 7. MISCELLANEOUS PROVISIONS

## 7.1 TERMINATION

This Agreement shall terminate on the date that the obligations of the Pool Parent under the Guarantee terminate (other than with respect to Special Guarantee Obligations (as defined in the Guarantee)), as provided in Section 10(a) of the Guarantee. Upon termination of this Agreement, any funds remaining in the Pool Cash Collateral Account, after application in accordance with this Agreement, including pursuant to the provisions of Section 6.2(b), and satisfaction of any Guaranteed Obligations to the extent of such funds, shall be released by the Collateral Agent to the Pool Parent.

## 7.2 FINANCIAL REPORTING

- (a) On each Distribution Date, CCC shall furnish to Landlord Representative a certificate of the chief financial officer, controller, chief accounting officer or treasurer of CCC setting forth the Tangible Net Worth, Consolidated Coverage Ratio, and Minimum Net Worth of CCC as of the last day of the immediately preceding Accounting Period and the CCC Limit of Liability under the Guarantee and each Other Guarantee as of such date.
- (b) CCC shall furnish to Landlord Representative, (i) within 40 days after the end of each of the first three quarters of a Fiscal Year, the most recent unaudited CCC Consolidated Financial Statements accompanied by a Financial Officer's Certificate and, (ii) within 80 days after the end of each Fiscal Year, the CCC Consolidated Financial Statements for such Fiscal Year, certified by Arthur Andersen LLP or another independent certified public accountant reasonably satisfactory to Landlord and accompanied by a Financial Officer's Certificate.
- (c) Except as otherwise provided herein, the cost for one consolidated audit with respect to CCC per Fiscal Year and one quarterly unaudited statement with respect to CCC for each of the first three fiscal quarters in the Fiscal Year as may be required by the Securities and Exchange Commission for regular reporting requirements of Landlord Representative or Host REIT under the Securities and Exchange Act of 1934 shall be borne by Pool Parent. Except as otherwise provided herein, the cost of any other audits or financial statements requested by Landlord Representative or any Landlord shall be borne by Landlord Representative and the Landlords. CCC Consolidated Financial Statements shall be accepted by Landlord Representative if prepared on a combined basis along with all of the other hotels leased from Landlord Representative or any of its Affiliates by a wholly owned Subsidiary of Pool Parent or CCC.
- (d) Any financial report or certificate delivered hereunder by CCC in a timely manner that includes the information required to be furnished under a Lease shall be deemed to satisfy the obligation to provide such report or certificate under such Lease.
- (e) If Landlord Representative does not object to the calculation of the CCC Limit of Liability under the Guarantee or any Other Guarantee within 30 days after receipt of the Financial Officer's Certificate accompanying any audited annual CCC

Consolidated Financial Statements, Landlord Representative will not be permitted to object to such calculation at any time thereafter.

## 7.3 NOTICES

- (a) Any and all notices, demands, consents, approvals, offers, elections and other communications required or permitted under this Agreement shall be deemed adequately given if in writing and the same shall be delivered either in hand, by telecopier with computer generated acknowledgment of receipt, or by mail or Federal Express or similar expedited commercial carrier, addressed to the recipient of the notice, postpaid and certified with return receipt requested (if by mail), or with all freight charges prepaid (if by Federal Express or similar carrier).
- (b) All notices required or permitted to be sent hereunder shall be deemed to have been given for all purposes of this Agreement upon the date of acknowledged receipt, in the case of a notice by telecopier, and, in all other cases, upon the date of receipt or refusal, except that whenever under this Lease a notice is either received on a day which is not a Business Day or is required to be delivered on or before a specific day which is not a Business Day, the day of receipt or required delivery shall automatically be extended to the next Business Day.
  - (c) All such notices shall be addressed:

if to any Landlord, addressed to such Landlord at:

10400 Fernwood Road Bethesda, Maryland 20817 Attn:

with a copy to:

Host Marriott, L.P. 10400 Fernwood Road Bethesda, Maryland 20817 Attn: General Counsel

If to Landlord Representative, to:

Host Marriott, L.P.

10400 Fernwood Road

Bethesda, Maryland 20817 Attn: General Counsel

If to Pool Parent or any Tenant, addressed to Pool Parent or such Tenant, as the case may be, at:

10400 Fernwood Road Bethesda, Maryland 20817 Attn: General Counsel

## with a copy to:

Crestline Capital Corporation 10400 Fernwood Road Bethesda, Maryland 20817 Attn: Chief Financial Officer

(d) By notice given as herein provided, the parties hereto and their respective successor and assigns shall have the right from time to time to change their respective addresses effective upon receipt by the other parties of such notice and each shall have the right to specify as its address any other address within the United States of America.

## 7.4 WAIVER; AMENDMENT

None of the terms and conditions of this Agreement may be changed, waived, modified or varied in any manner whatsoever unless in writing duly signed by CCC, Pool Parent and the Landlord Representative (with the consent of the Collateral Agent if such change or waiver imposes any new duties on the Collateral Agent or would adversely affect the Collateral Agent); provided, however, that no such change, waiver, modification or variance shall be made to Section 2.2(c) hereof or this Section 7.4 without the consent of each Tenant adversely affected thereby.

## 7.5 BENEFIT

This Agreement shall create a continuing security interest in the Account Collateral and shall (i) be binding upon Pool Parent, CCC, each Tenant, each Tenant Member, the Landlord Representative and each Landlord, and the Collateral Agent and their respective successors and assigns; and (ii) inure to the benefit of the Collateral Agent, the Landlord Representative and each Landlord and their respective successors and assigns; provided that no Landlord shall be permitted to assign its rights hereunder to any Person to which such Landlord sells or conveys any hotel subject to a Lease, other than a Subsidiary of the Landlord Representative.

## 7.6 GOVERNING LAW.

THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF MARYLAND, REGARDLESS OF THE LAWS THAT MIGHT GOVERN UNDER APPLICABLE PRINCIPLES OF CONFLICT OF LAWS THEREOF.

## 7.7 HEADINGS

The headings of the several sections and subsections in this Agreement are for purposes of reference only and shall not limit or define the meaning hereof.

#### 7.8 COUNTERPARTS

This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other party.

## 7.9 ASSIGNMENT

The rights of the Collateral Agent under this Agreement may be assigned to a successor Collateral Agent.

#### 7.10 WAIVER OF JURY TRIAL

THE PARTIES HERETO, TO THE FULLEST EXTENT PERMITTED BY LAW, EACH HEREBY IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

## 7.11 DELAY NOT A WAIVER

Neither any failure nor any delay on the part of the Collateral Agent in insisting upon strict performance of any term, condition, covenant or agreement, or exercising any right, power, remedy or privilege hereunder, shall operate as or constitute a waiver thereof, nor shall a single or partial exercise thereof preclude any other future exercise, or the exercise of any other right, power, remedy or privilege. In particular, and not by way of limitation, by accepting payment after the due date of any amount payable under this Agreement or the Guarantee, neither the Landlord Representative, any Landlord, nor the Collateral Agent shall be deemed to have waived any right either to require prompt payment when due of all other amounts due under this Agreement or the Guarantee or to declare a default for failure to effect prompt payment of any such other amount.

## 7.12 SEVERABILITY

Any term or provision of this Agreement which is invalid or unenforceable in any applicable jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other applicable jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.

## 7.13 PREFERENCES

Neither the Collateral Agent, the Landlord Representative nor any of the Landlords shall have any obligation to marshal any assets in favor of Pool Parent, CCC or any other party or against or in payment of any or all of the obligations of Pool Parent or CCC pursuant to the Guarantee or this Agreement. To the extent Pool Parent makes a payment or payments to the Collateral Agent, the Landlord Representative or the Landlords, which payment or proceeds or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid to a trustee, receiver or any other party under any bankruptcy law, state or federal law, common law or equitable cause, then, to the extent of such payment or proceeds received, the obligations hereunder or part thereof intended to be satisfied shall be revived and continue in full force and effect, as if such payment or proceeds had not been received by the Collateral Agent, the Landlord Representative or the Landlords.

## 7.14 WAIVER OF NOTICE.

Pool Parent shall not be entitled hereunder to any notices of any nature whatsoever from the Collateral Agent, the Landlord Representative or the Landlords except with respect to matters for which this Agreement specifically and expressly provides for the giving of notice by such Person to Pool Parent and except with respect to matters for which Pool Parent is not, pursuant to applicable legal requirements, permitted to waive the giving of notice. To the extent permitted by applicable legal requirements, Pool Parent hereby expressly waives the right to receive any notice hereunder from the Collateral Agent, the Landlord Representative or the Landlords with respect to any matter for which this Agreement does not specifically and expressly provide for the giving of notice by any one or more of such Persons to Pool Parent.

## 7.15 REMEDIES OF POOL PARENT, CCC, OR TENANTS

In the event that a claim or adjudication is made that the Collateral Agent has unreasonably delayed acting in any case where, by law or under this Agreement, the Collateral Agent has an obligation to act promptly, Pool Parent, CCC and Tenants agree that the Collateral Agent shall not be liable for any monetary damages, and their sole remedies shall be limited to commencing an action seeking injunctive relief or declaratory judgment, so long as the Collateral Agent acted without gross negligence, bad faith, fraud or willful misconduct.

IN WITNESS WHEREOF, the parties hereto have caused this Pooling and Security Agreement to be duly executed in their names, all as of the date first written above.

[POOL PARENT]			
By: Name: Title:			
CRESTLINE CAPITAL CORPORATION			
By: Name: Title:			
EACH OF THE TENANTS NAMED ON SCHEDULE 1 HERETO, BY THEIR DULY APPOINTED ATTORNEY IN FACT			
Name:Attorney in Fact			
HOST MARRIOTT, L.P., AS LANDLORD REPRESENTATIVE			
By: Host Marriott Corporation, its general partner			
By: Name: Title:			
- 27 -			

EACH OF THE LANDLORDS NAMED ON SCHEDULE 1 HERETO, BY THEIR DULY APPOINTED ATTORNEY IN FACT

Name:\_\_\_\_\_\_Attorney in Fact

-28-

## AMENDED AND RESTATED NONCOMPETITION AGREEMENT

THIS AMENDED AND RESTATED NONCOMPETITION AGREEMENT ("Agreement") is made and entered into as of \_\_\_\_\_\_, 1998 by and among HOST MARRIOTT CORPORATION, a Delaware corporation ("Host Marriott"), CRESTLINE CAPITAL CORPORATION, a Maryland corporation ("Crestline"), FORUM GROUP, INC., an Indiana corporation ("FGI"), MARRIOTT SENIOR LIVING SERVICES, INC., a Delaware corporation ("MSLS"), and MARRIOTT INTERNATIONAL, INC., a Delaware corporation ("MI").

#### **RECITALS**

WHEREAS, Host Marriott, FGI, MSLS and MI entered into that certain Noncompetition Agreement dated as of June 21, 1997 (the "Original Noncompetition Agreement") in connection with the Stock Purchase Agreement dated as of June 21, 1997 (the "Stock Purchase Agreement") by and between Host and MSLS; and

WHEREAS, it was the intent of the parties that from and after the closing under the Stock Purchase Agreement (x) MSLS would continue in its capacity as Operator of the Senior Living Business previously conducted by FGI and its Subsidiaries and Affiliates and (y) except as otherwise specifically set forth in this Agreement, Host Marriott and FGI would act solely in their capacities as Investors in the Senior Living Facilities owned by FGI and its Subsidiaries and Affiliates and other Senior Living Facilities that Host Marriott and FGI might acquire in the future; and

WHEREAS, MSLS and MI desired to grant Host Marriott certain rights to participate in the development of certain retirement communities and to acquire certain other retirement communities within limited geographic areas adjacent to the Senior Living Facilities owned by FGI and its Subsidiaries and Affiliates on the Closing Date; and

WHEREAS, Host Marriott has determined that it will restructure its business operations so that as to qualify as a real estate investment trust for federal income tax purposes, and, in connection therewith, on or about December 29, 1998, (i) Host Marriott will distribute approximately 82% of the outstanding common stock of Crestline to or on behalf of the stockholders of Host Marriott (the "Crestline Distribution") and will contribute the remaining 18% of such Crestline common stock to Host Marriott, L.P. for delivery to The Blackstone Group and certain affiliated entities thereof (or for return to Crestline if not delivered to The Blackstone Group and its affiliated entities) and (ii) Host Marriott will merge (the "Merger") into HMC Merger Corporation, a Maryland corporation ("Host REIT"); and

WHEREAS, following the Crestline Distribution (i) FGI will remain a subsidiary of Crestline and (ii) Crestline will conduct those certain activities as an Investor in Senior Living Facilities conducted by Host Marriott prior to the restructuring; and

WHEREAS, as a result of such restructuring, Crestline is obligated to become a party to this Agreement, as contemplated by Section 3.4 of the Original Noncompetition Agreement; and

WHEREAS, the parties hereto desire to amend and restate the Original Noncompetition Agreement to add Crestline as a party to the Agreement.

NOW, THEREFORE, in consideration of the premises and the mutual covenants contained herein and in the Stock Purchase Agreement and in the related agreements entered into pursuant to or related to the Stock Purchase Agreement, and for other valuable consideration, the receipt and sufficiency of which are hereby mutually acknowledged, the parties hereto agree as follows:

# ARTICLE ONE DEFINITIONS

The following terms when used herein shall have the meaning set forth

"AFFILIATES" means, with respect to any Person at any time, any other Person directly or indirectly controlling or controlled by, or under direct or indirect common control with, such Person at such time. For purposes of this definition "control", when used with respect to any Person, means the power to direct the management and policies of such Person, directly or indirectly, through the ownership of voting securities, by contract, or otherwise.

Notwithstanding the foregoing, from and after the Closing Date, (x) "Affiliates" of Host Marriott do not include MI, MSLS, Host Marriott Services Corporation, Crestline, FGI, any of their respective Subsidiaries or any of their officers, directors or stockholders (in such capacity only), (y) "Affiliates" of MI and MSLS do not include Host Marriott, Crestline, FGI, Host Marriott Services Corporation or any of their respective Subsidiaries or any of their officers, directors or stockholders (in such capacity only), and (z) "Affiliates" of Crestline and FGI do not include Host Marriott, Host Marriott Services Corporation, MI, MSLS or any of their respective Subsidiaries or any of their officers, directors or stockholders (in such capacity only).

"CLOSING DATE" means June 21, 1997.

below:

"CRESTLINE" has the meaning set forth in the first paragraph of this  $\ensuremath{\mathsf{Agreement}}.$ 

"CRESTLINE DISTRIBUTION DATE" means the effective date of the Crestline Distribution.

 $\hbox{\tt "FGI"}$  has the meaning set forth in the first paragraph of this Agreement.

"FORUM OPERATING AGREEMENTS" has the meaning set forth in Section 3.1 hereof

"HOST MARRIOTT" has the meaning set forth in the first paragraph of this Agreement.

"INVESTOR" means, with respect to any facility or property, any Person who, directly or indirectly, owns, has an equity interest in or lends money to or otherwise finances, such facility or property.

"MI" has the meaning set forth in the first paragraph of this Agreement.

"MSLS" has the meaning set forth in the first paragraph of this Agreement.

"OPERATOR" means, with respect to any facility or property, any Person who, in whatever capacity, operates, manages, franchises (as a franchisor), or in any other manner supervises the day-to-day conduct of, such facility or property.

"PERSON" means any person, firm, corporation, limited liability company, general or limited partnership, limited liability partnership, association or other entity.

"SENIOR LIVING BUSINESS" means (i) the business of operating, managing or franchising (as a franchisor), or in any other manner supervising the day-to-day conduct of, any Senior Living Facility (other than typical monitoring of an Operator of a Senior Living Facility done by an Investor in such Senior Living Facility) and (ii) operational or management services with respect to health care, therapy, home health care, assisted living, nursing and related medical, residential, supportive and personal care services conducted at or related to any Senior Living Facility; provided that "SENIOR LIVING BUSINESS" does not

include activities involving services that are ancillary to a Senior Living Facility (e.g. operation of a gift shop business), so long as such activities do not involve either (x) conduct as an Operator of any such Senior Living Facilities or (y) any of the activities described in clause (ii) above.

"SENIOR LIVING FACILITY" means any limited service or full service retirement or senior living service facilities or communities, including assisted living facilities, nursing homes, congregate care facilities and other health care facilities providing residential, recreational, personal care, home care, assisted living, nursing care, other health care and like services, in any combination, to the elderly.

"SUBSIDIARIES" means, with respect to any Person at any time, corporations or other entities which are more than fifty percent (50%) owned, directly or indirectly, by such Person at such time, and partnerships in which such Person or a subsidiary thereof is a general partner at such time.

"TERM" means that period commencing on the Closing Date and terminating on the tenth anniversary of the Closing Date, provided, however, that, with respect to Crestline and FGI,

"Term" shall mean that period commencing on the Closing Date and terminating on the thirteenth anniversary of the Closing Date.

"TERRITORY" shall mean the continental United States.

"TRANSFER" shall mean the sale, conveyance, disposal of or other transfer of ownership, title or other interest, by operation of law or otherwise.

# ARTICLE TWO AGREEMENTS

- 2.1 AGREEMENT NOT TO COMPETE. Except as otherwise provided in this
- Article II, Host Marriott, Crestline and FGI shall not, nor shall Host Marriott, Crestline or FGI permit any of their Subsidiaries or Affiliates to:
  - (i) during the Term, directly or indirectly, within the Territory (a) compete in the Senior Living Business by conducting, participating or engaging in, or bidding for or otherwise pursuing, whether as a principal, sole proprietor, partner, financing source, equity holder or agent of, or consultant to, or manager or operator for, another Person, any Senior Living Business, or (b) have any ownership interest whatsoever in, or lend money to or otherwise finance any Person that conducts, participates or engages in, or bids for or otherwise pursues any Senior Living Business; or
  - (ii) except as expressly permitted by the Stock Purchase Agreement or by a Forum Operating Agreement, use or permit any Subsidiary or Affiliate to use, either as an Investor or as an Operator, the name "Forum" (or any variant thereof) with respect to any Senior Living Facility or in any other capacity in the Senior Living Business, including using their corporate names, designs or logos (including the name "Forum") with respect to any such Senior Living Facility (including identifying a geographic location with respect to any such Senior Living Facility).
    - 2.2 AGREEMENT NOT TO TRANSFER. During the Term, Host Marriott,

Crestline and FGI shall not, nor shall Host Marriott, Crestline or FGI permit any of their Subsidiaries or Affiliates to, enter into (or agree to enter into) any transaction or series of transactions that would upon consummation result, directly or indirectly, in the Transfer of a controlling interest in ten (10) or more Senior Living Facilities set forth on Schedule 1 hereto to the same Person or affiliated group of Persons, unless contemporaneously therewith such Person(s) execute and deliver an agreement binding such Person or Persons during the remainder of the Term, to the restrictions in this Article II applicable to Host Marriott, Crestline and FGI (subject to the exceptions set forth in Section 2.3), in form and substance reasonably satisfactory to MSLS.

2.3 EXCEPTIONS. Notwithstanding anything herein to the contrary,

-----

clause (i) of Section 2.1 shall not prohibit Host Marriott, Crestline, FGI or any of their Subsidiaries and Affiliates from engaging in the following activities:

- (i) the ownership of equity interests in any Person which derives revenue from activities as an Operator in the Senior Living Business if (a) such equity interests are traded on a national or regional stock exchange in the United States or are traded on the National Association of Securities Dealers, Inc., Automated Quotation System, and (b) Host Marriott, Crestline, FGI or any of their Subsidiaries and Affiliates directly or indirectly, are not beneficial owners of more than five percent (5%) of such Person's outstanding equity interests, but only so long as Host Marriott, Crestline, FGI or any of their Subsidiaries and Affiliates do not seek to control the management or operations of any such Person; or
- (ii) activities as an Operator of any Senior Living Facility owned by any such Person for a period not to exceed twelve months in any such instance, so long as such conduct arises solely as a result of a change in management or the manager or operator of any such Senior Living Facility; or
- (iii) bidding for, otherwise pursuing the acquisition of, or acquiring either (A) all of the voting and ownership interests in a Person, or (B) voting and ownership interests sufficient to control a Person, in each case if such Person is both an Operator in the Senior Living Business and an Investor in Senior Living Facilities (including by means of acquiring equity or debt securities of any such Person or providing financing to any such Person); provided that (x) the principal purpose of

such acquisition is other than acquiring for its own account, or controlling, an Operator of a Senior Living Business and (y) the activities conducted by such Person as an Operator in the Senior Living Business are terminated, or such Senior Living Business (or Host Marriott's, Crestline's or FGI's investment therein) is divested by Host Marriott, Crestline, FGI and their Subsidiaries and Affiliates, within twelve months from the date Host Marriott, Crestline, FGI or their Subsidiaries and Affiliates (together with any Persons that may be acting in concert therewith) first acquire a majority of the voting interests of such Operator or otherwise are able to control such Operator; and provided further that, if requested

in writing by MSLS, Host Marriott or Crestline within five days confirms or reconfirms in writing to MSLS that the principal purpose of the acquisition is other than acquiring for its own account, or controlling, an Operator of a Senior Living Business (for purposes of this paragraph, the terms "control" and "controlling," when used with respect to any Person, mean the power to direct the management and policies of such Person, directly or indirectly, through the ownership of voting securities, by contract or otherwise); or

(iv) engagement of an Operator (other than Host Marriott, Crestline, FGI or any of their Subsidiaries or Affiliates) with respect to any Senior Living Facility owned by Host Marriott, Crestline, FGI or any of their Subsidiaries or Affiliates.

- (a) Host Marriott, Crestline and FGI shall not, nor shall either Host Marriott, Crestline or FGI permit any of their Subsidiaries or Affiliates to, at any time use for its own account or divulge to any Person whomsoever any MSLS Confidential Information (as defined below). Each of Host Marriott, Crestline and FGI shall use its best efforts to prevent the publication or disclosure of any such MSLS Confidential Information by any of their Subsidiaries or Affiliates. "MSLS Confidential Information" means confidential or proprietary information or material received directly or indirectly from MSLS or any of its Affiliates that is not publicly available relating to the following: (1) projects in the Senior Living Business or Senior Living Facilities that are under consideration or development by MSLS or any of its Affiliates at any time, (2) current and future operating systems and procedures employed by MSLS and its Affiliates in their capacities as Operators in the Senior Living Business, (3) marketing and feasibility analyses, studies and materials used and developed from time to time by MSLS and its Affiliates, (4) development of new or derivative products for use in the Senior Living Business or Senior Living Facilities and (5) such other types of information and materials as are reasonably requested by MSLS from time to time in writing to be treated as confidential or proprietary information hereunder. Notwithstanding the foregoing, Host Marriott, Crestline and FGI may disclose MSLS Confidential Information (x) if and to the extent required by law, so long as prior thereto (to the extent permitted by law) MSLS is provided the opportunity to contest any such required disclosure, and (y) if and to the extent required by existing or potential lenders, existing or potential investors or potential acquirors of any Senior Living Facility owned or leased by Host Marriott, Crestline, FGI or any of their Subsidiaries or Affiliates, so long as such Persons acknowledge the confidential and proprietary nature of such MSLS Confidential Information and agree, for the benefit of MSLS, to keep such information confidential to the same extent as required hereby.
- (b) MSLS shall not, nor shall MSLS permit any of its Subsidiaries or Affiliates to, at any time use for its own account or divulge to any Person whomsoever any Host Confidential Information (as defined below). MSLS shall use its best efforts to prevent the publication or disclosure of any Host Confidential Information by any of its Subsidiaries or Affiliates. "Host Confidential Information" means confidential or proprietary information or material received directly or indirectly from Host Marriott, Crestline, FGI or any of their Subsidiaries or Affiliates that is not publicly available relating to (i) projects that are under consideration or development by Host Marriott, Crestline, FGI or any of their Subsidiaries or Affiliates at any time in their capacities as Investors in Senior Living Facilities, (ii) marketing and feasibility analyses, studies and materials used and developed from time to time by Host Marriott, Crestline, FGI and their Subsidiaries and Affiliates in their capacities as Investors in Senior Living Facilities or (iii) such other types of information and materials as are reasonably requested by Host Marriott or Crestline from time to time in writing to be treated as confidential or proprietary information hereunder. Notwithstanding the foregoing, MSLS may disclose Host

Confidential Information (x) if and to the extent required by law, so long as prior thereto (to the extent permitted by law) Host Marriott, Crestline and FGI are provided the opportunity to contest any such required disclosure and (y) if and to the extent required by existing or potential lenders, existing or potential investors or potential acquirors of MSLS's Senior Living Business, so long as such Persons acknowledge the confidential and proprietary nature of such Host Confidential Information and agree, for the benefit of Host Marriott or Crestline, to keep such information confidential to the same extent as required hereby.

2.5 INTERPRETATION. The parties agree that the period of restriction

and the geographical area of restriction imposed upon Host Marriott, Crestline and FGI in this Article II are fair and reasonable and are reasonably required for the protection of MSLS. If the provisions of this Agreement relating to the area of restriction or the period of restriction shall be deemed to exceed the maximum area or period which a court having jurisdiction over the matter would deem enforceable, such area or period shall, for purposes of this Agreement, be deemed to be the maximum area or period which such court would deem valid and enforceable.

# ARTICLE THREE DEVELOPMENT AND ACQUISITION RIGHTS

3.1 DEVELOPMENT RESTRICTIONS. MSLS is a party to operating

agreements with respect to all Senior Living Facilities owned indirectly by FGI on the date hereof and listed on Schedule 1 to this Agreement (the "Forum Operating Agreements"), and, in accordance therewith, the parties thereto have agreed upon certain restrictions regarding development of Similar Properties within specified regions. These regions are referred to as "Area A" and "Area B" in each Forum Operating Agreement. As used herein, "Similar Property" means a then-existing full service retirement community offering the full care continuum of independent living (predominantly), assisted living, and health care services, and "Assisted Living Facility" means a retirement community or facility dedicated in its entirety to assisted living accommodations and services, Alzheimer's facilities and nursing care facilities, so long as the Alzheimer's and nursing care facilities constitute less than half of such community or facility.

3.2 ACQUISITION OF SIMILAR PROPERTY. Until the earlier of (x) the

seventh anniversary of the Closing Date or (y) the date on which the Forum Operating Agreement for the relevant Senior Living Facility is terminated (unless such termination results from a breach by MSLS thereunder), if MI, MSLS or any of their Affiliates has the opportunity to acquire a Similar Property in either Area A or Area B and MSLS is permitted under the Forum Operating Agreement to acquire Similar Properties in Area A or Area B, at MI's or MSLS's option, MI or MSLS shall either (i) offer Crestline the opportunity to acquire such Similar Property by providing prompt written notice to Crestline of such opportunity or (ii) attempt to acquire such Similar Property itself (or through one of its Affiliates) and, if successful in acquiring such similar Property, thereafter offer Crestline the opportunity to acquire such Similar Property on substantially similar terms and conditions by providing prompt written notice to Crestline of such

opportunity as soon as practicable following the acquisition of such Similar Property. Within thirty (30) days of this initial notice, Crestline must provide written notice to MI or MSLS of its intention to exercise this right. Failure to deliver such notice in a timely manner will terminate Crestline's rights hereunder as to the Similar Property in question. In connection with its acquisition of a Similar Property in accordance with this Section 3.2, Crestline will cause the owner of such Similar Property to enter into an operating agreement with MSLS having terms substantially similar to a Forum Operating Agreement. Crestline's rights under this Section 3.2 will not apply to any Similar Properties constructed or developed by MI, MSLS or any of their Affiliates.

### 3.3 ASSISTED LIVING DEVELOPMENT PARTICIPATION RIGHTS. Until the

earlier of (x) the seventh anniversary of the Closing Date or (y) the date on which the Forum Operating Agreement for the relevant Senior Living Facility is terminated (unless such termination results from a breach by MSLS thereunder), MI and MSLS will allow Crestline the opportunity to negotiate to participate in the development (as an Investor) of any Assisted Living Facility that MI, MSLS or their Affiliates may seek to develop in such Area A or Area B. The right of Crestline to negotiate with MI or MSLS to participate in the development of an Assisted Living Facility in such Area A or Area B shall commence upon its receipt of written notice from MI or MSLS, specifically referencing this Section 3.3, of its intent to develop an Assisted Living Facility in such area. Crestline must provide written notice to MI or MSLS within ten (10) days of this initial notice if it intends to exercise this right. Failure to deliver such notice in a timely manner will terminate Crestline's rights hereunder as to the Assisted Living Facility in question. If such notice is timely received, MI or MSLS and Crestline shall negotiate the terms and conditions upon which MI or MSLS would permit Crestline to invest in such Assisted Living Facility (it being understood that any such terms and conditions would require the owner of such Assisted Living Facility to enter into an operating agreement with MI or MSLS having terms substantially similar to a Forum Operating Agreement). If the terms of the proposed investment are not mutually agreed upon within forty-five (45) days of MI's or MSLS' initial notice, MI or MSLS may enter into agreements with any other parties for investment in such Assisted Living Property upon any terms whatsoever. In no event shall MI or MSLS be obligated to offer to Crestline the opportunity to invest in such Assisted Living Property on the same terms to which MI or MSLS agrees with any other party so long as MI or MSLS has complied with the other provisions of this Section. Crestline's rights under this Section 3.3 will not apply to (x) any of the Senior Living Facilities set forth on Schedule 2 hereto or (y) any Assisted Living Facility acquired or sought to be acquired by MI, MSLS or any of their Affiliates.

by Crestline, except to (i) one of its Affiliates or (ii) any successor to all or substantially all of Crestline's business as an Investor in Senior Living Facilities; provided that neither Host Marriott nor Crestline is released from

its obligations in connection with any such assignment.

3.5 NO EFFECT ON FORUM OPERATING AGREEMENTS. The terms of this

Article III do not modify or otherwise affect the restrictions on MI, MSLS and their Affiliates set forth in Section 19.14 of each Forum Operating Agreement.

## ARTICLE FOUR MISCELLANEOUS

- 4.1 ENTIRE AGREEMENT. This Agreement, the Stock Purchase Agreement
- and the other Transaction Documents (as defined in the Stock Purchase Agreement) constitute the entire agreement of the parties concerning the subject matter hereof. Additional restrictions on competition may be contained in the Forum Operating Agreements, and restrictions contained therein shall be in addition to the restrictions contained herein.
  - 4.2 DOCUMENTARY CONVENTIONS. This Agreement shall be governed by the

Documentary Conventions. Documentary Conventions means, with respect to any document or agreement that states in substance that it is governed thereby, that such document or agreement shall be deemed to include the following provisions and all references in any such provisions to "this Agreement," "hereunder," "hereby" or similar phrases shall refer to the document or agreement in which such provisions are incorporated:

- (i) MODIFICATIONS. No modification to this Agreement shall be valid unless in writing and signed by all parties thereto. No purported waiver of any of the provisions of this Agreement shall be valid or effective unless in writing signed by the party against whom such waiver is sought to be enforced.
- (ii) SURVIVAL. All representations, warranties and covenants in this Agreement shall survive and not be merged in the execution of this Agreement.
- (iii) GOVERNING LAW. The Agreement shall be governed by and construed in accordance with the internal laws of the State of Maryland, without reference to conflicts of laws principles.
- (iv) CAPTIONS; PRONOUNS. Captions in this Agreement are for convenience of reference only and shall not be considered in construing this Agreement. Whenever the context shall so require, the singular shall include the plural, the male gender shall include the female, and vice versa. "Include," "includes" and "including" shall be deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of like import.
- (v) EXHIBITS. All exhibits to this Agreement are incorporated in this Agreement as though set forth in full in the text of this Agreement.
- (vi) COUNTERPARTS. Multiple originals of this Agreement may be executed, each of which shall constitute one and the same agreement. This Agreement may be executed in counterparts, and it shall not be necessary that the original signature of each party to this Agreement appear on each such counterpart.

- (vii) SEVERABILITY. In the event that one or more of the provisions of this Agreement shall be held to be illegal, invalid or unenforceable, such provisions shall be deemed severable and the remaining provisions of this Agreement shall continue in full force and effect.
- (viii) NOT CONSTRUED AGAINST DRAFTER. Each party to this Agreement acknowledges that it was represented by counsel in connection with this Agreement, and that it and its counsel reviewed and participated in the preparation and negotiation of this Agreement. Consequently, any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement.
- (ix) BUSINESS DAY. To the extent that the date of any performance required under this Agreement falls on a date which is not a business day, the date of performance shall be extended to the next succeeding business day.
- (x) WAIVERS. No waiver of any provision or right set forth in this Agreement shall be valid unless it is in writing signed by the party against which such waiver is sought to be enforced. The failure of any party to insist on strict performance of any of the provisions of this Agreement or to exercise any right granted to it under this Agreement shall not be construed as a waiver of the requirement of such performance.
- (xi) PARTIES IN INTEREST. This Agreement shall be binding upon and inure to the benefit of each party, and nothing in this Agreement, express or implied, is intended to confer upon any other person or entity any rights or remedies of any nature whatsoever under or by reason of this Agreement. Nothing in this Agreement is intended to relieve or discharge the obligation of any third person to any party of this Agreement.
- 4.3 REMEDIES. The parties agree that irreparable damage would occur in the event any of the provisions of this Agreement were not to be performed in accordance with the terms bereof, and that their remedy at law for any breach of

accordance with the terms hereof, and that their remedy at law for any breach of the other party's obligations hereunder would be inadequate. The parties agree and consent that, in addition to any other rights or remedies that may be available of law or in equity, temporary and permanent injunctive relief may be granted in any proceeding which may be brought to enforce any provision hereof without the necessity of proof of actual damage.

4.4 ENFORCEABILITY; SUCCESSORS AND ASSIGNS. Subject to the terms of

Section 4.5 hereof, the terms, conditions and promises contained in this Agreement shall be binding upon and shall inure to the benefit of each of the parties hereto, their heirs, personal representatives or successors and assigns, including, in the case of MSLS, any successor to all or substantially all of MSLS's Senior Living Business. Each party hereto covenants and agrees to cause its Subsidiaries and Affiliates to comply with such party's obligations hereunder. Nothing herein,

expressed or implied, shall be construed to give any other Person any legal or equitable rights hereunder.

- 4.5 ASSIGNMENT. Except in the case of MSLS as provided in Section
- 4.4 and in the case of Crestline as provided in Section 3.4, no party to this Agreement shall, without the prior written consent of the others assign any rights or delegate any obligations under this Agreement.
  - 4.6 CONSENT TO JURISDICTION. Any suit, action or proceeding under or

in connection with this Agreement shall be brought in any federal or state court of competent jurisdiction located in the State of Maryland. By execution of this Agreement, each party consents to the exclusive jurisdiction of such courts, and waives any right to challenge the jurisdiction of such courts or the appropriateness of venue in such courts. EACH PARTY HEREBY WAIVES ANY RIGHT TO TRIAL BY JURY IN CONNECTION WITH ANY SUIT, ACTION OR PROCEEDING UNDER OR IN CONNECTION WITH THIS AGREEMENT.

4.7 NOTICES. All notices and other communications hereunder shall be  $\overline{\phantom{a}}$ 

in writing and shall be delivered by hand, by facsimile, delivered by nationally recognized overnight courier, or mailed by registered or certified mail (return receipt requested) to the parties at the following addresses (or at such other addresses for a party as shall be specified by like notice) and shall be deemed given on the date on which such notice is received:

To Host Marriott:

Host Marriott Corporation Asset Management Department 908 10400 Fernwood Road Bethesda, Maryland 20817 Attention: Director, Senior Living Facilities

FAX NO. 301/380-6338

with a copy to:

Host Marriott Corporation Law Department 903 10400 Fernwood Road Bethesda, Maryland 20817 Attention: General Counsel

FAX NO. 301/380-3588

with a copy to:

Arnold & Porter 555 12th Street, N.W. Washington, D.C. 20004 Attention: Michael D. Goodwin, Esq. FAX NO. 202/942-5999

#### To Crestline or FGI:

c/o Crestline Capital Corporation 10400 Fernwood Road Bethesda, Maryland, 20817 Attention: General Counsel

FAX NO.

with a copy to:

Crestline Capital Corporation 10400 Fernwood Road Bethesda, Maryland, 20817 Attention: Elizabeth Lieberman

FAX NO.

#### To MSLS:

Marriott Senior Living Services, Inc.
One Marriott Drive
Washington, D.C. 20058 (registered or certified mail)
Attention: Chief Financial Officer
Dept. 52/923
FAX NO. 301/380-6540

and

10400 Fernwood Road Bethesda, Maryland 20817 (express mail or courier) Attention: General Counsel Dept. 52/923

FAX NO. 301/380-6727

with a copy to:

O'Melveny & Myers LLP 555 13th Street, N.W. Washington, D.C. 20004 Attention: David G. Pommerening, Esq. FAX NO. 202/383-5414

4.8 RELATIONSHIP OF PARTIES. It is understood and agreed that

nothing in this Agreement shall be deemed or construed by the parties or any third party as creating an employer-employee principal/agent, partnership or joint venture relationship between the parties. It is further understood that the obligations of each of Host Marriott, on the one hand, and Crestline and FGI, on the other hand, hereunder, and any liability arising therefrom, is several and not joint.

4.9 HOST REIT AS SUCCESSOR. The parties acknowledge that, upon the effectiveness of the Merger, Host REIT shall succeed to all the rights and obligations of Host Marriott under this Agreement.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered, all as of the day and year first above written.

ву:		
Name: Title:		
MARRIOTT INTERNATIONAL, INC.		
Ву:		
Name: Title:		
HOST MARRIOTT CORPORATION		
Ву:		
Name: Title:		
HMC MERGER CORPORATION		
By:		
Name: Title:		

MARRIOTT SENIOR LIVING SERVICES, INC.

By:
Name:
Title:
CRESTLINE CAPITAL CORPORATION
By:
Name:
Title:

S-2

FORUM GROUP, INC.

## COMMUNITIES SUBJECT TO FORUM OPERATING AGREEMENTS

The Forum at Brookside, Louisville, Kentucky.

Coral Oaks Retirement Community, Palm Harbor, Florida.

Forum at Deer Creek, Deerfield Beach, Florida.

Desert Harbor, Peoria, Arizona.

Forwood Manor, Wilmington, Delaware.

The Forum at the Crossing, Indianapolis, Indiana.

Foulk Manor North, Wilmington, Delaware.

Foulk Manor South, Wilmington Delaware.

Fountainview, West Palm Beach, Florida.

Springwood Court, Ft. Myers, Florida.

The Forum at Knightsbridge, Columbus, Ohio.

Lafayette at Country Place, Lexington, Kentucky.

Lexington at Country Place, Lexington, Kentucky.

Forum at Lincoln Heights, San Antonio, Texas.

The Forum at Memorial Woods Healthcare, Houston, Texas.

Millcroft Retirement & Nursing Home, Wilmington, Delaware.

Montebello on Academy, Albuquerque, New Mexico.

Montevista at Coronado, El Paso, Texas.

Myrtle Beach Manor, Myrtle Beach, South Carolina.

The Forum at Overland Park, Overland Park, Kansas.

The Forum at Park Lane, Dallas, Texas.

Park Summit at Coral Springs, Coral Springs, Florida.

The Forum - Pueblo Norte, Scottsdale, Arizona.

Remington Club I at Rancho Bernardo, San Diego, California.

Remington Club II at Rancho Bernardo, San Diego, California.

Shipley Manor, Wilmington, Delaware.

Tiffany House, Ft. Lauderdale, Florida.

Forum at Tucson, Tucson, Arizona.

Forum at The Woodlands, Montgomery County, Texas.

### SCHEDULE 2 Assisted Living Sites Under Development

	EXISTING COMMUNITY	APPROXIMATE LOCATIONS OF EXISTING DEVELOPMENT SITES
1. E	BROOKSIDE	None
2. 0	CORAL OAKS	None
3. F	FORUM @ CROSSING	Approximately 400 feet west of Forum @ Crossing at NWC of E. 86th St. & Elrico
	DEER CREEK	None
	DESERT HARBOR	None
6. F	FORWOOD MANOR	S/W corner of New Murph Rd. (Rt. 141) and Rockland Rd.
7. F	FOULK MANOR NORTH	S/W corner of New Murph Rd. (Rt. 141) and Rockland Rd.
8. F	FOULK MANOR NORTH	. S/W corner of New Murph Rd. (Rt. 141) and Rockland Rd. . S/E corner of Lancaster Pike (Rt. 48) and Centerville Rd.
	-OUNTAINVIEW	None
	KNIGHTSBRIDGE	None
	LAFAYETTE	None
	LEXINGTON	None
13.	LINCOLN HEIGHTS	None
14.	MEMORIAL WOODS	None
15.	MILLCROFT	None
16.	MONTEBELLO	None
17.	MONTEVISTA	None
18.	MYRTLE BEACH MANOR	None
19.	OVERLAND PARK	East Side of Mission Road South of 71st Street
20.	PARK LANE	Across the street from Park Lane Forum
21.	PARK SUMMIT	None
22.	PUEBLO NORTE	. Hayden Avenue and E. Cactus Rd. . 91st St.; South of Shea Blvd.
23.	REMINGTON I	None
24.	REMINGTON II	None
	SHIPLEY MANOR	S/W corner of New Murph Rd. (Rt. 141) and Rockland Rd.
26.	SPRINGWOOD COURT (formerly known as Independence Court of Kentwood)	None
	TIFFANY HOUSE	None
	TUCSON	None
	WOODLANDS (formerly known as Chambrel)	None

### CONSENT OF INDEPENDENT PUBLIC ACCOUNTANTS

As independent public accountants, we hereby consent to the use of our reports (and to all references to our Firm) included in or made a part of this registration statement.

Arthur Andersen LLP

Washington, D.C.

November 19, 1998

To HMC Merger Corporation:

Pursuant to Rule 438 promulgated under the Securities Act of 1933, as mended, I hereby consent to the references in the Registration Statement of HMC Merger Corporation ("Host REIT") on Form S-4, and amendments thereto, which indicate that I have accepted a nomination to become a director of Host REIT subsequent to the consummation of the merger of Host Marriott Corporation with and into Host REIT as described in the Proxy Statement/Prospectus that forms a part of the Registration Statement.

/s/ Richard E. Marriott

Richard E. Marriott

To HMC Merger Corporation:

Pursuant to Rule 438 promulgated under the Securities Act of 1933, as amended, I hereby consent to the references in the Registration Statement of HMC Merger Corporation ("Host REIT") on Form S-4, and amendments thereto, which indicate that I have accepted a nomination to become a director of Host REIT subsequent to the consummation of the merger of Host Marriott Corporation with and into Host REIT as described in the Proxy Statement/Prospectus that forms a part of the Registration Statement.

/s/ J.W. Marriott, Jr.
J.W. Marriott, Jr.

To HMC Merger Corporation:

Pursuant to Rule 438 promulgated under the Securities Act of 1933, as amended, I hereby consent to the references in the Registration Statement of HMC Merger Corporation ("Host REIT") on Form S-4, and amendments thereto, which indicate that I have accepted a nomination to become a director of Host REIT subsequent to the consummation of the merger of Host Marriott Corporation with and into Host REIT as described in the Proxy Statement/Prospectus that forms a part of the Registration Statement.

/s/ R. Theodore Ammon

-----

R. Theodore Ammon

To HMC Merger Corporation:

Pursuant to Rule 438 promulgated under the Securities Act of 1933, as amended, I hereby consent to the references in the Registration Statement of HMC Merger Corporation ("Host REIT") on Form S-4, and amendments thereto, which indicate that I have accepted a nomination to become a director of Host REIT subsequent to the consummation of the merger of Host Marriott Corporation with and into Host REIT as described in the Proxy Statement/Prospectus that forms a part of the Registration Statement.

/s/ Robert M. Baylis -----Robert M. Baylis

To HMC Merger Corporation:

Pursuant to Rule 438 promulgated under the Securities Act of 1933, as amended, I hereby consent to the references in the Registration Statement of HMC Merger Corporation ("Host REIT") on Form S-4, and amendments thereto, which indicate that I have accepted a nomination to become a director of Host REIT subsequent to the consummation of the merger of Host Marriott Corporation with and into Host REIT as described in the Proxy Statement/Prospectus that forms a part of the Registration Statement.

/s/ Ann Dore McLaughlin
Ann Dore McLaughlin

To HMC Merger Corporation:

Pursuant to Rule 438 promulgated under the Securities Act of 1933, as amended, I hereby consent to the references in the Registration Statement of HMC Merger Corporation ("Host REIT") on Form S-4, and amendments thereto, which indicate that I have accepted a nomination to become a director of Host REIT subsequent to the consummation of the merger of Host Marriott Corporation with and into Host REIT as described in the Proxy Statement/Prospectus that forms a part of the Registration Statement.

/s/ Harry L. Vincent, Jr.
Harry L. Vincent, Jr.

To HMC Merger Corporation:

Pursuant to Rule 438 promulgated under the Securities Act of 1933, as amended, I hereby consent to the references in the Registration Statement of HMC Merger Corporation ("Host REIT") on Form S-4, and amendments thereto, which indicate that I have accepted a nomination to become a director of Host REIT subsequent to the consummation of the merger of Host Marriott Corporation with and into Host REIT as described in the Proxy Statement/Prospectus that forms a part of the Registration Statement.

/s/ John G. Schreiber John G. Schreiber

To HMC Merger Corporation:

Pursuant to Rule 438 promulgated under the Securities Act of 1933, as amended, I hereby consent to the references in the Registration Statement of HMC Merger Corporation ("Host REIT") on Form S-4, and amendments thereto, which indicate that I have accepted a nomination to become a director of Host REIT subsequent to the consummation of the merger of Host Marriott Corporation with and into Host REIT as described in the Proxy Statement/Prospectus that forms a part of the Registration Statement.

/s/ Terence C. Golden

Terence C. Golden

[LOGO OF BT WOLFENSOHN APPEARS HERE]

We hereby consent to (i) the inclusion of our opinion letter, dated November 23, 1998, to the Board of Directors of Host Marriott Corporation as an Exhibit to the Proxy Statement/Prospectus forming part of this Registration Statement on Form S-4, and (ii) references made to our firm and such opinion in such Proxy Statement/Prospectus under the captions entitled "Opinion of BT Wolfensohn, Financial Advisor to Host" and "Opinion of BT Wolfensohn." In giving such consent, we do not admit that we come within the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as Amended, and the Rules and Regulations Promulgated thereunder, and we do not admit that we are experts with respect to any part of the Registration Statement within the meaning of the term "expert" as used in the Securities ACt of 1933, as amended, or the rules and regulations of the Securities and Exchange Commission promulgated thereunder.

BT Wolfensohn is engaged in the merger and acquisition and client advisory business of Bankers Trust and, for legal and regulatory purposes, is a division of BT Alex. Brown Incorporated, a registered broker-dealer and member of the New York Stock Exchange.

/s/ BT Wolfensohn BT Wolfensohn

November 18, 1998